Parvaiz Ahmad M.N.V. Prasad *Editors*

Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change



Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change

Parvaiz Ahmad • M.N.V. Prasad Editors

Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change



Editors Parvaiz Ahmad Department of Botany Amar Singh College University of Kashmir Srinagar, Jammu and Kashmir India parvaizbot@rediffmail.com

M.N.V. Prasad Department of Plant Sciences University of Hyderabad Andhra Pradesh, Hyderabad 500 046 India mnvsl@uohyd.ernet.in, prasad_mnv@yahoo.com

ISBN 978-1-4614-0814-7 e-ISBN 978-1-4614-0815-4 DOI 10.1007/978-1-4614-0815-4 Springer New York Dordrecht Heidelberg London

Library of Congress Control Number: 2011938457

© Springer Science+Business Media, LLC 2012

All rights reserved. This work may not be translated or copied in whole or in part without the written permission of the publisher (Springer Science+Business Media, LLC, 233 Spring Street, New York, NY 10013, USA), except for brief excerpts in connection with reviews or scholarly analysis. Use in connection with any form of information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed is forbidden.

The use in this publication of trade names, trademarks, service marks, and similar terms, even if they are not identified as such, is not to be taken as an expression of opinion as to whether or not they are subject to proprietary rights.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Preface

Any external factor that imposes negative impact on growth and development of the plant is known as stress. Plants often experience abiotic stress like drought, salinity, alkalinity, temperature, UV-radiations, oxygen deficiency, etc. Abiotic stress is responsible for the huge crop loss and reduced yield more than 50% of some major crops. Ion imbalance and osmotic stress is the primary effect of abiotic stress. Prolonged exposure to primary stress causes secondary stress through the generation of reactive oxygen species (ROS). These are deleterious for the plants as it causes oxidative damage by reacting with biomolecules. Plants are able to perceive the external and internal signals and are then used by the plant to regulate various responses to stress. Plants respond the abiotic stress by up- and downregulation of genes responsible for the synthesis of osmolytes, osmoprotectants, and antioxidants. Stress-responsive genes and gene products including proteins are expressed and provide tolerance to the plant. To understand the physiological, biochemical, and molecular mechanisms for abiotic stress, perception, transduction, and tolerance is still a challenge before plant biologists.

The chapters in this book deal with the effect of different abiotic stresses on plant metabolism and responses of the plants to withstand the stress. Chapter 1 describes involvement of different osmolytes, osmoprotectants, and antioxidants during abiotic stress. Chapter 2 deals with the role of halophytes in understanding and managing abiotic stress. Chapter 3 addresses the effect and defense mechanisms in plants under UV stress. Chapter 4 throws light on the potassium uptake and its role under abiotic stress. Chapters 5-7 deal with the effect of temperature (heat, chilling) on plants and their responses. Chapter 8 deals with the formation and function of roots under stress. Chapter 9 is concerned with role of ROS and NO under abiotic stress. Chapter 10 throws light on nitrogen inflow and nitrogen use efficiency (NUE) under stress. Chapter 11 addresses Am symbiosis and soil interaction under abiotic stress. Chapter 12 deals with the role of small RNA in abiotic stress. Chapter 13 describes the involvement of transcription factors (TFs) under abiotic stress. Chapters 14-17 deal with the involvement of different signaling molecules (Ca2+, H2O2, and phytohormones) under abiotic stress. Chapter 18 covers the role of ethylene and plant growth-promoting bacteria under environmental stress. Chapter 19 throws light on new approaches about metal-induced stress. Chapters 20 and 21 address the role of sulfur and salicylic acid in alleviating heavy metal-induced stress. Chapters 22 and 23 cover the bioremediation of organic contaminants and utilization of different weeds in removal of heavy metals. We hope that this volume will provide the background for understanding abiotic stress tolerance in plants.

Srinagar, Jammu & Kashmir, India Hyderabad, Andhra Pradesh, India Parvaiz Ahmad M.N.V. Prasad

Contents

1	Abiotic Stress Responses in Plants: An Overview Hans-Werner Koyro, Parvaiz Ahmad, and Nicole Geissler	1
2	Prospects of Halophytes in Understanding and Managing Abiotic Stress Tolerance Vinayak H. Lokhande and Penna Suprasanna	29
3	UV-B Radiation, Its Effects and Defense Mechanisms in Terrestrial Plants Fernando E. Prado, Mariana Rosa, Carolina Prado, Griselda Podazza, Roque Interdonato, Juan A. González, and Mirna Hilal	57
4	K ⁺ Nutrition, Uptake, and Its Role in Environmental Stress in Plants Manuel Nieves-Cordones, Fernando Alemán, Mario Fon, Vicente Martínez, and Francisco Rubio	85
5	Temperature Stress and Responses of Plants Anna Źróbek-Sokolnik	113
6	Responses and Management of Heat Stress in Plants Abdul Wahid, Muhammad Farooq, Iqbal Hussain, Rizwan Rasheed, and Saddia Galani	135
7	Understanding Chilling Tolerance Traits Using Arabidopsis Chilling-Sensitive Mutants Dana Zoldan, Reza Shekaste Band, Charles L. Guy, and Ron Porat	159
8	Root Form and Function in Plant as an Adaptation to Changing Climate Maria Rosa Abenavoli, Maria Rosaria Panuccio, and Agostino Sorgonà	175

9	Reactive Oxygen Species and Nitric Oxide in Plants Under Cadmium Stress: From Toxicity to Signaling Luisa M. Sandalio, Maria Rodríguez-Serrano, Dharmendra K. Gupta, Angustias Archilla, Maria C. Romero-Puertas, and Luis A. del Río	199
10	Reactive Nitrogen Inflows and Nitrogen Use Efficiency in Agriculture: An Environment Perspective Khalid Rehman Hakeem, Ruby Chandna, Altaf Ahmad, and Muhammad Iqbal	217
11	Arbuscular Mycorrhizal Symbiosis and Other Plant–Soil Interactions in Relation to Environmental Stress Patrick Audet	233
12	MicroRNAs and Their Role in Plants During Abiotic Stresses Praveen Guleria, Deepmala Goswami, Monika Mahajan, Vinay Kumar, Jyoti Bhardwaj, and Sudesh Kumar Yadav	265
13	Transcription Factors Involved in Environmental Stress Responses in Plants Haibo Xin, Feng Qin, and Lam-Son Phan Tran	279
14	Plant Signaling Under Abiotic Stress Environment Parvaiz Ahmad, Renu Bhardwaj, and Narendra Tuteja	297
15	Calcium Signalling in Plant Cells Under Environmental Stress Sylvia Lindberg, Md. Abdul Kader, and Vladislav Yemelyanov	325
16	Role of H₂O₂ as Signaling Molecule in Plants M.A. Matilla-Vázquez and A.J. Matilla	361
17	Role of Phytohormone Signaling During Stress Mohammad Miransari	381
18	Ethylene and Abiotic Stress Tolerance in Plants Elisa Gamalero and Bernard R. Glick	395
19	New Approaches to Study Metal-Induced Stress in Plants M.C. Cia, F.R. Capaldi, R.F. Carvalho, P.L. Gratão, and R.A. Azevedo	413
20	Sulfur in the Alleviation of Cadmium-Induced Oxidative Stress in Plants Noushina Iqbal, Nafees A. Khan, Md. Iqbal R. Khan, Rahat Nazar, Asim Masood, and Shabina Syeed	429

21	Role of Salicylic Acid in Alleviating Heavy Metal Stress Losanka P. Popova, Liliana T. Maslenkova, Albena Ivanova, and Zhivka Stoinova	447
22	Bioremediation and Mitigation of Organic Contaminants in the Era of Climate Changes Laura Coppola, Edoardo Puglisi, Costantino Vischetti, and Marco Trevisan	467
23	Exploitation of Weeds and Ornamentals for Bioremediation of Metalliferous Substrates in the Era of Climate Change M.N.V. Prasad	487
Ind	ex	509

Contributors

Maria Rosa Abenavoli Dipartimento di Biotecnologie per il Monitoraggio Agro-Alimentare ed Ambientale, Università Mediterranea di Reggio Calabria, Contrada Melissari – Lotto D, 89124 Reggio Calabria, Italy

Altaf Ahmad Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard, New Delhi 110062, India

Parvaiz Ahmad Department of Botany, A.S. College, Srinagar 190008, Jammu & Kashmir, India

Fernando Alemán Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain

Angustias Archilla Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Científicas (CSIC), Mail box 419, E-18080 Granada, Spain

Patrick Audet Centre for Mined Land Rehabilitation, Sustainable Minerals Institute, The University of Queensland, Brisbane, QLD 4072, Australia

R. A. Azevedo Departamento de Genética, Escola Superior de Agricultura Luiz de Queiroz, Universidade de São Paulo, Piracicaba 13418-900, SP, Brazil

Reza Shekaste Band Department of Environmental Horticulture, University of Florida, Gainesville, FL 32611, USA

Jyoti Bhardwaj Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

Renu Bhardwaj Department of Botanical and Environmental Sciences, Guru Nanak Dev University, Amritsar, Punjab, India

F. R. Capaldi Departamento de Genética, Escola Superior de Agricultura Luiz de Queiroz, Universidade de São Paulo, Piracicaba 13418-900, SP, Brazil **R. F. Carvalho** Departamento de Biologia Aplicada à Agropecuária, Universidade Estadual Paulista Júlio de Mesquita Filho, Jaboticabal 14884-900, SP, Brazil

Ruby Chandna Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard, New Delhi 110062, India

M. C. Cia Departamento de Genética, Escola Superior de Agricultura Luiz de Queiroz, Universidade de São Paulo, Piracicaba 13418-900, SP, Brazil

Laura Coppola Dipartimento di Scienze Ambientali e delle Produzioni Vegetali, Università Politecnica delle Marche, Via Brecce Bianche, Ancona, Italy

Muhammad Farooq Department of Agronomy, University of Agriculture, Faisalabad 38040, Pakistan

Mario Fon Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain

Saddia Galani Khan Institute of Biotechnology and Genetic Engineering, University of Karachi, Karachi, Pakistan

Elisa Gamalero Dipartimento di Scienze dell'Ambiente e della Vita, Università del Piemonte Orientale, Viale Teresa Michel 11, Alessandria 15121, Italy

Nicole Geissler Institute of Plant Ecology, Justus Liebig University Giessen, Heinrich Buff-Ring 2632, 35392 Giessen, Germany

Bernard R. Glick Department of Biology University of Waterloo N2L 3G1, Waterloo, ON, Canada

Juan A. González Instituto de Ecología, Fundación Miguel Lillo, Miguel Lillo 251, CP 4000 Tucumán, Argentina

Deepmala Goswami Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

P. L. Gratão Departamento de Genética, Escola Superior de Agricultura Luiz de Queiroz, Universidade de São Paulo, Piracicaba 13418-900, SP, Brazil

Praveen Guleria Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

Dharmendra K. Gupta Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Cientificas (CSIC), Mail box 419, E-18080, Granada, Spain **Charles L. Guy** Department of Environmental Horticulture, University of Florida, Gainesville, FL 32611, USA

Khalid Rehman Hakeem Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard, New Delhi 110062, India

Mirna Hilal Cátedra de Fisiología Vegetal, Facultad de Ciencias Naturales e IML, Miguel Lillo 205, CP 4000 Tucumán, Argentina

Iqbal Hussain Department of Botany, University of Agriculture, Faisalabad 38040, Pakistan

Roque Interdonato Cátedra de Fisiología Vegetal, Facultad de Ciencias Naturales e IML, Miguel Lillo 205, CP 4000, Tucumán, Argentina

Muhammad Iqbal Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard, New Delhi 110062, India

Noushina Iqbal Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

Albena Ivanova Bulgarian Academy of Sciences, Institute of Plant Physiology, Acad. G. Bonchev str, BL. 21 1113, Sofia, Bulgaria

Md. Abdul Kader Department of Agronomy, Bangladesh Agricultural University Mymensingh, Mymensingh 2202, Bangladesh

Md. Iqbal R. Khan Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

Nafees A. Khan Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

Hans-Werner Koyro Institute of Plant Ecology, Justus Liebig University Giessen, Heinrich Buff-Ring 2632, 35392 Giessen, Germany

Vinay Kumar Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

Sylvia Lindberg Department of Botany, SU, SE-106 91 Stockholm, Sweden

Vinayak H. Lokhande Functional Plant Biology Section, Nuclear Agriculture and Biotechnology Division, Bhabha Atomic Research Centre, Mumbai 400 085, Maharashtra, India

Monika Mahajan Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

Vicente Martínez Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain **Liliana T. Maslenkova** Bulgarian Academy of Sciences, Institute of Plant Physiology, Acad. G. Bonchev str, BL. 21 1113, Sofia, Bulgaria

Asim Masood Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

A. J. Matilla Departamento de Fisiología Vegetal, Facultad de Farmacia, Universidad de Santiago de Compostela (USC), 15782, Santiago de Compostela, Spain

M.A. Matilla-Vázquez Department of Biochemistry, University of Cambridge, Tennis Court Road, Cambridge CB2 1QW, UK

Mohammad Miransari Department of Soil Science, Shahed University, College of Agricultural Sciences, 18151/159 Tehran, Iran

Rahat Nazar Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

Manuel Nieves-Cordones Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain

Maria Rosaria Panuccio Dipartimento di Biotecnologie per il Monitoraggio Agro-Alimentare ed Ambientale, Università Mediterranea di Reggio Calabria, Contrada Melissari – Lotto D, 89124 Reggio Calabria, Italy

Griselda Podazza Instituto de Ecología, Fundación Miguel Lillo, Miguel Lillo 251, CP 4000 Tucumán, Argentina

Losanka P. Popova Bulgarian Academy of Sciences, Institute of Plant Physiology, Acad. G. Bonchev str, BL. 21, 1113 Sofia, Bulgaria

Ron Porat Department of Postharvest Sciences of Fresh Produce, ARO, the Volcani Center, P.O. Box 6, Bet Dagan 50250, Israel

Carolina Prado Cátedra de Fisiología Vegetal, Facultad de Ciencias Naturales e IML, Miguel Lillo 205, CP 4000 Tucumán, Argentina

Fernando E. Prado Cátedra de Fisiología Vegetal, Facultad de Ciencias Naturales e IML, Miguel Lillo 205, CP 4000 Tucumán, Argentina

M.N.V. Prasad Department of Plant Sciences, University of Hyderabad, Prof. C.R. Rao Road, Gachibowli, Central University P.O., Hyderabad, AP 500 046, India

Edoardo Puglisi Istituto di Chimica Agraria ed Ambientale, Università Cattolica del Sacro Cuore, Via Emilia Parmense 84, Piacenza 29122, Italy **Feng Qin** Key Laboratory of Photosynthesis and Environmental Molecular Physiology, Institute of Botany, Chinese Academy of Sciences, Beijing 100049, China

Rizwan Rasheed Biology Department, Foreman Christian College, Lahore, Pakistan

Luis A. del Río Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Científicas (CSIC), Mail box 419, E-18080 Granada, Spain

Maria Rodríguez-Serrano Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Científicas (CSIC), Mail box 419, E-18080 Granada, Spain

Maria C. Romero-Puertas Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Científicas (CSIC), Mail box 419, E-18080 Granada, Spain

Mariana Rosa Cátedra de Fisiología Vegetal, Facultad de Ciencias Naturales e IML, Miguel Lillo 205, CP 4000 Tucumán, Argentina

Francisco Rubio Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain

Luisa M. Sandalio Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Científicas (CSIC), Mail box 419, E-18080 Granada, Spain

Agostino Sorgonà Dipartimento di Biotecnologie per il Monitoraggio Agro-Alimentare ed Ambientale, Università Mediterranea di Reggio Calabria, Contrada Melissari – Lotto D, 89124 Reggio Calabria, Italy

Zhivka Stoinova Bulgarian Academy of Sciences, Institute of Plant Physiology, Acad. G. Bonchev str, BL. 21, 1113 Sofia, Bulgaria

Penna Suprasanna Functional Plant Biology Section, Nuclear Agriculture and Biotechnology Division, Bhabha Atomic Research Centre, Mumbai 400 085, Maharashtra, India

Shabina Syeed Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India

Lam-Son Phan Tran Signaling Pathway Research Unit, Plant Science Center, RIKEN Yokohama Institute, 1-7-22, Suehiro-cho, Tsurumi, Yokohama 230-0045, Japan

Marco Trevisan Istituto di Chimica Agraria ed Ambientale, Università Cattolica del Sacro Cuore, Via Emilia Parmense 84, Piacenza 29122, Italy Narendra Tuteja Plant molecular Biology Group, International Centre for Genetic Engineering and Biotechnology, Aruna Asaf Ali Marg, New Delhi, India

Costantino Vischetti Dipartimento di Scienze Ambientali e delle Produzioni Vegetali, Università Politecnica delle Marche, Via Brecce Bianche, Ancona, Italy

Abdul Wahid Department of Botany, University of Agriculture, Faisalabad 38040, Pakistan

Haibo Xin Key Laboratory of Photosynthesis and Environmental Molecular Physiology, Institute of Botany, Chinese Academy of Sciences, Beijing 100049, China

Sudesh Kumar Yadav Plant Metabolic Engineering Laboratory, Biotechnology Division, Institute of Himalayan Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061, Himachal Pradesh, India

Vladislav Yemelyanov Department of Genetics and Breeding, St. Petersburg State University, St. Peterburg 199034, Russia

Dana Zoldan Department of Postharvest Sciences of Fresh Produce, ARO, the Volcani Center, P.O. Box 6, Bet Dagan 50250, Israel

Anna Źróbek-Sokolnik Department of Botany and Nature Protection, University of Warmia and Mazury in Olsztyn, Plac Łódzki 1, 10-727 Olsztyn, Poland

Abiotic Stress Responses in Plants: An Overview

Hans-Werner Koyro, Parvaiz Ahmad, and Nicole Geissler

Abstract

Plants are more and more affected by environmental stresses, especially by the devastating consequences of desertification and water scarcity which can be seen and felt all over the world. About 3.6 billion of the world's 5.2 billion hectares of dryland used for agriculture have already suffered erosion, soil degradation, and salinization. Desertification can hinder efforts for sustainable development and introduces new threats to human health, ecosystems, and national economies. This problem is catalyzed by global climate change which exacerbates desertification and salinization. Therefore, solutions are desperately needed, such as the improvement of drought and salinity tolerance of crops, which in turn requires a detailed knowledge about tolerance mechanisms in plants. These mechanisms comprise a wide range of responses on molecular, cellular, and whole plant levels, which include amongst others the synthesis of compatible solutes/osmolytes and radical scavenging mechanisms. Regarding global change, elevated atmospheric CO₂ concentrations can enhance salt and drought tolerance because oxidative stress is alleviated and more energy can be provided for energy-dependent tolerance mechanisms such as the synthesis of compatible solutes and antioxidants, thus increasing the suitability of plants as crops in future. A detailed knowledge of the physiological and biochemical basis of drought and salt tolerance and its interaction with elevated CO₂ concentration can provide a basis for the cultivation of suitable plants in regions threatened by desertification and water scarcity under sustainable culture conditions. Even the drylands could offer tangible economic and ecological opportunities.

H.-W. Koyro (⊠) • N. Geissler
Institute of Plant Ecology, Justus Liebig University
Giessen, Heinrich-Buff-Ring 26-32,
35392 Giessen, Germany
e-mail: Hans-Werner.Koyro@bot2.bio.uni-giessen.de

P. Ahmad

Department of Botany, Amar Singh College, Srinagar 190008, Jammu & Kashmir, India

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_1, © Springer Science+Business Media, LLC 2012

The aim of this chapter is to uncover how compatible solutes and antioxidants alleviate environmental stress, especially drought and salt stress, and the role elevated CO_2 concentrations can play in this context, so that early indicators allowing successful breeding can be identified and the potential of plants as crops in a CO, rich world can be assessed.

Keywords

Abiotic stress • Antioxidants • Osmolytes • Oxidative stress

1 Introduction

Plants are continuously affected by a variety of environmental factors. Whereas biotic environmental factors are other organisms such as symbionts, parasites, pathogens, herbivores, and competitors, abiotic factors include parameters and resources which determine plant growth like temperature, relative humidity, light, availability of water, mineral nutrients, and CO₂, as well as wind, ionizing radiation, or pollutants (Schulze et al. 2002). The effect each abiotic factor has on the plant depends on its quantity or intensity. For optimal growth, the plant requires a certain quantity of each abiotic environmental factor. Any deviation from such optimal external conditions, that is, an excess or deficit in the chemical or physical environment, is regarded as abiotic stress and adversely affects plant growth, development, and/or productivity (Bray et al. 2000). Abiotic stress factors include, for example, extreme temperatures (heat, cold, and freezing), too high or too low irradiation, water logging, drought, inadequate mineral nutrients in the soil, and excessive soil salinity. As especially drought and salt stress are becoming more and more serious threats to agriculture and the natural status of the environment, this chapter will focus on these stress factors. They are recurring features of nearly all the world's climatic regions since various critical environmental threats with global implications have linkages to water crises (Gleick 1994, 1998, 2000). These threats are collaterally catalyzed by global climate change and population growth.

The latest scientific data confirm that the earth's climate is rapidly changing. Due to rising concentrations of CO₂ and other atmospheric trace gases, global temperatures have increased by about 1°C over the course of the last century, and will likely rise even more rapidly in coming decades (IPCC 2007). Scientists predict that temperatures could rise by another 3-9°C by the end of the century with far-reaching effects. Increased drought and salinization of arable land are expected to have devastating global effects (Wang et al. 2003b). Abiotic stress is already the primary reason of crop loss worldwide, reducing average yields for most major crop plants by more than 50% (Bray et al. 2000; Wang et al. 2003b). It will soon become even more severe as desertification will further increase and the current amount of annual loss of arable area may double by the end of the century because of global warming (Evans 2005; Vinocur and Altman 2005). Simultaneously, rapid population growth increasingly generates pressure on existing cultivated land and other resources (Ericson et al. 1999). Population migration to those arid and semiarid areas increases the problems of water shortage and worsens the situation of land degradation in the destination, and in turn causes severe problems of poverty, social instability, and population health threats (Moench 2002). Water scarcity and desertification could critically undermine efforts for sustainable development, introducing new threats to human health, ecosystems, and national economies of various countries. Therefore, solutions to these problems are desperately needed, such as the improvement of salt and drought tolerance of crops, which in turn

requires a detailed knowledge about salt and drought tolerance mechanisms in plants.

The viability of plants in both dry and saline habitats depends on their ability to cope with (I) water deficit due to a low water potential of the soil and (II) restriction of CO_2 uptake. Plants growing on saline soils are additionally confronted with (III) ion toxicity and nutrient imbalance.

Water deficit (I) causes detrimental changes in cellular components because the biologically active conformation and thus the correct functioning of proteins and biomembranes depends on an intact hydration shell. As a consequence, severe osmotic stress can lead to an impairment of amino acid synthesis, protein metabolism, the dark reaction of photosynthesis or respiration and can cause the breakdown of the osmotic system of the cell (Larcher 2001; Schulze et al. 2002). Water deficit can be counteracted by compatible solutes, organic compounds which are highly soluble and do not interfere with cellular metabolism. They serve as a means for osmotic adjustment and also function as chaperons by attaching to proteins and membranes, thus preventing their denaturation. This protective function of compatible solutes can also alleviate ion specific effects of salt stress caused by ion toxicity and ion imbalance such as the precipitation of proteins due to changes in charge or the destruction of membranes caused by alterations of the membrane potential.

Regarding the restriction of CO₂ uptake (II), the negative effects of osmotic stress described earlier force plants to minimize water loss; growth depends on the ability to find the best tradeoff between a low transpiration and a high net photosynthetic rate (Koyro 2006). However, various plant species show a clearly reduced assimilation rate under osmotic stress conditions due to stomatal closure (Huchzermeyer and Koyro 2005). A consequence can be an excessive production of reactive oxygen species (ROS) which are highly destructive to lipids, nucleic acids, and proteins (Kant et al. 2006; Türkan and Demiral 2009; Geissler et al. 2010). However, generated ROS can be scavenged by the antioxidative system which includes nonenzymatic antioxidants and antioxidative enzymes (Blokhina et al. 2003).

Ion toxicity (III) on saline habitats is caused by ion specific effects on membranes and proteins: On the one hand, changes of the ionic milieu lead to alterations of the membrane potential and thus to a destruction of biomembranes (Schulze et al. 2002). On the other hand, the hydration and charge of proteins are negatively influenced, so that their precipitation is promoted, but their activity is reduced (Kreeb 1996). These effects of salt stress can be alleviated by the protective chaperone function of compatible solutes, similarly as explained above for osmotic stress.

When looking at drought and salt tolerance of plants in the face of global climate change, another important aspect should be considered: Compared to salinity and drought, elevated atmospheric CO₂ concentrations have contrary effects on plants: They often improve photosynthesis while reducing stomatal resistance in C₃ plants, thus increasing water use efficiency, but decreasing photorespiration and oxidative stress (Urban 2003; Kirschbaum 2004; Rogers et al. 2004). Furthermore, more energy can be provided for energy-dependent tolerance mechanisms such as the synthesis of compatible solutes and antioxidants. Therefore, the salt and drought tolerance and the productivity of these plants can be enhanced under elevated CO₂ (Ball and Munns 1992; Wullschleger et al. 2002; Urban 2003), increasing their future suitability as crops. Against the background described earlier, this review uncovers how compatible solutes and antioxidants alleviate environmental stress, especially drought and salt stress, and the role elevated CO_2 concentrations can play in this context.

2 Compatible Solutes Which Can Prevent Detrimental Changes Under Environmental Stress

Severe osmotic stress can cause detrimental changes in cellular components. The best characterized biochemical response of plant cells to osmotic stress is the accumulation of high concentrations of either organic ions or other low





molecular weight organic solutes termed compatible solutes. These compounds are highly soluble in water, electrically neutral in the physiological pH range, and noninhibitory to enzymes even at high concentrations, so that they do not interfere with essential metabolic (enzymatic) reactions (Rhodes et al. 2002). The structure of some important compatible solutes is shown in Fig. 1.1.

Organic solutes play a crucial role in higher plants grown under dry or saline conditions. However, their relative contribution varies among species, cultivars, and even between different compartments within the same plant (Ashraf and Harris 2004). A wide range of metabolites which can prevent these detrimental changes in cellular components have been identified, including mono-, di-, oligo-, and polysaccharides (glucose, fructose, sucrose, trehalose, raffinose, and fructans), sugar alcohols (mannitol, glycerol, and methylated inositols), quaternary amino acid derivatives (Pro, GB, β-alaninebetaine and prolinebetaine), tertiary amines (1,4,5,6-tetrahydro-2-mehyl-4-carboxyl pyrimidine), and sulfonium compounds (choline-O-sulphate, dimethylsulphoniopropionate) (Flowers and Colmer 2008; Vinocur and Altman 2005). The primary function of compatible solutes is to reduce water potential, to maintain turgescent cells, and to ensure balanced water relations (Wang et al. 2003a).

In addition, high concentration of compatible solutes exists primarily in the cytosol to balance the low water potentials achieved by high apoplasmic and vacuolar Na+ and Cl- concentration (Türkan and Demiral 2009). Recent studies indicate that compatible osmolytes also protect subcellular structures and mitigate oxidative damage caused by free radicals produced in response to salt stress (Slama et al. 2008; Smirnoff and Cumbes 1989). In many halophytes, organic osmolytes such as Pro or GB accumulate at suitably high concentrations to create osmotic potentials even below 0.1 MPa. In contrast to halophytes, in many glycophytes the concentrations of compatible solutes do not seem to be high enough to generate sufficiently low osmotic potentials (Türkan and Demiral 2009). This difference between halophytes and glycophytes can be used as an early indicator for salt resistance. Therefore, in the next chapters, the most important compatible solutes are described in detail.

2.1 Betaines

The quaternary ammonium compounds that function as effective compatible osmolytes in plants subject to salt stress are GB, β -alaninebetaine, prolinebetaine, choline-*O*-sulphate, hydroxyprolinebetaine, and pipecolatebetaine (Ashraf and



Harris 2004). GB occurs most abundantly in response to a variety of abiotic stress conditions by numerous organisms including bacteria, cyanobacteria, algae, fungi, animals, and many plant families such as Chenopodiaceae and Gramineae (Türkan and Demiral 2009). This metabolite is mainly located in chloroplasts and plays a vital role in the stroma adjustment and protection of thylakoid membranes, thereby maintaining the photosynthetic activity (Jagendorf and Takabe 2001). GB protects the photosystem II (PS-II) complex at high salinity (Murata et al. 1992) and at extreme temperatures or pH (Mohanty et al. 1993). GB also protects membranes against heat-induced destabilization and enzymes, such as RUBISCO, against osmotic stress (Mäkelä et al. 2000). In higher plants, GB is synthesized from serine via ethanolamine, choline by two-step oxidation reactions that were catalyzed by choline monooxygenase and betaine aldehyde dehydrogenase, respectively (Russell et al. 1998; Ahmad and Sharma 2008; see Fig. 1.2). The insertion of serine and glycine can be taken as an indicator for the close relationship of the photorespiration (peroxisomes) to the synthesis of GB. Besides this, recently a biosynthetic pathway of GB from glycine with the involvement of two N-methyl transferase enzymes has been reported (Waditee et al. 2005). Highly tolerant genera such as Spartina and Distichlis accumulated the highest levels of GB, moderately tolerant species intermediate levels, and sensitive species hardly any GB (Rhodes and Hanson 1993). Genetic evidence that GB improves salinity tolerance has been obtained for many important

agronomical crops such as tobacco, tomato, potato, barley, maize, and rice. These plants have long been a potential target for engineering GB biosynthesis pathway and thus for resistance against different environmental stress conditions (Sairam and Tyagi 2004; Türkan and Demiral 2009). The importance of N-methyltransferase for stress tolerance could also be shown for Arabidopsis. Genetically modified plants of this genus accumulated betaine to significant levels at different environmental stress conditions and hence improved seed yield (Waditee et al. 2005). A moderate stress tolerance was noted in some transgenic lines based on relative shoot growth in response to salinity, drought, and freezing. Huang et al. (2000) reported metabolic limitation in betaine production in transgenic plants. In fact, Arabidopsis thaliana, Brassica napus, and Nicotiana tabacum were transformed with bacterial choline oxidase cDNA, and their levels of GB were only between 5 and 10% of the levels found in natural betaine producers.

Beyond this, choline-fed transgenic plants synthesized substantially more GB. This result was taken as a hint that these plants require a distinct endogenous amount of choline to synthesize an adequate amount of GB (Sairam and Tyagi 2004).

The protective effect of GB at salinity or drought could also be demonstrated by exogenous application at rice seedlings, soybean, and common beans (Ashraf and Foolad 2007; Demiral and Türkan 2006). GB pretreatment also alleviated salinity-induced peroxidation (oxidative damage) of lipid membranes of rice cultivars. Besides rice,



the correlation between the protective effect of GB and the antioxidative defense system has been observed in chilling-stressed tomato (Park et al. 2006), drought- or salt-stressed wheat (Raza et al. 2007), and salt-stressed suspension cultured tobacco BY2 cells (Hoque et al. 2007).

2.2 Amino Acids, Proline, and Amides

It has been reported that amino acids (such as alanine, arginine, glycine, serine, leucine, and valine, the nonprotein amino acids citrulline and ornithine (Orn)), together with the imino acid Pro, and the amides such as glutamine and asparagine are accumulated in higher plants under salinity and drought stress (Dubey 1997; Mansour 2000). Pro is known to occur widely in higher plants and can be accumulated in considerable amounts in response to salt stress, water deficit, and other abiotic stresses (Ali et al. 1999; Kavi Kishore et al. 2005; Koca et al. 2007; Ahmad and Sharma 2008). The Pro concentration is metabolically controlled. This imino acid is synthesized in plastids and cytoplasm while degraded to L-glutamate (Glu) in mitochondria. There are two different precursors of Pro in plants: Glu and Orn (Fig. 1.3). Pro is synthesized from Glu via glutamic- γ -semialdehyde (GSA) and Δ^1 -pyrroline-5-carboxylate (P5C). P5C synthase (P5CS) catalyses the conversion of Glu to P5C, followed by P5C reductase (P5CR), which reduces P5C to Pro (Ashraf and Foolad 2007). The other precursor for Pro biosynthesis is Orn, which is transaminated to P5C by a mitochondrial Orn-y-aminotransferase (OAT) enzyme (Verbruggen and Hermans 2008). In the reverse reaction, Pro is metabolized to Glu in a feedback manner, via P5C and GSA with the aid of Pro dehydrogenase followed by P5C dehydrogenase (P5CDH) (Wang et al. 2003a).

The contribution of Glu and Orn pathways to stress-induced Pro synthesis differs between species, and it has been shown that stress-tolerant plants are able to accumulate Pro in higher concentrations than stress-sensitive plants. Slama et al. (2008) showed a positive correlation between Pro accumulation and tolerance to salt, drought, and the combined effects of these stresses. Osmotic stress (particularly mannitol stress) led to a considerable increase of the Pro concentration in the obligatory halophyte Sesuvium portulacastrum, while the contents in soluble sugars and in Na⁺ remained unchanged. In drought-stressed plants, the concentration of K⁺, Na⁺, Cl⁻, and Pro, as well as ornithine- δ aminotransferase (δ -OAT) activity increased significantly. Inversely, Pro dehydrogenase activity was impaired. Re-watering leads to a recovery of these parameters at values close to those of plants permanently irrigated with 100% of field capacity. The presence of NaCl and mannitol in the culture medium (ionic and osmotic stress) led to a significant increase of the Na⁺ and Pro concentration in the leaves, but it had no effect on leaf soluble sugar content. Slama et al. (2007a, b) assumed that the ability of NaCl to improve plant performance under mannitol-induced water stress is caused by an improved osmotic adjustment through Na⁺ and Pro accumulation, which is coupled with the maintenance of the photosynthetic activity. Similarly, the Pro concentration in the roots of salt tolerant alfalfa plants rapidly doubled under salt stress and was significantly higher than in salt sensitive genotypes (Petrusa and Winicov 1997). In addition to its role as an osmolyte for osmotic adjustment, Pro contributes to stabilizing subcellular structures (membranes and proteins) by forming clusters with water molecules which attach to proteins and membranes and prevent their denaturation (Koca et al. 2007; Ashraf and Foolad 2007; Lee et al. 2008). Due to its protective function on membranes it can also improve cell water status and ion homeostasis (Gadallah 1999; Gleeson et al. 2005), and it can scavenge free radicals and buffer cellular redox potential under stress conditions (Koca et al. 2007; Ashraf and Foolad 2007; Lee et al. 2008). Pro is also involved in alleviation of cytoplasmic acidosis and sustaining NADP⁺/NADPH ratios at required levels for metabolism (Hare and Cress 1997), thus supporting redox cycling (Babiychuk et al. 1995).

Transgenic approaches proved an enhancement of plant stress tolerance via overproduction of Pro. For instance, transgenic tobacco (N. tabacum), overexpressing the *p5cs* gene that encodes P5CS, produced 10- to 18-fold more Pro and exhibited better tolerance under salt stress (Kavi Kishor et al. 2005). In Arabidopsis, the overexpression of an antisense Pro dehydrogenase cDNA resulted in an increased accumulation of Pro and a constitutive tolerance to freezing and a higher salt tolerance (Nanjo et al. 2003). Similarly, Borsani et al. (2005) reported that the Arabidopsis P5CDH (Δ^1 -pyrroline-5-carboxylate dehydrogenase) and SRO5, an overlapping gene of unknown function in the antisense orientation, produced two types of siRNAs, 24-nt siRNA and 21-nt siRNA. In fact, they compared the levels of salt stress-induced Pro accumulation in various mutant plants (dcl2, sgs3, rdr6, and nrpd1a) which lacked SRO5-P5CDH nat-siRNAs and cleavage of the P5CDH transcript, Pro accumulation was not significantly induced by salt stress or was induced to a lesser extent than in the corresponding wild type. This result is consistent with their inability to downregulate P5CDH under stress, thereby causing a continued Pro catabolism and reduced Pro accumulation. In contrast, the dcl1 and rdr2 mutants, which were able to degrade P5CDH mRNA, had the same Pro level as the wild type under salt stress. The wild-type level of Pro accumulation in dcl1 indicates that although the 21-nt P5CDH natsiRNAs were not produced, the 24-nt SRO5-P5CDH nat-siRNA alone was sufficient to cause the downregulation of P5CDH (Fig. 1.4).

An alternative approach to improve plant stress tolerance is the exogenous application of Pro, which can lead to either osmoprotection or cryoprotection. For example, in various plant species growing under salt stress, among them the halophyte *Allenrolfea occidentalis*, exogenous application of Pro led to a higher osmoprotection and an increased growth (Yancey 1994).



2.3 **Sugars and Sugar Alcohols**

Several studies have been attempted to relate the magnitude of changes in soluble carbohydrates to salinity tolerance. Parida and Das (2005) found out that carbohydrates such as sugars (glucose, fructose, sucrose, and fructans) and starch are accumulated under salt stress. Furthermore, Megdiche et al. (2007) and Geissler et al. (2009a) proved that Cakile maritima and Aster tripolium plants accumulate high amounts of total soluble carbohydrates and Pro at high salinity (400 and 500 mM NaCl, respectively). The major functions of sugars and sugar alcohols are osmoprotection, osmotic adjustment, carbon storage, and radical scavenging (Adams et al. 2005; Ashraf et al. 2006; Messedi et al. 2006; Lee et al. 2008; Ahmad and Sharma 2008). Furthermore, there is a discussion about that they serve as molecular chaperones (Hasegawa et al. 2000; Liu et al. 2006).

There is a difference between starch and sugar accumulation in short- and long-term reaction (da Silva and Arrabaca 2004). In short-term water stress experiments, a decrease in sucrose and starch content was observed for Setaria sphacelata, a naturally adapted C₄ grass while in longterm experiments, a higher amount of soluble sugars and a lower amount of starch were found. da Silva and Arrabaca (2004) assumed that the shift of metabolism towards sucrose might occur because starch synthesis and degradation are more affected than sucrose synthesis.

Trehalose, a rare, nonreducing sugar, is present in several bacteria and fungi and in some desiccation-tolerant higher plants (Vinocur and Altman 2005). Under various abiotic stresses, the disaccharide trehalose accumulates in many organisms as an osmolyte and osmoprotectant, protects membranes and proteins in cells, and reduces the aggregation of denatured proteins

phased processing of SRO5-P5CDH nat-siRNAs and its role in a salt-stress regulatory loop (Borsani

(Ashraf and Harris 2004). In the transgenic plants, a comparatively moderate increase in trehalose levels lead to a higher photosynthetic rate and to a decrease in photooxidative damage during stress. Trehalose is thought to protect biological molecules from environmental stress (such as desiccation-induced damage), as suggested by its reversible water-absorption capacity (Penna 2003). It was shown that the contents of reducing and nonreducing sugars and the activity of sucrose phosphate synthase increase under salt stress, whereas starch phosphorylase activity decreases (Dubey and Singh 1999).

In general, the sugar alcohols are divided in acyclic (e.g., mannitol) and cyclic (e.g., pinitol) polyols. Polyols can make up a considerable percentage of all assimilated CO₂ and can have several functions such as compatible solutes, low molecular weight chaperones, and scavengers of stress-induced oxygen radicals (Bohnert et al. 1995). Polyols act in two indistinguishable ways, namely, osmotic adjustment and osmoprotection (Parida and Das 2005). In osmotic adjustment they act as osmolytes, facilitating the retention of water in the cytoplasm and enabling the sequestration of sodium into the vacuole or apoplast (cell wall). These osmolytes protect cellular structures by interacting with membranes, protein complexes, or enzymes. For instance, mannitol, a sugar alcohol that accumulates upon salt and water stress can alleviate abiotic stress. Transgenic wheat expressing the mannitol-1-phosphatase dehydrogenase gene (mtlD) of Escherichia coli was significantly more tolerant to water and salt stress (Abebe et al. 2003). Consequently, the transgenic wheat plants showed an increase in biomass, plant height, and number of secondary stems (tillers). The cyclic sugar alcohols pinitol and ononitol were accumulated in tolerant species such as the facultative halophyte Mesembryanthemum crys*tallinum* when exposed to salinity or water deficit (Bohnert and Jensen 1996). Pinitol can be synthesized from myoinositol by the sequential catalysis of inositol methyl transferase and ononitol epimerase. An inositol methyl transferase (Imt) cDNA was isolated from transcripts in M. crystallinum growing under saline conditions (Vernon and Bohnert 1992), and transgenic tobacco for Imt has been obtained (Vernon et al. 1993).

2.4 Polyamines

Under stressful conditions, different plant species respond differently towards levels of polyamines. Some might accumulate polyamines in response to stress, while others do not increase or even decrease their endogenous polyamine contents when exposed to harsh environments. It is proposed that PA play a defensive role during biotic stress responses (Walters 2003a, b). One of the examples is the hypersensitive response (HR) which consists of rapid cell death at the sight of pathogen entry, typically develops in the interaction between tobacco mosaic virus (TMV) and N resistance gene carrying N. tabacum and leads to enhanced polyamine synthesis and accumulation (Kusano et al. 2008). It is also believed that stress-induced polyamines tend to modulate the activity of a certain set of ion channels to adapt ionic fluxes in response to environmental changes. Many more examples of responses to biotic stress have been quoted by Kusano et al. (2008).

Various abiotic stress conditions have been reported to alter the concentration of polyamines (Bouchereau et al. 1999; Walters 2003a). Exogenous polyamine application and/or inhibitors of enzymes involved in polyamine biosynthesis pointed out a possible role of these compounds in plant adaptation/defense to several environmental stresses (Bouchereau et al. 1999; Alcázar et al. 2006; Groppa and Benavides 2008; Alcázar et al. 2010). More recent studies using either transgenic overexpression or loss-offunction mutants support this protective/adaptive/defensive role of PAs in plant response to abiotic stress (Alcázar et al. 2006; Kusano et al. 2008; Gill and Tuteja 2010). For example, Arabidopsis plants overexpressing Cucurbita ficifolia Spd synthase gene were tolerant of multistresses (chilling, freezing, salinity, drought, and paraquat toxicity) (Kusano et al. 2007; Tassoni et al. 2010). According to Rhee et al. (2007), the basic principle underlying polyamine adaptive responses appears to be shared by the prokaryotic stringent response and the eukaryotic unfolded protein response (UPR). UPR is triggered when unfolded proteins and uncharged tRNAs accumulate in the endoplasmic reticulum (ER) due to ER stress or nutrient starvation.

As a result of this, cap-dependent translation of many mRNAs is suppressed and the expression of a certain set of genes including the luminal binding protein gene BiP is induced. The underlying mechanisms of UPR in yeasts and mammals have been well researched (Rutkowski and Kaufman 2004), although those in plants have not (Kamauchi et al. 2005; Urade 2007; Kusano et al. 2008). Recently, nitric oxide (NO), an endogenous signaling molecule in plants and animals, has gained considerable importance in the PA studies. It is known to mediate responses to biotic and abiotic stresses. It has been reported by Tun et al. (2006) that spermine and spermidine are potent inducers of NO in Arabidopsis, but putrescine and its biosynthetic precursor arginine are not. There are many more examples of NO affecting the concentrations of PAs and over the past few years studies on polyamines and NO are gaining attention (Kusano et al. 2008).

3 Oxidative Stress and Antioxidative Responses to Environmental Stress

3.1 Production of ROS

Environmental stresses are responsible for the production of ROS. The production and removal of ROS is thought to be at equilibrium under normal conditions, whereas environmental stress disturbs this equilibrium by enhancing the production of ROS. ROS are very toxic for the organism as they affect the structure and function of the biomolecules. The main source of ROS production in plants is chloroplasts, mitochondria, and peroxisomes (Fig. 1.5).

Mitochondria are responsible for the generation of oxygen radicals and hydrogen peroxide due to the overreduction of the electron transport chain.



Fig. 1.5 Sites of reactive oxygen species (ROS) and the biological consequences leading to a variety of physiological dysfunctions that can lead to cell death (adopted from Ahmad et al. 2008)

Chloroplasts are found to be the major producer of O_2 and H_2O_2 (Davletova et al. 2005). This is because the oxygen pressure in the chloroplast is higher than in other organelles. Photoreduction of O_2 to O_2^{-} during the photosynthetic electron transport takes place and is called Mehler reaction. The production of superoxides is due to the reduction of molecular oxygen in the plastoquinone pool. This reduction may be due to the plastosemiquinone, by ferredoxin (Fd) or by sulfur redox centers in the electron transport chain within PSI (Dat et al. 2000). These superoxides are converted to hydrogen peroxide either spontaneously or by the action of the enzyme SOD. Hydrogen peroxide is also responsible for the production of hydroxyl radicals (OH[•]).

The major producer of H_2O_2 in plant cells are peroxisomes. It has been reported that peroxisomes are also responsible for the production of superoxides (O_2^{-}) . In peroxisomes, the production of O_2^{-} occurs in the peroxisomal matrix and the peroxisomal membrane. In the peroxisomal matrix, the oxidation of xanthine and hypoxanthine to uric acid in the presence of the enzyme xanthine oxidase generates O₂⁻radicals (Halliwell and Gutteridge 2000). Peroxisomes have got two pathways for the production of H₂O₂. One is the disproportionation of O₂⁻ generated in this organelle and the other is a direct pathway. During photorespiration glycolate is catalyzed by glycolate oxidase, yielding H_2O_2 . Fatty acid β -oxidation, the enzymatic reaction of flavin oxidases, can also produce H₂O₂ (Baker and Graham 2002; del Rio et al. 2002).

ROS include ${}^{1}O_{2}$, O_{2}^{-} , $H_{2}O^{+}$, $H_{2}O_{2}$, OH⁺, RO⁺ organic hydroperoxide (ROOH), excited carbonyl (RO⁺), etc. They cause damage to biomolecules like proteins, lipids, carbohydrates, and DNA, which ultimately results in cell death (Foyer and Noctor 2005). Fortunately, plants are equipped with an antioxidant machinery that scavenges the ROS and helps the plant to tolerate the stress conditions. The antioxidants include enzymatic antioxidants, viz., superoxide dismutase (SOD), catalase (CAT), ascorbate peroxidase (APX), glutathione reductase (GR), etc., and nonenzymatic antioxidants like ascorbic acid (AsA), vitamin E (α -tocopherol), reduced glutathione (GSH), etc.

3.2 Enzymatic Antioxidants

3.2.1 Superoxide Dismutase

SOD is one of the ubiquitous enzymes in aerobic organisms and plays a key role in cellular defense mechanisms against ROS. Within a cell, the SODs constitute the first line of defense against ROS. Its activity modulates the reactive amounts of O_2 ⁻⁻ and H_2O_2 , the two Haber–Weiss reaction substrates, and decreases the risk of OH radical formation, which is highly reactive and may cause severe damage to membranes, proteins, and DNA (reviewed by Ahmad et al. 2010b). SOD was for the first time reported by Cannon et al. (1987) in maize and it catalyzes the dismutation of superoxide into hydrogen peroxide and molecular oxygen.

$$2H^+ + 2^{\circ}O_2^- \rightarrow 2H_2O_2 + O_2$$

Different types of SOD isoforms have been observed in plants on the basis of metal cofactors attached to the active site. The isozyme containing Mn II at its active site is known as Mn-SOD. Similarly, the isozyme the active site of which contains Cu II and Zn II is known as Cu/Zn-SOD. The third isozyme contains Fe III and is referred to as Fe-SOD. The fourth SOD isoform contains Ni at the active site, is called Ni-SOD and is found in several Streptomyces species (Youn et al. 1996) and cyanobacteria (Palenik et al. 2003). Ni-SOD has not been reported in plants yet. Whereas only one type of SOD is found in most organisms, plants have multiple form of each type, which are encoded by more than one gene, indicating that plants have more complex antioxidant defense systems than other organisms.

Several studies have reported enhanced stress tolerance related to overproduction of chloroplastic SOD (Pastori and Foyer 2002). In maize leaves, GR and DHAR were exclusively localized in mesophyll cells whereas most of the SOD and APX were localized in mesophyll and bundle sheath cells. Increased SOD activity was reported in *Radix astragali* under water deficit stress, which varied in three different genotypes (Tan et al. 2006). Chilling stress has a significant effect in the enhancement of SOD activity in cucumber seedlings (Feng et al. 2003). The increase in SOD activity under drought stress was about 25% in soybean plants (Zhang et al. 2007). SOD activity was doubled in water stressed citrus plants when compared to well-watered control plants during seedling stage (Wu et al. 2006).

SOD activity increased under drought stress in Euphorbia esula (Davis and Swanson 2001), maize (Pastori et al. 2000), Cassia angustifolia (Agarwal and Pandey 2003), wheat (Singh and Usha 2003), rice (Wang et al. 2005), P. acutifolius (Turkan et al. 2005), and Camellia sinensis (Chen et al. 2011), and the SOD activity was higher under salinity stress in C. roseus (Jaleel et al. 2008) and Morus alba (Ahmad et al. 2010a). While subjecting higher plants to water deficit stress SOD activity increases (Reddy et al. 2004). Koca et al. (2007) have shown that elevated SOD activity is accompanied with an increase in the activity of major H_2O_2 scavenging enzymes like APX, CAT, and POX in salt tolerant sesame cultivar Cumhuriyat as compared to cultivar Orhangazi. SOD activity increased by 1.6-fold in a salt tolerant mutant of Chrysanthemum compared to a nontolerant one under NaCl stress (Hossain et al. 2006). An increased activity of SOD enzyme has also been reported under different abiotic stresses in Catharanthus roseus (Jaleel et al. 2007), Pisum sativum (Ahmad et al. 2008), M. alba (Ahmad et al. 2010a), and Brassica juncea (Ahmad 2010; Ahmad et al. 2011). SOD activity has also been observed to increase by the application of heavy metals such as cadmium (Shah et al. 2001; John et al. 2009; Ahmad et al. 2011), lead (Verma and Dubey 2003; John et al. 2009), and copper (Lombardi and Sebastiani 2005). Canola overexpressing Mn-SOD confers tolerance to aluminum stress (Basu et al. 2001). Overexpression of Mn-SOD in transgenic Arabidopsis showed a twofold increase in Mn-SOD activity and higher tolerance to salt as compared to nontransgenic plants (Wang et al. 2004). Tanaka et al. (1999) demonstrated that expression of yeast mitochondrial Mn-SOD in rice chloroplasts led to a 1.7fold increase in Mn-SOD as compared to nontransgenic plants. Transgenic Arabidopsis with Mn-SOD confers tolerance to heat (Im et al. 2009). Wang et al. (2005) demonstrated that transgenic rice plants expressing Mn-SOD have shown reduced injury and sustained photosynthesis under PEG stress. Overexpression of Cu/Zn-SOD and APX in transgenic tobacco enhanced seed longevity and germination rates after various environmental stresses (Lee et al. 2010). Transgenic tobacco expressing Cu/Zn-SOD have been shown to tolerate chilling and heat stress (Gupta et al. 1993) and enhanced tolerance to salt, water, and PEG stress (Badawi et al. 2004). Prashanth et al. (2008) have also demonstrated that Cu/Zn-SOD confers tolerance to salinity in rice plants.

3.2.2 Catalase

Plant catalases are tetrameric iron porphyrins and play a role in stress tolerance against oxidative stress. Catalases are produced in peroxisomes and glyoxysomes. Catalases are involved in eliminating hydrogen peroxide generated by different environmental stresses (Kim et al. 2008; Ahmad et al. 2010b). Catalases decompose hydrogen peroxide to water and molecular oxygen without consuming reductants and may thus provide plant cells with an energy efficient mechanism to remove hydrogen peroxide (reviewed by Ahmad et al. 2010b). The enzyme is abundant in the glyoxysomes of lipid-storing tissues in germinating barley, where it decomposes H₂O₂ formed during the β -oxidation of fatty acids (Jiang and Zhang 2002) and in the peroxisomes of the leaves of C_3 plants, where it removes H₂O₂ produced during photorespiration by the conversion of glycolate into glyoxylate (Kiani et al. 2008). This is also due to the fact that there is a proliferation of peroxisomes during stress, which might help in scavenging H₂O₂ which can diffuse from the cytosol (Lopez-Huertas et al. 2000; Kusaka et al. 2005).

High temperatures affect the structure of most proteins and thus the activity of many enzymes. Hertwig et al. (1992) have demonstrated that the translation of catalase was hampered at 40°C. Anderson (2002) showed that high temperature is responsible for the decrease in catalase activity in pepper plants. In comparison, the desert plant *Retama raetam* exposed to heat shock temperature showed only a minor inactivation of catalase activity (Streb et al. 1997). Scandalios et al. (2000) have also observed a reduced catalase activity in maize on exposure to temperatures of $35-40^{\circ}$ C.

Sublethal doses of NaCl induce catalase activity in *Nicotiana plumbaginifolia* through activation of *cat2* and *cat3* genes (Savoure et al. 1999). However, catalase activity was found to decrease due to the salt stress because of accumulation of salicylic acid (Shim et al. 2003). Vaidyanathan et al. (2003) have demonstrated that salt tolerant rice cultivars contain higher levels of catalase activity compared to susceptible cultivars. Increase in catalase activity during salt stress has also been shown by other workers in maize (Azevedo-Neto et al. 2006; Arora et al. 2008), in sesame (Koca et al. 2007), and in mulberry (Ahmad et al. 2010a).

Catalase activity has also been found to decrease in presence of heavy metal stress (Mishra et al. 2006; Khan et al. 2007; Mobin and Khan 2007; Ahmad et al. 2011). Verma and Dubey (2003) also demonstrated that the activity of catalase declines in rice plants with increasing concentration of Pb. John et al. (2009) also reported that an increase in Cd and Pb concentrations decreases the catalase activity in mustard. Decrease in catalase may be due to the inhibition of enzyme synthesis or change in assembly of enzyme subunits (Shah et al. 2001).

3.2.3 Ascorbate Peroxidase

APX is an important antioxidant enzyme mainly found in higher plants and algae (Raven 2003). APX helps to detoxify H_2O_2 in the ascorbateglutathione (= Halliwell-Asada) pathway. APX utilizes ascorbic acid and reduces H_2O_2 to water and monodehydroascorbate (MDA).

$$2AsA + H_2O_2 \rightarrow 2MDA + 2H_2O$$

APX was first isolated from chloroplasts and algae (Shigeoka et al. 1980; Nakano and Asada 1981). Different isoforms of APX which include thylakoid (tAPX), glyoxisomal (gmAPX), stromal (sAPX), and cytosolic (cAPX) have been reported (Shigeoka et al. 2002; Mittler et al. 2004). In comparison to other antioxidants, APX and guaiacol peroxidase (GPX) have a high affinity towards H₂O₂ (Mittler and Poulos 2005). APX isozymes have been found to be most stress responsive among the APX gene family during environmental stress (Mittler and Poulos 2005). APX1 has been found to enhance in response to environmental stress (Mittler 2002; Shigeoka et al. 2002). APX2 is expressed under stressful conditions and its expression is elevated in response to light stress or heat shock (Mullineaux and Karpinski 2002; Panchuk et al. 2002). Cytosolic APX1 has been found to protect Arabidopsis plants from a combination of stresses (Koussevitzky et al. 2008). Lu et al. (2007) demonstrated that cAPX improves salt tolerance in transgenic Arabidopsis.

Mittler et al. (1999) have demonstrated that suppression of APX1 in tobacco leads to a higher sensitivity of the plant to pathogen attacks. Overexpression of APX1 resulted in enhanced tolerance to oxidative stress in tobacco (Yabuta et al. 2002). Biologists have demonstrated the importance of APX1 by using APX1 knockout mutants. The plants lacking APX1 have showed delayed growth, no response of guard cells towards light, and light stress resulted in an induction of catalase and heat shock proteins (Pnueli et al. 2003). The accumulation of H_2O_2 is responsible for the abnormal closure of stomata in knockout APX1 plants (Pnueli et al. 2003). The induction of heat shock proteins in knockout APX1 plants may be due to an enhanced level of H₂O₂ which is considered as an essential signaling molecule during abiotic stress (Mittler 2002; Neill et al. 2002).

3.2.4 Glutathione Reductase

GR is a flavo-protein oxidoreductase and is found in both prokaryotes and eukaryotes (Romero-Puertas et al. 2006). GR is an important enzyme of the ascorbate–glutathione system and maintains the balance between reduced glutathione (GSH) and the ascorbate pool (reviewed by Ahmad et al. 2010b). Meldrum and Tarr (1935) for the first time reported GR in eukaryotes and yeast, and in 1951 it was also observed in plants (Conn and Vennesland 1951; Mapson and Goddard 1951). Later on GR has been isolated from different plants and bacteria (Creissen et al. 1991; Creissen and Mullineaux 1995). GR is mainly found in chloroplasts (70-80%), and small amounts have been found in mitochondria, cytosol, and peroxisomes (Edwards et al. 1990; Romero-Puertas et al. 2006). GR catalyzes the reduction of glutathione in the cell. GSH is oxidized to GSSG which should be converted back to GSH in normal cells. Rapid recycling of GSH is more important than the synthesis of GSH. Hence GR and GSH have been found to play a very crucial role in stress tolerance in plants. GR plays an important role in alleviating oxidative stress in plants as evidenced by increased activities of GR during oxidative stress (Contour-Ansel et al. 2006; Khan et al. 2007; Mobin and Khan 2007; Hsu and Kao 2007). Increased activities of GR during drought stress were observed in different plants, e.g. in wheat (Selote and Khanna-Chopra 2006) and in rice (Selote and Khanna-Chopra 2004; Srivalli et al. 2003; Sharma and Dubey 2005). Salt stress also increased the GR activity in rice (Demiral and Turkan 2005; Tsai et al. 2005) and wheat (Sairam et al. 2005). A positive correlation between the increased activity of GR and chilling tolerance has been reported in rice (Guo et al. 2006), in maize (Hodges et al. 1997a, b) and tomato (Walker and McKersie 1993). Heavy metal stress is also responsible for the increase in the activity of GR in plants. Mulberry plants exposed to copper stress exhibit an increased GR activity (Tewari et al. 2006). A positive correlation of increased GR activity in presence of Cd has been reported in potato, radish, soybean, sugarcane, and mustard (Stroinski et al. 1999; Vitoria et al. 2001; Ferreira et al. 2002; Fornazier et al. 2002; Mobin and Khan 2007). Transgenic tobacco plants expressing the gor2 gene from E. coli showed an increase in GR activity (Stevens et al. 2000). Pilon-Smith et al. (2000) observed that cytosol GR increases by 2-fold and chloroplast GR increases by 50-fold in transgenic plants of B. juncea expressing the gor2 gene from E. coli. These transgenic plants showed an enhanced tolerance to Cd stress up to 100 µM. Expression of the gor2 gene from E. coli in tobacco (cv. Belw3) showed an increased activity of GR and increases

tolerance to paraquat and H_2O_2 stress (Lederer and Böger 2003).

3.3 Nonenzymatic Antioxidants

3.3.1 Ascorbic Acid

Among the small molecular antioxidants in plants, ascorbate is most abundant and is most concentrated in leaves and meristems (reviewed by Ahmad et al. 2010b). It is about five to ten times more concentrated than GSH in leaves (Ishikawa et al. 2006). AsA is present in high concentration in fruits, especially citrus fruits, but the concentration in fruits is not always higher than in leaves (Davey et al. 2000). Some fruits such as blackcurrants and rose hips are famous for their exceptionally high ascorbate content (Ishikawa et al. 2006). AsA occurs in all subcellular compartments, and the concentration varies from 20 mM in the cytosol to 300 mM in chloroplasts (Noctor and Foyer 1998). The synthesis of AsA takes place in mitochondria and is transported to other cell compartments through a proton electrochemical gradient or through facilitated diffusion (Horemans et al. 2000). Franceschi and Tarlyn (2002) reported the presence of ascorbate in the phloem sap of A. thaliana. Other species of plants have also been reported to contain ascorbate in the phloem sap, e.g. cucurbita (Hancock et al. 2008). This led to the conclusion that ascorbate is transported from source (leaves) to sink (meristem) (Ishikawa et al. 2006).

Ascorbate plays an important role in plants as an antioxidant and as a cofactor of many enzymes (Ishikawa et al. 2006). As an antioxidant, ascorbate protects plants from oxidative stress. Ascorbate peroxidase utilizes ascorbic acid and reduces H_2O_2 to water, thereby generating monodehydroascorbate (MDA) in the ascorbate– glutathione cycle (Pan et al. 2003). MDA can also be reduced directly to AsA in the presence of the catalytic enzyme MDAR and the electron donor NADPH (Asada 1999). Maddison et al. (2002) have reported that ascorbate plays a role in the defense against ozone. AsA has the capability of donating electrons in various enzymatic and nonenzymatic reactions and is thus a powerful radical scavenger. It can directly scavenge ${}^{1}O_{2}$, O2-, and -OH radicals produced in the cell and can protect membranes against oxidative stress. In plant cells, the most important reducing substrate for H₂O₂ detoxification is ascorbic acid (Turkan et al. 2005). An increase in oxidized ascorbate during Cd stress has been reported by Demirevska-Kepova et al. (2006) in Hordeum vulgare. Yang et al. (2008) also reported that drought stress increases the ascorbate content in *Picea asperata*. Water stress results in significant increases in antioxidant AsA concentration in turfgrass (Zhang and Schmidt 2000; Vranova et al. 2002; Jaleel et al. 2007). Ascorbic acid shows a reduction under drought stress in maize and wheat, suggesting its vital involvement in oxidative response (Vertovec et al. 2001; Nayyar and Gupta 2006).

3.3.2 α-Tocopherol

Plants have the capacity to synthesize a lipophylic antioxidant known as α -tocopherol or vitamin E. α-tocopherol scavenges free radicals in combination with other antioxidants (Munne-Bosch and Algere 2003; Massacci et al. 2008). It has also been reported that α -tocopherol protects the structure and function of PSII as it chemically reacts with O₂ in chloroplasts (Lopez-Huertas et al. 2000; Nordberg and Arner 2001). Munne-Bosch and Algere (2003) reported that α -tocopherol helps in membrane stabilization and alleviates the tolerance of plants during oxidative stress. Environmental stresses are responsible for the generation of low molecular mass antioxidants such as α -tocopherol (Lowlor and Cornic 2002; Munne-Bosch and Algere 2003; Mahajan and Tuteja 2005; Martinez et al. 2007).

Falk et al. (2003) reported the upregulation of genes of α -tocopherol synthesis during oxidative stress. Water stress resulted in elevated levels of α -tocopherol in *Vigna* plants (Manivannan et al. 2007) and turfgrass (Zhang and Schmidt 2000).

3.3.3 Reduced Glutathione

Glutathione (L-glutamyl-L-cysteinylglycine, GSH) is a thiol compound composed of L-glutamic acid,

L-cysteine, and glycine. GSH is distributed universally in animals, plants, and microorganisms and has an established role as an essential compound of a free radical scavenger (Monneveux et al. 2006). GSH participates in numerous cellular processes and protects cells from the toxic effects of many ROS (Petropoulos et al. 2008). Additionally, GSH is involved in other biological functions, such as regulation of protein and DNA synthesis, protein activities, and maintaining membrane integrity (Cabuslay et al. 2002). Meyer et al. (2005) reported that levels of H_2O_2 are controlled by the action of glutathione. Reduction of glutathione (GSH) and oxidation of glutathione (GSSG) are necessary for controlling H₂O₂ levels in cells and have an important role in redox signaling (Pastori and Foyer 2002. Reduced glutathione is directly involved in the reduction of ROS in plants. Transgenic tobacco expressing glutathione gene withstands oxidative stress (Singh and Verma 2001).

Glutathione is a tripeptide (α -glutamyl cysteinylglycine) and is found in the cytosol, chloroplasts, ER, vacuoles, and mitochondria (Sankar et al. 2007a, b). The nonprotein thiols are nucleophilic in nature and thus are important for the formation of mercaptide bonds with metals and for reacting with selective electrophiles (Rodriguez et al. 2005). In most plants, the major source of these nonprotein thiols is glutathione. Glutathione is considered the most important nonenzymatic antioxidant due to its relative stability and high water solubility (Samarah 2005). It can protect plant cells from environmental stress-induced oxidative stress (Samarah 2005).

The Effect of Elevated Atmospheric CO₂ Concentration on Antioxidants and Osmolytes Under Environmental Stress

Elevated atmospheric CO_2 concentration leads to a higher CO_2 concentration gradient between the outside air and the intercellular spaces of the leaves, so that the diffusion of CO_2 into the leaves and the pCO_2/pO_2 ratio at the sites of photoreduction is increased (Robredo et al. 2007). Therefore, usually photorespiration and the rates of oxygen activation and ROS formation are reduced due to an increased NADPH utilization, whereas the net photosynthetic rate and thus the carbon supply is enhanced, especially in C₃ plants (Polle 1996; Urban 2003; Kirschbaum 2004; Long et al. 2004; Hikosaka et al. 2005; Ignatova et al. 2005). Furthermore, we often find a lower stomatal resistance (Hsiao and Jackson 1999; Li et al. 2003; Marchi et al. 2004; Rogers et al. 2004), which together with the higher net assimilation also leads to a better water use efficiency of photosynthesis (Amthor 1999; Morgan et al. 2001; Urban 2003). As a consequence of these effects, on the one hand there might be less need for antioxidants as elevated CO₂ ameliorates oxidative stress (Schwanz et al. 1996). On the other hand more energy can be provided for energy-dependent stress tolerance mechanisms such as the synthesis of osmolytes and antioxidants. Due to both effects mentioned earlier, elevated CO₂ can increase plant survival under abiotic stress conditions (Ball and Munns 1992; Rozema 1993; Drake et al. 1997; Fangmeier and Jäger 2001; Wullschleger et al. 2002; Urban 2003; Geissler et al. 2010).

Regarding oxidative stress, the antioxidant system can respond differently to elevated CO_2 depending on species or even genotype as well as on treatment duration and growth conditions such as mineral nutrition (Schwanz et al. 1996; Polle et al. 1997; Sanità di Toppi et al. 2002; Pérez-López et al. 2009). Varying responses can be related to a species-specific differential regulation in order to maintain an adequate balance between ROS formation and antioxidant ability under the actual conditions (Pérez-López et al. 2009). However, many studies have reported an increased tolerance to various abiotic stresses under elevated CO_2 due to an alleviation of oxidative stress:

In chestnut trees, photoinhibition due to high irradiance stress was ameliorated, and higher GSH levels were found in juvenile leaves (Carvalho and Amâncio 2002). Sgherri et al. (2000) reported that CO₂ enrichment led to an improved water use efficiency and a decreased photorespiration in *Medicago* sativa under drought stress. As a consequence, the cells showed a higher reducing status, increased ascorbate/dehydroascorbate and GSH/GSSG ratios. There was no demand for a higher GR activity (no CO₂ effect) and less requirement for Ca²⁺ ATPase activity to maintain Ca²⁺ homeostasis under stress conditions. Similarly, in cold stressed maize elevated CO₂ had no effect on SOD, CAT, and APX activities, but the formation of superoxide radicals and membrane injury was reduced (Baczek-Kwinta and Ko cielniak 2003). The alleviation of oxidative stress was probably due to a higher CO₂ assimilation, overcoming the Rubisco limitation under low temperature. In some cases elevated CO₂ even leads to reduced activities of antioxidative enzymes because there is less need for antioxidants: Pérez-López et al. (2009) reported that barley plants exposed to NaCl stress under ambient CO₂ exhibited enhanced activities of SOD, APX, CAT, GR, and dehydroascorbate reductase (DHAR), which was accompanied by ion leakage and lipid peroxidation. Furthermore, the expression ratio of enzyme isoforms changed, e.g. a relatively higher contribution of GR1 relative to GR2 and of Cu/Zn-SOD (which seems to be especially important for salt tolerance in Hordeum vulgare) was observed. Elevated CO₂ ameliorated ion leakage and lipid peroxidation, while the plants showed a lower upregulation of the antioxidant enzymes and an even higher relative contribution of GR1 and of Cu/Zn-SOD. The authors explain these results with less ROS generation and a better maintenance of redox homeostasis due to an enhanced photosynthesis and a reduced photorespiration. Similar results were found for Solanum lycopersicum by Takagi et al. (2009). NaCl salinity decreased plant biomass, net assimilation, and the transport of assimilates to the sink (stem), while CAT and APX activities increased. Under elevated CO₂ the negative effects of salinity were alleviated, especially when the sink activity was relatively high, and CAT and APX activities decreased compared to ambient CO₂. The improvement of oxidative stress (and of water relations) seemed to cause an activation of sink activity under elevated CO₂.



Fig. 1.6 Antioxidant enzyme expressions (relative volume percentages of the spots) in controls and salt treatments (75% seawater salinity) of *Aster tripolium* under ambient and elevated CO_2 . (a) Superoxide dismutase, (b) ascorbate peroxidase, (c) glutathione-*S*-transferase. Values represent mean±SD values of eight gels per

In contrast to the studies mentioned earlier, in some cases antioxidant activities are enhanced by elevated CO₂. In ozone stressed *Betula pendula*, elevated CO₂ eliminated the chloroplastic accumulation of H_2O_2 , which could be explained by a higher photosynthetic rate, leading to a higher NADPH formation and a more efficient enzymatic detoxification (e.g., via the ascorbate-glutathione cycle; Oksanen et al. 2005). Marabottini et al. (2001) found a higher activity of catalase (CAT), ascorbate peroxidase (APX), and superoxide dismutase (SOD) in drought-stressed Quercus under elevated CO₂. Rao et al. (1995) observed a more persistent high activity of glutathione reductase (GR), APX, und SOD in ozone-stressed wheat. Schwanz and Polle (2001) examined the drought tolerant species Quercus robur and the sensitive *Pinus pinaster*. They found out that Q. robur generally exhibits a higher activity of several antioxidative enzymes; furthermore, elevated CO₂ concentration ameliorated damage caused by drought stress in both species due to a higher stability of antioxidative enzymes and an enhanced SOD activity. Similar results were reported for the facultative halophyte A. tripolium. Under ambient CO₂ concentration

treatment. Significant differences ($P \le 0.05$) between the salinity treatments (within one CO₂ treatment) are indicated by *different letters*, significant differences between the CO₂ treatments (within one salt treatment) are indicated by an *asterisk. ctr* control, *sal* salt treatment

salt stress led to an overexpression and thus to higher relative activities of the antioxidative enzymes APX, SOD, and glutathione-Stransferase (GST), while under elevated CO_2 the expression and activities of these enzymes were further increased (Fig. 1.6; Geissler et al. 2010). Similarly, elevated CO₂ concentration led to a significantly higher content of carotenoids nonenzymatic antioxidants – in the salt treatments (Geissler et al. 2009b). These results implicate that the enhancement of enzyme expression and activity and the carotenoid content were not high enough to sufficiently eliminate ROS under ambient CO₂ concentration. Under elevated CO₂, however, a higher supply of energy-rich organic substances due to a significantly enhanced net assimilation rate (Geissler et al. 2009a, b) enabled the plants to invest more energy in the energydependent synthesis of enzymatic and nonenzymatic antioxidants. Therefore, ROS could be detoxified more effectively, so that salinity tolerance could be improved, manifesting itself in a higher survival rate of the salt-treated plants (Geissler et al. 2009a).

Furthermore, investigations about *A. tripolium* showed that elevated CO₂ concentration does not



Fig. 1.7 Content of osmolytes in controls and salt treatments (75% seawater salinity) of *Aster tripolium* under ambient and elevated CO_2 . (a) Proline, (b) total soluble carbohydrates. Values represent mean ± SD values of six measurements per treatment. Significant differences

 $(P \le 0.05)$ between the salinity treatments (within one CO₂ treatment) are indicated by *different letters*, significant differences between the CO₂ treatments (within one salt treatment) are indicated by an *asterisk. ctr* control, *sal* salt treatment

only have an effect on antioxidants, but on osmolytes as well. This halophyte employed its additional carbon gain under elevated CO_2 concentration also for a higher synthesis of compatible solutes (Geissler et al. 2009a): Salinity (under ambient CO_2) led to an accumulation of proline in all plant organs and of soluble carbohydrates in the main roots (Fig. 1.7). Under elevated CO_2 concentration, the plants accumulated a higher amount of proline, especially in the leaves which are the primary areas of influence of CO_2 . In the main root, there was no necessity of an additional accumulation of proline because this organ is well protected against salt damage due to a high content of compatible solutes even at ambient CO_2 concentration. Furthermore, a higher amount of soluble carbohydrates under elevated CO_2 was found in all plant organs due to the increased photosynthesis and a lower conversion of saccharides to starch. These results are in accordance with the study of Abdel-Nasser and Abdel-Aal (2002) investigating *Carthamus mareoticus* under drought stress: Elevated CO_2 concentration increased the accumulation of total soluble carbohydrates in well watered as well as in stressed plants due to a higher amount of assimilates. The drought-induced inhibition of the sucrose phosphate synthase activity was annihilated under elevated CO_2 , and the droughtinduced increase in sucrose content was further enhanced. The content of total amino acids and especially of proline behaved similarly to sucrose, as well as the activities of the proline synthesizing enzymes 1-pyrroline-5-carboxylate reductase (P5CR) and the ornithine aminotransferase (OAT). In contrast, the activity of the proline degrading enzyme proline dehydrogenase (PDH) was reduced by drought stress and further decreased under elevated CO_2 .

In contrast to C. mareoticus, proline (and other amino acids) do not seem to contribute to salt tolerance in barley, but to reflect a reaction to stress damage, as shown by Pérez-López et al. (2010): Although a better osmotic adjustment (more negative osmotic potential) of salt-stressed plants was recorded under elevated CO₂, the proline content decreased, showing less stress damage. Instead, the accumulation of soluble sugars and other unidentified osmolytes (possibly polyols and/or quaternary nitrogen compounds) was actively enhanced under elevated CO₂, and these substances played an important role in osmotic adjustment and as compatible solutes under saline conditions. Elevated CO₂ provided a higher carbon and ATP supply for salt tolerance mechanisms, enabling the plants to actively increase their compatible solute concentration, which in turn leads to a better water uptake and turgor maintenance for plant growth.

As a summary, it can be concluded that elevated CO_2 concentration can enhance salt and drought tolerance of plants by alleviating oxidative stress, increasing the activity of the antioxidative system, and/or increasing the accumulation of compatible substances, having a positive effect on their suitability as crops on dry and saline soils in future.

5 Conclusion and Future Perspective

Abiotic stresses, especially osmotic and ionic stresses, are responsible for the decrease in yield especially in arid and semiarid regions. It is estimated that 45% of the world's agricultural land

experience drought and 19.5% of the irrigated land are affected by salinity. These problems will be further catalyzed by global climate change. Prolonged environmental stresses are responsible for the production of ROS in different cell compartments like chloroplasts, mitochondria, peroxisomes, etc. ROS attack biomolecules, viz., DNA, lipids, proteins, carbohydrates, and disturb the normal functioning of the cell. Under severe stress conditions, ROS ultimately lead to cell death. In order to withstand oxidative stress, plants are equipped with enzymatic and nonenzymatic antioxidants. Many workers have reported the positive effects of SOD, CAT, APX, GR, MDHAR, AsA, glutathione, etc., in combating oxidative damage to the cell. To overcome the deleterious effects of abiotic stresses, plants also accumulate osmolytes and osmoprotectants such as proline and glycine betaine. These compounds are thought to play a role in osmotic adjustment and protect subcellular structures. Elevated atmospheric CO₂ concentration can alleviate oxidative stress, increase the activity of the antioxidative system, and/or increase the accumulation of compatible substances, so it can enhance salt and drought tolerance of plants and their suitability as crops in a future world of climate change.

The biggest challenge to the modern plant scientists is to develop stress-tolerant plants without compromising yield. There can be no doubt that transgenic plants will be invaluable in assessing precisely the role that main antioxidants, ROS, and osmolytes play in the functional network that controls stress tolerance. Researchers should look for defined sets of markers to predict tolerance towards a particular type of stress. While manipulating genes for stress tolerance in important crops, the genes incorporated should contribute to tolerance not only at a certain plant growth stage of interest but also at the whole plant level, because achieving maximum crop yield under saline conditions is the principal objective of all agriculturists. Modern techniques like genomics, proteomics, ionomics, and metabolomics will be helpful to study plant responses to abiotic stresses. Regarding global climate change, it would be desirable to develop model plants not only for understanding stress tolerance mechanisms, but
also their interaction with elevated atmospheric CO_2 concentration in order to assess the suitability of plants as crops in future.

Acknowledgments The authors would like to thank Mr. Jürgen Franz, Mr. Wolfgang Stein, Mr. Gerhard Mayer, Mrs. Angelika Bölke, Prof. Dr. Edwin Pahlich, PD Dr. Christian Zörb, Mrs. Anneliese Weber (Giessen University), and Mr. Steffen Pahlich (Zürich University) for technical assistance and scientific advice regarding the experiments with *Aster tripolium*.

References

- Abdel-Nasser LE, Abdel-Aal AE (2002) Effect of elevated CO₂ and drought on proline metabolism and growth of saf-flower (*Carthamus mareoticus* L.) seedlings without improving water status. Pak J Biol Sci 5:523–528
- Abebe T, Guenzi AC, Martin B, Cushman JC (2003) Tolerance of mannitol accumulating transgenic wheat to water stress and salinity. Plant Physiol 131:1748–1755
- Adams MA, Richter A, Hill AK, Colmer TD (2005) Salt tolerance in *Eucalyptus* spp.: identity and response of putative osmolytes. Plant Cell Environ 28:772–787
- Agarwal S, Pandey V (2003) Stimulation of stress-related antioxidative enzymes in combating oxidative stress in *Cassia* seedlings. Ind J Plant Physiol 8:264–269
- Ahmad P (2010) Growth and antioxidant responses in mustard (*Brassica juncea* L.) plants subjected to combined effect of gibberellic acid and salinity. Arch Agron Soil Sci 56:575–588
- Ahmad P, Sharma S (2008) Salt stress and phytobiochemical responses of plants. Plant Soil Environ 54:89–99
- Ahmad P, Sarwat M, Sharma S (2008) Reactive oxygen species, antioxidants and signaling in plants. J Plant Biol 51:167–173
- Ahmad P, Jaleel CA, Sharma S (2010a) Antioxidative defence system, lipid peroxidation, proline metabolizing enzymes and biochemical activity in two genotypes of *Morus alba* L. subjected to NaCl stress. Russ J Plant Physiol 57:509–517
- Ahmad P, Jaleel CA, Salem MA, Nabi G, Sharma S (2010b) Roles of enzymatic and non-enzymatic antioxidants in plants during abiotic stress. Crit Rev Biotechnol 30:161–175
- Ahmad P, Nabi G, Ashraf M (2011) Cadmium-induced oxidative damage in mustard [*Brassica juncea* (L.) Czern. & Coss.] plants can be alleviated by salicylic acid. S Afr J Bot 77:36–44
- Alcázar R, Marco F, Cuevas JC, Patrón M, Ferrando A, Carrasco P, Tiburcio AF, Altabella T (2006) Involvement of polyamines in plant response to abiotic stress. Biotechnol Lett 28:1867–1876

- Alcázar R, Altabella T, Marco F, Bortolotti C, Reymond M, Koncz C, Carrasco P, Tiburcio AF (2010) Polyamines: molecules with regulatory functions in plant abiotic stress tolerance. Planta 231:1237–1249
- Ali G, Srivastava PS, Iqbal M (1999) Proline accumulation, protein pattern and photosynthesis in regenerants grown under NaCl stress. Biol Planta 42: 89–95
- Amthor JS (1999) Increasing atmospheric CO₂ concentration, water use and water stress: scaling up from the plant to the landscape. In: Luo Y, Mooney HA (eds) Carbon dioxide and environmental stress. Academic, San Diego, pp 33–59
- Anderson JA (2002) Catalase activity, hydrogen peroxide content and thermotolerance of pepper leaves. Sci Hortic 95:277–284
- Arora N, Bhardwaj R, Sharma P, Arora HK (2008) Effects of 28-homobrassinolide on growth, lipid peroxidation and antioxidative enzyme activities in seedlings of Zea mays L under salinity stress. Acta Physiol Plant 30:833–839
- Asada K (1999) The water–water cycle in chloroplasts: scavenging of active oxygens and dissipation of excess photons. Annu Rev Plant Physiol Plant Mol Biol 50:601–639
- Ashraf M, Foolad MR (2007) Roles of glycine betaine and proline in improving plant abiotic stress resistance. Environ Exp Bot 59:206–216
- Ashraf M, Harris PJC (2004) Potential biochemical indicators of salinity tolerance in plants. Plant Sci 166:3–16
- Ashraf M, Hameed M, Arshad M, Ashraf Y, Akhtar K (2006) Salt tolerance of some potential forage grasses from Cholistan Desert of Pakistan. In: Khan MA, Weber DJ (eds) Ecophysiology of high salinity tolerant plants. tasks for vegetation science, vol 40. Springer, Dordrecht, pp 31–54
- Azevedo-Neto AD, Prisco JT, EnasFilho J, Braga de Abreu CE, GomesFilho E (2006) Effect of salt stress on antioxidative enzymes and lipid peroxidation in leaves and roots of salt tolerant and salt sensitive maize genotypes. Environ Exp Bot 56:87–94
- Babiychuk E, Kushnir S, Belles-Boix E, Van Montagu M, Inzé D (1995) Arabidopsis thaliana NADPH oxidoreductase homologs confer tolerance of yeasts towards the thiol-oxidizing drug diamide. J Biol Chem 270:26224–26231
- Baczek-Kwinta R, Ko cielniak J (2003) Anti-oxidative effect of elevated CO_2 concentration in the air on maize hybrids subjected to severe chill. Photosynthetica 41:161–165
- Badawi GH, Yamauchi Y, Shimada E, Sasaki R, Kawano N, Tanaka K, Tanaka K (2004) Enhanced tolerance to salt stress and water deficit by overexpressing superoxide dismutase in tobacco (*Nicotiana tabacum*) chloroplasts. Plant Sci 166:919–928
- Baker A, Graham I (2002) Plant peroxisomes, biochemistry, cell biology and biotechnological applications. Kluwer, Dordrecht

- Ball MC, Munns R (1992) Plant responses to salinity under elevated atmospheric concentrations of CO₂. Aust J Bot 40:515–525
- Basu U, Good AG, Taylor GJ (2001) Transgenic *Brassica* napus plants overexpressing aluminium-induced mitochondrial manganese superoxide dismutase cDNA are resistant to aluminium. Plant Cell Environ 24:1269–1278
- Blokhina O, Virolainen E, Fagerstedt KV (2003) Antioxidants, oxidative damage and oxygen deprivation stress: a review. Ann Bot 91:179–194
- Bohnert HJ, Jensen RG (1996) Strategies for engineering water stress tolerance in plants. Trends Biotechnol 14:89–97
- Bohnert HJ, Nelson DE, Jensen RG (1995) Adaptations to environmental stresses. Plant Cell 7:1099–1111
- Borsani O, Zhu J, Verslues PE, Sunkar R, Zhu JK (2005) Endogenous siRNAs derived from a pair of natural cis antisense transcripts regulate salt tolerance in *Arabidopsis*. Cell 123:1279–1291
- Bouchereau A, Aziz A, Larher F, Martin-Tanguy J (1999) Polyamines and environmental challenges: recent development. Plant Sci 140:103–125
- Bray EA, Bailey-Serres J, Weretilnyk E (2000) Responses to abiotic stresses. In: Buchanan BB, Gruissem W, Jones RL (eds) Biochemistry and molecular biology of plants. American Society of Plant Physiologists, Rockville, pp 1158–1203
- Cabuslay GS, Ito O, Alejal AA (2002) Physiological evaluation of responses of rice (*Oryza sativa L.*) to water deficit. Plant Sci 163:815–827
- Cannon RE, White JA, Scandalios JG (1987) Cloning of cDNA for maize superoxide dismutase (SOD-2). Proc Natl Acad Sci USA 84:179–183
- Carvalho LC, Amâncio S (2002) Antioxidant defence system in plantlets transferred from in vitro to ex vivo: effects of increasing light intensity and CO₂ concentration. Plant Sci 162:33–40
- Chen Q, Yang L, Ahmad P, Wan X, Hu X (2011) Proteomic profiling and redox status alteration of recalcitrant tea (*Camellia sinensis*) seed in response to desiccation. Planta 233:593–609
- Conn EE, Vennesland B (1951) Glutathione reductase of wheat germ. J Biol Chem 192:17–28
- Contour-Ansel D, Torres-Franklin M, Carvalho MHC, D'arcy-Lameta A, Zuily-Fodil Y (2006) Glutathione reductase in leaves of cowpea: cloning of two cDNAs, expression and enzymatic activity under progressive drought stress, desiccation and abscisic acid treatment. Ann Bot 98:1279–1287
- Creissen GP, Mullineaux PM (1995) Cloning and characterisation of glutathione reductase cDNAs and identification of two genes encoding the tobacco enzyme. Planta 197:422–425
- Creissen G, Edwards EA, Enard C, Wellbern A, Mullineaux P (1991) Molecular characterization of glutathione reductase cDNAs from pea (Pisum sativum L). Plant J 2:129–131
- da Silva JM, Arrabaca MC (2004) Contributions of soluble carbohydrates to the osmotic adjustment in the

C4 grass *Setaria sphacelata*: a comparison between rapidly and slowly imposed water stress. J Plant Physiol 161:551–555

- Dat J, Vandenabeele S, Vranova E, Van Montagu M, Inze D, Van Breusegem F (2000) Dual action of the active oxygen species during plant stress responses. Cell Mol Life Sci 57:779–795
- Davey MW, Montagu MV, Inze D, Sanmartin M, Kanellis A, Smirnoff N, Benzie IJJ, Strain JJ, Favell D, Fletcher J (2000) Plant L-ascorbic acid: chemistry, function, metabolism, bioavailability and effects of processing. J Sci Food Agric 80:825–860
- Davis DG, Swanson HR (2001) Activity of stress-related enzymes in the perennial weed leafy spurge (*Euphorbia esula L*.). Environ Exp Bot 46:95–108
- Davletova S, Rizhsky L, Liang H, Shengqiang Z, Oliver DJ, Coutu J, Shulaev V, Schlauch K, Mittler R (2005) Cytosolic ascorbate peroxidase 1 is a central component of the reactive oxygen gene network of *Arabidopsis*. Plant Cell 17:268–281
- del Río LA, Corpas FJ, Sandalio LM, Palma JM, Gómez M, Barroso JB (2002) Reactive oxygen species, antioxidant systems and nitric oxide in peroxisomes. J Exp Bot 53:1255–1272
- Demiral T, Turkan I (2005) Comparative lipid peroxidation, antioxidant defense systems and proline content in roots of two rice cultivars differing in salt tolerance. Environ Exp Bot 53:247–257
- Demiral T, Türkan I (2006) Exogenous glycinebetaine affects growth and proline accumulation and retards senescence in two rice cultivars under NaCl stress. Environ Exp Bot 56:72–79
- Demirevska-Kepova K, Simova-Stoilova L, Stoyanova ZP, Feller U (2006) Cadmium stress in barley: growth, leaf pigment, and protein composition and detoxification of reactive oxygen species. J Plant Nutr 29:451–468
- Drake BG, Gonzalez-Meler MA, Long SP (1997) More efficient plants. a consequence of rising atmospheric CO,? Annu Rev Plant Physiol 48:609–639
- Dubey RS (1997) Photosynthesis in plants under stressful conditions. In: Pessarakli M (ed) Handbook of photosynthesis. Marcel Dekker, New York, pp 859–875
- Dubey RS, Singh AK (1999) Salinity induces accumulation of soluble sugars and alters the activity of sugar metabolizing enzymes in rice plants. Biol Planta 42:233–239
- Edwards EA, Rawsthorne S, Mullineaux PM (1990) Subcellular distribution of multiple forms of glutathione reductase in leaves of pea (*Pisum sativum* L.). Planta 180:278–284
- Ericson J, Freudenberger M, Boege E (1999) Population dynamics, migration, and the future of the Calakmul Biosphere Reserve. American Association for the Advancement of Science, Washington
- Evans LT (2005) Is crop improvement still needed? J Crop Improv 14:1–7
- Falk J, Andersen G, Kernebeck B, Krupinska K (2003) Constitutive overexpression of barley 4-hydroxyphenylpyruvate dioxygenase in tobacco results in elevation

of the vitamin E content in seeds but not in leaves. FEBS Lett 540:35-40

- Fangmeier A, Jäger HJ (2001) Wirkungen erhöhter CO₂-Konzentrationen. In: Guderian R (ed) Handbuch der Umweltveränderungen und Ökotoxikologie. Volume 2a: Terrestrische Ökosysteme: Immissionsökologische Grundlagen – Wirkungen auf Boden – Wirkungen auf Pflanzen. Springer, Berlin, pp 382–433
- Feng Z, Guo A, Feng Z (2003) Amelioration of chilling stress by triadimefon in cucumber seedlings. Plant Growth Regul 39:277–283
- Ferreira RR, Fornazier RF, Vitoria AP, Lea PJ, Azevedo RA (2002) Changes in antioxidant enzyme activities in soybean under cadmium stress. J Plant Nutr 25:327–342
- Flowers TJ, Colmer TD (2008) Salinity tolerance in halophytes. New Phytol 179:945–963
- Fornazier RF, Ferreira RR, Pereira GJG, Molina SMG, Smith RJ, Lea PJ, Azevedo RA (2002) Cadmium stress in sugar cane callus cultures: Effect on antioxidant enzymes. Plant Cell Tissue Org Cult 71:125–131
- Foyer CH, Noctor G (2005) Redox homeostasis and antioxidant signaling: a metabolic interface between stress perception and physiological responses. Plant Cell 17:1866–1875
- Franceschi VR, Tarlyn NM (2002) L-Ascorbic acid is accumulated in source leaf phloem and transported to sink tissues in plants. Plant Physiol 130:649–656
- Gadallah MAA (1999) Effects of proline and glycinebetaine on *Vicia faba* responses to salt stress. Biol Plant 42:249–257
- Geissler N, Hussin S, Koyro HW (2009a) Interactive effects of NaCl salinity and elevated atmospheric CO₂ concentration on growth, photosynthesis, water relations and chemical composition of the potential cash crop halophyte *Aster tripolium* L. Environ Exp Bot 65:220–231
- Geissler N, Hussin S, Koyro HW (2009b) Elevated atmospheric CO₂ concentration ameliorates effects of NaCl salinity on photosynthesis and leaf structure of Aster tripolium L. J Exp Bot 60:137–151
- Geissler N, Hussin S, Koyro HW (2010) Elevated atmospheric CO₂ concentration enhances salinity tolerance in *Aster tripolium* L. Planta 231:583–594
- Gill SS, Tuteja N (2010) Polyamines and abiotic stress tolerance in plants. Plant Signal Behav 5:26–33
- Gleeson D, Lelu-Walter MA, Parkinson M (2005) Overproduction of proline in transgenic hybrid larch (*Larix x leptoeuropaea* (Dengler)) cultures renders them tolerant to cold, salt and frost. Mol Breed 15:21–29
- Gleick PH (1994) Water, war, and peace in the Middle East. Environment 36:7–41
- Gleick PH (1998) Water in crisis: Paths to sustainable water use. Ecol Appl 8:571–579
- Gleick PH (2000) The World's Water 2000–2001. The Biennial Report on Freshwater Recources. Island Press, Washington

- Groppa MD, Benavides MP (2008) Polyamines and abiotic stress: recent advances. Amino Acids 34:35–45
- Guo YP, Zhou HF, Zhang L (2006) Photosynthetic characteristics and protective mechanisms against photooxidation during high temperature stress in two citrus species. Sci Hortic 108:260–267
- Gupta AS, Heinen JL, Holaday AS, Burke JJ, Allen RD (1993) Increased resistance to oxidative stress in transgenic plants that overexpress chloroplastic Cu/Zn superoxide dismutase. Proc Natl Acad Sci USA 90:1629–1933
- Halliwell B, Gutteridge JMC (2000) Free radicals in biology and medicine, 3rd edn. Oxford University Press, Oxford
- Hancock RD, Chudek JA, Walker PG, Pont SDA, Viola R (2008) Ascorbic acid conjugates isolated from the phloem of *Cucurbitaceae*. Phytochemistry 69: 1850–1858
- Hare PD, Cress WA (1997) Metabolic implications of stress-induced proline accumulation in plants. Plant Growth Regul 21:79–102
- Hasegawa PM, Bressan RA, Zhu JK, Bohnert HJ (2000) Plant cellular and molecular responses to high salinity. Annu Rev Plant Physiol Plant Mol Biol 51:463–499
- Hertwig B, Streb P, Feierabend J (1992) Light dependence of catalase synthesis and degradation in leaves and the influence of interfering stress conditions. Plant Physiol 100:1547–1553
- Hikosaka K, Onoda Y, Kinugasa T, Nagashima H, Anten NPR, Hirose T (2005) Plant responses to elevated CO₂ concentration at different scales: leaf, whole plant, canopy, and population. Ecol Res 20:243–253
- Hodges DM, Andrews CJ, Johnson DA, Hamilton RI (1997a) Antioxidant enzyme responses to chilling stress in differentially sensitive inbred maize lines. J Exp Bot 48:1105–1113
- Hodges DM, Andrews CJ, Johnson DA, Hamilton RI (1997b) Antioxidant enzyme and compound responses to chilling stress and their combining abilities in differentially sensitive maize hybrids. Crop Sci 37: 857–863
- Hoque MA, Banu MNA, Okuma E, Amako K, Nakamura Y, Shimoishi Y, Murata Y (2007) Exogenous proline and glycinebetaine increase NaCl-induced ascorbateglutathione cycle enzyme activities, and proline improves salt tolerance more than glycinebetaine in tobacco Bright Yellow-2 suspension cultured cells. J Plant Physiol 164:1457–1468
- Horemans N, Foyer CH, Asard H (2000) Transport and action of ascorbate at the plant plasma membrane. Trends Plant Sci 5:263–267
- Hossain Z, Mandal AKA, Datta SK, Biswas AK (2006) Isolation of a NaCl tolerant mutant of *Chrysanthemum morifolium* by gamma radiation: in vitro mutagenesis and selection by salt stress. Funct Plant Biol 33:91–101
- Hsiao TC, Jackson RB (1999) Interactive effects of water stress and elevated CO₂ on growth, photosynthesis, and water use efficiency. In: Luo Y, Mooney HA (eds)

Carbon dioxide and environmental stress. Academic, San Diego, pp 3–31

- Hsu YT, Kao CH (2007) Cadmium-induced oxidative damage in rice leaves is reduced by polyamines. Plant Soil 291:27–37
- Huang J, Hijri R, Adam L, Rozwadowski KL, Hammerlindl JK, Keller WA, Selvaraj G (2000) Genetic engineering of glycinebetaine production toward enhancing stress tolerance in plants: metabolic limitations. Plant Physiol 122:747–756
- Huchzermeyer B, Koyro HW (2005) Salt and drought stress effects on photosynthesis. In: Pessarakli M (ed) Handbook of photosynthesis, 2nd edn. CRC, Boca Raton, pp 751–777
- Ignatova LK, Novichkova NS, Mudrik VA, Lyubimov VY, Ivanov BN, Romanova AK (2005) Growth, photosynthesis, and metabolism of sugar beet at an early stage of exposure to elevated CO₂. Russ J Plant Physiol 52:158–164
- Im YJ, Ji M, Lee A, Killens R, Grunden AM, Boss WF (2009) Expression of *Pyrococcus furiosus* superoxide reductase in Arabidopsis enhances heat tolerance. Plant Physiol 151:893–904
- IPCC (2007) Climate change 2007. The physical science basis. contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, NY
- Ishikawa T, Dowdle I, Smirnoff N (2006) Progress in manipulating ascorbic acid biosynthesis and accumulation in plants. Physiol Plant 126:343–355
- Jagendorf AT, Takabe T (2001) Inducers of glycinebetaine synthesis in barley. Plant Physiol 127:1827–1835
- Jaleel CA, Manivannan P, Kishorekumar A, Sankar B, Gopi R, Somasundaram R, Panneerselvam R (2007a) Alterations in osmoregulation, antioxidant enzymes and indole alkaloid levels in *Catharanthus roseus* exposed to water deficit. Colloids Surf B Biointerfaces 59:150–157
- Jaleel CA, Sankar B, Murali PV, Gomathinayagam M, Lakshmanan GMA, Panneerselvam R (2008) Water deficit stress effects on reactive oxygen metabolism in *Catharanthus roseus*; impacts on ajmalicine accumulation. Colloids Surf B Biointerfaces 62(1):105–111
- Jiang M, Zhang J (2002) Water stress-induced abscisic acid accumulation triggers the increased generation of reactive oxygen species and up-regulates the activities of antioxidant enzymes in maize leaves. J Exp Bot 53:2401–2410
- John R, Ahmad P, Gadgil K, Sharma S (2009) Cadmium and lead-induced changes in lipid peroxidation, antioxidative enzymes and metal accumulation in *Brassica juncea* L. at three different growth stages. Arch Agron Soil Sci 55:395–405
- Kamauchi S, Nakatani H, Nakano C, Urade R (2005) Gene expression in response to endoplasmic reticulum stress in *Arabidopsis thaliana*. FEBS J 272: 3461–3476

- Kant S, Kant P, Raveh E, Barak S (2006) Evidence that differential gene expression between the halophyte, *Thellungiella halophila*, and *Arabidopsis thaliana* is responsible for higher levels of the compatible osmolyte proline and tight control of Na⁺ uptake in *T. halophila*. Plant Cell Environ 29:1220–1234
- Kavi Kishor PB, Sangam S, Amrutha RN, Laxmi PS, Naidu KR, Rao KRSS, Reddy KJ, Theriappan P, Sreenivasulu N (2005) Regulation of proline biosynthesis, degradation, uptake and transport in higher plants: its implications in plant growth and abiotic stress tolerance. Curr Sci 88:424–438
- Khan N, Samiullah A, Singh S, Nazar R (2007) Activities of antioxidative enzymes, sulphur assimilation, photosynthetic activity and growth of wheat (*Triticum aestivum*) cultivars differing in yield potential under cadmium stress. J Agron Crop Sci 193:435–444
- Kiani SP, Maury P, Sarrafi A, Grieu P (2008) QTL analysis of chlorophyll fluorescence parameters in sunflower (*Helianthus annuus* L.) under well-watered and water-stressed conditions. Plant Sci 175: 565–573
- Kim YH, Kim CY, Song WK, Park DS, Kwon SY, Lee HS, Bang JW, Kwak SS (2008) Overexpression of sweetpotato swpa4 peroxidase results in increased hydrogen peroxide production and enhances stress tolerance in tobacco. Planta 227:867–881
- Kirschbaum MUF (2004) Direct and indirect climate change effects on photosynthesis and transpiration. Plant Biol 6:242–253
- Koca M, Bor M, Ozdemir F, Turkan I (2007) The effect of salt stress on lipid peroxidation, antioxidative enzymes and proline content of sesame cultivars. Environ Exp Bot 60:344–351
- Koussevitzky S, Suzuki N, Huntington S, Armijo L, Sha W, Cortes D, Shulaev V, Mittler R (2008) Ascorbate peroxidase 1 plays a key role in the response of *Arabidopsis thaliana* to stress combination. J Biol Chem 283:34197–34203
- Koyro HW (2006) Effect of salinity on growth, photosynthesis, water relations and solute composition of the potential cash crop halophyte *Plantago coronopus* (L.). Environ Exp Bot 56:136–146
- Kreeb KH (1996) Salzstreß. In: Brunold C, Rüegsegger A, Brändle R (eds) Streß bei Pflanzen. UTB, Bern, pp 149–172
- Kusaka M, Ohta M, Fujimura T (2005) Contribution of inorganic components to osmotic adjustment and leaf folding for drought tolerance in pearl millet. Physiol Plant 125:474–489
- Kusano T, Yamaguchi K, Berberich T, Takahashi Y (2007) The polyamine spermine rescues *Arabidopsis* from salinity and drought stresses. Plant Signal Behav 2:250–251
- Kusano T, Berberich T, Tateda C, Takahashi Y (2008) Polyamines: essential factors for growth and survival. Planta 228:367–381
- Larcher W (2001) Okophysiologie der Pflanzen, 5th edn. Eugen Ulmer-Verlag, Stuttgart

- Lederer B, Böger P (2003) Antioxidative responses of tobacco expressing a bacterial glutathione reductase. Z Naturforsch 58:843–849
- Lee G, Carrow RN, Duncan RR, Eiteman MA, Rieger MW (2008) Synthesis of organic osmolytes and salt tolerance mechanisms in *Paspalum vaginatum*. Environ Exp Bot 63:19–27
- Lee YP, Baek KH, Lee HS, Kwak SS, Bang JW, Kwon SY (2010) Tobacco seeds simultaneously over-expressing Cu/Zn-superoxide dismutase and ascorbate peroxidase display enhanced seed longevity and germination rates under stress conditions. J Exp Bot 61:2499–2506
- Li JH, Dugas WA, Hymus GJ, Johnson DP, Hinkle CR, Drake BG, Hungate BA (2003) Direct and indirect effects of elevated CO₂ on transpiration from *Quercus myrtifolia* in a scrub-oak ecosystem. Glob Change Biol 9:96–105
- Liu X, Duan D, Li W, Tadano T, Khan A (2006) A comparative study on responses of growth and solute composition in halophytes *Suaeda salsa* and *Limonium bicolor* to salinity. In: Khan MA, Weber DJ (eds) Ecophysiology of high salinity tolerant plants. Tasks for vegetation science, vol 40. Springer, Dordrecht, pp 135–143
- Lombardi L, Sebastiani L (2005) Copper toxicity in *Prunus cerasifera*: Growth and antioxidant enzymes responses of in vitro grown plants. Plant Sci 168: 797–802
- Long SP, Ainsworth EA, Rogers A, Ort DR (2004) Rising atmospheric carbon dioxide: plants FACE the future. Annu Rev Plant Biol 55:591–628
- Lopez-Huertas E, Charlton WL, Johnson B, Graham IA, Baker A (2000) Stress induces peroxisome biogenesis genes. EMBO J 19:6770–6777
- Lowlor DW, Cornic G (2002) Photosynthetic carbon assimilation and associated metabolism in relation to water deficits in higher plants. Plant Cell Environ 25:275–294
- Lu Z, Liu D, Liu S (2007) Two rice cytosolic ascorbate peroxidases differentially improve salt tolerance in transgenic. Arabidopsis 26:1909–1917
- Maddison J, Lyons T, Plochl M, Barnes J (2002) Hydroponically cultivated radish fed 1-galactono-1,4lactone exhibit increased tolerance to ozone. Planta 214:383–391
- Mahajan S, Tuteja N (2005) Cold, salinity and drought stresses an over view. Arch Biochem Biophys 444:139–158
- Mäkelä P, Kärkkäinen J, Somersalo S (2000) Effect of glycinebetaine on chloroplast ultrastructure, chlorophyll and protein content, and RUBISCO activities in tomato grown under drought or salinity. Biol Plant 43:471–475
- Manivannan P, Jaleel CA, Kishorekumar A, Sankar B, Somasundaram R, Sridharan R, Panneerselvam R (2007) Changes in antioxidant metabolism of *Vigna unguiculata* (L.) Walp. By propiconazole under water deficit stress. Colloids Surf B Biointerfaces 57:69–74
- Mansour MMF (2000) Nitrogen containing compounds and adaptation of plants to salinity stress. Biol Plant 43:491–500

- Mapson LW, Goddard DR (1951) The reduction of glutathione by plant tissues. Biochem J 49:592–601
- Marabottini R, Schraml C, Paolacci AR, Sorgona A, Raschi A, Rennenberg H, Badiani M (2001) Foliar antioxidant status of adult Mediterranean oak species (*Quercus ilex* L. and *Q. pubescens* Willd.) exposed to permanent CO₂-enrichment and to seasonal water stress. Environ Pollut 113:413–423
- Marchi S, Tognetti R, Vaccari FP, Lanini M, Kaligarič M, Miglietta F, Raschi A (2004) Physiological and morphological responses of grassland species to elevated atmospheric CO₂ concentrations in FACE-systems and natural CO₂ springs. Funct Plant Biol 31: 181–194
- Martinez JP, Silva H, Ledent JF, Pinto M (2007) Effect of drought stress on the osmotic adjustment, cell wall elasticity and cell volume of six cultivars of common beans (*Phaseolus vulgaris* L.). Eur J Agron 26:30–38
- Massacci A, Nabiev SM, Pietrosanti L, Nematov SK, Chernikova TN, Thor K, Leipner J (2008) Response of the photosynthetic apparatus of cotton (*Gossypium hirsutum*) to the onset of drought stress under field conditions studied by gas-exchange analysis and chlorophyll fluorescence imaging. Plant Physiol Biochem 46:189–195
- Megdiche W, Ben Amor N, Debez A, Hessini K, Ksouri R, Zuily-Fodil Y, Abdelly C (2007) Salt tolerance of the annual halophyte *Cakile maritima* as affected by the provenance and the developmental stage. Acta Physiol Plant 29:375–384
- Meldrum NU, Tarr HLA (1935) The reduction of glutathione by the Warburg-Christian system. Biochem J 29:108–115
- Messedi D, Slama I, Labidi N, Ghnaya T, Savouré A, Soltani A, Abdelly C (2006) Proline metabolism in *Sesuvium portulacastrum* under salinity and drought. In: Öztürk M, Waisel Y, Khan MA, Görk G (eds) Biosaline agriculture and salinity tolerance in plants. Birkhäuser, Basel, pp 65–72
- Meyer A, Hansen DB, Gomes CSG, Hobley TJ, Thomas ORT, Franzreb M (2005) Demonstration of a strategy for product purification by high-gradient magnetic fishing: recovery of superoxide dismutase from unconditioned whey. Biotechnol Prog 21:244–254
- Mishra S, Srivastava S, Tripathi RD, Govidarajan R, Kuriakose SV, Prasad MNV (2006) Phytochelatin synthesis and response of antioxidants during cadmium stress in Bacopa monnieri L. J Plant Physiol Biochem 44:25–37
- Mittler R (2002) Oxidative stress, antioxidants and stress tolerance. Trends Plant Sci 7:405–410
- Mittler R, Poulos TL (2005) Ascorbate peroxidase. In: Smirnoff N (ed) Antioxidants and reactive oxygen species in plants. Blackwell, Oxford, pp 87–100
- Mittler R, Hallak-Herr E, Orvar BL, Camp WV, Willekens H, Inzé D, Ellis BE (1999) Transgenic tobacco plants with reduced capability to detoxify reactive oxygen intermediates are hyper responsive to pathogen infection. Proc Natl Acad Sci USA 96:14165–14170
- Mittler R, Vanderauwera S, Gollery M, Van Breusegem F (2004) Reactive oxygen gene network of plants. Trends Plant Sci 9:490–498

- Mobin M, Khan NA (2007) Photosynthetic activity, pigment composition and antioxidative response of two mustard (*Brassica juncea*) cultivars differing in photosynthetic capacity subjected to cadmium stress. J Plant Physiol 164:601–610
- Moench M (2002) Water and the potential for social instability: livelihoods, migration and the building of society. Nat Resour Forum 26:195–204
- Mohanty P, Hayashi H, Papageorgiou GC, Murata N (1993) Stabilization of the Mn-cluster of the oxygenevolving complex by glycinebetaine. Biochim Biophys Acta 1144:92–96
- Monneveux P, Sánchez C, Beck D, Edmeades GO (2006) Drought tolerance improvement in tropical maize source populations: evidence of progress. Crop Sci 46:180–191
- Morgan JA, Lecain DR, Mosier AR, Milchunas DG (2001) Elevated CO_2 enhances water relations and productivity and affects gas exchange in C3 and C4 grasses of the Colorado shortgrass steppe. Glob Change Biol 7:451–466
- Mullineaux P, Karpinski S (2002) Signal transduction in response to excess light: getting out of the chloroplast. Curr Opin Plant Biol 5:43–48
- Munne-Bosch S, Algere L (2003) Drought induced changes in the redox state of α -tocopherol, ascorbate and the diterpene carnosic acid in the chloroplasts of labiatae species differing in carnosic acid contents. Plant Physiol 131:1816–1825
- Murata N, Mohanthy PS, Hayashi H, Papageorgiou GC (1992) Glycinebetaine stabilizes the association of extrinsic proteins with the photosynthetic oxygen involving PS-II complex against the inhibitory effects of NaCl. FEBS Lett 296:187–189
- Nakano Y, Asada K (1981) Hydrogen peroxide is scavenged by ascorbate peroxidase in spinach chloroplasts. Plant Cell Physiol 22:867–880
- Nanjo T, Fujita M, Seki M, Kato T, Tabata S, Shinozaki K (2003) Toxicity of free proline revealed in an *Arabidopsis* T-DNA-tagged mutant deficient in proline dehydrogenase. Plant Cell Physiol 44:541–548
- Nayyar H, Gupta D (2006) Differential sensitivity of C3 and C4 plants to water deficit stress: Association with oxidative stress and antioxidants. Environ Exp Bot 58:106–113
- Neill SJ, Desikan R, Clarke A, Hancock JT (2002) Nitric oxide is a novel component of abscisic acid signaling in stomatal guard cells. Plant Physiol 128:13–16
- Noctor G, Foyer CH (1998) Ascorbate and glutathione: keeping active oxygen under control. Annu Rev Plant Physiol Plant Mol Biol 49:249–279
- Nordberg J, Arner ESJ (2001) Reactive oxygen species, antioxidants, and the mammalian thioredoxin system. Free Radic Biol Med 31:1287–1312
- Oksanen E, Riikonen J, Kaakinen S, Holopainen T, Vapaavuori E (2005) Structural characteristics and chemical composition of birch (*Betula pendula*) leaves are modified by increasing CO₂ and ozone. Glob Change Biol 11:732–748

- Palenik B, Brahamsha B, Larimer FW, Land M, Hauser L, Chain P, Lamerdin J, Regala W, Allen EE, McCarren J, Paulsen I, Dufresne A, Partensky F, Webb EA, Waterbury J (2003) The genome of a motile marine *Synechococcus*. Nature 424:1037–1042
- Pan XY, Wang GX, Yang HM, Wei XP (2003) Effect of water deficits on within-plot variability in growth and grain yield of spring wheat in north west china. Field Crop Res 80:195–205
- Panchuk II, Volkov RA, Schoffl F (2002) Heat stress and heat shock transcription factor dependent expression and activity of ascorbate peroxidase in Arabidopsis. Plant Physiol 129:838–853
- Parida AK, Das AB (2005) Salt tolerance and salinity effects on plants: a review. Ecotoxicol Environ Saf 60:324–349
- Park EJ, Jeknic Z, Chen THH (2006) Exogenous application of glycinebetaine increases chilling tolerance in tomato plants. Plant Cell Physiol 47:706–714
- Pastori GM, Foyer CH (2002) Common components, networks, and pathways of cross- tolerance to stress. The central role of "redox" and abscisic acid-mediated controls. Plant Physiol 129:460–468
- Pastori GM, Mullineaux PM, Foyer CH (2000) Post transcriptional regulation prevents accumulation of glutathione reductase protein and activity in the bundle sheath cells of maize. Implication on the sensitivity of maize to temperatures. Plant Physiol 122:667–675
- Penna S (2003) Building stress tolerance through overproducing trehalose in transgenic plants. Trends Plant Sci 8:355–357
- Pérez-López U, Robredo A, Lacuesta M, Sgherri C, Muňoz-Rueda A, Navari-Izzo F, Mena-Petite A (2009) The oxidative stress caused by salinity in two barley cultivars is mitigated by elevated CO₂. Physiol Plant 135:29–42
- Pérez-López U, Robredo A, Lacuesta M, Sgherri C, Muňoz-Rueda A, Mena-Petite A (2010) Atmospheric CO₂ concentration influences the contributions of osmolyte accumulation and cell wall elasticity to salt tolerance in barley cultivars. J Plant Physiol 167:15–22
- Petropoulos SA, Daferera D, Polissiou MG, Passam HC (2008) The effect of water deficit stress on the growth, yield and composition of essential oils of parsley. Sci Hortic 115:393–397
- Petrusa LM, Winicov I (1997) Proline status in salt tolerant and salt sensitive alfalfa cell lines and plants in response to NaCl. Plant Physiol Biochem 35:303–310
- Pilon-Smith EAH, Zhu YL, Sears T, Terry N (2000) Overexpression of glutathione reductase in *Brassica juncea*: Effects on cadmium accumulation and tolerance. Physiol Plant 110:455–460
- Pnueli L, Liang H, Rozenberg M, Mittler R (2003) Growth suppression, altered stomatal responses, and augmented induction of heat shock proteins in cytosolic ascorbate peroxidase (Apx1)-deficient Arabidopsis plants. Plant J 34:187–203
- Polle A (1996) Protection from oxidative stress in trees as affected by elevated CO₂ and environmental stress.

In: Koch G, Mooney H (eds) Terrestrial ecosystem response to elevated CO_2 . Academic, New York, pp 299–315

- Polle A, Eiblmeier M, Sheppard L, Murray M (1997) Responses of antioxidative enzymes to elevated CO₂ in leaves of beech (*Fagus sylvatica* L.) seedlings grown under a range of nutrient regimes. Plant Cell Environ 20:1317–1321
- Prashanth SR, Sadhasivam V, Parida A (2008) Over expression of cytosolic copper/zinc superoxide dismutase from a mangrove plant Avicennia marina in indica Rice var Pusa Basmati-1 confers abiotic stress tolerance. Transgenic Res 17:281–291
- Rao MV, Hale BA, Ormrod DP (1995) Amelioration of ozone-induced oxidative damage in wheat plants grown under high carbon dioxide. Plant Physiol 109:421–432
- Raven EL (2003) Understanding functional diversity and substrate specificity in haem peroxidases; what can we learn from ascorbate peroxidase? Nat Prod Rep 20:367–381
- Raza SH, Athar HR, Ashraf M, Hameed A (2007) Glycinebetaine-induced modulation of antioxidant enzymes activities and ion accumulation in two wheat cultivars differing in salt tolerance. Environ Exp Bot 60:368–376
- Reddy AR, Chaitanya KV, Vivekanandan M (2004) Drought induced responses of photosynthesis and antioxidant metabolism in higher plants. J Plant Physiol 161:1189–1202
- Rhee HJ, Kim EJ, Lee JK (2007) Physiological polyamines: simple primordial stress molecules. J Cell Mol Med 11:685–703
- Rhodes D, Hanson AD (1993) Quaternary ammonium and tertiary sulfonium compounds in higher plants. Annu Rev Plant Physiol Plant Mol Biol 44:357–384
- Rhodes D, Nadolska-Orczyk A, Rich PJ (2002) Salinity, osmolytes and compatible solutes. In: Läuchli A, Lüttge U (eds) Salinity: environment – plants – molecules. Kluwer, Dordrecht, pp 181–204
- Robredo A, Pérez-López U, de la Maza HS, González-Moro B, Lacuesta M, Mena-Petite A, Muňos-Rueda A (2007) Elevated CO₂ alleviates the impact of drought on barley improving water status by lowering stomatal conductance and delaying its effects on photosynthesis. Environ Exp Bot 59:252–263
- Rodriguez P, Torrecillas A, Morales MA, Ortuno MF, Blanco MJS (2005) Effects of NaCl salinity and water stress on growth and leaf water relations of Asteriscus maritimus plants. Environ Exp Bot 53:113–123
- Rogers A, Allen DJ, Davey PA, Morgan PB, Ainsworth EA, Bernacchi CJ, Cornic G, Dermody O, Dohleman FG, Heaton EA, Mahoney J, Zhu XG, Delucia EH, Ort DR, Long SP (2004) Leaf photosynthesis and carbohydrate dynamics of soybeans grown throughout their life-cycle under Free-Air Carbon Dioxide Enrichment. Plant Cell Environ 27:449–458
- Romero-Puertas MC, Corpas FJ, Sandalio LM, Leterrier M, Rodríguez-Serrano M, del Río LA, Palma JM (2006) Glutathione reductase from pea leaves: response to abiotic stress and characterization of the peroxisomal isozyme. New Phytol 170:43–52

- Rozema J (1993) Plant responses to atmospheric carbon dioxide enrichment: Interactions with some soil and atmospheric conditions. Vegetatio 104(105):173–190
- Russell BL, Rathinasabapathi B, Hanson AD (1998) Osmotic stress induces expression of choline monooxygenase in sugar beet and amaranth. Plant Physiol 116:859–865
- Rutkowski DT, Kaufman RJ (2004) A trip to the ER: Coping with stress. Trends Cell Biol 14:20–28
- Sairam RK, Tyagi A (2004) Physiology and molecular biology of salinity stress tolerance in plants. Curr Sci 86:407–421
- Sairam RK, Srivastava GC, Agarwal S, Meena RC (2005) Differences in antioxidant activity in response to salinity stress in tolerant and susceptible wheat genotypes. Biol Plant 49:85–91
- Samarah NH (2005) Effects of drought stress on growth and yield of barley. Agron Sustain Dev 25:145–149
- Sanità di Toppi LS, Marabottini R, Badiani M, Raschi A (2002) Antioxidant status in herbaceous plants growing under elevated CO₂ in mini-FACE rings. J Plant Physiol 159:1005–1013
- Sankar B, Jaleel CA, Manivannan P, Kishorekumar A, Somasundaram R, Panneerselvam R (2007a) Effect of paclobutrazol on water stress amelioration through antioxidants and free radical scavenging enzymes in *Arachis hypogaea* L. Colloids Surf B Biointerfaces 60:229–235
- Sankar B, Jaleel CA, Manivannan P, Kishorekumar A, Somasundaram R, Panneerselvam R (2007b) Drought induced biochemical modifications and praline metabolism in *Abelmoschus esculentus* (L.) Moench. Acta Bot Croat 66:43–56
- Savoure A, Thorin D, Davey M et al (1999) NaCl and CuSO₄ treatments trigger distinct oxidative defense mechanisms in *Nicotiana plumbaginifolia*. Plant Cell Environ 22:387–396
- Scandalios JG, Acevedo A, Ruzsa S (2000) Catalase gene expression in response to chronic high temperature stress in maize. Plant Sci 156:103–110
- Schulze ED, Beck E, Müller-Hohenstein K (2002) Pflanzenökologie. Spektrum Akademischer, Heidelberg
- Schwanz P, Polle A (2001) Differential stress responses of antioxidative systems to drought in pendunculate oak (*Quercus robur*) and maritime pine (*Pinus pinaster*) grown under high CO₂ concentrations. J Exp Bot 52:133–143
- Schwanz P, Picon C, Vivin P, Dreyer D, Guehl J, Polle A (1996) Responses of antioxidative systems to drought stress in pendunculate oak and maritime pine as modulated by elevated CO₂. Plant Physiol 110:393–402
- Selote DS, Khanna-Chopra R (2004) Drought-induced spikelet sterility is associated with an inefficient antioxidant defence in rice panicles. Physiol Plant 121:462–471
- Selote DS, Khanna-Chopra R (2006) Drought acclimation confers oxidative stress tolerance by inducing co-ordinated antioxidant defense at cellular and subcellular level in leaves of wheat seedlings Physiol. Plant 127:494–506

- Sgherri CLM, Salvateci P, Menconi M, Raschi A, Navari-Izzo F (2000) Interaction between drought and elevated CO₂ in the response of alfalfa plants to oxidative stress. J Plant Physiol 156:360–366
- Shah K, Ritambhara GK, Verma S, Dubey RS (2001) Effect of cadmium on lipid peroxidation, superoxide anion generation and activities of antioxidant enzymes in growing rice seedlings. Plant Sci 161:1135–1144
- Sharma P, Dubey RS (2005) Drought induced oxidative stress and enhances the activities of antioxidant enzymes in growing rice seedlings. Plant Growth Regul 46:209–221
- Shigeoka S, Nakano Y, Kitaoka S (1980) Metabolism of hydrogen peroxide in *Euglena gracilis* Z by L-ascorbic acid peroxidase. Biochem J 186:377–380
- Shigeoka S, Ishikawa T, Tamoi M, Miyagawa Y, Takeda T, Yabuta Y, Yoshimura K (2002) Regulation and function of ascorbate peroxidase isoenzymes. J Exp Bot 53:1305–1319
- Shim IS, Momose Y, Yamamoto A, Kim DW, Usui K (2003) Inhibition of catalase activity by oxidative stress and its relationship to salicylic acid accumulation in plants. Plant Growth Regul 39:285–292
- Singh B, Usha K (2003) Salicylic acid induced physiological and biochemical changes in wheat seedlings under water stress. Plant Growth Regul 39:137–141
- Singh BG, Verma DPS (2001) Glutathione: An antioxidant to withstand oxidative stress in transgenic lines of tobacco. Ind J Plant Physiol 6:229–232
- Slama I, Ghnaya T, Hssini K, Messedi D, Savouré A, Abdelly C (2007a) Comparative study of mannitol and PE osmotic stress effects on growth, and solute accumulation in *Sesuvium portulacastrum*. Environ Exp Bot 61:10–17
- Slama I, Ghnaya T, Messedi D, Hssini K, Labidi N, Savoure A, Abdelly C (2007b) Effect of sodium chloride on the response of the halophyte species *Sesuvium portulacastrum* grown in mannitol-induced water stress. J Plant Res 120:291–299
- Slama I, Ghnaya T, Savouré A, Abdelly C (2008) Combined effects of long-term salinity and soil drying on growth water relations, nutrient status and proline accumulation of *Sesuvium portulacastrum*. C R Biol 331:442–451
- Smirnoff N, Cumbes QJ (1989) Hydroxyl radical scavenging activity of compatible solutes. Phyotochemistry 28:1057–1060
- Srivalli B, Chinnusamy V, Khanna-Chopra R (2003) Antioxidant defense in response to abiotic stresses in plants. J Plant Biol 30:121–139
- Stevens RG, Creissen GP, Mullineaux PM (2000) Characterization of pea cytosolic glutathione reductase expressed in transgenic tobacco. Planta 211: 537–545
- Streb P, Tel-Or E, Feierabend J (1997) Light stress effects and antioxidative protection in two desert plants. Funct Ecol 11:416–424
- Stroinski A, Kubis J, Zielezinska M (1999) Effect of cadmium on glutathione reductase in potato tubers. Acta Physiol Plant 21:201–207

- Takagi M, El-Shemy H, Sasaki S, Toyama S, Kanai S, Saneoka H, Fujita K (2009) Elevated CO₂ concentration alleviates salinity stress in tomato plant. Acta Agric Scand B Soil Plant 59:87–96
- Tan Y, Liang Z, Shao H, Du F (2006) Effect of water deficits on the activity of anti-oxidative enzymes and osmoregulation among three different genotypes of *Radix Astragali* at seeding stage. Colloids Surf B Biointerfaces 49:60–65
- Tanaka Y, Hibino T, Hayashi Y, Tanaka A, Kishitani S, Takabe T, Yokota S, Takabe T (1999) Salt tolerance of transgenic rice overexpressing yeast mitochondrial Mn-SOD in chloroplasts. Plant Sci 148:131–138
- Tassoni A, Franceschetti M, Bagni N (2010) Polyamines and salt stress response and tolerance in Arabidopsis thaliana flowers. Plant Physiol Biochem 46:607–613
- Tewari RK, Kumar P, Sharma PN (2006) Antioxidant responses to enhanced generation of superoxide anion radical and hydrogen peroxide in the copper-stressed mulberry plants. Planta 223:1145–1153
- Tsai YC, Hong CY, Liu LF, Kao CH (2005) Expression of ascorbate peroxidase and glutathione reductase in roots of rice seedlings in response to NaCl and H₂O₂. J Plant Physiol 162:291–299
- Tun NN, Santa-Catarina C, Begum T, Silveira V, Handro W, Floh EIS, Scherer GFE (2006) Polyamines induce rapid biosynthesis of nitric oxide (NO) in *Arabidopsis* thaliana seedlings. Plant Cell 47:346–354
- Türkan I, Demiral T (2009) Recent developments in understanding salinity tolerance. Environ Exp Bot 67:2–9
- Turkan I, Bor M, Ozdemir F, Koca H (2005) Differential responses of lipid peroxidation and antioxidants in the leaves of drought tolerant *P. acutifolius* Gray and drought sensitive *P. vulgaris* L. subjected to polyethylene glycol mediated water stress. Plant Sci 168:223–231
- Urade R (2007) Cellular response to unfolded proteins in the endoplasmic reticulum of plants. FEBS J 274: 1152–1171
- Urban O (2003) Physiological impacts of elevated CO₂ concentration ranging from molecular to whole plant responses. Photosynthetica 41:9–20
- Vaidyanathan H, Sivakumar P, Chakrabarty R, Thomas G (2003) Scavenging of reactive oxygen species in NaCl stressed rice (*Oryza sativa*) differential response in salt tolerant and sensitive varieties. Plant Sci 165: 1411–1418
- Verbruggen N, Hermans C (2008) Proline accumulation in plants: a review. Amino Acids 35:753–759
- Verma S, Dubey RS (2003) Lead toxicity induces lipid peroxidation and alters the activities of antioxidant enzymes in growing rice plants. Plant Sci 164: 645–655
- Vernon DM, Bohnert HJ (1992) A novel methyltransferase induced by osmotic stress in the facultative halophyte *Mesembryanthemum crystallinum*. EMBO J 11:2077–2085
- Vernon DM, Taraczynski MC, Jensen RG, Bohnert HJ (1993) Cyclitol production in transgenic tobacco. Plant J 4:199–205

- Vertovec M, Sakcali S, Ozturk M, Salleo S, Giacomich P, Feoli E, Nardini A (2001) Diagnosing plant water status as a tool for quantifying water stress on a regional basis in Mediterranean drylands. Ann For Sci 58: 113–125
- Vinocur B, Altman A (2005) Cellular basis of salinity tolerance in plants. Environ Exp Bot 52:113–122
- Vitoria AP, Lea PJ, Azevedo RA (2001) Antioxidant enzymes responses to cadmium in radish tissues. Photochem 57:701–710
- Vranova E, Inze D, Van Brensegem F (2002) Signal transduction during oxidative stress. J Exp Bot 53: 1227–1236
- Waditee R, Bhuiyan NH, Rai V, Aoki K, Tanaka Y, Hibino T, Suzuki S, Takano J, Jagendorf AT, Takabe T, Takabe T (2005) Genes for direct methylation of glycine provide high levels of glycine betaine and abiotic-stress tolerance in *Synechococcus* and *Arabidopsis*. Proc Natl Acad Sci USA 102:1318–1323
- Walker MA, McKersie BD (1993) Role of ascorbateglutathione antioxidant system in chilling resistance of tomato. J Plant Physiol 141:234–239
- Walters DR (2003a) Resistance to plant pathogens: possible roles for free polyamines and polyamine catabolism. New Phytol 159:109–115
- Walters DR (2003b) Polyamines and plant disease. Phytochem 64:97–107
- Wang HY, Huang YC, Chen SF, Yeh KW (2003a) Molecular cloning, characterization and gene expression of a water deficiency and chilling induced proteinase inhibitor I gene family from sweet potato (*Ipomoea batatas* Lam.) leaves. Plant Sci 165:191–203
- Wang W, Vinocur B, Altman A (2003b) Plant responses to drought, salinity and extreme temperatures: towards genetic engineering for stress tolerance. Planta 218:1–14
- Wang Y, Ying Y, Chen J, Wang XC (2004) Transgenic Arabidopsis overexpressing Mn-SOD enhanced salttolerance. Plant Sci 167:671–677

- Wang FZ, Wang QB, Kwon SY, Kwak SS, Su WA (2005) Enhanced drought tolerance of transgenic rice plants expressing a pea manganese superoxide dismutase. J Plant Physiol 162:465–472
- Wu QS, Zou YN, Xia RX (2006) Effects of water stress and arbuscular mycorrhizal fungi on reactive oxygen metabolism and antioxidant production by citrus (*Citrus tangerine*) roots. Eur Soil Biol 42:166–172
- Wullschleger SD, Tschaplinski TJ, Norby RJ (2002) Plant water relations at elevated CO₂-implications for water-limited environments. Plant Cell Environ 25:319–331
- Yabuta Y, Motoki T, Yoshimura K, Takeda T, Ishikawa T, Shigeoka S (2002) Thylakoid membrane-bound ascorbate peroxidase is a limiting factor of antioxidative systems under photo-oxidative stress. Plant J 32: 915–925
- Yancey PH (1994) Compatible and counteracting solutes. In: Strange K (ed) Cellular and molecular physiology of cell volume regulation. CRC, Boca Raton, pp 81–109
- Yang Y, Han C, Liu Q, Lin B, Wang J (2008) Effect of drought and low light on growth and enzymatic antioxidant system of *Picea asperata* seedlings. Acta Physiol Plant 30:433–440
- Youn HD, Kim EJ, Hah YC, Kang SO (1996) A novel nickel-containing superoxide dismutase from *Streptomyces* spp. Biochem J 318:889–896
- Zhang X, Schmidt RE (2000) Hormone containing products impact on antioxidant status of tall fescue and creeping bent grass subjected to drought. Crop Sci 40:1344–1349
- Zhang M, Duan L, Tian X, He Z, Li J, Wang B, Li Z (2007) Uniconazole-induced tolerance of soybean to water deficit stress in relation to changes in photosynthesis, hormones and antioxidant system. J Plant Physiol 164:709–717

Prospects of Halophytes in Understanding and Managing Abiotic Stress Tolerance

2

Vinayak H. Lokhande and Penna Suprasanna

Abstract

Halophytes are a diverse group of plants with tolerance to high salinity. While most of our crops are glycophytes lacking the genetic makeup for salt tolerance, halophytes are endowed with ability to seize NaCl into their cell vacuoles as an osmoticum. The sensitivity of crops to environmental extremities has become a major limitation to worldwide food production. Study of halophytes can be rewarding as the mechanisms by which halophytes survive and maintain productivity on saline water can be understood to define and manage adaptations in glycophytes. The adaptation mechanisms include ion compartmentalization, osmotic adjustment, succulence, ion transport and uptake, antioxidant systems, maintenance of redox and energetic status, and salt inclusion/excretion. Real benefits can be accrued if sustained efforts are in place to investigate the speciesspecific regulation during abiotic stresses and extend genetic resource and manipulate stress tolerance mechanisms. Halophytes are also an important plant species with potential for the purposes of desalination and restoration of saline soils, withstand high soil salinity and saline water irrigation, phytoremediation and wetland restoration. It will be invaluable to develop these strategies to ensure sustainability, and future efforts to improve crop performance on marginal and irrigated land.

Keywords

Halophytes • Abiotic stress • Compatible solutes • Antioxidants • Phytoremediation

Functional Plant Biology Section, Nuclear Agriculture and Biotechnology Division, Bhabha Atomic Research

P. Suprasanna (⊠) • V.H. Lokhande

Centre, Mumbai 400085, Maharashtra, India

e-mail: prasanna@barc.gov.in

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_2, © Springer Science+Business Media, LLC 2012

1 Introduction

Environment basically consists of balanced interaction between biotic and abiotic factors, and often abrupt perturbations in abiotic factors surrounding the biotic organisms lead to change the homeostasis, consequently creating a stressful condition for the survival of living organisms. Environmental stresses represent the most limiting factors for agricultural productivity. The abiotic stresses such as shade or high light levels, subzero, low or high temperatures, drought, flooding, high salinity, inorganic nutrient imbalance, infection, predation, and natural or man-made toxic compounds and oxidative stress (Bohnert and Sheveleva 1998) cause losses worth hundreds of million dollars each year due to reduction in crop productivity and crop failure (Zhu 2001; Flowers 2004). In view of their sessile nature, plants should have developed some adaptation strategies to manage the changing environmental conditions particularly with the available resources. Therefore, foremost adaptation carried out by terrestrial plants to its surrounding is adjustment in their water potential as low as that of soil in which they are able to grow. In the course of evolution, some plants have evolved and adapted to freshwater habitat for acquiring nutrients from the low concentrations of minerals present in fresh water such as glycophytes, whereas the plants which retained their habitat in nutrientrich marine environment were found more successful to combat the adverse abiotic stresses and are referred as "halophytes" (Flowers et al. 2010). These plants can be grown using land and water unsuitable for conventional crops and can provide food, fuel, fodder, fiber, resins, essential oils, and pharmaceutical feedstocks (Table 2.1).

2 Halophytes

Soil salinity and irrigated agriculture have coexisted since ancient times, and ever since the problem of salinity in agriculture has become a challenge. Soils are generally classified as saline when the electrical conductivity of the saturated paste extract (ECe) is 4 dS m⁻¹ or more (which is equivalent to 40 mM NaCl) and generate an osmotic pressure of approximately -0.2 MPa. Based on this, plants differ greatly in their growth response to saline conditions and therefore classified as "glycophytes" or "halophytes" referring to their capacity to grow on highly saline environments (Munns and Tester 2008). Halophytes are remarkable plants which have the ability to complete their life cycle in a substrate rich in NaCl that normally found toxic to other species and destroy almost 99% of their population (Flowers and Colmer 2008). These are highly evolved and specialized organisms with welladapted morphological, anatomical, and physiological characteristics allowing them to proliferate in the soils possessing high salt concentrations (Flowers et al. 1977; Flowers and Colmer 2008). Moreover, some halophytes consistently require a particular concentration of NaCl in the growth medium are referred as "obligate halophytes" or "true mangroves" and, apart from their growth in highly saline environment, some halophytes have capacity to grow on the soil devoid of salt are called as "facultative halophytes" or "mangrove associates." This presence or absence of substrate in the form of salt offers advantages for the halophytes in the competition with salt-sensitive plants (glycophytes) for the management of abiotic stress tolerance and utilization of these species for the improvement of crop yield.

In this regard, it is essential to understand the adverse effects of abiotic stresses and tolerance mechanisms developed by the halophytes and exploit such knowledge for the improvement of crop plants which can meet the demand of food, feed, fodder, and industrial raw material. The standard approach to this problem would be to increase the tolerance capacity of conventional crop plants, which otherwise are high yielders. An alternative strategy is to make use of halophytes that already have the required level of stress tolerance and are still productive at high external adverse conditions. Salinity is one of the major abiotic constraints, affecting almost every aspect of plant's physiology at both whole plant and cellular level through osmotic stress in an earlier phase and ionic stress at a later stage of

Uses	Plant species	
Food	A. hortensis, Aizoon canariense, Apium graveolens, Arundo donax, Atriplex halimus, Avicennia marina, Cocos nucifera, Cynamorium coccinium, Echinochloa crusgalli, Glinus lotoides, Glossonema varians, H. stocksii, Haloxylon griffithii ssp griffithii, N. schoberi, Neurada procumbens, Nitraria retusa, Ochradenus baccatus, Oxystelma esculentum, P. sylvestris, Pedalium murex, Pentatropis nivalis, Pheonix dactylifera, Pisonia grandis, Polypogon monspeliensis, Portulaca oleracea, Rumex vesicarius, S. brachiata, S. persica, Salicornia bigellovi, Salvadora oleoides, Sesuvium portulacastrum, Solanum incanum, Suaeda fruticosa, Triglochin maritime, Zizyphus nummularia, Zygophyllum simplex	
Fodder	 A. griffithii, A. halimus, A. leucoclada, A. tatarica, Aegiceras corniculatus, Alhaji maurorum, Anagallis arvensis, Artemisia scoparia, Arthrocnemum indicum, Atriplex canescens, Avicennia marina, B. glaucus, Beta vulgaris ssp maritma, Bienertia cycloptera, Bolboschoenus affinis, Caesalpinea bonduc, Camphorosma monspelictum, Carex divisa, Chloris virgata, Cressa cretica, Dalbergia sissoo, Glinus lotoides, Halocnemum strobila- ceum, Haloxylon stocksii, Lobularia maritime, Lolium multiflorum, Neurada procumbens, Orthochloa compressa, P. farcta, P. juliflora, Populus euphratica, Prosopis cineraria, Raphanus raphanistrum, Rhizophora mucronata, Salsola tragus, Seidlitzia florida, Seriphidium quettense, Suaeda fruticosa, T. repens, T. triquetra, Trianthema portulacastrum, Trifolium fragiferum, Vicia sativa, Zaleya pentandara, Zygophyllum simplex 	
Forage	 A. littorali, A. macrostachys, Aeluropus lagopoide, Agrostis stolonifera, Aristida adsceshoines, Aristida mutabilis, Atriplex dimorphostegia, C. ciliaris, C. pennesittiformis, Cenchrus biflorus, Chloris gayana, Cynodon dactylon, D. aristatum, D. scindicum, Dactyloctenium aegyptium, Desmostachya bipinnata, Dichantheum annulatum, Diplachne fusca, E. crusgalli, E. japonica, E. superba, Echinochloa colona, Eleusine indica, Eragrostis curvula, Festuca rubra, Halocharis hispida, Halopyrum mucronatum, Haloxylon persicum, Lasiurus scindicus, Nitraria retusa.Oligomeris linifolia, P. minor, P. pratensis, Paspalum pasploides, Phalaris arundinacea, Poa bulbosa, S. helvolus, S. ioclados, S. kentrophyllus, S. tourneuxii, S. tremulus, S. virginicus, Sacchraum bengalense, Salvadora persica, Sporobolus coromandelianus, Urochondra setulosa 	
Ornamental	 Achillea millefolium, Alhaji maurorum, Ammi visnaga, Artemisia scoparia, Avicennia marina, Caesalpinea bonduc, Calotropis procera, Camphorosma monspelictum, Cassia italic, Centella asiatica, Ceriops tagal, Chenopodium ambrosoides, Corchorus depressus, Cressa cretica, Cynamorium coccinium, Erythrina herbacea, Evolvulus alsinoides, Glinus lotoides, Halogeton glomeratus, Imperata cylindrical, Inula brittanica, Ipomoea alba, L. gilsei, L. sinuatum, L. stocksii, Leptadenia pyrotechnica, Limonium axillare, Melhania denhamii, Microcephala lamellate, Neurada procumbens, olanum surrattense, Oligomeris linifolia, Oxystelma esculentum, P. oleracea, Pedalium murex, Pentatropis nivalis, Populus euphratica, Portulaca quadrifida, Psylliostachys spicata, Rumex vesicarius, S. quettense, Seriphidium brevifolium, Solanum incanum, Sonneratia caseolaris, Thespesia populneoides, Trianthema portulacastrum, Tribulus terrestris, Urginea indica, Verbena officinalis, Withania sominifera, Z. simplex, Zaleya pentandara, Zygophyllum propinquum 	
Chemicals	Aeluropus lagopoides, Ardisia solanacea, Calotropis procera, Cenchrus ciliaris, Clerodendrum inerme, Dalbergia sissoo, Euphorbia thymifolia, Ficus microcarpa, Halocnemum strobilaceum, Ipomoea pes-caprae, K.iranica, Knorringia sibirica subsp. Kochia indica, Mesembryanthemum crystallinum, N. schoberi, Nitraria retusa, Phyla nodiflora, Polypogon monspeliensis, Raphanus raphanistrum, S. taccada, Scaevola plumier, Sessuvium sessuvioides, T. passernioides, T. ramosissima, T. szovitsiana, T. tetragyna, Tamarix mascatensis, Thomsonii, Trianthema portulacastrum	

 Table 2.1
 List of halophytes used for saline agriculture in Pakistan and other countries (modified from Khan and Qaiser 2006)

plant growth (Munns and Tester 2008) and leads to a series of morphological, physiological, biochemical, and molecular changes. In the past 2–3 decades, considerable progress has been made in the evaluation of halophytes to understand their survival mechanisms to be used as crop plants. In the present article, we document different aspects of halophytes, with an emphasis on mechanism of tolerance to salinity, drought and heavy metal tolerance, and their exploitation to manage the problems associated with the abiotic stresses as well as for environmental protection.

Halophytes respond to salt stress at cellular, tissue, and the whole plant level (Epstein 1980). In response to salt stress, the general physiology of halophytes has been reviewed occasionally (Flowers et al. 1977; Epstein 1980; Flowers 1985, 2004) and since then other reviews have examined their eco-physiology (Ball 1988; Rozema 1991; Breckel 2002), photosynthesis (Lovelock and Ball 2002), response to oxidative stress (Jitesh et al. 2006), and flooding tolerance (Colmer and Flowers 2008). Therefore, studies on the halophytes can be instructive from three prospects: first, the mechanism by which halophytes survive and maintain productivity under abiotic constraints can be used to define a minimal set of adaptations required in tolerant germplasm. This knowledge can help to focus the efforts of plant breeders and molecular biologists working with conventional crop plants (Glenn and Brown 1999). Second, halophytes grown in an agronomic setting can be used to evaluate the overall feasibility of high-salinity agriculture, which depends on more than finding a source of tolerant germplasm (Glenn et al. 1997). Third, halophytes may become a potential source of new crops.

3 Halophytes Diversity

Halophytes show immense diversity in habitat and behavior to tolerate the abiotic stress conditions with uneven distribution across the taxa of flowering plants (Flowers et al. 2010). This group of plants has been classified based on their tolerance capacity to salinity stress. Aronson (1989) listed approximately 1,550 species as salt-tolerant based on their capacity to tolerate the salt concentration more than 80 mM NaCl (equivalent to EC 7.8 dS m⁻¹), whereas, plants limiting the growth beyond this concentration were categorized as glycophytes or salt-sensitive. Using similar features, Menzel and Lieth (2003) recorded total 2,600 species as salt-tolerant. However, considering the salt tolerance limit proposed by Aronson (1989) and Menzel and Lieth (2003) which is found to be significantly lower than the salt concentration of seawater (~480 mM Na⁺ and 580 mM Cl⁻), Flowers and Colmer (2008) defined the halophytes as plants that have evolved and tolerate to complete their life cycle in at least ~200 mM NaCl. Applying the new definition to Aronson's database, Flowers et al. (2010) further classified a total of 350 species as halophytes with major species distributed in 20 orders including 256 families. It has also been suggested that salt tolerance was widely distributed among flowering plant families and had a polyphyletic origin. The authors also stated that distribution and development of evolutionary link of halophytes may account to not more than ~0.25% of the known species of angiosperms.

4 Adaptations to Abiotic Stresses

Halophytes have evolved a number of adaptive traits which allow them to germinate, grow, and achieve their complete life cycle of development under such harsh conditions (Flowers et al. 1977). A variety of studies performed on glycophytes and halophytes subjected to abiotic stresses has demonstrated that impairment in growth under stress condition results from various responses induced through both osmotic effects related to disturbance in plant-water relationships and ionic effects associated with mineral toxicity and deficiency (Lauchli and Epstein 1990). Associated with these primary stresses, higher plants also suffer from secondary stresses provoked by cellular damages especially those induced by oxidative stresses due to imbalance between production and destruction of reduced reactive oxygen species (Zhu 2001).

In order to achieve the tolerance status, three interconnected aspects of plant activity are significant: damage must be prevented, homeostatic condition must be re-established, and growth must resume. At present, there are different mechanisms of abiotic stress tolerance in halophytes that have been proposed which include ion compartmentalization, osmotic adjustment through osmolytes accumulation, succulence, selective transport and uptake of ions, enzymatic and nonenzymatic antioxidant response, maintenance of redox and energetic status, salt inclusion/excretion and genetic control (Flowers and Colmer 2008). Understanding the mechanism of tolerance in halophytes at morphological, anatomical, physiological, biochemical, and molecular levels is crucial to improve the tolerance of the crop plants and their adoption under abiotic stress conditions to exploit such problem soils. A generalized scheme for the plant's response to abiotic stresses and mechanism of stress tolerance is presented (Fig. 2.1).

4.1 Ion Compartmentation

The sensitivity of cytosolic enzymes to salt is similar in both glycophytes and halophytes, indicating that the maintenance of high cytosolic K⁺/ Na⁺ ratio is a key requirement for plant growth under salt conditions (Glenn and Brown 1999). While dealing with Na⁺, the cell must also acquire nutrient K⁺. The Na⁺ ion is the foremost inorganic ion and a cheap source of osmoticum in the halophytes to maintain the osmotic balance under abiotic stresses. Under typical physiological conditions, plant cells require high K⁺ (100-200 mM) and lower Na⁺ (less than 1 mM) and accordingly the high cytosolic K⁺/Na⁺ ratio to maintain the osmotic balance (Tester and Davenport 2003) for proper functioning of the cell. Na⁺ competes with K⁺ for intracellular influx since both these are transported by common channels present on the membranes and, thus, subsequently increase K⁺ efflux from intracellular stores as against the higher Na⁺ stress built up outside the cell. To maintain a high K⁺/Na⁺ ratio in the cytosol, plant cell employs primary active transport, mediated by channels and cotransporters for Na⁺ extrusion and/or the intracellular compartmentalization of Na+ into the vacuole (Blumwald 2000). When halophytes are exposed to saline condition, a large increase in extracellular Na⁺ level establishes the Na⁺ electrochemical potential gradient more than the actual negative electrical membrane potential difference at the plasma membrane (-140 mV)which favor the passive transport of sodium ions from the outer environment inside the cell. Recently, uniporter or ion channel type transporters have been identified for the entry of Na⁺ into the cell; these are high-affinity potassium transporter (HKT), low-affinity cation transporter (LCT1), nonselective cation channels (NSCC) like cyclic nucleotide-gated channels (CNGCs) and glutamate-activated channels (GLRs) (Apse and Blumwald 2007). HKTs have been shown to function as Na⁺/K⁺ symporter and as Na⁺ selective uniporters (Horie and Schroeder 2004). In the process of elevated levels of Na⁺ outside the cell, the electrochemical gradient makes the sodium uptake passive; however, the efflux of Na⁺ outside cell is an active process and requires energy in the form of ATP. In this process, the Na⁺/H⁺ antiporter (NHX) present on the plasma membrane facilitates the Na⁺ efflux. This electroneutral exchange of sodium for protons to facilitate efflux is the only mode of transport that has been measured for efflux under physiological conditions (Apse and Blumwald 2007). Besides the efflux of Na⁺, some halophytes have developed mechanism to sequester the Na⁺ into the vacuoles as an efficient mechanism to avoid the toxic effects of Na⁺ in the cytosol. The transport of Na⁺ into the vacuoles is mediated by cation/H⁺ antiporters that are driven by the electrochemical gradient of protons generated by the vacuolar H⁺ translocating enzymes such as H+-ATPase and H⁺-PPiase (Gaxiola et al. 2007). These transporters play an important role in the sequestration of Na⁺ ions into the vacuole or exclusion outside the cell of the halophytes.



Abiotic stresses

 Maintenance of redox homeostasis and energetics: GSH/GSSG,ASC/DHA, NADP/NADPH and ATP/ADP ratios

Û

Re-establishment of cellular homeostasis, protection of functional and structural proteins and membranes

Stress tolerance to abiotic constraints

Fig. 2.1 Generalized scheme for plant responses to abiotic stresses and mechanism of stress tolerance. Plants exposed to various abiotic stresses (salinity, drought, extreme temperatures, toxic metals, etc.) initiate the cascade of changes in plants' functioning such as imbalanced water and nutrient uptake, stomatal closure, altered gaseous exchange, improper functioning of photosynthetic systems due to over-reduction of electron transport chains in chloroplast and mitochondria causing generation of reactive oxygen species (ROS). The integrative effect of these factors leads to induce the oxidative damage to functional and structural molecules (DNA, proteins, lipids, and carbohydrates) making the changes in the redox, osmotic,

ionic, and energetic homeostasis of the plant. These stress signals triggers the downstream signaling processes and gene activation through transcription factors. Activation mechanisms involve enzymatic and nonenzymatic antioxidants for detoxification of ROS, osmolytes (proline, glycine betaine, sugar polyols) synthesis for osmotic balance and protection to structural molecules, ion compartmentation for ionic homeostasis and maintenance of redox and energetics through the higher ratios of GSH/GSSG, ASC/DHA, NADP/NADPH, and ATP/ADP. The coordinated action leads to re-establish the cellular homeostasis, protection of functional and structural proteins and membranes, and ultimately the tolerance to abiotic stresses

4.2 Succulence

Succulence is commonly called as halosucculence and found to occur within a range of salt concentrations optimal for growth. The sequestration of saline ions into the vacuoles leads to the plant to increase the succulence, one of the common characteristics of the halophytes (Flowers et al. 1977). Succulence minimizes the toxic effects of excessive ion accumulation and has been reported to be associated with accretion of osmotically active solutes for the maintenance of cell turgor pressure. The succulent halophytes unlike glycophytes tend to accumulate sodium in the vacuole to higher levels than in the cytoplasm and as the volume of the vacuole is much greater than that of the cytoplasm in fully expanded cells, the total sodium content of the root will approximate to the sodium content of the vacuole (Yeo and Flowers 1986). Succulent halophytes generally have thick leaves and stems, mainly associated with an increase in the size of their mesophyll cells along with smaller intracellular spaces. It has also been shown that succulent leaves have more and large-sized mitochondria because the succulent halophytes require excess energy for salt compartmentalization and excretion. Whether succulence is a response to salinity or adaptation to salinity is debatable. But as halophytes tend to become succulent in response to salinity (due to physiologically less available water which affects the changes in the integral part of the plant development), the succulence might be the adaptation to salinity stress (Waisel 1972). This adaptive nature of succulence made the halophytes more successful in the course of evolution exposing to various environmental constraints. Most of the halophytes such as Sesuvium portulacastrum, Suaeda sps., Lobularia maritime, Mesembryanthemum crystallinum, Halosarcia pergranulata subsp. Pergranulata, etc. were found more amenable to accumulate the excess Na⁺ in their leaves and stems and increase the succulence under optimum NaCl concentrations in the range of 100-400 mM which leads to sequester these saline ions into the vacuole and become more successful for their growth in saline environment (Qi et al. 2009; Lokhande et al.

2010a). Thus it appears that increased succulence could be due to a "diluting" effect on the ion content of cells which might otherwise rise to toxic levels, and sodium acts as a specific stimulant of growth which can be considered as tending to reduce the turgor pressure component of the water potential of the cell (Jennings 1968).

The succulent halophytes are able to balance the growth and ion accumulation through its sequestration into the vacuole; however, some of the halophytes were adapted to saline environment through secretion of salts from salt glands, cuticles or in guttation fluid, re-transported back to the roots and soil via the phloem or become concentrated in salt hairs. Salt glands act as transient cells because they are devoid of vacuole and have a large number of mitochondria and other organelles. The halophytes which secretes the saline ions include Limonium latifolium, Spartina sps., Sporobolus spicatus, Atriplex sps., etc. (Ramadan 2000). However, not all halophytes have salt glands; neither do they all discard salt saturated tissue, demonstrating that individual halophytes utilize different salt tolerance traits under different stress periods.

4.3 Osmotic Adjustment

Osmotic adjustment in response to abiotic stresses is an adaptive mechanism in the halophytes in order to maintain their water balance (Flowers and Colmer 2008). Besides the accumulation of inorganic ions and its sequestration in the vacuole, the osmotic balance between vacuole and cytoplasm is also maintained through the synthesis of organic solutes to retain the stability of the proteins in cells in response to drop in the water potential of the environment (Glenn and Brown 1999). Plant cells synthesize a variety of organic solutes such as proline, sucrose, polyols, trehalose and quaternary ammonium compounds (QACs) such as glycine betaine, alaninebetaine, prolinebetaine, choline-O-sulfate, hydroxyprolinebetaine, and pipecolatebetaine (Rhodes and Hanson 1993). These are low molecular weight, highly soluble compounds and are nontoxic even at high cellular concentrations (Ashraf and Foolad 2007) without

disturbing intracellular biochemistry and cellular functions (Cushman 2001), protects subcellular structures, mitigate oxidative damage caused by free radicals (Attipali et al. 2004), maintains the enzyme activities under salt stress and protection of cellular components from dehydration injury (Ashraf and Foolad 2007). The osmolytes accumulation is frequently reported in glycophytes and halophytes being continuously exposed to abiotic stresses; however, synthesis of these osmolytes is an energy-dependent process which consumes large number of ATP molecules (Raven 1985), thus affecting the growth. Osmolytes synthesis and their overproduction in transgenic plants has been achieved in transgenic crop plants, however little success has been achieved on the desired protective levels of these osmolytes in plants. In contrast, some plants showed increased tolerance to abiotic stresses after exogenous application of these organic solutes (Ashraf and Foolad 2007). Although increased accumulation of these osmolytes by the plants exposed to abiotic stresses has been reported, not all plant species synthesize the all kinds of osmolytes at a time; some plant species synthesize and accumulate very low quantity of these compounds while some plant species not do so (Ashraf and Foolad 2007).

4.3.1 Proline

Similar to glycophytes, proline accumulation is a common adaptive response to various abiotic stresses. Several studies using transgenic plants or mutants demonstrated that proline metabolism has a complex effect on development and stress responses, and that proline accumulation is important for the tolerance to certain adverse environmental conditions (Hong et al. 2000; Miller et al. 2010). In plants, proline is mainly synthesized from glutamate using two important enzymes such as pyrroline-5-carboxylate synthetase (P5CS) and pyrroline-5-carboxylate reductase (P5CR). Proline is synthesized in cytoplasm; however in mitochondria, the catabolism occurs via sequential action of proline dehydrogenase (PDH) producing pyrroline-5-carboxylate (P5C) and its conversion to glutamate using P5C dehydrogenase (P5CDH) (Szabados and Savoure 2009). Halophytes have shown vast diversity for the accumulation of proline in response to abiotic stresses, wherein plants from the Aizoaceae family accumulate large quantities of proline showing its role in osmoprotection (Delauney and Verma 1993). Proline concentrated in the cytosol, chloroplast and vacuoles and compatible with enzyme activity in the cytoplasm showed its significant contribution to osmotic adjustment. Besides being an osmoprotectant, proline also has a role in detoxification of reactive oxygen species and act as an antioxidant, stabilization of proteins and protein complexes and as a signaling/regulatory molecule (Szabados and Savoure 2009). It also function as a protein-compatible hydrotrope (Srinivas and Balasubramanian 1995), alleviating cytoplasmic acidosis, and maintaining appropriate NADP+/NADPH ratios compatible with metabolism (Hare and Cress 1997). Also, rapid breakdown of proline upon relief of stress provides sufficient reducing agents that support mitochondrial oxidative phosphorylation and generation of ATP for recovery from stress and repairing of stress-induced damages (Hare and Cress 1997). In halophytic plant species in response to abiotic stresses, proline accumulation in the cytosol has been shown to contribute substantially to cytoplasmic osmotic adjustment. For example, in cells of Distichlis spicata treated with 200 mM NaCl, the cytosolic proline concentration was estimated to be more than 230 mM (Ketchum et al. 1991). In Sesuvium portulacastrum, Lokhande et al. (2010a, b, 2011a) found an extensive increase in proline content when the callus and axillary shoot cultures exposed to salt and drought stress alone or under iso-osmotic stress conditions of NaCl and PEG. Higher proline accumulation has also been shown in S. portulacastrum plants exposed to various abiotic constraints that include salinity, drought, and heavy metals (Messedi et al. 2004; Slama et al. 2008; Ghnaya et al. 2007; Moseki and Buru 2010; Lokhande et al. 2011b). Such an osmotic adjustment through proline accumulation is also evident in other species like Plantago crassiflora, Salicornia europaea, Atriplex halimus, A. halimus subsp. schweinfurthii, Avicennia marina, Hordeum maritimum, Ipomoea pes-caprae, Paspalum vaginatum, Phragmites australis, and Suaeda sps.

(Vicente et al. 2004; Reda et al. 2004; Nedjimi and Daoud 2009; Pagter et al. 2009; Lefevre et al. 2009; Sucre and Suarez 2010). Among different halophytic plants, *S. portulacastrum* has been reported as a high proline accumulator, with levels reaching 300 μ mol g⁻¹ leaf dry matter (Slama et al. 2008). Such a pronounced accumulation of proline and its physiological role in osmotic adjustment may have made the halophytes more successful to grow under adverse environmental stresses.

4.3.2 Glycine Betaine

Among the variety of quaternary ammonium compounds, glycine betaine (GB) is one of the most abundantly occurring and synthesized at higher concentrations in the plants exposing to dehydration stress due to adverse environmental calamities. GB is located in chloroplast where it plays an important role in osmotic adjustment and protection of thylakoid membrane, by maintaining the photosynthetic machinery in active state (Robinson and Jones 1986). GB is synthesized mainly from choline, which is converted to betaine aldehyde and then to GB through sequential enzymatic action of choline monooxyegenase (CMO) and betaine aldehyde dehydrogenase (BADH), respectively. Although other pathways such as direct N-methylation of glycine are also known, the pathway from choline to GB has been identified in all GB-accumulating plant species (Ashraf and Foolad 2007). It is widely believed that synthesis and accumulation of GB protects cytoplasm from ion toxicity, dehydration and temperature stress and helps normal functioning of the metabolic machineries in the cell during stressed conditions by stabilizing macromolecule structures, protecting chloroplast and photosynthesis system II (PSII) by stabilizing the association of the extrinsic PSII complex proteins and indirectly interacting with phosphatidylcholine moieties of membranes to alter their thermodynamic properties (Subbarao et al. 2001). It has been shown that tolerant species are more amenable to accumulate higher GB in comparison to sensitive species as a response to abiotic stress imposition. Based on the GB and proline accumulation potential,

Tipirdamaz et al. (2006) categorized the halophytes from inland and salt marsh habitats of Turkey. The studies have shown that the species that behaved as GB accumulators appeared poor proline accumulators and vice versa. The GB accumulation reported in the halophytes is generally in the range of 1.5–400 µmol g⁻¹ DW and some of the highest GB accumulating halophytes are members of the Chenopodiaceae (Halocnemum strobilaceum, Petrosimonia brachiata, Suaeda confuse), Compositae (Artemisia santonicum), and Frankeniaceae (Frankenia hirsuta). Increased accumulation of GB has also been demonstrated in other halophytes such as Beta vulgaris (Subbarao et al. 2001), Spartina anglica (Mulholland and Otte 2002), Atriplex halimus (Martinez et al. 2005), A. Nummularia (Silveira et al. 2009), and S. portulacastrum (Lokhande et al. 2010a, b). Increased GB accumulation has also been correlated with increased betaine aldehyde dehydrogenase gene expression (BADHmRNA) in Salicornia europaea and Suaeda maritima leaves exposed to salt stress (Moghaieb et al. 2004). Considering the significance of GB in the osmotic balance of the halophytes under stressful environment, different methods can be derived to enhance the concentration of this compound in crop plants to increase their stress tolerance. The approaches can include breeding of sensitive cultivars with their tolerant relatives from halophytes with natural abilities to produce high levels of GB or genetically engineer the sensitive species through transformation of the genes responsible for GB synthesis. Although some progress has been made in introducing the genes for the production of these compounds in naturally accumulating or low-accumulating plant species, levels of these compounds' accumulation in transgenic plant have often been low or insufficient to the plant stress tolerance (Ashraf and Foolad 2007).

4.3.3 Soluble Sugars

In general, modulations in the carbon metabolism and the levels of carbohydrates (sugars) are seen due to changes occurring in the process of photosynthesis and carbon partitioning of the plant at organ level and in whole plants exposing to abiotic stresses (Gonzalez et al. 2009). Soluble sugars function as metabolic resources and structural constituents of cells, besides acting as signals regulating various processes associated with plant growth and development. Such signaling can modulate stress pathways into a complex network to further orchestrate metabolic plant responses. A variety of sugar compounds such as sucrose, glucose, mannose, maltose, trehalose, and many other sugar alcohols have been studied in response to abiotic stresses (Briens and Larher 1982; Yuanyuan et al. 2009) and the accumulation of soluble sugars has been attributed as an important parameter of osmotic adjustment in the halophytes. Briens and Larher (1982) screened different organs of 16 halophyte species for soluble carbohydrates and other osmolytes and found that all the species accumulated sucrose, fructose and glucose whereas Plantago maritime, Juncus maritimus, Phrgamites communis and Scripus maritimus showed the highest accumulation of soluble sugars. The presence of higher amounts of soluble sugars has been reported as main contributors to osmotic adjustment in the Atriplex halimus plants exposed to PEG and NaCl stresses and it is correlated with the response of NaCl stress on soluble sugar synthesis (Martinez et al. 2005). The accumulation of total soluble sugars has also been correlated with the variations at genotypic level among two genotypes of Cakile maritime namely Jebra and Tabarka which showed differences in the total soluble carbohydrate concentrations. While the content of the sugars was unaffected in the leaves of Jerba plants at moderate salinity, the plants of the saltsensitive Tabarka showed a slight increase in soluble carbohydrate contents during leaf development. The contribution of this compatible solute group to the "osmotic pool" was found higher in the salt-tolerant Jerba than in the salt-sensitive Tabarka seedlings exposed to 400 mM NaCl stress (Megdichi et al. 2007). Further, Sesuvium portulacastrum axillary shoots exposed to salinity stress showed optimum growth at 200 mM NaCl in comparison to control and exhibited increased synthesis of total soluble sugars over proline and glycine betaine (Lokhande et al. 2010b). Salinity-induced soluble sugar accumulation has also been observed in *P. euphratica* (Zhang et al. 2004). Accumulation of soluble sugars has been observed in plants undergoing drought, flooding, and water logging conditions (Chai et al. 2001; Munns 2002; Li and Li 2005). *Chenopodium quinoa* exposed to water deficit and water-logging stresses showed no changes in starch, sucrose, or fructose content but showed increased glucose and total soluble sugar content in stressed plants in comparison to control (Gonzalez et al. 2009). These studies in halo-phytes demonstrate that soluble sugars play a significant role besides other osmolytes in the osmotic adjustment.

4.4 Antioxidant Systems

The halophytic plants display a cascade of events upon exposure to environmental stresses leading to metabolic disturbance. The cascade of events include physiological water-deficit abscisic acidregulated stomatal closure in leaves, limited CO, availability, over-reduction of electron transport chain in the chloroplast and mitochondria and finally generation of reactive oxygen species (ROS). These ROS are highly toxic and in the absence of protective mechanism in the plant can cause oxidative damage to proteins, DNA, and lipids (Mittler 2002; Miller et al. 2010). Additionally, this may also lead to alteration in the redox state resulting in further damage to the cell (Mittler et al. 2004). To regulate the ROS levels, plant cells are evolved with complex enzymatic and nonenzymatic antioxidant defense mechanisms, which together help to control the cellular redox state under changing environmental conditions. A correlation between enzymatic and nonenzymatic antioxidant capacitance and abiotic stress tolerance has been reported in several plant species such as Crithmum maritimum, C. maritime, Plantago genus, Sesuvium portulacastrum, Mesembryanthemum crystallinum (Ben Amor et al. 2005; Jitesh et al. 2006; Sekmen, Turkan and Takio 2007; Ashraf 2009; Lokhande et al. 2010a, b, 2011a-c). Superoxide dismutase (SOD) constitutes the first line of defense converting O_2^{-} to H_2O_2 , which is further reduced to

Plant species	Enzyme/protein/gene studied	References
Avicennia marina	SOD, CAT, POX, APX, MDHAR	Cherian et al. (1999), Jitesh et al. (2006), Kavitha et al. (2008), and Kavitha et al. (2010)
Bruguiera parviflora, B. gymnorrhiza	SOD, CAT, APX	Takemura et al. (2000) and Parida et al. (2004)
Beta vulgaris, B. maritime	SOD, CAT, APX, GR	Bor et al. (2003)
Crithmum maritimum	SOD, CAT, POX	Ben Amor et al. (2005)
Hordeum vulgare	SOD, POX, CAT, GR	Patra and Panda (1998)
H. maritimum	SOD, CAT, GPX, APX, MDHAR, DHAR, GR	Hafsi et al. (2010)
M. crystallinum	SOD, ferritin, CAT, Mn, Fe, Cu/Zn SOD	Slesak et al. (2002, 2008), Slesak and Miszalski (2003), Hurst et al. (2004), and Parmonova et al. (2004)
Nitraria tangutorum	SOD, CAT, APX	Yang et al. (2010)
Phragmites australis	SOD, CAT, APX, POX, DHAR, GST	Carias et al. (2008)
Phaseolus vulgaris	SOD, APX, CAT, GPX	Jebara et al. (2005)
Sesuvium portulacastrum	SOD, CAT, APX, GR	Lokhande et al. 2010a, b, 2011a-c)
Suaeda nudiflora, S. salsa	SOD, CAT, POX	Cherian and Reddy (2003), Wang et al. (2004a), and Fang et al. (2005)
T. halophila	SOD, APX, POX	Taji et al. (2004), Wang et al. (2004a, b), and M'rah et al. (2006)

Table 2.2 Examples of halophytic plant species studied for the antioxidant responses in response to abiotic stresses

water and oxygen by ascorbate peroxidase (APX) and catalase (CAT). Monodehydroascorbate reductase (MDHAR), dehydroascorbate reductase (DHAR), glutathione reductase (GR), and glutathione peroxidase (GPX) are an important enzymes involved in regeneration of ascorbate and GSH for the proper functioning of ASC-GSH cycle (Noctor and Foyer 1998; Mittler 2002; Miller et al. 2010).

Halophytes have evolved various mechanisms of adaptations of which increased antioxidant enzyme activities was found one of an important mechanism of stress tolerance. Halophytes have capacity to maintain high metabolic activity even at inhibitory concentrations of intracellular Na⁺ due to enhanced antioxidant mechanism (Jitesh et al. 2006). Most of the early studies in halophytes were on photosynthesis and respiration and related to ion compartmentation, osmotic adjustment (Flowers et al. 1977; Fukushima et al. 1997). However, recently, more emphasis has been given on abiotic stresses in relation to antioxidant enzymes in halophytes (Takemura et al. 2000; Cherian and Reddy 2003; Parida et al. 2004; Slesak et al. 2002, 2008; Slesak and Miszalski 2003; Jitesh et al. 2006; Lokhande et al. 2010a–b, 2011a, c). The response of antioxidant enzyme systems in the halophytes exposed to abiotic stresses has been reviewed by Jitesh et al. (2006) and is summarized in Table 2.2. Most of the halophytes have shown increased efficiency of antioxidant enzyme machinery thus removing the ROS levels to a greater extent and maintain the plants survival under stressful conditions.

4.5 Redox and Energetics

The cellular redox state is made tangible in terms of the redox state of the individual redox-active molecules in a cell. For each redox-active molecule, its redox state can be defined as the proportion of reduced molecules relative to the total pool size, or alternatively as the ratio between reduced and oxidized molecules within a pool (Potters et al. 2010). A large number of redox-active compounds such as ascorbate (ASC), glutathione

V.H. Lokhande and P. Suprasanna

(GSH), pyridine nucleotides (NADH and NADPH), carotenoids, tocopherols, distinct redox-active phenolics, polyamines and proteins carrying redox-active S-groups are contained in plant cells (Smirnoff 2005; Queval and Noctor 2007). The enzymatic and nonenzymatic antioxidants involved in ROS scavenging significantly contribute to the redox state maintenance of the cellular environment through continuous channeling of these redox-active components which facilitate the proper functioning of the cell under stressful conditions. In general, the maintenance of redox state is correlated with the energy metabolism of the plant cell in terms of ATP/ ADP ratio. Under stressful conditions, the ROS produced due to oxidative stress requires more energy in the form of ATP to maintain the cellular homeostasis such as ion compartmentation, osmolytes synthesis, etc. The ATP requirement for these processes is different, such as 3.5, 41 and 50 ATP molecules are required for the synthesis or accumulation of one molecule of Na⁺, proline, and glycine betaine, respectively (Raven 1985). Thus, the energetic status of the plant is also dependent on the type of molecule synthesized or accumulated by the plant. It is possible that halophytes evolved to survive under abiotic stress conditions through proper maintenance of a higher redox and energetic status which could have conferred a plasticity to grow under stress. Not much information is available on the redox signaling and energetics in case of the halophytes. Recently, Lokhande et al. (2010c) for the first time demonstrated that maintenance of redox and energy state plays a major role in mediating salinity tolerance and in achieving a balance between tolerance and growth in Sesuvium portulacastrum. The plants under optimum levels of NaCl (250 mM) showed retention in the growth whereas significantly toxic levels of NaCl (1,000 mM) disturbed the homeostasis of the plant due to abrupt changes in the ratios of redox-active com-(ASC/DHA, GSH/GSSG, pounds NADP/ NADPH) and energy molecules (ATP/ADP) (Fig. 2.2a-f). Further to stick this work, more efforts could be initiated to unravel the redox and energetic of the halophytes to gain the knowledge of redox control. The concept of a cellular redox

state, as proposed by Foyer and Noctor (2005), is very useful in terms of elucidating the importance of redox reactions in gene expression, metabolism control, signal transduction, and cellular defense. The concept must now be developed to identify and quantify those redox-active compounds that are the specific regulators of cellular responses (Potters et al. 2010).

4.6 Genomic Approaches

Plant adaptation to environmental stresses is controlled by a cascade of molecular networks. In this regard, the application of genomic technologies has made more impact on understanding the plant responses to the abiotic stresses (Cushman 2003). The technology has made remarkable success in understanding the abiotic stress tolerance at genome level with potential to modify plants' tolerance for increasing yield under stressful conditions (Bohnert et al. 2006). In contrast to traditional breeding and marker-assisted selection programs, the direct introduction of a small number of genes by genetic engineering has also become tangible and attractive as a rapid approach to improve the plants' stress tolerance (Cushman and Bohnert 2000; Popova et al. 2008) to re-establish homeostasis and to protect and repair damaged proteins and membranes (Wang et al. 2003).

In the course of studies on mechanism of abiotic stress tolerance, Arabidopsis thaliana has emerged as an excellent model system (Zhu 2001) because most of the crop plants are glycophytes. However, study of some novel mechanisms unique to halophytes or stress-tolerant plants may be difficult with Arabidopsis and this has been made possible by the available genome information on Mesembryanthemum crystallinum, which, when compared with the Arabidopsis genome, seems to contain a number of transcripts that have no counterparts (Wang et al. 2004b). Thereafter, several halophytes such as M. crystallinum, Suaeda species, Atriplex species have been employed to dissect out the molecular basis of stress tolerance mechanism of the halophytes. Recently, Thellungiella halophila (salt cress), a member of the Brassicaceae, has emerged as a



Fig. 2.2 The salinity stress responses (growth, redox, and energy status) of *Sesuvium portulacastrum* exposed to optimum (250 mM) and supra-optimal (1,000 mM) concentrations of NaCl

model for understanding adaptation of the halophytes to abiotic stress tolerance due to its homology with the glycophyte model, *A. thaliana* (Wang et al. 2004b; Amtmann 2009). This halophyte has the ability to grow in high salt concentrations which otherwise become inhibitory for the growth of its salt-sensitive relative *A. thaliana* and other crop plants (Zhu 2001; Nah et al. 2009). The salient features of *T. halophila* such as small diploid genome (240 Mb and 2n = 14), short and self-fertile life cycle and ease of floral dipping method of transformation have enabled it as a successful candidate for molecular detailing of its response to abiotic stress tolerance and relative

comparison with A. thaliana. The comparative genomics of T. halophila and A. thaliana revealed extensive and novel information on presence of differential genes responsible for abiotic stress tolerance in T. halophila in comparison to A. thaliana (Nah et al. 2009). Taji et al. (2004) studied the differences in the regulation of salt tolerance between salt cress and Arabidopsis by analyzing the gene expression profiles using a full-length Arabidopsis cDNA microarray. Only a few genes were induced by 250 mM NaCl in salt cress stress compared to Arabidopsis. Even in the absence of stress, a large number of known abiotic- and biotic-stress inducible genes, including Fe-SOD, P5CS, PDF1.2, AtNCED, P-protein, b-glucosidase, and SOS1, were expressed at high levels. The study also found salt cress to be more tolerant to oxidative stress than Arabidopsis. The salt tolerance mechanisms between salt-sensitive glycophytes and salt-tolerant halophytes could result from alterations in the regulation of the same basic set of genes involved in salt tolerance among these plants. Kant et al. (2006) used gene-specific primers of Arabidopsis that showed similar realtime PCR amplification efficiencies with both A. thaliana and T. halophila cDNA and concluded that the expression of specific salt tolerance orthologues differs between unstressed and stressed plants of both species.

The development of expressed sequence tags (ESTs) and cDNA libraries using various genomic approaches such as suppressive subtractive hybridization (SSH), differential display reverse transcription-polymerase chain reaction (DDRT-PCR), representational difference analysis (RDA), serial analysis of gene expression (SAGE), and cDNA microarray (Breyne and Zabeau 2001) provided an enormous databases for understanding the genetic network involved in abiotic stress tolerance mechanism of halophytes (Wang et al. 2004b; Kore-eda et al. 2004; Popova et al. 2008). Using this approach, various genes responsible for stress tolerance have been isolated from halophytes and cloned or overexpressed in the bacterial systems as well as sensitive cultivars of glycophytes to enhance the stress tolerance capacity and improve crop yield (Table 2.3).

Transcript-profiling experiments in *Arabidopsis* in response to drought, cold, or salinity stresses using the Arabidopsis GeneChip array or full-length cDNA microarrays have shown that extensive changes occur in the transcriptome of Arabidopsis (Fowler et al. 1999; Kreps et al. 2002; Seki et al. 2002). It is known that approximately 30% of the transcriptome on the Arabidopsis GeneChip 8 K oligoarray changed in stress treatments (Kreps et al. 2002). The expressed sequence tag analyses of Thellungiella clones revealed 90-95% identities between Thellungiella and Arabidopsis cDNA sequences (Wang et al. 2004a, b; Wong et al. 2006). In a comparison of three stresses (cold, low water availability, and saline conditions) as well as recovery from water deficits in Thellungiella, Wang et al. (2006) employed an expression profiling strategy to identify stress responses. There was not much degree of overlap among genes responsive to drought, cold, or salinity suggesting relatively few common end responses triggered by these stresses existed in this halophyte. While Thellungiella had shown activation of the expression of some well-known stress-responsive genes, it was found to downregulate a large number of biotic stress-related genes under drought and salinity treatments. The study has made a significant step in showing the emergence of Thellungiella as a model species for the molecular elucidation of abiotic stress tolerance, and that Thellungiella responds precisely to environmental stresses, thereby conserving energy and resources and maximizing its survival potential.

SOS1 (Salt Overly Sensitive 1) is known to play key role in the ion homeostasis mechanism movement (Shi et al. 2000). Although SOS1 has been intensely studied in Arabidopsis, its involvement in the salt tolerance of halophytes is not much known. Oh et al. (2009) investigated the role(s) by which ThSOS1, the SOS1 homolog in Thellungiella, was involved in modulating the halophytic character using ectopic expression of the gene in yeast and in Arabidopsis and Thellungiella SOS1-RNA interference (RNAi) lines. The knockdown of SOS1 expression totally altered Thellungiella into a salt-sensitive plant like Arabidopsis. The authors found that the activity of ThSOS1 could limit Na⁺ accumulation and the distribution of Na⁺ ions.

Source organism	Genes	Trait improved	Target organism	References
Atriplex gmelini	Vacuolar Na ⁺ /K ⁺ antiporters Ag <i>NHX1</i>	Eightfold higher activity of the vacuolar-type Na ⁺ /H ⁺ antiporter	Oryza sativa	Ohta et al. (2002)
Medicago sativa	Vacuolar Na ⁺ /K ⁺ antiporters <i>MsNHX1</i>	Increased osmotic adjustment and MDA content	A. thaliana	Bao-Yan et al. (2002)
Aeluropus littoralis	Vacuolar Na'/K⁺ antiporters AlNHXI	Compartmentalize more Na ⁺ in roots and keep a relative high K ⁺ /Na ⁺ ratio in the leaves	N. tabacum	Zhang et al. (2008)
Atriplex centralasiatica	Betaine aldehyde dehydrogenase <i>AcBADH</i>	Improved synthesis of glycine betaine	N. tabacum	Yin et al. (2002)
Suaeda liaotungensis, Beta vulgaris, Atriplex hortensis, A. nummularia	Choline monooxygenase CMO	Three- to sixfold increased activity of CMO increases glycine betaine synthesis	N. tabacum	Russell et al. (1998), Shen et al. (2001), Tabuchi et al. (2005), and Li et al. (2003, 2007)
Avicennia marina	Monodehydroascorbate reductase (MDHAR)	Ascorbate regeneration and ROS scavenging	N. tabacum	Kavitha et al. (2010)
Sesuvium portulacastrum	Fructose-1,6-bisphosphate aldolase <i>SpFBA</i>	Strongly expressed in roots than in leaves and stems under abiotic stresses	Escherichia coli	Fan et al. (2009)
Mesembryanthemum crystallinum	IMT1, myo-Inositol O-methyl-transferase	Inositol methylation	E. coli	Rammesmayer et al. (1995)
Thellungiella halophila	FLC gene	Controls vernalization response pathway	T. halophila	Fang et al. (2006)
Suaeda salsa	Peroxiredoxin Q gene SsPrx Q	Thioredoxin-dependent peroxidase activity	E. coli	Guo et al. (2004)

 Table 2.3
 Source of genes from halophytes for the improvement of abiotic stress tolerance

5 Role of Halophytes in Abiotic Stress Management

5.1 Desalination and Stabilization of Saline Soils

The problem of salinity is widespread covering at least 75 countries (Goudie 1990). Various physical, chemical, and biological approaches have been developed for the reclamation of such saline soils (Shahid 2002). Biological methods include organic manure, crop rotation, salt-tolerant crops (Shahid 2002), as well as vegetative bioreclamation (Qadir and Oster 2004). The reclamation of saline soil using such biological means is also referred as desalinization (Zhao 1991), biodesalination, and desalination of salt-affected soils by halophytes (Rabhi et al. 2009). The potential of plants to accumulate enormous salt quantities depends often on the capacity of their aboveground biomass (hyper-accumulating plants) (Rabhi et al. 2010). This ability could be significant particularly in the arid and semi-arid regions where insufficient precipitations and inappropriate irrigation systems are unable to reduce the salt burden in the rhizosphere of plants and suitable physicochemical methods are too expensive (Shahid 2002). The plant-based method of saline soil stabilization is of importance especially in several developing countries where chemical amendments are getting more and more expensive. In this regard, the use of Na⁺ and Cl⁻ hyperaccumulating plants for soil desalination is often suggested as a strategy (Ravindran et al. 2007). A large number of species has been utilized for soil desalination based on their suitability and capacity to accumulate the salt. Halophytes are one of the important categories of plant species extensively used for this purpose with rice as the only one glycophytic exception (Iwasaki 1987). In order to be useful for desalination purpose, the plant species to be used should have high salt resistance, high biomass production, considerable shoot sodium content, and high degree of economic utilization (such as fodder, fuel, fiber, essential oil, and oil seeds) (Rabhi et al. 2010). Shoot-succulent halophytes such as Sesuvium

portulacastrum and *Suaeda* sps. meet these criteria since they are able to accumulate enormous Na⁺ quantities within their above-ground organs. They can become useful candidates for the desalination of salt-affected soils under nonleaching conditions. Zhao (1991) calculated that *Suaeda salsa* produced about 20 ton DW ha⁻¹ and withdraw 3–4 ton NaCl. Ravindran et al. (2007) estimated that *Suaeda*, *Sesuvium*, *Excoecaria*, *Clerodendron*, *Ipomoea*, and *Heliotropium* species could remove 504, 474, 396, 360, 325, and 301 kg NaCl, respectively, from 1 ha land in 4 months.

Selection of suited species is the first step for affordable soil desalination at a wider scale in the arid and semi-arid regions (Rabhi et al. 2009). In a case study, a significant decrease in electrical conductivity of the soil having a 50% saturation percentage was recorded from 33 to 20 dS m⁻¹ in the presence of single growth cycle of J. rigidus in Egypt (Zahran and Wahid 1982). Suaeda salsa showed its potential to reduce the soil Na⁺ content at depth 0-10 cm by 2.4 with a density of 15 plants m⁻² and by 3.8 with a density of 30 plants m⁻² (Zhao 1991). It has also been demonstrated that the growth of annual glycophytes (Medicago ssp.) was much better on the soil previously desalinated with perennial halophytes in saline ecosystems (Abdelly et al. 1995). It was concluded that perennial halophytes desalinize and fertilize the rhizosphere, offering a favorable microhabitat for a better growth of annual glycophytes. Ravindran et al. (2007) evaluated the capacity of six halophytic species (Suaeda maritima, Sesuvium portulacastrum, Clerodendron inerme, Ipomoea pes-caprae, Heliotropium curassavicum and one tree species Excoecaria agallocha) to desalinize the upper 40 cm of soil under field conditions in India. This study demonstrated that after 120 days of cultivation of the halophytes, Suaeda maritima and Sesuvium portulacastrum showed a decrease in electrical conductivity of saline soil from 4.9 to 1.4 and 2.5 dS m⁻¹, respectively. The potential of native halophytes Arthrocnemum indicum and Suaeda fruticosa to desalinize saline soils was compared with that of an introduced halophyte, S. portulacastrum. In this study, Rabhi et al. (2009) confirmed S. portulacastrum as the most suitable

plant with higher accumulation of Na⁺ in its shoot parts for desalination purpose in arid and semiarid regions where precipitation is too low to leach salts from rhizosphere. Similarly, successful germination and growth of *Hordeum vulgare* (barley) was observed on the soil desalinated with salt accumulator halophyte *S. portulacastrum* (Rabhi et al. 2010). Taken together, the reports suggest that salt accumulator halophytes can be exploited as a potential source for desalination of agricultural land in the arid and semi-arid regions as well as for the stabilization of saline lands along the coastal regions of the world.

5.2 Phytoremediation

Contamination of agricultural soil by heavy metals (such as Cu, Cd, Zn, Mn, Fe, Pb, Hg, As, Cr, Se, Ur, etc.) has become a serious environmental concern due to their potential impact on the ecosystems. Such toxic elements are considered as soil and water pollutants due to their widespread occurrence, and their acute and chronic toxic effect on plants grown in such soils as well as on humans living in their surrounding (Yadav 2010). Plants, as sessile organisms have developed detoxification mechanisms diverse against absorbing a diversity of natural and man-made toxic compounds. Pollutant-degrading enzymes in plants are a natural defense system against a variety of allelochemicals released by competing organisms, including microbes, insects and other plants. Therefore, plants act as natural, solarpowered pump-and-treat systems for cleaning up contaminated environments, leading to the concept of phytoremediation (Aken 2008). A variety of plant systems have been studied for phytoremediation practices of contaminated soil; however, each species has limitations to accumulate the toxic metals and detoxify to nontoxic compounds through the enzymatic actions. In the course of evolution from marine to freshwater habitat, halophytes are found most successful group of plants which have shown adaptations to a variety of abiotic stresses, tolerance to heavy metal stress is one of these. In recent years, more emphasis has been placed to remove the toxic metals from contaminated soil and water bodies and reclamation of such lands for sustainable agriculture. In this regard, extensive research is undertaken to exploit the use of metal hyperaccumulating plants and search for a suitable plants that can significantly accumulate heavy metals and metalloids (Zabłudowska et al. 2009). However, phytoremediation constitutes a group of strategies meant not only to reduce the metal load at the contaminated site but also to stabilize the site. These strategies are referred as "phytoextraction" or "phytostabilization" and the selection of a plant may depend on the level of contamination at the site of concern. Both strategies can be integrated into operation at highly contaminated mine sites with a plant that may not be a hyper-accumulator but can tolerate even very high concentrations of toxic metals (Lokhande et al. 2011b). Various halophytes have evolved distinct morphological specializations for dealing with abiotic stressed environments such as presence of "aerial stilts" in the members of families Rhizophoraceae and "pneumatophores" in the members of Avicennaceae and Sonneratiaceae which enable gaseous exchange and oxygenation for respiration in an anoxic environment (Hutchings and Saenger 1987); however, the members of Myrsinaceae, possess no aerial roots. Table 2.4 presents phytoremediation potential of some halophytes.

Numerous laboratory-based trials suggested that the concentrations of metals required to show significant negative effects on halophytes may be significantly higher when compared to their aquatic and terrestrial floral counterparts (MacFarlane et al. 2007). For example, there were no adverse effects on the growth of Rhizophora mucronata and Avicennia alba seedlings treated with Zn (10-500 µg ml⁻¹) and Pb (50-250 µg ml⁻¹). In Kandelia candel seedlings, only at the highest applied metal concentrations (400 mg kg⁻¹ Cu and Zn) inhibition of leaf and root development was observed (Chiu et al. 1995). Similarly, Pb (0–800 μ g g⁻¹) had little negative effect on Avicennia marina seedlings (MacFarlane and Burchett 2002). Studies have demonstrated the accumulation of metals (Cu, Zn, Pb, Fe, Mn, and Cd) predominantly in root

Plant species	Phytostabilization/phytoextraction/ phytoexcretion of heavy metals	References
Sesuvium portulacastrum	Cd, Pb and As	Ghnaya et al. (2007), Nouairi et al. (2006), Zaier et al. (2010a, b), and Lokhande et al. (2011b)
Mesembryanthemum crystallinum	Cd	Ghnaya et al. (2007) and Nouairi et al. (2006)
Halimione portulacoides, Spartina maritima	Cd, Cu, Pb, and Zn	Reboreda and Caçador (2007, 2008)
Arthrocnemum macrostachyum, Spartina argentinensis	Cd and Cr	Redondo-Gómez et al. (2010a, b)
Triglochin maritima, Juncus maritimus, Sarcocornia perennis, Halimione portulacoides	Hg	Castro et al. (2009)
Atriplex halimus subsp. Schweinfurthii	Cd	Nedjimi and Daoud (2009) and Lefevre et al. (2009)
A. halimus	Pb and Cd	Manousaki and Kalogerakis (2009)
Spartina densiflora, S. maritima	As, Cu, Fe, Mn, Pb, and Zn	Cambrolle et al. (2008)
Aster tripolium	Cu and Pb	Fitzgerald et al. (2003)
Sarcocornia perennis	Fe, Mn, and Hg	Lilebo et al. (2010)
Halimione portulacoides	Zn, Pb, Co, Cd, Ni, and Cu	Sousa et al. (2008) and Almeida et al. (2009)
Tamarix smyrnensis	Pb and Cd	Kadukova and Kalogerakis (2007), Kadukova et al. (2008), and Manousaki et al. 2008
Juncus maritimus	Al, Cd, Cr, Cu, Fe, Mn, Ni, Pb, and Zn	Almeida et al. (2006)
Sporobolus virginicus, Spartina patens, and Atriplex nammularia	Zn, Cu, and Ni	Eid and Eisa (2010)
Salicornia europaea	Cd	Ozawa et al. (2009)

 Table 2.4
 Examples of halophytic plant species used for the purpose of phytoremediation

tissue, rather than in foliage, in numerous mangrove species grown in the field conditions, such as Avicennia sps., Rhizophora sps. and Kandelia sps. (Peters et al. 1997). It has also been observed that for some mangroves, concentrations of translocated metals are low, with bio-concentration factors (BCF; ratio of leaf metal to corresponding sediment metal concentration) ranging from <0.01 in Rhizophora mangle to 0.06 for other species such as A. marina (Lacerda 1997). However, other studies suggest that mangroves may accumulate and translocate some metals with leaf BCFs greater than 1, for example, 1.5-2.4 for A. marina (Sadiq and Zaidi 1994), 1.7 for Aegiceras corniculatum and 1.2 for Kandelia candel (Chen et al. 2003) and behaved as hyper-accumulating species. Lokhande et al. (2011b) recently demonstrated the arsenic (As) accumulation potential of Sesuvium exposed to As(V) (100–1,000 μ M) for 30 days, wherein the growth of the plant was not affected even after prolonged exposure to arsenic stress with the significant As accumulation (155 µg g⁻¹ dry weight) and a bioaccumulation factor of more than ten at each concentration. On the basis of total As accumulation, bioaccumulation factor and known biomass production capacities, the Sesuvium like other As hyper-accumulator plants has been suggested to use as potential candidates for application in arsenic removal and land re-vegetation/ reclamation projects in the As-contaminated sites of the world.

Heavy metal uptake in halophytes is generally regulated at the root endodermis through modifying uptake from predominantly apoplastic to selective symplastic transport. The contribution of each tissue type is dependent on the molecular properties of the plamsalemma (i.e., specific membrane transport proteins) and on the metal in question (MacFarlane and Burchett 2000). Some halophytic genera such as Aegiceras and Avicennia secrete excessive Na⁺ and K⁺ through specialized glands or glandular trichomes on abaxial and adaxial leaf surfaces, while such specialized structures are absent in nonsecretors, for example, Rhizophora and Sonneratia (MacFarlane and Burchett 1999). Indeed mangroves and a number of other estuarine halophytes with glandular tissue are known to excrete heavy metals concomitantly with other solutes (MacFarlane and Burchett 2000). The variation in morphology/function of nutritive root tissue and glandular tissue to deal with the challenges of excess cations in saline environments could become significant for metal accumulation, transport, partitioning, and excretion among halophytic plant species (MacFarlane et al. 2007).

5.3 Wetland Restoration and Re-vegetation

Mangrove forests are ecologically important coastal ecosystem currently covering 146,530 km of the tropical shorelines of the world (FAO 2003). There has been a steady decline from 198,000 km of mangroves in 1980 and 157,630 km in 1990 (FAO 2003) which represents about 2 and 1% losses per year between 1980-1990 and 1990–2000, respectively (Lewis 2005). Mangroves have provided ecological benefits in terms of shorelines stabilization, reduction in wave and wind energy against shorelines thus protecting inland structures, supporting coastal fisheries for fish and shellfish through direct and indirect food support and provisions for habitat, and support of wildlife. However, increasing human activities and adverse environmental conditions have led to destruction of mangrove forests thus limiting the available resources.

Therefore, restoration and re-vegetation of mangrove forest has become important for the development of sustainable agriculture and to avoid the destructive natural calamities. In this context, restoration or rehabilitation of saline lands using potential halophytes will act as an effective strategy. Restoration or rehabilitation is recommended when an ecosystem is altered to such an extent that it can no longer self-correct or self-renew. This could result in total disturbance to ecosystem homeostasis and permanent stopping of the normal processes of secondary succession or natural recovery from damage. Wetlands play an important role in nutrient cycling, sediment accretion, pollution filtration, and erosion control in the world. In addition, they are known for their distinctive flora and rich spectrum of wildlife, especially waterfowl, which makes them more valuable and more prone to human impact than other ecosystems (Mitsch and Gosselink 2007). However, only a small percentage of the original wetlands have remained around the world after over two centuries of intensive development and urbanization. Having lost so many wetlands, it seems that there are many opportunities for wetland restoration along coastal lines, rivers, lakes, etc. The deteriorated saline wetland can be restored using halophytes and salt-tolerant plants. The research on salt-tolerant plants and halophytes has attracted attention of many scientists because these plants could be used for saline agriculture and biomass energy on saline land, and for overcoming the worldwide problem of food shortage and energy crisis. Restoration of areas of damaged or destroyed mangrove forests has been previously discussed by many workers (Brockmeyer et al. 1997; Lewis and Streever 2000; Saenger 2002). Saenger and Siddiqi (1993) described one of the largest mangrove afforestation programs which covered plantings of primarily one species (Sonneratia apetala) over 1,600 km² on newly accreting mud flats in Bangladesh.

Research in this field of wetland restoration has demonstrated the potential of salt-tolerant plants and halophytes (Wang et al. 2008a, b; Zhang et al. 2008) for restoration and re-vegetation of barren lands and wetlands along the coastal regions. Use of halophytes from the ancient period for restoration of wetlands is summarized here. Scirpus species have been extensively used in constructed wetlands for wastewater treatment as the species has ability to efficiently remove nutrients and pathogens from effluent (Huang et al. 2000; Coleman et al. 2001). Besides S. robustus showed efficiency for removal of selenium (Se) from contaminated water demonstrating potential for Se phytoremediation by wetlands (Pilon-Smits et al. 1999). Tissue culture mode of plant regeneration has been suggested as an efficient tool for producing plants required in wetland creation and restoration (Wang et al. 2003). Seliskar and Gallagher (2000) have further proposed that tissue culture-induced somaclonal variation can be advantageous to produce plants with particular characteristics for use in wetland restoration. Further, Wang et al. (2006) evaluated the wetland restoration potentials of selected tissue culture regenerants of ecologically important salt marsh monocots, Spartina patens, S. alterniflora, Juncus gerardi, J. roemerianus and Scirpus robustus in a simulated marsh fields.

5.4 Saline Agriculture

The rapidly increasing human population in the arid and semi-arid regions of the world has tremendously increased the pressure on the availability of good quality water and land resources for human usage, industry and agriculture. In addition, improper and poor quality irrigation practices have increased the level of under-ground water and large areas have become water logged which eventually results into higher salinities of the soil (Yensen 2006). Salinity affects the growth of the plant to a sever extent thus reducing the crop productivity in the arable lands. It becomes a serious threat of growing any conventional crops which are otherwise sensitive to high salt concentrations and expect the yield at higher level in such saline lands to fulfill the demand of ever-growing human population for food, fodder, shelter and as a raw material for the industrial purposes. The research work on engineering the salt tolerance of the crop plants has not yet produced the successful transgenic plants which could tolerate the excess saline stress and yield more productivity in such conditions. Therefore, extensive efforts should have been undertaken to protect the available resources of freshwater and arable lands in the arid and semi-arid regions of the world through the application of halophytes for the variety of uses as a source of nonconventional cash-crops such as food, fodder, forage, medicinal, ornamental, chemical, timber, and other usage of wood and fibers (Khan and Qaiser 2006). Introduction of these potential halophytes has led to cover the barren saline land along the coastal zones of the world and provided economic benefits to the humans.

A variety of halophytic plant species have been categorized into different groups such as euhalophytes, xerohalophytes, and hydrohalophytes on the basis of their growth performance in variable climatic conditions and salt concentrations in the soil which has been utilized as a source of nonconventional cash-crops (Khan and Qaiser 2006; Khan and Ansari 2008). Hollington et al. (2001) described the successful stories of the utilization of halophytic species for the improvement of sustainable agriculture as well as sources of economy. Atriplex species showed the highest productivity and increases water uptake whereas tree species of Acacia and Prosopis have shown its role in re-vegetation and for biological drainage. Further, the use of raised-bed technology and on-farm seed priming have improved the production and efficiency of a range of halophytic plant species in saline conditions. Zhao et al. (2002) have screened the halophytic species distributed along the coast of China and categorized into different groups and suggested their uses in saline agriculture for the economic purposes. The variety of halophytic species so far used as a nonconventional source for various purposes in Pakistan have been reviewed extensively (Khan and Qaiser 2006; Khan and Ansari 2008) and presented in Table 2.1. Similarly, various strategies have been utilized for managing the saline or alkaline soil for sustainable agricultural production in South Africa (Sharma and Minhas 2005). Besides, Masters et al. (2007) also reviewed the utilization of halophytic grasses and shrubs (such as Medicago sativa, M. polymorpha, Trifolium sps., Hordeum vulgare, Distichlis spicata, Suaeda sps., Sporobolus sps., etc.) in biosaline agriculture for the production of forage and livestock which showed their growth potentials in highly saline soil with salt concentrations >25 dS m⁻¹ and produce 0.5–5 ton of edible dry matter year-1. The potentials of five halophytic plant species namely Diplachne fusca, Spartina patens, Sporobolus virginicus (Smyrna-smooth), Sporobolus virginicus (Dixe-coarse), and Medicago sativa have been studied as a source of forage plants in Egypt on the soils irrigated with different concentrations of seawater. The studies showed that S. virginicus (Dixe) produced the highest biomass upon irrigated with either 25 or 37.5% sea water, followed by S. patens and D. fusca whereas S. virginicus showed the lowest yield. The studies conducted on cultivation of Salvadora persica in semi-arid saline and alkali soils showed its efficiency for growth and as a source of industrial oil on both saline and alkali soils for economic and ecological benefits which is otherwise not suitable for conventional arable farming (Reddy et al. 2008). Therefore, considering the approach of biosaline agriculture more efforts should have been undertaken on the cultivation of nonarable lands with nonconventional plant resources such as halophytes to bring the uncultivable land for the use of economic purposes of the human being. This strategy will help to improve the gross economy of the developing countries.

6 Conclusions and Future Perspectives

Worldwide food production is affected to a large extent by environmental extremities and the sensitivity of crops is a major limitation for achieving higher plant productivity. Plants have evolved adoptive mechanisms which can be understood and exploited as an important resource for development of crops tolerant to extremities. Halophytes show a diversity of growth responses to increasing salinity. Halophytes have evolved to the changing environmental conditions through developing

variety of tolerance mechanisms such as halosucculence, ion compartmentation (exclusion/inclusion), osmoregulation, enzymatic and nonenzymatic antioxidants and maintenance of redox and energy status. Research on understanding the abiotic stress tolerance mechanism of halophytes has been on the upfront using wide array of physiological, biochemical, and molecular tools. Some of the halophytes (e.g., Thellungiella halophila) that tolerate adverse conditions have become the choice model systems for unraveling the different pathways associated with halophytic behavior. However, research in the context of progress in metabolomics, genomics, and proteomics has to be initiated in diverse halophytic species with the use of advanced techniques to gain detailed knowledge of abiotic stress tolerance.

Halophytes have also been utilized practically for managing the stressful environment and shown to be involved in increasing the economy of developing countries in many parts of the world. Halophytes have shown their role in desalination of saline lands from arid and semi-arid regions as well as the stabilization of saline lands along the coastal sides, phytoremediation of heavy metal contaminated sites and wetland restoration and revegetation through introduction of variety of halophyte species. This has led to developing agriculture on saline lands for supporting the sources of food, forage, fodder, medicine, ornamental and important plant-based chemicals to ever-growing human population. In addition, this may also help in reducing the burden on the crop plants which are facing the productivity problems due to exposure to various abiotic stresses. Considering these applications, constructive strategies have to be developed and implemented for the protection of world's nonrenewable resources, wherein available halophyte diversity can be utilized as an important source. Investigation of the gene regulation and the balance of individual stress tolerance mechanisms will aid in translating the information to other salt-sensitive crops. Such studies will be helpful for ensuring sustainability of future research efforts to improve and manage crop performance on marginal and irrigated land through genetic manipulation.

References

- Abdelly C, Lachaal M, Grignon C, Soltani A, Hajji M (1995) Association episodique d'halophytes strictes et de glycophytes dans un ecosysteme hydromorphe sale en zone semi-aride. Agronomie 15:557–568
- Aken BV (2008) Transgenic plants for phytoremediation: helping nature to clean up environmental pollution. Cell 26:225–227
- Almeida CMR, Mucha AP, Vascancelos MTSD (2006) Variability of metal contents in the sea rush *Juncus maritimus*-estuarine sediment system through one year of plant's life. Mar Environ Res 61:424–438
- Almeida CMR, Dias AC, Mucha AP, Bordalo AA, Vascancelos MTSD (2009) Study of the influence of different organic pollutants on Cu accumulation by Halimione portulacoides. Estuar Coast Shelf Sci 85:627–632
- Amtmann A (2009) Learning from evolution: *Thellungiella* generates new knowledge on essential and critical components of abiotic stress tolerance in Plants. Mol Plant 2:3–12
- Apse MP, Blumwald E (2007) Na⁺ transport in plants. FEBS Lett 581:2247–2254
- Aronson JA (1989) HALOPH a data base of salt tolerant plants of the world. Office Arid Land Studies, University of Arizona, Tucson
- Ashraf M (2009) Biotechnological approach of improving plant salt tolerance using antioxidants as markers. Biotechnol Adv 27:84–93
- Ashraf M, Foolad MR (2007) Roles of glycine betaine and proline in improving plant abiotic stress resistance. Environ Exp Bot 59:206–216
- Attipali RR, Kolluru VC, Munusamy V (2004) Drought induced responses of photosynthesis and antioxidant metabolism in higher plants. J Plant Physiol 161:1189–1202
- Ball MC (1988) Ecophysiology of mangroves. Trees 2:129–143
- Bao-Yan AN, Yan L, Jia-Rui L, Wei-Hua Q, Xian-Sheng Z, Xin-Qi Z (2002) Expression of a vacuolar Na⁺/H⁺ antiporter gene of alfalfa enhances salinity tolerance in transgenic Arabidopsis. Acta Agron Sin 34:557–564
- Ben Amor N, Ben Hamed K, Debez A, Grignon C, Abdelly C (2005) Physiological and antioxidant responses of the perennial halophyte *Crithmum maritimum* to salinity. Plant Sci 68:889–899
- Blumwald E (2000) Sodium transport and salt tolerance in plants. Curr Opin Cell Biol 12:431–434
- Bohnert HJ, Sheveleva E (1998) Plant stress adaptationsmaking metabolism move. Curr Opin Plant Biol 1:267–274
- Bohnert HJ, Gong Q, Li P, Ma S (2006) Unraveling abiotic stress tolerance mechanisms – getting genomics going. Curr Opin Plant Biol 9:180–188
- Bor MF, Ozdemir F, Turkan I (2003) The effect of salt stress on lipid peroxidation and antioxidants in leaves of sugar beet *Beta vulgaris* L. and wild beet *Beta maritime* L. Plant Sci 164:77–84

- Breckel SW (2002) Salinity, halophytes and salt affected natural ecosystems. In: Lauchli A, Luttge U (eds) Salinity: environment-plants-molecules. Kluwer, Dordrecht, pp 53–77
- Breyne P, Zabeau M (2001) Genome-wide expression analysis of plant cell cycle modulated genes. Curr Opin Plant Biol 4:136–142
- Briens M, Larher F (1982) Osmoregulation in halophytic higher plants: a comparative study of soluble carbohydrates, polyols, betaines and free proline. Plant Cell Environ 5:287–292
- Brockmeyer RE Jr, Rey JR, Virnstein RW, Gilmore RG, Ernest L (1997) Rehabilitation of impounded estuarine wetlands by hydrologic reconnection to the Indian River Lagoon, Florida (USA). Wetland Ecol Manag 4:93–109
- Cambrolle J, Redondo-Gomez S, Mateos-Naranjo E, Figueroa ME (2008) Comparison of the role of two *Spartina* species in terms of phytostabilization and bioaccumulation of metals in the estuarine sediment. Mar Pollut Bull 56:2037–2042
- Carias CC, Novais JM, Martins-Dias S (2008) Are *Phragmites australis* enzymes involved in the degradation of the textile azo dye acid orange 7? Bioresour Technol 99:243–251
- Castro R, Pereira S, Lima A, Corticeiro S, Valega M, Pereira E, Duarte A, Figueira E (2009) Accumulation, distribution and cellular partitioning of mercury in several halophytes of a contaminated salt marsh. Chemosphere 76:1348–1355
- Chai CL, Li SH, Xu YC (2001) Carbohydrate metabolism in peach leaves during water stress and after stress relief. Plant Physiol Commun 37:495–498
- Chen XY, Tsang EPK, Chan ALW (2003) Heavy metal contents in sediments, mangroves and bivalves from Ting Kok, Hong Kong. China Environ Sci 23:480–484
- Cherian S, Reddy MP, Pandya JB (1999) Studies on salt tolerance in Avicennia marina (Forsk.) Vierh.: effect of NaCl salinity on growth, ion accumulation and enzyme activity. Indian J Plant Physiol 4:266–270
- Cherian S, Reddy MP (2003) Evaluation of NaCl tolerance in the callus cultures of *Suaeda nudiflora* Moq. Biol Plant 46:193–198
- Chiu CY, Hsiu FS, Chen SS, Chou CH (1995) Reduced toxicity of Cu and Zn to mangrove seedlings (*Kandelia candel* (L.) Druce.) in saline environments. Bot Bull Acad Sin 36:19–24
- Coleman J, Hench K, Garbutt K, Sexstone A, Bissonnette G, Skousen J (2001) Treatment of domestic wastewater by three plant species in constructed wetlands. Water Air Soil Pollut 128:283–295
- Colmer TD, Flowers TJ (2008) Flooding tolerance in halophytes. New Phytol 179:964–974
- Cushman JC (2001) Osmoregulation in plants: implications for agriculture. Am Zool 41:758–769
- Cushman JC (2003) Functional genomics of plant abiotic stress tolerance. In: Prade RA, Bohnert HJ (eds) Genomics of plants and fungi. Marcel Dekker, New York, pp 315–357, 18

- Cushman JC, Bohnert HJ (2000) Genomic approaches to plant stress tolerance. Curr Opin Plant Biol 3:117–124
- Delauney AJ, Verma DPS (1993) Proline biosynthesis and osmoregulation in plants. Plant J 4:215–223
- Eid MA, Eisa SS (2010) The use of some halophytic plants to reduce Zn, Cu and Ni in soil. Aust J Basic Appl Sci 4:1590–1596
- Epstein E (1980) Response of plants to saline environments. In: Rains DW, Valentine RC, Hollaender A (eds) Genetic engineering of osmoregulation. Plenum, New York, pp 7–21
- Fan W, Zhang Z, Zhang Y (2009) Cloning and molecular characterization of fructose-1,6-bisphosphate aldolase gene regulated by high-salinity and drought in *Sesuvium portulacastrum*. Plant Cell Rep. doi:10.1007/ s00299-009-0702-6
- Fang ZQ, Yuan LY, Hong PC, Ming LC, Shan WB (2005) NaCl enhances thylakoid-bound SOD activity in the leaves of C3 halophyte *Suaeda salsa* L. Plant Sci 168:423–430
- Fang Q, Xu Z, Song R (2006) Cloning, characterization and genetic engineering of FLC homolog in *Thellungiella halophila*. Biochem Biophys Res Commun 347:707–714
- Fitzgerald EJ, Caffrey JM, Nesaratnam ST, McLoughlin P (2003) Copper and lead concentrations in salt marsh plants on the suir Estuary. Ireland Environ Pollut 123:67–74
- Flowers TJ (1985) Physiology of halophytes. Plant Soil 89:41–56
- Flowers TJ (2004) Improving crop salt tolerance. J Exp Bot 55:307–319
- Flowers TJ, Colmer TD (2008) Salinity tolerance in halophytes. New Phytol 179:945–963
- Flowers TJ, Troke PF, Yeo AR (1977) The mechanism of salt tolerance in halophytes. Annu Rev Plant Physiol 28:89–121
- Flowers TJ, Galal HK, Bromham L (2010) Evolution of halophytes: multiple origins of salt tolerance. Funct Plant Biol 37:604–612
- Food and Agricultural Organization (FAO) (2003) New global mangrove estimate. http://www.fao.org/ forestry/foris/webview/forestry2/index.jsp%3Fgeold= 0%26langid
- Fowler S, Lee K, Onouchi H, Samach A, Richardson K, Morris B, Coupland G, Putterill J (1999) GIGANTEA: a circadian clock-controlled gene that regulates photoperiodic flowering in Arabidopsis and encodes a protein with several possible membrane-spanning domains. EMBO J 18:4679–4688
- Foyer CH, Noctor G (2005) Oxidant and antioxidant signaling in plants: a re-evaluation of the concept oxidative stress in a physiological context. Plant Cell Environ 28:1056–1071
- Fukushima Y, Sasamoto H, Baba S, Ashihara H (1997) The effect of salt stress on the catabolism of sugars in leaves and roots of a mangrove plant *Avicennia marina*. Z Naturforsch 52:187–192
- Gaxiola RA, Palmgren MG, Schumachner K (2007) Plant proton pumps. FEBS Lett 581:2204–2214

- Ghnaya T, Slama I, Messedi D, Grignon C, Ghorbel MH, Abdelly C (2007) Effect of Cd²⁺ on K⁺, Ca⁺ and N uptake in two halophytes *Sesuvium portulacastrum* and *Mesembrynathemum crystallinum*: consequences on growth. Chemosphere 67:72–79
- Glenn EP, Brown JJ (1999) Salt tolerance and crop potential of halophytes. Crit Rev Plant Sci 18(2):227–255
- Glenn E, Miyamoto M, Moore D, Brown JJ, Thompson TL, Brown P (1997) Water requirements for cultivating *Salicornia bigelovii* Torr. with seawater on sand in a coastal desert environment. J Arid Environ 36:711–730
- Gonzalez JA, Gallardo M, Hilal M, Rosa M, Prado FE (2009) Physiological responses of quinoa (*Chenopodium quinoa* Willd.) to drought and water logging stresses: dry matter partitioning. Bot Stud 50:35–42
- Goudie AS (1990) Soil salinity-causes and controls. In: Techniques for desert reclamation (ed) J. Wiley and Sons Ltd., Wiley, Chichester, England, pp 110–111
- Guo XL, Cao YR, Cao ZY, Zhao YX, Zhang H (2004) Molecular cloning and characterization of a stressinduced peroxiredoxin Q gene in halophyte *Suaeda salsa*. Plant Sci 167:969–975
- Hafsi C, Romero-Puertas MC, Gupta DK, del Riob LA, Sandalio LM, Abdelly C (2010) Moderate salinity enhances the antioxidative response in the halophyte *Hordeum maritimum* L. under potassium deficiency. Environ Exp Bot 69:129–136
- Hare PD, Cress WA (1997) Metabolic implications of stress-induced proline accumulation in plants. Plant Growth Regul 21:79–102
- Hollington PA, Hussain Z, Kahlown MA, Abdullah M (2001) Success stories in saline agriculture in Pakistan: from research to production and development. In: BAC saline agriculture conference Dubai, 19–21 Mar 2001
- Hong Z, Lakkineni K, Zhang Z, Verma DPS (2000) Removal of feedback inhibition of Δ¹-Pyrroline-5-Carboxylate Synthetase results in increased proline accumulation and protection of plants from osmotic stress. Plant Physiol 122:1129–1136
- Horie T, Schroeder JI (2004) Sodium transporters in plants. Diverse genes and physiological functions. Plant Physiol 136:2457–2462
- Huang J, Reneau RB Jr, Hagedorn C (2000) Nitrogen removal in constructed wetlands employed to treat domestic wastewater. Water Res 34:2582–2588
- Hurst AC, Grams TEE, Ratajczak R (2004) Effects of salinity, high irradiance, ozone, and ethylene on mode of photosynthesis, oxidative stress and oxidative damage in the C3/CAM intermediate plant *Mesembryanthemum crystallinum* L. Plant Cell Environ 27:187–197
- Hutchings P, Saenger P (1987) Ecology of mangroves. University of Queensland Press, New York
- Iwasaki K (1987) The effectiveness of salt-accumulating plants in reclaiming salinized soils. Jpn J Trop Agric 31:255–261
- Jebara S, Jebara M, Limam F, Aouani ME (2005) Changes in ascorbate peroxidase, catalase, guaiacol peroxidase and superoxide dismutase activities in common bean

(*Phaseolus vulgaris*) nodules under salt stress. J Plant Physiol 162:929–936

- Jennings DH (1968) Microelectrode experiments with potato cells: a re-interpretation of the experimental findings. J Exp Bot 19:13
- Jitesh MN, Prashanth SR, Sivaprakash KR, Parida AK (2006) Antioxidative response mechanism in halophytes: their role in stress defence. J Genet 85:237–254
- Kadukova J, Kalogerakis N (2007) Lead accumulation from non-saline and saline environment by *Tamarix smyrnensis* Bunge. Eur J Soil Biol 43:216–223
- Kadukova J, Manousaki E, Kalogerakis N (2008) Pb and Cd accumulation and phytoexcretion by salt cedar (*Tamarix smyrnensis* Bunge). Int J Phytoremediation 10:31–46
- Kant S, Kant P, Raveh E, Barak S (2006) Evidence that differential gene expression between the halophyte, *Thellungiella halophila*, and *Arabidopsis thaliana* is responsible for higher levels of the compatible osmolyte proline and tight control of Na+ uptake in *T. halophila*. Plant Cell Environ 29:1220–1234
- Kavitha K, Venkataraman G, Parida A (2008) An oxidative and salinity stress induced peroxisomal ascorbate peroxidase from *Avicennia marina*: molecular and functional characterization. Plant Physiol Biochem 46:794–804
- Kavitha K, George S, Venkataraman G, Parida A (2010) A salt-inducible chloroplastic monodehydroascorbate reductase from halophyte Avicennia marina confers salt stress tolerance on transgenic plants. Biochimie 92:1321–1329
- Ketchum REB, Warren RC, Klima LJ, Lopez-Gutierrez F, Nabors MW (1991) The mechanism and regulation of proline accumulation in suspension cultures of the halophytic grass *Distichlis spicata* L. J Plant Physiol 137:368–374
- Khan MA, Ansari R (2008) Potential use of halophytes with emphasis on fodder production in coastal areas of Pakistan. In: Abdelly C, Ozturk M, Ashraf M, Grignon C (eds) Biosaline agriculture and high salinity tolerance. Birkhauser Verlag, Basel, pp 157–162
- Khan MA, Qaiser M (2006) Halophytes of Pakistan: distribution, ecology, and economic importance. In: Khan MA, Barth H-J, Kust GC, Boer B (eds) Sabkha ecosystems: Vol II, The South and Central Asian countries. Springer, Drodrecht, pp 135–160
- Kore-eda S, Cushman MA, Akselrod I, Bufford D, Fredrickson M, Clark E, Cushman JC (2004) Transcript profiling of salinity stress responses by large-scale expressed sequence tag analysis in *Mesembryanthemum crystallinum*. Gene 341:83–92
- Kreps JA, Wu Y, Chang HS, Zhu T, Wang X, Harper JF (2002) Transcriptome changes for Arabidopsis in response to salt, osmotic, and cold stress. Plant Physiol 130:2129–2141
- Lacerda LD (1997) Trace metals in mangrove plants: why such low concentrations? In: Kjerfve B, Lacerda LD, Diop HS (eds) Mangrove ecosystem studies in Latin America and Africa. UNESCO, Paris, pp 171–178

- Lauchli A, Epstein E (1990) Plant responses to saline and sodic conditions. In: Tanji KK (ed) Agricultural salinity assessment and management. American Society of Civil Engineering, New York, pp 113–137
- Lefevre I, Marchal G, Meerts P, Correal E, Lutts S (2009) Chloride salinity reduces cadmium accumulation by the Mediterranean halophyte species *Atriplex halimus* L. Environ Exp Bot 65:142–152
- Lewis RR (2005) Ecological engineering for successful management and restoration of mangrove forests. Ecol Eng 24:403–418
- Lewis RR, Streever W (2000) Restoration of mangrove habitat. Tech Note ERDC TN-WRP-VN-RS-3. US Army, Corps of Engineers, Waterways Experiment Station, Vicksburg, MS
- Li TH, Li SH (2005) Leaf responses of micropropagated apple plants to water stress: nonstructural carbohydrate composition and regulatory role of metabolic enzymes. Tree Physiol 25:395–404
- Li QL, Liu DW, Gao XR, Su Q, An LJ (2003) Cloning of cDNA encoding choline mono-oxygenase from *Suaeda liaotungensis* and salt tolerance of transgenic tobacco. Acta Bot Sin 45:242–247
- Li Q, Yin H, Li D, Zhu H, Zhang Y, Zhu W (2007) Isolation and characterization of *CMO* gene promoter from halophyte *Suaeda* liaotungensis K. J Genet Genom 34:355–361
- Lilebo AI, Valega M, Otero M, Pardal MA, Pereira E, Duarte AC (2010) Daily and inter-tidal variations of Fe Mn and Hg in the water column of a contaminated salt marsh: halophytes effect. Estuar Coast Shelf Sci 88:91–98
- Lokhande VH, Nikam TD, Suprasanna P (2010a) Biochemical, physiological and growth changes in response to salinity in callus cultures of *Sesuvium portulacastrum* L. Plant Cell Tissue Organ Cult 102: 17–25
- Lokhande VH, Nikam TD, Suprasanna P (2010b) Differential osmotic adjustment to iso-osmotic salt and PEG stress *in vitro* in the halophyte Sesuvium portulacastrum L. J Crop Sci Biotechnol 13:251–256
- Lokhande VH, Srivastava AK, Srivastava S, Nikam TD, Suprasanna P (2010c) Regulated alterations in redox and energetic status are the key mediators of salinity tolerance in the halophyte *Sesuvium portulacastrum* (L.) L. Plant Growth Regul DOI 10.1007/s10725-011-9600-3
- Lokhande VH, Nikam TD, Patade VY, Ahire ML, Suprasanna P (2011a) Effects of optimal and supraoptimal salinity stress on antioxidative defence, osmolytes and in vitro growth responses in *Sesuvium portulacastrum* L. Plant Cell Tissue Organ Cult 104:41–49
- Lokhande VH, Srivastava S, Patade VY, Dwivedi S, Tripathi RD, Nikam TD, Suprasanna P (2011b) Investigation of arsenic accumulation and tolerance in *Sesuvium portulacastrum* (L.). Chemosphere 82:529–534
- Lokhande VH, Srivastava AK, Srivastava S, Nikam TD, Suprasanna P (2011c) Regulated alterations in redox and energetic status are the key mediators of salinity

tolerance in the halophyte *Sesuvium portulacastrum* (L.) L. Plant Growth Regul DOI 10.1007/s10725-011-9600-3

- Lovelock CE, Ball MC (2002) Influence of salinity on photosynthesis of halophytes. In: Lauchli A, Luttge U (eds) Salinity: environment-plant-molecules. Kluwer, Dordrecht, pp 315–339
- M'rah S, Ouerghi Z, Berthomieu C, Havaux M, Jungas C, Hajji M, Grignon C, Lachaal M (2006) Effects of NaCl on the growth, ion accumulation and photosynthetic parameters of *Thellungiella halophila*. J Plant Physiol 163:1022–1031
- MacFarlane GR, Burchett MD (1999) Zinc distribution and excretion in the leaves of the grey mangrove, *Avicennia marina* (Forsk.) Vierh. Environ Exp Bot 41:167–175
- MacFarlane GR, Burchett MD (2000) Cellular distribution of copper, lead and zinc in the grey mangrove, *Avicennia marina* (Forsk.) Vierh. Aquat Bot 68:45–59
- MacFarlane GR, Burchett MD (2002) Toxicity, growth and accumulation relationship of copper, lead and zinc in the grey mangrove, *Avicennia marina* (Forsk.) Vierh.: biological indication potential. Environ Pollut 123:139–151
- MacFarlane GR, Koller CE, Blomberg SP (2007) Accumulation and partitioning of heavy metals in mangroves: a synthesis of field-based studies. Chemosphere 69:1454–1464
- Manousaki E, Kalogerakis N (2009) Phytoextraction of Pb and Cd by the Mediterranean saltbush (*Atriplex halimus* L.): metal uptake in relation to salinity. Environ Sci Pollut Res 16:844–854
- Manousaki E, Kadukova J, Papadonatonakis N, Kalogerakis and N (2008) Phytoextraction and Phytoexcretion of Cd by *Tamarix smyrnensis* growing on contaminated non-saline and saline soils. Environ Res 106:326–332
- Martinez JP, Kinet JM, Bajji M, Lutts S (2005) NaCl alleviates polyethylene glycol-induced water stress in the halophyte species *Atriplex halimus* L. J Exp Bot 56:2421–2431
- Masters DG, Benes SE, Norman HC (2007) Biosaline agriculture for forage and livestock production. Agric Ecosyst Environ 119:234–248
- Megdichi W, Ben Amor N, Debez A, Hessini K, Ksouri R, Zuily-Fodil Y, Abdelly C (2007) Salt tolerance of the annual halophyte *Cakile maritima* as affected by the provenance and the developmental stage. Acta Physiol Plant 29:375–384
- Menzel U, Lieth H (2003) HALOPHYTE Database V. 2.0 update. In: Lieth H, Mochtchenko M (eds) Cash crop halophytes. Kluwer, Dordrecht, CD-ROM
- Messedi D, Labidi N, Grignon C, Abdelly C (2004) Limits imposed by salt to the growth of the halophyte Sesuvium portulacastrum. J Plant Nutr Soil Sci 167:720–725
- Miller G, Suzuki N, Ciftci-Yilmaz S, Mittler R (2010) Reactive oxygen species homeostasis and signaling during drought and salinity stresses. Plant Cell Environ. 33(4):453–467

- Mitsch WJ, Gosselink JG (2007) Wetlands, 4th edn. Wiley, New York
- Mittler M (2002) Oxidative stress, antioxidants and stress tolerance. Trends Plant Sci 7:405–410
- Mittler R, Vanderauwera S, Gollery M, Breusegem FV (2004) Reactive oxygen gene network of plants. Trends Plant Sci 9:490–498
- Moghaieb REA, Saneoka H, Fujita K (2004) Effect of salinity on osmotic adjustment, glycine betaine accumulation and the betaine aldehyde dehydrogenase gene expression in two halophytic plants, Salicornia europaea and Suaeda maritime. Plant Sci 166:1345–1349
- Moseki B, Buru JC (2010) Ionic and water relations of Sesuvium portulacastrum (L). Sci Res Essay 5:35–40
- Mulholland MM, Otte ML (2002) The effects of nitrogen supply and salinity on DMSP, glycine betaine- and proline concentrations in leaves of *Spartina anglica*. Aquat Bot 72:193–200
- Munns R (2002) Comparative physiology of salt and water stress. Plant Cell Environ 25:239–250
- Munns R, Tester M (2008) Mechanisms of salinity tolerance. Annu Rev Plant Biol 59:651–681
- Nah G, Pagliarulo CL, Mohr PG, Luo M, Sisneros N, Yu Y, Collura K, Currie J, Goicoechea JL, Wing RA et al (2009) Comparative sequence analysis of the SALT OVERLY SENSITIVE1 orthologous region in Thellungiella halophila and Arabidopsis thaliana. Genomics 94:196–203
- Nedjimi B, Daoud Y (2009) Cadmium accumulation in Atriplex halimus subsp. schweinfurthii and its influence on growth, proline, root hydraulic conductivity and nutrient uptake. Flora 204:316–324
- Noctor G, Foyer CH (1998) Ascorbate and glutathione. Keeping active oxygen under control. Ann Rev Plant Physiol Mol Biol 49:249–279
- Nouairi I, Ghnaya T, Youssef NB, Zarrouk M, Ghorbel MH (2006) Changes in content and fatty acid profiles of total lipids of two halophytes: Sesuvium portulacastrum and Mesembryanthemum crystallinum under cadmium stress. J Plant Physiol 163:1198–1202
- Oh D-H, Leidi E, Zhang Q, Hwang S-M, Li Y, Quintero FJ, Jiang X, D'Urzo MP, Sang Lee Y, Zhao Y, Bahk JD, Bressan RA, Yun D-J, Pardo JM, Bohnert HJ (2009) Loss of halophytism by interference with SOS1 expression. Plant Physiol 15:210–222
- Ohta M, Hayashi Y, Nakashima A, Hamada A, Tanaka A, Nakamura T, Hayakawa T (2002) Introduction of a Na⁺/H⁺ antiporter gene from *Atriplex gmelini* confers salt tolerance to rice. FEBS Lett 532:279–282
- Ozawa T, Miura M, Fukuda M, Kakuta S (2009) Cadmium tolerance and accumulation in a halophyte *Salicornia europaea* as a new candidate for phytoremediation of saline soils. Sci Rep Grad Sch Life Environ Sci Osaka Pref Univ 60:1–8
- Pagter M, Bragato C, Malagoli M, Brix H (2009) Osmotic and ionic effects of NaCl and Na₂SO₄ salinity on *Phragmites australis*. Aquat Bot 90:43–51
- Paramonova NV, Shevyakova NI, Kuznetsov VV (2004) Ultrastructure of chloroplasts and their storage inclusions in the primary leaves of *Mesembryanthemum*

crystallinum affected by putrescine and NaCl. Russ J Plant Physiol 1:86–96

- Parida AK, Das AB, Mohanty P (2004) Investigations on the antioxidative defense responses to NaCl stress in a mangrove, Bruguirea parviflora: differential, regulations of isoforms of some antioxidant enzymes. Plant Growth Regul 42:213–226
- Patra J, Panda BB (1998) A comparison of biochemical responses to oxidative and metal stress in seedlings of barley, *Hordeum vulgare* L. Environ Pollut 101:99–105
- Peters EC, Gassman NJ, Firman JC, Richmond RH, Power EA (1997) Ecotoxicology of tropical marine ecosystems. Environ Toxicol Chem 16:12–40
- Pilon-Smits EAH, de Souza MP, Hong G, Amini A, Bravo RC et al (1999) Selenium volatilization and accumulation by twenty aquatic plant species. J Environ Qual 28:1011–1017
- Popova OV, Yang O, Dietz KJ, Golldack D (2008) Differential transcript regulation in *Arabidopsis thaliana* and the halotolerant *Lobularia maritima* indicates genes with potential function in plant salt adaptation. Gene 423:142–148
- Potters G, Horemans N, Jansen MAK (2010) The cellular redox state in plant stress biology – A charging concept. Plant Physiol Biochem 48:292–300
- Qadir M, Oster J (2004) Crop and irrigation management strategies for saline–sodic soils and waters aimed at environmentally sustainable agriculture. Sci Total Environ 323:1–19
- Qi CH, Chen M, Song J, Wang BS (2009) Increase in aquaporin activity is involved in leaf succulence of the euhalophyte *Suaeda salsa*, under salinity. Plant Sci 176:200–205
- Queval G, Noctor G (2007) A plate-reader method for the measurement of NAD, NADP, glutathione and ascorbate in tissue extracts. Application to redox profiling during Arabidopsis rosette development. Anal Biochem 363:58–69
- Rabhi M, Hafsi C, Lakhdar A, Barhoumi Z, Hamrouni MH, Abdelly C, Smauoi A (2009) Evaluation of the capacity of three halophytes to desalinize their rhizosphere as grown on saline soils under nonleaching conditions. Afr J Ecol 47:463–468
- Rabhi M, Ferchichi S, Jouini J, Hamrouni MH, Koyro HW, Ranieri A, Abdelly C, Smaoui A (2010) Phytodesalination of a salt-affected soil with the halophyte Sesuvium portulacastrum L. to arrange in advance the requirements for the successful growth of a glycophytic crop. Bioresour Technol 101: 6822–6828
- Ramadan T (2000) Dynamics of salt secretion by Sporobolus Spicatus (Vahl) kunth from sites of differing salinity. Ann Bot 87:259–266
- Rammesmayer G, Pichorner H, Adams P, Jensen RG, Bohnert HJ (1995) Characterization of IMT1, myoinositol O-methyltrasferase, from Mesembryanthemum crystallinum. Arch Biochem Biophys 322:183–188
- Raven JA (1985) Regulation of pH and generation of osmolarity in vascular land plants: costs and benefits

in relation to efficiency of use of water, energy and nitrogen. New Phytol 101:25-77

- Ravindran KC, Venkatesan K, Balakrishnan V, Cehllappan KP, Balasubramanian T (2007) Restoration of saline land by halophytes for Indian soils. Soil Biol Biochem 39:2661–2664
- Reboreda R, Caçador I (2007) Halophyte vegetation influences in salt marsh retention capacity for heavy metals. Environ Pollut 146:147–154
- Reboreda R, Caçador I (2008) Enzymatic activity in the rhizosphere of *Spartina maritima*: potential contribution for phytoremediation of metals. Mar Environ Res 65:77–84
- Reda EAM, Saneoka H, Fujita K (2004) Effect of salinity on osmotic adjustment, glycinebetaine accumulation and the betaine aldehyde dehydrogenase gene expression in two halophyte plants, Salicornia europaea and Suaeda maritime. Plant Sci 166:1345–1349
- Reddy MP, Shah MT, Patolia JS (2008) Salvadora persica, a potential species for industrial oil production in semiarid saline and alkali soils. Ind Crops Prod 28:273–278
- Redondo-Gómez S, Mateos-Naranjo E, Andrades-Moreno L (2010a) Accumulation and tolerance characteristics of cadmium in a halophytic Cd-hyperaccumulator, *Arthrocnemum macrostachyum*. J Hazard Mater 184:299–307
- Redondo-Gómez S, Mateos-Naranjo E, Figueroa ME, Davy AJ (2010b) Salt stimulation of growth and photosynthesis in extreme halophyte, Arthrocnemum macrostachyum. Plant Biol 12:79–87
- Rhodes D, Hanson AD (1993) Quaternary ammonium and tertiary sulfonium compounds in higher plants. Ann Rev Plant Physiol Mol Biol 44:357–384
- Robinson SP, Jones GP (1986) Accumulation of glycine betaine in chloroplasts provides osmotic adjustment during salt stress. Aust J Plant Physiol 13:659–668
- Rozema J (1991) Growth, water and ion relationships of halophytic monocotyledonae and dicotyledonae – a unified concept. Aquat Bot 39:17–33
- Russell BL, Rathinasabapathi B, Hanson AD (1998) Osmotic stress induces expression of choline monooxygenase in sugar beet and amaranth. Plant Physiol 116:859–865
- Sadiq M, Zaidi TH (1994) Sediment composition and metal concentrations in mangrove leaves from the Saudi coast of the Arabian Gulf. Sci Total Environ 155:1–8
- Saenger P (2002) Mangrove ecology, silviculture and conservation. Kluwer, Dordrecht
- Saenger P, Siddiqi NA (1993) Land from the seas: the mangrove afforestration program of Bangladesh. Ocean Coast Manag 20:23–39
- Seki M, Ishida J, Narusaka M, Fujita M, Nanjo T, Umezawa T, Kamiya A, Nakajima M, Enju A, Sakurai T, Satou M, Akiyama K, Yamaguchi-Shinozaki K, Carninci P, Kawai J, Hayashizaki Y, Shinozaki K (2002) Monitoring the expression profiles of 7000 Arabidopsis genes under drought, cold and highsalinity stresses using a full-length cDNA microarray. Plant J 3:279–292

- Sekmen AH, Turkan I, Takio S (2007) Differential responses of antioxidative enzymes and lipid peroxidation to salt stress in salt-tolerant Plantago maritima and salt-sensitive Plantago media. Physiol Plant 131:399–411
- Seliskar DM, Gallagher JL (2000) Exploiting wild population diversity and somaclonal variation in the salt marsh grass *Distichlis spicata* (Poaceae) for marsh creation and restoration. Am J Bot 87:141–146
- Shahid SA (2002) New technologies for soil reclamation and desert greenery In: Nader MA, Faisal KT (eds) Proceedings of the joint KISR – PEC symposium Yazd, Iran, pp 308–329
- Sharma BR, Minhas PS (2005) Strategies for managing saline/alkali waters for sustainable agricultural production in South Asia. Agric Water Manag 78:136–151
- Shen YG, Du BX, Zhang JS, Chen SY (2001) Cloning and characterization of CMO gene from Atriplex hortensis. Chin J Biotechnol 17:1–6
- Shi H, Ishitani M, Kim C, Zhu JK (2000) The Arabidopsis thaliana salt tolerance gene SOS1 encodes a putative Na+/H+antiporter. Proc Natl Acad Sci USA 97:6896–6901
- Silveira JAG, Araujo SAM, Lima JPMS, Viegas RA (2009) Roots and leaves display contrasting osmotic adjustment mechanisms in response to NaCl-salinity in *Atriplex nummularia*. Environ Exp Bot 66:1–8
- Slama I, Ghnaya T, Savou e A, Abdelly C (2008) Combined effects of long-term salinity and soil drying on growth, water relations, nutrient status and proline accumulation of *Sesuvium portulacastrum*. C R Biol 331:442–451
- Slesak I, Miszalski Z (2003) Superoxide dismutase-like protein from roots of the intermediate C3-CAM plant *Mesembryanthemum crystallinum* L. in *in vitro* culture. Plant Sci 164:497–505
- Slesak I, Miszalski Z, Karpinska B, Niewiadomska E, Ratajczak R, Karpinski S (2002) Redox control of oxidative stress responses in the C3-CAM intermediate plant *Mesembryanthemum crystallinum*. Plant Physiol Biochem 40:669–677
- Slesak I, Slesak H, Libik M, Miszalski Z (2008) Antioxidant response system in the short-term postwounding effect in *Mesembryanthemum crystallinum* leaves. J Plant Physiol 165:127–137
- Smirnoff N (2005) Ascorbate, tocopherol and carotenoids: metabolism, pathway engineering and functions. In: Smirnoff N (ed) Antioxidants and reactive oxygen species in plants. Blackwell, Oxford, pp 53–86
- Sousa AI, Caçador I, Lillebo AI, Pardal MA (2008) Heavy metal accumulation in *Halimione portulacoides*: intra -and extra-cellular metal binding sites. Chemosphere 70:850–857
- Srinivas V, Balasubramanian D (1995) Proline is a protein-compatible hydrotrope. Langmuir 11:2830–2833
- Subbarao GV, Levine LH, Wheeler RM, Stutte GW (2001) Glycine betaine accumulation: its role in stress resistance in crop plants. In: Pessarakli M (ed) Handbook

of plant and crop physiology. Marcel Dekker, New York, pp 881–907

- Sucre B, Suarez N (2010) Effect of salinity and PEGinduced water stress on water status, gas exchange, solute accumulation, and leaf growth in *Ipomoea pescaprae*. Environ Exp Bot 70:192–203
- Szabados L, Savoure A (2009) Proline: a multifunctional amino acid. Trends Plant Sci 15:89–97
- Tabuchi T, Kawaguchi Y, Azuma T, Nanmori T, Yasuda T (2005) Similar regulation patterns of choline monooxygenase, phosphoethanolamine N-methyltransferase and S-adenosyl-L-methionine synthetase in leaves of halophyte *Atriplex nummularia* L. Plant Cell Physiol 46:505–513
- Taji T, Seki M, Satou M, Sakurai T, Kobayashi M, Ishiyama K, Narusaka Y, Narusaka M, Zhu JK, Shinozaki K (2004) Comparative genomics in salt tolerance between *Arabidopsis* and *Arabidopsis*-related halophyte salt cress using *Arabidopsis* microarray. Plant Physiol 135:1697–1709
- Takemura T, Hanagata N, Sugihara K, Baba S, Karube I, Dubinsky Z (2000) Physiological and biochemical responses to salt stress in the mangrove, *Bruguiera* gymnorrhiza. Aquat Bot 68:15–28
- Tester M, Davenport R (2003) Na⁺ tolerance and Na⁺ transport in higher plants. Ann Bot 91:503–5027
- Tipirdamaz R, Gagneul D, Duhaze C, Ainouche A, Monnier C, Zokum D, Larher F (2006) Clustering of halophytes from an inland salt marsh in Turkey according to their ability to accumulate sodium and nitrogenous osmolytes. Environ Exp Bot 57:139–153
- Vicente O, Boscaiu M, Naranjo MA, Estrelles E, Belles JM, Soriano P (2004) Responses to salt stress in the halophyte *Plantago crassifolia* (Plantaginaceae). J Arid Environ 58:463–481
- Waisel Y (1972) Biology of halophytes. Academic, New York
- Wang W, Vinocur B, Altman A (2003) Plant responses to drought, salinity and extreme temperatures: towards genetic engineering for stress tolerance. Planta 218:1–14
- Wang B, Luttge U, Ratajczak R (2004a) Specific regulation of SOD isoforms by NaCl and osmotic stress in leaves of the C3 halophyte *Suaeda salsa* L. J Plant Physiol 161:285–293
- Wang LW, Showalter AM (2004b) Cloning and saltinduced, ABA-independent expression of choline monooxygenase in Atriplex prostrate. Physiol Plant 120:405–412
- Wang B, Davenport RJ, Volkov V, Amtmann A (2006) Low unidirectional sodium influx into root cells restricts net sodium accumulation in *Thellungiella halophila*, a salt-tolerant relative of *Arabidopsis thaliana*. J Exp Bot 57:1161–1170
- Wang CQ, Xu C, Wei JG, Wang HB, Wang SH (2008a) Enhanced tonoplast H⁺ -ATPase activity and superoxide dismutase activity in the halophyte *Suaeda salsa* containing high level of betacyanin. J Plant Growth Regul 27:58–67
- Wang KS, Huang LC, Lee HS, Chen PY, Chang SH (2008b) Phytoextraction of cadmium by *Ipomoea aquatica* (water spinach) in hydroponic solution: effects of cadmium speciation. Chemosphere 72:666–672
- Wong CE, Yong L, Aurelie L, David G, Paulo N, Brett W, Claudia D, Brian GG, Gray GR, Weretilnyk EA, Griffith M, Moffatt BA (2006) Transcriptional profiling implicates novel interactions between abiotic stress and hormonal responses in *Thellungiella*, a close relative of *Arabidopsis*. Plant Physiol 140:1437–1450
- Yadav SK (2010) Heavy metal toxicity in plants: an overview on the role of glutathione and phytochelatins in heavy metal stress tolerance of plants. S Afr J Bot 76:167–179
- Yang YL, Shi R, Wei X, Fan Q, An L (2010) Effect of salinity on antioxidant enzymes in calli of the halophyte *Nitraria tangutorum* Bobr. Plant Cell Tissue Organ Cult 102:387–395
- Yensen NP (2006) Halophyte uses for the twenty-first century. In: Khan MA, Weber DJ (eds) Ecophysiology of high salinity tolerant plants. Springer, Dordrecht, pp 367–396
- Yeo AR, Flowers TJ (1986) Salinity resistance in rice (*Oryza sativa* L.) and a pyramiding approach to breeding varieties for saline soils. Aust J Plant Physiol 13:161–173
- Yin X, Zhao Y, Luo D, Zhang H (2002) Isolating the promoter of a stress-induced gene encoding betaine aldehyde dehydrogenase from the halophyte *Atriplex centralasiatica* Iljin. Biochim Biophys Acta 1577:452–456
- Yuanyuan M, Yali Z, Jiang L, Hongbo S (2009) Roles of plant soluble sugars and their responses to plant cold stress. Afr J Plant Biotechnol 8:2004–2010
- Zabłudowska E, Kowalska J, Jedynak L, Wojas S, Skłodowska A, Antosiewicz DM (2009) Search for a

plant for phytoremediation – What can we learn from field and hydroponic studies? Chemosphere 77:301–307

- Zahran MA, Wahid AAA (1982) Halophytes and human welfare. In: Sen DN, Rajpurohit KS (eds) Contributions to the ecology of halophytes. D.W. Junk Publishers, Boston, pp 235–257
- Zaier H, Ghnaya T, Lakhdar A, Baioui R, Ghabriche R, Mnasri M, Sghair S, Lutts S, Abdelly C (2010a) Comparative study of Pb-phytoextraction potential in *Sesuvium portulacastrum* and *Brassica juncea*: tolerance and accumulation. J Hazard Mat. doi:10.1016/j. jhazmat.2010.07.068
- Zaier H, Mudarra A, Kutscher D, Fernandez de la Campa MR, Abdelly C, Sanz-Medel A (2010b) Induced lead binding phytochelatins in *Brassica juncea* and *Sesuvium portulacastrum* investigated by orthogonal chromatography inductively coupled plasma-mass spectrometry and matrix assisted laser desorption ionization-time of flight-mass spectrometry. Anal Chim Acta. doi:10.1016/j.aca.2010.04.054
- Zhang F, Yang YL, He WL, Zhao X, Zhang LX (2004) Effects of salinity on growth and compatible solutes of callus induced from *Populus euphratica*. In Vitro Cell Dev Biol Plant 40:491–494
- Zhang Y, Lai J, Sun S, Li Y, Liu Y, Liang L, Chen M, Xie Q (2008) Comparison analysis of transcripts from the halophyte *Thellungiella halophila*. J Integr Plant Biol 50:1327–1335
- Zhao KF (1991) Desalinization of saline soils by *Suaeda* salsa. Plant Soil 135:303–305
- Zhao K, Hai F, Ungar IA (2002) Survey of halophytes species in China. Plant Sci 163:491–498
- Zhu J-K (2001) Plant salt tolerance. Trends Plant Sci 6:66–71

3

UV-B Radiation, Its Effects and Defense Mechanisms in Terrestrial Plants

Fernando E. Prado, Mariana Rosa, Carolina Prado, Griselda Podazza, Roque Interdonato, Juan A. González, and Mirna Hilal

Abstract

The UV-B is an important component of solar radiation to which all terrestrial and aquatic plants were exposed during the early evolutionary phase of the Earth. Hence the plants, principally terrestrial, have evolved different mechanisms to avoid and repair the UV-B damage; therefore, it is not surprising that photomorphogenic responses to the solar UV-B are erroneously assumed to be adaptations to the harmful UV radiation. The responses to UV-B enhancement include changes in the leaf area, leaf thickness, stomatal density, wax deposition, stem elongation, and branching pattern, as well as in the synthesis of secondary metabolites, alterations in plant-pathogen and plant-predator interactions, and in gene expression. However, under field conditions the ambient solar UV-B provides an important signal for the normal plant development and may be perceived by the plants through nondestructive processes involving both UV-B specific and UV-B nonspecific signaling pathways. The specific signaling pathways include the components UVR8 and COP1 which regulate the expression of a set of genes that are essential for the plants' protection. The nonspecific signaling pathways involve DNA damage, reactive oxygen species (ROS), hormones, and wound/defense signaling molecules. Indeed under the field conditions, the ambient UV-B might more properly be viewed as a photomorphogenic signal than as a stressor. Therefore, it might not be appropriate to evaluate the adaptive roles of plant responses to UV-B cues upon stress tolerance by the simultaneous application of both solar radiation and supplemental UV-B. In this chapter, we analyzed the information regarding

G. Podazza • J.A. González Instituto de Ecología, Fundación Miguel Lillo, Miguel Lillo 251, CP 4000 Tucumán, Argentina

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_3, © Springer Science+Business Media, LLC 2012

<sup>F.E. Prado (⊠) • M. Rosa • C. Prado
• R. Interdonato • M. Hilal
Cátedra de Fisiología Vegetal, Facultad de Ciencias
Naturales e IML, Miguel Lillo 205, CP 4000
Tucumán, Argentina
e-mail: fepra@csnat.unt.edu.ar</sup>

physiological and morphogenic responses of the terrestrial plants to the UV-B radiation, as well as the events related to UV-B perception, signal transduction, gene expression, and ROS formation from different studies carried out in greenhouses, growth chambers, and field conditions.

Keywords

UV-B radiation • DNA damage • DNA repair • Metabolites • Signaling

• Secondary metabolites • Morphogenic responses

1 Introduction: Knowing the Solar UV-B Radiation, a Historical Background

Within the electromagnetic radiation spectrum, the UV radiation describes a spectral range between 200 and 400 nm, which borders on the visible range. The UV radiation is divided into three effective types: UV-A (320-400 nm), UV-B (280-320 nm), and UV-C (200-280 nm). Less than 7% of the sun's radiation reaching the Earth's surface falls approximately in the range between 295 and 400 nm (UV-A and UV-B); the shorter UV wavelengths get filtered out by the stratospheric ozone. Therefore, the lower limit of shorter wavelengths of the solar UV radiation reaching the surface is determined by the stratospheric ozone layer. The stratospheric ozone absorbs virtually all the UV radiation ranging approximately 295 nm and lesser. Although the stratospheric ozone determines the amount of UV-B radiation that reaches the surface of the Earth, its level is significantly affected by variations in latitude and altitude. The level of UV-B radiation over tropical latitudes is higher than in temperate regions due to lesser atmospheric UV-B absorption determined by the solar angle and the ozone layer itself which is thinner in equatorial regions. Thus, the UV-B radiation is relatively high in tropical areas and relatively low in the polar regions. Increases of the UV-B irradiance with increasing elevation above sea level is also known, i.e., measurements of the UV-B irradiance show an average increase between 10 and 19% for every 1,000 m increase in elevation. Besides geographical factors, the atmospheric pollutants (e.g., smoke, aerosols) and especially the weather factors (e.g., clouds, haze) greatly decrease the level of UV-B reaching the Earth's surface. Depending upon the type and height of clouds, liquid water content, and particle distribution, the cover of clouds can attenuate over 70% of the incident UV-B radiation (McKenzie et al. 2007).

Over 35 years ago, it was warned that manmade nitrous oxide and chlorine-containing compounds (e.g., chlorofluorocarbons, CFCs) produce the breakdown of large amounts of ozone in the stratosphere (Crutzen 1972; Molina and Rowland 1974; in Velders et al. 2007). This fact causes the depletion of the stratospheric ozone layer increasing the UV-B radiation at the ground level, especially in Antarctic and Arctic regions as well as in high altitude areas (Ryan and Hunt 2005). By their contributions on chlorine-containing compounds and the depletion of the ozone layer, Crutzen, Molina, and Rowland were awarded with the Nobel Prize for chemistry in 1995. After the Molina and Rowland's work, Farman, Gardiner, and Shanklin, scientists of the British Antarctic Survey, shocked the scientific community during the early middle 1980s by publishing the results of a study showing a springtime ozone hole in the Antarctic ozone layer (Farman et al. 1985, in Ryan and Hunt 2005). This fact rang alarm bells worldwide and within the same year, 20 nations including most of the major CFCs producers signed the Vienna Convention, which established a framework for negotiating an international regulation on the ozone-depleting substances. After that, on September 16, 1987 the Montreal Protocol on "Substances that Deplete the Ozone Layer" was opened for nations signature and entered into force on January 1, 1989 (Velders et al. 2007). At present, after several amendments, all the countries in the United Nations have ratified the Montreal Protocol (http://ozone.unep.org/ Meeting_Documents/). However, today the ozone depletion is a global phenomenon and according to the European Ozone Research Coordinating Unit (EORCU) its amount approximately reaches 0.6% per year. The level of UV-B radiation in the biosphere varies, spatially and temporally, quite considerably but the depletion of the stratospheric ozone strongly affects its penetration (Ryan and Hunt 2005). Thus, despite reductions in the production and use of ozone-depleting chemicals, the potential of ozone depletion by anthropogenic emissions or natural causes (e.g., volcanoes) still remains. In this scenario, the level of stratospheric ozone will continue decaying with a severe decline occurring between 2010 and 2019 in the northern hemisphere that may result up to 50-60% increase in the springtime UV-B radiation according to the Global Climate Model (GCM) that is based on the simplified ozone-depletion chemistry (Taalas et al. 2000). Furthermore, the recovery of the stratospheric ozone to early 1980s levels is not predicted until roughly 2050.

2 Solar UV-B Radiation and the Life of Terrestrial Plants

Since the discovery of the ozone layer depletion (~30 years ago) the responses of microorganisms, animals, and terrestrial plants to solar UV-B radiation have been active subjects of many studies (Rozema 2000; Björn et al. 2002; Ryan and Hunt 2005). Prior to building of the atmospheric oxygen and then the stratospheric ozone layer, the UV-C radiation and high levels of both UV-B and UV-A would probably have reached the Earth's surface relatively unattenuated affecting all the living organisms. In this context, the UV radiation seems to be a ubiquitous factor in the course of terrestrial biota? Plant evolution from the early Archean era began as solitary photosynthetic cells (Cockell and Horneck 2001). The UV effects on terrestrial plants, which are principally detrimental, have been demonstrated

with some of the most essential components of the biochemical machinery, i.e., DNA molecule and photosystem II (PSII) (Singh et al. 2008). However, when the stratospheric ozone layer developed, the UV-A radiation and a minor portion of the UV-B wavelengths only could reach the Earth's surface due to the atmospheric absorption and scattering of the UV-C radiation. Therefore, how the UV radiation has altered the Earth's environment over geological time periods is essential for understanding the evolutionary history of the earth and also to understand how the UV-B radiation has contributed as selection pressure on the development of terrestrial plants (Björn and McKenzie 2007). In fact, Sagan (Sagan 1973 in Singh et al. 2008) first considered the UV radiation as a selection pressure on the early photosynthetic organisms, when our knowledge on biological effects of the UV radiation on plants was in its infancy. During the evolutionary history of the Earth, the terrestrial plants coevolved under different solar UV-B levels and may have experienced significantly higher UV-B irradiances than the current surface UV-B level (Cockell and Horneck 2001; Rozema et al. 2002). Thereby, the UV-B tolerance acquired earlier probably helps to explain why plants are distributed at lower latitudes or higher elevations, where UV-B irradiances are greater, are less sensitive to high levels of the UV-B radiation than those at higher latitudes and/or lower elevations (Turunen and Latola 2005; Ren et al. 2010). The present rate of atmospheric changes is so rapid that evolution may not keep up with it, particularly in long-living plants like trees. The UV-B environment of terrestrial plants is presently quite variable in both time and space, and thus organisms experience different UV-B doses and adapt to UV-B radiation at different levels (Rozema 2000). In this context it is expected that terrestrial plants respond differentially to increasing solar UV-B. Nevertheless, although studies assessing possible consequences of the ozone depletion have greatly increased our understanding on how living organisms are affected by the UV-B radiation, the focus of these researches may also have distracted the attention from the UV-B radiation as a component of the ambient light environment involved in the evolution of life on the Earth's surface (Cockell and Horneck 2001).

3 UV-B Radiation as a Modulator of the Plant Function

Due to their absolute sunlight requirement for survival, the plants are inevitably exposed to solar UV-B radiation. However, from equatorial to the polar regions and from the sea level to high mountains, the terrestrial plants are exposed to greatly different UV-B irradiances, given the geographical differences in UV-B irradiances is much greater than corresponding differences in the total solar radiation (Rozema 2000). The plant chemical photoprocesses respond differently to different UV wavelengths as the biological damage exacerbated as wavelength becomes shorter. Thus, the relative effectiveness of UV-B-ranging wavelengths (effective UV-B irradiance) must be known in order to assess the responses to ozone changes. The effective UV irradiance (E) or dose rate exposure is given by

$$\mathbf{E} = \int \mathbf{F}(\lambda) \mathbf{W}(\lambda) d\lambda,$$

where $W(\lambda)$ is the weighting function (action spectrum) for a specific biological or chemical effect and $F(\lambda)$ is the spectral irradiance, either computed or measured, for a given time (e.g., hour, day, year) and location. As a result, the biological effectiveness of the weighted UV-B irradiances (UV-BBE, biologically effective UV-B radiation) related to different action spectra has different responses to atmospheric ozone changes (Flint and Caldwell 2003). It has been estimated that 1% decrease in the stratospheric ozone concentration would result in nearly 2% increase in the UV-BBE at mid-latitudes. Therefore, the recently projected 15% stratospheric ozone reduction would result in up to 30% increase in the value of UV-BBE in the next three decades (McKenzie et al. 2007).

4 Solar UV-B Radiation: Stress Factor or Beneficial Signal Factor?

From the ozone depletion perspective, the UV-B radiation is considered as an environmental stressor of photosynthetic organisms (Jordan 2002;

Rozema et al. 2002; Ballaré 2003; Caldwell et al. 2007). However, from an evolutionary perspective this assumption is questionable. The terrestrial plants have always developed under the solar UV-B and then their genetic machinery coevolved together with the ambient UV-B level. Therefore, it can be hypothesized that the metabolic machinery of plants contains all the necessary elements for a normal coexistence with the current UV-B level and so the solar UV-B radiation should not be considered as an environmental stress factor. In fact, the current level of the ambient UV-B radiation should be considered as a signal factor that induces the expression of genes related to the normal plant development (Jenkins 2009). While the UV-B exclusion must be considered as an anomalous signal factor that induces the expression and/or repression of another set of genes (Brosché et al. 2002; Stratmann 2003; Hectors et al. 2007). In this context, the solar UV-B radiation appears as a reliable plant effector, but it is not always possible to identify a unique particular reason as explanation of the underlying UV-B effects. In nature, the terrestrial plants are seldom affected by only a single environmental factor; they typically respond to several environmental factors acting in concert (Bruno et al. 2003). Therefore, the influence of changing UV-B in natural ecosystems must be evaluated considering two opposite processes: (a) facilitation; (b) competition. These processes have been recognized as key drivers in a wide range of natural communities and hence, effectiveness of the UV-B radiation will be greatly modified by other environmental factors, in some cases aggravating and in others, ameliorating the overall UV-B effect (Bruno et al. 2003).

5 Responses of Terrestrial Plants to Ambient Solar UV-B

The most extended researches relating to the effects of increasing solar UV-B radiation on terrestrial plants have been performed in both austral and boreal polar regions (Day et al. 2001; Phoenix et al. 2003; Robson et al. 2003; Rozema et al. 2006; Newsham and Robinson 2009). The depletion of the stratospheric ozone is greater in both Antarctic and Arctic regions than in other nonpolar latitudes where it is less pronounced and subject to other atmospheric factors such as horizontal and vertical ozone transport. In the Antarctic zone, the complete breakdown of the stratospheric ozone occurs only during few springtime days, but the springtime ozone depletion reaches 50-60% on average (Rozema et al. 2005). This event has occurred uninterruptedly for at least 30 years, leading to a marked increase of the solar UV-B irradiance. Since the 1990s frequent occurrence of the springtime ozone hole over the Arctic also occurs resulting in significant ozone depletion and increasing the UV-B irradiance at the ground level (Rex et al. 2004). Similar to the Antarctic area, the ozone loss over the Arctic area is higher in the early springtime than in the growing plant season (late springtime and summer). The Arctic springtime ozone depletion is lower than the Antarctic one and rarely reaches 40–50% on average (Rex et al. 2004). Also, the Arctic ozone loss is extremely sensitive to frequency of sudden stratospheric warming due to the greenhouse effect. Because of the influence of increasing greenhouse gases, the ozone holes may worsen leading to greater ozone depletion over the Arctic and increasing the severity and duration of the Antarctic ozone depletion (Rozema et al. 2005). At present both Antarctic and Arctic polar regions represent one the most extreme UV-B environment and constitute an excellent site to study the responses of terrestrial plants to increased solar UV-B. Although almost all the investigations were carried out in the Antarctic area this ecosystem only has two species of higher plants: Deschampsia antarctica and Colobanthus quitensis (Convey and Smith 2006). While the terrestrial Arctic ecosystem has more than 160 higher plant species, allowing more species interactions and feedbacks and perhaps providing a more general representative ecosystem response to enhanced UV-B than the more simple two-species Antarctic ecosystem (Rozema et al. 2006). Short- and longterm studies have shown different and controversial effects of both enhanced and excluded solar UV-B radiation on the Antarctic and Arctic flora species (Searles et al. 2002; Phoenix et al. 2003; Robson et al. 2003). However, in three extensive overviews, Dormann and Woodin (2002) and Rozema et al. (2005, 2006) claimed the finding that neither flowering plants nor mosses and lichen species of the polar ecosystems are markedly affected by the enhanced solar UV-B. Almost all the plant parameters related to the growth and photosynthesis were not significantly affected by elevated UV-B simulating 15, 30, or may be higher (e.g., 50%) of the ozone depletion (Rozema et al. 2005). In fact, these overviews contradict many authors who hypothesized that stressful harsh climatic and environmental polar conditions would make the polar plants vulnerable to the enhanced UV-B, and that the repair of UV-Binduced damage could be hampered by the low polar temperatures (Newsham and Robinson 2009; Snell et al. 2009). The absence of significant UV-B effects on polar plants could imply that they are better adapted to high UV-B regimes and capable of preventing and/or effectively repairing the UV-B damage (Rozema et al. 2005). Although this fact may be interpreted that terrestrial plants from polar ecosystems are particularly tolerant to the ozone depletion, in a more generalized way it has been applied to all the plants and ecosystems, especially those located in high UV-B environments (e.g., tropical and subtropical mountain areas). This assumption implies that terrestrial plants occurring naturally in the high UV-B habitats would undoubtedly have evolved specific adaptations that protect them against the deleterious effects of the UV-B radiation. Hence such plants could show a reduced responsiveness mainly due to their reduced sensitivity to UV-B radiation. Similarly, the plants growing in habitats with low UV-B irradiances (e.g., forest underground) could suffer changes even under small variations in the stratospheric ozone layer (Turunen and Latola 2005).

The solar UV-B radiation cannot be regarded as merely an environmental factor causing plant damages because it can also act as an informational signal leading to morphogenic effects on the structure of plants and the overall function of forest ecosystems (Julkunen-Tiitto et al. 2005). For many years both field and laboratory experiments have focused on the UV-B increased scenario, being scarce those on the responses of plants to the current level of solar UV-B (Searles et al. 2001). This lack of information constitutes an important gap that impedes us to understand the responses of terrestrial plants to solar UV-B changes completely. Many reports consider the solar UV-B as an environmental stressor that affects the development of plants (Láposi et al. 2002; Kadur et al. 2007), while others have communicated no detrimental effects of the solar UV-B on the plant growth (Amudha et al. 2010). Some have even reported protective and/ or beneficial effects of the solar UV-B radiation (Winter and Rostás 2008). Although there is no conclusive explanation for these contradictory effects, they could obey to variations in the UV-B sensitivity among different species and even among cultivars of the same species (Gilbert et al. 2009; González et al. 2009). In this context, the terrestrial plants have developed different strategies to avoid UV-B radiation reaching the most sensitive cellular targets. A major strategy against penetration of the solar UV-B is based on epidermal screening of the incident radiation (Tattini et al. 2005). The mechanisms that inhibit the penetration of UV-B radiation inside the leaf tissues comprise different leaf structural features such as leaf surface reflectance due to the leaf surface wax and hairs (trichomes) (Liakopoulos et al. 2006; González et al. 2007), epidermal thickness (Hilal et al. 2004), epidermal terpenoids (resin) accumulation (Zavala and Ravetta 2002), and epidermal accumulation of UV-absorbing compounds (Burchard et al. 2000; Agati and Tattini 2010). However, despite largely evolved UV-protection mechanisms, complete UV-B protection is not achieved and a small percentage of the solar UV-B radiation penetrates inside the leaf (Krauss et al. 1997). It is generally accepted that a gradient exists in the ability to screening of UV; the herbaceous plants (being least efficient) towards woody and perennial plants, with the conifers being the most efficient (Krauss et al. 1997). Moreover, the proportion of UV-B radiation reaching the leaf photosynthetic mesophyll is significantly higher in the deciduous broadleaf trees than in the evergreen conifer trees (Julkunen-Tiitto et al. 2005; Turunen and Latola 2005). This indicates a greater susceptibility of the deciduous trees to the enhanced UV-B radiation as well as a greater cost of maintenance (Snell et al. 2009). The reason for the low UV-B transmittance in the conifer needles is that UV-absorbing compounds are located in both vacuoles and epidermal cell walls, whereas in herbaceous plants these are located primarily inside the vacuoles of epidermal cells (Julkunen-Tiitto et al. 2005). Moreover, the soluble flavonoids can be actively and rapidly mediated by the exposure to UV-B radiation whereas the cell-wall bound insoluble phenyl-propanoids represent a more passive UV-screening mechanism (Krauss et al. 1997; Clarke and Robinson 2008). These compounds absorb the UV-B wavelengths effectively, but they also transmit the visible PAR inside the mesophyll cells (Krauss et al. 1997). Interestingly, excess of penetrating UV radiation (UV-A and UV-B) could be converted into visible PAR radiation through both yellow and green fluorescence emission from the epidermal cell-wall bound UV-absorbing compounds (Hoque and Remus 1999). Although the epidermal thickness and concentration of UV-absorbing compounds seems to be the strongest predictors of epidermal transmittance and depth of the UV-B penetration, clear relationships between effectiveness of the accumulation of UV-absorbing compounds and epidermal morphological changes have still not been established, suggesting that other intrinsic plant factors are also important in determining the UV-B screening efficiency. Moreover, the endogenous constitution of plants can affect the chemical composition at both whole and organ level. Even within an individual plant the quality and quantity of secondary metabolites may differ between young and old leaves, as well as between the leaves exposed to the sun and those that remain in the shade (Brenes-Arguedas et al. 2006). Alteration in the accumulation of species-specific UV-absorbing compounds may result in changes in the tissue attractiveness or palatability to insects and herbivores (Izaguirre et al. 2007), pathogen attacks (Stratmann 2003), plant-plant interactions (Sullivan 2005), and changes in litter decomposition processes (Pancotto et al. 2003). Because most of these studies have been conducted on crop monocultures or isolated pot grown plants, the extrapolation of their responses to natural ecosystems is difficult (Phoenix et al. 2003). One previous ecosystem study found little effect of the ambient solar UV-B on Sphagnum bog and Carex fen in Tierra del Fuego-Argentina (Searles et al. 2002). Moreover, in related studies the solar UV-B reduced the herbivory, but increased the damage of DNA in the perennial herb Gunnera magellanica and reduced both the leaf number and length of the Antarctic species D. antarctica and C. quitensis (Ballaré et al. 2001). Although overall these studies expand our limited knowledge on how the exposure to natural ambient UV-B can modify the biomass accumulation, population dynamics, and competitive interactions in nonagricultural species and thereby how ecosystems may respond to future UV-B fluctuations. Presently long-time studies are very scarce, only a few studies with more than 4 years under continuous monitoring have been communicated (Robson et al. 2003; Rozema et al. 2006; Trošt-Sedej and Gaberščik 2008). Although the visible radiation can often penetrate dense canopies deeper than the UV-B because of its higher transmittance through leaves, in less dense canopies the situation may be reversed. This implies that the UV-B/PAR ratio should change with the canopy leaf area and leaf architecture (Shulski et al. 2004). In order to understand how the natural ecosystems respond to the ambient solar UV-B radiation, many additional well-designed long-term studies with various plant species are needed in order to understand the different behavior of UV-B and PAR inside the canopy as well as to obtain a complete picture of the gene-environment interactions.

6 Effects of Artificially Enhanced UV-B Radiation

Different to polar studies, earlier researches on nonpolar terrestrial plants were mainly focused on the effects of artificially increased UV-B radiation on crop species rather than ecosystems (Flint et al. 2003). Although, such researches were important to understand the physiological responses and identify possible targets for the UV-B radiation, extensive recent studies have shown that effects of the artificially manipulated UV-B have been often overestimated (Rozema 2000). Moreover, the responses of terrestrial plants to simulated solar UV-B enhancement vary greatly due to artifacts derived from the experimental conditions (Musil et al. 2002a; Flint et al. 2003). A critical point besides the variability of experimental conditions in the evaluation of simulated solar UV-B enhancement is the use of lamps to provide the UV-B radiation. Both UV-fluorescent and broadspectrum xenon-arc lamps are the most commonly used sources of UV radiation in UV-B enhancement experiments (Flint et al. 2009). In terrestrial studies, the UV-fluorescent lamps are usually used, but in aquatic experiments the xenon-arc lamps are preferred. Although the UV-fluorescent lamps are widely used, they supply more short- than long-wave UV-B radiation compared with the solar spectrum (Musil et al. 2002a). In addition, all the UV lamps emit small but biologically effective UV-C radiation, which is not present in the solar radiation reaching the Earth's surface (Flint et al. 2009). Other debatable question in the studies on UV-B effects is the use of UV filters. The major filters used to exclude either UV-A or UV-B in UV-exclusion studies are: (a) cellulose diacetate, CA, that is commonly used to exclude the UV-C radiation and transmit both UV-A and UV-B; (b) polyester, the generic name for Mylar (trade name of the DuPont Co.) that is used to exclude both UV-C and UV-B and transmit the UV-A only; (c) polychlorotrifluoroethylene, PCTFE (Aclar 22 C) that transmits all the UV radiation (UV-A, UV-B, and UV-C); (d) copolymers of tetrafluoroethylene and hexafluoropropylene, Teflon FEP (trade name of the DuPont Co.) that transmits the radiation at 245 nm and above; (e) polyvinyl fluoride, Tedlar TUT (trade name of the DuPont Co.) that blocks wavelengths in the UV-B region; (f) clear polyethylene, Dura-Film Super 4 (trade name of the AT Plastics Inc.) that blocks the UV radiation up to 380 nm; polymethylmethacrylate, Plexiglas (trade name

of the Arkema): the standard Plexiglas excludes the UV-B wavelengths and a portion of the UV-A region, whereas the UV-T Plexiglas transmits all the wavelengths in both UV-B and UV-A regions (Krizek et al. 2005). Although these filters have been widely used in UV-B studies, their transmittance properties vary leading to erroneous interpretations of the UV effects in long-term experiments (Day et al. 2001). Moreover, CA, the most widely used UV filter produces detrimental effects on plants (Krizek and Mirecki 2004). On the other hand, in greenhouses or growth chambers unrealistic balances frequently occur among the different light spectral regions: UV-B/UV-A/PAR (photosynthetic active radiation (PAR), 400-700 nm) and often levels of PAR lower than in the field conditions are also observed. Low levels of PAR increase the sensitivity of plants to UV-B-induced damages (Pradhan et al. 2006). Additionally, to calculate and compare the doses of UV-B under different spectral regimes the UV-B radiation is weighted (UV-BBE) according to a suitable biological action spectrum or biological weighting function (BWF). To obtain the UV-BBE there are different BWFs available but there is a generalized consensus for the use of the Caldwell's BWF (Flint and Caldwell 2003). Several results, however, have shown that this very steep action spectrum may lead to over- or underestimation of the UV-B effects (Micheletti et al. 2003). Moreover, differences in climatic conditions can also affect the interpretations and comparisons among different studies based on the BWF (Musil et al. 2002b; Flint et al. 2009). Therefore, all the quantitative predictions relating to UV-B enhancement effects could be greatly affected. On the other hand, more recent studies have proposed that increases of the ambient solar UV-B radiation at magnitudes anticipated under the current stratospheric ozone projections will not significantly have large-scale deleterious effects on terrestrial plants even though some species may suffer photosynthesis decreases and growth reductions (Rozema et al. 2006; Xu and Qiu 2007; Newsham and Robinson 2009).

7 Physiological and Morphological Responses to UV-B Enhancement

From indoor and outdoor studies, there is a general consensus that UV-B enhancement produces physiological, biochemical, morphological, and anatomical changes in the plants (Searles et al. 2001). According to the literature, the enhancement of UV-B radiation can affect the terrestrial plants at different functional levels involving conformational changes and damages to different molecules such as DNA, proteins, and lipids (Li et al. 2010). As a result, if damage to macromolecules, that is, DNA is not effectively repaired, the UV-B effect will be translated to the biochemical level with the consequent alteration and/or impairment of the plant functionality (e.g., photosynthetic process, growth, yield). Although it is clear that there is a wide range of both intra- and interspecific sensitivity to UV-B radiation (Gilbert et al. 2009) the terrestrial plants through the evolution have acquired different protective strategies to avoid the adverse effects of UV-B radiation. The two major protective mechanisms are: (a) shielding through the production of soluble phenolics (e.g., flavonoids, anthocyanins, hydroxycinnamic acid derivatives), insoluble polyphenols (e.g., lignin), and cell-wall bound UV-absorbing compounds (Hilal et al. 2004; Clarke and Robinson 2008), as well as by reflection of the UV-B radiation by epicuticular waxes and cuticular structures (Hada et al. 2003; Schmitz-Hoerner and Weissenböck 2003; Agati and Tattini 2010); (b) removal and direct reversion of the DNA lesions induced by UV-B radiation (Tuteja et al. 2001; Britt 2004; Kimura et al. 2004).

7.1 DNA: Damage and Repair

The more important UV-B-induced DNA alterations are the formation of cyclobutane pyrimidine dimers (CPDs) and pyrimidine (6–4) pyrimidine dimers (6–4 photoproducts, 6-4PPs) (Dany et al. 2001). The DNA repair mechanisms operating in plants include the following processes: (a) direct reversal (DR); (b) photoreactivation that induces photolyases; (c) dark repair (Tuteja et al. 2001; Britt 2004). The DR is a simple mechanism that involves a single-enzyme reaction for the removal of certain types of DNA damage. Alkyltransferases simply extract alkyl groups from the alkylated bases that are transferred to internal cysteine residues and thus inactivate themselves. The best example for DR is the correction of miscoding alkylation lesion O^6 -methylguanine, which is generated endogenously in small amounts by the reactive cellular catabolites. This reaction is catalyzed by a specific enzyme, called methylguanine methyltransferase (MGMT), which removes a methyl group from a guanine residue of the DNA molecule and transferring it to one of its own cysteine residues in a rapid and error-free repair process (Tuteja et al. 2001). The photoreactivating enzyme DNA photolyase (PRE) is a DR phenomenon performed by the combined action of one or more photolyases and the visible light (blue, violet, or long-wave UV) (Hidema et al. 2007). Photolyases specifically recognize and bind the pyrimidine dimers to form a complex molecular structure which is stable in absence of the light. After absorbing a blue light photon the pyrimidine dimers are reversed to pyrimidine monomers without excision of the damaged base (Tuteja et al. 2001). The repair reaction is fast and requires about 1 h for completion (Takeuchi et al. 2007). In plants, two specific types of photolyases have been characterized: (a) CPDphotolyase; (b) 6-4PP-photolyase (Tuteja et al. 2001). The dark repair processes include the nucleotide excision repair (NER), base excision repair (BER), mismatch repair (MMR), and other DNA repair pathways. These mechanisms have been observed in several plant species and some of the genes required for the processes were identified (Kimura et al. 2004). The wide class of helix-distorting lesions such as CPDs and 6-4PPs are repaired by the NER process. It is one of the most versatile DNA repair pathway operating in plants. Unlike other DNA repair pathways that are specific repair processes, the NER pathway is capable of removing various DNA damage classes, including those induced by the UV-B radiation (pyrimidine dimers) and chemical agents (bulky DNA adducts) (Kimura et al. 2004). The NER pathway sequentially involves recognition of the DNA damage, incision on the damaged strand, excision of the damage-containing oligonucleotides, and the DNA synthesis and ligation (Liu et al. 2003). Also, the NER pathway is a slow process (about 24 h for completion) and includes several enzymes. There are two subpathways of the NER process that are designated as: (a) global genomic repair (GGR); (b) transcription-coupled repair (TCR). While the GGR pathway repairs the DNA damage over the entire genome, the TCR pathway is selective for the transcribed DNA strand in expressed genes (Kimura et al. 2004). Oxidized or hydrated bases and single-strand breaks are repaired by the BER pathway that is considered an essential process for maintenance of the DNA molecule. The BER mainly removes the DNA damages that are arising spontaneously in the cell from hydrolytic events such as deamination or base loss, fragmented bases resulting from ionizing radiation (e.g., UV-B radiation), and oxidative damage or methylation of the ring nitrogen by endogenous agents. The process that involves the BER mechanism is initiated by DNA glycosylases that release the damaged base by cleavage of the sugarphosphate chain followed by excision of the abasic residue or abasic residue containing oligonucleotides and then the synthesis and ligation of the DNA occurs. The BER pathway involves several enzymatic steps and depends strongly on the presence of nicotinamide adenine dinucleotide (NAD+). Also, the BER process comprises two subpathways that are designated as: (a) BERshort-path; (b) BERlong-path. The BERshort-patch is a DNA polymerase beta-dependent mechanism, while the BERlong-patch is a DNA polymerase delta/ epsilon-dependent mechanism (Kimura et al. 2004). The major difference between BER and NER pathways is the way by which the DNA damage is removed. The NER pathway cuts out the damage as a part of an oligonucleotide fragment, while the BER mechanism excises only one nucleotide (Tuteja et al. 2001). The excision repair processes (NER and BER) are very important for maintaining the genome stability

and essential for the survival of plants. The mismatch repair pathway (MMR) is also important in the DNA repair processes when by errors of replication or homologous recombination can be produced mismatched bases. The MMR pathway basically discriminates between correct and incorrect bases and after DNA synthesis the error is corrected (Tuteja et al. 2001). Although most of the present understanding of the eukaryotic MMR has come from studies of the E. coli MutS and MutL proteins (Kolodner and Marsischky 1999), recent studies carried out in Arabidopsis and rice have reported interesting findings on the MMR pathway operating in plant cells (Tuteja et al. 2001; Kimura et al. 2004). According to the E. coli model, the MutS dimer recognizes mispairs and then binds on it followed by the MutL binding, which activates the MutH (endonuclease) that makes a single-strand incision (nick). The MutH incision can be done on either side of the mismatch. Subsequent to incision the excision is initiated and proceeds toward mismatch. To fill the gap (100–1,000 nucleotide gap), the original template strand can then be replicated and finally sealed by ligation. The proteins involved in the last step of eukaryotic MMR are: (a) DNA polymerase δ , RP-A (replication protein); (b) PCNA (proliferating cell nuclear antigen); (c) RFC (replication factor) (Kolodner and Marsischky 1999).

Although the UV-A wavelengths can mediate the photoxidative damage (Turcsányi and Vass 2000) the UV-B radiation is the most important photooxidant agent for terrestrial plants. The DNA damage can also be caused by reactive oxygen species (ROS) and free radicals produced by the UV-B radiation. This damage includes several modifications such as cross-linking, aggregation, denaturation, and degradation (Hidema et al. 2007). The formation of 7,8-dihydro-8oxoguanine (GO) is a common oxidative DNA lesion generated by a direct modification mediated by ROS. The GO is mutagenic and can mispair with adenine (A) during the DNA replication (Yang et al. 2001). If the resulting A/GO is not repaired before the next round of the DNA replication, a C/G \rightarrow A/T transversion occurs and the opportunity for repair is lost. The A/GO is repaired via the BER which is initiated by the DNA repair enzyme adenine-DNA glycosylase (Yang et al. 2001). The UV-absorbing compounds (e.g., flavonoids, anthocyanins, hydroxycinnamic acid derivatives, phenolics) accumulating in epidermal and subepidermal cell layers have traditionally been thought to function as UV-B filters, but also play an important role as quenchers of the ROS and free radicals in the amelioration of the UV-B-induced DNA oxidative damage (Agati and Tattini 2010). The UV-absorbing compounds are also effective in reducing the induction of cyclobutane pyrimidine dimers (CPDs) in plants exposed to high UV-B levels (Hidema et al. 2007). The inhibition of CDP formation seems to be high enough to compensate the DNA damage arising even from unusually strong solar irradiations (Tuteja et al. 2001). Other related UV-absorbing compounds, that is, anthocyanins through an anthocyanin-DNA complex could also provide protection against the oxidative damage. Since both anthocyanins and DNA mutually protect each other in vitro, it is likely that such protection mechanism may also operate in vivo (Sarma and Sharma 1999). In the plant cells, anthocyanins are predominantly localized inside the vacuoles and thus their putative role in the protection of DNA should be critically examined. Accepting this fact, it has been demonstrated that the excess accumulation of anthocyanins reduces the amount of blue/UV-A radiation reaching the cell and may sometimes lower the ability to photorepair the damaged DNA. For example, the purple rice is a highly UV-B sensitive species despite possessing an elevated level of anthocyanins in their leaves (Hada et al. 2003). Although significant amounts of flavonoids have been found in the chloroplasts or etioplasts isolated from a wide range of plants growing under both ambient and enhanced UV-B irradiances (Tattini et al. 2005; Agati et al. 2007), it is also likely that some amount of anthocyanins can be present in the nuclei and organelles and then may associate with the DNA molecule, offering to it a certain protection against the oxidative damage (Feucht et al. 2004). In this context, the UV-absorbing compounds seem to have an important protective function against the DNA

damage induced by shorter solar wavelengths (Schmitz-Hoerner and Weissenböck 2003) and so the speculations concerning a great biological risk with regard to increases in solar UV-B radiation after the depletion of the ozone layer are presumably premature.

In the natural populations, both protection and DNA repair are complementary and necessary processes for the plant development. Thereby, it is expected that the plants growing under different UV-B irradiances can exhibit different levels of the DNA protective mechanisms (Turunen and Latola 2005). Under field conditions, the observed DNA damage can often be modified by climatic conditions and then a direct extrapolation of the DNA changes obtained in controlled-environment experiments under artificially enhanced UV-B radiation to plants growing under the ambient solar UV-B is complex and unrealistic. Differences between damage, repair, and defense can be subtle and identification of a particular mechanism does not always occur as the explanation underlying a given phenomenon. For example, the UV-induced degradation of the D1 protein of the PSII can be seen either as damage or as a part of the repair mechanism leading to substitution of the damaged components of the PSII (Turcsányi and Vass 2000). Consequently, understanding these differences and potentially using the DNA repair mechanisms could become very important for producing UV-B-tolerant plants.

7.2 Secondary Metabolites: Flavonoids and Anthocyanins

The increase of secondary metabolites synthesis has been recognized as one of the most frequently observed plant response to UV-B enhancement (Searles et al. 2001; Rozema et al. 2002; Bassman 2004). Considerable attention has been focused, over the past two decades, on the UV-B-induced biosynthesis of phenylpropanoid-derivative compounds, particularly flavonoids and hydroxycinnamic acid derivatives (Jordan 2002; Rozema et al. 2002; Bassman 2004). Although these compounds exhibit important interspecific differences induced by the UV-B radiation, they are often derivatives of the flavonols quercetin and kaempferol (Buer et al. 2010). Quercetin- and kaempferol-derivative flavonoids are usually glycosylated and frequently contain a hydroxycinnamic acid moiety esterified to one of the (orthodihydroxy glycosyl groups B-ringsubstituted flavonoids) (Tattini et al. 2004). The flavonoids are ubiquitous molecules occurring in the vacuoles and cell walls of epidermal cells and in nonsecretory and glandular trichomes, and it has been assumed that they primarily have the function of attenuating the shorter solar wavelengths due to their good quantum efficiency (Burchard et al. 2000). In this way, the location of flavonoids in trichomes (Tattini et al. 2004), cuticular wax layers (Fukuda et al. 2008), and epidermal cells (Burchard et al. 2000) may largely prevent that the UV-B radiation reaches sensitive targets within the leaf. However, the flavonoids also have another protective function against the shorter solar wavelengths. Considering that the flavonoids with orthodihydroxylated B-ring may efficiently dissipate the excess of energy through tautomeric interconversions (Smith and Markham 1998), scavenge the ROS through the quenching mechanism (Yamasaki et al. 1997; Hilal et al. 2008), and inhibit the formation of free radicals (Neill and Gould 2003; Xu et al. 2008), they can also act as effective antioxidant molecules (Jordan 2002; Tattini et al. 2005; Buer et al. 2010). However, a major criticism regarding functions of the flavonoids is the use of mutants that lack or possess the ability to synthesize flavonoids, which may oversimplify the plant model system for UV-B-tolerance/flavonoidquantifying the biosynthesis relationships (Bieza and Lois 2001). In this context, contrary to determination of the flavonoid concentration at the whole-leaf level, less attention has been devoted to analyzing the tissue-specific location of individual flavonoids, which may clarify their complex functional roles in both attenuation and antioxidant mechanisms against the high UV-B irradiances (Tattini et al. 2004). Furthermore, the short-term experiments and inappropriate microscopy techniques for visualizing the flavonoids also greatly contributed to this superficial conclusion. More recent studies, however, suggest that the biosynthesis of

flavonoids, particularly internal flavonoid glycosides may be largely controlled by constitutive morphoanatomical and biochemical features, primarily intended both to prevent the light penetration (Burchard et al. 2000) and to remove the consequent oxidative damage (Apel and Hirt 2004). Agreeing with these findings, Semerdjieva et al. (2003) showed in the *Vaccinium* spp. an inverse relationship between cuticule thickness (primary barrier to UV-B penetration) and the mesophyll accumulation of UV-B-induced flavonoids.

Regarding to ROS scavenging activity of the flavonoids, Yamasaki et al. (1997) proposed a model to address major criticisms on the antioxidant functions of the flavonoids compartmentalized in epidermal vacuoles, and at the same time to explain the preferential UV-B-induced synthesis of flavonoids with effective antioxidant properties in vitro. According to Yamasaki's model, the orthodihydroxy B-ring-substituted flavonoids, not their monohydroxy B-ring-substituted counterparts, are effective substrates for the class III peroxidases, which quench the H₂O₂ freely diffusing from the mesophyll cellular organelles to vacuoles of the epidermal cells. The model was remarkable in calling out the question whether vacuolar flavonoids could be effective in protecting underlying tissues from the damaging shorter solar wavelengths, while not protecting the epidermal cells from the oxidative damage. Epidermal cells and glandular trichomes usually contain much higher concentrations of flavonoids than the mesophyll cells (Burchard et al. 2000), then the H_2O_2 leaked out from the mesophyll cells under high UV-B irradiances can be scavenged by the flavonoid-peroxidase system in the epidermal cells according to the proposal of Yamasaki et al. (1997). Consistent with this idea blackening of the epidermis after a severe light stress is frequently observed in many species under the field conditions. This phenomenon has been ascribed to the polymerization of vacuolar phenolics that result from the penetration of H_2O_2 inside the epidermal cells (Yamasaki et al. 1997). Nevertheless, it cannot be excluded the possibility that other apoplastic flavonoid-depending peroxidases such as guaiacol peroxidase and

syringaldazine peroxidase associated with the process of lignification may be involved in the mechanism of ROS scavenging (Hilal et al. 2004). The concept of delocalized scavenging of H_2O_2 by the vacuoles can be applied not only to the organelle-organelle interactions but also to the cell-cell interactions (Yamasaki et al. 1997). It is also acknowledged as a controversy matter whether the ability to accumulate flavonoids, particularly flavonoids with orthodihydroxylated B-rings, and the tolerance to UV-B radiation is highly correlated (Dixon et al. 2001; Musil et al. 2002b; Hofmann et al. 2003). The orthodihydroxylated B-ring-derivative flavonoids such as quercetin and luteolin glycosides are accumulated in the vacuoles of the mesophyll cells in Ligustrum vulgare leaves exposed to the full sunlight, in presence or absence of the UV-B radiation (Agati and Tattini 2010). This finding, which is consistent with previous reports indicating that the UV-B radiation is not a prerequisite for the synthesis of flavonoids (Tattini et al. 2004, 2005; Jenkins 2009), leads to the conclusion that the light-induced oxidative damage may regulate the biosynthesis of flavonoids, irrespective of the presence of UV-B radiation. The flavonoids and other UV-absorbing phenolics (e.g., hydroxycinnamic acid derivatives) are also synthesized in other abiotic/biotic unfavorable conditions such as drought, salinity, low temperature, heavy metal pollution, pathogen attack, and as feeding deterrent. Besides their antioxidant abilities the flavonoids might exert modulatory effects in the cell through selective actions at different components by cell interactions (Buer et al. 2010). This fact has become increasingly important because attention focuses on the new concept of flavonoids as potential modulators of the intracellular signaling cascades that are vital for the cell functionality (Jenkins 2009).

The anthocyanins, other members of the phenol family, have generally been included into photodamaging-protective compounds (Gould 2004). The anthocyanins show a weak absorption in the shorter UV region (270–290 nm), but their acylated counterparts (hydroxycinnamic acid derivatives) exhibit an increased absorption in the longer UV-B region (310–320 nm) (Neill and Gould 2003). Because anthocyanins are photoinduced many researchers surmise that they must either have a photoprotective function against the light-induced photooxidation or against the UV-B damage (Hughes et al. 2005). Even without acylation the anthocyanins can significantly attenuate the visible radiation. In fact, the more clear evidences really support the theory of the photooxidative protection, while the role in UV-B protection seems to be much less apparent (Kytridis and Manetas 2006). This assumption, however, contradicts the theory that anthocyanins have a UV-B-filtering role (Neill and Gould 2003). Disagreeing with the last theory the UV-B vulnerability is poorly correlated with the content of anthocyanins. For example, an Arabidopsis mutant with enhanced sensitivity to UV-B radiation was found deficient in certain flavonoids, whereas the amount of anthocyanins displayed unchanged. Similarly the responses of a Brassica rapa mutant to the supplementary UV-B treatment were mostly independent of the anthocyanin level in leaves (Gould 2004). Agreeing with these findings the anthocyanins often occur in very low concentrations compared to other UV-absorbing compounds, and require a long exposure to the UV-B radiation to be synthesized (Neill and Gould 2003). On the other hand, the red-leafed plants of Impatiens capensis and rice displayed significantly worse performances under the UV-B enhancement than their green-leafed counterparts (Dixon et al. 2001). Moreover, it has been communicated that the accumulation of anthocyanins can cause deleterious effects on terrestrial plants after a long-term UV-B exposure (Gould 2004). It has been noted that the DNA damage after a prolonged UV-B treatment was substantially greater in the purple-leafed rice than in the nearisogenic green line. Anthocyanins in the purple rice prevented the photoactivation of photolyases by absorbing some of the incident blue/UV-A light on leaves. Thus, any short-term gain from the absorption of UV-B radiation by anthocyanins would be offset by their property to absorb the visible light and thereby limit the rate of DNA repair (Hada et al. 2003). Furthermore, it has been demonstrated that other abiotic and biotic stresses produce changes in the chemical pattern

of anthocyanins (Close and Beadle 2003), then it is obvious that such changes will influence the absorption spectra of anthocyanins under the UV-B enhancement. Also absorptive artifacts due to the dissociation of covalent bonds can occur during the improper isolation of anthocyanins and misread absorption spectra will be generated (Gould 2004).

The accumulation of anthocyanins is usually transient and generally occurs in the vacuoles of peripheral tissues such as palisade and/or spongy mesophyll exposed to high light irradiances, but there are some exceptions (e.g., accumulation in the abaxial leaf tissues and in obligatory shade plants) (Kytridis and Manetas 2006). Perhaps the improved solubility of anthocyanins that in contrast to other flavonoids are nearly always glycosylated allows them to be stored in the vacuole more efficiently than the nonglycosylated flavonoids (Winefield 2002). In fact, the importance of flavonoids should not be overlooked in the discussion of anthocyanin production and UV-B protection. In this context, the flavonoids induced by the UV-B radiation (Agati and Tattini 2010; Buer et al. 2010) are recognized as strong UV-B absorbers, and their UV-B absorption capacity is much stronger than that of anthocyanins (Bieza and Lois 2001). Since the production of anthocyanins represents a conversion of flavonoid precursors that themselves are strong UV-B absorbers, a conundrum appears: if one of the effects of UV-B radiation on plants is to induce the UV-B-protective pigments, why are anthocyanins produced instead of their flavonoid precursors? Characteristics of both flavonoids and anthocyanins absorption spectra must be analyzed to respond this conundrum (Solovchenko and Merzlyak 2008). The flavonoids exhibit two bands in the UV region: (a) short-wave peaking around 280 nm; (b) long-wave situated in the range of 300-360 nm. However, the exact positions of the maxima vary for different flavonoid derivatives. The anthocyanins also have two maxima: one in the UV-B region (270-320 nm) and another in the visible region with a maximum located in the blue-green part of the visible wavelengths (500–540 nm) (Gould 2004). In this way, the UV-B component of the solar spectrum can

be screened by both flavonoids and anthocyanins. However the UV-A radiation, whose proportion in the solar spectrum could be tenfold higher as compared with the UV-B spectrum region also exerts significant effects on plants (Krizek 2004). For example, maximum inhibition of the photosynthesis under natural radiation fluxes is induced by radiation in the UV-A region (Ivanova et al. 2008). Although this fact supports the importance of the UV-protection provided by the flavonoids in the range of 300-360 nm, the high visible fluxes (400-700 nm) also induce a photodamage in plant tissues, especially in the chloroplast (Krizek 2004). The anthocyanins are able to intercept a great proportion of the solar radiation in the range of 500-600 nm, which correspond to the maximum solar energy reaching the Earth's surface (Gould 2004). This finding therefore contributes to support the role that anthocyanins play photoprotection of plant in the tissues (Solovchenko and Merzlyak 2008). In this context, the accumulation of anthocyanins requires visible light and generally coincides with the period of high excitation pressure and the increased potential for the photooxidative damage. The photooxidative damage is produced by an imbalance between the light capture, CO₂ assimilation, and carbohydrate utilization (e.g., greening of developing tissues, senescence, adverse environmental conditions) (Hughes et al. 2005). Thereby, the attenuation of light by anthocyanins may help to reestablish this balance and to reduce the excitation pressure (Kytridis and Manetas 2006). Then, the risk of cellular photooxidative damage is lowered. Also, it would seem that the anthocyanin biosynthesis can enhance under the high light, but it is not usually a prerequisite for the protection against the oxidative stress (Gould 2004).

Like the colorless flavonoids the colored anthocyanins may scavenge the free radicals and ROS (Gould 2004). The anthocyanins diminish the oxidative trend in the leaf simply by filtering out the yellow-green light, because most of the reactive oxygen in plant cells is derived from excitation of the chlorophyll molecule (Neill and Gould 2003). Agreeing with this theory, in juvenile and senescing plants the regulation of photosynthetic apparatus functions is often impaired, making it less efficient in utilization of the absorbed light and therefore prone to the photodamage (Merzlyak et al. 2008a). As a general rule, these situations are accompanied by an increased generation of ROS causing photooxidative damage to the plant and, eventually, its death (Bukhov 2004). Under these conditions, the anthocyanins may afford a detoxifying sink for some ROS when the chloroplast, the first line of the antioxidative defense, is surpassed (Kytridis and Manetas 2006). It is not clear, however, whether the ROS scavenging occurs predominantly through the anthocyanins found inside the vacuole or through their counterparts located in the cytosol. Both anthocyanin forms have impressive antioxidant potentials (Neill and Gould 2003), but due to their proximity to the chloroplastic source of ROS it is more probable that anthocyanins located in the cytosol (mesophyll tissue) than in the vacuole (epidermal tissue) provide the major contribution to antioxidant defense (Kytridis and Manetas 2006). In a similar trend, recent evidences suggest that flavonoids may scavenge the ROS within or near sites of its generation (Schmitz-Hoerner and Weissenböck 2003; Tattini et al. 2005; Agati et al. 2007). Interestingly, equal effectiveness as antioxidant molecules of other colorless phenolics suggests that the putative photooxidative protection afforded by the anthocyanins should be unrelated to their ability to quench oxidants.

Noteworthy the accumulation of anthocyanins in terrestrial plants has always been a contentious issue of the special interest. They often appear in juvenile plants but mature plants usually lack them or display transiently levels under stressful conditions (Merzlyak et al. 2008a). Obviously, upon maturation of the photosynthetic apparatus or its acclimation to stressors the photoprotective screen of anthocyanins is no longer required and the juvenile reddish pigmentation disappears. However, unfavorable environmental conditions such as low temperatures, heavy metals, drought, wounding, and pollutants can also predispose the photosynthetic apparatus to photoinhibition and photooxidation, and then the plants may increase, although not necessarily ascribed to, the accumulation of anthocyanins in vegetative organs (Gould 2004). Accordingly, the production of anthocyanins would fit neatly into the definition of Leshem and Kuiper's (1996) general adaptation syndrome (GAS). The GAS indicates that different types of stress evoke similar adaptation responses. In this context, along with compounds such as tocopherols, flavonoids, glutathione, and ascorbate, the anthocyanins may function as general mitigators of the oxidative damage. However, it should be addressed that there is no direct evidence that terrestrial plants benefit from the antioxidant properties of anthocyanins yet (Neill and Gould 2003). Although anthocyanins are of special importance for the photoprotection in senescing leaves, it seems not to be the only function of anthocyanins. It has been suggested, for example, that the red color may also deter aphids from laying their eggs or from feeding on the sugar-rich sap in the phloem. Noteworthy, despite that the autumnal color may be an extravagancy without a vital function. This phenomenon that enchants so many tourists each year may hold a vital key to the survival of deciduous trees (Archetti et al. 2009). Also the anthocyanins are involved in the photoprotection of ripening fruits, for example, the chlorophyll in apple fruit peel with high anthocyanin content showed a very high resistance to the photobleaching as compared with the anthocyanin-free zones of the same fruit (Merzlyak et al. 2008b). Despite its function as photoprotective molecules, the anthocyanins may instead serve to decrease the leaf osmotic potential. The resulting depression of leaf water potential could increase the water uptake and/or reduce transpirational losses. This phenomenon may allow to the anthocyanincontaining leaves to tolerate suboptimal water levels. The often transitory nature of the foliar anthocyanin accumulation may allow plants to respond quickly and temporarily to environmental variability rather than through more permanent anatomical or morphological modifications (Chalker-Scott 2002). Interestingly, the anthocyanins also fulfill the less common but important function of avoiding the photodegradation of sensitive molecules. The Ambrosia chamissonis, for example, hold strands of laticifers surrounded by an anthocyanin sheath. These laticifers contain

thiarubrines, toxic chemicals that are believed to deter herbivory and prevent both fungal and bacterial infections. The thiarubrines are photolabile molecules and are degraded by both visible and UV light giving thiophenes that are less toxic. Page and Towers (2002) have shown that the anthocyanin sheath, by absorbing a proportion of the rays that would otherwise strike the laticifers, protects these light-sensitive defensive chemicals from degradation, and thus provides a mechanism for the antiherbivory under conditions of strong sunlight.

However, although the role of anthocyanins in protecting plant tissues under stress conditions, including the photodamage mediated by both UV-B and visible light, as well as in the pollinator attractiveness and seed dispersion seems to be important, it is clearly evident that the adaptive significance of anthocyanins is still not fully understood (Close and Beadle 2003). Meanwhile two poorly explored areas became interesting: (a) how the increase of anthocyanin production is integrated to tissue responses to UV-B; (b) how the UV-B-induced anthocyanins contribute to the plant survival.

7.3 Morphogenic Responses

Studies carried out in greenhouses or in growth chambers using ultraviolet lamps and filters to simulate different solar UV-B enhancements have been conducted on a variety of terrestrial plants, including economically important crops (Santos et al. 2004) and wild plant species (Zu et al. 2010). Overall these studies showed that the UV-B enhancement besides physiological effects induces a range of morphological changes including: (a) increase/decrease of the leaf area and leaf thickness (González et al. 2002; Hilal et al. 2004); (b) reduction of the plant height (Santos et al. 2004) and increase/decrease of the shoot/root ratio (Furness and Upadhyaya 2002); (c) axillary branching (Kakani et al. 2003); (d) increase of the leaf glandular and uniseriate trichome density (Liakopoulos et al. 2006); (e) deposition of the waxy surface structures (Fukuda et al. 2008); (f) opening of the cotyledon curling (Boccalandro

et al. 2001; Barnes et al. 2005); (g) inhibition of the hypocotyl and stem elongation (Shinkle et al. 2004; Gerhardt et al. 2005); (h) premature leaf senescence (Pradhan et al. 2006). The effects of UV-B also include changes (increase/decrease) in the number and size of flowers as well as in the size of seeds (Kakani et al. 2003; Qaderi and Reid 2005). While some of the UV-B responses constitute a stimulation of the growth (e.g., axillary branching, leaf thickening), others reflect a growth inhibition (e.g., reduced hypocotyl elongation). However, in these experimental setups, frequently unrealistic balances between UV-B/ UV-A/PAR are obtained, and in some cases the plants have been exposed to relatively high shortterm doses of UV-B, which lack the ecological relevance (Newsham and Robinson 2009). Additionally, the levels of UV-A or PAR as well as other experimental conditions also affect the morphogenic responses, making it difficult to compare the results from different indoor studies. In addition, it is clear that not all the plant species respond in the same way to UV-B exposure (Pliura et al. 2008). In general, the monocots are more morphologically responsive to UV-B than the dicots (Pal et al. 1997). Closely related species or ecotypes, especially when occupy different habitats, also differ with respect to their morphogenic responses (Hofmann et al. 2003). Plant species also differ in the use of PAR and UV-B radiation; while some species use the PAR to trigger responses others use the UV-B radiation. Then the plants responding mainly to PAR radiation will probably be more sensitive to UV-B radiation than the UV-responding ones (Rozema et al. 2005). A critical factor in the UV-B studies is the visible light irradiance, which in growth chambers and greenhouses can be quite different to the natural sunlight (Flint et al. 2009). Indeed, it has been shown that as a result of the insufficient visible wavelengths and, therefore, of unrealistically high UV-B/PAR ratios in indoor studies, the morphogenic effects of the UV-B radiation are magnified (Musil et al. 2002a, b). In fact, even if realistic levels of the UV-B radiation in simulating ozone reductions are used the indoor responses of plants to UV-B radiation may be quite variable and exaggerated in relation to

the field. Microclimatic conditions and the interactions of different abiotic and biotic environmental factors additionally contribute inconsistency between the results obtained in growth chambers or greenhouses with those obtained under the field conditions (Flint et al. 2003; Caldwell et al. 2007). Furthermore, the plant responses to above ambient UV-B radiation (e.g., from stratospheric ozone depletion) have rarely been assessed in the broader context of the possible effects emerging from variations in the UV-B radiation within the ambient range. Also there is a significant knowledge gap between field and laboratory studies, which has two major components: (a) the occurrence of certain effects of the UV-B radiation under laboratory conditions has not yet been demonstrated in the field studies; (b) although some indoor responses are known to occur in the field, their functional implications are still unclear. Therefore, the obvious corollary from greenhouses or growth chambers studies is: study methodologies are as varied as results. In fact, from the field grown plants, the consensus that effects of artificially changed spectral UV-B irradiances are less pronounced (Searles et al. 2001). While under UV-B enhancement among other changes, leaf thickness, reduced leaf area, decreased plant height, changes in plant architecture, and biomass/yield reduction have been observed (Searles et al. 2001; Flint et al. 2003; Barnes et al. 2005). Nevertheless, the more recent studies have suggested that in the field, primary effects of the most realistic solar UV-B enhancements are subtle morphological and chemical changes with altered carbon partitioning and allocation, but doubt reveals such changes show significant effects on both plant growth and biomass accumulation (Gilbert et al. 2009; González et al. 2009; Morales et al. 2010; Ren et al. 2010; Zu et al. 2010). The morphogenic effects of the realistic UV-B enhancements are not usually considered as primary ecological factors influencing both species abundance and species distribution in relation to other abiotic environmental factors (e.g., drought, temperature, salinity). There are, however, situations where the UV-B-induced morphogenic effects can be ecologically important, giving changes in the competitive ability with a significant impact on the composition of the plant community (Flint et al. 2003). The UV-B enhancement alters the leaf angle and differential transmission, and absorbance of the UV-B radiation through stands of erectophilous or planophilous plant species may have an important consequence on terrestrial plant responses to the UV-B radiation (Rozema 2000). In a model study, it was predicted that a more planophilous leaf angle in erectophilous species would reduce the UV-B/PAR ratio and therefore the UV-B damage. Of course, this effect may affect the competitive relations among species and also the ecosystem composition (Deckmyn 1996 in Rozema et al. 1997). The morphogenic effects often can be pronounced on different organisms at other trophic levels (Bassman 2004). The UV-B radiation also affects the decomposition of plant materials into ecosystems. Plants grown under the enhanced solar UV-B showed a reduced rate of the litter decomposition when compared to control plants grown under the ambient solar UV-B. The accumulation of UV-B-induced lignin and/or tannin accounts for the reduced litter decomposition rate (Cybulski et al. 2000). Nevertheless, the reduced rate of the litter decomposition can be produced as consequence of detrimental effects of the enhanced UV-B radiation on decomposing fungi and other decomposer organisms (Pancotto et al. 2003). In opposite trend, the plant litter material exposed to the enhanced solar UV-B can be decomposed by photodegradation more rapidly than under the ambient solar UV-B (Gallo et al. 2006). Moreover, it has also been demonstrated that the species growing for several generations under enhanced UV-B radiation show accumulation and exacerbation of the UV-B effects and likelihood they might be heritable (Mpoloka et al. 2007).

Much of the UV-B research on terrestrial plants has concentrated on vegetative plant parts, but fitness of the organisms depend mainly on their successful reproduction. Of particular concern is the detrimental effect of UV-B on the pollen quality observed for some species (Koti et al. 2004; He et al. 2007). This finding suggests that pollination may be an ecologically critical developmental stage vulnerable to the UV-B damage, even in the UV-B-tolerant species. The pollen surface of some species may transmit up to 20% of the incident UV-B radiation (Stadler and Uber 1942 in He et al. 2007), despite the presence of a variety of UV-B-absorbing pigments (Rozema et al. 2001). Thus, the mature pollen grains are potentially susceptible to the UV-B damage during a short period between the dehiscence of anthers and the penetration of the pollen tube into the stigmatic tissue (Koti et al. 2004). This fact may lead to both reduced pollen quality and altered patterns of competition among species affecting the composition of the ecosystem. Furthermore, the UV-B enhancement can alter the production and/or the temporal availability of flowers so as to make the plant a less attractive host for the pollinators and impinge upon competition of plants for the pollinator service, as well as on the reproductive success of the plant/pollinator system (Sampson and Cane 1999).

8 UV-B Radiation and Signaling Pathways

The UV-B radiation triggers diverse responses involving a differential regulation of the genes and participate in several protective pathways including the DNA repair, detoxification of ROS, and production of secondary metabolites as well as in photomorphogenic events (Agrawal et al. 2009). The UV-B responses can be elicited with either high fluence (HF-UV-B, over 15 kJ m⁻²), intermediate fluence (IF-UV-B, 5-7 kJ m⁻²), low fluence (LF-UV-B, 1-3 kJ m⁻²), or very low fluence (VLF-UV-B, less than 1 kJ m⁻²) (Brosché et al. 2002). However, the exposure of plants to low fluence UV-B promotes the expression of varying genes involved in the UV-B protection, and genes responsible for the production of flavonoids and several phenolic compounds, while as the low fluence photomorphogenic responses seem to be initiated by photoreceptors and no alternative UV-B-absorbing molecules seem to mediate the photomorphogenic UV-B responses (Ulm and Nagy 2005). Also, many components of the protective pathways which lead to the changes of the gene expression in response to both UV-B radiation and pathogens are similar or identical (upregulation of PDF1.2), except the pathways which are distinct (upregulation of PR-1), signifies the effects of the high fluence UV-B radiation on the gene expression are unlikely to be due to nonspecific damage and a yet unidentified UV-B photoreceptor (Brosché et al. 2002). The response of plants to UV-B radiation depends upon the adaptation and acclimation to UV-B irradiances, as well as of the interactions with other environmental factors. Moreover, studies carried out with Arabidopsis plant suggest that some genes are differentially responsive to UV-B in both 280-290 nm and 300-310 nm ranges, hence could be multiple UV-B photoreception mechanisms (Kalbina et al. 2008). Consequently, the exposure of plants to UV-B radiation can cause multiple responses on the primary and secondary metabolisms as well as different changes on the growth and overall performances. Although other light-dependent photoreceptors (e.g., phytochrome, cryptochrome) have been described (Carvalho et al. 2010), presently a UV-B-specific photoreceptor has still not been described and therefore the basic mechanism of UV-B perception and the signal transduction remain still poorly understood. Nevertheless, the chromophores that could act as photoreceptors to absorb the UV-B radiation exist. Pterins or flavins in their reduced forms are candidates where some experimental work supports the involvement of the perception of UV-B radiation (Galland and Senger 1988a, b in Jenkins 2009). For example, the compounds that antagonize the flavins and pterins impair the UV-B induced anthocyanin synthesis in maize (Jenkins 2009) along with UV-B suppression of the hypocotyl elongation in tomato plants (Ballaré et al. 1995). In addition, other possible chromophore can be a phenolic molecule, with this assumption the *p*-coumaric acid chromophore present in the photoactive yellow protein (PYP), a photoreceptor found in the purplephotosynthetic bacteria Ectothiorhodospira halophila, enables to absorb in UV-A and blue regions of the solar radiation spectrum (Imamoto and Kataoka 2007). Moreover, an alternative possibility is that the UV-B radiation be sensed through some form of direct activation of a cellular component. Whether or not the UV-B photoreceptor exists the responses to UV-B radiation could be mediated by nonspecific signaling pathways involving the DNA damage, ROS production, hormone synthesis (e.g., salicylic acid, ethylene, jasmonic acid), and wound/defense signaling molecules (e.g., flavonoids, phenolics) (Apel and Hirt 2004; Demkura et al. 2010); or by specific UV-B signaling pathways mediated by the UV-B-specific component UV RESISTANCE LOCUS8 (UVR8) (Cloix and Jenkins 2008.). The UVR8 acts specifically to mediate the UV-B response, together with the expression of genes to establish the UV-B protection (Jenkins 2009). Moreover, the UVR8 also mediates the expression of genes activated at low UV-B fluence level, showing consistency with their involvement in the photomorphogenic UV-B signaling pathway (Brown and Jenkins 2008). No other component is known to act specifically in the photomorphogenic UV-B responses. The transcriptome analysis revealed that a set of approximately 70 indentified genes are stimulated by UV-B under control of the UVR8. Among these several genes are known to have key roles in the UV-B protection mechanism, including those encoding principal enzymes of the flavonoid biosynthetic pathway, as well as DNA photolyases and enzymes involved in amelioration of the photooxidative damage (Jenkins 2009). The findings demonstrate that the Arabidopsis UVR8 mutant shows severe necrosis under exposure to UV-B levels found in the bright sunlight, whereas it is indistinguishable from the wild type in the absence of UV-B (Brown and Jenkins 2008). The UVR8 regulates the expression of both ELONGATED HYPOCOTYL5 (HY5) and HY5 HOMOLOG (HYH) transcription factors at low UV-B fluence levels. The transcriptome analysis shows approximately the half of genes regulated by the UVR8 is also regulated by the HY5 transcriptor factor, but this is an underestimate and does not take into account the functional redundancy between HY5 and HYH (Brown et al. 2005). Further analysis, however, suggests that the HY5 and HYH transcription factors may regulate all the genes of the UVR8 component, and therefore are pivotal downstream effectors of the UVR8 signaling pathway. In fact, the HY5 is evidently a very important regulator of the UV-B responses because the *HY5* mutant, similar to the *UVR8* mutant, is very sensitive to UV-B, while the *HYH* mutant is less sensitive indicating that it has a subsidiary role (Brown and Jenkins 2008). These findings demonstrate that the UVR8 is a key regulator of the UV-B protection and therefore helps to promote the survival of terrestrial plants exposed to UV-B radiation (Jenkins 2009).

Another important component of the low UV-BsignalingpathwayistheCONSTITUTIVELY PHOTOMORPHOGENIC1 (COP1) (Oravecz et al. 2006). Contrarily to UVR8, the COP1 represses the expression of photomorphogenic genes and the plant development in the darkness. The COP1 acts as an E3 ubiquitin ligase, destroys HY5 and other positive regulators of the expression of photomorphogenic genes (Yi and Deng 2005). Following illumination, the COP1 is inactivated and moves slowly out of the nucleus, enabling the HY5 and other transcription factors to accumulate and promote the photomorphogenesis. In contrast to this function, the COP1 is a positive regulator of the UV-B responses such as the accumulation of flavonoids. Nevertheless, nearly half of genes regulated by the COP1 are controlled by HY5, indicating HY5 a key effector of the COP1 pathway. Both COP1 and HY5 transcriptor factors must act together in the nucleus to evoke the UV-B responses. Furthermore, the positive role of the COP1 seems not to be specific to UV-B because some evidences show a comparable function in several responses to the red light that require the involvement of the phytochrome B. With regard to this theory, the COP1 is required for the nuclear accumulation of the transcriptor factor PHYTOCHROME INTERACTING FACTOR3 (PIF3) in the darkness, although it does not mediate its destruction following the red and far-red illumination (Oravecz et al. 2006). Moreover, a more recent study showed that exposure to supplemental far-red (FR) light compared to red (R) light under UV-B radiation leads to a fast elongation growth and a phenolic accumulation in leaves of the silver birch seedlings (Tegelberg et al. 2004). Whether the COP1

acts positively in other light responses, however, still remains unknown (Jenkins 2009). On the other hand, both UVR8 and COP1 regulate many of the same genes and are required for the low fluence UV-B induction of the HY5 transcriptor factor expression which plays a central role in the regulation of genes involved in the photomorphogenic UV-B responses (Brown and Jenkins 2008). Although both UVR8 and COP1 seem to function in the same pathway, little information is available to explain their functional relationships. Since UVR8 is a UV-B specific component, it may have a direct action of the COP1 in UV-B responses. To explain this fact one possibility is that the UVR8 regulates the nuclear accumulation of the COP1 or vice versa, while another can be that the UVR8 recruits the COP1 into a complex involved in the regulation of the transcription by UV-B. The last hypothesis is supported by a recent study demonstrating that the UVR8 colocalizes with the COP1 and directly interacts with a UV-B-dependent manner (Favory et al. 2009). Besides these, the expression of some genes at low UV-B fluence levels occurs independently of the action of both UVR8 and COP1 components (Jenkins 2009).

According to data of A-H-Mackerness et al. (2001), the expression of genes by intermediate UV-B fluence levels (IF-UV-B) may be regulated partly by the enzymatic ROS formation after the specific UV-B induction, whereas the changes in mRNA levels of the high fluence (HF-UV-B) genes could be due to the formation of ROS as a result of the nonspecific damage to plant cells. However, it seems unlikely that sufficient ROS would be generated by the exposure of plants to the current ambient solar UV-B to cause the activation of the signaling pathway leading to biosynthetic responses; thus, presumably, the activation by ROS would not be UV-B specific (Jenkins 2009). Nevertheless, evidence exists for the involvement of ROS in some morphological changes and gene expression responses initiated by the UV-B radiation. Furthermore, the Arabidopsis RADICAL-INDUCED CELL DEATH1 (RCD1) transcriptor factor is also involved in the UV-B signaling pathway. Interestingly, the expression of *RCD1* genes is not significantly changed by the UV-B radiation. Previous study has shown that the SALT TOLERANCE (STO) protein is interacting with RCD1 in vitro being the mRNA level of the *STO* (*SALT TOLERANCE*) gene greatly increased in the *Arabidopsis rcd1-1* mutant after UV-B irradiation. However, the expression of UV-Binduced*HY5* and *CHS*(CHALCONESYNTHASE) genes is partially inhibited in the *STO* mutant, subsequently, seems to be the RCD1, together with the STO, involved in the *Arabidopsis* UV-B signaling (Jiang et al. 2009).

The cotyledon curling in Brassica napus stimulated by both UV-B and H₂O₂ is also inhibited by ascorbate (Gerhardt et al. 2005). In addition, the exposure to relatively high fluence rates of UV-B decreases the abundance of transcripts of the Arabidopsis LHCB1 gene that encodes the major chlorophyll binding protein of the chloroplast, and this response is inhibited by ascorbate as well as by a scavenger of superoxide radicals (A-H-Mackerness et al. 1999). Interestingly, supplementation of the ambient UV-B under greenhouse conditions increased the formation of CPDs and reduced the leaf area in G. magellanica, however did not cause lipid peroxidation being the modulation of the ascorbate content that counters the oxidative stress (Giordano et al. 2004).

The exposure to UV-B stimulates the expression of a set of genes normally induced in response to the pathogen attack, insect predation, and wounding (Stratmann 2003; Ulm and Nagy 2005). It also reduces the level of insect herbivory in a range of species, probably because of increased production of the secondary metabolites, proteinase inhibitors, and other molecules that deter the herbivorous insects (Ibañez et al. 2008). An explanation for the overlap in responses to UV-B, wounding, and pathogens is that the UV-B stimulates the accumulation of ROS and other signaling molecules (e.g., jasmonic acid, ethylene, salicylic acid) that mediate the wounding defense responses (Izaguirre et al. 2007; Demkura et al. 2010). Molecules mediating the responses of some UV-Bregulated genes have been reported (Izaguirre et al. 2003). Alterations in the induction of defenserelated genes by the UV-B radiation have been observed in both ethylene ETR-1 and jasmonic acid JAR1 Arabidopsis insensitive mutants (A-H-Mackerness et al. 1999; Jenkins 2009). Also a transgenic Arabidopsis plant expressing the salicylate hydroxylase was unable to accumulate salicylic acid and showed a reduced UV-B induction of the several PR genes (Surplus et al. 1998). The expression of genes related to the pathogenesisrelated proteins (PR) and class I Endo-β-1,3glucanases (IBGlus I) are also induced by the UV-B radiation (Kucera et al. 2003). The PR genes have been grouped as: (a) intermediate UV-B level genes (PR-5); (b) high UV-B level genes (like PR-1). The IβGlus I are also assigned to PR proteins and constitute the PR-2 family. In addition, it has been demonstrated that the UV-B-induced DNA damage seems to be related to induction of the IBGlus I genes, but not with the synthesis of flavonoids under high levels of UV-B radiation (Kucera et al. 2003). The involvement of ROS in defense signaling is well established and evidences show that superoxide generated by the plasma membrane NADPH oxidase seems to be involved in UV-B-induced regulation of some defense genes, either directly or through the production of H₂O₂ (A-H-Mackerness et al. 2001). In agreement with this assumption, it has been reported that both redox activity of the plasma membrane and cytosolic-free Ca2+ homeostasis are involved in the induction of gene expressions by UV-B and blue/ UV-A wavelengths in Arabidopsis plants (Long and Jenkins 1998). Although the available data supports that ROS might be used by plants to modulate the expression of different genes in response to varying levels of UV-B, little information is available regarding the scope and nature of the ROS production and function in response to the solar UV-B radiation under natural growth conditions. However, the ROS are very dangerous to the cellular integrity and must be eliminated (Agrawal et al. 2009). Most plants scavenge the excessive amounts of ROS using a combination of enzymatic scavengers such as superoxide dismutase, ascorbate peroxidase and glutathione reductase, and nonenzymatic scavengers such as ascorbate, glutathione, carotenoids, tocopherols, and secondary metabolites (mainly flavonoids, hydroxycinnamic acids derivatives, and anthocyanins) (Xu et al. 2008).

Although it is clear that the UV-B radiation stimulates both defense and wound signaling, there is little information on how the UV-B activates components of the signaling pathways. Experiments in tomato indicate that the UV-B radiation initiates similar signaling processes to systemin, an 18-aminoacid peptide that stimulates the wound responses (Ulm and Nagy 2005). Although data suggests that the tomato systemin receptor might also function as a brassinosteroid receptor (Wang and He 2004), more recent evidences indicate otherwise (Holton et al. 2007). In fact, a better knowledge of the mechanisms involved in the signaling pathways can help to understand the functional roles of the solar UV-B radiation in the resistance of plants to environmental factors into the terrestrial ecosystems.

9 Reduced Solar UV-B: A Future Scenario?

Despite the influence that the ambient level of solar UV-B radiation can exert on the plant life, recent findings have shown that the increases of UV-absorbing tropospheric gases (e.g., ozone, SO_2 , NO_2) and aerosols can reduce the amount of solar UV-B radiation reaching the Earth's surface (McKenzie et al. 2001). Thus, a further attention is needed to understand how the reduced solar UV-B can have an effect on the dynamics of ecosystems. Similar cues can occur when species originated from highland regions such as mountains (Ren et al. 2010), the Bolivian Altiplano (González et al. 2009) and the Tibet Plateau (Yang et al. 2008) or those from the equatorial regions that naturally receive high UV-B irradiances, are grown in lowland areas and/or high latitudes (Turunen and Latola 2005).

So far the ecological cost of terrestrial plant responses to the solar UV-B radiation has not been studied in detail. Clearly, some morphological and physiological responses must have a cost in terms of resources that in absence of the UV-B could have been allocated elsewhere (Snell et al. 2009). Under this perspective seems to be an interesting point to study the question *how the terrestrial plants respond to the solar UV-B* radiation growing under UV-B-reduced irradiances, because till now data on this topic are very scarce. Like morphological features, emerging findings revealed that the UV-absorbing compounds could be affected by the reduced solar UV-B (Ibañez et al. 2008; González et al. 2009). In this context, the removal of UV-B from the natural solar radiation causes large increases of the growth in Glycine max and Cyamopsis plants and a marginal increase in the Vigna radiata but did not affect the growth of Vigna mungo plants (Varalakshmi et al. 2003; Amudha et al. 2010). In addition, it has also been demonstrated that the photomorphogenic regulatory mechanisms, rather than the photosynthesis seems to play key roles in the observed metabolic changes upon exposure to the reduced solar UV-B (Kadur et al. 2007). In agreement with these findings, the prolonged treatment with chronic low doses of UV-B caused changes in the morphology, gene expression, and biomass redistribution without cessation of the growth and in absence of the stress symptoms (Hectors et al. 2007). In fact, the reduced solar UV-B induces probably a plethora of key enzymes into the metabolic pathways transmitting a general plant response, which under a future long-time solar UV-B-reduced scenario will probably affect the plant productivity, species competition, trophic interactions, and ultimately the structure of ecosystems.

10 Conclusion and Future Perspective

The enhanced UV-B radiation produced important physiological and morphological effects on the terrestrial plants, but most of these studies were carried out in greenhouses and growth chambers. Then the extrapolation and quantification of the observed indoor effects to field experiments is very complex, because they can reflect exacerbated responses of the plants and confuse the interpretation of physiological and morphogenic responses, as well as the molecular analysis at both individual species and ecosystem composition. In fact, much remains to be done to define and establish the effects of both increased and reduced solar UV-B, as well as the signaling pathways to understand how they may be integrated to terrestrial plants growing in the natural environment. *However, although in the solar UV-B alchemy, each successive understanding produces a larger doubt, God does not play dice with the universe! (Einstein).*

Acknowledgments We are grateful to anonymous reviewers for their useful comments on the manuscript. We are also grateful for the financial support provided by the Consejo Nacional de Investigaciones de la Universidad Nacional de Tucumán (CIUNT, grants 26/G423 and 26/ G437). FEP and MR are members of the Career of Investigator from the Consejo Nacional de Investigaciones Científicas y Técnicas (CONICET, Argentina).

References

- Agati G, Tattini M (2010) Multiple functional roles of flavonoids in photoprotection. New Phytol 186:786–793
- Agati G, Matteini P, Goti A, Tattini M (2007) Chloroplastlocated flavonoids can scavenge singlet oxygen. New Phytol 174:77–89
- Agrawal SB, Singh S, Agrawal M (2009) Ultraviolet-B induced changes in gene expression and antioxidants in plants. Adv Bot Res 52:47–86
- A-H-Mackerness S, Surplus SL, Blake P, John CF, Buchanan-Wollaston V et al (1999) Ultraviolet-Binduced stress and changes in gene expression in *Arabidopsis thaliana*: role of signaling pathways controlled by jasmonic acid, ethylene and reactive oxygen species. Plant Cell Environ 22:1413–1423
- A-H-Mackerness S, John CF, Jordan B, Thomas B (2001) Early signalling components in ultraviolet-B responses: distinct roles for different reactive oxygen species and nitric oxide. FEBS Lett 489:237–242
- Amudha P, Jayakumar M, Kulandaivelu G (2010) Photosystems activities and polypeptide composition of *Cyamopsis tetragonoloba* and *Vigna mungo* thylakoids as affected by exclusion of solar UV radiation. Biol Plant 54:339–343
- Apel K, Hirt H (2004) Reactive oxygen species: metabolism, oxidative stress, and signal transduction. Annu Rev Plant Biol 55:373–399
- Archetti M, Döring TF, Hagen SB, Hughes NM, Leather SR et al (2009) Unraveling the evolution of autumn colours: an interdisciplinary approach. Trends Ecol Evol 24:166–173
- Ballaré CL (2003) Stress under the sun. Spotlight on ultraviolet-B responses. Plant Physiol 132:1725–1727
- Ballaré CL, Barnes PW, Flint SD (1995) Inhibition of hypocotyl elongation by UV-B radiation in deetiolating tomato seedlings. 1. The photoreceptor. Physiol Plant 93:584–592

- Ballaré CL, Rousseaux MC, Searles PS, Zaller JG, Giordano CV et al (2001) Impacts of solar ultraviolet-B radiation on terrestrial ecosystems of Tierra del Fuego (southern Argentina): an overview of recent progress. J Photochem Photobiol B Biol 62:67–77
- Barnes PW, Shinkle JR, Flint SD, Ryel RJ (2005) UV-B radiation, photomorphogenesis and plant–plant interactions. Prog Bot 66:313–340
- Bassman JH (2004) Ecosystem consequences of enhanced solar ultraviolet radiation: secondary plant metabolites as mediators of multiple trophic interactions in terrestrial plant communities. Photochem Photobiol 79:382–398
- Bieza K, Lois R (2001) An Arabidopsis mutant tolerant to lethal ultraviolet-B levels shows constitutively elevated accumulation of flavonoids and other phenolics. Plant Physiol 126:1105–1115
- Björn LO, McKenzie RL (2007) Attempts to probe the ozone layer and the ultraviolet-B levels of the past. Ambio 36:366–371
- Björn LO, Widell S, Wang T (2002) Evolution of UV-B regulation and protection in plants. Adv Space Res 30:1557–1562
- Boccalandro HE, Mazza CA, Mazzella MA, Casal JJ, Ballaré CL (2001) Ultraviolet B radiation enhances a phytochrome-B mediated photomorphogenic response in *Arabidopsis*. Plant Physiol 126:780–788
- Brenes-Arguedas T, Horton MW, Coley PD, Lokvam J, Rachel A, Waddell RA et al (2006) Contrasting mechanisms of secondary metabolite accumulation during leaf development in two tropical tree species with different leaf expansion strategies. Oecologia 149:91–100
- Britt AB (2004) Repair of DNA damage induced by solar UV. Photosynth Res 81:105–112
- Brosché M, Schuler MA, Kalbina I, Connor L, Strid (2002) Gene regulation by low level UV-B radiation: identification by DNA array analysis. Photochem Photobiol Sci 1:656–664
- Brown BA, Jenkins GI (2008) UV-B signalling pathways with different fluence-rate response profiles are distinguished in mature Arabidopsis leaf tissue by requirement for UVR8, HY5, and HYH. Plant Physiol 146:576–588
- Brown BA, Cloix C, Jiang GH, Kaiserli E, Herzyk P et al (2005) A UV-B-specific signaling component orchestrates plant UV protection. Proc Natl Acad Sci USA 102:18225–18230
- Bruno JF, Stachowicz JJ, Bertness MD (2003) Inclusion of facilitation into ecological theory. Trends Ecol Evol 18:119–125
- Buer CS, Imin N, Djordjevic MA (2010) Flavonoids: new roles for old molecules. J Integr Plant Biol 52:98–111
- Bukhov NG (2004) Dynamic light regulation of photosynthesis (A review). Russ J Plant Physiol 51:742–753
- Burchard P, Bilger W, Weissenböck G (2000) Contribution of hydroxycinnamates and flavonoids to epidermal shielding of UV-A and UV-B radiation in developing rye primary leaves as assessed by ultraviolet-induced chlorophyll fluorescence measurements. Plant Cell Environ 23:1373–1380

- Caldwell MM, Bornman JF, Ballaré CL, Flint SD, Kulandaivelu G (2007) Terrestrial ecosystems, increased solar ultraviolet radiation, and interactions with other climate change factors. Photochem Photobiol Sci 6:252–266
- Carvalho RF, Takaki M, Azevedo RA (2010) Plant pigments: the many faces of light perception. Acta Physiol Plant. doi:10.1007/s11738-010-0533-7
- Chalker-Scott L (2002) Do anthocyanins function as osmoregulators in leaf tissues? Adv Bot Res 37:103–127
- Clarke LJ, Robinson SA (2008) Cell wall-bound ultraviolet-screening compounds explain the high ultraviolet tolerance of the Antarctic moss, Ceratodon purpureus. New Phytol 179:776–783
- Cloix C, Jenkins GI (2008) Interaction of the Arabidopsis UV-B-specific signaling component UVR8 with chromatin. Mol Plant 1:118–128
- Close DC, Beadle CL (2003) The ecophysiology of foliar anthocyanin. Bot Rev 69:149–161
- Cockell CS, Horneck G (2001) The history of the UV radiation climate of the Earth–theoretical and spacebased observations. Photochem Photobiol 73:447–451
- Convey P, Smith RIL (2006) Responses of terrestrial Antarctic ecosystems to climate change. Plant Ecol 182:1–10
- Cybulski WJ, Peterjohn WT, Sullivan JH (2000) The influence of elevated ultraviolet-B radiation on tissue quality and decomposition of loblolly pine needles. Environ Exp Bot 44:231–241
- Dany AL, Douki T, Triantaphylides C, Cadet J (2001) Repair of the main UV-induced thymine dimeric lesions within *Arabidopsis thaliana* DNA: evidence for the major involvement of photoreactivation pathways. J Photochem Photobiol B Biol 65:127–135
- Day TA, Ruhland CT, Xiong F (2001) Influence of solar UV-B radiation on Antarctic terrestrial plants: results from a 4-year field study. J Photochem Photobiol B Biol 62:78–87
- Demkura PV, Abdala G, Baldwin IT, Ballaré CL (2010) Jasmonate-dependent and -independent pathways mediate specific effects of solar ultraviolet B radiation on leaf phenolics and antiherbivore defense. Plant Physiol 152:1084–1095
- Dixon P, Weinig C, Schmitt J (2001) Susceptibility to UV damage in *Impatiens capensis* (Balsaminaceae): testing for opportunity costs to shade-avoidance and population differentiation. Am J Bot 88:1401–1408
- Dormann CF, Woodin SJ (2002) Climate change in the Arctic: using plant functional types in a meta-analysis of field experiments. Funct Ecol 16:4–17
- Favory J, Stec A, Gruber H, Rizzini L, Oravecz A et al (2009) Interaction of COP1 and UVR8 regulates UV-B-induced photomorphogenesis and stress acclimation in *Arabidopsis*. EMBO J 28:591–601
- Feucht W, Treutter D, Polster J (2004) Flavanol binding of nuclei from tree species. Plant Cell Rep 22:430–436
- Flint SD, Caldwell MM (2003) A biological spectral weighting function for ozone depletion research with higher plants. Physiol Plant 117:137–144

- Flint SD, Ryel RJ, Caldwell MM (2003) Ecosystem UV-B experiments in terrestrial communities: a review of recent findings and methodologies. Agric For Meteorol 120:177–189
- Flint SD, Ryel RJ, Hudelson TJ, Caldwell MM (2009) Serious complications in experiments in which UV doses are affected by using different lamp heights. J Photochem Photobiol B Biol 97:48–53
- Fukuda S, Satoh A, Kasahara H, Matsuyama H, Takeuchi Y (2008) Effects of ultraviolet-B irradiation on the cuticular wax of cucumber (*Cucumis sativus*) cotyledons. J Plant Res 121:179–189
- Furness NH, Upadhyaya MK (2002) Differential susceptibility of agricultural weeds to ultraviolet-B radiation. Can J Plant Sci 82:789–796
- Gallo ME, Sinsabaugh RL, Cabaniss SE (2006) The role of ultraviolet radiation in litter decomposition in arid ecosystems. Appl Soil Ecol 34:82–91
- Gerhardt KE, Wilson MI, Greenberg BM (2005) Ultraviolet wavelength dependence of photomorphological and photosynthetic responses in *Brassica napus* and *Arabidopsis thaliana*. Photochem Photobiol 81:1061–1068
- Gilbert M, Pörs Y, Grover K, Weingart I, Skotnica J et al (2009) Intra- and interspecific differences of 10 barley and 10 tomato cultivars in response to short-time UV-B radiation: a study analysing thermoluminescence, fluorescence, gas-exchange and biochemical parameters. Environ Pollut 157:1603–1612
- Giordano CV, Galatro A, Puntarulo S, Ballaré CL (2004) The inhibitory effects of UV-B radiation (280–315 nm) on *Gunnera magellanica* growth correlate with increased DNA damage but not with oxidative damage to lipids. Plant Cell Environ 27:1415–1423
- González JA, Liberman-Cruz M, Boero C, Gallardo M, Prado FE (2002) Leaf thickness, protective and photosynthetic pigments and carbohydrate content in leaves of the world's highest elevation tree *Polylepis tarapacana* (Rosaceae). Phyton 71:41–53
- González JA, Gallardo MG, Boero C, Liberman-Cruz M, Prado FE (2007) Altitudinal and seasonal variation of protective and photosynthetic pigments in leaves of the world's highest elevation trees *Polylepis tarapacana* (Rosaceae). Acta Oecol 32:36–41
- González JA, Rosa MA, Parrado MF, Hilal M, Prado FE (2009) Morphological and physiological responses of two varieties of a highland species (*Chenopodium quinoa* Willd.) growing under near-ambient and strongly reduced solar UV-B in a lowland location. J Photochem Photobiol B Biol 96:144–151
- Gould KS (2004) Nature's swiss army knife: the diverse protective roles of anthocyanins in leaves. J Biomed Biotechnol 5:314–320
- Hada H, Hidema J, Maekawa M, Kumagai T (2003) Higher amounts of anthocyanins and UV-absorbing compounds effectively lowered CPD photorepair in purple rice (*Oryza sativa* L.). Plant Cell Environ 26:1691–1701
- He JM, Bai XL, Wang RB, Cao B, She XP (2007) The involvement of nitric oxide in ultraviolet-B-inhibited

pollen germination and tube growth of *Paulownia* tomentosa in vitro. Physiol Plant 131:273–282

- Hectors K, Prinsen E, De Coen W, Jansen MAK, Guisez Y (2007) Arabidopsis thaliana plants acclimated to low dose rates of ultraviolet B radiation show specific changes in morphology and gene expression in the absence of stress symptoms. New Phytol 175:255–270
- Hidema J, Taguchi T, Ono T, Teranishi M, Yamamoto K, Kumagai T (2007) Increase in CPD photolyase activity functions effectively to prevent growth inhibition caused by UVB radiation. Plant J 50:70–79
- Hilal M, Parrado MF, Rosa M, Gallardo M, Orce L et al (2004) Epidermal lignin deposition in quinoa cotyledons in response to UV-B radiation. Photochem Photobiol 79:205–210
- Hilal M, Rodríguez-Montelongo L, Rosa M, Gallardo M, González JA et al (2008) Solar and supplemental UV-B radiation effects in lemon peel UV-B absorbing compound content Seasonal variations. Photochem Photobiol 84:1480–1486
- Hofmann RW, Campbell BD, Bloor SJ, Swinny EF, Markham KR et al (2003) Responses to UV-B radiation in *Trifolium repens* L. – physiological links to plant productivity and water availability. Plant Cell Environ 26:603–612
- Holton N, Cano-Delgado A, Harrison K, Montoya T, Chory J, Bishop GJ (2007) Tomato BRASSINOSTEROID INSENSITIVE1 is required for systemin-induced root elongation in Solanum pimpinellifolium but is not essential for wound signaling. Plant Cell 19:1709–1717
- Hoque E, Remus G (1999) Natural UV-screening mechanisms of Norway spruce (*Picea abies* [L.] Karst.) needles. Photochem Photobiol 69:177–192
- Hughes NM, Neufeld HS, Burkey KO (2005) Functional role of anthocyanins in high-light winter leaves of the evergreen herb *Galax urceolata*. New Phytol 168:575–587
- Ibañez S, Rosa M, Hilal M, González JA, Prado FE (2008) Leaves of *Citrus aurantifolia* exhibit a different sensibility to solar UV-B radiation according to development stage in relation to photosynthetic pigments and UV-B absorbing compounds production. J Photochem Photobiol B 90:163–169
- Imamoto Y, Kataoka M (2007) Structure and photoreaction of photoactive yellow protein, a structural prototype of the PAS domain superfamily. Photochem Photobiol 83:40–49
- Ivanova PI, Dobrikova AG, Taneva SG, Apostolova EL (2008) Sensitivity of the photosynthetic apparatus to UV-A radiation: role of light-harvesting complex II-photosystem II supercomplex organization. Radiat Environ Biophys 47:169–177
- Izaguirre MM, Scopel AL, Baldwin IT, Ballaré CL (2003) Convergent responses to stress: solar ultraviolet-B radiation and Manduca sexta herbivory elicit overlapping transcriptional responses in field-grown plants of *Nicotiana longiflora*. Plant Physiol 132:1755–1767
- Izaguirre MM, Mazza CA, Svat s A, Baldwin IT, Ballaré CL (2007) Solar ultraviolet-B radiation and insect herbivory trigger partially overlapping phenolic responses

in *Nicotiana attenuata* and *Nicotiana longiflora*. Ann Bot 99:103–109

- Jenkins GI (2009) Signal transduction in responses to UV-B radiation. Annu Rev Plant Biol 60:407–431
- Jiang L, Wang Y, Björn LO, Li S (2009) Arabidopsis RADICAL-INDUCED CELL DEATH1 is involved in UV-B signaling. Photochem Photobiol Sci 8:838–846
- Jordan BR (2002) Molecular response of plant cells to UV-B stress. Funct Plant Biol 29:909–916
- Julkunen-Tiitto R, Häggman H, Aphalo PJ, Lavola A, Tegelberg R, Veteli T (2005) Growth and defense in deciduous trees and shrubs under UV-B. Environ Pollut 137:404–414
- Kadur G, Swapan B, Sunita K, Sanjeev Y, Arjun T et al (2007) Growth enhancement of soybean (*Glycine max*) upon exclusion of UV-B and UV-B/A components of solar radiation: characterization of photosynthetic parameters in leaves. Photosynth Res 94:299–306
- Kakani VG, Reddy KR, Zhao D, Mohammed AR (2003) Effects of ultraviolet-B radiation on cotton (*Gossypium hirsutum* L.) morphology and anatomy. Ann Bot 91:817–826
- Kalbina I, Li S, Kalbin G, Björn LO, Strid Å (2008) Two separate UV-B radiation wavelength regions control expression of different molecular markers in *Arabidopsis thaliana*. Funct Plant Biol 35:222–227
- Kimura S, Tahira Y, Ishibashi T, Mori Y, Mori T et al (2004) DNA repair in higher plants; photoreactivation is the major DNA repair pathway in non proliferating cells while excision repair (nucleotide excision repair and base excision repair) is active in proliferating cells. Nucleic Acids Res 32:2760–2767
- Kolodner D, Marsischky GT (1999) Eukaryotic DNA mismatch repair. Curr Opin Genet Dev 9:89–96
- Koti S, Reddy KR, Kakani VG, Zhao D, Reddy VR (2004) Soybean (*Glycine max*) pollen germination characteristics, flower and pollen morphology in response to enhanced ultraviolet-B radiation. Ann Bot 94:855–864
- Krauss P, Markstädter C, Rieder M (1997) Attenuation of UV radiation by plant cuticles from woody species. Plant Cell Environ 20:1079–1085
- Krizek DT (2004) Influence of PAR and UV-A in determining plant sensitivity and photomorphogenic responses to UV-B radiation. Photochem Photobiol 79:307–315
- Krizek DT, Mirecki RM (2004) Evidence for phytotoxic effects of cellulose acetate in UV exclusion studies. Environ Exp Bot 51:33–43
- Krizek DT, Clark HD, Mirecki RM (2005) Spectral properties of selected UV-blocking and UV-transmitting covering materials with application for production of high-value crops in high tunnels. Photochem Photobiol 81:1047–1051
- Kucera B, Leubner-Metzger G, Wellmann E (2003) Distinct ultraviolet-signaling pathways in bean leaves. DNA damage is associated with β-1,3 glucanase gene induction, but not with flavonoid formation. Plant Physiol 133:1445–1452
- Kytridis VP, Manetas Y (2006) Mesophyll versus epidermal anthocyanins as potential *in vivo* antioxidants: evidence linking the putative antioxidant role to the

proximity of oxy-radical source. J Exp Bot 57: 2203–2210

- Láposi R, Veres S, Mile O, Mészáros I (2002) Photosynthesis-ecophysiological properties of beech (*Fagus sylvatica* L.) under the exclusion of ambient UV-B radiation. Acta Biol Szegediensis 46:243–245
- Leshem YY, Kuiper PJC (1996) Is there a GAS (general adaptation syndrome) response to various types of environmental stress? Biol Plant 38:1–18
- Li FR, Peng SL, Chen BM, Hou YP (2010) A meta-analysis of the responses of woody and herbaceous plants to elevated ultraviolet-B radiation. Acta Oecol 36:1–9
- Liakopoulos G, Stavrianakou S, Karabourniotis G (2006) Trichome layers versus dehaired lamina of *Olea europaea* leaves: differences in flavonoid distribution, UV-absorbing capacity, and wax yield. Environ Exp Bot 55:294–304
- Liu Z, Hong SW, Escobar M, Vierling E, Mitchell D et al (2003) *Arabidopsis UVH6*, a homolog of human *XPD* and yeast *RAD3* DNA repair genes, functions in DNA repair and is essential for plant growth. Plant Physiol 132:1405–1414
- Long JC, Jenkins GI (1998) Involvement of plasma membrane redox activity and calcium homeostasis in the UV-B and UV-A/blue light induction of gene expression in *Arabidopsis*. Plant Cell 10:2077–2086
- McKenzie RL, Seckmeyer G, Bais AF, Kerr JB, Madronich S (2001) Satellite retrievals of erythemal UV dose compared with ground–based measurements at northern and southern midlatitudes. J Geophys Res 106:24051–24062
- McKenzie RL, Aucamp PJ, Bais AF, Björn LO, Ilyas M (2007) Changes in biologically-active ultraviolet radiation reaching the Earth's surface. Photochem Photobiol Sci 6:218–231
- Merzlyak MN, Chivkunova OB, Solovchenko AE, Naqvi KR (2008a) Light absorption by anthocyanins in juvenile, stressed, and senescing leaves. J Exp Bot 59:3903–3911
- Merzlyak MN, Melø TB, Naqvi KR (2008b) Effect of anthocyanins, carotenoids, and flavonols on chlorophyll fluorescence excitation spectra in apple fruit: signature analysis, assessment, modelling, and relevance to photoprotection. J Exp Bot 59:349–359
- Micheletti MI, Piacentini RD, Madronich S (2003) Sensitivity of biologically active UV radiation to stratospheric ozone changes: effects of action spectrum shape and wavelength range. Photochem Photobiol 78:456–461
- Morales LO, Tegelberg R, Brosché M, Keinánen M, Lindfor A, Aphalo PJ (2010) Effects of solar UV-A and UV-B radiation on gene expression and phenolic accumulation in *Betula pendula* leaves. Tree Physiol 30:923–934
- Mpoloka SW, Abratt VA, Mundree SG, Thomson JA, Musil CF (2007) Potential effects of prolonged ultraviolet radiation exposure in plants: chloroplast DNA analysis. Am Eur J Agric Environ Sci 2:437–441
- Musil CF, Björn LO, Scourfield MWJ, Bodeker GE (2002a) How substantial are ultraviolet B supplementation

inaccuracies in experimental square-wave delivery systems? Environ Exp Bot 47:25–38

- Musil CF, Chimphango SBM, Dakora FD (2002b) Effects of elevated ultraviolet-B radiation on native and cultivated plants of Southern Africa. Ann Bot 90:127–137
- Neill SO, Gould KS (2003) Anthocyanins in leaves: light attenuators or antioxidants? Funct Plant Biol 30:865–873
- Newsham KK, Robinson SA (2009) Responses of plants in polar regions to UVB exposure: a meta-analysis. Glob Change Biol 15:2574–2589
- Oravecz A, Baumann A, Máté Z, Brzezinska A, Molinier J et al (2006) CONSTITUTIVELY PHOTOMOR PHOGENIC1 is required for the UV-B response in *Arabidopsis*. Plant Cell 18:1975–1990
- Page JE, Towers GHN (2002) Anthocyanins protect lightsensitive thiarubrine phototoxins. Planta 215:478–484
- Pal M, Sharma A, Abrol YP, Sengupta UK (1997) Exclusion of UV-B radiation from normal solar spectrum on the growth of mung bean and maize. Agric Ecosyst Environ 61:29–34
- Pancotto VA, Sala OE, Cabello M, López NI, Robson TM et al (2003) Solar UV-B decreases decomposition in herbaceous plant litter in Tierra del Fuego, Argentina: potential role of an altered decomposer community. Glob Change Biol 9:1465–1474
- Phoenix GK, Gwynn-Jones D, Lee JA, Callaghan TV (2003) Ecological importance of ambient solar ultraviolet radiation to a sub-arctic heath community. Plant Ecol 165:263–273
- Pliura A, Baliuckiene A, Baliuckas V (2008) Phenogenetic response of silver birch populations and half-sib families to elevated ozone and ultraviolet-B radiation at juvenile age. Environ Pollut 156:152–161
- Pradhan MK, Joshi PN, Nair JS, Ramaswamy NK, Iyer RK et al (2006) UV-B exposure enhances senescence of wheat leaves: modulation by photosynthetically active radiation. Radiat Environ Biophys 45:221–229
- Qaderi MM, Reid DM (2005) Growth and physiological responses of canola (*Brassica napus*) to UV-B and CO₂ under controlled environment conditions. Physiol Plant 125:247–259
- Ren J, Duan B, Zhang X, Korpelainen H, Li C (2010) Differences in growth and physiological traits of two poplars originating from different altitudes as affected by UV-B radiation and nutrient availability. Physiol Plant 138:278–288
- Rex M, Salawitch RJ, von der Gathen P, Harris NRP, Chipperfield MP, Naujokat B (2004) Arctic ozone loss and climate change. Geophys Res Lett 31:L04116
- Robson TM, Pancotto VA, Flint SD, Ballaré CL, Sala OE et al (2003) Six years of solar UV-B manipulations affect growth of *Sphagnum* and vascular plants in a Tierra del Fuego peatland. New Phytol 160:379–389
- Rozema J (2000) Effects of solar UV-B radiation on terrestrial biota. In: Hester RE, Harrison RM (eds) Causes and environmental implications of increased UV-B radiation. issues in environmental science and technology, vol 14. RSC, Cambridge, pp 86–105

- Rozema J, van de Staaij JWM, Tosserams M (1997) Effects of UV-B radiation on plants from agro- and natural ecosystems. In: Lumsden P (ed) Plants and UV-B. Responses to environmental change. Cambridge University Press, Cambridge, pp 213–232
- Rozema J, Noordijk AJ, Broekman RA, van Beem A, Meijkamp BM et al (2001) (Poly)phenolic compounds in pollen and spores of Antarctic plants as indicators of solar UV-B. Plant Ecol 154:11–26
- Rozema J, Björn LO, Bornman JF, Gaberščik A, Häder DP et al (2002) The role of UV-B radiation in aquatic and terrestrial ecosystems–an experimental and functional analysis of the evolution of UV-absorbing compounds. J Photochem Photobiol B Biol 66:2–12
- Rozema J, Boelen P, Blokker P (2005) Depletion of stratospheric ozone over the Antarctic and Arctic: responses of plants of polar terrestrial ecosystems to enhanced UV-B, an overview. Environ Pollut 137:428–442
- Rozema J, Boelen P, Solheim B, Zielke M, Buskens A et al (2006) Stratospheric ozone depletion: high arctic tundra plant growth on Svalbard is not affected by enhanced UV-B after 7 years of UV-B supplementation in the field. Plant Ecol 182:121–135
- Ryan KG, Hunt JE (2005) The effects of UVB radiation on temperate southern hemisphere forests. Environ Pollut 137:415–427
- Sampson BJ, Cane JH (1999) Impact of enhanced ultraviolet-B radiation on flower, pollen, and nectar production. Am J Bot 86:108–114
- Santos I, Fidalgo F, Almeida JM, Salema R (2004) Biochemical and ultrastructural changes in leaves of potato plants grown under supplementary UV B radiation. Plant Sci 167:925–935
- Sarma AD, Sharma R (1999) Anthocyanin–DNA copigmentation complex: mutual protection against oxidative damage. Phytochemistry 52:1313–1318
- Schmitz-Hoerner R, Weissenböck G (2003) Contribution of phenolic compounds to the UV-B screening capacity of developing barley primary leaves in relation to DNA damage and repair under elevated UV-B levels. Phytochemistry 64:243–245
- Searles PS, Flint SD, Caldwell MM (2001) A meta-analysis of plant field studies simulating stratospheric ozone depletion. Oecologia 127:1–10
- Searles PS, Flint SD, Diaz SB, Rousseaux MC, Ballaré CL, Caldwell MM (2002) Plant response to solar ultraviolet-B radiation in a southern South American *Sphagnum* peatland. J Ecol 90:704–713
- Semerdjieva SI, Sheffield E, Phoenix GK, Gwynn-Jones D, Callaghan TV, Johnson GN (2003) Contrasting strategies for UV-B screening in sub-Arctic dwarf shrubs. Plant Cell Environ 26:957–964
- Shinkle JR, Atkins AK, Humphrey EE, Rodgers CW, Wheeler SL, Barnes PW (2004) Growth and morphological responses to different UV wavebands in cucumber (*Cucumis sativum*) and other dicotyledonous seedlings. Physiol Plant 120:240–248
- Shulski MD, Walter-Shea EA, Hubbard KG, Yuen GY, Horst G (2004) Penetration of photosynthetically

active radiation and ultraviolet radiation into alfalfa and tall Fescue canopies. Agron J 96:1562–1571

- Singh SP, Kumari S, Rastogi RP, Singh KL, Sinha RP (2008) Mycosporine-like amino acids (MAAs): chemical structure, biosynthesis and significance as UV-absorbing/screening compounds. Indian J Exp Biol 46:7–17
- Smith GJ, Markham KR (1998) Tautomerism of flavonol glucosides – relevance to plant UV protection and flower colour. J Photochem Photobiol A Chem 118:99–105
- Snell KRS, Kokubun T, Griffiths H, Convey P, Hodgson DA, Newsham KK (2009) Quantifying the metabolic cost to an Antarctic liverwort of responding to UV-B radiation exposure. Glob Change Biol 15:2563–2573
- Solovchenko AE, Merzlyak MN (2008) Screening of visible and UV radiation as a photoprotective mechanism in plants. Russ J Plant Physiol 55:803–822
- Stratmann J (2003) Ultraviolet-B radiation co-opts defense signaling pathways. Trends Plant Sci 8:526–533
- Sullivan JH (2005) Possible impacts of changes in UV-B radiation on North American trees and forests. Environ Pollut 137:380–389
- Surplus SL, Jordan BR, Murphy AM, Carr JP, Thomas B, A-H-Mackerness S (1998) Ultraviolet-B induced responses in *Arabidopsis thaliana*: role of salicylic acid and reactive oxygen species in the regulation of transcripts encoding photosynthetic and acidic pathogenesis-related proteins. Plant Cell Environ 21:685–694
- Taalas P, Kaurola J, Kylling A, Shindell D, Sausen R et al (2000) The impact of greenhouse gases and halogenated species on future solar UV radiation doses. Geophys Res Lett 27:1127–1130
- Takeuchi Y, Inoue T, Takemura K, Hada M, Takahashi S et al (2007) Induction and inhibition of cyclobutane pyrimidine dimer photolyase in etiolated cucumber (*Cucumis sativus*) cotyledons after ultraviolet irradiation depends on wavelength. J Plant Res 120:365–374
- Tattini M, Galardi C, Pinelli P, Massai R, Remorini D, Agati G (2004) Differential accumulation of flavonoids and hydroxycinnamates in leaves of *Ligustrum vul*gare under excess light and drought stress. New Phytol 163:547–561
- Tattini M, Guidi L, Morassi-Bonzi L, Pinelli P, Remorini D et al (2005) On the role of flavonoids in the integrated mechanisms of response of *Ligustrum vulgare* and *Phillyrea latifolia* to high solar radiation. New Phytol 167:457–470
- Tegelberg R, Julkunen-Tiitto R, Aphalo PJ (2004) Red:farred light ratio and UV-B radiation: their effects on leaf phenolics and growth of silver birch seedlings. Plant Cell Environ 27:1005–1013
- Trošt-Sedej T, Gaberščik A (2008) The effects of enhanced UV-B radiation on physiological activity and growth of Norway spruce planted outdoors over 5 years. Trees 22:423–435
- Turcsányi E, Vass I (2000) Inhibition of photosynthetic electron transport by UV-A radiation targets the photosystem II complex. Photochem Photobiol 72:513–520

- Turunen M, Latola K (2005) UV-B radiation and acclimation in timberline plants. Environ Pollut 137:390–403
- Tuteja N, Singh MB, Misra MK, Bhalla PL, Tuteja R (2001) Molecular mechanisms of DNA damage and repair: progress in plants. Crit Rev Biochem Mol Biol 36:337–397
- Ulm R, Nagy F (2005) Signalling and gene regulation in response to ultraviolet light. Curr Opin Plant Biol 8:477–482
- Varalakshmi D, Lakshmi N, Guruprasad KN (2003) Physiological changes in soybean Cv. JS 71–05 after the exclusion of UV-A and UV-B from the solar radiation. Indian J Plant Physiol (special issue):602–606
- Velders GJM, Andersen SO, Daniel JS, Fahey DW, McFarland M (2007) The importance of the Montreal Protocol in protecting climate. Proc Natl Acad Sci USA 104:4814–4819
- Wang ZY, He JX (2004) Brassinosteroid signal transduction – choices of signals and receptors. Trends Plant Sci 9:91–96
- Winefield C (2002) The final steps in anthocyanin formation: a story of modification and sequestration. Adv Bot Res 37:55–74
- Winter TR, Rostás M (2008) Ambient ultraviolet radiation induces protective responses in soybean but does not attenuate indirect defense. Environ Pollut 155:290–297
- Xu K, Qiu S (2007) Responses of superhigh-yield hybrid rice Liangyoupeijiu to enhancement of ultraviolet-B radiation. Plant Sci 172:139–149

- Xu C, Natarajan S, Sullivan JH (2008) Impact of solar ultraviolet-B radiation on the antioxidant defense system in soybean lines differing in flavonoid contents. Environ Exp Bot 63:39–48
- Yamasaki H, Sakihama Y, Ikehara N (1997) Flavonoidperoxidase reaction as a detoxification mechanism of plant cells against H₂O₂. Plant Physiol 115:1405–1412
- Yang H, Clendenin WM, Wong D, Demple B, Slupska MM et al (2001) Enhanced activity of adenine-DNA glycosylase (Myh) by AP endonuclease (Ape1) in mammalian base excision repair of an A/GO mismatch. Nucleic Acids Res 29:743–752
- Yang Y, Yao Y, He H (2008) Influence of ambient and enhanced ultraviolet-B radiation on the plant growth and physiological properties in two contrasting populations of *Hippophae rhamnoides*. J Plant Res 121:377–385
- Yi C, Deng XW (2005) COP1 from plant photomorphogenesis to mammalian tumorigenesis. Trends Cell Biol 15:618–625
- Zavala JA, Ravetta DA (2002) The effect of solar UV-B radiation on terpenes and biomass production in *Grindelia chiloensis* (Asteraceae), a woody perennial of Patagonia. Argent Plant Ecol 161:185–191
- Zu YG, Pang HH, Yu JH, Li DW, Wei XX et al (2010) Responses in the morphology, physiology and biochemistry of *Taxus chinensis* var. mairei grown under supplementary UV-B radiation. J Photochem Photobiol B Biol 98:152–158

K⁺ Nutrition, Uptake, and Its Role in Environmental Stress in Plants

Manuel Nieves-Cordones, Fernando Alemán, Mario Fon, Vicente Martínez, and Francisco Rubio

Abstract

The increasing world population makes high yield crop production a necessity in future agriculture. However, the negative effects of the emission of greenhouse gases into Earth's atmosphere and the resulting climate change may impede reaching this goal. New stresses will appear and the existing ones will be exacerbated. Important plant processes such as the acquisition of K⁺, which is an essential macronutrient for plants, will be negatively affected. The development of new crop varieties with enhanced capacities in the acquisition of K⁺, especially under the future environmental conditions, is an important challenge. One of the first steps may be the identification of the K⁺ uptake systems operating in the roots, which may be later improved to enhance K⁺ acquisition under stress conditions. Some gene families encoding K⁺ transporters, that is, the HAK1-type, and channels, that is, the AKT1-type, key pieces for root K⁺ uptake, have been identified. Members of other families of transport systems, such as the cation proton antiporter (CPA) family, or the cyclic nucleotidegated channel (CNGC) family may also participate in that process. The use of T-DNA insertion lines in the model species Arabidopsis thaliana has allowed the demonstration of the role of some of these transport systems and, in some cases, the results obtained in the model plant also apply to crop species. Important points of the regulation of these transport systems are found at the transcriptional and the translational level. Internal K⁺ concentrations, plasma membrane potential, reactive oxygen species (ROS), hormones, kinases, and phosphatases are involved in their regulation. The knowledge accumulated to date and that to be obtained in the future could be used in biotechnological approaches to produce more efficient K⁺ transporters that endow plants with enhanced performance to face the future environmental challenges. Genetic engineering, natural variability, and the development of -omic technologies are valuable tools for the achievement of these objectives.

Departamento de Nutrición Vegetal, CEBAS-CSIC, Campus de Espinardo, Murcia 30100, Spain

F. Rubio (🖂) • M. Nieves-Cordones • F. Alemán

[•] M. Fon • V. Martínez

e-mail: frubio@cebas.csic.es

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_4,

[©] Springer Science+Business Media, LLC 2012

Keywords

Potassium • Transporters: Signaling molecules • Abiotic stress • Genetic engineering

1 Introduction

The increasing world population makes high yield crop production a necessity in future agriculture. One important factor that will affect agriculture worldwide and that prominent climate scientists have been warning of is the dangerous effects of the continual emission of greenhouse gases into Earth's atmosphere. Atmospheric CO₂ concentration, average temperature, and tropospheric ozone concentration will be higher in the near future. Droughts will be more frequent and severe, more intense precipitation events will lead to increased flooding, some soils will degrade, and climatic extremes will be more likely to occur. Existing abiotic stresses such as salinity will be exacerbated. All these changes will produce an environmental stress for plants and will have many important implications for plant physiology and crop yield. Among the processes that will be altered, K⁺ nutrition deserves special attention. K⁺ is an essential macronutrient that is absorbed from the soil solution by the roots. Environmental stress may reduce K⁺ availability and at the same time may produce physiological alterations that impair K⁺ acquisition. The development of plant varieties with increased capacity to absorb K⁺ may contribute to reducing the negative effects of the abiotic stress conditions derived from climate change. Characterizing the systems involved in the process of K⁺ absorption and how they are regulated and the effect that abiotic stresses will have on their functioning will help to obtain the required new plant varieties (Fedoroff et al. 2010).

2 Role of K⁺ in Plants

 K^+ is an essential macronutrient for plants, composing up to 10% of the total plant dry weight (Taiz and Zeiger 1991) and fulfilling important functions in metabolism, growth, and stress adaptation.

K⁺ functions can be classified into those that rely on high and relatively stable concentrations of the nutrient in certain cellular compartments or tissues and those that rely on its movement between different compartments, cells, or tissues (Amtmann et al. 2004). The first class of functions includes enzyme activation, stabilisation of protein synthesis, neutralization of negative charges on proteins, and maintenance of cytoplasmic pH homeostasis (Marschner 1995). Other roles of K⁺ are linked to its high mobility. This is particularly evident where K⁺ movement is the driving force for osmotic changes - as, for example, in stomatal movement, light-driven and seismonastic movements of organs, and phloem transport. In other cases, K⁺ movement provides a charge balancing counter-flux essential for sustaining the movement of other ions. Thus, sugar, amino acid, and NO3⁻ transport can be accompanied by a flux of K⁺ (Marschner 1995).

The most general phenomenon that requires directed movement of K^+ is growth. Accumulation of K^+ (together with an anion) in plant vacuoles creates the required osmotic potential for cell expansion which relies on its high mobility, and therefore only few other inorganic ions can replace K^+ in this role (Reckmann et al. 1990). Once cell growth has come to a halt, maintenance of osmotic potentials can be carried out by less mobile sugars, and K^+ ions can be partially recovered from vacuoles (Marschner 1995; Poffenroth et al. 1992).

2.1 Range of K⁺ Levels Inside the Plant

As for other nutrients, three ranges in internal K^+ concentrations can be distinguished (Marschner 1995): a deficient, an adequate, and a toxic range. K^+ internal levels found in plants may vary depending on the tissue considered and the

nutritional status of the plant, but they are always in the millimolar range. An optimum K⁺ concentration of around 50-100 mM in the cytosol is required for the performing of the functions mentioned above (Jones 1983). Many of the K+activated enzymes require K⁺ at concentrations between 10 and 50 mM to achieve their maximum activity and plant cells keep the cytosolic K⁺ concentration around this value (Walker et al. 1996). In contrast, vacuolar concentrations are more flexible and mirror the external supply of K⁺. Under K⁺ limiting conditions, cytosolic K⁺ activity can be efficiently maintained around its optimum value due to the release of vacuolar K⁺ ions into the cytosol. Once the electrochemical gradient from the vacuole reaches its limit, cytosolic K⁺ activity starts to decrease (Nieves-Cordones et al. 2008; Walker et al. 1996, 1998).

 K^+ homeostasis is not necessarily effective in all cell types. In barley epidermal leaf cells, cytoplasmic K^+ levels as low as 15 mM have been reported under saline conditions (Cuin et al. 2003). These cells were still alive but they probably showed very low metabolic activity. Indeed, low epidermal K^+ concentrations reflect selective tissue allocation of K^+ within leaves, which allows the plant to protect metabolically active mesophyll cells against K^+ deficiency.

2.2 Problems Related to K⁺ Deficiency

Although current agriculture is based on high levels of fertilization, soils of some intensively cultured lands may became K⁺ deficient (Dobermann et al. 1998; Pal et al. 2001; Rengel and Damon 2008; Yang et al. 2004) because of the withdrawal of K⁺ during crop harvests (Pal et al. 2001) or due to K⁺ losses as a result of lixiviation in sandy soil (Kayser and Isselstein 2005). In other cases, the importance of K⁺ has sometimes been overlooked, and financial constraints have forced farmers to prioritize applications of nitrogen (N) over K⁺ (Armengaud et al. 2009). As a result, a considerable area of farmland has become K⁺ deficient (Andrist-Rangel et al. 2007; Dobermann et al. 1998; Hoa et al. 2006).

K⁺ deficiency entails important effects on plant physiology such as limited cellular expansion and reduction of photosynthesis which ultimately results in growth reduction and development impairment. When external K⁺ is limiting, its translocation from mature leaves and stem is activated. Under conditions of severe deficiency, these organs become chlorotic and even necrotic. At the cellular and metabolic levels, important negative effects are observed. It has been reported that in barley, K⁺-limiting conditions give rise to a decrease in K⁺ activity and pH in the cytosol of root cells (Walker et al. 1996). These reductions positively correlated with a decrease in protein synthesis and a subsequent decline in growth (Walker et al. 1998). At the metabolic level, increase in sugars and depletion of pyruvate has been described in roots and shoots (Armengaud et al. 2009). Other changes also take place to (1)maintain carbon flux into amino acids and proteins, (2) decrease negative metabolic charge, and (3) increase the nitrogen-carbon ratio in amino acids. In addition, K+-deficient plants are more susceptible to abiotic and biotic stresses such as drought, cold, salinity, and fungal attacks (Marschner 1995).

3 K⁺ Uptake Systems

Root K⁺ uptake occurs through specialized epidermal cells (root hairs) which increase the surface in contact with the soil solution as well as through cortical cells. K⁺ must get through root tissues to reach the stele where the xylem vessels will distribute K⁺ to the rest of the plant.

There are two possible parallel pathways that K^+ can follow to reach the stele: the apoplastic and the symplastic pathways. The apoplastic pathway is blocked by the Casparian strip that limits K^+ flux into the stele. Therefore, K^+ needs to enter into the symplast to reach the xylem vessels by crossing the plasma membrane of an epidermal or cortical root cell and through K^+ transport systems of different capacities, with the latter constituting crucial pieces in the control of K^+ acquisition.

3.1 High- and Low-Affinity K⁺ Uptake

In the 1950s, of last century Epstein reported that in barley roots, K⁺ uptake exhibited biphasic kinetics in response to increasing external K⁺ concentrations (Epstein and Hagen 1952; Epstein et al. 1963). The first system showed high-affinity for K⁺ (K_m for K⁺ of 21 µM) and was not inhibited by Na⁺. The second system showed low-affinity for K⁺ (K_m of 11.4 mM) and was inhibited by Na⁺. In maize roots, the low-affinity component was linear (Kochian and Lucas 1982), and traditionally, it has been thought that ion channels are mostly responsible for the low-affinity component (Maathuis and Sanders 1996).

The high-affinity system has been thought to be mediated by transporters because with the reported data on plasma membrane potentials and gradients for K⁺ concentrations for barley and Arabidopsis (Maathuis and Sanders 1993, 1994; Walker et al. 1996), a channel would not be operative for K⁺ influx. A K⁺:H⁺ symport with a 1:1 stoichiometry has been suggested as the transport mechanism. In Neurospora crassa, a fungus which exhibited high-affinity K⁺ uptake similar to that present in plants, the high-affinity K⁺ uptake was shown to be mediated by a K⁺:H⁺ symport (Rodriguez-Navarro et al. 1986). In plant cells, as in fungal cells, the membrane potential is sustained by the H⁺-ATPase, supporting the idea of the plant K+:H+ symport. Other features observed in the high-affinity K⁺ uptake in plants are the upregulation by K⁺ starvation, the lack of discrimination between K⁺ and Rb⁺ and the inhibition by NH_4^+ (Kochian and Lucas 1988; Maathuis and Sanders 1996; Rodríguez-Navarro 2000; Rodríguez-Navarro and Rubio 2006).

3.2 Molecular Entities Mediating Low- and High-Affinity K⁺ Uptake

3.2.1 Shaker Channels

The identification of the genes involved in K⁺ uptake began in the 1990s by using the functional complementation of *Saccharomyces cerevisiae*

mutants defective in the endogenous K⁺ uptake systems *TRK1* and *TRK2*. In 1992, the Arabidopsis *Shaker* K⁺ channels *AKT1* and *KAT1* were isolated (Schachtman 1992; Sentenac et al. 1992). Both AKT1 and KAT1 did not exhibit inactivation through time which suggested that they mediated long-term K⁺ supply (Gaymard et al. 1996; Schachtman 1992). *AtAKT1* expression was preferentially localized in the peripheral cell layers of the root mature regions, which was consistent with a role of AtAKT1 in root K⁺ uptake (Lagarde et al. 1996). However, *AtKAT1* is mainly expressed in leaves discarding its participation in this process (Szyroki et al. 2001).

After the cloning of AtAKT1, several cDNAs encoding K⁺ channels with homology to AKT1 were obtained from other species. For instance, SKT1 from potato (Zimmermann et al. 1998), LKT1 from tomato (Hartje et al. 2000), TaAKT1 from wheat (Buschmann et al. 2000), OsAKT1 from rice (Golldack et al. 2003), DKT1 from carrot (Formentin et al. 2004), CaAKT1 from pepper (Martínez-Cordero et al. 2005), NKT1 from tobacco (Sano et al. 2007), and VvK1.1 from grapevine (Cuellar et al. 2010).

Intriguingly, another cDNA from Arabidopsis, *AtKC1*, encoding a protein with high homology to K⁺ channels was cloned, but it never exhibited ion currents when expressed alone in heterologous expression systems (Reintanz et al. 2002). Later, it was shown that AtKC1 together with the syntaxin SYP121, regulated AtAKT1 function by forming a ternary complex (Honsbein et al. 2009) (see below). Homologues of AtKC1 have also been found in other plant species (Downey et al. 2000; Wang et al. 2002).

3.2.2 HAK/KT/KUP Transporters

High-affinity K⁺ uptake in plants resembles that mediated by SoHAK1, a high-affinity K⁺ transporter from the yeast *Schwanniomyces occidentals*. By using an RT-PCR approach, a plant homolog, HvHAK1, was isolated from barley (Santa-María et al. 1997). Yeast expression of HvHAK1 demonstrated that it was a high-affinity K⁺ transporter that did not discriminate between K⁺ and Rb⁺ and was inhibited by Na⁺ and NH₄⁺, in agreement with the characteristics of high-affinity K⁺ uptake previously described in barley roots. Homologous transporters were later isolated in Arabidopsis (Fu and Luan 1998; Kim et al. 1998; Quintero and Blatt 1997). One of them, AtHAK5, showed a high homology to HvHAK1 and no discrimination between K⁺ and Rb⁺ (Rubio et al. 2000). Phylogenetic analyses show that the HAK family (also named KT or KUP family) of transporters is composed of four clusters and that all high-affinity K⁺ transporters from this family characterized so far belong to Cluster I. These Cluster I transporters are encoded by genes that are rapidly upregulated in root cells when the supply of the external K⁺ is removed (Rodríguez-Navarro and Rubio 2006). When expressed in yeast they mediate K⁺ uptake of similar characteristics to plant high-affinity K⁺ uptake such as low $K_{\rm m}$ values for K⁺ or sensitivity to NH₄⁺, suggesting that they are major components of this system (Rodríguez-Navarro and Rubio 2006).

After AtHAK5 cloning, numerous genes encoding K⁺ transporters of the HAK family were identified in other plant species: 17 genes in rice (Bañuelos et al. 2002), GsKT1 in cotton (Ruan et al. 2001), LeHAK5 in tomato (Wang et al. 2001, 2002), *CnHAK1* and CnHAK2 in Cymodocea nodosa (Garciadeblas et al. 2002), four genes in Mesembrianthemum cristallinum (Su et al. 2002), LjKUP in Lotus japonica (Desbrosses et al. 2004), VvKUP1 and VvKUP2 in Vitis vinifera (Davies et al. 2006), AlHAK in Aeluropus littoralis (Su et al. 2007), five genes in Phragmites australis (Takahashi et al. 2007b, c), CaHAK1 in pepper (Martínez-Cordero et al. 2004), and ThHAK5 in Thellungiella halophila (Alemán et al. 2009b).

3.2.3 HKT Transporters

In 1994, a wheat cDNA, *TaHKT1*, was isolated, and it encoded a protein with homology to the TRK transporters from yeast (Schachtman and Schroeder 1994). Its heterologous expression showed that TaHKT1-mediated high-affinity K⁺ uptake coupled with Na⁺ symport in addition to low-affinity Na⁺ uptake (Rubio et al. 1995). Because Na⁺-activated high-affinity K⁺ uptake has never been demonstrated in higher plants, the involvement of this system in high-affinity K⁺ uptake seems unlikely. Later studies have involved HKT transporters from different plant species to Na⁺ uptake from the external solution and Na⁺ movements within the plant through loading and unloading at the xylem and the phloem (Berthomieu et al. 2003; Garciadeblas et al. 2003; Golldack et al. 2002; Haro et al. 2005; Horie et al. 2007; Laurie et al. 2002; Rus et al. 2001, 2004; Uozumi et al. 2000). This has focused the research on the physiological roles of plant HKT towards adaptation to saline environments rather than K⁺ nutrition.

3.2.4 Cation Proton Antiporters

This superfamily of antiporters is composed of three families: the monovalent cation:proton antiporter-1 (CPA1) family (eight members), the monovalent cation:proton antiporter-2 (CPA2) family also referred to as the CHX family (28 members), and the NhaD family (two members) (Maser et al. 2001; Saier et al. 1999). In plants, Na⁺/H⁺ and K⁺/H⁺ antiporters belonging to the first family (CPA1) are likely critical determinants of salt tolerance and K⁺ homeostasis either by compartmentalization in cellular organelles, for instance NHX transporters, or by extrusion from the cell, for example SOS1 transporters (Apse et al. 1999; Leidi et al. 2010; Pardo et al. 2006; Rodriguez-Rosales et al. 2008; Shi et al. 2003). On the other hand, some members of the CPA2 family have been related to K⁺ acquisition. KEA5, AtCHX17, and AtCHX13 are upregulated in K+-starved roots (Cellier et al. 2004; Shin and Schachtman 2004; Zhao et al. 2008). AtCHX13 was indeed shown to complement a K+ uptake deficient yeast mutant showing a $K_{\rm m}$ $K^+(Rb^+)$ of 136 μM (Zhao et al. 2008), although its expression in the root tip suggested a role in sensing external K⁺ rather than mediating K⁺ uptake.

3.2.5 Cyclic Nucleotide-Gated Channels

Cyclic nucleotide-gated channels (CNGCs) share structural homology with *Shaker* channels, although they lack a GYGD motif (Szczerba et al. 2009; Talke et al. 2003). Published data has shown that their activation is cGMP- and/or cAMP-dependent and that most CNGCs do not discriminate among monovalent cations, have a limited Ca²⁺ permeability, and are blocked by NH_4^+ and external Mg^{2+} (Balague et al. 2003; Demidchik and Maathuis 2007; Leng et al. 2002). A notable exception is AtCNGC2 that exhibited a high degree of K⁺ selectivity as opposed to Na⁺ (Hua et al. 2003), a feature that is unknown in animal CNGCs. In particular, some CNGCs have shown features related to K⁺ uptake. For instance, AtCNGC10 was shown to rescue K⁺ channel mutants of Escherichia coli (LB650), yeast (CY162), and atakt1 Arabidopsis plants (Kaplan et al. 2007). In heterologous systems, AtCNGC3 can function as a Na⁺ and K⁺ uptake pathways (Gobert et al. 2006). Moreover, promoter-driven GUS activity data has shown that, in seedlings, AtCNGC3 is mainly expressed in epidermal and cortical root tissues, something consistent with a role in K⁺ uptake.

3.3 Advances in the Characterization of Entities Mediating Root K⁺ Uptake

Although the characterization in heterologous systems together with gene expression patterns strongly suggested that many of the aforementioned entities participated in root K⁺ uptake, a clear demonstration was still pending. It was necessary to demonstrate the transport activity of such systems in vivo, in their native environment. Furthermore, a subcellular localization compatible with K⁺ uptake (plasma membrane localization) was also required.

The first successful study on this topic demonstrated that an Arabidopsis T-DNA knock-out mutant in AKT1, *atakt1-1*, had reduced plant growth in the presence of 2 mM NH₄⁺ when external K⁺ was below 1 mM (Hirsch et al. 1998). These results suggested that K⁺ channels could also mediate K⁺ uptake in the high-affinity range of concentrations. Further studies showed the presence of two components of K⁺ uptake: an NH₄⁺-insensitive AKT1-mediated component and an NH₄⁺-sensitive component (Spalding et al. 1999). A later study showed that AKT1 fused to GFP localized to the plasma membrane in tobacco mesophyll protoplasts (Hosy et al. 2005). AtAKT1 function is regulated by the AtKC1 subunit, whose gene is strongly expressed in roots (Reintanz et al. 2002) and upregulated by salt stress in the shoot (Pilot et al. 2003). When AtKC1 is coexpressed together with AtAKT1 in heterologous systems, a shift in the activation threshold toward more negative values is observed (Duby et al. 2008; Geiger et al. 2009), probably preventing AKT1 from mediating K⁺ efflux. Recently, it has been shown that AtKC1 forms a ternary complex with AtAKT1 and a syntaxin (SYP121) that mediates K⁺ uptake in Arabidopsis roots (Honsbein et al. 2009). When expressed in Xenopus oocytes, the characteristics observed in this ternary complex are more similar to those observed in native Arabidopsis roots than those recorded without the syntaxin (Honsbein et al. 2009), indicating that *in planta*, the functional unit is the ternary complex.

As for HAK transporters, experiments performed with T-DNA insertion mutants in Arabidopsis demonstrated that root high-affinity K^+ uptake was impaired in the *athak5-1*, athak5-2, and athak5-3 mutants which lacked a functional AtHAK5, (Gierth et al. 2005; Rubio et al. 2008). Moreover, in the athak5 plants, NH_{4}^{+} did not inhibit Rb⁺ uptake, while this cation did so in WT plants, indicating that AtHAK5 mediates the NH4+-sensitive high-affinity K+ uptake in plants, in agreement with the sensitivity to NH₄⁺ that AtHAK5 shows when expressed in yeast (Rubio et al. 2000). Recently, it has been shown that adult athak5-3 plants display lower plant biomass due to reduced K⁺ uptake when they are grown for several weeks at $10 \,\mu M$ K^+ (Nieves-Cordones et al. 2010), evidencing that this transporter supports growth at very low-external K⁺ concentrations. These results are in agreement with another study performed in agarose plates in which AtHAK5 mutants seedlings exhibited reduced root growth when grown below 10 μ M K⁺ (Qi et al. 2008). Moreover, in this study, it was also reported that AtHAK5 fused to an epitope of the human influenza virus hemaglutinin protein (HA epitope) localizes to the plasma membrane.



Fig. 4.1 Molecular entities participating in K⁺ uptake in Arabidopsis roots. The graphic shows the external K⁺ concentrations at which the high- and the low-affinity K⁺ uptake systems operate. In the high-affinity range, AtHAK5 is the only system mediating K⁺ uptake when the external concentration is below 10 μ M. At higher concentrations, AtHAK5, AtAKT1, and AtCHX13 may be the main systems participating in this process. In the low-affinity range of K⁺ concentrations, AtAKT1, AtCHX13, unidentified members of the CNGC or unknown systems could contribute to K⁺ uptake. *Dashed horizontal lines* depict the range of external K⁺ concentrations in which the system just above plays its important role in root K⁺ uptake

As stated above, AtAKT1 also contributes to K⁺ uptake in the high-affinity range of concentrations. Recent studies with single athak5, atakt1 and double athak5, atakt1 mutants in combination with NH⁺ and Ba²⁺ that inhibit AtHAK5 and AtAKT1, respectively, have allowed the establishment of the relative contributions of each of these two systems to K+ uptake from diluted solutions (Rubio et al. 2010). AtHAK5 is the only system-mediating K⁺ uptake at concentrations below 10 µM. Between 10 and 50 µM K⁺, AtHAK5 and AtAKT1 have been demonstrated to be the only systems contributing to K⁺ uptake. Above 50 μ M K⁺, both systems are thought to act, and at concentrations higher than 200 µM, the contribution AtHAK5 decreases and the only system operating is probably AtAKT1, although the contribution of other unknown systems cannot be ruled out (Fig. 4.1).

Recent research has demonstrated that members of the HKT transporter/channel family

mediate important Na⁺-tolerance mechanisms in plants mainly by improving K⁺/Na⁺ homeostasis. Previous work based on the identification of a quantitative trait locus (QTL) determining salt tolerance showed that Kna1 in Triticum aestivum controls the selectivity of Na⁺ and K⁺ transport to shoots, resulting in a high K⁺:Na⁺ ratio in leaves (Dubcovsky et al. 1996; Gorham et al. 1987, 1990; Luo et al. 1996). Other loci identified by QTL analyses in durum wheat, Nax1 and Nax2, also contributed to salt tolerance (Lindsay et al. 2004; Munns et al. 2003). Furthermore, the presence of Nax1 and Nax2 was shown to enhance K⁺ accumulation in leaf blades and sheaths, leading to the model that Nax1 and Nax2 mediate K+loading into the xylem (James et al. 2006). These QTLs are attributable to polymorphisms in copies of wheat HKT genes, TmHKT1;4-A1 and TmHKT1;4-A2 (also named TmHKT7-A1 and -A2) for Nax1 and TmHKT1;5 and TaHKT1;5 for Nax2 and Kna1 (Byrt et al. 2007; Huang et al. 2006). Independent analysis of another QTL in rice, named SKC1 (shoot K⁺ content), resulted in an identical model for the function of the rice gene OsHKT1;5 (Ren 2005). The SKC1 QTL was due to point mutations in OsHKT1;5 that replace several amino acid residues in the salttolerant cultivar NonaBokra (Ren 2005).

Studies performed in different null mutant types of the class I HKT transporter AtHKT1 has shed light into the mechanisms by which this entity controls Na⁺/K⁺ homeostasis under salt stress. According to these studies, AtHKT1 may be involved in Na⁺ recirculation in plants by operating in Na⁺ loading in the phloem (Berthomieu et al. 2003) or removal of Na⁺ from the xylem preventing the accumulation of Na⁺ in the leaves (Davenport et al. 2007; Horie et al. 2006; Sunarpi et al. 2005). In agreement with this model, AtHKT1 overexpression in the root pericycle improves salt tolerance (Moller et al. 2009).

Concerning K⁺ homeostasis, mutations in OsHKT1;5 and AtHKT1 have also been found to reduce K⁺ and enhance Na⁺ accumulation in shoots during salt exposure, contributing to enhanced salinity stress (Ren 2005; Sunarpi et al. 2005). Interestingly, the disruption of AtHKT1 in mutants of the salt overly sensitive (SOS) pathway

92

prevented the K⁺-deficiency symptoms observed in *sos* mutants when grown at low K⁺ concentrations and improved cellular K⁺/Na⁺ ratios when compared with single *sos* mutants under saline conditions (Rus et al. 2001, 2004).

All the results described above refer to HKT transporters belonging to the class I of this family. They have been shown to mediate Na⁺ transport and the effect upon K⁺ nutrition may be indirect. On the other hand, some members of the Class II of HKT transporters can operate as Na⁺-K⁺ symporters under some conditions (Haro et al. 2005; Jabnoune et al. 2009; Rubio et al. 1995), and a contribution to K⁺ uptake may be expected. However, mutations in the OsHKT2;1 gene do not have a strong impact on high-affinity K⁺ (Rb⁺) uptake into intact rice roots (Horie et al. 2007). Many of the presently characterized HKT class two genes are also induced by K⁺ starvation, including those in wheat, barley, and rice (Garciadeblas et al. 2003; Horie et al. 2001; Wang et al. 1998). Therefore, in addition to mediating K⁺ uptake, this system mediates Na⁺ uptake, allowing Na⁺ to act as a substitute nutrient for K⁺ in K⁺-starved rice plants under moderate Na⁺ concentrations (Horie et al. 2007), supporting the long-standing hypothesis that Na⁺ may substitute for K⁺ when this nutrient is scarce (Flowers et al. 1983; Mengel and Kirkby 1982).

Regarding CHX transporters, it has been shown that disruption of AtCHX17 led to lower root K⁺ concentrations under saline and K⁺deprivation conditions, although a subcellular localization and a functional characterization for AtCHX17 remain to be assessed (Cellier et al. 2004). On the other hand, AtCHX13 was localized to the plasma membrane and AtCHX13 knock-out and over-expressing mutant plants showed impairment and enhancement of K⁺ uptake, respectively (Zhao et al. 2008). All these results suggested that this transporter may be involved in K⁺ uptake *in planta*.

With respect to CNGCs, AtCNGC10 is targeted to the plasma membrane, transports both K^+ and Na⁺ and partially rescues Arabidopsis *akt1* mutant plants (Kaplan et al. 2007). Results obtained in antisense lines of this channel, which displayed lower expression than WT plants, indicated a role in Na⁺/K⁺ homeostasis in roots by providing a pathway for Na⁺ influx and K⁺ efflux at the root/soil interface (Guo et al. 2008). Similar results were obtained for AtCNGC3. A null mutation in this channel altered both short-term Na⁺ influx and K⁺ uptake at high-external K⁺ conditions, suggesting an alternate role in nonselective monovalent cation uptake at the plasma membrane level (Gobert et al. 2006).

4 Regulation of the Transport Systems

4.1 Transcriptional Regulation

In general terms, genes encoding AKT1-like K⁺ channels do not show great differences in expression levels in response to the external supply of K⁺ as it has been observed for AtAKT1 and CaAKT1 (Lagarde et al. 1996; Martínez-Cordero et al. 2005; Pilot et al. 2003). However, K⁺ withdrawal from the growth solution increased *TaAKT1* transcript levels (Buschmann et al. 2000), and NaCl treatments or hormonal addition to the external medium produce changes in *AtAKT1* expression (Kaddour et al. 2009; Pilot et al. 2003).

Substantial differences are found in the expression pattern of the genes encoding HAK1-type K^+ transporters in comparison to that observed in the aforementioned genes for K^+ channels. It seems that an important point in the regulation of this type of transporters resides in the control of gene expression.

K⁺-starvation is a common inducer in the gene expression of the HAK1-type K⁺ transporters. This induction has been observed in *HvHAK1* (Santa-María et al. 1997), *AtHAK5* (Ahn et al. 2004; Armengaud et al. 2004; Gierth et al. 2005; Qi et al. 2008; Shin and Schachtman 2004), *OsHAK1* (Bañuelos et al. 2002), *LeHAK5* (Nieves-Cordones et al. 2007; Wang et al. 2002), *CaHAK1* (Martínez-Cordero et al. 2004), and *ThHAK5* (Alemán et al. 2009b). The reduction in root K⁺ concentration below a threshold level has been proposed as the stimulus that could trigger the increase in the transcription of these genes (Martínez-Cordero et al. 2005).
Some other factors have been shown to considerably modify gene transcription of the HAK1type genes. CaHAK1 expression in K⁺-starved plants was reduced, if half of the nitrogen of the growth solution was supplied as NH⁺ (Martínez-Cordero et al. 2005). Similarly, AtHAK5 promoter activity was diminished in K⁺-starved plants after exposure to NH_4^+ (Qi et al. 2008). Conversely, HvHAK1 transcript levels were increased by the presence of NH₄⁺ in the growth solution in K⁺-sufficient plants (Fulgenzi et al. 2008). Intriguingly, *LeHAK5* was downregulated when NO₃⁻ was supplied again after a withdrawal period (Wang et al. 2001). Another case in the regulation of HAK transporters was the increase in AtHAK5 mRNA levels after exposing plants to sucrose in the absence of light (Lejay et al. 2008). Similar results were found previously in Arabidopsis when plants were grown in the sucrose-containing media MS (Rubio et al. 2000): It was observed that in K⁺-sufficient plants AtHAK5 expression was high; in this medium, K⁺ starvation did not further increase the expression levels, probably because K⁺ was substituted for NH_4^+ in these experiments.

Another important factor affecting the expression of these genes seems to be the presence of salinity. This was illustrated, for example, in the decrease of *LeHAK5* (Nieves-Cordones et al. 2007), *AtHAK5*, and *ThHAK5* (Alemán et al. 2009b) transcripts when plants were starved of K^+ in the presence of Na⁺. Importantly, salinity decreased, to a lesser extent, the levels of *ThHAK5* mRNA in *T. halophila* than those of *AtHAK5* transcripts in Arabidopsis.

There is not much information about the signal transduction elements involved in the induction of HAK-type K⁺ transporters expression. In 2004, the important role of reactive oxygen species (ROS) in the signaling events after removing K⁺ from the growth solution was described for Arabidopsis (Shin and Schachtman 2004). Recently, it has been proposed that ethylene signaling acts upstream the increase of ROS during K⁺ deprivation (Jung et al. 2009).

In tomato plants, *LeHAK5* expression levels correlated with steady plasma membrane potentials registered in root cells (Nieves-Cordones et al. 2008) and high-affinity K⁺ uptake (Nieves-Cordones et al. 2007). Changes in plasma membrane potentials are one of the first signals that root cells sense after a stress is applied (Wang and Wu 2010). In tomato roots, the recorded plasma membrane potentials were importantly affected by long-term changes in the composition of the growth solution. For instance, the presence of NH⁺₄ and K⁺ starvation hyperpolarized and Na⁺ depolarized plasma membrane potentials, which produced an increase and a decrease in the *LeHAK5* mRNA levels, respectively. Short-term exposure of depolarizing agents such as CCCP or Vanadate to K⁺-starved roots also decreased LeHAK5 expression. These changes in LeHAK5 expression at both long- and short-term denoted a tight regulation at the transcription level and they also indicated that LeHAK5 contribution to K⁺ uptake could be limited to some specific conditions even if K⁺ deficiency was still present (Nieves-Cordones et al. 2008).

4.2 Post-translational Regulation

Recent studies have gained insights into AtAKT1 regulation. By analyzing mutants sensitive to low K⁺ stress, a CBL-interacting protein kinase, CIPK23, turned out to be essential in the activation of AtAKT1, therefore permitting Arabidopsis plants to grow under low K⁺ conditions (Xu et al. 2006). Two positive regulators of CIPK23, CBL1, and CBL9, which are two Calcineurin B-like proteins, were also found. Both CBL's could phosphorylate CIPK23 which became active after this phosphorylation (Fig. 4.2). Activation of the CBL's depended on Ca^{2+} as demonstrated by patch-clamp recordings (Li et al. 2006). Later on, it was shown that the network was more complex, being CIPK6 and CIPK16, in addition to CIPK23, able to interact with CBL1, CBL2 and CBL3 and CBL9 (Lee et al. 2007). Furthermore, a PP2C phosphatase (AIP1) inactivated AKT1 by dephosphorylating the latter. AIP1 bound AKT1 through AKT1's ankyrin domain. In addition, formation of heterotetramers of AKT1 subunits with AtKC1 also serves as another mode of regulation in which such heterotetramers display different

Low K⁺-CBL а CIPK23 Cytoplasm b AIP1 – High K⁺

Fig. 4.2 Regulation of the activity of AtAKT1 by the CBL-CIPK complex and by the phosphatase AIP1. (a) Low K⁺ stress provokes an increase in ROS levels that can possibly be translated into a cytosolic Ca2+ wave. This wave mainly activates CBL1, CBL9, and CIPK23. The activated CBL/CIPK complex phosphorylates the AtAKT1 channel, resulting in its activation (channel open). (b) At high-external K⁺, the AIP1 phosphatase is thought to dephosphorylate AtAKT1, resulting in the inactivation of the channel (channel closed)

voltage dependence and sensitivity to external K⁺ (Duby et al. 2008; Geiger et al. 2009).

The only report in which the regulation of HAK1-like K⁺ transporters at the protein level was discussed revealed that HvHAK1-mediated Rb⁺ uptake in yeast cells was modulated by the HAL4/5 kinases and the PPZ1 phosphatase (Fulgenzi et al. 2008). Both types of enzymes seemed to negatively regulate HvHAK1 activity in K+-starved yeast cells.

4.3 Signaling Molecules: ROS and Hormones

ROS and hormones represent two of the most important signaling mechanisms in plants. They are produced when stimuli act upon certain types of cells and they constitute an effective means of communication between cells, adjusting plant to the new environmental conditions. In recent years, several hormonal activities and ROS have been related to responses that take place after changes in the K⁺ status or to direct/indirect regulation of K⁺ uptake, suggesting a role in this process. In some cases, significant advances in the mechanisms by which some of these molecules regulate K⁺ nutrition have been reported (Fig. 4.3).

4.3.1 **ROS and Ethylene**

ROS have been shown to accumulate in a discrete region of roots active in K⁺ uptake and they are translocated during low K⁺ stress (Shin and Schachtman 2004). Knock-out of an NADPH oxidase (rhd2 plants) prevented upregulation of genes that are normally induced by K⁺ starvation, such as AtHAK5, but the high-affinity K⁺ uptake remained unchanged. Application of H₂O₂ restored the expression of those genes induced by K⁺ deficiency in *rhd2* plants. Both ethylene production and the transcription of genes involved in ethylene biosynthesis also increased when plants were deprived of K⁺ (Jung et al. 2009; Shin and Schachtman 2004). However, in the ethylene insensitive2-1 (ein2-1) mutant, the ethylenemediated low K⁺ responses were not completely eliminated, suggesting that some K⁺ deprivationinduced responses are either ethylene independent or EIN2 independent. Ethylene signaling stimulated the production of ROS and thereby it seems to constitute an earlier step in low K⁺ response. Nevertheless, elements acting upstream ethylene signaling in the onset of low K⁺ responses are still unknown. These results suggested that K+-dependent regulation of AtHAK5 mRNA levels relied, at least to some extent, on this ethylene-ROS pathway.

Abscissic Acid 4.3.2

Abscissic acid (ABA)-mediated control of ROS levels has been invoked to explain its function in protecting plants against oxidative conditions caused by many stress conditions, including nutritional deficiencies (Rubio et al. 2009). For instance, K⁺ deprivation increases ABA levels both in the root and in the shoot (Kim et al. 2009) that would produce, among the responses of this hormone, a decrease in plant transpiration. Moreover, exposure of Arabidopsis roots to ABA evoked a dramatic reduction in the transcript levels of SKOR, the





Fig. 4.3 Hormonal responses in relation to K⁺ uptake. Root apex is an important source of cytokinins and when it is removed, rapid efflux of K⁺ is observed. Such efflux can be reversed by adding cytokinins. K⁺ transport through TRH1 greatly contributes to auxin gradients in the root and therefore supporting auxin responses such as gravitropic growth and root hair development. Low-external K⁺ concentrations trigger stress-related pathways by increasing ethylene, abscisic acid (ABA), and Jasmonic acid

levels. Ethylene activates reactive oxygen species (ROS) production, which mediates AtHAK5 upregulation and, in turn, high-affinity K⁺ uptake. The ABA increase reduces stomatal aperture in leaves and in roots, it induces SKOR downregulation and thereby decreases K⁺ xylem load. Jasmonic acid production activates several pathways which aim at enhancing nutrient recycling, storage, and allocation processes

outward-rectifier K⁺ channel involved in K⁺ release in the xylem (Gaymard et al. 1998). Interestingly, important reductions in SKOR mRNA levels in the roots were also observed due to K⁺-starvation (Pilot et al. 2003), suggesting that this reduction could be related to the ABA increase under low K⁺ growth conditions.

4.3.3 Jasmonic Acid

Upregulation of the genes controlling the first three steps of jasmonic acid (JA) biosynthesis has been observed in K⁺-deprived plants (Armengaud et al. 2004). In contrast, K⁺ resupply

to these plants downregulated the expression of 19 JA biosynthesis-related genes. Interestingly, these transcriptional responses were observed mainly in the shoot, but not in the root, indicating that there is organ specificity in the K⁺mediated control of JA accumulation. This specificity could be related to the increased nutrient recycling observed in senescent leaves of K⁺-deprived plants, a process that is triggered by JA (He et al. 2002). If we consider the exact position of ethylene and jasmonate signals within the K⁺ starvation response, there is still a gap to be filled.

4.3.4 Auxins

Both physiological and molecular evidences show a relation between K⁺ transport and polar auxin transport. For example, it was observed that Arabidopsis mutant plants that lack activity of the K⁺ transporter TRH1 (AtKUP4/AtKT3) displayed agravitropic root growth and impaired root hair development and that these defects correlated with altered auxin transport or perception, as shown by the fact that addition of auxins rescued these trh1 phenotypes (Vicente-Agullo et al. 2004). K⁺ status may also affect auxin accumulation, as observed by reduced expression of genes controlling auxin biosynthesis, such as CYP79B2 and CYP79B3, when K+-deprived Arabidopsis plants were resupplied with this nutrient (Armengaud et al. 2004). Conversely, a role for auxins in the control of K⁺ homeostasis is supported by the fact that treatment of maize plants with auxins increased expression of the gene encoding the K⁺ transporter ZMK1 (Philippar et al. 1999). Such transcriptional induction is in agreement with the increase in K⁺ transport observed in maize coleoptiles protoplasts upon auxin application. These results contrast with those found in Arabidopsis in which exposure to the synthetic auxin 2,4 D drastically reduced AKT1, AtKC1, and SKOR root mRNA levels (Pilot et al. 2003).

4.3.5 Cytokinins

Cytokinins have long been implicated in the regulation of K⁺ transport in plants (Abutalybov et al. 1980; Abutalybov and Akhundova 1982; Alizade et al. 1988; Green and Muir 1979; Shabala et al. 2009). It has been described that K⁺ release into the xylem (as measured by ⁸⁶Rb⁺ release into root exudates) is inhibited by micromolar concentrations of kinetin (Collins and Kerrigan 1974; Hong and Sucoff 1976; Rains 1969). These observations are in agreement with the important decrease in the SKOR root transcript levels after exposure to benzyladenine (Pilot et al. 2003). In another study (Albacete et al. 2009), it was shown that xylem K^+ (but not Na⁺) concentration was strongly correlated with leaf size and maintenance of the photosynthetic apparatus in tomato under salt stress, as was leaf xylem zeatin concentration, highlighting another interesting cytokinin-K⁺ interaction.

At the root level, cytokinins seemed to play a role in the regulation of K⁺ fluxes in root epidermis (Shabala et al. 2009). Removal of the root apex, an important source of cytokinins, evoked significant K⁺ efflux from root segments that was rapidly reversed by the addition of exogenous kinetin. Regarding regulation of K⁺ transport systems, exposure of roots to the synthetic cytokinin, benzyladenine, produced a rapid decrease in the mRNA levels of the Shaker subunits AKT1 and AtKC1 in addition to the one previously mentioned of SKOR (Pilot et al. 2003). This effect of benzyladenine mirrors that observed after the addition of exogenous auxins. Nonetheless, studies revealing the underlying mechanisms of this interaction cytokinins-K⁺ are lacking.

5 The Arabidopsis Model

5.1 Integration of Uptake Systems

Presently, root K⁺ uptake systems from Arabidopsis are the best characterized and it is possible to integrate all the information available to generate a comprehensive model. Plants show a high concentrative capacity for K⁺ (Martínez-Cordero et al. 2005; Nieves-Cordones et al. 2007), mediated by an active transport mechanism (Maathuis and Sanders 1994) through HAK transporters. As described in a previous section, the disruption of AtHAK5 led to lower K⁺ uptake rates and growth impairment at 10 µM K⁺ and lower K⁺ concentrations (Nieves-Cordones et al. 2010). Together with the micromolar K_m obtained when AtHAK5 was expressed in yeast (Rubio et al. 2000), this suggested that at the lowest K⁺ concentrations, AtHAK5 is the only system mediating root K⁺ uptake, in agreement with the higher K_m values in athak5 plants (Gierth et al. 2005). AtHAK5mediated uptake probably occurs via H⁺–K⁺ symport, as membrane potentials registered in root cells were not negative enough to drive passive flux of K⁺ into root cells. Currently, the mechanism of transport through AtHAK5 remains elusive.

When external K⁺ concentrations are above 10 μ M K⁺, AtAKT1 becomes an active player. In such conditions, the electrochemical gradient is

compatible with K⁺ fluxes through channels and therefore AtAKT1 could mediate K⁺ uptake. Indeed, this idea is further supported by the fact that athak5 mutants did not display a defective phenotype at 20 µM K⁺ (Pyo et al. 2010), exhibited Rb⁺ uptake above that K⁺ concentration (Rubio et al. 2008) and only double mutants atakt1, athak5 lacked any of these features (Pyo et al. 2010; Rubio et al. 2010). These results strongly suggested that the channel was mediating K⁺/Rb⁺ uptake in this context. It is worth noting that in this range of K⁺ concentrations, there is an overlap in the K⁺ uptake capacities between AtHAK5 and AtAKT1 which made the phenotypic study of their corresponding single mutants difficult. AtAKT1 contribution to K+ uptake starts in the low micromolar range, but the upper limit is not well defined. At 1.4 mM K⁺, atakt1 plants exhibited lower plant biomass and lower tissue K⁺ concentrations, including an upregulation of AtHAK5 that reflects clear symptoms of K⁺ deficiency stress (Rubio et al. 2008). Therefore, with growth solutions containing around 1 mM K⁺, the AtAKT1 function was not compensated by other systems such as AtHAK5, and thus AtAKT1mediated uptake seems to be unique. At higher concentrations, for example, 10 mM K⁺, atakt1 mutant plants exhibited normal growth and tissue K⁺ concentrations, denoting that other systems were indeed able to support growth under these conditions (Nieves-Cordones et al. 2010; Rubio et al. 2010). Similar results were obtained in the athak5, atakt1 double mutants showing that AtHAK5 was not important in this range of concentrations. Interestingly, a complete inhibition of K⁺ uptake and plant growth were not observed in the aforementioned double mutant at 0.5 mM K⁺ (Rubio et al. 2010), suggesting that unknown systems are mediating K⁺ nutrition in this athak5, atakt1 line. Candidate systems thought to mediate K⁺ uptake in this mutant line are AtCHX13 (Zhao et al. 2008) or non-selective channels regulated by cyclic nucleotides (CNGC's) (Demidchik and Maathuis 2007; Li et al. 2005).

5.2 Usefulness of This Model to Understand Other Species

Arabidopsis has been proven to be a useful plant model for the understanding of physiological processes in other plant species such as tomato, pepper, and T. halophila (Alemán et al. 2009b; Martínez-Cordero et al. 2005; Nieves-Cordones et al. 2007; Rubio et al. 2008). With this in mind, it has been shown that lower upregulation of HAK transporters due to salt stress when plants are deprived of K⁺ is a common feature observed in barley, tomato, T. halophila, and A. thaliana. In contrast, regulation by NH⁺ of HAK transcript levels was found to be different in Arabidopsis with respect to tomato but similar to barley and pepper. Moreover, CIPK-CBL pathways from Arabidopsis (like that regulating K⁺ uptake through AKT1) are conserved among species and can interact with other Shaker channels from other species such as VvK1.1 from grapevine and ZmK1 from maize (Cuellar et al. 2010; Geiger et al. 2009). This interaction has allowed the production and characterization of functional channels in Xenopus oocytes.

On the other hand, comparing Arabidopsis with monocots species becomes more difficult. For instance, genome complexity in monocots implies that some K⁺ transport systems are found as different isoforms (e.g., HvHAK1A B) that could exhibit differential regulation and contrisomething bution, not yet observed in Arabidopsis. Moreover, HKT transporters belonging to subfamily II are present in monocots and absent in dicots such as Arabidopsis. Therefore, although some general assumptions regarding K⁺ uptake could be valid for monocots, for example, occurrence of high- and low-affinity components with similar properties or regulation of HAK transporters by K⁺, NH₄⁺, and Na⁺, indepth studies addressing the contribution of the different transport systems deserves a monocot model.

6 Future Environmental Stresses Affecting K⁺ Acquisition

Current agriculture faces many environmental abiotic stresses such as drought, salinity, or high temperatures. In the future, climate change will worsen the effects of these stresses. In addition, the continuous over-fertilization of crops to maintain production and the irrigation with low-quality water will also produce negative effects. All these abiotic stress conditions will have an important impact on K^+ acquisition by the plant (Fig. 4.4).

6.1 High CO₂

Stomatal movements and, in turn, plant transpiration are regulated by CO_2 levels in the atmosphere. These elevated atmospheric CO_2 concentrations could induce stomatal closure and thereby reduce plant transpiration. Implications of low transpiration rates in K⁺ uptake were studied in short-term experiments (hours) and it was found that the accumulation of K⁺ by the cells of the root was unaffected by water flux, whereas the passage through the root to the shoot via the vessels was linearly related to it (Bowling and Weatherley 1964). Similarly, K⁺ transport to the shoot was shown to be increasingly reduced at low transpiration rates (high relative humidity) in comparison to normal conditions, whereas K⁺ absorption by the root was less affected by such changes in transpiration (Hooymans 1969).

A long-term study concluded that, in response to sustained exposure to elevated CO₂ concentration, biomass is enhanced by 47% in C3 plants, 21% in CAM plants, and 11% in C4 plants (Poorter and Navas 2003). Generally, elevated CO₂ alters root architecture and fine-root turnover (Tingey et al. 2000) and increases the proportions of fine roots and secondary roots, implying an expansion of the rhizosphere (Curtis et al. 1994; Norby 1994; Pregitzer et al. 1995). Several studies have demonstrated that elevated CO₂ increases the root to shoot ratio (Norby 1994; Rogers et al. 1996; Rogers et al. 1994; Stulen and den Hertog 1993), thereby improving the capacity of the root system to acquire nutrients from the soil. In fact, it was described that after 2 years of exposure to elevated concentrations of CO_{2} (around 700 ppm), an increase in the K⁺, Pi, and N accumulation was observed in every organ in Larix kaempferi, together with an increase in dry matter (Shinano et al. 2007). The enhanced

K* Pi
Ca2*Heavy
MetalsFig. 4.4 Predicted effects of global warming and consequences over K* nutrition. An increase in CO_2 , O_3 , and
temperature is expected, enhancing stomatal closure, and
reducing transpiration. Irrigation with low-quality watermay cau
periods no
imbalance

may cause secondary salinization. Drought and flooding periods may also alter soil moisture triggering nutrient imbalance



uptake of nutrients in plants exposed to elevated CO_2 resulted from increased root biomass rather than increased root activity.

Although, as stated above, elevated CO_2 may have a positive effect on biomass production, important adverse effects also take place. Some plant species could respond to high CO_2 by a downregulation of net photosynthesis rate, which may be attributed to an end-product inhibition by elevated carbohydrate concentration in the leaf, combined N and P deficiency. In addition, photoinhibition, possibly due to the damage of PS core complexes, may also occur (Reddy 2010).

6.2 High Temperature

High temperatures may be the main factor in global change that could decrease crop yield (Battisti and Naylor 2009). In general, increased soil temperature appears to have a more consistent effect on root nutrient uptake kinetics than elevated CO₂ (Bassirirad 2000). A number of studies have demonstrated that changes in soil temperature can directly affect root transport properties for K⁺ (Siddiqi et al. 1984), NH₄⁺ (Chapin et al. 1986), NO_3^- (Bassirirad et al. 1991, 1993; Cumbus and Nye 1982), and PO_4^{3-} (Chapin 1974, 1977; Cumbus and Nye 1985). Temperature sensitive processes, such as root respiration, which affects ion movement across the root, could partially explain the observed changes in nutrient uptake with increasing soil temperature. However, in maize roots, it was observed that the absorption rates for K⁺ increased with temperature but they reached a maximum between 25 and 35°C, whereas root respiration did not reach its optimum even at 40°C (Bravo and Uribe 1981). Interestingly, the enhanced uptake with temperature was observed in both high- and low-affinity components of K⁺ uptake, revealing a general improvement in K⁺ uptake capacity by the root.

6.3 High and Low Soil Moisture

Variations in moisture levels are mainly driven by precipitation, and are also affected by temperature and other factors that determine evapotranspiration rates. As for water supply by rain, climate change may lead to lower rainfall values in some lands, which would ultimately produce drought and episodes of water logging and floods in other areas. These opposite water disorders evoke a series of stress responses in plants. First, drought leads plants to a stand-by state in which general physiological processes such as transpiration and growth are greatly diminished to prevent further water loss. K⁺ homeostasis can also be affected, as K⁺ deficiency symptoms often appear on drought stressed maize, grain sorghum, and soybeans (Wiebold and Scharf 2003). These symptoms may occur even if the soil tests high for K⁺. If root growth is inhibited by dry soil, K⁺ uptake decreases and deficiency symptoms are induced. There are additional problems derived from low water potentials in the soil, like salinity. Water loss from the soil increases salt concentrations (for instance, sodium salts) leading to toxic levels of these salts in the plant and further impairing nutrient homeostasis.

Conversely, soil flooding causes oxygen deficiency (hypoxia), leading to a reduction in adequate soil conditions for plant growth (Ponnamperuma 1972), and adversely affecting nutrient uptake in plants (Pezeshki et al. 1999). The mineral nutrition of plants in response to flooding depends on plant species and soil type (Kozlowski 1984; Pezeshki 2001). In floodintolerant species, the concentrations of N, P, and K⁺ in foliages are often, but not always reduced by flooding (Kozlowski 1984; Pezeshki 1995).

Furthermore, flooding and drought together with extreme temperatures could modify chemical and physical properties of the soil, which can lead to reduced nutrients mobility and absorbance or leaching of some individual nutrients.

The final outcome resulting from the interaction of these climate changes described above on K^+ uptake and nutrition requires further investigation. Nevertheless, it clearly indicates that K^+ uptake is going to be negatively affected and special attention is needed in this issue to ensure plant performance in the field.

6.4 Salinity

Most crops are glycophytes (Greenway and Munns 1980) and have been grown in soils with

low salinity. Thus, the mechanisms developed to absorb, transport, and manage nutrients may not operate as efficiently under salinity as they would under non-saline conditions. Salinity negatively affects plant development and growth and, therefore, reduces crop yield and quality (Laüchli and Epstein 1990).

Over 800 million hectares of land worldwide are affected by salinity (Munns 2005), comprising nearly 7% of the world's total land area and approximately 5% of cultivated land (Amtmann et al. 2004). The expected drought periods that global warming is predicted to cause will entail more water used by farmers. As water resources are becoming more scarce, secondary salinization will be increased as the water quality is likely to be reduced.

There are two types of effects of salt stress on plants: osmotic effect and the specific effect. The latter includes the toxic effect, the nutritional imbalance and the oxidative stress. The osmotic effect results from the reduction of the soil water potential due to salt accumulation. Plant cells respond with osmotic adjustment by synthesizing compatible organic solutes and by accumulating ions from the external environment (Niu et al. 1995). By doing this, plants can revert water flow and permit growth (Kurt et al. 1986). The specific effect depends on the salt species present. The most abundant salt under salinity conditions is NaCl and therefore, the specific effect is mainly derived from the high Na⁺ and Cl⁻ concentrations. Na⁺ specifically affects K⁺ nutrition because Na⁺ and K⁺ share physicochemical properties and Na⁺ competes for the K⁺ binding sites that are essential for the cellular function. More than 50 enzymes are activated by K⁺ and Na⁺ cannot replace its function (Bhandal and Malik 1988). Therefore, K⁺ nutrition under salt stress is greatly impaired.

6.4.1 Salt Stress Affects K⁺ Nutrition

One of the key physiological processes disrupted by high Na⁺ concentrations is the maintenance of cellular and whole-plant K⁺ homeostasis (Cakmak 2005; Flowers et al. 1983; Gaxiola et al. 1992; Kader and Lindberg 2008; Kronzucker et al. 2006; Peng et al. 2004; Rains and Epstein 1967; Santa-María and Epstein 2001; Takahashi et al. 2007a; Zhu et al. 1998). Reductions in K⁺ tissue concentrations can be a consequence of the inhibition of K⁺ uptake by Na⁺ (Alemán et al. 2009a; Kochian et al. 1985; Kronzucker et al. 2006), stimulation of root K⁺ efflux (Lynch and Lauchli 1984; Nieves-Cordones et al. 2010; Shabala et al. 2006) (Cramer et al. 1985; Nassery and Baker 1975), and differential allocation between organs (Graifenberg et al. 1995; Naidoo and Rughunanan 1990). Inhibition of K⁺ uptake by high-external Na⁺ concentrations is a consequence of the competition between K⁺ and Na⁺ for the K⁺ uptake systems. The characteristics of such an inhibition may depend on the conditions and species. Thus, the addition of 3 mM Na⁺ to the growth media resulted in a 50% suppression of K⁺ influx in maize plants (Kochian et al. 1985), and this effect was limited to the lowaffinity transport range for K⁺. In barley, highexternal Na⁺ was shown to inhibit both high- and low-affinity K⁺ uptake (Kronzucker et al. 2006, 2008) and the same has been observed in many other studies (Botella et al. 1997; Fu and Luan 1998; Martínez-Cordero et al. 2005; Rains and Epstein 1967; Rubio et al. 2000; Santa-María et al. 1997). Differences in Na⁺ effects may originate from differences in experimental conditions, but in some cases they are due to differences between species. For example, the inhibition of K⁺ uptake by Na⁺ has been observed in Arabidopsis and in its salt-tolerant relative T. halophila when grown under several saline conditions. However, the inhibition was less intense in T. halophila in comparison to Arabidopsis (Alemán et al. 2009a). The lower reduction of K⁺ uptake in T. halophila was concomitant to lower Na+ absorption rates, whereas the contrary was observed in Arabidopsis. It could be concluded that plant species that differ in salt tolerance may have different K⁺ uptake systems that results in different sensitivities to high-external Na⁺ which may constitute important tools for improving plant K⁺ nutrition efficiency under salinity conditions.

 K^+ efflux from salinity-induced plants is another process that contributes to the reduction of tissue K^+ concentration. The strong membrane depolarization caused by high Na⁺ uptake favors K^+ efflux via depolarization-activated outwardrectifying K^+ channels (Shabala et al. 2006). This is a Na⁺-specific effect because isotonic mannitol solution causes significant membrane hyperpolarization, resulting in increased K⁺ uptake (Chen et al. 2005; Cuin and Shabala 2007; Shabala et al. 2000; Shabala and Lew 2002). In agreement with this, net K⁺ balances in Arabidopsis roots grown for 14 days in the presence of 30 mM NaCl and low-micromolar K⁺ concentrations reflected a prominent K⁺ loss (Nieves-Cordones et al. 2010), although other studies have shown that under steady-state conditions high-external Na⁺ reduced K⁺ efflux (Kronzucker et al. 2008).

Salinity also interferes with K⁺ sensing because high Na⁺ suppresses the induction of high-affinity K⁺ uptake by low K⁺ (Alemán et al. 2009b; Nieves-Cordones et al. 2007, 2010) leading to lower K⁺ uptake capacity under these conditions.

The reduction in tissue K⁺ concentrations due to salinity also leads to a redistribution of this macronutrient to maintain its concentration buffered in metabolic active cells. For example, it has been observed that in leaves of plants grown at 50 mM external NaCl, K⁺ concentrations decreased preferentially in mesophyll cells, whereas at higher salt levels, K⁺ concentrations decreased only in epidermal cells (Fricke et al. 1996). Since K⁺ is more compatible with cellular (i.e., cytoplasmic) processes than Na⁺, K⁺ is preferentially retained in metabolically active (i.e., mesophyll) compared with metabolically less active (i.e., epidermis) tissues or cell compartments (Cuin et al. 2003).

6.5 Effect of Other Nutrients

The increasing world population makes high yield crop production a necessity in agriculture. The use of fertilizers has raised crop yield considerably (Stewart et al. 2005). The expansion of agriculture has lead to an important increase in global K⁺ consumption [4.4% per year between 1999 and 2005] and K⁺ fertilization to maintain crop production is a regular cultural practice. Global K⁺ consumption reached 33.9 Mt K₂O in 2008.

In some cases, over-fertilization occurs, which implies a financial and an environmental cost. Moreover, the high input of fertilizers to crops may lead to the inhibition of K⁺ acquisition because of the presence of high concentrations of other nutrients. It is, therefore, important to optimize the efficiency of fertilizer usage. Cultivating crops that acquire and/or utilize K^+ more efficiently can reduce the use of K^+ -fertilizers which would be environment and economic friendly (White and Brown 2010). Efforts to minimize fertilizer input and to develop nutrient-efficient, high-quality crops rely on detailed understanding of the exact interaction among the nutrients for uptake and the possible deficiency effects of a given nutrient caused by the supply of other nutrients (Amtmann and Armengaud 2009).

6.5.1 NH,+

Interaction between NH_4^+ and K^+ has been studied since long. Several studies have shown that external NH_4^+ decreases K^+ uptake (Deanedrummond and Glass 1983; Kirkby 1968; Martínez-Cordero et al. 2004; Pyo et al. 2010; Rubio et al. 2010), which is probably due to the fact that NH_4^+ and K^+ share some features such as charge value, hydrated ion diameter, and their effect on membrane electric potentials (Wang et al. 1994). In agreement with this, amelioration of NH_4^+ toxicity by K^+ supply has been shown (Cao et al. 1993).

Since the description showing that NH_4^+ specifically inhibits the non-AKT1 pathway of root K⁺ transport (Santa-María et al. 1997; Spalding et al. 1999), some advances have been made in the understanding of the $NH_4^+-K^+$ interaction in K⁺ uptake. The use of Arabidopsis mutants lead to the demonstration that the non-AKT1, NH_4^+ -sensitive component of high-affinity was exclusively mediated by AtHAK5 (Pyo et al. 2010; Rubio et al. 2010).

6.5.2 Cross-Talk in the Responses to K⁺, NH₄⁺, NO₃⁻, and Pi

The development of molecular biology techniques and the availability of plant genome sequences have accelerated the identification of transport systems for all plant nutrients. Now the research focus is moving towards the understanding of the regulation of these systems and cross-talk between them (Ohkama-Ohtsu and Wasaki 2010).

A relationship between K^+ supply and regulation of NO_3^- and NH_4^+ transporters seems to exist. NO_3^- transporters are downregulated in K⁺-starved plants (Armengaud et al. 2004) and NH_4^+ transporters are upregulated by K⁺ deprivation (Maathuis et al. 2003). Short-term (6 h) K⁺ deprivation in maize plants leads to the rapid upregulation of a maize NO₃⁻ transporter NRT2 homolog (Schachtman and Shin 2007), suggesting that factors other than changes in NO₃⁻ metabolism may act on NO₃⁻-sensing mechanisms. Moreover, it has been recently described that K⁺ and NO₃⁻ sensing share a protein kinase in the signaling pathway. As mentioned previously CIPK23 is known to be involved in K⁺ signaling through phosphorylation of the AKT1 channel (Krouk et al. 2010), and CIPK23 is also an NO₃⁻-inducible gene which is downregulated in *chl1 (nrt1.1)* mutants.

Regulation of Pi and K⁺ transporters may also share signaling elements. Sucrose supply induce the Pi transporters, PHT1;4 and PHT3;1, as well as the K⁺ transporter HAK5 upstream of hexokinase (HXK) sugar sensing pathways (Lejay et al. 2008). Other NO₃⁻, SO₄²⁻, and K⁺ transporters are also upregulated by sucrose but downstream of the HXK sugar sensing pathway. On the other hand, genes encoding an MAP kinases, transcription factors, and nutrient transporters are induced by K⁺ and Pi deprivation (Wang et al. 2002).

ROS may be a common signaling element of the plant response to the deficiency of several nutrients. ROS in roots is a common feature in response to NO_3^- , Pi, K⁺, and SO_4^{2-} deprivation. Although this response occurs because of the deprivation of several macronutrients, it appears that there are some differences in localization as well as differences in the molecules that produce the ROS (Schachtman and Shin 2007), thus allowing some specificity.

7 Biotechnological Perspectives

To improve crop productivity, it is necessary to understand the mechanisms of plant responses to environmental changes with the ultimate goal being the increase of food availability. There are concerns about our ability to increase, or even maintain, crop yield and production in the context of global environmental change and its associated abiotic stresses (Tester and Langridge 2010). Furthermore, the current increment in biofuels production adds more doubts about our capacity to produce enough food.

The correlation between the increased frequency of extreme environmental events and global warming, underlie an urgent need for protective measures including the development and introduction of new crop cultivars with enhanced tolerance to environmental stresses (Etterson and Shaw 2001; Mittler 2006). Different strategies could be carried out for this purpose, and each of them will present certain advantages and inconveniences that need to be taken into consideration. Two of these strategies, natural variability exploitation and genetic engineering, that could be used for the improvement of the K⁺ uptake systems in plant roots, will be discussed below.

7.1 Natural Variability and QTL Mapping

Natural variability offers a large resource of polymorphism, which is often explored to identify traits with environmental adaptive value or quality properties. Responses to environmental conditions depend on numerous genes and are typically controlled by QTLs. Genomic mapping of such QTLs may lead to the identification and cloning of important regulatory genes or allelic variants. It could also provide genetic markers for molecular breeding and/or cloned genes for genetic engineering for the improvement of stress tolerance in plants (Papdi et al. 2009).

Natural variability is often based on minor genetic changes, generating small quantitative alterations in responses to environmental conditions. One single nucleotide change, so-called single nucleotide polymorphism (SNP), could lead to different plant yield (Fleury et al. 2010; Papdi et al. 2009; Rafalski 2002; Xing and Zhang 2010).

The identification of genetic variability which affects stress responses requires phenotypic screenings capable of distinguishing between plants with different stress tolerance. Although previously tedious and time-consuming, nextgeneration sequencing of natural accessions can reveal sequence variability at the genome scale and will facilitate the large-scale identification of SNPs in different ecotypes and varieties. This is one of the goals of the "1001 Genomes Project," spearheaded by Magnus Nordborg, Joe Ecker,



Fig. 4.5 Proposed points of genetic engineering for obtaining plants with improved K^+ nutrition: Upregulation of known root K^+ uptake transporters (HAK5) and channels (AKT1); selective upregulation

and Detlef Weigel among others, http:// 1001genomes.org/where 617 accessions have been committed as of 2010-6-2.

There are many examples of successful uses of genetic markers and QTL mapping. ERECTA was the first Arabidopsis gene that was mapped as a main QTL, and it regulates transpiration efficiency by controlling leaf photosynthesis efficiency and stomatal conductance (Masle et al. 2005). Freezing tolerance is controlled by seven QTLs in Arabidopsis. QTL mapping revealed that the C-repeat binding factor (CBF) locus is the most important component in cold acclimation (Alonso-Blanco et al. 2005).

As mentioned above, the identification of QTLs for determining salt tolerance or K⁺ accumulation have highlighted the importance of HKT transporters in these processes. Highthroughput ionomic coupled with genomic analysis allowed the identification of the genetic alteration that drives the natural variation in shoot Na⁺ accumulation in Arabidopsis populations (Rus et al. 2006). Polymorphism of the AtHKT1 gene and sensitive wild populations of Arabidopsis illustrate the importance of this transporter in salt tolerance. Other examples are the already mentioned SKC1, a rice HKT-type Na⁺-selective transporter involved in unloading Na⁺ from the xylem, characterized as a QTL for salt tolerance (Ren 2005); or the durum wheat Nax1 and Nax2 loci, linked to Na⁺ exclusion which correspond to the Na⁺ transporters HKT1;4 (HKT7) and

of Na⁺/K⁺ transporters (HKT); upregulation of unknown K⁺ transporters (CPAs, CNGCs). In addition, protein activity and selectivity may be also modified to enhance K⁺ uptake

HKT1;5 (HKT8), respectively (Byrt et al. 2007; Huang et al. 2006). Recently, RAS1 (Response to ABA and Salt 1) were found by using QTL mapping of a recombinant inbred population derived from Landsberg erecta (Ler; salt and ABA sensitive) × Shakdara (Sha; salt and ABA resistant). This transcription factor have been shown to play an important role in salt tolerance and ABA sensitivity (Ren et al. 2010).

As we can see, the analysis of natural variation in crop plants and Arabidopsis has provided an unprecedented amount of information on the genetic and molecular mechanisms that determine intraspecific variation and adaptation. It can be anticipated that this trend will continue in the next decade, especially with the broad implementation of "-omics" technologies for the precise analysis of natural variation at different levels (Alonso-Blanco et al. 2009).

7.2 Genetic Engineering

Two main strategies can be developed by researchers to attain biotechnologically improved plants: Modifying the protein quantity of interest or modifying the protein quality to make it more selective or active. These two strategies may be achieved with modifications at the nucleic acid level that will encode the protein. These strategies may be applied to K^+ uptake systems to improve K^+ nutrition (Fig. 4.5).

7.2.1 Modification of Gene Expression

Although all cells in an organism contain essentially the same DNA, cell types and cell functions vary because of qualitative and quantitative differences in their gene expression. Overexpression and downregulation of key genes will provide plants with the necessary resources to grow properly in the global warming challenge. The nutrition of K⁺, one of the most important macronutrients presently found in most commercial fertilizers, could be improved in this way. Thus, enhanced expression of K⁺ transporters and channels could result in an increase in plant K⁺ uptake which could lead to high-yield production. At the same time, improved K⁺ nutrition could lead to enhanced salt tolerance, because of the aforementioned competition between K⁺ and Na⁺, or to alleviation of drought stress as K⁺ is a known osmolite (Amtmann et al. 2004; Römheld and Kirkby 2010).

To our knowledge, only two plasma membrane K^+ or channels have transporters been overexpressed. Overexpression of the barley high-affinity K+ transporter HvHAK1 in Arabidopsis transgenic plants has been characterized (Fulgenzi et al. 2008). They showed increased K⁺ uptake when plants were deprived of K⁺ but not in K⁺-sufficient conditions. Also a correlation between HvHAK1 transcript level and K⁺ uptake was absent. The cation:proton antiporter AtCHX13 has also been overexpressed (Zhao et al. 2008). Driven by the 35S promoter, AtCHX13 increases K⁺ (Rb⁺) uptake in intact plants with a K_m of 196 μ M, although it is not clear whether AtCHX13 localizes to the plasma membrane of the root cortex in native tissue. More K⁺ transporters and/or channels have to be upregulated to unravel their biotechnological potential. Biotechnological success has been recently reported by cell-specific overexpression of an HKT transporter. Specific expression of AtHKT1 in the mature root stele leads to a reduction of root-to-shoot transfer of Na+ and increased salinity tolerance (Moller et al. 2009). Interestingly, not only Arabidopsis but also rice plants exhibit salinity tolerance with the specific expression of AtHKT1 in root cortex cells (Plett et al. 2010).

7.2.2 Modification of Protein Activity

Future modification of K⁺ transporters and channels may contribute to K⁺ nutrition under adverse conditions such as high-external concentrations of K⁺ uptake inhibitors. The finding of transporters and channels with higher affinity for K^+ (lower K_m) and/or higher transport capacity (higher V_{max}) could lead to obtaining plants better suited to cope with the problems derived from climate change. Some advances have been made for this purpose and several random and sitedirected mutagenesis have been developed. S. cerevisiae, as a screening system model, has allowed the identification of modifications in the gene sequence encoding the transporters that confer salt tolerance or enhanced K⁺ uptake (Garciadeblas et al. 2007; Mangano et al. 2008; Rubio et al. 1995, 1999). None of these studies have shown whether the enhanced yeast growth was due to a higher V_{max} or to an increase of transporter quantity at the plasma membrane. Only Garciadeblas and colleagues (2007) have reported on a mutated transporter of the moss Physcomitrela patens (PpHAK1), with a lower $K_{\rm m}$ for K⁺, but still high for high-affinity K⁺ uptake (200 μ M).

Although good candidates for increasing K⁺ uptake in salt environments have been described for some K⁺ transporters, no biotechnological advances in plants have been made to date (or to our knowledge). Promising candidates would finally confirm *in planta* the hypothesis that the K⁺/Na⁺ ratio and not absolute Na⁺ concentration in the cytoplasm is critical for stress tolerance (Amtmann et al. 2004; Maathuis and Amtmann 1999) and that by modifying root K⁺ uptake systems salt tolerance can be increased. In addition, the possibility that improved K⁺ uptake ameliorates the plant response to other stresses such as drought and cold should be also considered.

8 Conclusion and Future Perspective

The development of new crop cultivars with enhanced tolerance to environmental stresses such as salinity, heat, etc., is a necessity for modern agriculture, as these stresses will be exacerbated as a consequence of future climate change. One of the targets for improvement is the plant's capacity to maintain/control its mineral nutrition. As regards to the acquisition of K⁺, the research developed in the past decades has allowed the identification of key pieces to this process, namely the identification of genes encoding the transport systems, the regulation of these genes and the elements involved in modulating the activities of these transport systems. Relevant transcription factors and promoter regions will hopefully be identified in the near future, which in turn will constitute objectives and targets of further research. Another set of targets in need of study are composed by the regulatory proteins that have been shown to interact with structural proteins involved in K⁺ acquisition. In this sense, elements of Ca²⁺ signaling pathways such as kinases, phosphatases, and Ca²⁺ binding proteins are of special importance. In addition, key players involved in hormone and ROS homeostasis should be also considered. The development of genetic engineering tools, the -omics approaches and the exploitation of natural variability should produce advances in the understanding of the process of K⁺ acquisition that could lead to the development of improved crop varieties better suited for facing future challenges.

Acknowledgements This work was supported by grants 08696/PI/08 from Fundación Séneca of Region de Murcia (Spain) and AGL2009-08140 form MICINN of Spain to FR.

References

- Abutalybov MG, Akhundova TS (1982) Cytokinin participation in regulation of potassium-ion activity in root epidermal-cells. Soviet Plant Physiol 29:395–402
- Abutalybov M, Melnikov P, Mardanov A, Achundova T, Goring H (1980) Influence of kinetin on the membranepotential and potassium activity in root-cells of trianeabogotensis. Biochem Physiol Pflanz 175:529–536
- Ahn SJ, Shin R, Schachtman DP (2004) Expression of KT/KUP genes in Arabidopsis and the role of root hairs in K⁺ uptake. Plant Physiol 134:1135–1145
- Albacete A, Martinez-Andujar C, Ghanem ME, Acosta M, Sanchez-Bravo J, Asins MJ, Cuartero J, Lutts S, Dodd IC, Perez-Alfocea F (2009) Rootstock-mediated

changes in xylem ionic and hormonal status are correlated with delayed leaf senescence, and increased leaf area and crop productivity in salinized tomato. Plant Cell Environ 32:928–938

- Alemán F, Nieves-Cordones M, Martínez V, Rubio F (2009a) Potassium/sodium steady-state homeostasis in *Thellungiella halophila* and *Arabidopsis thaliana* under long-term salinity conditions. Plant Sci 176:768–774
- Alemán F, Nieves-Cordones M, Martinez V, Rubio F (2009b) Differential regulation of the hak5 genes encoding the high-affinity K⁺ transporters of *Thellungiella halophila* and *Arabidopsis thaliana*. Environ Exp Bot 65:263–269
- Alizade VM, Akhundova TS, Alieva FK (1988) Role of kinetin in regulation of potassium-transport motive forces in root epidermal-cells. Soviet Plant Physiol 35:504–509
- Alonso-Blanco C, Gomez-Mena C, Llorente F, Koornneef M, Salinas J, Martinez-Zapater JM (2005) Genetic and molecular analyses of natural variation indicate *CBF2* as a candidate gene for underlying a freezing tolerance quantitative trait locus in arabidopsis. Plant Physiol 139:1304–1312
- Alonso-Blanco C, Aarts MGM, Bentsink L, Keurentjes JJB, Reymond M, Vreugdenhil D, Koornneef M (2009) What has natural variation taught us about plant development, physiology, and adaptation? Plant Cell 21:1877–1896
- Amtmann A, Armengaud P (2009) Effects of N, P, K and s on metabolism: New knowledge gained from multilevel analysis. Curr Opin Plant Biol 12:275–283
- Amtmann A, Armengaud P, Volkov V, Michael RB (2004) Potassium nutrition and salt stress. Membrane transport in plants. Blackwell, Oxford, pp 293–339
- Andrist-Rangel Y, Edwards AC, Hillier S, Öborn I (2007) Long-term K dynamics in organic and conventional mixed cropping systems as related to management and soil properties. Agric Ecosyst Environ 122:413–426
- Apse MP, Aharon GS, Snedden WA, Blumwald E (1999) Salt tolerance conferred by overexpression of a vacuolar Na⁺/K⁺ antiport in arabidopsis. Science 285:1256–1258
- Armengaud P, Breitling R, Amtmann A (2004) The potassium-dependent transcriptome of arabidopsis reveals a prominent role of jasmonic acid in nutrient signaling. Plant Physiol 136:2556–2576
- Armengaud P, Sulpice R, Miller AJ, Stitt M, Amtmann A, Gibon Y (2009) Multilevel analysis of primary metabolism provides new insights into the role of potassium nutrition for glycolysis and nitrogen assimilation in arabidopsis roots. Plant Physiol 150:772–785
- Balague C, Lin BQ, Alcon C, Flottes G, Malmstrom S, Kohler C, Neuhaus G, Pelletier G, Gaymard F, Roby D (2003) HLM1, an essential signaling component in the hypersensitive response, is a member of the cyclic nucleotide-gated channel ion channel family. Plant Cell 15:365–379
- Bañuelos MA, Garciadeblas B, Cubero B, Rodriguez-Navarro A (2002) Inventory and functional characterization of the hak potassium transporters of rice. Plant Physiol 130:784–795

- Bassirirad H (2000) Kinetics of nutrient uptake by roots: responses to global change. New Phytol 147:155–169
- Bassirirad H, Radin JW, Matsuda K (1991) Temperaturedependent water and ion-transport properties of barley and sorghum roots.1. Relationship to leaf growth. Plant Physiol 97:426–432
- Bassirirad H, Caldwell MM, Bilbrough C (1993) Effects of soil-temperature and nitrogen status on kinetics of ¹⁵NO₃ – uptake by roots of field-grown *Agropyron desertorum* (Fisch ex Link) Schult. New Phytol 123: 485–489
- Battisti DS, Naylor RL (2009) Historical warnings of future food insecurity with unprecedented seasonal heat. Science 323:240–244
- Berthomieu P, Conejero G, Nublat A, Brackenbury WJ, Lambert C, Savio C, Uozumi N, Oiki S, Yamada K, Cellier F, Gosti F, Simonneau T, Essah PA, Tester M, Very AA, Sentenac H, Casse F (2003) Functional analysis of athkt1 in arabidopsis shows that Na⁺ recirculation by the phloem is crucial for salt tolerance. EMBO J 22:2004–2014
- Bhandal IS, Malik CP (1988) Potassium estimation, uptake, and its role in the physiology and metabolism of flowering plants. Int Rev Cytol 110:205–254
- Botella MA, Martinez V, Nieves M, Cerda A (1997) Effect of salinity on the growth and nitrogen uptake by wheat seedlings. J Plant Nutr 20:793–804
- Bowling DJF, Weatherley PE (1964) Potassium uptake transport in roots of *Ricinus communis*. J Exp Bot 15:413
- Bravo P, Uribe EG (1981) Temperature-dependence of the concentration kinetics of absorption of phosphate and potassium in corn roots. Plant Physiol 67:815–819
- Buschmann PH, Vaidyanathan R, Gassmann W, Schroeder JI (2000) Enhancement of Na⁺ uptake currents, timedependent inward-rectifying K⁺ channel currents, and K⁺ channel transcripts by K⁺ starvation in wheat root cells. Plant Physiol 122:1387–1398
- Byrt CS, Platten JD, Spielmeyer W, James RA, Lagudah ES, Dennis ES, Tester M, Munns R (2007) HKT1;5-like cation transporters linked to Na⁺ exclusion loci in wheat, NAX2 and KNA1. Plant Physiol 143:1918–1928
- Cakmak I (2005) The role of potassium in alleviating detrimental effects of abiotic stresses in plants. J Plant Nutr Soil Sci 168:521–530
- Cao Y, Glass AD, Crawford NM (1993) Ammonium inhibition of arabidopsis root growth can be reversed by potassium and by auxin resistance mutations aux1, axr1, and axr2. Plant Physiol 102:983–989
- Cellier F, Conejero G, Ricaud L, Luu DT, Lepetit M, Gosti F, Casse F (2004) Characterization of atchx17, a member of the cation/H⁺ exchangers, CHX family, from *Arabidopsis thaliana* suggests a role in K⁺ homeostasis. Plant J 39:834–846
- Chapin FS (1974) Morphological and physiological mechanisms of temperature compensation in phosphate absorption along a latitudinal gradient. Ecology 55:1180–1198
- Chapin FS (1977) Temperature compensation in phosphate absorption occurring over diverse time scales. Arctic Alp Res 9:139–148

- Chapin FS, Vancleve K, Tryon PR (1986) Relationship of ion absorption to growth-rate in taiga trees. Oecologia 69:238–242
- Chen Z, Newman I, Zhou M, Mendham N, Zhang G, Shabala S (2005) Screening plants for salt tolerance by measuring K⁺ flux: a case study for barley. Plant Cell Environ 28:1230–1246
- Collins JC, Kerrigan AP (1974) Effect of kinetin and abscisic-acid on water and ion-transport in isolated maize roots. New Phytol 73:309–314
- Cramer GR, Lauchli A, Polito VS (1985) Displacement of Ca²⁺ by Na⁺ from the plasmalemma of root cells: a primary response to salt stress? Plant Physiol 79:207–211
- Cuellar T, Pascaud F, Verdeil JL, Torregrosa L, dam-Blondon AF, Thibaud JB, Sentenac H, Gaillard I (2010) A grapevine shaker inward K⁺ channel activated by the calcineurin b-like calcium sensor 1-protein kinase CIPK3 network is expressed in grape berries under drought stress conditions. Plant J 61:58–69
- Cuin TA, Shabala S (2007) Compatible solutes reduce ros-induced potassium efflux in arabidopsis roots. Plant Cell Environ 30:875–85
- Cuin TA, Miller AJ, Laurie SA, Leigh RA (2003) Potassium activities in cell compartments of saltgrown barley leaves. J Exp Bot 54:657–661
- Cumbus IP, Nye PH (1982) Root zone temperature effects on growth and nitrate absorption in rape (Brassicanapus cv. Emerald). J Exp Bot 33:1138–1146
- Cumbus IP, Nye PH (1985) Root zone temperature effects on growth and phosphate absorption in rape Brassicanapus cv. Emerald. J Exp Bot 36:219–227
- Curtis PS, Oneill EG, Teeri JA, Zak DR, Pregitzer KS (1994) Belowground responses to rising atmospheric CO₂ – implications for plants, soil biota and ecosystem processes – executive summary. Plant Soil 165:1–6
- Davenport RJ, Muñoz-Mayor A, Jha D, Essah PA, Rus A, Tester M (2007) The Na⁺ transporter AtHKT1;1 controls retrieval of Na⁺ from the xylem in arabidopsis. Plant Cell Environ 30:497–507
- Davies C, Shin R, Liu W, Thomas MR, Schachtman DP (2006) Transporters expressed during grape berry (Vitis vinifera L.) development are associated with an increase in berry size and berry potassium accumulation. J Exp Bot 57:3209–3216
- Deanedrummond CE, Glass ADM (1983) Short-term studies of nitrate uptake into barley plants using ionspecific electrodes and ³⁶ClO₃⁻¹ II. Regulation of NO₃⁻ efflux by NH₄⁺. Plant Physiol 73:105–110
- Demidchik V, Maathuis F (2007) Physiological roles of nonselective cation channels in plants: from salt stress to signalling and development. New Phytol 175:387–404
- Desbrosses G, Kopka C, Ott T, Udvardi MK (2004) Lotus Japonicus LjKUP is induced late during nodule development and encodes a potassium transporter of the plasma membrane. Mol Plant Microbe Interact 17:789–797
- Dobermann A, Cassman KG, Mamaril CP, Sheehy JE (1998) Management of phosphorus, potassium, and sulfur in intensive, irrigated lowland rice. Field Crops Res 56:113–138
- Downey P, Szabo I, Ivashikina N, Negro A, Guzzo F, Ache P, Hedrich R, Terzi M, Schiavo FL (2000) Kdc1,

a novel carrot root hair K^+ channel. Cloning, characterization, and expression in mammalian cells. J Biol Chem 275:39420–39426

- Dubcovsky J, Santa-Maria GE, Epstein E, Luo MC, Dvor kJ (1996) Mapping of the K⁺/Na⁺ discriminaton locus Kna1 in wheat. Theor Appl Genet 92:448–454
- Duby G, Hosy E, Fizames C, Alcon C, Costa A, Sentenac H, Thibaud JB (2008) AtKC1, a conditionally targeted shaker-type subunit, regulates the activity of plant K⁺ channels. Plant J 53:115–123
- Epstein E, Hagen CE (1952) A kinetic study of the absorption of alkali cations by barley roots. Plant Physiol 27:457–474
- Epstein E, Rains DW, Elzam OE (1963) Resolution of dual mechanisms of potassium absorption by barley roots. Proc Natl Acad Sci USA 49:684–692
- Etterson JR, Shaw RG (2001) Constraint to adaptive evolution in response to global warming. Science 294:151–154
- Fedoroff NV, Battisti DS, Beachy RN, Cooper PJM, Fischhoff DA, Hodges CN, Knauf VC, Kobell D, Mazur BJ, Molden D, Reynolds MP, Ronald PC, Rosegrant W, Sanchez PA, Vonshak A, Zhu JK (2010) Radically rethinking agriculture for the 21st century. Science 327:833–834
- Fleury D, Jefferies S, Kuchel H, Langridge P (2010) Genetic and genomic tools to improve drought tolerance in wheat. J Exp Bot 61:3211–3222
- Flowers TJ, Lauchli A, Bieleski RL (1983) Sodium versus potassium: substitution and compartmentation. In: Pirson A, Zimmermann MH (eds) Inorganic plant nutrition. Springer, Berlin, pp 651–681
- Formentin E, Varotto S, Costa A, Downey P, Bregante M, Naso A, Picco C, Gambale F, Lo SF (2004) DKT1, a novel K⁺ channel from carrot, forms functional heteromeric channels with KDC1. FEBS Lett 573:61–67
- Fricke W, Leigh RA, Tomos AD (1996) The intercellular distribution of vacuolar solutes in the epidermis and mesophyll of barley leaves changes in response to NaCl. J Exp Bot 47:1413–1426
- Fu HH, Luan S (1998) Atkup1: a dual-affinity K⁺ transporter from arabidopsis. Plant Cell 10:63–73
- Fulgenzi FR, Peralta ML, Mangano S, Danna CH, Vallejo AJ, Puigdomenech P, Santa-Maria GE (2008) The ionic environment controls the contribution of the barley HvHAK1 transporter to potassium acquisition. Plant Physiol 147:252–262
- Garciadeblas B, Benito B, Rodriguez-Navarro A (2002) Molecular cloning and functional expression in bacteria of the potassium transporters CnHAK1 and CnHAK2 of the seagrass *Cymodocea nodosa*. Plant Mol Biol 50:623–633
- Garciadeblas B, Senn ME, Bañuelos MA, Rodriguez-Navarro A (2003) Sodium transport and HKT transporters: the rice model. Plant J 34:788–801
- Garciadeblas B, Barrero-Gil J, Benito B, Rodriguez-Navarro A (2007) Potassium transport systems in the moss *Physcomitrella patens*: PpHAK1 plants reveal the complexity of potassium uptake. Plant J 52:1080–1093
- Gaxiola R, Delarrinoa IF, Villalba JM, Serrano R (1992) A novel and conserved salt-induced protein is an

important determinant of salt tolerance in yeast. EMBO J 11:3157-3164

- Gaymard F, Cerutti M, Horeau C, Lemaillet G, Urbach S, Ravallec M, Devauchelle G, Sentenac H, Thibaud JB (1996) The baculovirus/insect cell system as an alternative to xenopus oocytes. First characterization of the AKT1 K⁺ channel from *Arabidopsis thaliana*. J Biol Chem 271:22863–22870
- Gaymard F, Pilot G, Lacombe B, Bouchez D, Bruneau D, Boucherez J, Michaux-Ferriere N, Thibaud JB, Sentenac H (1998) Identification and disruption of a plant shaker-like outward channel involved in K⁺ release into the xylem sap. Cell 94:647–655
- Geiger D, Becker D, Vosloh D, Gambale F, Palme K, Rehers M, Anschuetz U, Dreyer I, Kudla J, Hedrich R (2009) Heteromeric AtKC1/AKT1 channels in arabidopsis roots facilitate growth under K⁺ limiting conditions. J Biol Chem 284:21288–21295
- Gierth M, Maser P, Schroeder JI (2005) The potassium transporter AtHAK5 functions in K⁺ deprivationinduced high-affinity K⁺ uptake and akt1 K⁺ channel contribution to K⁺ uptake kinetics in arabidopsis roots. Plant Physiol 137:1105–1114
- Gobert A, Park G, Amtmann A, Sanders D, Maathuis FJM (2006) Arabidopsis thaliana cyclic nucleotide-gated channel 3 forms a non-selective ion transporter involved in germination and cation transport. J Exp Bot 57:791–800
- Golldack D, Su H, Quigley F, Kamasani UR, Munoz-Garay C, Balderas E, Popova OV, Bennett J, Bohnert HJ, Pantoja O (2002) Characterization of a HKT-type transporter in rice as a general alkali cation transporter. Plant J 31:529–542
- Golldack D, Quigley F, Michalowski CB, Kamasani UR, Bohnert HJ (2003) Salinity stress-tolerant and -sensitive rice (oryza sativa l.) regulate AKT1-type potassium channel transcripts differently. Plant Mol Biol 51:71–81
- Gorham J, Hardy C, Wyn Jones RG, Joppa LR, Law CN (1987) Chromosomal location of the Kna discriminating character in the d genome of wheat. Theor Appl Genet 74:584–588
- Gorham J, Wyn Jones RG, Bristol A (1990) Partial characterization of the trait for enhanced K⁺-Na⁺ discrimination in the d genome of wheat. Planta 180:590–597
- Graifenberg A, Giustiniani L, Temperini O, di Paola ML (1995) Allocation of Na, Cl, K and Ca within plant tissues in globe artichoke (Cynara scolimus l.) under saline-sodic conditions. Sci Hort 63:1–10
- Green JF, Muir RM (1979) Analysis of the role of potassium in the growth effects of cytokinin, light and abscisic-acid on cotyledon expansion. Physiol Plant 46:19–24
- Greenway H, Munns R (1980) Mechanisms of salt tolerance in non-halophytes. Annu Rev Plant Physiol Plant Mol Biol 31:149–190
- Guo K, Babourina O, Christopher D, Borsics T, Rengel Z (2008) The cyclic nucleotide-gated channel, AtCNGC10, influences salt tolerance in Arabidopsis. Physiol Plant 134:499–507
- Haro R, Banuelos MA, Senn ME, Barrero-Gil J, Rodriguez-Navarro A (2005) HKT1 mediates sodium

uniport in roots. Pitfalls in the expression of HKT1 in yeast. Plant Physiol 139:1495–1506

- Hartje S, Zimmermann S, Klonus D, Mueller-Roeber B (2000) Functional characterisation of LKT1, a K⁺ uptake channel from tomato root hairs, and comparison with the closely related potato inwardly rectifying K⁺ channel SKT1 after expression in xenopus oocytes. Planta 210:723–731
- He YH, Fukushige H, Hildebrand DF, Gan SS (2002) Evidence supporting a role of jasmonic acid in arabidopsis leaf senescence. Plant Physiol 128:876–884
- Hirsch RE, Lewis BD, Spalding EP, Sussman MR (1998) A role for the AKT1 potassium channel in plant nutrition. Science 280:918–921
- Hoa NM, Janssen BH, Oenema O, Dobermann A (2006) Comparison of partial and complete soil K budgets under intensive rice cropping in the mekong delta, vietnam. Agric Ecosyst Environ 116:121–131
- Hong SG, Sucoff E (1976) Effects of kinetin and root tip removal on exudation and potassium (rubidium) transport in roots of honey locust. Plant Physiol 57:230–236
- Honsbein A, Sokolovski S, Grefen C, Campanoni P, Pratelli R, Paneque M, Chen Z, Johansson I, Blatt MR (2009) A tripartite snare-K⁺ channel complex mediates in channel-dependent K⁺ nutrition in arabidopsis. Plant Cell 21:2859–2877
- Hooymans JJM (1969) Influence of transpiration rate on uptake and transport of potassium ions in barley plants. Planta 88:369–371
- Horie T, Yoshida K, Nakayama H, Yamada K, Oiki S, Shinmyo A (2001) Two types of hkt transporters with different properties of Na⁺ and K⁺ transport in *Oryza* sativa. Plant J 27:129–138
- Horie T, Horie R, Chan W-Y, Leung H-Y, Schroeder JI (2006) Calcium regulation of sodium hypersensitivities of sos3 and athkt1 mutants. Plant Cell Physiol 47:622–633
- Horie T, Costa A, Kim TH, Han MJ, Horie R, Leung HY, Miyao A, Hirochika H, An G, Schroeder JI (2007) Rice OsHKT2;1 transporter mediates large Na⁺ influx component into K⁺-starved roots for growth. EMBO J 26:3003–3014
- Hosy E, Duby G, Very A-A, Costa A, Sentenac H, Thibaud J-B (2005) A procedure for localisation and electrophysiological characterisation of ion channels heterologously expressed in a plant context. Plant Methods 1:14
- Hua BG, Mercier RW, Leng Q, Berkowitz GA (2003) Plants do it differently. A new basis for potassium/ sodium selectivity in the pore of an ion channel. Plant Physiol 132:1353–1361
- Huang S, Spielmeyer W, Lagudah ES, James RA, Platten JD, Dennis ES, Munns R (2006) A sodium transporter (HKT7) is a candidate for Nax1, a gene for salt tolerance in durum wheat. Plant Physiol 142:1718–1727
- Jabnoune M, Espeout S, Mieulet D, Fizames C, Verdeil JL, Conejero G, Rodriguez-Navarro A, Sentenac H, Guiderdoni E, Abdelly C, Very AA (2009) Diversity in expression patterns and functional properties in the rice HKT transporter family. Plant Physiol 150:1955–1971
- James RA, Davenport RJ, Munns R (2006) Physiological characterisation of two genes for Na⁺ exclusion in

durum wheat: NAX1 and NAX2. Plant Physiol 142:1537–1547

- Jones RG (1983) Proteins, enzymes and inorganic ions encyclopedia of plant physiology. In: Lauchli A, Bielski RL (eds) Encyclopaedia of plant physiology (New Series), vol xv, Inorganic plant nutrition. Springer, Berlin, pp 528–562
- Jung JY, Shin R, Schachtman DP (2009) Ethylene mediates response and tolerance to potassium deprivation in arabidopsis. Plant Cell 21:607–621
- Kaddour R, Narsi N, Mïrah S, Berthomieu P, Lachaal M (2009) Comparative effect of potassium on K⁺ and Na⁺ uptake and transport in two accessions of *Arabidopsis thaliana* during salinity stress. Comptes Rendus Biologies 332:784–794
- Kader MA, Lindberg S (2008) Cellular traits for sodium tolerance in rice (Oryza sativa l.). Plant Biotechnol 25: 247–255
- Kaplan B, Sherman T, Fromm H (2007) Cyclic nucleotide-gated channels in plants. FEBS Lett 581: 2237–2246
- Kayser M, Isselstein J (2005) Potassium cycling and losses in grassland systems: a review. Grass Forage Sci 60:213–224
- Kim EJ, Kwak JM, Uozumi N, Schroeder JI (1998) Atkup1: an Arabidopsis gene encoding high-affinity potassium transport activity. Plant Cell 10:51–62
- Kim MJ, Shin R, Schachtman DP (2009) A nuclear factor regulates abscisic acid responses in arabidopsis. Plant Physiol 151:1433–1445
- Kirkby EA (1968) Influence of ammonium and nitrate nutrition on cation-anion balance and nitrogen and carbohydrate metabolism of white mustard plants grown in dilute nutrient solutions. Soil Sci 105:133–141
- Kochian LV, Lucas WJ (1982) Potassium transport in corn roots: I. Resolution of kinetics into a saturable and linear component. Plant Physiol 70:1723–1731
- Kochian LV, Lucas WJ (1988) Potassium transport in roots. Adv Bot Res 15:93–178
- Kochian LV, Xin-Zhi J, Lucas WJ (1985) Potassium transport in corn roots. Iv. Characterization of the linear component. Plant Physiol 79:771–776
- Kozlowski TT (1984) Plant responses to flooding of soil. Bioscience 34:162–167
- Kronzucker HJ, Szczerba MW, Moazami-Goudarzi MARY, Britto DT (2006) The cytosolic Na⁺:K⁺ ratio does not explain salinity-induced growth impairment in barley: a dual-tracer study using ⁴²K⁺ and ²⁴Na⁺. Plant Cell Environ 29:2228–2237
- Kronzucker HJ, Szczerba MW, Schulze LM, Britto DT (2008) Non-reciprocal interactions between K⁺ and Na⁺ ions in barley (Hordeum vulgare l.). J Exp Bot 59:2793–2801
- Krouk G, Crawford NM, Coruzzi GM, Tsay Y-F (2010) Nitrate signaling: adaptation to fluctuating environments. Curr Opin Plant Biol 13:265–272
- Kurt E, Cramer GR, Lauchli A, Epstein E (1986) Effects of NaCl and CaCl₂ on root cell enlargement and cell production in cotton roots. Plant Physiol 82:1102–1106
- Lagarde D, Basset M, Lepetit M, Conejero G, Gaymard F, Astruc S, Grignon C (1996) Tissue-specific expression

of arabidopsis AKT1 gene is consistent with a role in K^{\star} nutrition. Plant J 9:195–203

- Laüchli A, Epstein E (1990) Plant responses to saline and sodic conditions. In: TAnji K (ed) Agricultural salinity assessment and management. American Society of Civil Engineers, New York, pp 113–185
- Laurie S, Feeney KA, Maathuis FJ, Heard PJ, Brown SJ, Leigh RA (2002) A role for HKT1 in sodium uptake by wheat roots. Plant J 32:139–149
- Lee SC, Lan WZ, Kim BG, Li L, Cheong YH, Pandey GK, Lu G, Buchanan BB, Luan S (2007) A protein phosphorylation/dephosphorylation network regulates a plant potassium channel. Proc Natl Acad Sci 104:15959–15964
- Leidi EO, Barragán V, Rubio L, El-Hamdaoui A, Ruiz MT, Cubero B, Fernández JA, Bressan RA, Hasegawa PM, Quintero FJ, Pardo JM (2010) The atNHX1 exchanger mediates potassium compartmentation in vacuoles of transgenic tomato. Plant J 61:495–506
- Lejay L, Wirth J, Pervent M, Cross JM-F, Tillard P, Gojon A (2008) Oxidative pentose phosphate pathwaydependent sugar sensing as a mechanism for regulation of root ion transporters by photosynthesis. Plant Physiol 146:2036–2053
- Leng Q, Mercier RW, Hua BG, Fromm H, Berkowitz GA (2002) Electrophysiological analysis of cloned cyclic nucleotide-gated ion channels. Plant Physiol 128:400–410
- Li X, Borsics T, Harrington HM, Christopher DA (2005) Arabidopsis atCNGC10 rescues potassium channel mutants of *E. coli*, yeast and Arabidopsis and is regulated by calcium-/-calmodulin and cyclic gmp in *E. coli*. Funct Plant Biol 32:643–653
- Li L, Kim BG, Cheong YH, Pandey GK, Luan S (2006) A Ca²⁺ signaling pathway regulates a K⁺ channel for low-K response in arabidopsis. Proc Natl Acad Sci USA 103:12625–12630
- Lindsay MP, Lagudah ES, Hare RA, Munns R (2004) A locus for sodium exclusion (NAX1), a trait for salt tolerance, mapped in durum wheat. Funct Plant Biol 31:1105–1114
- Luo MC, Dubcovsky J, Goyal S, Dvorak J (1996) Engineering of interstitial foreign chromosome segments containing the K⁺/Na⁺ selectivity gene Kna1 by sequential homoeologous recombination in durum wheat. Theor Appl Genet 93:1180–1184
- Lynch J, Lauchli A (1984) Potassium transport in saltstressed barley roots. Planta 161:295–301
- Maathuis FJM, Amtmann A (1999) K⁺ nutrition and Na⁺ toxicity: the basis of cellular K⁺/Na⁺ ratios. Ann Bot 84:123–133
- Maathuis FJM, Sanders D (1993) Energization of potassium uptake in Arabidopsis thaliana. Planta 191:302–307
- Maathuis FJM, Sanders D (1994) Mechanisms of highaffinity potassium uptake in roots of *Arabidopsis thaliana*. Proc Natl Acad Sci USA 91:9272–9276
- Maathuis FJM, Sanders D (1996) Mechanisms of potassium absorption by higher plant roots. Physiol Plant 96:158–168

- Maathuis FJM, Filatov V, Herzyk P, Krijger GC, Axelsen KB, Chen SX, Green BJ, Li Y, Madagan KL, Sanchez-Fernandez R, Forde BG, Palmgren MG, Rea PA, Williams LE, Sanders D, Amtmann A (2003) Transcriptome analysis of root transporters reveals participation of multiple gene families in the response to cation stress. Plant J 35:675–692
- Mangano S, Silberstein S, Santa-Maria GE (2008) Point mutations in the barley HvHAK1 potassium transporter lead to improved K⁺-nutrition and enhanced resistance to salt stress. FEBS Lett 582:3922–3928
- Marschner H (1995) Mineral nutrition of higher plants. Springer, New York
- Martínez-Cordero MA, Martínez V, Rubio F (2004) Cloning and functional characterization of the highaffinity K⁺ transporter HAK1 of pepper. Plant Mol Biol 56:413–421
- Martínez-Cordero MA, Martinez V, Rubio F (2005) Highaffinity K⁺ uptake in pepper plants. J Exp Bot 56:1553–1562
- Maser P, Thomine S, Schroeder JI, Ward JM, Hirschi K, Sze H, Talke IN, Amtmann A, Maathuis FJ, Sanders D, Harper JF, Tchieu J, Gribskov M, Persans MW, Salt DE, Kim SA, Guerinot ML (2001) Phylogenetic relationships within cation transporter families of arabidopsis. Plant Physiol 126:1646–1667
- Masle J, Gilmore SR, Farquhar GD (2005) The erecta gene regulates plant transpiration efficiency in arabidopsis. Nature 436:866–870
- Mengel K, Kirkby EA (1982) Principles of plant nutrition. International Potash Institute, Worblaufen-Bern
- Mittler R (2006) Abiotic stress, the field environment and stress combination. Trends Plant Sci 11:15–19
- Moller IS, Gilliham M, Jha D, Mayo GM, Roy SJ, Coates JC, Haseloff J, Tester M (2009) Shoot Na⁺ exclusion and increased salinity tolerance engineered by cell type-specific alteration of Na⁺ transport in arabidopsis. Plant Cell 21:2163–2178
- Munns R (2005) Genes and salt tolerance: bringing them together. New Phytol 167:645–663
- Munns R, Rebetzke GJ, Husain S, James RA, Hare RA (2003) Genetic control of sodium exclusion in durum wheat. Aust J Agric Res 54:627–635
- Naidoo G, Rughunanan R (1990) Salt tolerance in the succulent, coastal halophyte, *Sarcocornia natalensis*. J Exp Bot 41:497–502
- Nassery H, Baker DA (1975) Interactions in the absorption of monovalent cations by excised radish roots. Ann Bot 39:621–625
- Nieves-Cordones M, Martinez-Cordero MA, Martinez V, Rubio F (2007) An NH₄⁺-sensitive component dominates high-affinity K⁺ uptake in tomato plants. Plant Sci 172:273–280
- Nieves-Cordones M, Miller A, Alemán F, Martínez V, Rubio F (2008) A putative role for the plasma membrane potential in the control of the expression of the gene encoding the tomato high-affinity potassium transporter hak5. Plant Mol Biol 68:521–532
- Nieves-Cordones M, Aleman F, Martinez V, Rubio F (2010) The Arabidopsis thaliana HAK5 K⁺ transporter

is required for plant growth and K^+ acquisition from low K^+ solutions under saline conditions. Mol Plant 3:326–333

- Niu X, Bressan RA, Hasegawa PM, Pardo JM (1995) Ion homeostasis in NaCl stress environments. Plant Physiol 109:735–742
- Norby RJ (1994) Issues and perspectives for investigating root responses to elevated atmospheric carbon-dioxide. Plant Soil 165:9–20
- Ohkama-Ohtsu N, Wasaki J (2010) Recent progress in plant nutrition research: cross-talk between nutrients, plant physiology and soil microorganisms. Plant Cell Physiol 51:1255–1264
- Pal Y, Gilkes R, Wong M (2001) Soil factors affecting the availability of potassium to plants for western Australian soils: a glasshouse study. Aust J Soil Res 39:611–625
- Papdi C, Joseph MP, Salamó IP, Vidal S, Szabados L (2009) Genetic technologies for the identification of plant genes controlling environmental stress responses. Funct Plant Biol 36:696–720
- Pardo JM, Cubero B, Leidi EO, Quintero FJ (2006) Alkali cation exchangers: roles in cellular homeostasis and stress tolerance. J Exp Bot 57:1181–1199
- Peng YH, Zhu YF, Mao YQ, Wang SM, Su WA, Tang ZC (2004) Alkali grass resists salt stress through high [K⁺] and an endodermis barrier to Na⁺. J Exp Bot 55:939–949
- Pezeshki SR (1995) Plant responses to flooding. In: Wilkinson RE (ed) Plant-environment interactions. Marcel Dekker, New York, pp 289–321
- Pezeshki SR (2001) Wetland plant responses to soil flooding. Environ Exp Bot 46:299–312
- Pezeshki SR, DeLaune AD, Anderson PH (1999) Effect of flooding on elemental uptake and biomass allocation in seedlings of three bottomland tree species. J Plant Nutr 22:1481–1494
- Philippar K, Fuchs I, Luthen H, Hoth S, Bauer CS, Haga K, Thiel G, Ljung K, Sandberg G, Bottger M, Becker D, Hedrich R (1999) Auxin-induced K⁺ channel expression represents an essential step in coleoptile growth and gravitropism. Proc Natl Acad Sci USA 96:12186–12191
- Pilot G, Gaymard F, Mouline K, Cherel I, Sentenac H (2003) Regulated expression of Arabidopsis shaker K⁺ channel genes involved in K⁺ uptake and distribution in the plant. Plant Mol Biol 51:773–787
- Plett D, Safwat G, Gilliham M, Skrumsager MI, Roy S, Shirley N, Jacobs A, Johnson A, Tester M (2010) Improved salinity tolerance of rice through cell typespecific expression of *AtHKT1*;1. PLoS One 5:e12571
- Poffenroth M, Green DB, Tallman G (1992) Sugar concentrations in guard-cells of vicia-faba illuminated with red or blue-light – analysis by high-performance liquid-chromatography. Plant Physiol 98:1460–1471
- Ponnamperuma FN (1972) The chemistry of submerged soils. Advan Agron 24:29–96
- Poorter H, Navas ML (2003) Plant growth and competition at elevated CO₂: on winners, losers and functional groups. New Phytol 157:175–198

- Pregitzer KS, Zak DR, Curtis PS, Kubiske ME, Teeri JA, Vogel CS (1995) Atmospheric CO₂, soil-nitrogen and turnover of fine roots. New Phytol 129:579–585
- Pyo YJ, Gierth M, Schroeder JI, Cho MH (2010) Highaffinity K⁺ transport in Arabidopsis: AtHAK5 and AKT1 are vital for seedling establishment and postgermination growth under low-potassium conditions. Plant Physiol 153:863–875
- Qi Z, Hampton CR, Shin R, Barkla BJ, White PJ, Schachtman DP (2008) The high affinity K⁺ transporter AtHAK5 plays a physiological role in planta at very low K⁺ concentrations and provides a caesium uptake pathway in Arabidopsis. J Exp Bot 59:595–607
- Quintero FJ, Blatt MR (1997) A new family of K⁺ transporters from Arabidopsis that are conserved across phyla. FEBS Lett 415:206–211
- Rafalski A (2002) Applications of single nucleotide polymorphisms in crop genetics. Curr Opin Plant Biol 5:94–100
- Rains DW (1969) Sodium and potassium absorption by bean stem tissue. Plant Physiol 44:547–554
- Rains DW, Epstein E (1967) Sodium absorption by barley roots: its mediation by mechanism 2 of alkali cation transport. Plant Physiol 42:319–323
- Reckmann U, Scheibe R, Raschke K (1990) Rubisco activity in guard-cells compared with the solute requirement for stomatal opening. Plant Physiol 92:246–253
- Reddy AR (2010) The impact of global elevated CO_2 concentration on photosynthesis and plant productivity. Current Sci 99:46–57
- Reintanz B, Szyroki A, Ivashikina N, Ache P, Godde M, Becker D, Palme K, Hedrich R (2002) AtKC1, a silent Arabidopsis potassium channel alpha-subunit modulates root hair K⁺ influx. Proc Natl Acad Sci USA 99:4079–4084
- Ren ZH (2005) A rice quantitative trait locus for salt tolerance encodes a sodium transporter. Nat Genet 37:1141–1146
- Ren Z, Zheng Z, Chinnusamy V, Zhu J, Cui X, Iida K, Zhu J-K (2010) Ras1, a quantitative trait locus for salt tolerance and aba sensitivity in arabidopsis. Proc Natl Acad Sci USA 107:5669–5674
- Rengel Z, Damon PM (2008) Crops and genotypes differ in efficiency of potassium uptake and use. Physiol Plant 133:624–636
- Rodríguez-Navarro A (2000) Potassium transport in fungi and plants. Biochim Biophy Acta 1469:1–30
- Rodríguez-Navarro A, Rubio F (2006) High-affinity potassium and sodium transport systems in plants. J Exp Bot 57:1149–1160
- Rodriguez-Navarro A, Blatt MR, Slayman CL (1986) A potassium-proton symport in *Neurospora crassa*. J Gen Physiol 87:649–674
- Rodriguez-Rosales MP, Jiang XY, Galvez FJ, Aranda MN, Cubero B, Venema K (2008) Overexpression of the tomato K⁺/H⁺ antiporter LeNHX2 confers salt tolerance by improving potassium compartmentalization. New Phytol 179:366–377
- Rogers HH, Runion GB, Krupa SV (1994) Plant response to atmospheric CO₂ enrichment with emphasis on roots and rhizosphere. Environ Pollut 83:155–189

- Rogers HH, Prior SA, Runion GB, Mitchell RJ (1996) Root to shoot ratio of crops as influenced by CO₂. Plant Soil 187:229–248
- Römheld V, Kirkby E (2010) Research on potassium in agriculture: needs and prospects. Plant Soil 335: 155–180
- Ruan Y-L, Llewellyn DJ, Furbank RT (2001) The control of single-celled cotton fiber elongation by developmentally reversible gating of plasmodesmata and coordinated expression of sucrose and K⁺ transporters and expansin. Plant Cell 13:47–60
- Rubio F, Gassmann W, Schroeder JI (1995) Sodiumdriven potassium uptake by the plant potassium transporter HKT1 and mutations conferring salt tolerance. Science 270:1660–1663
- Rubio F, Schwarz M, Gassmann W, Schroeder JI (1999) Genetic selection of mutations in the high affinity K⁺ transporterHKT1 that define functions of a loop site for reduced Na⁺ permeability and increased Na⁺ tolerance. J Biol Chem 274:6839–6847
- Rubio F, Santa-Maria GE, Rodríguez-Navarro A (2000) Cloning of Arabidopsis and barley cDNAs encoding HAK potassium transporters in root and shoot cells. Physiol Plant 109:34–43
- Rubio F, Nieves-Cordones M, Alemán F, Martinez V (2008) Relative contribution of AtHAK5 and AtAKT1 to K⁺ uptake in the high-affinity range of concentrations. Physiol Plant 134:598–608
- Rubio V, Bustos R, Irigoyen ML, Cardona-Lopez X, Rojas-Triana M, Paz-Ares J (2009) Plant hormones and nutrient signaling. Plant Mol Biol 69:361–373
- Rubio F, Alemán F, Nieves-Cordones M, Martínez V (2010) Studies on Arabidopsis *athak5*, *atak11* double mutants disclose the range of concentrations at which athak5, atak11 and unknown systems mediate K⁺ uptake. Physiol Plant 139:220–228
- Rus A, Yokoi S, Sharkhuu A, Reddy M, Lee BH, Matsumoto TK, Koiwa H, Zhu JK, Bressan RA, Hasegawa PM (2001) AtHKT1 is a salt tolerance determinant that controls Na⁺ entry into plant roots. Proc Natl Acad Sci USA 98:14150–14155
- Rus A, Lee Bh, Muñoz-Mayor A, Sharkhuu A, Miura K, Zhu JK, Bressan RA, Hasegawa PM (2004) Athkt1 facilitates Na⁺ homeostasis and K⁺ nutrition in planta. Plant Physiol 136:2500–2511
- Rus A, Baxter I, Muthukumar B, Gustin J, Lahner B, Yakubova E, Salt DE (2006) Natural variants of athkt1 enhance Na⁺ accumulation in two wild populations of Arabidopsis. PLoS Genet 2:1964–1973
- Saier MH, Eng BH, Fard S, Garg J, Haggerty DA, Hutchinson WJ, Jack DL, Lai EC, Liu HJ, Nusinew DP, Omar AM, Pao SS, Paulsen IT, Quan JA, Sliwinski M, Tseng TT, Wachi S, Young GB (1999) Phylogenetic characterization of novel transport protein families revealed by genome analyses. Biochim Biophys Acta Rev Biomembr 1422:1–56
- Sano T, Becker D, Ivashikina N, Wegner LH, Zimmermann U, Roelfsema MRG, Nagata T, Hedrich R (2007) Plant cells must pass aK⁺ threshold to re-enter the cell cycle. Plant J 50:401–413

- Santa-María GE, Epstein E (2001) Potassium/sodium selectivity in wheat and the amphiploid cross wheat x *Lophopyrum elongatum*. Plant Sci 160:523–534
- Santa-María GE, Rubio F, Dubcovsky J, Rodriguez-Navarro A (1997) The HAK1 gene of barley is a member of a large gene family and encodes a high-affinity potassium transporter. Plant Cell 9:2281–2289
- Schachtman DP (1992) Expression of an inward-rectifying potassium channel by the arabidopsis KAT1 cDNA. Science 258:1654–1658
- Schachtman DP, Schroeder JI (1994) Structure and transport mechanism of a high-affinity potassium uptake transporter from higher plants. Nature 370:655–658
- Schachtman DP, Shin R (2007) Nutrient sensing and signaling: NPKS. Annu Rev Plant Biol 58:47–69
- Sentenac H, Bonneaud N, Minet M, Lacroute F, Salmon JM, Gaymard F, Grignon C (1992) Cloning and expression in yeast of a plant potassium ion transport system. Science 256:663–665
- Shabala SN, Lew RR (2002) Turgor regulation in osmotically stressed Arabidopsis epidermal root cells. Direct support for the role of inorganic ion uptake as revealed by concurrent flux and cell turgor measurements. Plant Physiol 129:290–299
- Shabala S, Babourina O, Newman I (2000) Ion-specific mechanisms of osmoregulation in bean mesophyll cells. J Exp Bot 51:1243–1253
- Shabala S, Demidchik V, Shabala L, Cuin TA, Smith SJ, Miller AJ, Davies JM, Newman IA (2006) Extracellular Ca²⁺ ameliorates nacl-induced K⁺ loss from arabidopsis root and leaf cells by controlling plasma membrane K⁺permeable channels. Plant Physiol 141:1653–1665
- Shabala S, Pang JY, Zhou MX, Shabala L, Cuin TA, Nick P, Wegner LH (2009) Electrical signalling and cytokinins mediate effects of light and root cutting on ion uptake in intact plants. Plant Cell Environ 32: 194–207
- Shi HZ, Lee BH, Wu SJ, Zhu JK (2003) Overexpression of a plasma membrane Na⁺/H⁺ antiporter gene improves salt tolerance in *Arabidopsis thaliana*. Nat Biotechnol 21:81–85
- Shin R, Schachtman DP (2004) Hydrogen peroxide mediates plant root cell response to nutrient deprivation. Proc Natl Acad Sci USA 101:8827–8832
- Shinano T, Yamamoto T, Tawaraya K, Tadokoro M, Koike T, Osaki M (2007) Effects of elevated atmospheric CO₂ concentration on the nutrient uptake characteristics of japanese larch (*Larix kaempferi*). Tree Physiol 27:97–104
- Siddiqi MY, Memon AR, Glass ADM (1984) Regulation of K⁺ influx in barley – effects of low-temperature. Plant Physiol 74:730–734
- Spalding EP, Hirsch RE, Lewis DR, Qi Z, Sussman MR, Lewis BD (1999) Potassium uptake supporting plant growth in the absence of akt1 channel activity: inhibition by ammonium and stimulation by sodium. J Gen Physiol 113:909–918
- Stewart WM, Dibb DW, Johnston AE, Smyth TJ (2005) The contribution of commercial fertilizer nutrients to food production. Agron J 97:1–6

- Stulen I, den Hertog J (1993) Root growth and functioning under atmospheric CO₂ enrichment. Vegetatio 104: 99–115
- Su H, Golldack D, Zhao C, Bohnert HJ (2002) The expression of hak-type K⁺ transporters is regulated in response to salinity stress in common ice plant. Plant Physiol 129:1482–1493
- Su Q, Feng S, An L, Zhang G (2007) Cloning and functional expression in saccharomyces cereviae of a K⁺ transporter, alhak, from the graminaceous halophyte, *Aeluropus littoralis*. Biotechnol Lett 29: 1959–1963
- Sunarpi HT, Motoda J, Kubo M, Yang H, Yoda K, Horie R, Chan WY, Leung HY, Hattori K, Konomi M, Osumi M, Yamagami M, Schroeder JI, Uozumi N (2005) Enhanced salt tolerance mediated by athkt1 transporter-induced Na⁺ unloading from xylem vessels to xylem parenchyma cells. Plant J 44:928–938
- Szczerba MW, Britto DT, Kronzucker HJ (2009) K⁺ transport in plants: physiology and molecular biology. J Plant Physiol 166:447–466
- Szyroki A, Ivashikina N, Dietrich P, Roelfsema MR, Ache P, Reintanz B, Deeken R, Godde M, Felle H, Steinmeyer R, Palme K, Hedrich R (2001) Kat1 is not essential for stomatal opening. Proc Natl Acad Sci USA 98:2917–2921
- Taiz L, Zeiger E (1991) Plant physiology. Benjamin/ Cummings Publishing Company Inc., Redwood City, CA
- Takahashi R, Liu S, Takano T (2007a) Cloning and functional comparison of a high-affinity K⁺ transporter gene PhaHKT1 of salt-tolerant and salt-sensitive reed plants. J Exp Bot 58:4387–4395
- Takahashi R, Nishio T, Ichizen N, Takano T (2007b) Cloning and functional analysis of the K⁺ transporter, PhaHAK2, from salt-sensitive and salt-tolerant reed plants. Biotechnol Lett 29:501–506
- Takahashi R, Nishio T, Ichizen N, Takano T (2007c) High-affinity K⁺ transporter Phahak5 is expressed only in salt-sensitive reed plants and shows Na⁺ permeability under nacl stress. Plant Cell Rep 26: 1673–1679
- Talke IN, Blaudez D, Maathuis FJM, Sanders D (2003) Cngcs: prime targets of plant cyclic nucleotide signalling? Trends Plant Sci 8:286–293
- Tester M, Langridge P (2010) Breeding technologies to increase crop production in a changing world. Science 327:818–822
- Tingey DT, Phillips DL, Johnson MG (2000) Elevated CO_2 and conifer roots: effects on growth, life span and turnover. New Phytol 147:87–103
- Uozumi N, Kim EJ, Rubio F, Yamaguchi T, Muto S, Tsuboi A, Bakker EP, Nakamura T, Schroeder JI (2000) The Arabidopsis HKT1 gene homolog mediates inward Na⁺ currents in xenopus laevis oocytes and Na⁺ uptake in *Saccharomyces cerevisiae*. Plant Physiol 122:1249–1259

- Vicente-Agullo F, Rigas S, Desbrosses G, Dolan L, Hatzopoulos P, Grabov A (2004) Potassium carrier TRH1 is required for auxin transport in Arabidopsis roots. Plant J 40:523–535
- Walker DJ, Leigh RA, Miller AJ (1996) Potassium homeostasis in vacuolate plant cells. Proc Natl Acad Sci USA 93:10510–10514
- Walker DJ, Black CR, Miller AJ (1998) The role of cytosolic potassium and ph in the growth of barley roots. Plant Physiol 118:957–964
- Wang Y, Wu W-H (2010) Plant sensing and signaling in response to K⁺-deficiency. Mol Plant 3:280–287
- Wang MY, Glass ADM, Shaff JE, Kochian LV (1994) Ammonium uptake by rice roots.3. Electrophysiology. Plant Physio 104:899–906
- Wang TB, Gassmann W, Rubio F, Schroeder JI, Glass ADM (1998) Rapid up-regulation of HKT1, a highaffinity potassium transporter gene, in roots of barley and wheat following withdrawal of potassium. Plant Physiol 118:651–659
- Wang YH, Garvin DF, Kochian LV (2001) Nitrate-induced genes in tomato roots. Array analysis reveals novel genes that may play a role in nitrogen nutrition. Plant Physiol 127:345–359
- Wang Y-H, Garvin DF, Kochian LV (2002) Rapid induction of regulatory and transporter genes in response to phosphorus, potassium, and iron deficiencies in tomato roots. Evidence for cross talk and root/rhizospheremediated signals. Plant Physiol 130:1361–1370
- White PJ, Brown PH (2010) Plant nutrition for sustainable development and global health. Ann Bot 105: 1073–1080
- Wiebold B, Scharf P (2003) K⁺ deficiency symptoms in drought stressed crops. Integrated Pest Crop Management Newsletter 13:2–14
- Xing Y, Zhang Q (2010) Genetic and molecular bases of rice yield. Annu Rev Plant Biol 61:421–442
- Xu J, Li HD, Chen LQ, Wang Y, Liu LL, He L, Wu WH (2006) A protein kinase, interacting with two calcineurin b-like proteins, regulates K⁺ transporter AKT1 in Arabidopsis. Cell 125:1347–1360
- Yang X, Liu J, Wang W, Ye Z, Luo A (2004) Potassium internal use efficiency relative to growth vigor, potassium distribution, and carbohydrate allocation in rice genotypes. J Plant Nutr 27:837–852
- Zhao J, Cheng NH, Motes CM, Blancaflor EB, Moore M, Gonzales N, Padmanaban S, Sze H, Ward JM, Hirschi KD (2008) AtCHX13 is a plasma membrane K⁺ transporter. Plant Physiol 148:796–807
- Zhu JK, Liu J, Xiong L (1998) Genetic analysis of salt tolerance in Arabidopsis. Evidence for a critical role of potassium nutrition. Plant Cell 10:1181–1191
- Zimmermann S, Talke I, Ehrhardt T, Nast G, Muller-Rober B (1998) Characterization of SKT1, an inwardly rectifying potassium channel from potato, by heterologous expression in insect cells. Plant Physiol 116: 879–890

Temperature Stress and Responses of Plants

Anna Źróbek-Sokolnik

Abstract

Among the abiotic environmental factors temperature is the most important factor which significantly affects life processes of all organisms. Temperature stresses experienced by plants are usually classified into three types: (a) chilling stress (occurring at temperatures below freezing), (b) freezing stress (occurring at low temperatures above freezing), and (c) high temperature stress. This chapter shows the influence of low and high temperature to physiological and metabolic processes in plants. The consequences of chilling and freezing or heat stresses are presented as well as mechanisms of plant resistance to low or high temperature and adaptation or/and acclimatization possibilities is reported in this chapter.

Keywords

Temperature • Vernalization • Stratification • Metabolism • Freezing • Acclimatization • Adaptation

1 Introduction

Temperature is an abiotic environmental factor that significantly affects life processes in all organisms by modifying membrane properties, enzyme activity levels, the rate of chemical reactions and diffusion, viscosity of vacuole

Department of Botany and Nature Protection, University of Warmia and Mazury in Olsztyn, Plac Łódzki 1, 10-727 Olsztyn, Poland e-mail: a.zrobeksokolnik@uwm.edu.pl solution and the cytoplasm, phloem, and xylem solutions in plants (Sung et al. 2003). Living organisms can be classified into three groups, subject to the preferred temperature of growth (Fig. 5.1). This chapter analyzes the impact of temperature on plant growth with emphasis on plant response to temperature stress.

It is believed that land plants evolved in a tropical climate. This evolution process was spurred not so much by a warm climate, but by the stability of ambient temperature. Plants gradually migrated into temperate regions both north and south of the equator as they developed mechanisms that allowed them to accommodate

© Springer Science+Business Media, LLC 2012

5

A. Źróbek-Sokolnik (⊠)



Fig. 5.1 Classification of the living organisms, subject to their preferred temperature of growth

wider variations in temperature on both a daily and a seasonal basis (Fitter and Hay 2002). The growth and development of plants involves a countless number of biochemical reactions that are sensitive to temperature. Plant life is generally limited by the freezing point of water at the low end of the temperature scale and the irreversible denaturation of proteins at the high end. Temperature is a critical factor in the plant environment, and it may play a significant role in growth and development. Growth is defined as an increase in dry weight, while development is the increase in the number and/or dimension of organs by cell division and/or expansion: leaves, branches, spikelets, florets, root apices, etc., including those present in seed embryos. It also seems that the rate of plant development tends to be controlled primarily by temperature, and it is less sensitive to other environmental factors. The development of vegetation is determined by a broad variety of environmental factors that exert combined effects. Plant organisms are rarely affected by individual factors, and temperature stress is usually accompanied by water stress and, in consequence, oxidative stress (Fitter and Hay 2002). Temperature can also play a part in controlling the pattern and timing of plant development, and this accounts for the below phenomena.

1.1 Vernalization

In some plant species, a period of low temperatures is required to induce flowering, while in other plants, low temperatures only accelerate flowering or have no effect at all. Plants with a vernalization requirement experience a period of low temperatures in late fall and/or winter at the stage of seed imbibition or young seedlings (annual winter crops) or upon reaching vegetative maturity (biennial and perennial plants) (Kim et al. 2009). Flowering is induced in the temperature range of 0 to +10°C. The duration of the vernalization period, that is, the required number of days with low temperatures, varies subject to species, and it usually reaches from 2 weeks to several weeks (Dennis et al. 1996; Amasino 2006). In seeds, temperature stimuli are perceived by the embryo, while in seedlings and matured plants, this signal is sensed by apical meristems. A vernalized meristem retains competence following the reception of the inductive signal. When the signal is absent for a longer period of time, the plant is de-vernalized, and a similar effect can be achieved by exposing the plant to higher temperatures (around 40°C for 1–2 days) (Tretyn et al. 2003). The mechanisms underlying vernalization have not yet been fully explained. It is believed that low temperatures lead to changes in the permeability of cell membranes and/or the level of expression of "vernalization" genes. Phytohormones, in particular gibberellin, significantly contribute to this process (Sheldon et al. 2000; Amasino 2005).

1.2 Stratification

Stratification is a popular method of breaking seed dormancy that has been used for centuries. This technique involves the storage of seeds in a moist and well-ventilated environment at relatively low temperatures in the range of $1-10^{\circ}$ C. Stratification is generally defined as the process of subjecting seeds to cold or warm and cold conditions in a moist and ventilated environment to break the dormancy stage. Low temperature, high moisture content, and oxygen supply during the treatment induce deep physiological and biochemical changes in seeds. Stratification leads to the decomposition of germination inhibitors in seeds, and it induces the production of growth stimulators: cytokinin, gibberellin, and auxin. At various stages of the dormancy breaking period, changes are noted in the quantitative ratio of various stimulators which modify the seeds' sensitivity to light and temperature and support dormancy breaking in various dormancy mechanisms (e.g., Baskin and Baskin 1998; Opik and Rolfe 2005; Wróbel et al. 2005).

1.3 The Effect of Temperature on Membranes, Enzymes, and Metabolic Processes

An increase or a decrease in temperature changes the kinetic energy of particles, accelerating their motion and weakening hydrogen bonds in macromolecules. All of the reactions contributing to growth are catalyzed by enzymes whose activity depends on their precise, three-dimensional, tertiary structures, to which the reacting molecules must bind exactly for each reaction to proceed. As the temperature rises, tertiary structures are damaged, reducing enzyme activity and reaction rates (Price and Stevens 1999). The asymmetry of response curves, such as Fig. 5.2a, b, is the net result of an exponential increase in the reaction rate, caused by increased collision frequency, and increasingly modified by the thermal denaturation of macromolecules (Fitter and Hay 2002).

The effect of temperature on enzyme activity is not a simple correlation. Activity levels rise with an increase in temperature, but only within a temperature range that guarantees the enzyme's stability (Cornish-Bowden 2004). When the critical temperature is exceeded, enzymes undergo thermal denaturation, and their activity drops rapidly. The average rate of enzymatic reactions increases twofold with every 10°C increase in temperature within the range that does not cause enzyme denaturation (Fig. 5.3). The correlation between temperature and the increase in enzymatic activity is described by temperature coefficient Q_{10} which illustrates changes in reaction rate when the temperature increases by 10°C:

$$Q_{10} = \frac{v_{(t+10)}}{v_t}$$

Parameter Q_{10} applies only in a nondenaturing range of temperatures, it is enzyme specific and determined by the activation energy of the catalyzed reaction. Enzyme activity reaches the highest level at optimal temperature. The representative values of temperature coefficients (Q_{10}) for selected plant processes measured at varying intervals within the range 0-30°C are determined at 1-2.3 (e.g., light reactions of photosynthesis ~1; diffusion of small molecules in water: 1.2–1.5; water flow through seed coat: 1.3-1.6; water flow into germinating seeds: 1.5-1.8; hydrolysis reactions catalyzed by enzymes: 1.5-2.3; root axis extension: 2.3). Coefficient value reaches 2-3 for dark reactions of photosynthesis, 0.8-3 for phosphate ion uptake into storage tissue, and 2-5 for potassium ion uptake into seedlings. Grass leaf extension is characterized by Q_{10} of 3.2, and the relative growth rate is marked by coefficient value of 7.2 (Fitter and Hay 2002). The observed optimal temperature is the product of two processes: an increase in the reaction rate related to an increase in kinetic energy and an increase in the rate of thermal denaturation of an enzyme above a critical temperature point. When the



Fig. 5.2 Schematic illustrations of plant responses to temperature (adapted from Fitter and Hay 2002). (a) The response of plant growth rate; (b) the influence of temperature on the rate of photosynthesis and respiration

second parameter is higher, a drop in activity levels is noted. For most enzymes, the optimal temperature falls within the range of 30–45°C. Enzymes are irreversibly denatured and inactivated at temperatures higher than 60°C. The enzymes of thermophilous organisms (such as thermal spring bacteria) remain active and attain maximum reaction rates at higher temperatures. The highest temperature at which an enzyme is not thermally inactivated under given conditions determines the enzyme's thermal stability. An alternative approach involves the application of the Arrhenius equation (from chemical kinetics) to plant processes:

$$k = A \exp(-E_a / RT),$$

where k is the rate constant, E_a is the activation energy for the process, A is the constant, R is the gas constant, and T is the temperature expressed on the absolute temperature scale.

Arrhenius constants (E_a/R for the process) can be useful in biochemical comparisons between



Fig. 5.3 The influence of temperature on enzyme activity

species (Criddle et al. 1994; Levine 2005) and in analyses of plant membrane changes during cooling and freezing.

Higher temperatures increase the liquidity of membrane lipid layers. A temperature drop has the opposite effect: biological membranes become more rigid and the activation energy of membrane enzymes increases. The above phenomenon is as the result of thermotropic changes in the lipid phase. Temperature modifies the organization of fatty acid residues in phospholipids and galactolipids, the components of various membranes. The configuration of polyunsaturated fatty acid residues is more difficult to reorganize at lower temperatures than that of saturated fatty acids, but polyunsaturated fatty acids residues "melt" more easily at higher temperatures. Temperature-induced changes in the liquidity of the cell membrane or its selected domains that modify the structure and function of membrane proteins. Cell membrane's response to temperature variations may also be determined by its sterol content or interactions with other nonlipid organic compounds (Sung et al. 2003; Alberts et al. 2004). Temperature-induced changes in membrane properties also significantly affect water regulation in cells, and secondary water stress may occur when the rate of water uptake by the roots is slower than leaf transpiration. At temperatures below 0°C, liquid water changes into solid ice. Ice crystals are formed inside the protoplast which could lead to structural damage. Extracellular formation of ice may cause cell dehydration. The component processes of plant

growth do not all respond to temperature in the same way. For example, in most crop species, gross photosynthesis ceases at temperatures just below 0°C (minimum) and above 40°C (maximum), with the highest rates being achieved in the range of 20–35°C. In contrast, rates of respiration tend to be low below 20°C but, owing to the thermal disruption of metabolic controls and compartmentation at higher temperatures, they rise sharply up to the compensation temperature, at which the rate of respiration equals the rate of gross photosynthesis, and there can be no net photosynthesis (Wilkinson 2000; Fitter and Hay 2002; Jenks and Hasegawa 2005; Wahid et al. 2007).

Temperature stress in plants has been broadly researched, and the problem has been widely addressed by review articles (Wang et al. 2003; Wahid et al. 2007; Jan et al. 2009), books discussing various types of stress (Wilkinson 2000; Fitter and Hay 2002; Jenks and Hasegawa 2005), studies investigating the negative effects of extreme temperatures (Iba 2002; Sung et al. 2003), etc. It should be noted that unlike homeothermic animals, plants are unable to maintain their cells and tissues at a constant optimum temperature, therefore, their metabolism, growth, and development are profoundly affected by changes in environmental temperature. This suggests that as sessile organisms, plants must be able to sense transient fluctuations as well as seasonal changes in temperature and respond to these changes by actively adjusting their biology to fit the subsequent temperature regime. Temperature is a major environmental factor that changes from season to season and undergoes daily fluctuations and short, erratic lows and highs. For this reason, the stress-inducing role of temperature is difficult to define unambiguously since the response to various temperatures is determined by the plants' ability to adapt to different climate regimes. Vegetation occurs in climate zones characterized by extreme temperatures of -50 to $+50^{\circ}$ C, that is, within a range of 100°C. The margin of thermal tolerance that conditions the stability of life processes in most plants is relatively wide, ranging from several degrees above zero to around 35°C, and it is genetically determined. Many genotypes specific

Leaf/above-ground organs	-	The amount of solar radiation intercepted		
	_	The potential for energy exchange with the environment	nt	
	_	Time of day (regular diurnal variation of solar elevation)		
	_	Month (typical seasonal variation)		
	_	Cloud cover		
	_	Wind force and origin of air mass (irregular, short-term variation)		
	_	Position in the canopy (e.g., "sun" or "shade" leaf)		
	_	Altitude above soil surface		
	_	Canopy characteristics, including leaf shape, dimensions and surface properties		
Roots/under-ground organs	-	Seasonal and diurnal variations in energy exchange The interception of solar radiation by the canopy The account of the depth below the soil surface	Determine how much energy reaches the soil surface	
	-	Soil properties which influence the energy balance at the soil surface, and the transfer of heat through the soil (e.g., moisture content, bulk density, color/albedo, and the vegetative or litter cover)		

 Table 5.1 Factors which determinate temperature of above- or under-ground organs (adapted from Fitter and Hay 2002)

to extreme climate conditions, from arctic to tropical, have a much wider tolerance margin. In principle, plants in the dormant state (dry germs and seed embryos, dehydrated dormant organs) are far less sensitive to temperature change, and they are able to survive through periods of extreme temperature unharmed. Metabolically active tissues have thermal activity limits which, when exceeded, lead to a reversible drop in the rate of life processes to a minimum level. Further temperature change (referred to as critical or lethal temperature) causes permanent damage to cell structures, it affects cell metabolism, impairs vital life processes, and kills the protoplasm. During evaluations of plant response to extreme temperatures, special attention should be paid to the temperature of the plant which often differs from ambient temperature. In the summer, leaf temperature often exceeds ambient temperature by up to several degrees. Higher differences are noted in plants whose leaves are positioned horizontally, such as apple trees. In the spring and autumn, the night temperature of leaves, in particular when the sky is clear, may be even several degrees lower than ambient temperature (Wilkinson 2000; Fitter and Hay 2002; Jenks and Hasegawa 2005). At a given moment, leaf temperature is determined by several factors (Table 5.1). Roots demonstrate a stronger growth

response to extreme temperatures than the above-ground parts of plants, and the above applies to both extreme cold and extreme heat (Fig. 5.4). During the evolution process, roots became adapted to more stable temperatures. Nonetheless, the temperature of both the roots and other under-ground organs is also determined by factors presented in Table 5.1.

Plants can adapt to changes in the temperature regime through the evolution of genotypes with more appropriate morphologies, life histories, physiological and biochemical characteristics, or by plasticity. Plants also adapt to changing temperatures during the growing season by plastic responses.

2 Low Temperature

Periodic temperature drops below zero degrees are reported on around 64% of the Earth's surface. The lowest temperatures are noted in Antarctica, reaching around -50° C in coastal areas and up to -90° C in the interior. The minimum temperature at which a given species can survive is one of the main criteria determining plant distribution on our planet. In a temperate climate, low-temperature stress eliminates or inhibits the growth and yield of valuable plants and crops (Xin and Browse



Fig. 5.4 Schematic illustration of the influence on growth and morphology of roots and above-ground organs of potato seedlings (adapted from Marschner 1995)

2000; Jan et al. 2009). Plants indigenous to colder regions are usually well adapted to chilling temperatures and are, therefore, not significantly impaired by cold periods, apart from a general slowing down of the metabolic rate and growth. In a temperate climate, plants respond differently to freezing temperatures and the winter environment than other factors that occur irregularly. In the winter, chilling temperatures do not come as a surprise for plants that have adapted to the periodic, adverse vegetation factors in the course of evolution. Low temperatures are accompanied by short daytime and low radiation intensity. The adaptation to growth inhibiting factors is characteristic of the dormant state (Jan et al. 2009).

There are two types of injuries a plant can sustain through exposure to low temperatures (Fig. 5.5). On the other hand, many plants that are native to cold climates can survive extremely low temperatures without injury (Levitt 1980).

An analysis of freezing winter temperatures as an environmental stressor should also account for the impact of other adverse factors such as low light intensity and short daytime. The above conditions arrest the growth and development of vegetation (Hopkins 2006). The plants' ability to survive freezing and other adverse temperature changes differs from the remaining stressors. Levitt's stress avoidance theory (1980) does not apply in this case. Plants are unable to avoid freezing temperatures, and they can only protect themselves from the negative consequences of cold by increasing their tolerance to chilling. Many plants enter the dormant state to survive harsh winter weather. This is a typical feature of adaptation to freezing which is a genetically inherited trait.

Plants can be classified into three categories based on the range of lethal temperatures and the characteristics of mechanisms conditioning their resistance to low temperatures (Fig. 5.6).

2.1 Consequences of Chilling and Freezing Stress

There are two theories explaining the plants' primary response to temperature stress. The first concept, formulated by Lyons (1973), states that low temperatures induce the phase transition of cell membranes where a liquid-crystal structure is transformed into a crystal (gel) phase.



 Occurs in the temperature range of approximately 20 to 0⁰C

• The resultant injures may include a variety of physiological disruptions in germination, flower and fruit development, and storage life.

• Minor chilling stress at non-lethal temperatures is normally reversible. In frost-sensitive plants, those changes are often irreversible

freezinginjury

• Occurs when the external temperature drops below the freezing point of water

 Some species that are not resistant to chill may be killed by the first touch of frost.

Fig. 5.5 Two types of injuries a plant can sustain through exposure to low temperatures (adapted from Stushnoff et al. 1984)



Fig. 5.6 Classification of the plants, subject to their range of lethal temperatures and the characteristics of mechanisms conditioning their resistance to low temperatures (adapted from Stushnoff et al. 1984)

121

Thermotropic phase changes are the primary cause of membrane dysfunctions that lead to irreversible damage and cell death. The above may produce reactive oxygen species and the accompanying oxidative stress. According to recent research, the phospholipid which initiates the phase transition of the cell membrane is phosphatidylglycerol (PG). If a PG molecule contains fatty acids with a high melting point, that is, saturated fatty acids, then the phase transition of this lipid takes place relatively easily at low temperatures and this, in turn, induces the transformation of other phospholipids and galactolipids adjacent to PG (Los and Murata 2004; Wang et al. 2006). According to the second chilling injury theory, the primary cause of damage is the sudden increase in the concentration of free calcium ions in the cytosol (Minorsky 1989). Calcium ion concentrations increase as calcium channels in the plasmalemma become opened due to sudden depolarization (Lecourieux et al. 2006). In chilling-sensitive plants, calcium opens the stomata, and transpiration significantly exceeds water uptake by the roots (Liang et al. 2009). In many sensitive species, the first indication of cold stress is striking wilting of the leaves, despite optimal water supply in the soil (Mahajan and Tuteja 2005; Solanke and Sharma 2008). The release of calcium ions into the cytosol has many secondary effects, including induced gene expression which could result from changes in the content or distribution of cell hormones, mainly abscisic acid (ABA). This phenomenon is in particularly related to the acidification of the cytoplasm at low temperatures (and the corresponding alkalization of the vacuoles) which, at least in part, is actively controlled by H+-transport from the cytoplasm to the vacuole catalyzed by H⁺-ATPase located on the vacuolar membrane. The inactivation of this enzyme has been reported to occur much earlier than other symptoms of cell injury (Yoshida et al. 1999; Lindberg et al. 2005). Chilling affects the entire internal environment of each cell and each molecule within the cells (Kartsch and Wise 2000). The rate and extent of injury is determined by temperature, its duration as well as the chilling rate. Sudden temperature

drops (thermal shock) have particularly damaging consequences. The lower the temperature and the longer its effect, the greater the extent of the sustained injury (Mahajan and Tuteja 2005; Solanke and Sharma 2008). Plant structures and physiological cell processes have varied sensitivity to chilling temperatures (Fig. 5.7). Most injuries are sustained in the cell membrane which may represent a potential site of perception and/or injury (Lindberg et al. 2005). There are changes in the viscosity and liquidity of the membrane, leading to an increase in diffusion resistance and, in many cases, enzyme inactivation. The reversibility of those effects is determined by the severity of damage. Changes in chemical composition may be observed as the result of lipid degradation, the release of fatty acids and changes in the activity of metabolizing enzymes, peroxidation, disintegration of lipid-protein bonds, and higher membrane permeability. The chemical composition of the cytoplasm and differences in lipid quality in various chilling-sensitive species determine the phase transition point, that is, the point at which the membrane is transformed from a liquid-crystal state into a gel state (Solanke and Sharma 2008; Jan et al. 2009). This change in the membrane's physical state impairs its normal functioning. In most chilling-sensitive plants, the phase transition point is around 10°C. Chilling sensitivity is mostly related to a higher content of saturated fatty acid residues in lipids, while the cold-hardiness mechanism is explained by the desaturation of fatty acids which enables the plant to quickly acclimatize to low temperatures. The above is only one of the factors explaining variations in the plants' response to temperature stress (Lindberg et al. 2005; Zhang and Tian 2010). Interactions between membrane components, including lipid-lipid and lipid-protein, are also believed to play an important role. Higher sterol concentrations increase membrane rigidity. The role of membrane proteins during chilling is also a source of controversy, but there is general agreement that conformational changes in protein-lipid systems may lead to membrane disintegration and dysfunction (Los and Murata 2004; Lindberg et al. 2005). Frost-induced



Fig. 5.7 Functional disturbance occurred in chilling-sensitive plants, subjected to stress duration (adapted from Kacperska 1998)

changes may lead to inhibited protoplast movement, excessive protoplast vacuolization, damage to the endoplasmic reticulum, drop in turgidity, and higher membrane permeability. Cytoplasmic streaming and photosynthesis, including thylakoid functioning in chloroplasts (as demonstrated by enhanced in vivo chlorophyll fluorescence), are most susceptible to reversible disruptions. Irreversible damage, including injuries caused by stressors other than temperature, is also most likely to affect thylakoid membranes, mostly photosystem II. Chloroplast lipids undergo various metabolic changes in both chilling-sensitive and cold-hardy plants. Higher levels of galactolipase activity and, consequently, higher free fatty acid concentrations are noted in the chloroplasts of chilling-sensitive species (faba beans, beans, tomatoes, maize) than in cold-hardy plants (spinach, pea). Lower temperatures disrupt the maintenance of the proton gradient in thylakoid membranes conditioning ATP synthesis. Powerful radiation during or directly after chilling intensifies the relevant injuries and retards, or even disables, damage repair in both chilling-sensitive and cold-hardy plants. Long-term frost inhibits the synthesis of chlorophyll and starch (Muller et al. 2005; Liang et al. 2009; Sun et al. 2010). Other membranes (plasmalemma and tonoplast) are damaged after relatively longer exposure, as demonstrated by membrane cells' ability to plasmolyze and vital staining. Those injuries are irreversible. Other metabolic functions are marked by varied sensitivity to low temperatures which cause metabolic disorders and lead to toxin accumulation, for example, respiration efficiency may be higher or lower subject to environmental factors that accompany freezing temperatures. Chilling may also inhibit the activity of many oxidoreductive enzymes, such as catalase, leading to the accumulation of hydrogen peroxide and the production of free radicals (Suzuki and Mittler 2006; Liang et al. 2009; Sun et al. 2010). In sublethal cold stress, fruit ripening and seed germination are most severely inhibited (Kumar and Bhatla 2006).

Frost leads to the appearance of stress which is linked not directly to low temperature, but to freezing (crystallization) of water in the plant (Mahajan and Tuteja 2005). Intracellular and extracellular crystallization produces different effects. Ice crystals are formed readily in those parts of the plant where temperature drops most rapidly and where water freezes most easily (due to high water potential), mostly vascular bundles and intracellular spaces in above-ground parts where water vapor undergoes condensation. Ice crystals spread quickly via vessels and other tissues with uniform structure. The presence of air-filled intercellular spaces as well as tissues with lignified or cutinized walls slows down crystallization. Ice formation is accelerated by icenucleation active bacteria of the genera Ervinia and Pseudomonas. The proteins formed on the outer bacterial cell wall react with water particles and facilitate the formation of ice crystals at temperatures just below 0°C. In the absence of icenucleation active bacteria on the surface of tissues and on the walls of intracellular spaces, ice formation would begin at temperatures several degrees lower due to the supercooling of water solutions.

If tissue is supercooled rapidly (e.g., faster than 5 K min⁻¹) and the cells have high water potential, or if cell water had been first deeply supercooled, ice may be formed in the protoplast. The above invariably leads to cytoplasm destruction and cell death (Fitter and Hay 2002; Rajashekar 2000; Jan et al. 2009; Janska et al. 2010). Water freezing in intracellular spaces is a less dangerous phenomenon. In nature, where temperature decline is generally slow (1-5 Kmin⁻¹), crystallization usually takes place outside the protoplast in intracellular spaces and between the cell wall and the protoplast (partly due to the extracellular fluid having a higher freezing point, i.e., lower solute concentration, than intracellular fluid). The above leads to extracellular crystallization. Vapor pressure decreases in the spaces above ice, and a water potential gradient is created between the unfrozen interior of the cell and the extracellular environment. Water moves along this gradient into extracellular spaces where it is crystallized (Fitter and Hay 2002; Jan et al. 2009; Janska et al. 2010). Cells are dehydrated (secondary stress) and they contract due to desiccation. The lower the surrounding temperature, the longer it takes for an equilibrium to be reached between the water potential above ice and inside cells, and the greater the effect of cell dehydration (Solanke and Sharma 2008).

Multiple forms of membrane damage can occur as a consequence of freeze-induced cellular dehydration including expansion-induced-lysis, lamellar-to-hexagonal-II chase transitions, and fracture jump lesions. The above leads to cell contraction and the associated changes in reactions between the plasmalemma and the cell wall, partial loss of plasmalemma due to exocytosis and endocytosis, changes in the structure of the plasmalemma and other cell membranes, and the creation of protein-deprived lipid areas in the membrane. The greatest damage is done to the plasmalemma. Dehydration also increases the concentration of solutions in the cytosol and the cell sap, leading to higher salinity (Mahajan and Tuteja 2005; Solanke and Sharma 2008; Jan et al. 2009). Conformational changes in proteins found in the plasmalemma and other membranes lead to changes in the activity of various membrane enzymes, including ATPases responsible for the movement of protons and other ions through membranes (Lindberg et al. 2005). Some ions, accumulated in cells by ion transporters (e.g., potassium ions), are diffused after thawing into intracellular spaces together with water, for example, in leaf tissue. Certain proteins, such as the thylakoid coupling factor, become dissociated in the process. The effect of chill injury on life processes is often visible when plants resume their normal growth after freezing temperatures subside. Even partial degradation of thylakoid membranes inhibits photosynthesis, and the process may be reversible. PS II activity may be partially or completely inhibited, and the balance between the light-dependent phase and CO₂ assimilation may be upset. There is a rise in photorespiration intensity (Alam et al. 2005). Changes in the mitochondria and the respiration process are not as profound. In strongly dehydrated cells, the membrane undergoes lyotropic phase transitions, and hexagonal arrangements are formed in lipid bilayers of a single membrane or two layers of two adjoining membranes (e.g., plasmalemma and endoplasmic reticulum). The membranes' primary structure is not always restored after thawing, and water is diffused into the extracellular environment together with ions through membrane channels. Cell dehydration caused by extracellular crystallization increases the concentrations of salt and organic acids in the protoplast which, in turn, may lead to protein denaturation and enzyme inactivation (Mahajan and Tuteja 2005; Solanke and Sharma 2008). Few enzymes remain active at below zero temperatures, but some of them are activated, such as phospholipase D which catalyzes the hydrolysis of phospholipids (Ruelland et al. 2002). The degradation of membrane lipids begins during freezing and after thawing, releasing unsaturated fatty acids which are peroxidized. Chlorophyll may be also be photooxidized in green tissues exposed to light (Sung et al. 2003). Chill injuries may occur not only during freezing, but also during the thawing of tissue. Plant survival is also determined by post-thawing environmental factors rapid temperature growth and high light intensity may disturb metabolic pathways in cells and cause additional damage. During rapid melting of ice, the cell is rehydrated, and it quickly increases its volume. The above leads to tension and cracks in cell structures, mostly in the cytoplasm which is the site of primary cell injury. The above changes have less damaging consequences for dormant plants. In a temperate climate, winter frost is not a typical stressor for plants, but freezing temperatures could be a source of stress if they occurred in the spring or summer (Muller et al. 2005).

2.2 Resistance to low Temperature, Acclimatization

Resistance is related to frost tolerance, that is, the ability of the organism to survive low temperatures without damage. In regions characterized by seasonal climate change, plants' resistance to freezing fluctuates periodically – it is the lowest during intensive elongation growth in the spring, and it rises significantly in the fall when growth is arrested by the direct effect of low temperature or the combined effect of shorter daytime and temperature drop (Li et al. 2005a, b). Frost resistance is usually achieved by preventing ice formation in the symplast. An important mechanism

preventing or delaying symplastic ice formation is frost plasmolysis. Poorly hydrated plants which are acclimatized to water stress usually show increased cold resistance, for example, plants which are extremely tolerant to drying out, for example, embryos of ripened seeds, can be conserved alive at -200°C without damage (Jan et al. 2009). Species-specific cold resistance is a genetically programmed trait that can be modified by both endogenous and exogenous factors. For a vast number of species, frost tolerance is not a static feature, but it is closely correlated with season, it fluctuates in various growing periods, and it is not identical for all organs (Rorat et al. 2006; Hekneby et al. 2006). The above-ground parts of wheat seedlings were acclimatized even to -20° C, but the roots' sensitivity to frost did not change. Acclimatization can be accelerated by hardening the plants, that is, exposing them to increasingly lower temperatures on successive days, initially above zero, followed by insignificantly below zero (Li et al. 2005a; Zhang and Tian 2010). This process is continued for several weeks. Plants are characterized by the greatest frost resistance 1-3 weeks from the beginning of exposure to freezing temperatures. The period of deacclimatization, that is, dehardening, is much shorter, and it usually lasts several days. The higher the ambient temperature, the faster the deacclimatization process. After dehardening, repeated exposure to frost can severely damage many plants (Li et al. 2004; Burbulis et al. 2008).

Plant organs are also marked by varied sensitivity to frost (Li et al. 2005a; Rorat et al. 2006). Roots are most susceptible to the damaging effects of freezing temperatures, shoots are less sensitive, while tree trunks and older branches are characterized by the highest frost resistance (Muller et al. 2005; Kato-Noguchi 2007). Snow cover minimizes the temperature drop in the soil, and it protects crops from freezing. The cold sensitivity of flowers is determined by the given species' phenological growth stages (Thakur et al. 2009; Ohnishi et al. 2010).

ABA stimulates and speeds-up plant hardening. According to Weiser (1970), acclimatization, and perhaps also hardening, is determined by modifications in gene expression. In this case, ABA can enhance cold resistance if it is able to induce the expression of the respective genes (Gusta et al. 2005). Gibberellins and auxins deliver an opposite effect. Substances that retard gibberellin synthesis accelerate hardening. Intensive nitrogen fertilization generally delays dormancy and increases susceptibility to freezing. Heavy potassium fertilization has the opposite effect by increasing the frost resistance of both herbaceous and arborescent plants. The concentrations of sugar and other osmoprotectors that protect the cell from dehydration increases in the cytosol and vacuoles (Liang et al. 2009).

Fluid supercooling inside the cell is yet another factor that increases the plants' cold resistance by delaying crystallization in the cell. The presence of substances dissolved in the vacuolar sap lowers crystallization temperature. In small, weakly vacuolated cells, water may undergo deep supercooling. In large and hydrated parenchymal cells and xylem vessels, the supercooled state is very unstable, and it rarely lasts longer than several hours. Supercooling provides temporary protection against freezing caused by, for example, strong ground frost. In tissues comprising small, densely packed and weakly vacuolated cells whose walls prevent ice crystals from spreading, a supercooled state may persist until the temperature drops below a threshold value. The accumulation of nonpolar lipids on the surface of the plasmalemma also prevents ice penetration from the apoplast to the cell interior. In herbaceous plants, the supercooling of water is observed at -1 to -15° C, and in arborescent plants at -30°C, and even -50°C. Such a high degree of supercooling is observed only in some living tissues, such as core parenchymal cells, meristematic tissue, leaf bud scales, and flower buds. When ambient temperature drops below the critical supercooling point, this meta-stable state is rapidly disrupted, and ice is formed inside the cells, ultimately leading to their death. In some extremely frost-resistant tree species, the protoplasm is able to vitrify. Vitrification is stimulated by a high concentration of sucrose and other sugars. In this relatively stable condition, it is possible to cool cells almost to absolute zero without destruction

(Rajashekar 2000; Hopkins 2006; Jan et al. 2009).

Membranes are restructured under exposure to cold before the temperature drops below zero. In these conditions, the water potential is gradually lowered with a simultaneous drop in the osmotic potential due to the accumulation of carbohydrates in vacuoles. ABA is accumulated, and it induces the synthesis of specific proteins. The next stage brings intensified changes in the cell membrane - degradation of phosphatidylcholine and phosphoinositol, accompanied by a continued increase in ABA levels and protein synthesis modifications (Gusta et al. 2005; Lindberg et al. 2005). Cryoprotectants, substances that directly protect the membrane from damage, are also synthesized at this stage. Rigid membranes are less likely to be deformed during frost-induced dehydration, and they protect cells against freezing more effectively. This parameter is largely dependent on the sterol content of cells (Hopkins 2006; Janska et al. 2010).

In addition to membrane unsaturation, it appears that lipid asymmetry in the membrane also contributes to the physical structure of the membrane at low temperature (Gomès et al. 2000).

The mechanism protecting chloroplast membranes enables the plant to begin photosynthesis as soon as ambient temperature increases.

The cold resistance of plants is also determined by the following mechanisms:

- Thermal insulation which delays and minimizes heat loss, for example, shoot apices are often covered with dense foliage (rosette plant habit) or they winter under a layer of leaves or litter (geophytes). Frost tender organs are often rejected before the onset of very low temperatures (deciduous plants shed leaves in the fall). In high mountainous regions of tropical zones, the leaves of large rosette plants close above the tip at night to protect the interior from freezing (Hopkins 2006).
- Water freezing in intertissue spaces, for example, between the seed coat and the embryo or between bud scales, where extensive areas are covered with ice.

- 3. Cell structures are protected against excessive dehydration with an accompanying increase in the effectiveness of barriers that prevent ice crystals from propagating from the apoplast inside the cell. The following mechanisms are involved:
 - (a) Osmotic pressure increases to keep water inside the cell, and the water potential decreases due to the accumulation of osmotically active compounds (simple sugars and oligosaccharides, polyols, low-molecular-weight nitrogen compounds, such as selected amino acids) in vacuoles and hydrophilic proteins in the cytoplasm (Rorat 2006; Liang et al. 2009). The share of highly polar lipids in the membrane structure increases, such as phosphatidylcholine and phosphatidylethanolamine in the plasmalemma and cytoplasmic membranes or digalactosyldiacylglycerol in chloroplast membranes, which increases matrix interactions inside the cell.
 - (b) The membrane is enriched with more stable lipids containing polyunsaturated fatty acid residues, selected sterols, and cryoprotectants are accumulated in the cytoplasm to protect cell structures against strong dehydration (Lindberg et al. 2005; Zhang and Tian 2010). These substances stabilize membrane structure and prevent conformational protein changes. They counteract the accumulation of salt ions and selected organic acids in the cell, and they protect proteins against denaturation. Small proteins, whose synthesis is enhanced or induced under exposure to low temperatures, play a protective role. Some of them show significant homology to proteins synthesized in response to water stress, for example, to dehydrin (Rorat et al. 2006). The cell wall plays an important role in protecting the cell against the adverse consequences of dehydration, and it is the main barrier to ice penetration.

In addition to mechanisms responsible for resistance to the primary consequences of frost, cold-resistant plants develop acclimatization mechanisms that enable them to avoid secondary thermal stress at below zero temperatures, such as photoinhibition, draught, oxygen deficiency (under ice cover), or mechanical effects of ice load (Alcázar et al. 2011).

3 High Temperature

Heat stress occurs when a rise in temperature has negative consequences for a plant. It is a complex function of intensity (temperature in degrees), duration and the rate of temperature increase. For plants inhabiting very cold climates such as the Arctic, temperatures in the region of 15°C can already be a source of heat stress. In a temperate climate, heat stress takes place in the temperature range of 35-40°C. In scientific literature, heat stress denotes temperatures that exceed the optimum values by around 10-15°C (Larkindale et al. 2005). Plants can be divided into three groups, subject to their sensitivity to high temperature (Fig. 5.8). In geographic zones with a hot climate, in habitats marked by high fluctuations in daily temperature (soil surface, littoral zone, shallow waters) or seasonal fluctuations and in volcanic areas, temperature levels can be lethal for vascular land plants. High absorption of solar energy during windless weather can increase the temperature inside plant tissues in excess of the ambient temperature. Creeping grass shoots, the runners and tillers of young plants can also be subjected to heat stress. The lethal temperature range (thermal death point) is determined by the duration of tissue exposure to high temperature (Table 5.2). Only single-celled organisms can complete their life cycle during continued exposure to temperatures higher than 50°C, while only prokaryotic organisms can survive in temperatures higher than 60°C.

3.1 Consequences of Heat Exposure

At very high temperatures, severe cellular injury and even death may occur within minutes or even seconds (due to denaturation and/or aggregation of proteins), while at moderately high temperatures,



Fig. 5.8 Classification of the plants, subject to their sensitivity to high temperature (adapted from Stushnoff et al. 1984)

Table 5.2 The lethal temperature range (thermal death point) characteristic for varied types of plants

Type of plants	Thermal death point
Aquatic plants and plants growing in shaded habitats	38–42°C following several hours of exposure
Temperate plants with hydrated and metabolically active organs	45–55°C following several hours of exposure
Desert plants	Higher than 60–65°C for more than several hours per day

injuries or death may occur only after long-term exposure (due to disruptions in basic metabolic processes). The adverse effects of overheating are directly noticeable. The morphological symptoms of heat stress include scorching of leaves and twigs, sunburns on leaves, branches and stems, leaf senescence and abscission, shoot and root growth inhibition, fruit discoloration and damage, and reduced yield. Cell size reduction, closure of stomata, and curtailed water loss is observed at the tissue and cellular level. At the subcellular level, major modifications occur in chloroplasts (changing the structural organization of thylakoids, loss of grana stacking or its swelling) (Wahid et al. 2007; Mitra and Bhatia 2008). In vascular land plants, the negative consequences of elevated temperature are often related to secondary stress, namely, a negative water balance (leading to the perturbation of many physiological processes) due to intensive leaf transpiration during daytime.

Under field conditions, high temperature stress is frequently associated with reduced water availability (higher during daytime than at night). Heat stress may secondarily induce oxidative stress via the generation and the reactions of activated oxygen species (Xu et al. 2006; Almeselmani et al. 2006).

Metabolic pathways and processes show varied sensitivity to temperature which may result in a deficit or an excess of selected metabolites. It is generally believed that the processes taking place in membranes are most sensitive to temperature change. A heat-induced increase in membrane liquidity (either by denaturation of proteins or an increase in unsaturated fatty acids) and changes in reactions between lipid and protein components impair membrane functions (Savchenko et al. 2002; Wahid et al. 2007), including the functioning of ion and water channels, ion transporters, metabolite transport, energy generation, and other processes. Ion leakage from the cell is observed; photosynthesis and respiration are also impaired (Wahid et al. 2007; Wang et al. 2009). It has also been suggested that changing membrane fluidity plays a central role in sensing (plant thermometer) and influencing gene expression both under high and low temperatures (Plieth 1999). Photochemical reactions in thylakoid lamellae and carbon metabolism in the stroma of chloroplast have been suggested as the primary sites of injury at high temperatures (Yang et al. 2006a; Wang et al. 2009). Thylakoid membranes are particularly sensitive to high temperature, and this especially applies to photosystem II whose activity is greatly reduced or even partially stopped under high temperatures (Salvucci and Crafts-Brandner 2004; Camejo et al. 2005; Marchand et al. 2005). High temperature has a greater influence on the photosynthetic capacity of C3 plants than C4 plants (Crafts-Brandner and Salvucci 2002). Heat shock reduces the amount of photosynthetic pigments (Wang et al. 2009), soluble proteins, rubisco binding proteins (RBP), large-subunits (LS), and smallsubunits (SS) of rubisco in darkness but increases them in light (Kepova et al. 2005). Moreover, heat stress greatly affects starch and sucrose synthesis, as demonstrated by the reduced activity of sucrose phosphate synthase, ADP-glucose pyrophosphorylase, and invertase (Wahid et al. 2007; Sumesh et al. 2008). In any plant species, the ability to sustain leaf gas exchange under heat stress is directly correlated with heat tolerance. During the vegetative stage, high daytime temperature can cause damage to compensated leaf photosynthesis, reducing CO₂ assimilation rates (Crafts-Brander and Salvucci 2002; Morales et al. 2003). Photosynthesis is more sensitive to heat than dark respiration which could have additional consequences under prolonged stress, including the depletion of carbohydrate reserves and plant starvation (Sumesh et al. 2008). Heat stress rapidly increases selected phytohormone levels, including ABA, ethylene, and salicylic acid (SA), and it decreases cytokinin and gibberellin concentrations (Dat et al. 2000; Talanova et al. 2003; Larkindale and Huang 2004). The overlapping effects of the above changes in hormone levels speed-up plant aging.

3.2 Mechanism of Plant Resistance to High Temperature

Plants rely on two adaptation mechanisms to survive high temperatures: the ability to prevent excessive temperature growth in tissues or alleviate its effects and the heat tolerance of the protoplasm.

Survival in hot, dry environments can be achieved in a variety of ways, by combinations of adaptations (Fitter and Hay 2002). Plants growing in a hot climate avoid heat stress by reducing the absorption of solar radiation. This ability is supported by the presence of small hairs (tomentose) that form a thick coat on the surface of the leaf as well as cuticles, protective waxy covering. In such plants, leaf blades often turn away from light and orient themselves parallel to sun rays (paraheliotropism). Solar radiation may also be reduced by rolling leaf blades. Plants with small leaves are also more likely to avoid heat stress: they evacuate heat to ambient more quickly due to smaller resistance of the air boundary layer in comparison with large leaves. Plants rely on the same anatomical and physiological adaptive mechanisms that are deployed in a water deficit to limit transpiration. In well-hydrated plants, intensive transpiration prevents leaves from heat stress, and leaf temperature may be 6 K or even 10–15 K lower than ambient temperature. Many species have evolved life histories which permit them to avoid the hottest period of the year. This can be achieved by leaf abscission, leaving heatresistant buds, or in desert annuals, by completing the entire reproductive cycle during the cooler months (Fitter and Hay 2002). Such morphological and phonological adaptations are commonly associated with biochemical adaptations favoring net photosynthesis at high temperatures (in particular C₄ and CAM photosynthetic pathways), although C₃ plants are common in desert floras (Fitter and Hay 2002).

Heat tolerance is generally defined as the ability of the plant to grow and produce economic yield under high temperatures. This is a highly specific trait, and closely related species, even different organs and tissues of the same plant, may vary significantly in this respect. The above
is affected by climate conditions and the species' geographic origin. Plants native to cold regions (tundra, high mountain ranges) are much more sensitive to heat than temperate flora. The latter, in turn, are more susceptible to high temperatures than desert and tropical plants. The highest heat tolerance is demonstrated by selected sedge and grass species, mainly C₄ plants. Heat tolerance is associated with greater enzyme thermostability and a higher share of saturated fatty acids in membrane lipids which increases the lipid phase transition (melting) temperature and prevents a heat-induced increase in the membrane's liquidity. It is believed that PG is the phospholipid initiating phase transitions in thylakoid membranes. Heat tolerance leads to a rapid genome reaction even during short-term overheating. The biosynthesis of heat stress proteins (HSP) which prevent macroparticle denaturation is induced (Kotak et al. 2007; Al-Whaibi 2010). During exposure to high temperature, plants synthesize two groups of HSP: four high-molecular weight HSPs (HSP 100, HSP 90, HSP 70, HSP 60) and several lowmolecular weight HSPs (smHSPs). Those proteins remain stable over a certain period of time, and they are probably the main factor enabling plants to survive a temperature increase. HSPs are found in the cytoplasm and organelles such as the nucleus, mitochondria, chloroplasts, and endoplasmic reticulum. The tolerance conferred by HSPs results in improved physiological phenomena such as photosynthesis, assimilate partitioning, water and nutrient-use efficiency, and membrane stability. Those improvements make plant growth and development possible under heat stress (Wang et al. 2004). The HSPs/chaperones may be involved in stress signal transduction, gene activation, and the regulation of the cellular redox state. They also interact with other stress-response mechanisms such as the production of osmolytes and antioxidants (Kotak et al. 2007; Wahid et al. 2007; Al-Whaibi 2010). In heat-stressed plants, the induction of HSP synthesis inhibits the biosynthesis of other proteins. A plant's resistance to heat is determined by protein synthesis in cells that are lost with age. For this reason, aging organs (and organisms) have impaired ability to acclimatize to high

temperature. Smaller quantities of HSPs are also determined at optimal temperature, but in this environment, they play a different role than during and after stress. Under optimal conditions, HSPs regulate the formation of protein structures from newly emerged polypeptide strings to protect the cell from proteins that are nonfunctional due to synthesis "errors." At excessively high temperatures, HSPs minimize cell injuries by protecting cell proteins from denaturation and creating chelate bonds with ions leaking from the vacuoles into the cytosol (Kotak et al. 2007; Wahid et al. 2007; Al-Whaibi 2010). An increased content of ABA mediates the acclimation/adaptation of plants to desiccation by modulating the up- or downregulation of numerous genes (Talanova et al. 2003; Wahid et al. 2007). It is suggested that the induction of several HSPs (e.g., HSP70) is regulated by ABA (Snyman and Cronje 2008). Increased ethylene secretion at high temperatures leads to the abscission of reproductive organs; this is accompanied by both reduced levels and transport capacity of auxins to reproductive organs (Wahid et al. 2007). Among other hormones, SA has been suggested to be an important component of signaling pathways in response to systemic acquired resistance (SAR) and the hypersensitive response (HR) during heat stress (Kawano et al. 1998; Wang and Li 2006). Gibberellins and cytokinins have an opposite effect on high temperature tolerance than ABA. The potential roles of other phytohormones in plant thermotolerance are yet unknown (Wahid et al. 2007). Under stress, different plant species may accumulate a variety of osmolytes such as sugars and sugar alcohols (polyols), proline, tertiary and quaternary ammonium compounds, and tertiary sulphonium compounds (Singh and Grover 2008). The accumulation of such solutes may contribute to enhanced stress tolerance of plants, for example, proline and glycinebetaine may buffer the cellular redox potential under heat and other environmental stresses (Wahid and Close 2007); gama-4-aminobutyric acid (GABA) has a physiological role in the mitigation of stress effects (Kinnersley and Turano 2000). Hightemperature stress induces the production of phenolic compounds such as flavonoids and phenylpropanoids. The heat-induced increase in the activity of phenylalanine ammonia-lyase (PAL) is considered to be the cell's main acclamatory response to heat stress (Wahid and Ghazanfar 2006; Wahid 2007). Carotenoids of the xanthophylls family and selected terpenoids, such as isoprene or tocopherol, stabilize and photoprotect the lipid phase of thylakoid membranes during exposure to strong light and/or elevated temperatures (Wahid and Ghazanfar 2006; Wahid 2007). The expression of stress proteins is an important adaptive mechanism for environmental stress tolerance. Most stress proteins are soluble in water and, therefore, they contribute to stress tolerance, presumably by hydrating cellular structures (Wahid and Close 2007). Heat stress also induces the synthesis of other plant proteins, including ubiquitin (Sun and Callis 1997), cytosolic and chloroplasts Cu/Zn-SOD (Tang et al. 2006) and Mn-POD (Brown et al. 1993), cytosolic (Iba 2002) and chloroplasts APX (Tang et al. 2006), and other antioxidant enzymes (Sairam et al. 2000; Almeselmani et al. 2006), proteins of late embryogenesis abundant (LEA) (Goyal et al. 2005), and dehydrins. Their main function is to protect cellular and sub-cellular structures against oxidative damage and dehydrative forces.

3.3 Adaptation to High Temperature

Plants adapt to heat already after several hours of exposure to a temperature that evokes a stress response, but remains below the lethal temperature level (Xu et al. 2006). For most land plants, heat stress is triggered at temperatures slightly above 35°C, and in grasses at 38–40°C. The loss of resistance (dehardening) is a slower process that lasts several days in optimal growth conditions (Sung et al. 2003; Burke and Chen 2006). During acclimatization, the structure of the cell membrane changes by increasing the share of saturated fatty acids in the lipid layer. More unsaturated acyl residues are removed from the sn-2 position in a glycerolipid molecule by the respective hydrolases. They are replaced with saturated fatty acid residues (mostly 18-carbon chains) with the involvement of the respective

acetyltransferases and lipid transport proteins. At the current state of knowledge, it remains unknown whether a higher or a lower degree of membrane lipid saturation is beneficial for hightemperature tolerance (Klueva et al. 2001; Rahman et al. 2004). It is believed that the synthesis of HSP is also an effective mechanism protecting the plant from high temperature and other HSP synthesis-inducing stressors. In many plants, heat tolerance varies on a seasonal basis in view of their growth cycle and changes in seasonal temperature (Froux et al. 2004). During active growth, all plants are highly sensitive to temperature stress. Selected species of land plants increase their resistance to heat only in the summer, while others demonstrate the highest level of tolerance during winter dormancy. Dormant plants become resistant to stress upon reaching a developmental stage induced by factors other than high environmental temperature. In many land plant species, noticeable changes in heat tolerance are not observed. Due to the close correlation between drought and high temperature, the effects of each stressor on field-grown plants can be difficult to distinguish, and adaptations to arid environments can be effective only if they lead to avoidance or tolerance of both stresses (Fitter and Hay 2002).

4 Conclusion and Future Perspective

Temperature is a major environmental factor that changes from season to season and undergoes daily fluctuations. For this reason, the stressinducing role of temperature is difficult to define unambiguously since the response to various temperatures is determined by the plants' ability to adapt to different climate regimes. Plants exhibit a variety of responses to different (high and low) temperatures, which are depicted by symptomatic and quantitative changes in growth and morphology. However, it must be remembered the ability of the plant to cope with or adjust to the temperature stress varies across and within species as well as at different developmental stages. Generally heat or low temperature stress induces structural changes in tissues and cell organelles, disorganization of cell membranes, disturbance of leaf water relations, and impedance of photosynthesis by effects on photochemical and biochemical reactions and photosynthetic membranes. In response to temperature stress, plants manifest numerous adaptive changes. Metabolic pathways and processes show varied sensitivity to temperature which may result in a deficit or an excess of selected metabolites (such as HSPS, osmoprotectants, antioxidative enzymes, etc.). According to this, it is important to discover the induction of signaling cascades leading to profound changes in specific gene expression is also considered an important temperature-stress adaptation. Molecular knowledge of response and tolerance mechanisms will pave the way for engineering plants that can tolerate high or low temperatures and could be the basis for production of crops which can produce economic yield under temperature-stress conditions (Iba 2002; Wahid et al. 2007).

Although physiological mechanisms of heat and low temperature tolerance are relatively well understood, further studies are essential to determine physiological basis of assimilate partitioning from source to sink, plant phenotypic flexibility which leads to tolerance, and factors that modulate plant temperature-stress response. It is known that complex traits of abiotic stress phenomena in plants make genetic modification for efficient stress tolerance difficult to achieve. However, the modification of a single trait resulted in several cases in significant improvements in stress tolerance. By now, little is also known about the molecular mechanisms underlying signaling components during stress response and adaptation. It must be also remembered, alteration of further upstream molecules in the pathway often activates a much wider network of genes, other than stress-specific ones. The discovery and use of new stress-tolerance-associated genes, as well as heterologous genes, to confer plant stress tolerance (including those unique to extreme growth-environment organisms, e.g., halophytes, thermophilic organisms), has been the subject of ongoing efforts to obtain tolerant plants. It must be also remembered, plants receive

various stresses from their surrounding environment, which affect them in a complex manner (plant is usually subjected to many abiotic and biotic stresses at the same time). That means, it is necessary to identify the key strategies that plants use to deal with complex stresses of both biotic and abiotic origin (Iba 2002; Wahid et al. 2007).

References

- Alam BD, Nair B, Jacob J (2005) Low temperature stress modifies the photochemical efficiency, of a tropical tree species *Hevea brasiliensis*: effects of varying concentration of CO2 and photon flux density. Photosynthetica 43:247–252
- Alberts B, Bray D, Johnson A, Lewis J, Raff M, Roberts K, Walter P (2004) Essential cell biology, 2nd edn. Garland Science, New York
- Alcázar R, Cuevas JC, Planas J, Zarza X, Bortolotti C, Carrasco P, Salinasd J, Tiburcio AF, Altabella T (2011) Integration of polyamines in the cold acclimation response. Plant Sci 180:31–38
- Almeselmani M, Deshmukh PS, Sairam RK, Kushwaha SR, Singh TP (2006) Protective role of antioxidant enzymes under high temperature stress. Plant Sci 171:382–388
- Al-Whaibi MH (2010) Plant heat-shock proteins: a mini review. J King Saud Univ (Sci). doi:10.1016/ j.jksus.2010.06.022
- Amasino RM (2005) Vernalization and flowering time. Curr Opin Biotechnol 16:154–158
- Amasino RM (2006) Vernalization: remembering winter with an environmentally induced epigenetic switch. Dev Biol 295:323
- Baskin CC, Baskin JM (1998) Seeds. Ecology, biogeography, and evolution of dormancy and germination. Academic, San Diego
- Brown JA, Li D, Ic M (1993) Heat shock induction of manganese peroxidase gene transcription in *Phanerochaete chryosporium*. Appl Environ Microbiol 59:4295–4299
- Burbulis N, Kupriene R, Blinstrubiene A (2008) The effect of de-acclimation and re-acclimation treatments on winter rapeseed cold resistance *in vitro*, Scientific Works Of The Lithuanian Institute Of Horticulture And Lithuanian University Of Agriculture. Sodininkyste Ir Daržininkyste 27:233–240
- Burke J, Chen J (2006) Changes in cellular and molecular processes in plant adaptation to heat stress. In: Huang B (ed) Plant–environment interactions. CRC Press, New York
- Camejo D, Rodriguez P, Morales MA, Dellamico JM, Torrecillas A, Alarcon JJ (2005) High temperature effects on photosynthetic activity of two tomato cultivars with different heat susceptibility. J Plant Physiol 162:281–289

- Cornish-Bowden A (2004) Fundamentals of enzyme kinetics, 3rd edn. Portland Press, London
- Crafts-Brander C, Salvucci ME (2002) Sensitivity to photosynthesis in the C4 plant, maize to heat stress. Plant Cell 12:54–68
- Crafts-Brandner SJ, Salvucci ME (2002) Sensitivity of photosynthesis in a C4 plant, maize, to heat stress. Plant Physiol 129:1773–1780
- Criddle RS, Hopkin MS, McArthur ED, Hansen LD (1994) Plant distribution and the temperature coefficient of metabolisms. Plant Cell Environ 17:233–243
- Dat JF, Lopez-Delgado H, Foyer CH, Scott IM (2000) Effects of salicylic acid on oxidative stress and thermotolerance in tobacco. J Plant Physiol 156:659–665
- Dennis ES, Finnegan EJ, Bilodeau P, Chaudhury A, Genger R, Helliwell CA, Sheldon CC, Bagnall DJ, Peacock WJ (1996) Vernalization and the initiation of flowering. Cell Dev Biol 7:441–448
- Fitter AH, Hay RKM (2002) Environmental physiology of plants, 3rd edn. Academic, London
- Froux F, Ducrey M, Epron D, Dreyer E (2004) Seasonal variations and acclimation potential of the thermo stability of photochemistry in four Mediterranean conifers. Ann For Sci 61:235–241
- Gomès E, Jakobsen MK, Axelsen KB, Geisler M, Palmgreen MG (2000) Chilling tolerance in Arabidopsis involves ALA1, a member of a new family of putative aminophospholipid translocases. Plant Cell 12:2441–2453
- Goyal K, Walton LJ, Tunnacliffe A (2005) LEA proteins prevent protein aggregation due to water stress. Biochem J 388:151–157
- Gusta LV, Trischuk R, Weiser CJ (2005) Plant cold acclimation: the role of abscisic acid. J Plant Growth Regul 24:308–318
- Hekneby M, Antolin MC, Sanchez-Diaz M (2006) Frost resistance and biochemical changes during cold acclimation in different annual legumes. Environ Exp Bot 55:305–314
- Hopkins WG (2006) The green word. Plant development. Chelsea House an Imprint of Infobase Publishing, New York
- Iba K (2002) Acclimative response to temperature stress in higher plants: approaches of gene engineering for temperature tolerance. Annu Rev Plant Biol 53:225–245
- Jan N, Mahboob-ul H, Andrabi KI (2009) Cold resistance in plants: a mystery unresolved. Electronic J Biotechnol 12. doi:10.2225/vol12-issue3-fulltext-3
- Janska A, Marsik P, Zelenkova S, Ovesna J (2010) Cold stress and acclimation – what is important for metabolic adjustment? Plant Biol 12:395–405
- Jenks MA, Hasegawa PM (eds) (2005) Plant abiotic stress. Blackwell, Oxford
- Kacperska A (1998) Plant responses to stress factors. In: Kopcewicz J, Lewak S (eds) Basis of plant physiology. Wydawnicto Naukowe PWN, Warszawa, pp 575–633
- Kartsch HA, Wise RR (2000) The ultrastructure of chilling stress. Plant Cell Environ 23:337–350

- Kato-Noguchi H (2007) Low temperature acclimation to chilling tolerance in rice roots. Plant Growth Regul 51:171–175
- Kawano T, Sahashi N, Takahashi K, Uozumi N, Muto S (1998) Salicylic acid induces extracellular superoxide generation followed by an increase in cytosolic calcium ion in tobacco suspension culture: the earliest events in salicylic acid signal transduction. Plant Cell Physiol 39:721–730
- Kepova KD, Holzer R, Stoilova LS, Feller U (2005) Heat stress effects on ribulose-1,5-bisphosphate carboxylase/oxygenase, Rubisco bindind protein and Rubisco activase in wheat leaves. Biol Plant 49:521–525
- Kim DH, Doyle MR, Sung S, Amasino RM (2009) Vernalization: winter and the timing of flowering in plants. Annu Rev Cell Dev Biol 25:277–299
- Kinnersley AM, Turano FJ (2000) Gamma aminobutyric acid (GABA) and plant responses to stress. Crit Rev Plant Sci 19:479–509
- Klueva NY, Maestri E, Marmiroli N, Nguyen HT (2001) Mechanisms of thermotolerance in crops. In: Basra AS (ed) Crop responses and adaptations to temperature stress. Food Products Press, Binghamton, pp 177–217
- Kotak S, Larkindale J, Lee U, von Koskull-Doring P, Vierling E, Scharf K-D (2007) Complexity of the heat stress response in plants. Curr Opin Plant Biol 10:310–316
- Kumar A, Bhatla SC (2006) Polypeptide markers for low temperature stress during seed germination in sunflower. Biol Plantarum 50:81–86
- Larkindale J, Huang B (2004) Thermo-tolerance and antioxidant systems in Agrostis stolonifera: involvement of salicylic acid, abscisic acid, calcium, hydrogen peroxide, and ethylene. J Plant Physiol 161:405–413
- Larkindale J, Mishkind M, Vierling E (2005) Plant responses to high temperature. In: Jenks MA, Hasegawa PM (eds) Plant abiotic stress. Blackwell, Oxford, pp100–144
- Lecourieux D, Ranjeva R, Pugin A (2006) Calcium in plant defence-signalling pathways. New Phytol 171:249–269
- Levine RD (2005) Molecular reaction dynamics. Cambridge University Press, Cambridge
- Levitt J (1980) Responses of plants to environment al stress. Vol. I. Chilling, freezing, and high temperature stresses, 2nd edn. Academic, New York
- Li Ch, Junttila O, Palva ET (2004) Environmental regulation and physiological basis of freezing tolerance in woody plants. Acta Physiol Plant 26:213–222
- Li C, Wu N, Liu S (2005a) Development of freezing tolerance in different altitudinal ecotypes of *Salix paraplesia*. Biol Plantarum 49:65–71
- Li Ch, Yongqing Y, Junttila O, Palva ET (2005b) Sexual differences in cold acclimation and freezing tolerance development in sea buckthorn (Hippophae rhamnoides L.) ecotypes. Plant Sci 168:1365–1370
- Liang W, Wang M, Ai X (2009) The role of calcium in regulating photosynthesis and related physiological indexes of cucumber seedlings under low light intensity and suboptimal temperature stress. Sci Hortic 123:34–38

- Lindberg S, Banas A, Stymne S (2005) Effects of different cultivation temperatures on plasma membrane ATPase activity and lipid composition of sugar beet roots. Plant Physiol Biochem 43:261–268
- Los DA, Murata N (2004) Membrane fluidity and its roles in the perception of environmental signals. Biochim Biophys Acta 1666:142–157
- Lyons JM (1973) Chilling injury in plants. Annu Rev Plant Physiol 24:445–466
- Mahajan Sh, Tuteja N (2005) Cold, salinity and drought stresses: an overview. Arch Biochem Biophys 444:139–158
- Marchand FL, Mertens S, Kockelbergh F, Beyens L, Nijs I (2005) Performance of high arctic tundra plants improved during but deteriorated after exposure to a simulated extreme temperature event. Glob Change Biol 11:2078–2089
- Marschner H (1995) Mineral nutrition of higher plants, 2nd edn. Academic, New York
- Minorsky PV (1989) Temperature sensing by plants: a review and hypothesis. Plant Cell Environ 12:119–135
- Mitra R, Bhatia CR (2008) Bioenergetic cost of heat tolerance in wheat crop. Curr Sci 94:1049–1053
- Morales D, Rodriguez P, Dellamico J, Nicolas E, Torrecillas A, Sanchez-Blanco MJ (2003) High-temperature preconditioning and thermal shock imposition affects water relations, gas exchange and root hydraulic conductivity in tomato. Biol Plant 47:203–208
- Muller O, Hikosaka K, Hirose T (2005) Seasonal changes in light and temperature affect the balance between light harvesting and light utilization components of photosynthesis in an evergreen understory shrub. Oecologia 143:501–508
- Ohnishi S, Miyoshi T, Shirai S (2010) Low temperature stress at different flower developmental stages affects pollen development, pollination, and pod set in soybean. Environ Exp Bot 69:56–62
- Opik H, Rolfe S (2005) Reproductive development. In Öpik H, Rolfe S (eds) The physiology of flowering plants, 4th edn. Cambridge University Press, Cambridge UK, pp 270–317
- Plieth C (1999) Temperature sensing by plants: calciumpermeable channels as primary sensors – a model. J Membr Biol 172:121–127
- Price N, Stevens L (1999) Fundamentals of enzymology: cell and molecular biology of catalytic proteins. Oxford University Press, Oxford
- Rahman H, Malik SA, Saleem M (2004) Heat tolerance of upland cotton during the fruiting stage evaluated using cellular membrane thermostability. Field Crops Res 85:149–158
- Rajashekar CB (2000) Cold response and freezing tolerance in plants, In: Wilkinson RE (ed) Plant– environment interactions, 2nd edn. Marcel Dekker, Inc. New York USA, pp 321–342
- Rorat T (2006) Plant dehydrins tissue location, structure and function. Cell Mol Biol 11:536–556
- Rorat T, Szabala BM, Grygorowicz WJ, Wojtowicz B, Zhimin Yin Z, Rey P (2006) Expression of SK3-type

dehydrin in transporting organs is associated with cold acclimation in Solanum species. Planta 224:205–221

- Ruelland E, Cantrel C, Gawer M, Kader JC, Zachowski A (2002) Activation of phospholipases C and D is an early response to a cold exposure in *Arabidopsis* suspension cells. Plant Physiol 130:999–1007
- Sairam RK, Srivastava GC, Saxena DC (2000) Increased antioxidant activity under elevated temperature: a mechanism of heat stress tolerance in wheat genotypes. Biol Plant 43:245–251
- Salvucci ME, Crafts-Brandner SJ (2004) Inhibition of photosynthesis by heat stress: the activation state of Rubisco as a limiting factor in photosynthesis. Physiol Plant 120:179–186
- Savchenko GE, Klyuchareva EA, Abrabchik LM, Serdyuchenko EV (2002) Effect of periodic heat shock on the membrane system of etioplasts. Russ J Plant Physiol 49:349–359
- Sheldon CC, Finnegan EJ, Rouse DT, Tadege M, Bagnall DJ, Helliwell CA, Peacock WJ, Dennis ES (2000) The control of flowering by vernalization. Curr Opin Plant Biol 3:418–422
- Singh A, Grover A (2008) Genetic engineering for heat tolerance in plants. Physiol Mol Biol Plant 14:155–166
- Snyman M, Cronje MJ (2008) Modulation of heat shock factors accompanies salicylic acid-mediated potentiation of Hsp70 in tomato seedlings. J Exp Bot 59:2125–2132
- Solanke AU, Sharma AK (2008) Signal transduction during cold stress in plants. Physiol Mol Biol Plants 14:69–79
- Stushnoff C, Flower DB, Brule-Babel A (1984) Breeding and selection for resistance to low temperature. In: Vose PB (ed) Plant breeding – a contemporary basis. Pergamon Press, Oxford, pp 115–136
- Sumesh KV, Sharma-Natu P, Ghildiyal MC (2008) Starch synthase activity and heat shock protein in relation to thermal tolerance of developing wheat grains. Biol Plantarum 52:749–753
- Sun CW, Callis J (1997) Independent modulation of Arabidopsis thaliana polyubiquitin m RNAs in different organs of and in response to environmental changes. Plant J 11:1017–1027
- Sun W-H, Duan M, Li F, Shu D-F, Yang S, Meng Q-W (2010) Overexpression of tomato *tAPX* gene in tobacco improves tolerance to high or low temperature stress. Biol Plantarum 54:614–620
- Sung D, Kaplan F, Lee K, Guy CL (2003) Acquired tolerance to temperature extremes. Trends Plant Sci 8:179–187
- Suzuki N, Mittler R (2006) Reactive oxygen species and temperature stresses: a delicate balance between signaling and destruction. Phisiol Plant 126:45–51
- Talanova VV, Akimova TV, Titov AF (2003) Effect of whole plant and local heating on the ABA content in cucumber seedling leaves and roots and on their heat tolerance. Russ J Plant Physiol 50:90–94
- Tang L, Kwon S-Y, Kim S-H, Kim J-S, Choi JS, Cho KY, Sung CK, Kwak S-S, Lee H-S (2006) Enhanced tolerance of transgenic potato plants expressing both

superoxide dismutase and ascorbate peroxidase in chloroplasts against oxidative stress and high temperature. Plant Cell Rep 25:1380–1386

- Thakur P, Kumar S, Malik JA, Berge JD, Nayyar H (2009) Cold stress effects on reproductive development in grain crops: an overview. Environ Exp Bot 67:429–443
- Tretyn A, Głowacka K, Galoch E, Kasprzak K, Kulikowska-Gulewska H, Kopcewicz J (2003) Investigation on the chemical nature of flowerinducing factor(s) in short-day plant Pharbitis nil. In: Machackova I (ed) Phytohormones in plant biotechnology and agriculture. Kluwer, Dordrecht, pp 67–78
- Wahid A (2007) Physiological implications of metabolites biosynthesis in net assimilation and heat stress tolerance of sugarcane sprouts. J Plant Res 120:219–228
- Wahid A, Close TJ (2007) Expression of dehydrins under heat stress and their relationship with water relations of sugarcane leaves. Biol Plant 51:104–109
- Wahid A, Ghazanfar A (2006) Possible involvement of some secondary metabolites in salt tolerance of sugarcane. J Plant Physiol 163:723–730
- Wahid A, Gelani S, Ashraf M, Foolad MR (2007) Heat tolerance in plants: an overview. Environ Exp Bot 61:199–223
- Wang L-J, Li S-H (2006) Thermotolerance and related antioxidant enzyme activities induced by heat acclimation and salicylic acid in grape (Vitis vinifera L.) leaves. Plant Growth Regul 48:137–144
- Wang W, Vinocur B, Altman A (2003) Plant responses to drought, salinity and extreme temperatures: towards genetic engineering for stress tolerance. Planta 218:1–14
- Wang W, Vinocur B, Shoseyov O, Altman A (2004) Role of plant heat-shock proteins and molecular chaperones in the abiotic stress response. Trends Plant Sci 9:244–252

- Wang X, Li W, Li M, Welti R (2006) Profiling lipid changes in plant response to low temperatures. Physiol Plant 126:90–96
- Wang JZ, Cui LJ, Wang Y, Li JL (2009) Growth, lipid peroxidation and photosynthesis in two tall fescue cultivars differing in heat tolerance. Biol Plantarum 53:237–242
- Weiser CJ (1970) Cold resistance and injury in woody plants. Science 169:1269–1278
- Wilkinson RE (ed) (2000) Plant–environment interactions, 2nd edn. Marcel Dekker, Inc. New York USA
- Wróbel M, Karamac M, Amarowicz R, Frączek E, Weidner S (2005) Metabolism of phenolic compounds in Vitis riparia seeds during stratification and during germination under optimal and low temperature stress conditions. Acta Physiol Plant 27:313–320
- Xin Z, Browse J (2000) Cold comfort farm: the acclimation of plants to freezing temperatures. Plant Cell Environ 23:893–902
- Xu S, Li J, Zhang X, Wei H, Cui L (2006) Effects of heat acclimation pretreatment on changes of membrane lipid peroxidation, antioxidant metabolites, and ultrastructure of chloroplasts in two cool-season turfgrass species under heat stress. Environ Exp Bot 56:274–285
- Yang X, Chen X, Ge Q, Li B, Tong Y, Zhang A, Li Z, Kuang T, Lu C (2006a) Tolerance of photosynthesis to photoinhibition, high temperature and drought stress in flag leaves of wheat: a comparison between a hybridization line and its parents grown under field conditions. Plant Sci 171:389–397
- Yoshida S, Hotsubo Y, Kawamura Y, Murai M, Arakawa K (1999) Alternations in intracellular pH in response to low temperature stress. J Plant Res 112:225–236
- Zhang Ch, Tian Sh (2010) Peach fruit acquired tolerance to low temperature stress by accumulation of linolenic acid and N-acylphosphatidylethanolamine in plasma membrane. Food Chem 120:864–872

Responses and Management of Heat Stress in Plants

6

Abdul Wahid, Muhammad Farooq, Iqbal Hussain, Rizwan Rasheed, and Saddia Galani

Abstract

On the eve of global climate change, temperature increase, is the most evident phenomenon. This temperature increase is posing severe threat for sustainable crop production in many countries across the globe in the form of heat stress. Plants respond in many ways to the prevailing high temperature environment, and several inter- and intraspecific differences are reported. Heat stress produces quite tangible changes at cell, tissue, and organ levels. Photosynthetic acclimation to heat stress, synthesis and accumulation of primary and secondary metabolites, induction of stress proteins are among the major adaptive responses to heat stress. The important genes expressed in response to heat stress include heat shock protein (hsp) genes, dehydrins (dhn), senescence-associated (sag) genes, staygreen (sgr) genes. As mechanisms of heat stress tolerance, plants display the maintenance of membrane stability, scavenging of ROS, production of enzymatic and nonenzymatic antioxidants and adjustment of compatible solutes. Plant thermotolerance can be improved by various means; major being the mass screening and morphological and biochemical markersassisted selection, identification, and mapping of QTLs conferring heat resistance, conventional and molecular breeding, and exogenous use of

A. Wahid (⊠) • I. Hussain Department of Botany, University of Agriculture, Faisalabad 38040, Pakistan e-mail: drawahid2001@yahoo.com

M. Farooq Department of Agronomy, University of Agriculture, Faisalabad 38040, Pakistan

R. Rasheed Biology Department, Foreman Christian College, Lahore, Pakistan

S. Galani Khan Institute of Biotechnology and Genetic Engineering, University of Karachi, Karachi, Pakistan

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_6, © Springer Science+Business Media, LLC 2012

osmoprotectants and stress-signaling agents. Although pretty well understood, more research efforts are required to understand novel aspects of heat tolerance including molecular cloning and characterization of genes/ proteins and understanding the basis of growth improvements with seed pretreatments and plant acclimations. In this chapter, we discuss the plant responses to high temperature stress and integrated approaches, such as genetics, breeding and management options to improve the resistance in plants against heat stress.

Keywords

Heat stress • Metabolites • Molecular breeding • Osmolytes • Pretreatment strategies

1 Introduction

Plant growth and development throughout the globe is controlled, in one way or the other, by the prevailing environmental conditions. Abiotic stresses, including temperature extremes, salinity and drought, are serious intimidation to the sustainability and productivity of economic plants. Current climatic model predicts that global air temperature may increase by 1.1-6.4°C with doubling of atmospheric CO₂ (Kim et al. 2007; Lobell and Field 2007). According to Intergovernmental Panel on Climatic Change (IPCC), there was an increase of 0.5°C during the past 100 years in ambient temperature, which is expected to rise by 0.2°C per decade for the next two decades and 1-3.4°C per decade warmer in the year 2100 (IPCC 2007). Such a rise in the global temperatures greatly influences the agricultural production, specifically in terms of aggravating the associated effects of salinity, drought, mineral toxicity stresses or as the case may be.

Global warming is negatively affecting the agricultural activities. On exposure to high temperature, crop yield is decreased because of shortened life cycle and accelerated senescence in different agro-climatic zones (Porter 2005). Higher temperatures, either of days and nights or of soil and air, hamper plant growth or cause considerable pre- and postharvest losses (Hall 2001). Injuries occur due to short- and long-term exposure to high temperature. Severe heat stress for

short-terms can permanently damage to cells and tissues due to calamitous collapse of cellular organization (Schöffl et al. 1998). Long-term effects may be reduction in the size of tissues and organs and hampered morphological development (Gilani 2007; Rasheed 2009).

Heat stress substantially affects yields of many economically important cereals such as wheat, rice, maize, etc., and effects are quite often registered at the reproductive stages. Photosynthesis is the fundamental basis for carbon accumulation, growth and biomass yield in plants. Photosynthetic response of terrestrial plants can potentially change ecosystem carbon balance and cycling under global warming (Gunderson et al. 2000). Increased ambient temperature affects plant productivity by damaging photosynthesis (Al-Khatib and Paulsen 1990). According to Berry and Bjorkman (1980), at moderately higher temperatures suppression of photosynthetic rate is reversible; nonetheless upon exposure to extremely high temperature, the whole system of photosynthesis may be permanently damaged. It may also decrease chlorophyll content, net photosynthetic rate, and stomatal conductance (Morales et al. 2003). Likewise, upon exposure to severe heat stress, net photosynthesis is substantially inhibited due to impaired supply of Rubulose-1, 5-biphosphate (RUBP) (Law and Crafts-Brandner 1999).

Sudden heat stress may either denature the membrane proteins or increase the unsaturated fatty acids, leading to increased ion-leakage and thus the loss of cellular functions (Savchenko et al. 2002). Such damages occur due to production of activated (AOS) and reactive oxygen species (ROS) and dehydration-induced changes in phase transitions due to high temperature (Nishida and Murata 1996; Liu and Huang 2000). Both the AOS and ROS react with pigments, membranes, enzymes, and nucleic acids, and change their structure and functions (Smirnoff 1993; Scandalios 1993; Sairam et al. 2000). Electrolyte leakage, a measure of stress injury to membranes, varies in relation to membrane abilities to take up and retain solutes and reflect stress-induced changes in their potential. Studies indicated that finding genotypic variability for heat tolerance based on leaf electrolyte leakage might be more effective to screen plants for relatively hot areas (Li et al. 1991; Rahman et al. 2004; Wahid and Shabbir 2005). Heat stress decreases the activities of antioxidant enzymes leading to increase in injury to cell membranes by lipid peroxidation and leaf senescence (Liu and Huang 2000).

Heat stress tolerance in plants is an intricate phenomenon involving a great variety of response, mechanisms, and management practices. Determination of responses and possible strategies to improve heat stress tolerance is important to grow crop plants in the heat-stressed areas of the world. In this perspective, this chapter presents explicit responses and some pragmatic management options to avert the high temperature stress effects in plants.

2 High Temperature Stress: Responses

Plants subjected to heat stress show a range of responses and manifest mechanisms to cope with its adversaries. These changes can be discerned at whole plant to subcellular and molecular levels. An individualistic account of all these responses is briefly described below.

2.1 Growth and Phenology

High temperature is a major determinant of agricultural production throughout the world and its effects are evident at all critical growth stages starting from seed germination to final yield harvested. An account of changes in the phenology of plants has been described in the following lines.

Seeds put to germinate at supraoptimal temperatures show reduced or even inhibited germination. In soybean, heat stress changed the protein expression profiles and reduced the seed germination and seedling vigor, which appeared to determine the seed quality attributes (Egli et al. 2005; Ren et al. 2009). Seed germination, seedling emergence and its establishment is prone to increased temperature in most of the plant species (Grass and Burris 1995; Burke 2001; Ashraf and Hafeez 2004; Wahid et al. 2007). Columbo and Timmer (1992) demonstrated that black spruce plant seedlings are more susceptible to high temperature stress than adult plants. Maize shows optimal germination and growth at 20-30°C and 28-31°C, respectively (Hughes 1979; Medany et al. 2007; Farooq et al. 2008a, b, 2009).

There are conflicting reports about the postemergence seedling growth in maize under heat stress. For instance, some studies show that maize coleoptile was more heat tolerant at all stages of seedling development (Venter et al. 1997; Momcilovic and Ristic 2007), while in some other studies on maize, upon exposure to 40°C, there was a substantial reduction in coleoptile growth and at 45°C growth was completely stopped (Weaich et al. 1996, Akman 2009). Heat stress lowered the activity of specific enzymes and thus reduced the protein synthesis in germinating maize embryos (Riley 1981). Likewise, seedling growth and development of cotton (Gossypium hirsutum L.) was also reduced under heat stress (Mahan and Mauget 2005).

High temperature is a major environmental factor that determines the sustainability of crop growth and yield in some regions (Blum 1988; Al-Khatib and Paulsen 1999). Plants grown in warmer environments have much lower biomass than those grown at optimum or lower temperature (Kim et al. 2007). High temperature reduced the plant growth by affecting different mechanisms (Sibley et al. 1999; Wollenweber et al. 2003). For example, it decreased the dry weight,

Genotype	Control	Hat stress	Genotype x temperature
Grains per spike	Connor		Centryperitemperature
Bread wheat mean	70±0.85	70 ± 0.88	ns
Golia 69	69±1.31 a	70±1.43 a	ns
Sever	71±1.10 a	71±1.05 a	ns
Durum wheat mean	63±0.75	64 ± 0.74	ns
Acalou	66±0.98 a	65±0.95 a	ns
TE 9306	60±0.99 b	63±1.15 a	ns
Individual grain weight (mg)			
Bread wheat mean	56.54 ± 1.08	48.73 ± 0.90	***
Golia 69	47.11±0.84 a	43.75±0.90 a	**
Sever	64.67±1.03 b	53.53±1.23 b	***
Durum wheat mean	72.06±0.60	59.97 ± 0.61	***
Acalou	72.42±0.70 a	57.69±0.86 a	***
TE 9306	71.74±0.95 a	62.55±0.70 b	***
Yield per spike (g)			
Bread wheat mean	3.97 ± 0.09	3.43 ± 0.07	**
Golia 69	3.26±0.08 a	3.06±0.09 a	ns
Sever	4.58±0.11 b	3.78±0.09 b	***
Durum wheat mean	4.52 ± 0.06	3.84 ± 0.06	***
Acalou	4.76±0.08 a	3.77±0.08 a	***
TE 9306	4.29±0.09 b	3.92±0.08 a	**

Table 6.1 Effect of heat stress on yield and yield components of the bread and durum wheat genotypes

ns nonsignificant, significant at the **0.01, and ***0.001 levels of probability, respectively; for each *Triticum* species, different letters in the same column refer to significant differences between genotypes. *Source*: Dias and Lidon (2009)

growth and net assimilation rates of shoot (Wahid 2007). Likewise, heat shock affected the meristematic activity and reduced the growth of various parts mainly the leaves (Salah and Tardieu 1996). Applied heat stress arrested the cell wall elongation and altered cell differentiation (Potters et al. 2007).

Reproductive growth is more critically affected by the prevailing high temperature stress during anthesis and seed growth. Pollination is especially sensitive to heat stress. The mature pollens are more sensitive, and quite often fail to fertilize (Dupuis and Dumas 1990). Heat stress interferes with the development of pollen mother cell and microspore and causes male sterility (Sakata et al. 2000; Sato et al. 2006; Abiko et al. 2005). In the event of successful pollination, heat stress affected the kernel development in maize (Monjardino et al. 2005) and reduced the kernel density and reproductive growth in maize, wheat and *Suneca* during kernel development (Wilhelm et al. 1999; Maestri et al. 2002). Dias and Lidon (2009) did not find any effect of heat stress on number of grains per spike in both durum and bread wheat; nonetheless upon exposure to heat stress during grain growth, grain size was substantially reduced in both bread and durum wheats (Table 6.1). Likewise, heat stress also reduced the grain yield in both wheats (Table 6.1). However, different genotypes responded variably in terms of grain size and grain yield and a strong relationship between genotypes and temperature has been observed (Table 6.1).

High temperature also results in the boll and flower bud abortion in cotton, pea, and brassica, possibly owing to limited water supply and nutrients during reproductive development (Hall 1992; Guilioni et al. 1997; Young et al. 2004). During seed development, heat stress was found to affect seed storage process and kernel quality like starch and protein metabolism during grain filling in maize (Wilhelm et al. 1999; Maestri et al. 2002).



Fig. 6.1 Hypocotyl elongation phenotype of different pea varieties for 36 h in the dark at 22°C after heat acclimation and stress [(a) local variety, (b) Shandong variety, (c) Taiwan variety]. *Left* is expressed as control (22°C); *middle*

is expressed as the germinated seeds were acclimated at 37° C for 1 h, followed at 22° C for 1 h, and then stressed at 48° C for 2 h. *Right* is expressed as the seeds were stressed at 48° C for 2 h. After Tian et al. (2009) with permission

2.2 Anatomical and Developmental Responses

Like other abiotic stresses, heat stress brings about quite a few morphogenetic and histological modifications. At whole plant level, generally cell size is reduced (Santarius 1973; Berry and Bjorkman 1980). There may be several morpho-anatomical modifications in cells and tissues such as increased densities of stomata and trichomes and greater xylem vessels area of shoot and root in Lotus creticus seedlings (Banon et al. 2004). On exposure of grapes to heat stress, cell membrane permeability was substantially increased and mesophyll cells were severely damaged (Zhang et al. 2005). High temperature also causes various changes at subcellular level. For instance, in chloroplast, it changed the thylakoids structure in maize (Karim et al. 1997) and resulted in loss of swelling and stacking of grana (Gounaris et al. 1984). In grapes, heat stress damaged the mesophyll cells, which showed

round-shaped chloroplasts, swollen stroma lamellae, badly affected the antenna complex of photosystem (PS) II (Carpentier 1999), clump formation of vacuolar contents, disrupted cristae and deformed mitochondria (Zhang et al. 2005).

Heat stress restricted the emergence and elongation of hypocotyls in three pea (*Pisum sativum*) varieties (Tian et al. 2009). Nonetheless, heat acclimation for 1 h at 37°C improved the germination and hypocotyl development (Fig. 6.1). In the sprouting buds of sugarcane, heat stress badly affected the differentiation of various cells and tissues (Rasheed 2009). Here the major changes were noted on mesophyll cell expansion and development of vascular connections (Fig. 6.2).

2.3 Physiological and Metabolic Responses

In hot environments, plants exhibit various physiological and metabolic responses. The most



Fig. 6.2 Sprouting bud of sugarcane under control condition (*left*). Effect of heat stress on the histological changes in sprouting buds of sugarcane after 36 h of exposure (*middle*) and role of proline in mitigating heat stress effect (*right*). *Source*: Rasheed (2009)

important of those may be the changes in carbon fixation, oxidative stress; tissue water status and metabolites accumulation. These responses are briefly discussed below.

2.3.1 Photosynthesis

Heat stress causes photosynthetic acclimation and alters the physiological processes directly and changes the developmental patterns indirectly (Downton and Slatyer 1972). All the steps, processes, and aspects of photosynthesis are prone ambient temperature to increased (Al-Khatib and Paulsen 1990). The photosynthesis in C₃ plants is more affected by high temperature than C_4 plants (Wahid and Rasul 2005). The maize seedling grown at 25°C and transferred to 35°C for 20 min led to 50% inhibition in photosynthesis (Sinsawat et al. 2004). Maize showed maximum net photosynthesis near 31°C, decreased at temperature above 37°C and was completely inhibited near 45°C (Crafts-brandner and Salvucci 2002). Heat stress diminished the net photosynthetic (Pn) and stomatal conductance substantially in many plant species (Ranney and Peet 1994; Crafts-Brandner and Salvucci 2002; Morales et al. 2003); in this regard, Pn in developed leaves was more sensitive than mature leaves (Karim et al. 1997, 1999).

Photosynthetic apparatus are highly sensitive to heat temperature and inhibited when leaf temperature exceed 38°C in most plants (Edwards and Walker 1983). PS-II, water splitting and oxygen evolving complex (OEC) in photosynthesis are more heat-sensitive components of photosynthesis (Havaux 1993; Pastenes and Horton 1996a; Heckathorn et al. 1998a). Extensive studies show that both PS-I and PS-II are damaged by increased temperature. In barley and potato, heat stress damaged PS-I and PS-II and affected electron transport (Havaux 1998; Szilvia et al. 2005). Heat stress damaged the antenna complex of PS-II and reduced photosynthetic behavior (Carpentier 1999; Rokka et al. 2000; Zhang et al. 2005). High temperature during greening led to the inactivation of PS-I and PS-II (Sasmita and Narendranath 2002). High temperature increased chlorophyll a:b ratio and decreased chlorophyll:carotenoid ratio in sugarcane (Wahid 2007).

High temperature alters the energy sharing by changing the action of Calvin cycle and other metabolic processes such as photorespiration, synthesis and stability of the Rubisco enzyme (Holaday et al. 1992; Pastenes and Horton 1996b), disruption of electron transport activity and bound RUBP supply by heat stress (Ferrar et al. 1989). Extreme temperature reduced the activation state of Rubisco enzyme in the exposed leaf tissue and increased the RUBP (Feller et al. 1998; Crafts-Brandner and Law 2000), which inhibited photosynthesis as compared to control plants (Sharkey et al. 2001). High temperature enhances chlorophyllase activity and decreases the quantities of photosynthetic pigments (Todorov et al. 2003). The loss of chlorophyll is good indicator of heat tolerance in wheat (Ristic et al. 2007, 2008). High temperature modifies the activities of carbon metabolism enzymes,

especially the Rubisco (Ferrar et al. 1989; Holaday et al. 1992; Pastenes and Horton 1996a, b). Moreover, activities of starch and sucrose synthesis enzymes are greatly influenced (Chaitanya et al. 2001; Vu et al. 2001).

2.3.2 Reactive Oxygen Species and Oxidative Damage

Like other abiotic stresses, heat stress evokes the ROS generation including hydrogen peroxide (H_2O_2) , superoxide radical (O^{2-}) , singlet oxygen $({}^{1}O_{2})$ and hydroxyl radical (OH⁻), and induces oxidative stress (Mittler 2002; Taiz and Zeiger 2006; Potters et al. 2007). Chloroplast and mitochondria are the major sites where superoxide radicals are regularly produced, whereas some quantities are also produced in microbodies. Principally, ROS causes peroxidation of membrane lipids, destruction of pigments, and modification of membrane functions (Xu et al. 2006). The OH⁻ appears to be more damaging than other ROS, which is formed with the combination of O_2^- and H_2O_2 in the presence of Fe²⁺ and Fe³⁺ in trace amounts in Haber-Weiss reaction (Apel and Hirt 2004). The OH⁻ is greatly damaging to chlorophyll, proteins, lipids, DNA, and other important macromolecules (Sairam and Tyagi 2004).

Tolerant plants have the tendency to protect themselves from the damaging effects of ROS with the synthesis of various antioxidant systems (Apel and Hirt 2004). This protection starts with the conversion of O^{2-} by superoxide dismutase (SOD) into H_2O_2 , with the help of ascorbate peroxidase (APX) or catalase (CAT). A number of physiological processes are affected by the overexpression of SOD in plants, including removal of H_2O_2 , toxic reductants, biosynthesis and degradation of lignin in cell walls, auxin catabolism, etc. (Scandalios 1993). Activation of APX is due to the physiological injuries occurring in plants under heat stress (Mazorra et al. 2002).

Increased levels of ROS under high temperature cause cellular injury due to reduced antioxidant activity in the stressed tissues (Fadzillah et al. 1996; Mittler et al. 2004). In order to increase the heat tolerance, the levels and activities of antioxidants must be increased to protect against high temperature-induced oxidative stress. Studies conducted in this regard revealed that heat acclimated turf grass showed reduced ROS production owing to enhanced ascorbate and glutathione synthesis (Xu et al. 2006). It is suggested that antioxidant capacity of cells can be increased by some signaling molecules (Gong et al. 1997; Dat et al. 1998). Nonetheless, research is imperative to add to the list of potential signaling molecules, which may enhance the antioxidant production in cells exposed to heat temperature stress (Wahid et al. 2007).

2.3.3 Water Relations

Heat stress drives the rapid loss of water from the plant surface, causes tissue and organ dehydration, and restricts growth in plant species, for example, sorghum (Machado and Paulsen 2001), tomato (Mazorra et al. 2002), and sugarcane (Wahid and Close 2007). Heat stress produces osmotic strain on the growing tissues due to diminished root hydraulic conductance and tissue water status (Jiang and Huang 2001; Morales et al. 2003). Likewise, it may result in substantial reduction in sorghum (*Sorghum bicolor*) leaf growth and leaf water content and water potential in wheat (Shah and Paulsen 2003).

Heat stress also disrupts the uptake and translocation of water, ions, and organic solutes across the plant membranes, interferes with photosynthesis and respiration, increases evapo-transpiration rate, reduces the leaf osmotic potential and increases the chlorophyll fluorescence (Tsukaguchi et al. 2003; Huve et al. 2005; Taiz and Zeiger 2006). It results in stomatal closure and reduces the tissue water contents (Berry and Bjorkman 1980; Wahid et al. 2007). Heat stress-induced water stress thus is closely associated with reduction of soil water contents (Talwar et al. 1999).

2.3.4 Osmolytes Accumulation

Accumulation of certain low molecular mass organic compounds, generally called compatible solutes or osmoprotectants, is an important adaptive mechanism in plants subjected to abiotic stresses including temperature extremes (Hare et al. 1998; Sakamoto et al. 1998). Several osmolytes, including sugars and sugar alcohols (polyols), proline, tertiary, and quaternary ammonium compounds and tertiary sulphonium compounds, are reported to accumulate in different plant species exposed to stress conditions (Sairam and Tyagi 2004; Wahid et al. 2007).

Among different compatible solutes, enhanced synthesis of soluble sugars, free proline and glycinebetaine (GB) has been more frequently studied for their osmoregulatory and protective roles (Matysik et al. 2002; Bohnert et al. 2006; Wahid 2007; Wahid et al. 2008; Farooq et al. 2008a). GB plays a great role as osmoprotectant in plants under a range of abiotic stresses including high temperature (Sakamoto and Murata 2002). However, the ability of plants to synthesize GB under stressful conditions varies among species (Ashraf and Foolad 2007). For example, sugarcane under heat stress (Wahid and Close 2007) and maize under drought (Quan et al. 2004) and chilling (Farooq et al. 2008c) and rice under drought (Farooq et al. 2008a) are reported to accumulate large amounts of GB. Like GB, increased free proline accumulation in higher plants in response to abiotic stresses has also been reported (Kavi Kishore et al. 2005). Biosynthesis of GB or proline may buffer the cellular redox potential under heat and other abiotic stresses, suggesting their functional significance (Rontein et al. 2002). The accumulation of soluble sugars was greatly implicated for improved heat tolerance of sugarcane (Wahid and Close 2007). In view of the importance of osmoprotectants accumulation, more concerted efforts on engineering pathways for enhanced biosynthesis of osmolytes may be fruitful (Ashraf and Foolad 2007).

2.3.5 Metabolite Synthesis

Heat stress leads to the accumulation of a range of primary and secondary metabolites. Primary metabolites are either direct products of carbon fixation (e.g., sugars, organic acids) or are synthesized after preliminary transformations of primary metabolites (e.g., amino acids, betaines alcohol sugars). Like other abiotic stresses, the accumulation of primary metabolites under heat stress has also been well documented (Iba 2002; Zhu 2003). Important primary metabolites showing accumulation under heat stress include free proline, GB, soluble sugars, etc., (Wahid 2007; Wahid and Close 2007; Wahid et al. 2008). In a recent study, using cluster and principal component analyses, it was revealed that out of 122 primary and secondary metabolites determined using advanced techniques like GC-MS and amino acids analyzer, only sucrose, quinate, trans-aconitate, guanine, γ -amino butyric acid (GABA), and ethanolamine held relationships with the high temperature tolerance of sugarcane bud chips (Rasheed 2009).

On the contrary, the synthesis and accumulation of secondary metabolites are less well understood under high temperature stress. Secondary metabolites are biosynthesized in plants from the intermediates of primary carbon metabolism via phenylpropanoic acid, shikmic acid, mevalonic acid, and methyl erythritol phosphate pathways (Taiz and Zeiger 2006). Recently, it is reported that heat stress induces production of secondary metabolites including phenolics, flavonoids, phenyl propanoids, and plant steroids (Bharti and Khurana 1997; Wahid 2007; Wahid et al. 2008). Carotenoids show a role in protecting cellular structures in various plant species under different stress types (Havaux 1998). Studies show that lipid layer of the thylakoid membranes are stabilized and photoprotected by various carotenoids and some terpenoids such as isoprene and α -tocopherol. Exposure of plants to strong light and high temperatures caused the partitioning of xanthophylls (violaxanthin, anthraxanthin, zeaxanthins, etc.) between the light-harvesting complexes and lipid phase of thylakoid membranes and increases membrane thermostability (Havaux 1993, 1998).

Isoprenoids are low molecular weight volatile compounds, synthesized via mevalonic acid pathway (Taiz and Zeiger 2006); their emission from leaves confers their role in heat tolerance (Loreto et al. 1998; Sharkey 2005). Although their synthesis is cost intensive, they show compensatory benefits in terms of heat resistance (Funk et al. 2004). Plants, capable of emitting higher amounts of isoprene, photosynthesize better under heat stress, which indicates a relationship between isoprene emission and heat tolerance (Velikova et al. 2004). Isoprene emission protects PSII under high temperature (Sharkey 2005), whereas the endogenous production of isoprene protects the biological membranes by directly binding with singlet oxygen $({}^{1}O_{2})$ by virtue of isoprene-conjugate double bond (Velikova et al. 2004).

Phenolics are the largest class of secondary metabolites and include flavonoids, lignin, anthocyanin, etc. Accumulation of soluble phenolics under heat stress is accompanied with increased activity of phenyl ammonia lyase (PAL) but decreased activity of peroxidase polyphenyl lyase (Taiz and Zeiger 2006). Acclimation to heat stress is triggered by the biosynthesis of phenolic compounds induced by high temperature (Rivero et al. 2001). They act as efficient antioxidants in plant tissues under stressful conditions (Dixon and Paiva 1995; Sgherri et al. 2004). Levels of flavonoid (e.g., anthocyanins) are greatly altered in plant tissues under heat stress (Oren-Shamir and Nissim-Levi 1999; Sachray et al. 2002; Wahid et al. 2008).

Plant steroids, a class of secondary metabolites, also influence a variety of functions under stressful conditions. Brassinosteroids (BRs) and ginsenosides are important plant steroids whose physiological importance to high temperature tolerance in plants has been explored. Studies confirm that BRs confer tolerance to high temperature stress in brassica and tomato seedlings, by inducing the biosynthesis of major heat shock proteins (Dhaubhadel et al. 1999). Production of ginsenosides, another important plant steroid, has been reported in all organs of Panax quinquefolius. It is recently reported that growing season had a great effect on the ginsenosides biosynthesis *P. quinquefolius* plants grown at high temperatures had 49% higher concentrations of storage root ginsenosides than respective control plants (Jochum et al. 2007; Wahid and Tariq 2008).

2.4 Molecular Responses

Transcritional regulation plays an important role in plant defense from heat stress (Singh et al. 2002). Heat stress induces numerous genes encoding transcriptional factors, which are involved in heat stress response and tolerance (Chen and Zhu 2004; Kotak et al. 2007). Different studies revealed that several genes are up- and downregulated by abiotic stresses (Kawasaki et al. 2001; Provart et al. 2003; Nogueira et al. 2003). Elevated temperature affects the gene expression in storage protein synthesis and starch metabolism during grain filling stage in rice (Yamakawa et al. 2007). Heat stress changes the pattern of gene expression, which is important for thermotolerance (Yang et al. 2006). An account of various genes and proteins showing expression under heat stress is given below.

2.4.1 Heat Shock Genes and Proteins

Several transcriptome studies have identified many stress-responsive genes and encoding transcriptional factors during environmental stresses. Recent transgenic approaches suggest that heat tolerance is a multigenic character. Heat shock induces many genes, which are attributed to heat shock elements (HSE). These HSE are situated in the promoter region of hsp genes (Hubel and Schöffl 1994). Transgenic approach confirmed that Heat-shock transcription factor (HSF) binding to pentameric nucleotides (5'nGAAn-3') of HSE sequences (Perisic et al. 1989; Sung et al. 2003). This HSF-HSE interaction and transcriptional activation is quite conserved in nature. This multigenic phenomenon modifying the expression pattern of transcription factors motivate a series of genes (Dong et al. 2003). Studies show that *Hot1-Hot4* genes of Arabidopsis may function to improve heat tolerance; Hot1 is identified as Hsp101 in Arabidopsis thaliana (Hong and Vierling 2000). HsfA1 acts as master regulator of thermotolerance in tomato by reducing the expression of heat shock genes in co-suppression lines (Mishra et al. 2002).

Hsfs are essential for gene expression in response to high temperature (Nover et al. 2001). Various studies show that distinctive *hsp* genes are not expressed in germinating pollen. Only *hsp18* and *hsp70* genes are transcribed in response to heat stress (Wahid et al. 2007). A defective heat shock response of mature maize pollen was due to inefficient induction of heat shock gene transcription (Hopf et al. 1992). Enhanced expression of HSP70 assisted in translocation,

proteolysis, protein translation, protein folding, aggregation, and refolding of denatured proteins (Zhang et al. 2005; Iba 2002; Weggle et al. 2004; Gorantla et al. 2007). Recent studies revealed that α -amylase genes in seeds of rice reduced the seed weight and chalkiness during ripening under heat stress (Asatsuma et al. 2006; Yamakawa et al. 2007).

The synthesis and accumulation of heat shock proteins (HSPs) through heat shock factors (HSFs) network play great role in plant responses to heat stress (Wang et al. 2004; Kotak et al. 2007). Amounts of specific mRNA synthesis, mRNA stability, translation efficiency and alteration in protein activity increase in plants as a result of gene expression (Sullivan and Green 1993). All organisms synthesize HSPs upon exposure to high temperature. Heat stress altered gene expression in reproductive organ of plant (Dupuis and Dumas 1990; Oshino et al. 2007). Abortion of development and demarcation of pollen mother cell due to heat shock is due to tissue specific alterations in gene expression (Sakata et al. 2000; Abiko et al. 2005). In plants, a heat shock of 8-10°C above ambient temperatures induces the synthesis of both high (60-110 kDa) and low (15-30 kDa) molecular weight HSPs (Vierling 1991; Waters et al. 1996; Sun et al. 2002). These HSPs were induced either to protect the plant from injury or to help repair the injury caused by the heat stress (Leshem and Kuiper 1996). The synthesis of HSPs occurred in different plant species when they were exposed 10-15°C above growing temperatures (Dubey 1999). Their synthesis is extremely fast, diverse, and intensive in a variety of organisms (Parsell and Lindquist 1993; Wahid et al. 2007).

Both cytosolic and organelle synthesis of HSPs has been well studied. Some HSPs that accumulate in the cytosol at 27°C and in the chloroplast at 43°C and 37°C respectively, appeared to play a role in photosynthesis and thermotolerance (Heckathorn et al. 1998b). In maize, high temperature induced the synthesis and accumulation of chloroplast protein elongation factor *EF-TU*, which defended the chloroplasts proteins from heat-induced damage (Ristic

et al. 2004; Momcilovic and Ristic 2007). Maize *EF-TU* is a 45–46 kDa HSP confined to chloroplast stroma is involved in development at heat tolerance in maize (Ristic and Cass 1992; Bhadula et al. 2001; Moriarty et al. 2002). In maize, heat shock of 40°C induces the synthesis of HSPs18 (Nieto-Sotelo et al. 2002). Interaction of HSPs 22 kDa with the *Chenopodium album* and common bean chloroplast membranes affects the composition of membrane and decreases its fluidity; thus increasing the efficiency of ATP transport (Barua et al. 2003; Simões-Araújo et al. 2003).

Mitochondrial HSPs have been isolated from pumpkin (Cucurbita pepo) cotyledons under high temperature stress (Tsugeki et al. 1992; Kuzmin et al. 2004). They act as molecular chaperones in vitro (Schöffl et al. 1998; Guo et al. 2001; Kim and Schöffl 2002), preaggregation of denatured proteins vent (Sheffield et al. 1990), aid in folding of nascent polypeptides and refolding of denatured proteins (Lee et al. 1994; Goloubinoff et al. 1999). They also resolubilize the denatured aggregated proteins (Parsell et al. 1994). HSP68 synthesis was restricted to mitochondria as a precursor protein, but its synthesis increased during heat shock in cell (Neumann et al. 1993). When wheat, maize, and rye seedling were exposed at 42°C, five mitochondrial LMW HSPs (19, 20, 22, 23, and 28 kDa) were induced in maize and only one (20 kDa) in rye and wheat mitochondria each; the tolerance of maize was higher than wheat and rye (Korotaeva et al. 2001). The specific nucleusencoded HSPs have been identified in potato, maize, soybean, barley, and tomato (Neumann et al. 1993; Nautiyal and Shono 2010), peas (Ko et al. 1992; Watts et al. 1992) under heat stress.

Although with less certainty, some putative functions have been assigned to HSPs when produced under normal or high temperature conditions. The rapid accumulation of HSPs may play a significant role in the safety of metabolic apparatus of the cell. Some HSPs are produced in some developing cells under control condition (Hopf et al. 1992) during embryogenesis, germination,



Fig.6.3 Effect of heat acclimation and stress on the expression of HSP70 in hypocotyls of different pea varieties. After Tian et al. (2009) with permission

pollen formation, fruit set and its maturation (Vierling 1991; Sun et al. 2002 Prasinos et al. 2005; Wahid et al. 2007). For instance, HSPs were produced in greater amounts in etiolated maize seedling after 5-h exposure to high temperature stress (Lund et al. 1998). Acquired thermotolerance depends upon the synthesis of HSPs and their cellular localization (Heckathorn et al. 1999; Korotaeva et al. 2001). In arid and semi arid areas, plants may accumulate significant amount of HSPs in response to high leaf temperatures. In 2-day-old soybean seedlings, HSPs appeared to maintain the conformation of other proteins, as an aid for the acquired thermotolerance (Jinn et al. 1997). The wide diversity and abundance of HSPs is important for altering the plant response to high temperature stress (Waters et al. 1996). The mature pollen was susceptible to high temperature and pollen viability was extremely reduced due to nonproduction of HSPs. A distinct set of HSPs was induced in male tissues of maize under heat stress (Dupuis and Dumas 1990). HSPs (64 and 72 kDa) were induced in germinating pollens under heat stress (Frova et al. 1989). In a recent study, Tian et al. (2009) reported the improved heat tolerance of young pea seedlings due to enhanced synthesis of HSP70 (Fig. 6.3).

2.4.2 Dehydrins

Dehydrins (DHNs), belonging to subclass of LEA group II (Dure et al. 1989), are produced at the later stages of seed development in various plant species under drought, salinity, low temperature, heat stress, nutrients deficiency, and ABA application (Close 1996; Campbell and Close 1997; Svensson et al. 2002; Wahid and Close 2007; Pulla et al. 2007; Rurek 2010). D-11 from cotton (Baker et al. 1988), RAB16 (responsive to ABA) in rice (Mundy and Chua 1988) and RAB17 in maize (Vilardell et al. 1990) were cloned and characterized as DHN genes (Campbell and Close 1997; Ismail and Hall 1999 Koag et al. 2003). Immunological evidence indicated that DHNs are expressed in cyanobacteria (Close and Lammers 1993), brown algae (Li et al. 1997), liverworts (Hellwege et al. 1994), ferns (Reynolds and Bewley 1993), ginkgo (Close and Lammers 1993), and conifers (Jarvis et al. 1996). Using immunological studies, DHNs were detected in the nucleus, cytoplasm, mitochondria, chloroplasts, and vacuole (Close 1996; Campbell



SC, sclerenchyma; PH, phloem; MX, metaxylem; PX, protoxylem; BFC, Buliform cells; LE, lower epidermis; UE, upper epidermis

Fig. 6.4 Immunohistochemical expression of dehydrins in the leaf of sugarcane clone HSF-240 under control and heat stress. The dehydrins were found to associate to stele,

and Close 1997; Wahid et al. 2007) and found to be associated with cytoplasmic membranes system under abiotic stresses (Koag et al. 2003). Immuno-histolocalization studies revealed that the DHNs are associated with the mesophyll, vascular, and dermal tissues of heat-stressed sugarcane (Gilani 2007, Fig. 6.4). In maize, all parts of mature embryos show dehydrin accumulation (Godoy et al. 1994). In recent studies, three low molecular weight dehydrins were reported to be expressed in sugarcane leaves in response to heat stress (Wahid and Close 2007).

2.4.3 Senescence-Associated Genes

Temperature, pathogenic infection, drought, and nutrient deficiency; wounding and shading may increase leaf senescence (He et al. 2001). Thus about 183 senescence-associated genes (SAGs) are involved in energy metabolism, gene expression regulations, protein biosynthesis regulations, pathogenicity, stress and flower development (Liu et al. 2008). QTLs for some senescencerelated traits have been mapped on chromosome 2A, 3A, 3B, 6A, 6B, and 7A in winter wheat subjected to heat stress (Vijayalakshmi et al. 2010). A number of encoding SAGs for proteinases such as serine proteinase in parsley (Jiang et al. 1999), cysteine proteinase in *Arabidopsis* (Lohman et al. lower and upper epidermis, and buliform cells during heat stress, as evident from golden brown color in staining. *Source*: Gilani (2007)

1994) and aspartic proteinase in *Brassica* (Buchanan-Wollaston and Ainsworth 1997) are associated with leaf senescence. A large number of SAGs and defense genes has been reported to express during leaf senescence in maize (Smart et al. 1995), barley (Kleber-Janke and Krupinska 1997), rice (Lee et al. 2001), *A. thaliana* (Lohman et al. 1994; Oh et al. 1996; Gepstein et al. 2003), tomato (John et al. 1997; Drake et al. 1996), rad-ish (Azumi and Watanabe 1991), and *Brassica napus* (Buchanan-Wollaston and Ainsworth 1997).

Heat stress accelerates the senescence and results in decreased assimilation partitioning to grains (Spano et al. 2003). For instance, high temperature induced the expression of dehydration responsive genes (ERD1), which is known as SAG15. This gene also protects the cells from injury (Weaver et al. 1999). A combine effect of heat-shock and drought induced a senescenceassociated gene (SAG12), at least in Nicotiana tabacum, which improved the stress tolerance in plants (Rizhsky et al. 2002). Heat shock (40°C) induced tmr genes in Agrobacterium, which delays the senescence. This was achieved by an inducible promoter such as HS6871 from soybean (Smart et al. 1991). Chen et al. (2002) identified 18 transcription factors such as WRKY

genes and its protein in response to senescence and environmental stresses, including heat stress, which improved the agronomic characters of crop plants.

2.4.4 Stay-Green Gene

Photosynthetic responses of annual plants can be improved by extending duration of vegetative growth and delaying leaf senescence (Thomas and Howarth 2000). Stay-green (Sgr) proteins are responsible for the green-flesh and retention of chlorophyll during senescence (Park et al. 2007; Barry et al. 2008). The trait stay-green is divided into five types such as type A, B, C, D, and E on the basis of its chlorophyll retention during leaf senescence (Thomas and Howarth 2000). Overexpression of Sgr gene reduces the loss of chlorophyll and delays early senescence of developing leaves (Park et al. 2007). Sgr synthesis has been reported in many plants such as sorghum (Tao et al. 2000), maize (Rajcan and Tollenaar 1999), rice (Cha et al. 2002; Park et al. 2007), durum wheat (Spano et al. 2003), tomato (Akhtar et al. 1999; Barry et al. 2008), pea (Sato et al. 2007) A. thaliana (Oh et al. 2000; Ren et al. 2007), oat (Helsel and Frey 1978), and Festuca pratensis (Armstead et al. 2006).

A stay-green protein potentially downregulates the chlorophyll degradation at transcriptional level and delays senescence (Nam 1997; Park et al. 2007). Delaying leaf senescence resulted in about 11% increase in carbon fixation in Lolium temulentum (Thomas and Howarth 2000). Tollenaar and Daynard (1978) demonstrated that some maize varieties such as L087602 shows stay-green phenotype, which increases the water, carbohydrates, and protein contents in the husks, cobs, and seeds. Nguyen (1999) demonstrated that stay-green genes delay leaf senescence in sorghum and reduce lodging in heat-stressed and low moisture areas. In fact stay-green is used as a selection criterion in warm areas (Acevedo et al. 1991; Kohli et al. 1991). For instance, most lines of wheat are sensitive to heat stress while some lines are heat tolerant due to stay-green character (Rehman et al. 2009).

3 High Temperature Stress: Management

Despite the fact that heat stress affects all the aspects of growth and development in plants, its effects may be mitigated by adopting various approaches. Some of the pragmatic strategies in this regard are detailed below.

3.1 Exploitation of Genetic Variability

As mentioned above, heat tolerance is a multigenic trait, which offers the opportunity of improving plants against heat stress. Temperate genotypes show less dry matter production and reduced yield due to high temperature stress as compared to tropical ones (Giaveno and Ferrero 2003). Attempts have been made to find the genetic differences in plants based on morphological and physiological criteria (Wahid et al. 2007; Khan et al. 2008). For instance, high temperature stress during grain filling can reduce setting and filling of seed by accelerating senescence thereby reducing crop yields (Harding et al. 1990; Siddique et al. 1999). This is because, resources required are utilized by plants for heat stress tolerance and limited amount is available for reproductive growth (Hall 1992). Search for genotypic variation in heat resistance on the basis of leaf electrolyte leakage is important for improving heat tolerance (Li et al. 1991). It is known that membrane stability is positively associated with crop yield under heat stress (Rahman et al. 2004). Such association was important for survival of wheat when exposed to high temperature at anthesis stage (Saadalla et al. 1990).

3.2 Conventional Breeding and Molecular Strategies

Conventional and modern breeding methods have been practiced for improving plant stress tolerance for the last many decades. For conventional breeding, major challenges include germplasm mass screening, selection criteria, and identification of heat tolerant materials (Wahid et al. 2007). Among the several screening and selection methods, heat tolerance index (HTI) based on recovery of plants from heat stress has been proposed in case of sorghum (Young et al. 2001). In tomato, for example, under stress, there was a positive relationship between fruit set and yield. In this regard, many studies suggested that heat stress is not the only reason for impaired seed setting, nonetheless heat stress driven reduction in pollen tube growth and fertilization are rather more important (Wahid et al. 2007).

Present-day molecular and biotechnological tools have contributed enormously to understand the complexity and cellular pathways of stress responses and signal transduction mechanisms under abiotic stress responses (Sreenivasulu et al. 2006). Molecular studies involving cDNA arrays have revealed several genes, which are upregulated by several biotic and abiotic stresses (Kawasaki et al. 2001; Provart et al. 2003; Nogueira et al. 2003). Many of these genes are involved in signaling pathways via encoding proteins particularly mitogen-activated protein kinase (MAPK), histidine kinase, Ca²⁺-dependent protein kinase (CDPK), SOS3 Ca²⁺ sensor family and numerous transcription factors.

Marker-assisted selection (MAS) and genetic transformation are two main approaches to improve heat stress tolerance in plants, which proved worthwhile in improving our understanding of stress tolerance mechanisms at molecular level (Foolad 2005). The use of MAS approach requires identifying genetic markers associated with genes or QTLs contributing to whole or individual stress tolerance. A number of research efforts have identified genetic markers related to different environmental stresses at various growth stages of plants to facilitate understanding of genetic relationships for stresses tolerance (Foolad 2005). However, limited studies have identified genetic markers related to high temperature tolerance in plant. Available studies show that in a number of Arabidopsis mutants, four QTLs were found to be involved in acquiring thermotolerance in heat-sensitive mutants (Hong and Vierling 2000). Use of restriction fragment length polymorphism (RFLP) mapped 11 QTLs for pollen germination and pollen tube growth in heat-stressed maize plants (Frova and Sari-Gorla 1994; Wahid et al. 2007).

3.3 Seed Treatments and Plant Acclimation

In addition to the above, seed treatment and planting materials and foliar spray of with various organic and inorganic agents has proven their worth in enhancing heat tolerance in a number of plant species. For instance, presowing GB treatment in barley (Wahid and Shabbir 2005), H₂O₂ treatment in maize (Wahid et al. 2008), and presprouting soaking of sugarcane bud chips with GB and proline (Rasheed 2009) have been implicated with great success in improving high temperature tolerance at germination and subsequent growth stages (Fig. 6.2). Epibrassinolide, a subclass of brassinostreoids, treatment modulated the translational machinery, resulting in higher HSPs synthesis and rapid resumption of protein synthesis during and after the application of high temperature stress (Dhaubhadel et al. 1999; Kagale et al. 2007). More so, precondition of tomato plants (Morales et al. 2003) and acclimation of turfgrass (Xu et al. 2006) to heat stress resulted in better growth of plants in heat-stressed environments.

4 Conclusion and Future Perspective

Responses of plants to heat stress may be symptomatic to quantitative. Despite the fact that heat stress responses are well evident at all growth stages, reproductive growth stages are more prone to heat episodes. Other heat stress effects may entail structural changes in tissues and cell organelles, disturbance of leaf water relations, and decline in the rate of photosynthesis, production of ROS and lipid peroxidation, changes in enzymatic and nonenzymatic antioxidants, and synthesis of secondary metabolites are also important. The adaptive mechanisms of plants in response to heat stress include the induction of signaling cascades resulting in expression of specific gene. Synthesis of HSPs is a universal heat stress response in plants. Recently, evidence is pouring in on the synthesis and accumulation of some other stress-related proteins. All such proteins function as molecular chaperones and maintain three-dimensional structure of membrane proteins for sustained cell metabolism and plant survival under heat stress. Use of classical and modern breeding protocols, hunting genetic diversity for high temperature tolerance, use of presowing seed treatments and planting materials, and preconditioning/hardening of plants for high temperature tolerance has been beneficial. Although a considerable progress has been achieved in understanding the heat responses of plants, yet there is need for further understanding the biochemical and molecular basis of heat tolerance for improvement of yield benefits from hot environments. Recent molecular biology and gene transfer protocols can play important roles in this regard. Since benefits of seed treatments have been accrued at advanced growth stages (Wahid and Shabbir 2005), the basis of such changes needs to be explored.

References

- Abiko M, Akibayashi K, Sakata T, Kimura M, Kihara M, Itoh K, Asamizu E, Sato S, Takahashi H, Higashitani A (2005) High-temperature induction of male sterility during barley (*Hordeum vulgare* L.) anther development is mediated by transcriptional inhibition. Sex Plant Reprod 18:91–100
- Acevedo E, Nachit M, Ferrar GO (1991) Effects of heat stress on wheat and possible selection tools for use in breeding for tolerant. In: Saunders DA (ed) Wheat for the nontraditional warmer areas areas. CIMMYT, Mexico, pp 401–421
- Akhtar MS, Goldschmidt E, John I, Rodoni S, Matile P, Grierson D (1999) Altered patterns of senescence and ripening in gf, a stay-green mutant of tomato (*Lycopersicum esculentum* Mill.). J Exp Bot 50:1115–1122
- Akman Z (2009) Comparison of high temperature tolerance in maize, rice and sorghum seeds by plant growth regulators. J Anim Vet Adv 8:358–361
- Al-Khatib K, Paulsen GM (1990) Photosynthesis and productivity during high temperature stress of wheat genotypes from major world regions. Crop Sci 30:1127–1132

- Al-Khatib K, Paulsen GM (1999) High temperature effects on photosynthesis process in temperate and tropical cereals. Crop Sci 39:119–125
- Apel K, Hirt H (2004) Reactive oxygen species metabolism, oxidative stress, a signaling transduction. Annu Rev Plant Biol 55:373–399
- Armstead I, Donnison I, Aubry S, Harper J, Hortensteiner S, James C, Mani J, Moffet M, Ougham H, Robert L (2006) From crop to model to crop: identifying the genetic basis of the stay-green mutation in the *Lolium/Festuca* forage and amenity grasses. New Phytol 172:592–597
- Asatsuma S, Sawada C, Kitajima A, Asakura T, Mitsui T (2006) α-Amylase affects starch accumulation in the rice grain. J Appl Glycosci 53:187–192
- Ashraf M, Foolad MR (2007) Roles of glycine betaine and proline in improving plant abiotic stress resistance. Environ Exp Bot 59:206–216
- Ashraf M, Hafeez M (2004) Thermotolerance of pearl millet and maize at early growth stages: growth and nutrient relations. Biol Plantarum 48:81–86
- Azumi Y, Watanabe A (1991) Evidence for a senescenceassociated gene induced by darkness. Plant Physiol 95:577–583
- Baker J, Steele C, Dure L (1988) Sequence and characterization of 6 Lea proteins and their genes from cotton. Plant Mol Biol 11:277–291
- Banon S, Fernandez JA, Franco JA, Torrecillas A, Alarcon JJ, Sanchez-Blanco MJ (2004) Effects of water stress and night temperature preconditioning on water relations and morphological and anatomical changes of *Lotus creticus* plants. Sci Hortic 101:333–342
- Barry CS, McQuinn RP, Chung M-Y, Besuden A, Giovannoni JJ (2008) Amino acids substitutions in homologos of the stay-green protein are responsible for the green-flash and chlorophyll retainer mutations of tomato and pepper. Plant Physiol 147:179–187
- Barua D, Downs CA, Heckathorn SA (2003) Variation in chloroplast small heat-shock protein function is a major determinant of variation in thermotolerance of photosynthesis electron transport among ecotypes of *Chenopodium album*. Funct Plant Biol 30:1071–1079
- Berry J, Bjorkman O (1980) Photosynthesis response and adaptation to temperature in higher plants. Annu Rev Plant Physiol 31:491–543
- Bhadula SK, Elton TE, Habben JE, Helentjaris TG, Jiao S, Ristic Z (2001) Heat-stress induced synthesis of chloroplast protein synthesis elongation factor (EF-TU) in a heat-tolerant maize line. Planta 212:359–366
- Bharti AK, Khurana JP (1997) Mutant of Arabidopsis as tools to understand the regulation of phenylpropanoids pathway and UVB protection mechanism. Photochem Photobiol 65:765–776
- Blum A (1988) Plant breeding for stress environments. CRC, Boca Raton, FL
- Bohnert HJ, Gong Q, Li P, Ma S (2006) Unraveling abiotic stress tolerance mechanisms – getting genomics going. Curr Opin Plant Biol 9:180–188

- Buchanan-Wollaston V, Ainsworth C (1997) Leaf senescence in *Brassica napus*: cloning of senescence-related genes by subtractive hybridization. Plant Mol Biol 33:821–834
- Burke JJ (2001) Identification of genetic diversity and mutations in higher plant acquired thermotolerance. Physiol Plantarum 112:167–170
- Campbell SA, Close TJ (1997) Dehydrins: genes, proteins, and association with phenotypic traits. New Phytol 137:61–74
- Carpentier R (1999) The effect of high temperature stress on photosynthetic apparatus. In: Pessarakli M (ed) Handbook of plant and crop stress. Marcel Dekker, New York, pp 337–348
- Cha K-W, Lee Y-J, Koch H-J, Lee B-M, Nam Y-W, Paek N-C (2002) Isolation, characterization, and mapping of the stay green mutant in rice. Theor Appl Genet 104:525–532
- Chaitanya KV, Sundar D, Reddy AR (2001) Mulberry leaf metabolism under high temperature stress. Biol Plantarum 44:379–384
- Chen WJ, Zhu T (2004) Networks of transcription factors with roles in environmental stress response. Trends Plant Sci 9:591–596
- Chen W, Provart NJ, Glazebrook J, Katagiri F, Chang H-S, Eulgem T, Mauch F, Luan S, Zou G, Whitham SA, Budworth PR, Tao Y, Xie Z, Chen X, Lam S, Kreps JA, Harper JF, Si-Ammour A, Mauch-Mani B, Heinlein M, Kobayashi K, Hohn T, Dangl JL, Wang X, Zhu T (2002) Expression profile matrix of Arabidopsis transcription factor genes suggests their putative functions in response to environmental stresses. Plant Cell 14:559–574
- Close TJ (1996) Dehydrins: emergence of a biochemical role of a family of plant dehydration proteins. Physiol Plantarum 97:795–803
- Close TJ, Lammers PJ (1993) An osmotic stress protein of cyanobacteria is immunologically related to plant dehydrins. Plant Physiol 101:773–779
- Columbo SJ, Timmer VR (1992) Limits of tolerance to high temperature causing direct and indirect damage to black spruce. Tree Physiol 11:95–104
- Crafts-Brandner SJ, Law RD (2000) Effect of heat stress on the inhibition and recovery of the ribulose-1, 5-bisphoshate carboxylase/oxygenase activation state. Planta 212:67–74
- Crafts-Brandner SJ, Salvucci ME (2002) Sensitivity to photosynthesis in the C4 plant, maize to heat stress. Plant Cell 12:54–68
- Dat J, Foyer C, Scott I (1998) Change in salicylic acid and antioxidants during induced thermo tolerance in mustard seedlings. Plant Physiol 118:1455–1461
- Dhaubhadel S, Chaudhary S, Dobinson KF, Krishna P (1999) Treatment with 24-epibrassinolide, a brassinosteroid, increases the basic thermotolerance of *Brassica napus* and tomato seedlings. Plant Mol Biol 40:333–342
- Dias AS, Lidon FC (2009) Evaluation of grain filling rate and duration in bread and durum wheat, under heat stress after anthesis. J Agron Crop Sci 195:137–147
- Dixon RA, Paiva NL (1995) Stress induced phenylpropanoid metabolism. Plant Cell 7:1085–1097

- Dong A, Zhu Y, Yu Y, Cao K, Sun C, Shen WH (2003) Regulation of biosynthesis and intracellular localization of rice and tobacco homologues of nucleosome assembly protein 1. Planta 216:561–570
- Downton J, Slatyer RO (1972) Temperature dependence of photosynthesis in cotton. Plant Physiol 50:518–522
- Drake R, John I, Farrell A, Cooper W, Schunch W, Grierson D (1996) Isolation and analysis of cDNAs encoding tomato cysteine proteases expressed during leaf senescence. Plant Mol Biol 30:755–767
- Dubey RS (1999) Protein synthesis by plants under stressful conditions. In: Pessarakli M (ed) Handbook of plant and crop stress, 2nd edn. Marcel Dekker, New York, pp 365–397
- Dupuis I, Dumas C (1990) Influence of temperature stress on maize (*in vitro*) fertilization and heat shock protein synthesis in maize (*Zea mays* L.) reproductive tissues. Plant Physiol 94:665–670
- Dure L, Crouch M, Harada J, Ho T-HD, Mundy J, Quatrano R, Thomas T, Sung ZR (1989) Common amino acid sequence domains among the LEA proteins of higher plants. Plant Mol Biol 12:475–486
- Edwards GE, Walker D (1983) C3, C4: Mechanisms and cellular and environmental regulation of photosynthesis. University of California Press, Berkeley, CA
- Egli DB, TeKrony DM, Heitholt JJ, Rupe J (2005) Air temperature during seed filling and soybean seed germination and vigor. Crop Sci 45:1329–1335
- Fadzillah NM, Gill V, Finch RP, Burdon RH (1996) Chilling, oxidative stress and antioxidant responses in shoot cultures of rice. Planta 199:552–556
- Farooq M, Basra SMA, Wahid A, Cheema ZA, Cheema MA, Khaliq A (2008a) Physiological role of exogenously applied glycinebetaine in improving drought tolerance of fine grain aromatic rice (*Oryza sativa* L.). J Agron Crop Sci 194:325–333
- Farooq M, Aziz T, Basra SMA, Wahid A, Khaliq A, Cheema MA (2008b) Exploring the role of calcium to improve the chilling tolerance in hybrid maze. J Agron Crop Sci 194:350–359
- Farooq M, Aziz T, Hussain M, Rehman H, Jabran K, Khan MB (2008c) Glycinebetaine improves chilling tolerance in hybrid maize. J Agron Crop Sci 194:152–160
- Farooq M, Wahid A, Aziz T, Lee DJ, Siddique KHM (2009) Chilling tolerance in maize: physiological and agronomic implications. Crop Pasture Sci 60:501–516
- Feller U, Crafts-Brandner SJ, Salvucci ME (1998) Moderately high temperature inhibits ribulose-1, 5-bisphosphate carboxylase/oxygenase (Rubisco) activase-mediated activation of Rubisco. Plant Physiol 116:539–546
- Ferrar PJ, Slattyer RO, Vranjic NA (1989) Photosynthetic temperature acclimation in *Eucalyptus* species from diverse habitats and a comparison with *Nerium oleander*. Aust J Plant Physiol 16:199–217
- Foolad MR (2005) Breeding for abiotic stress tolerances in tomato. In: Ashraf M, Harris PJC (eds) Abiotic stresses: plant resistance through breeding and

molecular approaches. The Haworth Press, New York, pp 613-648

- Frova C, Sari-Gorla M (1994) Quantitative trait loci (QTLs) for pollen thermotolerance detected in maize. Mol Gen Genet 245:424–430
- Frova C, Taramino G, Binelli G (1989) Heat-shock proteins during pollen development in maize. Dev Genet 10:324–332
- Funk JL, Mak JE, Lerdau MT (2004) Stress-induced changes in carbon sources for isoprene production in *Populus deltoids*. Plant Cell Environ 27:747–755
- Gepstein S, Sabehi G, Carp MJ, Hajouj T, Nesher MF, Yariv I, Dor C, Bassani M (2003) Large-scale identification of leaf senescence-associated gene. Plant J 36:629–642
- Giaveno C, Ferrero J (2003) Introduction of tropical maize genotypes to increase silage production in the central area of Santa Fe, Argentina. Crop Breed Appl Biotechnol 3:89–94
- Gilani S (2007) Some physiological and biochemical responses of sugarcane (*Saccharum officinarum*) to high temperature stress. Ph.D. thesis, Department of Botany, University of Agriculture, Faisalabad, Pakistan
- Godoy JA, Lunar R, Torresschumann S, Moreno J, Rodrigo RM, Pintortoro JA (1994) Expression, tissue distribution and subcellular-localization of dehydrin Tas14. Plant Mol Biol 26:1921–1934
- Goloubinoff P, Mogk A, Zvi AP, Tomoyasu T, Bukau B (1999) Sequential mechanism of solubilization and refolding of stable protein aggregates by a bichaperone network. Proc Natl Acad Sci USA 96:13732–13737
- Gong M, Chen SN, Song YQ, Li ZG (1997) Effect of calcium and calmodulin on intrinsic heat tolerance in relation to antioxidant systems in maize seedlings. Aust J Plant Physiol 24:371–379
- Gorantla M, Babu PR, ReddyLachagari VB, Reddy AMM, Wusirika R, Bennetzen JL (2007) Identification of stress-responsive genes in an indica rice (*Oryza sativa* L.) using ESTs generated from drought-stressed seedlings. J Exp Bot 58:253–265
- Gounaris K, Brain APR, Quinn PJ, Williams WP (1984) Structural reorganization of chloroplast thylakoid membranes in response to heat stress. Biochim Biophys Acta 766:198–208
- Grass L, Burris I (1995) Effect of heat stress during seed development and maturation on wheat (*Triticum durum*) seed quality. I. Seed germination and seedling vigor. Can J Plant Sci 75:821–829
- Guilioni L, Wery J, Tardieu F (1997) Heat Stress-induced abortion of buds and flowers in pea: Is sensitivity linked to organ age or to relations between reproductive organs? Ann Bot 80:159–168
- Gunderson CA, Norby RJ, Wullschleger SD (2000) Acclimation of photosynthesis and respiration to simulated climatic warming in northern and southern population of *Acer saccharum*: laboratory and field evidence. Tree Physiol 20:87–95
- Guo YL, Guettouche T, Fenna M, Boellmann F, Pratt WB, Toft DO, Smith DF, Voellmy R (2001) Evidence for a mechanism of repression of heat shock factor I tran-

scriptional activity by a multichaperone complex. J Biol Chem 276:45791–45799

- Hall AE (1992) Breeding for heat tolerance. Plant Breed Rev 10:129–168
- Hall AE (2001) Crop responses to environment. CRC, Boca Raton, FL
- Harding SA, Guikema JA, Paulsen GM (1990) Photosynthetic decline from high temperature stress during maturation of wheat: I. interaction with senescence processes. Plant Physiol 92:648–653
- Hare PD, Cress WA, van Staden J (1998) Dissecting the roles of osmolyte accumulation during stress. Plant Cell Environ 21:535–553
- Havaux M (1993) Rapid photosynthetic adaptation to heat stress triggered in potato leaves by moderately elevated temperatures. Plant Cell Environ 16:461–467
- Havaux M (1998) Carotenoids as membrane stabilizers in chloroplasts. Trends Plant Sci 3:147–151
- He Y, Tang W, Swain JD, Green AL, Jack TP, Gan S (2001) Networking senescence-regulating pathways by using Arabidopsis enhancer trap lines. Plant Physiol 126:707–716
- Heckathorn SA, Downs CA, Coleman JS (1998a) Nuclearencoded chloroplast proteins accunulated in the cytosol during severe heat stress. Int J Plant Sci 159:39–45
- Heckathorn SA, Downs CA, Sharkey TD, Coleman JS (1998b) The small methionine-rich chloroplast heatshock protein protects photosystem II electron transport during heat stress. Plant Physiol 116:439–444
- Heckathorn SA, Downs CA, Coleman JS (1999) Small heat shock proteins protect electron transport in chloroplasts and mitochondria during stress. Am Zool 39:865–876
- Hellwege EM, Dietz KJ, Volk OH, Hartung W (1994) Abscisic acid and the induction of desiccation tolerance in the extremely xerophilic liverwort *Exormotheca holstii*. Planta 194:525–531
- Helsel GB, Frey RL (1978) Grain yield variations in oats associated with differences in leaf area during oat lines. Crop Sci 18:765–769
- Holaday AS, Martindale W, Alred R, Brooks AR, Leegood RC (1992) Changes in activities of enzymes of carbon metabolism in leaves during exposure of plants to low temperature. Plant Physiol 98:1105–1114
- Hong S-W, Vierling E (2000) Mutants of Arabidopsis thaliana defective in the acquisition of tolerance to high temperature stress. Proc Natl Acad Sci USA 97:4392–4397
- Hopf N, Plesofskv-Vig N, Brambl R (1992) The heat shock response of pollen and other tissues of maize. Plant Mol Biol 19:623–630
- Hubel A, Schöffl F (1994) Arabidopsis heat-shock factor: isolation and characterization of the gene and the recombinant protein. Plant Mol Biol 26:353–362
- Hughes RM (1979) Effect of temperature and moisture stress on germination and seedling growth of four tropical species. J Aust Inst Agric Sci 45:125–130
- Huve K, Bichele I, Tobias M, Niinemets U (2005) Heat sensitivity of photosynthetic electron transport varies

during the day due to changes in sugars and osmotic potential. Plant Cell Environ 29:212–218

- Iba K (2002) Acclimative response to temperature stress in higher plants: approaches of gene engineering for temperature tolerance. Annu Rev Plant Biol 53:225–245
- IPCC (2007) Climate change 2007: The physical science basis: summery for policymakers, IPCC WG1 Fourth Assessment Report
- Ismail AM, Hall AE (1999) Reproductive-stage, heat tolerance, leaf membrane thermostability and plant morphology in cowpea. Crop Sci 39:1762–1768
- Jarvis SB, Taylor MA, Macleod MR, Davies HV (1996) Cloning and characterization of the cDNA clones of three genes that are differentially expressed during dormancy-breakage in the seeds of Douglas fir (*Pseudotsuga menziesii*). J Plant Physiol 147:559–566
- Jiang Y, Huang B (2001) Drought and heat stress injury to two cool-season turf grasses in relation to antioxidant metabolism and lipid peroxidation. Crop Sci 41:436–442
- Jiang WB, Lers A, Lomaniec E, Aharoni N (1999) Senescence related serine protease in parsley. Phytochemistry 50:377–382
- Jinn TL, Chang PFL, Chen YM, Key JL, Lin CY (1997) Tissue type-specific heat-shock response and immunolocalization of class I low-molecular-weight heat-shock proteins in soyabean. Plant Physiol 114:429–438
- Jochum GM, Mudge KW, Thomas RB (2007) Elevated temperatures increase leaf senescence and root secondary metabolite concentrations in the understory herb *Panax quinquefolius* (Araliaceae). Am J Bot 94:819–826
- John I, Hackett R, Cooper W, Drake R, Farrel A, Grierson D (1997) Cloning and characterization of tomato leaf senescence-related cDNAs. Plant Mol Biol 33:641–651
- Kagale S, Divi UK, Krochko JE, Keller WA, Krishna P (2007) Brassinosteroids confers tolerance in *Arabidopsis thaliana* and *Brassica napus* to a range of abiotic stresses. Planta 225:353–364
- Karim MA, Fracheboud Y, Stamp P (1997) Heat tolerance of maize with reference of some physiological characteristics. Ann Bangladesh Agric 7:27–33
- Karim MA, Fracheboud Y, Stamp P (1999) Photosynthetic activity of developing leaves maize (*Zea mays* L.) is less affected by heat stress than that of developed leaves. Physiol Plantarum 105:685–693
- Kavi Kishore PB, Sangam S, Amrutha RN, Laxmi PS, Naidu KR, Rao KRSS, Rao S, Reddy KJ, Theriappan P, Sreenivasulu N (2005) Regulation of proline biosynthesis, degradation, uptake and transport in higher plants: its implications in plant growth and abiotic stress tolerance. Curr Sci 88:424–438
- Kawasaki S, Borchert C, Deyholos M, Wang H, Brazille S, Kawai K, Galbraith D, Bohnert HJ (2001) Gene expression profiles during the initial phase of salt stress in rice. Plant Cell 13:889–905
- Khan AI, Khan IA, Sadaqat HA (2008) Heat tolerance is variable in cotton (*Gossypium hirsutum* L.) and can be exploited for breeding of better yielding cul-

tivars under high temperature regimes. Pak J Bot 40:2053-2058

- Kim BH, Schöffl F (2002) Interaction between Arabidopsis heat shock transcription factor I and 70 kDa heat shock proteins. J Exp Bot 53:371–375
- Kim S-H, Dennis CG, Richard CS, Jeffrey TB, Dennis JT, Vangimalla RR (2007) Temperature dependence of growth, development and photosynthesis in maize under elevated CO₂. Environ Exp Bot 61:224–236
- Kleber-Janke T, Krupinska K (1997) Isolation of cDNA clones for genes showing enhanced expression in barley leaves during dark-induced senescence as well as during senescence under field conditions. Planta 203:332–340
- Ko K, Bornemisza O, Kourtz L, Ko ZW, Plaxton WC, Cashmore AR (1992) Isolation and characterization of a cDNA clone encoding a cognate 70-kDa heat shock protein of the chloroplast envelope. J Biol Chem 267:2986–2993
- Koag M-C, Fenton RD, Wilkens S, Timothy JC (2003) The binding of maize DHN1 to lipid vesicles. Gain of structure and lipid specificity. Plant Physiol 131:309–316
- Kohli MM, Mann C, Rajaram S (1991) Global status and recent progress in breeding wheat for the warmer areas. In: Saunders DA (ed) Wheat for the nontraditional warmer areas. CIMMYT, Mexico, pp 225–241
- Korotaeva NE, Antipina AI, Grabelynch OI, Varakina NN, Borovskii GB, Voinikov VK (2001) Mitochondrial low-molecular-weight heat shock proteins and tolerance of crop plant's mitochondria to hyperthermia. Russ J Plant Physiol 48:798–803
- Kotak S, Larkindale J, Lee U, von Koskull-Doring P, Vierling E, Scharf K-D (2007) Complexity of the heat stress response in plants. Curr Opin Plant Biol 10:310–316
- Kuzmin EV, Kapova OV, Elothn TE, Newton KJ (2004) Mitochondrial respiratory deficiencies signal up-regulation of genes for heat shock proteins. J Biol Chem 279:20672–20677
- Law RD, Crafts-Brandner SJ (1999) Inhibition and acclimation of photosynthesis to heat stress is closely correlated with activation of ribulose-1,5-bisphosphate carboxylase/oxygenase. Plant Physiol 120:173–181
- Lee Y-RJ, Nagao RT, Key JL (1994) A soybean 101-kDa HSPs complements a yeast HSP 104 deletion mutant in acquiring thermotolerance. Plant Cell 6:1889–1897
- Lee R-H, Wang C-H, Huang L-T, Chen S-CG (2001) Leaf senescence in rice plants: cloning and characterization of senescence up-regulated genes. J Exp Bot 52:1117–1121
- Leshem Y, Kuiper P (1996) Is there a GAS (general adaption syndrome) response to various types of environmental stress? Biol Plantarum 38:1–18
- Li PH, Daris DW, Shen Z (1991) High temperature acclimation potential of the common bean: can it be used as a selection criterion for improving crop performance in high-temperature environments? Field Crops Res 27:241–256
- Li R, Brawley SH, Close TJ (1997) Dehydrin-like proteins in fucoid algae. Plant Physiol 114:479

- Liu X, Huang B (2000) Heat stress injury in relation to membrane lipid peroxidation in creeping bent grass. Crop Sci 40:503–510
- Liu L, Zhou Y, Zhou G, Ye R, Zhao L, Li X, Lin Y (2008) Identification of early senescence-associated genes in rice flag leaves. Plant Mol Biol 67:37–55
- Lobell DB, Field CB (2007) Global scale climate-crop yield relationships and the impacts of recent warming. Environ Res Lett 2:1–7
- Lohman KN, Gan S, John MC, Amasino RM (1994) Molecular analysis of natural leaf senescence in Arabidopsis thaliana. Plant Physiol 92:322–328
- Loreto F, Forster A, Durr M, Csiky O, Seufert G (1998) On the monoterpene emission under heat stress and on the increased thermotolerance of leaves of *Quercus ilex* L. fumigated with selected monoterpenes. Plant. Cell Environ 21:101–107
- Lund AA, Blum PH, Bhattramakki D, Elthon TE (1998) Heat-stress response of maize mitochondria. Int J Plant Sci 159:39–45
- Machado S, Paulsen GM (2001) Combined effects of drought and high temperature on water relations of wheat and sorghum. Plant Soil 233:179–187
- Maestri E, Klueva N, Perrotta C, Gulli M, Nguyen HT, Marmiroli N (2002) Molecular genetics of heat tolerance and heat shock proteins in cereals. Plant Mol Biol 48:667–681
- Mahan JR, Mauget SA (2005) Antioxidant metabolism in cotton seedlings exposed to temperature stress in the field. Crop Sci 45:2337–2345
- Matysik J, Bhalu AB, Mohanty P (2002) Molecular mechanism of quenching of reactive oxygen species by proline under water stress in plants. Curr Sci 82:525–532
- Mazorra LM, Nunez M, Echerarria E, Coll F, Sanchez-Blanco MJ (2002) Influence of brassinosteriods and antioxidant enzymes activity in tomato under different temperatures. Plant Biol 45:593–596
- Medany MA, Hegazy AK, Kabiel HK, Maez MM (2007) Prediction of seed germination and seedling growth of four crop plants as affected by root zone temperature. World J Agric Sci 3:714–720
- Mishra SK, Tripp J, Winkelhaus S, Tschiersch B, Theres K, Nover L, Scharf K-D (2002) In the complex family of heat stress transcription factors, HsfA1 has a unique role as master regulator of thermotolerance in tomato. Genes Dev 16:1555–1567
- Mittler R (2002) Oxidative stress, antioxidants and stress tolerance. Trends Plant Sci 7:405–410
- Mittler R, Vanderauwera S, Gollery M, Van Breusegem F (2004) Reactive oxygen gene network of plants. Trends Plant Sci 9:490–498
- Momcilovic I, Ristic Z (2007) Expression of chloroplast protein synthesis elongation factor, EF-Tu, in two lines of maize with contrasting tolerance to heat stress during early stages of plant development. J Plant Physiol 164:90–99
- Monjardino P, Smith AG, Jones RJ (2005) Heat stress effects on protein accumulation of maize endosperm. Crop Sci 45:1203–1210

- Morales D, Rodrigues P, Bellamico J, Nichotes E, Torrecillas A, Sanchez-Blanco MJ (2003) High temperature preconditioning and thermal shock imposition affects water relations, gas exchanges and root hydraulic conductivity in tomato. Biol Plantarum 47:203–208
- Moriarty T, West R, Small G, Rao D, Ristic Z (2002) Heterologous expression of maize chloroplast protein synthesis elongation factor (EF-TU) enhances *Escherichia coli* viability under heat stress. Plant Sci 163:1075–1082
- Mundy J, Chua NH (1988) Abscisic acid and water-stress induce the expression of a novel rice gene. EMBO J 7:2279–2286
- Nam HG (1997) The molecular genetic analysis of leaf senescence. Curr Opin Biotechnol 8:200–207
- Nautiyal PC, Shono M (2010) Analysis of the role of mitochondrial and endoplasmic reticulum localized small heat shock proteins in tomato. Biol Plantarum 54:715–719
- Neumann D, Emmerman M, Thierfelder JM, Nieden UZ, Clericus M, Braun HP, Nover L, Schmitz UK (1993) HSP68: a DnaK-like heat-stress protein of plant mitochondria. Planta 190:32–43
- Nguyen HT (1999) Sorghum gene mapping develops stay-green line. Sci Daily. http://www.sciencedaily. com/releases/1999/07/990726101241.htm. Accessed 2 July 2009
- Nieto-Sotelo J, Martínez LM, Ponce G, Cassab GI, Alagón A, Meeley RB, Ribaut J-M, Yang R (2002) Maize HSP101 plays important roles in both induced and basal thermotolerance and primary root growth. Plant Cell 14:1621–1633
- Nishida I, Murata N (1996) Chilling sensitivity in plants and cyanobacteria: The crucial contribution of membrane lipids. Annu Rev Plant Physiol Plant Mol Biol 47:541–568
- Nogueira FTS, De Rosa VE, Menossi M, Ulian EC, Arruda P (2003) RNA expression profiles and data mining of sugarcane response to low temperature. Plant Physiol 132:1811–1824
- Nover L, Bharti K, Doring P, Mishra SK, Ganguli A, Scharf K-D (2001) Arabidopsis and the Hsf world: How many heat stress transcription factors do we need? Cell Stress Chapterones 6:177–189
- Oh SA, Lee SY, Chung IK, Lee C-H, Nam HG (1996) A senescence-associated genes of Arabidopsis thaliana is distinctively regulated during natural and artificially induced leaf senescence. Plant Mol Biol 30:739–754
- Oh M-H, Kim Y-J, Lee C-H (2000) Leaf senescence in a stay green mutant of Arabidopsis thaliana: Disassembly process of photosystem I and II during dark-incubation. J Biochem Mol Biol 33:256–262
- Oren-Shamir M, Nissim-Levi A (1999) Temperature and gibberellin effect on growth and anthocyanin pigmentation in *Photinia* leaves. J Hortic Sci Biotechnol 74:355–360
- Oshino T, Abiko M, Saito R, Ichiishi E, Endo M, Kawagishi-Kobayashi M, Higashitani A (2007) Premature progression of anther early developmental programs accompanied by comprehensive alterations

in transcription during high-temperature injury in barley plants. Mol Genet Genomics 278:31–42

- Park SY, Yu J-W, Park J-S, Li J, Yoo S-C, Lee N-Y, Lee S-K, Jeong S-W, Seo HS, Koh H-J, Jeon J-S, Park Y-II, Paek N-C (2007) The senescence-induced staygreen protein regulates chlorophyll degradation. Plant Cell 19:1649–1664
- Parsell PA, Lindquist S (1993) The function of heat-shock proteins in stress tolerance: degradation and reactivation of damaged proteins. Annu Rev Genet 27:437–496
- Parsell DA, Kowal AS, Singer MA, Lindquist S (1994) Protein disaggregation mediated by heat-shock protein Hsp104. Nature 372:475–478
- Pastenes C, Horton P (1996a) Effect of high temperature on photosynthesis in beans. I. Oxygen evolution and chlorophyll fluorescence. Plant Physiol 112:1245–1251
- Pastenes C, Horton P (1996b) Effect of high temperature on photosynthesis in beans. II. CO₂ assimilation and metabolite contents. Plant Physiol 112:1253–1260
- Perisic O, Xiao H, Lis JT (1989) Stable binding of Drosophila heat shock factor to head-to-head and tailto-tail repeats of a conserved 5 bp recognition unit. Cell 59:797–806
- Potters G, Pasternak TP, Guisez Y, Palme KJ, Jansen MAK (2007) Stress-induced morphogenic responses: growing out of trouble. Trends Plant Sci 12:98–105
- Porter JR (2005) Rising temperatures are likely to reduce crop yields. Nature 436:174
- Prasinos C, Krampis K, Samakovli D, Hatzopoulos P (2005) Tight regulation of expression of two Arabidopsis cytosolic Hsp90 genes during embryo development. J Exp Bot 56:633–644
- Provart NJ, Gil P, Chen W, Han B, Chang H-S, Wan X, Zhu T (2003) Gene expression phenotypes of *Arabidopsis* associated with sensitive to low temperatures. Plant Physiol 132:893–906
- Pulla RK, Kim Y-J, Kim MK, Senthil KS, In J-G, Yang DC (2007) Isolation of a novel dehydrin gene from *Codonopsis lanceolata* and analysis of its response to abiotic stresses. BMB Reprod 41:338–343
- Quan R, Shang M, Zhang H, Zhao Y, Zhang J (2004) Engineering of enhanced glycine betaine synthesis improves drought tolerance in maize. Plant Biotechnol J 2:477–486
- Rahman H, Malik SA, Saleem M (2004) Heat tolerance and upland cotton during fruiting stage evaluated using cellular membrane thermostability. Field Crops Res 85:149–158
- Rajcan I, Tollenaar M (1999) Source: sink ratio and leaf senescence in maize: II. Nitrogen metabolism during grain filling. Field Crops Res 60:255–265
- Ranney TG, Peet MM (1994) Heat tolerance of five taxa of Brich (*Betula*). Physiological responses to supraoptimal leaf temperatures. J Am Soc Hortic Sci 119:243–248
- Rasheed R (2009) Salinity and extreme temperature effects on sprouting buds of sugarcane Saccharum

officinarum L.): Some histological and biochemical studies. Ph.D. thesis, Department of Botany, University of Agriculture, Faisalabad, Pakistan

- Rehman AU, Habib I, Ahmad N, Hussain M, Khan MA, Farooq J, Ali MA (2009) Screening wheat germplasm for heat tolerance at terminal growth stage. Plant Omics J 2:9–19
- Ren GD, An K, Liao Y, Zhou X, Cao YJ, Zhao HF, Ge XC, Kuai BK (2007) Identification of novel chloroplast protein AtNYE1 regulating chlorophyll degradation during leaf senescence in Arabidopsis. Plant Physiol 144:1429–1441
- Ren C, Bilyeu KD, Beuselinck PR (2009) Composition, vigor, and proteome of mature soybean seeds developed under high temperature. Crop Sci 49:1010–1022
- Reynolds TL, Bewley JD (1993) Abscisic acid enhances the ability of the dessication – tolerant fern *Polypodium virginianum* to withstand drying. J Exp Bot 269:1771–1779
- Riley GJP (1981) Effects of high temperature on the germination of maize (Zea mays L.). Planta 151:68–74
- Ristic Z, Cass DD (1992) Chloroplast structure after water and high temperature stress in two lines of maize that differ in endogenous levels of abscisic acid. Int J Plant Sci 153:186–196
- Ristic Z, Wilson K, Nelsen C, Momcilovic I, Kobayashi S, Meeley R, Muszynski M, Habben J (2004) A maize mutant with decreased capacity to accumulate chloroplast protein synthesis elongation factor (EF-Tu) displays reduced tolerance to heat stress. Plant Sci 167:1367–1374
- Ristic Z, Bukovnik U, Prasad PVV (2007) Correlation between heat stability of thylakoid membranes and loss of chlorophyll in winter wheat under heat stress. Crop Sci 47:2067–2073
- Ristic Z, Bukovnik UK, Momcilovic I, Fu J, Prasad PVV (2008) Heat-induced accumulation of chloroplast protein synthesis elongation factor, EF-Tu, in winter wheat. J Plant Physiol 165:192–202
- Rivero RM, Ruiz JM, Garcýá PC, Lopez-Lefebre LR, Sańchez E, Romero L (2001) Resistance to cold and heat stress: accumulation of phenolic compounds in tomato and watermelon plants. Plant Sci 160:315–321
- Rizhsky L, Hongjian L, Mittler R (2002) The combined effect of drought stress and heat shock on gene expression in tobacco. Plant Physiol 130:1143–1151
- Rokka A, Aro E-M, Herrmann RG, Andersson B, Vener AV (2000) Dephosphorylation of photosynthetic membranes as an immediate response to abrupt elevation of temperature. Plant Physiol 123:1525–1535
- Rontein D, Basset G, Hanson AD (2002) Metabolic engineering of osmoprotectant accumulation in plants. Metab Eng 4:49–56
- Rurek M (2010) Diverse accumulation of several dehydrin-like proteins in cauliflower (*Brassica oleracea* var. botrytis), *Arabidopsis thaliana* and yellow lupin (*Lupinus luteus*) mitochondria under cold and heat stress. BMC Plant Biol 10:181

- Saadalla MM, Quick JS, Shanahan JF (1990) Heat tolerance in winter wheat. II. Membrane thermostability and field performance. Crop Sci 30:1248–1251
- Sachray L, Weiss D, Reuveni M, Nissim-Levi A, Shamir MO (2002) Increased anthocyanin accumulation in aster flowers at elevated temperatures due to magnesium treatment. Physiol Plantarum 114:559–565
- Sairam RK, Tyagi A (2004) Physiology and molecular biology of salinity stress tolerance in plants. Curr Sci 86:407–421
- Sairam RK, Srivastava GC, Saxena DC (2000) Increased antioxidant activity under elevated temperature: a mechanism of heat tolerance in wheat genotypes. Biol Plantarum 43:245–251
- Sakamoto A, Murata N (2002) The role of glycine betaine in the protection of plants from stress: clues from transgenic plants. Plant Cell Environ 25:163–171
- Sakamoto K, Urishidani T, Nago T (1998) Translocation of *Hsp* 27 to cytoskeleton by repetitive hypoxia–reoxygenation in the rat myoblast cell line, H9c2. Biochem Biophys Res Commun 251:576–579
- Sakata T, Takahashi H, Nishiyama I, Higashitani A (2000) Effects of high temperature on the development of pollen mother cells and microspores in barley (*Hordeum* vulgare L.). J Plant Res 113:395–402
- Salah HBH, Tardieu F (1996) Quantitative analysis of the combined effects of temperature, evaporative demand and light on leaf elongation rate in well watered field and laboratory grown maize plants. J Exp Bot 47:1689–1698
- Santarius KA (1973) The protective effect of sugars on chloroplast membranes during temperature and water stress and its relation to frost, desiccation and heat resistance. Planta 113:105–114
- Sasmita D, Mohanty N (2002) Response of seedlings to heat-stress in cultivars of wheat: Growth temperaturedependent differential modulation of photosystem 1 and 2 activity, and foliar antioxidant defense capacity. J Plant Physiol 159:49–59
- Sato S, Kamiyama M, Iwata T, Makita N, Furukawa H, Ikeda H (2006) Moderate increase of mean daily temperature adversely affects fruit set of *Lycopersicon esculentum* L. by disrupting specific physiological processes in male reproductive development. Ann Bot 97:731–738
- Sato Y, Morita R, Nishimura M, Yamaguchi H, Kusaba M (2007) Mendel's green cotyledon gene encodes a positive regulator of the chlorophyll-degrading pathway. Proc Natl Acad Sci USA 104:14169–14174
- Savchenko GE, Klyuchareva EA, Abramck LM, Serdyuchenko EV (2002) Effect of periodic heat shock on the inner membrane system of etioplasts. Russ J Plant Physiol 49:349–359
- Scandalios JG (1993) Oxygen stress and superoxide dismutases. Plant Physiol 101:7–12
- Schöffl F, Praendl R, Reindl A (1998) Regulation of heatshock response. Plant Physiol 117:1135–1141
- Sgherri C, Stevanovic B, Navari-Izzo F (2004) Role of phenolics in the antioxidative status of the resurrection

plant *Ramonda serbica* during dehydration and rehydration. Physiol Plantarum 122:478–485

- Shah NH, Paulsen GH (2003) Interaction of drought and high temperature on photosynthesis and grain filling of wheat. Plant Soil 257:219–226
- Sharkey TD (2005) Effects of moderate heat stress on photosynthesis: importance of thylakoid reactions, rubisco deactivation, reactive oxygen species, and thermotolerance provided by isoprene. Plant Cell Environ 28:269–277
- Sharkey TD, Badger MR, Von Caemmerer S, Andrews TJ (2001) Increased heat sensitivity of photosynthesis in tobacco plants with reduced Rubisco activase. Photosynth Res 67:147–156
- Sheffield WP, Shore GC, Randall SK (1990) Mitochondrial precursor protein: Effect of 70-kDa heat shock protein on polypeptide folding, aggregation and import competence. J Biol Chem 265:11069–11076
- Sibley JZ, Rutter JM, Eaks DJ (1999) High temperature tolerance of roots container grown red maple cultivars. Proc South Nurserymens Assoc Res Conf 44:24–28
- Siddique KHM, Loss SP, Regan KL, Jettner RL (1999) Adaptation and seed yield of cool season grain legumes in Mediterranean environments of south-western Australia. Aust J Agric Res 50:375–387
- Simões-Araújo JL, Rumjanek NG, Margis-Pinheiro M (2003) Small heat shock proteins genes are differentially expressed in distinct varieties of common bean. Braz J Plant Physiol 15:33–41
- Singh K, Foley RC, Oñate-Sánchez L (2002) Transcriptional factors in plant defense and stress responses. Curr Opin Plant Biol 5:430–436
- Sinsawat V, Leipner J, Stamp P, Fracheboud Y (2004) Effect of heat stress on the photosynthetic apparatus in maize (*Zea mays* L.) grown at control or high temperature. Environ Exp Bot 52:123–129
- Smart CM, Scofield SR, Bevan MW, Dyer TA (1991) Delayed leaf senescence in tobacco plants transformed with tmr, a gene for cytokinin production in *Agrobacterium*. Plant Cell 3:647–656
- Smart CM, Hosken SE, Thomas H, Greaves JA, Blair BC, Schuch W (1995) The timing of maize leaf senescence and characterization of senescence-related cDNAs. Physiol Plantarum 93:637–682
- Smirnoff N (1993) The role of active oxygen in the response of plant to water deficient and desiccation. New Phytol 125:27–58
- Spano G, Fonzo ND, Perrotta C, Platani C, Ronga G, Lawlor DW, Napier JA, Shewry PR (2003) Physiological characterization of 'staygreen' mutants in durum wheat. J Exp Bot 54:1415–1420
- Sreenivasulu N, Radchuk V, Strickert M, Miersch O, Weschke W, Wobus U (2006) Gene expression patterns reveal tissue-specific signalling networks controlling programmed cell death and ABA-regulated maturation in developing barley seeds. Plant J 47:310–327
- Sullivan ML, Green PJ (1993) Post-transcriptional regulation of nuclear-encoded genes in higher plants: the

roles of mRNA stability and translation. Plant Mol Biol 23:1091–1104

- Sun W, Montagu MV, Verbruggen N (2002) Small heat shock proteins and stress tolerance in plants. Biochim Biophys Acta 1577:1–9
- Sung D-Y, Kaplan F, Lee K-J, Guy CL (2003) Acquired tolerance to temperature extremes. Trends Plant Sci 8:179–187
- Svensson J, Ismail AM, Palva ET, Close TJ (2002) Dehydrins. In: Storey KB, Storey JM (eds) Sensing, signaling and cell adaptation. Elsevier, Amsterdam, pp 155–171
- Szilvia ZT, Schansker G, Kissimon J, Kovács L, Garab G, Strasser RJ (2005) Biophysical studies of photosystem II-related recovery processes after a heat pulse in barley seedlings (*Hordeum vulgare* L.). J Plant Physiol 162:181–194
- Taiz L, Zeiger E (2006) Plant physiology, 4th edn. Sinauer Associates, Sunderland, MA
- Talwar HS, Takeda H, Yashima S, Senboku T (1999) Growth and photosynthetic responses of groundnut genotypes o high temperature. Crop Sci 39:460–466
- Tao YZ, Henzell RG, Jordan DR, Butler DG, McIntyre AM (2000) Identification of genomic regions associated with stay-green in sorghum by testing RILs in multiple environments. Theor Appl Genet 100:1225–1232
- Thomas H, Howarth CJ (2000) Five ways to stay green. J Exp Bot 51:329–337
- Tian X-J, Tao ZH, Luo J, Zhang X-D, Tao F-Q (2009) Physiological effect of heat stress on pea (*Pisum sativum*) hypocotyls. Acta Bot Yunnan 31:363–368
- Todorov DT, Karanov EN, Smith AR, Hall MA (2003) Chlorophyllase activity and chlorophyll content in wild type and eti5 mutant of *Arabidopsis thaliana* subjected to low and high temperatures. Biol Plantarum 46:633–636
- Tollenaar M, Daynard TB (1978) Leaf senescence in short season maize hybrids. Can J Plant Sci 58:869–874
- Tsugeki R, Mori H, Nishimura M (1992) Purification, cDNA cloning and northern blot analysis of mitochondrial chaperonin 60 from pumpkin cotyledons. Eur J Biochem 209:453–458
- Tsukaguchi T, Kawamitsuy Y, Takada H, Suzuki K, Egawa Y (2003) Water status of flower buds and leaves as affected by high temperature in heat tolerant and heat sensitive cultivar of snap bean (*Phaseolus vul*garis L.). Plant Prod Sci 6:24–27
- Velikova V, Edreva A, Loreto F (2004) Endogenous isoprene protects *Phragmites australis* leaves against singlet oxygen. Physiol Plantarum 122:219–225
- Venter HA, Hoffman R, Wolfaardt M (1997) Thermotolerance of mesocotyls and coleoptiles of selected maize (*Zea mays* L.) cultivar seed lots. Appl Plant Sci 11:39–42
- Vierling E (1991) The role of heat shock proteins in plants. Annu Rev Plant Physiol Plant Mol Biol 42:579–620
- Vijayalakshmi K, Fritz AK, Paulsen GM, Bai GH, Pandravada S, Gill BS (2010) Modeling and mapping

QTL for senescence-related traits in winter wheat under high temperature. Mol Breed 26:163–175

- Vilardell J, Goday A, Freire MA, Torrent M, Martinez C, Torne JM, Pages M (1990) Gene sequence, development expression, and protein phosphorylation of RAB-17 in maize. Plant Mol Biol 14:423–432
- Vu JCV, Gesch RW, Pennanen AH, Allen LHJ, Boote KJ, Bowes G (2001) Soybean photosynthesis, Rubisco and carbohydrate enzymes function at supra-optimal temperatures in elevated CO₂. J Plant Physiol 158:295–307
- Wahid A (2007) Physiological implications of metabolites biosynthesis in net assimilation and heat stress tolerance of sugarcane (*Saccharum officinarum*) sprouts. J Plant Res 120:219–228
- Wahid A, Close TJ (2007) Expression of dehydrins under heat stress and their relationship with water relations of sugarcane leaves. Biol Plantarum 51:104–109
- Wahid A, Shabbir A (2005) Induction of heat stress tolerance in barley seedlings by pre-sowing seed treatment with glycinebetaine. Plant Growth Regul 46:133-141
- Wahid A, Tariq A (2008) Plant phenolics and their role in abiotic stress tolerance. In: Khan NA, Singh S (eds) Plant responses to abiotic stresses. Narosa, New Dehli, pp 44–55
- Wahid A, Gelani S, Ashraf M, Foolad MR (2007) Heat tolerance in plants: An overview. Environ Exp Bot 61:199–223
- Wahid A, Sehar S, Perveen M, Gelani S, Basra SMA, Farooq M (2008) Seed pretreatment with hydrogen peroxide improves heat tolerance in maize at germination and seedling growth stages. Seed Sci Technol 36:633–645
- Wahid A, Rasul E (2005) Photosynthesis in leaf stem, flower and fruit. In: Pessarakli M (ed) Handbook of Photosynthesis, 2nd edition. CRC press, Boca Raton, Florida, pp 479–497
- Wang W, Vinocur B, Shoseyov O, Altman A (2004) Role of plant heat shock proteins and molecular chaperones in the abiotic stress response. Trends Plant Sci 9:244–252
- Waters ER, Lee GJ, Vierling E (1996) Evolution, structure and function of the small heat shock proteins in plants. J Exp Bot 47:325–338
- Watts FZ, Walters AJ, Moore AL (1992) Characterisation of PHSP1, a cDNA encoding a mitochondrial HSP70 from *Pisum sativum*. Plant Mol Biol 18:23–32
- Weaich K, Briston KL, Cass A (1996) Modeling preemergent maize shoot growth. II. High temperature stress conditions. Agron J 88:398–403
- Weaver LM, Froehlich JE, Amasino RM (1999) Chloroplast-targeted ERD1 protein declines but its mRNA increases during senescence in Arabidopsis. Plant Physiol 119:1209–1216
- Weggle H, Muller L, Buchner J (2004) Hsp70 and Hsp90 a relay team for protein folding. Rev Physiol Biochem Phamacol 151:1–44
- Wilhelm EP, Mullen RE, Keeling PL, Singletary GW (1999) Heat stress during grain filling in maize: Effects

on kernel growth and metabolism. Crop Sci 39:1733–1741

- Wollenweber B, Porter JR, Schellberg J (2003) Lack of interaction between extreme high temperature events at vegetative and reproductive growth stages in wheat. J Agron Crop Sci 189:142–150
- Xu S, Li J, Zhang X, Wei H, Cui L (2006) Effects of heat acclimation pretreatment on changes of membrane lipid peroxidation, antioxidant metabolites, and ultrastructure of chloroplasts in two cool-season turfgrass species under heat stress. Environ Exp Bot 56:274–285
- Yamakawa H, Hirose T, Kuroda M, Yamaguchi T (2007) Comprehensive expression profiling of rice grain filling-related genes under high temperature using DNA microarray. Plant Physiol 144:258–277
- Yang KA, Lim CJ, Hong JK, Park CY, Cheong YH, Chung WS, Lee KO, Lee SY, Cho MJ, Lim CO (2006) Identification of cell wall genes modified by a

permissive high temperature in Chinese cabbage. Plant Sci 171:175–182

- Young TE, Ling J, Geisler-Lee CJ, Tanguay RL, Caldwell C, Gallie DR (2001) Developmental and thermal regulation of maize heat shock protein, HSP101. Plant Physiol 127:777–791
- Young LW, Wilen RW, Bonham-Smith P-C (2004) High temperature stress of *Brassica napus* during flowering reduces micro-and megagametophyte fertility, induces fruit abortion, and disrupts seed production. J Exp Bot 55:485–495
- Zhang J-H, Huang WD, Liu Y-P, Pan Q-H (2005) Effects of temperature acclimation pretreatment on the ultrastructure of mesophylls in young grape plants (*Vitis vinifera* L. cv. Jingxiu) under cross-temperature stresses. J Integr Plant Biol 47:959–970
- Zhu JK (2003) Regulation of ion homeostasis under salt stress. Curr Opin Plant Biol 6:441–445

Understanding Chilling Tolerance Traits Using *Arabidopsis* Chilling-Sensitive Mutants

7

Dana Zoldan, Reza Shekaste Band, Charles L. Guy, and Ron Porat

Abstract

Many plants of tropical and subtropical origin are severely damaged when exposed to chilling temperatures between 2 and 15°C. In contrast, the cruciferous plant Arabidopsis thaliana is chilling tolerant and, therefore provides an alternative model plant system for the identification of chilling tolerance traits. In this chapter, we describe physiological, biochemical, and molecular responses of Arabidopsis class 1 chilling-sensitive (chs) mutants to low temperatures. These mutants, including chs1, chs2 and chs3, are extremely chilling-sensitive and wilt and turn yellow in just a few days after transfer to low temperatures of 4-13°C. Overall, following exposure to chilling, class 1 chs mutants suffer from: (1) loss of chlorophyll and decrease in photosynthetic efficacy resulting in lack of starch accumulation, (2) damage to cellular membranes resulting in increased electrolyte leakage, and (3) accumulation of the reactive oxygen species (ROS) hydrogen peroxide (H_2O_2) . At the molecular level, transcriptome analysis studies following exposure to 10°C for 48 h using the Affymetrix ATH1 genome array reveal remarkable changes in expression patterns of between 1,500 and 3,000 genes, which are significantly differentially expressed ($p \le 0.05$ and up- or down-regulated by a factor of at least 4) in chs1, chs2, and chs3 mutants compared to wild-type (WT) plants. The main functional categories of up-regulated genes by chilling include "stress," "protein," and "signaling," whereas the main categories down-regulated by chilling were "photosynthesis," "tetrapyrrole synthesis," "carbohydrate metabolism," "cell wall," and "lipid metabolism". Overall, these and other studies using Arabidopsis chilling-sensitive mutants allow the recognition of major genetic traits crucial for plant survival under chilling conditions.

R. Porat (⊠) • D. Zoldan Department of Postharvest Sciences of Fresh Produce, ARO, the Volcani Center, P.O. Box 6, Bet Dagan 50250, Israel e-mail: rporat@volcani.agri.gov.il

R.S. Band • C.L. Guy Department of Environmental Horticulture, University of Florida, Gainesville, FL 32611, USA

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_7, © Springer Science+Business Media, LLC 2012

Keywords

Arabidopsis • Low temperature stress • ROS • Transcriptome • Physiological response • Biochemical response • Gene mapping

1 Introduction

Low temperature is an important environmental factor that greatly influences the growth, development, survival, and geographical distribution of plants (Levitt 1980). Whereas most plant species from temperate regions can acclimatize to cold and can survive exposures to deep freezing temperatures, plants of tropical and subtropical origin are severely injured when exposed to low nonfreezing temperatures between 0 and 15°C (Lyons 1973; Lynch 1990; Wang 1990). Symptoms of chilling injury often include cessation of growth, wilting, chlorosis, necrosis, and eventually plant death (Lyons 1973; Graham and Patterson 1982; Maruyama et al. 1990; Allen and Ort 2001). In addition to its adverse effects on plant growth and development, chilling sensitivity also imposes major limitations on the postharvest storage and handling of many fruits and vegetables, because it necessitates storage at relatively high temperatures that do little to delay deterioration and spoilage (Paull 1990).

In contrast to our knowledge of plant responses to other abiotic stresses, such as freezing, drought, salinity, and heat, little is known regarding the molecular basis in regulating chilling tolerance or about the signal transduction networks involved in its acquisition. In previous reviews, the occurrence of chilling damage was attributed mainly to the general disruption or dysfunction of cellular metabolic processes (Lyons 1973; Graham and Patterson 1982; Markhart 1986). In this respect, it has been suggested that an important primary event in the occurrence of chilling injury is an alteration in the physical state of the cellular membranes, which leads to dysfunctional selective permeability and increases in solute leakage from the cells (Lyons 1973; Markhart 1986; Nishida and Murata 1996). Another factor that affects susceptibility to chilling is the status of

the cellular antioxidant defensive system required to avoid accumulation of toxic reactive oxygen species (ROS). Indeed, overexpression of antioxidant defensive genes, such as ascorbate peroxidase, superoxide dismutase, and catalase (CAT) enhanced chilling tolerance (Van Breusegem et al. 1999; Payton et al. 2001), whereas repression of catalase gene expression reduced chilling tolerance (Kerdnaimongkol and Woodson 1999). Finally, several reports have suggested that various stress genes, usually related to other types of stress responses, may also contribute to the acquisition of chilling tolerance. For instance, it has been suggested that heat shock proteins (HSPs) (Sabehat et al. 1998) and dehydrin genes (Ismail et al. 1999) may also be factors in chilling tolerance in plants.

In this chapter, we will summarize data obtained from studies on *Arabidopsis* chillingsensitive (*chs*) mutants, and will suggest how the adoption of *Arabidopsis* as a plant model system could improve our basic understanding regarding the molecular and biochemical nature of chilling tolerance. Overall, *Arabidopsis* has many advantages as a plant model system and its chilling tolerance makes it an excellent study subject for the identification of major genetic traits important for low temperature survival of plants.

2 Using Arabidopsis as a Genetic Resource for Identification of Chilling Tolerance Traits

It is believed that chilling-sensitive species evolved in warm tropical regions where there was no selection pressure favoring growth at low temperatures. In contrary, the dispersal of plants to cooler climates necessitated the acquisition of chilling tolerance and other low temperature tolerance traits. Indeed, trees and shrubs in temperate regions are able to grow at low chilling temperatures, and become dormant during the coldest portion of the winter when temperatures are below 0°C (Levitt 1980). Many herbaceous annual species such as Arabidopsis normally overwinter in their vegetative state, survive both chilling and freezing conditions, and switch to a reproductive stage during the spring, when temperatures rise. Thus, over-wintering herbaceous annual species can be very resistant to chilling, and continue their growth and development, albeit more slowly, at low temperatures. Indeed, it has been shown that Arabidopsis can complete its entire life cycle and produce fertile seeds when sown and grown under continuous chilling temperatures of 4-6°C (Hasdai et al. 2006). Therefore, naturally chilling-resistant plants, such as Arabidopsis, may provide valuable genetic resources for the identification of chilling tolerance traits which, in the future, may be incorporated into horticultural important chilling-sensitive crops (Tokuhisa and Browse 1999; Porat and Guy 2007).

Arabidopsis mutants and forward genetic approaches have become useful tools to study the molecular and physiological responses and traits of plants to low temperature stress. Because Arabidopsis is freezing tolerant and can modulate its freezing tolerance by cold acclimation, and is easily genetically modified, creating and characterization of mutants deficient in freezing tolerance has been a productive strategy to uncover major determinants of signaling pathways and regulators of gene expression at low temperatures. Forward genetic analysis has identified a number of transcription factors like HOS9, HOS10 and other regulators of freezing tolerance like ESKIMO1 (ESK1). HOS9 and HOS10 encode a homeodomain and MYB transcription factors, respectively, (Zhu et al. 2007). Loss-offunction mutations in HOS9 and HOS10 cause significant decreases in basal and acquired freezing tolerance. Mutations in ESK1 (Xin and Browse 1998) result in constitutive freezing tolerance. Also, mutations in a transcriptional adaptor protein ADA2b causes constitutive freezing tolerance (Vlachonasios et al. 2003). Another

example of forward genetics includes the *chy1–10* mutant that is freezing-sensitive after cold acclimation. The *chy1* mutant accumulates ROS. Mapbased cloning of *CHY1* revealed that the mutant encodes a peroxisomal β -hydroxyisobutyryl–CoA hydrolase needed for valine catabolism and fatty acid β -oxidation, and suggests a role in cold stress signaling, and freezing tolerance for peroxisomal metabolism (Dong et al. 2009).

Arabidopsis mutants have also helped to bring light to the relationships of lipid metabolism and low-temperature exposure particularly as the isolation of mutants with altered lipid compositions has assisted in biochemical and molecular approaches to understanding lipid metabolism and membrane function (Wallis and Browse 2002). Further, the availability of a variety of plant lines with specific changes in membrane lipids has afforded valuable resources to study the structural and adaptive roles of lipids. Presently, there are at least five types of Arabidopsis fatty acid metabolism mutants that grow well at 22°C, but are injured at low temperatures (2-6°C). The mutant lines include fab1 (Wu et al. 1997), fad2 (Miquel et al. 1993), fad5, fad6 (Hugly and Somerville 1992), and the fad3-2 fad7-2 fad8 triple mutant (Routaboul et al. 2000). Some examples that could be equated to chilling injury include the *fab1* mutant of *Arabidopsis* thaliana that was derived from the Columbia ecotype following mutagenesis with ethane methyl sulfonate (EMS). The mutant contains increased levels of saturated fatty acids, particularly increased proportions of 16:0 in all of the major membrane lipids of the leaf tissue (Lightner et al. 1994). When subjected to 2°C, after 14 days, major changes in chloroplast ultrastructure were observed (Wu et al. 1997). The mutant chloroplasts were irregular in shape with poorly defined and broken envelopes. By 21 days at a low temperature, chloroplasts had largely disappeared from the mutant, but some chloroplast remnants remained visible in most mesophyll cells. The loss of chloroplast ultrastructure correlated with a major loss of photosynthetic function as indicated by Chlorophyll a fluorescence data. Nevertheless, the *fab1* plants were able to largely recover upon return to 22°C. Given that fab1 plants are unable to maintain photosynthetic function at 2°C and will die after 5-7 weeks, the sensitivity of *fab1* plants to 2°C may result from increased levels of phosphatidylglycerol (PG) having high-melting-point molecular species (containing only 16:0, 18:0, and 16:1, Δ^3 -trans fatty acids) (Kim et al. 2010). A fab1 suppressor line, S7, had reduced levels of 16:3 fatty acid in leaf galactolipids compared to WT and was identified by map-based cloning as a hypomorphic allele of lysophosphatidic acid acyltransferase1 (lpat1), lpat1-3. The lpat1-3 mutation was found to strongly affect fatty acid composition of PG, with the proportion of highmelting-point molecular species in PG being reduced from 48.2% in fab1 to 10.7% in fab1 *lpat1-3* (S7), a value that was close to the 7.6%found in wild type.

The ability to modulate membrane fatty acid unsaturation has long been thought to be critical to membrane function and cell viability for poikilothermic organisms like plants. The Arabidopsis fad3-2, fad7-2, fad8 triple fatty acid desaturase mutant was found to be only subtly impacted upon short-term exposure to low temperatures, with small decreases in photosynthetic quantum yield, ΦII , were observed in the mutant (Routaboul et al. 2000). However, long-term exposure to 4°C resulted in lower fluorescence parameters, chlorophyll content, photosynthetic processes, and thylakoid membranes in the triplemutant. When taken together, the examples outlined above demonstrate the value in forward genetic approaches to understanding plant lowtemperature responses.

Overall, in contrast to nearly all chillingsensitive species, with *Arabidopsis* it is relatively easy to screen large mutagenized populations for individuals sensitive to chilling phenotype resulting from a loss-of-function mutation in an important gene. Furthermore, there are *Arabidopsis* populations with insertional mutations, in which the compromised genetic locus can be readily obtained, and the mutated gene can be isolated and sequenced. Additional follow-up complementation studies can confirm the linkage of a mutant gene with the ability of the plant to tolerate chilling.

3 *Arabidopsis* Chilling-Sensitive Mutants

In order to study the molecular basis of chilling tolerance in Arabidopsis, EMS-mutagenized M₂ populations of Arabidopsis were grown at 22°C for 2 weeks after which they were transferred to 10 or 15°C and were screened for the appearance of chilling-sensitive mutants (Schneider et al. 1995a). Of about 20,000 M₂ plants examined, 21 mutants were identified that appeared normal at 22°C, but developed chlorosis or necrosis when shifted to lower temperatures (Schneider et al. 1995a). The chilling-sensitive mutants were categorized into four different phenotypic classes according to the severity of their chilling damage symptoms: class 1 mutants (chs1-3) turned yellow, wilted and died, in class 2 mutants (chs4) only the mature leaves became necrotic, and class 3 mutants (chs5-6) developed yellow chlorotic patches, but continued to grow and develop at low temperatures; and in class 4 (chs7-15) only part of the leaf near the rosette turned yellow (Schneider et al. 1995a). Crosses among mutants in different phenotypic classes showed that those in the first three classes were found only in a small number of loci (Schneider et al. 1995a).

Detailed characterization of the chs1 mutant revealed that it was sensitive to temperatures below 18°C, and that exposure to low temperatures resulted in defects in chloroplast maintenance and integrity, including disruption of chloroplast protein accumulation and altered steryl-ester metabolism (Hugly et al. 1990; Patterson et al. 1993; Schneider et al. 1995b). Furthermore, it was found that only the leaf tissues of chs1 plants were injured at low temperatures whereas germination, root and callus growth were unaffected by chilling. All these findings suggest that the function of the chs1 gene product may be required to maintain chloroplast function at low temperatures. Overall, the chs1 mutant was found to be extremely sensitive to low temperatures, and after 3 days at 13°C, as the plants become irreversibly injured and could not be rescued upon returning them to normal temperatures (Schneider et al. 1995b). Interestingly, it was also noted that the *chs1* mutants were much more sensitive in terms of leaf yellowing and time till the initiation of wilting to exposure to 15° C than to 5° C (Schneider et al. 1995b).

Transcriptome profiling studies among approximately 8,000 Arabidopsis genes that compared gene expression patterns in wild-type (WT) plants with those in 12 chilling-sensitive mutants showed that the expression of more than 1,000 genes in normal plants was unaffected by chilling at 13°C but was affected by at least twofold in class 1 chs mutants (Provart et al. 2003; Zhu and Provart 2003). In the light of these microarray expression data, it was suggested that the normal functions of the mutated chs1, chs2, and chs3 genes might be to prevent widespread chilling damage effects on transcriptional regulation (Provart et al. 2003). It was further observed that the profiles of gene expression of the various class 1 chs mutants (chs1, chs2, and chs3) at 13°C were very similar to each other, which supports the idea that the products of these genes might perform related biological functions. In the future, identification of the gene products of class 1 chs mutants by means of mapping and chromosome walking technologies will certainly provide important insights into the molecular basis of at least some major factors that are crucial for plant survival at chilling temperatures.

A recent report on the chs3 mutant has demonstrated the expected arrested growth and chlorosis phenotype when grown at 16°C or when shifted from 22 to 4°C. chs3 plants exhibited chlorotic and spontaneous lesion phenotypes when grown at 16°C as older leaves turned yellow and died, and emerging leaves became watersoaked (Yang et al. 2010). Evidence presented indicated that chs3 plants exhibit an activated defense response at 16°C, which was suppressed to WT levels at 22°C. Map-based cloning of chs3 gene revealed an unusual disease resistance protein belonging to the TIR-NB-LRR class and also having a zinc-binding LIM domain at the carboxyl terminus. The mutation of a G-to-A substitution at the ninth intron-exon junction appears to lead to abnormal splicing and the formation of a truncated protein. Thus, chs3 seems to be a gain-of-function mutation as the mutation led to

the constitutive activation of the TIR-NB-LRR domain. An activated defense response was supported by hydrogen peroxide (H_2O_2) analyses with 3,3'-diaminobenzidine (DAB) staining. Strong staining was observed in *chs3* plants grown at 16°C, but not in WT plants under the same conditions indicating H_2O_2 levels were high in the mutant. The *chs3* growth and defense phenotypes could be suppressed by *eds1*, *sgt1b* and *rar1*, and partially suppressed by *pad4* and *nahG*, but not by *npr1* and *ndr1*. These findings reveal an unexpected linkage between defense responses and cold stress, and points to a mutual interaction between cold signaling and defense responses (Yang et al. 2010).

Beside *chs3*, the only other gene among the 21 chs mutants identified by Schneider et al. (1995a) that has been cloned so far is chs5; it belongs to class 3 of chs mutants, which become chlorotic at low temperatures but otherwise continue to grow and develop normally. Genetic and sequence analysis demonstrated that the chs5 mutation occurred in the coding region of 1-deoxy-D-xylulose 5-phosphate synthase (DXS), an enzyme belonging to the nonmevalonate pathway localized in the chloroplast (Araki et al. 2000). DXS functions in the synthesis of isoprenoid compounds like carotenoids, xanthophylls, sterols, and isopentynyl chains of cytokinins and chlorophylls. Once again, as in the case of the chs1 mutation, it seems that among the diverse components of cellular machinery that chloroplast is especially vulnerable to low chilling temperatures.

In another independent study, Tokuhisa et al. (1997) identified additional *Arabidopsis* EMS and T-DNA insertion chilling-sensitive mutants that were indistinguishable from WT plants when grown at 22°C, but exhibited visible chilling symptoms after 42 days of growth at 5°C. Under these conditions, the chilling symptoms that were identified included chlorosis, reduced and impaired growth (small stature, reduced leaf growth, high anthocyanin content, and distorted leaf morphology), necrosis, and death. Thus, two independent populations of chilling-sensitive mutants have been identified so far in *Arabidopsis*: class 1 *chs* mutants described by Somerville and coworkers (Schneider et al. 1995a) which showed

apparent chilling damage symptoms after a short exposure (3-7 days) to mild temperatures (10–15°C), and the mutants described by Browse and coworkers (Tokuhisa et al. 1997) which showed apparent chilling symptoms only after a much longer period of 6 weeks at a very low temperature of 5°C (Schneider et al. 1995a; Tokuhisa et al. 1997). The first type of mutants, especially class 1 chs mutants, most likely encode proteins crucial for basic aspects of cell survival at chilling temperatures, whereas the second type of mutants probably involves governing particular mechanisms of adaptation to low temperatures of older cells and organs or are characteristic of older cells. One of the latter T-DNA-tagged mutants, paleface1 (pfc1), that becomes chlorotic during growth at 5°C, was cloned and encodes a specific 16 S rRNA methylase, which is required for maintenance of a particular step in pre-RNA processing in the chloroplasts, and which is apparently sensitive to low temperatures (Tokuhisa et al. 1998).

Further cloning and isolation of additional *Arabidopsis* chilling-sensitive genes will greatly improve our understanding regarding the molecular basis of chilling tolerance (Tokuhisa et al. 1998; Araki et al. 2000). In the following sections, we will analyze in more details the physiological, biochemical and molecular responses of class 1 chilling-sensitive mutants, including *chs1*, *chs2*, and *chs3*, to low temperatures, and based on that will aim to identify crucial traits required for survival of *Arabidopsis* plants under chilling conditions.

4 Physiological and Biochemical Responses of *chs1*, *chs2*, and *chs3* Mutants to Chilling

Class 1 chilling-sensitive mutants include *chs1*, *chs2*, and *chs3*. In addition to previous observations regarding the responsiveness of *chs1* plants to chilling (Hugly et al. 1990; Patterson et al. 1993; Schneider et al. 1995b), we hereby show that upon transfer of 2-week-old plants from a normal temperature of 22° C to a moderate chilling temperature of 10°C all class 1 chs mutants, including chs1, chs2, and chs3, turned yellow, wilted and eventually died (Fig. 7.1). The yellowing and growth arrest phenotypes become visible in 7 days after transfer to chilling temperatures, and these phenotypes become more severe as the time of exposure to chilling increased up to 3 weeks (Fig. 7.1). Furthermore, we found that in addition to what has been reported previously for chs1 (Hugly et al. 1990), chs3 mutants are also extremely sensitive to low temperatures, as they showed a severe dwarfism phenotype when grown at a moderate chilling temperature of 18°C (Fig. 7.2). Overall, all class 1 chilling-sensitive mutants are very sensitive to chilling, and turn yellow and wilt in just a few days after exposure to low temperatures.

To further evaluate the effects of chilling on chs1, chs2, and chs3 plants, we examined their chlorophyll content, photosynthesis efficacy and electrolyte leakage rates at various periods after exposure to 10°C. It can be seen that the earliest event that was observed within 4 days of exposure to chilling was a sharp decline in chlorophyll content (Fig. 7.3a), and this was followed by a gradual decrease in photosynthetic efficacy (determined by measuring chlorophyll fluorescence; Fv/Fm ratio) which continued up to 21 days of exposure to 10°C (Fig. 7.3b). An increase in electrolyte leakage rates was evident after 7 days at 10°C and gradually became more severe as the plants were kept for longer periods at 10°C (Fig. 7.3c).

In accordance with the observed decreases in chlorophyll contents and photosynthetic efficacy, we found that *chs1*, *chs2*, and *chs3* plants were unable and did not accumulate starch in their leaves as observed in WT plants (Fig. 7.4). Following exposure to low temperatures, WT plants continued to produce assimilates by photosynthesis, but since growth was slowed down they accumulated starch. In contrast, the *chs1*, *chs2*, and *chs3* mutants were defected in their photosynthetic machinery (as observed by the decrease in chlorophyll content and photosynthesis efficacy) and, therefore, did not produce sugars and did not accumulate starch following exposure to low temperatures (Fig. 7.4).



Fig. 7.1 Phenotypes of wild-type and *chs1*, *chs2*, and *chs3* mutants following transfer to chilling conditions. Plants were grown for 2 weeks at 22° C and afterwards transferred to a chilling temperature of 10° C



Fig.7.2 Effects of a moderate growth temperature of 18° C on the phenotypes of wild-type and *chs1*, *chs2*, and *chs3* mutants. Plants were grown from sowing at 22 or 18° C, and photographs were taken after 3 weeks of growth at each temperature


Fig. 7.3 Effects of chilling on chlorophyll content, photosynthesis efficacy, and electrolyte leakage rates of wildtype and *chs1*, *chs2*, and *chs3* mutants. Plants were grown for 2 weeks at 22°C and afterwards transferred to a chilling

Another response of *chs1*, *chs2*, and *chs3* mutants to chilling was enhanced accumulation of ROS in general, and H_2O_2 particularly, as observed by DAB staining experiments (Fig. 7.5). It can be seen that WT plants remained clear and did not accumulate H_2O_2 following exposure to chilling, whereas leaves of *chs1*,

temperature of 10°C. (a) Chlorophyll content, (b) photosynthesis efficacy, and (c), electrolyte leakage. Data are means \pm S.E. of five replications

chs2, and *chs3* mutants were stained by brown spots indicating presence of H_2O_2 (Fig. 7.5). The accumulation of H_2O_2 was somewhat more pronounced in *chs2* and *chs3* plants as compared with *chs1*, and after exposure to the lower temperature of 4°C as compared with 13°C (Fig. 7.5).



Fig. 7.4 Effects of chilling on starch accumulation in wild-type and *chs1*, *chs2*, and *chs3* mutants. Plants were grown for 2 weeks at 22°C and afterwards transferred to

chilling temperatures of 13 and 4°C. Starch accumulation was indicated by iodine staining after 7 days at chilling conditions

5 Mapping of the *chs1*, *chs2*, and *chs3* Genes

Identification of the gene products of class 1 *chs* mutants by means of mapping and chromosome walking technologies will provide important insights into the molecular basis of chilling tolerance traits crucial for plant survival at low temperatures. To facilitate the mapping of *chs1*, *chs2*, and *chs3* genes, we crossed the mutants present in Colombia (Col) background with Lansdberg

erecta (Ler) WT plants, and scanned populations of F_2 plants for chilling sensitivity phenotypes upon transfer to low temperatures. Afterwards, we extracted DNA from chilling-resistant plants and searched for possible linkage of PCR markers with the chilling tolerance phenotype. Accordingly, based on polymorphism between Col and Ler ecotypes, we mapped the *chs1*, *chs2*, and *chs3* genes to the following chromosomal locations (Fig. 7.6):

(1) *chs1* – was mapped to chromosome 1 in the region between 5.9 and 7.0 Mb.



H₂O₂ accumulation (DAB staining)

Fig. 7.5 Effects of chilling on H_2O_2 accumulation in wild-type and *chs1*, *chs2*, and *chs3* mutants. Plants were grown for 2 weeks at 22°C and afterwards transferred to

chilling temperatures of 13 and 4°C. Accumulation of hydrogen peroxide was indicated by DAB staining after 7 days at chilling conditions

- (2) *chs2* was mapped to chromosome 4 in the region between 9.02 and 9.95 Mb.
- (3) *chs3* was mapped to chromosome 5 in the region between 5.41 and 7.37 Mb.

This initial mapping of the *chs1*, *chs2*, and *chs3* genes to a final resolution of approximately 1 Mb will facilitate in the future the final cloning and isolation of these important genes that are crucial for survival of *Arabidopsis* plant at low chilling temperatures. As indicated in Sect. 3, the *chs3* gene was recently mapped to the top of chromosome 5 and its sequence was identified (Yang et al. 2010).

6 Effects of Chilling on the Transcriptome of *chs1, chs2,* and *chs3* Mutants

To identify transcripts that exhibited significant changes in their abundance after exposure to chilling, we performed pair-wise comparisons, and selected transcripts that had ANOVA values of $p \le 0.05$ and that were induced or repressed by a factor of at least 4 after exposure of *chs1*, *chs2*, and *chs3* mutants to 10°C for 48 h, as compared with their corresponding expression levels in WT



Fig. 7.6 Chromosomal locations of the *chs1*, *chs2*, and *chs3* loci. The *chs1* gene was mapped to chromosome 1 in the region between 5.9 and 7.0 Mb, *chs2* was mapped to

chromosome 4 in the region between 9.02 and 9.95 Mb, and *chs3* was mapped to chromosome 5 in the region between 5.41 and 7.37 Mb

Table 7.1 Effects of chilling (48 h at 10°C) on the transcriptome of chs1, chs2, and chs3 mutants

	Probe sets differentially expressed at $p \le 0.05$ and <i>induced or repressed by a factor of at least 4</i>				
Pair-wise comparison	Up-regulated Down-regulated		Total		
chs1/wild type	913	591	1,504		
chs2/wild type	1,523	1,717	3,240		
chs3/wild type	1,293	1,426	2,719		

Data include numbers of probe sets on the Affymetrix ATH1 Genome Array differentially expressed at $p \le 0.05$ and induced or repressed by a factor of at least 4

plants. Doing so, led to the identification of 1,504, 3,240, and 2,719 probe sets whose expression significantly changed after exposure to chilling in *chs1*, *chs2*, and *chs3* mutants, respectively (Table 7.1).

To define the degree of similarity and differences between gene expression patterns in the various class 1 *chs* mutants in response to chilling, we conducted a Venn diagram comparison (Fig. 7.7). It was found that the expression of a large group of 1,426 probe sets was similarly affected by chilling in all class 1 *chs* mutants, and that of 1,207 additionally genes were similarly affected in both *chs2* and *chs3* mutants (Fig. 7.7). Thus, *chs1*, *chs2*, and *chs3* mutants endure similar molecular responses following exposure to chilling stress.

In order to assign the chilling-induced differentially expressed genes in *chs1*, *chs2*, and *chs3* mutants into corresponding molecular functions, we performed functional categorization analysis using the MapMan software (Thimm et al. 2004). Doing so revealed that the main functional categories up-regulated by chilling in the mutants were "stress," "protein," and "signaling," whereas the main categories down-regulated by chilling



Fig. 7.7 Venn diagram illustrating the overlapping and differences in gene expression patterns among the various *chs1*, *chs2*, and *chs3* chilling-responsive regulons. The numbers on the diagram indicate the amount of overlapped probe sets

were "photosynthesis" and "tetrapyrrole synthesis," "major carbohydrate metabolism," "cell wall," and "lipid metabolism" (Table 7.2). A more or less similar response consisting differential expression of stress, photosynthesis, and protein, carbohydrate and lipid metabolism genes, following exposure to chilling were reported also in several other chilling-sensitive commodities, such as rice, sunflower and potato as well as chilling-tolerant poplar trees (Yan et al. 2006; Fernandez et al. 2008; Oufir et al. 2008; Maestrini et al. 2009). The up-regulated category of "protein" included massive up-regulation of genes involved in protein degradation, and particularly transcripts encoding members of RING finger and F-BOX proteins belonging to the ubiquitin protein degradation pathway. In addition, the "protein" category further included massive down-regulation of genes involved in protein synthesis. The up-regulated category of "signaling" included mainly up-regulation of receptor kinase genes and calcium signaling genes, as previously reported (Bhattacharjee 2009). The observed down-regulation in "lipid metabolism"

genes included mainly suppression of transcripts involved in fatty acid synthesis and desaturation; the latter is known to be a crucial factor required for adaptation to chilling (Murata et al. 1992; Nishida and Murata 1996).

The overall meanings of these findings are that under chilling conditions class 1 *chs* mutants are defective in normal gene expression related to the photosynthesis machinery and carbohydrate metabolism, and cell wall and lipid metabolism. In addition, the *chs* mutants were in severe stress as indicated by massive up-regulation of stress genes, and suffered from imbalanced protein metabolism (suppression of protein synthesis and induction of protein degradation).

7 Conclusion and Future Perspective

From the current evaluations of the physiological, biochemical and molecular responses of class 1 *chs* mutants to chilling, we have arrived at the following main conclusions:

- Under chilling conditions, class 1 *chs* mutants are defective in normal gene expression related to photosynthesis, chlorophyll synthesis and major carbohydrate metabolism (Table 7.2). Furthermore, these molecular observations were confirmed by biochemical measurements indicating a rapid loss of chlorophyll content and decrease in photosynthetic efficacy (Fig. 7.3), leaf yellowing (Fig. 7.1), and inability to accumulate starch as observed in WT plants (Fig. 7.4).
- (2) Under chilling conditions, class 1 *chs* mutants are defective in normal gene expression related to lipid metabolism including downregulation of fatty acid synthesis and desaturation (Table 7.2). Furthermore, these molecular observations were supported by conductivity measurements indicating a continuous increase in electrolyte leakage rates upon transfer of the *chs* mutants to chilling, thus indicating accumulated damage to cellular membranes (Fig. 7.3).
- (3) Under chilling conditions, class 1 *chs* mutants activated a battery of stress-related genes,

	chs1 regu	ılon	chs2 regu	ılon	chs3 regi	ulon
Functional categorization	Up	Down	Up	Down	Up	Down
Photosynthesis	1	16	2	48	2	38
Major CHO metabolism	2	12	4	26	2	21
Minor CHO metabolism	9	3	15	16	12	12
Glycolysis	2	3	4	5	4	3
Fermentation	0	0	2	1	1	0
Gluconeogenesis	1	0	2	0	1	0
OPP	0	1	2	3	2	2
TCA/transformation	0	0	0	5	0	5
Electron transport/ATP	4	2	5	3	4	3
Cell wall	15	35	27	83	22	77
Lipid metabolism	9	31	24	57	18	49
N-metabolism	1	0	3	3	2	3
Amino acid metabolism	10	9	31	36	19	24
S-assimilation	0	3	0	4	0	4
Metal handling	4	3	6	8	5	7
Secondary metabolism	19	17	33	46	39	27
Hormone metabolism	27	25	47	46	41	42
Tetrapyrrole synthesis	0	11	0	21	0	17
Stress	73	15	99	47	81	44
Redox regulation	12	7	21	17	19	12
Polyamine metabolism	0	1	1	1	1	1
Nucleotide metabolism	6	2	9	19	7	11
Misc	59	48	91	113	83	101
RNA and transcription	64	48	120	133	105	113
DNA	4	6	9	18	7	13
Protein	115	32	211	140	177	105
Signaling	103	21	141	72	120	66
Cell	18	7	28	44	27	35
"Micro RNA"	0	0	0	0	0	0
Development	11	10	19	23	15	18
Transport	43	32	68	61	54	56
Not assigned	296	223	496	579	433	495

 Table 7.2
 Functional categorization of chs1, chs2, and chs3 regulon genes

Functional categorization was performed according to MapMan software (http://gabi.rzpd.de/projects/MapMan/). The functional groups of "Photosynthesis," "Major CHO metabolism," "Cell wall," "Lipid metabolism," and "Tetrapyrrole synthesis" down-regulated in *chs* mutants are marked by light shading, while the groups of "Stress," "Protein," and "Signaling" up-regulated in *chs* mutants are marked by dark shading

indicating that the plants were under severe stress conditions (Table 7.2). This observation was further supported by our findings regarding enhanced accumulation of ROS, as observed by DAB staining for detection of H_2O_2 levels (Fig. 7.5).

(4) Following exposure to chilling, class 1 *chs* mutants suffered from imbalanced protein metabolism, demonstrated by suppression of transcripts involved in protein synthesis and massive induction of transcripts belonging to the ubiquitin protein degradation pathway. These processes obviously lead to progressive destruction of normal cellular activity.

Overall, based on our studies with *Arabidopsis* chilling-sensitive mutants, we conclude that several biochemical and molecular traits are apparently crucial for plant survival under chilling

temperatures; these include maintenance of photosynthetic activity and carbohydrate metabolism, maintenance of normal lipid metabolism, maintenance of stress tolerance and capability to detoxify accumulation of ROS, and maintenance of normal cellular function including normal protein turnover and cellular signaling processes.

Acknowledgements This chapter is a contribution from the Agricultural Research Organization, the Volcani Center, Bet Dagan, Israel, no. 597/10.

References

- Allen DJ, Ort DR (2001) Impacts of chilling temperatures on photosynthesis in warm climate plants. Trends Plant Sci 6:36–42
- Araki N, Kusumi K, Masamoto K, Niwa Y, Iba K (2000) Temperature-sensitive Arabidopsis mutant defective in 1-deoxy-D-xylulose 5-phosphate synthase within the plastid non-mevalonate pathway of isoprenoid biosynthesis. Physiol Plant 108:19–24
- Bhattacharjee S (2009) Involvement of calcium and calmodulin in oxidative and temperature stress of *Amaranthus lividus* L. during early germination. J Environ Biol 30:557–562
- Dong CH, Zolman BK, Bartel B, Lee BH, Stevenson B, Agarwal M, Zhu JK (2009) Disruption of *Arabidopsis CHY1* reveals an important role of metabolic status in plant cold stress signaling. Mol Plant 2:59–72
- Fernandez P, Rienzo D, Fernandez L, Hopp HE, Paniego N, Heinz RA (2008) Transcriptomic identification of candidate genes involved in sunflower responses to chilling and salt stresses based on cDNA microarray analysis. BMC Plant Biol 8:11
- Graham D, Patterson BD (1982) Responses of plants to low nonfreezing temperatures: proteins, metabolism, and acclimation. Annu Rev Plant Physiol 33:47–72
- Hasdai M, Weiss B, Levi A, Samach A, Porat R (2006) Differential responses of *Arabidopsis* ecotypes to cold, chilling and freezing temperatures. Ann Appl Biol 148:113–120
- Hugly S, Somerville C (1992) A role for membrane lipid polyunsaturation in chloroplast biogenesis at low temperature. Plant Physiol 99:197–202
- Hugly S, McCourt P, Browse J, Patterson GW, Somersville C (1990) A chilling sensitive mutant of *Arabidopsis* with altered steryl-ester metabolism. Plant Physiol 93:1053–1062
- Ismail AM, Hall AE, Close TJ (1999) Allelic variation of a dehydrin gene cosegregates with chilling tolerance during seedling emergence. Proc Natl Acad Sci USA 96:13566–13570
- Kerdnaimongkol K, Woodson WR (1999) Inhibition of catalase by antisense RNA increases susceptibility to

oxidative stress and chilling injury in tomato plants. J Amer Soc Hort Sci 124:330–336

- Kim HU, Vijayan P, Carlsson AS, Barkan L, Browse J (2010) A Mutation in the *LPAT1* gene suppresses the sensitivity of *fab1* plants to low temperature. Plant Physiol 153:1135–1143
- Levitt J (1980) Responses of Plants to Environmental Stresses. Academic, New York, NY
- Lightner J, Wu J, Browse J (1994) A mutant of *Arabidopsis* with increased levels of stearic acid. Plant Physiol 106:1443–1451
- Lynch DV (1990) Chilling injury in plants: The relevance of membrane lipids. In: KattermanF(ed) Environmental Injury to Plants. Academic, San Diego, CA, pp 17–34
- Lyons JM (1973) Chilling injury in plants. Annu Rev Plant Physiol 24:445–466
- Maestrini P, Cavallini A, Rizzo M, Giordoni T, Bernardi R, Durante M, Natalim L (2009) Isolation and expression of low temperature-induced genes in white poplar (*Poplus* alba). J Plant Physiol 166:1544–1556
- Markhart AH (1986) Chilling injury: a review of possible causes. HortScience 21:1329–1333
- Maruyama S, Yatomi M, Nakamura Y (1990) Response of rice leaves to low temperature. I. Changes in basic biochemical parameters. Plant Cell Physiol 31:303–309
- Miquel M, James D, Dooner H, Browse J (1993) *Arabidopsis* requires polyunsaturated lipids for low-temperature survival. Proc Natl Acad Sci USA 90:6208–6212
- Murata N, Ishizaki-Nishizawa O, Higashi S, Hayashi H, Tasaka Y, Nishida I (1992) Genetically engineered alteration in the chilling sensitivity of plants. Nature 356:710–713
- Nishida I, Murata N (1996) Chilling sensitivity in plants and cyanobacteria: the crucial roles of membrane lipids. Annu Rev Plant Physiol Plant Mol Biol 47:541–568
- Oufir M, Legay S, Nicot N, Van Moer K, Hoffmann L, Renaut J, Hausman JF, Evers D (2008) Gene expression in potato during cold exposure: changes in carbohydrate and polyamine metabolisms. Plant Sci 175:839–852
- Patterson GW, Hugly S, Harrison D (1993) Sterols and phytyl esters of *Arabidopsis thaliana* under normal and chilling temperatures. Phytochemistry 33:1381–1383
- Paull RE (1990) Chilling injury of crops of tropical and subtropical origin. In: Wang CY (ed) Chilling Injury of Horticultural Crops. CRC Press, Boca Raton, FL, pp 17–36
- Payton P, Webb R, Kornyeyev D, Allen R, Holiday S (2001) Protecting cotton photosynthesis during moderate chilling at high light intensity by increasing chloroplastic antioxidant enzyme activity. J Exp Bot 52:2345–2354
- Porat R, Guy CL (2007) Arabidopsis as a model system to study chilling tolerance mechanisms in plants. Plant Stress 1:85–92
- Provart NJ, Gil P, Chen W, Han B, Chang HS, Wang X, Zhu T (2003) Gene expression phenotypes of *Arabidopsis* associated with sensitivity to low temperatures. Plant Physiol 132:893–906

- Routaboul JM, Fischer SF, Browse J (2000) Trienoic Fatty Acids Are Required to Maintain Chloroplast Function at Low Temperatures. Plant Physiol 124:1697–1705
- Sabehat A, Lurie S, Weiss D (1998) Expression of small heat shock proteins at low temperature: A possible role in protecting against chilling injuries. Plant Physiol 117:651–658
- Schneider JC, Hugly S, Somerville CR (1995a) Chillingsensitive mutants of Arabidopsis. Plant Mol Biol Rep 13:11–17
- Schneider JC, Nielsen E, Somerville CR (1995b) A chillingsensitive mutant of *Arabidopsis* is deficient in chloroplast protein accumulation at low temperature. Plant Cell Environ 18:23–32
- Thimm O, Bläsing O, Gibon Y, Nagel A, Mayer S, Krüger P, Selbig J, Müller LA, Rhee SY, Stitt M (2004) MAPMAN: a user-driven tool to display genomics data sets onto diagrams of metabolic pathways and other biological processes. Plant J 37:914–939
- Tokuhisa J, Browse J (1999) Genetic engineering of plant chilling tolerance. Genet Eng 21:79–93
- Tokuhisa JG, Feldmann KA, LaBrie ST, Browse J (1997) Mutational analysis of chilling tolerance in plants. Plant Cell Environ 20:1391–1400
- Tokuhisa JG, Vijayan P, Feldmann KA, Browse J (1998) Chloroplast development at low temperatures requires a homolog of *DIM1*, a yeast gene encoding the 18 S rRNA dimethylase. Plant Cell 10:699–711
- Van Breusegem F, Slooten L, Stassart J, Botterman J, Moens T, Van Montagu M, Inzé D (1999) Effects of overproduction of tobacco MnSOD in maize chloroplasts on foliar tolerance to cold and oxidative stress. J Exp Bot 50:71–78

- Vlachonasios KE, Thomashow MF, Triezenberg SJ (2003) Disruption mutations of ADA2b and GCN5 transcriptional adaptor genes dramatically affect Arabidopsis growth, development, and gene expression. Plant Cell 15:626–38
- Wallis JG, Browse J (2002) Mutants of Arabidopsis reveal many roles for membrane lipids. Prog Lipid Res 41:254–278
- Wang CY (1990) Chilling Injury of Horticultural Crops. CRC Press, Baca-Raton, FL
- Wu J, Lightner J, Warwick N, Browse J (1997) Lowtemperature damage and subsequent recovery of fab1 mutant *Arabidopsis* exposed to 2 degrees C. Plant Physiol 113:347–356
- Xin Z, Browse J (1998) Eskimo1 mutants of Arabidopsis are constitutively freezing-tolerant. Proc Natl Acad Sci USA 95:7799–7804
- Yan SP, Zhang QY, Tang ZC, Su WA, Sun WN (2006) Comparative proteomic analysis provides new insights into chilling stress responses in rice. Mol Cell Proteom 5:484–496
- Yang H, Shi Y, Liu J, Guo L, Zhang X, Yang S (2010) A mutant CHS3 protein with TIR-NB-LRR-LIM domains modulates growth, cell death and freezing tolerance in a temperature-dependent manner in *Arabidopsis*. Plant J 63:283–296
- Zhu T, Provart NJ (2003) Transcriptional responses to low temperature and their regulation in *Arabidopsis*. Can J Bot 81:1168–1174
- Zhu J, Dong CH, Zhu JK (2007) Interplay between coldresponsive gene regulation, metabolism and RNA processing during plant cold acclimation. Curr Opin Plant Biol 10:290–295

Root Form and Function in Plant as an Adaptation to Changing Climate

Maria Rosa Abenavoli, Maria Rosaria Panuccio, and Agostino Sorgonà

Abstract

Climate variables including temperature, atmosphere CO2, and precipitation are expected to change during this century. As consequence, in the short- and long-term, the increase of soil temperature, salinity, drought, and waterlogging stresses could be the more exceeding problems for agricultural productivity and the functioning of the natural ecosystems. Root system represents the first and more sensitive target of the climate change, being seriously damaged in its form and function, and consequently strongly contributes to limit plant growth, development and crop productivity. This review focuses on changes of root morphology, architecture, distribution and dynamics, and on essential root physiological processes, such as water and nutrient uptake in response to soil warming, salinity, drought, and waterlogging. The literature appear sometimes to be controversial due to the complexity of root system characterized by different root types, genetically, developmentally, and functionally distinct, and by diverse root morphological parameters such as total root length, biomass, specific root length (SRL), and tissue density and fineness differently involved on root stress responses. For example, the change on total root length and dry weight, the lateral root formation, the depth of rooting and the root dynamics represent the preferential strategy for plant species in waterlimited environments. Whereas, the development of aerenchyma, tissue containing enlarged gas spaces with a low-resistance pathway to oxygen, often accompanied by the aerotropic and extensive lateral roots formation, the herringbone-type root architecture, the emergence of adventitious roots and the presence of anatomical barriers are expressed in flooded root. Salinity reduces plant growth and yield by two mechanisms, osmotic stress and ion cytotoxicity. It is difficult to separate the osmotic effect

8

A. Sorgonà (🖂) • M.R. Abenavoli • M.R. Panuccio

Dipartimento di Biotecnologie per il Monitoraggio

Agro-Alimentare ed Ambientale, Università Mediterranea

di Reggio Calabria, Contrada Melissari - Lotto D,

⁸⁹¹²⁴ Reggio Calabria, Italy

e-mail: asorgona@unirc.it

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_8,

[©] Springer Science+Business Media, LLC 2012

from specific ion effects that overlap during the development of salinity stress, thereby some uncertainty exists regarding the relative importance of both mechanisms. The responses of root cells are finalized to maintain their own correct functionality, despite the condition of elevated Na+ concentration. The genetic diversity of the root system in the plant response to climate change was also reported.

Behind the root "form" changes, many metabolic and physiological pathways are involved in the plant adaptation to climate change. The maintenance of lower respiration rate, carbohydrate metabolism and cell expansion and elongation, often mediated by hormones, are expressed in roots grown in dry soil. At molecular level, the deposition of proline, metabolite responsible of the total osmotic adjustments, the higher expansin and xyloglucan endotransglycolase/hydrolase (XTH) activities, enzymes of the cell wall extension, represent the physiological mechanisms implemented by plants for improving their drought tolerance. In waterlogging soils and also in presence of salinity, the adaptation of physiological mechanisms are addressed to improve the cellular energy status and reduce the accumulation of toxic end products that acidify the cytosol or damage membrane integrity. Remarks on proteomic and molecular aspects which represent a future approach to individuate the plant strategies for their adaptation to the climate change are also included.

Keywords

Root system • Drought • Precipitation • Salinity • Temperature • Waterlogging

1 Introduction

Climate change variables such as temperature (warming), precipitation (drought and flooding), and atmospheric CO₂ concentrations (CO₂ fertilization) have greater impact on agricultural productivity, ecosystem structure and function. Plant has a strong photohydraulic system involved in water and nutrient uptake, a sink for the photoassimilates in which the root system plays a key role in determining plant responses to climate change. Further, the root system, by its respiration, turnover, exudation processes, and interactions with the soil biota, plays a critical role in controlling the soil C storage and cycling and ultimately in the feedbacks of terrestrial C cycling to climate change. Hence, the root system could be considered as sink and source of carbon dioxide, the main driver of climate change, and a better understanding

of its responses could provide useful information in the plant adaptation to the future atmospheric composition. In the present chapter, we will mainly focus on how direct (temperature, drought, flooding) and indirect (salinity) components of climate changes influence the root form and function in higher plants.

Let us consider the root form as a "photographical description" of the root system at the tri-, bi-, and one-dimensional levels, as determined by biometric parameters. The root form includes the root morphology, architecture, distribution, and dynamics. Root morphology means "superficial features of the whole and single root axis" (Lynch and Nielsen 1996) such as the length, mass, surface area, volume, and diameter. Further morphological parameters derived from the formers and having a functional significance (Ryser 1998) are the root length ratio (RLR) (root length per unit of the plant's dry mass), root mass ratio (RMR) (root mass per unit of the plant's dry mass), specific root length (SRL) (root length per unit of root dry weight), root fineness (RF) (root length per unit root volume) and root tissue density (RTD) (root dry mass per unit root volume). The root architecture is defined as the spatial configuration of the root system (Lynch and Nielsen 1996) and is generally estimated in terms of topology (Robinson et al. 2003). Root topology, which refers to the distribution of the branches within the system, can lie within two extreme types: the "herringbone" type in which branching is confined to the main axis and the "dichotomous" type exhibiting a more random branching (Fitter and Stickland 1991). The root distribution, which refers to the deployment of the root axis in terms of biomass or length along the soil profile is described by root mass density (RMD) (root mass per unit soil volume) and root length density (RLD) (root length per unit soil volume). Finally, the root dynamics includes the root production, mortality, and turnover (ratio of root number present at a time point to the number of roots produced up to that time) and life span.

The root system as defined by Robinson et al. (1991) "... is the result of an evolution strategy to solve the problems of soil resources acquisition...." Hence, the main function of the roots is the capture of belowground resources, such as water and nutrients, from that "...heterogeneous and porous system ..." (Robinson et al. 1991) which is the soil environment. The climate change has impact on the plant C allocation and respiration, water and nutrients uptake which are important physiological processes in terms of C (Hinsinger 1998, 2001; Hinsinger et al. 2003). Finally, we will discuss the impact of climate changes on the root's "secondary functions" such as storage of carbon and nutrients and supply of energy to belowground food web and microorganisms (Hinsinger et al. 2005, 2006).

2 Roots and High Temperature

The Fourth Assessment Report (AR4) of the Inter governmental Panel on Climate Change (IPCC) of United Nations predicted an approximately 1.8–4°C increase in global mean air temperature during this century (IPCC 2007). Soil temperatures are also expected to increase reflecting the future atmospheric temperature trend (Pollack et al. 1998). The plant root system will be affected by soil temperatures which will have a significant impact on its form and function and therefore on plant development and productivity. Lots of study have been done on the effects of temperature on the root system (Cooper 1973; Voorhes et al. 1981; Kaspar and Bland 1992; McMichael and Burke 2002), relatively few information have focused on integrated root development, growth, metabolic responses to soil warming and how roots preserve its form and function under warming soil conditions is not completely clarified.

Considerable evidence indicates that the root growth increased in response to increased soil temperature up to an optimum threshold, typical for each species and depending partly on their native temperature regime (McMichael and Burke 1998), beyond which root growth decreased. Faster elongation rates of whole root system were observed in the temperature range of 5-23°C for Eucalyptus species (Misra 1999), 10-30°C for sunflower (Seiler 1998), 10-15°C for winter wheat (Gavito et al. 2001) and bog and fen plant communities (Weltzin et al. 2000). Supraoptimal soil temperatures, on the other hand, reduced the root growth in many species such as Agrostis stolonifera (>35°C, Huang et al. 1998), Lactuca sativa (>35°C, Qin et al. 2007; He et al. 2009) and wheat (>38°C, Tahir et al. 2008). McMichael and Burke (2002) grouped the different taxa in relation to optimum of temperature for the root growth, pointing out the influence (incidence) of diverse genetic background on temperature-dependent root growth pattern among the plant species due to their different acclimation strategies. For example, a diverse response and adaptation, in terms of both root length and mass, was evident among genetically diverse sunflower (Seiler 1998) wheat genotypes (Tahir et al. 2008) and in two Agrostis species; where the root system of A. scabra was more thermotolerant growing up to 45°C (Terceck et al. 2003) than that of A. stolonifera which grew upto 23°C only (Pote et al. 2006). These observations suggested that genetic diversity in root growth contributes to the survival of plant species and to improve their productivity under high soil temperature conditions and deserves further studies.

To better understand the temperature-induced root responses between and within plant species is needed to consider that the root systems comprise of different root types which are distinct genetically, developmentally, and functionally and differently respond to soil environmental stresses (Waisel and Eshel 2002). Several examples can be mentioned in this regard: the first root axes of pearl millet showed a higher elongation rate in response to the increase of temperature (from 20 to 32°C) with respect to the second one (Gregory 1986), the primary root of sunflower was less inhibited at temperature above 35°C than lateral roots (Seiler 1998), the fine roots were more sensitive to soil warming (Pregitzer et al. 2000), the SRL (root length/root mass) and specific root area (root area/root mass) increased in warmer soil in the root finest fraction (<0.5 mm) only (Bjork et al. 2007). However, more studies are needed for gaining a better knowledge on how the different root types/orders respond to high soil temperatures.

The morphological responses and the acclimation of the root to higher temperatures involve the integration of many metabolic and physiological pathways. It is well-known that the root growth depends on the supply of carbohydrates which are sharply consumed by higher root maintenance respiration in warmer soil conditions. Therefore, the maintenance of lower respiration rate may represent an important basis for the thermotolerance of the root systems and, ultimately, for the plant adaptation to the higher soil temperatures. Indeed, the lower energy required for root maintenance permitted to Agrostis scabra to grow up to 45°C while the root growth of Agrostis stolonifera, heat-sensispecies was inhibited above 27°C tive (Rachmilevitch et al. 2008). Other plant species such as Citrus volkameriana (Bouma et al. 1997), Bellis perennis, and Poa annua (Gunn and Farrar 1999), adapted to warmer soil exhibited a lower maintenance respiration rate of their root systems.

However, several authors pointed out that the temperature-induced root growth patterns also depend on radiation flux which influencing the photosynthesis determines a variation of the carbohydrate supply to root system. Indeed, the tap and lateral root elongation rate of sunflower were weakly correlated with soil temperature and sharply dependent on the amount of radiation intercepted (Aguirrezabal et al. 1994). Further, the root biomass and length of temperate northern grassland species dominated by *Holcus lanatus* were strongly affected by incident radiation and not by soil temperature (Edwards et al. 2004). Consequently, in order to understand root responses to warming soil, it is necessary to separate the effects of photosynthetically active radiation by soil temperature.

Root growth is not only associated with the carbohydrate metabolism but is also correlated with other cellular processes, such as cell expansion and elongation. By analyzing the spatial distribution of expansion growth along the primary root axis of Zea mays, Walter et al. (2002) observed a greater extension accompanied by a maximal expansion activity of the growing zone with the rising temperature (from 21 to 26°C). Furthermore, several lines of evidence suggested that the root morphological changes to high temperatures might be mediated by hormones. Qin et al. (2007) demonstrated that the application of the ethylene precursor (1-aminocyclopropane-1-carboxylic acid, ACC), in lettuce seedlings, mimicked the high temperature-induced root morphological changes (inhibition and increase of the root length and diameter, respectively) and the addition of ethylene biosynthesis inhibitors such as aminooxyacetic acid (AOA) or aminoisobutyric acid (AIB) relieved these effects.

The soil temperatures fluctuated over a wide range of temporal (diurnal and seasonal) and spatial (depth) scales (Kaspar and Bland 1992) determining soil-zone temperatures. This soil gradient affects the orientation of the root axis, mainly nodal and seminal roots which observed a plagiotropic growth (growth at angles from the vertical). For example, the seminal roots of *Zea mays* which at 17°C grew horizontally, above and below this temperature showed a more vertical growth (Onderdonk and Ketcheson 1973). The result of the diverse root orientation was a different root distribution in terms of length and mass within the soil profile. Root mass of a community constituted by *Cardamine hirsuta*, *Poa annua*, *Senecio vulgaris*, and *Spergula arvensis* was reduced at the surface soil layers by elevated temperatures (Kandeler et al. 1998). Similar results were pointed out by Soussana et al. (1996) in the root of perennial ryegrass (*Lolium perenne*).

Different studies have been focused on the influence of the soil temperature on the root dynamics although no consistent patterns have been observed. The root production and mortality, especially of the fine-roots, determining the soil C inputs and soil microbial activity, play a critical role in regulating agro- and ecosystem C balance and, ultimately, in the below-ground CO₂ efflux. An increase of root turnover in response to soil warming were observed in maple tree seedlings (Wan et al. 2004), Norway spruce stand (Majdi and Ohrvik 2004) and Eriophoretum vaginatum (Sullivan and Welker 2005) unlike of temperate steppe perennial species (Bai et al. 2010), Douglas-fir (Johnson et al. 2006), maple (Côté et al. 1998) and oak trees (Joslin et al. 2001) whose root dynamics were not affected by high soil temperatures.

Warming is one of the main factors of climate change along with altered precipitation, elevate atmospheric CO₂ and N deposition and each of these may be expected to vary independently as well as to be interdependent. For example, elevated CO₂ reducing the evapotranspiration increased the soil moisture (Nelson et al. 2004), the increase of the soil temperature stimulated the soil microorganism activity causing a higher N availability for the plant (Rustad et al. 2001) and finally soil warming interacted with the concomitant drought stress also (Kramer and Boyer 1995). In this respect, combinations of different climate change factors are possible and it would be interesting to know the root responses to the interactive effects of these factors. For example, Tierney et al. (2003) reported a strong relationship between root production and soil temperature in a hardwood forest which contrasted with the results of Joslin et al. (2001). Tierney et al. (2003) justified this discrepancy by difference in water availability between their site with that of Joslin et al. (2001) pointing out the water availability and soil temperature interactions in the root growth responses. Bai et al. (2010) revealed that the temperature-induced inhibition on the root production and mortality of temperate perennial steppe species was observed with increased precipitation while the root dynamics was improved under ambient precipitation. Further, the effects of the increase of temperature (from 10 to 15°C) and N supply on the total root length of winter wheat were highly significant and additive (Gavito et al. 2001), Majdi and Ohrvik (2004) observed that the addition of N reduces the risk of root mortality in Norway spruce contrasting the effect of soil warming. The reduction of the root biomass of a mixing plant species (Cardamine hirsuta, Poa annua, Senecio *vulgaris*, and *Spergula arvensis*) at 0–10 cm of soil layers (Kandeler et al. 1998) and the greater induction of the root production and mortality in Acer spp. (Wan et al. 2004) by higher soil temperatures were observed at elevated but not at ambient CO₂. Soil warming influences several root functions such as nutrient and water captures (St.Clair and Lynch 2010) which are fundamental physiological processes of the plant development and productivity and the functioning of the terrestrial ecosystems. In general, high temperatures increased the nutrient acquisition up to a peak of a maximum activity and then decline. Over the range of 14-34°C two cultivars of red maple increased the net nitrate uptake reaching a maximum absorption at 24°C (Adam et al. 2003). Roots of *Eucalyptus nitens* treated with 20°C showed a greater nitrate and ammonium uptake rates than that exposed to 10°C (Garnett and Smethurst 1999). The K (Ching and Barber 1979) and the P uptake (Mackay and Barber 1984) were increased in response to the moderate increase of temperature up to 29°C. However, Gavito et al. (2001) pointed out that the specific absorption rates are temperature dependent for each nutrient, varying from 10 to 15°C. During the vegetative growth, winter wheat reduced both the root and shoot N concentrations leaving unchanged the P concentration. Further, E. nitens showed the different Q_{10} values of the NO₃⁻ and NH₄⁺ uptake rates estimated as 1.88 and 1.31 between 10 and 20°C, respectively (Garnett and Smethurst 1999).

Moreover, nutrient uptake appeared to be controlled by warmer temperatures through a direct and an indirect effect, that is, via root system changes and/or via plant-soil interactions, respectively. Direct changes in root morphology could explain the temperature-induced effects on K uptake (Ching and Barber 1979) and P uptake (Mackay and Barber 1984), as well as changes of the fluidity of fatty acids and thermostability of plasma membrane (Clarkson et al. 1988; Sibley et al. 1999), the uptake kinetics (BassiriRad et al. 1993, 1996; BassiriRad 2000; Adam et al. 2003) the root cell energy by respiration process (Atkin et al. 2000), plant nutrient demand (Lainè et al. 1993; Gavito et al. 2001). Thus, suggesting that temperature directly effects on root physiology of nutrient uptake. On the other hand, the temperature-induced indirect effects on nutrient uptake were mainly correlated with the capacity of the soil warming to influence the nutrient availability through changes on the biogeochemical, nodulation, and mycorhizzation processes and/or nutrient transport, at the rhizosphere level. Mineral weathering, decomposition of organic matter and exchange reactions between soil solid-solution phases, the biogeochemical processes involved in the nutrient availability, occurred at accelerated rates in warmer soils (Pregitzer and King 2005). For example, an increase of N mineralization was observed in a litter and a sandy mineral soil from forest of *Pinus sylvestris* (Ross et al. 1999) and in a boreal Norway spruce stand (Stromgren and Linder 2002). The nutrient movement toward the root system was improved at moderate supraoptimal soil temperatures which increased the ion diffusion and transpiration-driven mass flow. Therefore, the nutrient uptake was enhanced by the higher nutrient concentration around the root axis in warming soil. In this respect, Ching and Barber (1979) observed that temperature changes (from 15 to 29°C) determined a sharp improvement of the K uptake due to increase of the diffusion flux (+160%) rather than the enlargement of the root surface area (+70%).

An important agricultural and ecological function of the root system is the biological nitrogen fixation of the legume species by bacterial infection considering that half of the 320 Tg of N input to terrestrial ecosystems annually comes from the biological fixation of N₂ (Paul and Clark 1996). Symbiotic nitrogen fixation was positively influenced by increased soil temperature which raised nodule mass and activity. However, although Rhizobium can tolerate high soil temperature $(30-35^{\circ}C)$, the extreme temperatures (>40°C) strongly affected the bacterial infection and N₂ fixation, at different degree among the host plant species (Zaharan 1999). The total N_2 fixation of Trifolium repens was enhanced by an increase of the temperature in the 7–13°C range while it was not influenced by temperatures above 13°C (MacDuff and Dhanoa 1990). Warmer soil also affected other main symbiotic relationship of the higher plants, that is, the plant host-mycorrhizal fungi interactions. Increasing the soil temperature, root length colonization (LRC) was improved (Heinemeyer and Fitter 2004) and an increase in development of arbuscular mycorrhizal hyphae was observed which consequently helped in higher P uptake by roots of pea plants (Gavito et al. 2003). However, Olsrud et al. (2004) pointed out that the positive relationship between the mycorrhizal development and warmer soil was an indirect effect due to an increased C allocation toward the roots in response of the concomitant low soil moisture content rather than a direct temperature effect on the root system (Olsrud et al. 2004).

3 Roots and Altered Precipitation

The climate change models of the Intergovernmental Panel on Climate Change predicted that, as a consequence of temperature increases, the precipitation pattern will vary (IPCC 2007). At high latitudes, it should increase in winter and decrease in summer, and in various regions of Central and South Europe will also increase the frequency and duration of summer precipitation. Thus, the climate change will be responsible for determining an increased risk of both soil drought and waterlogging.

Generally, water is believed to be available for plant uptake at soil water potential greater than wilting point, -1.6 MPa, which is considered the limiting value for the growth and development of many mesophytic plants. Suboptimal water availability, that is, drought soil condition is in fact considered a major constraint limiting the crop productivity. Soil drought more severely affect shoots than roots (Spollen et al. 1993) which have the ability to grow under mild stress condition. This differential sensitivity represents an important advantage to plants and allows them a greater exploration of soil for ensuring a supply of water and increasing their probability to survive at dry condition.

Drying soils produce a range of effects on the plant root system and varies from species to species. The total root length and dry weight, the lateral root production, the depth of rooting and the root dynamics represent the major traits influenced by water deficit. Several evidences indicated that the primary root length of maize was inhibited at -0.5 MPa water potential after 24 h (Fan and Neumann 2004). The reduction in the primary root elongation of Arabidopsis thaliana was observed at moderate water stress (-0.2 MPa)(van der Weele et al. 2000), similar results were observed in Pinus pinaster whose root system was increased at -0.15 MPa and reduced at -0.66 MPa) (Triboulot et al. 1995). In White oak (less sensitive) whose root elongation rate was reduced in the range of -0.4 and -0.8 MPa and completely ceased at -1.2 MPa soil water potential (Kuhns et al. 1985). The rapid soil drying conditions determined an intense reduction of the elongation rate of the Opuntia ficus-indica roots after 3 days of water deficit while the gradual drought stress caused the root inhibition after 9 days of water stress (Dubrovsky et al. 1998). Additionally, as response to soil drying, lateral roots varied in length in a specie-specific manner. The length of the first-order lateral roots was more reduced in maize than in wheat making this species less sensitive to water deficit (Ito et al. 2006).

What physiological mechanisms are involved for maintaining the root elongation at low water potential? Over the last 20 years, Sharp and coworkers (Sharp et al. 2004; Ober and Sharp 2007) pointed out detailed results on the regulation of maize primary root growth under soil drought conditions. In particular, they observed that elongation rate of the apical region (<3 mm, region 1) of the water-stressed roots (-1.6 MPa)was maintained same as well-watered roots but was reduced and completely inhibited in the basal root regions (3-7 mm region 2; and >7 mm, region 3, respectively). The mechanisms involved in the maintenance of the elongation rate in the region 1 may be due to the following reasons: (1) accumulation of the plant hormone abscissic acid (ABA), (2) the osmotic adjustments, and (3) the cell wall extension properties. Saab et al. (1990) and Sharp et al. (1994) showed that endogenous ABA accumulated in the apical region of waterstressed maize roots causing prevention of excess ethylene production and playing a regulatory role in the ion homeostasis (Ober and Sharp 2003). Accumulation of proline was responsible for as much as 45% of the total osmotic adjustments in the apical root region of maize seedlings under water stress (Voetberg and Sharp 1991). Further, Wu et al. (1996) reported differential responses of cell wall extension properties in maize seedlings. The expansion and xyloglucan endotransglycolase/hydrolase (XTH) activities were higher in the apical region of water-stressed than wellwatered roots. It is interesting to note that ABA accumulation was involved in both processes increasing the proline transport and the XTH activity in the apical region of water-stressed roots underlying its regulative role in drought stress (Ober and Sharp 1994; Wu et al. 2001). In addition, Fan and Neumann (2004) found that the water stress induced an acidification in the apical region of maize roots which maintained the growth in presence of drought stress. Consistent with these results, transcriptomic and proteomic analyses of root growth of maize seedlings in responses to soil drought revealed that the water stress induced gene expression and the cell wall protein abundance were largely region specific. The apical region exhibited a greater gene and protein numbers involved for the root adaptation to drought (Yamaguchi and Sharp 2010).

Under water stress conditions, beyond the root length, other root morphological parameters were modified such as root diameter or SRL. Water-stressed roots became thinner than that of well-watered plant such as, A. thaliana (van der Weele et al. 2000), rice (Trillana et al. 2001), maize ((Liang et al. 1997; Hund et al. 2009) and rootstocks of Vitis vinifera (Baurle et al. 2008). Root thinning as adaptive traits in dry environments could imply: (1) a lower construction cost for unit of root length (Sharp et al. 1988; Ho et al. 2005), (2) a faster elongation rate of root axis (Thaler and Pages 1996; van der Weele et al. 2000), (3) a greater root proliferation in moisturedisturbed soils (Eissenstat 1991), and (4) a limitation on radial expansion to conserve water (van der Weele et al. 2000). However, since the thicker roots were positively correlated with an increase of water transport (Passioura 1988; Doussan et al. 1998) and a greater RLD in deep soil layers (Azhiri-Sigari et al. 2000; Kato et al. 2006). The reduction of the root diameter as plant adaptive traits in drying soils was still questioned.

Generally, the soil dried gradually from the upper layers producing a sharply vertical soilmoisture gradient. Padilla and Pugnaire (2007) observed that in a semi-natural field during the summer, the soil moisture was <8% and >20% at 5 and below 30 cm of soil depths, respectively, and this difference disappeared in the spring season. This vertical soil-moisture gradient in the rainfed or non-irrigated conditions was also observed in field experiments by Songsri et al. (2008), Bucci et al. (2009), and Cheng et al. (2009). As the soil environment became drier, root system changed its distribution becoming deeper since deep-rooted plants had an improved ability to extract water from well-watered deep soil layers compared to shallow-rooted. This shift on root distribution in response to water stress has been confirmed by Schenk and Jackson (2005) who collected data of >1,300 records of root distribution of individual plants from deserts, scrublands, grassland, and savannas along the soil depths, showed that the "absolute" rooting depth was more strongly correlated with the mean annual precipitation in all plant growth except shrubs and trees. However, Bucci et al. (2009) observed that shallow-rooted shrub species of Patagonia steppe were correlated with leaf negative water potential than deeply rooted ones. Padilla and Pugnaire (2007) observed that species characterized by deep roots (Salsola oppositifolia and *Retama sphaerocarpa*) had consistent access to deep water content and pointed out a successful survival compared with shallow-rooted plants (Ephedra fragilis, Olea europaea, and Pinus halepensis) which died as drought progressed. Further, root depth was found to be the preferential strategy for plant species in dry ecosystems (Canadell et al. 1996), long dry periods (Paz 2003), dry sandy soils (Yamada et al. 2005), and seedling of dry forest (Markesteijn and Poorter 2009). The root depth as drought-adaptive trait was exploited in crop breeding programs for improving the yield in water-limited environments. Two drought-tolerant maize hybrids exhibited around three times more axile roots in the deeper soil layers compared drought-sensitive ones (Wan et al. 2000). Drought-tolerant genotypes of sorghum were characterized by deeper roots (Ludlow et al. 1990) and the ability to produce roots in deeper soil layers could markedly improve the drought tolerance of wheat cultivars (Manschadi et al. 2006). Change in root distribution under water stress was reported among genotypes in cowpea (Matsui and Singh 2003), white clover (Annicchiarico and Piano 2004), and chickpea (Kashiwagi et al. 2006). Beside the deep rooting, the higher RLD at lower soil depths has been identified as a drought adaptative trait that permits to stabilize the pod yield and the harvest index in drought-avoiding peanut genotypes (Songsri et al. 2008) and the ability to produce roots in deeper soil layers could markedly improve the drought tolerance of wheat cultivars (Manschadi et al. 2006). However, the direct relationship between the deeply rooted system and increased water absorption from deeper soil layers has not been clearly demonstrated and it was generally based on indirect evidences with the above-ground biomass and/or yield parameters. This consideration pointed out the following question: did root form changes root functional modification? Hund et al. (2009) demonstrated that the root system of CML444, drought-tolerant maize germoplasm, compared with that of SC-Malawy, moderate drought-tolerant, exhibited a deeper roots accompanied by higher ability to absorb water from deep layers. Further, the deep-rooting is an important root architecture strategy for efficient water uptake in dry environments. Indeed, Manschadi et al. (2008) observed that the narrower angle of the seminal roots causing a deeply rooted architecture was a trait to exploit in breeding for improved wheat cultivars for water-stressed environments. Further, the investment in terms of carbon allocation toward specific root types, such as tap root in bean, determining a deeply rooted architecture improved the water acquisition efficiency (Ho et al. 2005).

In the agro-ecosystems, plants usually faced with several different environmental stresses acting in combination or in sequence and then it must optimize their resource allocation for the construction and maintenance of root systems. For example, the reduction of the soil water status caused the copresence of both water stress and mechanical impedance to root growth (Whitmore and Whalley 2009). Ho et al. (2005) demonstrated that the "dimorphic" root system exhibited the best performance in environments characterized by multiple stresses. The bean genotype BAT477, exhibiting root architecture with both shallow and deep root localization, was well adapted in the presence of suboptimal water and phosphorus availability.

Besides the root morphological and architectural traits, the regulation of root water flow represented a physiological adaptive mechanism to soil water limitation. Root hydraulic conductance (Kr) was generally reduced when soil dried (North and Nobel 1991, 1992; Nardini et al. 2002). However, it has been observed that, when seedlings were exposed to moderate water stress, drought-sensitive plants increased the Kr (Nardini et al. 1998) and drought-tolerant plants decreased it (Lo Gullo et al. 1998). The capacity and the time-course necessary to recover Kr was a key factor of plant adaptation to seasonality in water availability under water stress, especially in Mediterranean species. The timing of recovery seemed to be correlated with the recovery of growth of root tip pre-existing and the new lateral root formation (Lo Gullo et al. 1998; Dubrovsky et al. 1998). Other than apoplastic pathway, the root hydraulic conductance was also related with water movement occurred along the cell-to-cell path by the aquaporins, water channel proteins (Javot and Maurel 2002; Tyerman et al. 2002; Maurel 2007). Several studies suggested that aquaporin-mediated transport was important in the regulation of root water flow under drought stress (Siemens and Zwiazek 2004), although their actual function was still unclear. Indeed roots of sunflower exposed to water limitation exhibited an up- and downregulation of different aquaporin genes (Sarda et al. 1999) and transgenic plants of *Arabidopsis* and tobacco, that constitutively over-expressed PIP1b, PIP1;4 or PIP2;5, had adverse effects on plant growth under drought stress (Aharon et al. 2003; Jang et al. 2007).

The drought stress altered the root functionality such as the nutrient uptake. Substantially, drought reduced the uptake of phosphorous (P) in barley (Shone and Flood 1983) and rye grass (Jupp and Newman 1987) and nitrogen (N) in maize (Buljovicic and Engels 2001), Pseudoregneria spicata (BassiriRad and Caldwell 1992a), and Artemisia tridentata (BassiriRad and Caldwell 1992b). Conversely, N uptake was not affected by drought stress in Agropyron desertorum (BassiriRad and Caldwell 1992a) while P uptake was increased in Artemisia tridentata (Matzner and Richards 1996). These controversial results could be due to the wide influence of drought stress on all other factors/components involved in actual rate of nutrient uptake from soil, such as (1) soil processes that provide the nutrient availability at root surface; (2) anatomical (endo- and exodermis), morphological (length and surface area), and architectural (shallow and deep rooting) root characteristics; and (3) physiological (nutrient transporters) root characteristics.

4 Root and Excess Water

Excess water in soil determines the full filling of the pore space restricting the diffusion of oxygen by 10⁴-fold than in air (Drew and Armstrong 2002) and causing a stress, named waterlogging, with dramatic impact on plant growth and productivity. Climate change provisions provide that, as a consequence of anomalous weather patterns, waterlogging could be a more exceeding problem for many plant communities. Under oxygen absence (anoxic soil) or under severe hypoxic conditions, the cytochrome oxidase activity of the plants became oxygen limited with a consequent reduction of ATP and pH of the cytoplasm, carbohydrate starvation, and accumulation of the toxic products due to a switch to a fermentative pathway (Drew 1997; Geigenberger 2003; Bailey-Serres and Chang 2005). Within short-time (few minutes or hours), these toxic effects caused severe damages to the plant growth and ultimately leads to death of many plant species. However, the supply of small amount of oxygen (hypoxic soil) stimulated several acclimative mechanisms which allowed the plants to survive to the transient waterlogging.

In waterlogging soils, root system represented the first and more sensitive target of plants which could be seriously damaged in its form and function. Significant inhibition of the root growth, exposed to waterlogging stress were observed in Arabidopsis (van Dongen et al. 2009), Trifolium glomeratum (Gibberd et al. 1999), wheat (Malik et al. 2002), maize (Wei and Li 2000; Qiu et al. 2007), woody species (Poot and Lambers 2003; Nicoll and Ray 1996; Nicoll and Coutts 1998; Coutts and Philipson 1978). At the same time, the root system was also able to engender several adaptative responses to waterlogging that could enhance the plant survival in flooded soils. The root adaptation mechanisms were addressed to improve the cellular energy status and reduce the accumulation of toxic end products that acidify the cytosol or damage membrane integrity.

To provide sufficient oxygen for maintaining the root respiration and, consequently, the ATP production it was essential to improve the root energy status and, ultimately, the root growth in chemically anaerobic and reduced soils. Aerenchyma, a plant tissue containing enlarged gas spaces, is an important trait for the root growth and function which provides a low-resistance pathway to obtain the oxygen from the atmosphere to the flooded below-ground organ. The development of aerenchyma has been associated with the tolerance to waterlogging in many plant species (Colmer et al. 1998). Evans (2003) distinguished two types of aerenchyma: (1) the schizogeneous derived from a differential cell expansion and

specific pattern of cell separation with subsequent creation of cell spaces and (2) the lysigenous produced by the death and dissolution of the root cortical cells. While the former was common in various wetland species as Rumex (Laan et al. 1989), the lysigenous was typical of many crop species as soybean (Thomas et al. 2005), rice (Kaway et al. 1998), maize (Drew et al. 1979; Gunawardena et al. 2001), wheat (Huang et al. 1994), and pasture species (Gibberd et al. 2001; Ashi-Smiti et al. 2003). However, there was a third type of aerenchyma defined secondary aerenchyma, a white spongy tissue filled with gas spaces, which was found in stem, hypocotyls, tap roots, and root nodules of Glycine max (Shimamura et al. 2003), Lotus uliginosus (James and Sprent 1999), and Sesbania rostrata (Shiba and Daimon 2003). Generally, the aerenchyma was constitutively expressed in rice and wetland species. Recently, Seago et al. (2005) described the pattern of aerenchyma formation in 85 species representing 41 families of wetland plants. On the other hand, the aerenchyma development was also induced by flooding and other stresses, e.g., nutritional and drought in many field crops. For example, soybean, very sensitive species to flooding stress during the vegetative stage developed a secondary aerenchyma in stems, roots, and root nodules within few weeks of stress (Shimamura et al. 2003). Complex physiological and molecular mechanisms were involved in the development of aerenchyma in plants subjected to the flooding stress (Colmer 2003a; Evans 2003). In maize roots, the hypoxia conditions (3-12% oxygen) promoted the ethylene biosynthesis which triggered a signal transduction cascade involving Ca²⁺ and protein kinases, inducing a programmed cell death in target cells of the root cortex (Drew et al. 2000).

The genetic variability of the plant species in the differing tolerance to waterlogging was associated with the aerenchyma formation which determined higher root porosity. Indeed, the superior tolerance to waterlogging of *Trifolium tomentosum*, *T. fragiferum* and *T. repense* than *T. subterraneum* var. *subterraneum* and *T. glomeratum*, more sensitive species were due to the development of aerenchyma in adventitious roots (Gibberd et al. 1999). The different ability to form aerenchyma and, hence, to exhibit a greater tolerance to the waterlogging stress were observed among the genotypes of soybean (Bacanamwo and Purcell 1999), maize (Zaidi et al. 2004), wheat (Boru et al. 2003), and woody species (Aguilar et al. 1999). Further, the wetland *Rumex* species tolerant produced a higher root porosity than sensitive ones (Laan et al. 1989).

Beside the aerenchyma, other root mechanisms which improved the presence of oxygen in plant tissues subjected to the flooding stress, such as the aerotropic roots and extensive lateral roots (Gibberd et al. 2001), the herringbone-type root architecture (Bouma et al. 2001), the emergence of adventitious roots (Mergemann and Sauter 2000) and the anatomical barriers such as exoderism (Colmer 2003b) were suggested to be part of the basis of tolerance to waterlogging. The recovery of the root cellular energy status in oxygen deficiency conditions was obtained through different metabolic adaptative mechanisms: the switch to a fermentative pathway (Sachs et al. 1996; Chang et al. 2000), the global depression of ATPconsuming processes (van Dongen et al. 2009), the death of metabolically intensive tissues as root tip (Subbaiah and Sachs 2000; Subbaiah and Sachs 2001) and the induction of anaerobic protein synthesis (Lal et al. 1998; Mujer et al. 1993; Chang et al. 2000). During waterlogging, the roots were more prone to oxidative stress which caused the generation of "reactive oxygen species" (ROS), including superoxide anion radicals (O₂⁻), hydroxyl radicals (OH), hydrogen peroxide (H₂O₂), alkoxy radicals (RO), and singlet oxygen (O1/2) (Munné-Bosch and Peñuelas 2003). The production of ROS led to enhanced peroxidation of membrane lipids and degradation of nucleic acids, and both structural and functional proteins. In this respect, the induction of free radical scavenging enzymes observed in wheat roots subjected to hypoxia determining tolerance to the waterlogging stress (Biemelt et al. 1998).

Proteomic and genomic analysis supported the morphological and physiological mechanisms that made up the root acclimative responses to the waterlogging stress. Indeed, the roots stressed by water shortage exhibited a large-scale reprogramming of gene expression and metabolism: (1) up- and downregulation of genes in *Arabidopsis* that mainly encoded proteins involved in fermentative process and energy-consuming processes (transport, lipid and secondary metabolism), respectively (van Dongen et al. 2009), (2) a quantitative trait locus containing the ethylene response factor-like genes was regulated by submergence in rice (Fukao et al. 2006), and (3) several quantitative trait loci associated with waterlogging tolerance were detected in maize plants (Qiu et al. 2007).

5 Root and Salinity

Salinity reduces plant growth and yield by two mechanisms, osmotic stress and ion cytotoxicity (Munns and Tester 2008). Munns 2002 proposed a two-phase model of salt injury where growth is initially reduced by osmotic stress and then by Na⁺ toxicity. According to this biphasic model, growth is first reduced by the decrease in soil osmotic potential (ψ_{α}), caused by salt outside the plant rather than within it. The induced osmotic stress is controlled by inhibitory signals from the roots and, genotypes differing in salt resistance, respond identically in this first phase. Ionic stress develops over time and is due to a combination of ion accumulation in the shoot and to an inability to tolerate ions that have accumulated. The adaptation responses to ionic stress by plants are of two distinct types: Na⁺ exclusion and tissue tolerance, and is the result of different abilities by plants to exclude or to sequester toxic ions into vacuoles. In this second phase of growth reduction, genotypes varying in salt resistance may respond differently. Secondary stresses induced by salinity, such as nutritional imbalances and oxidative stress, are also responsible of reduced plant growth.

It is difficult to separate the osmotic effect from specific ion effects that overlap during the development of salinity stress, thereby some uncertainty exists regarding the relative importance of both mechanisms. Rengasamy (2010) conducted a pot experiment on wheat growth where the plants were exposed to NaCl or to a Hoagland nutrient solution at different salinity levels. The results evidenced that the osmotic effect is continuous but, at low level of salinity, the ionic effect may be significant in reducing growth and, the application of nutrients (Hoagland solution), alleviates the salinity stress on plants. However, above a threshold value of soil solution salinity, the osmotic effect becomes the dominant mechanism limiting the growth. Although the term salinity implies high concentration of salts in soil NaCl contributes the most part in soil salinity and this explains why all plants have evolved some mechanisms to regulate NaCl accumulation or exclusion. Moreover, the specific-ion toxicity of NaCl is not only the result of an excessive Na uptake, but a combined contribution of both Na⁺ and Cl⁻ as well. In fact Cl⁻ concentrations may be higher than those of Na⁺ ions, cations that can be adsorbed by soil particles. Generally, anions like Cl⁻ are repelled from soil surface and retained in soil solution where they can accumulate also at large amount, controlling the overall salt concentration of the soil solution. For most species Na⁺ appears to reach a toxic

concentration before Cl⁻ does; however, for some crops, such as soybean, citrus, and grapevine, Cl⁻ is considered to be the more toxic ion (Storey and Walker 1999).

5.1 Hydroponic vs. Soil Systems

The majority of works regarding salt effects and developing selection criteria for improved salt tolerance in plants has been done using solution culture, assuming that responses in hydroponics mimic those in soil. In a recent study on barley, Tavakkoli et al. (2010) show that the effects of salinity on plants differed between the hydroponic and soil systems. The salt concentration in the rhizosphere may increase, as a result of decreasing water content in the vicinity of the roots, due to the high transpiration demand and low hydraulic conductivity of soil. This does not occur in solution culture where no ion gradients will build up and neither depletion nor salt accumulation in the rhizosphere will occur. In soil sudden changes of salt concentration are unlikely

because of the soil-buffering capacity, associated with the cation exchange soil complex (Vetterlein et al. 2004). Thus, plants in soil have more time to adapt to the increase of salinity than plants in hydroponic system. This is of a particular importance for cellular homeostasis adjustments that require ion uptakes and compatible solute accumulation. A result consistent with the two systems is the more significant negative effect of osmotic stress on plant growth, in comparison with the specific ion effects (Tavakkoli et al. 2010). This agrees with assumptions by Munns (2005) and Rengasamy (2010) which indicated that the biggest reduction in growth is caused by the osmotic stress and a relatively smaller effect is due to the genetic differences in ion exclusion.

5.2 Soil Constraints on Root Growth in Saline Environment

The evaluation of the average soil salinity and water content of a specific soil layer cannot be considered comprehensive to calculate the effective soil solution salinity roots are exposed to. In fact, it does not consider aspects concerning interactions between roots and soil at the root/ soil interface. Driven by transpiration of the shoot, saline soil solution moves from the bulk soil to the root surface where water uptake occurs but most ions are excluded. Consequently, rhizospheric soil can be up to 15 times more saline than the bulk soil and this gradient is also more expressed under conditions of higher ET demand. The osmotic water potentials of the soil solution contacting the root surface are significantly lower than the bulk soil and this gradient initiates a flow of soil solution directed to the root surface (mass flow). Then, the increase in soil salinity, as result of evaporation, occurs at the soil interface, while the site of separation of salts from the soil water, due to root water uptake, takes place at the soilroot interface.

In many saline soils a deterioration of the structure leads some physical constraints that in the root zone appear principally in the form of compaction and crusting. Low porosity restricts rates of water and nutrient uptake by roots as well as gas exchange, whereas high soil strength directly inhibits root elongation and expansion. Soil oxygen movement to roots is critical to maintain adequate respiration for plant growth. Under anoxic conditions some bacteria shift metabolic pathways so as to utilize alternative terminal electron acceptors and produce some substances, such as hydrogen sulfide, that are toxic for plants. Roots need nutrition, water aeration, and low mechanical strength to grow and function in the soil environment. The study of interactions between root properties (morphology and activity) and soil conditions are relevant to assess the water supply of plants and the salt tolerance of plants. If root growth and physiological processes in the root are affected, adverse leaf water status and top growth can occur through both hydraulic and biochemical signals.

5.3 Na⁺ Uptake and Accumulation in Roots

In most plants, roots should exclude 98% of the salt in soil solution allowing only 2% to be transported to the shoot. Then roots filter out most of the salt in the soil while taking up water and play a fundamental role in protecting the plants from excessive uptake of salts. Furthermore, roots have a remarkable ability to control their Na⁺ and Cl⁻ concentration, which is rarely much higher than in external solution. (Munns 2005)

Unidirectional influx and efflux provide the two main components of the currently accepted model of Na⁺ uptake in plants. Na⁺ ions passively enter the cell, down the ion's electrochemical potential gradient, and exit the cell via a secondarily active proton-driven sodium with a probable Na⁺:H⁺ stoichiometry of 1:1 (SOS1; Shi et al. 2002). This process consumes significant cellular energy. Recently, Malagoli et al. (2008) showed that the energy predicted to drive active Na⁺ efflux in rice roots was much greater than the measured one. This discrepancy may indicate the involvement of more Na⁺-specific transport systems and interestingly, a sodium– potassium–chloride transporter has recently been discovered in *A. thaliana* (Colmenero-Flores et al. 2007). Then, it was suggested a possible mechanism in which active Na⁺ efflux is energized differently from current models, possibly via its coupling to passive fluxes of ions other than protons.

The compartmentation in root vacuoles of remaining Na⁺ is achieved by tonoplast Na⁺/H⁺ antiporters. A passive leakage of Na⁺ back to the cytosol (possibly via tonoplast nonselective cation channels) requires a constant resequestration of Na⁺ into vacuoles (Apse et al. 1999). This mechanism allows plants to minimize or delay the toxic effects of high concentrations of salts, so genotypes with a poor ability to sequester salts have a greater rate of leaf death. Therefore, an efficient sequestration system may improve tissue tolerance by plants, perhaps by reducing cytosolic Na⁺ concentrations. As the water moves from the soil across the root cortex ions are transported by this stream toward the stele. Some X-ray microanalysis on roots of wheat plants showed that the root cortex is the main barrier to Na⁺ transport into the stele, rather than the endodermis (Läuchli et al. 2005) and the highest concentration of Na⁺ was in the cell layer of pericycle. Similar results show substantial sequestering of large amount of Na⁺ and Cl⁻ in vacuoles of pericycle cells in grapevine roots, grown at relatively low salinity (25 mM NaCl), suggesting an important role of pericycle in the radial transport of Na⁺ and resulting xylem loading (Storey et al. 2003).

5.4 How Salinity Is Sensed in Roots

The perception of salinity is achieved by both ionic and osmotic stress signals in plants. The responses of root cells are finalized to maintain their own correct functionality, despite the condition of elevated Na⁺ concentration. Long distance signals to shoots are activated in the form of hormones or their precursors; in fact the reduction of leaf growth under salinity is independent of carbohydrate supply and water status (Turka and Demiral 2009). ABA plays a central role in root-to-shoot and cellular signaling, but gibberellins are also involved. ABA can inhibit leaf elongation by lowering the content of active GA, as observed in barley leaves (Munns et al. 2006). Root growth is usually less affected by salinity than leaf growth. Root elongation rates recover remarkably well after exposure to NaCl or other osmotica and, unlike leaves, the recovery takes place despite turgor. Changes in wall properties must occur, but the mechanism is unknown. With time, reduced initiation of new lateral or seminal roots is evident.

Signals within root cells are likely independent from ABA. Plants respond directly and specifically to the addition of Na⁺ within seconds. Then, a plasma membrane protein must be the sensor, but this is still obscure. The first recorded response in roots is an increase in [Ca⁺²]_{cvt} from an influx across the plasma membrane and also from the tonoplast. This perturbation in Ca⁺² level activates salt stress signaling, sensed by a protein (SOS_2) that interacts with a protein kinase, identified as SOS₂. The complex SOS₃/ SOS₂, enabling the phosphorylation, activates the membrane bound Na⁺/H⁺ antiporter, SOS₁, that is responsible of Na⁺ efflux. The discovery of the SOS (Salt-Overlay-Sensitive) pathway in Arabidopsis clarified how Na⁺ (ionic stress) is sensed and the relationship between ion homeostasis and salinity tolerance. However, Arabidopsis is a glycophite species, sensitive to moderate levels of NaCl, and the adaptive responses to Na⁺ in this plant should be extrapolated with caution. In fact, if Arabidopsis remains a useful model to study and discover plant Na⁺ transport processes, the identification of signaling pathways in salt tolerant species is more relevant to define adaptive rather than dysfunctional responses to salinity. The relationship between Na⁺ tolerance and Na⁺ accumulation is different in Arabidopsis and cereals (Tester and Davenport 2003). More work is necessary for the identification of the different mechanisms that are fundamental to specific aspects of salinity tolerance, and also the evaluation of the time of exposure and the severity of salt treatment are important, because they determine the physiological and molecular changes that are detected.

5.5 Root Form and Function in Saline Environment

Root system is the main interface between plants and their environment, and shows a high degree of plasticity in its development in response to local heterogeneity of the soil. On the level of the individual root and the entire root system, various morphological parameters such as length, section, surface area, root hairs are used as potential indicator of root plasticity. Moreover, responses of biomass allocation patterns and structural traits such as SRL, RTD, and root diameter distribution, are associated with acquisition capacities for below-ground resources and respond to stresses and environmental changes. Therefore, some morphological modifications, at the individual root level can affect the structural and physiological characteristics of the entire root system and this can change water uptake and nutrient supply by plants.

Rice is considered as a moderately salt-sensitive crop, although a large variability exists among cultivars, as well as between developmental stages (Bahaji et al. 2002). A delay in the emergence of primary, adventitious, and lateral roots and a subsequent inhibition of root development, in terms of number and length were common responses to osmotic and saline treatments. However, some specific NaCl responses were detected and concerned lateral root development. In particular, lateral roots were thicker, as well as more densely arranged, and more irregular spaced than those of control plants. Furthermore, some bifurcations were occasionally noticed in primary and adventitious roots of NaCl stressed rice seedlings. In rice under salinity silicon can accumulate to high levels and it reduces Na⁺ loading to xylem in plants. X-ray microanalysis of root transverse sections showed that the greatest silicon deposition was in the endodermis. Silicon deposition restricted the movement of water and ions through the apoplast so the Na⁺ uptake was reduced by blocking the influx through the apoplastic pathway. It has been reported a positive role for silicon in reduction of salt stress in many crop grasses including wheat, maize, and barley (Munns 2002; Munns et al. 2006; Flowers 2004).

Wang et al. (2009) showed that high salt exposure suppressed lateral root initiation and organogenesis in Arabidopsis thaliana, resulting in the abortion of lateral root development but, on the other hand, salt stress markedly promoted lateral root elongation. The lateral root shaping is considered a prime example of developmental plasticity because both number and placement of lateral roots are highly responsive to external cues. This indicates that there must be a signal transduction pathway that interprets complex environmental conditions and makes the "decision" to form a lateral root at a particular time and place. Auxin plays a key role in shaping plant architecture and it mediates responses to a broad range of external signals. Histochemical staining, physiological experiments using transport inhibitors and genetic analysis revealed that the quantity of auxin and its patterning in roots were both greatly altered by exposure to high concentrations of salt stress and auxin transport pathway is important for adaptive root system development under salt stress (Malamy 2005). Root hairs can make up 70-80% of the root surface area. They play an important role in nutrient uptake, and root hair number and density generally increase as a consequence of a nutrient stress (Glory and Jones 2000). Wang et al. (2008) showed that in Arabidopsis thaliana root hair number and density decreased significantly under salinity, in a dose-dependent manner, and they reported a physiological mechanism for root hair development in response to salt stress. They hypothesize that salt stress may affect cell-fate specification and the reduction in root hair number is likely caused by a decrease in the epidermal cells differentiating into trichoblasts. The inhibition was sensitive to ions but not to osmotic stress, and was considered an adaptive mechanism to avoid excessive ion uptake, by reducing the absorptive area when ion disequilibrium occurs in roots. Furthermore, because of high sensitivity of root hairs toward salt, they suggest a possible role of root hair alteration as an early indicator of salt stress and plant response.

Salinity stress is also responsible for thickening of roots. Some studies were carried out on growth and changes in structure of root cells in Kikuyu seedlings grown in Hoagland nutrient solution with different salt concentrations (Panuccio et al. 2002, 2003; Muscolo et al. 2003). The cross sections of the primary structure of Kikuyu grass roots exposed to 50-100 mM NaCl did not show significant changes in the cortex growth and stele development; in contrast, 200 mM NaCl caused a significant reduction in the relative volume of the endodermis around the central cylinder, a thickness of the Casparian band and an increase in the number and diameter of root metaxylem vessels. These anatomical modifications may increase the mechanical resistance and decrease the root permeability to avoid the toxic effects of ions in excess. Lens culinaris has always been considered as a salt-sensitive species, but the microsperma landrace "Ustica" is a genotype that behaves like a salt-tolerant one because of its adaptation to the particular environment of the homonymous little island (North of Sicily). Some studies have been conducted to evaluate salt effects and plant responses in Ustica seedlings, grown for 20 days in microcosms, using agrilite as solid substrate and in the presence of different salt concentrations (0, 50, 100, 200 mM NaCl) (unpublished data). Various morphological root parameters such as length (cm), diameter (mm) and surface area (cm²) were tested, by using an image analysis system. The results were also compared with those of a commercial cultivar (Eston), as they are valuable parameters when describing and comparing root systems. In both cultivars, the length of lateral roots was more affected than that of primary roots, but to different extent (Fig. 8.1). The parameter "root length" is considered more important than the "root weight" to indicate the root functionality, because it expresses the potential for solute and water uptake (Ryser 2006). In Eston seedlings, exposed to 100 mM NaCl, no lateral roots were expressed while seedling of Ustica showed an inhibition of the lateral root length even though the number was not significantly influenced. Generally, a water supply reduction in plants brings to a lower lateral root production (Fig. 8.1). The SRL values were higher in Ustica than in Eston seedlings (Table 8.1). SRL is the length-to-mass ratio; it is believed to characterize economic aspect of the



Fig. 8.1 Effects of salinity (0–200 mM NaCl) on primary and lateral root length and on lateral number of a landrace (Ustica) and a commercial cultivar (Eston) of *Lens culinaris* seedlings

Table 8.1	Effects of salinity (0-	200 mM NaCl) on	specific root	length (cm	g-1) and root	t tissue density	(g cm ⁻³) of a
landrace (U	Jstica) and a commerci	al cultivar (Eston)	of Lens culind	aris seedling	gs		

	NaCl (mM)	Specific root length (cm g ⁻¹)	Root tissue density (g cm ⁻³)
Ustica	0	8.0 ± 1.9	48.1±6.1
	50	10.6 ± 2.9	42.0±9.3
	100	2.8±0.3	73.1±1.9
	200	4.5 ± 2.0	82.2±10.7
Eston	0	4.9 ± 0.3	47.9 ± 1.7
	50	3.5 ± 0.5	63.3 ± 4.7
	100	1.1 ± 0.1	106.1 ± 11.2
	200	_	_



Fig. 8.2 Effects of salinity (0–200 mM NaCl) on the frequency (%) of different root diameter classes of primary (**a**) and lateral roots (**b**) of *Lens culinaris* landrace (Ustica)

root system and is frequently used as indicator of root fineness. Then, higher SRL results from longer and thinner roots per unit construction cost (root mass) and this root apparatus is more effective in water and nutrient uptake (Fitter 2002). SRL is a complex parameter that includes variations in root diameter and RTD, which respond to environmental conditions differently (Ryser 2006). In the Ustica variety, a salinity increase leads to an increase in SRL (Table 8.1), due to thicker lateral roots and the root diameter distribution was shifted toward larger diameter classes (Fig. 8.2a, b). Root diameter distribution is usually expressed as the mean diameter but sometimes it does not necessarily characterize a response of root system structure adequately. In fact, fine and coarse roots show different responses, indicating that root diameter classes should be considered as functionally distinct and regarded separately to fully understand stress responses of root systems. It is known that roots with a smaller root diameter can contact a larger soil volume per

unit root surface area; however, the maintenance carbon cost of producing finer roots may be higher as these will have to be replaced more frequently (Fitter 2002). In Ustica plants, coarse roots, for both principal and lateral prevailed under high salinity conditions. This result can be explained by considering that under salinity, the construction costs per root length should be minimized because of the onset of growth-limiting conditions, and the root development resulted further inhibited to counter water stress and ion toxicity due to the salt around the root. Apart from the effects on root biomass production, contrasting root morphological responses of ecotypes to salt treatments might be partially responsible for dissimilar abilities to tolerate salinity. Structural and morphological differences in roots certainly play an essential role for nutrient and water uptake by plants from saline soil and the study of these parameters can help to determine different mechanisms underlying salt toxicity and the way plants can cope with saline conditions. Some modifications of root morphology should not be considered a simple growth stopping, but rather an induced reorientation of growth which is related to stress avoidance. This information could be considered an important tool in studies that involve salt tolerance improvements in plants.

6 Conclusion and Future Perspectives

As detailed above, the root system may have a fundamental role in relieving the disturbances caused by the variables of the climate change on the plant growth, development, and production. However, an exciting challenge will be to understand the following key aspects regarding the impact of the root system on plant adaptation to the climate change:

- An increase of the knowledge on the root responses to the interactive effects of the climate variables change (high temperature vs. drought and/or salinity and/or drought) that usually occur together
- 2. An greater understanding, in an integrated view, of physiological processes (development,

growth, metabolic) involved in root responses to the warmer, drought, and salinity environments

- 3. A study of changes in proteins, metabolites, and other compounds inside the root cells by advanced genomic techniques for better understanding the molecular mechanisms implemented by the plant in response to temperature, water availability, and salinity change
- 4. An increase of the knowledge on the genetic diversity of the root system in plant response to climate variables change

An improved understanding of these aspects together with genomics, proteomics, and transcriptomic approaches are likely to pave the way for engineering roots that can withstand and give satisfactory economic yield under climate change.

References

- Adam ML, Kelly JM, Graves WR, Dixon PM (2003) Net nitrate uptake by red maple is a function of root-zone temperature. J Plant Nutr 26:203–222
- Aguilar EA, Turner DW, Sivasithamparam K (1999) Aerenchyma formation in roots of four banana (*Musa* spp.) cultivars. Sci Hortic 80:57–72
- Aguirrezabal LAN, Deleens E, Tardieu F (1994) Root elongation rate is accounted for by intercepted PPFD and source-sink relations in field and laboratory-grown sunflower. Plant Cell Environ 17:443–450
- Aharon R, Shahak Y, Winiger S, Bendov R, Kapulnik Y, Galili G (2003) Overexpression of a plasma membrane aquaporin in transgenic tobacco improves plant vigor under favorable growth conditions but not under drought or salt stress. Plant Cell 15:439–447
- Annicchiarico P, Piano E (2004) Indirect selection for root development of white clover and implication for drought tolerance. J Agric Crop Sci 190:28–34
- Apse MP, Aharon GS, Snedden WA, Blumwald E (1999) Salt tolerance conferred by overexpression of a vacuolar Na⁺/H⁺ antiport in *Arabidopsis*. Science 285:1256–1258
- Ashi-Smiti S, Chaibi W, Brouquisse RB, Saglio P (2003) Assessment of enzyme induction and aerenchyma formation as mechanisms for flooding tolerance in *Trifolium subterraneum* 'Park'. Ann Bot 91: 195–204
- Atkin OW, Edwards EJ, Loveys BR (2000) Response of root respiration to changes in temperature and its relevance to global warming. New Phytol 147:141–154
- Azhiri-Sigari T, Yamauchi A, Kamoshita A, Wade LJ (2000) Genotypic variation in response of rainfed lowland rice to drought and rewatering: II. Root growth. Plant Prod Sci 3:180–188

- Bacanamwo M, Purcell LC (1999) Soybean root morphological and anatomical traits associated with acclimation to flooding. Crop Sci 39:143–149
- Bahaji A, Mateu I, Sanz A, Corneo MJ (2002) Common and distinctive responses of rice seedlings to salineand osmotically-generated stress. Plant Growth Regul 38:83–94
- Bai W, Wan S, Niu S, Liu W, Chen Q, Wang Q, Zhang W, Han X, Li L (2010) Increased temperature and precipitation interact to affect root production, mortality and turnover in a temperate steppe: implications for ecosystem C cycling. Glob Change Biol 16:1306–1316
- Bailey-Serres J, Chang R (2005) Sensing and signalling in response to oxygen deprivation in plants and other organisms. Ann Bot 96:507–518
- BassiriRad H (2000) Kinetics of nutrient uptake by roots: responses to global change. New Phytol 147:155–169
- BassiriRad H, Caldwell MM (1992a) Temporal changes in root growth and ¹⁵N-uptake and water relations of two tussock grass species recovering from water stress. Physiol Plant 86:525–531
- BassiriRad H, Caldwell MM (1992b) Root growth, osmotic adjustment and NO₃- uptake during and after a period of drought in Artemisia tridentata. Aust J Plant Physiol 19:493–500
- BassiriRad H, Caldwell MM, Bilbrough C (1993) Effects of soil temperature and nitrogen status on kinetics of ¹⁵NO₃-uptake by roots of field-grown *Agropyron desertorum* (Fisch. ex Link) Schult. New Phytol 123: 485–489
- BassiriRad H, Tissue DT, Reynolds JF, Chapin FS III (1996) Response of *Eriophorum vaginatum* to CO_2 enrichment at different soil temperatures: effects on growth, root respiration and PO_4^{3-} uptake kinetics. New Phytol 133:423–430
- Baurle TL, Smart DR, Baurle WL, Stockert C, Eissenstat DM (2008) Root foraging in response to heterogeneous soil moisture in two grapevines that differ in potential growth rate. New Phytol 179:857–866
- Biemelt S, Keetman U, Albrecht G (1998) Re-aeration following hypoxia or anoxia leads to activation of the antioxidative defence system in roots of wheat seedlings. Plant Physiol 116:651–658
- Bjork RG, Majdi H, Klemedtsson L, Lewis-Jonsson SC, Molau U (2007) Long-term warming effects on root morphology, root mass distribution, and microbial activity in two dry tundra plant communities in northern Sweden. New Phytol 176:862–873
- Boru G, van Ginkel M, Trethowan RM, Boersma L, Kronstad WE (2003) Oxygen use from solution by wheat genotypes differing in tolerance to waterlogging. Euphytica 132:151–158
- Bouma TJ, Nielsen KL, Eissenstat DM, Lynch JP (1997) Estimating respiration of roots in soil: interactions with soil CO₂, soil temperature, and soil water content. Plant Soil 195:221–232
- Bouma TJ, Nielsen KL, van Hal J, Koutstaal B (2001) Root system topology and diameter distribution of species from habitats differing in inundation frequency. Funct Ecol 15:360–369

- Bucci SJ, Scholz FG, Goldstein G, Meinzer FC, Arce ME (2009) Soil water availability and rooting depth as determinants of hydraulic architecture of Patagonian woody species. Oecologia 160:631–641
- Buljovicic Z, Engels C (2001) Nitrate uptake ability by maize roots during and after drought stress. Plant Soil 22:125–135
- Canadell J, Jackson RB, Ehleringer JB, Mooney HA, Sala OE, Schulze ED (1996) Maximum rooting depth of vegetation types at the global scale. Oecologia 108:583–595
- Chang WWP, Huang L, Shen M, Webster C, Burlingame AL, Oberts JKM (2000) Patterns of protein synthesis and tolerance to anoxia in root tips of maize seedlings acclimated to a low-oxygen environment, and identification of proteins by mass spectrometry. Plant Physiol 122:295–317
- Cheng X, Huang M, Shao M, Warrington DN (2009) A comparison of fine root distribution and water consumption of mature *Caragana korshinkii* Kom grown in two soils in a semiarid region, China. Plant Soil 315:149–161
- Ching PC, Barber SA (1979) Evaluation of temperature effects on K uptake by corn. Agron J 71:1040–1044
- Clarkson DT, Earnshaw MJ, White PJ, Cooper HD (1988) Temperature dependent factors influencing nutrient uptake: an analysis of responses at different levels of organization. In: Long SP, Woodward FI (eds) Plants and temperature, vol 42. The Society for Experimental Biology, Cambridge, pp 281–309
- Colmenero-Flores JM, Martinez G, Gamba G, Vazquez N, Iglesias DJ, Brumos J, Talon M (2007) Identification and functional characterization of cation-chloride cotransporters in plants. Plant J 50:278–292
- Colmer TD (2003a) Long-distance transport of gases in plants: a perspective on internal aeration and radial oxygen loss from roots. Plant Cell Environ 26:17–36
- Colmer TD (2003b) Aerenchyma and an inducible barrier to radial oxygen loss facilitate root aeration in up-land paddy and deep-water rice (*Oryza sativa* L.). Ann Bot 91:301–309
- Colmer TD, Gibberd MR, Wiengweera A, Tinh TK (1998) The barrier to radial oxygen loss from roots of rice (*Oryza sativa* L.) is induced by growth in stagnant solution. J Exp Bot 49:1431–1436
- Cooper AJ (1973) Root temperature and plant growth. A review. Research review No. 4. Commonwealth Bureau of Horticulture and Plantation Crops, Commonwealth Agricultural Bureau, Farnham Royal, UK
- Côté B, Hendershot WH, Fyles JW, Roy AG, Bradley R, Biron PM, Courchesne F (1998) The phenology of fine root growth in a maple-dominated ecosystem: relationships with some soil properties. Plant Soil 201:59–69
- Coutts MP, Philipson JJ (1978) The tolerance of tree roots to waterlogging. I. Survival of Sitka spruce and Lodgepole pine. New Phytol 80:63–69
- Doussan C, Vercambre G, Pages L (1998) Modelling of the hydraulic architecture of root systems: An integrated approach to water absorption – Distribution of axial and radial conductances in maize. Ann Bot. 81:225–232

- Drew MC (1997) Oxygen deficiency and root metabolism: injury and acclimation under hypoxia and anoxia. Annu Rev Plant Physiol Plant Mol Biol 48:223–250
- Drew MC, Armstrong W (2002) Root growth and metabolism under oxygen deficiency. In: Waisel Y, Eshel A, Kafkafi U (eds) Plant roots: the hidden half, 3rd edn. Marcel Dekker, New York, pp 729–761
- Drew MC, Jackson MB, Giffard S (1979) Ethylenepromoted adventitious rooting and development of cortical air spaces in roots may be adaptive responses to flooding in *Zea mays* L. Planta 147:83–88
- Drew MC, He CJ, Morgan PW (2000) Programmed cell death and aerenchyma formation in roots. Trends Plant Sci 5:123–127
- Dubrovsky JG, North GB, Nobel PS (1998) Root growth developmental changes in the apex and hydraulic conductivity for *Opuntia ficus-indica* drought. New Phytol 138:75–82
- Edwards EJ, Benham DG, Marland LA, Fitter AH (2004) Root production is determined by radiation flux in a temperate grassland community. Glob Change Biol 10:209–227
- Eissenstat DM (1991) On the relationship between specific root length and the rate of root proliferation: a field study using citrus root stocks. New Phytol 118:63–68
- Evans DE (2003) Aerenchyma formation. New Phytol 161:35–49
- Fan L, Neumann PM (2004) The spatially variable inhibition by water deficit of maize root growth correlates with altered profiles of proton flux and cell wall pH. Plant Physiol 135:2291–2300
- Fitter AH (2002) Characteristics and functions of root systems. In: Waisel Y, Eshel A, Kafkafi U (eds) Plant roots: the hidden half, 3rd edn. Marcel Dekker, New York, pp 15–32
- Fitter AH, Stickland TR (1991) Architectural analysis of plant root systems. 2. Influences of nutrient supply on architecture in contrasting plant species. New Phytol 118:383–389
- Flowers TJ (2004) Improving crop salt tolerance. J Exp Bot 55:307–319
- Fukao T, Xu K, Ronald PC, Bailey-Serres J (2006) A variable cluster of ethylene response factor-like genes regulates metabolic and development acclimation responses to submergence in rice. Plant Cell 18:2021–2034
- Garnett TP, Smethurst PJ (1999) Ammonium and nitrate uptake by *Eucalyptus nitens*: effect of pH and temperature. Plant Soil 214:133–140
- Gavito ME, Curtis PS, Mikkelsen TN, Jakobsen I (2001) Interactive effects of soil temperature, atmospheric carbon dioxide and soil N on root development, biomass and nutrient uptake of winter wheat during vegetative growth. J Exp Bot 52:1913–1923
- Gavito ME, Schweiger P, Jakobsen I (2003) P uptake by arbuscular mycorrhizal hyphae: effect of soil temperature and atmospheric CO₂ enrichment. Glob Change Biol 9:106–116
- Geigenberger P (2003) Response of plant metabolism to too little oxygen. Curr Opin Plant Biol 6:223–250
- Gibberd MR, Colmer TD, Cocks PS (1999) Root porosity and oxygen movement in waterlogging-tolerant

Trifolium tomentosum and -intolerant *Trifolium glomeratum*. Plant Cell Environ 22:1161–1168

- Gibberd MR, Gray JD, Cocks PS, Colmer TD (2001) Waterlogging tolerance among a diverse range of trifolium accessions is related to root porosity, lateral root formation and 'aerotropic rooting'. Ann Bot 88:579–589
- Glory S, Jones DL (2000) Through form to function: root hair development and nutrient uptake. Trends Plant Sci 5:56–60
- Gregory PJ (1986) Response to temperature in a stand of pearl millet (*Pennisetum typhoides* S. & H.). VIII. Root growth. J Exp Bot 37:379–388
- Gunawardena AHLAN, Pearce DM, Jackson MB, Awes CR, Evans DE (2001) Characterization of programmed cell death during aerenchyma formation induced by ethylene or hypoxia in roots of maize (*Zea mays* L.). Planta 212:205–214
- Gunn S, Farrar JF (1999) Effects of a 4°C increase in temperature on partitioning of leaf area and dry mass, root respiration and carbohydrates. Funct Ecol 13:12–20
- He J, Tan LP, Lee SK (2009) Root-zone temperature effects on photosynthesis, 14C-photoassimilate and growth of temperate lettuce (Lactuca sativa cv. Panama) in the tropics. Photosynthetica 47:95–103
- Heinemeyer A, Fitter AH (2004) Impact of temperature on the arbuscular mycorrhizal (AM) symbiosis: growth responses of the host plant and its AM fungal partner. J Exp Bot 55:525–534
- Hinsinger P (1998) How do plant roots acquire mineral nutrients ? Chemical processes involved in the rhizosphere. Adv Agron 64:225–265
- Hinsinger P (2001) Bioavailability of soil inorganic P in the rhizosphere as affected by root-induced chemical changes: a review. Plant Soil 237:173–195
- Hinsinger P, Plassard C, Tang C, Jaillard B (2003) Origins of root-induced pH changes in the rhizosphere and their responses to environmental constraints: a review. Plant Soil 248:43–59
- Hinsinger P, Gobran GR, Gregory PJ, Wenzel WW (2005) Rhizosphere geometry and heterogeneity arising from root-mediated physical and chemical processes. New Phytol 168:293–303
- Hinsinger P, Plassard C, Jaillard B (2006) Rhizosphere: A new frontier for soil biogeochemistry. J Geochem Explor 88:210–213
- Ho MD, Rosas JC, Brown KM, Lynch JP (2005) Root architectural tradeoffs for water and phosphorus acquisition. Funct Plant Biol 32:737–748
- Huang X-F, Binh N-Q, Prem SC, Serge Y (1994) Complete nucleotide sequence of the maize (*Zea mays* L.) sucrose synthase 2 cDNA. Plant Physiol 104:293–294
- Huang B, Liu X, Fry JD (1998) Effects of high temperature and poor soil aeration on root growth and viability of creeping bentgrass. Crop Sci 40:1618–1622
- Hund A, Ruta N, Liedgens M (2009) Rooting depth and water use efficiency of tropical maize inbred lines, differing in drought tolerance. Plant Soil 318:311–325
- IPCC (2007) Working group I Fourth Assessment Report. Cambridge University Press, Cambridge
- Ito K, Tanakamaru K, Morita S, Abe J, Inanaga S (2006) Lateral root development, including responses to soil

drying, of maize (Zea mays) and wheat (Triticum aestivum) seminal roots. Physiol Plant 127:260–267

- James EK, Sprent JI (1999) Development of N₂-fixing nodules on the wetland legume *Lotus uliginosus* exposed to conditions of flooding. New Phytol 142:219–231
- Jang J, Lee S, Rhee J, Chung G, Ahn S, Kang H (2007) Transgenic Arabidopsis and tobacco plants overexpressing an aquaporin respond differently to various abiotic stresses. Plant Mol Biol 64:621–632
- Javot H, Maurel C (2002) The role of aquaporins in root water uptake. Ann Bot 90:301–313
- Johnson MG, Rygiewicz PT, Tingey DT, Phillips DL (2006) Elevated CO₂ and elevated temperature have no effect on Douglas-fir fine-root dynamics in nitrogenpoor soil. New Phytol 170:345–356
- Joslin JD, Wolfe MH, Hanson PJ (2001) Factors controlling the timing of root elongation intensity in a mature upland oak stand. Plant Soil 228:201–212
- Jupp P, Newman EI (1987) Phosphorus uptake from soil by *Lolium perenne* during and after drought. J Appl Ecol 24:979–990
- Kandeler E, Tscherko D, Bardgett RD, Hobbs PJ, Kampichler C, Jones TH (1998) The response of soil microorganisms and roots to elevated CO₂ and temperature in a terrestrial model ecosystem. Plant Soil 202:251–262
- Kashiwagi J, Krishnamuthy JH, Crouch JH, Seraj R (2006) Variability of root length density and its contributions to seed yield in chickpea (*Cicer arietinum* L.) under terminal drought stress. Field Crops Res 95:171–181
- Kaspar TC, Bland WL (1992) Soil temperature and root growth. Soil Sci 152:290–299
- Kato Y, Abe J, Kamoshita A, Yamagishi J (2006) Genotypic variation in root growth angle in rice (*Oryza sativa* L.) and its association with deep root development in upland fields with different water regimes. Plant Soil 287:117–129
- Kaway M, Samarajeewa PK, Barrero RA, Nishiguchi M, Uchimiya H (1998) Cellular dissection of the degradation pattern of cortical cell death during aerenchyma formation of rice roots. Planta 204:277–287
- Kramer PJ, Boyer JS (1995) Water relations of plants and soils. Academic Press, San Diego USA
- Kuhns MR, Garrett HE, Teskey RO, Hinckley TM (1985) Root growth of black walnut trees related to soil temperature, soil water potential, and leaf water potential. For Sci 31:617–629
- Laan P, Berrevoets MJ, Lythe S, Armstrong W, Blom CWM (1989) Root morphology and aerenchyma formation as indicators of the flood-tolerance of *Rumex* species. J Ecol 77:693–703
- Lainè P, Ourry A, MacDuff J, Boucaud J, Salette J (1993) Kinetic parameters of nitrate uptake by different catch crop species: effects of low temperatures or previous nitrate starvation. Physiol Plant 88:85–92
- Lal SK, Lee C, Sachs MM (1998) Differential regulation of enolase during anaerobiosis in maize. Plant Physiol 118:1285–1293
- Läuchli A, James RA, Munns R (2005) Salt exclusion and the cell-specific localization of Na in wheat roots. In: Proceedings of the Society for Experimental Biology, annual main meeting, Barcelona, Spain, July 2005

- Liang BM, Sharp RE, Baskin TI (1997) Regulation of growth anisotropy in well-watered and water-stressed maize roots. Plant Physiol 115:101–111
- Lo Gullo MA, Nardini A, Salleo S, Tyree MT (1998) Changes in root hydraulic conductance (Kr) of *Olea oleaster* seedling following drought stress and irrigation. New Phytol 140:25–31
- Ludlow MM, Santamaria JM, Fukai S (1990) Contribution of osmotic adjustment to grain-yield in *Sorghum bicolor* (L) Moench under water-limited conditions. 2. Water-stress after anthesis. Aust J Agric Res 41:67–78
- Lynch JP, Nielsen KL (1996) Simulation of root system architecture. In: Waisel Y, Eshel A, Kafkafi U (eds) Plant roots: the hidden half, 2nd edn. Marcel Dekker, New York, pp 247–258
- Macduff JH, Dhanoa MS (1990) N_2 fixation and nitrate uptake by white clover swards in response to root temperature in flowing solution culture. Ann Bot 65:325–335
- Mackay AD, Barber SA (1984) Soil-temperaure effects on roots growth and phosphorus uptake by corn. Soil Sci Soc Am J 48:818–823
- Majdi H, Ohrvik J (2004) Interactive effects of soil warming and fertilization on root production, mortality, and longevity in a Norway spruce stand in Northern Sweden. Glob Change Biol 10:182–188
- Malagoli P, Britto T, Schulze L, Kronzucker HJ (2008) Futile Na⁺ cycling at the root plasma membrane in rice (*Oryza sativa* L.): kinetics, energetics, and relationship to salinity tolerance. J Exp Bot 59:4109–4117
- Malamy JE (2005) Intrinsic and environmental response pathways that regulate root system architecture. Plant Cell Environ 28:67–77
- Malik AI, Colmer TD, Lambers H, Setter TL, Schortemeyer M (2002) Short-term waterlogging has long-term effects on the growth and physiology of wheat. New Phytol 153:225–236
- Manschadi AM, Christopher J, Devoil P, Hammer GL (2006) The role of root architectural traits in adaptation of wheat to water-limited environments. Funct Plant Biol 33:823–837
- Manschadi AM, Hammer GL, Christopher JT, deVoil P (2008) Genotypic variation in seedling root architectural traits and implications for drought adaptation in wheat (*Triticum aestivum* L.). Plant Soil 303:115–129
- Markesteijn L, Poorter L (2009) Seedling root morphology and biomass allocation of 62 tropical tree species in relation to drought- and shade-tolerance. J Ecol 97:311–325
- Matsui T, Singh BB (2003) Root characteristics in cowpea related to drought tolerance at the seedling stage. Exp Agric 39:29–38
- Matzner ST, Richards JH (1996) Sagebrush (Artemisia tridentata Nutt.) roots maintain nutrient uptake capacity under water stress. J Exp Bot 47:1045–1056
- Maurel C (2007) Plant aquaporins: novel functions and regulation properties. FEBS Lett 581:2227–2236
- McMichael BL, Burke JJ (1998) Soil temperature and root growth. Hortic Sci 33:947–951
- McMichael BL, Burke JJ (2002) Temperature effects on root growth. In: Waisel Y, Eshel A, Kafkafi U (eds)

Plant roots: the hidden half, 3rd edn. Marcel Dekker, New York, pp 717–728

- Mergemann H, Sauter M (2000) Ethylene induces epidermal cell death at the site of adventitious root emergence in rice. Plant Physiol 124:609–614
- Misra RK (1999) Root and shoot elongation of rhizotrongrown seedlings of *Eucalyptus nitens* and *Eucalyptus* globulus in relation to temperature. Plant Soil 206:37–46
- Mujer CV, Rumpho ME, Lin JJ, Kennedy RA (1993) Constitutive and inducible aerobic and anaerobic stress proteins in the *Echinochloa* complex and rice. Plant Physiol 101:217–226
- Munné-Bosch S, Peñuelas J (2003) Photo- and antioxidative protection, and a role for salicylic acid during drought and recovery in field-grown *Phillyrea* angustifolia plants. Planta 217:758–766
- Munns R (2002) Comparative physiology of salt and water stress. Plant Cell Environ 25:239–250
- Munns R (2005) Genes and salt tolerance: bringing them together. New Phytol 167:645–663
- Munns R, Tester M (2008) Mechanisms of salinity tolerance. Annu Rev Plant Biol 59:651–681
- Munns R, Richard A, Lauchli A (2006) Approaches to increasing the salt tolerance of wheat and other cereals. J Exp Bot 57:1025–1043
- Muscolo A, Panuccio MR, Sidari M (2003) Effect of salinity on growth, carbohydrate metabolism and nutritive properties of Kikuyu grass (*Pennisetum clandestinum*, Hochst). Plant Sci 164:1103–1110
- Nardini A, Salleo S, Lo Gullo MA (1998) Root hydraulic conductance of six forest trees: possible adaptive significance of seasonal changes. Plant Biosyst 132:97–104
- Nardini A, Salleo S, Tyree M (2002) Ecological aspects of water permeability of roots. In: Waisel Y, Eshel A, Kafkafi U (eds) Plant roots: the hidden half, 3rd edn. Marcel Dekker, New York, pp 683–698
- Nelson JA, Morgan JA, LeCain DR, Mosier AR, Milchunas DG, Parton BA (2004) Elevated CO₂ increases soil moisture and enhances plant water relations in a longterm field study in semi-arid shortgrass steppe of Colorado. Plant Soil 259:169–179
- Nicoll BC, Coutts MP (1998) Timing of root dormancy and tolerance to root waterlogging in clonal Sitka spruce. Trees 12:241–245
- Nicoll BC, Ray D (1996) Adaptive growth of tree root systems in response to wind action and site conditions. Tree Physiol 16:899–904
- North GB, Nobel PS (1991) Changes in hydraulic conductivity and anatomy caused by drying and rewetting roots of Agave deserti (Agavaceae). Am J Bot 78:906–915
- North GB, Nobel PS (1992) Drough-induced change in hydraulic conductivity and structure in roots of *Ferocactus acanthodes* and *Opuntis ficus-indica*. New Phytol 120:9–19
- Ober ES, Sharp RE (1994) Proline accumulation in maize (Zea mays L.) primary roots at low water potentials. I. Requirement for increased levels of abscisic acid. Plant Physiol 105:981–987
- Ober ES, Sharp RE (2003) Electrophysiological responses of maize roots to low water potentials: relationship to growth and ABA accumulation. J Exp Bot 54:813–824

- Ober ES, Sharp RE (2007) Regulation of root growth responses to water deficit. In: Jenks MA, Hasegawa PM, Jain SM (eds) Advances in molecular breeding toward drought and salt tolerant crops. Springer, Dordrecht, pp 33–53
- Olsrud M, Melillo JM, Christensen TR, Michelsen A, Wallander H, Olsson PA (2004) Response of ericoid mycorrhizal colonization and functioning to global change factors. New Phytol 162:459–469
- Onderdonk JJ, Ketcheson JW (1973) Effect of soil temperature on direction of corn root growth. Plant Soil 39:177–186
- Onderdonk JJ, Ketcheson JW (1973) Effect of soil temperature on direction of corn root growth. Plant Soil 39:177–186
- Padilla FM, Pugnaire FI (2007) Rooting depth and soil moisture control Mediterranean woody seedling survival during drought. Funct Ecol 21:489–495
- Panuccio MR, Sidari M, Muscolo A (2002) Effect of different salt concentrations and pH conditions on growth of *Pennisetum clandestinum* Hochst (Kikuyu grass) Fresenius. Environ Bull 11:295–299
- Panuccio MR, Sidari M, Muscolo A (2003) Tolerance of Kikuyu grass to long term salt stress is associated with the induction of antioxidant defences. Plant Growth Regul 41:57–62
- Passioura JB (1988) Water transport in and to roots. Annu Rev Plant Physiol Plant Mol Biol 39:245–265
- Paul EA, Clark FE (1996) Soil microbiology and biochemistry. Academic Press, London
- Paz H (2003) Root/shoot allocation and root architecture in seedlings: variation among forest sites, microhabitats, and ecological groups. Biotropica 35:318–332
- Pollack HN, Huang S, Shen P-Y (1998) Climate change record in subsurface temperatures: a global perspective. Science 282:279–281
- Poot P, Lambers H (2003) Growth responses to waterlogging and drainage of woody *Hakea* (Proteaceae) seedlings, originating from contrasting habitats in south-western Australia. Plant Soil 253:57–70
- Pote J, Wang ZL, Huang BR (2006) Timing and temperature of physiological decline for creeping bentgrass. J Am Soc Hortic Sci 131:608–615
- Pregitzer KS, King JS, Burton AJ, Brown SE (2000) Responses of tree fine roots to temperature. New Phytol 147:105–115
- Pregitzer KS, King JS (2005) Effects of soil temperature on nutrient uptake. In: BassiriRad (ed) Nutrient acquisition by plants: an ecological perspective. Springer, Heidelberg, pp 277–310
- Qin L, He J, Lee SK, Dodd IC (2007) An assessment of the role of ethylene in mediating lettuce (*Lactuca sativa*) root growth at high temperatures. J Exp Bot 58:3017–3024
- Qiu F, Zheng Y, Zhang Z, Xu S (2007) Mapping of QTL associated with waterlogging tolerance during the seedling stage in maize. Ann Bot 99:1067–1081
- Rachmilevitch S, Lambers H, Huang B (2008) Shortterm and long-term root respiratory acclimation to elevated temperatures associated with root thermotolerance for two Agrostis grass species. J Exp Bot 59:3803–3809

- Rengasamy P (2010) Osmotic and ionic effects of various electrolytes on the growth of wheat. Australian Journal of Soil Research 37:255–263
- Robinson D, Linehan DJ, Caul S (1991) What limits nitrate uptake from soil? Plant Cell Environ 14:77–85
- Robinson D, Hodge A, Fitter A (2003) Constraints on the form and function of root system. In: de Kroon H, Visser EJW (eds) Root ecology. Springer, Berlin, pp 1–32
- Ross DJ, Kelliher FM, Tate KR (1999) Microbial process in relation to carbon, nitrogen and temperature regimes in litter on a sandy mineral soil from a central Siberian Pinus sylvestris L. forest. Soil Biol Biochem 31:757–767
- Rustad LE, Campbell JL, Marion GM, Norby RJ, Mitchell MJ, Hartley AE, Cornelissen JHC, Gurevitch J (2001) A meta-analysis of the response of soil respiration, net nitrogen mineralization, and aboveground plant growth to experimental ecosystem warming. Oecologia 126:542–562
- Ryser P (1998) Intra- and interspecific variation in root length, root turnover and the underlying parameters. In: Lambers H, Poorter H, Van Vuuren MMI (eds) Inherent variation in plant growth: physiological mechanisms and ecological consequences. Backhuys, Leiden, pp 441–465
- Ryser P (2006) The mysterious root length. Plant Soil 286:1–6
- Saab IM, Sharp RE, Pritchard J, Voetberg GS (1990) Increased endogenous abscisic acid maintains primary root growth and inhibits shoot growth of maize seedlings at low water potentials. Plant Physiol 93:1329–1336
- Sachs MM, Subbaiah CC, Saab IN (1996) Anaerobic gene expression and flooding tolerance in maize. J Exp Bot 47:1–15
- Sarda X, Tousch D, Ferrare K, Cellier F, Alcon C, Dupuis JM, Casse F, Lamaze T (1999) Characterization of closely related δ-TIP genes encoding aquaporins which are differentially expressed in sunflower roots upon water deprivation through exposure to air. Plant Mol Biol 40:179–191
- Schenk HJ, Jackson RB (2005) Mapping the global distribution of deep roots in relation to climate and soil characteristics. Geoderma 126:129–140
- Seago JL Jr, Marsch LC, Stevens KJ, Soukup A, Votrubova O, Enstone DE (2005) A re-examination of the root cortex in wetland flowering plants with respect to aerenchyma. Ann Bot 96:613–623
- Seiler GJ (1998) Influence of temperature on primary and lateral root growth of sunflower seedlings. Environ Exp Bot 40:135–146
- Sharp RE, Silk WK, Hsiao TC (1988) Growth of the maize primary root at low water potential. I. Spatial distribution of expansive growth. Plant Physiol 87:50–57
- Sharp RE, Wu Y, Voetberg GS, Saab IN, LeNoble ME (1994) Confirmation that abscisic acid accumulation is required for maize primary root elongation at low water potentials. J Exp Bot 45:743–751
- Sharp RE, Poroyko V, Hejlek LG, Spollen WG, Springer GK, Bohnert HJ, Nguyen HT (2004) Root growth maintenance during water deficits: physiology to functional genomics. J Exp Bot 55:2343–2351

- Shi HZ, Quinterno FJ, Pardo JM, Zhu JK (2002) The putative plasma membrane Na⁺/H⁺ antiporter SOS1 controls longdistance Na⁺ transport in plants. Plant Cell 14:465–477
- Shiba H, Daimon H (2003) Histological observation of secondary aerenchyma formed immediately after flooding in *Sesbania cannabina* and *S. rostrata*. Plant Soil 255:209–215
- Shimamura S, Mochizuki T, Nada Y, Fukuyama M (2003) Formation and function of secondary aerenchyma in hypocotyls, roots and nodules of soybean (*Glycine max*) under flooded conditions. Plant Soil 151:351–359
- Shone MGT, Flood AV (1983) Effects of periods of localized water stress on subsequent nutrient uptake by barley roots and their adaptation by osmotic adjustment. New Phytol 94:561–572
- Sibley JL, Ruter JM, Eakesb DJ (1999) Root membrane thermostability of red maple cultivars J. Therm Biol 24:79–89
- Siemens JA, Zwiazek JJ (2004) Changes in root water flow proprieties of solution culture-grown trembling aspen (*Populus tremuloides*) seedling under different intensities of water-deficit stress. Physiol Plant 121:44–49
- Songsri P, Jogloy S, Vorasoot N, Akkasaeng C, Patanothai A, Holbrook CC (2008) Root distribution of droughtresistant peanut genotypes in response to drought. J Agron Crop Sci 194:92–103
- Soussana JF, Casella E, Loiseau P (1996) Long-term effects of CO₂ enrichment and temperature increase on temperate grass sward. II. Plant nitrogen budgets and root fraction. Plant Soil 182:101–114
- Spollen WG, Sharp RE, Saab IN, Wu Y (1993) Regulation of cell expansion in roots and shoots at low water potentials. In: Smith JAC, Griffiths H (eds) Water deficits. Plant responses from cell to community. Bios Scientific, Oxford, pp 37–52
- St.Clair SB, Lynch JP (2010) The opening of Pandora's Box: climate change impacts on soil fertility and crop nutrition in developing countries. Plant Soil 335:101–115
- Storey R, Walker RR (1999) Citrus and salinity. Sci Hortic 78:39–81
- Storey R, Schachtman DP, Thomas MR (2003) Root structure and cellular chloride, sodium and potassium distribution in salinized grapevines. Plant Cell Environ 26:789–800
- Stromgren M, Linder S (2004) Effects of nutrition and soil warming on stemwood production in a boreal Norway spruce stand. Glob Change Biol 8:1194–1204
- Subbaiah CC, Sachs MM (2000) Maize cap1 encodes a novel SERCA-type calcium ATPase with a calmodulinbinding domain. J Biol Chem 275:21678–21687
- Subbaiah CC, Sachs MM (2001) Altered patterns of sucrose synthase phosphorylation and localization precede callose induction and root tip death in anoxic maize seedlings. Plant Physiol 125:585–594
- Sullivan PF, Welker JM (2005) Warming chambers stimulate early growth of an arctic sedge: results of a minirhizotron field study. Oecologia 142:616–626
- Tahir ISA, Nakata N, Yamaguchi T, Nakano J, Ali AM (2008) Influence of high shoot and root-zone temperatures on growth of three wheat genotypes during early vegetative stages. J Agron Crop Sci 194:141–151

- Tavakkoli E, Rengasamy P, McDonald GK (2010) The response of barley to salinity stress differs between hydroponics and soil systems. Funct Plant Biol 37:621–633
- Terceck MT, Hauber DP, Darwin SP (2003) Genetic and historical relationships among the geothermally adapted *Agrostis* (bentgrass) of North America and Kamchatka: evidence for a previously unrecognized, thermally adapted taxon. Am J Bot 90:1306–1312
- Tester M, Davenport R (2003) Na⁺ tolerance and Na⁺ transport in higher plants. Ann Bot 91:503–527
- Thaler P, Pages L (1996) Root apical diameter and root elongation rate of rubber seedlings (*Hevea brasilien*sis) show parallel responses to photoassimilate availability. Physiol Plant 97:365–371
- Thomas AL, Guerreiro SMC, Sodek L (2005) Aerenchyma formation and recovery from hypoxia of the flooded root system of nodulated soybean. Ann Bot 96:1191–1198
- Tierney GL, Fahey TJ, Groffman PM, Hardy JP, Fitzhugh RD, Driscoll CT (2003) Environmental control of fine root dynamics in a norther hardwood forest. Glob Change Biol 9:670–679
- Triboulot MB, Pritchard J, Tomos D (1995) Stimulation and inhibition of pine root growth by osmotic stress. New Phytol 130:169–175
- Trillana N, Inamura T, Chaudhary R, Horie T (2001) Comparison of root system development in two rice cultivars during stress recovery from drought and the plant traits for drought resistance. Plant Prod Sci 4:155–159
- Turka I, Demiral T (2009) Recent developments in understanding salinity tolerance. Environ Exp Bot 67:2–9
- Tyerman SD, Niemietz CM, Bramley H (2002) Plant aquaporins: multifunctional water and solute channels with expanding roles. Plant Cell Environ 25:173–194
- van der Weele CM, Spollen WG, Sharp RE, Baskin TI (2000) Growth of *Arabidopsis thaliana* seedlings under water deficit studied by control of water potential in nutrient-agar media. J Exp Bot 51:1555–1562
- van Dongen JT, Frohlich A, Ramirez-Aguilar SJ, Schauer N, Fernie AR, Erban A, Kopka J, Clark J, Langer A, Geigenberg P (2009) Transcript and metabolite profiling of the adaptive response to mild decreases in oxygen concentration in the roots of *Arabidopsis* plants. Ann Bot 103:269–280
- Vetterlein D, Kuhn K, Schubert S, Jahn R (2004) Consequences of sodium exclusion for the osmotic potential in the rhizosphere - Comparison of two maize cultivars differing in Na⁺ uptake. Journal of Plant Nutrition and Soil Science 167:337–344
- Voetberg GS, Sharp RE (1991) Growth of the maize primary root at low water potentials. III. Role of increased proline deposition in osmotic adjustment. Plant Physiol 96:1125–1130
- Voorhes WB, Allmaras RR, Johnson CE (1981) Alleviating temperature stress. In: Arkin GF, Taylor HM (eds) Modifying the root environment to reduce crop stress. American Society of Agricultural Engineers, St. Joseph, pp 217–266
- Waisel Y, Eshel A (2002) Functional diversity of various constituents of a single root system. In: Waisel Y,

Eshel A, Kafkafi U (eds) Plant roots: the hidden half, 3rd edn. Marcel Dekker, New York, pp 157–174

- Walter A, Spies H, Terjung S, Küsters R, Kirchge
 ßner N, Schurr U (2002) Spatio-temporal dynamics of expansion growth in roots: Automatic quantification of diurnal course and temperature response by digital image sequence processing. J Exp Bot 53:689–698
- Wan C, Xu W, Sosebee RE, Machado S, Archer T (2000) Hydraulic lift in drought-tolerant and -susceptible maize hybrids. Plant Soil 219:117–126
- Wan S, Norby RJ, Pregitzer KS, Ledford J, O'Neil EG (2004) CO₂ enrichment and warming of the atmosphere enhance both productivity and mortality of maple tree fine roots. New Phytol 162:437–446
- Wang Y, Zhang W, Li K, Sun F, Han C, Wang Y, Li X (2008) Salt-induced plasticity of root hair development is caused by ion disequilibrium in *Arabidopsis thaliana*. J Plant Res 121:87–96
- Wang Y, Li K, Li X (2009) Auxin redistribution modulates plastic development of root system architecture under salt stress in *Arabidopsis thaliana*. J Plant Physiol 166:1637–1645
- Wei H-P, Li R-Q (2000) Effect of flooding on morphology, structure and ATPase activity in adventitious root apical cells of maize seedlings. Acta Phytoecol Sin 24:293–297
- Weltzin JF, Pastor J, Harth C, Bridgham SD, Updegraff K, Chapin CT (2000) Response of bog and fern plant communities to warming and water table manipulations. Ecology 81:3464–3478
- Whitmore AP, Whalley WR (2009) Physical effects of soil drying on roots and crop growth. J Exp Bot 60:2845–2857
- Wu Y, Sharp RE, Durachko DM, Cosgrove DJ (1996) Growth maintenance of the maize primary root at low water potentials involves increases in cellwall extension properties, expansin activity, and wall susceptibility to expansins. Plant Physiol 111:765–772
- Wu Y, Thorne ET, Sharp RE, Cosgrove DJ (2001) Modification of expansin transcript levels in the maize primary root at low water potentials. Plant Physiol 126:1471–1479
- Yamada T, Suzuki E, Yamakura T, Tan S (2005) Taproot depth of tropical seedlings in relation to speciesspecific edaphic preferences. J Trop Ecol 21:155–160
- Yamaguchi M, Sharp RE (2010) Complexity and coordination of root growth at lower water potential: recent advances from transcriptomic and proteomic analyses. Plant Cell Environ 33:590–603
- Zaharan HH (1999) Rhizobium–legume symbiosis and nitrogen fixation under severe conditions and in an arid climate. Microbiol Molecul Biol Rev 63: 968–989
- Zaidi PH, Rafique S, Rai PK, Singh NN, Srinivasan G (2004) Tolerance to excess moisture in maize (*Zea mays* L.): susceptible crop stages and identification of tolerant genotypes. Field Crops Res 90:189–202

9

Reactive Oxygen Species and Nitric Oxide in Plants Under Cadmium Stress: From Toxicity to Signaling

Luisa M. Sandalio, Maria Rodríguez-Serrano, Dharmendra K. Gupta, Angustias Archilla, Maria C. Romero-Puertas, and Luis A. del Río

Abstract

The toxicity of heavy metals as a result of increasing environmental pollution in living organisms has become a major focus of research in recent decades. Among the heavy metals cadmium is one of the most dangerous heavy metals because of its high mobility in plants. It causes severe disturbances in plant metabolism that affect photosynthesis and water/ nutrient balance, and it also causes oxidative damage. Although there is an enormous literature on the tolerance and accumulation of cadmium in plants, very little research has been performed on the molecular mechanisms and signaling events underlying plant responses to Cd toxicity. The dual role as both oxidative damage inducers and signaling molecules of ROS and NO in heavy metal toxicity has been demonstrated by many workers. In this chapter, we review the contribution of different ROS and NO sources in cells and their role in regulating cellular responses to Cd.

Keywords

Cadmium stress • ROS • NO • Photosynthesis • Metal transporters • GSH metabolism • Signaling

1 Introduction

Heavy metals such as Cd, Hg, Pb, and Al are major environmental pollutants, particularly in industrial areas. The heavy metals are generated as a result of anthropogenic activities such as metal working industries, cement factories, smelting plants, refineries, traffic, and heating systems (Sanitá di Toppi and Gabbrielli 1999). Much of the arable soil around the world has been moderately contaminated by Cd through the

L.M. Sandalio (🖂) • M. Rodríguez-Serrano

[•] D.K. Gupta • A. Archilla • M.C. Romero-Puertas • L.A. del Río

Department of Biochemistry and Molecular and Cellular Biology of Plants, Estación Experimental del Zaidín, Consejo Superior de Investigaciones Cientificas (CSIC), Mail box 419, E-18080 Granada, Spain e-mail: luisamaria.sandalio@eez.csic.es

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_9,

[©] Springer Science+Business Media, LLC 2012

use of phosphate fertilizers, sludge, and irrigation water (Sanitá di Toppi and Gabbrielli 1999). In polluted soils, Cd is generally present as a free ion or in other soluble forms, and its mobility depends on pH as well as the presence of chelating substances and other cations. Cadmium has a toxic impact on all living organisms by entering the food chain and accumulating by humans and animals (Nordberg 2004). Prolonged exposure to Cd by humans can cause renal dysfunction, lung damage, acute gastrointestinal problems, depression of immune system, increased cancer risk, and anemia (Nordberg 2004). The accumulation of cadmium in plants causes chlorosis, growth reduction, and even cell death. The cellular toxicity of this metal results from its various direct and indirect effects on cell metabolism and can be explained by its chemical characteristics. Cd can bind to SH groups of proteins and enzymes, leading to misfolding, enzyme inhibition, and interferences in redox regulation. Cadmium can also displace other cations from proteins and enzymes, which affects their functioning (Van Assche and Clijsters 1990). Like most heavy metals, cadmium induces oxidative stress by generating reactive oxygen species which causes oxidative damage to biomolecules such as membrane lipids, proteins, nucleic acids, etc. (Sandalio et al. 2009).

Cadmium is one of the most dangerous heavy metals in nature, and at low concentrations it adversely affects the plant growth and development. Strong evidence have shown that Cd-induced generation of reactive oxygen species plays an important role in cellular toxicity, and the effects and species-dependent produced are dose-(Benavides et al. 2005; Sandalio et al. 2009). However, ROS are double-faced molecules acting as signal molecules that regulate a large gene network involved in cell response to biotic and abiotic stress. Nitric oxide (NO) is a gaseous reactive molecule with a pivotal signaling role in many developmental and cell response processes (Besson-Bard et al. 2008). This molecule can also interfere with the ROS metabolism. There are many reports that showed the role of NO in the alleviation of the toxicity caused by heavy metals including Cd and As (Xiong et al. 2010). Several defense strategies to avoid metal toxicity have been developed by plants, which include preventing the entry of metal through exudation of metal-complexing agents (citrates and phytosiderophores) by roots and metal immobilizing pectic sites and histidinyl groups in the cell wall (Sanitá di Toppi and Gabbrielli 1999; Clemens 2006). A second line of defense involves the induction of specific peptides called phytochelatins (PCs) which chelate the metal. PC--Cd complexes are transported into the vacuole to protect cells from toxicity (Cobett 2000). The isolation of an Arabidopsis cad1 mutant, which is defective in PC activity and hypersensitive to Cd, has demonstrated the importance of this mechanism in defending plants against Cd (Howden et al. 1995). Cd and other metals can also be complexed by metallothioneins and nicotianamine (Sharma and Dietz 2006).

2 Cadmium Toxicity in Plants

The toxic effects of cadmium on several plant species have been reported by different authors (Sanitá di Toppi and Gabbrielli 1999; Sandalio et al. 2001; Schützendübel et al. 2001; Benavides et al. 2005), although the mechanisms involved in cadmium toxicity are still not fully understood. Cadmium inhibits seed germination, decreases plant growth, induces premature senescence, and can even trigger cell death in cell suspension cultures (Fotjová and Kovařik 2000; McCarthy et al. 2001; Rodríguez-Serrano et al. 2009; De Michele et al. 2009). At cellular levels, Cd produces alterations in membrane functionality by inducing changes in lipid composition and by promoting lipid peroxidation (Ouariti et al. 1997; Hernández and Cooke 1997; Sandalio et al. 2001); it also produces disturbances in photosynthesis by affecting CO₂ fixation and by inhibiting PSII photoactivation because of competition with essential Ca²⁺ sites (Faller et al. 2005; Baryla et al. 2001). Cadmium toxicity is associated with modifications in both the uptake and distribution of macro- and micronutrients (Hernández et al. 1998; Rogers et al. 2000; Sandalio et al. 2001; Tsyganov et al. 2007) and can therefore compete with other cations for protein- and transporter-binding sites (Clemens 2006). Cd uptake occurs through plasma membrane transporters similar to those used for other cations such as K⁺, Ca²⁺, Mg²⁺, Fe²⁺, Mn²⁺ or Cu²⁺ (Clemens 2006). Cadmium reduces Ca²⁺ content, which can then affect the activity of calmodulindependent proteins (Rivetta et al. 1997; Rodríguez-Serrano et al. 2009). Cd tolerance and Ca²⁺ homeostasis in a Cd-resistant pea mutant (SGECdt) have been observed to be interrelated (Tsyganov et al. 2007), while in radish and Arabidopsis seedlings, calcium has been reported to alleviate Cd toxicity by reducing Cd uptake (Rivetta et al. 1997; Suzuki 2005). The role of oxidative stress in Cd toxicity has been established in different plant species by analyzing oxidative damage to proteins and lipids as well as by studying disturbances in antioxidative defenses caused by this metal (Benavides et al. 2005; Sandalio et al. 2009; Remans et al. 2010). Although Cd is a bivalent cation unable to participate in redox reactions in the cell, most transcriptome studies show upregulation of genes encoding proteins involved in defense against oxidative stress and ROS production (Suzuki et al. 2001; Zhao et al. 2009). These results suggest that oxidative stress is one of the primary effects of Cd exposure. Reactions involving oxygen free radicals are an intrinsic feature of plant senescence and stimulate the process of oxidative deterioration that leads to cell death (del Río et al. 2009). Cd induces senescence and cell death in both cell culture and plant tissues characterized by the induction of the glyoxylate cycle, protein oxidation, and proteolytic activities (McCarthy et al. 2001; Romero-Puertas et al. 2002). Senescence is considered to be a type of plant programmed cell death (PCD), and various studies have indicated that Cd induces PCD in cell cultures (Fotjová and Kovařik 2000; De Michele et al. 2009). There are evidence to support the possibility of dose dependence and intensity of the onset of the senescence process and the final cell death event (De Michele et al. 2009). Condensation of chromatin, fragmentation of DNA, as visualized by TUNEL assay, and induction of SAG12 expression are some of the symptoms of PCD and have been observed in

Arabidopsis and tobacco cell cultures exposed to Cd (Fotjová and Kovařik 2000; De Michele et al. 2009). Cadmium-dependent senescence and PCD are regulated by ROS and NO, although the mechanisms involved are not fully understood (Yakimova et al. 2006; De Michele et al. 2009; Rodríguez-Serrano et al. 2009). Lipid signaling and Ca²⁺ also play an important role in Cd-induced cell death (Yakimova et al. 2006).

3 Sources of ROS in Plants Exposed to Cadmium

The reactive oxygen species are mainly singlet oxygen ($^{1}O_{2}$), superoxide radical ($O_{2}\cdot^{-}$), hydroxyl radical (\cdot OH) and hydrogen peroxide (H₂O₂) which are by-products of normal aerobic metabolism such as respiration and photosynthesis. Their steady-state levels are determined by the interplay of different ROS-producing and ROSscavenging mechanisms. This balance maintained by enzymes such as superoxide dismutase (SOD), which remove O_2^- radicals, and catalase (CAT), peroxidase (POX), and peroxiredoxin, which decompose H_2O_2 and use metabolites such as glutathione (GSH) and ascorbate (ASC), to control ROS accumulation in different subcellular compartments. An excess of ROS is dangerous mainly because of reactions with lipids, proteins, and nucleic acids, giving rise to lipid peroxidation, membrane leakage, enzyme inactivation and DNA breaks or mutations, which can cause severe damage to cell viability. Subtle control of ROS production enables these species to act as signaling molecules which are involved in the regulation of processes such as mitosis, tropism, cell death and cell response to biotic and abiotic stresses. Compared with other ROS, H₂O₂ is a relatively long-lived molecule that is able to diffuse across cell membranes and acts as a signaling molecule during growth and development (Van Breusegem and Dat 2006). However, although we have a clear understanding of the toxic effects of ROS induced by metals as well as detoxification mechanisms, information on their role in regulation and signal transduction under metal stress remains quite limited.

The electron transfer chains associated with chloroplasts and mitochondria are the main sources of ROS generation. However, this view has changed and the oxidative metabolism of peroxisomes is now seen as a very important source of ROS under different stress conditions (del Río et al. 2006, 2009). In peroxisomes purified from pea leaves, a Cd-dependent increase in the H₂O₂ concentration was observed, mainly as a result of the activation of glycolate oxidase, a key enzyme in the photorespiration cycle (Romero-Puertas et al. 1999). In pea leaves, it has been demonstrated through the use of a cytochemical approach that Cd-dependent H₂O₂ production occurs in peroxisomes, in the outer mitochondrial membrane, and mainly in the plasma membrane, where the NADPH oxidase (NOX) is the main source of ROS (Romero-Puertas et al. 2004). In peroxisomes, H₂O₂ was located in close contact with other organelles, which suggests possible cross-talk with other cell compartments (Romero-Puertas et al. 2004). In mitochondria, the Cd-dependent H₂O₂ produced could be because of increased O_2^- production at the complex III site of the electron transport chain, as reported in animals treated with Cd (Wang et al. 2004) and also suggested for soybean roots (Heyno et al. 2008). H₂O₂ was also observed in the tonoplast from bundle sheet cells and plasma membrane from epidermal and transfer cells (Romero-Puertas et al. 2004). Cd-dependent superoxide radical accumulation was demonstrated in the tonoplast from bundle sheet cells and plasma membrane from mesophyll cells, although the source has not been identified (Romero-Puertas et al. 2004). Accumulation of both H_2O_2 and $O_2^$ was also observed in vascular tissues from Cd-treated pea plants using confocal laser microscopy, electron microscopy, and cytochemistry (Romero-Puertas et al. 2004; Rodríguez-Serrano et al. 2006, 2009). This ROS accumulation is associated with lignifications processes which are highly active in vascular tissue under physiological conditions and are also induced in response to metal toxicity (Schützendübel et al. 2001; Rodríguez-Serrano et al. 2009). Results using different inhibitors and modulators of signal transduction demonstrated that the earliest

control point in ROS production induced by Cd is at the level of protein phosphorylation/dephosphorylation. A comparative transcriptomic study using different metals and sodium chloride in *Arabidopsis thaliana* showed that Cd specifically induced genes coding kinases (Zhao et al. 2009), which demonstrates the importance of these processes in regulating cell response to Cd. Calcium ions are also important in the regulation of ROS production induced by Cd, and cGMP is also involved in this process, probably as result of a transient increase in Ca²⁺ concentration (Romero-Puertas et al. 2004).

NOXs are located in the plasma membrane and catalyze the production of O_2^- , which can be converted into H₂O₂, spontaneously or in the reaction catalyzed by SOD. Ten genes encoding NOXs in Arabidopsis have been described and are termed respiratory burst oxidase homologs A–J (rbohA–J) given their homology with the catalytic subunit gp91 phox (Nox2) of the NOX complex of mammalian phagocytes (Torres and Dangl 2005). The role of NOX as the main source of ROS under Cd stress has also been demonstrated in tobacco cell cultures (Olmos et al. 2003; Garnier et al. 2006; Horemans et al. 2007) and alfalfa roots (Ortega-Villasante et al. 2005). In Arabidopsis plants, the analysis of transcript levels of different NOXs shows a transient increase in the expression of rbohF in response to Cd, while the expression of rbohC and rbohD remained unchanged (Horemans et al. 2007). However, the contribution of NOXs to cadmiuminduced ROS production is a subject of debate (Heyno et al. 2008). In tobacco cell cultures, Cd induced cell death, which was preceded by three successive waves of ROS production. The first wave was because of an NOX followed by an accumulation of O₂⁻ and fatty acid hydroperoxides (Garnier et al. 2006). Before the first oxidative burst induced by Cd, a rapid and transient induction of cytosolic Ca²⁺ concentration takes place, which requires protein phosphorylation and IP3mediated release of calcium from internal stores (Garnier et al. 2006). Downstream, protein phosphorylation, calmodulin, and Ca2+ may directly regulate NtrbohD activity (Garnier et al. 2006). The disturbances caused by Cd in the mitochondrial


Fig. 9.1 Sources of reactive oxygen species activated in response to cadmium. Cd-dependent ROS production takes place in different compartments: plasma membrane-associated NADPH oxidase, electron transport chain from

mitochondria and peroxisomes (mainly glycolate oxidase). Overaccumulation of H_2O_2 produces redox imbalance and oxidative damage, but also can regulate gene expression in order to improve plant survival

electron transport chain stimulate a second wave of ROS production possibly because of an increase in the semi-ubiquinone radical concentration (Garnier et al. 2006). A third wave of ROS coincides with cell death and involved membrane peroxidation as a result of increase in ROS production caused by mitochondria (Garnier et al. 2006). Remans et al. (2010) have recently demonstrated a Cd-dependent induction of NOX and differential regulation of gene expression by Cd and Cu in Arabidopsis plants and have suggested a link between NOX and lipoxygenase gene expression. A diagram showing the different subcellular locations of ROS production is provided in Fig. 9.1.

The inhibition of antioxidative enzymes may also lead to a cadmium-mediated increase in the level of cellular ROS (Sandalio et al. 2001, 2009; Romero-Puertas et al. 2002; Schützendübel and Polle 2002; Benavides et al. 2005). One of the consequences of plant cell exposure to cadmium is the rapid consumption of GSH for sequestration of the metal and synthesis of PCs. This limits the GSH level required to maintain the redox balance of the cell which then increases ROS accumulation (Romero-Puertas et al. 2007a, b).

4 NO Production in Plants Exposed to Cadmium

Nitric oxide is a simple gaseous signaling molecule which, in many plant tissues, regulates a wide range of physiological and biochemical processes as well as plant responses to biotic and abiotic stresses (del Río 2011; Delledonne 2005; Siddiqui et al. 2010). An increasing number of studies have reported the role played by NO in plant response to heavy metals including cadmium, although the source of NO and its role in metal toxicity and plant responses are not yet clearly established (Xiong et al. 2010). NO can be generated enzymatically by nitrate reductase and nitric oxide synthase (NOS)-like activities and can also be produced nonenzymatically by reduction of apoplastic nitrite under acid conditions and by reduction of nitrite to NO in the mitochondria (Neill et al. 2008; del Río 2011). There is still some uncertainty concerning NOS in plants. Although there is strong evidence to show the presence of L-arginine-dependent NOS activity in plants (Barroso et al. 1999; del Río 2011), the only NOS from the plant kingdom to be fully characterized so far is the enzyme from the Ostreococcus tauri green alga (Foresi et al. 2010). NOS activity has been shown to be present in peroxisomes from pea leaves (Barroso et al. 1999). This enzyme uses L-arginine as substrate and requires NADPH, Ca²⁺/calmodulin, BH, FAD, and FMN, although its gene has not yet been characterized (Corpas et al. 2004; del Río 2011). The generation of NO in peroxisomes by this NOS activity has also been reported by Corpas et al. (2004). In addition, chloroplasts have recently been identified as a source of NO via arginine and nitrite, although the enzyme involved has not been characterized yet (Jasid et al. 2006; del Río 2011). A nitrite-NO oxidoreductase enzyme (Ni-NOR) associated with root plasma membrane may also contribute to NO production (Stöhr and Stremlau 2006). However, there are other potential enzymatic sources of NO in plants (del Río et al. 2004; del Río 2011) such as xanthine oxidase which can produce NO under hypoxic conditions (Millar et al. 1998; Harrison 2002). Regardless of the source of NO involved, the mechanisms determining the effects of NO are far from being fully understood, while a number of downstream signaling pathways involving Ca²⁺, cyclic GMP, and cyclic ADP-Rib have been described (Besson-Bard et al. 2008). NO is able to react with oxygen radicals such as O_2^{-} , generating peroxynitrite (ONOO⁻), and also to control ROS levels in cells and vice versa (Delledonne et al. 2001). NO can also react with GSH to produce S-nitrosoglutathione (GSNO), which is regarded as a long-distance-signaling molecule and a natural reservoir of NO (del Río 2011). NO directly or indirectly can regulate gene expression and protein functions. It therefore reacts very rapidly with heme groups and thiols, thus regulating enzymatic activities (Moreau et al. 2010). The protein S-nitrosylation of cystein residues has been demonstrated to be very important in regulating the enzymatic activity of certain proteins (Lindermayr et al. 2006; Lindermayr and Durner 2009; Romero-Puertas et al. 2007b, 2008). Some studies of NO production during the exposure of plants to heavy metals have reached controversial conclusions. The cell cultures of soybean (Kopyra et al. 2006) and Arabidopsis (De Michele et al. 2009) exposed to Cd showed an increase in NO and was dependent on NOS-like activity (De Michele et al. 2009), whereas, in pea leaves and roots, prolonged exposure to 50 µM Cd reduced NO accumulation (Rodríguez-Serrano et al. 2006, 2009). Bartha et al. (2005) reported increased NO in the roots of *Brassica juncea* and *Pisum* sativum exposed to 100 µM Cd, Cu, and Zn. Besson-Bard et al. (2009) have shown that NO production in Cd-treated roots is related to Cd-induced Fe deficiency. The discrepancies observed in these results could be because of differences in Cd exposure duration, with NO increasing after a short period of Cd treatment and decreasing after a prolonged treatment (Fig 9.2). The metal concentrations, plant ages, and plant tissues used could also explain these discrepancies (Rodríguez-Serrano et al. 2009). Exogenously supplied NO has been demonstrated to alleviate heavy metal toxicity (Kopyra and Gwóždž 2003; Hsu and Kao 2004; Yu et al. 2005; Wang and Yang 2005; Laspina et al. 2005; Xiong et al. 2010) possibly because of its ability to act as an ROS-scavenging antioxidant such as SOD and CAT (Wang and Yang 2005; Rodríguez-Serrano et al. 2006; Singh et al. 2008; Siddiqui et al. 2010). Exogenous NO application also affects root cell walls and helps in metal accumulation. NO cause increases in cytosolic Ca²⁺ concentrations by regulating Ca²⁺ channels and transporters, which may be involved in the signaling cascade that regulates gene expression under stress conditions (Besson-Bard et al. 2008). Recently, cross-talk between Cd, Ca^{2+} , ROS, and NO has been detected in pea leaves (Rodríguez-Serrano et al. 2009). The supply of exogenous Ca2+ to pea plants exposed to Cd reduced Cd-dependent O₂⁻ accumulation and restored NO accumulation to the level observed in control plants (Rodríguez-Serrano et al. 2009).



Fig. 9.2 Cadmium induces differential response in plants depending on the period of treatment. A short period of treatment produces oxidative and NO burst, which induces gene expression to prevent oxidative damages caused by the metal. Long-term treatment produces overaccumulation of ROS and a reduction of NO giving rise to severe damages. Gene regulation in long-term treatment is focused on repairing oxidative damages and cell death. *PCs* phytochelatins, *HSPs* heat shock proteins

5 Plant Responses to Cadmium

Information on molecular mechanisms and signaling events underlying plant transcriptional responses to Cd is rather limited as compared to research in the field of cadmium toxicity. The mechanism by which Cd modulates the levels of expression of most genes is not clearly understood, while our knowledge of global changes in the expression of Cd-responsive genes is also limited. A number of studies have been carried out involving both small-scale experiments and whole-genome approaches. Their findings suggest that gene expression is time- rather than dose-regulated in response to Cd and is differentially regulated in roots and leaves (Herbette et al. 2006; Ogawa et al. 2009). Genes regulated by cadmium can be categorized into different protein groups in terms of photosynthetic processes, signal transduction, and transcriptional regulation, cellular defenses, ROS detoxification and repair, hydric balance, metal transport, cell wall metabolism, sulfate and GSH metabolism and protein degradation.

5.1 Photosynthesis Regulation

The decrease in chlorophyll content has been considered to be one of the early symptoms of cadmium toxicity. The inhibition of chlorophyll biosynthesis has been suggested to be a primary event in Cd toxicity (Baryla et al. 2001). A substantial number of genes involved in photosynthesis were downregulated in the leaves of Arabidopsis plants grown with 5–50 µM Cd. For example, the genes involved in the photochemical process of photosynthesis, such as the chlorophyll synthesis pathway, glutamyl tRNA reductase, hydroxymethylbilane synthase, and Mg chelatase, some proteins of PSI and PSII, electron transporters, enzymes involved in Calvin cycle and Rubisco are downregulated (Herbette et al. 2006). Genes encoding enzymes in the pentose phosphate pathway were also downregulated by Cd (Herbette et al. 2006). These results have been corroborated by proteomic approaches (Alvarez et al. 2009) and correlate with the reduction observed in the photosynthesis net rates for different plant species (Sandalio et al. 2001; Faller et al. 2005). Downregulation of photosynthesis-related genes is a primary response under different stress conditions probably to avoid oxidative damage (Mittler 2002).

5.2 Signal Transduction and Transcriptional Regulation

Numerous genes involved in signal transduction were regulated in response to Cd in different plant species showing that signal transduction pathways are rapidly activated by the presence of Cd (Suzuki et al. 2001; Herbette et al. 2006; Ogawa et al. 2009). These include genes encoding mitogen-activated protein kinases (MAPKs), calmodulins and calcium-dependent protein kinases (CDPKs) (Suzuki et al. 2001; Herbette et al. 2006; Ogawa et al. 2009), which suggests that Cd interferes with the Ca²⁺ signaling pathway, as demonstrated by Rodríguez-Serrano et al. (2009). MAPKs and CDPKs are involved in biotic and abiotic stress responses and participate in cross-talk with ROS production activities (Kobayashi et al. 2007). Transcription factors belonging to different families, such as WRKY, bZip, MYB, DREB, NAC, and AP2, are induced by Cd in different plant species (Herbette et al. 2006; Weber et al. 2006; Ogawa et al. 2009). The inductions by Cd of transcripts for bZIP, MYB, and zinc finger transcriptional factors have also been demonstrated in the root of the metal accumulator B. juncea (Fusco et al. 2005). Genes involved in hormone signaling, mainly ABA and ethylene and jasmonic acid, have also been shown to be regulated in response to Cd (Herbette et al. 2006; Minglin et al. 2005).

5.3 Cellular Detoxification and Repair

Several genes associated with cellular detoxification and repair have been shown to be induced by treatment with cadmium. Chitinases and heat shock proteins (HSPs) are induced in response to heavy-metal stress and are regarded as a second line of defense under these stress conditions (Metwally et al. 2003; Békésiová et al. 2008; Rodríguez-Serrano et al. 2009; Zhao et al. 2009). Transgenic plants expressing fungal chitinases actually showed enhanced tolerance to metals (Dana et al. 2006), while chitinase isoforms are differentially modified by certain metals (Békésiová et al. 2008). Chitinases are regulated by ROS and are possibly part of the general defense response program of cells under heavymetal stress (Békésiová et al. 2008; Rodríguez-Serrano et al. 2009). Other pathogenesis-related proteins (PRPs) are upregulated by Cd (Fusco et al. 2005; Rodríguez-Serrano et al. 2009). ROSdependent up-regulation of PRP4A has been demonstrated in pea plants exposed to Cd, whose transcripts were specifically accumulated in palisade mesophyll cells, as evidenced by in situ hybridization (Rodríguez-Serrano et al. 2009). These results point to an overlap in the regulatory mechanisms underlying these processes, with ROS production being a common event in these situations.

HSPs are upregulated by heat stress and can act as molecular chaperones favoring the transport of proteins to organelles and preventing protein aggregation (Ma et al. 2006). Induction of HSPs by Cd has been observed in different plant species (Sanitá di Toppi and Gabbrielli 1999; Rodríguez-Serrano et al. 2009) and is regulated by H₂O₂ overproduction (Rodríguez-Serrano et al. 2009). The transcription factors involved in HSP expression can act as H_2O_2 sensors (Miller and Mittler 2006). In B. juncea, Cd upregulates a DNAJ HSP (BjCdR57), a chaperone involved in protein protection against stress, which confirms that protein denaturation is one of the effects of Cd toxicity (Suzuki et al. 2001; Fusco et al. 2005). GSH S-transferases catalyze the conjugation of xenobiotics with GSH and participate in the removal of ROS and are upregulated in response to Cd (Suzuki et al. 2002; Fusco et al. 2005; Ogawa et al. 2009). Antioxidative defenses such as glutaredoxin, thioredoxin, GSH reductase, monodehydroascorbate reductase, SOD, CAT, and POXs are upregulated by Cd in order to deal with oxidative damage caused by this metal (Lemaire et al. 1999; Herbette et al. 2006; Romero-Puertas et al. 2007a, b; Smeets et al. 2005; Ogawa et al. 2009). Enzymes involved in vitamin E biosynthesis are upregulated in response to Cu and Cd in Arabidopsis plants, while vitamin E-deficient mutants (vte1) showed enhanced oxidative stress and sensitivity to both metals, suggesting that Vitamin E also contributes to defense against heavy metals (Collin et al. 2008). The regulation of these antioxidative enzymes is mainly dependent on H₂O₂ (Romero-Puertas et al. 2007a, b; Rodríguez-Serrano et al. 2009), although GSH metabolism also plays an important role in controlling the gene regulation of antioxidants in response to Cd stress (Cuypers et al. 2011). The activity of glucose-6-P dehydrogenase (G6PDH), malic enzyme (ME), and NADP isocitrate dehydrogenase (NADP-ICDH) is stimulated by Ni, Zn, and Cd (Van Assche and Clijsters 1990; León et al. 2002), while, in pepper cultivars with different levels of sensitivity to Cd, tolerance to this heavy metal was more dependent on the availability of NADPH than on its antioxidant capacity (León et al. 2002).

5.4 Metal Transporters

Some of the genes regulated by Cd, such as AtPcr1 (Song et al. 2004) and those belonging to the ABC, MATE, cation diffusion facilitator (CDF), heavy metal P-type ATPase (HMA) and ZIP families, are involved in Cd transport (Ogawa et al. 2009). Fe and Zn transporters are also often involved in Cd transport because of their low substrate specificity. The iron transporters ZIP, AtIRT1, OsIRT1111, and OsIRT2 as well as the Zn transporter OsZIP1 have been shown to transport Cd. The HMA family is also involved in Cd detoxification in addition to CDF transporters and natural resistance-associated macrophage protein (NRAMP) family transporters (Ogawa et al. 2009). Pleiotropic drug resistance (PDR) family proteins are involved in Cd tolerance via export out of the cytoplasm (Kim et al. 2007). AtPDR8 is a cadmium extrusion pump, while AtOSA1 could be involved in the signal transduction pathway in response to oxidative stress (Kim et al. 2007; Jasinski et al. 2008). Cd-binding proteins such as Cdl19 could be involved in maintaining heavy-metal homeostasis and/or detoxification (Suzuki et al. 2002).

5.5 Cell Wall Metabolism

The cell wall is one of the first structures to be directly exposed to Cd and has the ability to bind metals, which is regarded as a mechanism of metal tolerance. Most of the heavy metals associated with the cell wall are linked to polygalacturonic acids, whose metal ion affinities vary depending on the metal in question. The plant cell wall is mainly composed of cellulose and matrix polysaccharides, which are divided into pectins and hemicelluloses, both of which are rich in polygalacturonic acids (Xiong et al. 2010). Cellulose is a key component in plant cell walls, and it has been reported that NO affects the cellulose content of tomato roots in a dose-dependent manner. Low concentrations of sodium nitroprusside (SNP) increase cellulose content in roots, while higher concentrations have the opposite effect (Correa-Aragunde et al. 2008). Exogenous NO increases Cd tolerance in rice plants by increasing pectin and hemicelluloses content in the root cell wall and by decreasing Cd accumulation in the soluble fraction of cells in rice leaves (Xiong et al. 2009). H₂O₂ may also trigger secondary defenses, causing cell wall rigidification and lignifications in Cd-exposed cells (Schützendübel and Polle 2002). The transcript levels of genes involved in cell wall metabolism are modulated in response to Cd. The proteins involved in lignification and extension were therefore upregulated (Fusco et al. 2005; Herbette et al. 2006), whereas expansins and pectin esterases were downregulated (Herbette et al. 2006).

5.6 Sulfate and GSH Metabolism

One of the best described mechanisms induced under heavy-metal toxicity is the chelation of the metal by PCs and GSH. PCs have the general formula (γ Glu-Cys)_n-Gly (with n=2-11) and are synthesized enzymatically through the transpeptidation of γ Glu-Cys moieties of GSH onto another GSH molecule by the phytochelatin synthase (PCS) enzyme, which is known to be activated posttranslationally by a range of heavy metal metalloids (Grill et al. 2006). Chelation of metals by PCs and the compartmentalization of PC-metal complexes in vacuoles (Clemens 2006; Grill et al. 2006) are generally considered as firstline defense mechanisms.

The rate-limiting step for PCs and GSH biosynthesis is the availability of reduced sulfur to the roots. Various genes involved in the sulfate metabolism are induced in response to Cd, which include sulfate transporters from roots (Sultr1; 1; Sultr1; 2), enzymes involved in sulfate reduction to sulfide (ATP sulfurylase), and those involved in PC biosynthesis (PC synthases) (Herbette et al. 2006; Ramos et al. 2007). One of the steps in PC biosynthesis is the synthesis of cysteine catalyzed by *O*-acetylserine(thiol)lyase (OASTL) which is upregulated by Cd (Fusco et al. 2005). Arabidopsis plants over expressing OASTL were highly Cd resistant, which suggests that cysteine pool required for GSH biosynthesis is one of the principal factors affecting Cd tolerance (Domínguez-Solis et al. 2001). A deficiency in the major OASTL isoform in the cytosol from Arabidopsis plants, OAS-A1, causes αH_2O_2 homeostasis imbalance (López-Martín et al. 2008).

5.7 Hydric Balance

The plant-water balance is also disturbed by Cd, and the stomatal opening is inhibited (Poschenrieder et al. 1989; Sandalio et al. 2001; Perfus-Barbeoch et al. 2002). Sequence analysis of Cd-responsive genes in the metal accumulator B. juncea revealed the induction of genes encoding aquaporins, which facilitates the movement of water through cellular membranes. In addition, other drought and ABA-responsive genes, such as BjCdR39 (the aldehyde dehydrogenase) and BjCdR55, (RNA-binding protein), are also upregulated by Cd, which confirms the existence of cross-talk between Cd-induced and water stressinduced signaling using ABA as a signal transducer. Stomatal closure, a symptom of water stress mediated by ABA, is one of the principal responses of higher plants to Cd (Sanitá di Toppi and Gabbrielli 1999).

5.8 Protein Degradation

Oxidative damage to proteins has been observed in different plant species exposed to Cd and is regarded as an oxidative stress marker (Sandalio et al. 2001; Pena et al. 2007; Djebali et al. 2008; Paradiso et al. 2008). Some of the proteins undergoing oxidative modification have been identified in pea leaves and include CAT, GR, Rubisco, and Mn-SOD (Romero-Puertas et al. 2004). Increased proteolytic activity in leaves following Cd treatment and more efficient degradation of the oxidized proteins have been observed (McCarthy et al. 2001; Romero-Puertas et al. 2004). Similar results have been reported by Pena et al. (2006) in Helianthus annus and by Djebali et al. (2008) in Solanum lycopersicum. A proteomic study of A. thaliana cells has also reported an increase in several proteases after Cd treatment (Sarry et al. 2006). Cd treatment has also been shown to increase polyubiquitinated protein accumulation (Pena et al. 2007). The proteasome-ubiquitin system is the major proteolytic pathway in eukaryotes and is also involved in the depredation of oxidized proteins (Pena et al. 2007). The plant proteasome was upregulated at transcriptional and translational levels under oxidative conditions caused by cadmium stress (Pena et al. 2006, 2007; Djebali et al. 2008; Polge et al. 2009). Using in vivo experiments with A. thaliana mutants, it has been demonstrated that 20S proteasomes are preferentially involved in the degradation of oxidized proteins (Kurepa et al. 2008). The remobilization of oxidized proteins may be a protective mechanism under stress conditions to prevent further damage to other macromolecules and to facilitate the recycling of amino acids for protein biosynthesis.

6 Signal Transduction Under Cadmium Stress

The response to heavy metals depends on a complex signal transduction pathway within the cell which begins with the sensing of heavy metal and converges in transcription regulation of metal-responsive genes (Sing et al. 2002), although much remains to be learned about the molecular components of metal-induced signal transduction. Various transcription factors (TFs) involved in the regulation of cell response to metal stress have recently been identified (Sect. 21.5.2). The modulation of different groups of TFs highlights the complex response of plants to Cd (DalCorso et al. 2008). ROS and NO are important players in the regulation of plant response from signal perception to the intracellular transduction cascade, triggering the activation of genes involved in the induction of different metabolic pathways to deal with Cd toxicity. Hydrogen peroxide governs the transduction of cellular response in different abiotic stresses including those caused by heavy metals. The transduction of H₂O₂ signals into biologically relevant information is coordinated by a complex network of sensors and receptors, such as MAPKs, and transcription factors and is thought to be evolutionarily conserved (Vandenbroucke et al. 2008), although there are around 400 H₂O₂-responsive protein families in A. thaliana and may vary depending on the plant species in question (Vandenbroucke et al. 2008). There are several elements in the signal transduction pathway of ROS-sensitive plants which include the MAPK, MAPKK, MAPKKK, AtMPK3/6, AtANP1, NtNPK1, Ntp46MAPK, and calmodulin (Mittler 2002; Vanderauwera et al. 2009). The increase in H_2O_2 levels induced by Cd can be perceived by oxidative protein modifications. The protein thiol groups tyrosine, tryptophan, and histidine can be oxidised by H_2O_2 and O_2^{-1} . The redox changes in the Cys residues of transcription factors directly regulate nuclear gene expression. However, transcriptional modifications may also require additional upstream sensing and transduction of ROS and ROSderived signals, being involved MAPKs and several protein phosphatases (Vanderauwera et al. 2009). Salicylic and jasmonic acid as well as ethylene can also participate in signal transduction under Cd stress (Rodríguez-Serrano et al. 2009; Ogawa et al. 2009). Salicylic acid (SA) acts as an important signaling element in plants and has been observed to alleviate Cd-induced growth inhibition and oxidative damage (Metwally et al. 2003). Although this mechanism is not fully understood, it has been suggested that SA may induce H₂O₂ signals involved in Cd tolerance, such as repair processes, Cd binding and compartmentation (Metwally et al. 2003). The Cd-induced ethylene biosynthesis has been reported to occur in various plant species (Sanitá di Toppi and Gabbrielli 1999; Rodríguez-Serrano et al. 2006), although the molecular relationships between ethylene biosynthesis and Cd stress have yet to be clearly determined. Transcriptomic studies of Arabidopsis plants have detected Cd-dependent up-regulation of ACC oxidase and ACC synthase as well as the ethylene responsive factors ERF2 and ERF5 (Herbette et al. 2006). JA content increases in response to heavy metals in various plant species (Wang and Wu 2005; Rodríguez-Serrano et al. 2006, 2009). JA regulates genes involved in GSH and PCS in Arabidopsis plants under Cd treatment (Xiang and Oliver 1998). In different plant species, Ca²⁺, calmodulin, CDPK and an MAPK act as signaling molecules which regulates cell response to cadmium stress (Romero-Puertas et al. 2004, 2007a, b; Herbette et al. 2006; Yeh et al. 2007; Rodríguez-Serrano et al. 2009). Several studies have provided genetic evidence for the importance of NO in gene regulation. Two studies involving large-scale transcriptional analysis of Α. thaliana have revealed NO-dependent regulation of genes involved in signal transduction, disease resistance, stress response, photosynthesis, and basic metabolism (Grün et al. 2006); however, the intracellular signaling pathway involved has not yet been defined. Most of the information available relates to plant defense and wounding and suggests that NO and salicylic and jasmonic acids are interrelated (Grün et al. 2006). The activity of different nuclear regulatory proteins is dramatically affected by NO. Modification by S-nitrosylation can regulate the activity and function of some regulatory proteins and transcription factors. Although no plant transcription factor has been observed to be regulated by this process, some regulatory proteins could be S-nitrosylated (Grün et al. 2006). NO can regulate cell signaling by controlling Ca²⁺ homeostasis. Most Ca²⁺ channels are regulated by NO either directly through S-nitrosylation or indirectly through cyclic ADP-ribose (cADPR) involving GMP (Courtois et al. 2008). NO-dependent activation of protein kinases, MAPK, and CDPK has been reported in various plant species. The activation of these kinases by NO is thought to be involved in defense responses and/or cell death (Courtois et al. 2008).

7 Conclusion and Future Perspectives

A hypothetical model depicting some of the players involved in ROS, NO perception, and signal transduction pathways in response to cadmium is shown in Fig. 9.3. Cadmium promotes an increase of ROS production in different cell compartments and their accumulation could give rise to oxidative damages affecting to lipids and proteins. However, ROS can also trigger defense cellular responses indirectly acting as signaling molecules, by promoting changes in Ca²⁺ concentration, through GMP or by altering the redox status of several proteins, and further activating the MAPK cascade. But ROS can also directly regulate nuclear gene expression by affecting the redox statu of transcription factor Cys residues. In this scenario, NO plays an important role in the regulation of cell responses to this metal, but further works are needed to better understand the coordinated role of ROS and NO in both toxicity and regulation of cell response to cadmium. The role of some hormones, such as JA and ET, in the cell response to Cd is also an interesting point which deserves more research in order to understand the cross-talk between hormone balance/ ROS and NO in the regulation of plant defense against heavy metals. In response to Cd the upregulation of some defense genes takes place, although some of them are also induced during pathogen attack, which suggests an overlap in the regulatory mechanisms governing these processes. However, the role of genes such as HSPs or chitinases in the mechanisms of tolerance to Cd has not been explored in depth so far, and they could be important in the development of



Fig. 9.3 Hypothetical model showing signal transduction of cell response to cadmium toxicity. Cd-dependent changes in H_2O_2 and NO levels can be perceived by changes in Ca²⁺ concentration, and oxidation or *S*-nitrosylation of

proteins or transcription factors. The transcriptional response can also require more upstream transduction involving mitogen-activated protein kinases (MAPKs) and hormones such as jasmonic acid (JA) and ethylene (ET)

new strategies for phytoremediation. Further studies will be necessary to understand the role of the post-translational modification of proteins in the perception of metal toxicity and also in the transduction and regulation of cell response to Cd. An integrated study of all the players mentioned in this chapter at biochemical, molecular, and cellular levels is needed in order to understand the complex network involved in perception, transduction, and development of cell responses to cope with adverse conditions caused by heavy metals. This could allow the development of new and more efficient strategies for phytoremediation.

Acknowledgments This work was supported by ERDFcofinanced grants from the Ministry of Education and Science (Grant BIO2008-040067) and Junta de Andalucía (Project P06-CVI-01820), Spain

References

- Álvarez S, Berla BM, Sheffiel J, Cahoon RE (2009) Comprehensive analysis of the *Brassica juncea* root proteome in response to cadmium exposure by complementary proteomic approaches. Proteomics 9:2419–2431
- Barroso JB, Corpas FJ, Carreras A, Sandalio LM, Valderrama R, Palma JM, Lupiáñez JA, del Río LA (1999) Localization of nitric oxide synthase in plant peroxisomes. J Biol Chem 274:36729–36733
- Bartha B, Kolbert Z, Erdei L (2005) Nitric oxide production induced by heavy metals in *Brassica juncea* L. Czern. and *Pisum sativum* L. Acta Biol Szeg 49:9–12
- Baryla A, Carrier P, Franck F, Coulomb C, Sahut C, Havaux M (2001) Leaf chlorosis in oilseed rape plants (*Brassica napus*) grown on cadmium-polluted soil: causes and consequences for photosynthesis and growth. Planta 212:696–709
- Békésiová B, Hraška S, Libantová J, Moravcikova J, Matusšiková I (2008) Heavy-metal stress induced accumulation of chitinase isoforms in plants. Mol Biol Rep 35:579–588
- Benavides MP, Gallego SM, Tomaro M (2005) Cadmium toxicity in plants. Brazil J Plant Physiol 17:21–34
- Besson-Bard A, Pugin A, Wendehenne D (2008) New insights into nitric oxide signalling in plants. Ann Rev Plant Biol 59:21–39
- Besson-Bard A, Gravot A, Richaud P, Auroy P, Duc C, Gaymard F, Taconnat L, Renou JP, Pugin A, Wendehenne D (2009) Nitric oxide contributes to cadmium toxicity in Arabidopsis by promoting cadmium accumulation in roots and by up-regulating genes related to iron uptake. Plant Physiol 149:1302–1315

- Clemens S (2006) Toxic metal accumulation, responses to exposure and mechanisms of tolerance in plants. Biochimie 88:1707–1719
- Cobett CS (2000) Phytochelatins and their roles in heavy metal detoxification. Plant Physiol 123:825–832
- Collin V, Eymery F, Genty B, Rey P, Havaux M (2008) Vitamin E is essential for the tolerance of *Arabidopsis thaliana* to metal-induced oxidative stress. Plant Cell Environ 31:244–257
- Corpas FJ, Barroso JB, Carreras A, Quirós M, León AM, Romero-Puertas MC, Esteban J, Valderrama R, Palma JM, Sandalio LM, Gómez M, del Río LA (2004) Cellular and subcellular localization of endogenous nitric oxide in young and senescent pea plants. Plant Physiol 136:2722–2733
- Correa-Aragunde N, Lombardo C, Lamattina L (2008) Nitric oxide: an active nitrogen molecule that modulates cellulose synthesis in tomato roots. New Phytologist 179:386–396
- Courtois C, Besson A, Dahan J, Bourque S, Dobrowolska G, Pugin A, Wendehenne D (2008) Nitric oxide signalling in plants: interplays with Ca²⁺ and protein kinases. J Exp Bot 59:155–163
- Cuypers A, Smeets K, Ruytinx J, Opdenakker K, Keunen E, Remans T, Horemans N, Vanhoudt N, Van Sanden S, Van Belleghem F, Guisez Y, Colpaert J, Vangronsveld J (2011) The cellular redox states as a modulator in cadmium and copper responses in *Arabidopsis thaliana* seedlings. J Plant Physiol 168:309–316
- DalCorso G, Farinati S, Maistri S, Furini A (2008) How plants cope with cadmium: staking all on metabolism and gene expression. J Int Plant Biol 50:1268–1280
- Dana MM, Pintor-Toro JA, Cubero B (2006) Transgenic tobacco plants overexpressing chitinases of fungal origin show enhanced resistance to biotic and abiotic stress agents. Plant Physiol 142:722–730
- De Michele R, Vurro E, Rigo C, Costa A, Elviri L, Di Valentin M, Careri M, Zottini MM, Sanità di Toppi L, Lo Schiavo F (2009) Nitric oxide is involved in cadmium-induced programmed cell death in *Arabidopsis* suspension cultures. Plant Physiol 150:217–228
- del Río LA (2011) Peroxisomes as a cellular source of reactive nitrogen species signal molecules. Arch Biochem Biophys 506:1–11
- del Río LA, Corpas FJ, Barroso JB (2004) Nitric oxide and nitric oxide synthase activity in plants. Phytochemistry 65:783–792
- del Río LA, Sandalio LM, Corpas FJ, Palma JM, Barroso JB (2006) Reactive oxygen species and reactive nitrogen species in peroxisomes. Production, scavenging and role in cell signalling. Plant Physiol 141:330–335
- del Río LA, Sandalio LM, Corpas FJ, Romero-Puertas MC, Palma JM (2009) Peroxisomes as a cellular source of ROS signal molecules. In: del Río LA, Puppo A (eds) Reactive oxygen species in plant signaling. Springer-Verlag, Belin-Heidelberg, pp 95–111
- Delledonne M (2005) NO news is good news for plants. Curr Opin Plant Bio 1 8:390–396
- Delledonne M, Zeier J, Marocco A, Lamb C (2001) Signal interactions between nitric oxide and reactive oxygen

intermediates in the plant hypersensitive disease resistance response. Proc Natl Acad Sci USA 98: 13454–13459

- Djebali W, Gallusci P, Polge C, Boulila L, Galtier N, Raymond P, Chaibi W, Brouquisse R (2008) Modifications in endopeptidase and 20S proteasome expression and activities in cadmium treated tomato (Solanum lycopersicum L.) plants. Planta 227:625–639
- Domínguez-Solis JR, Gutiérrez-Alcalá G, Romero LC, Gotor C (2001) The cytosolic O-acetylserine (thiol) lyase gene is regulated by heavy metal and can function in cadmium tolerance. J Biol Chem 276:9297–9303
- Faller P, Kienzler K, Krieger-Liszkay A (2005) Mechanism of Cd²⁺ inhibits photoactivation of photosyntem II by competitive binding to the essential Ca²⁺ site. Biochim Biophys Acta 1706:158–164
- Foresi N, Correa-Aragunde N, Parisi G, Caló G, Salerno G, Lamattina L (2010) Characterization of a nitric oxide synthase from the plant kingdom: NO generation from the green alga Ostreococcus tauri is light irradiance and growth phase dependent. Plant Cell 22:3816–3830
- Fotjová M, Kovařik A (2000) Genotoxic effect of cadmium is associated with apoptotic changes in tobacco cells. Plant Cell Environ 23:531–537
- Fusco N, Micheletto L, Dal Corso G, Borgato L, Furini A (2005) Identification of cadmium-regulated genes by cDNA-AFLP in the heavy metal accumulator *Brassica juncea* L. J Exp Bot 56:3017–3027
- Garnier L, Simon-Plas F, Thuleau P, Agnel JP, Blein JP, Ranjeva R, Montillet JL (2006) Cadmium affects tobacco cells by a series of three waves of reactive oxygen species that contribute to cytotoxicity. Plant Cell Environ 29:1956–1969
- Grill E, Mishra S, Srivastava S, Tripathi RD (2006) Role of phytochelatins in phytoremediation of heavy metals. In: Singh SN, Tripathi RD (eds) Environmental bioremediation technologies. Springer, Heidelberg, pp 101–145
- Grün S, Lindermayr C, Sell S, Durner J (2006) Nitric oxide and gene regulation in plants. J Exp Bot 57:507–516
- Harrison R (2002) Structure and function of xanthine oxidoreductase: where are we now? Free Radic Biol Med 33:774–797
- Herbette S, Taconnat L, Hugouvieux V, Piette L, Magniette ML, Cuine S, Auroy P, Richaud P, Forestier C, Bourguignon J, Renou JP, Vavasseur A, Leonhardt N (2006) Genome-wide transcriptome profiling of the early cadmium response of *Arabidopsis* roots and shoots. Biochimie 88:1751–1765
- Hernández LE, Cooke DT (1997) Modification of the root plasma membrane lipid composition of cadmiumtreated *Pisum sativum*. J Exp Bot 48:1375–1381
- Hernández LE, Lozano-Rodriguez E, Garate A, Carpena-Ruiz R (1998) Influence of cadmium on the uptake, tissue accumulation and subcellular distribution of manganese in pea seedlings. Plant Science 132:139–151

- Heyno E, Klose C, Krieger-Lyzkay A (2008) Origin of cadmium-induced reactive oxygen species production: mitochondrial electron transfer versus plasma membrane NADPH oxidase. New Phytol 179:687–699
- Horemans N, Raeymaekers T, Van Beek K, Nowocin A, Blust R, Broos K, Cuypers A, Vangronsveld J, Guisez Y (2007) Dehydroascorbate uptake is impaired in the early response of *Arabidopsis* plant cell cultures to cadmium. J Exp Bot 16:4307–4317
- Howden R, Goldsbrough PB, Andersen CS, Cobbett CS (1995) Cadmium-sensitive, cad1 mutants of *Arabidopsis thaliana* are phytochelatin deficient. Plant Physiol 107:1059–1066
- Hsu YT, Kao CH (2004) Cadmium toxicity is reduced by nitric oxide in rice leaves. Plant Growth Regul 42: 227–238
- Jasid S, Simontacchi M, Bartoli CG, Puntarulo S (2006) Chloroplasts as a nitric oxide cellular source. Effect of reactive nitrogen species on chloroplastic lipids and proteins. Plant Physiol 142:1246–1255
- Jasinski M, Sudre D, Schanske G, Schellenberg M, Constant S, Martinoia E, Bovet L (2008) AtOSA1, a member of the Abc1-like family, as a new factor in cadmium and oxidative stress response. Plant Physiol 147:719–731
- Kim DY, Bovet L, Maeshima M, Martinoia E, Lee Y (2007) The ABC transporters AtPDR8 is a cadmium extrusion pump conferring heavy metal resistance. Plant J 50:207–218
- Kobayashi M, Ohura I, Kawakita K, Yokota N, Fujiwara M, Shimamoto K (2007) Calcium-dependent protein kinases regulate the production of reactive oxygen species by potato NADPH oxidase. Plant Cell 19:1065–1080
- Kopyra M, Gwóždž EA (2003) Nitric oxide stimulates seed germination and counteracts the inhibitory effect of heavy metals and salinity on root growth of *Lupinus luteus*. Plant Physiol Biochem 41:1011–1017
- Kopyra M, Stachon-Wilk M, Gwozdz EA (2006) Effects of exogenous nitric oxide on the antioxidant capacity of cadmium-treated soybean cell suspension. Acta Physiologiae Plantarum 28:525–536
- Kurepa J, Toh-E A, Smalle JA (2008) 26S proteasome regulatory particle mutants have increased oxidative stress tolerance. Plant J 53:102–114
- Laspina NV, Groppa MD, Tomaro ML, Benavides MP (2005) Nitric oxide protects sunflower leaves against Cd-induced oxidative stress. Plant Sci 169:323–330
- Lemaire S, Keryer E, Stein M, Schepens I, Issakidis-Bourguet E, Gérard-Hirme C, Miginiac-Maslow M, Jacquot JP (1999) Heavy-metal regulation of thioredox in gene expression in *Chlamydomonas reinhardti*. Plant Physiol 120:773–778
- León AM, Palma JM, Corpas FJ, Gomez M, Romero-Puertas MC, Chatterjee D, Mateos RM, del Río LA, Sandalio LM (2002) Antioxidative enzymes in cultivars of pepper plants with different sensitivity to cadmium. Plant Physiol Biochem 40:813–820
- Lindermayr C, Durner J (2009) *S*-nitrosylation in plants: pattern and function. J Proteomics 73:1–9

- Lindermayr C, Saalbach G, Bahnweg G, Durner J (2006) Differential inhibition of *Arabidopsis* methionine adenosyltransferases by protein *S*-nitrosylation. J Biol Chem 281:4285–4291
- López-Martín MC, Becana M, Romero LC, Goto C (2008) Knocking out cytosolic cysteine synthesis compromises the antioxidant capacity of the cytosol to maintain discrete concentrations of hydrogen peroxide in *Arabidopsis*. Plant Physiol 147:562–572
- Ma CH, Haslbeck M, Babujee L, Jahn O, Reumann S (2006) Identification and characterization of a stressinducible and a constitutive small heat-shock protein targeted to the matrix of plant peroxisomes. Plant Physiol 141:47–60
- McCarthy I, Romero-Puertas MC, Palma JM, Sandalio LM, Corpas FJ, Gómez M, del Río LA (2001) Cadmium induces senescence symptoms in leaf peroxisomes of pea plants. Plant Cell Environ 24:1065–1073
- Metwally A, Finkemeier I, Georgi M, Dietz KJ (2003) Salicylic acid alleviates the cadmium toxicity in barley seedling. Plant Physiol 132:272–281
- Millar TM, Stevens CR, Benjamin N, Eisenthal R, Harrison R, Blake DR (1998) Xanthine oxidoreductase catalyses the reduction of nitrates and nitrite to nitric oxide under hypoxic conditions. FEBS Lett 427:225–228
- Minglin L, Yuxiu Z, Tuanyao C (2005) Identification of genes up-regulated in response to Cd exposure in *Brassica juncea* L. Gene 363:151–158
- Mittler R (2002) Oxidative stress, antioxidants and stress tolerance. Trends Plant Sci 7:405–410
- Mittler G, Mittler R (2006) Could heat shock transcription factor function as hydrogen peroxide sensor in plants? Ann Bot 98:279–288
- Moreau M, Lindermayr C, Durner J, Klessig DF (2010) NO synthesis and signaling in plants – where do we stand? Physiol Plant 138:372–383
- Neill S, Bright J, Desikan R, Hancock Harrison J, Wilson I (2008) Nitric oxide evolution and perception. J Exp Bot 59:25–35
- Nordberg GF (2004) Cadmium and health in the 21st century-historical remark and trends for the future. Biometals 17:485–489
- Ogawa I, Nakanishi H, Mori S, Nishizawa NK (2009) Time course analysis of gene regulation under cadmium stress in rice. Plant Soil 325:97–108
- Olmos E, Martinez-Solano JR, Piqueras A, Hellin E (2003) Early steps in the oxidative burst induced by cadmium in cultured tobacco cells (BY-2 line). J Exp Bot 54:291–301
- Ortega-Villasante C, Rellán-Álvarez ZZ, Del Campo FF, Carpena-Ruíz RO, Hernández LE (2005) Cellular damage induced by cadmium and mercury in *Medicago* sativa. J Exp Bot 56:2239–2251
- Ouariti O, Boussama N, Zarrouk M, Cherif A, Ghorbal MH (1997) Cadmium- and copper-induced changes in tomato membrane lipids. Phytochemistry 45:1343–1350
- Paradiso A, Berardino R, de Pinto MC, di Toppi LS, Storelli MM, Tommasi F, De Gara L (2008) Increase in ascorbate-glutathione metabolism as local and precocious systemic responses induced by cadmium in

durum wheat plants. Plant and Cell Physiology 49:362–374

- Pena LB, Pasquini LA, Tomaro ML, Gallego SM (2006) Proteolytic system in sunflower (Helianthus annuus L.) leaves under cadmium stress. Plant Science 171:531–537
- Pena LB, Pasquini LA, Tomaro ML, Gallego SM (2007) 20S proteasome and accumulation of oxidized and ubiquitinated proteins in maize leaves subjected to cadmium stress. Phytochemistry 68:1139–1146
- Perfus-Barbeoch L, Leonhardt N, Vavasseu A, Forestier C (2002) Heavy metal toxicity: cadmium permeates through calcium channels and disturbs the plant water status. Plant J 32:539–548
- Polge C, Jaquinod M, Holzer F, Bourguignon J, Walling L, Brouquisse R (2009) Evidence for the existence in *Arabidopsis thaliana* of the proteasome proteolytic pathway. Activation in response to cadmium. J Biol Chem 284:5412–35424
- Poschenrieder C, Gunsé B, Barceló J (1989) Influence of cadmium on water relations, stomatal resistance, and abscisic acid content in expanding bean leaves. Plant Physiol 90:1365–1371
- Ramos J, Clemente MR, Naya L, Pérez-Rontomé C, Sato S, Tabata S, Becana M (2007) Phytochelatin synthases of the model legume *Lotus japonicus*. A small multigene family with differential response to cadmium and alternatively spliced variants. Plant Physiol 143:1110–1118
- Remans T, Opdenakker K, Smeets K, Matgijsen D, Vangronsveld J, Cuypers A (2010) Metal-specific and NADPH oxidase dependent changes in lipoxygenase and NADPH oxidase gene expression in *Arabidopsis thaliana* exposed to cadmium or excess copper. Funct Plant Biol 37:532–544
- Rivetta A, Negrini N, Cocucci M (1997) Involvement of Ca²⁺ – calmodulin in Cd²⁺ toxicity during the early phases of radish (*Raphanus sativus* L.) seed germination. Plant Cell Environ 20:600–608
- Rodríguez-Serrano M, Romero-Puertas MC, Zabalza A, Corpas FJ, Gómez M, del Río LA, Sandalio LM (2006) Cadmium effect on oxidative metabolism of pea (*Pisum sativum* L.) roots. Imaging of reactive oxygen species and nitric oxide accumulation *in vivo*. Plant Cell Environ 29:1532–1544
- Rodríguez-Serrano M, Romero-Puertas MC, Pazmiño DM, Testillano PS, Risueño MC, del Río LA, Sandalio LM (2009) Cellular response of pea plants to cadmium toxicity: cross talk between reactive oxygen species, nitric oxide, and calcium. Plant Physiol 150:229–243
- Rogers EE, Eide DJ, Guerinot ML (2000) Altered selectivity in an Arabidopsis metal transporter. Proc Natl Acad Sci USA 97:12356–12360
- Romero-Puertas MC, McCarthy I, Sandalio LM, Palma JM, Corpas FJ, Gómez M, del Río LA (1999) Cadmium toxicity and oxidative metabolism of pea leaf peroxisomes. Free Rad Res 31(Suppl):S25–S32
- Romero-Puertas MC, Palma JM, Gómez M, del Río LA, Sandalio LM (2002) Cadmium causes the oxidative

modification of proteins in pea plants. Plant Cell Environ 25:677–686

- Romero-Puertas MC, Rodríguez-Serrano M, Corpas FJ, Gómez M, del Río LA, Sandalio LM (2004) Cadmium-induced subcellular accumulation of O₂⁻ and H₂O₂ in pea leaves. Plant Cell Environ 27:1122–1134
- Romero-Puertas MC, Corpas FJ, Rodríguez-Serrano M, Gomez M, del Río LA, Sandalio LM (2007a) Differential expression and regulation of antioxidative enzymes by cadmium in pea plants. J Plant Physiol 164:1346–1357
- Romero-Puertas MC, Laxa M, Mattè A, Zanninotto F, Finkemeier I, Jones AME, Perazzolli M, Vandelle E, Dietz KJ, Delledonne M (2007b) S-nitrosylation of peroxiredoxin II E promotes peroxynitrite-mediated tyrosine nitration. Plant Cell 19:4120–4130
- Romero-Puertas MC, Campostrini N, Mattè A, Righetti PG, Perazzolli M, Zolla L, Roepstorff P, Delledonne M (2008) Proteomic analysis of S-nitrosylated proteins in Arabidopsis thaliana undergoing hypersensitive response. Proteomics 8:1459–1469
- Sandalio LM, Dalurzo HC, Gómez M, Romero-Puertas MC, del Río LA (2001) Cadmium-induced changes in the growth and oxidative metabolism of pea plants. J Exp Bot 52:2115–2126
- Sandalio LM, Sandalio LM, Rodríguez-Serrano M, del Río LA, Romero-Puertas MC (2009) Reactive oxygen species and signalling in cadmium toxicity. In: del Río LA, Puppo A (eds) Reactive oxygen species in plant signaling. Springer-Verlag, Belin-Heidelberg, pp 175–190
- Sanitá di Toppi L, Gabbrielli R (1999) Response to cadmium in higher plants. Environ Exp Bot 41:105–130
- Sarry JE, Kuhn L, Ducruix C, Lafaye A, Junot C, Hugouvieux V, Jourdain A, Bastien O, Fievet JB, Vailhen D, Amekraz B, Moulin C, Ezan E, Garin J, Bourguignon J (2006) The early responses of Arabidopsis thaliana cells to cadmium exposure explored by protein and metabolite profiling analyses. Proteomics 6:2180–2198
- Schützendübel A, Polle A (2002) Plant responses to abiotic stresses: heavy metal-induced oxidative stress and protection by mycorrhization. Journal of Experimental Botany 53:1351–1365
- Schützendübel A, Schwanz P, Terchmann T, Gross K, Langenfeld-Heyger R, Godbold DL, Polle A (2001) Cadmium-induced changes in antioxidative systems, hydrogen peroxide content, and differentiation in scots pine roots. Plant Physiol 127:887–898
- Sharma SS, Dietz KJ (2006) The significance of amino acids and amino acid-derived molecules in plant responses and adaptation to heavy metal stress. J Exp Bot 57:711–726
- Siddiqui M, Al-Whaibi MH, Basalah MO (2010) Role of nitric oxide in tolerance of plants to abiotic stress. Protoplasma. doi:10.1007/s00709-010-0206-9
- Sing BK, Foley RC, Onate-Sanchez L (2002) Transcription factors in plant defense and stress responses. Curr Opin Plant Biol 5:430–436

- Singh HP, Batish DR, Kaur G, Arora K, Kohli RK (2008) Nitric oxide (as sodium nitroprusside) supplementation ameliorates Cd toxicity in hydroponically grown wheat roots. Environ Exp Bot 63:158–167
- Smeets K, Cuypers A, Lambrechts A, Semane B, Hoet P, Van Laere A, Vangronsveld J (2005) Induction of oxidative stress and antioxidative mechanisms in *Phaseolus vulgaris* after Cd application. Plant Physiol Biochem 43:437–444
- Song WY, Martinoia E, Lee J, Kim D, Kim DY, Vogt E, Shim D, Choi KS, Hwang I, Lee Y (2004) A novel family of cys-rich membrane proteins mediates cadmium resistance in *Arabidopsis*. Plant Physiol 135:1027–1039
- Stöhr C, Stremlau S (2006) Formation and possible roles of nitric oxide in plant roots. J Exp Bot 57:463–470
- Suzuki N (2005) Alleviation by calcium of cadmiuminduced root growth inhibition in *Arabidopsis* seedlings. Plant Biotech 22:19–25
- Suzuki N, Koizumi N, Sano H (2001) Screening of cadmium-responsive genes in Arabidopsis thaliana. Plant Cell and Environment 24:1177–1188
- Suzuki N, Yamaguchi Y, Koizumi N, Sano H (2002) Functional characterization of a heavy metal binding protein CdI19 from *Arabidopsis*. Plant J 32: 165–173
- Torres MA, Dangl JL (2005) Functions of the respiratory burst oxidase in biotic interactions, abiotic stress and development. Curr Opin Plant Biol 8:397–403
- Tsyganov VE, Belimov AA, Borisov AY, Safronova VI, Georgi M, Dietz KJ, Tikhonovich IA (2007) A chemically induced new pea (*Pisum sativum*) mutant SGECdt with increased tolerance to, and accumulation of, cadmium. Ann Bot 99:227–237
- Van Assche F, Clijsters H (1990) Effects of metals on enzyme activity in plants. Plant Cell Environ 13:195–206
- Van Breusegem F, Dat JF (2006) Reactive oxygen species in plant cell death. Plant Physiol 141:384–390
- Vandenbroucke K, Robbens R, Vandepoele K, Inzé D, Van de Peer Y, Van Breusegem F (2008) Hydrogen peroxide-induced gene expression across kingdoms: a comparative analysis. Mol Biol Evol 25:507–516
- Vanderauwera S, Hoeberichts FA, Van Breusegem F (2009) Hydrogen peroxide-reponsive genes in stress acclimation and cell death. In: del Río LA, Puppo A (eds) Reactive oxygen species in plant signaling. Springer-Verlag, Belin-Heidelberg, pp 149–164
- Wang JW, Wu JY (2005) Nitric oxide is involved in methyl jasmonate-induced defense responses and secondary metabolism activities of *Taxus xells*. Plant Cell Physiol 46:923–930
- Wang Y, Fang J, Leonard SS, Rao KM (2004) Cadmium inhibits the electron transfer chain and induces reactive oxygen species. Free Radic Biol Med 36:1434–1443
- Wang YS, Yang ZM (2005) Nitric oxide reduces aluminum toxicity by preventing oxidative stress in the roots of Cassia tora L. Plant and Cell Physiology 46:1915–1923

- Weber M, Trampczynska A, Clemens S (2006) Comparative transcriptome analysis of toxic metal responses in *Arabidopsis thaliana* and the Cd²⁺ hypertolerant facultative metallophyte *Arabidopsis* halleri. Plant Cell Environ 29:950–963
- Xiang C, Oliver DJ (1998) Glutathione metabolic genes co-ordinately respond to heavy metals and jasmonic acid in *Arabidopsi*. Plant Cell 1:1539–1550
- Xiong J, An L, Zhu C (2009) Exogenous nitric oxide enhances cadmium tolerance of rice by increasing pectin and hemicellulose contents in root cell wall. Planta 230:755–765
- Xiong J, Fu G, Tao L, Zhu C (2010) Roles of nitric oxide in alleviating heavy metal toxicity in plants. Arch Biochem Biophys 497:13–20
- Yakimova ET, Kapchina-Toteva VM, Laarhoven LJ, Harren FM, Woltering EJ (2006) Involvement of

ethylene and lipid signalling in cadmium-induced programmed cell death in tomato suspension cells. Plant Physiol Biochem 44:581–589

- Yeh CM, Chien PS, Huang HJ (2007) Distinct signalling pathways for induction of MAP kinase activities by cadmium and copper in rice roots. J Exp Bot 58:659–671
- Yu CC, Hung KT, Kao CH (2005) Nitric oxide reduces Cu toxicity and Cu-induced NH₄⁺ accumulation in rice leaves. J Plant Physiol 162:1319–1330
- Zhao C-R, Ikka T, Sakawi Y, Kobayashi Y, Suzuki Y, Shigeru S, Sakurai N, Shibata D, Koyama H (2009) Comparative transcriptomics characterization of aluminum, sodium chloride, cadmium and copper rhizotoxicities in *Arabidopsis thaliana*. BMC Plant Biol 9:32–47

Reactive Nitrogen Inflows and Nitrogen Use Efficiency in Agriculture: An Environment Perspective

10

Khalid Rehman Hakeem, Ruby Chandna, Altaf Ahmad, and Muhammad Iqbal

Abstract

Increased use of nitrogenous (N) fertilizer has significantly altered the global N-cycle and produced nitrogenous gases of environmental consequence. While nitrous oxide (N₂O) emissions contribute to global greenhouse gas accumulation and the stratospheric ozone depletion, degradation of groundwater quality by N use in agriculture is fundamentally a nitrate leaching problem. Despite these evident negative environmental impacts, consumption of N fertilizer cannot be reduced in view of the food security for teeming population in the developing countries. Various strategies, from agronomic to genetic engineering, have been tried to tackle this problem. Split application of N, use of slow-release fertilizers, nitrification inhibitors, and the use of organic manures are some agronomic techniques adopted. One of the important goals to reduce N-fertilizer application can be effectively achieved by choosing N-efficient (i.e., which can grow under low N conditions), ensuring their optimum uptake of applied N by application of adequate amounts of fertilizer nutrients in a balanced manner and knowing the molecular mechanisms for their uptake as well as assimilatory pathways. Newer approaches like quantitative trait locus and proteomics could also help us in understanding these processes fully, hence could contribute greatly in enhancing nitrogen use efficiency and reduction of N pollution in the environment.

Keywords

Reactive nitrogen • NUE • QTL • Proteomics • N pollution

K.R. Hakeem (⊠) • R. Chandna • A. Ahmad • M. Iqbal Molecular Ecology Laboratory, Department of Botany, Faculty of Science, Jamia Hamdard, New Delhi 110062, India e-mail: Kur.hakeem@gmail.com

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_10, © Springer Science+Business Media, LLC 2012

1 Introduction

Nitrogen (N) represents one of the most important nutrients found in terrestrial ecosystems. It is an important constituent of a number of complex organic molecules viz., proteins, nucleic acids, etc. Atmosphere is the main reservoir of nitrogen (N_{2}) , which stores around one million times more N than contained in all the organisms. Oceans and organic matter in soil are the other major store houses of nitrogen. N is often considered as an important limiting nutrient for plant growth and development, despite its remarkable abundance in the atmosphere. This is the reason for the past half a century, supply of nitrogen through fertilizers has been an influential application for increasing the growth and yield of cultivated plants such as cereals. To meet the increasing demand for food, farmers apply more fertilizers in their bid to increase the agricultural productivity. Fertilizer nitrogen has provided food security particularly to developing nations including India, as the cereal production has kept pace with its ever-increasing population. Today, India occupies the third rank in the world in fertilizer N consumption and second in fertilizer N production (FAI 2008). The consumption of fertilizer nitrogen in India increased from a mere 55,000 metric tons in 1950-1951 to over 14.2 million tons in 2007–2008 and is still increasing (FAI 2008). With the current rate of N fertilization, the requirement of nitrogen will be 22-25 million tons/year in 2020 (FAI 2008). However, it is remarkable that utilization of applied fertilizer nitrogen in field by most cereal crops does not exceed 50% and around 70% of the total nitrogenous fertilizer is applied for rice and wheat cultivation (Abrol et al. 1999). Therefore, with the increase in agricultural food production worldwide in last 50 years, the N fertilization of crop plants has increased more than 20-fold (Shrawat and Good 2008). However, the use of this fertilizer is generally inefficient, as lesser amount of applied N (around 30-40%) is actually utilized by cereal crops, and the major part (60-70%)is lost from the plant-soil system which has caused severe impacts on the ecosystems of the non-agricultural neighboring bacteria, animals, and plants. As a result of leaching, the unused N fertilizer causes impacts like eutrophication of freshwater (London 2005) and marine ecosystems (Beman et al. 2005). In addition, gaseous augmentation of N oxides reacting and affecting with stratospheric ozone and the volatilization of toxic ammonia into the atmosphere (Stulen et al. 1998) has also been linked to unused N fertilizers. The toxic effects of nitrate are due to its endogenous conversion to nitrite and this ion has been implicated in the occurrence of methaemoglobinemia, gastric cancer, and many other diseases (Anjana et al. 2007).

Presently the human population is more than 6.5 million, which is expected to increase around 10 billion by 2025 (Hirel et al. 2007), therefore, the major challenge will be to reach a highly productive agriculture without degrading the quality of our environment. Efficient farming techniques and choosing plant varieties/genotypes that have better nitrogen use efficiency (NUE) could be the tools to tackle this problem. The development of such varieties/genotypes, through conventional plant breeding techniques or by using recombinant DNA technology, will be more proficient with a better understanding the physiological, genetic, and molecular bases of NUE among cereal crops. Therefore, there is an urgent need of a "second green revolution" that does not rely on exhaustive use of inorganic fertilizers rather would aim at improving crop yields in soils by developing varieties with better adaptation to low-fertility soils (Yan et al. 2006). In the present chapter, we have discussed the inflow and effects of reactive N in the environment and then summarized the strategies adopted to develop the crop varieties/genotypes with high NUE.

2 Reactive Nitrogen Inflow

Reactive nitrogen (Nr) is usually referred to all the nitrogen species that are biologically active, photochemically reactive, and radiatively important N compounds in the atmosphere and biosphere of the earth (Galloway 1995). Thus, Nr includes reduced inorganic forms of N (NH₂, NH₄⁺), oxidized inorganic forms (NO₂, HNO₂, N₂O, NO₃⁻), and organic compounds (urea, amines, proteins, nucleic acids). There are numerous sources in environment that contribute to Nr and total nitrate content of natural waters, e.g., atmogeological sphere, features, anthropogenic sources, atmospheric nitrogen fixation, and soil nitrogen. However, detailed hydro geological investigations conducted have indicated a heterogeneous pattern of nitrate distribution. Soils with low water-holding capacity (sandy soil) and high permeability, movement of pollutants like chloride and nitrate is much quicker than in clayey soil. This is probably the main cause for high nitrates in areas with sandy soil. Vegetables account for more than 70% of the nitrates ingested in the human diet. The remainder of nitrate in a typical diet comes from drinking water (21%), meat and meat products (6%) (Prasad 1999).

The form of added N plays a role in regulating N losses and influencing NUE. Among these forms, NO₃ is the most susceptible to leaching, NH₄ the least, and urea moderately susceptible. Ammonia and urea are more susceptible to volatilization loss of N than fertilizers containing NO₃. Urea is the most widely used N fertilizer in India. The studies showed the importance of selecting ammonium-based N fertilizer early in the season to reduce N leaching due the mobility of urea and nitrate source in irrigated rice and wheat systems (Prasad and Prasad 1996). Nitrate containing fertilizers when applied to rice proved less efficient because nitrate is prone to be lost via denitrification and leaching under submerged soil conditions in normal and alkali soils (Prasad 1998). In saline soils, however, it is beneficial to use NO₃ containing N fertilizers as it compensates the adverse effects of Cl- and SO42- on absorption of NO₃ by plants (Choudhary et al. 2003).

Nitrogen losses from soil–plant system. Once inorganic N has appeared in the soil, it can be absorbed by the roots of higher plants or still metabolized by other microorganism during nitrification. This process is carried out by a specialized series of actions in which a few species of microorganisms oxidize NH_4^+ to NO_2 or NO_2^- to NO_3^- . Ammonium ion reacts with excess hydroxyls in soil solutions, which leads to N losses to the atmosphere by NH₃ volatilization (Wood et al. 2000). This represents an important source of N loss in agricultural soils under favorable conditions. Due to extensive use of N fertilizers and nitrogenous wastes, the amount of N available to plants significantly exceeds the N returned to the atmosphere by gaseous losses of N through volatilization and denitrification (Martre et al. 2003). Minimizing drying of surface soil and providing additional source of urease enzyme can minimize NH₂ volatilization. A portion of this excess N is leached out in the soil profile as NO₃⁻ or carried in runoff waters. These are conductive conditions for N losses in agricultural soils, thus reducing the NUE (Delgado et al. 2001). With transport of N in water ways and neighboring ground-water systems, the N concentration could exceed the levels acceptable for human consumption. Nitrate in soil profile may be leached into groundwater when percolating water moves below the rooting depths of crop and provides leaching potential. Paramasivam et al. (2002) have reported a potential leaching of NO₂⁻ in arid regions and sandy soils. Losses of N by leaching are affected by local differences in rainfall, water-holding capacity of soil, soil-drainage properties, and rates of mineralization of soil organic N (Delgado et al. 1999). Processes such as adsorption, fixation, immobilization, and microbial assimilation of added NH₄-N in soils are of great importance as they affect NUE and have the corresponding environmental repercussions (Kissel et al. 2004).

In many field situations, more than 60% of applied N is lost due in part to the lack of synchrony of plant N demand with N supply. The remainder of the N is left in the soil, or is lost to other parts of the environment through leaching, runoff, erosion, NH_3 volatilization, and denitrification. The cereal NUEs are 42% in developed and 29% in developing countries (Raun and Johnson 1999). Many ¹⁵N studies have reported N fertilizer losses in cereal production from 20 to 50% with higher values in rice than in wheat (Ladha 2005). Prasad (1998) reported that apparent recovery of N applied to wheat varies from 40 to 91%. It has been estimated that rice and wheat N recovery efficiency ranging from 30 to 40% are occurring



Fig. 10.1 Diagrammatic representation of N inflow and N loss in ecosystem

in irrigated conditions. An N recovery efficiency exceeding 40% is expected to occur in response to improved N management practices. In a rice–wheat cropping system of Punjab, recovery of ¹⁵N by the first wheat crop was 30–41%, the soil at wheat harvest retained 19–26%, and the succeeding rice recovered 5.2% of the 120 kg N ha⁻¹ applied (Singh and Singh 2001). Total losses of applied N (not recovered from soil–plant system) were about 42% in rice and 33% in wheat grown on a typical sandy loam soil in north-west India.

The main causes of for low N recovery are usually attributed to (1) ammonia volatilization, (2) denitrification, (3) leaching, and (4) runoff and erosion (Fig. 10.1). Loss of N via NH₂ volatilization can be substantial from surface-applied urea in both rice and wheat, which can exceed 40%, and generally greater with increasing soil pH, temperature, electrical conductivity, and surface residue (Singh et al. 2003; Choudhary et al. 2003). Water management in rice and wheat fields influences the extent of N losses due to nitrification-denitrification and NH, volatilization. Available research results from ideal rice soils suggest that NH₃ volatilization rather than denitrification is an more important gaseous loss mechanism for fertilizer N applied to continuously flooded, puddled rice soils of the tropics. The picture is quite opposite in highly permeable porous soils under rice. There exist two mechanisms in such soils due to which losses due to denitrification assume more importance than NH₃

K.R. Hakeem et al.

volatilization losses. Firstly, in porous soils under rice it is difficult to maintain continuous flooding. Rather there occur very frequent alternate aerobicanaerobic cycles, which lead to very fast formation of nitrate under aerobic conditions and their subsequent denitrification under anaerobic conditions that develop due to application of irrigation (Singh and Singh 2001). Secondly, due to high permeability of coarse textured porous soils, urea as such is rapidly transported to subsoil where even after it is hydrolyzed to NH₄, it is not prone to losses via NH₃ volatilization (Sangwan et al. 2004a). Sangwan et al. (2004a, b) have shown that NH₃ volatilization losses from urea increases with the increase in soil salinity, sodicity, and the rate of N applied. The losses of fertilizer N as NH₂ in rice decreased with increasing floodwater depth and depth of placement (Singh et al. 1995a), and with the application of organic manures (Sihag and Singh 1997). Alkalinity, pH, and NH₂ concentration in flood water control the extent of NH₃ loss from flooded soils (Singh et al. 2003). Sarkar et al. (1991) reported a loss of 15-20% of applied N when urea was broadcast in a wheat field. Prasad (1999) reported a marked reduction in the loss of applied N when the urea was deep placed as compared with surface broadcast on a moist soil. They have reported 13.5% N losses as ammonia after 1 week of urea application under submerged conditions. The high pH or alkalinity resulted in high losses of ammonia by volatilization, which can be nearly 60% of applied N at field capacity. Submergence decreases pH as well as losses to 35% of applied N. The reclamation of sodic soils using gypsum has been found to decrease N losses through ammonia volatilization (Choudhary et al. 2003). The timing of fertilization and irrigation could further influence the losses of urea applied to porous soils. If applied on the wet soil surface following irrigation, as much as 42% of the applied ¹⁵N was lost, most likely due to volatilization (Sangwan et al. 2004a). Singh et al. (1995b) showed that application of urea before irrigation increased the NUE by 20% as compared to its surface application after irrigation or broadcast application and surface mixing of urea at field capacity in a clay loam soil.

In nonideal porous soils under rice, there exist every possibility that applied urea N is preferentially lost via denitrification rather than NH₃ volatilization. Direct measurement of denitrification losses made by Aulakh et al. (2001) showed that denitrification is a significant N loss process under wetland rice amounting to 33% of the applied N. In excessive N fertilizer application (i.e., at rates in excess of that needed for maximum yield in cereal crops), NO₃ leaching can be significant, particularly from the coarse-textured soils. Residual N is then available in soil profile for potential leaching. High levels of NO₃-N in the region's groundwater have been reported by Singh et al. (1995b). There is not much information available on leaching losses of N. In a pot culture study, the leaching loss was 11.5% of the applied urea N and was reduced to 8.7% when urea was coated with neem cake (Prasad and Prasad 1996). In a field study at Pantnagar on a silty clay loam soil, 12% of the applied N was lost by leaching and these losses were reduced to 8% when urea was blended with neem cake (Singh et al. 1995b).

3 Nitrogen Removal by Crops

From the human nutrition point of view, rice and wheat are the most important cereals and their production in north-west India in rice-wheat cropping system, which covers about 10 million hectares, is the backbone of the India's food security (Prasad 2005). Rice-wheat cropping system produces 5-14 ton/ha/year grain and this depends heavily on nitrogen fertilization which ranges from 100 to 150 kg N/ha/crop or even more, especially in rice. From the animal nutrition point of view, maize, sorghum, and pearl millet stovers which contain 27-51% of nitrogen harvested by the crop in stover are more important both for milch as well as draught cattle. On the contrary, rice and wheat straw is low in nitrogen content and is a poor protein source. Nevertheless, they meet majority calorie requirements of the cattle. Also most sorghum and pearl millet is grown in rainfed areas where nitrogen application rates are low and even response to N application is low. Nitrogen removal per metric ton as well as its percentage in grain in pulses depends very much upon the plant stature and its vegetative growth. For example, Prasad (2004) reported a removal of 50.6 kg/ton in chickpea and 92.1 kg/ton in pigeon pea; these are the two major pulse crops in India. Most of this nitrogen is obtained by N fixation by Rhizobia as very little fertilizer N is applied to pulses. Again depending upon the plant stature and vegetative growth, 63.3% of total N removed by chickpea was contained in its grain, while the values for pigeon pea, a tall and heavily fertilized plant, was 31.6%. The protein-rich pulse foliage is widely used for enriching rice or wheat straw fed to cattle. Before the mechanization of Indian agriculture which is even now limited mostly to north-western India, draught animals were the major source of farm power and the Indian agriculture provided a characteristic "humans-animals-crops" ecosystem where man survived on the grains and the animals on the straw/stover. Taking an average N contribution by grain legumes at 30 kg N/ha, about 0.66 million metric tons of N is annually added to soil on 22 million hectares occupied by them. Another 0.34 million metric tons N may be added by leguminous trees and plants in forests and grasslands, and by leguminous oilseed crops such as groundnut. Thus the N contribution of legumes in Indian soils can be roughly estimated at least at 1 million metric tons, it is likely to be much more. In addition, some N is added by rains and use of N-fixing biofertilizer such as Azotobacter, Azospirillum, Acetobacter, Bluegreen algae, and Azolla.

4 Concept of NUE

NUE at the plant level is its ability to utilize the available nitrogen (N) resources to optimize its productivity (Raghuram et al. 2006). As a concept, NUE includes N uptake, utilization, or acquisition efficiency, expressed as a ratio of the total plant N, grain N, biomass yield, grain yield (output) and total N, soil N, or N-fertilizer applied (input) (Pathak et al. 2008). NUE is quantified based on apparent nitrogen recovery using physiological and agronomic parameters. Agronomic

efficiency is an integrative index of total economic outputs relative to the available soil N (native and applied). Apparent nitrogen recovery is related to the efficiency of N uptake; physiological NUE deals with N utilization to produce grain or total plant dry matter. NUE in the context of photosynthesis is called as photosynthetic nitrogen use efficiency (PNUE), which is determined by the rate of carbon assimilation per unit leaf nitrogen (Kumar et al. 2002). The most suitable way to estimate NUE depends on the crop, its harvest product, and the processes involved in it.

5 Strategies for Minimizing N Pollution in Agriculture

Various strategies were adopted to minimize the N loss from the agricultural fields. Split application of N, use of slow-release fertilizers, nitrification inhibitors (NIs), and the use of organic manures are some agronomic techniques used. Bulk of the fertilizer nitrogen in India is broadcast on surface and both surface runoff (on sloppy lands) and ammonia volatilization lead to N losses. This can be easily overcome by deep placement of N a few centimeters below soil surface. For example, Sarkar (2005) showed that in wheat surface broadcast application of urea as band or top dressing caused 15-20% loss of N due to agriculture volatilization. Surface broadcast application followed by its mixing with top soil reduced the volatilization loss to 10%, while side band placement of urea reduced it further to only 5%. Thus the farmers need to be told about the advantage of incorporation in surface soil or if possible its placement using a *ferti-drill* or a pore in upland crops. Split application is a wellestablished technique for increasing NUE. In wheat and maize, studies with ¹⁵N showed that application of 40 kg N/ha as basal followed by 60 kg N/ha at crown root initiation (CRI) gave significantly higher yield than all basal application and other split application combinations (Sachdev et al. 2000; Narang et al. 2000). Havangi and Hegde (1983) showed in pearl millet also two or three split applications were found to be better than a single application. In rice, two split applications are recommended for short and medium duration varieties, while three split applications are recommended for long duration varieties (Prasad 1999). Another way is NIs, these are a group of chemicals that are toxic to Nitrosomonas sp. and Nitrosomonas sp. involved in the conversion of NH_4 to NO_2^- as well as to *Nitrobacter* sp. involved in the conversion of NO₂ to NO₃ and therefore, inhibits nitrification, which reduces losses due to leaching and denitrification. The most widely tested NIs are 2-chloro-6-trichloromethyl pyridine (N-serve), 2 amino-4-chloro-6 methyl pyrimidine (AM), dicyandiamide (DCD), and sulfathiazole (ST) (Prasad and Power 1995). Research on the use of NIs for reducing N losses and increasing NUE from the soil was initiated in India by Prasad (1999) at the Indian Agricultural Research Institute (IARI), New Delhi, with a field experiment on rice. Treatment of ammonium sulfate with N-serve significantly increased rice yield and nitrogen uptake by the rice crop. Prasad (2005) showed from a laboratory experiment that N losses due to denitrification could be considerably reduced by treating ammonium sulfate with NIs N-serve and AM. Prasad and Prasad (1996) showed through field experiments that treatment of urea with NIs, N-serve, and AM significantly increased rice yield and N uptake. Das et al. (2004) showed the effect of N-serve and AM on nitrification under field capacity moisture (upland) and water-logged (low-land paddy) conditions at New Delhi. Both the NIs were effective in retarding nitrification. The nitrification rate (nitrates expressed as percentage of total mineral N) after 40 days of incubation was 78% with N-serve at 2 ppm and 76% with AM at 10 ppm (mg/kg) as against 100% with untreated urea. Slow-release N fertilizers (SRFs) were developed with an aim to slowdown the dissolution of applied N so that most of it is taken up by crop plants rather than be subjected to N-loss mechanisms. There are two kinds of SRFs, namely, coated fertilizers and inherently slow dissolution rate materials. The examples of coated SRFs are sulfur-coated urea (SCU) (developed by TVA, USA), lac-coated urea (developed by Indian Lac Research Institute), polymer-coated urea, and to some extent neem cake-coated urea. The other kind of slow-release fertilizers are generally urea-aldehyde condensates, e.g., urea-form (urea and formaldehyde products developed in USA), isobutylidene diurea (IBDU, urea and isobutyraldehyde product developed in Japan and USA), and CD-urea (urea and crotonaldehyde product developed in Germany) (Prasad 2005). After 20 days of incubation under field capacity conditions, the mineral N ($NH_4^+ NO_3^-$) in soil was 67, 43, 31, and 27 ppm (mg/kg soil) with urea, oxamide, IBDU, and SCU, respectively. As would be expected under submerged conditions, NO₂-N was not detected and the NH₄⁺-N content in soil after 20 days of incubation was 67, 61, 46, and 15 ppm with urea, oxamide, IBDU, and SCU, respectively. Thus, of the three SRFs oxamide released, the N the fastest and SCU the slowest.

6 Physiological and Molecular Aspects for Improving NUE

NUE at the plant level is its ability to utilize the available nitrogen (N) resources to optimize its productivity. In terms of agriculture, it is the optimal utilization of nitrogenous manures or fertilizers for plant growth, yield, and protein content, as atmospheric nitrogen gas is not utilized by higher plants, except symbiotic legumes. The inherent efficiency of the plant to utilize available N for higher productivity needs to be tackled biologically (Abrol et al. 1999; Abdin et al. 2005). This includes uptake, assimilation, and redistribution of nitrogen within the cell and balance storage and current use at the cellular and whole plant level. Moreover, since N demand and its actual availability tend to vary in time, space, and environmental conditions, the regulation of plant nitrogen metabolism must be responsive to nutritional, metabolic, and environmental cues.

6.1 Regulation of Nitrate Uptake

Plants have evolved an active, regulated, and multiphasic transport system making their NO_3^- uptake scheme efficient enough to transport

sufficient NO₃⁻ to satisfy total nitrogen demand of the plant in face of varying external NO₃⁻ concentrations. Plants can also take up other forms of nitrogen, such as amino acids and ammonium ions. Root NH₄⁺ uptake is carried out by both high-affinity and low-affinity NH₄⁺ transporters that are encoded by a multigene family (Glass et al. 2002). However, nitrate is the most abundant form of nitrogen available to the plant roots in aerated soils. Nitrate influx is an active process driven by the H⁺ gradient and can work against an electrochemical potential gradient (Vidmar et al. 2000). The uptake involves high- and low-affinity transport systems, also known as HATS and LATS, respectively (Forde 2000). One of the high-affinity systems is strongly induced in presence of NO₃⁻ and is known as inducible highaffinity transport system (or iHATS), while the second high-affinity system (the cHATS) and LATS are constitutively expressed (Aslam et al. 1993; Glass and Siddiqi 1995; Forde 2002). The Km values of iHATS, cHATS, and LATS for nitrate are in the ranges of 13-79 µM, 6-20 µM, and >1 mM, respectively.

The iHATS is a multicomponent system encoded partly by genes of the NRT2 family or nitrate-nitrite porter family of transporters. Recently, two dual affinity transporters have been identified in Arabidopsis, AtKUP1, and AtNRT1.1, of which the latter is induced as HATS by phosphorylation at threonine residue 101. This family of transporters is recognized as being exceptional in both the variety of different substrates which its members can mobilize (oligopeptides, amino acids, NO₃⁻, chlorate) and in the ability of individual transporters to handle substrates of very different sizes and charges. Nitrate acts as a regulator for its own uptake, a specific property which is not seen in other ion transport systems such as phosphate, sulfate, etc. On exposure of the cells to external NO_3^{-} , the uptake capacity increases after a lag period of 0.5-1.5 h and reaches a new steady state after 4–6 h. Use of RNA and protein synthesis inhibitors provided early evidence that induction of the iHATS involves gene expression and the synthesis of new transporter protein (Aslam et al. 1993). The evidence that the inducer of iHATS is indeed nitrate ion and not its downstream metabolite came from NR-deficient mutants of Arabidopsis and N. plumbaginifolia (Krapp et al. 1998; Lejay et al. 1999). Studies in the last decade have shown that enhancing the uptake of N by overexpressing transporters may not necessarily improve NUE. For example, transgenic overexpression of a CHL1 cDNA (representing the constitutive HATS) driven by the cauliflower mosaic virus 35S promoter in a chl1 mutant, recovered the phenotype for the constitutive phase but not for the induced phase (Liu et al. 2003). Similarly, the NO₃⁻ contents in transgenic tobacco plants overexpressing the NpNRT2.1 gene (encoding HATS), were remarkably similar to their wildtype levels, despite an increase in the NO_3^- influx. These findings indicate that genetic manipulation of nitrate uptake may not necessarily lead to associated improvement in nitrate retention, utilization, or NUE, though it remains to be seen whether different plants respond differently to the overexpression of different transporters (Pathak et al. 2008). Light as an important abiotic factor is known to enhance NO₃⁻ uptake in a number of plant species and diurnal changes in nitrate uptake have been observed (Anjana et al. 2007). These changes seem to be linked to the imbalance between nitrate uptake and reduction due to the light regime and as well as to the rate of photosynthesis in shoots. Reduced nitrate uptake during darkness could be reversed by exogenous supply of sugars (Raghuram and Sopory 1995). Recent evidence on the upregulation of AtNRT1.1 gene expression by auxin (Li et al. 2007) suggests that nitrate transporters may also be regulated by hormones.

6.2 Physiology of Nitrate Reduction in Crops

A portion of the nitrate taken up is utilized/ stored in the root cells, while the rest is transported to other parts of the plant. Due to the abundant availability of photosynthetic reductants, leaf mesophyll cells are the main sites of nitrate reduction. This is initiated by the NAD/ NADP-dependent NR enzyme, which converts nitrate to nitrite by catalytic reaction in the cytosol. Nitrite is transported into the chloroplast, where it is further reduced into ammonium ion by a ferredoxin-dependent NiR. Being the first, irreversible, and often rate-determining step of the N-assimilatory pathway, nitrate reduction has been a favorite step for physiological and biochemical approaches to optimize fertilizer N use.

6.3 Developing Plants with Transport Gene Systems Using Genetic Engineering Tools

Plants receive N from the soil in the form of nitrate or ammonia, however, some may utilize amino acid as an important sources of N. Specific transporters located in the root cell membrane are responsible for uptake of N from the soil. Subsequent to its uptake, NO₃⁻ is assimilated via a series of enzymatic steps. Nitrate reductase being the first enzyme in nitrate assimilatory pathway and thus an important gene for manipulation. NR activity in leaf blades, express either as seasonal average or converted into seasonal input of reduced N, has been related to total reduced N, grain N, and grain yield of cereals. The pattern of nitrate assimilation from different plant parts, viz. the main shoot of wheat, developing ear of wheat plants grown at different soil N levels, and in the leaf blades at different stages of growth has revealed a direct positive correlation between increasing NR activity and increasing rates of nitrogenous fertilization. Most plant tissues have the capacity to assimilate nitrate, though their NR activity varies widely. Several endogenous as well as exogenous factors have been found to influence the expression of NR genes at both translational as well as transcriptional levels.

Andrews et al. (2004) reported that overexpression of either the NR or the NiR gene often affects N uptake by increasing mRNA levels in the plants. However, this does not seem to increase the growth or yield of plants, irrespective of N source. It is believed to be due, in part, to the complex regulation of both NR and the pathway as a whole. Transcriptional regulation of NR has only minor influence on the levels of free amino acids, ammonium, and nitrate, whereas posttranslational regulation of NR strongly affects these compounds (Lea et al. 2006). The light/dark conditions affect NR activity; heterotrophic nitrate assimilation in darkness is closely linked to the oxidative pentose phosphate pathway and the supply of glucose-6-phosphate. Under photoautotrophic conditions, glucose-6phosphate dehydrogenase is inhibited by reduction with thioredoxin in light, thus replacing the heterotrophic dark nitrate assimilatory pathway with regulatory reactions functioning in light. These studies as well as bioenergetic calculations have indicated that both yield and N harvest or protein can be increased to some extent with adequate nitrogen supply by altered management practices, thus improving the fertilizer NUE. Genotypic differences in the NR levels also provide insight in the relation of varietal differences in N assimilation. The genotypic differences in NR expression have been reported in corn, wheat, sorghum, and barley. In sorghum, a positive relationship between decline in the height of the plant and enhancement of NR activity was observed, though no such relationship was evident in tall and dwarf cultivars of wheat, T. aestivum. Wheat genotypes revealed over twofold variability in NR activity, which supports genetic findings that the enzyme level is highly heritable, its differences are reflected in N harvest and that hybrids could be bred with predictable NR levels by selecting parents appropriately. In the high NR genotypes, higher levels of NR activity were found under low N levels, often with significantly higher N concentration in the grains. They also have sustained activity at later stages of growth, such as flag leaf emergence and anthesis. The reasons for these genetic differences are not fully understood, except that the regulation operated at the level of gene expression and that low levels of NADH might limit NR activity in low NR genotypes. Similarly, overexpressing NiR genes in Arabidopsis and tobacco resulted in increased NiR transcript levels but decreased enzyme activity levels, which were attributed to posttranslational modifications.

6.4 Glutamine Synthetase and Glutamate Synthase (GOGAT) Gene Systems

Glutamine synthetase (GS) catalyzes the critical incorporation of inorganic ammonium into glutamine. In higher plants, it is represented by two groups of protein - the cystolic and plastidic forms (Miffin and Habash 2002). Cystosolic GS (GS1) is known to be encoded by a complex multigene family, whereas plastidic GS (GS2) is encoded by a single gene. Glutamate synthase (Glutamine (amide): 2-oxoglutarate aminotransferase, GOGAT) catalyses the reductive transfer of the amide group of glutamine (produced by GS) to 2-oxoglutarate (α -keto glutarate) to form two glutamate molecules (Lea and Ireland 1999). GS/GOGAT pathway is of crucial importance since the glutamine and glutamate produced are donors of amino groups for the biosynthesis of major N-containing compounds, including amino acids, nucleotides, chlorophylls, polyamines, and alkaloids (Lea and Ireland 1999; Hirel and Lea 2001). A direct correlation was reported between an enhanced GS activity in transgenic plants in some cases, which is depicted by an increase in biomass or yield by transforming novel GS1 construct. Similarly, Kozaki and Takeba (1996) constructed transgenic tobacco plants enriched or reduced in plastidic glutamine synthetase (GS2, a key enzyme in photorespiration). Ectopic expression of GS1 has been shown to alter plant growth (Fuentes et al. 2001; Oliveira et al. 2002) and the overexpression of GS1 in transgenic plants could cause the enhancement of photosynthetic rates, higher rates of photorespiration and enhanced resistance to water stress (Fuentes et al. 2001). The overexpression of soybean cytosolic GS1 in the shoots of Lotus corniculatus was reported to accelerate plant development, leading to early senescence and premature flowering, particularly when plants were grown under conditions of high ammonium (Vincentz et al. 1993). Man et al. (2005) provided additional empirical evidence for enhanced nitrogen-assimilation efficiency in GS1 transgenic lines. However, differences in the degree of ectopic GS1 expression have been reported (Fuentes et al. 2001) and attributed to positional effects, effectiveness of chimeric constructs, or differences in growth conditions. This may be due to lack of correlation between the enhanced expression of GS1 and concomitant growth (Vincentz et al. 1993; Ortega et al. 2001). A significant increase in leaf area, plant area, plant height, and dry weight has been recorded in poplar trees transformed with conifer gs1a gene. Striking differences were observed at low nitrate concentration. Furthermore, higher rates at ¹⁵N incorporation into the transgenic plants demonstrate that the transformed plants have increased NUE (Man et al. 2005). Transgenic overexpression and antisense technology have been employed recently to modulate the expression of NADH-GOGAT in alfalfa and rice plants (Yamaya et al. 2002). The studies on transgenic rice plants expressing antisense RNA for either GS1 or NADH-GOGAT point towards the possible involvement of GS1 in the export of N via phloem in senescing leaves. On the other hand, in case of developing leaf blades and spikelets, NADH-GOGAT was implicated in the utilization of glutamine transported from senescing organs (Yamaya 2003). While these genes appear to be good candidates for improving NUE in the short run, the degree of improvement may vary with the crop and cropping conditions. Therefore, the utility of transgenic overexpression of N-assimilatory genes for major improvements of NUE remains uncertain, though the possibility that different crops respond differently cannot be ruled out yet.

6.5 Other Gene Systems Regulating N Metabolism and Their Manipulation

Enzymes like asparagine synthetase (AS), that catalyzes the formation of asparagine (Asn) and glutamate from glutamine (Gln) and aspartate. In higher plants, AS is encoded by a small gene family (Lam et al. 1998). Together with GS, AS is believed to play a crucial role in primary N metabolism. The observation made by Carvalho et al. (2003) that the levels of AS transcripts and polypeptides in the transgenic nodules of Medicago truncatula increase when GS is reduced suggests that AS can compensate for the reduced GS ammonium assimilatory activity. However, it was also demonstrated that GS activity is essential for maintaining the higher level of AS. Thus, GS is required to synthesize enough Gln to support Asp biosynthesis via NADH-GOGAT and AspAT (Carvalho et al. 2003). A reduction in GS activity in transgenic Lotus japonicas is also correlated with an increase in Asn content (Harrison et al. 2007), supporting the hypothesis that when GS becomes limiting, AS may be important in controlling the flux of reduced N into plants. With the aim of increasing Asn production in plants and to study the role of AS, several researches have attempted to clone AS genes and to examine the corresponding gene expression in plants. Lam et al. (2009) showed overexpression of the ASN1 gene in Arabidopsis and demonstrated that the transgenic plants have enhanced soluble seed protein content, enhanced total protein content, and better growth on N-limiting medium. Arabidopsis plants overexpressing the ASN2 gene accumulate less endogenous ammonium than wild-type plants when grown on medium containing 50-mM ammonium. This study indicates that signaling processes may provide an attractive route for metabolic engineering. In comparison to GS/GOGAT enzymes, the physiological role of glutamate dehydrogenase (GDH) has been less clear (Dubois et al. 2003). In an attempt to investigate the role of GDH by expressing a bacterial gdhA gene from E. coli in tobacco, Ameziane et al. (2000) found that biomass production is consistently increased in gdhA transgenics, regardless of whether they are grown under controlled conditions or in the field.

7 Signaling and Regulation of Nitrogen Metabolism

It is a well-known concept in signal transduction that whenever multiple genes are subject to transcriptional regulation by a common signal, it is mediated through a regulatory sequence that exists in all the genes that respond to the signal. These signature sequences, commonly known as response elements, are identified by mutations that abolish their function, and their conserved nature as revealed by homology comparisons. Early experiments in transgenic Nicotiana plants using GUS gene fused to NR and NiR promoter sequences clearly demonstrated for the first time that nitrate induction of gene expression requires some sequence(s) associated with the NR and NiR promoters (Raghuram et al. 2006). Subsequent studies in transgenic tobacco incorporating the 5' flanking regions of the nitrate reductase genes NR1 and NR2 (designated NP1 and NP2), in case of Arabidopsis thaliana, demonstrated that 238 and 330 bp of NP1and NP2, respectively, are sufficient for nitrate-dependent transcription (Lin and Demain 2006). These nitrate-responsive elements (NREs) are composed of several copies of a core A[G/C]TCA sequence motif preceded by an ~7-bp AT-rich sequence present in the 5' flanking regions of nitrate reductase (NR1 and NR2) genes. This particular sequence motif was also found to be very well conserved in the 5' flanking regions of NR and NiR genes from eight other plants. Sarkar (2003) compared the flanking sequences of all available plant nitrate-responsive genes and found that the NRE core sequence (A[C/G] TCA) was present in multiple copies on both strands in all the known nitrate-responsive genes in many dicots, monocots, and cyanobacteria. Though most of the NREs examined contained both the core sequence and a proceeding AT-rich sequence, there were some cases which had GC-rich regions or did not reveal any AT/GC bias. A more detailed bioinformatic analysis of the entire Arabidopsis genome in our lab revealed that the proposed NREs are randomly distributed, with no difference between nitrate-responsive genes and the presumably nonresponsive genes and intergenic regions in the rest of the genome (Raghuram et al. 2006). These findings raise doubts on the validity of the proposed NRE as comprising of (A[C/G] TCA) elements preceded by AT-rich sequence. Further work in this area will need a combination of bioinformatic and experimental approaches to redefine the NREs that mediate the expression of all

nitrate-responsive genes in all plants. The discovery of NREs is important, as it provides an end point for nitrate signal transduction.

8 QTL Approach to NUE

NUE in plants is a complex quantitative trait that depends on a number of internal and external factors in addition to soil nitrogen availability, such as photosynthetic carbon fixation to provide precursors required for amino acid biosynthesis or respiration to provide energy. Although this trait is controlled by a large number of loci acting individually or together, depending on nutritional, environmental, and plant developmental conditions, it is possible to find enough phenotypic and genotypic variability to partially understand the genetic basis of NUE and thus identify some of the key components of yield for markerassisted breeding. Thus the development of molecular markers has facilitated the evaluation of the inheritance of NUE using specific quantitative trait loci (QTLs) that could be identified. In maize, Hirel et al. (2001) and Masclaux et al. (2001) analyzed recombinant inbred lines for physiological traits such as nitrate content, NR and GS activities. When the variation in these traits and yield components were compared, it was found that there was a positive correlation between nitrate content, GS activity, and yield. When the loci that govern quantitative traits were determined on the map of the maize genome, the positions of QTLs for yield components and the locations of the genes for cytosolic GS (GS1) coincided. In maize, studies on different genotypes or populations of recombinant inbred lines based on NUE components, chromosomal regions, and putative candidate genes have hinted at some factors that might control yield and its components directly or indirectly, when the amount of N fertilizers provided to the plant is varied (Hirel et al. 2007).

Similar results were obtained in rice by Obara et al. (2001), confirming the earlier indications that the GS1 enzymatic activity in the leaf cytosol is one of the major steps controlling organic matter reallocation from source to sink organs during senescence and for grain-filling in cereals. Previous studies have already demonstrated that when GS1 is over expressed in Lotus, nitrogen remobilization was prematurely induced leading to early senescence of the plant (Vincentz et al. 1993). In rice (Yamaya et al. 2002) and wheat (Habash et al. 2001), preliminary investigations with enhanced or decreased GS1 activity indicated that grain yield and grain nitrogen content were modified. In other species such as tobacco (Migge et al. 2000) or poplar (Gallardo et al. 1999), overexpression of GS2 or GS1 significantly increased plant biomass production at early stages of plant development. With these experiments, two out of seven QTLs for GS1 protein content were detected in different regions from other physiological and biological traits. In maize, QTLs for the activities of acid-soluble invertase and sucrosephosphate-synthase were detected in the regions where each structural gene was mapped (Ishimaru et al. 2001).

Thus, quantitative studies of genetic variability for NUE using molecular markers and combining agronomic and physiological studies will be increasingly used in the future to identify new genes or loci involved in the regulation of these metabolic pathways and their interconnection with carbon assimilation and recycling and to select genotypes that assimilate or remobilize nitrogen more efficiently.

9 Proteomics Approach to NUE

The ability of crop plants to cope up with the variety of environmental stresses depends upon a number of changes in their proteins, which may be up- and downregulated as a result of altered gene expression. Under a stressful condition, the modifications in the expression levels of these proteins could provide us valuable information about the nature of stress factor as well as the physiological and molecular state of a biological system. Hence, provides us some clues to understand the nature of defensive mechanism and adaptability, besides stress monitoring in these biological systems.

Proteomic-based technologies have been recently applied for the systematic analysis of the induced gene products in a number of plant species subjugated to a wide range of abiotic and biotic challenges. Proteome analysis is becoming a powerful tool in the functional characterization of plants. Due to the availability of vast nucleotide sequence information and based on the progress achieved in sensitive and rapid protein identification by mass spectrometry, proteome approaches open up new perspectives to analyze the complex functions of model plants and crop species at different levels. Improvements in proteomic technology regarding protein separation and detection, as well as mass spectrometry-based protein identification, have an increasing impact on the study of plant responses to salinity stress (Parker et al. 2006; Qureshi et al. 2007; Caruso et al. 2008). Proteomics has provided valuable information in various fields of plant biology. Construction of several plant protein databases is in progress for Arabidopsis, rice, maize, and some trees, where different genetic, cellular, and physiological information is available, such as expression in various organs or tissues, response to treatments, cellular localization, and genetic bases (Thiellement et al. 1999). Recent advances in MS techniques will facilitate protein identification so that in the future this will not be a limiting factor in the interpretation of variations detected on 2D gels. By providing information on affected and unaffected proteins, large-scale protein identification will simplify determining the consequences of mutations, plant transformation, or natural polymorphism for plant metabolism, as well as interpreting the effects of protein changes on development, or in response to biotic and abiotic stress. Studies in Saccharomyces cerevisiae, for which hundreds of proteins have been identified, show the power of the proteomic approach in the study of the regulation of metabolic pathways. Schiltz et al. (2005) studied that during seed filling, the accumulation of proteins in the seeds relies on the nitrogen supply from the mother plant, and a proteomic approach was used to study the mobilization of proteins from the leaves to the filling seeds in pea. Two contrasting N-responsive wheat varieties have differential expressions of root as well as leaf

proteins when grown under controlled conditions at different N levels (Bahrman et al. 2004, 2005). These proteins were grouped into two categories, one involved in carbon metabolism and the other associated with other pathways and functions like thiol-specific antioxidant proteins, etc. This study revealed that levels of gene expressions are modified with the varying levels of nitrate supply, even if only a few polypeptides appear, disappear, or change. Sarry et al. (2006) have demonstrated the protein level changes associated with nitrogen and sulfur metabolism, and their interaction. With the help of high throughput proteomic tools, they were able to detect various enzymes including ATP sulfurylase, sulfite reductase, cysteine synthase, S adenosylmethionine synthase, glutamine synthase, aspartate aminotransferase, GDH, etc., involved directly or indirectly in S and N metabolism. Recently a study for the detection of low nitrogen-responsive proteins in cultivated rice species was done by Kim et al. (2009). Studies at constructing 2-D gel reference map for use in comparative proteomics among cultivars for N-responsive proteins might provide an insight for precise identification of potential molecular protein markers to assist the breeders for screening N-efficient genotypes and help in understanding how crop adapts to low N availability. Correlations between the level of expression and NUE might bring information on the possible role of the genes involved in nitrogen metabolism.

10 Conclusion and Future Perspectives

Present review provides an overview of plant nutriomics, which is still at a conceptual stage. Although considerable efforts are in progress with the aim at enhancing plant nutrient efficiency through molecular and genetic approaches. We have focused here largely on nitrogen with which we have been working on along molecular biology lines. Crop response to N and NUE is very low in developing countries including India. Use of NIs and slow-release nitrogen fertilizers and efficient crop and fertilizer management can significantly increase NUE. It is clearly evident that optimizing the plants, NUE goes beyond the primary process of uptake and reduction of nitrate, involving quality of events, including metabolite partitioning, secondary remobilization, C-N interactions, as well as signaling pathways and regulatory controls outside the metabolic cascades. Despite the various attempts to manipulate each of the above steps in some plant or the other, we are far from finding a universal switch that controls NUE in all plants. However, transgenic studies, QTL, and proteomics approaches seem to increasingly suggest that the enzymes of secondary ammonia remobilization are better targets for manipulation, followed by regulatory processes that control N-C flux, rather than the individual genes/enzymes of primary nitrate assimilation. There is an urgent need of large-scale, co-ordinated research on plant nutriomics, involving sincere efforts from both national and international researchers to develop the nutrient-efficient, high-yielding, and stresstolerant genotypes/varieties that will contribute to both environmental safety as well as food security worldwide.

References

- Abdin MA, Dwivedi RK, Abrol YP (2005) Nitrogen in agriculture. In: Singh RP, Shankar N, Jaiwal PK (eds) Nitrogen nutrition and plant productivity. Studium Press, Houston, TX, pp 1–41
- Abrol YP, Chatterjee SR, Kumar PA, Jain V (1999) Improvement in nitrogenous fertilizer utilization – physiological and molecular approaches. Curr Sci 76:1357–1364
- Ameziane RK, Bernhard K, Bates R, Lightfoot D (2000) Expression of the bacterial gdhA gene encoding NADPH glutamate dehydrogenase in tobacco affects plant growth and development. Plant Soil 221:47–57
- Andrews M, Lea PJ, Raven JA, Lindsey K (2004) Can genetic manipulation of plant nitrogen assimilation enzymes result in increased crop yield and greater N-use efficiency? An assessment. Ann Appl Biol 145:25–40
- Anjana US, Iqbal M, Abrol YP (2007) Are nitrate concentrations in leafy vegetables within safe limits? Curr Sci 92:355–360
- Aslam M, Travis RL, Huffaker RC (1993) Comparative induction of nitrate and nitrite uptake and reduction systems by ambient nitrate and nitrite in intact roots of barley (*Hordeum vulgare* L.) seedlings. Plant Physiol 102:811–819

- Aulakh MS, Khera TS, Doran JW, Bronson KF (2001) Denitrification, N₂O and CO₂ fluxes in rice-wheat cropping system as affected by crop residues, fertilizer N and legume green manure. Biol Fertil Soils 34:375–389
- Bahrman N, Le Gouis J, Negroni L, Amilhat L, Leroy P, Lainé A-L, Jaminon O (2004) Differential protein expression assessed by two-dimensional gel electrophoresis for two wheat varieties grown at four nitrogen levels. Proteomics 4:709–719
- Bahrman N, Gouy A, Devienne-Barret F, Hirel B, Vedele F, Le Gouis J (2005). Differential change in root protein patterns of two wheat varieties under high and low nitrogen nutrition levels. Plant Sci 168:81–87
- Beman JM, Arrigo K, Matson PM (2005) Agricultural runoff fuels large phytoplankton blooms in vulnerable areas of the ocean. Nature 434:211–214
- Caruso G, Cavaliere C, Guarino C, Gubbiotti R, Foglia P, Laganà A (2008) Identification of changes in Triticum durum L. leaf proteome in response to salt stress by two-dimensional electrophoresis and MALDI-TOF mass spectrometry. Anal Bioanal Chem 39:381–390
- Carvalho HG, Lopes-Cardoso IA, Lima LM, Melo PM, Cullimore JV (2003) Nodule-specific modulation of glutamine synthetase in transgenic *Medicago truncatula* leads to inverse alterations in asparagine synthetase expression. Plant Physiol 133:243–252
- Choudhary OP, Bajwa MS, Josan AS (2003) Fertilizer management in salt-affected soils – a review. J Res (PAU) 40(2):153–171
- Das A, Prasad M, Shivay YS, Subha KM (2004) Productivity and sustainability of cotton (*Gossypium hirsutum* L.)–wheat (*Triticum aestivum* L.) cropping system as influenced by prilled urea, farmyard manure and *Azotobacter*. J Agron Crop Sci 190:298–304
- Delgado JA, Sparks RT, Follett RF, Sharkoff JL, Riggenbach RR (1999) Use of winter cover crops to conserve soil and water quality in the San Luis Valley of south central Colorado. In: Lal R (ed) Soil quality and soil erosion. CRC Press, Boca Raton, FL, pp 125–142
- Delgado JA, Riggenbach RR, Sparks RT, Dillon MT, Kawanabe LM, Ristau RJ (2001) Evaluation of nitratenitrogen transport in a potato-barley rotation soil science. Soc Am J 65:878–883
- Dubois F, Terce-Laforgue T, Gonzalez-Moro MB, Estavillo JM, Sangwan R, Gallis A, Hirel B (2003) Glutamae dehydrogenase in plants: is there a new story for an old enzyme? Plant Physiol Biochem 41:565–576
- FAI (2008) Fertiliser statistics 2007–08. Fertiliser Association of India, New Delhi
- Forde BG (2000) Nitrate transporters in plants: structure, function and regulation. Biochim Biophys Acta 1465:219–235
- Forde BG (2002) Local and long-range signalling pathways regulating plant responses to nitrate. Annu Rev Plant Biol 53:203–224
- Fuentes SI, Allen DJ, Ortiz-Lopez A, Hernández G (2001) Over-expression of cytosolic glutamine synthetase

increases photosynthesis and growth at low nitrogen concentrations. J Exp Bot 52:1071–1081

- Gallardo F, Fu J, Canton FR, Garcia-Gutierez A, Canovas FM, Kirby EG (1999) Expression of a conifer glutamine synthetase gene in transgenic poplar. Planta 210:19–26
- Galloway JN (1995) Acid deposition: perspectives in time and space. Water Air Soil Pollut 85:15–24
- Glass ADM, Siddiqi MY (1995) Nitrogen absorption by plant roots. In: Srivastava HS, Singh RP (eds) Nitrogen nutrition in higher plants. Associated Publishers, New Delhi, pp 21–56
- Glass ADM, Britto DT, Kaiser BN, Kinghorn JR, Kronzucker J, Kumar A, Okamoto M, Rawat S, Siddiqi MY, Unckles SE (2002) The regulation of nitrate and ammonium transport systems in plants. J Exp Bot 53:855–864
- Habash DZ, Massiah AJ, Rong HL, Wallsgrove RM, Leigh RA (2001) The role of cytosolic glutamine synthetase in wheat. Ann Appl Biol 138:83–89
- Harrison GA, Tricarico JM, Meyer MD, Dawson KA (2007) Effects of optigen on fermentation, digestion, and N partitioning in rumen-simulating fermenters. J Anim Sci 85:98–105
- Havangi GV, Hegde BR (1983) Response of pearl millet and finger millet crop to fertilizer management under rainfed conditions. Fertil News 28:62–66
- Hirel B, Lea PJ (2001) Ammonium assimilation. In: Lea PJ, MorotGaudry JF, eds. Plant nitrogen. Berlin: Springer-Verlag, pp 79–99
- Hirel B, Bertin P, Quillere I, Bourdoncle W, Attagnant C, Dellay C, Gouy A, Cadiou S, Retailliau C, Falque M, Gallis A (2001) Towards better understanding of the genetic and physiological basis for nitrogen use efficiency in Maize. Plant Physiol 125:1258–1270
- Hirel B, Le Gouis J, Ney B, Gallais A (2007) The challenge of improving nitrogen use efficiency in crop plants: towards a more central role for genetic variability and quantitative genetics within integrated approaches. J Exp Bot 58:2369–2387
- Ishimaru K, Kobayashi N, Ono K, Yano M, Ohsugi R (2001) Are contents of Rubisco, soluble protein and nitrogen in flag leaves of rice controlled by the same genetics? J Exp Bot 52:1827–1833
- Kim DH, Shibato J, Kim D-W, Oh MK, et al (2009) Gelbased proteomics approach for detecting low nitrogenresponsive proteins in cultivated rice species. Physiol Mol Bio Plants 15:31–41
- Kissel J, Krueger FR, Silen J, Clark BC (2004) The cometary and interstellar dust analyzer at comet 81P/ Wild 2. Science 304:1774–1776
- Kozaki A, Takeba G (1996) Photorespiration protects C3 plants from photooxidation. Nature 384:557–560
- Krapp A, Fraisier V, Scheible WR, Quesada A, Gojon A, Stitt M, Caboche M, Daniel-Vedele F (1998) Expression studies of *Nrt2:1Np*, a putative high affinity nitrate transporter: evidence for its role in nitrate uptake. Plant J 14:723–731
- Kumar PA, Parry MAJ, Mitchell RAC, Ahmad A, Abrol YP (2002) Photosynthesis and nitrogen use-efficiency.

In: Foyer CH, Noctor G (eds) Photosynthetic nitrogen assimilation and associated carbon and respiratory metabolism. Kluwer, Dordrecht, pp 23–34

- Ladha JK (2005) Improving the recovery efficiency of fertilizer nitrogen in cereals. J Indian Soc Soil Sci 53:472–483
- Lam HM, Hsieh MH, Coruzzi G (1998) Reciprocal regulation of distinct asparagine synthetase genes by light and metabolites in Arabidopsis thaliana. Plant J 16:345–353
- Lam QD, Schmalz B, Fohrer N (2009) Ecohydrological modelling of water discharge and nitrate loads in a mesoscale lowland catchment, Germany. Adv Geosci 21:49–55
- Lea PJ, Ireland RJ (1999) "Nitrogen Metabolism in higher plants" in "Plant Amino Acids" edited by B. Singh. Marcel Dekker publ pp 1–47
- Lea US, Leydecker MT, Quilleré I, Meyer C, Lillo C (2006) Posttranslational regulation of nitrate reductase strongly affects the levels of free amino acids and nitrate, whereas transcriptional regulation has only minor influence. Plant Physiol 140:1085–1094
- Lejay L, Tillard P, Lepetit M, Olive F, Filleur S, Daniel-Vedele F, Gojon A (1999) Molecular and functional regulation of two NO₃⁻ uptake systems by N- and C-status of Arabidopsis plants. Plant J 18:509–519
- Li W, Wang Y, Okamoto M, Crawford NM, Siddiqi MY, Glass ADM (2007) Dissection of the AtNRT2.1:AtNRT2.2 inducible high affinity nitrate transporter gene cluster. Plant Physiol 143:425–433
- Lin TF, Demain AL (2006) Negative effect of ammonium nitrate as nitrogen source on the production of watersoluble red pigments by *Monascus* sp. Appl Microbiol Biotechnol 43:701–705
- Liu LH, Ludewig U, Frommer WB, von Wirén N (2003) AtDUR3 encodes a new type of high-affinity urea/H⁺ symporter in Arabidopsis. Plant Cell 15:790–800
- London JG (2005) Nitrogen study fertilizes fears of pollution. Nature 433:791
- Man HM, Boriel R, El-Khatib R, Kirby EG (2005) Characterization of transgenic poplar with ectopic expression of pine cytosolic glutamine synthetase under conditions of varying nitrogen availability. New Phytol 167:31–37
- Martre P, Porter JR, Jamieson PD (2003) Modeling grain nitrogen accumulation and protein composition to understand the sink/source regulations of nitrogen remobilization for wheat. Plant Physiol 133:1959–1967
- Masclaux C, Quillere I, Gallis A, Hirel B (2001) The challenge of remobilization in plant nitrogen economy. A survey of physio-agronomic and molecular approaches. Ann Appl Biol 138:69–81
- Miffin BJ, Habash DZ (2002) The role of glutamine synthetase and glutamate dehydrogenase in nitrogen assimilation and possibilities for improvement in nitrogen utilization of crops. J Exp Bot 53:979–987
- Migge A, Carrayol E, Hirel B, Becker TW (2000) Leaf specific overexpression of plastidic glutamine synthetase stimulates the growth of transgenic tobacco seedlings. Planta 2:252–260

- Narang RS, Gill MS, Gosal KS, Mahal SS (2000) Irrigation and N-fertilizer requirements for maximum yield potential of wheat. J Res 37:20–27
- Obara M, Kajiura M, Fukuta Y, Yano M, Hayashi M, Yamaya T, Sato T (2001) Mapping of QTLs associated with cytosolic glutamine synthetase and NADHglutamate synthase in rice (Oryza sativa L.). J Exp Bot 52:1209–1217
- Oliveira IC, Brears T, Knight TJ, Clark A, Coruzzi GM (2002) Overexpression of cytosolic glutamine synthetase. Relation to nitrogen, light, and photorespiration. Plant Physiol 129:1170–1180
- Ortega JL, Temple SJ, Sengupta GC (2001) Constitutive overexpression of cytosolic glutamine synthetase (GS₁) gene in transgenic alfalfa demonstrates that GS₁ may be regulated at the level of RNA stability and protein turnover. Plant Physiol 126:109–121
- Paramasivam S, Alva AK, Fares A, Sajwan KS (2002) Fate of nitrate and bromide in an unsaturated zone of a sandy soil under citrus production. J Environ Qual 31:671–681
- Parker R, Flowers TJ, Moore AL, Harpham NVJ (2006) An accurate and reproducible method for proteome profiling of the effects of salt stress in the rice leaf lamina. J Exp Bot 57:1109–1118
- Pathak RR, Ahmad A, Lochab S, Raghuram N (2008) Molecular physiology of plant nitrogen use efficiency and biotechnological options for its enhancement. Curr Sci 94:1394–1403
- Prasad R (1998) Fertilizer urea, food security and the environment. Curr Sci 75:677–683
- Prasad R (1999) Text book of rice agronomy. In: Prasad R (ed) Nutrient management. Jain Brothers, New Delhi, pp 99–130
- Prasad R (2005) Research on nitrification inhibitor and slow release nitrogen fertilizer in India – a review. Proc Natl Acad Sci USA 75:149–157
- Prasad R, Power JF (1995) Nitrification inhibitor for agriculture, health and the environment. Adv Agron 54:233–281
- Prasad M, Prasad R (1996) Response of various levels of nitrogen, zinc and magnesium on seed cotton yield. Indian J Plant Physiol 1:286–287
- Prasad R (2004) Integrated nitrogen management for sustained productivity, health and environment. Natl Symp Res Conserv and Agri Producti. (PAU, Ludiana November 22–25, 2004) extended summaries pp 299–300
- Qureshi MI, Qadir S, Zolla L (2007) Proteomics-based dissection of stress-responsive pathways in plants. J Plant Physiol 164:1239–1260
- Raghuram N, Sopory SK (1995) Light regulation of NR gene expression: mechanism and signal-response coupling. Physiol Mol Biol Plants 1:103–114
- Raghuram N, Pathak RR, Sharma P (2006) Signalling and the molecular aspects of N-use efficiency in higher plants. In: Singh RP, Jaiwal PK (eds) Biotechnological approaches to improve nitrogen use efficiency in plants. Studium Press, Houston, TX, pp 19–40
- Raun and Johnson (1999) Improving nitrogen use efficiency for cereal production. Agron J 91:357–363

- Sachdev MS, Sachdev P, Deb DL (2000) Fate of fertilizer on irrigated wheat and its pollution potential. In: Optimizing nitrogen fertilizer application to irrigated wheat. IAEA-TECDOC-1164, IAEA, Vienna, pp 129–142
- Sangwan PS, Kumar V, Singh JP, Dahiya SS (2004a) Ammonia volatilization losses from surface applied urea in saline soils. Ann Biol 20:157–160
- Sangwan PS, Singh JP, Kumar V, Mehta SS (2004b) Ann Biol 20:153–156
- Sarkar R (2003) Nitrate-induced gene expression and metabolic regulation in rice. M.Sc. thesis. University of Mumbai, Mumbai
- Sarkar R (2005) Assessment of sustainability of ricewheat system under different combinations of tillage, crop residue and fertilizer nitrogen applications using crop simulation model. Ph.D. thesis. Indian Institute of Technology, Kharagpur, West Bengal
- Sarkar MC, Banerjee NK, Rana DS, Uppal KS (1991) Field measurements of ammonia volatilization losses of nitrogen from urea applied to wheat. Fertil News 36(11):25–28
- Sarry JE, Kuhn L, Lay PL, Garin J, Bourguignon J (2006) Dynamics of Arabidopsis thaliana soluble proteome in response to different nutrient culture conditions. Electrophoresis 27:495–507
- Schiltz S, Munier-Jolain N, Jeudy C, Burstin J, Salon C (2005) Dynamics of exogenous nitrogen partitioning and nitrogen remobilization from vegetative organs in pea revealed by ¹⁵N in vivo labeling throughout seed filling. Plant Physiol 137:1463–1473
- Shrawat AK, Good AG (2008) Genetic engineering approaches to improving nitrogen use efficiency. ISB News Report, May 2008
- Sihag D, Singh JP (1997) Effect of organic materials on ammonia volatilization losses from urea under submerged condition. Indian Soc Soil Sci 45:822–825
- Singh Y, Singh B (2001) Efficient management of primary nutrients in the rice-wheat system. J Crop Prod 4:23–86
- Singh JP, Dahiya DJ, Kumar V (1995a) Effect of water depth and initial soil water content on the volatilization losses of ammonia. Indian Soc Soil Sci 43:177–182

- Singh JP, Kumar V, Singh M (1995b) Effect of different nitrogen carriers and methods of urea application on ammonia volatilization losses in soils. Indian Soc Soil Sci 43:269–271
- Singh Y, Singh B, Gupta RK, Khind CS, Ladha JK (2003) Managing pressmud cake for nitrogen and phosphorus nutrition of crops in a rice-wheat rotation. Int Rice Res Notes 28(1):59–61
- Stulen I, Perez-Soba M, De Kok LJ, Van Der Eerden L (1998) Impact of gaseous nitrogen deposition on plant functioning. New Phytol. 139:61–67
- Thiellement H, Bharmann N, Damerval C (1999) Proteomics for genetic and physiological studies in plants. Electrophoresis 20:2013–2026
- Vidmar JJ, Zhuo D, Siddiqi MY, Schjoerring JK, Touraine B, Anthony DMG (2000) Regulation of high-affinity nitrate transporter genes and high-affinity nitrate influx by nitrogen pools in roots of barley. Plant Physiol 123:307–318
- Vincentz M, Moureaux T, Leydecker MT, Vaucheret H, Caboche M (1993) Regulation of nitrate and nitrite reductase expression in *Nicotiana plumbaginifolia* leaves by nitrogen and carbon metabolites. Plant J 3:315–324
- Wood CW, Marshall SB, Cabrera ML (2000) Improved method for field scale measurement of ammonia volatilization. Commun Soil Sci Plant Anal 31:581–590
- Yamaya T (2003) Biotechnological approaches for modification of nitrogen assimilation in rice. In: Jaiwal PK, Singh RP (eds) Plant gernetic engineering, vol 2: Improvement of food crops. SciTech Publishing, Houston, TX, pp 79–88
- Yamaya T, Obara M, Nakajima H, Sasaki S, Hayakawa T, Sato T (2002) Genetic manipulation and quantitativetrait loci mapping for nitrogen recycling in rice. J Exp Bot 53:917–925
- Yan JB, Tang H, Huang YQ, Zheng YL, Li JS (2006) Quantitative trait loci mapping and epistatic analysis for grain yield and yield components using molecular markers with an elite maize hybrid. Euphytica 149:121–131

Arbuscular Mycorrhizal Symbiosis and Other Plant– Soil Interactions in Relation to Environmental Stress

11

Patrick Audet

Abstract

In this chapter, focused on the arbuscular mycorrhizal (AM) fungi and their mostly mutualistic association with the vast majority of herbaceous plant species, we examine the cellular, molecular, and physiological mechanisms by which the mycorrhizal symbiosis can enhance plant stress tolerance in relation to a number of abiotic environmental stressors, such as macro- and micronutrient deficiency, drought, and metal toxicity. Overall, the primary mechanisms of interaction discussed here include: (1) the enhanced uptake of macro- and micronutrients and water; and (2) the stabilization of the soil architecture via mycorrhizal-enhanced soil aggregation and metal biosorption processes. A key facet of this analysis involves the identification of direct vs. indirect benefits of interactions, and their distinctive impacts toward plant development as well as the proximal growth environment. Accordingly, due to the significant and widespread effects of these direct and indirect processes toward plant physiological and soil ecological function, it is suggested that the mycorrhizal symbiosis should constitute an extrinsic stress tolerance strategy that could complement the inherent resistance mechanisms of plants when subjected to an array of potential stressors, and also buffer the growth environment. For this reason, it is recommended that future studies take into account such multitrophic interactions (e.g., above- and belowground relationships) to better depict physiological and ecological phenomena in relation to environmental stress.

Keywords

Mutualism • Macro- and micronutrients • Drought • Soil stabilization

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_11, © Springer Science+Business Media, LLC 2012

P. Audet (🖂)

Centre for Mined Land Rehabilitation, Sustainable Minerals Institute, The University of Queensland, Brisbane, QLD 4072, Australia e-mail: p.audet@uq.edu.au

1 Introduction

1.1 Intrinsic vs. Extrinsic Stress Tolerance

In the field of plant physiology, *environmental stress* (or *strain*) refers to a state or event causing a deviation in plant growth relative to its optimal

development (Larcher 1987, 2003). As shown in other chapters of this volume, potential environmental stress factors are many (e.g., radiation, temperature, water, nutrient availability) and often grouped according to their abiotic or biotic origins (Fig. 11.1). From this definition, such stressors can be either localized (root vs. shoot predation) or systemic (heat shock, frost), and typically incur a complex cascade of physiological effects ranging



Fig. 11.1 Abiotic and biotic environmental stress factors (adapted from Marschner 1995)

from temporary (reversible) to permanent (irreversible) adaptive responses depending on the relative intensity of the given stress factor. In order to increase their survivorship and reproductive success, plants have developed a remarkable array of stress tolerance (endurance), resistance (acclimation), and (or) avoidance (prevention) mechanisms to circumvent a number of environmental challenges and avoid any permanent associatedstress injuries. Further to these more conventional descriptors of adaptive stress tolerance (e.g., tolerance, resistance, and avoidance), two alternative descriptors have been proposed relating to plant investment (or resource and energy allocation) in either intrinsic (e.g., metabolic) or extrinsic (e.g., symbiotic) tolerance strategies1 (Audet and Charest 2007b, 2008). Here, intrinsic stress tolerance refers to plant investment in inherent (or built-in) metabolic systems that are inducible when subjected to stress. For example, the production of metallothioneins which bind ion-free radicals to prevent metal-induced cellular oxidative stress (Chaps. 9, 20, and 21), the production of secondary metabolites to thwart herbivores (Howe and Jander 2008; Mole 1994), the dispatch of heat shock proteins to prevent enzyme denaturing under temperature extremes (Chaps. 5-7), or the production of anti-microbial proteins to inhibit viral, fungal, and (or) bacterial pathogens (Dangl and Jones 2001; Fritig et al. 1998; Ganz and Lehrer 1999), to name just a few. More broadly, intrinsic stress tolerance can also include constitutive systems such as the processes of cell lignification, the development of trichomes and glandular hairs, or the exudation of resins and waxes which, together, offers mechanical defenses to some of these environmental stressors (Bhuiyan et al. 2009;

Duke 1994; Langenheim 2003; Wagner 1991). By contrast, extrinsic stress tolerance refers to plant investment in external systems, particularly symbiotic mutualism, to circumvent environmental stress. In this regard, the symbiotic mutualism encompasses an intimately co-operative relationship between two different species (referred to as symbionts) contributing to the mutual benefit of both individuals (Leung and Poulin 2008). Notably, the balance between the benefits of association among the symbionts is critical for defining the symbiotic mutualism since symbiotic relationships are believed to function along a continuum ranging from parasitism to mutualisms (Boucher et al. 1982; Bronstein 2001; Johnson et al. 1997). From this definition, it can be argued that plants have developed the widest assortment of mutualism in the natural world, whereby host plants typically exchange essential resources (e.g., plant carbohydrates, soil nutrients) and (or) ecological services (e.g., pollination services, shelter) with individuals from another species to reciprocally enhance their tolerance to environmental stressors and thereby increase their survivorship (Boucher 1988; Boucher et al. 1982). Examples of plant mutualism include plant-pollinator interactions with insects, birds, and mammals to ensure plant fertilization and sexual reproduction (Rønsted et al. 2005, 2008; Wiebes 1979), plantfungus (mycorrhizal) interactions to increase the root system's resource acquisition capacity (Douds and Johnson 2007; Marschner and Dell 1994), and plant-rhizobial interactions for the fixation of inorganic soil nitrogen (Long 1996, 2001; Young and Johnston 1989). Among these interactions, the mycorrhizal symbiosis is considered to be one of the most widespread and well-studied ecological associations having key implications at the scale of plant physiological and whole-ecosystem function. In this chapter, focused on the arbuscular mycorrhizal (AM) fungi and their symbiotic association with herbaceous plant species, we examine how some plants invest in mycorrhizal symbiosis as an extrinsic stress tolerance strategy in relation to a number of abiotic environmental stressors, such as macro- and micronutrient deficiency, drought, and metal toxicity. It is also suggested that such an investment can contribute

¹In their review of "Heavy metal tolerance in plants," Antonovics et al. (1971), and later Baker and Walker (1990), were first to allude to and distinguish between internal and external tolerance mechanisms. Similarly to the investment of plant resources toward intrinsic versus extrinsic strategies proposed here, the internal and external tolerance mechanisms suggested by Antonovics et al. refer to central (e.g., metabolic) and peripheral (e.g., rhizospheric) processes, respectively, which can impact plant development when faced with critical metal toxicity conditions.



Fig. 11.2 Defining the mycorrhizosphere and its zone of influence (adapted from Beare et al. 1995)

in shaping plant development by influencing edaphic conditions within the proximal growth environment.

1.2 Mycorrhizae and the AM Symbiosis

Aptly referred to as "fungus roots" (Frank 1885), the mycorrhizae are mostly nonpathogenic soil fungi living intimately with terrestrial plant roots that, together, form a symbiotic mutualism characterized by the direct exchange of plant carbohydrates for soil resources, such as mineral nutrients and water (Allen 1991). The mycorrhizae are ubiquitous organisms having adapted to and successfully colonized nearly all known terrestrial ecosystems by forming symbiotic associations with the broad majority of all plant families (Peterson et al. 2004). For this reason, the mycorrhizal fungi are classified into three primary assemblages depending on their respective morphologies and specific plant hosts: the ectomycorrhizae (primarily associated with Pinaceae, Fagaceae, Betulaceae, and Salicaceae species), the endomycorrhizae (associated with the majority of angiosperms and some gymnosperms), and the ectendomycorrhizae (primarily associated with

Orchidaceae and Ericaceae species). A common feature among all mycorrhizae is the development of the mycorrhizosphere² (Fig. 11.2) consisting in the combined zones of influence of the roots (rhizosphere) and extraradical hyphae (hyphosphere), and encompassing a highly active and multilateral interface between the host plants, mycorrhizal fungi, and proximal soil environment (Duponnois et al. 2008; Garbaye 1991). In line with the notion of plant investment in extrinsic systems for the purpose of stress tolerance, plant investment toward the development of the mycorrhizospheric network involves a considerable plant carbon allocation (occasionally representing up to and possibly well over 20% of the plant's total carbon budget) which is required for

²Fitzpatrick's (1984) characterization of the micromorphology of soils indicates that the pedosphere (e.g., soil realm) is constituted of four essential "spheres": atmosphere (e.g., soil air), biosphere (e.g., litter and microorganisms), lithosphere (e.g., rocks and minerals), and hydrosphere (e.g., soil water). Accordingly, Beare et al. (1995) have subclassified arenas of interaction to identify further interfaces within the pedospheric framework, such as the drilosphere (e.g., worm castings), detritusphere (e.g., saprotrophs), rhizosphere (e.g., plant roots), mycorrhizosphere (e.g., combined roots and extraradical hyphae), etc.

actively sustaining the symbiotic infrastructure and maintaining the functional viability of the mycorrhizal symbiont (Douds et al. 2000; Tinker et al. 1994). In exchange, this extrinsic investment provides the host plant with a number of ecological services typically pertaining to the enhancement of the plant's resource acquisition capability and the stabilization of the proximal soil environment.

Falling within the grouping of the endomycorrhizae, the AM fungi and their symbiosis with herbaceous plants are particularly well studied in the field of plant physiology and mycology, and widely recognized for benefiting host plants when subjected to various environmental stressors (Table 11.1). Having originated an estimated 450 million years ago, the AM fungi comprise species of the Glomeromycota phylum which are believed to form associations with up to 90% of all herbaceous plants (Redecker et al. 2000; Remy et al. 1994; Schüßler et al. 2001). As characterized by a unique morphology consisting of intra- and extraradical hyphae, arbuscules, vesicles, and spores (Fig. 11.3), the AM fungi behave in a peculiar manner compared to other mycorrhizal phyla since they penetrate between the cortical cells of vascular plant roots in order to develop an intracellular exchange network (Garcia-Garrido et al. 2009): a behavior believed to be rooted in ancient parasitic origins (Purin and Rillig 2008). Following reciprocal signaling processes between the symbionts resulting in the successful colonization of host roots (Harrison 1999, 2005; Vierheilig and Piché 2002), the intraradical hyphae proliferate within the root architecture to interact with roots cells across a slender periarbuscular zone formed between the fungal arbuscular structures and plant cell membranes. It is across the periarbuscular zone where soil resources (e.g., phosphorus, nitrogen, or mineral nutrients) and plant carbohydrates (e.g., glucose, hexose) are actively exchanged between the symbionts (Hahn and Mendgen 2001). Meanwhile, the extraradical hyphae typically scavenge beyond the root depletion zone to form an expansive mycorrhizospheric network, thereby increasing the host roots' resource acquisition capabilities and zone of influence compared to the rhizosphere alone (Koide 1991, 2000; Koide and Elliott 1989). With the development of this active and bidirectional symbiotic exchange network, the AM fungi then shift their developmental allocation to the production of extraradical spores and vesicles which are involved, respectively, in fungal reproduction and lipid storage (Bago et al. 2000; Dalpé et al. 2005). Under these circumstances, the AM fungi are generally considered to be "true" mutualists due to their *host obligate* status which requires that they maintain an active symbiosis in order to ensure an influx of plant carbon allocations for the completion their life cycle (Johnson et al. 1997; Jones and Smith 2004).

As stated previously, numerous advances have been made over the past decades demonstrating the beneficial role of the AM fungi in plant physiology and soil ecology; this being attributed especially to the dynamic function of the mycorrhizosphere in relation to various edaphic processes (Fig. 11.4). In order to accurately depict such dynamic interactions, the classification of the mycorrhizospheric processes presented here distinguishes specifically between two types of interaction depending on the nature of the benefits of association being either direct or indirect. In the present context, the direct benefits of interaction refer to processes that directly enhance the plant health status as mediated by the dynamics of bidirectional exchange between the symbionts described above (Cushman and Beattie 1991; Schwartz and Hoeksema 1998). For example, the process of mycorrhizal enhanced uptake in which the extraradical hyphae increase the uptake of limiting soil resources in exchange for plant carbohydrates, then enabling the host plant to supplement its nutrient status when subjected to deficiency conditions. Alternatively and occasionally overlooked from a plant physiological perspective, the indirect benefits³ of interaction refer to processes that indirectly enhance the plant growth or survival status by altering the proximal growth environment thereby providing

³The notion of "indirect benefits" derived from mutualism has previously been used within the context of species community structure.

Table 11.1 Sun	nmary of the impact of AM syn	abiosis on plant physiology	and soil ecology	
Mechanism		Target	Description	Reference
Direct benefit	Enhanced resource acquisition capability	Essential soil resources (nonmetals, metals, and water)	Preferential uptake of nitrogen (NO_3/NH_4) and phosphorus (P_i)	Bolan (1991) ^a , Chapman et al. (2006) ^a , Schachtman et al. (1998) ^a , Smith et al. (2003) ^a , and George et al. (1995) ^a
			Mobilization and uptake of trace essential elements having low bioavailability (e.g., Zn, Ni, Co, Cu, Mn, Fe) particularly under nutrient-deficiency conditions	Jeffries et al. $(2003)^a$, Koide $(1991)^a$, and Marschner and Dell $(1994)^a$
			Enhanced water use efficiency and drought recovery	Augé et al. (2001) , Augé $(2001)^a$, and Montaño et al. $(2007)^a$
Indirect benefit	Soil structure stabilization	Metal bioavailability	Metal binding due to negatively charged surface constituents of extraradical hyphae (e.g., carboxyls, hydroxides, oxy-hydroxides, sulfhydryls); Reduction of plant metal uptake to delay phytotoxicity, particularly at high soil exposure levels (e.g., Zn Pb, Cd, Ni)	Leyval et al. $(1997)^{\mu}$, Galli et al. $(1994)^{\mu}$, Gadd $(1993)^{\mu}$, Meharg $(2003)^{\mu}$, and González-Guerrero et al. $(2009)^{\mu}$
		Soil retention capacity	Enhanced soil aggregation properties; increased water and nutrient retention capacity, decrease of nutrient leaching	Augé et al. (2001), Augé (2001) ^a , Bearden (2001), Bearden and Petersen (2000), Miller and Jastrow (1990)
	Mycorrhizal exudation	Soil microbia	Promotion of beneficial bacteria (i.e., nitrogen fixation), competitive exclusion of soil pathogens	Chapman et al. $(2005)^a$, Barea et al. $(1998)^a$, and Johanson et al. $(2003)^a$
		Metal bioavailability	Precipitation of metal-ligands, modulation of soil pH	Leyval et al. $(1997)^a$, Galli et al. $(1994)^a$, and Gadd $(1993)^a$
^a Denotes review	publications			

5 1 LANA F C 11ΰ hlo 11 1



Fig. 11.3 Defining structures of an arbuscular mycorrhizal fungus (*Glomus intraradices* Schenck & Smith) in association with Ri T-DNA carrot roots (*Daucus carota* L.) grown under aseptic conditions. Shown from (a) to (e) are: vesicles

(Ve), arbuscules (Ar), host roots (HR), spores (Sp), spore clusters (Sc), extraradical hyphae (Eh), and intraradical hyphae (Ih). Roots are stained with an aniline blue 0.02% dye solution and observed under a compound microscrope

more favorable developmental conditions (Bertness and Callaway 1994; Stachowicz 2001; Müller and Krauss 2005). For instance, the process of mycorrhizospheric-enhanced soil aggregation which contributes in stabilizing the proximal growth environment to increase its resource retention capacity. A key aspect of the indirect benefits of interaction is the notion that such processes can benefit host plants as well as nonassociated species within the mycorrhizosphere's zone of influence, unlike the direct benefits which suggest an intimate exchange occurring exclusively between the symbionts. By distinguishing between the direct and indirect benefits of interaction, it is intriguing that a combination of such AM-induced mycorrhizospheric processes can complement many intrinsic tolerance mechanisms when subjected to a broad


Fig. 11.4 Summary of potential mycorrhizospheric interactions

range of environmental conditions and abiotic stress. For this reason, it is considered that these processes likely play a key role in enhancing plant stress tolerance, as well as shaping the proximal growth environment to increase the soil's resilience, in relation to a number of potential ecological stressors.

2 Direct Benefits of Association

2.1 Macro- and Micronutrient Uptake

2.1.1 N Acquisition

Plant productivity in temperate agro-ecosystems is most commonly limited by soil nitrogen (N) bioavailability (Vitousek and Howarth 1991; Chapin et al. 2002). Due to its principal role in protein biosynthesis and nucleic acid metabolism, N deficiency typically results in stunted plant growth and increased leaf senescence thereby detrimentally affecting the plant's photosynthetic potential and overall growth yield. A particular environmental challenge regarding plant N assimilation exists with regard to the source of N in soils, whether it is in the form of nitrate (NO₂⁻) which is readily assimilated in roots and (or) shoots but easily leached from the rhizosphere or, instead, in the form of ammonium (NH_{4}^{+}) which can be more abundant than the former but requires detoxification prior to its assimilation (Gutschick 1981; Bloom 1997). The primary benefit of mycorrhizal associations pertains to the increase in belowground surface area (e.g., roots and extraradical hyphae) which enhances the host plant's soil resource acquisition capability compared to the rhizosphere alone. In this regard, two complementary mechanisms describing the role of AM fungi in plant N assimilation have been presented suggesting that the



Fig. 11.5 AM fungal nitrogen uptake and transfer pathway (adapted from Govindarajulu et al. 2005; modified according to Jin et al. 2005, Chalot et al. 2006, and Cruz et al. 2007). Bidirectional transfer is indicated by solid and dashed lines. Refer to text for abbreviations

mycorrhizosphere contributes first by generally increasing the plant N acquisition capability (both NO₃⁻ and NH₄⁺ sources), and second by specifically increasing the uptake of NH_{4}^{+} as a result of fungal pre-assimilation and detoxification processes (Chapman et al. 2005; Marschner and Dell 1994). Together, these mechanisms are believed to increase the bioavailability of N within and often beyond the rhizosphere's resource depletion zone to improve plant stress tolerance when subjected to N-deficiency conditions. Among others, Haystead et al. (1988), Faure et al. (1998), and Subramanian and Charest (1997, 1998, 1999) investigated these hypotheses using greenhouse experimental systems in the objective of assessing the nutritional status of plants in relation to the amendment of NO₃⁻ and NH₄⁺ fertilizers, and then comparing the N assimilation pathways of AM vs. non-AM plants by measuring the activity of key assimilation enzymes: namely, nitrate reductase (NR), glutamine synthase (GS), glutamate dehydrogenase (GDH), and glutamate synthase (GOGAT). Collectively, these studies have shown that AM-colonized rye grasses (Lolium perenne), "field" clover (Trifolium repens), and maize (Zea mays) all gained considerable increases in total nitrogen uptake leading to an overall greater amino acid composition compared to non-AM plants. Notably, these physiological effects coincided with increases in the activity of NR, GDH, and GS-GOGAT assimilation enzymes measured in the AM roots and shoots. As predicted, the AM-plant nutritional status was supplemented by increasing the overall uptake of N, especially by increasing the uptake of its less labile form, [NH₄⁺]. More recently, the studies of Toussaint et al. (2004), Govindarajulu et al. (2005), Jin et al. 2005; Chalot et al. (2006), and Cruz et al. (2007) have corroborated these general findings using in vitro culture tools to further characterize the AM fungal uptake, assimilation, and translocation pathways as soil-N travels from the extraradical hyphae to the host root (Fig. 11.5).

Accordingly, it has been elucidated that NO₃⁻ and NH⁺ are actively taken up by extraradical hyphae via transporters of the AMT/Mep/Rh protein superfamily (Khademi et al. 2004). Having reached the cytosol of the extraradical mycelium, NO_3^{-} and NH_4^{+} are converted to glutamine and then arginine via the fungal NR, GDH, and GS-GOGAT enzyme cycles. In this form, arginine travels to the intraradical mycelium through cytoplasmic streaming via coenocytic channels to be further broken down through the action of ornithine aminotransferase and urease, thereby releasing ornithine, urea, and ultimately NH_{4}^{+} . Finally, the NH_4^+ is either catabolized via AM fungal amino acid synthesis or transferred to roots across the periarbuscular interface apparently via ammonium transport proteins. To complete the bidirectional exchange between the symbionts, plant carbohydrates in the form of hexose are transferred to the AM fungus via putative transporters (HXT1, potentially among others – Hahn and Mendgen 2001) and likely mediated via plasma membrane H+-ATPases $(GmHA_{1-5} - Ferrol et al. 2000)$. Once in the fungal cytosol, the hexose is converted to trehalose, glycogen, and glucose for usage in various fungal metabolisms. As such, the characterization of the AM nitrogen uptake pathway provides key evidence as to the active role of AM fungi in supplementing the plant N nutritional status in the enhancement of plant nutrient stress tolerance.

2.1.2 P Acquisition

After soil-N, plant productivity in temperate agroecosystems is limited by phosphorus (P) bioavailability⁴: a key bioenergetic constituent (e.g., ATP) and cellular structural component (e.g., phospholipids, DNA, RNA). In this regard, P deficiency is prevalent in areas of high rainfall due to extensive nutrient leaching and (or) acidic soil conditions conducive to reciprocal antagonisms (also known as phosphorus-induced micronutrient deficiencies) which can cause leaf chlorosis and stunted growth (Cleveland et al. 2002; Mengel and Kirkby 2001). Similarly to the case of N assimilation described above, it has been hypothesized that the AM fungi hold a significant role in plant P acquisition by increasing the plant's soil resource uptake capacity, particularly by enhancing the uptake of phosphates and inorganic P which typically have slow soil diffusion rates (Marschner 1995; Picone et al. 2003; Saito 2000). In this regard, these physiological mechanisms are among the most thoroughly investigated in the study of the AM symbiosis and have been well reviewed by Bolan (1991), Koide (1991), George et al. (1995), Schachtman et al. (1998), and Smith et al. (2003). The consensus from these studies is that host plants benefit from an investment in AM symbiosis for the supplementation of their P nutritional status, especially when subjected to soil-P deficiency, due to the expansive mycorrhizosphere's ability to increase soil-surface contact and reduce soil-P diffusion distances. Consistent with the notion of plant investment in extrinsic systems to circumvent environmental stress, the mycorrhizal investment is often inversely correlated with the bioavailability of soil-P such that AM root colonization and symbiotic activity are believed to be highest under low P conditions (Smith et al. 2003, 2004). Consequently, this relationship would suggest that P bioavailability is a key factor dictating the plant's relative symbiotic investment (or mycor*rhizal responsiveness*) in order to maximize the reciprocal benefits of association (Graham et al. 1991; Janos 2007; Tawaraya 2003). Under such environmental conditions, soil-P can be actively taken up by the extraradical hyphae and efficiently transferred to host roots (Fig. 11.6 – Schachtman et al. 1998; Smith et al. 2003; Javot et al. 2007). This process is characterized by the activity of extraradical hyphae which increase the solubility of soil-P due to the exudation of organic chelators to then facilitate the uptake of both organic (P_{org}) and inorganic (P_i) forms of P. Coinciding with these events, a number of AM fungal phosphate

⁴Unlike temperate environments, plant productivity in tropical agro-ecosystems is primarily limited by phosphorus bioavailability followed by less labile soil micronutrients due to their slow diffusion rates and subsequently low bioavailability to plants. In addition, the high rates of plant photosynthesis and evapo-transpiration in this ecosystem typically cause the bioavailable nutrient pool to be rapidly assimilated (Brams 1973; Baligar and Bennett 1986; Ewel 1986).



Fig. 11.6 AM fungal phosphorus uptake and transfer pathway (adapted from Schachtman et al. 1998; modified according to Smith et al. 2003 and Javot et al. 2007). Bidirectional transfer is indicated by solid and dashed lines. Refer to text for abbreviations

transporters have been isolated from the mycelium of a number of AM fungi and found to be upregulated under P-deficiency conditions, such as: Glomus intraradices (GiPT - Maldonado-Mendoza et al. 2001), G. mosseae (GmosPT -Benedetto et al. 2005), and G. versiforme (GvPT-Harrison and van Buuren 1995). Once taken up into the cytosol of the extraradical mycelium, the Pore and Pi are converted to polyphosphate complexes (poly-P - e.g., glucose-6-phosphate) through the enzymatic activity of polyphosphate glucokinase (Capaccio and Callow 1982; Cox et al. 1980). In this more stable cytosolic form, the poly-P complexes are either stored in fungal vacuoles or transferred to the intraradical mycelium through cytoplasmic streaming via coenocytic channels: a process which is putatively linked with H⁺-ATPase co-transport (Gauthier and Turpin 1994). Here, the polyphosphate complexes can be broken down by polyphosphatases for use in AM fungal ATP, DNA, RNA, and (or) phospholipid syntheses, or transferred to the host plant across species-specific phosphate transporters in exchange for plant carbohydrates in the form of hexose. To date, advanced molecular analyses have identified an array of phosphate transporters (Table 11.2) isolated especially in the periarbuscular interface region of various model study organisms, such as barley (Hordeum vulgare), deervetches (Lotus japonica), tomato (Lycopersicon esculentum), alfalfa (Medicago truncatula), rice (Oryza sativa), potato (Solanum tuberosum), and maize (Zea mays). Accordingly, it has been reported that these species also experience an increase in phosphorus assimilation activity (e.g., acid phosphatase, alkaline phosphatase, and H+-ATPase) in their roots which can contribute in increasing P assimilation and P nutritional status following the enhanced uptake and transfer of soil-P from the extraradical hyphae to the host plant (Capaccio and Callow 1982; Dexheimer et al. 1982; Schwab et al. 1991), which was later

Taxon	Nomencla	ture			Reference
Arbuscular mycorrhizal fungi					
Glomus intraradices Schenck & Smith	GiPT				Maldonado-Mendoza et al. (2001)
<i>Glomus mosseae</i> (T.H. Nicolson & Gerd.) Gerd. & Trappe	GmosPT				Benedetto et al. (2005)
Glomus versiforme (P. Karst.) S. M. Berch	GvPT				Harrison and Van Buuren (1995)
Plant host					
Hordeum vulgare L.	HvPT ₁ HvPT ₂ HvPT ₂				Smith et al. (1999), Rae et al. (2003), and Glassop et al. (2005)
Lotus japonicus L.	LjPT ₁ LjPT ₂ LjPT ₃				Maeda et al. (2006)
Lycopersicon esculentum L.	LePT ₁ LePT ₂	LePT ₃ LePT ₄	LePT ₅		Daram et al. (1998), Liu et al. (1998a), Rosewarne et al. (1999), and Nagy et al. (2005)
Medicago truncatula L.	$\frac{MtPT_1}{MtPT_2}$ $MtPT_4$				Karandashov et al. (2004), Liu et al. (1998b), and Harrison et al. (2002)
Oryza sativa L.	OsPT ₁ OsPT ₂ OsPT ₃ OsPT ₄	OsPT ₅ OsPT ₆ OsPT ₇	OsPT ₈ OsPT ₉ OsPT ₁₀	OsPT ₁₁ OsPT ₁₂ OsPT ₁₃	Paszkowski et al. (2002) and Guimil et al. (2005)
Solanum tuberosum L.	StPT ₁ StPT ₂	StPT ₃ StPT ₄	StPT ₅		Nagy et al. (2005), Leggewie et al. (1997), Rausch et al. (2001), and Karandashov et al. (2004)
Zea mays L.	$ZmPT_1$ $ZmPT_2$ $ZmPT_3$				Glassop et al. (2005), Wright et al. (2005), and Nagy et al. (2006)

Table 11.2 Summary of known AM fungal and plant phosphate transporters (from Javot et al. 2007)

corroborated by molecular analyses of phosphate transporter activity (Karandashov and Bucher 2005). As in the case of AM–plant N uptake, the ongoing characterization of the mycorrhizal P uptake and translocation pathways provide key evidence as to the active and intricate role of AM fungi in the enhancement of plant nutrient stress tolerance.

2.1.3 Micronutrient (Metal) Uptake

Analogous to the processes of AM–plant N and P acquisition, there is a considerable body of

literature describing the beneficial role of the AM symbiosis in micronutrient uptake, particularly soil-metal⁵ deficiency conditions (Jeffries et al. 2003; Koide 1991; Marschner and Dell 1994).

⁵Macro- and micronutrients are alternatively classified according to their physicochemical properties to define them, respectively, as either non-metals (nitrogen, sulfur, phosphorus, boron, chlorine) which have a negative valence, or metals (potassium, calcium, magnesium, iron, manganese, zinc, copper, molybdenum, nickel) which have a positive valence (Foy et al. 1978; Larcher 2003).



Fig. 11.7 AM fungal micronutrient (metal) uptake and transfer pathway (adapted from Meharg 2003; modified according to Göhre and Pazkowski 2006 and González-Guerrero et al. 2009). Bidirectional transfer is indicated by solid and dashed lines. Refer to text for abbreviations

In this regard, soils of temperate environments (typically classified as alfisols and vertisols) and tropical environments (ultisols and oxisols) can suffer from suboptimal elemental compositions and (or) nutrient imbalances due to long-term weathering and soil erosion resulting in plant zinc (Zn), nickel (Ni), cobalt (Co), copper (Cu), manganese (Mn), and (or) iron (Fe) nutritional deficiencies and (or) mutual antagonisms (Blinkley and Vitousek 1989; White and Zasoski 1999). As described previously, the AM symbiosis contributes in circumventing nutrient deficiency stress by increasing nutrient bioavailability in the mycorrhizosphere due to an increased resource acquisition capability which helps in supplementing the plant nutritional status, especially poorly labile metal nutrients (Eckhard et al. 1994; Liu et al. 2000; Marschner 1998; Rengel et al. 1999; Sharma et al. 1994). A commonality in the mode-of-action seems to exist regarding the general AM-plant metal uptake pathway (Fig. 11.7) as metal nutrients are taken up,

translocated, and transferred from the extraradical hyphae to the host roots, as depicted in reviews by Meharg (2003), Göhre and Pazkowski (2006), and González-Guerrero et al. (2009). These recent studies have reported that the exudation of organic chelators by the extraradical hyphae contributes in solubilizing metal ions in the mycorrhizosphere followed by the mobilization of metal-chelator complexes across fungal transporters, such as GintZIP in Glomus intraradices (González-Guerrero et al. 2009). Free metal ions may also be taken up passively across trans-membrane ion channels depending on the soil metal concentration gradient. In the cytosol, metal ions are typically bound and (or) sequestered by metallothionein proteins or glutathione complexes. Notably, it has been shown that this course-of-action can correspond with an upregulation of GintMT₁ (encoding for fungal metallothioneins in G. intraradices - González-Guerrero et al. 2005, 2007; López-Pedrosa et al. 2006) as well as an increase in glutathione S-transferase activity (González-Guerrero et al. 2009), which could represent critical steps in limiting internal stress due to the production of reactive oxygen species in the fungal cytosol (González-Guerrero et al. 2010a). Subsequently, the production of such less reactive metal complexes enables the AM fungi to store excess metal ions into their vacuolar compartments (via GintZIP, GintABC₁, or GintZnT₁ membrane transporters), integrate them into their essential metabolic function, or transfer them to the host root across the periarbuscular interface (via MtZIP₂ or LeNramp_{1,3}) (González-Guerrero et al. 2005, 2007, 2009, 2010b). The identification of ion transporters common to both plants and fungi which are putatively involved in the regulation of metal uptake (Burleigh et al. 2003; López-Millán et al. 2004) could support the perspective that this process is actively comodulated by both symbionts and provides further evidence as to the fundamentally mutualistic nature of the association, as suggested regarding the symbiotic transfer of P. Altogether, these combined processes characterize the complex role of the mycorrhizosphere in plant mineral nutrition contributing by modulating plant nutrient uptake and thereby enhancing the host plant's physiological status compared to non-AM plants, particularly when subjected to a number of environmental stress factors.

2.2 Plant Water Relations

Besides the significant role of the AM symbiosis in plant mineral nutrition, there is a considerable body of literature describing the beneficial effects of the AM mycorrhizosphere in plant water relations across a broad range of water stress, for instance, from amply watered to droughted conditions (Augé 2001, 2004). Yet, unlike the dynamics of AM-plant nutrient acquisition described above, the specific mechanisms underlying the direct impact of the mycorrhizosphere on plant stress tolerance under such environmental conditions remain slightly ambiguous. A central question in this regard considers whether the AM symbiosis benefits plants more by enhancing their intrinsic drought resistance (i.e., survival at low internal water content) or, rather, by increasing their drought avoidance (i.e., maintenance of high internal water content) when subjected to a low external water potential, such as drought and drought recovery conditions. As outlined in recent reviews by Augé (2001, 2004), a wide array of AM-plants in association with a number of Glomus and allied AM species (refer to Augé 2001 for a more comprehensive list of plant and AM fungal species interactions) have been shown to develop a variety of beneficial physiological responses compared to non-AM plants, with such responses ranging from relatively higher stomatal conductance, leaf transpiration, and (or) osmotic potential (Allen et al. 1981; Augé 2000; Augé et al. 2003, 2007, 2008). Among AM-plants, such enhanced metabolic and physiological functions under strained water conditions have subsequently been linked to an increased photosynthetic potential due to a generally larger leaf area, relatively greater water potential (i.e., water content) in roots and shoots, and an increased overall growth status observed under both greenhouse and field conditions (Cho et al. 2006; Khalvati et al. 2005). Altogether, these physiological responses contribute to an increased AM-plant stress tolerance occurring especially, but not exclusively, during both drought and drought recovery conditions. In addition to imposing a direct physiological stress toward host plants as symptomatically expressed by a general loss of turgidity and down regulation of various essential metabolisms (Hsiao 1973; Nautiyal et al. 1994), strained water relations often also cause alterations in nutrient bioavailability in the soil solution which leads to significant nutrient imbalances and potential antagonisms between metal ions. This potential correlationality between environmental stressors (e.g., both water and nutrient deficiency) represents an important challenge faced by experimental investigators in determining the role(s) of AM fungi in plant water relations, particularly when attempting to distinguish the specific mechanisms of AM-enhanced plant stress tolerance (Koide 1993). Similar to the notion of enhanced mycorrhizospheric uptake regarding AM-plant nutrient acquisition, it has been suggested that the mycorrhizosphere should play a fundamental role in enhancing both the nutrient and water acquisition capabilities of host plants by increasing their resource acquisition pool compared to the rhizosphere alone; this, again through the general mechanism of actively scavenging the proximal soil environment for essential resources (Koide 1993). More specifically, extraradical hyphae could contribute directly in circumventing water deficiency conditions by penetrating soil micropores to improve hydraulic conductivity due to the expansive mycorrhizosphere, and thereby enhance plant stress avoidance during drought stress and drought recovery (Miller and Jastrow 1990). Accordingly, analyses comparing AM and non-AM root conductivity among plants under drought and amply watered conditions have also shown improvements in AM-plant water uptake that often coincide with increases in the host plant's mineral nutrition. Here, an improved N, P, and (or) micronutrient status could further benefit AM-plants by circumventing internal mineral deficiencies that may otherwise detrimentally affect their intrinsic drought resistance mechanisms, such as the accumulation and maintenance of high foliar concentrations of soluble sugars and free polyamines (Augé 2001, 2004). This improved mineral status has also been linked with lower amino acid accumulation in AM than non-AM plants which suggests that the former are generally less metabolically strained under these conditions (Augé 2001). The consensus from these overall findings suggests that AM colonization primarily contributes by bolstering intrinsic plant resistance mechanisms by circumventing internal deficiencies. In addition, water conductance and nutrient uptake are improved due to an increased bioavailable pool of soil resources within the mycorrhizosphere (Cho et al. 2006; Khalvati et al. 2005). Together, these mycorrhizospheric processes can increase AM-plant resilience in relation to stressful water relations by enabling a relatively more stable (or less strained) metabolic function, which is typically manifested through an increased overall photosynthetic potential and relative growth potential during drought stress and drought recovery. While these physiological

effects provide a significant environmental advantage to AM vs. non-AM plants when subjected to adverse water conditions, plants having overall increased photosynthetic activities and stomatal conductances also tend to have higher rates of evapo-transpiration. Ironically, this effect can impose further stress on the plants themselves by increasing the rate of soil-drying in the proximal soil environment. In adding further complexity to the role of the AM symbiosis in plant water relations, these plants could be more vulnerable compared to less photosynthetically active plants under such environmental conditions due to complications stemming from accelerated soil-drying. Nevertheless, investigations into the role of the mycorrhizosphere in stabilizing the proximal soil environment could shed light into these matters, especially regarding the impact of extraradical hyphae in soil aggregation and biosorption processes which can buffer a number of edaphic factors such as the water and nutrient retention capacities. In fact, such "indirect" mycorrhizospheric processes could represent equally important components of plant stress tolerance and ecosystem function compared to the "direct" processes presented here.

3 Indirect Benefits of Association

3.1 Soil Structure Stabilization

3.1.1 Soil Aggregation

Notwithstanding the direct role of the AM symbiosis in plant resource acquisition, the mycorrhizosphere also provides significant indirect benefits of interaction which can buffer and (or) stabilize the soil matrix. In this regard, processes such as mycorrhizal-induced soil aggregation and metal biosorption (Figs. 11.8 and 11.9) are considered, here, to fall within the category of *indirect benefits* since plant investment in the mycorrhizosphere can provide key ecological services which are not directly associated with the intimately co-modulated mechanism of resource exchange. Accordingly, it can also be argued that non-associated species could benefit



Fig. 11.8 Mycorrhizal-induced micro- and macroaggregate formation indicating affinities for binding soil nutrients and water. (adapted from Miller and Jastrow 1990 and Rillig and Mummey 2006)

from these processes due to their proximity to the mycorrhizosphere's zone of influence and its broad effects on the soil growth environment. Coincidentally, it has been suggested that such edaphic interactions should influence ecosystem function at various hierarchical scales due their biogeochemical implications ranging from the micro- (e.g., soil water and nutrient retention, enhanced resource acquisition) to the macroscopic levels (e.g., whole plant functionality, species abundance and distribution) (Beare et al. 1995; Rillig and Mummey 2006; Rillig et al. 2010). In this regard, and further to the role of the extraradical hyphae in directly improving nutrient uptake and hydraulic conductivity (as discussed previously), the proliferation of extraradical hyphae and their penetration into soil

micropores can significantly enhance the aggregation properties of soils to then improve their overall water and nutrient holding capacities (Piotrowski et al. 2004): a process akin to the enmeshment of roots throughout soils during rhizospheric expansion (Angers and Caron 1998). More specifically, the greater degree of mycorrhizal branching and ramification produces localized compression forces within the proximal soil environment which serve to increase the formation of micro- and macroaggregates (Fig. 11.8) (Miller and Jastrow 1990; Rillig and Mummey 2006). Consequently, mycorrhizospheric expansion can directly increase the rate of soil cluster formation and enhance the overall resilience of the soil structure in relation to stress (e.g., soil drying, flooding, compaction, nutrient leaching).



Fig. 11.9 Mycorrhizal-induced metal biosorption and metal complexation (adapted from Apak 2002, Gadd 1993, Galli et al. 1994, González-Chavez et al. 2002, and González-Guerrero et al. 2008)

Alternatively, under such conditions, bulk soils tend to have a comparatively lower particulate binding capacity and subsequently lower aggregation potential resulting in the relatively more rapid collapse of their matrix structure. In addition to the physical entanglement of soil aggregates by the mycorrhizosphere, mycorrhizal-induced soil aggregation can also be attributed to the exudation of organic acids by extraradical hyphae (Bais et al. 2006; Bertin et al. 2003; Rovira 1969). Fundamentally, these mucilage exudates - typically consisting in polysaccharides and other extra-cellular polymeric compounds - contribute to nutrient chelation for mineral solubilization within the mycorrhizosphere as well as protection of the extraradical mycelium from desiccation. Further to these essential roles, such organic exudates also adhere to soil particles and permit the physical entanglement of micro- and macroaggregates leading to the development of soil clusters within the mycorrhizosphere. Glomalin and glomalin-related soil proteins, considered to be effective biochemical markers of AM fungal growth and mycorrhizosphere development, are also believed to bind soil particles in the same manner leading to increased structure stability (Driver et al. 2005; Purin and Rillig 2007). Altogether, this improved potential for soil cluster formation can increase the incidence of micropores within the soil architecture leading to an overall increased colloidal surface area. As a result, the mycorrhizosphere can have a higher affinity for retaining water molecules as well as metal and nonmetal ions compared to bulk soil. Taking into account the processes of hyphal proliferation and chelator exudation, Augé (2004) and Rillig and Mummey (2006) have likened the mycorrhizospheric network to an essential skeletal structure and the production of mycorrhizalderived organic compounds as the "glue" which, together, contribute in holding together the soil matrix. Consequently, from a biogeochemical perspective, the mycorrhizospheric network should play central role in enhancing soil water and nutrient retention. When subjected to environmentally stressful conditions, these enhanced soil stabilization properties can significantly increase the soil's resilience to then buffer the

growth environment for plants and associated soil microorganisms. Notably, under drought stress, the development of soil aggregates increases water and nutrient retention to delay the effects of soil drying (Auge 2004; Rillig and Mummey 2006); meanwhile, these aggregates also increase water infiltration during drought recovery due to the more *hydratable* (or water stable) soil matrix (Rillig et al. 2010). This mycorrhizal-induced structural advantage benefits plant stress tolerance by increasing the soil's water storage capacity and increasing its resilience. Likewise, the increased water retention capacity within the mycorrhizosphere can also impact plant stress tolerance in relation to nutrient stress (e.g., reciprocal ion antagonisms leading to deficiency) since soil nutrient bioavailability is closely correlated with soil water potential. As a result, nutrient bioavailability may be increased within the mycorrhizosphere due to a greater retention capacity; meanwhile nutrient losses are decreased due to a reduction in leaching.

3.1.2 Metal Biosorption

The metal-binding capacity of soil is primarily dictated by its essential composition, whereby soils having a higher proportion of organic matter (e.g., humic and fluvic acids) typically tend to have a greater retention capacity and redox potential than other soil types (Bohn 1971; McBride 1994). Further to the role of the mycorrhizosphere in stabilizing the soil's structural matrix, the extraradical hyphae have also been shown to increase the biosorption potential of soils. This is attributed primarily to the preferential binding of metal ions to negatively charged mycelial and root surface constituents (Fig. 11.9), such as carboxyl, hydroxide, oxy-hydroxide, and sulfhydryl groups (Apak 2002; Gadd 1993; Galli et al. 1994; González-Chavez et al. 2002, 2008). Similar analyses of non-mycorrhizal fungi suggest that phenolic polymers and melanins should also be effective metal binding sites even among AM fungi (Baldrian 2003; Fogarty and Tobin 1996). Likewise, the exudation of organic chelators within the mycorrhizosphere (described above) has been shown to result in an enhanced binding capacity due to the formation of metal-ligand complexes and precipitates in the soil solution. For these reasons, the general processes of metal biosorption, including ion-exchange (i.e., CEC), metal complexation, and metal-ligand precipitation and crystallization occurring on and within the fungal cell wall (Gadd 1993; Galli et al. 1994), represent significant mechanisms regarding the modulation of metal bioavailability within the mycorrhizosphere. As in the case of mycorrhizalinduced soil aggregation, these enhanced metal biosorption properties can improve the soil's resilience by increasing its nutrient retention capacity, while reducing nutrient losses due to leaching (Giller et al. 1998; Leyval et al. 1997). Notably, there is considerable evidence suggesting that, when essential and nonessential metals occur at exceedingly high exposure levels representing potentially toxic metal conditions, such metal biosorption properties can significantly reduce the bioavailability of metals in the soil solution to reduce plant metal uptake and then delay the onset of metal phytotoxicity (Audet and Charest 2007b). In this regard, a wide array of plant species (refer to Audet and Charest 2007b for broad list plant species) subjected to increasingly high metal concentrations, both essential (e.g., Cu, Fe, Mn, Ni, and Zn) and nonessential elements (e.g., Cd, Co, Cr, and Pb), have repeatedly been shown to incur considerably lower (up to 50%) metal uptake among AM (Gl. caledonium, Gl. intraradices, Gl. mosseae, and a consortium of unidentified Glomus species) than non-AM plants; an effect often coinciding with an increased plant growth and (or) health status. As proposed by Leyval et al. (1997), and later Audet and Charest (2006, 2007a, b, 2008, 2009), Hildebrandt et al. (2007), and Giasson et al. 2008, these findings suggest that mycorrhizal-induced metal biosorption could represent a significant extrinsic plant stress avoidance strategy, whereby excess soil metals are bound and precipitated in the soil solution as well as sequestered in fungal tissues instead of being transferred to host roots. As such, plant investment in this extrinsic stress avoidance mechanism could complement known intrinsic plant detoxification mechanisms, for instance, metallothienin and phytochelatin metabolisms (Cobbett 2000; Cobbett and Goldsbrough 2002) by reducing



Fig. 11.10 Dynamics of AM–plant metal uptake as characterized by the mycorrhizospheric processes of enhanced uptake and metal biosorption. (adapted from Audet and Charest 2007b, 2008, 2009)

AM-plant metal uptake and subsequently reducing cellular oxidative stress and other physiological challenges associated with metal toxicity conditions. This entire perspective has been summarized by Audet and Charest (2007b, 2008, 2009) who described plant metal uptake (Fig. 11.10) and relative plant growth (Fig. 11.11) by AM and non-AM plants using a conceptual modeling strategy based on meta-analytical, in vitro, and greenhouse culture systems. When taking into account the many different roles of AM fungi in plant physiology and ecosystem function, the mycorrhizosphere is first believed to increase AM-plant metal uptake to supplement the plant nutritional status under nutrient deficiency conditions. Subsequently, the role of the mycorrhizo-



Extrinsic Metal Exposure

Fig. 11.11 AM–plant relative plant growth in relation to metal exposure corresponding to the "Dynamics of AM-plant metal uptake" (adapted from Audet and Charest 2007b, 2008, 2009)

sphere seems to shift toward the modulation of soil nutrients due to metal biosorption processes which can delay the effects of metal phytotoxicity when subjected to potentially toxic conditions. Here, the combination of the enhanced uptake and metal biosorption mechanisms occurring simultaneously and (or) independently causes a distinct metal uptake profile among AM than non-AM plants in relation to a wide range of metal exposure, albeit depending on the soil's fundamental composition and inherent properties. As such, AM-plant growth and stress tolerance are often significantly improved as evidenced by an increased growth status whether under trace or toxicity conditions. As for conditions within the proximal growth environment, the mycorrhizalinduced processes of metal biosorption and metalligand complexation are also believed to influence a number of edaphic factors, for instance, the soil pH and redox potential (Christie et al. 2004; Fourest and Roux 1992; Joner et al. 2000; Leyval et al. 1997). When considering the case of phosphorus and nitrogen acquisition by mycorrhizae, the extraradical hyphae are believed to cause the moderate alkalinization of the growth substrate due to their selective depletion of nutrients and specific exudation of organic chelators, unlike roots that tend to acidify it (Bago et al. 1996; Eckhard et al. 1995; Gahoonia and Nielsen 1992; Li et al. 1991; Rufyikiri et al. 2004). Similar outcomes have been reported by Li and Christie (2001) and Audet and Charest (2010 and unpublished results) who investigated the impact of the rhizosphere (roots), mycorrhizosphere (roots and extraradical hyphae), and hyphosphere environments (strictly extraradical hyphae) in relation to increasing soil-Zn exposure levels. Although these facets of plant-AM-soil interactions should require further and more in-depth investigation, these studies indicate preliminarily that the presence of roots and (or) extraradical hyphae should play an essential part in shaping edaphic conditions due (in part) to differential nutrient depletion zones caused by AM and non-AM plants, their exudation of organic chelators, and their retention (or biosorption) of metals within the proximal growth environment. These factors are relevant to the bioavailability of metal and non-metal nutrients in soils since the process of hyphal alkalization could favor metal biosorption and contribute in reducing metal bioavailability and toxicity, whereas root acidification may facilitate leaching by increasing metal solubility (Apak 2002; Bradl 2004; Tack et al. 1996).

3.2 Biotic Interactions

3.2.1 Biodiversity of Beneficial and Nonbeneficial Soil Microflora

Although this chapter has focused primarily on identifying the role(s) of the AM symbiosis in benefiting plant tolerance when subjected to a number of abiotic stressors (e.g., macro- and micronutrient deficiency, drought, metal toxicity), the role of the mycorrhizosphere toward biotic soil interactions is still worthy of mention - albeit discussed only briefly in the present context. In addition to shaping growth parameters such as soil nutrient bioavailability and other edaphic factors, mycorrhizal proliferation and exudation can also significantly impact belowground biodiversity by influencing soil microbial communities within the proximal soil environment (Brussaard et al. 1997; Newsham et al. 1995a, b; Wardle et al. 1998, 2004). In turn, such subsidiary mycorrhizospheric interactions can have important consequences toward aboveground species biodiversity (i.e., plant species abundance and distribution) due to the mutual feedback existing between above- and belowground symbionts (Bever 1999, 2003; Bever et al. 1997; Klironomos 2002; van der Heijden et al. 1998). As such, the AM fungi are believed to develop their own soil microflora apparently due to the exudation of organic compounds and chelators within the mycorrhizosphere (Andrade et al. 1997; Fitter and Garbaye 1994; Frey-Klett et al. 2007; Linderman 1988); this, in addition to their modulation of edaphic factors (as described previously) which can provide more favorable soil pH and nutrient bioavailability conditions for the development of such a microflora (Cavagnaro et al. 2006; Deubel and Merbach 2005; Villegas and Fortin 2001). This is not surprising considering that AM fungal spores are known to harbor their own internal bacterial flora within their sporocarps, meanwhile the extraradical hyphae maintain their own extensive array of bacterial biofilms (Andrade et al. 1997; Bianciotto and Bonfante 2002; Fitter and Garbaye 1994; Frey-Klett et al. 2007). Although the ecological role(s) of these bacteria in AM fungal development have yet to be fully understood, various in vitro analyses have demonstrated a number of interactive behaviors ranging from stimulated spore germination, induced hyphal branching, and enhanced sporulation due to the presence of volatile bacterial metabolites (Azcón-Aguilar and Barea 1992; Barea 1997; Bianciotto and Bonfante 2002; von Alten et al. 1993). In response, the AM fungi seem to reciprocate this ulterior "symbiotic" association by enriching the mycorrhizosphere environment with carbon in the form of organic exudates (Antoun and Prevost 2001; Artursson et al. 2006). For this reason, it is possible that such bacteria (also known as mycorrhiza helper bacteria - Garbaye 1994; Frey-Klett et al. 2007) and their exudates are fundamentally involved in AM fungal development and mycorrhizospheric function, and then potentially impacting plant health status due to their role as growth promoters (Bianciotto and Bonfante 2002; Bianciotto et al. 1996a, b; Söderberg et al. 2002). In terms of their potential effects on plant stress tolerance and extrinsic growth conditions within the mycorrhizosphere, the mycorrhiza helper bacteria are generally considered to influence edaphic parameters similar to the extraradical hyphae themselves: for instance, by increasing the solubilization of soil nutrient via chelation and (or) contributing in the modulation of nutrient bioavailability via metal biosorption and soil aggregation processes (Cavagnaro et al. 2006; Deubel and Merbach 2005; Villegas and Fortin 2001). Accordingly, these mechanisms should benefit plant stress tolerance by indirectly improving nutrient bioavailability and uptake, particularly when subjected to suboptimal soil nutrient conditions. Moreover, the AM fungi could also benefit plant stress tolerance in relation to biotic stressors by inhibiting and (or) impeding soilborne pathogens (Azcón-Aguilar and Barea 1996; Barea et al. 1998; Newsham et al. 1995a, b). In this regard, St. Arnaud and Elsen (2005) have meticulously summarized the interaction outcomes of a number of AM fungi cultured under in vitro conditions in the presence of soil bacteria (Table 11.3), other soil fungi (Table 11.4), and nematodes (Table 11.5). This meta-analysis as to the impact of the AM fungi on soil-borne pathogens, and vice versa, suggests that their interaction can be highly variable (i.e., having positive, neutral, or negative outcomes under experiment conditions); however, the AM fungi are still widely recognized for benefiting plant growth in relation to a number of highly persistent and destructive pathogens (Azcón-Aguilar and Barea 1996; Barea et al. 1998; Newsham et al. 1995a, b). Harrier and Watson (2003, 2004) as well as Mukerji and Ciancio (2007) have outlined a number of potential mechanisms (particularly in the context of integrated pest and disease management) in which AM fungi can interact with such soil microorganisms within the proximal growth environment to then enhance plant stress tolerance, which include: the competitive exclusion (or mechanical inhibition) of nonbeneficial soil microorganisms by competing for sites of infection and (or) colonization; the subsequent alteration of root architecture and anatomical structure due to AM colonization; the activation of plant defense responses such as antibiotics and phytoalexins due to AM-plant signaling; the modification of soil physicochemical parameters such as soil pH and nutrient bioavailability as well as alteration in carbon partitioning (or rhizo-deposition); and, finally, the increase in damage compensation due to an enhanced nutrient status. Although some of these perspectives have been described only briefly in this sections, it is evident that such mycorrhizospheric processes leading to the development of a beneficial belowground species biodiversity profile can then indirectly influence the composition of aboveground species biodiversity. Likewise, further experimental investigation to better elucidate specific mechanisms of interaction would certainly highlight the impact of AM fungi toward plant stress tolerance in regards to biotic stress and other biodiversity interactions.

4 Conclusions and Future Perspective

4.1 Assessing AM–Plant Interactions in Plant Stress Tolerance

In this chapter, it has been reported how the AM symbiosis is a widespread ecological association which is deeply rooted in the essential function of the vast majority of herbaceous plant species, as well as the function of soils and associated soil microorganisms. By describing some of the mechanisms underlying these plant physiological and soil ecological functions, it is clear that a number of such AM fungal processes are fundamentally involved in enhancing the stress tolerance of

Bacteria	AM fungi	Interaction outcome ^a	Reference
Azospirillum brasilense	Glomus intraradices	Ν	Hildebrandt et al. (2002)
<i>Bacillus chitinosporus; B. pabuli</i> and other spore-associated bacteria	G. clarum	P, n,N	Xavier and Germida (2003)
Clavibacter michiganensis ssp. michiganensis	G. intraradices	Ν	Filion et al. (1999)
Corynebacterium sp.	G. versiforme	Р	Mayo et al. (1986)
Escherichia coli	G. intraradices	Ν	Hildebrandt et al. (2002)
Paenibacillus validus	G. intraradices	Р	Hildebrandt et al. (2002)
Pseudomonas sp.	Endogone sp.	Р	Mosse (1962)
Pseudomonas sp.	G. versiforme	Р	Mayo et al. (1986)
P. aeruginosa	G. intraradices	Р	Villegas and Fortin (2001, 2002)
P. chlororaphis	G. intraradices	Р	Filion et al. (1999)
P. fluorescens	Gigaspora margarita	Р	Bianciotto et al. (1996a, b)
P. putida	G. intraradices	P, n	Villegas and Fortin (2001, 2002)
Rhizobium leguminosarum	Gi. margarita	Ν	Bianciotto et al. (1996a, b)
Serratia plymutica	G. intraradices	Р	Villegas and Fortin (2001, 2002)
Streptomyces avermitilis	G. mosseae	Ν	Tylka et al. (1991)
S. griseus	Scutellospora heterogama	Р	Tylka et al. (1991)
S. orientalis	Gi. margarita	Р	Tylka et al. (1991)
S. orientalis	G. mosseae	Р	Mugnier and Mosse (1987) and Tylka et al. (1991)
S. orientalis	S. heterogama	P, N	Tylka et al. (1991)
Spore-associated bacteria	G. versiforme	Р	Tylka et al. (1991)
Unidentified soil bacteria	G. mosseae	Р	Azcón (1987, 1989)

Table 11.3 Interactions between AM fungi and bacteria (from St. Arnaud and Elsen 2005)

P Positive, N negative, n neutral

^aRefers to the impact of the bacterial species toward the AM fungus (e.g., promotion of spore germination and hyphal growth)

plants and bolstering the resilience of soil in relation to a number of abiotic environmental stressors due especially to the dynamic function of the mycorrhizosphere. When taking into account the significant allocation of plant carbohydrates required for the development and maintenance of an expansive and prolific mycorrhizospheric infrastructure, it has been suggested that the AM symbiosis should represent an extrinsic plant stress tolerance strategy which could inherently complement other intrinsic plant stress tolerance strategies. To further depict the roles of the mycorrhizosphere in ecosystem function, the present description of the AM symbiosis has distinguished between the direct vs. indirect dynamics of interaction. Accordingly, this distinction refers to AM-induced activities which either directly benefit host plant due to intimately comodulated pathways or, alternatively, activities

which indirectly benefit the symbionts (potentially including non-associated species) due to their effects in buffering the proximal growth environment. This perspective is intriguing since it presumes that a number of essential mycorrhizospheric processes could occur simultaneously and (or) independently, and thereby have fundamental ecological functions across different trophic levels. This could account, in part, for the widespread abundance and distribution of mycorrhizal associations within the majority of terrestrial ecosystems, especially those subjected to highly stressful and (or) extreme environments. Still, there are so far very few modeling strategies depicting these combined, multilateral effects especially across a broad or continuous spectrum of stress (e.g., from trace to toxicity conditions or from droughted to amply water conditions). For instance, the case for AM-plant metal uptake is

		Interaction					
		outcome ^a ,				Interaction	
Fungus	AM fungi	P, n, N	Reference	Fungus	AM fungi	outcome	Reference
Alternaria alternate	G. mosseae	Z	McAllister et al. (1996)	Pyrenochaeta terrestris	Gi. margarita	n	Chabot (1991)
Aspergillus fumigatus	G. mosseae	Z	Calvet et al. (1992)	Rhizoctonia solani	Gi. margarita	n	Chabot (1991)
A. niger	G. mosseae	n	McAllister et al. (1996)	Rhodotorula mucilaginosa	G. mosseae	Ч	Francchia et al. (1998)
Bipolaris sorokiniana	Gi. margarita	P, n, N	Chabot (1991)	Sclerotinia scletotiorum	Gi. margarita	u	Chabot (1991)
Fusarium equiseti	G. mosseae	Z	McAllister et al. (1996)	Thievaliopsis basicola	Gi. margarita	u	Chabot (1991)
F. oxysporum f. sp. Chrysanthemi	G. intraradices	Z	Benhamou et al. (1994) and Filion et al. (1999)	Trichoderma aureoviride	G. mosseae	Ь	Calvet et al. (1992)
F.o. chrysanthemi	G. intraradices	P, N	St. Arnaud et al. (1996)	T. harzianum	G. intraradices	Ρ	Filion et al. (1999)
F. solani	Gi. margarita	N	Chabot (1991)	T. harzianum	G. mosseae	Ρ	Calvet et al. (1992)
F. solani	G. mosseae	P, n	McAllister et al. (1996)	T. harzianum	G. mosseae	n	Francchia et al. (1998)
Gaeumannomyces graminis	Gi. margarita	u	Chabot (1991)	T. harzianum	G. intraradices	Z	Rousseay et al. (1996)
Gliocladium roseum	G. mosseae	u	Francchia et al. (1998)	T. koningii	G. mosseae	N, n	McAllister et al. (1996)
Ophiostoma ulmi	Gi. margarita	n	Chabot (1991)	T. pseudokoningii	G. mosseae	n	Francchia et al. (1998)
Paedilomyces farinosus	G. mosseae	P, n	Francchia et al. (1998)	Verticillium albo-atrum	Gi. Margarita	u	Chabot (1991)
Penicillium decumbens	G. mosseae	Z	Calvet et al. (1992)	V. dahliae	Gi. margarita	n	Chabot (1991)
Phytophthora sp.	Gi. margarita	n	Chabot (1991)	Wardomyces inflatus	G. mosseae	Z	Francchia et al. (1998)
P. nicotianae	G. intraradices	Z	Lioussanna et al. (2003)	Unidentified soil fungi	G. mosseae	Р	Azcón-Aguilar et al. (1986)
Pythium ultimum	Gi. margarita	u	Chabot (1991)				

 Table 11.4
 Interactions between AM fungi and other fungi (from St. Armaud and Elsen 2005)

P Positive, N negative, n neutral "Refers to the impact of the bacterial species toward the AM fungus

Р	Ryan et al. (2000)
N	Elsen et al. (2001)
Ν	Elsen et al. (2003)
	N N

Table 11.5 Interactions between AM fungi and nematodes (from St. Arnaud and Elsen 2005)

P Positive, N negative, n neutral

^aRefers to the impact of the bacterial species toward the AM fungus

noteworthy since the AM fungi apparently hold two antithetical roles in plant metal uptake (i.e., enhanced uptake vs. metal biosorption) having different predicted outcomes depending on the soil metal conditions. From this viewpoint, a few lingering questions (among other) have yet to be fully addressed regarding the fundamental role(s) of the AM symbiosis in plant stress tolerance and ecosystem function, such as:

How do these combined mycorrhizospheric processes benefit plant stress tolerance? Under what environmental conditions do they occur? How are they regulated by the host plants, or the AM fungi?

Considering the abundance of high-quality data available within the published literatures which were developed primarily using reductionist strategies, future investigations could benefit from more holistic approaches in order to bridge our current mechanistic understanding of mycorrhizospheric function with other known soil ecological processes to better depict the impact of the AM symbiosis in whole-ecosystem function; in other words, taking into account multitrophic interactions and making attempts to quantify such relationships at the physiological and ecological levels. Such an approach would highlight the interconnectedness of above- and belowground species, as well as proving beneficial in re-assessing the balance between the costs of maintaining the symbiosis vs. the benefits of association among the symbionts. As mentioned previously, this resource allocation balance has been critical for plant physiologists and mycologists alike in defining the symbiotic mutualism since it is widely believed to function along a continuum potentially ranging from parasitism to mutualisms (Bronstein 2001; Johnson et al. 1997; Jones and Smith 2004). And so, an assessment of plant-AM-soil interactions using quantifiable descriptors could provide an insightful analysis of this continuum of interaction.

Another potential topic of concern regards the quantification of the mycorrhizosphere in relation to plant *mycorrhizal dependence*. With exception to the characterization of AM–plant nutrient uptake and symbiotic transfer using molecular tools, we have yet to quantitatively define the mycorrhizosphere's and (or) hyphosphere's actual zones of influence (e.g., their capacity to bind metals or stabilize the soil structure) other than by indirect methodologies. Such a quantification of the mycorrhizosphere could help address issues such as:

How prolific is the mycorrhizosphere compared to the rhizosphere? What are its biogeochemical properties and subsequent area of influence? How prolific is the mycorrhizosphere in relation to the plant's carbon allocation investment?

In turn, it may be feasible to quantitatively assess the mycorrhizal dependency (or resource allocation and extrinsic investment) of plants in relation to a wide-range stressors and broad-spectrum of environmental stress. Altogether, the potential focus-shift of future research objectives toward assessing the combined mycorrhizospheric impact in whole-ecosystem function as well as the quantification of the mycorrhizosphere in relation to plant symbiotic investment could highly benefit the effective integration of mycorrhizal technologies into agro-ecosystem management practices, especially sustainable agriculture and environmental remediation (Brussaard et al. 1997, 2007; Gosling et al. 2006; Jeffries et al. 2003; Mäder et al. 2002) - an intricate topic deserving of more lengthy discussion. Nevertheless, in the present context, it can be concluded that the AM fungi have a quintessential role in plant stress tolerance which could translate well toward enhancing the stress tolerance and stress resistance of whole ecosystems, particularly in the current *era of climate change* and in relation to anthropogenically derived environmental stressors.

References

- Allen MF (1991) The ecology of mycorrhizae. Cambridge University Press, Cambridge
- Allen MF, Smith WK, Moore TS, Christensen M (1981) Comparative water relations and photosynthesis of mycorrhizal and non-mycorrhizal *Bouteloua gracilis* H.B.K. New Phytol 88:683–693
- Andrade G, Mihara KL, Linderman RG, Bethlenfalvay GJ (1997) Bacteria from rhizosphere and hyphosphere soils of different arbuscular mycorrhizal fungi. Plant Soil 192:71–79
- Andrade G, Mihara KL, Linderman RG, Bethlenfalvay GJ (1998) Soil aggregation status and rhizobacteria in the mycorrhizosphere. Plant Soil 202:89–96
- Angers DA, Caron J (1998) Plant-induced changes in soil structure: processes and feedbacks. Biogeochemistry 42:55–72
- Antonovics J, Bradshaw AD, Turner RG (1971) Heavy metal tolerance in plants. In: Cragg JB (ed) Advances in ecological research. Academic, New York, pp 1–85
- Antoun H, Prevost D (2005) Ecology of plant growth promoting rhizobacteria. In: Siddiqu ZA (ed) PGPR: biocontrol and biofertilization. Springer, Dordrecht, pp 1–38
- Apak R (2002) Adsorption of heavy metal ions on soil surfaces and similar substances. In: Hubbard AT (ed) Encyclopedia of surface and colloid science. Dekker Encyclopedias, New York, pp 385–417
- Artursson V, Finlay RD, Jansson JK (2006) Interactions between arbuscular mycorrhizal fungi and bacteria and their potential for stimulating plant growth. Environ Microbiol 8:1–10
- Audet P, Charest C (2006) Effects of AM colonization on "wild tobacco" plants grown in zinc-contaminated soil. Mycorrhiza 16:277–283
- Audet P, Charest C (2007a) Heavy metal phytoremediation from a meta-analytical perspective. Environ Pollut 147:231–237
- Audet P, Charest C (2007b) Dynamics of AM symbiosis in heavy metal phytoremediation: meta-analytical and conceptual perspectives. Environ Pollut 147:609–614
- Audet P, Charest C (2008) Allocation plasticity & metalpartitioning: meta-analytical perspectives in phytoremediation. Environ Pollut 156:290–296
- Audet P, Charest C (2009) Contribution of AM symbiosis to *in vitro* root metal uptake: from trace to toxic metal conditions. Botany 87:913–921
- Audet P, Charest C (2010) Determining the impact of the AM-mycorrhizosphere on 'dwarf' sunflower Zn uptake and soil-Zn bioavailability. J Bot. doi:10.1155/2010/268540

- Audet P, Charest C (unpublished) Assessing AM-plant metal uptake and soil metal bioavailability among 'dwarf' sunflower in a stratified compartmental growth environment
- Augé RM (2000) Stomatal behavior of arbuscular mycorrhizal plants. In: Kapulnik Y, Douds DD (eds) Arbuscular mycorrhizas: physiology and function. Kluwer, Dordrecht, pp 201–237
- Augé RM (2001) Water relations, drought and VA mycorrhizal symbiosis. Mycorrhiza 11:3–42
- Augé RM, Stodola AJW, Tims JE, Saxton AM (2001) Moisture retention properties of a mycorrhizal soil. Plant Soil 230:87–97
- Augé RM (2004) Arbuscular mycorrhizae and soil/plant water relations. Canadian Journal of Soil Science 84:373–381
- Augé RM, Moore JL, Cho K, Stutz JC, Sylvia DM, Al-Agely AK, Saxton AM (2003) Relating foliar dehydration tolerance of mycorrhizal *Phaseolus vulgaris* to soil and root colonization by hyphae. J Plant Physiol 160:1147–1156
- Augé RM, Toler HD, Moore JL, Cho K, Saxton AM (2007) Comparing contributions of soil versus root colonization to variations in stomatal behavior and soil drying in mycorrhizal *Sorghum bicolor* and *Cucurbita pepo*. J Plant Physiol 164:1289–1299
- Augé RM, Toler HD, Sams CE, Nasim G (2008) Hydraulic conductance and water potential gradients in squash leaves showing mycorrhiza-induced increases in stomatal conductance. Mycorrhiza 18: 115–121
- Azcón R (1987) Germination and hyphal growth of *Glomus mosseae in vitro*: effects of rhizosphere bacteria and cell-free culture media. Soil Biol Biochem 19:417–419
- Azcón R (1989) Selective interaction between free-living rhizosphere bacteria and vesicular-arbuscular mycorrhizal fungi. Soil Biol Biochem 21:639–644
- Azcón-Aguilar C, Barea JM (1992) Interactions between mycorrhizal fungi and other rhizosphere microorganisms. In: Allen MJ (ed) Mycorrhizal functioning: an integrative plant–fungal process. Chapman & Hall, New York, pp 163–198
- Azcón-Aguilar C, Barea JM (1996) Arbuscular mycorrhizas and biological control of soil-borne plant pathogens: an overview of the mechanisms involved. Mycorrhiza 6:457–464
- Azcón-Aguilar C, Diaz-Rodriguez R, Barea JM (1986) Effects of soil microorganisms on spore germination and growth of the vesicular-arbuscular mycorrhizal fungus *Glomus mosseae*. Trans Br Mycol Soc 86:337–340
- Bago B, Vierheilig H, Piche Y, Azcón-Aguilar C (1996) Nitrate depletion and pH changes induced by the extraradical mycelium of the arbuscular mycorrhizal fungus *Glomus intraradices* grown in monoxenic culture. New Phytol 133:273–280
- Bago B, Pfeffer PE, Shachar-Hill Y (2000) Carbon metabolism and transport in arbuscular mycorrhizas. Plant Physiol 124:949–958

- Bais HP, Weir TL, Perry LG, Gilroy S, Vivanco JM (2006) The role of root exudates in rhizosphere interactions with plants and other organisms. Annu Rev Plant Biol 57:233–266
- Baker AJM, Walker PL (1990) Ecophysiology of metal uptake by tolerant plants. In: Shaw AJ (ed) Heavy metal tolerance in plants: evolutionary aspects. CRC, Boca Raton, pp 155–177
- Baldrian P (2003) Interactions of heavy metals with whiterot fungi. Enzyme Microb Technol 32:78–91
- Baligar VC, Bennett OL (1986) NPK-fertilizer efficiency: a situation for the tropics. Fert Res 10:147–164
- Barea JM (1997) Mycorrhiza-bacteria interactions on plant growth promotion. In: Ogoshi A, Kobayashi K, Homma Y, Kodama F, Kondo N, Akino S (eds) Plant growth promoting rhizobacteria. OECD, Paris, pp 150–158
- Barea JM, Andrade G, Bianciotto V, Dowling D, Lohrke S, Bonfante P, O'Gara F, Azcon-Aguilar C (1998) Impact on arbuscular mycorrhiza formation of *Pseudomonas* strains used as inoculants for biocontrol of soil-borne fungal plant pathogens. Appl Environ Microbiol 64:2304–2307
- Beare MH, Coleman DC, Crossley DA, Hendrix PF, Odum EP (1995) A hierarchical approach to evaluating the significance of soil biodiversity and no-tillage ultisols. Plant Soil 170:5–22
- Bearden BN (2001) Influence of arbuscular mycorrhizal fungi on soil structure and soil water characteristics of vertisols. Plant Soil 229:245–258
- Bearden BN, Petersen L (2000) Influence of arbuscular mycorrhizal fungi on soil structure and aggregate stability of a vertisol. Plant Soil 218:173–183
- Benedetto A, Magurno F, Bonfante P, Lanfranco L (2005) Expression profiles of a phosphate transporter gene (GmosPT) from the endomycorrhizal fungus *Glomus mosseae*. Mycorrhiza 15:620–627
- Benhamou N, Fortin JA, Hamel C, St. Arnaud M, Shatilla A (1994) Resistance responses of mycorrhizal Ri T-DNA transformed carrot roots to infection by *Fusarium* oxysporum f. sp. chrysanthemi. Phytopathology 84:958–968
- Bertin C, Yang X, Weston LA (2003) The role of root exudates and allelochemicals in the rhizosphere. Plant Soil 256:67–83
- Bertness MD, Callaway R (1994) Positive interactions in communities. Trends Ecol Evol 9:191–193
- Bever JD (1999) Dynamics within mutualism and the maintenance of diversity: inference from a model of interguild frequency dependence. Ecol Lett 2:52–61
- Bever JD (2003) Soil community feedback and the coexistence of competitors: conceptual frameworks and empirical tests. New Phytol 157:465–473
- Bever JD, Westover KM, Antonovics J (1997) Incorporating the soil community into plant population dynamics: the utility of the feedback approach. J Ecol 85:561–573
- Bhuiyan NH, Selvaraj G, Wei Y, King J (2009) Role of lignification in plant defense. Plant Signal Behav 4:158–159

- Bianciotto V, Bonfante P (2002) Arbuscular mycorrhizal fungi: a specialized niche for rhizospheric and endocellular bacteria. Antonie Van Leeuwenhoek J Microb 81:365–371
- Bianciotto V, Bandi C, Minerdi D, Sironi M, Tichy HV, Bonfante P (1996a) An obligately endosymbiotic mycorrhizal fungus itself harbors obligately intracellular bacteria. Appl Environ Microbiol 62:3005–3010
- Bianciotto V, Minerdi D, Perotto S, Bonfante P (1996b) Cellular interactions between arbuscular mycorrhizal fungi and rhizosphere bacteria. Protoplasma 193:123–131
- Blinkley D, Vitousek P (1989) Soil nutrient availability. In: Pearcy RW, Ehleringer JR, Mooney HA, Rundel PW (eds) Plant physiological ecology: field methods and instrumentation. Chapman and Hall, New York, pp 75–96
- Bloom AJ (1997) Nitrogen as a limiting factor: crop acquisition of ammonium and nitrate. In: Jackson LE (ed) Ecology in agriculture. Academic, London
- Bohn HL (1971) Redox potentials. Soil Sci 112:39-45
- Bolan NS (1991) A critical review on the role of mycorrhizal fungi in the uptake of phosphorus by plants. Plant Soil 134:189–207
- Boucher DH (1988) The biology of mutualism: ecology and evolution. Oxford University Press, Oxford
- Boucher DH, James S, Keeler KH (1982) The ecology of mutualism. Annu Rev Ecol Syst 13:315–347
- Bradl HB (2004) Adsorption of heavy metal ions on soils and soils constituents. J Colloid Interface Sci 277: 1–18
- Brams E (1973) Soil organic matter and phosphorus relationships under tropical forests. Plant Soil 39:465–468
- Bronstein JL (2001) The costs of mutualism. Integr Comp Biol 4:825–839
- Brussaard L, Behan-Pelletier V, Bignell D, Brown V, Didden W, Folgarait P, Fragoso C, Freckman D, Gupta V, Hattori T, Hawksworth D, Kopatek C, Lavelle P, Malloch D, Rusek J, Sodestrom B, Tiedje J, Virginia R (1997) Biodiversity and ecosystem function in soil. Ambio 26:563–570
- Brussaard L, de Ruiter PC, Brown GG (2007) Soil biodiversity for agriculture sustainability. Agric Ecosyst Environ 121:233–244
- Burleigh SH, Kristensen BK, Bechmann IE (2003) A plasma membrane zinc transporter from Medicago truncatula is up-regulated in roots by Zn fertilization, yet down-regulated by arbuscular mycorrhizal colonization. Plant Mol Biol 52:10–77–1088
- Calvet C, Barea JM, Pera J (1992) In vitro interactions between the vesicular-arbuscular mycorrhizal fungus Glomus mosseae and some saprophytic fungi isolated from organic substrates. Soil Biol Biochem 24:775–780
- Capaccio LCM, Callow JA (1982) The enzymes of polyphosphate metabolism in vesicular-arbuscular mycorrhizas. New Phytol 91:81–91
- Cavagnaro TR, Jackson LE, Six J, Ferris H, Goyal S, Asami D, Scow KM (2006) Arbuscular mycorrhizas,

microbial communities, nutrient availability and soil aggregates in organic tomato production. Plant Soil 282:209–225

- Chabot S (1991) Utilisation d'un system de culture monoxenique comme modele pour l'etude des endomycorhizes a vesiules et arbuscules. These MSc, Universite Laval, Quebec
- Chalot M, Blaudez D, Brun A (2006) Ammonia: a candidate for nitrogen transfer at the mycorrhizal interface. Trends Plant Sci 11:263–266
- Chapin FS, Matson PA, Mooney HA (2002) Principles of terrestrial ecosystem ecology. Springer, New York
- Chapman SK, Langley JA, Hart SC, Koch GW (2005) Plants actively control nitrogen cycling: uncorking the microbial bottleneck. New Phytol 169:27–34
- Cho K, Toler HD, Lee J, Ownley BH, Stutz JC, Moore JL, Augé RM (2006) Mycorrhizal symbiosis and response of sorghum plants to combined drought and salinity stresses. J Plant Physiol 163:517–528
- Christie P, Li XL, Chen BD (2004) Arbuscular mycorrhiza can depress translocation of zinc to shoots of host plants in soils moderately polluted with zinc. Plant Soil 261:209–217
- Cleveland CC, Townsend AR, Schmidt SK (2002) Phosphorus limitation of microbial processes in moist tropical forests: evidence from short-term laboratory incubations and field studies. Ecosystems 5:680–691
- Cobbett CS (2000) Phytochelatins and their roles in heavy metal detoxification. Plant Physiol 123:825–832
- Cobbett CS, Goldsbrough P (2002) Phytochelatins and metallothioneins: role in heavy metal detoxification and homeostasis. Annu Rev Plant Biol 53:159–182
- Cox GC, Moran KJ, Sanders FE, Nockolds C, Tinker PB (1980) Translocation and transfer of nutrients in vesicular arbuscular mycorrhizas: polyphosphate granules and phosphorus translocation. New Phytol 84:649–659
- Cruz C, Egsgaard H, Trujillo C, Ambus P, Requena N, Martins-Loucao MA, Jakobsen I (2007) Enzymatic evidence for the key role of arginine in nitrogen translocation by arbuscular mycorrhizal fungi. Plant Physiol 144:782–792
- Cushman JH, Beattie AJ (1991) Mutualisms: assessing the benefits to hosts and visitors. Trends Ecol Evol 6:193–195
- Dalpé Y, de Souza FA, Declerck S (2005) Life cycle of *Glomus* species in monoxenic culture. In: Declerck S, Strullu DG, Fortin FA (eds) *In vitro* culture of mycorrhizas. Springer, Berlin, pp 49–71
- Dangl JL, Jones JDG (2001) Plant pathogens and integrated defense responses to infection. Nature 411:826–833
- Daram P, Brunner S, Persson BL, Amrhein N, Bucher M (1998) Functional analysis and cell-specific expression of a phosphate transporter from tomato. Planta 206:225–233
- Deubel A, Merbach W (2005) Influence of microorganisms on phosphorus bioavailability in soils. In: Varma A, Buscot F (eds) Microorganisms in soils: roles in genesis and functions. Springer, Berlin, pp 177–191

- Dexheimer CM, Gianinazzi-Pearson V, Gianinazzi S (1982) Enzymatic studies on the metabolism of vesicular-arbuscular mycorrhizas: IV. Ultracytoenzymological evidence (ATPase) for active transfer processes in the host-arbuscule interface. New Phytol 90:37–43
- Douds DD, Johnson NC (2007) Contributions of arbuscular mycorrhizas to soil biological fertility. In: Abbott LK, Murphy DV (eds) Soil biological fertility: a key to sustainable land use in agriculture. Springer, Berlin, pp 129–162
- Douds DD, Pfeffer PE, Shachar-Hill Y (2000) Carbon partitioning, cost and metabolism of arbuscular mycorrhizae in arbuscular mycorrhizas: physiology and function. In: Kapulnick Y, Douds DD (eds) Arbuscular mycorrhizas: molecular biology and physiology. Kluwer, Dordrecht
- Driver JD, Holben WE, Rillig MC (2005) Characterization of glomalin as a hyphal wall component of arbuscular mycorrhizal fungi. Soil Biol Biochem 37: 101–106
- Duke SO (1994) Glandular trichomes: a focal point of chemical and structural interactions. Int J Plant Sci 155:617–620
- Duponnois R, Galiana A, Prin Y (2008) The mycorrhizosphere effect: a multitrophic interaction complex improves mycorrhizal symbiosis and plant growth. In: Anwar SZ, Sayeed AM, Kazuyoshi F (eds) Mycorrhizae: sustainable agriculture and forestry. Springer, Berlin, pp 227–240
- Eckhard G, Römheld V, Marschner H (1994) Contribution of mycorrhizal fungi to micronutrient uptake by plants.
 In: Manthey J, Crowley DE, Luster GL (eds) Biochemistry of metal micronutrients in the rhizosphere. CRC, Boca Raton, pp 93–110
- Eckhard G, Marschner H, Jakobsen I (1995) The role of arbuscular mycorrhizal fungi in uptake of phosphorus and nitrogen from soil. Crit Rev Biotechnol 15:257–270
- Elsen A, Declerck S, De Waele D (2001) Effects of *Glomus intraradices* on the reproduction the burrowing nematode (*Radopholus similis*) in dixenic culture. Mycorrhiza 11:49–51
- Elsen A, Declerck S, De Waele D (2003) Use of root organ cultures to investigate the interaction between *Glomus intraradices* and *Pratylenchus coffeae*. Appl Environ Microbiol 69:4308–4311
- Ewel JJ (1986) Designing agricultural ecosystems for the humid tropics. Annu Rev Ecol Syst 17:245–271
- Faure S, Cliquet JB, Thephany G, Boucaud J (1998) Nitrogen assimilation in *Lolium perenne* colonized by the arbuscular mycorrhizal fungus *Glomus fasciculatum*. New Phytol 138:411–417
- Ferrol N, Barea JM, Azcòn-Aguilar C (2000) The plasma membrane H⁺-ATPase gene family in the arbuscular mycorrhizal fungus *Glomus mosseae*. Curr Genet 37:112–118
- Filion M, St. Arnaud M, Fortin JA (1999) Direct interaction between the arbuscular mycorrhizal fungus *Glomus intraradices* and different rhizosphere microorganisms. New Phytol 141:525–533

- Fitter AH, Garbaye J (1994) Interactions between mycorrhizal fungi and other soil organisms. Plant Soil 159:123–132
- Fitzpatrick EA (1984) Micromorphology of soils. Chapman and Hall, London
- Fogarty RV, Tobin JM (1996) Fungal melanins and their interactions with metals. Enzyme Microb Technol 19:311–317
- Fourest E, Roux JC (1992) Heavy metal biosorption by fungal mycelial by-products: mechanism and influence of pH. Appl Microbiol Biotechnol 37:399–403
- Foy CD, Chaney RL, White MC (1978) The physiology of metal toxicity in plants. Annu Rev Plant Physiol 29:511–566
- Francchia S, Mujica MT, Garcia-Romera I, Garcia-Garrido JM, Martin J, Ocampo JA, Godeas A (1998) Interactions between *Glomus mosseae* and arbuscular mycorrhizal sporocarp-associated saprophytic fungi. Plant Soil 200:131–137
- Frank AB (1885) Über die auf Würzelsymbiose beruhende Ehrnährung gewisser Bäum durch unterirdische Pilze. Berichte der Deutschen Botanischen Gesellschaft 3:128–145
- Frey-Klett P, Garbate J, Tarkka M (2007) The mycorrhiza helper bacteria revisited. New Phytol 176:22–36
- Fritig B, Heitz T, Legrand M (1998) Antimicrobial proteins in induced plant defense. Curr Opin Immunol 10:16–22
- Gadd GM (1993) Interactions of fungi with toxic metals. New Phytol 124:25–60
- Gahoonia TS, Nielsen NE (1992) The effects of root induced pH changes on the depletion of inorganic and organic phosphorus in the rhizosphere. Plant Soil 143:185–191
- Galli U, Schüepp H, Brunold C (1994) Heavy metal binding by mycorrhizal fungi. Physiol Plant 92:364–368
- Ganz T, Lehrer R (1999) Antibiotic peptides from higher eukaryotes: biology and applications. Mol Med Today 5:292–297
- Garbaye J (1991) Biological interactions in the mycorrhizosphere. Cell Mol Life Sci 47:370–375
- Garbaye J (1994) Helper bacteria: a new dimension to the mycorrhizal symbiosis. New Phytol 128:197–210
- Garcia-Garrido JM, Lendzemo V, Castellanos-Morales V, Steinkellner S, Vierheilig H (2009) Strigolactones, signals for parasitic plants and arbuscular mycorrhizal fungi. Mycorrhiza 19:449–459
- Gauthier DA, Turpin DH (1994) Inorganic phosphate (P_i) enhancement of dark respiration in the P_i limited green alga *Selenastrum minutum*. Plant Physiol 104: 629–637
- George E, Marschner H, Jakobsen I (1995) Role of arbuscular-mycorrhizal fungi in uptake of phosphorus and nitrogen from soil. Crit Rev Biotechnol 15:257–270
- Giasson P, Kara A, Jaouich A (2008) Arbuscular mycorrhizae and alleviation of soil stress on plant growth. In: Siddiqui ZA, Akhtar MS, Futai K (eds) Mycorrhizae: sustainable agriculture and forestry. Springer, Berlin, pp 99–134

- Giller KE, Witter E, McGrath SP (1998) Toxicity of heavy metals to microorganisms and microbial processes in agricultural soils: a review. Soil Biol Biochem 30:1389–1414
- Glassop D, Smith S, Smith F (2005) Cereal phosphate transporters associated with the mycorrhizal pathway of phosphate uptake into roots. Planta 222:688–698
- Göhre V, Pazkowski U (2006) Contribution of the arbuscular mycorrhizal symbiosis to heavy metal phytoremediation. Plant 223:1115–1122
- González-Chavez C, d'Haen J, Vangronsveld J, Dodd JC (2002) Copper sorption & accumulation by the extraradical mycelium of different *Glomus* spp. (arbuscular mycorrhizal fungi) isolated from the same polluted soil. Plant Soil 240:287–297
- González-Guerrero M, Azcón-Aguilar C, Mooney M, Valderas A, MacDiarmid CW, Eide DJ, Ferrol N (2005) Characterization of a *Glomus intraradices* gene encoding a putative Zn transporter of the cation diffusion facilitator family. Fungal Genet Biol 42:130–140
- González-Guerrero M, Cano C, Azcón-Aguilar C, Ferrol N (2007) GintMT1 encodes a functional metallothionein in *Glomus intraradices* that responds to oxidative stress. Mycorrhiza 17:327–335
- González-Guerrero M, Melville LH, Ferrol N, Lott JNA, Azcon-Aguilar C, Peterson RL (2008) Ultrastructural localization of heavy metals in the extraradical mycelium and spores of the arbuscular mycorrhizal fungus *Glomus intraradices*. Can J Microbiol 54: 103–110
- González-Guerrero M, Benabdellah K, Ferrol N, Azcón-Aguilar C (2009) Mechanisms underlying heavy metal tolerance in arbuscular mycorrhizas. In: Azcón-Aguilar C, Barea JM, Gianinazzi S, Gianinazzi-Pearson V (eds) Mycorrhizas: functional processes and ecological impact. Springer, Berlin, pp 1–16
- González-Guerrero M, Benabdellah K, Valderas A, Azcón-Aguilar C, Ferrol N (2010a) GintABC1 encodes a putative ABC transporter of the MRP subfamily induced by Cu, Cd, and oxidative stress in *Glomus intraradices*. Mycorrhiza 20:137–146
- González-Guerrero M, Oger E, Benabdellah K, Azcón-Aguilar C, Lanfranco L, Ferrol N (2010b) Characterization of a CuZn superoxide dismutase gene in the arbuscular mycorrhizal fungus *Glomus intraradices*. Curr Genet 56:265–274
- Gosling P, Hodge A, Goodlass G, Bending GD (2006) Arbuscular mycorrhizal fungi and organic farming. Agric Ecosyst Environ 113:17–35
- Govindarajulu M, Pfeffer PE, Jin H, Douds DD, Allen JW, Bucking H, Lammers PJ, Shachar-Hill Y (2005) Nitrogen transfer in the arbuscular mycorrhizal symbiosis. Nature 435:819–823
- Graham JH, Eissenstat DM, Drouillard DL (1991) On the relationship between a plant's mycorrhizal dependency and rate of vesicular-arbuscular mycorrhizal colonization. Funct Ecol 5:773–779
- Guimil S, Chang HS, Zhu T (2005) Comparative transcriptomics of rice reveals an ancient pattern of

response to microbial colonization. Proc Natl Acad Sci USA 102:8066-8070

- Gutschick VP (1981) Evolved strategies in nitrogen acquisition by plants. Am Nat 118:607–637
- Hahn M, Mendgen K (2001) Signal and nutrient exchange at biotrophic plant-fungus interfaces. Curr Opin Plant Biol 4:322–327
- Harrier LA, Watson CA (2003) The role of arbuscular mycorrhizal fungi in sustainable cropping systems. Adv Agron 79:185–225
- Harrier LA, Watson CA (2004) The potential role of arbuscular mycorrhizal (AM) fungi in the bioprotection of plants against soil-borne pathogens in organic and/or other sustainable farming systems. Pest Manag Sci 60:149–157
- Harrison MJ (1999) Molecular and cellular aspects of the arbuscular mycorrhizal symbiosis. Annu Rev Plant Physiol Plant Mol Biol 50:361–389
- Harrison MJ (2005) Signaling in the arbuscular mycorrhizal symbiosis. Annu Rev Microbiol 59:19–42
- Harrison MJ, van Buuren ML (1995) A phosphate transporter from the mycorrhizal fungus *Glomus versiforme*. Nature 378:626–629
- Harrison MJ, Dewbre GR, Liu J (2002) A phosphate transporter from *Medicago truncatula* involved in the acquisition of phosphate released by arbuscular mycorrhizal fungi. Plant Cell 14:2413–2429
- Haystead A, Maljczuk N, Grove TS (1988) Underground transfer of nitrogen between pasture plants infected with vesicular-arbuscular mycorrhizal fungi. New Phytol 108:417–423
- Hildebrandt U, Janetta K, Bothe H (2002) Towards growth of arbuscular mycorrhizal fungi independent of a plant host. Appl Environ Microbiol 68:1919–1924
- Hildebrandt U, Regvar M, Bothe H (2007) Arbuscular mycorrhiza and heavy metal tolerance. Phytochemistry 68:139–146
- Howe GA, Jander G (2008) Plant immunity to insect herbivores. Annu Rev Plant Biol 59:41–66
- Hsiao TC (1973) Plant responses to water stress. Annu Rev Plant Physiol 24:519–570
- Janos DP (2007) Plant responsiveness to mycorrhizas differs from dependence upon mycorrhizas. Mycorrhiza 17:75–91
- Javot H, Pumplin N, Harrison MJ (2007) Phosphate in the arbuscular mycorrhizal symbiosis: transport properties and regulatory roles. Plant Cell Environ 30:310–322
- Jeffries P, Gianinazzi S, Perotto S, Turnau K, Barea JM (2003) The contribution of arbuscular mycorrhizal fungi in sustainable maintenance of plant health and soil fertility. Biol Fert Soils 37:1–16
- Jin H, Pfeffer PE, Douds DD, Piotrowski E, Lammers PJ, Shachar-Hill Y (2005) The uptake, metabolism, transport and transfer of nitrogen in an arbuscular mycorrhizal symbiosis. New Phytol 168:687–696
- Johanson JF, Paul LR, Finlay RD (2003) Microbial interactions in the mycorrhizosphere and their significance for sustainable agriculture. FEMS Microbiol Ecol 48:1–13

- Johnson NC, Graham JH, Smith FA (1997) Functioning of mycorrhizal associations along the mutualism-parasitism continuum. New Phytol 135:575–586
- Joner EJ, Briones R, Leyval C (2000) Metal-binding capacity of arbuscular mycorrhizal mycelium. Plant Soil 226:227–234
- Jones MD, Smith SE (2004) Exploring functional definitions of mycorrhizas: are mycorrhizas always mutualisms? Can J Bot 82:1089–1109
- Karandashov V, Nagy R, Wegmuller S, Amrhein N. & Bucher M (2004) Evolutionary conservation of a phosphate transporter in the arbuscular mycorrhizal symbiosis. Proceedings of the National Academy of Sciences of the United States of America 101: 6285–6290
- Karandashov V, Bucher M (2005) Symbiotic phosphate transport in arbuscular mycorrhizas. Trends Plant Sci 10:22–29
- Khademi S, O'Connell J, Remis J, Robles-Colmenares Y, Miercke LJW, Stroud RM (2004) Mechanism of ammonia transport by Amt/MEP/Rh: structure of AmtB at 1.35 Å. Science 605:1587–1594
- Khalvati MA, Hu Y, Mozafar A, Schmidhalter U (2005) Quantification of water uptake by arbuscular mycorrhizal hyphae and its significance for leaf growth, water relations, and gas exchange of barley subjected to drought stress. Plant Biology 7:706–712
- Klironomos JN (2002) Feedback with soil biota contributes to plant rarity and invasiveness in communities. Nature 417:67–70
- Koide RT (1991) Nutrient supply, nutrient demand and plant response to mycorrhizal infection. New Phytol 117:365–386
- Koide RT (1993) Physiology of the mycorrhizal plant. Adv Plant Pathol 9:33–54
- Koide RT (2000) Mycorrhizal symbiosis and plant reproduction. In: Kapulnik Y, Douds DD (eds) Arbuscular mycorrhizas: physiology and function. Kluwer, Dordrecht, pp 19–46
- Koide RT, Elliott G (1989) Cost, benefit and efficiency of the vesicular-arbuscular mycorrhizal symbiosis. Funct Ecol 3:252–255
- Langenheim JH (2003) Plant resins: chemistry, evolution, ecology, and ethnobotany. Timber, Portland
- Larcher W (1987) Stress bei pflanzen. Naturwissenschaften 74:158–167
- Larcher W (2003) Plant physiological ecology: ecophysiology and stress physiology of functional groups, 4th edn. Springer, Berlin
- Li X, Christie P (2001) Changes in soil solution Zn and pH and uptake of Zn by arbuscular mycorrhizal red clover in Zn-contaminated soil. Chemosphere 42:201–207
- Leggewie G, Willmitzer L, Riesmeier JW (1997) Two cDNAs from potato are able to complement a phosphate uptake-deficient yeast mutant: identification of phosphate transporters from higher plants. Plant Cell 9:381–392
- Leung TLF, Poulin R (2008) Parasitism, commensalism and mutualism: exploring the many shades of symbiosis. Vie et Milieu 58:107–115

- Leyval C, Turnau K, Haselwandter K (1997) Effect of heavy metal pollution on mycorrhizal colonization and function: physiological, ecological, and applied aspects. Mycorrhiza 7:139–153
- Li X, George E, Marschner H (1991) Phosphorus depletion and pH decrease at the root-soil and hyphae-soil interfaces of VA mycorrhizal white clover fertilized with ammonium. New Phytol 119:397–404
- Linderman RG (1988) Mycorrhizal interactions with the rhizosphere microflora: the mycorrhizosphere effect. Phytopathology 78:366–371
- Lioussanna L, Jolicoeur M, St. Arnaud M (2003) Effects of the alteration of tomato root exudation by *Glomus intraradices* colonization on *Phytophthora parasitica* var. *nicotianae* zoospores. In: Proceedings of the 4th international conference on mycorrhizae, Montreal
- Liu C, Muchhal US, Uthappa M, Kononowicz AK, Raghothama KG (1998a) Tomato phosphate transporter genes are differentially regulated in plant tissues by phosphorus. Plant Physiol 116:91–99
- Liu H, Trieu AT, Blaylock LA, Harrison MJ (1998b) Cloning and characterization of two phosphate transporters from *Medicago truncatula* roots: regulation in response to phosphate and to colonization by arbuscular mycorrhizal (AM) fungi. Mol Plant Microbe Interact 11:14–22
- Liu A, Hamel C, Hamilton RI, Ma BL, Smith DL (2000) Acquisition of Cu, Zn, Mn and Fe by mycorrhizal maize (*Zea mays* L.) grown in soil at different P and micronutrient levels. Mycorrhiza 9:331–336
- Long SR (1996) Rhizobium symbiosis: nod factors in perspective. Plant Cell 8:1885–1898
- Long SR (2001) Genes and signals in the rhizobiumlgume symbiosis. Plant Physiol 125:69–72
- López-Millán A-F, Ellis DR, Grusak MA (2004) Identification and characterization of several new members of the ZIP family of metal ion transporters in *Medicago trunculata*. Plant Mol Biol 54:583–596
- López-Pedrosa A, González-Guerrero M, Valderas A, Azcón-Aguilar C, Ferrol N (2006) GintAMT1 encodes a functional high-affinity ammonium transporter that is expressed in the extraradical mycelium of *Glomus intraradices*. Fungal Genet Biol 43:102–110
- Mäder P, Fließbach A, Dubois D, Gunst L, Fried P, Niggli U (2002) Soil fertility and biodiversity in organic farming. Science 296:1694–1697
- Maeda D, Ashida K, Iguchi K, Chechetka S, Hijikata A, Okusako Y, Deguchi Y, Izui K, Hata S (2006) Knockdown of an arbuscular mycorrhiza-inducible phosphate transporter gene of *Lotus japonicus* suppresses mutualistic symbiosis. Plant Cell Physiol 47:807–817
- Maldonado-Mendoza IE, Dewbre GR, Harrison MJ (2001) A phosphate transporter gene from the extraradical mycelilum of an arbuscular mycorrhizal fungus *Glomus intraradices* is regulated in response to phosphate in the environment. Mol Plant Microbe Interact 14:1140–1148
- Marschner H (1995) Mineral nutrition of higher plants, 2nd edn. Springer, Berlin

- Marschner H (1998) Role of root growth, arbuscular mycorrhiza, and root exudates for the efficiency in nutrient acquisition. Field Crop Res 56:203–207
- Marschner H, Dell B (1994) Nutrient uptake in mycorrhizal symbiosis. Plant Soil 159:89–102
- Mayo K, Davis R, Motta R (1986) Stimulation of germination spores of Glomus versiforme by spore-associated bacteria. Mycologia 78:426–431
- McAllister CB, Garcia-Romera I, Martin J, Godeas A, Ocampo JA (1996) In vitro interaction between Alternaria alternata, Fusarium solani and Glomus mosseae. Symbiosis 20:163–174
- McBride MB (1994) Environmental chemistry of soils. Oxford University Press, Oxford
- Meharg AA (2003) The mechanistic basis of interactions between mycorrhizal associations and toxic metal cations. Mycol Res 107:1253–1265
- Mengel K, Kirkby EA (2001) Principles of plant nutrition, 5th edn. Kluwer, Dordrecht
- Miller RM, Jastrow JD (1990) Hierarchy of root and mycorrhizal fungal interactions with soil aggregation. Soil Biol Biochem 22:579–584
- Mole S (1994) Trade-offs and constraints in plant-herbivore defense theory: a life-history perspective. Oikos 71:3–12
- Mosse B (1962) The establishment of vesicular-arbuscular mycorrhiza under aseptic conditions. J Genet Microbiol 27:509–520
- Mugnier J, Mosse B (1987) Vesicular-arbuscular mycorrhizal infections in transformed Ri T-DNA roots grown axenically. Phytopathology 77:1045–1050
- Mukerji K, Ciancio A (2007) Mycorrhizae in the integrated pest and disease management. In: Mukerji K, Ciancio A (eds) General concepts in integrated pest and disease management. Springer, Berlin, pp 245–266
- Müller CB, Krauss J (2005) Symbiosis between grasses and asexual fungal endophytes. Curr Opin Plant Biol 8:450–456
- Nagy F, Karandashov V, Chague W, Kalinkevich K, Tamasloukht M, Xu GH, Jakobsen I, Levy AA, Amrhein N, Bucher M (2005) The characterization of novel mycorrhiza-specific phosphate transporters from *Lycopersicon esculentum* and *Solanum tuberosum* uncovers functional redundancy in symbiotic phosphate transport in solanaceous species. Plant J 42:236–250
- Nagy R, Vasconcelos M, Zhao S, McElver J, Bruce W, Amrhein N, Raghothama K, Bucher M (2006) Differential regulation of five Pht1 phosphate transporters from maize (*Zea mays L.*). Plant Biol 8:186–197
- Nautiyal S, Badola HK, Pal M, Negi DS (1994) Plant responses to water stress changes in growth, dry matter production, stomatal frequency and leaf anatomy. Biol Plantarum 36:91–97
- Newsham KK, Fitter AH, Watkinson AR (1995a) Arbuscular mycorrhiza protect an annual grass from root pathogenic fungi in the field. J Ecol 83: 991–1000

- Newsham KK, Fitter AH, Watkinson AR (1995b) Multifunctionality and biodiversity in arbuscular mycorrhizas. Trends Ecol Evol 10:407–411
- Paszkowski U, Kroken S, Roux C, Briggs SP (2002) Rice phosphate transporters include an evolutionarily divergent gene specifically activated in arbuscular mycorrhizal symbiosis. Proc Natl Acad Sci USA 99: 13324–13329
- Peterson RL, Massicotte HB, Melville LH (2004) Mycorrhizas: anatomy and cell biology. NRC Research Press, Ottawa
- Picone LI, Zamuner EC, Berardo A, Marino MA (2003) Phosphorus transformations as affected by sampling date, fertilizer rate and phosphorus uptake in a soil under pasture. Nutr Cycl Agroecosyst 67:225–232
- Piotrowski JS, Denich T, Klironomos JN, Graham JM, Rillig MC (2004) The effects of arbuscular mycorrhizae on soil aggregation depend on the interaction between plant and fungal species. New Phytol 164:365–373
- Purin S & Rillig MC (2007) The arbuscular mycorrhizal fungal protein glomalin: limitations, progress, and a new hypothesis for its function. Pedobiologia 51: 123–130
- Purin S, Rillig MC (2008) Parasitism of arbuscular mycorrhizal fungi: reviewing the evidence. FEMS Microbiol Lett 279:8–14
- Rae AL, Cybinski DH, Jarmey JM, Smith FW (2003) Characterization of two phosphate transporters from barley; evidence for diverse function and kinetic properties among members of the Pht1 family. Plant Mol Biol 53:27–36
- Rausch C, Daram P, Brunner S, Jansa J, Laloi M, Leggewie G, Amrhein N, Bucher M (2001) A phosphate transporter expressed in arbuscule-containing cells in potato. Nature 414:462–466
- Redecker D, Kodner R, Graham LE (2000) Glomalean fungi from the Ordovician. Science 289: 1920–1921
- Remy W, Taylor TN, Hass H, Kerp H (1994) Four hundred-million-year-old vesicular arbuscular mycorrhizae. Proc Natl Acad Sci USA 91:11841–11843
- Rengel Z, Batten GD, Crowley DE (1999) Agronomic approaches for improving the micronutrient density in edible portions of field crops. Field Crop Res 60:27–40
- Rillig MC, Mummey DL (2006) Mycorrhizas and soil structure. New Phytol 171:41–53
- Rillig MC, Mardatin NF, Leifheit EF, Antunes PM (2010) Mycelium of arbuscular mycorrhizal fungi increases soil water repellency and is sufficient to maintain water-stable soil aggregates. Soil Biol Biochem 42:1189–1191
- Rønsted N, Weiblen GD, Cook JM, Salamin N, Machado CA, Savolainen V (2005) 60 million years of codivergence in the fig-wasp symbiosis. Proc R Soc Lond B Biol 272:2593–2599
- Rønsted N, Wiblen GD, Clement WL, Zerega NJC, Savolainen V (2008) Reconstring the phylogeny of

figs (*Ficus*, Moraceae) to reveal the history of the fig pollination mutualism. Symbiosis 45:45–56

- Rosewarne G, Barker S, Smith S, Smith F, Schachtman D (1999) A Lycopersicon esculentum phosphate transporter (LePT1) involved in phosphorous uptake form a vesicular-arbuscular mycorrhizal fungus. New Phytol 144:507–516
- Rousseay A, Benhamou N, Chet I, Piché Y (1996)
 Mycoparasitism of the extrametrical phase of *Glomus* intraradices by *Trichoderma harzianum*.
 Phytopathology 86:434–443
- Rovira AD (1969) Plant root exudates. Bot Rev 35:35–57
- Rufyikiri G, Declerck S, Thiry Y (2004) Comparison of ²³³U and ³³P uptake and translocation by the arbuscular mycorrhizal fungus *Glomus intraradices* in root organ culture conditions. Mycorrhiza 14:203–207
- Ryan NA, Duffy EM, Cassells AC, Jones PW (2000) The effect of mycorrhizal fungi on the hatch of potato cyst nematodes. Appl Soil Ecol 15:233–240
- Saito M (2000) Symbiotic exchange of nutrients in arbuscular mycorrhizas: transport and transfer of phosphorus. In: Douds DD, Kapunik Y (eds) Arbuscular mycorrhizas: physiology and function. Kluwer, Dordrecht, pp 85–106
- Schachtman DP, Reid RJ, Ayling SM (1998) Phosphorus uptake by plants: from soil to cell. Plant Physiol 116:447–453
- Schüßler A, Schwarzott D, Walker C (2001) A new fungal phylum, the Glomeromycota: phylogeny and evolution. Mycol Res 105:1413–1421
- Schwab SM, Menge JA, Tinker PB (1991) Regulation of nutrient transfer between host and fungus in vesiculararbuscular mycorrhizas. New Phytol 117:387–398
- Schwartz MW, Hoeksema JD (1998) Specialization and resource trade: biological markets as a model of mutualisms. Ecology 79:1029–1038
- Sharma AK, Srivastava PC, Johri BN (1994) Contribution of VA mycorrhiza to zinc uptake in plants. In: Manthey J, Crowley DE, Luster DG (eds) Biochemistry of metal micronutrients in the rhizosphere. CRC, Boca Raton, pp 111–124
- Smith FA, Cybinski D, Rae A (1999) Regulation of expression of genes encoding phosphate transporters in barley roots. In: Jensen A, Gissel-Nielsen G (eds) Plant nutrition: molecular biology and genetics. Kluwer, Dordrecht, pp 145–150
- Smith SE, Smith FA, Jakobsen I (2003) Mycorrhizal fungi can dominate phosphate supply to plants irrespective of growth responses. Plant Physiol 133:13–20
- Smith SE, Smith FA, Jakobsen I (2004) Functional diversity in arbuscular mycorrhizal (AM) symbioses: the contribution of the mycorrhizal P uptake pathway is not correlated with mycorrhizal responses in growth or total P uptake. New Phytol 162:511–524
- Söderberg KH, Olsson PA, Bääth E (2002) Structure and activity of the bacterial community in the rhizosphere of different plant species and the effect of arbuscular mycorrhizal colonisation. FEMS Microbiol Ecol 40:223–231

- St. Arnaud M, Elsen A (2005) Interaction of arbuscular mycorrhizal fungi with soil-borne pathogens and nonpathogenic rhizosphere micro-organisms. In: Declerck S, Strullu DG, Fortin FA (eds) *In vitro* culture of mycorrhizas. Springer, Springer, pp 217–234
- St. Arnaud M, Hamel C, Vimard B, Caron M, Fortin JA (1996) Enhanced hyphal and spore production of the arbuscular mycorrhizal fungus *Glomus intraradices* in an *in vitro* system in the absence of host roots. Mycol Res 100:328–332
- Stachowicz JJ (2001) Mutualism, facilitation, and the structure of ecological communities. Bioscience 51: 235–246
- Subramanian KS, Charest C (1997) Nutritional, growth, and reproductive responses of maize (Zea mays L.) to arbuscular mycorrhizal inoculation during and after drought stress at tasselling. Mycorrhiza 7:25–32
- Subramanian KS, Charest C (1998) Arbuscular mycorrhizae and nitrogen assimilation in maize after drought and recovery. Physiol Plant 102:285–296
- Subramanian KS, Charest C (1999) Acquisition of N by external hyphae of an arbuscular mycorrhizal fungus and its impact on physiological responses in maize under drought-stress and well-watered conditions. Mycorrhiza 9:69–75
- Tack FM, Callewaert OWJJ, Verloo MG (1996) Metal solubility as a function of pH in a contaminated, dredged sediment affected by oxidation. Environ Pollut 91:199–208
- Tawaraya K (2003) Arbuscular mycorrhizal dependency of different plant species and cultivars. Soil Sci Plant Nutr 49:655–668
- Tinker PB, Durall DM, Jones MD (1994) Carbon use efficiency in mycorrhizas: theory and sample calculations. New Phytol 128:115–122
- Toussaint JP, St. Arnaud M, Charest C (2004) Nitrogen transfer and assimilation between the arbuscular mycorrhizal fungus *Glomus intraradices* Schenck & Smith and Ri T-DNA roots *Daucus carota* L. in an *in vitro* compartmented system. Can J Microbiol 50:251–260
- Tylka GL, Hussey RS, Roncadori RW (1991) Axenic germination of vesicular-arbuscular mycorrhizal fungi: effects of selected *Streptomyces* species. Phytopathology 81:54–59
- van der Heijden MGA, Klironomos JN, Ursic M, Moutoglis P, Streitwolf-Engel R, Boller T, Wiemken A, Sanders IR (1998) Mycorrhizal fungal diversity deter-

mines plant biodiversity, ecosystem variability and productivity. Nature 396:69–72

- Vierheilig H, Piché Y (2002) Signaling in arbuscular mycorrhiza: facts and hypotheses. In: Buslig BS, Manthey JA JA (eds) Flavonoids in cell function. Kluwer, Dordrecht, pp 23–40
- Villegas J, Fortin JA (2001) Phosphorus solubilization and pH changes as a result of the interactions between soil bacteria and arbuscular mycorrhizal fungi on a medium containing NH₄⁺ as nitrogen source. Can J Bot 79:865–870
- Villegas J, Fortin JA (2002) Phosphorus solubilization and pH changes as a result of the interactions between soil bacteria and arbuscular mycorrhizal fungi on a medium containing NH₄⁺ as nitrogen source. Can J Bot 79:865–870
- Vitousek PM, Howarth RW (1991) Nitrogen limitation on land in the sea: how can it occur. Biogeochemistry 13:87–115
- von Alten H, Lindermann A, Schonbeck F (1993) Stimulation of vesicular-arbuscular mycorrhiza by fungicides or rhizosphere bacteria. Mycorrhiza 2:167–173
- Wagner GJ (1991) Secreting glandular trichomes: more than just hairs. Plant Physiol 96:675–679
- Wardle DA, Barker GM, Bonner KI, Nicholson KS (1998) Can comparative approaches based on plant ecophysiological traits predict the nature of biotic interactions and individual plant species effects in ecosystems? J Ecol 86:405–420
- Wardle DA, Bardgett RD, Klironomos JN, Setälä H, van der Putten WH, Wall DH (2004) Ecological linkages between above- and belowground biota. Science 304:1629–1633
- White JG, Zasoski RJ (1999) Mapping soil micronutrients. Field Crop Res 60:11–26
- Wiebes JT (1979) Co-evolution of figs and their insect pollinators. Annu Rev Ecol Syst 10:1–12
- Wright DP, Scholes JD, Read DJ, Rolfe SA (2005) European and African maize cultivars differ in their physiological and molecular responses to mycorrhizal infection. New Phytol 167:881–896
- Xavier LJC, Germida JJ (2003) Bacteria associated with *Glomus clarum* spores influence mycorrhizal activity. Soil Biol Biochem 35:471–478
- Young JPW, Johnston AWB (1989) The evolution of specificity in the legume-rhizobium symbiosis. Trends Ecol Evol 4:341–349

MicroRNAs and Their Role in Plants During Abiotic Stresses

12

Praveen Guleria, Deepmala Goswami, Monika Mahajan, Vinay Kumar, Jyoti Bhardwaj, and Sudesh Kumar Yadav

Abstract

Abiotic stresses have been considered as the growth-limiting factors affecting plants. Nutrient deficiency, drought, salinity, cold, submergence, and hypoxia are some of the severe types of abiotic stresses. Interdisciplinary research has been carried out to find stress-regulating mechanisms. MicroRNAs (miRNAs) are the newly discovered, 18–24 nucleotides long molecule of the genome. They have been considered as the key players against plant stress. They have been identified in plants, animals, humans, and even microbes. miRNAs have been shown to regulate various stressresponsive genes, proteins and transcription factors, thus helping to counteract adverse conditions. Various stress-inducible miRNAs have been identified and well characterized. Most of these miRNAs have been conserved among plants. This conservative nature has become the basis of development of computational methods of miRNA identifications, in additional to the traditional cloning approach. Presence of computational strategy has further simplified the miRNA prediction. Using this approach various stress-responsive miRNAs have been predicted, annotated and functionally validated from cotton, grapes, rice, maize, and soyabean. This chapter reviews the expanding world of miRNAs, methods unveiling miRNAs from various organisms, and specifically stress-induced miRNAs.

Keywords

miRNA • ESTs • GSSs • Drought • Salt • Nutrient stress • Hypoxia • Submergence • Plant responses

S.K. Yadav (⊠) • P. Guleria • D. Goswami • M. Mahajan
• V. Kumar • J. Bhardwaj
Plant Metabolic Engineering Laboratory,
Biotechnology Division, Institute of Himalayan
Bioresource Technology, Council of Scientific and Industrial Research, Palampur 176061,

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_12,

Himachal Pradesh, India

e-mail: sudeshkumar@ihbt.res.in; skyt@rediffmail.com

[©] Springer Science+Business Media, LLC 2012

1 Introduction

Gene regulation is the basic and foremost phenomenon assisting the possible survival of multicellular organisms in the adverse environmental conditions and varying evolutionary changes. Gene regulatory mechanisms are highly complex, diverse, variously linked, and flexible (Szymanski et al. 2003). This regulatory mechanism has been found to work at various important levels of genome function like chromosome segregation, transcription, RNA processing, and translation (Carthew and Sontheimer 2009). It has been the area of intense research in the recent years and has lead to the discovery of new mechanism termed as RNA interference (RNAi). RNAi was initially studied in plants followed by fungi and was later studied in almost each and every eukaryotic organism. It has been considered as the conserved mechanism of gene regulation among all eukaryotes. Most important feature of RNAi is the presence of RNA as a signal molecule that triggers gene regulation. These signaling RNA molecules are actually 22-25 nucleotides (nts) long small RNAs (sRNAs) (Mao et al. 2009).

sRNAs came into scenario when the hidden facts regarding eukaryotic gene regulation were explored. These are actually a part of noncoding RNAs (ncRNAs). ncRNAs are characterized as the RNA molecules lacking significant open reading frames that code for RNAs, not proteins. Genes coding for ncRNAs have been explored and annotated from certain genomes. This suggests that ncRNAs have a role to play during the cellular development, physiology, and maintenance. But still a lot is to be discovered about ncRNAs and genes encoding them. It has been expected that with the generation of sRNA libraries, EST analysis and new algorithms, sooner or later more and more ncRNAs and respective genes will be revealed. Thus, the horizons of plant genomics will be widened (MacIntosh et al. 2001).

ncRNAs are variously classified depending upon their origin and functions, out of which sRNAs have been extensively studied (Erdmann et al. 2001; Kim 2005; Rymarquis et al. 2008). sRNAs differ from rest of the RNAs of genome because of their small length and capability to bind to the Argonaute (AGO) family proteins (Ghildiyal and Zamore 2009). It has been observed that sRNAs mostly repress the expression of the target genes and hence, the phenomenon is also defined as RNA silencing. They have a sequence-specific mode of action. They direct the final effector proteins, i.e., AGO proteins to the target molecules by the approach of basepairing interactions (Carthew and Sontheimer 2009).

Several classes of sRNAs have emerged and a number of them have been identified and functionally characterized. On the basis of their origin, structure and functional identifications, three main categories of sRNAs have been identified: short interfering RNAs (siRNAs), microRNAs (miRNAs), and piwi-interacting RNAs (piRNAs). Out of these three, siRNAs and miRNAs have been broadly studied among plants whereas piR-NAs, found in mammalian testes, are still under exploration. The piRNAs are reported to have single-stranded RNA precursors, while miRNAs and siRNAs have double-stranded RNA precursors (Carthew and Sontheimer 2009). Both of these sRNAs have been characterized as riboregulators. They have the ability to act on DNA as well as on RNA (Vaucheret 2006).

miRNAs are 20-22 nts and are more abundant in the plant system. They act posttranscriptionally through mRNA degradation or translational repression. Unlike miRNAs, siRNAs are 21-24 nts long and act at both transcriptional and posttranscriptional levels by carrying out DNA methvlation, histone modification, and mRNA degradation (Sunkar and Zhu 2007). It has been mentioned that siRNAs and miRNAs have the tendency to adopt similar mechanism for gene regulation. But the kind of mechanism they adopt depends upon the level of complementarity between the target gene and sRNAs (Zeng et al. 2003). As far as their biogenesis is concerned, they differ on the basis of kind of precursors, mechanism of synthesis and maturation, formation of RNA-induced silencing complex (RISC) and finally mode of action on the target gene (Bartel 2004).

miRNAs are endogenous by origin. Genes encoding miRNAs are part of the host's genome and termed as miRNA genes. miRNA genes form a fold-back stem-loop structure which is around 10-100 nts long. This structure is known as primiRNAs. The double-stranded region created by the fold-back acts as a marker for Dicer-Like (DCL) enzyme, that cleaves pri-miRNAs and synthesizes mature miRNAs against the specific endogenous target mRNAs. Mature miRNAs bound to AGO proteins of RISC. This binding directs miRNAs toward the target mRNA, thus repressing the gene expression. siRNAs are endogenous as well as exogenous sRNAs. It has been found that siRNAs precursors are doublestranded RNAs but longer than miRNAs precursors. siRNAs precursors are again acted upon by certain unusual Dicer ribonuclease, giving rise to mature siRNAs (Zeng et al. 2003; Eckardt 2004).

It has been observed that among the known sRNAs, miRNAs have emerged as one of the important regulatory molecules of plant and animal system (Carrington and Ambros 2003; Bartel 2004). This very fact has initiated an era of evolutionary miRNA research in the scientific fraternity. It has lead to the identification and characterization of various miRNAs from plants as well as animals. This research, in turn, has enhanced the development of several new and novel experimental and computational approaches for miRNA prediction, isolation, and characterization. The miRNA study has revealed surprising facts related to genomics. It has also well played a part to understand metabolomics, proteomics, and stress biology. So, it can be concluded that miRNAs have initiated and well supported the era of interdisciplinary research.

2 miRNAs Isolation and Characterization

With the initiation of miRNA research, a lot of work has been carried out since then. Fifteen classes of miRNAs have been identified and well characterized from the model plant *Arabidopsis* (Bartel and Bartel 2003). Twenty miRNA families have been computationally predicted in case of *Oryza*, out of which 14 miRNAs have been experimentally isolated and characterized (Sunkar et al. 2005). Similarly, eight novel miRNAs have been isolated and characterized from *Medicago* (Szittya et al. 2008). miRNAs have been identified by cloning as well as computational approaches. The two strategies for miRNAs identification are:

2.1 In Silico Prediction and Characterization

The very first miRNAs were experimentally isolated and cloned by biochemical and genetic approaches (Lee et al. 1993). But in cases of low expression and tissue-specific miRNAs, cloning is not an appropriate approach. As a result, it has been replaced by the method of computational predictions and experimental validations (Zhang et al. 2006). Since then computational approaches have dominated the field of miRNAs research. The traditional computational strategy was based on the decoded genome sequences of certain model species. In that case only completely sequenced genomes could be analyzed for miR-NAs prediction, leaving behind the species with unsequenced genomes (Zhang et al. 2005). Because of the inefficiency associated with the traditional method, presently a new approach has been developed. It has been reported that within same kingdom as well as sometimes in between different kingdoms, miRNAs show evolutionary conservations. It gave the idea that comparative genomics could be a powerful strategy to identify miRNAs. So, this conservative nature was employed to predict miRNAs (Zhang et al. 2007). Most recently, genomic survey sequences (GSSs) and expressed sequence tags (ESTs) have been adopted for mining novel, undiscovered miRNAs (Zhang et al. 2005).

Various steps involved in the in silico miRNAs prediction are: (1) miRNAs prediction using ESTs and GSSs analysis, (2) identification of potential miRNAs, and (2) miRNAs target identification.

2.1.1 miRNAs Prediction Using ESTs and GSSs Analysis

It has been known that ESTs are the cDNA sequences of the expressed genes. In case of organisms with incomplete genomic draft, ESTs have been considered as an alternative tool for gene discovery. Also, ESTs and expressed genes are actually obtained from true gene expression. Thus it justifies the use of ESTs to predict miRNAs for species with undiscovered genomes (Matukumalli et al. 2004; Zhang et al. 2005). National Center for Biotechnology Information (NCBI) contains 22,165,266 entries of ESTs from various organisms in its EST database (Boguski et al. 1993). ESTs could be obtained from EST database of NCBI, dbEST (http://www.ncbi.nlm.nih.gov/ dbEST/). Initially, redundancy within the ESTs is removed by using software CAP3 (http:// pbil.univlyon1.fr/cap3.php). This software presents the overlapping sequences as contigs and nonoverlapping sequences as singletons. The contigs are used for miRNAs prediction. These processed sequences, i.e., contigs are submitted to miRNA-finder (http://bioinfo3. noble.org/mirna/). This software predicts the possible miRNAs from the query sequence. In order to further remove the redundant and overlapping miRNAs from the predicted ones, a reference set of miRNAs is used. This reference miRNAs set belongs to any of the closely related organisms and can be obtained from miRNA registry database (http://miRNA. sanger.ac.uk) (Manila et al. 2009). A computational tool MicroHARVESTER (http://www. ab.informatik.unituebingen.de/bribane/tb/ index/php) has been designed to search homology between the predicted miRNAs and previously detected miRNAs (Dezulian et al. 2006). While using GSSs, the complete sequence is used as miRNA precursor sequence. Overlapping as well as protein coding sequences is removed. The remaining nonprotein sequences are used for further proceedings (Sunkar and Jagadeeswaran 2008). The predicted miRNAs are then analyzed to determine the potential miRNAs among them.

P. Guleria et al.

2.1.2 Identification of Potential miRNAs

Potential miRNAs are predicted on the basis of data obtained from their secondary structures. For the predicted miRNAs or pre-miRNAs, secondary structure is determined by using software MFOLD 3.1 (http://www.bioinfo.rpi.edu/applications/mfold/rna/form1.cgi) (Mathews et al. 1999; Zuker 2003). In order to screen the miRNAs, following criteria are followed (Xie et al. 2007; Sunkar and Jagadeeswaran 2008):

- 1. Pre-miRNA sequences should fold into an appropriate stem-loop hairpin secondary structure that contains around 22 nts mature miRNA sequence within one arm of the hairpin.
- Predicted mature miRNAs are allowed to have only 0–3 nts mismatch in comparison to the previously known plant mature miRNAs.
- 3. Predicted secondary structures should have higher negative minimal free energies (MFEs) and minimal free energy index (MFEIs) than the other different types of RNAs.
- 4. No loop or break is allowed in the miRNAs sequence.
- 5. miRNAs sequence should have 30–70% A+U content.
- miRNAs should have no more than six mismatches with the opposite miRNA* sequence in the other arm.

miRNAs satisfying the following parameters are thus considered as the potential miRNAs.

2.1.3 miRNAs Target Prediction

It has been known that plant miRNAs have the tendency to bind to the protein coding regions of their mRNA targets. The binding of miRNA to target mRNA is based on the nature of complementarity between them, either perfect or nearly perfect (Xie et al. 2007). This reveals the usage of homology search to determine miRNA targets (Zhang et al. 2007). A software for miRNAs target prediction based upon above said approach is miRU2 (http://bioinfo3.noble.org/miRU2/). This tool has been applied for the prediction of several miRNA targets (Fig. 12.1) (Manila et al. 2009). Presently, computational approaches are being further explored to ease the process of novel



miRNAs identification. One of such modified computational approach is miRTRAP. miRTRAP stands for miRNA Tests for Read Analysis and Prediction. This tool simplifies the systemic and whole-genome prediction of miRNAs by making use of high-throughput sequencing data. This tool utilizes a system of binary decisions that are based on biochemical mechanisms behind miRNA biogenesis (Hendrix et al. 2010). This program has been able to predict 400 putative miRNAs loci from simplest chordate Ciona intestinalis. No doubt in silico prediction provides an efficient and easy method for miRNA prediction and target identification. For organisms with incomplete genomic draft, in silico methods offer a promising approach to study genetic regulations. But an important fact to note is that all the data obtained by in silico predictions require experimental validations.

2.2 sRNA Library Preparation and miRNA Characterization

The experimental approach of miRNAs identification has been termed as RNomics. It deals with the preparation of cDNA libraries of sRNAs (Huttenhofer et al. 2005). In order to develop sRNA libraries, a number of strategies have been adopted. Various commercial kits are available that specifically isolate sRNAs fraction from the sample. Other method is to isolate total RNA and then purification of sRNAs fraction from the large RNA fractions. sRNAs fraction is usually resolved on 15% polyacrylamide gel and then eluted. sRNAs range from 18 to 26 nts in length, as a result before cloning, adaptors are ligated to them. Before ligations, sRNAs are often polyadenylated to avoid circularization of linker RNAs (Devor et al. 2009). Ligation of adaptors on both

5' and 3' ends helps in the visualization of sRNAs in the further steps. Ligated product is reversetranscribed using adaptor-specific primers. Resulting PCR product is used for cloning purpose and cloned into the desired vector (Sunkar et al. 2005). Prepared sRNA clones are then sequenced.

Sequencing of clones using BLAST analysis against Genbank genome assemblies helps to determine its nature, whether it is a transposon, any degraded product or already known noncoding RNAs except miRNAs/siRNAs. Candidates with perfect matches against genome sets are further used for fold-back secondary structure predictions using MFOLD 3.1 software (Zuker 2003). sRNA sequences are folded with flanking sequences in five contexts: (1) 300 bp upstream and 20 bp downstream, (2) 150 bp upstream and 20 bp downstream; (3) 150 bp upstream and 150 downstream, (4) 20 bp upstream and 150 downstream, and (5) 20 bp upstream and 300 bp downstream (Lu et al. 2005). Candidate clones satisfying all the criteria are considered as potential miRNAs. Further target for these miRNAs is determined using software miRU2. sRNAs isolation and cloning is a laborious job. Also it does not always ensure that each clone is a miRNA. As a result, in silico prediction of miRNAs and their experimental validation outshine the above technique.

3 miRNAs Annotation Criteria

With the increasing rate of sRNA discovery, there is an increased requirement for miRNA annotations. A specific criterion has been managed for the purpose of annotations that include experimental as well as computational data. It has been considered that any sRNA should fulfill two conditions to be recognized as miRNA. These two conditions are expression and biogenesis criteria. Where expression criteria stands for the identification of miRNA by cloning or hybridization; biogenesis criteria includes folding of predicted precursors into a defined stem-loop hairpin structure, phylogenetic miRNA conservation, and increased precursor accumulation on decreased Dicer activity (Ambros et al. 2003). With the development of more sophisticated techniques for miRNAs prediction, it has been observed that some miRNAs followed the criteria but some not. This has led to the generation of new annotation parameters. These parameters are divided as: primary criteria and ancillary criteria.

Primary criterion defines the fundamental features of miRNAs. It states that biogenesis of around 21 nts miRNA/miRNA* duplex should occur by an excision of a qualifying stem-loop precursor. These criteria have been considered as the most necessary and sufficient parameters for miRNAs annotation. In order to fulfill this parameter, both miRNA as well as miRNA* should satisfy the conditions. In case of miRNA*-deficient clones, the potential miRNA candidate should be isolated and sequenced from multiple, independent libraries. During sequencing, often very low abundance of one or two sequencing reads of putative miRNA is obtained. In such cases, even RNA gel blots fail to satisfy the qualifying conditions of primary criteria. It is so because detection of sRNAs via blotting is unable to determine whether it is a miRNA, siRNA, or a decay product of larger precursors. To help out in such cases low depth sequencing and extensive blot analysis using multiple probes is utilized.

In addition to primary criteria, certain other features have also been determined for miRNAs annotation. However, satisfaction of ancillary parameters is not essentially required but fulfilling these features would further enhance the significance of miRNAs annotation. Ancillary criteria include miRNAs conservation, their targets, Dicer-like 1 (DCL1) dependence, RNA Dependent RNA polymerases (RDRs) independence. Conservation of stem-loop secondary structure and mature miRNA sequence among lineages is a sufficient proof for miRNAs annotation. Target prediction is not a necessary miRNAs annotation. There could be cases where conserved miRNAs are target-less or nonconserved miR-NAs have well characterized targets. So, in both cases, functional characterization of miRNAs is not required for its annotation. Plants require DCL1 enzyme for miRNAs generation but not in all cases. The dcl1 mutant has also been reported to possess miRNAs. So, DCL1 dependence cannot be a strict annotation requirement. Similar is with RDRs. RDRs generate dsRNA precursors. These precursors lead to siRNAs biogenesis and hence, RDR dependence is an essential feature of siRNAs. But miRNAs are not synthesized from RDR-generated dsRNA molecules. As a result, RDR independence should not be a necessary condition for miRNAs annotation (Meyers et al. 2008). This can be summed up as: the primary criterion, but not the ancillary, is a kind of eligibility for sRNAs to be annotated as miRNAs.

4 miRNAs Nomenclature Parameters

The process on submission of novel miRNAs to miRBase registry requires miRNAs to be well named and numbered. Certain rules have been formulated to ensure proper nomenclature of miRNAs (Griffiths-Jones et al. 2008; Meyers et al. 2008). The rules are discussed as follows:

- A three-letter prefix "miR" is written in every miRNAs name. This is followed by a number denoting the order in which they were discovered, e.g., miR1, miR2. In an organism, miRNA genes are named as "mir," e.g., mir1, mir2 in animals and MIR1, MIR2 in plants. Identical miRNAs are given the same numerical designations, no matter from which organism they were isolated.
- 2. Similar or identical miRNAs, originating from a common locus should be assigned that number followed by a sequential alphabetical suffix, e.g., MIR172a, MIR172b.
- miRNA is assigned a three-letter prefix that specifies the genus and species-generating miRNA, e.g., miRNA from *Brassica napus* designated as bna-miR167a, bna-miR167b.
- 4. In some cases, miRNA genes encode similar miRNAs, that for certain historical reasons have been assigned distinct identifiers. They are thus grouped under same families, e.g., miR156/157, miR170/171 families.
- While naming miRNAs derived from the same arm of stem-loop, miRNA designations should also consider the number of mismatches

among the miRNAs. Mismatches from zero to four are considered as typical and are acceptable up to four.

- miRNAs derived from same sequence but having different targets can be classified under different families.
- In case miRNAs originating in equal proportion from same miRNA/miRNA* duplex, suffixes 5p and 3p are used. 5p and 3p signify the end, either 5' or 3', giving rise to the miRNA sequences.

5 Role of miRNAs in Abiotic Stress

Abiotic stresses have already been known as the primary plant growth-limiting factors. Abiotic stress is a broad term including various physiochemical stresses like salinity, drought, temperature, and oxidative stress (Vinocur and Altman 2005). These stresses have been reported to reduce the average yield of most of the crops by 50% (Tuteja and Sopory 2008). It has been observed that plants possess the ability of monitoring fluctuations occurring around its niche and have the inbuilt capacity to respond against them. Plants survive in unfavorable stress conditions by carrying out variations at their physiological and genetic grounds. Various genes, proteins, transcription factors, DNA histone-modifying enzymes, and several metabolites have been reported responsible for plant stress tolerance (Cushman and Bohnert 2000; Vinocur and Altman 2005; Chinnusamy and Zhu 2009). Nowa-days, molecular studies are being done to decode the mechanism by which plants cope with these adverse conditions. These studies have unveiled that sRNAs, mostly miRNAs present within the plant system, have the ability of regulating stress-responsive factors (Sunkar and Zhu 2004). It has been reported that miRNAs either upregulate or downregulate in response to stress. Various miRNAs specifically responding to abiotic stresses have been reported from various plants as Arabidopsis, Oryza, Nicotiana, Brassica, Gossypium, etc. Role of several miRNAs in various kind of stresses are discussed.

5.1 miRNAs in Response to Nutrient Stress

Plants often grow in nutrient-deprived soils, where amount of macronutrients is less than the required concentration. In such cases, plants sense the internal and external mineral ion concentration and adapt in response to the nutrient deficiency. Various miRNAs have been reported to act during nutrient stress.

5.1.1 Phosphorus Starvation

Phosphorus is the structural unit behind nucleic acids, cellular membranes and energy currency ATP (Chiou 2007). Phosphorus is taken via plant roots in the form of inorganic phosphate $[Pi(HPO_4^{2-})]$ (Marschner 1995; Schachtman et al. 1998). It has been found that external phosphorus concentration is 100-1,000 times lower than internal concentration (Marschner 1995). It has lead to the development of adaptive responses by plants to maintain phosphate homeostasis. With the discovery of miR395 involvement during sulfate stress, it has been considered that miRNAs do play a role in nutrient starvation. So, afterwards miR399 was reported to act specifically in response to Pi deprivation (Fujii et al. 2005; Bari et al. 2006). Affymetrix GeneChip analysis in Arabidopsis has revealed that At2g33770 is the target gene of miR399. miR399 was reported to target multiple sites on the 5' untranslated regions (UTRs) of At2g33770 mRNA encoding a ubiquitin-conjugating E2 enzyme AtUBC24 (Sunkar and Zhu 2004; Kraft et al. 2005). It was reported that during low Pi concentration, miR399 was highly expressed with lower accumulation of UBC24 mRNA transcript and vice versa during higher Pi content. Thus, it is demonstrating a direct relation between E2 and miR399.

Role of miR399 was further confirmed by generating transgenic *Arabidopsis* overexpressing miR399. The transgenics showed an enhanced expression of miR399 both in decreased and increased Pi levels. Elevated levels of miR399 accumulation suggested a successful processing and expression of precursor RNAs. Further a detectable loss in the expression of E2 RNA was observed. The transgenics experimentally validated that miR399 down regulated the E2 mRNA levels (Chiou et al. 2006). The same study proved that overaccumulation of miR399 was responsible for the increased uptake of Pi. In wild plants, Pi normally constitutes 0.2% of plant dry matter (Schachtman et al. 1998). While in transgenics overexpressing miR399, Pi level was reported to increase up to 1.8–2.0%. The plants also showed symptoms of Pi toxicity like chlorosis, necrosis (Delhaize and Randall 1995; Shane et al. 2004). It was further reported that overaccumulation of Pi in transgenics was due to the increased uptake of Pi (Chiou et al. 2006).

5.1.2 Sulfur Deficiency

Cysteine and methionine are two important sulfur-containing amino acids. Animals and humans cannot synthesize them, so they are essential part of their diets. In plants, sulfur is a component of important anti-cancerous secondary metabolites (Talalay and Fahey 2001). Sulfur concentration keeps changing; as a result plants initiate their sulfate transport system for an enhanced uptake of sulfur. Sulfur is taken by the plants in the form of inorganic sulfate (Nikiforova et al. 2006). It has been reported that miR395 has a role in sulfur starvation (Kawashima et al. 2009). miR395 has been observed to accumulate in sulfur deficient conditions. It has been found to regulate the activities of low-affinity sulfate transporter (SULTR2;1) and ATP sulphurylases (APS1, APS3 and APS4) (Jones-Rhoades and Bartel 2004; Allen et al. 2005). ATP sulphurylases are the key enzymes of sulfate assimilation pathway and SULTR2;1 is a transporter helping sulfate translocation from roots to shoots (Takahashi et al. 2000; Allen et al. 2004; Jones-Rhoades and Bartel 2004). miR395mediated cleavage of APS1, APS4, and SULTR2;1 has been reported both in roots and leaves whereas for APS3 in the leaves only (Kawashima et al. 2009). While studying the relationship between miR395 and ATP sulphurylases, a negative correlation has been determined (Jones-Rhoades and Bartel 2004); whereas for SULTR2;1 contradictory results were obtained. It was found that in sulfate-deficient conditions, in leaves with miR395 upregulation, SULTR2;1 mRNA was downregulated. But at the same time, roots showed upregulation both in the levels of miR395 as well as SULTR2;1. Later, it was found that a transcription factor Sulfur Limitation 1 (SLIM1) was actually responsible for the induction of miR395. This was further experimentally validated by carrying out miR395 and SULTR2;1 mRNA expression analysis in *slim1* mutants (Kawashima et al. 2009).

5.1.3 Copper Maintenance

Copper is an essential micronutrient required for photosynthesis, ethylene perception, respiratory electron transport, and oxidative stress protection (Marschner 1995). Plant proteins requiring copper as an important cofactor are plastocyanin, laccases, and intracellular proteins formed by Cu, Zn-superoxide dismutase (SOD). Plastocyanin is an essential protein of electron transport chain, laccases are required for wound healing, lignin synthesis, stress responsiveness and Cu, Zn-SOD for regulating reactive oxygen species (ROS) (Abdel-Ghany and Pilon 2008).

It has been reported that CSD1, CSD2, and CSD3 are three Cu, Zn SOD encoding genes, out of which CSD1 and CSD2 have been considered as two important isoforms. Copper deficiency has been shown to downregulate the expression of CSD1 and CSD2. Later it has been proved that a miR398 is responsible for their downregulation (Yamasaki et al. 2007). It has been observed that copper deficiency causes upregulation of miR398 that resulted in the reduced expression of above two genes. This also gives a view of miR398 role toward oxidative stress. It has been found that oxidative stresses causes transcriptional downregulation of miR398. This downregulation is responsible for posttranscriptional CSD1 and CSD2 mRNA accumulation, thus providing tolerance to oxidative stress. By generating CSD2 overexpression Arabidopsis transgenics, it has been proved that plants neglecting miR398-mediated suppression of CSD2 are more tolerant to high light, heavy metals, and oxidative stress (Sunkar et al. 2006).

Further experimentation has shown that three more miRNAs, miR397, miR408, and miR857 are accumulated during copper starvation. Computational analysis has revealed that these miRNAs target laccase and plastocyanin mRNAs (Sunkar et al. 2005; Yamasaki et al. 2007). Cleavage site analysis has experimentally validated the computational predictions. Expression analysis data of miR397, miR408 and miR857 in presence of varying copper concentrations have evidenced an inverse relationship between these miRNAs and their target genes. It has been considered that miRNA-mediated downregulation of the respective genes offer a significant mechanism of copper homeostasis (Abdel-Ghany and Pilon 2008).

5.2 miRNAs in Response to Cold and Related Stresses

Cold stress is a house of other secondary stresses. Cold stress refers to variation in temperature ranges causing freezing $(T < 0^{\circ}C)$ and chilling $(T < 20^{\circ} C)$. This stress itself causes direct damage to the genetic potential of plant by causing inhibition of plant secondary metabolism. In addition, cold stress is responsible for causing further damage due to osmotic, oxidative, and other kind of abiotic stresses (Chinnusamy et al. 2007). Microanalysis has revealed that 17% of cold upregulated genes code for transcription factors. On the contrary, only 7% of cold downregulated genes code for transcription regulators. This has given a view that plants undergo cold acclimatization via post-transcriptional downregulation of genes. miRNAs have been observed as an efficient genetic regulators, so they might have a role to play in cold stress. Later, it was hypothesized that the miRNAs targeting growth and development responsive genes might be cold-responsive. Various miRNAs have been found responsive to cold and its related stresses. It has been reported that miR393, miR397b, and miR402 are upregulated in response to cold, ABA, dehydration, and salt stress; whereas miR389a.1 is downregulated under similar set of conditions. The cold-specific upregulation has also been observed in case of miR319c. Oxidative stress-responsive miR393 has been found to be upregulated during cold stress. miR393 targets E3 ubiquitin ligase mRNA causing their cleavage during cold acclimatization.

An F-Box protein similar to glucose repression resistance 1 (GRR1) has been known to undergo miR393-mediated cleavage (Sunkar and Zhu 2004). Sugar has been considered as a signaling molecule regulating the growth and developmental processes during various stresses. miRNAmediated targeting of sugar-related gene suggests that miR393 has the ability of integrating sugar signaling with cold and other related stresses (Sunkar et al. 2007). Most recently, 18 novel coldresponsive miRNA families have been reported from rice. Most of these have been shown to undergo downregulation. Out of these, three families have been characterized as miR167, miR319, and miR171. miR167 and miR319 have been observed to show downregulation in their expression pattern, whereas miR319 has shown variableexpression profiles (Lv et al. 2010).

5.3 miRNAs in Response to Mechanical Stress

Plants have various stress-sensing mechanisms. It has been observed that plants undergo certain developmental changes in xylem and cambium to increase its mechanical support. Two types of woody tissues, tension wood and opposite woods, synthesized in response to stress are termed as reaction woods (Wu et al. 2000). This long-term deposition of supporting woody tissues is one of the most critical plant defense mechanism against mechanical stress. In order to have a better understanding of plant developmental changes, miRNA networks ruling xylem tissues were explored in one of the species of *Populus* (Lu et al. 2005). Experimental analysis has revealed the presence of 21 novel miRNAs in the developing xylem and phloem tissues. Presence of miRNAs suggests their role in cambium differentiation activities. The cloned miRNAs were designated as miR156, miR472, miR160, miR164, miR171, miR473, miR477, miR478, miR479, and miR480. A comparative expression analysis of miRNAs was carried out in tension-stressed and normal-developing xylem tissues. A detectable difference was observed in the expression patterns of most of the miRNAs. Depending upon the expression patterns, miRNAs were categorized as:

- 1. Type A miRNAs: suppressed by tension and compression stresses. This type includes miR156, miR162, miR164, miR475, miR480, and miR481.
- Type B miRNAs: upregulated in tension and compression stresses. This type includes miR408.
- Type C miRNAs: upregulated specifically in compression tissues. This type includes miR159, miR476, and miR479.
- Type D miRNAs: suppressed only in compression tissues. This type includes miR160 and miR172.
- Type E miRNAs: attenuated induction specifically by tension stress. This comprises of miR168.

It has been reported that type A and B miR-NAs have the ability of counteracting against overall mechanical stress without differentiating between compression and tension stresses. Type C, D, and E miRNAs have been considered responsible for the specialized regulatory mechanisms behind development of reaction woods (Lu et al. 2005).

5.4 miRNAs in Response to Salt Stress

Salt stress has been considered as one of the most severe abiotic stress. Plants have the ability of altering their genetic profile in order to counteract against salinity. miRNAs microarray analysis has revealed the presence of 98 miRNAs belonging to 27 miRNA families during varying salt treatments (Ding et al. 2009). It has been observed that most of the miRNAs responding to salt stress are directly involved in the regulation of transcription factors (TFs). From Zea mays, miR159a/b, miR164a/b/c/d and miR1661/m have been cloned that targets TFs Myb, NAC1 and homeodomainleucine zipper protein (HD-ZIP) (Jones-Rhoades and Bartel 2004). Other salt-responsive TFs targeted by miRNAs were MADS-box proteins and zinc-finger proteins (Fang et al. 2006). Further experimentation has lead to the cloning of miR474,

miR395, and miR396 family miRNAs from *Zea* mays. miR474 and miR395 were reported to target negative regulators of salt tolerance. They were upregulated during salt stress, causing suppression of the respective factors. On the contrary, miR396 was reported to downregulate in presence of salt stress.

It has been observed that during salt stress miR395, miR474 initiated a salt-induced nonspecific pathway for the maintenance of continuous energy supply. Both of these miRNAs were reported to upregulate in salt-stressed plants, thus targeting the key enzymes of plant energy transduction pathways, NADP-ME and NAD-ME (Cheng and Long 2007; Ding et al. 2009). It has already been known that salt stress causes the excess accumulation of ROS in plants. It has been experimentally validated that miR528 from Oryza sativa targets Cu/Zn SOD, causing its upregulation and scavenging ROS (Kim et al. 2007). Another miRNA family, miR169 has been known to be induced by high salinity. It causes cleavage of CCAAT box-binding transcription factor and thus plays role in transcriptional regulation of large number of genes (Zhao et al. 2009).

5.5 miRNAs in Response to Drought Stress

Drought stress is considered as a moderate loss of water content. This stress causes extensive water loss, stomatal closure, hampered gas exchange, disruption of cellular metabolism, and leads to death in severe cases (Jaleel et al. 2009). Genomewide miRNA profiling has revealed 30 droughtresponsive miRNAs from Oryza sativa. Out of these, 16 miRNAs namely miR156, miR159, miR168, miR170, miR171, miR172, miR319, miR396, miR397, miR408, miR529, miR896, miR1030, miR1035, miR1050, miR1088, and miR1126 were downregulated in response to drought stress. The remaining miRNAs miR159, miR169, miR171, miR319, miR395, miR474, miR845, miR851, miR854, miR896, miR901, miR903, miR1026, and miR1125 were significantly upregulated during drought stress. It has

been considered that these cloned miRNAs are actually responsible for the physiological and developmental changes occurring during drought stress. miR854 and miR170 have been confirmed as novel drought-induced miRNAs from rice. Rest of the miRNAs requires experimental verifications (Zhou et al. 2010).

5.6 miRNAs in Response to Hypoxia

Hypoxia is a state of reduced availability of oxygen. But it should not be confused with flooding or water logging. Till date very less is known about hypoxia and its responsive signaling cascade. Recently, a study has been carried out to predict the role of TFs and miRNAs in response to hypoxia. More than 1,900 TFs and 180 miRNAs primary transcripts from Arabidopsis roots have been exposed to hypoxic conditions via quantitative PCR. It has been found that TFs do play a significant role to regulate the expression of hypoxiainduced genes, whereas only single miRNA, miR391 showed a potent activity against hypoxia. Thus, it has been concluded that TFs are the potent transcriptional regulators of hypoxia and miRNAs are a minor players (Licausi et al. 2010).

5.7 miRNAs in Response to Submergence

Submergence of plants or flooding of soil causes adverse anaerobic conditions. Plants carry out several morphological, physiological, and metabolic changes to survive the adversity but the underlying mechanism remains unknown. It has been thought that miRNAs might have submergence-responsive roles. Microarray analysis has resulted into the prediction of 39 submergenceresponsive miRNAs. It has been found that these miRNA-targeted enzymes are involved in carbohydrate and energy metabolism, ROS elimination pathways. Most of the targeted mRNAs possessed *cis*-acting elements that are considered as an essential requirement for anaerobic responses (Zhang et al. 2008).
6 Conclusion and Future Perspective

In the recent times of scientific research, efficient work has been done in the direction of exploring miRNAs, their mechanism of action and applications. But many questions still remain unanswered. miRNAs not only influence plants but humans, animals and microbes as well. It was a microbe from where miRNAs were first isolated. miRNAs have a broad area of influence. Talking about plants, miRNAs play an effective role to maintain plant's growth, development, stress tolerance and cellular integrity. It has been well known that miRNAs either upregulate or downregulate to influence the expression of genes. miRNAs help plants to survive the stressful conditions. An important fact to notice is that from the beginning of miRNAs research till date, scientists are still keenly involved in discovering more and more sRNAs and finding their various applications to help plant maintenance. Development of overexpression transgenics and knockout mutants has further expanded the knowledge bank. The field of miRNA research is blooming and is going to grow in the years to come. Discoveries of many more miRNAs and invention of technologies using these wonder molecules will help to fight adversity. Thus miRNAs can be rightly pronounced as "the miracle of genome."

References

- Abdel-Ghany SE, Pilon M (2008) MicroRNA-mediated systemic down-regulation of Copper protein expression in response to low copper availability in *Arabidopsis*. J Biol Chem 283:15932–15945
- Allen E, Xie Z, Gustafson AM, Sung GH, Spatafora JW, Carrington JC (2004) Evolution of microRNA genes by inverted duplication of target gene sequences in *Arabidopsis thaliana*. Nat Genet 36:1282–1290
- Allen E, Xie Z, Gustafson AM, Carrington JC (2005) microRNA-directed phasing during trans-acting siRNA biogenesis in plants. Cell 121:207–221
- Ambros V, Bartel B, Bartel DP, Burge CB, Carrington JC, Chen XM, Dreyfuss G, Eddy SR, Griffiths-Jones S, Marshall M, Matzke M, Ruvkun G, Tuschl T (2003) A uniform system for microRNA annotation. RNA 9:277–279

- Bari R, Datt PB, Stitt M, Scheible WR (2006) PHO2, microRNA399, and PHR1 define a phosphate-signaling pathway in plants. Plant Physiol 141:988–999
- Bartel DP (2004) MicroRNAs: genomics, biogenesis, mechanism, and function. Cell 116:281–297
- Bartel B, Bartel DP (2003) MicroRNAs: at the root of plant development? Plant Physiol 132:709–717
- Boguski MS, Lowe TM, Tolstoshev CM (1993) dbEST database for "expressed sequence tags". Nat Genet 4:332–333
- Carrington JC, Ambros V (2003) Role of microRNAs in plant and animal development. Science 301:336–338
- Carthew RW, Sontheimer EJ (2009) Origins and mechanisms of miRNAs and siRNAs. Cell 136:642–655
- Cheng Y, Long M (2007) A cytosolic NADP-malic enzyme gene from rice (*Oryza sativa L*.) confers salt tolerance in transgenic *Arabidopsis*. Biotechnol Lett 29:1129–1134
- Chinnusamy V, Zhu JK (2009) Epigenetic regulation of stress responses in plants. Curr Opin Plant Biol 12:1–7
- Chinnusamy V, Zhu JH, Zhu J-K (2007) Cold stress regulation of gene expression in plants. Trends Plant Sci 12:444–451
- Chiou TJ (2007) The role of microRNAs in sensing nutrient stress. Plant Cell Environ 30:323–332
- Chiou TJ, Aung K, Lin SI, Wu CC, Chiang SF, Sua CL (2006) Regulation of phosphate homeostasis by microRNA in Arabidopsis. Plant Cell 18:412–421
- Cushman JC, Bohnert HJ (2000) Genomic approaches to plant stress tolerance. Curr Opin Plant Biol 3:117–124
- Delhaize E, Randall PJ (1995) Characterization of a phosphate-accumulator mutant of *Arabidopsis thaliana*. Plant Physiol 107:207–213
- Devor EJ, Huang L, Abdukarimov A, Abdurakhmonov IY (2009) Methodologies for in vitro cloning OF Small RNAs and application for plant genome(s). Int J Plant Genomics 2009:1–10
- Dezulian T, Remmert M, Palatnik JF, Weigel D, Huson DH (2006) Identification of plant microRNA homologs. Bioinformatics 22:359–360
- Ding D, Zhang L, Wang H, Liu Z, Zhang Z, Zheng Y (2009) Differential expression of miRNAs in response to salt stress in maize roots. Ann Bot 103:29–38
- Eckardt NA (2004) Small RNA on the move. Plant Cell 16:1951–1954
- Erdmann VA, Barciszewska MZ, Hochberg A, De Groot N, Barciszewski J (2001) Regulatory RNAs. Cell Mol Life Sci 58:1–18
- Fang Q, Xu Z, Song R (2006) Cloning, characterization and genetic engineering of FLC homolog in *Thellungiella halophila*. Biochem Biophys Res Commun 347:707–714
- Fujii H, Chiou TJ, Lin SI, Aung K, Zhu JK (2005) A miRNA involved in phosphate-starvation response in *Arabidopsis*. Curr Biol 15:2038–2043
- Ghildiyal M, Zamore PD (2009) Small silencing RNAs: an expanding universe. Nat Rev Genet 10:94–108
- Griffiths-Jones S, Saini HK, van Dongen S, Enright AJ (2008) miRBase: tools for microRNA genomics. Nucleic Acids Res 36:D154–D158

- Hendrix D, Levine M, Shi W (2010) miRTRAP, a computational method for the systematic identification of miRNAs from high throughput sequencing data. Genome Biol 11:39–50
- Huttenhofer A, Schattner P, Polacek N (2005) Non-coding RNAs: hope or hype? Trends Genet 21:289–297
- Jaleel CA, Manivannan P, Wahid A, Farooq M, Al-Juburi HJ, Somasundaram R, Panneerselvam R (2009) Drought stress in plants: a review on morphological characteristics and pigments composition. Int J Agric Biol 11:100–105
- Jones-Rhoades MW, Bartel DP (2004) Computational identification of plant microRNAs and their targets, including a stressinduced miRNA. Mol Cell 14:787–799
- Kawashima CG, Yoshimoto N, Maruyama-Nakashita A, Tsuchiya YN, Saito K, Takahashi H, Dalmay T (2009) Sulphur starvation induces the expression of microRNA-395 and one of its target genes but in different cell types. Plant J 57:313–321
- Kim VN (2005) Small RNAs: classification, biogenesis, and function. Mol Cells 19:1–15
- Kim DW, Shibato J, Agrawal GK, Fujihara S, Iwahashi H, Kim DH, Shim IS, Rakwal R (2007) Gene transcription in the leaves of rice undergoing salt-induced morphological changes (*Oryza sativa L*). Mol Cells 24:45–59
- Kraft E, Stone SL, Ma L, Su N, Gao Y, Lau OS, Deng XW, Callis J (2005) Genome analysis and functional characterization of the E2 and RING-type E3 ligase ubiquitination enzymes of *Arabidopsis*. Plant Physiol 139:1597–1611
- Lee RC, Feinbaum RL, Ambros V (1993) The *C. elegans* heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14. Cell 75:843–854
- Licausi F, Weits DA, Pant BD, Scheible WR, Geigenberger P, van Dongen JT (2010) Hypoxia responsive gene expression is mediated by various subsets of transcription factors and miRNAs that are determined by the actual oxygen availability. New Phytol. doi:10.1111/j.1469-8137.2010.03451.x
- Lu S, Sun YH, Shi R, Clark C, Li L, Chiang VL (2005) Novel and mechanical stress-responsive microRNAs in *Populus trichocarpa* that are absent from *Arabidopsis*. Plant Cell 17:2186–2203
- Lv DK, Bai X, Li Y, Ding XD, Ge Y, Cai H, Ji W, Wu N, Zhu YM (2010) Profiling of cold-stress-responsive miRNAs in rice by microarrays. Gene 459:39–47
- MacIntosh GC, Wilkerson C, Green PJ (2001) Identification and analysis of *Arabidopsis* Expressed Sequence Tags characteristic of Non-Coding RNAs. Plant Physiol 127:765–776
- Manila TM, Riju A, Priya Darshini KL, Chandrasekar A, Eapen SJ (2009) *In silico* microRNA identification from paprika (*Capsicum annuum*) ESTs. Nat Proc 3737:1–10
- Mao Y, Xue X, Chen X (2009) Are small RNAs a big help to plants? Sci China C Life Sci 52:212–223
- Marschner H (1995) Mineral nutrition of higher plants, 2nd edn. Academic, London, pp 333–347

- Mathews DH, Sabina J, Zuker M, Turner DH (1999) Expanded sequence dependence of thermodynamic parameters improves prediction of RNA secondary structure. J Mol Biol 288:911–940
- Matukumalli LK, Grefenstette JJ, Sonstegard TS, van Tassell CP (2004) EST-PAGE – managing and analyzing EST data. Bioinformatics 20:286–288
- Meyers BC, Axtell MJ, Bartel B, Bartel DP, Baulcombe D, Bowman JL, Cao XF, Carrington JC, Chen X, Green PJ, Griffiths-Jones S, Jacobsen SE, Mallory AC, Marteinssen RA, Poethig RS, Qi Y, Vaucheret H, Voinnet O, Watanabe Y, Weigel D, Zhu J-K (2008) Criteria for Annotation of Plant MicroRNAs. Plant Cell 20:3186–3190
- Nikiforova VJ, Bielecka M, Gakière B, Krueger B, Rinder J, Kempa M, Scheible WR, Hesse H, Hoefgen R (2006) Effect of sulfur availability on the integrity of amino acid biosynthesis in plants. Amino Acids 30:173–183
- Rymarquis LA, Kastenmayer JP, Huttenhofer AG, Green PJ (2008) Diamonds in the rough: mRNA-like noncoding RNAs. Trends Plant Sci 13:329–334
- Schachtman DP, Reid RJ, Ayling SM (1998) Phosphorus uptake by plants: from soil to cell. Plant Physiol 116:447–453
- Shane MW, McCully ME, Lambers H (2004) Tissue and cellular phosphorus storage during development of phosphorus toxicity in *Hakea prostrata* (Proteaceae). J Exp Bot 55:1033–1044
- Sunkar R, Jagadeeswaran G (2008) In silico identification of conserved microRNAs in large number of diverse plant species. BMC Plant Biol 8:37–49
- Sunkar R, Zhu JK (2004) Novel and stress-regulated MicroRNAs and other small RNAs from *Arabidopsis*. Plant Cell 16:2001–2019
- Sunkar R, Zhu JK (2007) Micro RNAs and Shortinterfering RNAs in Plants. J Int Plant Biol 49:817–826
- Sunkar R, Girke T, Jain PK, Zhu JK (2005) Cloning and characterization of MicroRNAs from rice. Plant Cell 17:1397–1411
- Sunkar R, Kapoor A, Zhu JK (2006) Posttranscriptional induction of two Cu/Zn superoxide dismutase genes in *Arabidopsis* is mediated by downregulation of miR398 and important for oxidative stress tolerance. Plant Cell 18:2051–2065
- Sunkar R, Chinnusamy V, Zhu J, Zhu JK (2007) Small RNAs as big players in plant abiotic stress responses and nutrient deprivation. Trends Plant Sci 12:301–309
- Szittya G, Moxon S, Santos DM, Jing R, Fevereiro MPS, Moulton V, Dalmay T (2008) High-throughput sequencing of *Medicago truncatula* short RNAs identifies eight new miRNA families. BMC Genomics 9:593–601
- Szymanski M, Barciszewska MZ, Zywicki M, Barciszewski J (2003) Noncoding RNA transcripts. J Appl Genet 44:1–19
- Takahashi H, Watanabe-Takahashi A, Smith FW, Blake-Kalff M, Hawkesford MJ, Saito K (2000) The roles of

three functional sulphate transporters involved in uptake and translocation of sulphate in *Arabidopsis thaliana*. Plant J 23:171–182

- Talalay P, Fahey JW (2001) Phytochemicals from cruciferous plants protect against cancer by modulating carcinogen metabolism. J Nutr 131:3027S–3033S
- Tuteja N, Sopory SK (2008) Chemical signaling under abiotic stress environment in plants. Plant Signal Behav 3:525–536
- Vaucheret H (2006) Post-transcriptional small RNA pathways in plants: mechanism and regulations. Genes Dev 20:759–771
- Vinocur B, Altman A (2005) Recent advances in engineering plant tolerance to abiotic stress: achievements and limitations. Curr Opin Biotechnol 16:123–132
- Wu L, Joshi CP, Chiang VL (2000) A xylem-specific cellulose synthase gene from aspen (*Populus tremuloides*) is responsive to mechanical stress. Plant J 22:495–502
- Xie FL, Huang SQ, Guo K, Xiang AL, Zhu YY, Nie L, Yang ZM (2007) Computational identification of novel microRNAs and targets in *Brassica napus*. FEBS Lett 581:1464–1474
- Yamasaki H, Abdel-Ghany SE, Cohu CM, Kobayashi Y, Shikanai T, Pilon M (2007) Regulation of copper homeostasis by micro-RNA in *Arabidopsis*. J Biol Chem 282:16369–16378
- Zeng Y, Yi R, Cullen BR (2003) MicroRNAs and small interfering RNAs can inhibit mRNA expression by

similar mechanisms. Proc Natl Acad Sci USA 100:9779–9784

- Zhang BH, Pan XP, Wang QL, Cobb GP, Anderson TA (2005) Identification and characterization of new plant microRNAs using EST analysis. Cell Res 15:336–360
- Zhang BH, Pan XP, Wang QL, Cobb GP, Anderson TA (2006) Computational identification of microRNAs and their targets. Comput Biol Chem 30:395–407
- Zhang BH, Wang QL, Wang K, Pan XP, Liu F, Guo T, Cobb GP, Anderson TA (2007) Identification of cotton microRNAs and their targets. Gene 397:26–37
- Zhang Z, Wei L, Zou X, Tao Y, Liu Z, Zheng Y (2008) Submergence-responsive microRNAs are potentially involved in the regulation of morphological and metabolic adaptations in Maize Root Cells. Ann Bot 102:509–519
- Zhao B, Ge L, Liang R, Li W, Ruan K, Lin H, Jin Y (2009) Members of miR-169 family are induced by high salinity and transiently inhibit the NF-YA transcription factor. BMC Mol Biol 10:29–38
- Zhou L, Liu Y, Liu Z, Kong D, Duan M, Luo L (2010) Genome-wide identification and analysis of droughtresponsive microRNAs in *Oryza sativa*. J Exp Bot 61:4157–4168
- Zuker M (2003) Mfold web server for nucleic acid folding and hybridization prediction. Nucleic Acids Res 31:3406–3415

Transcription Factors Involved in Environmental Stress Responses in Plants

13

Haibo Xin, Feng Qin, and Lam-Son Phan Tran

Abstract

Environmental stresses such as drought, high salinity, high temperature, and cold stresses are the major factors which cause significant losses to the yields of economically important crops. Sequence-specific binding transcription factors (TFs) are known to be involved in regulation of stress responses. In this chapter, we review our current knowledge about the functions of various TFs in stress responses, including the major DREB/CBF, AREB/ABF, and HSF TFs, and their signaling cascades. Understanding the mechanisms of how the TFs act as "master regulators" in the signal transduction networks involved in the conversion of stress signal perception to stress-responsive gene expression will enable us to develop stress-tolerant crops by genetic engineering of the major TFs.

Keywords

Drought • Salinity • Cold, DREB • Transcription factors • Heat shock proteins, plant responses

1 Introduction

Environmental stresses such as drought, salinity, and extreme temperatures adversely affect the growth and production of plants, and have been

L.-S.P. Tran (🖂)

considered as major stressors because of severe desertification worldwide (Vinocur and Altman 2005). It has been reported that abiotic stresses may cause more than 40% yield loss for most of the major crop plants (Bray et al. 2000; Manavalan et al. 2009; Tran and Mochida 2010; Hadiarto and Tran 2011). Over the last several decades, botanists have been seeking various approaches including traditional breeding, promotion of cultivation environment, and adoption of rational cropping system to reduce the crop loss caused by these stresses.

However, conventional methods are known to face with difficulties to improve such complex

Signaling Pathway Research Unit, Plant Science Center, RIKEN Yokohama Institute, 1-7-22 Suehiro-cho, Tsurumi, Yokohama 230-0045, Japan e-mail: tran@psc.riken.jp

H. Xin • F. Qin

Key Laboratory of Photosynthesis and Environmental Molecular Physiology, Institute of Botany, Chinese Academy of Sciences, Beijing 100049, China

stress-related traits which are considered to be controlled by quantitative loci. For example, breeding strategies based on genetic variants, natural mutations, and intergeneric crosses often meet with the difficulties in eliminating undesired linkage drag (Rommens et al. 2007). In contrast with traditional breeding, genetic engineering takes an advantage in connecting the tolerance trait directly to determinant gene loci by manipulating expression level of key stress-associated genes (Ashraf 2010). Recent advances in functional genomics have enabled biotechnologists to identify and characterize many stress-responsive functional and regulatory genes which are key components in abiotic stress signaling pathways in various plant species (Chinnusamy et al. 2007; Nakashima et al. 2009). Among the regulatory proteins, transcription factors (TFs) have been known to play crucial roles in signal transduction by receiving the upstream signal and activating the expression of downstream stress-inducible genes.

In this chapter, we focus on the regulation of stress-responsive TFs under various abiotic stresses and summarize the recent advances in enhancing stress tolerance of plants by genetic engineering using TFs.

2 DREB1/CBF TFs and Cold Stress Response

Cold stress adversely affects all aspects of cellular function. The DREB1/CBF regulon is one of the most important regulatory systems at transcriptional level in cold signal transduction in plants. The DREB1/CBF TFs, which specifically bind to the dehydration-responsive cis-element (DRE), belong to the AP2_EREBP family that is unique to plants (Nakashima et al. 2009). The DREB1/ CBF TFs have been the most extensively studied TFs in the last 20 years aimed at improving plant tolerance to various abiotic stresses, including cold stress (Thomashow 2001; Shinozaki et al. 2003; Yamaguchi-Shinozaki and Shinozaki 2006; Nakashima et al. 2009). It has been shown that DREB1 genes are promising candidates for development of stress-tolerant plants by manipulation at transcriptional level (Shinozaki and YamaguchiShinozak 2000; Thomashow 2001). Among the DREB1 genes, the DREB1B/CBF1 and DREB1A/ CBF3 genes were first isolated using yeast onehybrid screening by two independent research groups (Stockinger et al. 1997; Liu et al. 1998). These two proteins specifically recognize and bind to the DRE motif containing the core sequence A/GCCGAC located in the promoters of cold and water stress-inducible genes (Yamaguchi-Shinozaki and Shinozaki 1994; Baker et al. 1994; Liu et al. 1998). In cold signal transduction pathway, the Arabidopsis DREB1A/CBF3, regulated positively by the Inducer of C-repeat binding factor Expression 1 (ICE1) and negatively by the MYB15, functions as a switch to control expressions of downstream COR genes (Chinnusamy et al. 2007). In addition to ICE1 and MYB15, members of the calmodulin binding protein transcriptional activator (CAMTA) family target directly the CM2 motif found in the promoters of the CBFs, and CAMTA3 is a positive regulator of DREB1C/CBF2 expression, which provides evidence of a link between calcium signaling and CBF-dependent cold accumulation (Doherty et al. 2009).

The Arabidopsis genome contains six DREB1/ CBF genes (Sakuma et al. 2002), including the well-studied DREB1B/CBF1, DREB1A/CBF3, and DREB1C/CBF2, which lie in tandem on chromosome 4 and are rapidly induced by cold and slightly by high-salinity stresses but not by dehydration (Liu et al. 1998; Gilmour et al. 2000; Nakashima et al. 2000). It should be noticed that although the transcripts of DREB1s/CBFs were specifically induced by cold stress, the majority of their downstream genes respond not only to cold stress but also to dehydration and high salinity. The enhanced tolerance to drought, high salinity, and freezing stresses was observed in transgenic Arabidopsis plants overexpressing DREB1B/ CBF1 or DREB1A/CBF3 under control of the cauliflower mosaic virus (CaMV) 35S promoter (Jaglo-Ottosen et al. 1998; Liu et al. 1998; Kasuga et al. 1999; Gilmour et al. 2000), indicating that DREB1s/CBFs target multiple genes which respond to various stresses. Genomewide analyses of the DREB1A transgenic Arabidopsis plants using both cDNA and GeneChip microarrays

identified more than 40 target genes of DREB1A/ CBF3, such as RD29A, Cor15B, Kin1, Kin2, RD17, LEA14, AtGolS3, and STZ/ZAT10 (Seki et al. 2002; Fowler and Thomashow 2002; Maruyama et al. 2004). These target genes contain DRE or DRE-related core sequences in their promoter regions, suggesting that the DREB1A/ CBF3 TF could bind to these motifs and directly activate expressions of these genes in plants (Maruyama et al. 2004). The majority of the target genes of DREB1A/CBF3 encode TFs, phospholipase C, RNA-binding proteins, sugar transport proteins, desaturase, LEA proteins, osmoprotectant biosynthetic proteins, which are known to play important roles in plant acclimation to stresses. However, constitutive overexpression of DREB1/CBF genes adversely affects plant growth, leading to severe growth retardation even under optimal conditions (Liu et al. 1998; Kasuga et al. 1999). To overcome the negative impact of DREB1 TFs on growth and development, Kasuga et al. (1999) used the stressinducible RD29A promoter instead of the constitutive 35S promoter. With this strategy, the DREB1/CBF genes have been successfully used to promote the tolerance in a number of transgenic plants to various abiotic stresses (Kasuga et al. 1999, 2004). The potential application of DREB1/CBF genes in genetic engineering of stress-tolerant plants has led to isolation and functional studies of DREB1/CBF orthologs in many plant species such as rice, maize, *Brassica* napus, tomato, wheat, rye, oat, sorghum, ryegrass, and dwarf apple (Jaglo et al. 2001; Gao et al. 2002; Dubouzet et al. 2003; Qin et al. 2004; Zhao and Bughrara 2008; Yang et al. 2010). The availability of DREB1/CBF genes in various plant species implies that the DREB1/CBF regulon is ubiquitous in higher plants.

DREB1/CBF genes have been widely studied in rice, which is not only an important crop worldwide but also a model for monocotic plants. Interestingly, the OsDREB1A–D genes, which were cloned as DREB1 orthologous genes from rice, showed different stress-inducible expression patterns in comparison with their Arabidopsis counterparts (Dubouzet et al. 2003). Both OsDREB1A and OsDREB1B expressions were induced by cold stress. In addition, OsDREB1A was also induced by salt stress. Additionally, OsDREB1C displayed a constitutive expression pattern, whereas OsDREB1D transcripts were undetected in plants (Dubouzet et al. 2003). Overexpression of OsDREB1A in Arabidopsis conferred freezing and high salinity tolerance, implying that this gene may have similar function as that of AtDREB1A, and play an important role in stress tolerance in rice. Furthermore, Ito et al. (2006) generated transgenic rice plants independently overexpressing the Arabidopsis DREB1A, 1B, and 1C using the 35S promoter, and the OsDREB1A and 1B using the maize Ubi promoter. Under the normal conditions, all the transgenic plants exhibited growth retardation and enhanced stress tolerance to drought, high salinity, and cold stresses in comparison with control plants. Microarray and RNA-blot analyses confirmed that 12 genes were up-regulated by either OsDREB1A or AtDREB1A in rice. Among the 12 up-regulated genes, 8 genes contained DRE sequences in their promoters, implying that they could be direct target genes of OsDREB1A and AtDREB1A proteins. Additionally, the majority of these downstream genes were induced not only by dehydration, but also by cold and high salinity.

It is worthy to mention that, in another independent study Oh et al. (2005) did not observe any negative effects on the growth of transgenic rice plants in which the *AtDREB1A/CBF3* was constitutively overexpressed using the maize Ubi promoter. These transgenic rice plants exhibited enhanced stress tolerance as well. This phenomenon was explained that stress-associated genes were not up-regulated under unstressed conditions in the transgenic plants (Oh et al. 2005).

In maize, Kizis and Pagès (2002) isolated two members of the AP2-EREBP TF family, and named *DBF1* and *DBF2*. Phylogenetic analysis indicated that these two genes were classified into the A-6 and A-4 subgroups of the DREB family proteins, respectively. Although both proteins are capable to bind to the DRE motif, only DBF1 functions as transactivator to activate DREBdependent gene expression. Overexpression of *DBF1* gene in *Arabidopsis* led to enhanced tolerance of transgenic plants to dehydration and salt stresses (Saleh et al. 2006). A maize DREB1 orthologous TF encoding gene, named *ZmDREB1A*, was identified by Qin et al. (2004). Its transcripts were increased greatly at 4°C and induced slightly by high salinity stress. An enhanced tolerance to dehydration and cold stresses was observed in transgenic Arabidopsis plants overexpressing ZmDREB1A which was accompanied with a dwarfed phenotype correlated to expression levels of the transgene. In two independent ZmDREB1A transgenic lines, downstream genes of DREB1 TFs, such as COR, KIN1, KIN2, RD29A, and RD17, were found to be upregulated. These data indicated that DREB1 regulon might play a conserved mechanism to protect different plant species under adverse conditions.

Skinner et al. (2005) reported that barley (Hordeum vulgare), an important crop and a diploid Triticeae plant, contains at least 20 DREB/ CBF genes comprising three multigene phylogenetic subgroups designated HvCBF1, HvCBF3, and HvCBF4 subgroups. In transgenic Arabidopsis, the temperature-independent CBFs, the members of the HvCBF1- and HvCBF3-subgroups, activated COR genes at warm temperature, whereas the cold-dependent CBFs, HvCBF4-subgroup TFs, did not. HsDREB1A from wild barley (Hordeum spontaneum) shows a high degree of sequence conversation to that of barley (H. vulgare) (James et al. 2008). Eight CBF genes from another cereal, the Einkorn wheat (Triticum monococcum), have dramatically different levels of induction after exposure to cold stress (Vágújfalvi et al. 2005). Differences in CBF expression were generally associated with a variation in frost tolerance. Pellegrineschi et al. (2004) introduced RD29Apro:AtDREB1A construct into wheat, and screened 12 T2 transgenic lines exhibiting waterdeficit stress tolerance. The AtDREB1A wheat plants displayed substantial tolerance to water stress in comparison with control plants under experimental greenhouse conditions. In the Triticeae, rye (Secale cereale) is one of the most low-temperature-tolerant species. Therefore, it has been considered as an excellent model to study and compare expression profiles of DREB/ CBF-type TF encoding genes under cold stress.

Eleven *ScCBF* genes spanning all four main subgroups were cloned. Several *ScCBF* genes were found to be expressed at warmer acclimation temperatures and be repressed at the end of an 8-h dark period at warmer temperatures, implying that the *ScCBF* genes might have not only temperature-dependent but also light-regulated diurnal response in rye (Campoli et al. 2009).

In horticultural plants, progress has been made in studies on isolation, functional and transformation analyses of DREB1 genes. The growth and yield of tomato (Lycopersicon esculentum), which is originated from subtropical area and sensitive to cold, are adversely affected by chilling. As a means to provide a solution to this negative impact, transgenic tomato plants overexpressing AtDREB1B/CBF gene were constructed (Hsieh et al. 2002). As expected, transgenic tomato exhibited enhanced tolerance to chilling by showing higher survival rate under stressed conditions in comparison with control plants. However, the growth rate, the fruit and seed numbers of transgenic plants tolerant to freezing were not observed to be improved. Three DREB1/CBF homologous genes, LeCBF1/2/3, were cloned and characterized in tomato (Zhang et al. 2004). Among these three genes, only expression of LeCBF1 is induced by chilling stress. Overexpression of LeCBF1 in Arabidopsis improved tolerance of transgenic plants. However, the same authors demonstrated that neither overexpression of AtDREB1A/CBF3 nor that of *LeCBF1* in tomato improved freezing stress tolerance. Recently, a dwarf apple MbDREB1 was isolated and functionally characterized (Yang et al. 2010). Expression of MbDREB1 was induced by cold, drought, salt stress, and exogenous ABA treatment. In comparison with wild-type plants, transgenic Arabidopsis over expressing MbDREB1 showed increased tolerance to low temperature, drought, and salt stresses. These data imply that *MbDREB1* might increase plant tolerance to low temperature, drought, and salt stresses in both ABA-dependent and ABA-independent manners in apple. It has also been reported that RD29pro:DREB1A construct was introduced into Chrysanthemum (Hong et al. 2006a, b). As expected, transgenic plants showed stronger resistance to drought, salt, and low temperature.

DREB1/CBF-type genes, BnCBF5, 7, 16, and 17 from B. napus were cloned by Gao et al. (2002). Deduced protein sequences of BnCBF5, 7, and 16 are very similar to the Arabidopsis CBF1 whereas that of BnCBF17 is different because of the presence of two extra regions in its acidic domain. All the BnCBF genes are induced by low temperature. Additionally, expression of BnCBF5, 7, and 16 is also slightly responsive to salt stress, but that of BnCBF17 is not. Furthermore, overexpression of either BnCBF5 or BnCBF17 improved tolerance of transgenic Brassica to freezing (Savitch et al. 2005).

Overall, the DREB1/CBF TFs regulate cold stress signaling at transcriptional level without posttranscriptional regulation. Overexpression of *DREB1/CBFs* can improve tolerance to cold stress in various transgenic plants. Noticeably, the adverse effect caused by overproduced DREB1/CBFs should be minimized by using inducible or controllable promoter such as *RD29A* promoter instead of the constitutive 35 S. Investigation of several upstream regulators of the DREB1/CBFs, such as ICE1 and MYB15, may provide a deeper understanding of DREB1/ CBF function in cold acclimation.

3 DREB2 and Drought/High Salinity Stress Response

Although DREB2A was isolated from the yeastone-hybrid screening together with DREB1A (Liu et al. 1998), its function has been less reported than that of DREB1A gene. The reason might be that because overexpression of DREB2A led to neither enhanced tolerant phenotype nor alteration in expression of stress-inducible genes. Domain analysis of DREB2A in Arabidopsis protoplasts revealed that DREB2A contains a negative regulatory region and the deletion of this region activated DREB2A protein. The deleted protein was called DREB2A-CA (Sakuma et al. 2006a). Overproduction of DREB2A-CA protein in Arabidopsis enhanced water stress tolerance and caused a dwarf phenotype. However, no obvious tolerance to freezing was observed. As is the case for DREB1s, overexpression of DREB2A-

CA using stress-inducible *RD29A* promoter eliminated the dwarfism and maintained a better tolerance (Sakuma et al. 2006a).

Further analyses of expression and sequences demonstrated differences in stress-responsive expression profiles and the AP2-EREBP domains between DREB1-type and DREB2-type TFs which might lead to their different response to different stresses and the discrepancy in their downstream target genes. In Arabidopsis, DREB1 genes are induced greatly by cold or slightly by salt stress; however, DREB2 genes are greatly increased by salt and dehydration stresses (Liu et al. 1998; Sakuma et al. 2002). Recently, it was reported that DREB2 genes are induced by heat stress (Sakuma et al. 2006b; Qin et al. 2007; Lim et al. 2007). It was also found that downstream target genes of DREB2A were different from those of DREB1A. This might be explained by the fact that the DNAbinding domains of DREB1A and DREB2A are slightly different, which causes a high affinity of DREB1A to A/GCCGACNT sequences and the preferential binding of DREB2A to ACCGAC motifs (Sakuma et al. 2006a).

Interestingly, in addition to drought and salt inducible genes, overexpression of DREB2A-CA in Arabidopsis also up-regulates heat shock related genes (Sakuma et al. 2006b). Consequently, the DREB2A-CA transgenic plants displayed enhanced thermotolerance. Thus, DREB2A was recognized to play an important regulatory role in response not only to drought but also to high temperature stress. Microarray analysis of DREB2A-CA-overexpressing plants identified heat shock protein 70 (HSP70), HSP18.2, and heat shock A3 factor encoding gene (HSFA3) among the up-regulated downstream genes (Sakuma et al. 2006b). The DREB2A protein was shown to bind to the DRE sequence located in the promoter of the AtHSFA3 and activate its transcription (Schramm et al. 2008; Yoshida et al. 2008). Additionally, Chen et al. (2009) presented data in regard to the regulation of HSFA3 by DREB2C in the heat shock signal transduction cascade. Similar to DREB2A, DREB2C interacts with two DREs located in the HSFA3 promoter. Deletion analysis of DREB2C indicated that its transactivation region is located in the C terminus.

A DREB2A orthologous TF encoding gene, the ZmDREB2A, was isolated from maize. Expression analysis revealed that ZmDREB2A is strongly induced by high salinity, slightly by drought, and unchanged by heat stress (Qin et al. 2007). Interestingly, in contrast with Arabidopsis DREB2A, ZmDREB2A gene contains 53 bp intron, causing a premature termination of translation. This fragment was shown to be involved in an alternative splicing under stress conditions. This phenomenon was also found in other monocotic plants, such as wheat and barley (Xue and Loveridge 2004; Terashima and Takumi 2009). Constitutive overexpression of ZmDREB2A in Arabidopsis enhanced water-deficit stress tolerance but caused dwarfism. Using the RD29A promoter instead of the 35 S for overexpression of the ZmDREB2A diminished plant growth retardation (Qin et al. 2007). In contrast with overexpression of ZmDREB2A, that of a Populus euphratica DREB2 gene, designated PeDREB, under the control of 35 S promoter improved salt stress tolerance but did not cause dwarf phenotype (Chen et al. 2009).

Matsukura et al. (2010) performed a comprehensive analysis of all five DREB2-type genes OsDREB2B, (OsDREB2A, OsDREB2C, OsDREB2E, and OsABI4) in rice. Among them, only OsDREB2A and OsDREB2B exhibited abiotic stress-inducible expression. OsDREB2B showed nuclear specific localization and the highest transactivation activity. In transgenic Arabidopsis, overexpression of OsDREB2B upregulated expression of AtDREB2A target genes, including both the drought and heat shock-inducible genes. Noticeably, OsDREB2B has functional and nonfunctional transcripts similar to its orthologues in the grass family. In the study on DREB2-type TFs in diploid progenitors and hexaploid wheat lines, it was found that allopolyploidization during wheat polyploid evolution could inhibit efficiently alternative splicing of WDREB2 transcripts (Terashima and Takumi 2009). These data suggest that DREB2 genes in grass family play an important role in response to drought and heat shock stress which is controlled by alternative splicing.

In contrast with alternative splicing found for *DREB2* genes in grass, AtDREB2 protein was shown to undergo posttranscriptional regulation

by degradation caused by DRIPs (DREB2Ainteracting proteins), which are C3HC4 ring domain-containing proteins. The DRIPs function as E3 ubiquitin ligases and are able to mediate DREB2A ubiquitination and proteolysis under favorable growing condition, which probably restricts the negative effects of DREB2A protein on plant growth and development (Qin et al. 2008).

Collectively, research on DREB2 TFs indicates that the DREB2 regulon forms an ABA-independent pathway which functions in both osmotic and heat shock responses (HSR). Alternative splicing and posttranscription regulation are the two mechanisms which control the stress-related DREB2 functions in grass species and *Arabidopsis*, respectively.

4 AREB TFs and ABA-Dependent Drought and High Salinity Stress Response

Under water stress conditions, such as cold, high salinity, and drought, endogenous ABA levels in plants increase, which triggers stomatal closure to avoid environmental stress damage (Leung and Giraudat 1998). In Arabidopsis and rice, many drought and high salinity-inducible genes also respond to ABA (Seki et al. 2002; Rabbani et al. 2003). The key transcriptional regulators of ABA-dependent gene expression are ABAresponsive element binding TFs (AREBs/ABFs) which belong to the basic-leucine zipper (bZIP) TF family (Finkelstein et al. 2005; Choi et al. 2005). AREBs/ABFs control ABA-dependent gene expression by binding to *cis*-acting regulatory elements which share the C/TACGTGGC consensus, designated as ABREs, in the promoters of ABA-inducible genes (Busk and Pages 1998). ABRE element is usually coupled with other sequences to realize its function. ABRE and GC-rich coupling elements, including coupling element 1 (CE1) and coupling element 3 (CE3), function together as a complex in the regulation of gene expression in wheat (Shen et al. 1996). The core sequence of known coupling elements, which is A/GCGT, is similar to that of ABREs (Hobo et al. 1999a). Narusaka et al. (2003) reported that the DRE/CRT may also function as a coupling element of ABRE in response to ABA in *Arabidopsis*.

In Arabidopsis, cDNAs encoding AREBs/ ABFs were successfully isolated using yeast one-hybrid system (Choi et al. 2000; Uno et al. 2000). Further genomic sequence analyses indicated that the Arabidopsis genome contains at least 75 distinct bZIP TFs, among which 13 members are identified as AREB/ABFs (Bensmihen et al. 2002; Jakoby et al. 2002). Among these AREB/ABFs encoding genes, AREB1/ABF2, AREB2/ABF4, and ABF3 are induced by ABA, drought, and high salinity in vegetative tissues but not in seeds (Fujita et al. 2005). However, ABI5, AREB3, DPBF2, and EEL (Enhanced Em Level) genes were expressed during seed maturation (Finkelstein and Lynch 2000; Lopez-Molina and Chua 2000; Bensmihen et al. 2002).

Different from DREB1/CBF TFs, AREB/ABF proteins contain many target sequences of protein kinases, and modification of AREB/ABF by phosphorylation is required to transform the native but inactive AREB/ABF TFs into their active form. A number of reports have revealed that phosphorylation of the AREB1 protein by SnRK2, a sucrose nonfermenting protein (SNF1)-related kinase, is important for its activation. Recent research showed that SnRK2.2/SnRK2D and SnRK2.3/SnRK2I are involved in the phosphorylation of AREB1 (Fujii et al. 2007). A strong lines of evidence indicated that OST1/SnRK2.6/ SnRK2E (an ABA-activated protein kinase, homologue of SnRK2.2/SnRK2D and SnRK2.3/ SnRK2I) was also able to directly target ABF/ AREBs to phosphorylation, and three of SnRK2 kinases cooperatively phosphorylate AREB1 in response to water stress or ABA signals (Fujii et al. 2007, 2009; Yoshida et al. 2010). Therefore, it was the reason why transformation of an unphosphorylated or inactive AREB1 gene into Arabidopsis has failed to cause remarkable improved expression of downstream genes and ABA-related phenotype (Fujita et al. 2005). In addition to the SNF-type kinases, Ca2+-dependent protein kinases, such as the CPK4 and CPK11, also phosphorylate ABF1 and ABF4/AREB2 (Zhu et al. 2007).

An independent study indicated that deletion of a fragment from 61 to 318 aa region between the transactivation and the DNA binding domain transforms the inactive AREB1 into a constitutive active form, which can promote its target gene expression, even in the absence of ABA (Fujita et al. 2005). Transgenic Arabidopsis plants overexpressing the active form of AREB1 under the control of 35 S promoter were shown to be hypersensitive to ABA and displayed enhanced drought tolerance. Microarray and RNA-gel blot analyses of transgenic plants discovered eight genes as downstream target genes of the active AREB1. These up-regulated genes were classified into two groups: the first group consists of genes encoding functional proteins, such as LEA or LEA-like proteins, including RD29B, At3g17520, and the other contains regulatory proteins, including RD20 and HIS1-3 (Fujita et al. 2005). Concurrently, the same lab demonstrated that overproduction of a phosphorylated AREB1 protein in Arabidopsis improved RD29B and other ABA-inducible gene expression in plants (Furihata et al. 2006). Taken together, these results support that AREB1 protein requires posttranscriptional modification to modulate its downstream gene expression.

In addition to AREB1, other AREB/ABF TFs also contribute to ABA-dependent stress response. ABI3, a member of B3 transcriptional regulators, interacts with ABI5 to enhance its action. An AP2-type TF, ABI4, and some MYC/MYB-type TFs (AtMYC2 and AtMYB2) function as positive regulators of ABA response (Yamaguchi-Shinozaki and Shinozaki 2006).

In rice and barely, the homologues of *Arabidopsis* AREB1/ABF2, the TRAB1, and HvAB15 encoding genes, respectively, were found to respond to ABA treatment and drought stress in seedlings (Hobo et al. 1999b; Casaretto and Ho 2003). Zou et al. (2008) reported the cloning of *OsAB15* and that its expression was induced by ABA and high salinity, but was down-regulated by drought and cold stresses in seedlings. Overexpression of *OsAB15* caused high sensitivity of transgenic rice to salt stress. However, down-regulation of *OsAB15* enhanced stress tolerance, but negatively influenced rice

fertility. Microarray analysis indicated that, among 89 *OsbZIP* genes surveyed, 26 genes were up-regulated and 11 genes were down-regulated under dehydration, high salinity, and cold conditions (Nijhawan et al. 2008).

In conclusion, AREB/ABF regulon is a common ABA-dependent signaling pathway in plants, functioning in response to dehydration and high salinity stresses. Phosphorylation of AREB/ABF TFs, such as AREB1, may be required for their full activation.

5 HSFs and HSP-Related Heat Stress Response

Heat stress is one of the key environmental stresses in fields around the world, especially in tropical areas. As sessile organisms, plants can acquire better thermotolerance against unavoidable high temperature challenges by a prior high temperature treatment (Sun et al. 2002). In this process, heat acclimation involves the accumulation of HSPs that are molecular chaperons (Sun et al. 2002), which are important for cellular homeostasis in cells under both optimal and adverse growth conditions (Kim and Schöffl 2002; Hartl and Hyer-Hartl 2002; Port et al. 2004; Wang et al. 2004). Expression of HSP encoding genes is regulated by a large family of HSFs which bind to the heat shock elements (HSE) "GAANNTTC" in the promoters of HSP encoding genes (Schöffl et al. 1998; Nover et al. 2001; von Koskull-Döring et al. 2007).

While there are only a few HSF encoding genes in yeast, *Drosophila* and vertebrates (yeast and *Drosopholia* have only 1; vertebrates possess 4), plants have multiple HSF encoding genes (for instance tomato, *Arabidopsis*, rice and soybean have 18, 21, 25, and 34 genes, respectively) divided into three classes of A, B and C (Nover et al. 1996, 2001; Nakai 1999; Baniwal et al. 2004; Kotak et al. 2004; Xing et al. 2005). Identification and detailed functional analyses of HSFs of *Arabidopsis* and tomato have revealed that the majority of class-A HSFs contain the AHA motifs vital for transcriptional activation of their downstream genes, while the class-B and C HSFs have no transactivational activity, and they function as repressors or co-activators (Kotak et al. 2004).

In the plant kingdom, cloning and genetic analysis of HSF encoding genes have been pioneered from tomato (Nover et al. 1996, 2001; Lyck et al. 1997). The excellent work of Mishra et al. (2002) revealed that HSFA1a functions as a master regulator of HSR in tomato. In their study, plants with tenfold overexpression of HSFA1 (OE) and cosuppression (CS) plants were produced. Under normal conditions, major developmental parameters of OE and CS plants are similar to that of wild type. However, CS plants and fruits were extremely sensitive to elevated temperatures because of reducing and lacking of heat-inducible HSFs and synthesis of HSP chaperons. These results indicated that HSFA1 has a unique function as master regulator for induced thermotolerance in tomato. In contrast with HSFA1a, tomato HSFA2 or HSFB subgroup members could not function independently. HSFA2 needs interaction with HSFA1 for efficient nuclear import (Scharf et al. 1998), and HSFB1 has been identified as a co-activator of class-A HSFs and other TFs for transactivation (Bharti et al. 2004).

Multiple plant HSFs display diversity in their biological functions. Completion of Arabidopsis genomic sequence has enabled us to find and analyze HSFs and their target genes such as HSPs encoding genes at whole genome scale. In Arabidopsis, 44 HSPs and 21 HSFs were identified (Swindell et al. 2007). Additionally, Guo et al. (2008) reported that the Arabidopsis genome contains 22 HSFs, and in contrast with TAIR annotation of HSFs, an additional HSF has been found to be probable member of HSFA2 class. Unlike tomato HSF system, no master HSF regulator has been found, suggesting that there may be more complicated HSF regulation in Arabidopsis. A number of studies on the functions of Arabidopsis HSFs have been conducted. HSFA1a and HSFA1b play important roles in the induction of some HSP encoding genes in the early phase of HSR and in acquired thermotolerance (Lohmann et al. 2004). Arabidopsis HSFA2, HSFA3, and HSFA7a also play critical role in this process (Charng et al. 2007; Larkindale and Vierling 2008; Schramm et al. 2008; Yoshida et al. 2008). There are five different *HSFB* genes in *Arabidopsis* genome. The HSFB3 and HSFB4 function at the early stage after heat shock, and HSFB1, HSFB2a, and HSFB2b are considered to function as late HSFs (Lohmann et al. 2004; Busch et al. 2005).

Although functionally similar to HSFA1a, several plant HSF encoding genes are HS-inducible genes themselves, such as HSFA2 proteins from tomato and Arabidopsis, which encode the major HSFs in thermotolerant cells (Scharf et al. 1998; Morimoto 2002; Busch et al. 2005). According to the data gained from transcriptome analyses, six HSF encoding genes, HSFA2, HSFA4a, HSFA7a, HSFB1, HSFB2a, and HSFB2b, are strongly up-regulated in Arabidopsis leaves by heat shock treatment at 37°C for 1 h (Busch et al. 2005). Among these up-regulated genes, function of AtHSFA2 was analyzed in detail. Loss-offunction studies of AtHSFA2 supported the direct evidence that AtHSFA2 is essential for acquired thermotolerance (AT) after long recovery but not short recovery (Charng et al. 2007). On the other hand, ectopic overexpression of OsHSFA2e led to increased thermotolerance not only in the cotyledons but also in rosette leaves, inflorescence stems, and seeds in transgenic Arabidopsis (Yokotani et al. 2008).

In addition to heat stress, numerous abiotic and biotic stressors can induce expression of HSF encoding genes (Swindell et al. 2007), suggesting that HSFs may play crucial role in crosstalk between multiple stress response pathways. Recently, it was reported that overexpression of AtHSFA2 conferred not only increased tolerance of transgenic Arabidopsis to heat and osmotic stresses but also enhanced callus growth (Nishizawa et al. 2006; Yokotani et al. 2008). High-level overexpression of the Arabidopsis HSFA2 gene revealed that HSFA2 plays, in addition to its role in heat and salt stress tolerance, an important role in cell proliferation (Ogawa et al. 2007). Interestingly, AtHSFA2 was shown to be one of the regulatory components of cytoplasmic protein response (CPR) (Sugio et al. 2009) and target of sumoylation which may regulate AtHSFA2 transcriptional activity (Cohen-Peer et al. 2010).

There is an ongoing effort to elucidate functions of the HSFs. However, detailed analyses of the HSFs are still mainly restricted to tomato and Arabidopsis (Panchuk et al. 2002; Baniwal et al. 2004; Zhang et al. 2009). During recent several years, functional analyses of HSFs from other species have been conducted. In rice, a mutant of HSFA4 exhibited a potted leaf phenotype under elevated temperature (Yamanouchi et al. 2002). It has been reported that GmHSFA1, a heat shock TF, was cloned and characterized from soybean (Glycine max), and its overexpression conferred enhanced thermotolerance of transgenic soybean plants (Zhu et al. 2006). Zhu et al. (2009) cloned a HSF gene named BhHSF1 from Boea hygrometrica. Their study revealed that this gene may play dual functions in mediating heat stress tolerance and growth retardation via regulation of target genes related to stress protection and mitotic cell cycle. A HSFA2 from lily, an important ornamental flower, has recently been isolated by Xin et al. (2010). This gene was found to be induced exclusively under heat shock. Ectopic overexpression of the lily HSFA2 obviously improved thermotolerance of transgenic Arabidopsis.

Recently, a CaM-binding protein kinase (CBK) was identified as an interacting protein of HSFA1a in *Arabidopsis* (Liu et al. 2008). The knockout lines of *AtCBK* failed to acquire thermotolerance and accumulate downstream HSP encoding genes, implying that the activity of HSFs can be modulated at posttranscriptional level (Liu et al. 2008). This case is similar to that of CAMTA regulation for cold induction.

Taken together, HSFs form a group of key regulators, controlling gene expression, thereby playing critical roles in heat signal transduction pathway. Manipulation of single *HSF* can change expression of a series of heat-inducible genes which may lead to thermotolerance.

6 Other Stress-Related TFs

In addition to the major TFs mentioned earlier, many other TFs are involved in plant response to abiotic stress.

AtMYC2 and AtMYB2 proteins are thought to function as transcriptional activators in the ABA-responsive gene expression of the RD22 (Abe et al. 1997). Overexpression of AtMYC2 increased the ABA-responsive expression of RD22. In plants overexpressing both AtMYC2 and AtMYB2, the expression of RD22 was induced by ABA significantly earlier than in 35S:AtMYB2 or 35S:AtMYC2 plants. At the same time, the AtADH1 transcript in plants overexpressing both two genes accumulated to high level under normal condition, implying that MYC2 and MYB2 act together as transcriptional activators in the ABA-responsive expression of AtADH1. Furthermore, microarray analysis of 35S:AtMYC2/MYB2 transgenic plants demonstrated that the up-regulated genes possess the MYC (CANNTG) and MYB (C/TAACNA/G) recognition sequences in their promoter regions (Abe et al. 2003).

Cys-2/His-2-type zinc-finger and zinc-finger homeodomain (ZFHD) proteins have been identified to be important as transcriptional regulators in plant abiotic stress responses. Unlike the abovementioned TFs, Cys-2/His-2-type zinc-finger TFs were observed to regulate target genes by repressing their expressions in plants (Sakamoto et al. 2004). Overproduction of Zat10/STZ, a Cys-2/His-2-type zinc-finger TF, in Arabidopsis enhanced drought stress tolerance (Sakamoto et al. 2004). Zat12 was also shown to be involved in response to both cold and oxidative stresses (Iida et al. 2000; Rizhsky et al. 2004; Vogel et al. 2005). As for ZFHD TFs, an Arabidopsis ZFHD protein, the ZFHD1, was characterized by a putative DNA-binding homeodomain in the C terminus which binds to the 62-bp region in the promoter region of ERD1 (early responsive to dehydration 1) gene. Expression of the ZFHD1 gene is induced by dehydration, high salt, and ABA treatment. Overexpression of ZFHD1 in transgenic plants conferred drought stress tolerance (Tran et al. 2007).

The NAC gene family, which is specific to plant kingdom, was originated from the names of three TFs (1) NAM (no apical meristem, *Petunia*), (2) ATAF1-2, and (3) CUC2 (cup-shaped cotyledon, *Arabidopsis*) (Nuruzzaman et al. 2010).

Arabidopsis and rice contains 117 and 151 nonredundant NAC TFs, respectively, which are characterized by the conserved N-terminal region known as the NAC domain (Ooka et al. 2003; Nuruzzaman et al. 2010). NAC proteins play critical role in plant developmental processes such as formation of apical shoots and in disease defense. Several NAC TFs were found to be regulators of abiotic stress response (Tran et al. 2010). Three NAC genes have been isolated from Arabidopsis by yeast one-hybrid screening (Tran et al. 2004). These NAC TFs were found to bind to a CATGTG motif in the promoter of ERD1. Expression of these three NAC genes was induced by drought, high salinity, ABA, and methyl jasmonate. Following this study, Hu et al. (2006) reported the isolation of a rice NAC gene, named SNAC1, whose expression was found to be induced by drought, salt, cold, and ABA treatment. The 35S:SNAC1 transgenic rice showed significantly increased drought resistance under field conditions and strong tolerance to salt stress. Another NAC TF encoding gene, OsNAC6, was also reported to be induced by cold, drought, and high salinity (Nakashima et al. 2007). When OsNAC6 was overexpressed in rice under the control of the maize Ubi promoter, the transgenic plants were more tolerant to both drought and high salt stresses accompanied by retarding of growth. Lip9 (low temperature inducible protein 9) promoter was subsequently used to minimize the adverse effect on plant growth. Recent research has revealed that Arabidopsis ATAF1, a NAC family member, plays important roles in both abiotic and biotic stress responses. ATAF1 overexpression lines exhibited enhanced drought tolerance, implying a positive regulation in plant drought response (Wu et al. 2009b). Recent studies indicated that some NAC members are membrane-associated proteins and involved in stress responses (Kim et al. 2006). It is worthy to mention that a membrane-bound bZIP TF, the bZIP28, was found to be involved in regulation of heat-stress-associated gene expression (Gao et al. 2008).

The WRKY TFs, which are regulators of downstream genes including *NAC* genes, were reported to be involved in the ABA signaling pathway (Zou et al. 2004). Transient expression

study using aleurone cells revealed that OsWRKY24 and OsWRKY45 acted as repressors of an ABA-inducible promoter; however, OsWRKY72 and OsWRKY77 were shown to be activators of the same promoter (Xie et al. 2005). Overexpression of OsWRKY11 using heat shockinducible HSP101 promoter led to enhanced heat and drought tolerance in transgenic rice (Wu et al. 2009a). In Arabidopsis, overexpression of either AtWRKY25 or AtWRKY33 increased salt tolerance (Jiang and Deyholos 2009). In B. hygrometrica, the BhGolS1 gene is inducible by both dehydration and ABA (Taji et al. 2002). The promoter of BhGolS1 encoding galactinol synthase, which plays an important role in drought and cold tolerance, contains four W boxes. Chromatin immunoprecipitation demonstrated that they are bound in vivo by the early dehydration- and ABA-inducible BhWRKY1 (Wang et al. 2009). These data provide direct evidence linking a dehydration-inducible WRKY TF with a downstream target gene that plays a vital role in drought response (Rushton et al. 2010).

Additionally, some nuclear TFs were identified to be important regulators of abiotic stress response. Nelson et al. (2007) identified a TF from the nuclear factor Y family, the AtNF-YB1, whose overexpression improved drought tolerance in *Arabidopsis*. ZmNF-YB2, an ortholog of AtNF-YB1, also has similar function in maize. The *Arabidopsis* NF-X1 has a heat-inducible expression pattern and promotes both acquired thermotolerance and salt tolerance (Larkindale and Vierling 2008; Hua 2009).

7 Conclusions and Perspectives

Owing to the power of genetics and genomics, more and more abiotic stress signaling pathways and key TFs acting as molecular regulatory switches are identified. The regulatory functions of the TFs involved in the stress signaling pathways have been elucidated. Commonly, each group of TFs responds mainly to a certain abiotic stress. However, a number of TFs, such as DREB2A, MBF1c, NAC, and NF-X1, play dual or multiple roles in different signaling pathways. Up-to-date, a plenty of valuable TFs have been identified and isolated for our original intention of studying their function in stress responses and using them to improve tolerance of plants to environmental stresses by genetic engineering. Importantly, the biological functions of some TF encoding genes have been characterized in transgenic plants. These findings have provided us a better understanding of transcriptional regulatory mechanisms in plants exposed to environmental stresses and a bright perspective in improvement of stress tolerance of plants by manipulating gene expression.

On the other hand, there is a long way to go for our ultimate goal of enhancing plant tolerance to environmental stresses by manipulating key stress-associated TFs in a field condition because of the complexity of regulatory systems in plant itself and uncontrollable testing conditions. The challenge ahead is to elucidate more detailed regulatory mechanism at transcriptional level, find internodes of multiple signals, and improve gene transfer strategies and evaluation systems. All these efforts will lead to an essential advance in development of stress-tolerant plant cultivars.

Acknowledgements The financial support from Chinese Academy of Sciences to F.Q. (grant number KSCX2-YW-N-097) is highly appreciated. Research in Tran's lab is supported by a Grants-in-Aid (Start-up) for Scientific Research grant (No. 21870046) from Ministry of Education, Culture, Sports, Science and Technology of Japan, and by a Start-up Support grant (No. M36-57000) from RIKEN Yokohama Institute Director Discretionary Funds.

References

- Abe H, Yamaguchi-Shinozaki K, Urao T, Iwasaki T, Hosokawa D, Shinozaki K (1997) Role of Arabidopsis MYC and MYB homologs in drought- and abscisic acidregulated gene expression. Plant Cell 10:1859–1868
- Abe H, Urao T, Ito T, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2003) Arabidopsis AtMYC2 (bHLH) and AtMYB2 (MYB) function as transcriptional activators in abscisic acid signaling. Plant Cell 15:63–78
- Ashraf M (2010) Inducing drought tolerance in plants: recent advances. Biotechnol Adv 28:169–183

- Baker SS, Wilhelm KS, Thomashow MF (1994) The 5'-region of Arabidopsis thaliana cor15a has cis-acting elements that confer cold-, drought- and ABA-regulated gene expression. Plant Mol Biol 24:701–713
- Baniwal SK, Bharti K, Chan KY, Fauth M, Ganguli A, Kotak S, Mishra SK, Nover L, Port M, Scharf KD (2004) Heat stress response in plants: a complex game with chaperons and more than twenty heat stress transcription factors. J Biosci 29:471–487
- Bensmihen S, Rippa S, Lambert G, Jublot D, Pautot V, Granier F, Giraudat J, Parcy F (2002) The homologous ABI5 and EEL transcription factors function antagonistically to fine-tune gene expression during late embryogenesis. Plant Cell 14:1391–1403
- Bharti K, von Koskull-Döring P, Bharti S, Kumar P, Tintschl-Körbitzer A, Treuter E, Nover L (2004) Tomato heat stress transcription factor HsfB1 represents a novel type of general transcription coactivator with a histone-like motif interacting with HAC1/CBP acetyltransferase-related proteins. Nucleic Acids Res 29:589–597
- Bray EA, Bailey-Serres J, Weretilnyk E (2000) Responses to abiotic stresses. In: Gruissem W, Buchannan B, Jones R (eds) Biochemistry and molecular biology of plants. American Society of Plant Physiologists, Rockville, MD, pp 1158–1249
- Busch W, Wunderlich M, Schöffle F (2005) Identification of novel heat shock factor-dependent genes and biochemical pathways in *Arabidopsis thaliana*. Plant J 41:1–14
- Busk PK, Pages M (1998) Regulation of abscisic acidinduced transcription. Plant Mol Biol 37:425–435
- Campoli C, Matus-Cádiz MA, Pozniak CJ, Cattivelli L, Fowler DB (2009) Comparative expression of Cbf genes in the Triticeae under different acclimation induction temperatures. Mol Genet Genomics 282:141–152
- Casaretto J, Ho TH (2003) The transcription factors HvABI5 and HvVP1 are required for the abscisic acid induction of gene expression in barley aleurone cells. Plant Cell 15:271–284
- Charng YY, Liu HC, Liu NY, Chi WT, Wang CN, Chang SH, Wang TT (2007) A heat-inducible transcription factor, HsfA2, is required for extension of acquired thermotolerance in *Arabidopsis*. Plant Physiol 143:251–262
- Chen J, Xia X, Yin W (2009) Expression profiling and functional characterization of a DREB2-type gene from Populus euphratica. Biochem Biophus Res Commun 16;378(3):483–487
- Chinnusamy V, Zhu J, Zhu JK (2007) Cold stress regulation of gene expression in plants. Trends Plant Sci 12:444–451
- Choi H, Hong J, Ha J, Kang J, Kim SY (2000) ABFs, a family of ABA-responsive element binding factors. J Biol Chem 275:1723–1730
- Choi HI, Park HJ, Park JH, Kim S, Im MY, Seo HH, Kim YW, Hwang I, Kim SY (2005) Arabidopsis calciumdependent protein kinase AtCPK32 interacts with ABF4, a transcriptional regulator of abscisic acidresponsive gene expression, and modulates its activity. Plant Physiol 139:1750–1761

- Cohen-Peer R, Schuster S, Meiri D, Breiman A, Avni A (2010) Sumoylation of Arabidopsis heat shock factor A2 (HsfA2) modifies its activity during acquired thermotolerance. Plant Mol Biol 74:33–45
- Doherty CJ, Van Buskirk HA, Myers SJ, Thomashow MF (2009) Roles for Arabidopsis CAMTA transcription factors in cold-regulated gene expression and freezing tolerance. Plant Cell 21:972–984
- Dubouzet JG, Sakuma Y, Ito Y, Kasuga M, Dubouzet EG, Miura S, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2003) OsDREB genes in rice, Oryza sativa L., encode transcription activators that function in drought-, high-salt- and cold-responsive gene expression. Plant J 33:751–763
- Finkelstein RR, Lynch TJ (2000) The Arabidopsis abscisic acid response gene ABI5 encodes a basic leucine zipper transcription factor. Plant Cell 12:599–609
- Finkelstein R, Gampala SS, Lynch TJ, Thomas TL, Rock CD (2005) Redundant and distinct functions of the ABA response loci ABA-INSENSITIVE(ABI)5 and ABRE-BINDING FACTOR (ABF)3. Plant Mol Biol 59:253–267
- Fowler S, Thomashow MF (2002) Arabidopsis transcriptome profiling indicates that multiple regulatory pathways are activated during cold acclimation in addition to the CBF cold response pathway. Plant Cell 14(8):1675–90.
- Fujii H, Verslues PE, Zhu JK (2007) Identification of two protein kinases required for abscisic acid regulation of seed germination, root growth, and gene expression in Arabidopsis. Plant Cell 19:485–494
- Fujii H, Chinnusamy V, Rodrigues A, Rubio S, Antoni R, Park SY, Cutler SR, Sheen J, Rodriguez PL, Zhu JK (2009) In vitro reconstitution of an abscisic acid signalling pathway. Nature 462:660–664
- Fujita Y, Fujita M, Satoh R, Maruyama K, Parvez MM, Seki M, Hiratsu K, Ohme-Takagi M, Shinozaki K, Yamaguchi-Shinozaki K (2005) AREB1 is a transcription activator of novel ABRE-dependent ABA signaling that enhances drought stress tolerance in Arabidopsis. Plant Cell 17:3470–3488
- Fujita Y, Nakashima K, Yoshida T, Katagiri T, Kidokoro S, Kanamori N, Umezawa T, Fujita M, Maruyama K, Ishiyama K, Kobayashi M, Nakasone S, Yamada K, Ito T, Shinozaki K, Yamaguchi-Shinozaki K (2009) Three SnRK2 protein kinases are the main positive regulators of abscisic acid signaling in response to water stress in Arabidopsis. Plant Cell Physiol 50:2123–2132
- Furihata T, Maruyama K, Fujita Y, Umezawa T, Yoshida R, Shinozaki K, Yamaguchi-Shinozaki K (2006) Abscisic acid-dependent multisite phosphorylation regulates the activity of a transcription activator AREB1. Proc Natl Acad Sci USA 103:1988–1993
- Gao MJ, Allard G, Byass L, Flanagan AM, Singh J (2002) Regulation and characterization of four CBF transcription factors from *Brassica napus*. Plant Mol Biol 49:459–471
- Gao H, Brandizzi F, Benning C, Larkin RM (2008) A membrane-tethered transcription factor defines a branch of the heat stress response in *Arabidopsis thaliana*. Proc Natl Acad Sci USA 105:16398–16403

- Gilmour SJ, Sebolt AM, Salazar MP, Everard JD, Thomashow MF (2000) Overexpression of the Arabidopsis CBF3 transcriptional activator mimics multiple biochemical changes associated with cold acclimation. Plant Physiol 124:1854–1865
- Guo J, Wu J, Ji Q, Wang C, Luo L, Yuan Y, Wang Y, Wang J (2008) Genome-wide analysis of heat shock transcription factor families in rice and Arabidopsis. J Genet Genomics 35:105–118
- Hadiarto T, Tran L-SP (2011) Progress studies of droughtresponsive genes in rice. Plant Cell Rep 30:297–310
- Hartl FU, Hyer-Hartl M (2002) Protein folding molecular chaperones in the cytosol: from nascent chain to folded protein. Science 295:1852–1858
- Hobo T, Asada M, Kowyama Y, Hattori T (1999a) ACGTcontaining abscisic acid response element (ABRE) and coupling element 3 (CE3) are functionally equivalent. Plant J 19:679–689
- Hobo T, Kowyama Y, Hattori T (1999b) A bZIP factor, TRAB1, interacts with VP1 and mediates abscisic acid-induced transcription. Proc Natl Acad Sci USA 96:15348–15353
- Hong B, Tong Z, Ma N, Kasuga M, Yamaguchi-Shinozaki K, Gao J (2006a) Expression of the Arabidopsis DREB1A gene in transgenic Chrysanthemum enhances tolerance to low temperature. J Hort Sci Biotechnol 81:1002–1008
- Hong B, Tong Z, Ma N, Li J, Kasuga M, Yamaguchi-Shinozaki K, Gao J (2006b) Heterologous expression of the *AtDREB1A* gene in chrysanthemum increases drought and salt stress tolerance. Sci China C Life Sci 49:1–10
- Hsieh TH, Lee JT, Charng YY, Chan MT (2002) Tomato plants ectopically expressing Arabidopsis CBF1 show enhanced resistance to water deficit stress. Plant Physiol 130:618–626
- Hu H, Dai M, Yao J, Xiao B, Li X, Zhang Q, Xiong L (2006) Overexpressing a NAM, ATAF, and CUC (NAC) transcription factor enhances drought resistance and salt tolerance in rice. Proc Natl Acad Sci USA 35:12987–12992
- Hua J (2009) From freezing to scorching, transcriptional responses to temperature variations in plants. Curr Opin Plant Biol 12:568–573
- Iida A, Kazuoka T, Torikai S, Kikuchi H, Oeda K (2000) A zinc finger protein RHL41 mediates the light acclimatization response in Arabidopsis. Plant J 24: 191–203
- Ito Y, Katsura K, Maruyama K, Taji T, Kobayashi M, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2006) Functional analysis of rice DREB1/CBF-type transcription factors involved in cold-responsive gene expression in transgenic rice. Plant Cell Physiol 47:141–153
- Jaglo KR, Kleff S, Amundsen KL, Zhang X, Haake V, Zhang JZ, Deits T, Thomashow MF (2001) Components of the Arabidopsis C-repeat/dehydration-responsive element binding factor cold-response pathway are conserved in *Brassica napus* and other plant species. Plant Physiol 127:910–917
- Jaglo-Ottosen KR, Gilmour SJ, Zarka DG, Schabenberger O, Thomashow MF (1998) Arabidopsis CBF1 overex-

pression induces *COR* genes and enhances freezing tolerance. Science 280:104–106

- Jakoby M, Weisshaar B, Dr ge-Laser W, Vicente-Carbajosa J, Tiedemann J, Kroj T, Parcy F (2002) bZIP transcription factors in Arabidopsis. Trends Plant Sci 7:106–111
- James VA, Neibaur I, Altpeter F (2008) Stress inducible expression of the DREB1A transcription factor from xeric, Hordeum spontaneum L in turf and forage grass (*Paspalum notatum* Flugge) enhances abiotic stress tolerance. Transgenic Res 17:93–104
- Jiang Y, Deyholos MK (2009) Functional characterization of Arabidopsis NaCl-inducible WRKY25 and WRKY33 transcription factors in abiotic stresses. Plant Mol Biol 69:91–105
- Kasuga M, Liu Q, Miura S, Yamaguchi-Shinozaki K, Shinozaki K (1999) Improving plant drought, salt, and freezing tolerance by gene transfer of a single stressinducible transcription factor. Nat Biotechnol 17:287–292
- Kasuga M, Miura S, Shinozaki K, Yamaguchi-Shinozaki K (2004) A combination of the Arabidopsis DREB1A gene and stress-inducible rd29A promoter improved drought- and low-temperature stress tolerance in tobacco by gene transfer. Plant Cell Physiol 45:346–350
- Kim BH, Schöffl F (2002) Interaction between Arabidopsis heat shock transcription factor 1 and 71 kD heat shock proteins. J Exp Bot 53:371–375
- Kim YS, Kim SG, Park JE, Park HY, Lim MH, Chua NH, Park CM (2006) A membrane-bound NAC transcription factor regulates cell division in Arabidopsis. Plant Cell 18:3132–3144
- Kizis D, Pagès M (2002) Maize DRE-binding proteins DBF1 and DBF2 are involved in rab17 regulation through the drought-responsive element in an ABAdependent pathway. Plant J 30:679–689
- Kotak S, Port M, Ganguli A, Bicker F, von Köskull-Döring P (2004) Characterization of C-terminal domains of *Arabidopsis* heat stress transcription factors (Hsfs) and identification of a new signature combination of plant class Ahsfs with AHA and HES motifs essential for activator function and intracelluler localization. Plant J 39:98–112
- Larkindale J, Vierling E (2008) Core genome responses involved in acclimation to high temperature. Plant Physiol 146:748–761
- Leung J, Giraudat J (1998) Abscisic acid signal transduction. Annu Rev Plant Physiol Plant Mol Biol 49:199–222
- Lim CJ, Hwang JE, Chen H, Hong JK, Yang KA, Choi MS, Lee KO, Chung WS, Lee SY, Lim CO (2007) Over-expression of the Arabidopsis DRE/CRTbinding transcription factor DREB2C enhances thermotolerance. Biochem Biophys Res Commun 362:431–436
- Liu Q, Kasuga M, Sakuma Y, Abe H, Miura S, Yamaguchi-Shinozaki K, Shinozaki K (1998) Two transcription factors, DREB1 and DREB2, with an EREBP/AP2 DNA-binding domain separate two cellular signal transduction pathways in drought- and low-temperature-

responsive gene expression in Arabidopsis. Plant Cell 10:1391–1406

- Liu HT, Gao F, Li GL, Han JL, Liu DL, Sun DY, Zhou RG (2008) The calmodulin-binding protein kinase 3 is part of heat-shock signal transduction in *Arabidopsis thaliana*. Plant J 55:760–773
- Lohmann C, Eggers-Schumacher G, Wunderlich M, Schoffl F (2004) Two different heat shock transcription factors regulate immediate early expression of stress genes in Arabidopsis. Mol Genet Genomics 271:11–21
- Lopez-Molina L, Chua NH (2000) A null mutation in a bZIP factor confers ABA-insensitivity in Arabidopsis thaliana. Plant Cell Physiol 41:541–547
- Lyck R, Harmening U, Höhfeld I, Scharf K-D, Nover L (1997) Intracellular distribution and identification of the nuclear localization signals of two tomato heat stress transcription factors. Planta 202:117–125
- Manavalan LP, Guttikonda SK, Tran L-SP, Nguyen HT (2009) Physiological and molecular approaches to improve drought resistance in soybean. Plant Cell Physiol 50:1260–1276
- Maruyama K, Sakuma Y, Kasuga M, Ito Y, Seki M, Goda H, Shimada Y, Yoshida S, Shinozaki K, Yamaguchi-Shinozaki K (2004) Identification of cold-inducible downstream genes of the Arabidopsis DREB1A/CBF3 transcriptional factor using two microarray systems. Plant J 38:982–993
- Matsukura S, Mizoi J, Yoshida T, Todaka D, Ito Y, Maruyama K, Shinozaki K, Yamaguchi-Shinozaki K (2010) Comprehensive analysis of rice DREB2-type genes that encode transcription factors involved in the expression of abiotic stress-responsive genes. Mol Genet Genomics 283:185–196
- Mishra SK, Tripp J, Winkelhaus S, Tschiersch B, Theres K, Nover L, Scharf KD (2002) In the complex family of heat stress transcription facors, HsfA1 has a unique role as master regulator of thermotolerance in tomato. Genes Dev 16:1555–1567
- Morimoto RI (2002) Dynamic remodeling of transcription complexes by molecular chaperones. Cell 110:281
- Nakai A (1999) New aspects in the vertebrate heat shock factor system: Hsf3 and Hsf4. Cell Stress Chaperones 4:86–93
- Nakashima K, Shinwari ZK, Sakuma Y, Seki M, Miura S, Shinozaki K, Yamaguchi-Shinozaki K (2000) Organization and expression of two Arabidopsis DREB2 genes encoding DRE-binding proteins involved in dehydration-and high-salinity-responsive gene expression. Plant Mol Biol 42:657–665
- Nakashima K, Tran LS, Van Nguyen D, Fujita M, Maruyama K, Todaka D, Ito Y, Hayashi N, Shinozaki K, Yamaguchi-Shinozaki K (2007) Functional analysis of a NAC-type transcription factor OsNAC6 involved in abiotic and biotic stress-responsive gene expression in rice. Plant J 51:617–630
- Nakashima K, Ito Y, Yamaguchi-Shinozaki K (2009) Transcriptional regulatory networks in response to abiotic stresses in Arabidopsis and grasses. Plant Physiol 149:88–95

- Narusaka Y, Nakashima K, Shinwari ZK, Sakuma Y, Furihata T, Abe H, Narusaka M, Shinozaki K, Yamaguchi-Shinozaki K (2003) Interaction between two cis-acting elements, ABRE and DRE, in ABAdependent expression of Arabidopsis rd29A gene in response to dehydration and high-salinity stresses. Plant J 34:137–148
- Nelson DE, Repetti PP, Adams TR, Creelman RA, Wu J, Warner DC, Anstrom DC, Bensen RJ, Castiglioni PP, Donnarummo MG, Hinchey BS, Kumimoto RW, Maszle DR, Canales RD, Krolikowski KA, Dotson SB, Gutterson N, Ratcliffe OJ, Heard JE (2007) Plant nuclear factor Y (NF-Y) B subunits confer drought tolerance and lead to improved corn yields on waterlimited acres. Proc Natl Acad Sci USA 104:16450–16455
- Nijhawan A, Jain M, Tyagi AK, Khurana JP (2008) Genomic survey and gene expression analysis of the basic leucine zipper transcription factor family in rice. Plant Physiol 146:333–350
- Nishizawa A, Yabuta Y, Yoshida E, Maruta T, Yoshimura K, Shigeoka S (2006) *Arabidopsis* heat shock transcription factor A2 as a key regulator in response to several types of environmental stress. Plant J 48:535–547
- Nover L, Scharf KD, Gagliardi D, Vergne P, Czarnecka-Verner E, Gurley WB (1996) The Hsf word: classification of plant heat stress transcription factors. Cell Stress Chaperones 1:215–223
- Nover L, Bharti K, Döring P, Mishra SK, Ganguli A, Scharf KD (2001) Arabidopsis and the heat stress transcription factor world: how many heat stress transcription factors do we need? Cell Stress Chaperones 6:177–189
- Nuruzzaman M, Manimekalai R, Sharoni AM, Satoh K, Kondoh H, Ooka H, Kikuchi S (2010) Genome-wide analysis of NAC transcription factor family in rice. Gene 465:30–44
- Ogawa D, Yamaguchi K, Nishiuchi T (2007) High-level overexpression of the Arabidopsis HsfA2 gene confers not only increased themotolerance but also salt/ osmotic stress tolerance and enhanced callus growth. J Exp Bot 58:3373–3383
- Oh SJ, Song SI, Kim YS, Jang HJ, Kim SY, Kim M, Kim YK, Nahm BH, Kim JK (2005) Arabidopsis CBF3/ DREB1A and ABF3 in transgenic rice increased tolerance to abiotic stress without stunting growth. Plant Physiol 138:341–351
- Ooka H, Satoh K, Doi K, Nagata T, Otomo Y, Murakami K, Matsubara K, Osato N, Kawai J, Carninci P, Hayashizaki Y, Suzuki K, Kojima K, Takahara Y, Yamamoto K, Kikuchi S (2003) Comprehensive analysis of NAC family genes in *Oryza sativa* and *Arabidopsis thaliana*. DNA Res 20:239–247
- Panchuk II, Volkov RA, Schoffl F (2002) Heat stress and heat shock transcription factor-dependent expression and activity of ascorbate peroxidase in Arabidopsis. Plant Physiol 129:838–853
- Pellegrineschi A, Reynolds M, Pacheco M, Brito RM, Almeraya R, Yamaguchi-Shinozaki K, Hoisington D

(2004) Stress-induced expression in wheat of the *Arabidopsis thaliana* DREB1A gene delays water stress symptoms under greenhouse conditions. Genome 47:493–500

- Port M, Tripp J, Zielinski D, Weber C, Heerklotz D, Winkelhaus S, Bubla KD, Scharf KD (2004) Role of HSP17.4-CII as coregulator and cytoplasmic retention factor of tomato heat stress transcription factor HsfA2. Plant Physiol 135:1457–1470
- Qin F, Sakuma Y, Li J, Liu Q, Li YQ, Shinozaki K, Yamaguchi-Shinozaki K (2004) Cloning and functional analysis of a novel DREB1/CBF transcription factor involved in cold-responsive gene expression in *Zea mays* L. Plant Cell Physiol 45:1042–1052
- Qin F, Kakimoto M, Sakuma Y, Maruyama K, Osakabe Y, Tran LS, Shinozaki K, Yamaguchi-Shinozaki K (2007) Regulation and functional analysis of ZmDREB2A in response to drought and heat stresses in *Zea mays* L. Plant J 50:54–69
- Qin F, Sakuma Y, Tran LS, Maruyama K, Kidokoro S, Fujita Y, Fujita M, Umezawa T, Sawano Y, Miyazono K, Tanokura M, Shinozaki K, Yamaguchi-Shinozaki K (2008) Arabidopsis DREB2A interacting proteins function as RING E3 ligases and negatively regulate plant drought stress-responsive gene expression. Plant Cell 20:1693–1707
- Rabbani MA, Maruyama K, Abe H, Khan MA, Katsura K, Ito Y, Yoshiwara K, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2003) Monitoring expression profiles of rice genes under cold, drought, and high-salinity stresses and abscisic acid application using cDNA microarray and RNA gel-blot analyses. Plant Physiol 133:1755–1767
- Rizhsky L, Davletova S, Liang H, Mittler R (2004) The zinc finger protein Zat12 is required for cytosolic ascorbate peroxidase 1 expression during oxidative stress in Arabidopsis. J Biol Chem 279: 11736–11743
- Rommens CM, Haring MA, Swords K, Davies HV, Belknap WR (2007) The intragenic approach as a new extension to traditional plant breeding. Trends Plant Sci 12:397–403
- Rushton PJ, Somssich IE, Ringler P, Shen QJ (2010) WRKY transcription factors. Trends Plant Sci 15:247–258
- Sakamoto H, Maruyama K, Sakuma Y, Meshi T, Iwabuchi M, Shinozaki K, Yamaguchi-Shinozaki K (2004) Arabidopsis Cys2/His2-type zinc-finger proteins function as transcription repressors under drought, cold, and high-salinity stress conditions. Plant Physiol 136:2734–2746
- Sakuma Y, Liu Q, Dubouzet JG, Abe H, Shinozaki K, Yamaguchi-Shinozaki K (2002) DNA-binding specificity of the ERF/AP2 domain of Arabidopsis DREBs, transcription factors involved in dehydration- and cold inducible gene expression. Biochem Biophys Res Commun 290:998–1009
- Sakuma Y, Maruyama K, Osakabe Y, Qin F, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2006a) Functional analysis of an Arabidopsis transcription

factor, DREB2A, involved in drought-responsive gene expression. Plant Cell 18:1292–1309

- Sakuma Y, Maruyama K, Qin F, Osakabe Y, Shinozaki K, Yamaguchi-Shinozaki K (2006b) Dual function of an Arabidopsis transcription factor DREB2A in water-stress-responsive and heat-stress-responsive gene expression. Proc Natl Acad Sci USA 103: 18822–18827
- Saleh A, Lumbreras V, Lopez C, Dominguez-Puigjaner E, Kizis D, Pagès M (2006) Maize DBF1-interactor protein 1 containing an R3H domain is a potential regulator of DBF1 activity in stress responses. Plant J 46:747–757
- Savitch LV, Allard G, Seki M, Robert LS, Tinker NA, Huner NP, Shinozaki K, Singh J (2005) The effect of overexpression of two Brassica CBF/DREB1-like transcription factors on photosynthetic capacity and freezing tolerance in *Brassica napus*. Plant Cell Physiol 46:1525–1539
- Scharf KD, Heider H, Höhfeld I, Lack R, Schmidt E, Nover L (1998) The tomato Hsf tomato Hsf system: HsfA2 needs interaction with HsfA1 for efficient nuclear import and may be localized in cytoplasmic heat stress granules. Mol Cell Biol 18:2240–2251
- Schöffl F, Prändl R, Reindl A (1998) Regulation of the heat-shock response. Plant Physiol 117:1135–1141
- Schramm F, Larkindale J, Kiehlmann E, Ganguli A, Englich G, Vierling E, von Koskull-Döring P (2008) A cascade of transcription factor DREB2A and heat stress transcription factor HsfA3 regulates the heat stress response of Arabidopsis. Plant J 53:264–274
- Seki M, Narusaka M, Ishida J, Nanjo T, Fujita M, Oono Y, Kamiya A, Nakajima M, Enju A, Sakurai T, Satou M, Akiyama K, Taji T, Yamaguchi-Shinozaki K, Carninci P, Kawai J, Hayashizaki Y, Shinozaki K (2002) Monitoring the expression profiles of 7000 Arabidopsis genes under drought, cold and high-salinity stresses using a fulllength cDNA microarray. Plant J 31:279–292
- Shen Q, Zhang P, Ho TH (1996) Modular nature of abscisic acid (ABA) response complexes: composite promoter units that are necessary and sufficient for ABA induction of gene expression in barley. Plant Cell 8:1107–1119
- Shinozaki K, Yamaguchi-Shinozak K (2000) Molecular responses to dehydration and low temperature: differences and cross-talk between two stress signaling pathways. Curr Opin Plant Biol 3:217–223
- Shinozaki K, Yamaguchi-Shinozaki K, Seki M (2003) Regulatory network of gene expression in the drought and cold stress responses. Curr Opin Plant Biol 6:410–417
- Skinner JS, von Zitzewitz J, Szűcs P, Marquez-Cedillo L, Filichkin T, Amundsen K, Stockinger EJ, Thomashow MF, Chen THH, Hayes PM (2005) Structural, functional, and phylogenetic characterization of a large CBF gene family in barley. Plant Mol Biol 59:533–551
- Stockinger EJ, Gilmour SJ, Thomashow MF (1997) Arabidopsis thaliana CBF1 encodes an AP2 domaincontaining transcriptional activator that binds to the C-repeat/DRE, a cis-acting DNA regulatory element

that stimulates transcription in response to low temperature and water deficit. Proc Natl Acad Sci USA 94:1035–1040

- Sugio A, Dreos R, Aparicio F, Maule AJ (2009) The cytosolic protein response as subcomponent of the wider heat shock response in Arabidopsis. Plant Cell 21:642–654
- Sun W, Van MM, Verbruggen N (2002) Small heat shock proteins and stress tolerance in plants. Biochim Biophys Acta 1577:1–9
- Swindell WR, Huebner M, Weber AP (2007) Transcriptional profiling of Arabidopsis heat shock proteins and transcription factors reveals extensive overlap between heat and non-heat stress response pathways. BMC Genomics 8:125
- Taji T, Ohsumi C, Iuchi S, Seki M, Kasuga M, Kobayashi M, Yamaguchi-Shinozaki K, Shinozaki K (2002) Important roles of drought- and cold-inducible genes for galactinol synthase in stress tolerance in *Arabidopsis thaliana*. Plant J 29:417–426
- Terashima A, Takumi S (2009) Allopolyploidization reduces alternative splicing efficiency for transcripts of the wheat DREB2 homolog, WDREB2. Genome 52:100–105
- Thomashow MF (2001) So what's new in the field of plant cold acclimation? Lots! Plant Physiol 125:89–93
- Tran L-SP, Mochida K (2010) Functional genomics of soybean for improvement of productivity in adverse conditions. Funct Integr Genomics 10:447–462
- Tran LS, Nakashima K, Sakuma Y, Simpson SD, Fujita Y, Maruyama K, Fujita M, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2004) Isolation and functional analysis of Arabidopsis stress-inducible NAC transcription factors that bind to a drought-responsive cis-element in the early responsive to dehydration stress 1 promoter. Plant Cell 16:2481–2498
- Tran L-SP, Nakashima K, Sakuma Y, Osakabe Y, Qin F, Simpson SD, Maruyama K, Fujita Y, Shinozaki K, Yamaguchi-Shinozaki K (2007) Co-expression of the stress-inducible zinc finger homeodomain ZFHD1 and NAC transcription factors enhances expression of the ERD1 gene in Arabidopsis. Plant J 49:46–63
- Tran L-SP, Nishiyama R, Yamaguchi-Shinozaki K, Shinozaki K (2010) Potential utilization of NAC transcription factors to enhance abiotic stress tolerance in plants by biotechnological approach. GM Crops 1:32–39
- Uno Y, Furihata T, Abe H, Yoshida R, Shinozaki K, Yamaguchi-Shinozaki K (2000) Arabidopsis basic leucine zipper transcription factors involved in an abscisic acid-dependent signal transduction pathway under drought and high-salinity conditions. Proc Natl Acad Sci USA 97:11632–11637
- Vágújfalvi A, Aprile A, Miller A, Dubcovsky J, Delugu G, Galiba G, Cattivelli L (2005) The expression of several Cbf genes at the Fr-A2 locus is linked to frost resistance in wheat. Mol Genet Genomics 274:506–514
- Vinocur B, Altman A (2005) Recent advances in engineering plant tolerance to abiotic stress: achievements and limitations. Curr Opin Biotechnol 16:123–132
- Vogel JT, Zarka DG, Van Buskirk HA, Fowler SG, Thomashow MF (2005) Roles of the CBF2 and ZAT12

transcription factors in configuring the low temperature transcriptome of Arabidopsis. Plant J 41: 195-211

- Von Koskull-Döring P, Scharf KD, Nover L (2007) The diversity of plant heat stress transcription factors. Trends Plant Sci 12:452–457
- Wang W, Vinocur B, Shoseyov O, Altman A (2004) Role of plant heat-shock proteins and molecular chaperones in the abiotic stress response. Trends Plant Sci 9(5):244–252
- Wang Z, Zhu Y, Wang L, Liu X, Liu Y, Phillips J, Deng X (2009) A WRKY transcription factor participates in dehydration tolerance in *Boea hygrometrica* by binding to the W-box elements of the galactinol synthase (BhGolS1) promoter. Planta 230:1155–1166
- Wu X, Shiroto Y, Kishitani S, Ito Y, Toriyama K (2009a) Enhanced heat and drought tolerance in transgenic rice seedlings overexpressing OsWRKY11 under the control of HSP101 promoter. Plant Cell Rep 28:21–30
- Wu YR, Deng ZY, Lai JB, Zhang YY, Yang CP, Yin BJ, Zhao QZ, Zhang L, Li Y, Yang CW, Xie Q (2009b) Dual function of Arabidopsis ATAF1 in abiotic and biotic stress responses. Cell Res 19:1279–1290
- Xie Z, Ruas P, Shen QJ (2005) Regulatory networks of the phytohormone abscisic acid. Vitam Horm 72:235–269
- Xin HB, Zhang H, Chen L, Li XX, Lian QL, Yuan X, Hu XY, Cao L, He XL, Yi MF (2010) Cloning and characterization of HsfA2 from lily (*Lilium longiflorum*). Plant Cell Rep 29:875–885
- Xing H, Wilkerson DC, Mayhew CN, Lubert EJ, Skaggs HS, Goodson ML, Hong Y, Park-Sarge O-K, Sarge KD (2005) Mechanism of HSP70i gene bookmarking. Science 307:421–423
- Xue GP, Loveridge CW (2004) *HvDRF1* is involved in abscisic acid-mediated gene regulation in barley and produces two forms of AP2 transcriptional activators, interacting preferably with a CT-rich element. Plant J 37:326–339
- Yamaguchi-Shinozaki K, Shinozaki K (1994) A novel cisacting element in an Arabidopsis gene is involved in responsiveness to drought, low-temperature, or highsalt stress. Plant Cell 6:251–264
- Yamaguchi-Shinozaki K, Shinozaki K (2006) Transcriptional regulatory networks in cellular responses and tolerance to dehydration and cold stress. Annu Rev Plant Biol 57:781–803
- Yamanouchi U, Yano M, Lin H, Ashikari M, Yamada K (2002) A rice spotted leaf gene, Spl7, encodes a heat stress transcription factor protein. Proc Natl Acad Sci USA 99:7530–7535
- Yang W, Liu XD, Chi XJ, Wu CA, Li YZ, Song LL, Liu XM, Wang YF, Wang FW, Zhang C, Liu Y, Zong JM, Li HY (2010) Dwarf apple MbDREB1 enhances plant tolerance to low temperature, drought, and salt stress via both ABA-dependent and ABA-independent pathways. doi 10.1007/s00425-010-1279-6
- Yokotani N, Ichikawa T, Kondou Y, Matsui M, Hirochika H, Iwabuchi M, Oda K (2008) Expression of rice heat stress transcription factor OsHsfA2e enhances tolerance to environmental stresses in transgenic Arabidopsis. Planta 227:957–967

- Yoshida T, Sakuma Y, Todaka D, Maruyama K, Qin F, Mizoi J, Kidokoro S, Fujita Y, Shinozaki K, Yamaguchi-Shinozaki K (2008) Functional analysis of an Arabidopsis heat-shock transcription factor HsfA3 in the transcriptional cascade downstream of the DREB2A stress-regulatory system. Biochem Biophys Res Commun 368:515–521
- Yoshida T, Fujita Y, Sayama H, Kidokoro S, Maruyama K, Mizoi J, Shinozaki K, Yamaguchi-Shinozaki K (2010) AREB1, AREB2, and ABF3 are master transcription factors that cooperatively regulate ABRE-dependent ABA signaling involved in drought stress tolerance and require ABA for full activation. Plant J 61:672–685
- Zhang X, Fowler SG, Cheng H, Lou Y, Rhee SY, Stockinger EJ, Thomashow MF (2004) Freezing-sensitive tomato has a functional CBF cold response pathway, but a CBF regulon that differs from that of freezing-tolerant Arabidopsis. Plant J 39:905–919
- Zhang LR, Li YS, Xing D, Gao CJ (2009) Characterization of mitochondrial dynamics and subcellular localization of ROS reveal that HsfA2 alleviates oxidative damage caused by heat stress in Arabidopsis. J Exp Bot 60:2073–2091
- Zhao H, Bughrara SS (2008) Isolation and characterization of cold-regulated transcriptional activator LpCBF3

gene from perennial ryegrass (*Lolium perenne* L). Mol Genet Genomics 279:585–594

- Zhu BG, Ye CJ, Lü HY, Chen XJ, Chai GH, Chen JN, Wang C (2006) Identification and characterization of a novel heat shock transcription factor gene, GmHsfA1, in soybeans (*Glycine max*). J Plant Res 119:247–256
- Zhu SY, Yu XC, Wang XJ, Zhao R, Li Y, Fan RC, Shang Y, Du SY, Wang XF, Wu FQ, Xu YH, Zhang XY, Zhang DP (2007) Two calcium-dependent protein kinases, CPK4 and CPK11, regulate abscisic acid signal transduction in Arabidopsis. Plant Cell 19:3019–3036
- Zhu Y, Wang Z, Jing YJ, Wang LL, Liu X, Liu YX, Deng X (2009) Ectopic over-expression of BhHsf1, a heat shock factor from the resurrection plant *Boea hygrometrica*, leads to increased thermotolerance and retarded growth in transgenic Arabidopsis and tobacco. Plant Mol Biol 71:451–467
- Zou X, Seemann JR, Neuman D, Shen QJ (2004) A WRKY gene from creosote bush encodes an activator of the abscisic acid signaling pathway. J Biol Chem 279:55770–55779
- Zou M, Guan Y, Ren H, Zhang F, Chen F (2008) A bZIP transcription factor, OsABI5, is involved in rice fertility and stress tolerance. Plant Mol Biol 66:675–683

Plant Signaling Under Abiotic Stress Environment

14

Parvaiz Ahmad, Renu Bhardwaj, and Narendra Tuteja

Abstract

Abiotic stress is one of the major factors that negatively affect crops yield; therefore, development of stress-tolerant crops is essential for future food security. The stress stimuli are perceived by the plasma membrane and different signals get activated. In response to the stress, expression of many genes gets altered which plays an important role in the transmission of the signals. Various chemicals are also responsible for these signals like calcium (Ca2+), nitric oxide (NO), sugars, abscisic acid (ABA), brassinosteroids (BRs), ethylene, jasmonates (JA), salicylic acid (SA), and auxins. Ca²⁺ acts as a secondary messenger to perceive the environmental stimuli and transduce them into downstream effectors in order to bring about changes leading to adaptations to stressful conditions or developmental effects. The phytohormones have a role in tolerance and adaptations to plants under abiotic stress. During abiotic stress, cross talk between different signaling pathways is very common. In the present review, we elucidated the role of these chemicals in plant signaling under abiotic stress. The signal transduction pathway involving mitogen-activated protein kinases (MAPK) under abiotic stress is also discussed.

Keywords

Abiotic stress • ABA • Ca²⁺ signaling • CaM • CDPK • MAPK • Phytohormones

P. Ahmad (🖂)

Department of Botany, A.S. College, Srinagar 190008, Jammu & Kashmir, India e-mail: parvaizbot@rediffmail.com

R. Bhardwaj

Department of Botanical and Environmental Sciences, Guru Nanak Dev University, Amritsar, Punjab, India

N. Tuteja

Plant molecular Biology Group, International Centre for Genetic Engineering and Biotechnology, Aruna Asaf Ali Marg, New Delhi, India

1 Introduction

Plants are forced to exposure to various abiotic stresses including salt, drought, extreme temperatures (cold and heat), UV radiations, etc., since they are unable to move to more favorable places. These abiotic stresses adversely affect the plant metabolic activities and are also responsible for crop loss to a greater extent (Mahajan and Tuteja 2005; Tuteja 2007a). During stress, the upregulation of many genes has been reported which helps the plant to withstand the stress conditions and lead to plant adaptation (Tuteja 2009a).

Plants perceive the external and internal signals and are used to regulate various responses for its development. Exposure to stimuli causes membrane depolarization of the plant cells with 10–30 s, which is coordinated with early Ca²⁺ influx (reviewed by Tuteja and Sopory 2008). Signals are perceived by the membranes and therefore membrane events are the likely routes for signal generation and transduction (Tuteja and Sopory 2008). Different signaling pathways can operate independently to each other and can modulate other pathways (Kaur and Gupta 2005). Sometimes, components of pathways are dependent on each other and can cross talk among them.

Different molecules have been reported to have a role in signal transduction. The present review throws light on various signaling molecules and pathways like calcium and MAPK signaling pathways during abiotic stress. We also tried to cover the role of phytohormones such as ABA, SA, JA, BR, etc., in abiotic stress signaling.

2 Calcium Signaling

Ca²⁺ regulates a range of activities within the cell such as cell division and elongation, cytoplasmic streaming, photomorphogenesis, and plant defense against environmental stresses (Song et al. 2008; Tuteja 2009b; Kader and Lindberg 2010). It functions as the central node in overall signaling web and has a promising role in stress tolerance (Tuteja and Sopory 2008). During abiotic stress, Ca²⁺ acts as a second messenger under various stress conditions (Knight 2000). Ca²⁺ signatures have a leading role in numerous physiological processes such as regulation of stomatal apertures in plants (Allen et al. 2001). Ca²⁺ signatures change according to the nature of stress (Kiegle et al. 2000), duration of stress (Plieth et al. 1999), earlier exposure to any stress (Knight et al. 1997), and the type of tissue exposed to stress (Kiegle et al. 2000). For studying the role of calcium and its dynamics, various pharmacological and transgenic approaches have been utilized. The calcium reporter protein aequorin has proved to be very useful for such studies in revealing calcium fluxes within one cell and in different tissues.

Calcium-binding proteins act as stress sensors, get conformationally transformed upon binding with Ca²⁺, which facilitates their interaction with downstream effector molecules (Clapham 2007; Gifford et al. 2007; Tuteja 2009b). Elongation factor (EF)-hand motif, which generally occurs in pairs, is the most common motif present in these Ca2+-binding proteins and helps in highaffinity binding of Ca2+. It has a helix-loop-helix structure and is reported in about 250 proteins of Arabidopsis (Day et al. 2002). In plants these EF-hand proteins comprise three different categories, namely, CDPKs (Ca²⁺-dependent protein kinases), CaMs (calmodulins) and CMLs (CaMlike proteins), and the CBLs (calcineurin B-like proteins) (Fig. 14.1). Of these, only CDPKs act as "responders," as they are capable of directly transducing signals through their catalytic activity. CaMs/CMLs and CBLs are only sensors for regulating the downstream targets. CMLs, CDPKs, and CBLs are restricted to plants and some protists, whereas CaM is universal to all eukaryotes.

2.1 Calcium-Dependent Protein Kinases

CDPKs have been reported from several plants (Kawasaki et al. 2001; Seki et al. 2002; Ozturk et al. 2002; Asano et al. 2011) and are specific for a particular osmotic stress response. Studies on salt-tolerant and salt-sensitive rice varieties revealed specific CDPKs to be induced earlier with a sustained expression in the tolerant variety in comparison to the sensitive variety (Kawasaki et al. 2001). *Arabidopsis* CDPKs are also highly specific (Sheen 1996). Out of several CDPKs tested, only AtCDPK1 and AtCDPK1a were able to transcriptionally activate selected reporter



genes. CDPKs are five domain-containing proteins, and range from ~40 to 90 kDa in size. Their catalytic domain consists of a highly conserved serine/threonine kinase region, whereas, their N-terminal variable domain ranges from 21 to 185 amino acids in length (Klimecka and Muszynska 2007). Adjacent to the kinase domain is a pseudosubstrate-containing autoinhibitory junction domain that interacts with the active site and inhibits its kinase activity. Next to the autoinhibitory domain is the CaM-like domain (CLD) which is responsible for its Ca²⁺-binding activity. Their C-terminal domain is relatively short and variable.

Plants have a large family of Ca²⁺-dependent protein kinase (CDPKs) and have an important role in signaling during abiotic stresses like drought, wounding, and cold (Fig. 14.2). Induction of CDPK by osmotic stress has been shown by many workers in various plants (Kawasaki et al. 2001; Seki et al. 2002; Ozturk et al. 2002; Witte et al. 2010; Franza et al. 2011; Wurzinger et al. 2011). Induction of CDPK exists in longer duration in tolerant varieties of rice than in the salt-sensitive variety (Kawasaki et al. 2001). Martin and Busconi (2001) have reported that cold stress activates the membrane-associated CDPK in rice plants. Saijo et al. (2001) have also



reported that overexpression of OsCDPK7 in rice provides cold and osmotic stress tolerance in these plants. Sheen (1996) has demonstrated that CDPK is involved in signal transduction. In his experiment, he demonstrated that in protoplast of maize leaf the expression of stress-responsive HVA1 was induced by active AtCDPK1. Sheen (1996) has also reported that AtCDPK1 activation is blocked by a protein phosphatase type 2C (AtPP2CA). Enhanced induction of CBF1, RAB18, RC12A, and LT178 (i.e., RD29A) gene expression has been observed in AtPP2CAsilenced Arabidopsis plants during cold and ABA treatment (Tahtiharju and Palva 2001). Romeis et al. (2001) also reported that pathogen infection also activated CDPK in plants.

2.2 Calmodulin and Other Calcium-Binding Proteins

Elevated levels of calcium within a cell activate various calcium-binding proteins which in turn induce specific kinases. On the basis of function, calcium-binding proteins are classified into two groups: (1) trigger proteins and (2) buffer proteins. The trigger proteins are activated when they bind with Ca^{2+} and after that they interact with other

proteins and alter their activity. The trigger-type CaBPs are calmodulin (CaM), CaM-binding proteins, Ca²⁺-dependent protein kinase, and phosphatase (Reddy 2001).

Stress causes increase in inositol 1,4,5-triphosphate (IP3). Activation of phospholipase C (PLC) results in hydrolysis of PIP2 to IP3. IP3 is regarded as the activator of vacuolar calcium channels in plants. During stress, the increase in cytosolic Ca²⁺ is due to the activation of IP3dependent calcium channels (Braam 2005; Boudsocq and Laurière 2005). Furthermore, calcium-binding proteins (CaBP) or calcium sensors recognize and translate the information provided in calcium signatures (Tuteja and Mahajan 2007) and pass the information downstream for regulation of gene expression.

These CaBPs are partially responsible in modulating the intracellular calcium levels. The calcium-binding proteins can be regulated either in a cell- or in a tissue-specific manner. NaCl-inducible Ca²⁺/calmodulin-dependent protein kinase in pea (PsCCaMK) was reportedly specific to roots (Pandey et al. 2002). A calmodulin-binding transcription activator family was shown to be specific only to the multicellular organisms (Bouche et al. 2002). Other examples of osmotic stress-activated calcium-binding proteins include



Arabidopsis protein AtCP1, the membraneassociated rice protein OsEFA27, and the *Arabidopsis* counterpart RD20 (Frandsen et al. 1996; Jang et al. 1998). Some CaBPs can be negative regulators of osmotic stress also; one example is the calmodulin-binding protein in *Arabidopsis*, AtCaMBP25. Although it is upregulated by osmotic stress, its overexpression renders plants sensitive to osmotic or salt stresses and its antisense transgenics show improved tolerance (Perruc et al. 2004).

Calmodulin, an important CaBP, is a small acidic protein and is responsible for the regulation of intracellular Ca2+ levels. Increased Ca2+ concentration activates calmodulin which then induces specific kinases. Calmodulin is a very important calcium-binding protein in Ca2+ signaling and has been found to be involved in biotic and abiotic stresses (Fig. 14.3) (Reddy 2001; Tuteja and Mahajan 2007; Tuteja and Sopory 2008). Plants have been found to possess unique Ca²⁺ sensors like calmodulin-like proteins (CMLs), and Arabidopsis contains 50 such proteins (Tuteja and Sopory 2008). The calmodulinlike proteins differ from calmodulin in having more than 148 amino acids and also have one to six EF-hand motifs. These CMLs have been found to play a role as Ca2+ sensor during stress in plants (Vanderbeld and Snedden 2007). Certain Ca²⁺-binding proteins do not contain EF-hand motifs, such as calreticulin, annexins, calnexin, phospholipase D (PLD), and pistil-expressed Ca²⁺-binding proteins.

PLD activity has been reported to be involved in ethylene and ABA responses, synthesis of α -amylase in aleurone cells, closing of stomata, responses to pathogens, leaf senescence, and drought tolerance (reviewed by Tuteja and Sopory 2008).

Annexins have been reported to be involved in biological membrane organization and functions (Tuteja and Mahajan 2007). The exact function of annexins is not clear yet but are thought to play a role in secretary processes and they have ATPase and peroxidase activities (Tuteja and Mahajan 2007). Annexins have been regarded to have a role in stress responses (Gorecka et al. 2007a). Annexin At1 of *Arabidopsis thaliana* (AnnAt1) plays a vital role in pH-mediated cellular responses to environmental stress (Gorecka et al. 2007b).

Calnexin (CNX) is an important calciumbinding protein and is an endoplasmic reticulum (ER) type 1 integral membrane protein. CNX behaves as a molecular chaperone and has a leading role in the recognition of misfolded proteins, lectin-like activity, and Ca²⁺ binding (Sarwat and Tuteja 2007). Till date, the role of calnexin in stress has not been reported but it is considered that it has a role in ER stress response in plants (reviewed by Tuteja and Sopory 2008).

In plants, another group of Ca²⁺ sensors is the SOS (salt overly sensitive) family and is responsible for calcium-mediated pathway for salinity stress tolerance (Mahajan et al. 2008). Zhu (2003) has isolated sos mutants (sos1, sos2, and sos3) from Arabidopsis which are hypersensitive to salt. The sos mutants have been shown to accumulate more proline under salt stress which gives protection to the salt-stressed plants (Liu and Zhu 1998). Cloning and characterization of sos genes (SOS1-SOS3) has opened new doors for ion homeostasis and plant tolerance to salt. It has been reported that sos1, sos2, and sos3 function in a common pathway leading to salt tolerance (Halfter et al. 2000; Zhu 2000). Sos1 is activated by sos3-sos2 complex and has been reported by many workers. Shi and Zhu (2002) demonstrated that if sos1 alone was allowed to express in yeast cells, slight enhancement in salt tolerance was observed. However, if sos1 was expressed with sos3 and sos2 the yeast cells showed more tolerance to salt. Qiu et al. (2002) have reported very low Na⁺/H⁺ exchange activity in sos mutant plants in comparison with wild type in plasma membrane vesicles. Addition of activated sos2 protein to isolated membrane vesicles of mutant plants showed that the exchange activity increased in sos2 and sos3 but remains unaffected in sos1 mutants. The results lead to the conclusion that Na⁺/H⁺ exchange activity of sos1 was stimulated by sos3 and sos2 (reviewed by Xiong et al. 2002). sos3-sos2 also regulate Na⁺ transporter AtHKT1, which is a salt tolerance effector (Uozumi et al. 2000). Homologs of HKT1 in other plant species revealed that it can be either a K⁺ transporter or a Na⁺/K⁺ cotransporter (Horie et al. 2001; Liu et al. 2001). Rus et al. (2001) demonstrated that mutation in AtHKT1 of Arabidopsis suppressed the salt hepersensitivity phenotype of sos3, leading to the concept that activity of AtHKT1, the Na⁺ influx transporter, may be inhibited by SOS3 (reviewed by Xiong et al. 2002).

As compared to caltractin and calmodulin, sos3 binds with Ca^{2+} with low affinity and has three EF-hand motifs (Ishitani et al. 2000). In sos3 mutation, one of the EF-hand motifs gets mutated, thus preventing it to bind Ca^{2+} (Ishitani et al. 2000). *sos4* and *sos5* have recently been characterized (Mahajan et al. 2008). *sos4* encodes a pyridoxal (PL) kinase which has a role in the biosynthetic pathway of pyridoxal-5-phosphate, which is an active form of vitamin B6. Sos5 is a putative cell surface adhesion protein and has a role in normal cell expansion. Under salt stress, *sos5* has been found to have a role of maintenance of cell wall integrity (reviewed by Tuteja and Sopory 2008).

2.3 Calcineurin B-Like Proteins

These proteins possess four Ca^{2+} -binding EF-hand domains and have significant identity to calcineurin B subunit and neural calcium sensor from yeast and animals (Kudla et al. 1999). The CBL protein family and their corresponding kinases (CIPKs) together form a complex and dynamic Ca^{2+} decoding signaling network (Fig. 14.4). Together they show Ca^{2+} -binding functionality and kinase activity (Mahajan et al. 2006a).

CIPKs have a conserved N-terminal kinase domain and C-terminal regulatory domain, separated from the kinase domain by a variable junction domain. A conserved NAF domain present in the divergent regulatory domain is required for interaction with CBLs. CBL binds to the NAF domain of CIPKs, releases the C-terminal (autoinhibitory) domain from the kinase domain, in turn transforming the kinase into its active state (Guo et al. 2001; Mahajan et al. 2006a).

Bioinformatic analysis shows a number of components of both proteins in the families of CBLs and CIPKs; 10 CBLs and 26 CIPKs in *Arabidopsis*, 10 CBLs and 30 CIPKs in rice (Albrecht et al. 2001; Kolukisaoglu et al. 2004; Weinl and Kudla 2009). However, several species of green algae possess single CBL and CIPK genes, other plant species can have multiples, like *Physcomitrella* contains four CBLs and seven CIPKs and the fern *Selaginella moellendorfii* has five CBLs and five CIPKs (Batistic and Kudla 2009); Weinl and Kudla 2009); thus showing evolutionary complexity of these CBL and CIPK protein families from lower to higher organisms.



CBLs are found to be localized all through the cell. In Arabidopsis itself, four CBLs are present at the plasma membrane, four at the vacuolar membrane, and two in the cytoplasm and nucleus (Batistic et al. 2008; Cheong et al. 2007; D' Angelo et al. 2006; Kim et al. 2007; Weinl and Kudla 2009). The pea CBL was reported to be exclusively localized in the cytosol whereas pea CIPK is localized in the cytosol and the outer membrane (Mahajan et al. 2006b). The pea CBL and CIPK were reported to be coordinately upregulated in response to high NaCl, cold, wounding and also in response to calcium and salicylic acid, whereas drought and abscisic acid had no effect on the expression of these genes (Mahajan et al. 2006b).

Our knowledge has been greatly facilitated by the reverse genetic approaches. A CIPK3 loss-offunction (LOF) mutant showed its involvement in regulating ABA-induced gene expression and ABA responses during seed germination (Kim et al. 2003). Through these studies, CBL1 has came out to function in an ABA-independent manner in controlling responses to drought, cold, and salinity (Albrecht et al. 2003; Cheong et al. 2003), whereas its closely related Ca²⁺ sensor CBL9 renders plants hypersensitive to ABA (Pandey et al. 2004). Interestingly, when CIPK1 complexes with CBL1, it mediates the ABA-dependent pathway (D' Angelo et al. 2006). CBL1 and CBL9 activate CIPK23, and the complex regulates the activity of the shaker-like K⁺ channel ARABIDOPSIS K⁺ TRANSPORTER1 (AKT1) and thus contributes in K⁺ homeostasis within the cell (Kudla et al. 1999). This complex also has a role in stomatal regulation under drought conditions.

CBL–CIPK network is a central and critical system functioning in response to a broad variety of stimuli in order to decode Ca²⁺ signals. Each CBL and each CIPK can make alternative protein interactions and are part of multifunctional signaling component, thus determining the flow of information (Mahajan et al. 2006a).

3 Signaling Through MAPK Kinases

3.1 Historical Background

Sturgill and Ray (1986) for the first time discovered MAPK in animal cells and named it as microtubule-associated protein-2 kinase (MAP-2 kinase). It was renamed as mitogen-activated protein kinase (MAP kinase) 1 by Rossomando et al. (1987)

as mitogen was found to activate the group of proteins and this kinase was found to be related to these proteins (reviewed by Sanan-Mishra et al. 2006; Sinha et al. 2011). In 1990, it was reported as serine/tyrosine kinase that belonged to a multigene family (Gotoh et al. 1990). MAP kinase genes (MsERK1) in plant system was for the first time reported from alfalfa in 1993 (Duerr et al. 1993) and D5 kinase in pea (Stafstrom et al. 1993). After that they were reported from various plants like tobacco, Arabidopsis, etc. (Jonak et al. 1994). Three major groups of MAPKs are found in yeast and animals: (1) extracellular signal-regulated kinases (ERK) (Cobb et al. 1994), (2) c-Jun amino (NH₂)-terminal kinases or stress-activated protein kinases (JNS/SAPK) (Davis 1994), and (3) high osmolarity glycerol response or p38 kinases (Hog/p38) (Landry and Huot 1995). MAP kinase genes reported in plants belong to the ERK subfamily (Hirt 2000) and transmit a broader range of stimuli (Ligterink and Hirt 2001). Stafstrom et al. (1993) reported that the first MAP kinase gene isolated from pea has 41% identity with plant cdc2 kinases and other kinases involved in osmosensing. MAP kinases generally function as a cascade in which MAPKKK phosphoralates and activates MAPKK which in turn activates MAPK. All the three kinases are interlinked together and are also called extracellular receptor kinases (Hirt 2000; Sanan-Mishra et al. 2006). Different plant MAPKs recognize different substrates (Jonak et al. 2002) because of the high similarity in catalytic domain and little similarity in N-termini. The activation domain of most of the plant MAPKs contains TEY (Thu-Glu-Tyr) sequence and is similar to ERK/MAP kinases group of mammals and yeast (Hirt 2000). The activation domain of some MAPKs possesses TDY (Thr-Asn-Tyr) sequence and is closer to the p38/Hog group of mammals and yeast (Tena et al. 2001). No plant MAPK has been found which has TPY (Thr-Pro-Tyr) sequence at its activation domain. There are three functionally linked protein kinase viz.: MAP3K, MAP2K, and MAPKs. 60 MAP3Ks, 10 MAP2Ks, and 20 MAPK have been reported in Arabidopsis thaliana by Group et al. (2002).

3.2 MAP3Ks

It is also known as MAPKKKs or MEKKs. MAP3Ks constitute a diverse family of kinases and are divided into two subfamilies viz. MEKK1 and RAF-like kinases. The MEKK1-like subfamily members are similar to mammalian MEKK1 and to yeast STE11 and BCK1 and RAF-like kinases are similar to mammalian RAF1 MAPK (Group et al. 2002). About 8-10 algal MAP3Ks are found in Chlamydomonas and Volvox and about 40-60 in Sorghum and Populous. One of the important things in MEKK-like kinases is having a conserved catalytic domain and Arabidopsis has ten members in this group. Arabidopsis has been reported to have 80 putative MAPKKKs whereas rice has 75 members (Rao et al. 2010). Plant species having homologs of MAPKKKs have been identified, including the MEKK-like protein kinases, oxidative stressactivated MAP triplekinase 1 (OMTK1) from alfalfa (Nakagami et al. 2004), ANP1, ANP2, ANP3 (Kovtun et al. 2000), YDA (Lukowitz et al. 2004) from Arabidopsis, NPK1 (Nicotiana protein kinase 1) from tobacco (Nishihama et al. 2001), Raf-like protein kinase, EDR1 (enhanced disease resistance 1), and CTR1 (constitutive triple response 1) from Arabidopsis (Frye et al. 2001; Kieber et al. 1993). MEKK-like proteins have been found to participate in canonical MAP kinase cascades that activate downstream MAP2Ks (Rodriguez et al. 2010). The two RAFlike MAP3Ks are CTR1 and EDR1, and these two RAF-like MAP3Ks are found to participate in ethylene-mediated signaling and defense responses (Frye et al. 2001; Huang et al. 2003). It has also been reported that CTR1 and EDR1 do not participate in a canonical MAPK cascade (Rodriguez et al. 2010).

3.3 MAP2Ks

It is also known as MEKs and MKKs and is divided into four groups viz. Groups A, B, C, and D (Hamel et al. 2006). MKK1 and MKK2 belong to Group A and act upstream of the MAPK MAK4 (Ichimura et al. 1998). There are several reports about the involvement of MKK2 in response to cold and salinity stress and apart from this both MKK1 and MKK2 mediate innate immunity responses (Meszaros et al. 2006; Qiu et al. 2008). Group B includes MKK3. One of the distinguishing features of this group is nuclear transfer factor (NTF) domain (Hamel et al. 2006). Steggerda and Paschal (2002) reported that NTF enhances the nuclear import of cargo proteins, which suggests that plant MAP2Ks with NTF domains are involved in cytoplasmic nuclear trafficking. MKK3 has been found to participate in cascades that are elicited by pathogens and are dependent on jasmonic acid (JA) signaling (Doczi et al. 2007; Takahashi et al. 2007). Group C has MKK4 and MKK5 and Group D has rest of the kinases from MKKs 7-10. In general, all plant phyla appear to use a more limited number of MKKs compared to other MAPK components. Rodriguez et al. (2010) have reported a single MKK each for the algae Chlamydomonas and Volvox. This indicates that the same MAP2K may function in several different MAPK modules. Genetic analysis has shown that closely related pairs of plant MAP2Ks have similar functions, e.g., MKK1 and MKK2 are proposed to activate MAPK MPK4 (Qiu et al. 2008), whereas MKK4 and MKK5 act upstream of MPK3 and MPK6, apparently in a redundant manner (Asai et al. 2002). In rice system, MKK genes exhibit differential regulation under different abiotic stresses (Kumar et al. 2008).

3.4 MAPKs

These are also known as MPKs and have been divided into four groups (A–D) (Group et al. 2002). The activation domain of Groups A, B, and C contains TEY (Thu-Glu-Tyr) sequence similar to ERK kinases of animals. The activation domain of Group D contains TDY (Thr-Asn-Tyr) sequence (Rodriguez et al. 2010). MPK3 and MPK6 belong to Group A and have been reported to have a role in developmental processes and also shows responses against biotic and abiotic stresses (Zhang and Klessig 2001; Seo et al. 2007; Sinha et al. 2011). MPK4 belongs to Group B and has a role in pathogen defense and abiotic

stress responses (Andreasson et al. 2005; Brodersen et al. 2006; Qiu et al. 2008). The characteristic feature of Group D MAPKs is a C-terminal docking domain that may act as a docking site for MAP2Ks (Yoo et al. 2008).

3.5 MAPKs and Abiotic Stress Signaling

The mitogen-activated protein kinase (MAPK/ MPKs) cascades transducer environmental cues into intracellular responses. The stimulated plasma membrane receptors activate MAP kinase kinase kinase and then the sequential phosphorylation ensues as MAP3Ks activate downstream MAP kinase kinase that in turn activates MAPKs (Sinha et al. 2011) (Fig. 14.5). MAPKs then target various effector proteins in the cytoplasm or nucleus, which include other kinases, enzymes, or transcription factors (Khokhlatchev et al. 1998; Sinha et al. 2011; Wurzinger et al. 2011).

Abiotic stress is responsible for the activation of MAPK genes and increased MAPK activity. MEKK1, MKK2, MPK4, and/or MPK6 have been activated during salt, drought, and cold stress (Teige et al. 2004; Sinha et al. 2011). Temperature stress (26-38°C) induces 50 kDa MAP kinase in tomatoes. Induction of MPK3 during cold and salinity in Arabidopsis has also been reported by different workers. Ichimura et al. (2000) reported that MPK4 and MPK6 are activated during low-temperature and osmotic stress. The role of several MAP kinases in response to salinity has also been reported by Sanan-Mishra et al. (2006). Matsuoka et al. (2002) demonstrated the role of MAPK kinase (MKK1) in abiotic stress signaling. Multiple abiotic stresses such as wounding, cold, drought, and salinity activated MKK1 which activates downstream MPK4 (Matsuoka et al. 2002). In Arabidopsis a specific MAPKK kinase, ANP1, was activated by H_2O_2 . This ANP1 activates MPK3 and MPK6 and its positive regulator nucleoside diphosphate kinase (NDP) 2 (Moon et al. 2003). Overexpression of ANP1 shows tolerance to heat shock, freezing, and salt stress tin transgenic plants (Kovtun et al. 2000). Plants



Fig. 14.5 The MAP kinase signal transduction pathway (Adopted from Ahmad et al. 2008)

expressing AtNDPK2 show reduction in H₂O₂ accumulation and tolerance to multiple stresses like cold, salt, and oxidative stress. Munnik et al. (1999) demonstrated that a 46 kDa MAP kinase named SIMK activated during moderate hyperosmotic stress in alfalfa. At severe hyperosmotic stress, a smaller kinase gets activated and the activation of SIMK was not observed at severe hyperosmotic stress suggesting that the two kinases function at different stress levels. Mikolajczyk et al. (2000) showed that a salicylic acid-induced protein kinase (SIPK) was activated in tobacco cells under hyperosmotic stress. In tobacco, MAP kinase is activated by multiple stresses like hyperosmotic, hypoosmotic, salicylic acid, and fungal elicitors. A 40 kDa kinase was activated in Arabidopsis by salt stress in a P. Ahmad et al.

calcium and ABA-independent manner (Hoyos and Zhang 2000). MPK4 acts as a negative regulator in plant defense mechanisms. Tang et al. (2005) demonstrated that Arabidopsis EDR1 acts as negative regulator of disease resistance and drought. The edr1 mutant containing a kinasedeficient form of the EDR1 gene exhibits enhanced resistance to pathogens. The *edr1* mutants could also enhance stress responses and spontaneous necrotic lesions under drought conditions (Tang et al. 2005). Accumulation of H₂O₂, superoxide anions, and hydroxyl radicals during abiotic stress causes oxidative burst in cells (Samuel and Ellis 2000). Plants can withstand this oxidative stress by production of antioxidant enzymes like catalase, which decompose H₂O₂ in the cells. Xing et al. (2008) demonstrated that AtMKK1 mediates ABA-induced CAT1 expression in Arabidopsis thaliana. The Arabidopsis mutants mkk1 and mpk6 were altered in their responses to ABA and desiccation stress and the results showed that MKK1–MPK6 regulate H₂O₂ metabolism by CAT1 (Xing et al. 2008). Nakagami et al. (2004) reported that MEKK1-MPK4 cascades are an important part of ROS metabolism. During abiotic stress, two signaling events must occur to induce defense responses in plant cells: one is the inhibition of negative regulators such as EDR1 and the other is activation of positive regulators (Tena et al. 2001). LeMPK3 isolated from Lycopersicon peruvianum is homologous to AtMPK3 and is activated by UV-B radiations (Holley et al. 2003). Mayrose et al. (2004) showed that LeMPK3 is involved in mechanical stress and wounding in tomatoes. CbMAPK3 isolated from Chorispora bungeana shows identity of MPK3 and is activated by cold, salt, and ABA. H₂O₂ activated a novel MAPKKK OMTK1 in alfalfa which activates the MMK3 pathway. In tobacco, the overexpression of NtMEK2 stimulates the gene expression for defense and the generation of reactive oxygen species, which are led by the stimulation of two endogenous MAPKs, SIPK and WIPK (Yang et al. 2001; Ren et al. 2002). In rice system, Rao et al. (2011) showed activation of OsMPK3 and OsMPK4 by arsenic stress. While OsMPK3 was activated only in leaves, both OsMPK3 and MPK4 showed activation in roots. Recently, MPK3 from rice and pea were

reported to function as effector molecules of the stress-regulated beta subunit of pea heterotrimeric G-proteins (Bhardwaj et al. 2011).

 H_2O_2 induces 2MAPK-like activities in *Arabidopsis* and are independent of ethylene and jasmonic acid (Grant et al. 2000). It is not clear that these MAPKs belong to AtMPK3 and AtMPK6. Jonak et al. (1999) have reported that AtMPK3 and AtMPK6 have similarity to wound and SA-induced protein kinases (WIPK and SIPK) of tobacco, respectively. It was also found that only SIPK is activated in tobacco by ozone and H_2O_2 .

4 Nitric Oxide

Nitric oxide (NO) is an important endogenous plant signaling molecule that is responsible for several developmental and physiological processes (Neill et al. 2003, 2008; Tuteja et al. 2004; Delledonne 2005; Lamotte et al. 2005; Erdei and Kolbert 2008; Molassiotis et al. 2010). Extensive work on mammalian system has revealed that NO is a crucial signaling molecule in animals. The physiological functions in plants that are influenced by NO include reduction in seed dormancy (Libourel et al. 2006; Bethke et al. 2007), plant growth regulation and senescence (Mishina et al. 2007), floral transition suppression (He et al. 2004), stomatal movements (Bright et al. 2006; Garcia-Mata and Lamattina 2007), and tolerance to abiotic and biotic stress responses (Uchida et al. 2002; Modolo et al. 2005; Zhao et al. 2007; Floryszak-Wieczorek et al. 2007; Molassiotis et al. 2010). It has also been reported that externally applied NO or NO donors enhance plant tolerance to environmental stresses (Uchida et al. 2002; Garcia-Mata and Lamattina 2007; Zhao et al. 2007a, b).

4.1 Biosynthesis of NO in Plants

In animals, NO is synthesized via a pathway where l-citrulline is formed from l-arginine with the help of nitric oxide synthase (NOS) (Fig. 14.6). The electron donor in this reaction is NADPH and cofactors FMN, FADH, and tetrahydropterin are also used in this reaction. For the biosynthesis of NO in plants, two enzymes NOS and nitrate reductase (NR) are involved (Crawford 2006). There may be other sources for NO biosynthesis apart from these two enzymes (Arnaud et al. 2006). In higher plants, the genes responsible for the upregulation of NOS proteins are yet to be identified. A unique plant NOS (AtNOS1) was isolated from Arabidopsis which encodes a protein associated with NO synthesis (Guo et al. 2003). Overexpression of AtNOS1 increases NO synthesis in E. coli. Reduced root growth and guard cell NO synthesis has been found in the Atnos1 mutant of Arabidopsis plant in response to ABA (Guo et al. 2003). Low amount of NO accumulation in Atnos1 mutants was also reported by many workers (Zhao et al. 2007b; Bright et al. 2006; Zottini et al. 2007). Recently, AtNOS1 was found not to have NOS activity and was not required for the normal synthesis of NO, and it now appears that it is not in fact an NOS at all (Crawford et al. 2006; Zemojtel et al. 2006).



Fig. 14.6 Synthesis of NO from 1-arginine catalyzed by nitric oxide synthase

AtNOS1 in the biosynthesis of NO is either indirect or regulatory and thus AtNOS1 was renamed as *Arabidopsis thaliana* nitric oxideassociated 1 (ATNOA1).

Another enzymatic source of NO generation is NAD(P)H-nitrate reductase (NR) that converts nitrite to nitric oxide (NO) (Yamasaki 2000; Rockel et al. 2002). The preliminary function of NR in plants is nitrogen assimilation, but in an NAD(P)H-dependent reaction NR can also convert nitrite to NO (Neill et al. 2003; Bright et al. 2006; Crawford 2006). Generation of NO by NR activity has been reported in different plants by different workers, e.g., cucumber (Haba et al. 2001), sunflower, maize (Rockel et al. 2002), and wheat (Xu and Zhao 2003). NR is encoded by two genes NIA1 and NIA2 in Arabidopsis. Double mutants (nia1nia2) deficient in NIA genes showed low NR activity and low NO production in guard cells while their stomata do not close in response to ABA or nitrite (Desikan et al. 2002). Modolo et al. (2006) have reported that due to the lack of NR activity, the nia1nia2 double mutants showed reduced levels of l-arginine and the exogenous l-arginine can restore NO generation in this mutant. Reduction in NOS-mediated NO production may be due to reduced levels of arginine. It is still to be identified how cooperation of two pathways of NO generation controls production of NO in plants.

Another source of NO is a plasma membranebound root-specific enzyme, nitrite:NO oxidoreductase (Ni-NOR). The electron donor in this enzyme is not NAD(P)H but cytochrome c and its optimum pH is more acidic than that of NR. The physiological role and genetic identity of this enzyme are not clear yet (Stohr and Stremlau 2006).

4.2 NO Signal Transduction

NO signaling involves cyclic GMP (cGMP)dependent and -independent pathways such as protein nitrosylation. Pfeiffer et al. (1994) demonstrated that NO stimulated cGMP formation in spruce needles. It is also reported that in tobacco cGMP synthesis is required for NO signaling (Durner et al. 1998). NO activates the cGMPdependent pathway leading to adventitious root formation in cucumber (Pagnussat et al. 2003). NO increases the cGMP content and inhibitors of cGMP synthesis through guanylate cyclase. It has also been observed that NO induces PCD in Arabidopsis and cGMP synthesis is also involved there. cGMP was suggested to be the likely target of NO signaling in guard cells. The inhibitor of cGMP synthesis ODQ attenuated ABA and NO-induced stomatal closure. Clarke et al. (2000) reported that the addition of cell-permeable cGMP analogue, 8-bromo cGMP relieved this inhibition. However, 8-bromo cGMP alone did not promote closing of stomata, suggesting that synthesis of cGMP is involved but is insufficient for stomotal closure (Neill et al. 2002). Desikan et al. (2004) reported that stomatal closure by H₂O₂ was not suppressed by ODQ which indicates that H₂O₂ and NO are different signaling pathways in terms of cGMP signaling. Neither guanylate cyclase nor a cGMP-dependent protein kinase has yet been isolated and cloned from plants (Neill et al. 2003). NO activates intracellular Ca2+ channels through cGMP/cADPRdependent signaling pathway. Besson-Bard et al. (2008) demonstrated that calcium and MAP kinase can mediate NO signaling. K/Na ratios in Populus eupharatica have been regulated by NO and H₂O₂ (Zhang et al. 2007). Activation of MAP kinases by NO has been reported in Arabidopsis (Clarke et al. 2000) and tobacco (Kumar and Klessig 2000). Other signaling molecules like H₂O₂ (Samuel et al. 2000) and SA (Kumar and Klessig 2000) have been reported to activate MAPK in tobacco which indicates that MAPK cascades should be a focal point of convergence of both H₂O₂ and NO signaling pathway activated in response to various stresses.

5 Sugar as Signaling Molecule

In plants, the biomolecules are very much essential for plant growth and development. Among biomolecules, sugars serve as energy source and as structural component in plants. Koch (2004) reported that 80% of the CO₂ assimilated during photosynthesis is used for synthesis of sucrose. Sucrose is the major transport form of organic carbon exported from the photosynthetic source to sink organs. Extensive losses in agricultural production have been observed due to environmental stresses (Bohnert et al. 2006; Mittler 2006). Sucrose is performing dual functions, one as transported carbohydrate in vascular part and the other as signaling molecule (Baier et al. 2004). Sugars are responsible for gene expression involved in plant metabolism viz. photosynthesis, glycolysis, N₂ metabolism, cell cycle regulation, etc. Soluble sugars like hormones can act as primary messengers and regulate signals that control the expression of different genes involved in plant growth and metabolism (Rolland et al. 2006; Chen 2007). Gupta and Kaur (2005) have reported that HXK is a sugar sensor in plants. Evolutionary conserved glucose sensor hexokinase 1 (HXK1) mixes different signals (nutrient and hormone signals) and use them for the expression of genes and plant growth in response to environmental stress (Cho et al. 2006). It is not fully understood how the HXK1 controls the gene expression for encoding proteins involved in photosynthesis. Rolland et al. (2006) reported two glucose signal transduction pathways in plants: hexokinase (HXK)-dependent -independent. HXK-dependent system and requires the phosphorylation of glucose while HXK-independent system do not need (Smeekens 2000). Recently, expression of specific photosynthetic genes by HXK1 nuclear complex has been observed and the process is glucose metabolism independent and requires two partners VHAB1 and RPT5B (Chen 2007). This leads to the conclusion that enzymes involved in metabolic activities can play a part in signal transduction by expression of genes in the nucleus. Arabidopsis thaliana mutant hsr8 (high sugar response 8) exhibited increased sugar-responsive growth and expression of genes (Li et al. 2007). Li et al. (2007) observed that hsr8 plants grown under light showed lower chlorophyll content and higher levels of starch and anthocyanin in response to glucose treatment. The hsr8 plant grown under dark showed glucose hypersensitivity. HXK transgenic Arabidopsis plants (AtHXK) have three distinct glucose signal transduction pathways. First is AtHXK1-dependent pathway, where the expression of gene was correlated with the AtHXK1-mediated signaling functions. The second pathway is glycolysis dependent and is regulated by catalytic activity of both AtHXK1 and the heterologous yeast HXK2. The third is AtHXK1-independent pathway, where the expression of gene was independent of AtHXK1 (Xiao et al. 2000). Hence, the role of HXK in sensing the sugar status is still under discussion (reviewed by Rosa et al. 2009).

Kempa et al. (2007) reported that MsK4 (*Medicago sativa* glycogen synthase kinase 3-like kinases) have been involved in stress signaling with carbon metabolism. MsK4 is located in plastids and is associated with starch granules. Kempa et al. (2007) have also reported that kinase activity of MsK4 is induced rapidly under high salt concentration. Overexpression of MsK4 in transgenic plants showed tolerance to salinity stress accompanied with more starch accumulation and modified carbohydrate content (Kempa et al. 2007). The protein kinases KIN10 and KIN11 connect abiotic stress signals, sugar signals, and developmental signals to regulate plant metabolism (Baena-Gonzalez et al. 2007).

Sucrose nonfermenting-1 (SNF-1)-related proteins, analogues of the protein kinase (SNF-1) yeast signaling pathway, have been reported in plants (Loreti et al. 2001). SNF-1-related proteins have a role in sugar sensing (Purcell et al. 1998).

6 Abscisic Acid in Signaling

Ohkuma et al. in 1963 for the first time purified and crystallized abscisic acid (ABA) from cotton fruits and named it as Abscisin II. It was also isolated from sycamore leaves and was named as dormin. Chemical characterization of both revealed identical nature of the two compounds and it was later named as abscisic acid. Apart from being having an inhibitory role in the plant system, the hormone is also known to possess a stress-protective function. ABA plays an important role in plant responses to drought and salt stresses (Tuteja 2007b; Bansal et al. 2011).

ABA has been shown to regulate many agronomical aspects of plant development like synthesis of proteins and lipids, seed desiccation tolerance and dormancy, and germinative, vegetative, and reproductive growth (Leung and Giraudat 1998; Rock 2000; Rohde et al. 2000). In addition, it mediates the responses of plants to many abiotic stresses like drought, salt, cold stress, and biotic stress like pathogen (Leung and Giraudat 1998; Rock 2000; Rohde et al. 2000; Shinozaki and Yamaguchi-Shinozaki 2000). This implies that ABA is involved in both long-term developmental processes as well as short-term physiological effects. Long-term processes involve changes in pattern of gene expression whereas short-term responses involve changes in the activity of various signaling molecules and fluxes of ion channels across the membranes. Both set of responses require the action of signaling elements which amplify the primary signal generated when the hormone binds to its receptors.

Signaling through ABA causes the production and accumulation of second messengers like Ca²⁺, phosphatidic acid (PA), or reactive oxygen species (ROS) in the cell which play an important role in ABA signal transduction (Bansal et al. 2011). Reversible protein phosphorylation is an early and central mechanism that occurs in ABA signal transduction (Leung et al. 1997: Himmelbach et al. 2003; Sokolovski et al. 2005). This mechanism involves several protein kinases and phosphatases (Leung and Giraudat 1998; Finkelstein et al. 2002). For example, ABAactivated serine-threonine protein kinase (AAPK) which is a guard cell-specific protein kinase in Vicia faba or orthologous OPEN STOMATA 1/ SNF1-RELATED PROTEIN **KINASE** 2.6 (OST1/SnRK2.6) which regulates ABAstimulated closure of stomata (Li et al. 2000; Mustilli et al. 2002). Fujii et al. (2007) reported other SnRK2, SnRK2.2, and SnRK2.3 that regulate ABA response in germination, growth, and gene expression. PKABA1 is another protein kinase involved in suppressing the gibberellin (GA)-inducible gene expression in barlet aleurone layers (Gomez-Cadenas et al. 1999). Apart from the above-mentioned calcium-independent protein kinases, the role of several calciumindependent protein kinases has also been revealed in ABA signaling. These either belong to the CDPK or to the SnRK3 family. CDPK1 and CDPK1a have been reported to activate ABA-dependent promoters (Sheen 1996). Also, CDPK3 and CDPK6 possess a role in controlling the aperture of guard cells under abiotic stress (Mori et al. 2006). Zhu et al. (2007) reported the involvement of CPK4 and CPK11 which positively regulate ABA signal transduction in seed germination, seedling growth, and stomatal movements. The mechanism involved is probably the phosphorylation of abscisic acid responsive element-binding factor 1 (ABF1) and abscisic acid responsive element-binding factor 4 (ABF4). Finally, several reports confirm the involvement of mitogen-activated protein kinases (MAPKs) in ABA-dependent response to different stresses (Boudsocq and Laurière 2005; Zhang et al. 2006) and germination (Lu et al. 2002). ABA, in case of stress, regulates gene expression in both positive and negative manner (Chandler and Robertson 1994). Under stress conditions, the gene expression results in the production of transcripts that are responsible for hardening or stress tolerance.

7 Phytohormones as Signaling Molecules

7.1 Brassinosteroids

Brassinosteroids (BRs) are plant steroidal horgrowth-promoting mones having activities (Hacham et al. 2011). Grove et al. (1979) discovered the brassinolide (BL) (the most active form of BR) from the pollens of Brassica napus. BRs play significant role in seed germination, photomorphogenesis, root and stem elongation, vascular differentiation, senescence, flowering, and resistance to biotic and abiotic stresses (Clouse and Sasse 1998; Divi and Krishna 2010; Divi et al. 2010). The biosynthetic pathway of BRs was elucidated through chemical analysis and isolation of additional BR-biosynthetic mutants, defective in genes encoding proteins which catalyze the plant steroid conversion to BR precursors (Asami et al. 2005). First BR biosynthesis inhibitor, brassinazole, is another powerful tool for elucidation of BR signaling pathway (Asami et al. 2000). Genetic, genomic, and proteomic approaches lead to the establishment of BR signaling pathway by providing an important role in the mechanism of receptor activation and regulating components by process of phosphorylation (Tang et al. 2010).

In Arabidopsis, extensive genetic screens for LOF BR signaling mutants are detected in one locus, BRI1 encoding LRR RLK (Clouse et al. 1996; Kauschmann et al. 1996; Li and Chory 1997; Noguchi et al. 1999). Phenotypes of BRI1 mutants are similar as that of BR-deficient mutants, but these are not rescued by the addition of BRs. Components of BR signaling pathway have been characterized in additional suppressor and gain-of-function screens, which involve second LRR RLK, the BRI1-associated receptor kinase-1 (BAK1) (Li et al. 2002); the glycogen synthase kinase-3 (GSK3)-like kinase, BR insensitive-2 (BIN2) (Li et al. 2001b; Li and Nam 2002), the serine/carboxypeptidase BRI1 suppressor-1 (BRS1) (Li et al. 2001a), the phosphatase BRI1 suppressor-1 (BSU1) (Mora-Garcia et al. 2004), and the transcription factors brassinazole-resistant 1 (BZR1) (Wang et al. 2002) and BZR2 (BRI1-EMS suppressor 1 (BES1)) (Yin et al. 2002).

Recently, BR signaling model has been refined by proteomic studies by identifying the components like BR signaling kinases (BSKs), which are not found in previous screens, generating a complete signaling pathway from an RLK to transcription factors in plants (Tang et al. 2008).

BRs regulate the signaling pathway identical to that of classic receptor tyrosine kinases (RTKs) and transform growth factor- β (TGF- β)-mediated signaling in plants (Feng and Derynck 1997; Schlessinger 2000, 2002). In *Arabidopsis*, genome sequence has more than 600 RLK members (Shiu et al. 2004) leading to identical signaling mechanisms. Plant RLKs and signaling pathways provide activation to signaling networks, which are controlled by plant hormones (Smet et al. 2009).

7.2 Ethylene

Ethylene is a gaseous plant hormone, plays significant role in developmental processes like seed germination, senescence, fruit ripening, root nodulation, leaf abscission, programmed cell death, stress, and pathogen attack (Johnson and Ecker 1998; Bleecker and Kende 2000; Binder et al. 2010). Ethylene has "triple response" effect on plant growth of etiolated dicotyledonous seedlings. This response leads to radial swelling of hypocotyl, inhibition of hypocotyls, and root cell elongation and exaggerated curvature of the apical hook. Genetic screens of Arabidopsis are based on the triple-response phenotype. More than dozen of mutants are divided into three distinct categories. Constitutive triple-response mutants, i.e., ethylene-insensitive overproduction (eto1), eto2, eto3, constitutive triple-response1 (ctr1) and responsive to antagonist1 (ran1)/ctr2; ethyleneinsensitive2 (ein2), ein3, ein4, ein6 and tissuespecific ethylene-insensitive mutants, i.e., hookless1 (hls1), ethylene insensitive root1 (eir1) and various auxin-resistant mutants (Johnson and Ecker 1998; Bleecker and Kende 2000; Stepanova and Ecker 2000). Ethylene belongs to the family of membrane-associated receptors, which include ETR1/ETR2, ethylene response Sensor1 (ERS1)/ ERS2 and EIN4 in Arabidopsis (Chang et al. 1993; Hua et al. 1995, 1998; Sakai et al. 1998). Ethylene attaches to its receptor by copper transporter RAN1-delivered copper cofactor. Functions of receptor are inactivated by the hormone binding (Hua and Meyerowitz 1998). EIN2, EIN3, EIN5, and EIN6 act downstream of CTR1 and positively regulate the ethylene response. EIN2 acts as an integral membrane protein, EIN3 acts as transcription factor and expression of intermediate target gene like ethylene response factor1 is regulated.

Ethylene belongs to the family of five receptors (ETR1, ETR2, ETS1, ERS2, and EIN4) and is divided into two subfamilies on the basis of structural similarities. Type-I subfamily contains ETR1 and ERS1 having amino-terminal ethylene-binding domain (which is also known as sensor domain) and carboxy-terminal histidine (His) kinase domain, whereas type-II subfamily receptors contain ETR2, ERS2, and EIN4, which involve an amino-terminal ethylene binding domain and a degenerate His kinase domain. Receptors of ethylene negatively regulates the ethylene responses (Bleecker and Kende 2000; Chang and Stadler 2001). Dominant ethylene in sensitivity mutations in receptor ETR1 leads to signaling (Schaller and Bleecker 1995). LOF mutants have no ethylene response phenotypes. Recently, LOF mutations were isolated in ERS1 gene (Zhao et al. 2002; Wang et al. 2003) with etr1, etr2, ers2, and ein4 mutants. Double LOF etr1ers1 mutants possess strong constitutive-ethylene response phenotypes (Wang et al. 2003). These phenotypes are present in plants containing strong allele of ran1, which cause loss-offunction of all receptors of ethylene (Woeste and Kieber 2000). ETR possesses His kinase activity in vitro, which is important for receptor function (Gamble et al. 1998). For other aspects of receptor functionality like localization, protein stability or interaction with other factors, His kinase activity is essential.

In the mechanism of ethylene signaling, ethylene perception and signaling occurs at endoplasmic reticulum (Chen et al. 2002; Gao et al. 2003). For ER association, the aminoterminal membrane-spanning sensor domain of ETR1 is essential. ER localization of ETR1 is not affected by the introduction of etr1-1 mutations or BR application. CTR is found at ER (Gao et al. 2003). CTR1 contains an aminoterminal domain and carboxy-terminal kinase domain that is linked with Raf-like mitogenactivated protein kinase (MAPK). CTR1 interacts with His kinase domains of ETR1 and ERS1 (Clark et al. 1998). ER-associated CTR1 level inhibits due to removal of ethylene receptors and distribution of CTR1 and receptor interactions. CTR1-ETR1 interaction depends on two lines of evidences in vivo. Co-purification of ETR1 leads to affinity purification of CTR1 from the Arabidopsis ER-membrane fraction, which describes the ETR1 and CTR1 presence in the protein complex (Gao et al. 2003). Overexpression of amino-terminal domain of CTR1 causes LOFctr1 mutant phenotype. Type-I receptors, i.e., ETR1 and ERS1 play

significant role in ethylene signaling. This role is not due to His kinase activity of type-I receptors.

7.3 Jasmonates

Jasmonates (JA) regulate plant growth and development. In the reproductive development of plants, jasmonate signaling plays an important role (Stintzi and Browse 2000; Avanci et al. 2010) by giving protection to plants from abiotic stresses (Traw and Bergelson 2003; Huang et al. 2004; Avanci et al. 2010; Lackman et al. 2011) and from pathogens and insects (Farmer and Ryan 1990; Engelberth et al. 2004; Smith et al. 2009; Ma et al. 2010). In Arabidopsis, three mutants namely *jar1*, *coi1*, and *jin1*, which are defective in JA response and one triple mutant defective in JA biosynthesis (fad3-2fad-72fad8) help in understanding the functioning of JA in plants (Staswick et al. 1992; Feys et al. 1994; Berger et al. 1996; McConn and Browse 1996). Disruption of biosynthetic pathway of JA causes susceptibility of plants to various insects and pathogens (Engelberth et al. 2004; Lewsey et al. 2010); for example, susceptibility of coil to Alternaria brassicicola and Pythium mastophorum (Feys et al. 1994). Oxo-phytodienoic acid, JA-amino acids, and JA-glucosyl are the intermediates of JA biosynthesis which act as the signaling molecule in JA pathway (Staswick et al. 2002).

7.4 Salicylic Acid

Salicylic acid (SA) is a naturally occurring phenolic compound having carboxylic acid group attached to the benzene ring. SA has important role in various aspects of plant development (Hayat et al. 2007, 2010). In mung bean, SA helps in the increase in yield and pod number (Singh and Kaur 1980). It also possesses tuberinducing capacity in potato (Koda et al. 1992). It has positive influence on productivity and nitrogen content in maize (Singh and Srivastava 1978; Asthana and Srivastava 1978), flowering, and helps in reducing transpiration by regulation of
stomata (Khurana and Maheswari 1978; Larque 1979) and alleviation of abiotic stress (Ahmad et al. 2011).

SA signaling has been evaluated in case of plants exposed to abiotic stress. Plant tissues when exposed to abiotic stress release more superoxide anions which further increase the level of hydrogen peroxide (Doke et al. 1994; Ahmad et al. 2010). The increased level of hydrogen peroxide has the ability to stimulate the accumulation of SA (Ahmad et al. 2011). Hence, there is a connection between increase in H₂O₂ level and SA accumulation (Rao et al. 1997). Role of SA has also been described by many workers during cold tolerance in plants like maize, rice, wheat, cucumber, tomato, banana, pea, and mung bean (Janda et al. 1999; yDing et al. 2002; Kang and Saltveit 2002; Kang et al. 2003; Tasgin et al. 2003, Krantev et al. 2009; Popova et al. 2009; Khan et al. 2010). Joseph et al. (2010) have reviewed the exogenous application of SA ant its protective role in different plants under salt stress. Scott et al. (2004) showed the inhibitory effect of SA on the growth of Arabidopsis exposed to chilling conditions. Under low temperatures, the salicylate is reported to accumulate as free and glucosyl SA. Based on studies on various wild species and mutants in Arabidopsis, it was proposed that SA induces low-temperature growth inhibition. Wang and Li (2006) showed increase in cytoplasmic Ca2+ levels after the pretreatment of grape plants with SA. This increased Ca²⁺ helps in maintaining the integrity of plasma membrane during the stress conditions. Also, it was shown that SA-treated plants had higher levels of antioxidants like glutathione and ascorbic acid.

7.5 Auxins

Auxin, the dynamic plant hormone, controls the growth and developmental processes by modulating the levels of auxin/indole acetic acid proteins (Mockaitis and Estelle 2008; Iglesias et al. 2011). Exogenous application of auxin to plants causes alteration in the transcription of gene families, changes in the rate of cell division and cell elongation, range of electrophysiological responses, and changes of tissue pattern and differentiation (Berleth and Sachs 2001; Perrot-Rechenmann 2010). Auxin signaling initiates with the interaction of auxin receptors. Auxin is considered as a multifunctional hormone and the signal is transduced through several signaling pathways. Iglesias et al. (2010) reported that auxin signaling participates in the adaptive response against oxidative stress and salinity in *Arabidopsis*. For wild-type auxin response, a large screen for mutants with changed auxin sensitivity was used to define genes for normal functioning. AXR1, AXR2, AXR3, AXR4, and AXR6 are five different loci and TIR1 is the sixth one (Leyser 2002).

8 Conclusion and Future Perspectives

Plants often experience a variety of changing environmental conditions like light conditions, temperature variations, water and nutrient availability, CO_2 levels, etc. These changes often lead to stress which hampers the plant growth and development. Plants recognize and respond differently to different stresses for their survival. Plants can sense the external stimulus and lead to induction of defense mechanism. How plants cope with these demands is the subject of intensive research and in this review we tried to throw light on different signaling molecules during abiotic stress.

Calcium signaling is involved in the regulation of plant cell cycle progression in response to abiotic stress. Our knowledge regarding Ca^{2+} signaling processes in plants have increased tremendously because of the reverse and forward genetic approaches. Most of this research is focused on Ca^{2+} signal decoders. The versatility of Ca^{2+} ion signaling is amazing. Major contributor to this phenomenon is its unequal distribution which leads to rapid Ca^{2+} oscillations resulting in major concentration changes. Other contributors being highly evolved group of calcium sensors which sense these Ca^{2+} signatures and transduce them to downstream targets for phosphorylation and transcriptional responses. These sensors and transducers together form intricate signaling networks. Ca2+-regulated transcription factors also have an important role in this information processing. CAMTAs are the ones which are most emerging in this field. So, the important parameters which give specificity to this whole mechanism of sensing and decoding of Ca2+ signatures are brought out as the specific Ca²⁺binding affinity, the specific cellular concentration and subcellular localization, and the specific interaction affinities of Ca2+ decoders. Now the important question is how calcium sensor proteins read the codes in calcium signatures and decode them downstream in such a precise manner. In order to answer this there is a need to analyze the structures of these sensors, their structural dynamics, the parameters which lead to association and dissociation of these Ca2+ sensors based on fluorescence resonance energy transfer, and the mathematical modeling approaches. In plants, studies can also be utilized to detect the dynamics of protein interactions and protein complex formation. Additionally, an important but not fully exploited area is the reverse genetics approach for exploring the functioning of Ca²⁺ sensor proteins. Gene knockouts prepared either via crosses or RNAi approaches can be proved beneficial to untangle the functional redundancy issues within and between Ca²⁺ sensor families. The mutant complementation analyses are useful to study specific pathways which may get influenced by disrupting the gene encoding Ca²⁺ sensor. MAPK cascades transduce the environmental and developmental signals into adaptivet and programmed responses. Physiological and developmental processes including stress and hormonal responses, innate immunity, etc. are regulated by MAPK cascades. More research is needed to identify substrates of MAPKs and crosstalk with other signaling molecules.

Chemicals such as NO, ABA, SA, JA, BRs, ethylene, and auxins are reported to respond to various environmental stresses and are reviewed in this chapter. Phytohormones SA, JA, and ET play a central role in plant metabolism. SA is known to play a central role in defense against biotrophic pathogens and systemic acquired resistance. JA plays a central role in defense responses to necrotrophic pathogens and herbivores, whereas ET plays a great role in fruit ripening and senescence and modulates defense responses. Intensive research is needed to know the role of different phytohormones in signaling. The overall progress of research on chemical-regulated stress-responsive genes and their products reflects their central role in plant growth and development under stress conditions.

Acknowledgments This work was partially supported by the Department of Biotechnology (DBT), and the Department of Science and Technology (DST), Government of India.

References

- Ahmad P, Sarwat M, Sharma S (2008) Reactive oxygen species, antioxidants and signaling in plants. J Plant Biol 5:167–173
- Ahmad P, Jaleel CA, Salem MA, Nabi G, Sharma S (2010) Roles of enzymatic and nonenzymatic antioxidants in plants during abiotic stress. Crit Rev Biotechnol 30:161–175
- Ahmad P, Nabi G, Ashraf M (2011) Cadmium-induced oxidative damage in mustard [*Brassica juncea* (L.) Czern. & Coss.] plants can be alleviated by salicylic acid. S Afr J Bot 77:36–44
- Albrecht V, Ritz O, Linder S, Harter K, Kudla J (2001) The NAF domain defines a novel protein-protein interaction module conserved in Ca²⁺-regulated kinases. EMBO J 20:1051–1063
- Albrecht V, Weinl S, Blazevic D, D'Angelo C, Batistič O, et al. (2003) The calcium sensor CBL1 integrates plant responses to abiotic stresses. Plant J 36:457–470
- Allen GJ, Chu SP, Harrington CL, Schumacher K, Hoffman T, Tang YY, Grill E, Schroeder JI (2001) A defined range of guard cell calcium oscillation parameters encodes stomatal movements. Nature 411:1053–1057
- Andreasson E, Jenkins T, Brodersen P, Thorgrimsen S, Petersen NH et al (2005) The MAP kinase substrate MKS1 is a regulator of plant defense responses. EMBO J 24:2579–2589
- Arnaud N, Murgia I, Boucherez J, Briat JF, Cellier F, Gaymard F (2006) An iron-induced nitric oxide burst precedes ubiquitin-dependent protein degradation for *Arabidopsis* AtFer1 ferritin gene expression. J Biol Chem 281:23579–23588
- Asai T, Tena G, Plotnikova J, Willmann MR, Chiu WL et al (2002) MAP kinase signaling cascade in *Arabidopsis* innate immunity. Nature 415: 977–983
- Asami T, Min YK, Nagata N, Yamagishi K, Takatsuto S, Fujioka S, Murofushi N, Yamaguchi I, Yoshida S

(2000) Characterization of Brassinazole, a triazoletype brassinosteriod biosynthesis inhibitor. Plant Physiol 123:93–99

- Asami T, Nakano T, Fujioka S (2005) Plant brassinosteroid hormones. Vitam Horm 72:479–504
- Asano T, Hakata M, Nakamura H, Aoki N, Komatsu S, Ichikawa H, Hirochika H, OhsugI R (2011) Functional characterisation of OsCPK21, a calcium-dependent protein kinase that confers salt tolerance in rice. Plant Mol Biol 75:179–191
- Asthana JS, Srivastava HS (1978) Effect of presowing treatment of maize seeds with salicylic acid and ascorbic acid on seedling and nitrogen content. Ind J Plant Physiol 21:150–155
- Avanci NC, Luche DD, Goldman GH, Goldman MHS (2010) Jasmonates are phytohormones with multiple functions, including plant defense and reproduction. Genet Mol Res 9:484–505
- Baena-Gonzalez E, Rolland F, Thevelein JM, Sheen J (2007) A central integrator of transcription network in plant stress and energy signaling. Nature 448:938–942
- Baier M, Hemmann G, Holman R, Corke F, Card R, Smith C, Rook F, Bevan MW (2004) Characterization of mutants in *Arabidopsis* showing increased sugar-specific gene expression, growth, and developmental responses. Plant Physiol 134:81–91
- Bansal KC, Lenka SK, Tuteja N (2011) Abscisic acid in abiotic stress tolerance: an 'omics' approach. In: Tuteja N, Gill SS, Tuteja R (eds) Omics and plant abiotic stress tolerance. Bentham Science, Sharjah, pp 143–150
- Batistic O, Kudla J (2009) Plant calcineurin B-like proteins and their interacting protein kinases. Biochim Biophys Acta 1793:985–992
- Batistic O, Sorek N, Schultke S, Yalovsky S, Kudla J (2008) Dual fatty acyl modification determines the localization and plasma membrane targeting of CBL/ CIPK Ca²⁺ signaling complexes in *Arabidopsis*. Plant Cell 20:1346–1362
- Berger S, Bell E, Mullet JE (1996) Two methyl jasmonateinsensitive mutants show altered expression of *AtVsp* in response to methyl jasmonate and wounding. Plant Physiol 111:525–531
- Berleth T, Sachs T (2001) Plant morphogenesis: longdistance coordination and local patterning. Curr Opin Plant Biol 4:57–62
- Besson-Bard A, Courtois C, Gauthier A, Dahan J, Dobrowolska G, Jeandroz S, Pugin A, Wendenhenne D (2008) Nitric oxide in plants: production and cross talk with Ca²⁺ signaling. Mol Plant 1:218–228
- Bethke PC, Libourel IG, Aoyama N, Chung YY, Still DW, Jones RL (2007) The Arabidopsis aleurone layer responds to nitric oxide, gibberellin, and abscisic acid and is sufficient and necessary for seed dormancy. Plant Physiol 143:1173–1788
- Bhardwaj D, Sheikh AH, Sinha AK, Tuteja N (2011) Stress induced beta subunit of heterotrimeric G-proteins from *Pisum sativum* interacts with mitogen activated protein kinase. Plant Signal Behav 6(2):287–292

- Binder BM, Rodríguez FI, Bleecker AB (2010) The copper transporter RAN1 is essential for biogenesis of ethylene receptors in *Arabidopsis*. J Biol Chem 285:37263–37270
- Bleecker AB, Kende H (2000) Ethylene: a gaseous signal molecule in plants. Annu Rev Cell Dev Biol 16:1–18
- Bohnert HJ, Gong Q, Li P, Ma S (2006) Unraveling abiotic stress tolerance mechanisms-getting genomics going. Curr Opin Plant Biol 9:180–188
- Bouche N, Scharlat A, Snedden W, Bouchez D, Fromm H (2002) A novel family of calmodulin-binding transcription activators in multicellular organisms. J Biol Chem 277:21851–21861
- Boudsocq M, Laurière C (2005) Osmotic signaling in plants. Multiple pathways mediated by emerging kinase families. Plant Physiol 138:1185–1194
- Braam J (2005) In touch: plant responses to mechanical stimuli. New Phytol 165:373–389
- Bright J, Desikan R, Hancock JT, Weir IS, Neill SJ (2006) ABA-induced NO generation and stomatal closure in *Arabidopsis* are dependent on H2O2 synthesis. Plant J 45:113–122
- Brodersen P, Petersen M, Bjorn Nielsen H, Zhu S, Newman MA et al (2006) Arabidopsis MAP kinase 4 regulates salicylic acid- and jasmonic acid/ethylene-dependent responses via EDS1 and PAD4. Plant J 47:532–546
- Chandler PM, Robertson M (1994) Gene expression regulated by abscisic acid and its relation to stress tolerance. Annu Rev Plant Physiol Plant Mol Biol 45:113–141
- Chang C, Stadler R (2001) Ethylene hormone receptor action in Arabidopsis. Bioessays 23:619–627
- Chang C, Kwok SF, Bleecker AB, Meyerowitz EM (1993) *Arabidopsis* ethylene-response gene ETR1: similarity of product to two component regulators. Science 262:539–544
- Chen JG (2007) Sweet sensor, surprising partners. Sci STKE 373:7
- Chen YF, Randlett MD, Findell JL, Schaller GE (2002) Localization of the ethylene receptor ETR1 to the endoplasmic reticulum of *Arabidopsis*. J Biol Chem 277:19861–19866
- Cheong YH, Kim KN, Pandey GK, Gupta R, Grant JJ, Luan S (2003) CBL1, a calcium sensor that differentially regulates salt, drought, and cold responses in *Arabidopsis*. Plant Cell 15:1833–1845
- Cheong YH, Pandey GK, Grant JJ, Batistic O, Li L et al (2007) Two calcineurin B-like calcium sensors, interacting with protein kinase CIPK23, regulate leaf transpiration and root potassium uptake in Arabidopsis. Plant J 52:223–239
- Cho YH, Yoo SD, Sheen J (2006) Regulatory functions of nuclear hexokinase1 complex in glucose signaling. Cell 127:579–589
- Clapham DE (2007) Calcium signaling. Cell 131: 1047–1058
- Clark KL, Larsen PB, Wang X, Chang C (1998) Association of the Arabidopsis CTR1 Raf-like kinase with the ETR1 and ERS ethylene receptors. Proc Natl Acad Sci USA 95:5401–5406

- Clarke A, Desikan R, Hurst RD, Hancock JT, Neill SJ (2000) NO way back: nitric oxide and programmed cell death in *Arabidopsis thaliana* suspension cultures. Plant J 24:667–677
- Clouse SD, Sasse JM (1998) Brassinosteriods: essential regulators of plant growth and development. Ann Rev Plant Physiol Plant Mol Biol 49:427–451
- Clouse SD, Langford M, McMorris TC (1996) A brassinosteroid-insensitive mutant in Arabidopsis thaliana exhibits multiple defects in growth and development. Plant Physiol 111:671–678
- Cobb MH, Hepler JE, Cheng M, Robbins D (1994) The mitogen-activated protein kinases, ERK1 and ERK2. Semin Cancer Biol 5:261–268
- Crawford NM (2006) Mechanisms for nitric oxide synthesis in plants. J Exp Bot 57:471–478
- Crawford NM, Galli M, Tischner R, Heimer YM, Okamoto M, Mack A (2006) Response to Zemojtel et al: plant nitric oxide synthase: back to square one. Trend Plant Sci 11:526–527
- D' Angelo C, Weinl S, Batistic O, Pandey GK, Cheong YH et al (2006) Alternative complex formation of the Ca²⁺-regulated protein kinase CIPK1 controls abscisic acid-dependent and independent stress responses in Arabidopsis. Plant J 48:857–872
- Davis RJ (1994) MAPKs: new JNK expands the group. Trends Biochem Sci 19:470–473
- Day I, Reddy V, Ali GS, Reddy ASN (2002) Analysis of EF-hand-containing proteins in Arabidopsis. Genome Biol 3(10):RESEARCH0056
- Delledonne M (2005) NO news is good news for plants. Curr Opin Plant Biol 8:390–396
- Desikan R, Griffiths R, Hancock J, Neill S (2002) A new role for an old enzyme: nitrate reductase-mediated nitric oxide generation is required for abscisic acidinduced stomatal closure in *Arabidopsis thaliana*. Proc Natl Acad Sci USA 99:16314–16318
- Desikan R, Cheung MK, Bright J, Henson D, Hancock JT, Neill SJ (2004) ABA, hydrogen peroxide and nitric oxide signalling in stomatal guard cells. J Exp Bot 55:205–212
- Ding CK, Wang CY, Gross KC, Smith DL (2002) Jasmonate and salicylate induce the expression of pathogenesisrelated-protein genes and increase resistance to chilling injury in tomato fruit. Planta 214:895–901
- Divi U, Krishna P (2010) Overexpression of the brassinosteroid biosynthetic gene AtDWF4 in Arabidopsis seeds overcomes abscisic acid-induced Inhibition of germination and increases cold tolerance in transgenic seedlings. J Plant Growth Regul 29:385–393
- Divi U, Rahman T, Krishna P (2010) Brassinosteroidmediated abiotic stress tolerance in Arabidopsis shows interactions with abscisic acid, ethylene and salicylic acid pathways. BMC Plant Biol 10:151
- Doczi R, Brader G, Pettko-Szandtner A, Rajh I, Djamei A et al (2007) The *Arabidopsis* mitogen-activated protein kinase kinase MKK3 is upstream of group C mitogenactivated protein kinases and participates in pathogen signaling. Plant Cell 19:3266–3279

- Doke N, Miura Y, Leandro MS, Kawakita K (1994) Involvement of superoxide in signal transduction: responses to attack by pathogens, physical and chemical shocks, and UV irradiation. In: Foyer CH, Mullineaux PM (eds) Causes of photo oxidative stress and amelioration of defense systems in plants. CRC, Boca Raton, FL, pp 177–179
- Duerr B, Gawienowski M, Ropp T, Jacobs T (1993) MsERK1: a mitogen-activated protein kinase from a flowering plant. Plant Cell 5:87–96
- Durner J, Wendehenne D, Klessig DF (1998) Defense gene induction in tobacco by nitric oxide, cyclic GMP and cyclic ADP-ribose. Proc Natl Acad Sci USA 95:10328–10333
- Engelberth J, Alborn HT, Schmeize EA, Tumlinson JH (2004) Airborne signals prime plants against insect herbivore attack. Proc Natl Acad Sci USA 101: 1781–1785
- Erdei L, Kolbert Z (2008) Nitric oxide as a potent signalling molecule in plants. Acta Biol Szeged 52:1–58
- Farmer EE, Ryan CA (1990) Interplant communication: airborne methyl jasmonate induces synthesis of proteinase inhibitors in plant leaves. Proc Natl Acad Sci USA 87:7713–7716
- Feng XH, Derynck R (1997) A kinase subdomain of transforming growth factor- β (TGF- β) type I receptor determines the TGF- β intracellular signaling specificity. EMBO J 16:3912–3923
- Feys BJF, Benedetti CE, Penfold CN, Turner JG (1994) Arabidopsis mutants selected for resistance to the phytotoxin coronative are male sterile, insensitive to methyl jasmonate, and resistant to bacterial pathogen. Plant Cell 6:751–759
- Finkelstein RR, Gampala SS, Rock CD (2002) Abscisic acid in seeds and seedlings. Plant Cell 14:S15–S45
- Floryszak-Wieczorek J, Arasimowicz M, Milczarek G, Jelen H, Jackowiak H (2007) Only an early nitric oxide burst and the following wave of secondary nitric oxide generation enhanced effective defense responses of pelargonium to a necrotrophic pathogen. New Phytol 175:718–730
- Frandsen GI, Müller-Uri F, Nielsen M, Mundy J, Skriver K (1996) Novel plant Ca²⁺-binding protein expressed in response to abscisic acid and osmotic stress. J Biol Chem 271:343–348
- Franza S, Ehlerta B, Liesea A, Kurtha J, Cazaléb AC, Romeisa T (2011) Calcium-Dependent Protein Kinase CPK21 Functions in Abiotic Stress Response in *Arabidopsis thaliana*. Mol Plant 4:83–96
- Frye CA, Tang D, Innes RW (2001) Negative regulation of defense responses in plants by a conserved MAPKK kinase. Proc Natl Acad Sci USA 98:373–378
- Fujii H, Verslues PE, Zhu JK (2007) Identification of two protein kinases required for abscisic acid regulation of seed germination, root growth, and gene expression in *Arabidopsis*. Plant Cell 19:485–494
- Gamble RL, Coonfield ML, Schaller GE (1998) Histidine kinase activity of the ETR1 ethylene receptor from *Arabidopsis*. Proc Natl Acad Sci USA 95:7825–7829

- Gao Z, Chen YF, Randlett MD, Zhao XC, Findell JL, Kieber JJ, Schaller GE (2003) Localization of the Raf-like kinase CTR1 to the endoplasmic reticulum of Arabidopsis through participation in ethylene receptor signaling complexes. J Biol Chem 278:34725–34732
- Garcia-Mata C, Lamattina L (2007) Abscisic acid (ABA) inhibits light-induced stomatal opening through calcium- and nitric oxide-mediated signaling pathways. Nitric Oxide 17:143–151
- Gifford JL, Walsh MP, Vogel HJ (2007) Structure and metal-ion-binding properties of the Ca²⁺ binding helixloop-helix EF-hand motifs. Biochem J 405:199–221
- Gomez-Cadenas A, Verhey SD, Holappa LD, Shen Q, Ho THD, Walker-Simmons MK (1999) An abscisic acid induced protein kinase, PKABA1, mediates abscisic acid-suppressed gene expression in barley aleurone. Proc Natl Acad Sci USA 96:1767–1772
- Gorecka KM, Thouverey C, Buchet R, Pikula S (2007a) Potential role of annexin AnnAt1 from Arabidopsis thaliana in pH-mediated cellular response to environmental stimuli. Plant Cell Physiol 48:792–803
- Gorecka KM, Trebacz K, Górecki R, Pikula S (2007b) Participation of annexin At1 in plant response to abiotic stress. Postepy Biochem (Polish) 53:154–158
- Gotoh Y, Nishida E, Yamashita T, Hoshi M, Kawakami M, Sakai H (1990) Microtubule-associated-protein (MAP) kinase activated by nerve growth factor and epidermal growth factor in PC12 cells. Eur J Biochem 193:661–669
- Grant JJ, Yun BW, Loake GJ (2000) Oxidative burst and cognate redox signalling reported by luciferase imaging: identification of a signal network that functions independently of ethylene, SA and Me-JA but is dependent on MAPKK activity. Plant J 24:569–582
- Group M, Ichimura K, Shinozaki K, Tena G, Sheen J et al (2002) Mitogen-activated protein kinase cascades in plants: a new nomenclature. Trends Plant Sci 7: 301–308
- Grove MD, Spencer GF, Rohwedder WK, Mandava N, Worley JF, Warthen JD, Steens G, Flippen-Anderson JL, Cook JC (1979) Brassinolide, a plant growth promoting steroid isolated from *Brassica napus* pollen. Nature 281:216–217
- Guo Y, Halfter U, Ishitani M, Zhu JK (2001) Molecular characterization of functional domains in the protein kinase SOS2 that is required for plant salt tolerance. Plant Cell 13:1383–1399
- Guo F, Okamoto M, Crawford NM (2003) Identification of a plant nitric oxide synthase gene involved in hormonal signaling. Science 302:100–103
- Gupta AK, Kaur N (2005) Sugar signalling and gene expression in relation to carbohydrate metabolism under abiotic stresses in plants. J Biosci 30:761–776
- Haba P, Agüera E, Benitez L, Maldonado JM (2001) Modulation of nitrate reductase activity in cucumber (*Cucumis sativus*) roots. Plant Sci 161:231–237
- Hacham Y, Holland N, Butterfield C, Ubeda-Tomas S, Bennett MJ, Chory J, Savaldi-Goldstein S (2011)

Brassinosteroid perception in the epidermis controls root meristem size. Development 138:839–848

- Halfter U, Ishitani M, Zhu JK (2000) The Arabidopsis SOS2 protein kinase physically interacts with and is activated by the calcium-binding protein SOS3. Proc Natl Acad Sci USA 97:3735–3740
- Hamel LP, Nicole MC, Sritubtim S, Morency MJ, Ellis M et al (2006) Ancient signals: comparative genomics of plant MAPK and MAPKK gene families. Trends Plant Sci 11:192–198
- Hayat S, Ali B, Ahmad A (2007) Salicylic acid: biosynthesis, metabolism and physiological role in plants. In: Hayat S, Ahmad A (eds) Salicylic acid, a plant hormone. Springer, Drodrecht, pp 1–14
- Hayat Q, Hayat S, Irfan M, Ahmad A (2010) Effect of exogenous salicylic acid under changing environment: a review. Environ Exp Bot 68:14–25
- He YK, Tang RH, Yi H, Stevens RD, Cook CW, Ahn SM et al (2004) Nitric oxide represses the *Arabidopsis* floral transition. Science 305:1968–1971
- Himmelbach A, Yang Y, Grill E (2003) Relay and control of abscisic acid signaling. Curr Opin Plant Biol 6:470–479
- Hirt H (2000) Results and problems in cell differentiation: MAP kinases in Plant Signal Transduction. Springer, Heidelberg
- Holley SR, Yalamanchili RD, Moura DS, Ryan CA, Stratmann JW (2003) Convergence of signaling pathways induced by systemin, oligosaccharide elicitors, and ultraviolet-B radiation at the level of mitogen activated protein kinases in *Lycopersicon peruvianum* suspension cultured cells. Plant Physiol 132:1728–1738
- Horie T, Yoshida K, Nakayama H, Yamada K, Oiki S, Shinmyo A (2001) Two types of HKT transporters with different properties of Na⁺ and K⁺ transport in *Oryza sativa*. Plant J 27:129–138
- Hoyos ME, Zhang S (2000) Calciumindependent activation of salicylic acid-induced protein kinase and a 40-kilodalton protein kinase by hyperosmotic stress. Plant Physiol 122:1355–1363
- Hua J, Meyerowitz EM (1998) Ethylene responses are negatively regulated by a receptor gene family in *Arabidopsis thaliana*. Cell 94:261–271
- Hua J, Chang C, Sun Q, Meyerowitz EM (1995) Ethylene insensitivity conferred by *Arabidopsis* ERS gene. Science 269:1712–1714
- Hua J, Sakai H, Nourizadeh S, Chen QG, Bleecker AB, Ecker JR, Meyerowitz EM (1998) EIN4 and ERS2 are members of the putative ethylene receptor gene family in *Arabidopsis*. Plant Cell 10:1321–1332
- Huang Y, Li H, Hutchison CE, Laskey J, Kieber JJ (2003) Biochemical and functional analysis of CTR1, a protein kinase that negatively regulates ethylene signaling in *Arabidopsis*. Plant J 33:221–233
- Huang X, Stettmaier K, Michel C, Hutzier P, Mueller MJ, Durner J (2004) Nitric oxide is induced by wounding and influences jasmonic acid signaling in *Arabidopsis thaliana*. Planta 218:938–946

- Ichimura K, Mizoguchi T, Irie K, Morris P, Giraudat J et al (1998) Isolation of ATMEKK1 (a MAP kinase kinase kinase)-interacting proteins and analysis of a MAP kinase cascade in *Arabidopsis*. Biochem Biophys Res Commun 253:532–543
- Ichimura K, Mizoguchi T, Yoshida R, Yuasa T, Shinozaki K (2000) Various abiotic stresses rapidly activate Arabidopsis MAP kinases ATMPK4 and ATMPK6. Plant J 24:655–665
- Iglesias MJ, Terrile MC, Bartoli CG, D'Ippólito S, Casalongué CA (2010) Auxin signaling participates in the adaptative response against oxidative stress and salinity by interacting with redox metabolism in Arabidopsis. Plant Mol Biol 74:215–222
- Iglesias MJ, Terrile MC, Casalongué CA (2011) Auxin and salicylic acid signalings counteract the regulation of adaptive responses to stress. Plant Signal Behav 6(3):452–454
- Ishitani M, Liu J, Halfter U, Kim CS, Shi W, Zhu JK (2000) SOS3 function in plant salt tolerance requires N-myristoylation and calcium-binding. Plant Cell 12:1667–1677
- Janda T, Szalai G, Tari I, Paldi E (1999) Hydroponic treatment with salicylic acid decreases the effects of chilling injury in maize (*Zea mays L.*) plants. Planta 208:175–180
- Jang HJ, Pih KT, Kang SG, Lim JH, Jin JB, Piao HL, Hwang I (1998) Molecular cloning of a novel Ca 2+ binding protein that is induced by NaCl stress. Plant Mol Biol 37:839–884
- Johnson PR, Ecker JR (1998) The ethylene gas signal transduction pathway: a molecular perspective. Annu Rev Genet 32:227–254
- Jonak C, Heberle-Bors E, Hirt H (1994) MAP kinases: universal multi-purpose signaling tools. Plant Mol Biol 24:407–416
- Jonak C, Ligterink W, Hirt H (1999) MAP kinases in plant signal transduction. Cell Mol Life Sci 55:204–213
- Jonak C, Ökrész L, Bögre L, Hirt H (2002) Complexity, cross talk and integration of plant MAP kinase signalling. Curr Opin Plant Biol 5:415–424
- Joseph B, Jini D, Sujatha S (2010) Insight into the role of exogenous salicylic acid on plants grown under salt environment. Asian J Crop Sci 2:226–235
- Kader MA, Lindberg S (2010) Cytosolic calcium and pH signaling in plants under salinity stress. Plant Signal Behav 5:233–238
- Kang HM, Saltveit ME (2002) Chilling tolerance of maize, cucumber and rice seedling leaves and roots are differentially affected by salicylic acid. Physiol Plant 115:571–576
- Kang GZ, Wang ZX, Sun GC (2003) Participation of H_2O_2 in enhancement of cold chilling by salicylic acid in banana seedlings. Acta Bot Sin 45:567–573
- Kaur N, Gupta AK (2005) Signal transduction pathways under abiotic stresses in plants. Curr Sci 88:1771–1780
- Kauschmann A, Jessop A, Koncz C, Szekeres M, Willmitzer L, Altmann T (1996) Genetic evidence for an essential role of brassinosteroids in plant development. Plant J 9:701–713

- Kawasaki S, Borchert C, Deyholos M, Wang H, Brazille S, Kawai K, Galbraith D, Bohnert H (2001) Gene expression profiles during the initial phase of salt stress in rice. Plant Cell 13:889–905
- Kempa S, Rozhon W, Samaj J, Erban A, Baluska F, Becker T, Haselmayer J, Schleiff E, Kopka J, Hirt H, Jonak C (2007) A plastid-localized glycogen synthase kinase 3 modulates stress tolerance and carbohydrate metabolism. Plant J 49:1076–1090
- Khan NA, Syeed S, Masood A, Nazar R, Iqbal N (2010) Application of salicylic acid increases contents of nutrients and antioxidative metabolism in mungbean and alleviates adverse effects of salinity stress. Inter J Plant Biol 1:e1. doi:10.4081/pb.2010.e1
- Khokhlatchev AV, Canagarajah B, Wilsbacher J, Robinson M, Atkinson M et al (1998) Phosphorylation of the MAP kinase ERK2 promotes its homodimerization and nuclear translocation. Cell 93:605–615
- Khurana JP, Maheswari SC (1978) Induction of flowering in *Lemna paucicostata* by salicylic acid. Plant Sci Lett 12:127–132
- Kieber JJ, Rothenberg M, Roman G, Feldmann KA, Ecker JR (1993) CTR1, a negative regulator of the ethylene response pathway in Arabidopsis, encodes a member of the raf family of protein kinase. Cell 72:427–441
- Kiegle E, Moore CA, Haseloff J, Tester MA, Knight MR (2000) Cell-type-specific calcium responses to drought, salt and cold in the *Arabidopsis* root. Plant J 23:267–278
- Kim KN, Cheong YH, Grant JJ, Pandey GK, Luan S (2003) CIPK3, a calcium sensor-associated protein kinase that regulates abscisic acid and cold signal transduction in Arabidopsis. Plant Cell 15:411–423
- Kim BG, Waadt R, Cheong YH, Pandey GK, Dominguez-Solis JR et al (2007) The calcium sensor CBL10 mediates salt tolerance by regulating ion homeostasis in Arabidopsis. Plant J 52:473–484
- Klimecka M, Muszynska G (2007) Structure and functions of plant calcium dependent protein kinases. Acta Biochim Pol 54:219–233
- Knight H (2000) Calcium signaling during abiotic stress in plants. Int Rev Cytol Surv Cell Biol 195:269–324
- Knight H, Trewavas AJ, Knight MR (1997) Calcium signalling in *Arabidopsis thaliana* responding to drought and salinity. Plant J 12:1067–1078
- Koch K (2004) Sucrose metabolism: regulatory mechanisms and pivotal roles in sugar sensing and plant development. Curr Opin Plant Biol 7:235–246
- Koda V, Koshi T, Kikuta Y (1992) Potato tuber inducing activities of salicylic acid and related compounds. Plant Growth Regul 11:215–219
- Kolukisaoglu U, Weinl S, Blazevic D, Batistic O, Kudla J (2004) Calcium sensors and their interacting protein kinases: genomics of the Arabidopsis and rice CBL-CIPK signaling networks. Plant Physiol 134:43–58
- Kovtun Y, Chiu W, Tena G, Sheen J (2000) Functional analysis of oxidative stress-activated mitogen-activated protein kinase cascade in plants. Proc Natl Acad Sci USA 97:2940–2945

- Krantev AP, Szalai G, Janda T (2009) Exogenous treatment with salicylic acid attenuates cadmium toxicity in pea seedlings. Plant Physiol Biochem 47:224–231
- Kudla J, Xu Q, Harter K, Gruissem W, Luan S (1999) Genes for calcineurin B-like proteins in Arabidopsis are differentially regulated by stress signals. Proc Natl Acad Sci USA 96:4718–4723
- Kumar D, Klessig DF (2000) Differential induction of tobacco MAP kinases by the defense signals nitric oxide, salicylic acid, ethylene and jasmonic acid. Mol Plant Microbe Interact 13:347–351
- Kumar K, Rao KP, Sharma P, Sinha AK (2008) Differential regulation of rice mitogen activated protein kinase kinase (MKK) by abiotic stress. Plant Physiol Biochem 46:891–897
- Lackman P, González-Guzmán M, Tilleman S, Carqueijeiro I, Cuéllar Pérez A, Moses T, Seo M, Kanno Y, Häkkinen ST, Van Montagu MCE, Thevelein JM, Maaheimo H, Oksman-Caldentey KM, Rodriguez PL, Rischer H, Goossens A (2011) Jasmonate signaling involves the abscisic acid receptor PYL4 to regulate metabolic reprogramming in*Arabidopsis* and tobacco. Proc Natl Acad Sci USA. doi:10.1073/pnas.1103010108
- Lamotte L, Courtois C, Barnavon L, Pugin A, Wendehenne D (2005) Nitric oxide in plants: the biosynthesis and cell signalling properties of a fascinating molecule. Planta 221:1–4
- Landry J, Huot J (1995) Modulation of actin dynamics during stress and physiological stimulation by a signaling pathway involving p38 MAP kinase and heatshock protein 27. Biochem Cell Biol 73:703–707
- Larque SA (1979) Stomatal closure in response to acetyl salicylic acid treatments. Pflanzen Physiol 93:371–375
- Leung J, Giraudat J (1998) Abscisic acid signal transduction. Annu Rev Plant Physiol Plant Mol Biol 49:199–222
- Leung J, Merlot S, Giraudat J (1997) The Arabidopsis ABSCISIC ACID-INSENSITIVE2 (ABI2) and ABI1 genes encode homologous protein phosphatases 2C involved in abscisic acid signal transduction. Plant Cell 9:759–771
- Lewsey MG, Murphy AM, MacLean D, Dalchau N, Westwood JH, Macaulay K, Bennett MH, Moulin M, Hanke DE, Powell G, Smith AG, Carr JP (2010) Disruption of two defensive signaling pathways by a viral RNA silencing suppressor. Mol Plant Microbe Interact 23:835–845
- Leyser O (2002) Molecular genetics of auxin signaling. Annu Rev Plant Biol 53:377–398
- Li J, Chory J (1997) A putative leucine-rich repeat receptor kinase involved in brassinosteroid signal transduction. Cell 90:929–938
- Li J, Nam KH (2002) Regulation of brassinosteroid signaling by a GSK3/SHAGGY-like kinase. Science 295:1299–1301
- Li J, Wang XQ, Watson MB, Assmann SM (2000) Regulation of abscisic acid-induced stomatal closure and anion channels by guard cells AAPK kinase. Science 287:300–303

- Li J, Nam KH, Vafeados D, Chory J (2001a) BIN2, a new brassinosteroid-insensitive locus in *Arabidopsis*. Plant Physiol 127:14–22
- Li J, Lease KA, Tax FE, Walker JC (2001b) BRS1, a serine carboxypeptidase, regulates BRI1 signaling in *Arabidopsis thaliana*. Proc Natl Acad Sci USA 98:5916–5921
- Li J, Wen J, Lease KA, Doke JT, Tax FE, Walker JC (2002) BAK1, an *Arabidopsis* LRR receptor-like protein kinase, interacts with BRI1 and modulates brassinosteroid signaling. Cell 110:213–222
- Li Y, Smith C, Corke F, Zheng L, Merali Z, Ryden P, Derbyshire P, Waldron K, Bevan MW (2007) Signaling from an Altered Cell Wall to the Nucleus Mediates Sugar-Responsive Growth and Development in *Arabidopsis thaliana*. Plant Cell 19:2500–2515
- Libourel IG, Bethke PC, De Michele R, Jones RL (2006) Nitric oxide gas stimulates germination of dormant *Arabidopsis* seeds: use of a flow-through apparatus for delivery of nitric oxide. Planta 223:813–820
- Ligterink W, Hirt H (2001) Mitogen-activated protein (MAP) kinase pathways in plants: versatile signaling tools. Int Rev Cytol 201:209–275
- Liu J, Zhu JK (1998) A calcium sensor homolog required for plant salt tolerance. Science 280:1943–1945
- Liu W, Fairbairn DJ, Reid RJ, Schachtman DP (2001) Characterization of two *HKT1* homologues from *Eucalyptus camaldulensis* that display intrinsic osmosensing capability. Plant Physiol 127:283–294
- Loreti E, de Bellis L, Alpi A, Perata P (2001) Why and how do plant cells sense sugars? Ann Bot 88:803–812
- Lu C, Han MH, Guevara-Garcia A, Fedoroff NV (2002) Mitogen-activated protein kinase signaling in postgermination arrest of development by abscisic acid. Proc Natl Acad Sci USA 99:15812–15817
- Lukowitz W, Roeder A, Parmenter D, Somerville C (2004) A MAPKK kinase gene regulates extra-embryonic cell fate in Arabidopsis. Cell 116:109–119
- Ma QH, Tian B, Li YL (2010) Overexpression of a wheat jasmonate-regulated lectin increases pathogen resistance. Biochimie 92:187–193
- Mahajan S, Tuteja N (2005) Cold, salinity and drought stresses: an overview. Arch Biochem Biophys 444: 139–158
- Mahajan S, Sopory SK, Tuteja N (2006a) CBL-CIPK paradigm: role in calcium and stress signalling in plants. Proc Indian Natl Sci Acad 72:63–78
- Mahajan S, Sopory SK, Tuteja N (2006b) Cloning and characterization of CBL-CIPK signalling components from a legume (*Pisum sativum*). FEBS J 273: 907–925
- Mahajan S, Pandey GK, Tuteja N (2008) Calcium- and salt-stress signaling in plants: shedding light on SOS pathway. Arch Biochem Biophy 471:146–158
- Martin ML, Busconi L (2001) A rice membrane-bound calcium-dependent protein kinase is activated in response to low temperature. Plant Physiol 125:1442–1449
- Matsuoka D, Nanmori T, Sato K, Fukami Y, Kikkawa U, Yasuda T (2002) Activation of AtMEK1, an

Arabidopsis mitogen-activated protein kinase kinase, in vitro and in vivo: analysis of active mutants expressed in *E. coli* and generation of the active form in stress response in seedlings. Plant J 29:637–647

- Mayrose M, Bonshtien A, Sessa G (2004) LeMPK3 is a mitogen-activated protein kinase with dual specificity induced during tomato defense and wounding responses. J Biol Chem 279:14819–14827
- McConn M, Browse J (1996) The critical requirement of linolenic acid is pollen development, not photosynthesis, in an Arabidopsis mutant. Plant Cell 8:403–406
- Meszaros T, Helfer A, Hatzimasoura E, Magyar Z, Serazetdinova L et al (2006) The *Arabidopsis* MAP kinase kinaseMKK1participates in defense responses to the bacterial elicitor flagellin. Plant J 48:485–498
- Mikolajczyk M, Olubunmi SA, Muszynska G, Klessig DF, Dobrowolska G (2000) Osmotic stress induces rapid activation of a salicylic acid-induced protein kinase and a homolog of protein kinase ASK1 in tobacco cells. Plant Cell 12:165–178
- Mishina TE, Lamb C, Zeier J et al (2007) Expression of a nitric oxide degrading enzyme induces a senescence programme in *Arabidopsis*. Plant Cell Environ 30:39–52
- Mittler R (2006) Abiotic stress, the field environment and stress combination. Trend Plant Sci 11:15–19
- Mockaitis K, Estelle M (2008) Auxin receptors and plant development: a new signaling paradigm. Annu Rev Cell Dev Biol 24:55–80
- Modolo LV, Augusto O, Almeida IM, Magalhaes JR, Salgado I et al (2005) Nitrite as the major source of nitric oxide production by *Arabidopsis thaliana* in response to *Pseudomonas syringae*. FEBS Lett 579:3814–3820
- Modolo LV, Augusto O, Almeida IM, Pinto-Maglio CAF, Oliveira HC et al (2006) Decreased arginine and nitrite levels in nitrate reductase-deficient *Arabidopsis thaliana* plants impair nitric oxide synthesis and the hypersensitive response to *Pseudomonas syringae*. Plant Sci 171:34–40
- Molassiotis A, Tanou G, Diamantidis G (2010) NO says more than 'YES' to salt tolerance: salt priming and systemic nitric oxide signaling in plants. Plant Signal Behav 5:209–212
- Moon H, Lee B, Choi G, Shin D, Prasad DT, Lee O, Kwak SS, Kim DH, Nam J, Bahk JC, Lee SY, Cho MJ, Lim CO, Yun DJ (2003) NDP kinase 2 interacts with two oxidative stress-activated MAPKs to regulate cellular redox state and enhances multiple stress tolerance in transgenic plants. Proc Natl Acad Sci USA 100:358–363
- Mori IC, Murata Y, Yang YZ, Munemasa S, Wang YF, Andreoli S, Tiriac H, Alonso JM, Harper JF, Ecker JR, Kwak JM, Schroeder JI (2006) CDPKs CPK6 and CPK3 function in ABA regulation of guard cell S-type anion- and Ca²⁺-permeable channels and stomatal closure. PLoS Biol 4:1749–1762
- Mora-Garcia S, Vert G, Yin Y, Cano-Delgado A, Cheong H, nd Chory J (2004) Nuclear protein phosphatases with Kelch-repeat domains modulate the response to brassinosteroids in *Arabidopsis*. Genes Dev 18:448–460

- Munnik T, Ligterink W, Meskiene I, Calderini O, Beyerly J et al (1999) Distinct osmosensing protein kinase pathways are involved in signalling moderate and severe hyperosmotic stress. Plant J 20:381–388
- Mustilli AC, Merlot S, Vavasseur A, Fenzi F, Giraudat J (2002) Arabidopsis OST1 protein kinase mediates the regulation of stomatal aperture by abscisic acid and acts upstream of reactive oxygen species production. Plant Cell 14:3089–3099
- Nakagami H, Kiegerl S, Hirt H (2004) OMTK1, a novel MAPKKK, channels oxidative stress signaling through direct MAPK interaction. J Biol Chem 279:26959–26966
- Neill SJ, Desikan R, Clarke A, Hancock JT (2002) Nitric oxide is a novel component of abscisic acid signaling in stomatal guard cells. Plant Physiol 128:13–16
- Neill SJ, Desikan R, Hancock JT (2003) Nitric oxide signalling in plants. New Phytol 159:11–35
- Neill S, Barros R, Bright JO, Desikan R, Hancock J, Harrison J, Morris P, Ribeiro B, Wilson I (2008) Nitric oxide, stomatal closure, and abiotic stress. J Exp Bot 59:165–176
- Nishihama R, Ishikawa M, Araki S, Soyano T, Asada T, Machida Y (2001) The NPK1 mitogen-activated protein kinase kinase kinase is a regulator of cell-plate formation in plant cytokinesis. Genes Dev 15:352–363
- Noguchi T, Fujioka S, Choe S, Takatsuto S, Yoshida S, Yuan H, Feldmann FA, Tax FE (1999) Brassinosteroidinsensitive dwarf mutants of Arabidopsis accumulate brassinosteroids. Plant Physiol 121:743–752
- Ohkuma K, Lyon JL, Addicott FT, Smith OE (1963) Abscisin II, an abscission-accelerating substance from young cotton fruit. Science 142:1592–1593
- Ozturk ZN, Talame V, Deyholos M, Michalowski CB, Galbraith DW, Gozukirmizi N, Tuberosa R, Bohnert HJ (2002) Monitoring large-scale changes in transcript abundance in drought- and salt-stressed barley. Plant Mol Biol 48:551–573
- Pagnussat GC, Lanteri ML, Lamattina L (2003) Nitric oxide and cyclic GMP are messengers in the indole acetic acid-induced adventitious rooting process. Plant Physiol 132:1241–1248
- Pandey GK, Reddy MK, Sopory SK, Singla-Pareek S (2002) Calcium homeostasis in plants: role of calcium binding proteins in abiotic stress tolerance. Ind J Biotechnol 1:135–157
- Pandey GK, Cheong YH, Kim KN, Grant JJ, Li LG et al (2004) The calcium sensor calcineurin B-like CBL9 modulates abscisic acid sensitivity and biosynthesis in Arabidopsis. Plant Cell 16:1912–1924
- Perrot-Rechenmann C (2010) Cellular responses to auxin: division versus expansion. Cold Spring Harb Perspect Biol 2(5):a00144
- Perruc E, Charpenteau M, Ramirez BC, Jauneau A, Galaud JP, Ranjeva R, Ranty B (2004) A novel calmodulin-binding protein functions as a negative regulator of osmotic stress tolerance inArabidopsis thaliana seedlings. Plant J 38:410–420
- Pfeiffer S, Janistyn B, Jessner G, Pichorner H, Eberman RE (1994) Gaseous nitric oxide stimulates guanosine

3', 5'-cyclic monophosphate formation in spruce needles. Phytochemistry 36:259–262

- Plieth C, Hansen UP, Knight H, Knight MR (1999) Temperature sensing by plants: the primary characteristics of signal perception and calcium response. Plant J 18:491–497
- Popova LP, Maslenkova LT, Yordanova RY, Ivanova AP, Krantev AP, Szalai G, Janda T (2009) Exogenous treatment with salicylic acid attenuates cadmium toxicity in pea seedlings. Plant Physiol Biochem 47:224–231
- Purcell PC, Smith AM, Halford NG (1998) Antisense expression of a sucrose non-fermenting-1-related protein kinase sequence in potato results in decreased expression of sucrose synthase in tubers and loss of sucrose-inducibility of sucrose synthase transcripts in leaves. Plant J 14:195–202
- Qiu QS, Guo Y, Dietrich MA, Schumaker KS, Zhu JK (2002) Regulation of SOS1, a plasma membrane Na⁺/ H⁺ exchanger *Arabidopsis thaliana*, by SOS2 and SOS3. Proc Natl Acad Sci USA 99:8436–8441
- Qiu JL, Zhou L, Yun BW, Nielsen HB, Fiil BK et al (2008) Arabidopsis mitogen-activated protein kinase kinases MKK1and MKK2 have overlapping functions in defense signaling mediated by MEKK1, MPK4, and MKS1. Plant Physiol 148:212–222
- Rao MV, Paliyath G, Ormrod P, Murr DP, Watkins CB (1997) Influence if salicylic acid on H_2O_2 production, oxidative stress and H_2O_2 -metabolizing enzymes. Plant Physiol 115:137–149
- Rao KP, Richa T, Kumar K, Raghuram B, Sinha AK (2010) *In silico* analysis reveals 75 members of mitogen-activated protein kinase kinase kinase gene family in rice. DNA Res 17:139–153
- Rao KP, Vani G, Kumar K, Wankhede DP, Mishra M, Gupta M, Sinha AK (2011) Arsenic stress activates MAP kinase in rice roots and leaves. Arch Biochem Biophys 506(1):73–82
- Reddy ASN (2001) Calcium: silver bullet in signaling. Plant Sci 160:381–404
- Ren D, Yang H, Zhang S (2002) Cell death mediated by MAPK is associated with hydrogen peroxide production in Arabidopsis. J Biol Chem 277:559–565
- Rock C (2000) Pathways to abscisic acid-regulated gene expression. New Phytol 148:357–396
- Rockel P, Strube F, Rockel A, Wildt J, Kaiser WM (2002) Regulation of nitric oxide (NO) production by plant nitrate reductase *in vivo* and *in vitro*. J Exp Bot 53:103–110
- Rodriguez MCS, Petersen M, Mundy J (2010) Mitogenactivated protein kinase signaling in plants. Annu Rev Plant Biol 61:621–649
- Rohde A, Kurup S, Holdsworth M (2000) ABI3 emerges from the seed. Trend Plant Sci 5:418–419
- Rolland F, Baena-Gonzalez E, Sheen J (2006) Sugar sensing and signaling in plants: conserved and novel mechanisms. Annu Rev Plant Biol 57:675–709
- Romeis T, Ludwig AA, Martin R, Jones JDG (2001) Calcium-dependent protein kinases play an essential role in a plant defense response. EMBO J 20:5556–5567
- Rosa M, Prado C, Podazza G, Interdonato R, González JA, Hilal M, Prado FE (2009) Soluble sugars, metabolism, sensing and abiotic stress-A complex

network in the life of plants. Plant Signal Behav 4:388-393

- Rossomando EF, Hadjimichael J, Varnum-Finney B, Soll DR (1987) HLAMP – a conjugate of hippuryllysine and AMP which contains a phosphoamide bond-stimulates chemotaxis in Dictyostelium discoideum. Differentiation 35:88–93
- Rus A, Yokoi S, Sharkhuu A, Reddy M, Lee BH, Matsumoto TK, Koiwa H, Zhu JK, Bressan RA, Hasegawa PM (2001) AtHKT1 is a salt tolerance determinant that controls Na⁺ entry into plant roots. Proc Natl Acad Sci USA 98:14150–14155
- Saijo Y, Kinoshita N, Ishiyama K, Hata S, Kyozuka J, Hayakawa T, Nakamura T, Shimamoto K, Yamaya T, Izui K (2001) A Ca²⁺ dependent protein kinase that endows rice plants with cold- and salt stress tolerance functions in vascular bundles. Plant Cell Physiol 42:1228–1233
- Sakai H, Hua J, Chen QG, Chang C, Medrano LJ, Bleecker AB, Meyerowitz EM (1998) ETR2 is an ETR1-like gene involved in ethylene signaling in *Arabidopsis*. Proc Natl Acad Sci USA 95:5812–5817
- Samuel MA, Ellis BE (2000) Double jeopardy both overexpression and suppression of a redox-activated plant mitogen-activated protein kinase render tobacco plants ozone sensitive. Plant Cell 14:2059–2069
- Samuel MA, Miles GP, Ellis BE (2000) Ozone treatment rapidly activates MAP kinase signalling in plants. Plant J 22:367–376
- Sanan-Mishra N, Tuteja R, Tuteja N (2006) Signaling through MAP kinase networks in plants. Arch Biochem Biophys 452:55–68
- Sarwat M, Tuteja N (2007) Calnexin: a versatile calcium binding integral membrane-bound chaperone of endoplasmic reticulum. Calcium Binding Proteins 2:36–50
- Schaller GE, Bleecker AB (1995) Ethylene-binding sites generated in yeast expressing the Arabidopsis ETR1 gene. Science 270:1809–1811
- Schlessinger J (2000) Cell signaling by receptor tyrosine kinases. Cell 103:211–225
- Schlessinger J (2002) Ligand-induced, receptor-mediated dimerization and activation of EGF receptor. Cell 110:660–672
- Scott IM, Clarke SM, Wood JE, Mur LAJ (2004) Salicylate accumulation inhibits growth at chilling temperature in *Arabidopsis*. Plant Physiol 135:1040–1049
- Seki M, Narusaka M, Ishida J et al (2002) Monitoring the expression profiles of 7000 Arabidopsis genes under drought, cold and high-salinity stresses using a full-length cDNA microarray. Plant J 31:279–292
- Seo S, Katou S, Seto H, Gomi K, Ohashi Y (2007) The mitogen-activated protein kinases WIPK and SIPK regulate the levels of jasmonic and salicylic acids in wounded tobacco plants. Plant J 49:899–909
- Sheen J (1996) Ca²⁺-dependent protein kinases and stress signal transduction in plants. Science 274:1900–1902
- Shi H, Zhu J-K (2002) Regulation of expression of the vacuolar Na⁺/H⁺ antiporter gene AtNHX1 by salt stress and ABA. Plant Mol Biol 50:543–550
- Shinozaki K, Yamaguchi-Shinozaki K (2000) Molecular responses to dehydration and low temperature: differences

and cross-talk between two stress signaling pathways. Curr Opin Plant Biol 3:217–223

- Shiu SH, Karlowski WM, Pan R, Tzeng YH, Mayer KFX, Li WH (2004) Comparative analysis of the receptorlike kinase family in *Arabidopsis* and rice. Plant Cell 16:1220–1234
- Singh G, Kaur M (1980) Effect of growth regulators on podding and yield of mung bean (*Vigna radiata* L.). Ind J Plant Physiol 23:366–370
- Singh RP, Srivastava HS (1978) Effect of salicylic acid on NADH: glutamate synthase activity in roots and leaf tissues of maize seedlings. Ind J Plant Physiol 30:60–65
- Sinha AK, Jaggi M, Raghuram B, Tuteja N (2011) Mitogen-activated protein kinase signaling in plants under abiotic stress. Plant Signal Behav 6(2):196–203
- Smeekens S (2000) Sugar induced signal transduction in plants. Annu Rev Plant Physiol Plant Mol Biol 52:49–81
- Smet ID, Voss U, Jurgens G, Beeckman T (2009) Receptor-like kinases shape the plant. Nat Cell Biol 11:1166–1173
- Smith JL, Moraes CMD, Mescher MC (2009) Jasmonateand salicylate-mediated plant defense responses to insect herbivores, pathogens and parasitic plants. Pest Manag Sci 65:497–503
- Song WY, Zhang ZB, Shao HB, Guo XL, Cao HX, Zhao HB, Fu ZY, Hu XJ (2008) Relationship between calcium decoding elements and plant abiotic stress resistance. Int J Biol Sci 4:116–125
- Sokolovski S, Hills A, Gay R, Garcia-Mata C, Lamattina L, Blatt MR (2005) Protein phosphorylation is a prerequisite for intracellular Ca²⁺ release and ion channel control by nitric oxide and abscisic acid in guard cells. Plant J 43:520–529
- Stafstrom JP, Altschuler M, Anderson DH (1993) Molecular cloning and expression of a MAP kinase homologue from pea. Plant Mol Biol 22:83–90
- Staswick PE, Su W, Howell SH (1992) Methyl jasmonate inhibition of root growth and induction of leaf protein are decreased in an *Arabidopsis thaliana* mutant. Proc Natl Acad Sci USA 89:6837–6840
- Staswick PE, Tiryaki I, Rowe ML (2002) Jasmonate response locus JAR1 and several related *Arabidopsis* genes encode enzymes of the firefly luciferase super family that show activity on jasmonic salicylic and indole-3-acetic acids in an assay for adenylation. Plant Cell 14:1405–1415
- Steggerda SM, Paschal BM (2002) Regulation of nuclear import and export by the GTPase Ran. Int Rev Cytol 217:41–91
- Stepanova AN, Ecker JR (2000) Ethylene signaling: from mutants to molecules. Curr Opin Plant Biol 3:353–360
- Stintzi A, Browse J (2000) The Arabidopsis male sterile mutant, opr3, lacks the 12-oxophytodienoic acid reductase required for jasmonate synthesis. Proc Natl Acad Sci USA 97:10625–10630
- Stohr C, Stremlau S (2006) Formation and possible roles of nitric oxide in plant roots. J Exp Bot 57:463–470
- Sturgill TW, Ray LB (1986) Muscle proteins related to microtubule associated protein-2 are substrates for an

insulin-stimulatable kinase. Biochem Biophys Res Commun 134:565–571

- Tahtiharju S, Palva T (2001) Antisense inhibition of protein phosphatase 2C accelerates cold acclimation in *Arabidopsis thaliana*. Plant J 26:461–470
- Takahashi F, Yoshida R, Ichimura K, Mizoguchi T, Seo S et al (2007) The mitogen-activated protein kinase cascade MKK3-MPK6 is an important part of the jasmonate signal transduction pathway in *Arabidopsis*. Plant Cell 19:805–818
- Tang D, Christiansen KM, Innes RW (2005) Regulation of plant disease resistance, stress responses, cell death, and ethylene signaling in Arabidopsis by the EDR1 protein kinase. Plant Physiol 138:1018–1026
- Tang W, Kim TW, Oses-Prieto JA, Sun Y, Deng Z, Zhu S, Wang R, Burlingame AL, Wang ZY (2008) BSKs mediate signal transduction from the receptor kinase BRI1 in *Arabidopsis*. Science 321:557–560
- Tang W, Deng Z, Wang ZY (2010) Proteomics shed light on the brassinosteroid signaling mechanisms. Curr Opin Plant Biol 13:27–33
- Tasgin E, Atici O, Nalbantoglu B (2003) Effects of salicylic acid and cold on freezing tolerance in winter wheat leaves. Plant Growth Regul 41:231–236
- Teige M, Scheikl E, Eulgem T, Doczi R, Ichimura K et al (2004) The MKK2 pathway mediates cold and salt stress signaling in *Arabidopsis*. Mol Cell 15:141–152
- Tena G, Asai T, Chiu WL, Sheen J (2001) Plant mitogenactivated protein kinase signaling cascades. Curr Opin Plant Biol 4:392–400
- Traw MB, Bergelson J (2003) Interactive effects of jasmonic acid, salicylic acid, and gibberellin on induction of trichomes in *Arabidopsis*. Plant Physiol 133:1367–1375
- Tuteja N (2007a) Abscisic acid and abiotic stress signalling. Plant Signal Behav 2:135–138
- Tuteja N (2007b) Mechanisms of high salinity tolerance in plants. Methods Enzymol 428:419–438
- Tuteja N (2009a) Cold, salt and drought stress. In: Hirt H (ed) Plant stress biology: from genomics towards system biology. Wiley, Weinheim, pp 137–159
- Tuteja N (2009b) Integrated calcium signaling in plants. In: Baluska F, Mancuso S (eds) Signaling in plants. Springer, Heidelberg, pp 29–49
- Tuteja N, Mahajan S (2007) Further characterization of calcineurin B-like protein and its interacting partner CBL-interacting protein kinase from *Pisum sativum*. Plant Signal Behav 2:358–361
- Tuteja N, Sopory SK (2008) Chemical signaling under abiotic stress environment in plants. Plant Signal Behav 3:525–536
- Tuteja N, Chandra M, Tuteja R, Misra MK (2004) Nitric oxide as a unique bioactive signaling messenger in physiology and pathophysiology. J Biomed Biotechnol 2004:227–237
- Uchida A, Jagendorf AT, Hibino T, Takabe T, Takabe T (2002) Effects of hydrogen peroxide and nitric oxide on both salt and heat stress tolerance in rice. Plant Sci 163:515–523

- Uozumi N, Kim EJ, Rubio F, Yamaguchi T, Muto S, Tsuboi A, Bakker EP, Nakamura T, Schroeder JI (2000) The Arabidopsis HKT1 gene homolog mediates inward Na⁺ currents in Xenopus laevis oocytes and Na⁺ uptake in Saccharomyces cerevisiae. Plant Physiol 122:1249–1259
- Vanderbeld B, Snedden W (2007) Developmental and stimulus-induced expression patterns of Arabidopsis calmodulin-like genes CML37, CML38 and CML39. Plant Mol Biol 64:683–697
- Wang LJ, Li SH (2006) Salicylic acid-induced heat or cold tolerance in relation to Ca²⁺ homeostasis and antioxidant systems in young grape plants. Plant Sci 170:685–694
- Wang ZY, Nakano T, Gendron J, He J, Chen M, Vafeados D, Yang Y, Fujioka S, Yoshida S, Asami T, Chory J (2002) Nuclear-localized BZR1 mediates brassinosteroid-induced growth and feedback suppression of brassinosteroid biosynthesis. Dev Cell 2:505–513
- Wang W, Hall AE, O'Malley R, Bleecker AB (2003) Canonical histidine kinase activity of the transmitter domain of the ETR1 ethylene receptor from Arabidopsis is not required for signal transmission. Proc Natl Acad Sci USA 100:352–357
- Weinl S, Kudla J (2009) The CBL-CIPK Ca²⁺-decoding network: function and perspectives. New Phytol 184:517–528
- Witte CP, Keinath N, Dubiella U, Demouliere R, Seal A, Romeis T (2010) Tobacco calcium-dependent protein kinases are differentially phosphorylated in vivo as part of a kinase cascade that regulates stress response. J Biol Chem 285:9740–9748
- Woeste KE, Kieber JJ (2000) A strong loss-of-function mutation in RAN1 results in constitutive activation of the ethylene response pathway as well as a rosettelethal phenotype. Plant Cell 12:443–455
- Wurzinger B, Mair A, Pfister B, Teige M (2011) Crosstalk of calcium-dependent protein kinase and MAP kinase signalling. Plant Signal Behav 6:1–5
- Xiao W, Sheen J, Jang JC (2000) The role of hexokinase in plant sugar signal transduction and growth and development. Plant Mol Biol 44:451–461
- Xing Y, Jia W, Zhang J (2008) AtMKK1 mediates ABAinduced CAT1 expression andH2O2 production via AtMPK6-coupled signaling in *Arabidopsis*. Plant J 54:440–451
- Xiong L, Lee H, Ishitani M, Zhu JK (2002) Regulation of osmotic stress responsive gene expression by LOS6/ABA1 locus in Arabidopsis. J Biol Chem 277:8588–8596
- Xu YC, Zhao BL (2003) The main origin of endogenous NO in higher non-leguminous plants. Plant Physiol Biochem 41:833–838
- Yamasaki H (2000) Nitrite-dependent nitric oxide production pathway: implications for involvement of active

nitrogen species in photoinhibition *in vivo*. Phil Trans R Soc Lond B Biol Sci 355:1477–1488

- Yang KY, Liu Y, Zhang S (2001) Activation of a mitogenactivated protein kinase pathway is involved in disease resistance in tobacco. Proc Natl Acad Sci USA 98:741–746
- Yin Y, Wang ZY, Mora-Garcia S, Li J, Yoshida S, Asami T, Chory J (2002) BES1 accumulates in the nucleus in response to brassinosteroids to regulate gene expression and promote stem elongation. Cell 109:181–191
- Yoo SD, Cho YH, Tena G, Xiong Y, Sheen J (2008) Dual control of nuclear EIN3 by bifurcate MAPK cascades in C2H4 signalling. Nature 451:789–795
- Zemojtel T, Frohlich A, Palmieri MC, Kolanczyk M, Mikula I, Wyrwicz LS et al (2006) Plant nitric oxide synthase: a never-ending story? Trend Plant Sci 11:524–525
- Zhang S, Klessig DF (2001) MAPK cascades in plant defense signaling. Trends Plant Sci 6:520–527
- Zhang A, Jiang M, Zhang J, Tan M, Hu X (2006) Mitogenactivated protein kinase is involved in abscisic acidinduced antioxidant defense and acts downstream of reactive oxygen species production in leaves of maize plants. Plant Physiol 141:475–487
- Zhang F, Wang Y, Yang Y, Wu H, Di W, Liu J (2007) Involvement of hydrogen peroxide and nitric oxide in salt resistance in the calluses from *Populus euphratica*. Plant Cell Environ 30:775–785
- Zhao XC, Qu X, Mathews DE, Schaller GE (2002) Effect of ethylene pathway mutations upon expression of the ethylene receptor ETR1 from *Arabidopsis*. Plant Physiol 130:1983–1991
- Zhao MG, Zhao X, Wu YX, Zhang LX (2007a) Enhanced sensitivity to oxidative stress in *Arabidopsis* nitric oxide synthesis mutant. J Plant Physiol 164:737–745
- Zhao MG, Tian QY, Zhang WH (2007b) Nitric oxide synthase-dependent nitric oxide production is associated with salt tolerance in *Arabidopsis*. Plant Physiol 144:206–217
- Zhu JK (2000) Genetic analysis of plant salt tolerance using Arabidopsis. Plant Physiol 124:941–948
- Zhu JK (2003) Regulation of ion homeostasis under salt stress. Curr Opin Plant Biol 6:441–445
- Zhu SY, Yu XC, Wang XJ, Zhao R, Li Y, Fan RC, Shang Y, Du SY, Wang XF, Wu FQ, Xu YH, Zhang XY, Zhang DP (2007) Two calcium-dependent protein kinases, CPK4 and CPK11, regulate abscisic acid signal transduction in *Arabidopsis*. Plant Cell 19:3019–3036
- Zottini M, Costa A, De Michele R, Ruzzene M, Carimi F, Lo Schiavo F (2007) Salicylic acid activates nitric oxide synthesis in *Arabidopsis*. J Exp Bot 58:1397–1405

Calcium Signalling in Plant Cells Under Environmental Stress

15

Sylvia Lindberg, Md. Abdul Kader, and Vladislav Yemelyanov

Abstract

A change of intracellular calcium concentration is an early event in a large array of biological processes in plants, such as cell division, polarity, growth and development at normal conditions and under adaptation to abiotic and biotic stresses. This chapter focuses on calcium signalling induced by different types of abiotic stresses, such as salt, cold, anoxia, aluminium and heavy metal stress, while a minor part deals with biotic stress signalling. Most investigations, so far, concerned Ca2+ signalling in the cytosol; however, signalling in the nucleus and other cell compartments such as mitochondria, ER and cell wall have also been reported. The specific "signature" of calcium, including duration, amplitude and frequency of the signalling, induced by different stresses is essential for a change of the physiological function. Different stores for calcium take part in the signalling under various types of stress. Of special interest is a comparison of signalling in tolerant and sensitive species and cultivars.

Keywords

Abiotic stress • Aluminium • Anoxia • Calcium • Membrane transport • Signal transduction • Stress

S. Lindberg (⊠) Department of Botany, SU, SE-106 91 Stockholm, Sweden e-mail: Sylvia.Lindberg@botan.su.se

Md. A. Kader Department of Agronomy, Bangladesh Agricultural University Mymensingh, Mymensingh 2202, Bangladesh

V. Yemelyanov Department of Genetics and Breeding, St. Petersburg State University, St. Peterburg 199034, Russia

1 Introduction

Calcium, (Ca²⁺), is involved in the signal transduction of almost all abiotic and biotic stresses in plants. It also mediates growth and links hormonal signals in plants under stress (Marschner 1995; Reddy 2001; White and Broadley 2003; Desikan et al. 2005; Hepler 2005; Covington and Harmer 2007; Kudla et al. 2010). The signalling network is very complex since it processes multiple signals simultaneously. Signals not only operate in the cytosol but also in mitochondria, chloroplasts, ER and nucleus (Logan and Knight 2003; Xiong et al. 2004; Mazars et al. 2009). The cytosolic calcium is maintained at submicromolar level to prevent precipitation of calcium with phosphate and other anions, or binding to negatively charged macromolecules. Different types of stress usually cause a temporary increase in free cytosolic calcium, (Ca2+cyt), concentration depending on the specific stimulus that induces it. Calcium is transported into the cytosol by channels in the plasma membrane, tonoplast, ER or mitochondria (Hetherington and Brownlee 2004; Dodd et al. 2010). The signal is transient or sustained and occurs as repetitive oscillations or spiking, or can last for a longer time. There is a great electrochemical potential difference for Ca²⁺ across the membrane systems. Regulation of calcium in the cytosol is important for the plant to maintain homeostasis through different channels, pumps and carriers (Dodd et al. 2010). An increase in Ca2+ concentration in the cytosol and nucleus is detected by many Ca2+-sensing proteins, present in these compartments as free proteins or attached to membranes. These proteins are able to decode the information present within the different calcium spikes or oscillations and process this information into alteration of the cell function. Different signalling pathways are regulated by the transcription machinery and expression of downstream target genes. The downstream reactions can cause an adaptation, helping the plant to survive the specific stress. In addition, if the stress is severe, or continues for a long time, the response could be growth inhibition or cell death.

2 Sensors for Calcium

The role of Ca^{2+} as a second messenger molecule depends on its unique physical and chemical properties (Jaiswal 2001). The free Ca^{2+} ion rapidly binds to anions and easily forms insoluble salts with inorganic and organic macromolecular anions (Williams 2006). At high concentration in the cell cytosol, calcium becomes toxic and the Ca²⁺ concentration thus is well regulated. The Ca²⁺ concentration in the apoplast and vacuoles usually range between 0.1 and 1 mM while the endoplasmic reticulum (ER) lumen contains approximately 1 mM, but these concentrations can increase up to 50 mM (Tuteja 2007). On the other hand, the stress-induced Ca²⁺ elevations in the cytosol never exceed micromolar concentration and even a sustained signal may be toxic to the cell and cause apoptosis (Levine et al. 1996).

Plants have a large family of Ca2+-sensor proteins. DeFalco et al. (2010) recently presented a detailed review of Ca2+ sensors. The different Ca²⁺-sensor proteins detect the Ca²⁺ signals and transform them to downstream pathways, either by binding or activating the different targets. Together with channels, pumps and carrier proteins these sensor proteins regulate the free calcium concentration in the cell cytosol. The calcium binding proteins (CaBPs) can function as "trigger" proteins as well as "buffer" proteins (Tuteja and Sopory 2008). The former are activated by forming complex with Ca²⁺ and then affect other proteins of the signalling pathway, while the buffer proteins bind to an elevated concentration of Ca²⁺, leading to a decrease of the free Ca2+ level. The trigger proteins identified so far are calmodulin (CaM), CaM-binding proteins, Ca2+-dependent protein kinases (CDPKs) and phosphatases (Reddy 2001).

An increase in Ca²⁺cyt concentration can be sensed by CaM, CDPKs and other calciumbinding proteins. These proteins can trigger downstream responses, which, in turn, can elicit specific physiological responses. The signals from CaM can induce mRNA and gene expression, either directly, or by binding to specific transcription factors (Tuteja and Mahajan 2007), that can be activated by phosphorylation or dephosphorylation. The CDPKs are Ser/Thr protein kinases found in plants and in some protozoans. Except for the N-terminal domain the CDPKs have a catalytic domain, an auto inhibitory region, and a calmodulin-like domain activated by calcium binding (Harper et al. 1991; Hrabak et al. 2003). The peroxisome-located CDPK may be involved in oxidative stress and lipid metabolism. In Arabidopsis, 9 CDPK isoforms have been characterized (Dammann et al. 2003). The subcellular locations were detected by expression of green fluorescent protein (GFP) fusion to root tips of transgene plants and fluorescence microscopy. The isoforms, AtCPK3 and AtCPK4, being primarily soluble, showed a nuclear and cytosolic distributions, respectively, while all the others were membrane bound. Except for AtCPK1, which was targeted to the peroxisomes, all the others were targeted to plasma membrane, and these membranes were associated with the kinase by potential N-terminal acylation sites.

Plants possess calmodulin-like proteins (CMLs), which differ from CaM in possessing approximately 148 amino acids and also have 1–6 EF hand motifs (Reddy 2001). Some of the CMLs are suggested to take part as sensors for calcium during stress and developmental responses (Vanderbeld and Snedden 2007).

Protein phosphatases are involved in various stress signals in plants, for instance, at low temperature-induced signal transduction (Sharma and Deswal 2004). Cold stress induces protein phosphatase C (PP2B) and transcripts in the ice plant (Miyazaki et al. 1999). Calcineurin is a Ca2+/CaM-dependent protein phosphatase (PP2B), belonging to the Ser/Thr-specific phosphatases. It has a structure similar to CaM with a catalytic subunit (CnA) and a regulatory subunit (Hashimoto et al. 1990). When the Ca^{2+} concentration is low in a cell, calcineurin becomes inactive, since it has an autoinhibitory domain in the active site, whereas at elevated Ca²⁺ concentration, Ca2+/CaM binds to the CnA subunit of calcineurin, gets activated, causing a conformational change in the enzyme. Calcineurin activity is present in animal cells, but has not yet been detected in plant cells. However, calcineurin-like proteins may be involved in stomatal guard-cell opening, as calcineurin inhibitors block this process (Luan et al. 1993). Calcineurin-like activity was also shown in Brassica juncea subjected to low temperature (Sharma and Deswal 2004).

There are calcium-binding proteins with no EF-hand motifs, such as PLD (phopholipase D), annexins, calreticulin, calnexin and PCP (Pistil-expressed Ca²⁺-binding protein). The PLDs can split the membrane phospholipids into a head

group and phosphatidic acid (Wang 2001). Many different isoforms exist with various affinities for Ca²⁺ and difference in their modulation of phosphoinositides, free fatty acids and lysolipids. The regulators of PLD activity are substrate or products of PLC (phospholipase C). The latter enzyme cleaves phospholipids into IP, and diacylglycerol (DAG). These enzymes together with phospholipase A2 and DAG kinase are supposed to be involved in plant signalling under stress. The annexins have been found in both animals and plants. They contain 4-8 repeats of about 70 amino acids and bind to phospholipids in a Ca²⁺depending way (Tuteja and Mahajan 2007). They are supposed to be involved in the organization and function of membranes. The Arabidopsis annexin (AnnAt1) may have an important impact on pH-mediated cellular response to environmental changes and in ion homeostasis in plant cells (Gorecka et al. 2007). CaM can bind to and activate transcription factors. A family of transcription factors called calmodulin-binding transcription activators (CAMTA) is probably involved in stress-induced gene expression in Arabidopsis, mediated by calcium (Yang and Poovaiah 2002). All CAMTAs have a specific DNA-binding domain, as well as an IQ motif which binds CaM. In salt stress signals in plants, the salt overly sensitive (SOS) pathway is involved and is discussed in this chapter.

3 Transporters of Calcium

In higher plant cells, the free Ca^{2+} level necessary for maintaining a proper metabolism in the cytosol usually varies between 30 and 400 nM. By hormone action, light and when plants are subjected to biotic or abiotic stresses, calcium ions are transported passively into the cytosol by channels in the plasma membrane, tonoplast, ER, chloroplasts and mitochondria. Calcium signals not only generate in the cytosol, but also in organelles with a double membrane, such as mitochondria, chloroplasts and nuclei (Xiong et al. 2006). To restore the low resting concentration, Ca^{2+} has to be transported out against an electrochemical gradient into the apoplast, or into intracellular organelles by Ca²⁺ ATPases or Ca²⁺/H+ antiporters. CaBPs play a vital role in regulating the free Ca²⁺ concentration, whereas the calcium-transporter proteins have different affinities for Ca²⁺. It was suggested by Hirschi (2001) that the Ca²⁺-ATPases, with high affinity ($K_m = 1-10 \mu$ M), although low capacity for transport, are responsible for maintaining the [Ca²⁺]cyt homeostasis in resting cells, but that the Ca²⁺/H⁺ antiporters, with lower affinity ($K_m = 10-15 \mu$ M) and high transport capacity, are active after Ca²⁺cyt signals.

3.1 Ca²⁺ ATPases

In plants the major families of Ca²⁺ ATPases energized by ATP are P-type ATPases IIA and IIB (Sanders et al. 2002). In Arabidopsis, the PIIA clade ATPases, present in the ER named ECAs, and the PIIB clade of autoinhibited Ca²⁺ ATPases, named ACAs, have been identified (Boursiac and Harper 2007). The nucleotide specificity of the PIIA (ECA) type is broad, as they can react with GTP and ITP and are inhibited by erythrosine B (Axelsen and Palmgren 2001). The ECA type lacks the N-terminal autoregulatory domain, present in the ACA type. The ECA group seems to be less important for calcium homeostasis (Dodd et al. 2010). The ACA type along with Ca-CaM binding and a serineresidue phosphorylation site can be modulated by Ca²⁺cyt, either through activation upon binding CaM or by inhibition upon phosphorylation by CDPK (Hwang et al. 2000). The isoforms of PIIB (ACAs) ATPases have different affinities for CaM and are active at different membranes in Arabidopsis, such as the plasma membrane, tonoplast and plastid membranes. It is suggested that they have different functions and react differently to different Ca²⁺ signals (Tuteja and Mahajan 2007). The expression of many Ca²⁺ ATPases increases under high salinity or high Ca²⁺cyt, and some of them are only expressed under stress (Garciadeblas et al. 2001). ACA12 and ACA13 transcripts were upregulated by pathogen stress (Boursiac and Harper 2007) and ACA8 and ACA10 were differently regulated by cold (Schiott and Palmgren 2005).

3.2 Calcium/Proton Exchangers

The Ca²⁺/H⁺ antiporters energized by a protonmotive force have less affinity for Ca²⁺ than Ca²⁺ ATPases. The proton gradient is generated by the tonoplast while the H⁺ ATPase and the protonpumping pyrophosphatases pump protons into the vacuole. The Ca²⁺/H⁺ antiporters are called cation exchangers (CAXs) 1-6 (Shigaki and Hirschi 2006). When the membrane location has been identified, for instance for CAX1, CAX2 and CAX4, the location was in the tonoplast (Hirschi 2001). The CAX1 antiporter has high affinity and high specificity for Ca²⁺, while CAX2 has high affinity for heavy metals and transports them into the vacuole. Especially CAX1 and CAX3 seem to be important for Ca²⁺ homeostasis. Mutants lacking CAX1, cax1, have reduced lateral roots and number, as well as reduced primary inflorescence length (Cheng et al. 2003). These mutants also have an increased capacity for cold acclimation and enhanced expression of CBF/DREB1 (Catala et al. 2003). CAX1 and CAX3 are expressed differently in the root and shoot, and the expression increases at a high Ca²⁺ supply (Hirschi 2001). In shoot, CAX1 expression is strong, while in roots, CAX3 expression dominates. It was shown that CAX3 mutants were more susceptible to salt stress (Zhao et al. 2008). The stoichiometry of the tonoplast antiporters was reported to be 3H+/1Ca2+ (Sanders et al. 2002).

3.3 Calcium Channels

Calcium transport into the cytosol is mediated by different Ca²⁺-permeable channels, such as depolarization-activated Ca²⁺ (DACC), hyperpolarization-activated Ca²⁺ (HACC) and voltageindependent (VICC) channels (Sanders et al. 2002). These channels can be activated by abscisic acid (ABA) and ROS, for instance in guard cells, and by ROS at the plasma membrane of root hairs during cell expansion (Foreman et al. 2003). In other membranes, calcium-permeable channels also take part in the signalling pathway. There are also voltage-dependent channels in the tonoplast and ER (Sanders et al. 2002; Demidchik and Maathuis 2007). Moreover, ligand-gated channels in the tonoplast and ER transport Ca^{2+} into the cytosol. Results have shown that inositol triphosphate (InsP₃) and cADPR can mobilize Ca^{2+} from internal stores. However, receptors for these substances are not encoded by higher plants as in animal cells (Dodd et al. 2010).

Other types of Ca²⁺-permeable channels are cyclic nucleotide-gated channels (CNGCs), glutamate receptor-like (GLR) channels, two-pore channels (TPCs) and annexins (Dodd et al. 2010 and references therein). Some of the CNGCs. besides permeable for monovalent cations, have also been found in the plasma membrane with both a nucleotide-binding domain and a CaMbinding one. The GLR channels are homologues of animal glutamate receptors and function as non-selective cation channels. Arabidopsis has 20 members of the GLR gene family and the channel is supposed to be a tetramer or pentamer (Davenport 2002). Addition of glutamate and five other amino acids, as well as glutathione, to Arabidopsis seedlings induces a Ca2+ spike in the cytosol, which is reduced in glr3.3 mutant (Qi et al. 2006).

The *Arabidopsis* genome contains of a single member of the TPC. This protein is supposed to be a homodimer, with two calcium-binding EF hands and a 14-3-3 binding domain and is probably voltage gated (Peiter et al. 2005). It has been suggested that during stress, annexins could form plasma membrane Ca^{2+} -permeable channels (Laohavisit et al. 2009). When annexin proteins were purified from maize and incorporated into planar lipid bilayers, they could function as Ca^{2+} channels. Also, it was proposed that calcium transport into the nucleus proceeds in different ways (Alonso et al. 2006).

4 Organ, Cell and Organelle-Specific Signalling

Different types of signalling occur in different plant organs, cell types and cell organelles. Moreover, root-to-shoot signalling often occurs under different types of stresses. The xylem is most likely involved in the transfer of signals, e.g. pH changes (Felle et al. 2005). To study calcium signalling in vivo, in intact plants and in different cell types, luminescence measurements of seedlings constitutively expressing the calciumbinding protein aequorin can be used (Plieth 2001). To investigate Ca²⁺cyt changes in single living cells or protoplasts, fluorescent probes specific for calcium, such as Fura 2 and Indo 1, and two-wavelength photometry together with epifluorescence microscopy can be used. For confocal laser-scanning microscopy, the fluorescent probe Fluo 3 is more often used. The sub-cellular distribution of soluble calcium can also be detected by transmission electron microscopy in combination with potassium pyroantimonate precipitation method (Ma et al. 2009).

4.1 Calcium Signalling in Different Cell Types

To detect changes in Ca2+cyt concentration in different root cells, Kiegle et al. (2000) used transgene Arabidopsis, so-called GAL4 expression in specific cell types. By fusing yellow fluorescent protein (YFP) to aequorin, the luminescence of targeted aequorin, reflecting the free calcium concentration, could be measured in the different cells when roots were subjected to salt, cold and osmotic stresses. Cold stress induced rapid peaks in all cell types tested. However, there were significant differences in their response to cold, osmotic (440 mM mannitol) and salt stresses (220 mM NaCl) in different cell types. The endodermis and pericycle cells displayed prolonged oscillations in Ca²⁺cyt concentration when subjected to osmotic and salt stresses, different from other cell types (Kiegle et al. 2000).

4.2 Calcium Signalling in Different Organelles

Calcium-dependent events take place in the cytosol, as well as in the organelles containing millimolar concentration of calcium, mainly sequestrated as bound calcium (Malho et al. 1998). Calcium signalling takes place in organelles originating from prokaryotes, such as mitochondria and chloroplasts (Johnson et al. 1995; Logan and Knight 2003; Xiong et al. 2006; McAinsh and Pittman 2009). The sub-cellular location of Ca²⁺ sensors plays an important role in defining their functions. Different organelles in the cell can communicate with each other by Ca²⁺. For instance, Ca²⁺ can be exchanged between chloroplasts and cytosol (Sai and Johnson 2002), and between mitochondria and ER (Rizutto et al. 2009). The nucleus has the potential to generate and regulate Ca²⁺ signals of its own (Pauly et al. 2000; Mazars et al. 2009). Since the nucleus has a Ca²⁺ signalling system, this could originate from prokaryotes too, as speculated by Galon et al. (2010b). In maize leaflets under drought stress, the level of Ca2+ increased in cytoplasm, chloroplasts and nucleus and decreased in vacuoles and intercellular spaces (Ma et al. 2009). It is evident that calcium signalling in plant cells is very complex.

4.2.1 Apoplastic Signalling

Gao et al. (2004) reported both intracellular and extracellular pH and calcium changes in Arabidopsis mutants under salinity, drought (mannitol) and cold stresses. They used ratiometric pH-sensitive derivatives of GFP as pH indicator and fused it with luminescent aequorin for simultaneous calcium measurements. An Arabidopsis chitinase signal sequence was used to deliver the indicator complex to the apoplast. Luminescence measurements from the root showed that chilling stress induced different calcium and pH changes in the cytosol compared with apoplast. Calcium changes were much more distinct in the cytosol than in the apoplast. Osmotic stress (200 mM mannitol) hardly affected the apoplastic calcium, but gave pronounced transients in cytosolic calcium. On the other hand, repeated additions/removals of isoosmotic NaCl (100 mM) induced calcium changes both in the apoplast and cytosol. The first NaCl addition showed a permanent increase of the cytoplasmic calcium, after a short transient peak, and this increase remained after several additions/

removals. The amplitude of Ca²⁺cyt response to hyperosmotic stimulus was much smaller as compared to hypo-osmotic stimulus. This finding corroborates result from experiments with tobacco culture cells (Pauly et al. 2001) showing that shrinking is less serious for the cells compared to swelling. Swelling could rupture the cell wall and/or membrane (Gao et al. 2004). Although calcium can activate cell-wall phosphatases (DeMarty et al. 1984) and apoplastic CaM plays an important role in signal transduction (Sun et al. 1994; Ma et al. 1999), it is still an open question whether the apoplast plays an important role in the signalling system under stress. Essah (2000) showed that external calcium had no effect on Na+ toxicity in Arabidopsis.

4.2.2 Calcium Signalling and Transport in Chloroplasts

It is now more clear that organelles with double membranes, such as mitochondria, chloroplasts and nuclei can generate calcium signals on their own (Xiong et al. 2006). It was already shown during 1980 that light induces a calcium influx into chloroplasts of both wheat and spinach (Muto et al. 1982; Kreimer et al. 1985). Kreimer et al. (1985) suggested that the influx across the envelope of intact chloroplasts was linked to the photosynthetic electron transport. Later on Shabala and Newman (1999) reported, by use of ion-selective vibrating microelectrodes close to the leaf surface, that light also induced changes in H⁺, K⁺, Cl⁻ and Ca²⁺ concentrations in the mesophyll cells of beans. These ion fluxes were related to changes in plasma membrane potential and a calcium influx was considered as a main depolarizing agent in the electrical response to light. Calcium influx started within 5 s, while net fluxes of H⁺, K⁺, Cl⁻ did not begin until after 2 min. The initial alkalinization found in the medium was suggested to depend on CO₂ uptake by the photosynthesizing tissue, while the activation of the H⁺ pump occurred 1.5-2 min later. Two different mechanisms for uptake of Ca2+ into chloroplasts were suggested; one by a potential-stimulated uniporter at the inner-envelope membrane and another by a H⁺/Ca²⁺ antiporter in the thylakoids fuelled by ATP (Xiong et al. 2006).

Elevated temperatures induce many changes in gene expression leading to thermotolerance and improve the cell survival to high temperature. A report of Gong et al. (1998) suggests that cytosolic Ca2+ is also involved in heat-shock (HS) signal transduction. Use of transgene tobacco, where the aequorin protein was targeted both to cytosol and chloroplasts, showed that HS induced a prolonged, transient increase of Ca²⁺ in the cytosol and not in the chloroplasts. Inhibitor analysis suggested that Ca²⁺ was mobilized from both intra as well as extracellular sources. There are some reports showing that chloroplast can control the cytosolic Ca²⁺ transients involved in stomatal closure (Nomura et al. 2008; Weini et al. 2008).

4.2.3 Calcium Signalling and Transport in Mitochondria

Logan and Knight (2003) described the first successful targeting of aequorin to plant mitochondria, where they found, that the resting values of free Ca²⁺ concentration differed in the cytosol (100 nM) and in mitochondria (200 nM). Treatment of Arabidopsis seedlings floating in water, with cold, osmotic (mannitol), touch and oxidative (hydrogen peroxide 10 mM) stresses showed almost the same calcium signature (kinetic pattern) in both cytosol and mitochondria. However, except for hydrogen peroxide addition, the amplitude was much higher in cytosolic than in mitochondrial calcium increase. Touch stress induced an immediate elevation, followed by a return to near-resting concentration within 20 s in the cytosol, but the return to resting level was much slower in mitochondria. The addition of hydrogen peroxide caused almost the same reaction in both compartments and hence indicated that mitochondria could be more sensitive to oxidative stress than to other stresses. Calcium and palmitic acid (Pal) induced a stable and prolonged partial depolarization of the mitochondrial membrane, pore opening, release of calcium and swelling of mitochondria (Mironova et al. 2007). Addition of inhibitors of Ca²⁺ uniporters, like ruthenium red and La³⁺, as well as EGTA, added 10 min after Pal/Ca2+-activated pore opening, prevented the release of Ca²⁺ and

re-polarized the membrane to initial level. The authors also found similar effects in mitochondria accumulating high strontium, Sr^{2+} , concentration in the absence of exogenous Pal, leading to activation of phospholipase A2 and formation of endogenous fatty acids. They concluded that Ca^{2+} is taken up into mitochondria by a uniporter and that Ca^{2+} efflux is mediated by a short-living Pal/Ca²⁺-activated pore. Under oxidative stress, an increase of electron transport in mitochondria triggers H_2O_2 production leading to depletion of ATP, opening of permeability transition pores (PTP) and finally causes death of the cell (Tiwari et al. 2002).

4.2.4 The Nucleus Can Generate and Regulate Calcium Signals on Its own

The nucleus can be considered as two compartments: the nucleoplasm and the nuclear envelope, in which calcium is stored, and possibly both have a calcium buffering capacity (Briere et al. 2006). The envelope lumen is connected with the lumen of the ER (Bach et al. 1992). The envelope also possesses pores that allow molecules up to 40 kDa to penetrate (Brandizzi et al. 2004). It is well known that calcium-dependent processes take place in the nuclei (Xiong et al. 2006). For instance, an increase in free calcium level is necessary for activation of nuclear sensing kinases, phosphatases in the nucleus and other steps in the signalling pathways both in animals (Carafoli 2002), as well as in plants (Harper et al. 2004). It has been suggested that the nucleus can be described as "a cell within the cell" (Bkaily 2006; Gomes et al. 2006).

Since 1999 it was possible to detect free calcium concentration changes in the nucleus, $[Ca^{2+}]n$. van der Luit et al. (1999) succeeded to fuse aequorin to nucleoplasmin, a nuclear protein. They also investigated the effects by wind and cold shock on both cytosolic and nuclear changes of calcium in tobacco seedlings, and found different dynamics in $[Ca^{2+}]n$ and $[Ca^{2+}]$ cyt. A simultaneous rapid increase of both $[Ca^{2+}]$ cyt and $[Ca^{2+}]n$ was obtained, but the increase in $[Ca^{2+}]n$ was delayed with respect to the cytosolic changes. Wind and cold also induced the expression of a CaM gene (NgCAM-1) and a comparison between calcium dynamics with gene expression indicated that wind-induced expression depends on nuclear calcium signalling, while cold shockinduced expression is mediated by cytosolic calcium elevation. When tobacco seedlings were subjected to decreased osmolarity of the culture medium, an increase of $[Ca^{2+}]cyt$ was followed by an increase of $[Ca^{2+}]cyt$ was followed by an increase of $[Ca^{2+}]cyt$ (Pauly et al. 2001). On the other hand, increase of the osmolarity of the culture medium elicited a smaller change in $[Ca^{2+}]cyt$, but did not modify the biphasic shape of the cytosolic response, and did not affect the $[Ca^{2+}]n$.

Bimodal long-lasting changes in [Ca²⁺]cyt were also found when proteinaceous elicitors were used (Lecourieux et al. 2005). Cryptogein, a polypeptidic elicitor induced a transient peak in [Ca²⁺]cyt after 5 min followed by a sustained increase for at least 2 h. On the other hand, a weak increase in $[Ca^{2+}]n$ was followed by a substantial increase for 1 h, with a maximum at 23 min, and thereby, peaked much later than the [Ca²⁺]cyt. It was suggested that a 1-s delay between the cytosolic and nuclear responses is enough to exclude the possibility of just a diffusion of Ca²⁺ from the cytosol to the nucleus of the cell (Meyer et al. 1995). Xiong et al. (2008) showed that sphingolipid metabolites selectively elicited an increase in nuclear calcium concentrations in a dose-dependent manner both in cell suspensions and in isolated nuclei of tobacco BY-2 cells. Thus, the nuclear calcium changes were independent of the cytosolic compartment.

In pollen tubes, calcium gradients in the tip were oscillating in the cytosol during normal polarized growth, but no measurable changes in $[Ca^{2+}]n$ were found (Watashiki et al. 2004). Thus, the available results so far demonstrate that different calcium signals proceed in different cell compartment and that the $[Ca^{2+}]n$ is not directly linked by diffusion to a change in $[Ca^{2+}]cyt$ (Mazars et al. 2009). Proteins that are important for the amplification or diminishing of calcium signals are found to shuttle to and from the nucleus. Rodriguez-Concepcion et al. (1999) showed that CaM 53 could be localized reversible

to the nucleus and plasma membranes. When localized to the plasma membrane it was isoprenylated at the C-terminus domain, but inhibition of isoprenoid biosynthesis caused an accumulation of CaM to the nucleus. They also showed that CaM was associated with the plasma membrane after light exposure, then transferred to the nucleus in darkness. It has been suggested that other calcium-binding or CAM-binding proteins, as well as calcium-dependent enzymes and different transcription factors are present in the nucleus and regulate the activity (Mazars et al. 2009). Calcium signals may also be generated via G-coupled receptors localized to the nucleus without involvement by the plasma membrane. It is suggested that the nucleus can produce inositol 1,4,5 triphosphate (IP3) and IP3 receptor-mediated Ca²⁺ release (Gomes et al. 2006). Nuclear activities are probably regulated by a cross-talk between ROS and calcium (Mazars et al. 2010).

5 Calcium Signalling Under Anoxia

Higher plants are strict aerobic organisms, and thus, directly depend on molecular oxygen for their respiration and other metabolic processes. Nevertheless, very often they suffer from oxygen shortage in different agricultural, horticultural and industrial areas. Availability of oxygen has a strong influence on distribution of plant species in ecosystems and severe economical impact. Yield loss due to deficiency of oxygen could reach up to 50% (Dennis et al. 2000). The lack of oxygen usually results from excess of water, ice crust and soil compaction. Waterlogging of rhizosphere and partial flooding of the aboveground parts of plants lead to gradual hypoxia (deficiency of oxygen) and complete submergence brings about anoxia (total absence of this gas).

5.1 Anoxic Injury in Plants

The lack of molecular oxygen leads to inhibition of aerobic respiration, which in turn ultimately causes energy starvation. After the switch from aerobic condition, ATP level in the cell exhausts within 1-2 min (Drew 1997). To provide ATP for energizing cellular metabolism, glycolysis is passed into fermentation by activation. Overconsumption of respirable sugars by these processes aggravates energy starvation, particularly in hypoxia-susceptible species (Vartapetian and Jackson 1997). On the other hand, stimulation of lactic fermentation and lack of ATP to energize transport ATPases account for cytosol acidification (Drew 1997). Alcoholic fermentation generates a toxic intermediate (acetic aldehyde) and end-products (ethanol) that together with acidosis self-poison plant under oxygen deficiency (Vartapetian and Jackson 1997; Gibbs and Greenway 2003; Bailey-Serres and Voesenek 2008). In natural habitats flooding leads to low soil-redox potential and production of reduced substances including Mn²⁺, Fe²⁺, H₂S, NO₂⁻ along with the intermediates of carbon metabolism such as methane, ethane, ethylene, acetylene, acetic and butyric acid (Drew 1997 and references wherein). Moreover, reestablishment of normoxic conditions triggers oxidation of these substances and synthesis of reactive oxygen species (ROS) resulting in post-anoxic injury.

5.2 Oxygen Deficiency Tolerance Mechanisms in Plants

Capacity to survive under oxygen deprivation depends on a number of developmental, morphological and metabolic adaptations in plants. Imposition of hypoxia accelerates growth of shoot axial organs, stimulates formation of adventitious roots and aerenchyma particularly in tolerant plant species, as a result, the shoot actively transports oxygen to a flooded root (for review, see Crawford and Braendle 1996; Drew 1997; Vartapetian and Jackson 1997; Kende et al. 1998; Sauter 2000; Gibbs and Greenway 2003; Bailey-Serres and Voesenek 2008). Simultaneous shifts occur in the metabolism, particularly underneath lack of oxygen. Metabolic adaptations mainly include avoidance of energy starvation, prevention of toxicity of anaerobic intermediate and end products, besides, post-anoxic injury, disposal of cytosol acidification

and use of alternative electron acceptors (like nitrate, nitrite, unsaturated lipids, etc.) (Crawford and Braendle 1996; Drew 1997; Vartapetian and Jackson 1997; Gibbs and Greenway 2003; Greenway and Gibbs 2003; Bailey-Serres and Voesenek 2008). Hypoxia-tolerant plants are notable for maintenance of their cell ultra structure (Vartapetian and Jackson 1997), membrane stability (Crawford and Braendle 1996) and synthesis of anaerobic stress proteins (Vartapetian and Jackson 1997; Greenway and Gibbs 2003). Most of the anaerobic stress proteins belong to enzymes of the glycolytic or fermentative pathways, carbohydrate mobilization and nitrogen metabolism (Vartapetian and Jackson 1997; Greenway and Gibbs 2003). All these metabolic adaptations mutually allow tolerant plants to generate sufficient amount of energy, maintain mineral uptake and even to grow in total absence of oxygen.

5.3 Perception of Anoxic Stress in Plants

The mechanism(s) of oxygen sensing by plant cell and sensor itself are still to be elucidated. Oxygen depletion could be detected directly by binding molecular oxygen (direct sensing) or recognized by altered cellular metabolism, i.e. indirect sensing (Bailey-Serres and Chang 2005; Licausi and Perata 2009). Only prokaryotes have direct oxygen sensors, including heme-containing protein kinases, Fe/S cluster- and SH-containing transcription factors, which induce anaerobic gene expression involved in aerotaxis (Green et al. 2009). None of the eukaryotes are reported to possess direct mechanism of oxygen sensing. In fungi and animals, anaerobic genes are regulated at transcriptional level. Hap-transcription factors involved in oxygen sensing (yeast) and hypoxiainducible factor (HIF) are widespread throughout the animal kingdom (Bailey-Serres and Chang 2005; Bailey-Serres and Voesenek 2008; Licausi and Perata 2009). Activity of these factors is regulated by the oxygen level, production of ROS and redox state of the cell. Furthermore, intracellular pH, ATP and sugar level are good candidates for indirect sensing signal.

5.4 Cytosolic Calcium Signalling in Plants Under Anoxia

The main effect of anoxia sensing in eukaryotic cells is an increase in $[Ca^{2+}]$ cyt. There are two major hypotheses to explain this reaction in animal physiology (Lahiri et al. 2006; Fähling 2008) (Fig. 15.1). The "mitochondrial" hypothesis states that mitochondrial electron-transport chain is retarded by depletion of oxygen, produces ROS and leads to a Ca^{2+} release into the cytosol from the mitochondria and other intracellular compartments. While as the "plasma membrane" or "NAD(P)H-oxidase" hypothesis concerns involvement of ROS generated by NAD(P) H-oxidase leading to suppression of outward

plasmalemma K^+ channels, a depolarization of the membrane and a Ca^{2+} influx from extracellular Ca^{2+} stores.

Subbaiah et al. (1994a) first reported an elevation of $[Ca^{2+}]cyt$ in maize suspension-cultured cells, which was completely reversible after a few seconds of reoxygenation. Treatment with Ca^{2+} -channel blockers prevented the anoxic induction of ADH1 and SuS1 genes in plant cell and post-anoxic seedling survival (Subbaiah et al. 1994b; Subbaiah 2009). In maize cells, the elevation of $[Ca^{2+}]cyt$ within the first minute of anoxia derived from efflux of Ca^{2+} from intracellular stores, as it was significantly inhibited by ruthenium red (RR, inhibitor of intracellular membrane Ca^{2+}/H^+ -antiporter), but not affected by



Fig. 15.1 A proposed model for Ca^{2+} signalling under anoxia. The "mitochondrial" pathway starts with retardation of the mitochondrial electron-transport chain by depletion of oxygen. It stimulates production of ROS that leads to a Ca^{2+} release into the cytosol from the mitochondria and other intracellular compartments. The "plasma membrane" or "NAD(P)H-oxidase" pathway is switched on by NAD(P) H-oxidase-dependent synthesis of ROS under oxygen shortage. This leads to a Ca²⁺ influx from extracellular Ca²⁺ stores. Increase of cytosolic Ca²⁺ concentration activates CDPKs, and ROS production stimulates a MAPK cascade, leading to activation of transcription factors and gene regulation EGTA (Ca^{2+} chelator) and various inhibitors of plasma membrane channels (Subbaiah et al. 1994a). Moreover, subsequent research showed that elevation of Ca^{2+} in the cytosol of maize cell originates from mitochondria (Subbaiah et al. 1998) and could be induced by production of ROS by the mitochondrial electron transport chain under lack of oxygen, according to "mitochondrial" hypothesis (Rhoads et al. 2006).

On the other hand, a transient spike and prolonged elevation of [Ca²⁺]cyt in transgenic apoaequorin-expressing Arabidopsis seedling upon imposition of anoxia depends on external as well as internal stores, since the downstream anaerobic gene expression is partially inhibited by EGTA, Gd³⁺, La³⁺ (non-specific inhibitors of plasma membrane Ca2+ channels) and RR (Sedbrook et al. 1996) supporting both the hypothesis. Our results showed an ultimate importance of both external and internal Ca²⁺ stores for anoxic signalling in rice, whereas the hypoxia-intolerant wheat does not require external sources for that purpose (Yemelyanov et al. 2011). Ca²⁺ was shown to be important for anoxic induction of ADH in Arabidopsis and rice seedlings (Dolferus et al. 1997; Tsuji et al. 2000). Aurisano et al. (1995, 1996) demonstrated involvement of Ca2+cyt, plasma membrane Ca2+ channels, CDPKs, CaM and G-proteins in anoxic signal transduction leading to accumulation of GABA and other amino acids in rice. Influx of Ca²⁺ via plasma membrane channels in Arabidopsis and rice is closely related with production of ROS on the cell surface. Blokhina et al. (2001) revealed production of ROS in the apoplast of different monocot species under anoxia. In Arabidopsis seedlings, induction of ADH gene depends on H₂O₂. Moreover, treatment of seedlings with diphenyleneiodonium, an inhibitor of ROS production by NAD(P)H-oxidase, blocked hypoxia-induced ADH activation (Baxter-Burrell et al. 2002). This links ROS production by NAD(P)H-oxidase, ADH induction and Ca²⁺ signalling. A scheme of Ca²⁺ signalling in plants under anoxia is shown in Fig. 15.1. In maize and wheat plants, accumulation of $[Ca^{2+}]$ cyt occurs mainly via "mitochondrial" pathway,

whereas in *Arabidopsis* and rice it passes through both "mitochondrial" and "plasma membrane" pathways.

6 Cold Temperature Stress Signalling

Many plant species have the possibility to enhance freeze tolerance after exposure to non-freezing temperatures. Cold acclimation is associated with alterations in plasma membrane lipid composition, increase in proline and sugar contents, synthesis of new polypeptides and changes in the mRNA populations (Steponkus and Lynch 1989; Guy 1990; Lin et al. 1990; Monroy et al. 1993; Uemura et al. 1995).

6.1 Cold-Induced Changes in Plasma Membrane Lipid Composition

The plasma membrane lipid composition is changed under cold stress in order to stabilize the membranes against freeze injury. Acclimation results in more unsaturated fatty acids causes a drop in the transition temperature. In *Arabidopsis*, maximal freeze tolerance was induced after 1 week at 2°C (Uemura et al. 1995). During that time, the proportions of phospholipids in the plasma membranes increased, while cerebrosides and free sterols decreased. Moreover, the diunsaturated species of phosphatidylcholine and phosphatidylethanolamine increased. Some proteins, like dehydrins and lipocalin may help the plant to prevent damage to the plasma membrane during freezing (Uemura et al. 2006).

6.2 Calcium Is Involved in Acclimatization to Cold Temperature

Calcium has a role in cold-induced changes in protein phosphorylation, gene expression and development of freeze tolerance (Monroy et al. 1993; Knight et al. 1996; Tähtiharju et al. 1997; Viswanathan et al. 2006). In the freeze-tolerant alpine plant, Chorispora bungeana, chilling induction at 0°C increased the calcium contents (Fu et al. 2006) and these Ca²⁺ levels were different in various tissues and organs. Knight et al. (1996) reported that inhibition of calcium influx caused a partial inhibition of cold-dependent kin1 expression in Arabidopsis. It was also shown that inhibitors of calcium channels, CaM action or protein kinases, inhibited development of freezing tolerance. In alfalfa plants, calcium elevation caused by low temperature induced expression of two cas (cold acclimation-specific) genes: CAS15 and CAS18 (Monroy and Dhindsa 1995). In Arabidopsis, cold stress induced the KIN1 and KIN2 genes but KIN2 mRNA accumulated to a higher level than KIN1 mRNA under acclimation (Kurkela and Franck 1990). Many cold-regulated genes have one or several copies of the DRE/ CRT cis-element (dehydration responsive element/C repeat) in their promoters. Other transcription factors bind to this element and activate transcription of downstream genes (Zhu 2001). The CBF/DREB1 genes are also induced directly by a low temperature and the induction precedes that of genes containing the DRE/CRT ciselement. Thus, a network of multiple signalling pathways is involved in cold stress response in plants, some of them are also induced by salt and drought stresses in a complicated way. It was proposed by Zhu (2001) that low temperature specifically induces the transcription of the CBF/ DREB1-based cascade. Calcium is required for full expression of the CRT/DRE controlled COR6 and KIN1 genes of Arabidopsis (Monroy and Dhindsa 1995; Knight et al. 1996). In the regulation of cold-responsive genes and freezing tolerance, also a CBF-independent pathway exists (Viswanathan et al. 2006). The transcription factors, HOS9 (a homeodomain type) and HOS10 (a myeloblastosis type), play an important role in this system.

Calcium is important not only for gene expression during cold acclimation, but also has an effect on resealing the membranes after cold stress (Yamazaki et al. 2008). In experiments with *Arabidopsis* protoplasts, extracellular calcium increased tolerance to electroporation that otherwise punctures the membrane. An antibody against a homolog of synaptotagmin, SYT1, inhibited the calcium-dependent freezing tolerance in *Arabidopsis* leaf protoplasts and the substance is a calcium sensor that causes exocytosis. Moreover, this inhibition indicates that the puncture allowing the antibody to enter into the cytoplasm occurs during freeze/thawing. The authors suggested that calcium-dependent freezing tolerance results from membrane resealing and that SYT1 is involved in this mechanism.

6.3 Calcium Signalling Induced by Cold or Chilling Stresses

In transgene *Arabidopsis*, wherein aequorin was targeted to different types of root cells, as well as to the whole plant, cold water addition caused a transient rise in [Ca²⁺cyt] (Kiegle et al. 2000). The maximum peak occurred at approximately 15–20 s in all cell types. However, the increase in [Ca²⁺cyt] was lower in the root cells, compared with shoot cells. The highest elevation was obtained in the deepest cell type, the pericycle, and the lowest peak level in the elongation zone. Hence, concluded, the size of the perceived agonist does not decide the magnitude of the calcium response, instead, there is a cell-specific response.

Calcium dynamics under chilling stress were also investigated in single mesophyll protoplasts from tomato plants loaded with the calcium specific dye Fura2, AM (Sebastiani et al. 1999). The protoplasts subjected to chilling stress were kept in a temperature-controlled (10-15°C) perfusion chamber. The results showed that different kinetics in [Ca²⁺cyt] occurred, depending on different resting calcium levels. In 84% of the investigated protoplasts, there was an increase in [Ca2+cyt] followed by a maximum increase obtained after 10-20 s. In 21% of the reacting protoplasts, the maximum [Ca²⁺cyt] was also obtained after 10-20 s, thus corroborating results from root cells of Arabidopsis (Kiegle et al. 2000); in 11% of the protoplasts both the increase and decrease in [Ca²⁺cyt] were slower; and in 32% a constant increase of [Ca2+cyt] was obtained 1 min after start of temperature decrease. When the resting calcium concentration was higher



Fig. 15.2 Cytosolic calcium changes, detected by Fura 2-AM and fluorescence intensity ratio 340/380 nm, in a single wheat protoplast upon cold shock. (a) First addition of cold (5°C) solution. (b) Second addition of cold

than normal, the increase in $[Ca^{2+}cyt]$ was constant. In these experiments, the cooling rate was constant, as the perfusion system was used. Therefore, it is likely that different cells have different competence and ability to maintain calcium homeostasis. A sustained high $[Ca^{2+}cyt]$ is implicated in apoptosis and in hypersensitive responses to pathogens (Levine et al. 1996).

When a cold solution (5°C) was added to mesophyll protoplasts of wheat leaves (Fig. 15.2), much larger changes in [Ca²⁺cyt] were obtained compared to chilling sensitive tomato protoplasts, in which a perfusion system was used (Sebastiani et al. 1999). A second addition of cold solution to the same wheat protoplast induced different [Ca²⁺cyt] kinetics. The first addition of cold solution gave a transient prolonged increase of

solution to the same protoplasts. (c) Reaction in the presence of erythrosine B (Lindberg and Sebastiani, unpublished results)

[Ca²⁺cyt], much smaller than that induced by a second addition (Fig. 15.2a). The second addition gave a peak response and thereafter; a prolonged calcium increase was observed (Fig. 15.2b). These results cannot be compared with results shown by Knight et al. (1996), since these authors report a mean of 3 or more cold additions at time zero. When the wheat protoplasts were treated with erythrosine B, an inhibitor of Ca2+ ATPase in the ER and plasma membrane, the calcium transient was prolonged and the magnitude was much higher than without the inhibitor (Fig. 15.2c). Thus, it is likely that the Ca²⁺ ATPase is involved in the signalling by pumping Ca²⁺ out from the cytosol. It has been proposed that [Ca2+cyt] signatures are modified by previous experience, indicating that the plant has a calcium "memory"

(Knight et al. 1996). The magnitude of $[Ca^{2+}cyt]$ elevation elicited by wind becomes progressively smaller upon repeated stimulation and for some stimulus several hours are needed for a second reaction to take place (Tuteja and Mahajan 2007). The calcium signature can also be modified during a second exposure to a stress as shown in Fig. 15.2. Moreover, the magnitude of $[Ca^{2+}cyt]$ increase can be changed by prior in vivo exposure to a contrasting stress. These observations imply a cross talk between the signalling cascades.

In response to cold shock, an immediate rise in [Ca²⁺cyt] was obtained in both chilling-sensitive tobacco as well as in chilling-tolerant Arabidopsis (Knight et al. 1996). In Arabidopsis, both EGTA and lanthanum caused a partial inhibition of [Ca²⁺cyt] and of cold-dependent KIN1 gene expression. To investigate if the vacuole was involved in the calcium signalling, aequorin was targeted to the cytosolic face of the tonoplast. As a result, a higher peak of calcium elevation was obtained in the cytosol than in tonoplast microdomain. In addition, the elevation in the microdomain, [Ca²⁺mic], was maintained for a longer time than [Ca²⁺cyt]. Pre-treatment with neomycin or lithium, which interferes with phosphoinositide cycling, diminished the calcium reactions, showing that some efflux of calcium occurred from the vacuoles. The magnitude of the cold-shock induced [Ca²⁺cyt] response was similar in Arabidopsis and tobacco, although it was more prolonged in tobacco. Upon repetitive addition of cold solution, the response after 3 and 10 min was weaker in both species. However, tobacco was able to recover its ability to respond fully to cold shock 30 min after the initial shock, whereas Arabidopsis was not. Another difference between the species was that they responded in different ways to cold shock after acclimation. Only in chilling-tolerant Arabidopsis, cold acclimation altered the signature of [Ca²⁺]cyt, so that cold shock caused a reduced peak but a prolonged profile. The different mechanisms in stress response could depend on the different sensitivity of Arabidopsis and tobacco to chilling stress. Cold shocks of different amplitude applied to protoplasts of freeze-sensitive olive tree caused transient increases in [Ca²⁺cyt] that

differed in non-acclimated and acclimated protoplasts (D'Angeli et al. 2003). Upon repeated cold-shock treatment and by using non-severe rate changes in temperature, the transient increases in $[Ca^{2+}]cyt$ fall, while in the acclimated protoplasts, the $[Ca^{2+}]cyt$ elevations were further reduced.

Temperature sensing in Arabidopsis depends both on the cooling rate (Plieth et al. 1999) and the final temperature to which cooling occurs (Knight 2002). Plieth (1999) presented a mathematical model of how cold is sensed by a plant based on a passive influx across the plasma membrane and an active efflux by a pump. A single peak in [Ca²⁺cyt] obtained with a cold shock, specifically decreases, at a very fast temperature change but at low cooling rate, the response lacks the initial peak more or less. The pump slows down at low temperature leading to a second slow phase of the increase in [Ca²⁺cyt]. A biphasic response to a single cooling step is thus obtained, when the sensitizing action by the temperature change on the pumps and channels is of equal magnitude (Plieth 1999). By applying patch clamp technique to mesophyll cells of Arabidopsis, Carpaneto et al. (2007) showed that cold induced a rapid increase in [Ca²⁺cyt], and that the influx of calcium could occur through non-selective cation channels.

6.4 Cold Shock Induces Change in the Membrane Potential

A drop in temperature can change the transmembrane potential. Krol et al. (2004) showed that the obtained transient depolarization of membrane potential in mesophyll cells of *Arabidopsis*, *Helianthus* and *Vicia*, induced by temperature decreases, depended on calcium influx both from apoplast and internal stores. It was verified later that the cold-induced depolarizations depended on calcium influx (Carpaneto et al. 2007). Verapamil, a calcium channel blocker, caused significant suppression of the cold-induced potential changes. Since the presence of CaM antagonists prolonged the repolarization, this could be attributed to activation of CaM-dependent Ca²⁺-ATPases (Krol et al. 2004). It was reported that



Fig. 15.3 A proposed model for cold-stress signalling in plants. Cold and chilling stress induce a transient or prolonged increase in $[Ca_{cyt}^{2^+}]$. Calcium influx is mainly mediated by non-selective-cation channels in the plasma membrane (PM) (Carpaneto et al. 2007), but also by channels in the tonoplast or other endomembranes (Knight et al. 1996; Krol et al. 2004). Calcium influx causes a depolarization of the PM potential, and then a re-polarization (Krol et al. 2004). Erythosine B is supposed to inhibit the PIIA ATPases (ECAs) channels in the ER, but may also inhibit PIIB ATPases (ACAs) in the PM. ACA8 and ACA10 were differently

this enzyme is involved in low temperature response in rice (Martin and Busconi 2001). A suggested model for cold signalling in plants is shown in Fig. 15.3.

7 Salinity; Sodium and Osmotic Stress Signalling

Soil salinity is a major environmental hazard worldwide as more than 40% of the earth is arid or semi-arid and most of the planet's water is saline. Currently, more than 6% of the world's land which exceeds 20% if only the irrigated area is considered, is affected by varying degrees of

regulated by cold stress (Schiott and Palmgren 2005). The ACA type channels contain a Ca–CaM binding site and a serine-residue phosphorylation site. Their activity can be modulated by Ca_{cyt}^{2+} , either through activation upon binding CaM, or by inhibition upon phosphorylation by CDPK (Hwang et al. 2000). The signals from CaM also induce mRNA and gene expressions, either directly, or by CaM-binding to specific transcription factors (Tuteja and Mahajan 2007). The *CBF/DREB1* genes are induced by low temperature and the induction precedes that of genes containing the *DRE/CRT cis*-element (Zhu 2001; Zhu et al. 2007)

soil salinity (Flowers and Yeo 1995; Munns 2002; Kader and Lindberg 2008). However, nearly 50% of the irrigated land in the arid and semi-arid regions of the world is facing most serious salinity problems.

7.1 Salinity Stress Has an Impact on Agricultural Productivity

Salinity stress adversely affects agricultural productivity by decreasing the crop yield in many ways. Furthermore, the saline area increases day by day due to sea-level rise and, thus, have profound harmful effects on agricultural productivity in many countries of the world. Long time ago, Buringh (1979) estimated that among 10 hectares, 3 hectares of arable land are lost in every minute due to soil salinization. In 2002, FAO reported that about 20-30 million hectares of irrigated land are seriously damaged by the increased levels of salts, moreover, 0.25-0.50 million hectares are lost from production every year (Martínez-Beltran and Manzur 2005). The disastrous consequence of this increasing salinity stress together with the growing world population is certainly threatening the future stable global food availability. Agricultural productivity in future will depend mostly on our ability to identify or develop salttolerant crop plants and to grow them in rapidly increasing salt-affected lands.

7.2 Salinity Stress Injury in Plants

The salinity in nature imposes two primary harmful effects on plants: one is osmotic stress and the other, ionic toxicity. Due to the presence of high salt, salinity stress increases the osmotic pressure in the soil solution over the osmotic pressure in plant cells. As a result, plant loses its ability for uptake of water and minerals, especially the uptake of K⁺ and Ca²⁺ (Glenn et al. 1997; Munns et al. 2006). Inhibition of plant growth by high amounts of Na⁺ and Cl⁻ is one of the main deleterious effects of salinity stress. When present in excess amount, Na⁺ and Cl⁻ ions enter into the plant cells and can exert toxic effects on cell membranes and metabolic activities in the cytosolic part of the cell (Greenway and Munns 1980; Hasegawa et al. 2000; Zhu 2001). The resultant effect of osmotic stress and ionic toxicity may lead to secondary effects in plants such as decreased cell expansion, production of assimilate and membrane functions, decreased cytosolic metabolism with raised production of ROS.

7.3 Salinity Stress Tolerance Mechanisms in Plants

As paleontological and molecular evidence suggest that some 500 million years ago, the embryophytes (terrestrial plants that are not algae) were evolved from the Streptophyta (Raven and Edwards 2001). The evolution of salinity tolerance mechanisms in halophytic plants has recently been reviewed by Flowers et al. (2010a). Though physiological foundations of salinity tolerance are present in all plants and show a very wide range of adaptability to salinity stress. For example, glycophytes like chickpea is very sensitive and suffer toxicity at 25 mM NaCl (Flowers et al. 2010b), whereas halophytes can tolerate salinity concentration as high as 1,000 mM (Khan et al. 2005). The recent advancement in molecular biology research is uncovering the mechanisms of salinity stress tolerance including the key genes involved in the molecular networks and the signalling cascade that mediates plant responses to salinity stress. Plants need different tolerance mechanisms to be adopted, since salinity stress elicits two different adverse effects like osmotic stress and ionic toxicity. To deal with the ionic toxicity, under salinity stress, the key mechanisms for tolerance are a diminished toxic ion uptake into the cytosol, the ability to limit the entry of these toxic ions into the transpiration stream, the ability to regulate transpiration in the presence of these toxic ions and compartmentalization of Na+ and Cl- ions into the apoplast/vacuole (Blumwald 2000; Tester and Davenport 2003; Kader and Lindberg 2005, 2008; Munns and Tester 2008; Flowers and Colmer 2008; Flowers et al. 2010a). Compartmentalization of toxic Na⁺ into the vacuole is advantageous, since it is no more toxic for the cell, and also a benefit for growth and adjustment of the osmotic potential (Flowers and Läuchli 1983; Zhu 2003; Subbarao et al. 2003; Rodríguez-Navarro and Rubio 2006). Jou et al. (2006) showed that excess Na⁺ also can be compartmentalized in ER and Golgi bodies. An important tolerance mechanism is also a plant's capability to increase and decrease the uptake of potassium (K⁺), and Na⁺ into the cytosol, respectively, under high sodium concentration. Expression analyses of transporter genes for K⁺ and Na⁺ transporters OsHKT1 and OsHKT2 showed that they were differently expressed in tolerant and sensitive rice cultivars (Kader et al. 2006).

To obtain osmotic homeostasis, the synthesis of compatible organic solutes, such as glycine betaine, mannitol, pinitol, proline, sorbitol, sucrose and trehalose in the cytosol, is of great importance (Bohnert and Jensen 1996; Chen and Murata 2002; Zhu 2002; Zhang et al. 2004; Chinnusamy et al. 2005; Taiz and Zeiger 2006; Liang et al. 2009).

7.4 Perception of Salinity Stress in Plants

Cellular perception of salinity stress by plants is prerequisite to start the activation of the whole cell-signalling cascade. This begins with an elevation of [Ca²⁺]cyt and ends with different tolerance mechanisms activated in the plant. Like other stresses, salinity stress (both osmotic stress and ionic toxicity) is perceived in plants at the cell membrane, either extracellular or intracellular by a protein spanning the plasma membrane and/or an enzyme within the cytosol (Zhu 2003). Under salinity stress, a low K⁺ level in the cytosol may also lead to cytosolic calcium signals (Luan et al. 2009). For sensing the osmotic component of salinity stress, probably several sensors are involved (Urao et al. 1999; Reiser et al. 2003; Tamura et al. 2003; Boudsocq and Lauriere 2005; Tran et al. 2007; Wohlbach et al. 2008). A substantial progress in understanding the signal transduction under Na⁺ toxicity was made by identification of the SOS pathway in Arabidopsis (Zhu 2002). In a recent review, Luan et al. (2009) suggested that CBL (calcineurin B-like proteins)-CIPK (CBL-interacting protein kinase) pathways regulate Na⁺ transport in plants and thus confer salinity tolerance. The CBLs also appear to be an important group for conferring salinity tolerance through enhanced K⁺ uptake under salinity stress (Luan et al. 2009). As shown in Fig. 15.4, the salinity tolerance mechanisms in plants entails SOS3, a Ca²⁺ sensor in the cytosol, that reads the changes in [Ca2+]cyt under salinity stress and specifically binds Ca²⁺ followed by protein interaction through SOS2 protein kinase. Thereafter, the SOS3-SOS2 complex in turn activates the plasma membrane Na⁺/H⁺ antiporter, the SOS1 protein,

and re-establishes the Na⁺ homeostasis of the cells. An elevation of [Ca²⁺]cyt is also detected by CBL10, interaction with SOS2 might trigger tonoplast Na⁺/H⁺ antiporter to transport Na⁺ from cytosol to vacuole. Furthermore, the increase in [Ca²⁺]cyt can also be perceived by CBL1 and CBL9, which then bind to CIPK23 and interact with the C terminus of AKT1. In this way the AKT1 channel is activated causing an increased K⁺ uptake into the cell, hence confer salinity tolerance. However, it is still necessary to clarify how Na⁺ toxicity is sensed by the plant cell. It was suggested that SOS1 protein, with its long C-terminal tail in the cytosol, might sense Na⁺ (Zhu 2003; Zhang et al. 2004; Shabala et al. 2005). Kader et al. (2007) showed that, for Na⁺ sensing in rice protoplasts, Na⁺ primarily must enter into the cytosol. The results corroborate with the earlier suggestion that the cytosolic tail of the SOS1 protein might sense Na⁺. Conversely, in the halophytic plant quince, Na⁺ entry in to the cell was not necessary for the shift in cytosolic Ca²⁺ (D'Onofrio and Lindberg 2009). Therefore, the question still remains to be answered concerning the sensors for Na⁺ toxicity in plants, and if they differ in different species, and/or in salinity sensitive as well as salinity tolerant cultivars.

7.5 Cytosolic Calcium Signalling in Plants Under Salinity Stress

It is clear, the salinity stress perception, irrespective of how the stress is perceived, triggers an intracellular signalling cascade starting with the elevation of secondary messenger molecules like calcium in the plant cytosol [Ca²⁺]cyt. Many studies were done to measure the [Ca²⁺]cyt changes in cells, organs or intact plants under salinity stress by use of different techniques. Fluorescence microscopy measurements were performed in root hairs (Halperin et al. 2003) and in individual mesophyll protoplasts (Kader et al. 2007; D'Onofrio and Lindberg 2009) Measurements in intact whole plants were made by aequorin luminescence detection (Knight et al. 1997; Gao et al. 2004; Henriksson and Henriksson 2005, Tracy et al. 2008). These studies suggest that the "signature" of [Ca²⁺]cyt



Fig. 15.4 A proposed model for calcium signalling under salinity stress. Salinity stress (both osmotic stress and ionic toxicity) is perceived in plants at the cell membrane, either extracellular or intracellular by a transmembrane protein, or within the cytosol by enzymes (Zhu 2003). Several osmo-sensors are involved in sensing the osmotic stress (Urao et al. 1999; Reiser et al. 2003; Tamura et al. 2003; Boudsocq and Lauriere 2005; Tran et al. 2007; Wohlbach et al. 2008). Na⁺ stress is sensed by the SOS pathway (Zhu 2002). The CBL (calcineurin B-like proteins)–CIPK (CBL-interacting protein kinase) pathways regulate Na⁺ transport in plants and thus confer

change, e.g. the amplitude, duration and frequency, is very important for transferring of specific downstream reactions leading to stress tolerance.

7.6 The Signature of Cytosolic Calcium Differs

The change in $[Ca^{2+}]$ cyt varies with species, cell type or tissue type and also with different techniques used (Cramer and Jones 1996; Kiegle et al. 2000; Kader et al. 2007; Tracy et al. 2008; D'Onofrio and Lindberg 2009). The change in

salinity tolerance (Luan et al. 2009). The SOS3, Ca²⁺ sensor in the cytosol, reads the changes in calcium level and binds Ca²⁺. This protein interacts with a SOS2 protein kinase, and then the SOS3–SOS2 complex activates the SOS1 protein, a plasma membrane Na⁺/H⁺-antiporter, thereby re-establishing the Na⁺ homeostasis in cells. The elevation of $[Ca^{2+}]_{cyt}$ is also detected by CBL10, which in interaction with SOS2 might trigger tonoplast Na⁺/H⁺ antiporter to transport Na⁺ from the cytosol into the vacuole. The increase in $[Ca^{2+}]_{cyt}$ can also be perceived by CBL1 and CBL9, which then bind to CIPK23 and activate the AKT1. This leads to increased K⁺ uptake into the cell

[Ca²⁺]cyt activates different downstream reactions, such as up-regulation or down-regulation of different genes. The reactions may also change with the particular stress (Kiegle et al. 2000), the stress development rate (Plieth et al. 1999; Tracy et al. 2008; D'Onofrio and Lindberg 2009), preexposure to the stress (Knight et al. 1997) and the tissue type (Kiegle et al. 2000; Tracy et al. 2008). Upon application of 100 mM NaCl to root cells of *Arabidopsis* (Cramer and Jones 1996; Halperin et al. 2003), or to corn root protoplast (Lynch and Läuchli 1988), a decrease in [Ca²⁺]cyt was obtained. On the other hand, most studies show an increase in [Ca²⁺]cyt upon salinity stress (Bittisnich et al. 1989; Lynch et al. 1989; Knight et al. 1997; Halfter et al. 2000; Kiegle et al. 2000; Knight 2000; Zhu 2001; Gao et al. 2004; Henriksson and Henriksson 2005; Kader et al. 2007; D'Onofrio and Lindberg 2009; Kader and Lindberg 2010).

7.7 Plant Roots and Shoots Signal Differentially Under Sodium Toxicity and Osmotic Stress

Contrasting results are reported whether osmotic stress increases or decreases $[Ca^{2+}]cyt$ (Cramer and Jones 1996; Knight et al. 1997; Kiegle et al. 2000; Kader et al. 2007). In *Arabidopsis* root cells, osmotic and ionic stresses induced different shifts in $[Ca^{2+}]cyt$ and the heterogenous $[Ca^{2+}]cyt$ changes were found only in the root (Tracy et al. 2008). Also in experiments with rice and quince protoplasts, different $[Ca^{2+}]cyt$ changes were induced under sodium and osmotic stresses (Kader et al. 2007; D'Onofrio and Lindberg 2009). A proposed model for calcium- and pH-signalling under salinity stress is reviewed in Kader and Lindberg (2008, 2010). A simplified model is shown in Fig. 15.4.

8 Aluminium and Heavy Metal Stress Signalling in Plants

8.1 Aluminium Toxicity to Plants

Aluminium (Al) toxicity in plants is a serious factor, limiting crop production in acidic soils and affecting up to 40% of the world's arable soils (Haug 1984; Foy 1984). When the soil pH decreases below 5, Al³⁺ is dissolved in the soil, causing harmful effects on plant roots. At pH 4, the dominating species is $Al(H_2O)^{3+}$ while at higher pH, $Al(OH)^{2+}$, $Al(OH)^{2+}$, $Al(OH)^{3-}$ and $Al(OH)^{4-}$ are present, besides sulphate complexes and polynuclear species (Lindsay 1979). When roots are subjected to Al, they become stunted and damaged with poor root hair development (Clarkson 1965). The uptake of water and minerals is severely inhibited. At a low pH, Al mainly

binds to the root apex and inhibits root elongation (Ryan et al. 1993; Kochian 1995; Matsumoto 2000: Barcelo and Poschenrieder 2002). Aluminium affects the transmembrane potential of root cells and inhibits ATPase activities. After cultivation of sugar beets in the presence of low pH and/or AlCl₃, the transmembrane potential, PD, between the vacuole and external medium, PDv, of root cells was largely depolarized (Lindberg et al. 1991). Since the effect of dinitrophenol was negligible, it was suggested that Al interacts with the active component of the PD. This was confirmed by experiments showing that Al inhibits the plasma membrane ATPase activity (Lindberg and Griffiths 1993) as well as proton transport (Matsumoto 1988). Lipid analysis of sugar beet plasma membranes showed that Al treatment during cultivation caused an increase in the ratio of phosphatidylcholine: phosphatidylethanolamine (Lindberg and Griffiths 1993). The lipid changes correlated with the observed change in the Km for the Mg ATPase, and, hence, it was concluded that Al could bind to the membranebound enzyme and/or modify the lipid environment. The inhibition of the ATPase activity causes a reduced uptake of minerals. In the soil, precipitation of phosphate with Al causes phosphate deficiency in plants (Horst et al. 1982). Addition of low concentrations of Al (10-50 µM) to plant roots cultivated without Al caused a fast hyperpolarization of PDv, and of PDc, the membrane potential across the plasma membrane. A depolarization of PDc was obtained at pH 6.5 (Lindberg et al. 1991). At the latter pH, the dominant species Al(OH),⁰ can easily penetrate membranes. Using artificial liposome vesicles it was shown that Al uptake was facilitated at neutral pH compared with pH 4 and 5 (Shi and Haug 1988). Therefore, toxic effects of Al on plants can also occur at a neutral pH. Most of the Al binds to the cell walls of root epidermal and cortical cells (Delhaize et al. 1993) and to the plasma membranes. It can also penetrate the plasma membrane (Lazof et al. 1994). When Al enters into a cytosol, it may inhibit cell division in the meristem and cell elongation in the elongation zone (Baluska et al. 1993) probably by binding to nucleic acids (Matsumoto et al. 1976).

8.2 Aluminium Interferes with Calcium Homeostasis

A disruption of cytosolic calcium homeostasis is a primary trigger of Al toxicity. Calcium plays an important role in cell division and cell expansion. For instance, transient changes in the cytosolic calcium concentration, [Ca2+cyt] have been observed to accompany the mitotic mechanism (Hepler 1994). Calcium promotes elongation in many plant cells (Takahashi et al. 1992; Levina et al. 1995) and calcium antagonists can block elongation (Cho and Hong 1995). Sustained gradients in Ca²⁺cyt are necessary for expansion of tip-growing plant cells, such as pollen tubes (Clarkson et al. 1988; Felle and Hepler 1997; Wymer et al. 1997). A maintained homeostatic control is, thus, necessary for cell viability (Bush 1995). Aluminium affects the calcium homeostasis in cell by inhibition of calcium uptake or efflux (Lindberg 1990). In 1-h experiments with intact sugar beet plants, the metabolic influx of ⁴⁵Ca²⁺ and K⁺(⁸⁶Rb⁺), and efflux of ⁴⁵Ca²⁺ were inhibited in the presence of Al (Lindberg 1990). Aluminium at a low concentration and low pH can elevate the [Ca2+cyt] (Lindberg and Strid 1997). Aluminium may also interact with the phosphoinositide signalling pathway. Both AlCl, and Al-citrate inhibited the phospholipase C (PLC) action in a dose-dependent manner (Jones and Kochian 1995).

8.3 Aluminium Induces Cytosolic Calcium Elevation

In protoplasts of wheat roots, AlCl₃ (80 μ M) induced immediate and transient oscillations in [Ca²⁺cyt], during 1–2 min in the presence of 0.1, 0.2 and 0.5 mM external calcium concentration (Lindberg and Strid 1997). No difference in the transients was obtained in the tolerant and sensitive wheat cultivar. Different cells may behave differently, since the reaction was found in only 60% of the protoplasts. Jones et al. (1998) reported an opposite effect of Al on [Ca²⁺cyt], using cultured cells of tobacco, BY-2. Upon addition of 200 μ M Al, the [Ca²⁺cyt] decreased, suggesting

a blockage of calcium channels. Addition of LaCl₃ or EGTA also decreased the resting calcium level, and thus inhibited the cell growth. The different results from wheat protoplasts and tobacco cells depend on the different concentrations of Al used, or that different plant material reacts differently. However, most investigations using other suspension cells, intact root tips, root hairs, excised whole roots and protoplasts showed an increase in [Ca²⁺cyt] (Rengel and Zhang 2003). The source of Ca²⁺ for increase in [Ca²⁺cyt] under Al exposure is partly extracellular, likely due to flux through DACC and through Ca2+-permeable non-selective cation channels in the plasma membrane (Rengel and Zhang 2003 and references therein). Lin et al. (2005) proposed that in Arabidopsis, the AtTPC1 is the only channel, sensitive to Al. This channel also is responsive to ROS and cryptogein, a fungal elicitor protein. It is believed that intracellular channels in the tonoplast and ER participate in the signalling. The latter channels could be activated by the increase in [Ca²⁺cyt]. An Al-related inhibition of the plasma membrane, endo-membrane Ca2+-ATPases and Ca2+ exchangers (CaX) are likely to exist. Kawano et al. (2003) showed that addition of AlCl₃ to tobacco cells triggers the generation of superoxide (O₂⁻) and that NADPH oxidase was involved in the production of O2-and was produced in a dose-dependent way concerning Al concentration. A fast spike in O2- generated a gradual increase in Ca2+cyt. A scavenger of O2 - and a calcium chelator significantly lowered the level of [Ca²⁺cyt] increase, indicating, that O₂⁻ stimulates the influx of Ca2+. Moreover, high concentrations of Al inhibited the [Ca²⁺cyt] increase. A change in [Ca²⁺cyt] is often connected with a change in cytosolic pH (pHcyt) (Kader and Lindberg 2010). In wheat-root protoplasts, a decrease in pHcyt, upon AlCl₂ additions was observed (Lindberg and Strid 1997). This pH decrease was most obvious in the Al-sensitive cultivar. Membrane potential measurements with intact roots of the same wheat cultivars showed similar results as with sugar beet roots (Lindberg et al. 1991). Additions of AlCl₃ both at pH 4 and 5 caused hyperpolarizations, and upon removal of Al depolarizations, the final depolarization was found larger in the

sensitive cultivar (Lindberg and Strid 1997). As concluded, the toxic effect of Al on the sensitive cultivar should be related to the larger change in pH, probably as a result of inhibition of the plasma membrane H⁺ ATPase.

8.4 Heavy Metal Toxicity and Uptake in Plants

Increased anthropogenic activities have caused rapid release of toxic compounds, such as trace elements (including heavy metals and metalloids) into the environment. Within the European community, the 11 trace elements of highest concern are arsenic, cadmium, cobalt, chromium, copper, mercury, manganese, nickel, lead, tin and thallium. Some of them, usually called "heavy metals", e.g. cadmium, mercury and lead, pose serious threat to microorganisms, plants, animals and humans, while other trace elements in low concentrations like zinc, copper and iron are essential for normal growth and functioning of organisms. Their toxicity depends on their available concentration, efficiency in uptake and translocation within the plant (Lindberg and Greger 2002). The uptake depends on pH, other ions and chelating agents present, as well as root microbes. Different trace elements are taken up and translocated in different ways depending on species and type of trace element. Cadmium is highly mobile in the soil and is easily taken up by the plant roots via active and passive pathways (Greger and Lindberg 1986; Hart et al. 1998; Lindberg et al. 2004). Most of the cadmium ions (Cd²⁺) are bound to the cell wall, but some Cd²⁺ enters into the cytosol by unspecific cation channels (Perfus-Barbeoch et al. 2002; Lindberg et al. 2004).

Cadmium, mercury and lead are considered as non-essential and are potentially toxic because of their reactivity with S and N atoms in amino acid chains of proteins (Chiang et al. 2006). Heavy metals also bind to oxygen atoms of membranes, or to histidine, tryptophan and tyrosine groups of polypeptides (Maksymiek 1997), interfere with several metabolic processes and alter the physiological functions of different enzymes. Cadmium and other trace metals strongly bind to thiol groups and thus to cystein-rich proteins (Vallee and Ulmer 1972; De Filippis 1979). Cadmium binding to plasma membrane sulfhydryl (SH) groups could be the reason for cadmium inhibition of plasma membrane ATPase activity and mineral uptake in sugar beet roots (Greger and Lindberg 1987; Lindberg and Wingstrand 1985; Greger et al. 1991). Cadmium and copper also inhibited the plasma membrane ATPase in root cells of cucumber and changed the lipid composition of the membrane (Kabata et al. 2008). Heavy metals inhibit photosynthesis and growth (Foy et al. 1978; Vangronsveld and Clijsters 1994; Maksymiec et al. 1994; Maksymiec and Baszynski 1999; Vinit-Dunand et al. 2002; Alaoui-Sossé et al. 2004; Rahman et al. 2005). Besides chlorophyll degradation, cadmium along with lead and mercury has been reported to interact with light-harvesting proteins of chloroplast's thylakoid membranes (Ahmed and Tajmir-Riahi 1993). Root growth inhibition in barley was suggested to depend on cadmium-induced water and oxidative stress (Tamas et al. 2008). Heavy metals can cause lipid peroxidation of photosynthetic membranes and affect the lipid composition of the plasma membranes (Chaoui et al. 1997; Quartacci et al. 2001; Nouairi et al. 2006).

8.5 Tolerance Mechanisms to Heavy Metals

Plant metal homeostasis must be tightly regulated in order to prevent toxic concentrations of nonessential metals and also to ensure sufficient micronutrients like Zn, Cu, Fe, etc. (Clemens 2001; Hall 2002). Plants have mechanisms that enable them to extrude cadmium and other heavy metals from the cytosol. Different transporters have been identified such as ATP-binding cassette (ABC) and Zinc–Iron-Like Proteins (ZIP) (Colangelo and Guerinot 2006; Clemens 2006). The P-ATPase HMA (heavy metal associated) also plays an important role in allocation and detoxification of heavy metals (Williams and Mills 2005). In *Arabidopsis*, eight genes belong to HMAs (Baxter et al. 2003). Among them the HMA1-4 specifically may transport Zn²⁺/Cd²⁺/ Pb²⁺/Co²⁺, while HMA5-8 transports Cu²⁺, and probably Ag⁺. In planta, AtHMA1 is expressed at the inner envelope membrane of the chloroplast (Seigneurin-Berny et al. 2006), while AtHMA4 is expressed in the plasma membrane (Hussein et al. 2004) involved in Zn and Cd xylem loading. The AtHMA3 was found to be a vacuolar transporter whose overexpression increased the Cd, Pb and Zn tolerance (Morel et al. 2009). Besides HMAs, CAX and metal chelators like phytochelatins, metallothioneins and organic acids could be important in plant tolerance to heavy metals, transporting metals into the vacuole (Clemens 2006; Morel et al. 2009). It was found that wheat protoplasts took up more cadmium when added to control than protoplasts from wheat seedlings pretreated 1 week with 1 µM CaCl₃ (Lindberg et al. 2007). In the control protoplasts, a H^+/Cd^{2+} antiport was suggested at the plasma membrane, but in the Cd-pretreated protoplasts, a phytochelatins-Cd transport was more likely. An ABC transporter AtPDR8 is also supposed to be involved in Cd²⁺ efflux at the plasma membrane (Kim et al. 2007).

8.6 Heavy Metal Signalling

Toxic effects of heavy metals on plants depend on direct effects on membranes, including photosynthetic membranes, as well as indirect effects caused by signalling pathways (Maksymiek 2007). Jasmonate, ethylene and H_2O_2 were reported to be involved in the signalling pathways (Maksymiek 2007, and references therein). Investigations showed that trace elements are able to induce oxidative stress in plants (Clijsters et al. 1999; Unyayar et al. 2006; Sharma and Dietz 2009). During oxidative stress, increased concentrations of superoxide anion (O₂⁻), hydrogen peroxide (H_2O_2) and its radical (OH) are produced and damage the cellular components. After the first hours of stress, H_2O_2 can be synthesized as a result of activation of the NADPH oxidase (Foreman et al. 2003) or indirectly, by increased lipid peroxidation and increased level of jasmonate (Maksymiec and Krupa 2002, 2006, Metwally et al. 2003). At prolonged exposure of cadmium to barley seedlings, lipid peroxides were formed (Metwally et al. 2003). It was suggested that Cd toxicity might be alleviated by salicylic acid (SA), a known factor that can block the jasmonate pathways (Gupta et al. 2000). Although many investigations deal with induction of different genes and proteins by heavy metals (see references in Maksymiek 2007) little information is available about calcium signalling in heavy metal transduction pathways. An increased level of H2O2 can alert plant cells against environmental stresses (Lamb and Dixon 1997; Foyer and Nocter 2003) and may increase the antioxidant response by expression of glutathione transferase gene through calcium signalling (Rentel and Knight 2004). Calcium and calcium channels may be involved in Cu²⁺ toxicity in bean plants (Maksymiec and Baszynski 1999). Recent results with protoplasts from Elodea canadensis showed that addition of low concentration of CdCl, induced an increase in both cytosolic and vacuolar pH, followed by a cytosolic calcium increase, and was found to be dose dependent (T. Javed, S. Lindberg, M. Greger 2011, unpublished results). Rodriguez-Serrano et al. (2009) reported that cadmium reduced the superoxide dismutase (SOD) iso-enzyme activity in pea plants. They suggested that the down-regulation of SOD could be due to cadmium-induced calcium deficiency. Production of superoxide radicals was also prevented by external calcium. On the other hand, nitric oxide (NO) production was strongly reduced by cadmium and treatment with calcium prevented this effect. Therefore, they proposed a complicated cross talk between ROS, NO and calcium.

9 Calcium Interactions with Hormones in Plants Under Stress

9.1 Abscisic Acid

ABA is a prime mediator of plant responses to cold, drought and salt stresses (Zhu 2002). It has many different functions, including regulation of

stomatal closure and activation of gene expression. Many second messengers, such as Ca²⁺, InsP3 and others are involved in ABA-mediated signalling. Both ABA-dependent and ABA-independent pathways are interacting in a complex way. Under drought stress, drought-responsive factors, such as DREBA2A and ERD1 were identified (Shinozaki and Yamaguchi-Shinozaki 2000, 2007; Yamaguchi-Shinozaki and Shinozaki 2006). InsP3 may act as a negative regulator in the DREB2A drought-signalling pathway (Perera et al. 2008). ABA-responsive elements, ARBREs, are present in the promoter of the transcription factors C-repeat/DRE binding factors (CBF or DREB1), that function as main regulators of abiotic stress responses (Finkler et al. 2007). These findings suggest a direct connection between Ca2+ regulation of transcription and abiotic stress responses. CAMTAs are Ca2+-dependent CaMbinding transcription factors, which may mediate the binding of CAMTAs to DNA. CAMTA gene expression in Arabidopsis responds rapidly to cold, salinity and hormones, such as ABA and jasmonic acid (Yang and Poovaiah 2002). ABA effects on plant-guard cells have been intensively studied. When maize seedlings were subjected to drought stress, ABA induced calcium increase in the mesophyll and guard cells of the leaves (Ma et al. 2009). The increase in calcium concentration occurred in cytosol, chloroplasts and nucleus under PEG treatment, while it decreased in vacuoles and intercellular spaces. The calcium elevation in the cytosol was ABA dependent and originated from extracellular Ca2+. CaM also participated in ABA signal transduction, since ABA induced changes in Ca²⁺/CaM concentration. In Arabidopsis guard cells, ABA induced an increase in ROS level (Pei et al. 2000). This increase in ROS would stimulate ABA-regulated HACC both in Arabidopsis and Vicia faba guard cells and regulate stomatal closure. It is suggested that ROS also activate other Ca²⁺ permeable channels (Mori and Schroeder 2004). Staxen et al. (1999) have shown that phosphoinositide-specific phospholipase C can be involved in ABA-induced calcium oscillations in guard cells. ABA can affect the transport of sucrose from mesophyll cells during apoplastic phloem loading. In protoplasts

of *Pisum sativum*, sucrose efflux was maximally increased at 10⁻⁷ M ABA. ABA also increased the pHcyt, independent of sucrose concentration (Opaskornkul et al. 1999). Not only calcium changes, but also pH changes are also induced in the cytosol of plant cells subjected to different stresses (Kader and Lindberg 2010).

9.2 Jasmonic Acid

Jasmonic acid (JA) plays an important role in protecting plants from pathogens or insect attacks, as well as from abiotic stress (Hu et al. 2009). Calcium influx, ROS and NO are often involved in JA signalling (Besson-Bard et al. 2008; Beligni and Lamattina 2008; Hu et al. 2009). The influx of calcium seemed to be necessary for elicitorinduced JA synthesis and secondary metabolite accumulation in cultured ginseng cells (Hu et al. 2009). Addition of JA to Arabidopsis leaves could induce an increase in [Ca²⁺cyt] (Sun et al. 2006). Moreover, JA induced gene expression could be mimicked by high concentration of extracellular Ca²⁺. It was proposed that the influx of Ca²⁺ through a Nif-sensitive channel in the plasmalemma may be responsible for the elevation of Ca2+ and the expression of genes encoding molecules downstream to JA. Increased concentrations of magnesium (Mg) and Ca in the apoplast of V. faba leaves significantly enhanced the biological activity of systemin (Liu et al. 2005). Liu et al. (2005) also showed that addition of Ca ionophores, which cause influx of calcium, also induced accumulation of proteinase inhibitors.

9.3 Ethylene

Ethylene concentration increases in plants under stress. Heavy metals, such as Cd, Cu, Fe and Zn can induce ethylene production (Gora and Clijsters 1989; Maksymiek 2007). Ethylene induces stomatal closure by a signalling system involving Ca²⁺ and NO. Treatment of *V. faba* with ethephon, an ethylene-released compound, caused a time- and dose-dependent stomatal closure. Ethylene is also involved in epinastic movements.

tobacco.

The ethylene-induced epinasty of tomato petioles was further stimulated by Ca²⁺ addition (Lee et al. 2008). Pre-treatment with NPA, an auxin-transport inhibitor prevented the effect by calcium ions on the petiolar epinasty. Ethylene caused accumulation of Ca2+ towards the lower (abaxial) side of the petioles, opposite to the ^{14C}IAA redistribution. These results suggest that the gravityinsensitive ethylene-induced Ca²⁺ redistribution and accumulation towards the abaxial side are closely coupled to the upper (adaxial) auxin redistribution and accumulation and, hereby, to the epinastic curvature. Addition of ethephon and ACC (precursor of ethylene), to tobacco protoplasts, activated the plasma membrane channels permeable to Ca²⁺, Mg²⁺ and Ba²⁺ (Zhao et al. 2007). Ethephon also modified the calcium signalling in the cytoplasm induced by herbivores (Arimura et al. 2008). This was necessary for both JA and terpene biosynthesis. Thus, elevation of [Ca²⁺cyt] is supposed to be an important component in ethylene signalling in plants. Ludwig et al. (2005) found an ethylene-mediated cross-

talk between CDPKs and mitogen-activated protein kinases (MAPKs) in response to stress in

9.4 **Auxin and Calcium Signalling**

Auxin, an important plant hormone, regulates the growth of plant and development. Early investigations showed that auxin induces elongation growth and causes oscillating kinetics of the transmembrane potential, starting with an immediate depolarization, and then changing to a hyperpolarization (Cleland et al. 1977; Poovaiah and Reddy 1987; Cho and Hong 1996; Polevoi et al. 1996 and others). Auxin addition to maizecoleoptile cells showed a rapid elevation in $[Ca^{2+}cyt]$ within the first 3–5 min (Felle 1988). Later on, similar reactions were found in maize roots, parsley hypocotyls, wheat and maize protoplasts (Gehring et al. 1990; Shishova and Lindberg 2004, 2010; Shishova et al. 2007). In wheat protoplasts, calcium elevation occurred within 5–10 s after addition of 1-naphthyl acetic acid (1-NAA), a synthetic and physiologically

active auxin (Shishova and Lindberg 2004). Inhibitor analyses showed that the shift in [Ca²⁺cyt] could depend on the activation of calcium channels, i.e. both in the plasma membrane and tonoplast. External calcium also increased the sensitivity to auxin. In several investigations, a simultaneous shift in pHcyt was obtained (Felle 1988; Irving et al. 1992; Shishova and Lindberg 1999). Shishova and Lindberg (1999) showed that verapamil, an inhibitor of calcium channels, blocked the cytosolic acidification, indicating the cell response to auxin depends on Ca²⁺-channel activation. Receptors for auxin are supposed to be located both at the plasma membrane and intracellular. A soluble receptor, the F-box transport inhibitor resistant 1 (TIR1) protein in the nucleus was discovered (Dharmasiri et al. 2005; Kepinski and Leyser 2005). For the plasma membrane receptor, two models have been proposed: either the receptor includes a protein kinase or it is connected with a G-protein. Based on the found auxin-induced reactions, such as changes in membrane potential, shifts in cytosolic concentrations of Ca2+ and pH as well as increase in cell sensitivity to auxin by external calcium, a third model was also proposed (Shishova and Lindberg 2010). This model suggests that an associative domain of the ABP1 protein is closely connected with the calcium-permeable ion channel. Little is known about the mechanisms of integrating auxin effects on growth and stress responses. However, it was recently shown that the calmodulin-binding transcription activator 1 (CAMTA 1) is involved in auxin signalling and responds to stress (Galon et al. 2010a). The expression pattern of CAMTA1 changed significantly and differentially on exposure of Arabidopsis to high salt concentration and heat.

9.5 Gibberellins

Calcium is also involved in gibberellin signalling. The homeostasis of gibberellins (GAs) in plant cells is maintained by a negative-feedback regulation. A transcriptional activator (RSG, Repression of Shoot Growth) is suggested to take part in the feedback mechanism causing RSG and regulating genes coding GA biosynthetic enzymes (Nakata et al. 2009). A 14-3-3 signalling protein negatively regulates RSG by moving into the cytoplasm in response to GAs. A phosphorylation is necessary for binding of 14-3-3 to RSG. It was found that a CDPK promoted the binding. This CDPK decodes the Ca²⁺ signal produced by GAs.

10 Biotic Stress Signalling

Pathogens, as well as symbiotic microorganisms induce Ca^{2+} signals in plant cells; however, the defence responses obtained can be both activated and suppressed by the signals. This suggests that Ca^{2+} -responsive, but antagonistic, signalling mechanisms are present (Dodd et al. 2010).

Molecules with microbe-associated molecular patterns, called MAMPs, can cause both cytosolic and nuclear calcium increases, but the signals may be more prolonged in the nucleus. Lecourieux et al. (2005) found the Ca^{2+} elevations induced by harpin, in cell cultures from tobacco cells, continued for 5 min in the cytosol and for 150 min in the nucleus. Calcium was mobilized from both extracellular (apoplast) and intracellular (vacuole or ER) stores. It was reported that MAMPinduced Ca2+ signals caused activation of SA and mitogen- and wound-activated protein kinases (Lecourieux et al. 2005; Ma and Berkowitz 2007). Usually Ca²⁺ increase encodes stimulus-specific information, but MAMP-induced signals cause similar defence responses irrespective of the type of elicitor (Ma and Berkowitz 2007 and references within). Moreover, Du et al. (2009) found the Ca²⁺ signals may also suppress the SA-mediated gain of systemic acquired resistance. When the Ca²⁺/CaM complex binds to the transcription activator CAMTA, EDS1 (enhanced disease susceptibility 1) is repressed. As EDS1 functions as a positive regulator of resistance and SA levels, Ca²⁺ signals in this case repress the resistance mechanisms. Calcium also plays an important role in the symbiosis between a plant and a microorganism. Since atmospheric nitrogen, N_2 , cannot be used by the plant, it needs to be converted to ammonia by certain prokaryotic bacteria in a process called nitrogen fixation. The microsymbiont is hosted in special organs, called root nodules, and are developed on the plant roots upon infection (reviewed by Mylona et al. 1995).

Two types of root-nodule symbiosis are known: symbiosis between legumes and gramnegative, unicellular soil bacteria, called rhizobia, and between actinorhizal plants (of the order Fagales and Cucurbitales) and gram-positive actinomycetes of the genus Frankia (reviewed by Sprent 2006). Less information is available on the actinorhizal, than on legume-rhizobia symbiosis. In the legume-rhizobia interaction, the plant secretes flavonoid compounds from the roots and induces the expression of rhizobial nodulation (nod) genes. The Nod genes encode proteins involved in the rhizobial signal transduction, such as Nod factors, which induce deformation of plant roots, plant gene expression and nodule development. Most terrestrial plants (80%) can also establish a symbiosis with a fungus of the phylum Glomeromycota (Parniske 2008). Such symbiosis is an advantage to the plant, since the mychorrhiza increases the availability and uptake of soil nitrogen and phosphorus into the plant. For the establishment of mycorrhiza, diffusible fungal signal factors, socalled Myc factors must be recognized by the plant. Both Myc and Nod factors are representatives of lipochito-oligosaccharides (Denarie et al. 2010) and play a great role of the signalling system for development of nodules and mycorrhizal infection (Kosuta et al. 2008; Oldroyd and Downie 2008). Kosuta et al. (2008) suggested that in legume plants, at least seven common proteins mediate infection by both mycorrhizal fungi and rhizobial bacteria. For instance in *Medicago* truncatula, the proteins DMI1, DMI2 and DMI3 are necessary for both types of symbiosis signalling. In symbiotic signalling, cytosolic calcium changes are supposed to be linked to nuclear calcium changes by a complicated system (Oldroyd and Downie 2006). Two receptor-like kinases at the plasma membrane, NFR1 and NFR5, are required for Nod-factor perception (Madsen et al. 2003) and an equivalent receptor-like kinase is supposed to exist for Myc factors. A third kinase
could also function in a common symbiotic pathway (Endre et al. 2002; Limpens et al. 2003). The phosphorylations at the plasma membrane following Nod factor recognition must be linked to the induction of calcium changes associated with the nucleus (Oldroyd and Downie 2006). The secondary messenger involved so far is unknown, but could be a product of phospholipase C and D (PLC and PLD). The perception of Nod and Myc factors at receptors in legume roots induces calcium oscillations, but the frequencies of the spikes differ depending on the elicitor factor (Kosuta et al. 2008). Kosuta and co-workers showed that Nod factors induced calcium spiking during longer periods and with higher amplitude than the mycorrhizal spiking. It is interesting that in both cases, the oscillation was chaotic in nature, that is, deterministic and predictable (Casdagli 1992), as shown by the Lyapunov method and other mathematical analyses (Kantz and Schreiber 1997). Chaos is not common in biological systems, but may provide a mechanism for flexibility (Kosuta et al. 2008). Chaotic systems can produce variable responses with very little energetic input.

Arbuscular mycorrhizal fungi were shown to induce calcium signalling in the host plant (Navazio et al. 2007). A difference in amplitude and frequency of calcium oscillations has been linked to differences in gene expression and enzymatic activities in several systems (Kummer et al. 2005). For nodule development and mycorrhizal association, an activation of CCaMK (calcium– calmodulin activated kinase) is necessary and the different outcome could depend on the difference in calcium oscillations (Kosuta et al. 2008). A sustained high cytosolic calcium concentration leads to apoptosis during the normal development and in hypersensitive responses to pathogens (Levine et al. 1996; Tuteja and Mahajan 2007).

11 Conclusion and Future Perspective

Calcium plays an important role in stress signalling. A change in the cytosolic calcium concentration is induced by all types of abiotic and biotic stresses investigated. Calcium signalling is very complex and takes place not only in the cytosol, but also in the nucleus, mitochondrion and ER. The specific "signature" of calcium, such as duration, frequency and amplitude of calcium concentration changes is important for the downstream reactions. Plants have many different calcium sensor proteins, such as CaM, CaM-binding proteins, CDPKs and phosphatases, annexins, as well as CaM-like and calcineurin-like proteins. They exist in the cell as free proteins or bound to membranes. Some of these proteins are "buffer" protein, that is, they regulate the free calcium concentration, others are so-called trigger proteins, which are activated by calcium binding and in turn activate other proteins. Calcium and CAM can also activate transcription factors and are supposed to be involved in stress-related gene expression.

In most stress reactions investigated so far, the change in calcium concentration is connected with a change in pHcyt. The pH change differs in salt-tolerant and -sensitive cultivars or species, but the reason for this is unknown. More information is needed about pH changes in plant cells under various stresses since the proteins and enzymes in the downstream reactions are pH sensitive.

Even if there is considerable knowledge about the tolerance mechanisms and gene expression under different types of stress, the nature of the sensors for different stresses is still enigmatic and needs clarification. Further research should also focus on the complex signalling system in different compartments of a cell and how they interact with each other.

References

- Ahmed A, Tajmir-Riahi HA (1993) Interaction of toxic metal ions Cd²⁺, Hg²⁺ and Pb²⁺ with light-harvesting proteins of chloroplast thylakoid membranes. An FTR studies. J Inorg Chem 50:235–243
- Alaoui-Sossé B, Genet P, Vinit-Dunand F, Toussaint M-L, Epron D, Badot P-M (2004) Effect of copper on growth in cucumber plants (*Cucumus sativus*) and its relationships with carbohydrate accumulation and changes in ion contents. Plant Sci 166:1213–1218

- Alonso MT, Villalobos C, Chamero P, Alvarez J, Garcia-Sancho J (2006) Calcium microdomains in mitochondria and nucleus. Cell Calcium 40:513–525
- Arimura G-I, Garms S, Maffei M, Bossi S, Schultze B, Leitner M, Mithöfer A, Boland W (2008) Herbivoreinduced terpenoid emission in *Medicago truncatula*: concerted action of jasmonate, ethylene and calcium signalling. Planta 227:453–464
- Aurisano N, Bertani A, Reggiani R (1995) Involvement of calcium and calmodulin in protein and amino acid metabolism in rice roots under anoxia. Plant Cell Physiol 36:1525–1529
- Aurisano N, Bertani A, Reggiani R (1996) Evidence for the involvement of GTP-binding proteins in the process of anaerobic gamma-aminobuirate accumulation in rice roots. J Plant Physiol 149:517–519
- Axelsen KB, Palmgren MG (2001) Inventory of the superfamily of P-type ion pumps in *Arabidopsis*. Plant Physiol 126:696–706
- Bach O, Agell N, Carafoli E (1992) Calcium and calmodulin function in the cell nucleus. Biochem Biophys Acta 1113:259–270
- Bailey-Serres J, Chang R (2005) Sensing and signalling in response to oxygen deprivation in plants and other organisms. Ann Bot 96:507–518
- Bailey-Serres J, Voesenek L (2008) Flooding stress: acclimations and genetic diversity. Annu Rev Plant Biol 59:313–339
- Baluska F, Parker JS, Barlow PW (1993) Specific patterns of cortical and endoplasmic microtubules associated with cell growth and tissue differentiation in roots of maize (Zea mays L). J Cell Sci 103:191–200
- Barcelo J, Poschenrieder C (2002) Fast root growth responses, root exudates and internal detoxification as clues to the mechanisms of aluminum toxicity and resistance. Environ Exp Bot 48:75–92
- Baxter I, Tchieu J, Sussman MR, Boutry M, Palmgren MG, Gribskov M, Harper JE, Axelsen KB (2003) Genomic comparison of P-type ATPase ion pumps in *Arabidopsis* and rice. Plant Physiol 132:618–628
- Baxter-Burrell A, Yang Z, Springer PS, Bailey-Serres J (2002) RopGAP4-dependent Rop GTPase rheostat control of *Arabidopsis* oxygen deprivation tolerance. Science 296:2026–2028
- Beligni MV, Lamattina L (2008) Nitric oxide counteracts cytotoxic processes mediated by reactive oxygen species in plant tissue. Planta 33:337–344
- Besson-Bard A, Courtois C, Gauthier A, Dahan J, Dobrowolska G, Jeandroz S, Pugin A, Wendehenne D (2008) Nitric oxide in plants: production and crosstalk with Ca²⁺ signalling. Mol Plant 1:218–228
- Bittisnich D, Robinson D, Whitecross M (1989) Membraneassociated and intracellular free calcium levels in root cells under NaCl stress. In: Dainty J, de Michelis MI, Marre E, Rasi-Caldogno F (eds) Plant membrane transport: the current position. Proceedings of eighth international workshop on plant membrane transport, Venice, Italy, 25–30 June 1989. Elsevier, New York, pp 681–682
- Bkaily G (2006) The nucleus: a cell within the cell. Can J Physiol Pharmacol 84:279–507

- Blokhina OB, Chirkova TV, Fagerstedt KV (2001) Anoxic stress leads to hydrogen peroxide formation in plant cells. J Exp Bot 52:1179–1190
- Blumwald E (2000) Sodium transport and salt tolerance in plants. Curr Opin Cell Biol 12:431–434
- Bohnert HJ, Jensen RG (1996) Strategies for engineering water-stress tolerance in plants. Trends Biotechnol 14:89–97
- Boudsocq M, Lauriere C (2005) Osmotic signalling in plants. Multiple pathways mediated by emerging kinase families. Plant Physiol 138:1185–1194
- Boursiac Y, Harper JF (2007) The origin and function of calmodulin regulated Ca²⁺ pumps in plants. J Bioenerg Biomembr 39:409–414
- Brandizzi F, Irons SL, Evans DE (2004) The plant nuclear envelope: new prospects for poorly understood structure. New Phytol 163:227–246
- Briere C, Xiong TC, Mazars C, Ranjeva R (2006) Autonomous regulation of free Ca²⁺ concentrations in isolated plant cell nuclei: a mathematical analysis. Cell Calcium 39:293–303
- Buringh P (1979) Introduction to the soils in tropical and subtropical regions, 3rd edn. PUDOC, Wageningen, 124p
- Bush DS (1995) Calcium regulation in plant cells and its role in signalling. Annu Rev Plant Physiol Plant Mol Biol 46:95–122
- Carafoli E (2002) Calcium signalling: a tale for all seasons. Proc Natl Acad Sci USA 99:1115–1122
- Carpaneto A, Ivashikina N, Levchenko V, Krol E, Jeworutzki E, Zhu J-K, Hedrich R (2007) Cold transiently activates calcium-permeable channels in *Arabidopsis* mesophyll cells. Plant Physiol 143:487–494
- Casdagli M (1992) Chaos and deterministic versus stochastic non-linear modelling. J R Statist Soc B 54:303–328
- Catala R, Santos E, Alonso JM, Ecker JR, Martinez-Zapater JM, Salinas J (2003) Mutations in the Ca²⁺/H⁺ transporter CAX1 increase CBF/DREB1 expression and the cold-accumulation response in *Arabidopsis*. Plant Cell 15:2940–2951
- Chaoui A, Mazhoudi S, Ghorbal MN, Ferjani EE (1997) Cadmium and zinc induction of lipid peroxidation and effects on antioxidant enzymes activity in bean (*Phaseolus vulgaris* L.). Plant Sci 127:139–147
- Chen THH, Murata N (2002) Enhancement of tolerance to abiotic stress by metabolic engineering of betaines and other compatible solutes. Curr Opin Plant Biol 5:250–257
- Cheng NH, Pittman JK, Barkla BJ, Shigaki T, Hirschi KD (2003) The Arabidopsis cax1 mutant exhibits impaired ion homeostasis, development, and hormonal responses and reveals interplay among vacuolar transporters. Plant Cell 15:347–364
- Chiang HEC, Lo JC, Yeh KC (2006) Genes associated with heavy metal tolerance and accumulation in Zn/Cd hyperaccumulator *Arabidopsis halleri*: a genomic survey with cDNA microarray. Environ Sci Technol 40:6792–6798
- Chinnusamy V, Jagendorf A, Zhu JK (2005) Understanding and improving salt tolerance in plants. Crop Sci 45:437–448

- Cho HT, Hong YN (1995) Effect of IAA on synthesis and activity of plasma membrane H⁺-ATPase of sunflower hypocotyls, in relation to IAA-induced cell elongation and H⁺ excretion. J Plant Physiol 145:717–725
- Cho HT, Hong YN (1996) Effect of calcium channel blockers on the IAA-induced cell elongation of sunflower hypocotyls segments. J Plant Physiol 149:377–383
- Clarkson DT (1965) The effect of aluminium and some trivalent metal cations on cell division in the root apices of *Allium cepa*. Ann Bot 29:309–315
- Clarkson DT, Brownlee C, Ayling SM (1988) Cytoplasmic calcium measurements in intact higher plant cells: results from fluorescence ratio imaging of fura-2. J Cell Sci 91:71–80
- Cleland RE, Prins HBA, Yarper JR, Higinbotham N (1977) Rapid hormone-induced hyperpolarisation of the oat coleoptile transmembrane potential. Plant Physiol 59:395–397
- Clemens S (2001) Molecular mechanisms of plant metal tolerance and homeostasis. Planta 212:475–486
- Clemens S (2006) Toxic metal accumulation, responses to exposure and mechanisms of tolerance in plants. Biochimie 88:1707–1719
- Clijsters H, Cuypers A, Vangronsveld J (1999) Physiological responses to heavy metals in higher plants: defense against oxidative stress. Z Naturforsch 54:730–734
- Colangelo EP, Guerinot ML (2006) Put the metal to the petal: metal uptake and transport throughout plants. Curr Opin Plant Biol 9:322–330
- Covington ME, Harmer SL (2007) The circadian clock regulates auxin signalling and responses in *Arabidopsis*. PLoS Biol 5:1773–1784
- Cramer GR, Jones RL (1996) Osmotic stress and abscisic acid reduce cytosolic calcium activities in roots of *Arabidopsis thaliana*. Plant Cell Environ 19:1291–1298
- Crawford RMM, Braendle R (1996) Oxygen deprivation stress in a changing environment. J Exp Bot 47:145–159
- D'Angeli S, Malho R, Altamura MM (2003) Lowtemperature sensing in olive tree: calcium signalling and cold acclimation. Plant Sci 165:1303–1313
- D'Onofrio C, Lindberg S (2009) Sodium induces simultaneous changes in cytosolic calcium and pH in salt-tolerant quince protoplasts. J Plant Physiol 166:1755–1763
- Dammann C, Ichida A, Hong B, Romanowsky SM, Hrabak E, Harmon AC, Pickard BG, Harper JF (2003) Subcellular targeting of nine calcium-dependent protein kinase isoforms from *Arabidopsis*. Plant Physiol 132:1840–1848
- Davenport R (2002) Glutamate receptors in plants. Ann Bot 90:549–557
- De Filippis LF (1979) The effect of heavy metal compounds on the permeability of *Chlorella* cells. Z Pflanzen Physiol 92:39–49
- DeFalco TA, Bender KW, Snedden WA (2010) Breaking the code: Ca²⁺ sensors in plant signalling. Biochem J 425:27–40
- Delhaize E, Ryan PR, Randall P (1993) Aluminium tolerance in wheat (*Triticum aestivum* L.). II. Aluminiumstimulated excretion of malic acid from root apices. Plant Physiol 103:695–702

- DeMarty M, Morvan C, Thellier M (1984) Calcium and the cell wall. Plant Cell Environ 7:441–448
- Demidchik V, Maathuis FJM (2007) Physiological roles of nonselective cation channels in plants: from salt stress to signalling and development. New Phytol 175:387–404
- Denarie J, Maillet F, Poinsot V, Andre O, Becard G, Gueunier M, Cromer L et al (2010) International application published under patent cooperation treaty (PCT) WO 2010/049817A2
- Dennis ES, Dolferus R, Ellis M, Rahman M, Wu Y, Hoeren FU, Grover A, Ismond KP, Good AG, Peacock WJ (2000) Molecular strategies for improving waterlogging tolerance in plants. J Exp Bot 51:89–97
- Desikan R, Hancock JT, Bright J et al (2005) A role for ETR1 in hydrogen peroxide signalling in stomatal guard cells. Plant Physiol 137:831–834
- Dharmasiri N, Dharmasiri S, Estelle M (2005) The F-box protein TIR1 is an auxin receptor. Nature 435: 441–445
- Dodd AN, Kudla J, Sanders D (2010) The language of calcium signalling. Annu Rev Plant Biol 61: 593–620
- Dolferus R, Ellis M, De Bruxelles G, Trevaskis B, Hoeren F, Dennis ES, Peacock WJ (1997) Strategies of gene action in *Arabidopsis* during hypoxia. Ann Bot 79(Suppl A):21–31
- Drew MC (1997) Oxygen deficiency and root metabolism: injury and acclimation under hypoxia and anoxia. Annu Rev Plant Physiol Plant Mol Biol 48:223–250
- Du L, Ali GS, Simons KA, Hou J, Yang T et al (2009) Ca^{2+/}calmodulin regulates salicylic-acid-mediated plant immunity. Nature 457:1152–1158
- Endre G, Kereszt A, Kevei Z, Mihacea S, Kalo P, Kiss GB (2002) A receptor kinase gene regulating symbiotic nodule development. Nature 417:962–966
- Essah PA (2000) Sodium transport in *Arabidopsis thaliana*. PhD thesis. University of Cambridge, UK
- Fähling M (2008) Cellular oxygen sensing, signalling and how to survive translational arrest in hypoxia. Acta Physiol 195:205–230
- Felle H (1988) Auxin causes oscillations of cytosolic free calcium and pH in Zea mays coleoptiles. Planta 174:495–499
- Felle H, Hepler PK (1997) The cytosolic Ca²⁺ concentration gradient of *Sinapis alba* root hairs revealed by Ca²⁺-selective microelectrode tests and Fura-dextran ratio imaging. Plant Physiol 114:39–45
- Felle HH, Herrmann A, Hückelhoven R, Kogel K-H (2005) Root-to-shoot signalling: apoplastic alkalinization, a general stress response and defense factor in barley (*Hordeum vulgare*). Protoplasma 227:17–24
- Finkler A, Kaplan B, Fromm H (2007) Ca²⁺-responsive cis elements in plants. Plant Signal Behav 2:17–19
- Flowers TJ, Colmer TD (2008) Salinity tolerance in halophytes. New Phytol 179:945–963
- Flowers TJ, Läuchli A (1983) Sodium versus potassium: substitution and compartmentalization. In: Läuchli A, Bieleski RL (eds) Inorganic plant nutrition, vol 15B. Springer, Berlin, pp 651–681

- Flowers TJ, Yeo AR (1995) Breeding for salinity resistance in crop plants: where next? Aust J Plant Physiol 22:875–884
- Flowers TJ, Galal HK, Bromham L (2010a) Evolution of halophytes: multiple origins of salt tolerance in land plants. Funct Plant Biol 37:604–612
- Flowers TJ, Gaur PM, Laxmipathi Gowda CL, Krishnamurthy L, Samineni S, Siddique KHM, Turner NC, Vadez V, Varshney RK, Colmer TD (2010b) Salt sensitivity in chickpea. Plant Cell Environ 33:490–509
- Foreman D, Demidchik V, Bothwell JHF, Myona P, Miedema H, Torres MA, Linsted P, Costa SE et al (2003) Reactive oxygen species produced by NADPH oxidase regulate plant cell growth. Nature 422:442–446
- Foy CD (1984) Physiological effects of hydrogen aluminium and manganese toxicities in acid soils. In: Adams FJ (ed) Soil acidity and liming, 2nd edn. American Society of Agronomy, Madison, WI, pp 57–97. ISBN 0-89118-080-X
- Foy CD, Chaney RL, White MC (1978) The physiology of metal toxicity in plants. Annu Rev Plant Physiol Plant Mol Biol 29:511–566
- Foyer CH, Nocter G (2003) Redox sensing and signalling associated with reactive oxygen in chloroplasts, peroxisomes and mitochondria. Physiol Plant 119:355–364
- Fu X, Chang J, An L, Zhang M, Xu S, Chen T, Liu Y, Xin H, Wang J (2006) Association of the cold-hardiness of *Chorispora bungeana* with the distribution and accumulation of calcium in the cells and tissues. Environ Exp Bot 55:282–293
- Galon Y, Aloni R, Nachmias D, Snir O, Feldmesser E, Scase-Field S, Boyse JM, Bouché N, Knight MR, Fromm H (2010a) Calmodulin-binding transcription activator 1 mediates auxin signalling and responds to stress in *Arabidopsis*. Planta 232:165–178
- Galon Y, Finkler A, Fromm H (2010b) Calcium-regulated transcription in plants. Mol Plant 3:653–669
- Gao D, Knight MR, Trewavas AJ, Sattelmacher B, Plieth C (2004) Self-reporting *Arabidopsis* expressing pH and [Ca²⁺] indicators unveil ion dynamics in the cytoplasm and in the apoplast under abiotic stress. Plant Physiol 134:898–908
- Garciadeblas B, Benito B, Rodriguez-Navarro A (2001) Plant cell express stress calcium ATPases but apparently no sodium ATPase. Plant Soil 235:189–192
- Gehring CA, Irving H, Parish RW (1990) Effects of auxin and abscisic acid on cytosolic calcium and pH in plant cells. Proc Natl Acad Sci USA 87:9645–9649
- Gibbs J, Greenway H (2003) Mechanisms of anoxia tolerance in plants. I. Growth, survival and anaerobic catabolism. Funct Plant Biol 30:1–47
- Glenn EP, Brown JJ, Khan MJ (1997) Mechanisms of salt tolerance in higher plants. In: Basra AS, Basra RK (eds) Mechanisms of environmental stress resistance in plants. Harwood Academic Publishers, Amsterdam, The Netherlands, pp 83–110
- Gomes DA, Leite MF, Bennett AM, Nathanson MH (2006) Calcium signalling in the nucleus. Can J Physiol Pharmacol 84:325–332

- Gong M, van der Luit AH, Knight MR, Trewavas AJ (1998) Heat-shock-induced changes in intracellular Ca²⁺ level in Tobacco seedling in relation to thermotolerance. Plant Physiol 116:429–437
- Gora L, Clijsters H (1989) Effect of copper and zinc on the ethylene metabolism in *Phaseolus vulgaris*L. In: Clijsters H (ed) Biochemical and physiological aspects of ethylene production in lower and higher plants. Kluwer, Dordrecht, The Netherlands, pp 219–228
- Gorecka KM, Thouverey C, Buchet R, Pikula S (2007) Potential role of annexin AnnAt1 from Arabidopsis thaliana in pH-mediated cellular response to environmental stimuli. Plant Cell Physiol 48:792–803
- Green J, Crack JC, Thomson AJ, LeBrun NE (2009) Bacterial sensors of oxygen. Curr Opin Microbiol 12:145–151
- Greenway H, Gibbs J (2003) Mechanisms of anoxia tolerance in plants. II. Energy requirements for maintenance and energy distribution to essential processes. Funct Plant Biol 30:999–1036
- Greenway H, Munns R (1980) Mechanisms of salt tolerance in non halophytes. Annu Rev Plant Physiol Plant Mol Biol 31:149–190
- Greger M, Lindberg S (1986) Effects of Cd²⁺ and EDTA on young sugar beets (*Beta vulgaris*). I. Cd²⁺ uptake and sugar accumulation. Physiol Plant 66:69–74
- Greger M, Lindberg S (1987) Effects of Cd²⁺ and EDTA on young sugar beets (*Beta vulgaris*). II. Net uptake and distribution of Mg²⁺, Ca²⁺ and Fe²⁺/Fe³⁺. Physiol Plant 69:81–86
- Greger M, Brammer E, Lindberg S, Larsson G, Idestam-Almquist J (1991) Uptake and physiological effects of cadmium in sugar beet (*Beta vulgaris*) related to mineral provision. J Exp Bot 239:729–737
- Gupta V, Willits MG, Galzebrook J (2000) Arabidopsis thaliana EDS4 contributes to salicylic acid (SA)dependent expression of defense responses. Evidence for inhibition of jasmonic acid signalling by SA. Mol Plant Microbe Interact 13:503–511
- Guy CL (1990) Cold acclimation and freezing stress tolerance: Role of protein metabolism. Annu Rev Plant Physiol Plant Mol Biol 41:187–223
- Halfter U, Ishitani M, Zhu JK (2000) The Arabidopsis SOS2 protein kinase physically interacts with and is activated by the calcium-binding protein SOS3. Proc Natl Acad Sci USA 97:3730–3734
- Hall JL (2002) Cellular mechanisms for heavy metal detoxification and tolerance. J Exp Bot 366:1–11
- Halperin SJ, Gilroy S, Lynch JP (2003) Sodium chloride reduces growth and cytosolic calcium, but does not affect cytosolic pH, in root hairs of *Arabidopsis thaliana* L. J Exp Bot 54:1269–1280
- Harper JF, Schaller GE, Sussman MR, Putnan-Evans C, Charbonneau H, Harmon AC (1991) A novel calciumdependent protein kinase with a regulatory domain similar to calmodulin. Science 252:951–954
- Harper JF, Breton G, Harmon A (2004) Decoding Ca²⁺ signals through plant protein kinases. Annu Rev Plant Biol 55:263–288

- Hart JJ, Welch RM, Norvell WA, Sullivan LA, Kochian LV (1998) Characterization of cadmium binding, uptake, and translocation in intact seedlings of bread and durum wheat cultivars. Plant Physiol 116: 1413–1420
- Hasegawa PM, Bressan RA, Zhu JK, Bohnert HJ (2000) Plant cellular and molecular responses to high salinity. Annu Rev Plant Physiol Plant Mol Biol 51:463–499
- Hashimoto Y, Perrino BA, Soderling TR (1990) Identification of an autoinhibitory domain in calcineurin. J Biol Chem 265:1924–1927
- Haug AR (1984) Molecular aspects of aluminium toxicity. CRC Crit Rev Plant Sci 1:345–373
- Henriksson E, Henriksson KN (2005) Salt-stress signalling and the role of calcium in the regulation of the Arabidopsis ATHB7 gene. Plant Cell Environ 28:202–210
- Hepler PK (1994) The role of calcium in cell division. Cell Calcium 16:322–330
- Hepler PK (2005) Calcium: a central regulator of plant growth and development. Plant Cell 17:2142–2155
- Hetherington AM, Brownlee C (2004) The generation of Ca²⁺ signals in plants. Annu Rev Plant Biol 55:401–427
- Hirschi K (2001) Vacuolar H⁺/Ca²⁺ transport: who's directing the traffic? Trends Plant Sci 6:100–104
- Horst WJ, Wager A, Marschner H (1982) Mucilage protects root meristems from aluminium injury. Z Pflanzen Physiol 105:435–444
- Hrabak EM, Chan CWM, Gribskov M, Harper JF, Choi JH, Halford N et al (2003) The *Arabidopsis* CDPKsnRK superfamily of protein kinases. Plant Physiol 132:666–680
- Hu X, Li W, Chen Q, Yang Y (2009) Early signal transduction linking the synthesis of jasmonic acid in plant. Plant Signal Behav 4:696–697
- Hussein D, Haydon MJ, Wang Y, Wong E, Sherson SM, Young J, Camakaris J, Harper JF, Cobett CS (2004) P-type ATPase heavy metal transporters with roles in essential zinc homeostasis in *Arabidopsis*. Plant Cell 16:1327–1339
- Hwang I, Sze H, Harper JF (2000) A calcium-dependent protein kinase can inhibit a calmodulin-stimulated Ca²⁺ pump (ACA2) located in the endoplasmatic reticulum of *Arabidopsis*. Proc Natl Acad Sci USA 97:6224–6229
- Irving H, Gehring CA, Parish RW (1992) Changes in cytosolic pH and calcium of guard cells proceed stomatal movements. Proc Natl Acad Sci USA 89:1790–1794
- Jaiswal JK (2001) Calcium-how and why? J Biosci 26:357–363
- Javed T, Lindberg S, Greger M; unpublished
- Johnson C, Knight M, Kondo T, Masson P, Sedbrook J, Haley A, Trewavas A (1995) Circadian oscillations of cytosolic and chloroplastic free calcium in plants. Science 269:1863–1865
- Jones DL, Kochian LV (1995) Aluminum inhibition of the inositol 1,4,5-triphoshate signal transduction pathway in wheat roots: a role in aluminum toxicity? Plant Cell 7:1913–1922
- Jones DL, Kochian LV, Gilroy S (1998) Aluminum induces a decrease in cytosolic calcium concentration in BY-2 tobacco cell cultures. Plant Physiol 116:81–89

- Jou Y, Chiang CP, Jauh GY, Yen HE (2006) Functional characterization of ice plant SKD1, an AAA-type ATPase associated with the endoplasmic reticulumgolgi network, and its role in adaptation to salt stress. Plant Physiol 141:135–146
- Kabata K, Janicka-Russak M, Burzynski M, Klobus G (2008) Comparison of heavy metal effect on the proton pumps of plasma membrane and tonoplast in cucumber root cells. J Plant Physiol 165:278–288
- Kader MA, Lindberg S (2005) Uptake of sodium in protoplasts of salt-sensitive and salt-tolerant cultivars of rice *Oryza sativa* L. determined by the fluorescent dye SBFI. J Exp Bot 56:3149–3158
- Kader A, Lindberg S (2008) Cellular traits for sodium tolerance in rice (*Oryza sativa* L). Plant Biotechnol 25:247–255
- Kader A, Lindberg S (2010) Cytosolic calcium and pH signalling in plants under salinity stress. Plant Signal Behav 5(3):1–7
- Kader A, Seidel T, Golldack D, Lindberg S (2006) Expressions of OsHKT1, OsHKT2 and OsVHA are differently regulated under NaCl stress in salt-sensitive and salt-tolerant rice (*Oryza sativa* L.) cultivars. Exp Bot 57:4257–4268
- Kader MA, Lindberg S, Seidel T, Golldack D, Yemelyanov V (2007) Sodium sensing induces different changes in free cytosolic calcium concentration and pH in salttolerant and salt-sensitive rice (*Oryza sativa* L.) cultivars. Physiol Plant 130:99–111
- Kantz H, Schreiber T (1997) Nonliniar time series analyses. Cambridge University Press, New York
- Kawano T, Kadono T, Furuichi T, Muto S, Lapeyrie F (2003) Aluminum-induced distortion in calcium signalling involving oxidative bursts and channel regulation in tobacco BY-2 cells. Biochem Biophys Res Comm 308:35–42
- Kende H, Van Der Knaap E, Cho HT (1998) Deepwater rice: a model plant to study stem elongation. Plant Physiol 118:1105–1110
- Kepinski S, Leyser O (2005) The Arabidopsis F-box protein TIR1 is an auxin receptor. Nature 435:446–451
- Khan MA, Ungar IA, Showalter AM (2005) Salt stimulation and tolerance in an intertidal stem-succulent halophyte. J Plant Nutr 28:1365–1374
- Kiegle E, Moore CA, Haseloff J, Tester MA, Knight MR (2000) Cell-type-specific calcium responses to drought, salt and cold in *Arabidopsis* root. Plant J 23:267–278
- Kim DY, Bovet L, Maeshima M, Martinoia E, Lee Y (2007) The ABC transporter AtPDR8 is a cadmium extrusion pump conferring heavy metal resistance. Plant J 50:207–218
- Knight H (2000) Calcium signalling during abiotic stress in plants. Int Rev Cytol 195:269–324
- Knight MR (2002) Signal transduction leading to lowtemperature tolerance in *Arabidopsis thaliana*. Philos Trans R Soc Lond B Biol Sci 357:871–875
- Knight H, Trewavas AJ, Knight MR (1996) Cold calcium signalling in *Arabidopsis* involves two cellular pools and a change in calcium signature after acclimation. Plant Cell 8:489–503

- Knight H, Trewavas AJ, Knight MR (1997) Calcium signalling in *Arabidopsis thaliana* responding to drought and salinity. Plant J 12:1067–1078
- Kochian KV (1995) Cellular mechanisms of aluminum toxicity and tolerance in plants. Annu Rev Plant Physiol Plant Mol Biol 46:237–260
- Kosuta S, Hazledine S, Sun J, Miwa H, Morris RJ et al (2008) Differential and chaotic calcium signatures in the symbiosis signalling pathway of legumes. Proc Natl Acad Sci USA 105:9823–9828
- Kreimer G, Melkonian M, Holtum JAM, Latzko E (1985) Characterization of calcium fluxes across the envelope of intact spinach chloroplasts. Planta 166:515–523
- Krol E, Dziubinska H, Trebacz K (2004) Low-temperatureinduced trans-membrane potential changes in mesophyll cells of Arabidopsis thaliana, Heliathus annuus and Vicia faba. Physiol Plant 120:265–270
- Kudla J, Batistic O, Hashimoto K (2010) Calcium signals: the lead currency of plant information processing. Plant Cell 22:541–563
- Kummer U, Krajnc B, Pahle J, Green AK, Dixon CJ, Marhl M (2005) Transition from stochastic to deterministic behavior in calcium oscillations. Biophys J 89:1603–1611
- Kurkela S, Franck M (1990) Cloning and characterisation of a cold- and ABA-induced *Arabidopsis* gene. Plant Mol Biol 15:137–144
- Lahiri S, Roy A, Baby SM, Hoshi T, Semenza GL, Prabhakar NR (2006) Oxygen sensing in the body. Prog Biophys Mol Biol 91:249–286
- Lamb C, Dixon R (1997) The oxidative burst in plant disease resistance. Annu Rev Plant Physiol Plant Mol Biol 48:251–275
- Laohavisit A, Mortimer JC, Demidchik V, Coxon KM, Stancombe MA et al (2009) Zea mays annexins modulate cytosolic free Ca²⁺ and generate a Ca²⁺ permeable conductance. Plant Cell 21:479–493
- Lazof DB, Goldsmith JG, Rufty TW, Linton RW (1994) Rapid uptake of aluminum into cells of soybean root tips. Plant Physiol 106:1107–1114
- Lecourieux D, Lamotte O, Bourque S, Wendehenne D, Mazars C et al (2005) Proteinaceous and oligosaccharidic elicitors induce different calcium signatures in the nucleus of tobacco cells. Cell Calcium 38:527–538
- Lee Y, Jung JW, Kim SK, Hwang YS, Lee JS, Kim SH (2008) Ethylene-induced opposite redistributions of calcium and auxin are essential components in the development of tomato petiolar epinastic curvature. Plant Physiol Biochem 46:685–693
- Levina NN, Lew RR, Hyde GJ, Heath IB (1995) The roles of Ca²⁺ and plasma membrane ion channels in hyphal tip growth of *Neurospora crassa*. J Cell Sci 108: 3405–3417
- Levine A, Pennell RI, Alvarez ME, Palmer R, Lamb C (1996) Calcium mediated apoptosis in a plant hypersensitive disease resistance response. Curr Biol 6:427–437
- Liang C, Zhang XY, Luo Y, Wang GP, Zou Q, Wang W (2009) Over-accumulation of glycine betaine alleviates the negative effects of salt stress in wheat. Russian J Plant Physiol 56:370–376
- Licausi F, Perata P (2009) Low oxygen signalling and tolerance in plants. Adv Bot Res 50:139–198

- Limpens E, Franken C, Smit P, Willemse J, Bisseling T, Guerts R (2003) LysM domain receptor kinases regulation rhizobial nod factor-induced infection. Science 302:630–633
- Lin C, Guo WW, Everson E, Thomashow MF (1990) Cold acclimation in *Arabidopsis* and wheat. A response associated with expression of related genes encoding boilingstable polypeptides. Plant Physiol 94:1078–1083
- Lin C, Yu Y, Kadono T, Iwata M, Uemura K, Furuichi T, Kuse M, Isobe M, Yamamoto Y et al (2005) Action of aluminum, novel TPC1-type channel inhibitor, against salicylate-induced and cold-shock-induced calcium influx in tobacco BY-2 cells. Biochem Biophys Res Commun 332:823–830
- Lindberg S (1990) Aluminium interactions with K⁺ (⁸⁶Rb⁺) and ⁴⁵Ca²⁺ fluxes in three cultivars of sugar beet (*Beta vulgaris*). Physiol Plant 79:275–282
- Lindberg S, Greger M (2002) Plant genotypic differences under metal deficient and enriched conditions. In: Prasad MNV, Strzaflka K (eds) Physiology and biochemistry of metal toxicity and tolerance in plants. Kluwer Academic Press, Dordrecht, The Netherlands
- Lindberg S, Griffiths G (1993) Aluminium effects on ATPase activity and lipid composition of plasma membranes in sugar beet roots. J Exp Bot 44(267):1543–1550
- Lindberg S, Strid H (1997) Aluminium induces rapid changes in cytosolic pH and free calcium and potassium concentrations in root protoplasts of wheat (*Triticum aestivum*). Physiol Plant 99:405–414
- Lindberg S, Wingstrand G (1985) Mechanism for Cd²⁺ inhibition of (K⁺+ Mg²⁺)ATPase activity and K⁺(⁸⁶Rb⁺) uptake in roots of sugar beet (*Beta vulgaris*). Physiol Plant 63:181–185
- Lindberg S, Szynkier K, Greger M (1991) Aluminium effects on transmembrane potential in cells of fibrous roots of sugar beet. Physiol Plant 79:275–282
- Lindberg S, Landberg T, Greger M (2004) A new method to detect cadmium uptake in protoplasts. Planta 219:526–532
- Lindberg S, Landberg T, Greger M (2007) Cadmium uptake and induction of phytochelatins in wheat protoplasts. Plant Physiol Biochem 45:47–53
- Lindsay WL (1979) Chemical equilibria in soils. Wiley, New York, pp 34–49
- Liu X, Shi WL, Zhang SQ, Lou CH (2005) Calcium involved in the signalling pathway of jasmonic acid induced stomatal closure of *Vicea faba*. J Exp Biol Acta 38:297–302
- Logan DC, Knight MR (2003) Mitochondrial and cytosolic calcium dynamics are differentially regulated in plants. Plant Physiol 133:21–24
- Luan S, Li W, Rusnak F, Assmann SM, Schreiber SL (1993) Immunosuppressants implicate protein phosphatase regulation of K⁺ channels in guard cells. Proc Natl Acad Sci USA 90:2202–2206
- Luan S, Lan W, Lee SC (2009) Potassium nutrition, sodium toxicity, and calcium signalling: connections through the CBL-CIPK network. Curr Opin Plant Biol 12:339–346
- Ludwig AA, Saitoh H, Felix G, Freymark G, Miersch O, Wasternack C, Boller T, Jones JD, Romeis T (2005)

Ethylene-mediated cross-talk between calcium-dependent protein kinase and MAPK signalling controls stress responses in plants. Proc Natl Acad Sci USA 102:10736–10741

- Lynch J, Läuchli A (1988) Salinity affects intracellular calcium in corn root protoplasts. Plant Physiol 87:351–356
- Lynch J, Polito VS, Läuchli A (1989) Salinity stress increases cytoplasmic calcium activity in maize root protoplasts. Plant Physiol 90:1271–1274
- Ma W, Berkowitz GA (2007) The grateful dead: calcium and cell death in plant innuate immunity. Cell Microbiol 9:2571–2585
- Ma L, Xu X, Cui S, Sun D (1999) The presence of a heterotrimeric G protein and its role in signal transduction of extracellular calmodulin in pollen germination and tube growth. Plant Cell 11:1351–1363
- Ma Y, Song W, Liu Z, Zhang H, Guo X, Shao H, Ni F (2009) The cell dynamic changing of Ca²⁺ cellular localization in maize leaflets under drought stress. C R Biol 332:351–362
- Madsen EB, Madsen LH, Radutoiu S, Olbryt M, Rakwalska M, Szczglowski K, Sato S et al (2003) A receptor kinase gene of the LysM type is involved in legume perception of rhizobial signals. Nature 425:637–640
- Maksymiec W, Baszynski T (1999) Are calcium and calcium channels involved in the mechanisms of Cu²⁺ toxicity in bean plants? The influence of leaf age. Photosynthetica 36:267–278
- Maksymiec W, Krupa Z (2002) Jasmonate and heavy metals in Arabidopsis plants-a similar physiological response to both stressors? J Plant Physiol 159:509–515
- Maksymiec W, Krupa Z (2006) The effects of short-term exposition to Cd, excess Cu ions and jasmonate on oxidative stress appearing in *Arabidopsis thaliana*. Environ Exp Bot 57:187–194
- Maksymiec W, Russa R, Urbanik-Sypniewska T, Baszynski T (1994) Effect of excess Cu on photosynthetic apparatus of runner bean leaves treated at two different growth stages. Physiol Plant 91:715–721
- Maksymiek W (1997) Effect of copper on cellular processes in higher plants. Photosynthetica 34:321–342
- Maksymiek W (2007) Signalling responses in plants to heavy metal stress. Acta Physiol Plant 29:177–187
- Malho R, Moutinho A, Vanderluit A, Trewavas AJ (1998) Spatial characterization of calcium signalling: the calcium wave as a basic unit in plant cell calcium signalling. Philos Trans R Soc Lond B Biol Sci 353:1463–1473
- Marschner H (1995) Mineral nutrition of higher plants, 2nd edn. Academic, London
- Martin ML, Busconi L (2001) A rice membrane-bound calcium-dependent protein kinase is activated in response to low temperature. Plant Physiol 125:1442–1449
- Martínez-Beltran J, Manzur CL (2005) Overview of salinity problems in the world and FAO strategies to address the problem. In: Managing saline soils and water: science, technology and social issues. Proceedings international salinity forum. Riverside, CA, pp 311–313
- Matsumoto H (1988) Inhibition of proton transport activity of microsomal membrane vesicles of barley roots by aluminum. Soil Sci Plant Nutr 34:499–506

- Matsumoto H (2000) Cell biology of aluminium toxicity and tolerance in higher plants. Int Rev Cytol 200:1–46
- Matsumoto H, Hirasawa E, Torikai H, Takahashi E (1976) Plant Cell Physiol 17:127–137
- Mazars C, Bourque S, Mithöfer A, Pugin A, Ranjeva R (2009) Calcium homeostasis in plant cell nuclei. New Phytol 181:261–274
- Mazars C, Thuleau P, Lamotte O, Bourque S (2010) Cross-talk between ROS and calcium in regulation of nuclear activities. Mol Plant 3:1–13
- McAinsh MR, Pittman JK (2009) Shaping the calcium signature. New Phytol 181:275–294
- Metwally A, Finkelmeier I, Georgi M, Dietz KJ (2003) Salicylic acid alleviates the cadmium toxicity in barley seedlings. Plant Physiol 132:272–281
- Meyer T, Allbritton NL, Oancea E (1995) Regulation of nuclear calcium concentration. Ciba Found Symp 188:252–262
- Mironova GD, Belosludtsev KN, Belosludtseva NV, Gritsenko EN, Khodorov BI, Saris N-E (2007) Mitochondrial Ca²⁺ cycle mediated by palmitate-activated cyclosporine A-insensitive pore. J Bioenerg Biomembr 39:167–174
- Miyazaki S, Koga R, Bohnert HJ, Fakuhara T (1999) Tissue and environmental response specific expression of 10 PP2C transcripts in *Mesembryanthemum crystallinum*. Mol Gen Genet 261:307–316
- Monroy AF, Dhindsa R (1995) Low-temperature signal transduction: induction of cold acclimation-specific genes of alfalfa by calcium at 25°C. Plant Cell 7:321–331
- Monroy AF, Sarhan F, Dhindsa RS (1993) Cold-induced changes in freezing tolerance, protein phosphorylation, and gene expression: evidence for a role of calcium. Plant Physiol 102:1227–1235
- Morel M, Crouzet J, Gravot A, Auroy P, Leonhardt N, Vavasseur A, Richaud P (2009) AtHMA3 a P_{1B}-ATPase allowing CD/Zn/Co/Pb vacuolar storage in *Arabidopsis*. Plant Physiol 149:894–904
- Mori IC, Schroeder JI (2004) Reactive oxygen species activation of plant Ca²⁺ channels: a signalling mechanism in polar growth, hormone transduction, stress signalling, and hypothetically mechanotransduction. Plant Physiol 135:702–708
- Munns R (2002) Comparative physiology of salt and water stress. Plant Cell Environ 25:239–250
- Munns R, Tester M (2008) Mechanisms of salinity tolerance. Annu Rev Plant Biol 59:651–681
- Munns R, James RA, Läuchli A (2006) Approaches to increasing the salt tolerance of wheat and other cereals. J Exp Bot 57:1025–1043
- Muto S, Izawa S, Miyachi S (1982) Light-induced Ca²⁺ uptake by intact chloroplasts. FEBS Lett 139:250–254
- Mylona P, Pawlowski K, Bisseling T (1995) Symbiotic nitrogen fixation. Plant Cell 7:869–885
- Nakata M, Yuasa T, Takahasi Y, Sarahmi I (2009) Plant Signal Behav 4:372–374
- Navazio L, Moscatiello R, Genre A, Novero M, Baldan B et al (2007) A diffusible signal from arbuscular mycorrhizal fungi elicits a transient cytosolic calcium elevation in host plant cells. Plant Physiol 144:673–681
- Nomura H, Komori T, Kobori M, Nakahira Y, Shiina T (2008) Evidence for chloroplast control of external

Ca²⁺-induced cytosolic Ca²⁺ transients and stomatal closure. Plant J 53:988–998

- Nouairi I, Ammar WB, Yousef NB, Daoud DBM, Ghorbal NH, Zarrouk M (2006) Comparative study of cadmium effects on membrane lipid composition of *Brassica juncea* and *Brassica napus* leaves. Plant Sci 170:511–519
- Oldroyd GED, Downie JA (2006) Nuclear calcium changes at the core of symbiosis signalling. Curr Opin Plant Biol 9:351–357
- Oldroyd GED, Downie JA (2008) Coordinating nodule morphogenesis with rhizobial infection in legumes. Annu Rev Plant Biol 59:519–546
- Opaskornkul C, Lindberg S, Tillberg J-E (1999) Effects of ABA on the distribution of sucrose and protons across the plasmalemma of pea mesophyll protoplasts. -Suggesting a sucrose/proton symport. J. Plant Physiol 154:447–453
- Parniske M (2008) Arbuscular mycorrhiza: the mother of all plant root endosymbiosis. Nat Rev Microbiol 6:765–777
- Pauly N, Knight MR, Thuleau P, van der Luit AH, Moreau M, Trewavas AJ, Ranjeva R, Mazars C (2000) Control of free calcium in plant cell nuclei. Nature 405:754–755
- Pauly N, Knight MR, Thuleau P, Graziana A, Muto S, Ranjeva R, Mazars C (2001) The nucleus together with the cytosol generates patterns of specific cellular calcium signatures in tobacco suspension culture cells. Cell Calcium 30:413–421
- Pei ZM, Murata Y, Benning G, Thomine S, Klusener B, Allen GJ, Grill E, Schroeder JI (2000) Calcium channels activated by hydrogen peroxide mediate abscisic acid signalling in guard cells. Nature 406:731–734
- Peiter E, Maathuis FJM, Mills LN, Knight H, Pelloux J et al (2005) The vacuolar Ca²⁺activated channel TPC1 regulates germination and stomatal movement. Nature 434:404–408
- Perera IY, Hung C-Y, Moore CD, Stevenson-Paulik J, Boss WF (2008) Transgenic Arabidopsis plants expressing the type 1 inositol 5-phosphatase exhibit increased drought tolerance and altered abscisic acid signalling. Plant Cell 20:2876–2893
- Perfus-Barbeoch L, Leonhardt N, Vavasseur A, Forestier C (2002) Heavy metal toxicity: Cadmium permeates through calcium channels and disturbs the plant water status. Plant J 32:539–548
- Plieth C (1999) Temperature sensing by plants: calciumpermeable channels as primary sensors – a model. J Membr Biol 172:121–127
- Plieth C (2001) Plant calcium signalling and monitoring: pros and cons and recent, experimental approaches. Protoplasma 218:1–23
- Plieth C, Hansen UP, Knight H, Knight MR (1999) Temperature sensing by plants: the primary characteristics of signal perception and calcium response. Plant J 18:491–497
- Polevoi V, Sinyutina NF, Salamatoma TS, Inge-Vechtomova NI, Tankelyun OV, Sharova EI, Shishova MF (1996) Mechanism of auxin action: second messengers. In: Smith AR et al (eds) Plant hormone signal perception

and transduction. Kluwer Academic Publishers, Amsterdam, pp 223–231. ISBN 0-7923-3768-9

- Poovaiah BW, Reddy ASN (1987) Calcium messenger system in plants. CRC Crit Rev Plant Sci 6:1–46
- Qi Z, Stephens NR, Spalding EP (2006) Calcium entry mediated by GLR3.3, an *Arabidopsis* glutamate receptor with a broad agonist profile. Plant Physiol 142:963–971
- Quartacci MF, Cosi E, Navari-Izzo F (2001) Lipids and NADPH-dependent superoxide production in plasma membrane vesicles from roots of wheat grown under copper deficiency or excess. J Exp Bot 52:77–84
- Rahman H, Sabreen S, Alam S, Kawai S (2005) Effect of nickel on growth and composition of metal micronutrients in barley plants grown in nutrient solution. J Plant Nutr 28:393–404
- Raven JA, Edwards D (2001) Roots: evolutionary origins and biogeochemical significance. J Exp Bot 52:381–401
- Reddy ASN (2001) Calcium: silver bullet in signalling. Plant Sci 160:381–404
- Reiser V, Raitt DC, Saito H (2003) Yeast osmosensor Sln1 and plant cytokinin receptor Cre1 respond to changes in turgor pressure. J Cell Biol 161:1035–1040
- Rengel Z, Zhang W-H (2003) Role of dynamics of intracellular calcium in aluminium-toxicity syndrome. New Phytol 159:295–314
- Rentel M, Knight MR (2004) Oxidative stress-induced calcium signalling in *Arabidopsis*. Plant Physiol 135:1471–1479
- Rhoads DM, Umbach AL, Subbaiah CC, Siedow JN (2006) Mitochondrial ROS: contribution to oxidative stress and inter-organellar signalling. Plant Physiol 141:357–366
- Rizutto R, Marchi S, Bonora M, Aguiari P, Bononi A, De Stefani D et al (2009) Ca²⁺ transfer from the ER to mitochondria: when, how and why. Biochem Biophys Acta 1787:1342–1351
- Rodriguez-Concepcion M, Yalovsky S, Zik M, Fromm H, Gruissem W (1999) The prenylation status of a novel plant calmodulin directs plasma membrane or nuclear localization of the protein. EMBO J 18:1996–2007
- Rodríguez-Navarro A, Rubio F (2006) High-affinity potassium and sodium transport systems in plants. J Exp Bot 57:1149–1160
- Rodriguez-Serrano M, Romero-Puertas MC, Pazmino DM, Testillano PS, Risueno MC, del Rio LA, Sandalio LM (2009) Cellular response of pea plants to cadmium toxicity: cross talk between reactive oxygen species, nitric oxide, and calcium. Plant Physiol 150:229–243
- Ryan PR, DiTomaso JM, Kochian LV (1993) Aluminum toxicity in roots: an investigation of spatial sensitivity and the role of the root cap. J Exp Bot 44:437–446
- Sai J, Johnson CH (2002) Dark-stimulated calcium ion fluxes in the chloroplast stroma and cytosol. Plant Cell 14:1279–1291
- Sanders D, Pelloux J, Brownlee C, Harper JF (2002) Calcium at the crossroads of signalling. Plant Cell 14:S401–S417
- Sauter M (2000) Rice in deep water: "How to take heed against a sea of troubles". Naturwissenschaften 87:289–303

- Schiott M, Palmgren MG (2005) Two plant Ca²⁺ pumps expressed in stomatal guard cells show opposite expression pattern during cold stress. Physiol Plant 124:278–283
- Sebastiani L, Lindberg S, Vitagliano C (1999) Cytoplasmic free calcium dynamics in single tomato (*Lycopersicon esculentum* L) protoplasts subjected to chilling temperatures. Physiol Plant 105:239–245
- Sedbrook JC, Kronebusch PJ, Borisy GG, Trewavas AJ, Masson PH (1996) Transgenic AEQUORIN reveals organ-specific cytosolic Ca²⁺ responses to anoxia in *Arabidopsis thaliana* seedlings. Plant Physiol 111:243–257
- Seigneurin-Berny D, Gravot A, Auroy P, Mazard C, Kraut A et al (2006) HMA1, a new Cu-ATPase of the chloroplast envelope, is essential for growth under adverse light conditions. J Biol Chem 281:2882–2892
- Shabala S, Newman I (1999) Light-induced changes in hydrogen, calcium, potassium and chloride ion fluxes and concentrations from the mesophyll and epidermal tissues of bean leaves. Understanding the ionic basis of light-induced bioelectrogenesis. Plant Physiol 119:1115–1124
- Shabala L, Cuin TA, Newman IA, Shabala S (2005) Salinity-induced ion flux patterns from the excised roots of *Arabidopsis* sos mutants. Planta 222:1041–1050
- Sharma P, Deswal R (2004) Detection and characterization of calcineurin-like activity in *Brassica juncea* and its activation by low temperature. Plant Physiol Biochem 42:579–584
- Sharma SS, Dietz KJ (2009) The relationship between metal toxicity and cellular redox imbalance. Trends Plant Sci 14:43–50
- Shi B, Haug A (1988) Uptake of aluminium by lipid vesicles. Toxicol Environ Chem 17:337–349
- Shigaki T, Hirschi KD (2006) Diverse functions and molecular properties emerging for CAX cation/H⁺ exchangers in plants. Plant Biol 8:419–429
- Shinozaki K, Yamaguchi-Shinozaki K (2000) Molecular responses to dehydration and low temperature: differences and cross-talk between two stress signalling pathways. Curr Opin Plant Biol 3:217–223
- Shinozaki K, Yamaguchi-Shinozaki K (2007) Gene networks involved in drought stress response and tolerance. J Exp Bot 58:221–227
- Shishova M, Lindberg S (1999) Auxin-induced cytosol acidification in wheat leaf protoplasts depends on external concentration of Ca²⁺. J Plant Physiol 155:190–196
- Shishova M, Lindberg S (2004) Auxin induces rapid increase of Ca²⁺ concentration in the cytosol of wheat leaf protoplasts. J Plant Physiol 161:937–945
- Shishova M, Lindberg S (2010) A new perspective on auxin perception. J Plant Physiol 167:417–422
- Shishova M, Yemelyanov V, Rudashevskaya E, Lindberg S (2007) A shift in sensitivity to auxin within development of maize seedlings. J Plant Physiol 164:1323–1330
- Sprent JI (2006) Evolving ideas of legume evolutional diversity: a taxonomic perspective on occurrence of nodulation. New Phytol 174:11–25

- Staxen II, Pical C, Montgomery LT, Gray JE, Hetherington AM, McAinsh MR (1999) Abscisic acid induces oscillations in guard-cell cytosolic free calcium that involve phosphoinositide-specific phospholipase C. Proc Natl Acad Sci USA 96:1779–1784
- Steponkus PL, Lynch DV (1989) The behavior of large unilamellar vesicles of rye plasma membrane lipids during freeze/thaw-induced osmotic excursions. Cryo Lett 10:43–50
- Subbaiah CC (2009) Ionic loops and rebounds: oxygendeprivation signalling in plants. In: Baluška F, Mancuso S (eds) Signalling in plants. Springer, Berlin, pp 195–207
- Subbaiah C, Bush DS, Sachs M (1994a) Elevation of cytosolic calcium precedes anoxic gene expression in maize suspension-cultured cells. Plant Cell 6:1747–1762
- Subbaiah C, Zhang J, Sachs M (1994b) Involvement of intracellular calcium in anaerobic gene expression and survival of maize seedlings. Plant Physiol 105:369–376
- Subbaiah CC, Bush DS, Sachs MM (1998) Mitochondrial contribution to the anoxic Ca²⁺ signal in maize suspension-cultured cells. Plant Physiol 118:759–771
- Subbarao GV, Ito O, Berry WL, Wheeler RM (2003) Sodium – a functional plant nutrient. Crit Rev Plant Sci 22:391–416
- Sun DY, Li HB, Cheng G (1994) Extracellular calmodulin accelerates the proliferation of suspension-cultured cells of *Angelica dahurica*. Plant Sci 99:1–8
- Sun QP, Guo Y, Sun Y, Sun DY, Wang XJ (2006) Influx of extracellular Ca²⁺ involved in jasmonic-acid-induced elevation of [Ca_{cy}²⁺] and JR1 expression in *Arabdopsis* thaliana. J Plant Res 119:343–350
- Tähtiharju S, Sangwan V, Monroy AF, Dhindsa RS, Borg M (1997) The induction of *kin* genes in cold-acclimating *Arabidopsis thaliana*. Evidence of a role for calcium. Planta 203:442–447
- Taiz L, Zeiger E (2006) Plant physiology, 4th edn. Sinauer Associates Inc. Publishers, Sunderland, MA, p 694
- Takahashi H, Scott TK, Suge H (1992) Stimulation of root elongation and curvature by calcium. Plant Physiol 98:246–252
- Tamas L, Dudikova J, Durcekova K, Haluskova L, Huttova J, Mistrik I, Ollé M (2008) Alterations of the gene expression, lipid peroxidation, proline and thiol content along the barley root exposed to cadmium. J Plant Physiol 165:1193–1203
- Tamura T, Hara K, Yamaguchi Y, Koizumi N, Sano H (2003) Osmotic stress tolerance of transgenic tobacco expressing a gene encoding a membrane-located receptor-like protein from tobacco plants. Plant Physiol 131:454–462
- Tester M, Davenport R (2003) Na⁺ tolerance and Na⁺ transport in higher plants. Ann Bot 91:503–527
- Tiwari BS, Belenghi A, Levine A (2002) Oxidative stress increased respiration and generation of reactive oxygen species, resulting in ATP depletion, opening of mitochondrial permeability transition, and programmed cell death. Plant Physiol 128:1271–1281

- Tracy FE, Gilliham M, Dodd AN, Webb AAR, Tester M (2008) Nacl-induced changes in cytosolic free Ca²⁺ in *Arabidopsis thaliana* are heterogeneous and modified by external ionic composition. Plant Cell Environ 31: 1063–1073
- Tran LS, Urao T, Qin F, Maruyama K, Kakimoto T, Shinozaki K, Yamaguchi-Shinozaki K (2007) Functional analysis of AHK1/ATHK1 and cytokinin receptor histidine kinases in response to abscisic acid, drought, and salt stress in *Arabidopsis*. Proc Natl Acad Sci USA 104:20623–20628
- Tsuji H, Nakazono M, Saisho D, Tsutsumi N, Hirai A (2000) Transcript levels of the nuclear-encoded respiratory genes in rice decrease by oxygen deprivation: evidence for involvement of calcium in expression of the alternative oxidase 1a gene. FEBS Lett 471:201–204
- Tuteja N (2007) Mechanisms of high salinity tolerance in plants. Methods Enzymol 428:419–438
- Tuteja N, Mahajan S (2007) Calcium signalling network in plants. Plant Signal Behav 2:79–85
- Tuteja N, Sopory SK (2008) Chemical signalling under abiotic stress environment in plants. Plant Signal Behav 3:525–536
- Uemura M, Joseph R-A, Steponkus P-L (1995) Cold acclimation of *Arabidopsis thaliana*. Effect on plasma membrane lipid composition and freeze-induced lesions. Plant Physiol 109:15–30
- Uemura M, Tominaga Y, Nakagawara C, Shigematsu S, Minami A, Kawamura Y (2006) Responses of the plasma membrane to low temperatures. Physiol Plant 126:81–89
- Unyayar S, Celik A, Cekic OF, Gozel A (2006) Cadmiuminduced genotoxicity, cytotoxicity, and lipid peroxidation in *Allium sativum* and *Vicia faba*. Mutagenesis 21:77–81
- Urao T, Yakubov B, Satoh R, Yamaguchi-Shinozaki K, Seki M, Hirayama T, Shinozaki K (1999) A transmembrane hybrid-type histidine kinase in *Arabidopsis* functions as an osmosensor. Plant Cell 11:1743–1754
- Vallee BL, Ulmer DD (1972) The biochemical effects of mercury, cadmium and lead. Annu Rev Biochem 41:91–128
- van der Luit AH, Olivari C, Haley A, Knight MR, Trewavas AJ (1999) Distinct calcium signalling pathways regulate calmodulin gene expression in tobacco. Plant Physiol 121:705–714
- Vanderbeld B, Snedden WA (2007) Developmental and stimulus-induced expression patterns of *Arabidopsis* calmodulin-like genes CML37, CML38, CML39. Plant Mol Biol 64:683–697
- Vangronsveld J, Clijsters H (1994) Toxic effects of metals. In: Farago M (ed) Plants and the chemical elements. VCH Verlagsgesellshaft, Weinheim, pp 149–177
- Vartapetian BB, Jackson MB (1997) Plant adaptations to anaerobic stress. Ann Bot 79(Suppl A):3–20
- Vinit-Dunand F, Epron D, Alaoui-Sossé B, Badot P-M (2002) Effects of copper on growth and on photosynthesis of mature and expanding leaves in cucumber plants. Plant Sci 163:53–58

- Viswanathan C, Zhu J, Zhu J-K (2006) Gene regulation during cold acclimation in plants. Physiol Plant 126:52–61
- Wang X (2001) Plant phospholipases. Annu Rev Plant Physiol Mol Biol 52:211–231
- Watashiki MK, Trewavas AJ, Parton RM (2004) Fluctuations in pollen tube tip-focused calcium gradient are not reflected in nuclear calcium level: a comparative analysis using recombinant yellow camelion reporter. Sexual Plant Reprod 17:125–130
- Weini S, Held K, Schlückling K, Steinhorst L, Kuhlgert S, Hippler M, Kudla J (2008) A plastid protein crucial for Ca²⁺-regulated stomatal responses. New Phytol 179:675–686
- White PJ, Broadley MR (2003) Calcium in plants. Ann Bot 92:487–511
- Williams RJP (2006) The evolution of calcium biochemistry. Biochem Biophys Acta 1763:1139–1146
- Williams LE, Mills RF (2005) P(1B)-ATPases: an ancient family of transition metal pumps with diverse functions in plants. Trends Plant Sci 10:491–502
- Wohlbach DJ, Quirino BF, Sussmand MR (2008) Analysis of the Arabidopsis histidine kinase ATHK1 reveals a connection between vegetative osmotic stress sensing and seed maturation. Plant Cell 20:1101–1117
- Wymer CL, Bibikova TN, Gilroy S (1997) Cytoplasmic free calcium distributions during the development of root hairs of *Arabidopsis thaliana*. Plant J 12:427–439
- Xiong TC, Jauneau A, Ranjeva R, Mazars C (2004) Isolated plant nuclei as mechanical and thermal sensors involved in calcium signalling. Plant J 40:12–21
- Xiong TC, Bourque S, Lecourieux D, Amelot N, Grat S, Briere C, Mazars C, Pugin A, Ranjeva R (2006) Calcium signalling in plant cell organelles delimited by a double membrane. Biochim Biophys Acta 1763:1209–2015
- Xiong TC, Coursol S, Grat S, Ranjeva R, Mazars C (2008) Sphingolipid metabolites selectively elicit increase in nuclear calcium concentration in cell suspension cultures and in isolated nuclei of tobacco. Cell Calcium 43:29–37
- Yamaguchi-Shinozaki K, Shinozaki K (2006) Transcriptional regulatory networks in cellular responses and tolerance to dehydration and cold stresses. Annu Rev Plant Biol 57:781–803
- Yamazaki T, Kawamura Y, Minami A, Uemura M (2008) Calcium-dependent freezing tolerance in *Arabidopsis* involves membrane resealing via synaptotagmin SYT1. Plant Cell 20:3389–3404
- Yang T, Poovaiah BW (2002) A calmodulin-binding/ CGCG box DNA-binding protein family involved in multiple signalling pathways in plants. J Biol Chem 277:45049–45058
- Yemelyanov V, Shishova M, Chirkova T, Lindberg S (2011) Anoxia-induced elevation of cytosolic Ca²⁺ concentration depends on different Ca²⁺ sources in rice and wheat protoplasts. Planta. doi:10.1007/s00425-011-1396-x
- Zhang JZ, Creelman RA, Zhu JK (2004) From laboratory to field. Using information from Arabidopsis to engineer

salt, cold, and drought tolerance in crops. Plant Physiol 135:615–621

- Zhao MG, Tian QY, Zhang WH (2007) Ethylene activates a plasma membrane Ca(²⁺)-permeable channel in tobacco suspension cells. New Phytol 174:507–515
- Zhao J, Barkla BJ, Marshall J, Pittman JK, Hirschi KD (2008) The *Arabidopsis cax3* mutants display altered salt tolerance, pH sensitivity and reduced plasma membrane H⁺-ATPase activity. Planta 227: 659–669
- Zhu JK (2001) Plant salt tolerance. Trends Plant Sci 6:66–71
- Zhu JK (2002) Salt and drought stress signal transduction in plants. Annu Rev Plant Biol 53:247–273
- Zhu JK (2003) Regulation of ion homeostasis under salt stress. Curr Opin Plant Biol 6:441–445
- Zhu J, Dong CH, Zhu JK (2007) Interplay between cold-responsive gene regulation, metabolism and RNA processing during plant cold acclimation. Curr Opin Plant Biol 10:290–295

Role of H₂O₂ as Signaling Molecule in Plants

16

M.A. Matilla-Vázquez and A.J. Matilla

Abstract

Hydrogen peroxide (H_2O_2) is continuously produced as a by-product of oxidative plant aerobic metabolism. It plays a dual role in cells. Under normal conditions, H₂O₂ acts as a key regulator of many biological processes because H2O2 have been identified as an important second messenger in signal transduction networks. On the other hand, H₂O₂ is a sharp oxidant triggering cell damages or even death during oxidative stress. Therefore, cellular "redox homeostasis" is tightly regulated by H₂O₂ production and scavenging. In these processes a great number of genes are involved. Organelles with a high oxidizing metabolic activity or with an intense rate of electron flow, such as chloroplasts, mitochondria, or peroxisomes are major sources of H2O2 production. However, its notable apoplastic production by means of a plasmalemma NADPH-oxidase must be taken in account. H₂O₂, interacting with thiol-containing proteins, can modulate the activities of many components in signaling, such as protein kinases, protein phosphatases, or transcription factors. Moreover, the H₂O₂ cascade can result in the downstream activation of Ca2+ channels, which may be the central step in many H₂O₂-mediated processes. All these summarized alterations trigger the regulation of important processes like gene expression, elongation growth, ABA-mediated stomatal closure, programmed cell death, defense against pathogens, seed dormancy, and germination. The molecular mechanisms by which these processes are affected by H_2O_2 signaling have not been completely clarified. Here, we describe several aspects of H₂O₂ production, scavenging and gene regulation, and cross-talk with ABA and ethylene during plant growth and development.

A.J. Matilla (🖂)

M.A. Matilla-Vázquez Department of Biochemistry, University of Cambridge, Tennis Court Road, Cambridge CB2 1QW, UK

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_16, © Springer Science+Business Media, LLC 2012

Departamento de Fisiología Vegetal, Facultad de Farmacia, Universidad de Santiago de Compostela (USC), 15782 Santiago de Compostela, Spain e-mail: angeljesus.matilla@usc.es

Keywords

 $H_2O_2 \bullet$ Production \bullet Scavenging \bullet Signaling \bullet Transcription factors \bullet Plant interactions \bullet Plant growth

1 Introduction

The evolution of oxygen (O₂) metabolism in higher plants led to the production of reactive oxygen species (ROS) in the organelles involved in this aerobic process (mitochondria, chloroplasts, and peroxisomes) (Apel and Hirt 2004; Slesak et al. 2007; Corpas et al. 2009). ROS appear under physiological conditions and they are highly reactive molecules that can disturb membranes and other cellular components. Hence, it is important to remove these toxic molecules from cells to prevent stress-induced injuries. The plants possess a complex battery of antioxidant defense systems to regulate ROS production with beneficial effects. ROS are generated in the cells by several enzymatic reactions and electron transport systems (Fig. 16.1). The ROS includes superoxide $(O_2^{\bullet-})$, hydroxyl (HO) radicals, singlet oxygen $({}^{1}O_{2})$, and hydrogen peroxide (H_2O_2) . HO[•] has indiscriminate reactivity towards biological molecules, whereas O2-- and H₂O₂ have preferential biological targets. Plants have an abundance of antioxidant defense systems to regulate their production with beneficial



Fig. 16.1 Sequence of electron transfer reactions at the O, molecule

effects. Under abiotic (e.g., drought, salinity, flooding, heat, and cold) and biotic stresses (Gechev et al. 2006; Gadjev et al. 2008; Miller et al. 2008), the overproduction of ROS can overcome the antioxidant systems (scavenging mechanisms) resulting in oxidative damage of lipids, sugars, proteins, and nucleic acids. In some cases, the inactivation of specific targets loss of physiological functions takes place, thereby leading to cell death (Gadjev et al. 2008). It became clear that ROS play a dual role in plants as toxic compounds as well as key regulators of many biological processes such as growth, cell cycle, hormone signaling, biotic and abiotic cell responses, programmed cell death (PCD), and plant development (Apel and Hirt 2004; Miller et al. 2008; Corpas et al. 2009). In the last decade, H₂O₂ has been identified as an important second messenger in signal transduction network of plants. Here, the latest data related to H_2O_2 signaling are described and summarized.

2 Chemical Properties of H₂O₂

In aqueous solution, $O_2^{\bullet-}$ is moderately reactive, but it can generate H_2O_2 by dismutation. The H₂O₂ belongs to non-radical ROS, carries no net charge, is the only ROS species that is stable in solution (cellular half-life ~1 ms, steady-state levels $\sim 10^{-7}$ M), and it can diffuse across biological membranes making H₂O₂ fit for signaling. Diffusion of H₂O₂ might be modulated by changes in lipid membrane permeability or by transport through aquaporins (Bienert et al. 2006). Its toxicity is essentially the consequence of its reduction to HO• by metal-catalyzed Fenton chemistry. H₂O₂ is actually a poor oxidant that reacts mildly with [Fe-S] (rate constant of 10^2-10^3 M⁻¹s⁻¹), it loosely binds metals $(10^3-10^4 \text{ M}^{-1} \text{ s}^{-1})$, and it also reacts with methionine residues (10² M⁻¹ s⁻¹). In contrast, its reactivity



towards Cys residues can significantly increase to 10–10⁶ M⁻¹ s⁻¹ depending on the protein environment (D'Autréaux and Toledano 2007). Sulfur-mediated nucleophilic attack of the peroxide O–O bond by the Cys thiol group (-SH) leads to H_2O_2 release and formation of sulphenic acid (-SOH). The -SOH is highly reactive, its stability is influenced by the availability of a proximal nitrogen to form a sulphenamide or by the presence of H_2O_2 , which further oxidizes it to form sulphinic (-SO₂H) or sulphonic (-SO₃H) acids (Spadaro et al. 2010) (Fig. 16.2).

Production of H₂O₂ by Plants 3

Close to 1% of O₂ consumed by plants is diverted to produce ROS in various subcellular sites (Bhattachrjee 2005). The multiple sites and sources of ROS production increase its complexity. The endogenous concentration of H_2O_2 is reported in a wide margin, ranging from nmol to several hundred µmol/gFW. Plant tissues can tolerate high concentrations of H₂O₂ in the range of $10^2-10^5 \mu$ M. Normally, H_2O_2 generation is often maintained at a constant basal level in healthy cells, but their levels increase transiently or persistently in response to stress (Desikan et al. 2003; Apel and Hirt 2004). Although it was previously hypothesized that H2O2 produced intracellularly diffuses to other cells, it now appears more probable that intracellularly produced H_2O_2 is consumed quickly and locally, and that extracellular metabolism uses H₂O₂ produced extracellularly. Transmembrane movements of H₂O₂ (e.g., from the apoplast to the cytosol) probably involve aquaporins (Bienert et al. 2007). The

H₂O₂ production requires a continuous Ca²⁺ entry to activate the plasmalemma-localized NADPHoxidase, leading to Ca2+-dependent cellular responses resulting from anion and K⁺ exit. This fate was also confirmed by the administration of Ca²⁺ inhibitors which prevent increases of cytosolic Ca^{2+} concentration. Under these conditions, the accumulation of endogenous H₂O₂ was prevented.

 H_2O_2 is produced by means of the following reactions:

$$O_2^{-} + 2H^+ \rightarrow H_2O_2 + O_2$$

 $O_2^{-} + H^+ + XH_2 \rightarrow H_2O_2 + XH$
(X = reductant molecule)

(

That is, H₂O₂ production is related to the synthesis of O₂⁻⁻. However, glycolate, glucose, amino-acid, and sulfite oxidases also release H_2O_2 following the oxidation of their respective substrates. Moreover, cell wall (CW)-bound peroxidases and oxalate, amine and plasmalemma NADPH-oxidases are other enzymatic sources of $O_2^{\bullet-}$ and H_2O_2 . Peroxisomes, mitochondria, and chloroplasts are cellular compartments involved in H₂O₂ production (Mittler et al. 2004). It is estimated that the chloroplast/peroxisome system generates about 90% of the total H_2O_2 in the photosynthetically active plant cell. Excess H₂O₂ leads to oxidative stress and it is capable to cause injury to cells. For that reason, during the course of evolution plants were able to achieve a high degree of control over H₂O₂ accumulation. It is suggested that the tolerance of plants to high H_2O_2 levels is due to the fact that plant antioxidant response systems are designed more for the control of the cellular redox state than for complete elimination of H₂O₂.

Table 16.1	H ₂ O ₂ (nmol mg ⁻¹	protein)	production	in			
different plant organelles							
Chloroplast	Mitochondria	Peroxisomes					

1.79 1.30 2.81

For more details see Corpas et al. (2009)



Cytosol

Fig. 16.3 Generation of H_2O_2 as a consequence of plant peroxisomal metabolism. *CA* catalase, *Cyt-b* cytochrome-b, *MDAR* mono-dehydro-ascorbate reductase, *SOD* super-oxide dismutase, *XOD* xanthine oxidase. For details see Corpas et al. (2009)

3.1 Peroxisomes (Glyoxisomes)

Peroxisomes are one of the main sites for the generation of H₂O₂ due to an essentially oxidative type of metabolism (Table 16.1). The H_2O_2 production is mainly through the photorespiration and fatty acid β -oxidation pathways. At least two sites of $O_2^{\bullet-}$ generation were demonstrated in the peroxisome (Fig. 16.3): (1) in the matrix by means of xanthine oxidase (XOD), which catalyzes the oxidation of xanthine and hypoxanthine to uric acid and it is a well-known producer of $O_2^{\bullet-}$; and (2) in the organelle membranes being dependent on NAD(P)H (Corpas et al. 2009). The superoxide dismutase (SOD) is a metalloenzyme that catalyzes the synthesis of H₂O₂ following the reaction: $O_2^{\bullet-} + O_2 + 2H^+ \rightarrow H_2O_2 + O_2$. The SOD is an important enzymatic component of a defense mechanism that protects cells from the damaging action of $O_2^{\bullet-}$ radicals. The various

Table 16.2 Peroxisomal enzymes involved in the H_2O_2 production

Fatty acid degredation
Short chain acyl-CoA oxidase
Medium chain acyl-CoA oxidase
Long chain acyl-CoA oxidase
Photorespiration
Glycolate oxidase
Purine metabolism
Urate oxidase
Xanthine oxidoreductase
Catalase (CAT)
Ascorbate-glutathione cycle
Ascorbate peroxidase (APX)
Peroxiredoxin
Others
Sarcosine oxidase
Sulfite oxidase

With the exception of APX (membrane-bound), all enzymes have been found in the peroxisomal matrix

SOD forms are induced with different kinetics during sustained stress conditions. The presence in peroxisomes of Cu/Zn-SOD was unequivocably demonstrated. However, the genes encoding this peroxisome Cu/Zn-SOD have not yet been identified. Other peroxisome enzymes involved in the H_2O_2 production are included in Table 16.2 (Corpas et al. 2009). On the other hand, the uricase or urate oxidase, essential in the plant catabolism of purines, catalyzes the oxidation of uric acid to allantoin with the concomitant generation of H₂O₂. Under physiological conditions, the peroxisomal level of H_2O_2 and $O_2^{\bullet-}$ are controlled by catalase (CAT) and ascorbate peroxidase (APX) and by SOD, respectively. In peroxisomes/glyoxisomes, CAT predominates, but its properties suggest that the enzyme is inefficient to remove low concentrations of H₂O₂. Taken together all the disposable data on the H_2O_2 -producing peroxisomal β -oxidation of fatty acids, they strongly suggest that any type of plant stress inducing the peroxisomal production of $O_2^{\bullet-}$ and •NO radicals inhibits the CAT and APX activities and, perhaps, increases the level of H_2O_2 by enhancing the peroxisomal fatty acid β -oxidation.



Fig. 16.4 Generation and scavenging of H_2O_2 in plant cells. *APX* ascorbate (AsA) peroxidase, *CAT* catalase, *Fd* ferredoxine, *GR* glutathione reductase, *GSSG*-oxidized glutathione, *GSH*-reduced glutathione, *MDA* monodehy-droascorbate, *MDHAR* MDA-reductase, *SOD* superoxide dismutase. Chloroplastic H_2O_2 is detoxified by APX with the AsA as an electron donor and transformed in GSH by means of the intervention of GR and DHAR activities, respectively. During photorespiration and fatty acid

3.2 Mitochondria

In the dark or in non-photosynthetic tissues, mitochondria can be the major source of ROS comparing with chloroplasts and peroxisomes (Rhoads et al. 2006). The mitochondrial contribution in H₂O₂ generation in plant cells under light conditions requires further study since photosynthesis and respiration in the light are strongly interdependent. About 1-5% of mitochondrial O₂ consumption leads to H2O2 production (Moller 2001). The plant mitochondria have five H_2O_2 generating enzymes that exist only in plants (i.e., one alternative oxidase and four NADPH dehydrogenases assembled to flavoproteins). During respiration, H₂O₂ production is carried out in the NADH dehydrogenase complex (CI) and in the ubiquinol-cytochrome bc1 complex (CIII), respectively. In CIII, which is inhibited by KCN, it takes place the conversion of $O_2 \rightarrow O_2^{\bullet-}$ and it

oxidation, peroxisomes and glyoxisomes, respectively, produce large amounts of H_2O_2 which is rapidly scavenged by CAT. The movement of H_2O_2 between different cellular compartments is facilitated by peroxoporins (specialized aquaporins). The excess of H_2O_2 leaking into cytosol from different compartments is metabolized by various peroxidases or may eventually be transported and detoxified into the vacuole. Adapted from Gechev et al. (2006)

appears to be the major site of mitochondrial H_2O_2 production. A plant mitochondrial-specific Mn-SOD is involved in the formation of H_2O_2 from $O_2^{\bullet-}$ produced in two complex of the electron transport chain (ETC), NAD(P)H dehydrogenase and cytochrome bc_i , respectively.

3.3 Chloroplasts

Much of the ROS generated in photosynthetic plant cells is produced in chloroplasts which are notable sites of H_2O_2 production as a by-product of the reduction of O_2 (Fig. 16.4). Chloroplasts produce singlet O_2 from the excited triplet state of chlophyll (photosystem II) and $O_2^{\bullet-}$ in the Mehler reaction (photosystem I). As in the mitochondria, H_2O_2 generation in chloroplasts is also linked to the ETC (pseudocyclic electron flow). In PSI, under limiting conditions the availability of NADP or in situations of overloading of the FTC, a part of the electron flow is diverted from and ferredoxin to O_2 , and the $O_2^{\bullet-}$ can be produced abive via a Mehler reaction. There seems to be a consensus that the Mehler reaction acts as an alternative sink for an excess of electrons, generated during excess excitation energy stress. Currently, NA this reaction is considered as the primary and the most powerful source of H_2O_2/ROS in chloroplasts (Logan et al. 2006). In the stroma side of the membrane, $O_2^{\bullet-}$ is spontaneously dismutated to H_2O_2 by Cu/Zn-SOD. On the other hand, is known that the acceptor side of PSII also provides sides (A_A, Q_B) with electron leakage to O_2 producing 3.4

signaling function in the chloroplast, H_2O_2 can be secreted from it and elicit signaling in the cytosol, elsewhere in the cell or even systematically. Thus, H_2O_2 may play an important role as initiator or transducer of signals from chloroplast to the nucleus in response to environmental factors (Gálvez-Valdivieso and Mullineaux 2010).

3.4 Other Sites Producing H₂O₂

3.4.1 Plasmalemma and Cell-Wall Apoplast

CW-associated peroxidases as well as plasmalemmabound NADPH-oxidases are the main $O_2^{\bullet-}$ and H_2O_2 producing apoplastic enzymes which are regulated by various developmental and environmental stimuli. The NADPH-dependent oxidase system (rboh, respiratory burst oxidase homologue) has received further attention. It catalyzes the production of $O_2^{\bullet-}$ by one-electron reduction of O₂ using NADPH as the electron donor (Apel and Hirt 2004). Genes of *rboh* have been cloned from several plant species (Desikan et al. 2003). The $O_2^{\bullet-}$ is most likely located in the CW apoplastic space and is converted either spontaneously or by extracellular SOD to H₂O₂. The role of H₂O₂ and other ROS in CW apoplast should not only be limited to defense responses, but their primary role is in the regulation of the synthesis of CW components (e.g., lignin). On the other hand, $O_2^{\bullet-}$ and H_2O_2 are also produced by XOD during purine catabolism, ribonucleotide reductase during deoxyribonucleotide synthesis and various other oxidases induced by biotic and abiotic stresses (Foreman et al. 2003; Gechev et al. 2006). Interestingly, Rops (Rho-like small G-proteins) signaling have an important role in regulating H_2O_2 production, potentially via NADPH-oxidase (Neill et al. 2002). The O_2 deprivation was found to activate Rop signaling that in turn activates NADPH-oxidase. This fate was attenuated by H_2O_2 -induction of *RopGAP* gene expression that leads to the deactivation of Rop (Neill et al. 2002).

3.4.2 Cytoplasm

The cytosol cannot be regarded as the major source of H_2O_2 in plant cells, but it may act as a sink for H_2O_2 leaking from other cellular compartments. The function of cytosol-generated H_2O_2 is not understood. The ETC associated with the endoplasmic reticulum is the main source of H_2O_2/ROS , where reduced forms of cytochrome P450 and cytochrome P450 reductase that are involved in oxidation and hydroxylation processes. Likewise, cytochrome b_5 and cytochrome b_5 reductase that are engaged in fatty acid desaturation, donate electrons to O_2 producing $O_2^{\bullet-}$, and a cytosolic form of SOD can convert $O_2^{\bullet-}$ to H_2O_2 .

4 Scavenging of H_2O_2

In plant cells, enzymes and redox metabolites act in synergy to carry out H2O2 scavenging. Plant antioxidant enzymes together with ROSproducing enzymes constitute a highly sophistiand redundant network, which in cated Arabidopsis thaliana consist of at least 289 genes (Gechev et al. 2006). This network maintains ROS homeostasis in all cellular compartments and regulates the adjustment of ROS levels according to the cellular need at a particular time (Fig. 16.4). SODs are the only plant enzymes capable of scavenging $O_2^{\bullet-}$, whereas high concentrations of H₂O₂ can be catabolized directly by CATs or with the help of various reductants like APXs, glutathione peroxidases (GPXs), and guaicol peroxidases. The balance between SOD and the different H₂O₂-scavenging enzymes in cells is considered to be crucial in determining the steady-state level of $O_2^{\bullet-}$ and H_2O_2 . APX has a high affinity for H_2O_2 and it is probably the most important H_2O_2 scavenging enzyme produced in chloroplasts; this enzyme is also present in cytoplasm, peroxisomes, and mitochondria (Narendra et al. 2006). Likewise, by combining APX with glutathione reductase (GR), H₂O₂ can also be removed by redox reactions promoted by glutathione (GSH), hence preventing cell damage. Lipid-soluble antioxidants like ascorbate (AsA; vitamin C), GSH, tocopherol (vitamin E), and carotenoids are non-enzymatic antioxidants that contribute to ROS homeostasis in plants (Dellapenna and Pogson 2006; Gechev et al. 2006). Since GSH appears in plant cells in millimolar concentrations, it is regarded as the key determinant of the cellular redox state. GSH levels, redox status, and biosynthesis can regulate the expression of a large number of genes including the antioxidant defense system and the pathogenesis-related 1 (PR1) gene (Ball et al. 2004). Taking it all together, all cellular compartments are well equipped with antioxidant enzymes and antioxidants. However, when this local antioxidant capacity cannot cope with ROS production, H_2O_2 can leak into cytosol and diffuse to other compartments, for example, vacuoles (Bienert et al. 2006). Vacuoles are enriched in flavonoids, AsA, and GSH, whereas the peroxidases are localized at the tonoplast inner surface. The main scavenging enzymes are described below.

1. Catalase was the first antioxidant enzyme found in peroxisomes $(2H_2O_2 \rightarrow 2H_2O + O_2)$. It degrades H_2O_2 without any reducing power and it is active only at relatively high H_2O_2 concentrations. Thus, lower H₂O₂ levels are eliminated by APX and other peroxidases with the aid of various reductants like AsA and GSH. CAT is of enormous importance for regulating H₂O₂ homeostasis, as it can function as a cellular sink for H_2O_2 . CAT is tightly regulated at transcriptional and translational levels and in plants CAT is encoded by a multigene family. Therefore, two genes in cotton, three in maize, and five in pea were described (Gechev et al. 2006; Corpas et al. 2009). In Arabidopsis, CAT is encoded by a multigene family consisting of three genes (CAT1, CAT2, CAT3) encoding individual subunits that associate to form at least six isoenzymes. Furthermore, peroxisomal CAT2 and CAT3 show circadian regulation (Du et al. 2008). On the other hand, a genetic system that controls H_2O_2 levels with the involvement of CAT has been identified and described in Arabidopsis (Mateo et al. 2004, 2006). A network of at least 152 genes is involved in managing the level of H₂O₂ (Mittler et al. 2004). Transcriptome analysis of CAT-deficient plants demonstrated that perturbation of the H₂O₂-scavenging capacity significantly affects nuclear gene expression (Queval et al. 2007). Likewise, in response to H_2O_2 , 349 transcripts were significantly induced (upregulated) and 88 were repressed (downregulated) in the above CAT-deficient plants (Vanderauwera et al. 2005). A detailed description on photorespiratory H₂O₂-dependent gene expression is reported by Vanderauwera et al. (2009).

- 2. Ascorbate peroxidase catalyzes in distinct cell compartments the electron transfer from AsA to H_2O_2 to render dehydroascorbate and H_2O_2 . APX is the main enzyme responsible for H_2O_2 removal in the chloroplast, peroxisome, and mitochondria. Expression of genes for the cytosolic APX1 and APX2 is controlled, at least in part, at the chloroplast level. The activity of APXs is thought to form a second barrier of defense against H₂O₂/ROS produced in the chloroplasts (Chang et al. 2004). Recently, it has been reported that H₂O₂-mediated inactivation of two chloroplastic peroxidases, APX and 2-cys peroxiredoxin, and it is discussed how the chloroplast become susceptible to H₂O₂ damage (Kiitajima 2008; Tripathi et al. 2009). It has also been described that the overexpression of APX in barley peroxisomal membrane increases the tolerance to heat stress, whereas the Arabidopsis peroxisomal APX is essential for plant growth and development (Narendra et al. 2006).
- 3. *Glutathione reductase (GR)* catalyzes the reduction of oxidized glutathione (GSSG) to glutathione (GSH). This ubiquitous enzyme is a component of the AsA–GSH cycle (AGC)

that controls the cellular level of H₂O₂ (Foyer and Noctor 2005). The peroxisomal isoform of GR has been characterized, but the gene encoding it has not yet been identified (Romero-Puertas et al. 2006). Other component of AGC is the monodehydroascorbate reductase (MDAR) which catalyzes the reaction: monodehydroAsA + NAD(P)H \rightarrow AsA + NAD(P)⁺. In Arabidopsis it has been shown that the membrane-bound MDAR4, component of the AsA-dependent electron transfer system, is necessary to detoxify H_2O_2 (Eastmond 2007). In relation to GR, the GSH peroxidase (GPX) uses GSH to reduce H₂O₂, thereby protecting cells against oxidative damage. In plants, GPX exist in the cytosol to reduce H_2O_2 to H_2O . However, its ability to scavenge H₂O₂ decrease due to its Cys residue without selenium. Therefore, the major functions of GPX are lignin biosynthesis, resistance to pathogens, and degradation of indole-3-acetic acid (IAA).

5 H₂O₂ Signaling

Oxidants utilized in cellular signaling must exhibit specificity for their target substrate. Furthermore, the given redox-based modification must be reversible to ensure signal transduction and is transient rather than constitutive. H_2O_2 , O₂⁻, and NO execute Cys-based modifications that are substrates-specific and generally reversible (D'Autréaux and Toledano 2007). The relative stability, rapid turnover, and higher concentrations of H₂O₂ in plant cells could point to the fact that H₂O₂ plays a key role as a signal transduction factor. Nowadays, there are many reports indicating that H₂O₂ is a cellular component at physiological conditions. However, signal molecules are usually present in very low concentrations. Therefore, it is thought that H_2O_2 is not only a signaling molecule, but also plays a key role in primary plant metabolism. Although H_2O_2 is a signal molecule affecting transcriptome, it is not clear whether H_2O_2 is actually the signal per se, or whether oxidation of other molecules by H_2O_2 is needed to generate an intracellular

signal. In order to H₂O₂ as a specific signaling molecule, its cellular alterations must be tightly perceived. To start the H₂O₂ signaling process it is necessary that a concrete protein specifically recognize H₂O₂, or alternatively a direct chemical interaction exist to propagate the H_2O_2 signal. The first possibility is unlikely due to the small size of H_2O_2 . One likely mechanism for cells to perceive H₂O₂ was proposed by Hancock et al. (2006). In this sense, mechanism should involve H_2O_2 -dependent protein modifications. H_2O_2 can directly interact with Cys residues within proteins. This modification of –SH groups is the key to H₂O₂ perception since it could alter protein conformation, affecting their activity and therefore initiating subsequent cellular responses (Foyer and Noctor 2005; Hancock et al. 2006). If two -SH groups are present in a specific protein, the formation of a S–S bridge could be a possible mechanism for protein conformational changes involving functional alterations. Alternatively, single -SH groups can be oxidized and reversal oxidation carried out under reducing conditions, for example, in the presence of GSH or thioredoxin (TRXH). Many cellular and extracellular proteins are potential targets of such oxidative modification.

5.1 H₂O₂ Signal Perception

The first genes involved in ROS perception and signal transduction have been identified (Lee et al. 2007; Vanderauwera et al. 2009). However, it still remains to be identified a cell surface protein that acts as H_2O_2 sensor. In contrast, two possible H_2O_2 sensor candidates exist in the cytosol (e.g., glycer-aldehide-3-P-dehydrogenase) and the endoplasmic reticulum (ethylene receptor), respectively. Miller and Mittler (2006) have argued that heat-shock transcription factors are the molecular sensors of ROS and their complexity, flexibility, and specialty allow them to control the expression of a wide range of stress-response genes, not only those involving in heat shock. But confirmation of this possibility has not been carried out.

1. *Glyceraldehide-3-P-dehydrogenase (GAPDH)*: using the 5-iodoacetamide fluorescein (IAF)

methodology is possible to mark a number of cytosolic proteins. Pretreatment with H_2O_2 severely reduced the IAF binding to one particular protein (Mw≈40 kDa) which corresponded to the glycolytic enzyme GAPDH (Hancock et al. 2005). GADPH inhibition with low concentrations of H_2O_2 (<250 µM) can be restored with GSH or dithiothreitol (DTT), suggesting that inhibition is indeed reversible. GADPH has also been inhibited by NO, and this inhibition could be reversed with DTT (Lindermayr et al. 2005). At present it is unknown whether NO and H_2O_2 are targeting the same -SH groups in GADPH. On the other hand, GAPDH has been shown in other different systems to plant being S-thiolated by the addition of GSH to the thiol side group. It is unknown whether this modification also occurs in plants.

2. Ethylene receptor (ETR1): in yeast, histidine kinases (HKs) of two-component signaling systems have been reported to function as sensors of oxidative stress. In Arabidopsis, the HK of the ETR1 appears to be essential for H_2O_2 perception during stomatal movement. Thus, mutation of a Cys residue in the N-terminal region of ETR1 disrupted the perception, indicating that this thiol group is important for the H₂O₂ signaling. As the ETR1 kinase domain is not required for H₂O₂ perception, the H₂O₂-provoked signaling through ETR1 was seemingly unrelated to its welldescribed role as an Et receptor. This fate suggests that the signaling through ETR1 provoked by H_2O_2 is different from Et induced. The S. cerevisiae mutant TM219 has not SLN1-SSK1, a functional two-component system and therefore it has enhanced susceptibility to H₂O₂ and shows a great growth inhibition. The transformation of SLN1-SSK1 with AtETR1 restored the yeast tolerance to H_2O_2 (Desikan et al. 2005). Likewise, etr1-1 which contains the Cys65Tyr mutation, has reduced stomatal closure in response to H₂O₂, suggesting that the -SH group of Cys65 is important for H_2O_2 signaling. On the other hand, the HK AtHK5 plays a crucial role in mediating H_2O_2 dependent processes in stomatal cells that are induced by environmental and hormonal signals (Desikan et al. 2008). It appears that modification of -SH groups belonging to ETR1 may be the key for the H_2O_2 signaling.

ETR1 may be the key for the H_2O_2 signaling. However, it is not currently known whether this modification is a direct effect of H_2O_2 or it is conducted by another H_2O_2 -sensing protein.

5.2 Transduction of H₂O₂ Signal

 H_2O_2 acts as a second messenger, mediating the acquisition of tolerance to biotic and abiotic stresses. One of the earliest events that follow elevation levels of H₂O₂ is alteration of K⁺ and Ca²⁺ ion fluxes. The transient Ca²⁺ oscillations are stress-specific and can lead to various downstream effects through the numerous Ca²⁺interacting proteins (e.g., calmodulins, CaM) and Ca²⁺-dependent protein kinases (PKs) or/and over-amplification of the H₂O₂ signal. Changes in plasmalemma permeability leading to Ca2+ and H⁺ influx appear to be necessary and sufficient to induce H₂O₂ production. Exogenous H₂O₂ could activate Ca2+ channels to concomitantly elevate the cytosolic $Ca^{+2} (Ca_{cvt}^{2+})$ content. The activation of these channels by endogenous H₂O₂ production could be an autocatalytic process in which NADPH-oxidase is also involved. This fact is a positive regulation of H₂O₂ level. The Ca²⁺/CaM complex has been supposed to increase H₂O₂ production through Ca2+/CaM-dependent NADkinase which affects the concentration of available NADPH during activation of NADPH-oxidase. On the other hand, it can also effect a negative regulation of H₂O₂ level in plants. That is, the Ca_{cvt}²⁺ elevation activates the CaM and subsequently the signal is transmitted to a downstream target protein (i.e., CAT). These findings suggest that Ca2+/CaM complex and CAT activation downregulates the H_2O_2 levels. Likewise, there are evidences indicating that Ca²⁺ has dual functions in the regulation of H_2O_2 homeostasis, influencing the redox signaling in response to environmental signals (Yang and Poovaiah 2002; Hung et al. 2005). CaM activated some plant CAT in the presence of Ca²⁺, thus, AtCAT3 can bind CaM in a Ca²⁺-dependent way (Du et al. 2008).

Recently, it was demonstrated that intraperoxisomal Ca²⁺ rise increases CAT activity, which in turn increases peroxisomal H₂O₂ scavenging efficiency (Costa et al. 2010). On the other hand, H_2O_2 increases when the plants are under deficit of K⁺. Therefore, H₂O₂ might play a role in cellular signaling of K⁺ deprivation (Shin and Schachtman 2004). During stomatal closure, in addition to regulate Ca²⁺ channels, H₂O₂ also inhibits K⁺ channel activity and induces cytosolic alkalinization in guard-cells. The overall effect of

exogenous H_2O_2 on the cellular redox state may mean that a multitude of proteins can be involved (Foyer and Noctor 2005). A variety of forward and reverse genetics studies revealed a number of components in the H2O2 signaling network, including PKs, protein phosphatases (PPs), and ROS-responsive transcription factors (Cheng and Song 2006). The activity of PKs has been a particular focus. Thus, studies of PKs have shown that the mitogen-activated protein kinases (MAPK) cascade in plants are activated by H₂O₂, in particular the Arabidopsis AtMPK3 and AtMPK6, thereby regulating the cellular redox state (Hancock et al. 2006). Whether H_2O_2 has a direct effect on MAPKs or activates upstream effectors is unclear. Interestingly, the stabilization of MKK4 and MKK5-MAPKKs by H₂O₂ was demonstrated and the authors hypothesized that H_2O_2 might have a general stabilizing effect on MAPKKs (Dòczi et al. 2007). In addition, the ANP1-MAPKKK and the Ser/Thr-PK OXI1 (oxidative signal-inducible), have also been shown to be induced specifically by H_2O_2 , being important for H_2O_2 sensing and the activation of MAPK cascade in Arabidopsis (Rentel et al. 2004). ANP1 activates two downstream MAPKs, AtMPK3 and AtMPK6 and eventually regulates gene expression of specific H₂O₂-inducible transcripts. OXI1 activity is required for full activation of AtMPK3 and AtMPK6 (Rentel et al. 2004). Another H_2O_2 -inducible PK is OMTK1 (oxidative stress-activated MAPK triple kinase), isolated from alfalfa. In contrast to OXI1, OMTK1 is H₂O₂-specific and not activated by phytohormones (Nakagami et al. 2004, 2005). In contrast, OMTK1 activated the downstream MAPK-

MMK3 which can be activated by ethylene (Et).

Summarizing, all existing data suggest that both environmental and cellular stress induce the generation of cytosolic H₂O₂, which in turn induces a MAPKs cascade that subsequently stimulates expression of antioxidants genes, reducing H₂O₂ levels and restoring cellular homeostasis. Besides everything described, PPs can be involved in H₂O₂ signaling. However, H₂O₂ inhibits PPs activities, probably by the direct oxidation of Cys in the active site of these enzymes. The Arabidopsis PP2C enzymes ABI1 and ABI2, negative regulators of ABA signaling, and the Tyr-PP AtPTP1 have been suggested to play a role (Gupta and Luan 2003). ABI1 and ABI2 could be receptors for the H_2O_2 signal in higher plans (Hung et al. 2005). Interestingly, AtPTP1 regulates the activity of MAPKs, suggesting a tight relationship among H₂O₂, PKs, and PPs. All these described effects provoked by H_2O_2 in plants raise the question: if H_2O_2 is involved in so many diverse responses, how can it be specific?

6 Expression of Transcription Factor Genes H₂O₂-Dependent

The control of gene expression is one of the most important changes induced by H₂O₂. In plants, the first report on a genome-wide expression analysis was provided by Desikan et al. (2001). By using a cDNA microarray from Arabidopsis, at least 113 and 62 transcripts were induced and repressed, respectively, in the presence of H₂O₂. The predicted function of all these genes was: cell signaling, transcription, cell death, and defense response. Among these genes are transcription factors (TFs) due to their capacity of activating the expression of downstream target genes. Several redox-controlled TFs have also been identified. In Arabidopsis, a TF called CBF1 (C-repeat binding factor) belonging to the APETALA/EREBP-family was described. CBF seems to be able to indirectly regulate the redox status of cytosol by activating downstream genes that encode antioxidants enzymes (e.g., CAT). Nevertheless, there is still no evidence that join the H_2O_2 signaling with CBF accumulation or activation. However, the overexpression in tomato



Fig. 16.5 The OxyR transcription factor regulates the $H_2O_2 E$. *coli* response. H_2O_2 oxidizes OxyR Cys199 to an R-SOH that reacts with Cys208 in an intramolecular disulphide bond. The resulting allosteric change activates OxyR DNA binding. OxyR is reduced by glutaredoxin-1 (Grx)

of AtCBF1 induced a high tolerance for water deficit stress, CAT activity increased and H₂O₂ concentration decreased comparing with wild type (wt) plants (Hsieh et al. 2002). The activity of TFs can be directly regulated by oxidative modifications as exemplified by the S. cerevisiae oxidative stress TF Yap1 (Herrero et al. 2008). However, oxidation of Yap1 is not directly by H_2O_2 , but mediated by GSH peroxidase like Gpx3 enzyme. In this process, Cys36 of Gpx3 forms an intermolecular disulphide with Cys598 of Yap1 (a TF that activates the expression of genes for antioxidants), forming a disulphide which is rearranged forming a disulphide in Yap1 itself and activating it (Hancock et al. 2006). Yap1 is a functional homologue of OxyR which is an E. coli (Fig. 16.5) transcriptional regulator with six Cys residues, being their Cys199 pivotal in the response of OxyR to redox cues. A Cys-rich domain of Yap1, that influenced in the nuclear localization of Yap1 when it was activated, it was also necessary for H₂O₂ tolerance. Therefore, alteration of key Cys residues of Yap1 provides a direct and powerful means to influence nuclear gene expression (Delaunay et al. 2002). In summary, Gpx3 acts as a protein that perceives H₂O₂ and transduces their signal to Yap1. On the other hand, the H_2O_2 gene network eventually transmits the signal to ROS-specific TFs. Some of these are LSD1 and LOL1 (negative and positive regulators of ROS-induced cell death), the senescencespecific WRKY53, and the ROS-inducible WRKY75 and some heat-shock TFs. Two other zinc finger TFs, Zat11 and Zat12, are also induced

by H_2O_2 (Gadjev et al. 2008). More alterations in TFs caused by H_2O_2 will be described later in the stomatal closure section (Sect. 7.1).

7 H₂O₂-Mediated Physiological Processes

7.1 H₂O₂-Signaling in Stomatal Movements

The exposure of guard-cells to H₂O₂ induced elevations of (Ca_{cvt}^{2+}) and stomatal closure. There is considerable interest in the study of the relationship of ABA/H_2O_2 . The discovery that ABAinduces H₂O₂ production in guard-cells was a significant finding. It is well established that H₂O₂ production is required to initiate the ABAinduced stomatal closure and that chloroplasts are the principal sources of H₂O₂ in guard-cells (Wang and Song 2008). It has been clearly demonstrated that: (1) stomatal closure induced by ABA is a different process that inhibition of stomatal opening induced by ABA; and (2) H_2O_2 is involved in inhibition of stomatal opening induced by ABA (Yang et al. 2007). Moreover, the mediation of H₂O₂ in ABA-induced stomatal closure by targeting inward K⁺ channels in plasmalemma was also suggested. Experiments with abi1, abi2, and abi2-1 suggest that the last double mutation impairs ABA signaling downstream of H₂O₂ production (Wang and Song 2008). An overlap exists between ABA- and H₂O₂-induced transcription genes, and both signaling molecules regulate many dowstream genes in a coordinated manner (Yang et al. 2007). Several TFs that are redox-controlled have also been identified during guard-cell signaling for H₂O₂ and ABA. Thus, the beforehand described Yap1 (Delaunay et al. 2002), AtMYB60, and AtMYB61 have been described to be specifically expressed in guardcells. However, the relationship between AtMYB60 and AtMYB61 with H₂O₂ signaling is under study (Liang et al. 2005). A cross-talk between ABA and H₂O₂-signaling cascades seem to be clear. On the other hand, various common components of ABA and H₂O₂-signaling cascades (e.g., cytosolic-free Ca2+ levels, MAPKs,

and NADPH-oxidase) are involved during guardcell movements. Two CDPK (i.e., CPK3 and CPK6) genes expressed in Arabidopsis guardcells have remarkable functions in the regulation of guard-cells ion channels, and provide genetic evidence for the existence of Ca²⁺ sensors that transduce stomatal ABA signaling (Mori et al. 2006). The results of Chen et al. (2004) strongly suggest that extracellular CaM activates a signaling pathway which involves the activation of G-protein, H₂O₂ production, and changes in (Ca_{cvt}^{2+}) , thus regulating the stomatal movements. Likewise, H₂O₂ and ABA can activate the same MAPK which mediates both ABA- and H₂O₂induced stomatal closure (Wang and Song 2008). Time-course analysis of both MAPK activation and ROS production indicated that the increase of H_2O_2 preceded the MAPK activation and subsequently takes place the activation of the ABAsignaling pathway. That is, the rise of H₂O₂ induced by ABA activates MAPK, which in turn leads to the upregulation of antioxidant defense systems. Recently, the MAPK-MPK3 was shown to be essential in Arabidopsis for ABA and H₂O₂inhibition of stomatal opening (Gudesblat et al. 2007). On the other hand, the PK-OST1 also regulates the production of H₂O₂ in guard-cells by means of signaling pathways requiring AtrbohD and AtrbohF (Kwak et al. 2003). As previously described in Sect. 5.1, one member of the Arabidopsis histidine kinase (HK) family (De la Torre et al. 2006) has a dual function in guardcells: (1) it is a possible target for H_2O_2 during stomatal closure, ETR1-Cys65 being essential for this function (Desikan et al. 2005, 2008); and (2) it perceives the hormone Et. Therefore, Et induces stomatal closure that is dependent on H_2O_2 production in guard-cells, generated by NADPH-oxidase AtrbohF. Moreover, Et and H_2O_2 signaling in guard-cells are mediated by ETR1 via EIN2 and ARR2-dependent pathways(s) (Desikan et al. 2006). Recently, AHK5, another member of the Arabidopsis HK family, was studied (Desikan et al. 2008). In addition to the predicted cytoplasmic localization of AHK5, it also appears to co-localize to the plasmalemma. Mutants lacking AHK5 show a reduced stomatal closure in response to H_2O_2 . The AHK5

over-expression results in a constitutive lower stomatal closure than wt plants. Interestingly, treatment with flagellin (flg 22) also exhibited reduced stomatal closure and H₂O₂ production in ahk5 mutants (Desikan et al. 2008). Finally, the two Arabidopsis NADPH-oxidase catalytic subunit genes AtrbohD and AtbohF function in the signal transduction of ABA in guard-cells (Kwak et al. 2003). In AtrbohD/F, the exogenous H_2O_2 restored the ability of the Ca2+ channel activation and stomatal closure in response to ABA stimulation. The rboh transposon-insertion mutants of Arabidopsis have been used to provide unequivocal evidence that NADPH-oxidase-mediated H₂O₂ production is required for ABA-induced stomatal closure (Kwak et al. 2003). Likewise, treatment with ABA increased the expression of AtrbohD in guard-cells. The activation of MAPKs can amplify H₂O₂ signals directly regulating NADPH-oxidase activity, or activating also TFs to increase the expression of NADPH-oxidase genes in the ROS signal transduction (Mittler et al. 2004). ABA can also enhance the gene expression (Kwak et al. 2003) and NADPHoxidase activity. Therefore, the MAPK activated by ABA might increase H₂O₂ signals via NADPHoxidase activity.

7.2 H₂O₂-Signaling in Germinating Seeds

Likely, the seeds have high levels of stress and accumulate free radicals in their desiccated state and show high levels of CAT and SOD (Bailly et al. 2008). This detoxification occurs during dormancy, after-ripening (AR) (Iglesias-Fernández et al. 2011), and germination to preirreversible alterations vent in proteins, membranes, and DNA (Kibinza et al. 2006). The H_2O_2 production during seed storage in the dry state was initially documented by Pukacka and Ratajezak (2006). More broadly, $O_2^{\bullet-}$ and other ROS appear to play roles during seed germination and dormancy (Oracz et al. 2007; Müller et al. 2009). In a recent update, it was proposed that ROS play a key role in the completion of germination and that they should be considered as messenger or transmitters of environmental cues during seed germination (Bailly et al. 2008). Thus, rice germination is inhibited by inhibiting NADPH-oxidase, and exogenous H₂O₂ stimulates the germination of dormant seeds belonging to several species. Likewise, exogenous H₂O₂ results in a decrease of ABA in dormant seeds and alleviation of dormancy by HCN induces an increase in H₂O₂ level and a decrease in ABA content (Bailly et al. 2008). On the other hand, Oracz et al. (2007) show that AR is associated with accumulation of H₂O₂ in the embryonic axes of sunflower seeds. In summary, H₂O₂ acts directly, or as a messenger of hormonal networks, signaling molecules involved in the transition from dormant to non-dormant state. But the H_2O_2 mechanism at the cellular level during the regulation of dormancy is far from being resolved. Data recently obtained in Arabidopsis seem to indicate that H₂O₂ could regulate seed dormancy by triggering ABA catabolism and GA synthesis (Liu et al. 2010). Thus, exogenous H₂O₂ decreased seed dormancy and rose transcription of genes involved in ABA conjugation (ZYP707A) and GAs synthesis (GA3ox and GA20ox), respectively. These data also support the possibility that H₂O₂ downregulates genes responsible of GA biosynthesis and ABA catabolism.

On the other hand, ROS can alter the function of seed-specific proteins through its carbonylation (i.e., introduction of CO into a molecule), thereby relieving dormancy (El-Maarouf-Bouteau and Bailly 2008; Oracz et al. 2007). The carbonylation is the most commonly occurring oxidative protein modification. Protein carbonylation inhibits or alters protein activities and intensifies their susceptibility to proteolytic attack. There are no indications that carbonylation is reversible. The identification of the factors causing the carbonylation is very complex. Several main possibilities include: (1) a decrease in the antioxidant defense system, (2) an increase in the ROS production, (3) a decreased capacity to remove oxidized proteins, or (4) an augment in the proteic sensitivity to oxidative attack. These four possibilities are not mutually exclusives. In sunflower seeds, ROS accumulation

leads to targeted protein carbonylation not only during dry AR but also during artificial breaking of dormancy by cyanide application (HCN) (compound that triggers ROS accumulation because it is an inhibitor of SOD, CAT, and carbonylation) or methylviologen (ROS-generating compound) (Oracz et al. 2007; El-Maarouf-Bouteau and Bailly 2008). It was hypothesized that the action of HCN in sunflower seed dormancy alleviation does not involve Et production (Oracz et al. 2008). However, the hypothesis that HCN interacts with ROS-producing pathways have been supported by data on intracellular ROS production in response to HCN treatment (Oracz et al. 2009).

It has been proposed that NADPH-oxidases are involved in H₂O₂ production in seed germination (Liu et al. 2007) and dormancy breaking (Oracz et al. 2009). Recently, Müller et al. (2009) have demonstrated that dry AR in Arabidopsis is impaired in the NADPH-oxidase (AtrbohB) mutant which also shows reduced protein oxidation. The ABA sensitivity was not affected in dry AR AtrbohB mutant seeds. It mentions that: (1) AtrbohB gene is expressed in the embryo but not in the endosperm, (2) AtrbohB pre-mRNA is spliced in seeds depending on ABA and the seed dry AR status, and (3) AtrbohD gene is more abundant in the endosperm than in the embryo of ABA-treated seeds. On the other hand, Oracz et al. (2009) have demonstrated that in sunflower the expression of genes related to ROS production (i.e., NADPHox, NADPH-oxidase, and POX, peroxidase) is differentially affected during dry AR or by HCN treatment, and the effect of HCN is likely to be mediated by ROS.

Bailly et al. (2008) propose a model to account for the dual role of ROS in seed physiology. Seed germination is only possible when the ROS level is in an oxidative threshold. Below this level (i.e., dormancy), the amount of ROS during imbibition is too low for allowing germination. Alleviation of dormancy (i.e., AR) leads to an increase of the cellular level of ROS during seed imbibition thus ensuring germination completion, owing to the ROS signaling role. Above this threshold, ROS content is too high, because seeds are aged or placed in inappropriate environmental conditions during their imbibition, and ROS become deleterious and cause cellular oxidative damages that prevent delay of germination.

7.3 Role of H₂O₂ in Plant-Microorganism Interaction

Plants have developed well-organized defense mechanisms against a wide range of pathogens. Therefore, plants can react actively to attack of pathogenic microorganisms with an array of inducible responses that lead to local and systemic expression of a broad spectrum of antimicrobial defenses. These defensive systems can be triggered by recognition, via cell surface receptors, of pathogens- or microbial-associated molecular patterns (PAMPs or MAMPs), such as flagellin, lipopolysaccharides, peptidoglycanes, or siderophores from bacteria or glucan and chitosan from fungi (Bolwell and Daudi 2009; Jones and Dangl 2006). Since Doke reported the production of $O_2^{\bullet-}$ during incompatible interactions between potato and the pathogen Phytophthora infestans (Doke 1983), the generation of ROS, especially H₂O₂, have been repeatedly associated with successful resistance responses (Torres 2010). This transient production of ROS is termed "oxidative burst" (Shetty et al. 2008) and it has been also observed through the direct application of purified siderophores (De Vleesschauwer et al. 2008), mycotoxis (Desmond et al. 2008), or cyclic lipopeptides (Jourdan et al. 2009) to the plant or plant cells inducing a timely and highly localized H₂O₂ production.

Although three phases have been observed in specific interactions (Shetty et al. 2008) and H_2O_2 production occurs in a biphasic manner during incompatible interactions (Apel and Hirt 2004). First, an unspecific and transitory phase usually takes place within minutes after the interaction followed by a second and prolongated production of H_2O_2 that occurs between 3 and 6 h after pathogen attack (Lamb and Dixon 1997). This second phase is usually associated with the establishment of the defenses and the hypersensitive response (HR), further characterized by a rapid cell death at the site of infection to restrict

pathogen growth and which is mainly effective against biotrophic pathogens (Apel and Hirt 2004; Shetty et al. 2008). In contrast, during compatible interactions only the first transient phase is induced (Jourdan et al. 2009). During these responses, H_2O_2 is produced by enhanced apoplastic CW-peroxidases or plasmalemma NADPH-oxidases (Almagro et al. 2009; Bolwell and Daudi 2009). Additionally, H_2O_2 can be directly generated by diamine and oxalate oxidases.

Several functions are attributed to H₂O₂ during plant-pathogen interactions (Shetty et al. 2008; Torres 2010). H_2O_2 could kill directly the pathogen functioning as an antimicrobial agent. However, this toxicity depends on the sensitivity of the pathogen to the concentration of H_2O_2 present because many pathogens can grow in high concentrations of H₂O₂ (mM) or possess effective detoxification mechanisms (Aguirre et al. 2005; Matilla et al. 2007; Molina and Kahmann 2007; Shetty et al. 2008). Interestingly, H₂O₂ production has been detected during the rhizobium-legume interaction, suggesting that symbiotic bacteria are initially recognized as pathogens by the host plant (Chang et al. 2009). Additionally, H₂O₂ production has been reported to benefit infection by necrotrophic pathogens because these usually kill host cells and extract nutrients from them (Shetty et al. 2008). Alternatively, H₂O₂ production has been associated with establishment of physical defensive barriers by reinforcing the CW to prevent plant invasion by pathogens, through cross-linking of CW structural proteins and by deposition of callose-rich papillae, lignin, and suberin in the sites (Almagro et penetration al. 2009;Hückelhoven 2007; Shetty et al. 2008). However, experimental data also suggest that H_2O_2 could have a signaling role mediating changes in gene expression (Shetty et al. 2008; Torres 2010). The fast production of H₂O₂ in response to pathogen attack and its capacity to diffuse across membranes suggest that it can exert this function directly through redox control of TFs or indirectly by interacting with other signaling components like phosphorylation cascades (Torres 2010). Furthermore, H_2O_2 and other ROS, in association with salicylic acid (SA), have been related with the establishment of SAR (Conrath 2006; Vlot et al. 2009), an inducible plant defense response triggered by necrotizing microbes and SA-dependent (Conrath 2006). Redox regulation has been reported for the nonexpressor of pathogenesis-related 1 (NPR1) and the leucine zipper TF TGA1 that are important mediators of systemic acquired resistance (SAR) during plantpathogen interactions. Reduction of key Cys residues in these proteins relocates NRP1 to the nucleus and modulates the DNA-binding activity of the NPR1/TGA1 protein complex, thereby affecting decisively downstream gene expression (Mou et al. 2003). Interestingly, the oxi1-null mutant has abnormal root hair growth and enhanced susceptibility to pathogen infection, two processess mediated by H₂O₂ (Rentel et al. 2004). Collectively, the modifications of ROS levels in the host might be a strategy that is extended among the pathogens to increase host susceptibility.

Following with the study on pathogenesis and oxidative stress, nitric oxide (NO) is often produced at the same time and in the same locations in plants as H₂O₂ and it is also involved in a plethora of responses and functions (Neill et al. 2003). S-nitrosylation, the covalent attachment of an NO group to a reactive Cys thiol to form an S-nitrosothiol (SNO), has emerged as a prototypic redox-based posttranslational modification (Spadaro et al. 2010). NO seems to collaborate with H₂O₂ in plant disease resistance. During plant-pathogen interactions: (1) NO can act as an antioxidant, scavenging excess H₂O₂, ending radical-mediated lipid peroxidation and inhibiting H_2O_2 signaling pathways, which leads to cell death; and (2) NO can act synergistically with H₂O₂ to induce SAR (Quan et al. 2008). NO may also react with thiol groups on proteins to yield a -S-NO group. Many of the proteins modified in this way by NO are also potentially modified by H_2O_2 (Lindermayr et al. 2005).

7.4 Plant Growth H₂O₂-Mediated

The root hair growth is one of the most important studied effects of H_2O_2 as developmental regulator. The initiation of root hair growth process is known to be a physiologically and genetically separate program from tip growth, suggesting that the lesion in rhd2 (root hair defective 2) mutant is selectively affecting the tip growth machinery. In the *rhd2* background, root hairs are able to initiate their developmental program. When various alleles of the *rhd2* mutation were cloned, they were found to reside in a gene encoding AtrohC of Arabidopsis, a homolog of the gp91phox domain of the neutrophyll respiratory burst NADPH-oxidase. At difference of mammals, plant Rbohs are not glycosilated. Rbohs NADPH-oxidase proteins are localized into the plasmalemma, where they oxidize cytosolic NADPH, transferring an electron across the membrane to generate apoplastic $O_2^{\bullet-}$. This $O_2^{\bullet-}$ dismutates to H_2O_2 which is thought to diffuse back into the cell, providing a possible cytosolic regulator (Swanson and Gilroy 2010). The gene family has ten known members in Arabidopsis; AtrbohA-J and its homologues have also been found in other angiospermic species. Each member of Rboh family differ in their expression pattern across plant and have been shown to be involved in a range of processes as root elongation and root hair development (Foreman et al. 2003; Swanson and Gilroy 2010). Thus, AtrbohC is present in root epidermal cells of the elongation zone but its expression becomes limited to those capable of making root hairs. In the *rhd2* mutant lacking AtrbohC, the hair roots fail to extend. Further evidence for the role of H₂O₂ in hair roots growth came from the studies of the *atrbohC* mutant which has low ROS levels in root hairs and is defective in activation of Ca2+ channels required for formation of Ca2+ gradient necessary for root hair growth (Foreman et al. 2003). According to these findings, the atoxi1-null mutant has reduced root hair growth (Rentel et al. 2004). AtOXI1 has been implicated in root hair development and AtOXI1 kinase is induced by H_2O_2 . Oxil null mutants are impaired in the activation of MAPKs, MPK3 and MPK6, upon oxidative stress, suggesting that OXI1 functions downstream of ROS but upstream of the MAPK module.

7.5 H₂O₂ in PCD

Cell death is essential for plant growth and development and responses to the environment. PCD can be initiated by all types of ROS. H₂O₂ is the key factor in PCD (Vanderauwera et al. 2009). However, the mechanisms by which H_2O_2 induces PCD are not yet established, although several studies point to effects on mitochondrial function. Hence, exogenous H₂O₂ increased subsequent H_2O_2 production in this organelle, altering mitochondrial function and inducing PCD. On the other hand, if MAPKs are long time activated, H_2O_2 production will increase and PCD will be induced. It is already known that H₂O₂ activates MAPK pathways, and it is tempting to speculate that increased H_2O_2 can stimulate further mitochondrial H₂O₂ production, perhaps via MAPK activation, in an amplifying oxidative death cycle. During endospermic seed germination, α -amylase GA-induced degredation of carbohydrates from aleurone layer and the PCD appears. This cell death is dependent on glyoxysomal production of H_2O_2 and there are evidences that CAT, APX, and SOD are downregulated by GA₃ to ensure sufficient accumulation of H₂O₂ prior to onset of cell death (Gechev et al. 2006; Gechev and Hille 2005). The application of H_2O_2 have confirmed the role of H_2O_2 as a triggering of cell death and also showed that high concentrations of H_2O_2 can cause necrosis instead of PCD. In agreement with these observations, overexpression of APX can suppress the cell death induced by H_2O_2 or NO (Murgia et al. 2004). In addition, CAT deficiency leads to the increase of H₂O₂ levels and the triggering of PCD (Vandenabeele et al. 2004; Gechev et al. 2006). Following pathogen infection, transient H₂O₂ overproduction and accumulation can promote a local defense response connected with NADPH-oxidase activation. The increase of H₂O₂ levels causes the HR as well, leading to rapid localized cell death at infection sites or can induce SAR (Slesak et al. 2007). Although, in many plant-pathogen interactions PCD is a welcome event for the plant host, there are examples of pathogen-triggered cell death that are mortal for the plant. On the other hand, ROS can be involved in allelopatic plant-plant interactions. This is the

case for *Centaurea maculosa* whose roots secrete the phytotoxin catechin that triggers ROS accumulation in root meristems of neighboring species and subsequent Ca^{2+} -dependent cell death (Gechev et al. 2006; Gechev and Hille 2005).

8 Conclusion and Future Perspective

As can be inferred from this review, a wealth of information about H₂O₂ was generated in the last ten years. The strategy used to study the mechanism and mode of action of H₂O₂ was similar to the one for all the signaling molecules known in plants. The H₂O₂ concentration in different subcellular compartments is not yet known and H₂O₂ accumulation is maintained at a very low level due to the existence of an antioxidant system that eliminates excess of H2O2 production. Sensitive intracellular imaging will be required to visualize H_2O_2 in cells. The NADPH-oxidase (*rboh* gene family) was reported to be the pivotal enzyme involved in apoplastic H₂O₂ production. H_2O_2 , as a signaling molecule, must possess some cellular metabolic molecule(s) that recognize(s) it and form(s) either a complex or a derived product that triggers the transduction cascade. However, data on the existence of H_2O_2 receptor are still unclear. The interaction of H_2O_2 with proteins having Cys-groups seems beyond doubt. The discovery of specific biological system with high sensitivity and rapid response to H_2O_2 is key to all above investigations. Tipgrowing cells and stomatal cells-guard movements have proven to be powerful tools in elucidation of physiological process under H₂O₂. It is very interesting to study in detail these biological systems. Biochemical evidences indicated that MAPK cascades are responsible for transmitting the H_2O_2 signal. The recent identification in Arabidopsis of the Ser/Thre kinase OXI1, as an essential component of the H_2O_2 signaling, provided new insights into the H₂O₂transmiting kinase network. In addition to MAPK cascade, the H₂O₂ signal can also be transmitted through changes in Ca²⁺ ion fluxes and cellular redox state. In addition, H₂O₂ seems to work

together with other molecules (e.g., fitohormones and NO) to control a variety of processes in plants. The use of transgenic plants with non- H_2O_2 , hormones, and NO products, together with the isolation of H_2O_2 -signaling mutants will be helpful in elucidating the biological roles of H_2O_2 in specific cells and in response to various stimuli. Finally, we must not forget different -omics approaches to facilitate further insights into H_2O_2 cellular response and to provide more clues to how plants sense and respond to environmental stress conditions.

Acknowledgments We would like to thank the Ministerio de Ciencia e Innovación (MICINN) (grant nº CGL2009-11425) for funding this research. We also thank Dra María del Carmen Rodríguez-Gacio for help in preparing the manuscript.

References

- Aguirre J, Rios-Momberg M, Hewitt D, Hansberg W (2005) Reactive oxygen species and development in microbial eukaryotes. Trends Microbiol 13:111–118
- Almagro L, Gómez-Ros LV, Belchi-Navarro S, Bru R, Ros-Barceló A, Pedreño MA (2009) Class III peroxidases in plant defence reactions. J Exp Bot 60:377–390
- Apel K, Hirt H (2004) Reactive oxygen species: metabolism, oxidative stress, and signal transduction. Annu Rev Plant Biol 55:373–399
- Bailly C, El-Maarouf-Bouteau H, Corbineau F (2008) From intracellular signalling networks to cell death: the dual role of reactive oxygen species in seed physiology. CR Biol 331:806–814
- Ball L, Accotto GP, Bechtold U, Creissen G, Funck D, Jiménez A, Kular B, Leyland N, Mejia-Carranza J, Reynolds H, Karpinski S, Mullineaux PM (2004) Evidence for a direct link between glutathione biosynthesis and stress defense gene expression in Arabidopsis. Plant Cell 16:2448–2462
- Bhattachrjee S (2005) Reactive oxygen species and oxidative burst: roles in stress, senescence and signal transduction in plant. Curr Sci 89:1113–1121
- Bienert GP, Schjoerring JK, Jahn TP (2006) Membrane transport of hydrogen peroxide. Biochim Biophys Acta 1758:994–1003
- Bienert GP, Moller ALB, Kristiansen KA, Schulz A, Moller IM, Schjoerring JK, Jahn TP (2007) Specific aquaporins facilitate the diffusion of hydrogen peroxide across membranes. J Biol Chem 282:1183–1192
- Bolwell GP, Daudi A (2009) Reactive oxygen species in plant-pathogen interactions. In: del Río LA, Puppo A (eds) Reactive oxygen species in plant signalling. Springer, Berlin, pp 113–133

- Chang CCC, Ball L, Fryer MJ, Baker NR, Karpinski S, Mullineaux PM (2004) Induction of ascorbate peroxidase 2 expression in wounded Arabidopsis leaves does not involve known wound-signalling pathways but is associated with changes in photosynthesis. Plant J 38:499–511
- Chang C, Damiani I, Puppo A, Frendo P (2009) Redox changes during the legume-rhizobium symbiosis. Mol Plant 2:370–377
- Chen YL, Huang RF, Xiao YM, Lu P, Chen J, Wang XC (2004) Extracellular calmodulin-induced stomatal closure is mediated by heterotrimeric G protein and H₂O₂. Plant Physiol 136:4096–4103
- Cheng Y, Song C (2006) Hydrogen peroxide homeostasis and signaling in plant cells. Sci China C Life Sci 49:1–11
- Conrath U (2006) Systemic acquired resistance. Plant Signal Behav 1:179–184
- Corpas FJ, Barroso JB, Palma JM, del Río LA (2009) Peroxisomes as key organelles in the metabolism of reactive oxygen species, and reactive sulfur species in plants. Research Signpost, Trivandrum, Kerala
- Costa A, Drago I, Behera S, Zottini M, Pizzo P, Schroeder J, Pozzan T, Lo Schiavio F (2010) H₂O₂ in plant peroxisomes: an *in vivo* analysis uncovers a Ca²⁺dependent scavenging system. Plant J 62:760–772
- D'Autréaux B, Toledano MB (2007) ROS as signalling molecules: mechanisms that generate specificity in ROS homeostasis. Nat Rev Mol Cell Biol 8:813–824
- De la Torre F, Rodríguez-Gacio MC, Matilla AJ (2006) How ethylene works in the reproductive organs of higher plants. A signaling update from the third millennium. Plant Sign Behav 1:231–242
- De Vleesschauwer D, Djavaheri M, Bakker M, Hofte M (2008) *Pseudomonas fluorescens* WCS374r-induced systemic resistance in rice against *Magnaporthe oryzae* is based on pseudobactin-mediated priming for a salicylic acid-repressible multifaceted defense response. Plant Physiol 148:1996–2012
- Delaunay A, Pflieger D, Barrault MB, Vinh J, Toledano MB (2002) A thiol peroxidase is an H₂O₂ receptor and redox-transducer in gene activation. Cell 111: 471–481
- Dellapenna D, Pogson BJ (2006) Vitamin synthesis in plants: tocopherol and carotenoids. Ann Rev Plant Biol 57:711–738
- Desikan R, Mackerness SA-H, Hancock JT, Neill SJ (2001) Regulation of the Arabidopsis transcriptome by oxidative stress. Plant Physiol 127:159–172
- Desikan R, Hancock JT, Neill SJ (2003) Oxidative stress signaling. In: Hirt H, Shinozaki K (eds) Plant responses to abiotic stress: topic in current genetics. Springer, Berlin, pp 121–148
- Desikan R, Hancock JT, Bright J, Harrison J, Weir I, Hooley R, Neill SJ (2005) A role for ETR1 in hydrogen peroxide signalling in stomatal guard cells. Plant Physiol 137:831–834
- Desikan R, Last K, Harrett-Williams R, Taglivia C, Harter K, Hooley R, Hancock JT, Neill SJ (2006) Ethyleneinduced stomatal closure in Arabidopsis occurs via

AtrbohFd-mediated hydrogen peroxide synthesis. Plant J 47:907–916

- Desikan R, Horák J, Chaban C, Mira-Rodado V, Witthöft J, Elgass K, Grefen C, Cheun M-K, Meixner AJ, Hooley R, Neill SJ, Hancock JT, Harter E (2008) The histidine kinase AHK5 integrates endogenous and environmental signals in Arabidopsis guards cells. PLoS One 3:e2491.1–e2491.15
- Desmond OJ, Manners JM, Stephens AE, Maclean DJ, Schenk PM, Gardiner DM, Munn AL, Kazan K (2008) The Fusarium mycotoxin deoxynivalenol elicits hydrogen peroxide production, programmed cell death and defence responses in wheat. Mol Plant Pathol 9:435–445
- Dòczi R, Brader G, Pettko-Szandtner A, Rajh I, Djmei A, Pitzshke A, Teige M, Hirt H (2007) The Arabidopsis mitogen-activated protein kinase kinase MKK3 is upstream of group mitogen-activated protein kinases and participates in pathogen signalling. Plant Cell 19:3266–3279
- Doke N (1983) Involvement of superoxide anion generation in hypersensitive response of potato tuber tissues to infection with an incompatible race of *Phytophthora infestans*. Physiol Plant Pathol 23:345–347
- Du YY, Wang PC, Chen J, Song CP (2008) Comprehensive functional analysis of the catalase gene family in *Arabidopsis thaliana*. J Integr Plant Biol 50:1318–1326
- Eastmond PJ (2007) Monodehydroascorbate reductase4 is required for seed storage oil hydrolysis and postgerminative growth in Arabidopsis. Plant Cell 219:1376–1387
- El-Maarouf-Bouteau H, Bailly C (2008) Oxidative signalling in seed germination and dormancy. Plant Signal Behav 3:175–182
- Foreman J, Demidchik V, Bothwell JHF, Mylona P, Miedema H, Torres MA, Linstead P, Costa S, Brownlee C, Jones JDG, Davies JM, Dolan L (2003) Reactive oxygen species produced by NADPH oxidase regulate plant cell growth. Nature 422:442–446
- Foyer CH, Noctor G (2005) Redox homeostasis and antioxidant signaling: a metabolic interface between stress perception and physiological responses. Plant Cell 17:1866–1875
- Gadjev I, Stone JM, Tsanko S (2008) Programmed cell death in plants: new insights into redox regulation and the role of hydrogen peroxide. Int Rev Cell Mol Biol 270:87–144
- Gálvez-Valdivieso G, Mullineaux PM (2010) The role of reactive oxygen species in signalling from chloroplasts to the nucleus. Physiol Plant 138:430–439
- Gechev TS, Hille J (2005) Hydrogen peroxide as a signal controlling plant programmed cell death. J Cell Biol 168:17–20
- Gechev TS, Van Breusegem F, Stone JM, Denev I, Laloi Ch (2006) Reactive oxigen species as signals that modulate plant stress responses and programmed cell death. Bioessays 28:1091–1101
- Gudesblat GE, Iusem ND, Morris PC (2007) Guard cellspecific inhibition of Arabidopsis MPK3 expression

causes abnormal stomatal responses to ABA and hydrogen peroxide. New Phytol 173:713–721

- Gupta R, Luan S (2003) Redox control of protein tyrosine phosphatases and mitogen activated protein kinases in plants. Plant Physiol 132:1149–1152
- Hancock JT, Henson D, Nyirenda M, Desikan R, Harrison J, Lewis L, Hughes J, Neill SJ (2005) Proteomic identification of glyceraldehyde 3-phosphate dehydrogenase as an inhibitory target of hydrogen peroxide in Arabidopsis. Plant Physiology Biochemistry 43:828–835
- Hancock J, Desikan R, Harrison J, Bright J, Hooley R, Neill S (2006) Doing the unexpected: proteins involved in hydrogen peroxide perception. J Exp Bot 57:1711–1718
- Herrero E, Ros J, Bellí G, Cabiscol E (2008) Redox control and oxidative stress in yeast cells. Biochim Biophys Acta 1780:1217–1235
- Hsieh TH, Lee JT, Charng YY, Chan MT (2002) Tomato plants ectopically expressing Arabidopsis CBF1 show enhanced resistance to water deficit stress. Plant Physiol 130:618–626
- Hückelhoven R (2007) Cell wall-associated mechanisms of disease resistance and susceptibility. Annu Rev Phytopathol 45:101–127
- Hung S-H, Yu Ch-W, Lin Ch-H (2005) Hydroge peroxide functions as a stress signal in plants. Bot Bull Acad Sin 46:1–10
- Iglesias-Fernández R, Rodríguez-Gacio MC, Matilla AJ (2011) Progress in research on dry after-ripening. Seed Sci Res. doi:10.1017/S096025851000036X
- Jones JDG, Dangl JL (2006) The plant immune system. Nature 444:323–329
- Jourdan E, Henry G, Duby F, Dommes J, Barthelemy JP, Thonart P, Ongena M (2009) Insights into the defense-related events occurring in plant cells following perception of surfactin-type lipopeptide from *Bacillus subtilis*. Mol Plant Microbe Interact 22:456–468
- Kibinza S, Vinel D, Côme C, Bailly C, Corbineau F (2006) Sunflower seed deterioration as related to moisture content during ageing, energy metabolism and active oxygen species scavenging. Physiol Plant 128: 496–506
- Kiitajima S (2008) Hydrogen peroxide-mediated inactivation of two chloroplastic peroxidases, ascorbate peroxidase and 2-cys peroxiredoxin. Photochem Photobiol 84:1404–1409
- Kwak JM, Mori IC, Pei ZM, Leonhardt N, Torres MA, Dangl JL, Bloom RE, Bodde S, Jones JD, Schroeder JI (2003) NADPH oxidase *AtrbohD* and *AtrbohF* genes function in ROS-dependent ABA signalling in Arabidopsis. EMBO J 22:2623–2633
- Lamb C, Dixon RA (1997) The oxidative burst in plant disease resistance. Annu Rev Plant Physiol Plant Mol Biol 48:251–275
- Lee KP, Kim C, Landgraf K, Apel K (2007) EXECUTER1and EXECUTER2-dependent transfer of stressrelated signals from the plastid to the nucleus of *Arabidospsis thaliana*. Proc Natl Acad Sci USA 104:10270–10275

- Liang YK, Dubos C, Dodd IC, Holroyd GH, Hetherington AM, Campbell MM (2005) AtMIB61, an R2R3-MYB transcription factor controlling stomatal aperture in *A. thaliana*. Curr Biol 15:1201–1206
- Lindermayr C, Saalbach G, Durner J (2005) Proteomic identification of S-nitrosylated proteins in Arabidopsis. Plant Physiol 137:921–930
- Liu X, Xing D, Li L, Zhang L (2007) Rapid determination of seed vigor based on the level of superoxide generation during early imbibition. Photochem Photobiol Sci 6:767–774
- Liu Y, Ye N, Liu R, Chen M, Zhang J (2010) H_2O_2 mediates the regulation of ABA catabolism and GA biosynthesis in Arabidopsis seed dormancy and germination. J Exp Bot 61:2979–2990
- Logan BA, Kornyeyev D, Hardison J, Holaday AS (2006) The role of antioxidant enzymes in photoprotection. Photosynth Res 88:119–132
- Mateo A, Mühlenbock P, Restérucci C, Chi-Chang C, Miszalski Z, Karpinska B, Parker JE, Mullineaux PM, Karpinski S (2004) LESION SIMULATING DISEASE 1 required for acclimation to conditions that promote excess excitation energy. Plant Physiol 136: 2818–2830
- Mateo A, Funck D, Mühlenbock P, Kular B, Mullineaux PM, Karpinski S (2006) Controlled levels of salicylic acid are required for optimal photosinthesis and redox homeostasis. J Exp Bot 57:1795–1807
- Matilla MA, Espinosa-Urgel M, Rodríguez-Herva JJ, Ramos JL, Ramos-González MI (2007) Genomic analysis reveals the major driving forces of bacterial life in the rhizosphere. Genome Biol 8:R179.1–R179.13
- Miller G, Mittler R (2006) Could heat shock transcription factors function as hydrogen peroxide sensors in plants? Ann Bot 98:279–288
- Miller G, Shulaev V, Mittler R (2008) Reactive oxygen signaling and abiotic stress. Physiol Plant 133:481–489
- Mittler R, Vanderauwera S, Gollery M, Van Breusegem F (2004) Reactive oxygen gene network of plants. Trends Plant Sci 132:1489–1498
- Molina L, Kahmann R (2007) An Ustilago maydis gene involved in H₂O₂ detoxification is required for virulence. Plant Cell 19:2293–2309
- Moller IM (2001) Plant mitochondria and oxidative stress electron transport, NADPH turnover, and metabolism of reactive oxygen species. Annu Rev Plant Physiol Plant Mol Biol 52:561–591
- Mori IC, Murata Y, Yang Y, Munemasa S, Wang YF, Andreoli S, Tiriac H, Alonso JM, Harper JF, Echer JR, Kwak JM, Schroeder JL (2006) CDPKs CPK6 and CPK3 function in ABA regulation of guard cell S-type anion- and Ca²⁺-permeable channels and stomatal closure. PLoS Biol 4:1749–1762
- Mou Z, Fan W, Dong X (2003) Inducers of plant systemic acquired resistance regulate NPR1 function through redox changes. Cell 113:935–944
- Müller K, Carstens AC, Linkies A, Torres MA, Leubner-Metzger L (2009) The NADPH-oxidase AtrbohB plays a role in Arabidopsis seed after-ripening. New Phytol 184:885–897

- Murgia I, Tarantin D, Vannini C, Bracale M, Caravieri S, Soave C (2004) Arabidopsis thaliana plants overexpressing thylakoidal ascorbate peroxidase show increased resistance to paraquat-induced photooxidative stress and to nitric oxide-induced cell death. Plant J 38:940–953
- Nakagami H, Kiegerl S, Hirt H (2004) OMTK1, a novel MAPKKK, channels oxidative stress signaling through direct MAPK interaction. J Biol Chem 279: 26959–26966
- Nakagami H, Pitzschke A, Hirt H (2005) Emerging MAP kinase pathways in plant stress signalling. Trends Plant Sci 10:339–346
- Narendra S, Venkataramani S, Shen G, Wang J, Pasapula V, Lin Y, Kornyeyev D, Holaday AS, Zhang H (2006) The Arabidopsis ascorbate peroxidase 3 is a peroxisomal membrane-bound antioxidant enzyme and is dispensable for *Arabidopsis thaliana* growth and development. J Exp Bot 57:3033–3042
- Neill SJ, Desikan R, Hancock JT (2002) Hydrogen peroxide signalling. Curr Opin Plant Biol 5:388–395
- Neill SJ, Desikan R, Hancock JT (2003) Nitric oxide signalling in plants. New Phytol 159:1483–1495
- Oracz K, El Maarouf-Bouteau H, Farrant J, Coopper K, Belgazhi M, Job C, Job D, Corbineau F, Bailly C (2007) ROS production and protein oxidation as novel mechanism of seed dormancy alleviation. Plant J 50:452–465
- Oracz K, El Maarouf-Bouteau H, Kranner I, Corbineau F, Bailly C (2008) Release of sunflower seed dormancy by cyanide: cross-talk with ethylene signalling pathway. J Exp Bot 59:2241–2251
- Oracz K, El Maarouf-Bouteau H, Kranner I, Bogatek R, Corbineau F, Bailly C (2009) The mechanisms involved in seed dormancy alleviation by hydrogen cyanide unravel the role of reactive oxygen species as key factors of cellular signalling during germination. Plant Physiol 150:494–505
- Pukacka S, Ratajezak E (2006) Antioxidative response of ascorbate-glutathione pathway enzymes andmetabolites to desiccation of recalcitrant Acer saccharinum seeds. J Plant Physiol 163:1259–1266
- Quan L-J, Zhang B, Shi W-W, Li HY (2008) Hydrogen peroxide in plants: a versatile molecule of the reactive oxygen species network. J Integr Plant Biol 50:2–18
- Queval G, Issakidis-Bourguet E, Hoeberichts FA, Vandorpe M, Gakière B, Vanacker H, Miginiac-Maslow M, Van Breusegem F, Noctor G (2007) Conditional oxidative stress responses in the Arabidopsis photorespiratory mutant *cat2* demonstrate that redox state is a key modulator of daylenghdependent gene expression, and define photoperiod as a crucial factor in the regulation of H₂O₂-induced cell death. Plant J 52:640–657
- Rentel MC, Lecourieux D, Ouaked F, Usher SL, Petersen L, Okamoto H, Knight H, Peck SC, Grierson CS, Hirt H, Knight MR (2004) OXI1 kinase is necessary for oxidative burst-mediated signalling in Arabidopsis. Nature 427:858–861
- Rhoads DM, Umbach AL, Subbaiah CC, Siedow JN (2006) Mitochondrial reactive oxygen species.

Contribution to oxidative stress and interorganellar signalling. Plant Physiol 141:357–366

- Romero-Puertas MC, Corpas FJ, Sandalio LM, Leterrier M, Rodríguez-Serrano M, del Río LA, Palma JM (2006) Glutathione reductase from pea leaves: response to abiotic stress and characterization of the peroxisomal isozyme. New Phytol 170:43–52
- Shetty NP, Jørgensen HJL, Jensen JD, Collinge DB, Shetty HS (2008) Roles of reactive oxygen species in interactions between plants and pathogens. Eur J Plant Pathol 121:267–280
- Shin R, Schachtman DP (2004) Hydrogen peroxide mediates plant root cell signaling in response to potassium deprivation. Proc Natl Acad Sci USA 101:8827–8832
- Slesak I, Libik M, Karpinska B, Karpinski S, Miszalski Z (2007) The role of hydrogene peroxide in regulation of plant metabolism and cellular signalling in response to environmental stresses. Acta Biochem Pol 54:39–50
- Spadaro D, Yun B-W, Spoel SH, Chu Ch, Wang Y-Q, Loake GJ (2010) The redox switch: dynamic regulation of protein function by cysteine modifications. Physiol Plant 138:360–371
- Swanson S, Gilroy S (2010) ROS in plant development. Physiol Plant 138:384–392
- Torres MA (2010) ROS in biotic interactions. Physiol Plant 138:414–429
- Tripathi BN, Bhatt I, Dietz K-J (2009) Peroxiredoxins: a less studied component of hydrogen peroxide detoxification in photosynthetic organisms. Protoplasm 235:3–15

- Vandenabeele S, Vanderauwer S, Vuylsteke M, Rombauts S, Langebartels C, Seidlitz HK, Zabeau M, Van Montagu M, Inzé D, Van Breusegem F (2004) Catalase deficiency drastically affects gene expression induced by high light in Arabidopsis thaliana. Plant J 39:45–58
- Vanderauwera S, Zimmermann P, Rombauts S, Vandenbeele S, Langerbartels C, Gruissem W, Inzé D, Van Breusegem F (2005) Genome-wide analysis of hydrogen peroxide-regulated gene expression in Arabidopsis reveals a high light-induced transcriptional cluster involved in anthocyanin biosynthesis. Plant Physiol 139:806–821
- Vanderauwera S, Hoeberichts FA, Van Breusegem F (2009) Hydrogen peroxide-responsive genes in stress acclimation and cell death. In: del Río LA, Puppo A (eds) Reactive oxygen species in plant signalling. Springer, Berlin, pp 149–164
- Vlot AC, Dempsey DA, Klessig DF (2009) Salicylic acid, a multifaceted hormone to combat disease. Annu Rev Phytopathol 47:177–206
- Wang P, Song Ch-P (2008) Guard-cell signalling for hydrogen peroxide and abscisic acid. New Phytol 178:703–718
- Yang T, Poovaiah BW (2002) Hydrogen peroxide homeostasis: activation of plant catalase by calcium/calmodulin. Proc Natl Acad Sci USA 99:4097–4102
- Yang J, Tsuichilhara N, Etoh T, Iwai S (2007) Reactive oxygen species and NO are involved in ABA inhibition of stomatal opening. Plant Cell Environ 30:1320–1325

Role of Phytohormone Signaling During Stress

17

Mohammad Miransari

Abstract

Plant hormones are among the most important plant components regulating different plant functions, which can equip the plant to survive under stress conditions. There has been extensive research work regarding the effects of plant hormones on plant growth and development. The notable phytohormones include auxin, abscisic acid, ethylene, cytokinins, gibberellins, jasmonates, salicylic acid, brassinosteroids, and strigolactones. Role of phytohormones may be useful for the production of transgenic plants, which are tolerant to stress. Selected plant hormone biosynthesis, their role on plant growth under stress, signal transduction pathways, and interactions are reviewed.

Keywords

Auxins • ABA • Gibberellins • Jasmonates • Brassinosteroids

1 Introduction

Plant hormones are a group of biochemical products handling different functions in plant. Plant hormones include auxin, abscisic acid (ABA), ethylene, cytokinins, gibberellins, jasmonates, salicylic acid, brassinosteroids, and strigolactones. They can regulate plant growth and development under different conditions

Department of Soil Science, Shahed University, College of Agricultural Sciences, 18151/159, Tehran, Iran e-mail: miransari1@gmail.com including stress (Table 17.1). There are different physiological alterations made by plant hormones at cellular and molecular level. At the time of hormonal activity plant genes are activated resulting in different morphological and physiological responses in plant (Table 17.1). Fluctuations in hormones can influence plant growth (Kagale et al. 2007; Jackson 2008; Hirayama and Shinozaki 2010).

The functions of plant hormones include: tissue organogenesis and development by affecting cell cycling, fruit ripening, controlling water behavior in plant by adjusting the stomata activity, and enhanced plant resistance to stress by activating different signaling pathways (Van der Knaap et al. 1999; Wang et al. 2007; Tuteja 2007).

M. Miransari (⊠)

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance* of Plants in the Era of Climate Change, DOI 10.1007/978-1-4614-0815-4_17,

[©] Springer Science+Business Media, LLC 2012

Hormone	Activities	Reference
Auxin	 (1) Regulation of phototropism, (2) regulation of embryo and fruit development, (3) organogenesis, formation and differen- tiation of vascular bundle, (4) root growth and development, (5) apical formation and dominance 	Han et al. (2009), Tomasz and Jiri (2006)
ABA	 (1) Dormancy and development, (2) stomatal activity, (3) morphogenesis of embryo, (4) protein and lipid production, (5) senescence of different tissues, and (6) tolerance to pathogens 	Tuteja (2007), Tripathi and Tuteja (2007)
Cytokinins	(1) Cell division and differentiation, (2) formation of membrane components, (3) carbon cycle in photosynthesis, (4) chlorophyll formation, (5) delaying leaf and chlorophyll senescence by decreasing the rate of protein and RNA degradation, (6) development of seed, (7) differentiation of vesicular bundle, (8) root and shoot growth, (9) balance of nutrients, and (10) stress resistance	Yordanov et al. (2000), Kulaeva and Prokoptseva (2004), Muller and Sheen (2007)
Ethylene	(1) Seed germination, (2) tissue senescence and abscission, and (3) tissue elongation under submerged conditions	Arteca and Arteca (2008)
Gibberellins	(1) Inducing plant systemic resistance, (2) seed germination, and (3) mediating plant response to environmental stresses	Ueguchi-Tanaka et al. (2005), Achard et al. (2006), Griffiths et al. (2006), Miransari and Smith (2009a, b)
Brassinosteroids	(1) Enhancing plant resistance to environmental stresses	Nunez et al. (2003), Vardhini and Rao (2003)
Jasmonates	(1) Enhancing plant resistance versus different environmental stresses and pathogen infection, (2) nodule morphogenesis	Wasternack (2007), Murray et al. (2007), Balbi and Devoto (2008), Hu et al. (2009)
Salicylic acid	Inducing plant systemic resistance	Lian et al. (2000), Sun et al. (2006), Chen et al. (2009)
Strigolactones	(1) Hyphal branching in AM fungi, (2) shoot branching in the host plant as well as the parasitic plant, <i>Striga</i>	Akiyama et al. (2005), López-Ráez et al. (2008), Miransari (2011)

 Table 17.1
 Plant hormones and their activities

Plant hormones regulate plant growth and development through a set of complex interaction between their signaling pathways. A signaling pathway is defined by a collection of elements, which eventually results in plant response to different parameters including stress Baron 1999; Klipp (Schwartz and and Liebermeister 2006). Research has indicated the importance of plant hormones during stress. During stress the activation of different signaling pathways mediated by different plant hormones and their interaction can enhance plant resistance to stress (Nakamura et al. 2006; Rolland et al. 2006; Truman et al. 2007). Accordingly, some of the most important findings regarding the effects of plant hormones through their signaling pathways on different stresses are presented.

2 Auxins

Auxin is among the most important plant hormones affecting plant growth and development. Auxin can modulate plant growth by affecting the process of phototropism in plant (Darwin 1880; Tomasz and Jiri 2006). In addition, auxin can regulate the development of embryo and fruit, organogenesis, formation and differentiation of vascular bundle, root growth and development, and apical formation and dominance (Han et al. 2009). The place of auxin synthesis is stem tip and young leaf, and it is then translocated to the site of action (Ljung et al. 2001).

Among the most important functions of auxin in plant is the formation of lateral roots

(Han et al. 2009), which is of special significance to plant growth under different conditions including stress. For example, root growth under compaction is adversely affected, decreasing plant growth as a result of reduction in the uptake of water and nutrients (Miransari et al. 2007, 2008, 2009a, b). Therefore, role of auxin under such conditions can be important in the alleviation of stress by stimulating root growth.

The expression of different auxin-responsive genes indicates that there is a cross-talk between auxin and signaling pathways (Jain and Khurana 2009). Auxin can rapidly induce the accumulation of a significant number of transcript factors related to different plant genes under different conditions including stress. Such plant genes include Aux/IAA, GH3, and small auxin-up RNA (SAUR) genes (Guilfoyle 1999). Molecular genetics and biochemical research have suggested that the Aux/IAA genes are related to auxin signaling (Leyser 2002; Woodward and Bartel 2005). These genes can activate the proteins, which can suppress the transcriptional activities regulated by auxin (Tiwari et al. 2004; Woodward and Bartel 2005).

The *GH3* genes are responsible for the production of enzymes, which produce amino acidrelated products by adenylation of indole 3-acetic acid and hence inhibit the production of extrafree auxin resulting in auxin homeostasis (Staswick et al. 2005). The *GH3* genes can also turn the produced amino acid-related products into salicylic acid and jasmonates (Staswick et al. 2002). The *SAUR* genes can result in the production of proteins, which may influence cell elongation regulated by auxin (Hagen and Guilfoyle 2002).

Auxin signaling pathway is related to the expression of different genes, which are mostly induced by two transcriptional factors including auxin response factors (ARFs) and the Aux/IAA repressors. The auxin response promoter elements in the responsive auxin genes are bound by ARFs. Reduction of auxin concentration to the amounts less than the threshold level results in the combined activation of ARFs and Aux/IAA repressors and hence the inhibition of the responsive genes. However, high concentration of auxin would adversely affect the Aux/IAA repressors and hence restores the activity of such responsive genes (Han et al. 2009; Jain and Khurana 2009; Ghanashyam and Jain 2009).

3 Abscisic Acid

There are a wide range of functions controlled and affected by ABA in plant including seed dormancy and development, stomatal activity, morphogenesis of embryo, protein and lipid production, senescence of different tissues, and tolerance to pathogens (Tuteja 2007; Tripathi and Tuteja 2007). Different stresses result in cell desiccation and osmotic imbalance and hence there may be similar signaling pathways and genes, expressed during the stress (Tuteja 2007).

Stresses such as drought and salinity result in the production of ABA in the roots and its eventual translocation to the shoots affecting stomata activities and cellular growth. In addition, ABA can also be produced in plant leaf and translocated to the other parts of plant (Wilkinson and Davies 2002; Chaves et al. 2009). The other parameter controlling ABA localization is the xylem/apoplastic pH. For example, when plant is subjected to drought stress the higher xylem/apoplastic pH prohibits the movement of ABA from the xylem/apoplastic to the symplastic space resulting in the enhancement of ABA concentration in the guard cells. Different stresses including drought, light, salinity, and nitrate can increase xylem sap pH and hence affect stomata activities (Jia and Davies 2007). ABA can also influence plant growth under stress by affecting gene expression. Furthermore, under drought and high light stress, the production of sugars and their translocation in the xylem can affect stomata response to ABA (Wilkinson and Davies 2002).

Chaves and Oliveira (2004) indicated that under stress the production of soluble sugars, which can also act as signal molecules in plant, is altered. In addition, such sugar molecules can also be interactive with plant hormones (Rolland et al. 2006). It has been indicated that nitrous oxide is also a signal molecule influencing the effects of plant hormones and other signal molecules in response to environmental parameters by enhancing the sensitivity of plant cells to such signal molecules. In addition, nitrous oxide can also influence ABA effects on the activity of stomata (Neill et al. 2003).

Under water stress the amount of ABA increases (Kulkarni et al. 2000; Liu et al. 2005). ABA can primarily control water transpiration from the leaf and then can alleviate the stress and enhance plant tolerance by activating the expression of different stress genes (Bray 2004; Zhang et al. 2006). Genes, which are expressed under drought stress, include functional and regulatory ones. Functional genes can perform some specific functions related to the alleviation of stress by inducing different transporters including detoxifying enzymes, enzymes related to the production of osmolyte, and different proteases. However, regulatory genes including transcription factors, phosphatases and protein kinases, and the ones related to the production of ABA, can regulate the activity of functional genes (Aroca et al. 2007).

With respect to the functions mentioned for ABA, it is the most important plant signal activated during stress (Zhang et al. 2006). Interestingly and similarly, the soil fungi, arbuscular mycorrhiza (AM), can alleviate drought stress in their host plants. AM fungi can establish symbiosis with their host plant and enhance its water and nutrient uptake by its extensive hyphal network in exchange for carbon. Mycorrhizal plants are able to regulate their ABA level more efficiently (Auge 2001; Aroca et al. 2007).

The *nced* genes, which are expressed in plant under drought stress (Wan and Li 2006), are induced by ABA, if available at a minimum amount (Cohen et al. 1999). The effects of small RNA pathways on ABA were evaluated by Zhang et al. (2008) using mutants. They found that adverse effects on small RNA pathways can increase the production of ABA indicating that there is a link between small RNA pathways and signaling pathways resulting in the production of ABA in plant cells.

Under submerged conditions there is some kind of interesting interactions between different

plant hormones. The accumulated amounts of ethylene down regulate the production of ABA through inhibiting the expression of 9-cisepoxycarotenoid dioxygenase and by turning ABA into pahseic acid. The 9-cis-epoxycarotenoid dioxygenase are among the genes, responsible for the production of ABA through the pathway of carotenoid cleavage dioxygenases (Kende et al. 1998; Benschop et al. 2005). Prevented production of endogenous ABA results in the activation of the enzyme, gibberellin 3-oxidase, which catalyses the pathway related to the production of gibberellins (Benschop et al. 2006), and in submerged rice the production of gibberellins by the internodes (Kende et al. 1998). Down regulation of gibberellins-related genes can induce the elongation of rice roots under submerged conditions. The activities of such genes include the loosening of the cell wall, cell cycle, and starch turnover (Van der Knaap et al. 1999, 2000).

4 Cytokinins

The plant hormone cytokinins can regulate the following important functions in plant: (1) cell division and differentiation, (2) formation of membrane components, (3) carbon cycle in photosynthesis, (4) chlorophyll formation, (5) delaying leaf and chlorophyll senescence by decreasing the rate of protein and RNA degradation, which is related to the controlling effects of hormone on the production of protease and activity of RNase, (6) development of seed, (7) differentiation of vesicular bundle, (8) root and shoot growth, (9) balance of nutrients, and (10) stress resistance (Yordanov et al. 2000; Muller and Sheen 2007). The receptors perceiving cytokinins were found in the cellular membrane. Cytokinins are able to act multifunctionally by affecting different physiological processes in plant and controlling stresses (Kulaeva and Prokoptseva 2004).

Under nonstressed or stressed conditions, decreased level of cytokinins result in leaf senescence (Naqvi 1995). Exogenous application of cytokinins delays the process of leaf senescence (Okamoto et al. 2010). The process of leaf senescence can also be delayed by regulating the related genes pathways, which are under the influence of cytokinins level in plant (Naqvi 1995). The gene, which is responsible for the production of cytokinins is *ipt*, which produces the enzyme isopentyl transferase, promoting the formation of isopentenyladenosine-5'-monophosphate (McGraw 1987). The *ipt* genes are expressed at low levels under controlled and drought-stress conditions, being highly specific in the cellular tissue (Vyroubalova et al. 2009).

Cytokinins are able to protect the process of photosynthesis in plants under stress. The related mechanism includes the interaction of cytokinins with receptor proteins resulting in the production of the signal pathway. Consequently, the genes are expressed and produce mRNAs, photosynthesisrelated proteins, the enzyme ribulose bisphosphate carboxylase/oxygenase, electrons, and carbon (Chernyadev 2009; Bianco et al. 2009).

It is likely to enhance plant tolerance to drought by genetically modifying the plant using the *ipt* gene regulated by the related promoter, which delays the process of leaf senescence. In addition, the rate of photosynthesis and production of antioxidant enzymes increase under such genetic modification. It must be mentioned that there is a kind of regulatory interactions between leaf transpiration and leaf photosynthesis while the leaf senescence takes place in the plant (Rivero et al. 2007, 2009).

While leaf senescence can increase plant tolerance to drought stress by significantly reducing the rate of leaf transpiration, the presence of old leaf in plant can contribute to the enhanced photosynthesis process in plant (Chaves et al. 2003; Rivero et al. 2007, 2009). Under stresses such as drought while the production and transport of cytokinins are prohibited, its degradation is encouraged resulting in a decrease in plant growth as well as plant-reduced tolerance to stress (Yang et al. 2002; Kudoyarova et al. 2006).

The two-component pathways related to cytokinins regulation may affect ABA activity and can alleviate the effects of osmotic stress. The three histidine kinases as cytokinins receptors can adversely regulate ABA activity, as well as the stress. Accordingly, it can be mentioned that during the stress control in plant, there is a crosstalk between cytokinin, ABA, and the osmotic stress indicating that cytokinin pathway and metabolism is of particular importance to plant growth and development, especially under stress (Tran et al. 2009).

5 Ethylene

Ethylene is a gaseous plant hormone with important functioning in plant. Although ethylene is among plant hormones with the simplest structures, it can regulate some of the most important functioning in plant including seed germination, tissue senescence, and abscission (Arteca and Arteca 2008). There are some complex pathways and signaling related to ethylene, stimulated by other plant hormones and parameters such as plant growth, pathogens, and sugars (Abeles et al. 2004; Stepanova and Alonso 2005).

There is a set of interactions between ethylene and ethylene receptors in the plasma membrane indicating the pathway of ethylene signaling, although such a pathway is adversely affected by the ethylene receptors. The membrane receptors are two-component histidine protein kinases (Mount and Chang 2002). The degradation of ethylene receptors after ethylene binding can further increase ethylene sensitivity, determined by the presence of a novel protein. The degradation of transcription factors in the nucleus is also controlled (Kendrick and Chang 2008).

In the recent years the signaling pathway of ethylene is among the most known pathways. The important transcription factor is ETHYLENE INSENSITIVE3 (EIN3). However, what has to be yet investigated is the biochemical mechanism by which the ethylene receptor signaling is performed (Kendrick and Chang 2008). A key process in the functioning of ethylene is the degradation of proteins, which control both ethylene biosynthesis and ethylene perceiving by the receptors. The other important component related to ethylene signaling is the degradation of EIN3 by the proteins EIN-BINDING F-BOXES (McClellan and Chang 2008).
Importantly, the elevated production of ethylene under stress adversely affects plant growth and development. If there is any way by which the degradation of ethylene is performed, the stress of ethylene on plant growth and development can be controlled. Plant growth promoting rhizobacteria (PGPR) have the ability to produce the enzyme, 1-aminocyclopropane-1carboxylate (ACC) deaminase, which can degrade the precursor for ethylene production by the ACC oxidase pathway. There has been extensive research work regarding such alleviating effects of ACC-deaminase on plant growth and development under stress, especially by professor Glick and his research team (Glick et al. 2007; Yang et al. 2008; Jalili et al. 2009).

The other important functioning regarding ethylene is its important role under submerged condition, which is considered stressful to most plant species. Under submerged conditions, the plant must own some unique abilities to be able to grow under such conditions. Increased rate of adventitious roots as well as formation of new leaf and hence increased photosynthesis under submerged conditions is among such abilities regulated by ethylene. Ethylene concentration can increase up to 100 times higher in plants under submerged compared with nonsubmerged conditions resulting in tissue elongation. Such kind of elongation is a result of processes, which would loosen the cell wall, extend and divide the cell, and is also under the influence of ethylene interaction with other plant hormones (Jackson 2008).

Under submerged, due to the efficiency of oxygen and high concentration of CO_2 the synthesis of ethylene can increase up to four times higher in the base of rice stem. The increased ethylene concentration is a result of higher biosynthesis of ethylene prerequisite 1-aminocyclopropane-1-carboxylic acid (ACC). Such alterations are accompanied with some other genetic processes including mRNA control on the activation of ACC oxidase genes. These activities would collectively result in the enhanced production of ethylene in rice under submerged conditions (Mekhedov and Kende 1996; Vriezen et al. 1999; Zhou et al. 2001).

6 Gibberellins

Gibberellins are plant hormones performing different functions in plant. The three enzymes of cyclases, monooxygenases, and dioxygenases catalyze the production of gibberellins, which are tetracyclic diterpenoids, from geranylgeranyl diphosphate (Sponsel and Hedden 2004). Gibberellins are able to enhance plant growth by degrading the negative growth regulators DELLA proteins (Ueguchi-Tanaka et al. 2005; Griffiths et al. 2006). Gibberellins can induce plant systemic resistance and result in seed germination. Fungi and bacteria are also able to produce gibberellins as secondary metabolites resulting in the signaling interaction with their host plant (MacMillan 2001; Miransari and Smith 2009a, b).

It has been recently indicated that *Arabidopsis* DELLA proteins, as negative regulator of gibberellins signaling, can influence plant systemic resistance by affecting jasmonates and salicylic acid signaling pathways (Navarro et al. 2008). The *Arabidopsis* mutant which does not have the DELLA genes is very susceptible to pathogenic fungi (Navarro et al. 2008). DELLA proteins are able to activate plant systemic resistance to pathogenic fungi by affecting the jasmonates/ethylene pathways. Accordingly, DELLA proteins can influence plant systemic resistance by affecting the combined response of salicylic acid, jasmonates, and ethylene pathways to the fungal infection.

It has also been indicated that the response of plant to the environmental stresses is also mediated by DELLA proteins affecting the combined response of plant hormonal pathways to the stress (Achard et al. 2006). Mutant, which do not have gibberellins receptors in their cellular membrane, accumulate higher rate of gibberellins enhancing plant response to pathogenic fungi relative to the wild types (Tanaka et al. 2006). In addition to fungi, viral infection can also affect gibberellins pathway in plant. For example, rice dwarf virus suppressed the expression of the enzyme *ent*kaurene oxidase, which results in the production of gibberellins in rice plants (Zhu et al. 2005).

7 Brassinosteroids

Brassinosteroids are plant hormones found in different parts of the plant including seed, pollen, flower, fruit, leaf, vesicular bundle, root, and shoot. Such steroidal compounds can be found in association with sugars and fatty acids. So far about 70 different plant brassinosteroids have been recognized and isolated, affecting plant growth and development (Sasse 2003; Yu et al. 2008). Under biotic (pathogen infection) and abiotic stresses the level of brassinosteroids in plant may increase. However, the related mechanisms are not known yet (Krishna 2003).

Different stresses such as salinity, drought, heavy metals, high or low temperature, etc., usually result in the similar cellular pathways and responses including the regulation of antioxidants, production of stress protein, and increased concentration of solutes altering the production of hormones in plant (Smirnoff 1995; Sajedi et al. 2011). There is a cross-talk between different hormones affecting plant growth and development. Under stress, the production of kinase protein in plant and the related responses by plant hormones is altered resulting in the induction of signals related to the production of reactive oxygen species, which is of high significance in the alleviation of stress. In addition, there is also some sort of interaction between different plant hormones under stress. For example, among the mechanisms by which brassinosteroids can affect plant response to stress is the production of jasmonates (Mussig et al. 2000; Schaller et al. 2000; Miller et al. 2010).

Under stress the production of reactive oxygen species can adversely affect cellular growth and development (Sajedi et al. 2011). The role of brassinosteroids in different plant physiological mechanisms, which results in the regulation of plant growth and development, has been indicated. However, there is not much known about the controlling effects of brassinosteroids on oxidative stress.

Under stress, use of exogenous brassinosteroids results in the modification of antioxidant enzymes including gluthatione peroxidase, ascorbate peroxidase, superoxide dismutase, and catalase as well as the non-enzymatic antioxidants including gluthatione, cartenoids, tocopherols, ascorbic acid, etc. (Nunez et al. 2003; Vardhini and Rao 2003). In sorghum (*Sorghum vulagre* L.) subjected to osmotic stress while brassinosteroids increased the activity of catalase, they decreased the activities of ascorbic acid oxidase and peroxidase (Vardhini and Rao 2003).

Under salinity stress, treating rice seedlings with brassinosteroids significantly enhanced the activity of catalase, superoxide dismutase, and glutathione reductase and slightly increased the activity of ascorbate peroxidase (Nunez et al. 2003). The fact that molecular oxygen is necessary for the production of brassinosteroids at different stages indicates that this hormone can mediate the effects of hypoxia on plant growth and development. Upreti and Murti (2004) indicated that brassinosteroids can increase root nodulation in *Phaseolus vulgaris* as well as ABA contents and cytokinin transzeatin riboside.

Brassinosteroids can also enhance root nodulation and crop yield under nonstressed and water-stressed conditions by affecting the content of cytokinin in the nodulated roots of *Phaseolus vulgaris*. Seedling growth of sorghum and sugar beet was increased by brassinosteroids under osmotic and water stress, respectively. Such alleviating effects were attributed to the enhanced level of acid invertase in the plant young leaf (Schilling et al. 1991). As a result of osmotic stress, the content of protein in three sorghum varieties decreased, which was completely restored and stimulated by brassinosteroids. The stress also increased the proline level (Vardhini and Rao 2003).

Use of brassinosteroids alleviated the stress of cadmium on the performance of chickpea (Hasan et al. 2008) and mustard (Hayat et al. 2007) and also the stresses of aluminum and nickel on the growth of mung bean (Ali et al. 2008) and mustard (Alam et al. 2007), respectively. Such alleviating effects may be attributed to the enhanced activities of antioxidant enzymes such as catalase, superoxidase, and peroxidase by brassinosteroids (Hasan et al. 2008).

In case of salinity stress, brassinosteroids alleviated the stress of salinity on rice (*Oryza sativa*), seed germination, and seedling growth. The hormone was also able to restore plant chlorophylls and enhance the activity of nitrate reductase under salinity. Brassinosteroids increased the cellular growth of rice seedlings under suboptimal temperature (15°C) as well as the germination of rice and corn seeds (He et al. 1991; Fujii and Saka 2001). Although brassinosteroids and ABA may affect plant systemic resistance at the time of pathogen infection, the induction of plant systemic resistance is mostly related to the combining effects of signaling pathways induced by salicylic acid, jasmonic acid, and ethylene.

8 Jasmonates

Jasmonates are lipid plant hormones affecting plant systemic resistance as well as plant growth and development and elasticity by their signaling pathways. The lipase enzymes synthesize jasmonates as oxylipins (oxygenated fatty acids). The enzymes, which are located in the chloroplast membrane, result in the release of linolenic acid, which is then oxygenated by lipoxygenases and produces hydroperoxide derivatives (Wasternack 2007; Schaller and Stintzi 2009).

Jasmonates are able to affect gene expression in plant positively or negatively while interacting with other plant hormones including salicylic acid, auxin, ABA, and ethylene (Wasternack 2007). Jasmonates can enhance plant resistance versus different environmental stresses and pathogen infection. There is a high rate of positive or negative cross-talk and interactions between jasmonates and the other plant hormone salicylic acid determining the ultimate response of plant to stress (Wasternack 2007; Balbi and Devoto 2008).

Plant mutants lacking the ability to synthesize jasmonates have been used to investigate the signaling pathways, which result in the production of jasmonates (Devoto and Turner 2005; Lorenzo and Solano 2005; Schilmiller et al. 2007). Jasmonates can also mediate plant response to stress by affecting the production of reactive oxygen species, nitrous oxide (NO), influx of calcium as well as activation of nitrogen protein kinase (Hu et al. 2009). The important role of jasmonates in nodule organogenesis has also been indicated by different researchers. Nodules are root organs developed during the symbiosis between the soil bacteria *Rhizobium* and their specific host plant from the leguminous family. Nodules are the place of rhizobium residence, for the fixation of atmospheric N by the production of rhizobium nitrogenase (Miransari and Smith 2007, 2008, 2009a, b). It has been recently indicated that for the onset of nodule development, the cytokinins signaling pathway is necessary (Murray et al. 2007).

The other plant hormones including auxin, ABA, ethylene, gibberellins, and brassinosteroids are also required for nodule development (Oldroyd et al. 2001; Ferguson et al. 2005; van Noorden et al. 2006). However, more research must be performed to indicate the other important details regarding the complex effects of hormonal signaling pathways on nodule organogenesis.

It has also been indicated that jasmonates can also influence nodule formation. For example, the antagonistic effects of jasmonates on the process of nodulation in Medicago truncatula and Lotus japonicus have been indicated (Sun et al. 2006; Nakagawa and Kawaguchi 2006). There is also some kind of positive and negative interactions between jasmonates and salicylic acid during the process of nodule formation (Sun et al. 2006). Jasmonates are able to alleviate the stress of salinity on barley growth. In plants treated with jasmonates, lower amounts of Na⁺ were found in plant shoot. Such alleviating effects were attributed to the performance of the following three genes including apoplastic invertase, arginine decarboxylase, and Rubisco, regulated by jasmonates (Tuteja and Sopory 2008).

9 Salicylic Acid

Salicylic acid is also another important plant hormone affecting plant systemic resistance to pathogen infection (Lian et al. 2000). During the activation of plant systemic resistance, transcriptional factors are activated and transcriptional repressors are inhibited (Dong 2001). It can adversely affect gene expression when affected by the stress hormone jasmonates. Borsani et al. (2001) indicated that salicylic acid can affect the production of reactive oxygen species as a result of osmotic stress by NaCl in Arabidopsis seedlings. According to Ndamukong et al. (2007), glutaredoxin is the protein regulating the pathways related to salicylic signaling. It has been indicated that the adverse effects of salicylic acid on pathogen growth in plants is by the suppressing effects of salicylic acid on the auxin signaling pathway (Wang et al. 2007). There are also positive and negative interactions between salicylic acid and jasmonates during the process of nodule organogenesis (Sun et al. 2006).

There are two different pathways by which salicylic acid is synthesized. The cinnamate pathway in which cinnamate is synthesized from the phenylalanine ammonia lyase (PAL) resulting in the production of salicylic acid. Silencing the PAL genes inhibits the production of salicylic acid in plant and inhibiting the activity of PAL genes chemically reduces the production of salicylic acid in plant. Salicylic acid is also produced in the isochorismate pathway catalyzed by isochorismate synthase (Chen et al. 2009).

In brief, it can be stated that plant parameters such as salicylic acid can enhance plant systemic resistance to stresses such as pathogen invasion by the following mechanisms: (1) expression of PAL genes, (2) expression of priming genes, (3) activation of pathways resulting in phytoalexin production, (4) deposition of callose, (5) oxidative burst, (6) phenolic compounds deposition, and (7) deposition of hydroxycinnamoyltyramine products (Goellner and Conrath 2008).

10 Strigolactones

Strigolactones are new classes of plant hormones produced from carotenoids, probably by carotenoid cleavage deoxygenase or 9-*cis* epoxycarotenoid deoxygenase. They affect the process of symbiosis between the soil fungi AM and the host plant as hyphal branching factors, shoot branching, and seed germination of parasitic weeds such as *Striga*. Their production by plants roots is significantly enhanced by phosphate starvation (Akiyama et al. 2005; López-Ráez et al. 2008; Miransari 2011).

During the process of symbiotic association between AM fungi and the host plant, an extensive network of hypha is developed, substantially increasing the uptake of water and nutrients by the host plant (Smith and Read 2008). Among different nutrients, phosphorous is more affected by AM fungi symbiosis with the host plant. Interestingly, strigolactones are more produced under phosphate starvation, which mediate the activation of strigolactones producing genes (López-Ráez and Bouwmeester 2008). However, other than P starvation, there is no any other details regarding the effects of strigolactones on stresses affecting plant growth.

11 Conclusion and Future Perspectives

Some of the most important details regarding the production of plant hormones, their signal pathways under different conditions including stress and the interactions between plant hormones were reviewed. Accordingly, plant hormones are among the most important plant components which can make the plant survive under different conditions including stress. Under stress plant hormones mediate plant genes, which can alleviate the stress by the production of stress proteins. There are yet more details that must be known regarding the functions of plant hormones under different conditions including stress. Elucidation of such details may result in the production of transgenic plants, which are more tolerant under stress.

References

- Abeles FB, Morgan PW, Saltveit ME Jr (2004) Ethylene in Plant Biology, 2nd edn. Academic, San Diego
- Achard P, Cheng H, De Grauwe L, Decat J, Schoutteten H, Moritz T, Van Der Straeten D, Peng J, Harberd NP (2006) Integration of plant responses to environmentally activated phytohormonal signals. Science 311:91–94

- Akiyama K, Matsuzaki KI, Hayashi H (2005) Plant sesquiterpenes induce hyphal branching in arbuscular mycorrhizal fungi. Nature 435:824–827
- Alam MM, Hayat S, Ali B, Ahmad A (2007) Effect of 28-homobrassinolide on nickel induced changes in Brassica juncea. Photosynthetica 45:139–142
- Ali B, Hasan SA, Hayat S, Hayat Q, Yadav S, Fariduddin Q, Ahmad A (2008) A role for brassinosteroids in the amelioration of aluminum stress through antioxidant system in mung bean (*Vigna radiata* L. Wilczek). Environ Exp Bot 62:153–159
- Aroca R, Porcel R, Ruiz-Lozano JM (2007) How does arbuscular mycorrhizal symbiosis regulate root hydraulic properties and plasma membrane aquaporins in *Phaseolus vulgaris* under drought, cold or salinity stresses? New Phytol 173:808–816
- Arteca RN, Arteca JM (2008) Effects of brassinosteroid, auxin, and cytokinin on ethylene production in Arabidopsis thaliana plants. J Exp Bot 59:3019–3026
- Auge RM (2001) Water relations, drought and vesiculararbuscular mycorrhizal symbiosis. Mycorrhiza 11:3–42
- Balbi V, Devoto A (2008) Jasmonate signalling network in Arabidopsis thaliana: crucial regulatory nodes and new physiological scenarios. New Phytol 177:301–318
- Benschop JJ, Jackson MB, Guhl K, Vreeburg RAM, Croker SJ et al (2005) Contrasting interactions between ethylene and abscisic acid in *Rumex* species differing in submergence tolerance. Plant J 44:756–768
- Benschop JJ, Bou J, Peeters AJM, Wagemaker N, Guhl K et al (2006) Long-term submergence-induced elongation in *Rumex palustris* requires abscisic acid-dependent biosynthesis of gibberellin. Plant Physiol 141:1644–1652
- Bianco C, Imperlini E, Defez R (2009) Legumes like more IAA. Plant Signal Behav 4:763–765
- Borsani O, Valpuesta V, Botella MA (2001) Evidence for a role of salicylic acid in the oxidative damage generated by NaCl and osmotic stress in Arabidopsis seedlings. Plant Physiol 126:1024–1030
- Bray EA (2004) Genes commonly regulated by waterdeficit stress in Arabidopsis thaliana. J Exp Bot 55:2331–2341
- Chaves MM, Oliveira MM (2004) Mechanisms underlying plant resilience to water deficits: prospects for water-saving agriculture. J Exp Bot 55:2365–2384
- Chaves MM, Maroco JP, Pereira JS (2003) Understanding plant responses to drought: from genes to whole plant. Funct Plant Biol 30:239–264
- Chaves MM, Flexas J, Pinheiro C (2009) Photosynthesis under drought and salt stress: regulation mechanisms from whole plant to cell. Ann Bot 103:551–560
- Chen Z, Zheng Z, Huang J, Lai Z, Fan B (2009) Biosynthesis of salicylic acid in plants. Plant Signal Behav 4:493–496
- Chernyadev I (2009) The protective action of cytokinins on the photosynthetic machinery and productivity of plants under stress. Appl Biochem Microbiol 45:351–362
- Cohen A, Moses MS, Plants AL, Bray EA (1999) Multiple mechanisms control the expression of abscisic acid (ABA) requiring genes in tomato plants exposed to soil water deficit. Plant Cell Environ 22:989–998

- Darwin C (1880) The Power of Movement in Plants. John Murray, London
- Devoto A, Turner JG (2005) Jasmonate-regulated *Arabidopsis* stress signaling network. Physiol Plant 123:161–172
- Dong X (2001) Genetic dissection of systemic acquired resistance. Curr Opin Plant Biol 4:309–314
- Ferguson BJ, Ross JJ, Reid JB (2005) Nodulation phenotypes of gibberellin and brassinosteroid mutants of pea. Plant Physiol 138:2396–2405
- Fujii S, Saka H (2001) The promotive effect of brassinolide on lamina joint-cell elongation, germination and seedling growth under low-temperature stress in rice (*Oryza sativa* L.). Plant Prod Sci 4:210–214
- Ghanashyam C, Jain M (2009) Role of auxin-responsive genes in biotic stress responses. Plant Signal Behav 4:846–848
- Glick BR, Todorovic B, Czarny J, Cheng Z, Duan J (2007) Promotion of plant growth by bacterial ACC deaminase. Crit Rev Plant Sci 26:227–242
- Goellner K, Conrath U (2008) Priming: it's all the world to induced disease resistance. Eur J Plant Pathol 121:233–242
- Griffiths J, Murase K, Rieu I, Zentella R, Zhang ZL, Powers SJ, Gong F, Phillips AL, Hedden P, Sun TP, Thomas SG (2006) Genetic characterization and functional analysis of the GID1 gibberellin receptors in Arabidopsis. Plant Cell 18:3399–3414
- Guilfoyle TJ (1999) Auxin-regulated genes and promoters. In: Hooykaas PJJ, Hall MA, Libbenga KR (eds) Biochemistry and Molecular Biology of Plant Hormones. Elsevier, Amsterdam, p 423
- Hagen G, Guilfoyle T (2002) Auxin-responsive gene expression: genes, promoters and regulatory factors. Plant Mol Biol 49:373–385
- Han H, Zhang S, Sun X (2009) A review on the molecular mechanism of plants rooting modulated by auxin. Afr J Biotechnol 8:348–353
- Hasan SA, Hayat S, Ali B, Ahmad A (2008) 28-homobrassinolide protects chickpea (*Cicer arieti-num*) from cadmium toxicity by stimulating antioxidants. Environ Pollut 151:60–66
- Hayat S, Ali B, Hasan SA, Ahmad A (2007) Brassinosteroid enhanced the level of antioxidants under cadmium stress in *Brassica juncea*. Environ Exp Bot 60:33–41
- He RY, Wang GJ, Wang XS (1991) Effect of brassinolide on growth and chilling resistance of maize seedlings. In: Cutler HG, Yokota T, Adam G (eds) Brassinosteroids: Chemistry, Bioactivity and Applications. American Chemical Society, Washington, DC, pp 220–230
- Hirayama T, Shinozaki K (2010) Research on plant abiotic stress responses in the post-genome era: past, present and future. Plant J 61:1041–1052
- Hu X, Li W, Chen Q, Yang Y (2009) Early signal transduction linking the synthesis of jasmonic acid in plant. Plant Signal Behav 4:696–697
- Jackson MB (2008) Ethylene-promoted elongation: an adaptation to submergence stress. Ann Bot 101:229–248
- Jain M, Khurana J (2009) Transcript profiling reveals diverse roles of auxin-responsive genes during

reproductive development and abiotic stress in rice. FEBS J 276:3148–3162

- Jalili F, Khavazi K, Pazira E, Nejati A, Rahmani HA, Sadaghiani HR, Miransari M (2009) Isolation and characterization of ACC deaminase producing fluorescent pseudomonads, to alleviate salinity stress on canola (*Brassica napus* L.) growth. J Plant Physiol 166:667–674
- Jia W, Davies WJ (2007) Modification of leaf apoplastic pH in relation to stomatal sensitivity to root-sourced abscisic acid signals. Plant Physiol 143:68–77
- Kagale S, Divi UK, Krochko JE, Keller WA, Krishna P (2007)Brassinosteroid confers tolerance in Arabidopsis thaliana and Brassica napus to a range of abiotic stresses. Planta 225:353–364
- Kende H, Van Der Knaap E, Cho HT (1998) Deepwater rice: a model plant to study stem elongation. Plant Physiol 118:1105–1110
- Kendrick M, Chang C (2008) Ethylene signaling: new levels of complexity and regulation. Curr Opin Plant Biol 11:479–485
- Klipp E, Liebermeister W (2006) Mathematical modeling of intracellular signaling pathways. BMC Neurosci 7:S10
- Krishna P (2003) Brassinosteroid-mediated stress responses. J Plant Growth Regul 22:289–297
- Kudoyarova GR, Vysotskaya LB, Cherkozyanova A, Dodd IC (2006) Effects of partial root zone drying on the concentration of zeatin type cytokinins in tomato (*Solanum lycopersicum* L.) xylem sap and leaves. J Exp Bot 58:161–168
- Kulaeva N, Prokoptseva S (2004) Recent advances in the study of mechanism of action of phytohormones. Biokhimiya 69:293–310
- Kulkarni MJ, Prasad TG, Sashidhar VR (2000) Genotypic variation in 'early warning signals' from roots in drying soil: intrinsic differences in ABA synthesising capacity rather than root density determines total ABA 'message' in cowpea (*Vigna unguiculata* L.). Ann Appl Biol 136:267–272
- Leyser O (2002) Molecular genetics of auxin signaling. Annu Rev Plant Biol 53:377–398
- Lian B, Zhou X, Miransari M, Smith DL (2000) Effects of salicylic acid on the development and root nodulation of soybean seedlings. J Agron Crop Sci 185:187–192
- Liu F, Jensen CR, Andersen MN (2005) A review of drought adaptation in crop plants: changes in vegetative and reproductive physiology induced by ABA-based chemical signals. Aust J Agr Res 56:1245–1252
- Ljung K, Bhalerao RP, Sandberg G (2001) Sites and homeostatic control of auxin biosynthesis in Arabidopsis during vegetative growth. Plant J 28:465–474
- López-Ráez JA, Bouwmeester H (2008) Fine-tuning regulation of strigolactone biosynthesis under phosphate starvation. Plant Signal Behav 3:963–965
- López-Ráez JA, Charnikhova T, Gómez-Roldán V, Matusova R, Kohlen W, De Vos R, Verstappen F, Puech-Pages V, Bécard G, Mulder P, Bouwmeester H (2008) Tomato strigolactones are derived from carotenoids and their biosynthesis is promoted by phosphate starvation. New Phytol 178:863–874

- Lorenzo O, Solano R (2005) Molecular players regulating the jasmonates signaling network. Curr Opin Plant Biol 8:532–540
- MacMillan J (2001) Occurrence of gibberellins in vascular plants, fungi, and bacteria. J Plant Growth Regul 20:387–442
- McClellan CA, Chang C (2008) The role of protein turnover in ethylene biosynthesis and response. Plant Sci 175:24–31
- McGraw BA (1987) Cytokinin biosynthesis and metabolism. In: Davies PJ (ed) Plant hormones and their role in plant growth and development. Martinus Nijhoff, Dordrecht, The Netherlands, pp 76–93
- Mekhedov SL, Kende H (1996) Submergence enhances expression of a gene encoding 1-aminocyclopropane-1-carboxylate oxidase in deep water rice. Plant Cell Physiol 37:531–537
- Miller G, Suzuki N, Ciftci-Yilmaz S, Mittler R (2010) Reactive oxygen species homeostasis and signalling during drought and salinity stresses. Plant Cell Environ 33:453–467
- Miransari M (2011) Interactions between arbuscular mycorrhizal fungi and soil bacteria. Appl Microbiol Biotechnol 89:917–930
- Miransari M, Smith DL (2007) Overcoming the stressful effects of salinity and acidity on soybean [*Glycine max* (L.) Merr.] nodulation and yields using signal molecule genistein under field conditions. J Plant Nutr 30:1967–1992
- Miransari M, Smith DL (2008) Using signal molecule genistein to alleviate the stress of suboptimal root zone temperature on soybean-Bradyrhizobium symbiosis under different soil textures. J Plant Interact 3:287–295
- Miransari M, Smith D (2009a) Alleviating salt stress on soybean (*Glycine max* (L.) Merr.) - *Bradyrhizobium japonicum* symbiosis, using signal molecule genistein. Eur J Soil Biol 45:146–152
- Miransari M, Smith DL (2009b) Rhizobial lipo-chitooligosaccharides and gebberelins enhance barley (Hoedum vulgare L.) seed germination. Biotechnology 8:270–275
- Miransari M, Bahrami HA, Rejali F, Malakouti MJ, Torabi H (2007) Using arbuscular mycorrhiza to reduce the stressful effects of soil compaction on corn (*Zea mays* L.) growth. Soil Biol Biochem 39:2014–2026
- Miransari M, Bahrami HA, Rejali F, Malakouti MJ (2008) Using arbuscular mycorrhiza to reduce the stressful effects of soil compaction on wheat (*Triticum aestivum* L.) growth. Soil Biol Biochem 40:1197–1206
- Miransari M, Rejali F, Bahrami HA, Malakouti MJ (2009a) Effects of soil compaction and arbuscular mycorrhiza on corn (*Zea mays* L.) nutrient uptake. Soil Till Res 103:282–290
- Miransari M, Rejali F, Bahrami HA, Malakouti MJ (2009b) Effects of arbuscular mycorrhiza, soil sterilization, and soil compaction on wheat (*Triticum aestivum* L.) nutrients uptake. Soil Till Res 104:48–55
- Mount SM, Chang C (2002) Evidence for a plastid origin of plant ethylene receptor genes. Plant Physiol 130:10–14
- Muller B, Sheen J (2007) Advances in cytokinin signaling. Science 318:68–69

- Murray JD, Karas BJ, Sato S, Tabata S, Amyot L, Szczyglowski K (2007) A cytokinin perception mutant colonized by *Rhizobium* in the absence of nodule organogenesis. Science 315:101–104
- Mussig C, Biesgen C, Lisso J, Uwer U, Weiler EW, Altmann T (2000) A novel stressinducible 12-oxophytodienoate reductase from Arabidopsis thaliana provides a potential link between brassinosteroid-action and jasmonicacid synthesis. J Plant Physiol 157:143–152
- Nakagawa T, Kawaguchi M (2006) Shoot-applied MeJA suppresses root nodulation in *Lotus japonicus*. Plant Cell Physiol 47:176–180
- Nakamura A, Nakajima N, Goda H, Shimada Y, Hayashi K, Nozaki H, Asami T, Yoshida S, Fujioka S (2006) *Arabidopsis Aux/IAA* genes are involved in brassinosteroid-mediated growth responses in a manner dependent on organ type. Plant J 45:193–205
- Naqvi SSM (1995) Plant/crop hormones under stressful conditions. In: Pessarakli M (ed) Handbook of plant and crop physiology. Marcel Dekker Inc, New York, pp 645–660
- Navarro L, Bari R, Achard P, Lison P, Nemri A, Harberd NP, Jones JD (2008) DELLAs control plant immune responses by modulating the balance of jasmonic acid and salicylic acid signaling. Curr Biol 18:650–655
- Ndamukong I, Abdallat AA, Thurow C, Fode B, Zander M, Weigel R, Gatz C (2007) SA-inducible Arabidopsis glutaredoxin interacts with TGA factors and suppresses JA-responsive PDF1.2 transcription. Plant J 50:128–139
- Neill SJ, Desikan R, Hancock JT (2003) Nitric oxide signalling in plants. New Phytol 159:11–35
- Nunez M, Mazzafera P, Mazorra LM, Siqueira WJ, Zullo MAT (2003) Influence of a brassinsteroid analogue on antioxidant enzymes in rice grown in culture medium with NaCl. Biol Plant 47:67–70
- Okamoto M, Tatematsu K, Matsui A et al (2010) Genomewide analysis of endogenous abscisic acid-mediated transcription in dry and imbibed seeds of arabidopsis using tiling arrays. Plant J 62:39–51
- Oldroyd GE, Engstrom EM, Long SR (2001) Ethylene inhibits the Nod factor signal transduction pathway of *Medicago truncatula*. Plant Cell 13:1835–1849
- Rivero RM, Kojima M, Gepstein A, Sakakibara H, Mittler R, Gepstein S, Blumwald E (2007) Delayed leaf senescence induces extreme drought tolerance in a flowering plant. Proc Natl Acad Sci USA 104:19631–19636
- Rivero RM, Shulaev V, Blumwald E (2009) Cytokinindependent photorespiration and the protection of photosynthesis during water deficit. Plant Physiol 150:1530–1540
- Rolland F, Baena-Gonzalez E, Sheen J (2006) Sugar sensing and signaling in plants: conserved and novel mechanisms. Ann Rev Plant Biol 57:675–709
- Sajedi NA, Ardakani MR, Rejali F, Mohabbati F, Miransari M (2011) Yield and yield components of hybrid corn (*Zea mays* L.) as affected by mycorrhizal symbiosis and zinc sulfate under drought stress. Physiol Mol Biol Plants 16:343–351

- Sasse JM (2003) Physiological actions of brassinosteroids: an update. J Plant Growth Regul 22:276–288
- Schaller A, Stintzi A (2009) Enzymes in jasmoante biosynthesis- Structure, function, regulation. Phytochemistry 70:1532–1538
- Schaller F, Biesgen C, Mussig C, Altmann T, Weiler EW (2000) 12-Oxophytodienoate reductase 3 (OPR 3) is the isoenzyme involved in jasmonate biosynthesis. Planta 210:979–984
- Schilling G, Schiller C, Otto S (1991) Influence of brassinosteroids on organ relations and enzyme activities of sugar-beet plants. In: Cutler HG, Yokota T, Adam G (eds) Brassinosteroids: Chemistry, Bioactivity and Applications. American Chemical Society, Washington, DC, pp 208–219
- Schilmiller AL, Koo AJ, Howe GA (2007) Functional diversification of acyl-coenzyme. A oxidases in jasmonic acid biosynthesis and action. Plant Physiol 143:812–824
- Schwartz MA, Baron V (1999) Interactions between mitogenic stimuli, or, a thousand and one connections. Curr Opin Cell Biol 11:197–202
- Smirnoff N (1995) Environment and plant metabolism: flexibility and acclimation. BIOS Scientific Publishers Ltd, Oxford
- Smith SE, Read DJ (2008) Mycorrhizal symbiosis. Academic, London
- Sponsel V, Hedden P (2004) Gibberellin biosynthesis and inactivation. In: Davies P (ed) Plant Hormones: Biosynthesis, Signal Transduction, Action. Kluwer Acad Pub, Dordrecht, the Netherlands, pp 63–94
- Staswick PE, Tiryaki I, Rowe ML (2002) Jasmonate response locus JAR1 and several related Arabidopsis genes encode enzymes of the firefly luciferase superfamily that show activity on jasmonic, salicylic, and indole-3-acetic acids in an assay for adenylation. Plant Cell 14:1405–1415
- Staswick PE, Serban B, Rowe M, Tiryaki I, Maldonado MT, Maldonado MC, Suza W (2005) Characterization of an Arabidopsis enzyme family that conjugates amino acids to indole-3-acetic acid. Plant Cell 17:616–627
- Stepanova AN, Alonso JM (2005) Ethylene signaling and response pathway: unique signaling cascade with a multitude of inputs and outputs. Physiol Plant 123:195–206
- Sun J, Cardoza V, Mitchell DM, Bright L, Oldroyd G, Harris JM (2006) Cross talk between jasmonic acid, ethylene and Nod factor signaling allows integration of diverse inputs for regulation of nodulation. Plant J 46:961–970
- Tanaka N, Matsuoka M et al (2006) gid1, a gibberellininsensitive dwarf mutant, shows altered regulation of probenazole-inducible protein (PBZ1) in response to cold stress and pathogen attack. Plant Cell Environ 29:619–631
- Tiwari SB, Hagen G, Guilfoyle TJ (2004) Aux/IAA proteins contain a potent transcriptional repression domain. Plant Cell 16:533–543

- Tomasz P, Jiri F (2006) Auxin signaling. J Cell Sci 119:1199–1202
- Tran L, Shinozaki K, Yamaguchi-Shinozaki K (2009) Role of cytokinin responsive two-component system in ANA and osmotic stress signaling. Plant Signal Behav 5:148–151
- Tripathi SK, Tuteja N (2007) Integrated signaling in flower senescence: an overview. Plant Signal Behav 2:437–445
- Truman W, Bennett MH, Kubigsteltig I, Turnbull C, Grant M (2007) Arabidopsis systemic immunity uses conserved defense signaling pathways and is mediated by jasmonates. Proc Natl Acad Sci USA 104:1075–1080
- Tuteja N (2007) Abscisic acid and abiotic stress signalling. Plant Signal Behav 2:135–138
- Tuteja N, Sopory S (2008) Chemical signaling under abiotic stress environment in plants. Plant Signal Behav 3:525–536
- Ueguchi-Tanaka M, Ashikari M et al (2005) Gibberellin Insensitive Dwarf1 encodes a soluble receptor for gibberellin. Nature 437:693–698
- Upreti KK, Murti GSR (2004) Effects of brassinosteroids on growth, nodulation, phytohormone content and nitrogenase activity in French bean under water stress. Biol Plant 48:407–411
- Van der Knaap E, Song WY, Ruan DL, Sauter M, Ronald PC, Kende H (1999) Expression of a gibberellininduced leucine-rich repeat receptor-like protein kinase in deepwater rice and its interaction with kinase-associated protein phosphatase. Plant Physiol 120:559–569
- Van der Knaap E, Kim JH, Kende H (2000) A novel gibberellin-induced gene from rice and its potential regulatory role in stem growth. Plant Physiol 122: 695–704
- Van Noorden GE, Ross JJ, Reid JB, Rolfe BG, Mathesius U (2006) Defective long-distance auxin transport regulation in the *Medicago truncatula* super numeric nodules mutant. Plant Physiol 140:1494–1506
- Vardhini BV, Rao SSR (2003) Amelioration of osmotic stress by brassinosteroids on seed germination and seedling growth of three varieties of sorghum. J Plant Growth Regul 41:25–31
- Vriezen WH, Hulzink R, Mariani C, Voesenek L (1999) 1-aminocyclopropane-1-carboxylate oxidase activity limits ethylene biosynthesis in *Rumex palustris* during submergence. Plant Physiol 121:189–195
- Vyroubalova S, Vaclavikova K, Tureckova V, Novak O, Smehilova M, Hluska T, Ohnoutkova L, Frebort I, Galuszka P (2009) Characterization of new maize

genes putatively involved in cytokinin metabolism and their expression during osmotic stress in relation to cytokinin levels. Plant Physiol 151:433–447

- Wan X, Li L (2006) Regulation of ABA level and waterstress tolerance of Arabidopsis by ectopic expression of a peanut 9-cisepoxycarotenoid dioxygenase gene. Biochem Biophys Res Commun 347:1030–1038
- Wang D, Pajerowska-Mukhtar K, Culler AH, Dong X (2007) Salicylic acid inhibits pathogen growth in plants through repression of the auxin signaling pathway. Curr Biol 17:1784–1790
- Wasternack C (2007) An update on biosynthesis, signal transduction and action in plant stress response, growth and development. Ann Bot 100:681–697
- Wilkinson S, Davies WJ (2002) ABA-based chemical signalling: the co-ordination of responses to stress in plants. Plant Cell Environ 25:195–210
- Woodward AW, Bartel B (2005) Auxin: regulation, action, and interaction. Ann Bot 95:707–735
- Yang J, Zhang J, Wang Z, Zhu Q, Liu L (2002) Abscisic acid and cytokinins in the root exudates and leaves and their relationship to senescence and remobilization of carbon reserves in rice subjected to water stress during grain filling. Planta 215:645–652
- Yang J, Kloepper JW, Ryu C (2008) Rhiospahre bacteria help palnts tolerate abiotic stress. Trends Plant Sci 14:1–4
- Yordanov I, Velicova V, Tsonev T (2000) Plant responses to drought acclimation and stress tolerance. Photosynthetica 38:171–186
- Yu X, Li L, Li L, Guo M, Chory J, Yin Y (2008) Modulation of brassinosteroid-regulated gene expression by jumonji domain-containing proteins ELF6 and REF6 in Arabidopsis. Proc Natl Acad Sci USA 105:7618–7623
- Zhang J, Jia W, Yang J, Ismail AM (2006) Role of ABA in integrating plant responses to drought and salt stresses. Field Crop Res 97:111–119
- Zhang JF, Yuan LJ, Shao Y, Du W, Yan DW, Lu YT (2008) The disturbance of small RNA pathways enhanced abscisic acid response and multiple stress responses in Arabidopsis. Plant Cell Environ 31:562–574
- Zhou ZY, Vriezen W, Van Caeneghem W, Van Montagu M, Van der Straeten D (2001) Rapid induction of a novel ACC synthase gene in deepwater rice seedlings upon complete submergence. Euphytica 121:137–143
- Zhu S, Gao F, Cao X et al (2005) The rice dwarf virus P2 protein interacts with ent-kaurene oxidases in vivo, leading to reduced biosynthesis of gibberellins and rice dwarf symptoms. Plant Physiol 139:1935–1945

Ethylene and Abiotic Stress Tolerance in Plants

18

Elisa Gamalero and Bernard R. Glick

Abstract

Plants are commonly exposed to large number of different environmental stresses including extremes of pH and temperature, flooding, drought, high salt, both organic and inorganic contaminants, and a variety of pathogenic organisms. As a consequence of these environmental stresses, plants typically synthesize increased levels of the phytohormone ethylene and are often unable to grow and proliferate to any great extent, at least until the stress is removed and the ethylene level is lowered. To reduce the deleterious effects of ethylene stress, plant growth-promoting bacteria (PGPB) that facilitate the proliferation of plants under stress conditions may be added to the system. These bacteria lower the level of growth inhibiting stress ethylene within the plant through the action of the enzyme ACC deaminase, and are also able to directly promote plant growth, usually by providing the plant with the phytohormone indoleacetic. The net result of adding PGPB to plants is a significant increase in both the number of seeds that germinate and the amount of biomass that the plants are able to attain under otherwise stressful and inhibitory conditions. In this chapter we provide a detailed overview regarding the functioning, the biochemistry and the regulation of ACC deaminase, with emphasis on application of PGPB synthesizing this enzyme and supporting plant growth under abiotic stress such as salinity, flooding, drought, organic, and inorganic pollution. Finally, recent developments on the exploitation of transgenic plants expressing ACC deaminase are discussed.

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_18, © Springer Science+Business Media, LLC 2012

E. Gamalero (⊠)

Dipartimento di Scienze dell'Ambiente e della Vita, Università del Piemonte Orientale, Viale Teresa Michel 11, Alessandria 15121, Italy e-mail: elisa.gamalero@mfn.unipmn.it

B.R. Glick Department of Biology University of Waterloo N2L 3G1, Waterloo, ON, Canada

Keywords

Ethylene • Plant growth-promoting bacteria • Salt • Flood • Drought • Metal stress • Transgenic plants

1 Introduction

Plants are characterized by a high level of physiological plasticity, which ensures a wide range of protective responses and enables them to survive a variety of different environmental stresses (Lusk et al. 2008; Schmidt 2008; Potters et al. 2009). The exposure of plants to mild chronic stress causes stress-induced morphogenic responses (SIMRs), with auxin, reactive oxygen species (ROS), and ethylene playing the role of response mediators. The visible consequences of the SIMRs are the inhibition of root and/or shoot elongation and the simultaneous enhancement of root and axillary branching.

Ethylene is a gaseous phytohormone that is involved in several phases of plant growth (e.g., fruit ripening, flower senescence, leaf and petal abscission) as well as in plant responses to biotic and abiotic stresses (Abeles et al. 1992). In fact, the term "stress ethylene" (Abeles 1973) indicates the increase in ethylene synthesis associated with environmental stresses such as extremes of temperature, high light, flooding, drought, heavy metal and organic pollution, radiation, wounding, insect predation, high salt, and various pathogens including viruses, bacteria, and fungi (Morgan and Drew 1997).

In higher plants, the synthesis of ethylene involves three enzymes: (1)*S*-adenosyl-I-methionine (SAM) synthetase, which catalyzes the conversion of methionine to SAM (Giovanelli et al. 1980), (2) 1-aminocyclopropane-1-carboxylic acid (ACC) synthase, which mediates the hydrolysis of SAM to ACC and 5'-methylthioadenosine (Yang and Hoffman 1984; Kende 1989), and (3) ACC oxidase which catalyzes the conversion of ACC to ethylene, carbon dioxide, and cyanide (John 1991).

Ethylene synthesized in response to environmental stresses can either exacerbate the symptoms of stress or it can lead to responses that enhance plant survival under adverse conditions. Thus, alleviation or exacerbation of the stress-induced effects of the stress depends on the plant species, its age and the nature of the stress (VanLoon and Glick 2004). This behavior has been explained by the two-phase model proposed by Glick et al. (2007a). A few hours after the onset of a stress, an initial small peak of ethylene occurs. This beneficial peak of ethylene initiates a protective response by the plant, such as transcription of pathogenesis-related genes and induction of acquired resistance (Ciardi et al. 2000; VanLoon and Glick 2004). Then, if the stress becomes chronic or more intense, 1-3 days later, the ACC synthase genes are transcribed; as a consequence of the synthesis of additional ACC, a second much larger ethylene peak is observed. This second ethylene peak is considered to be deleterious to the plant in that it induces processes such as senescence, chlorosis, and abscission that may lead to a significant inhibition of plant growth and survival.

2 Plant Growth-Promoting Bacteria

Due to the release of photosynthate by the plant roots, the rhizosphere (the zone of soil surrounding the roots of plants) is a microhabitat relatively rich in nutrients which favors the establishment of complex multitrophic interactions among the plant, the microorganisms, and the soil itself (Lynch and Whipps 1990). Consequently, the rhizosphere is considered as a "hot spot" that attracts large and active microbial populations able to affect plant growth by exerting beneficial, detrimental, or no effects. "Plant growth-promoting rhizobacteria" (PGPR) (Kloepper and Schroth 1978) are soil bacteria able to efficiently colonize plant roots after their inoculation onto seeds. Once the colonization of the roots by PGPR is well established plant biomass by direct or indirect stimulation of plant health are frequently achieved (Cook 2002). In order to discuss the modes of action of a wide range of plant beneficial bacteria, the term "plant growth-promoting bacteria" (PGPB) was proposed (Bashan and Holguin 1998). In agreement with this proposal, in this chapter we will use the term PGPB instead of PGPR.

Enhancement of plant development by PGPB has been demonstrated on a wide range of different crops under both controlled conditions and in the field, and a large number of papers have been published on this topic (Reed and Glick 2004). As an initial requirement for effective use of PGPB, efficient root colonization by the bacteria is required, in particular in open field conditions, where competition with autochtonous microorganisms is high (Gamalero et al. 2004, 2005; Pivato et al. 2008).

2.1 Mechanisms Used by PGPB

PGPB can promote plant growth in two different ways: either indirectly by plant disease suppression or through direct stimulation (Glick 1995). While plant disease suppression leads to improvement of plant health, the direct effects of many soil bacteria on plants is mediated by a range of mechanisms including improvement of mineral nutrition (P, N, Fe), enhancement of plant tolerance to biotic and abiotic stresses, and modification of root development (Glick 1995; Glick et al. 1999; Kloepper et al. 1989; Gamalero et al. 2002, 2004). The bacterial traits involved in these activities include nitrogen fixation, phosphate solubilization, iron sequestration, synthesis of phytohormones, and modulation of plant ethylene levels (Gamalero and Glick in press).

2.2 ACC Deaminase

In 1978, an enzyme capable of degrading the ethylene precursor, ACC, to ammonia and α -ketobutyrate was isolated from the bacterium *Pseudomonas* sp. strain ACP (Honma and Shimomura 1978). Since then, numerous studies have demonstrated ACC deaminase activity in a wide range of soil microorganisms including the fungus Penicillium citrinum (Honma 1993) and various bacteria (Jacobson et al. 1994; Glick et al. 1995; Burd et al. 1998; Belimov et al. 2001; Ghosh et al. 2003; Ma et al. 2003a; Sessitsch et al. 2005; Blaha et al. 2006; Madhaiyan et al. 2007; Kuffner et al. 2008; Chinnadurai et al. 2009). Bacteria that have ACC deaminase facilitate plant growth under different ethylene-producing environmental stresses, including flooding (Grichko and Glick 2001a, b; Farwell et al. 2007), pollution caused by organic toxicants such as polycyclicaromatic hydrocarbons (PAHs), polycyclic biphenyls (PCBs) and total petroleum hydrocarbons (TPHs) (Huang et al. 2004; Saleh et al. 2004; Reed and Glick 2005) and by heavy metals including nickel, lead, zinc, copper, cadmium, cobalt and arsenic (Burd et al. 1998, 2000; Belimov et al. 2001, 2005; Nie et al. 2002; Glick 2003; Reed and Glick 2005; Farwell et al. 2006; Rodriguez et al. 2008; Ma et al. 2009), salinity (Mayak et al. 2004b; Saravanakumar and Samiyappan 2006; Cheng et al. 2007; Gamalero et al. 2010), drought (Mayak et al. 2004a; Arshad et al. 2008; Belimov et al. 2009), and phytopathogen attack (Wang et al. 2000; Hao et al. 2007; Toklikishvili et al. 2010).

In addition to the above-mentioned stresses, other natural processes such as nodulation of legumes and mycorrhizal establishment in the host plant induce local increases in ethylene content which has the potential to decrease various aspects of plant growth and development. In this context, ACC deaminase-producing bacteria, lowering the ethylene content in the plants, can increase nodulation in legumes such as pea and alfalfa (Ma et al. 2003a, 2004) and mycorrhizal colonization in cucumber (Gamalero et al. 2008).

2.2.1 Model

In a model that was previously reported by Glick et al. (1998), PGPB that synthesize ACC deaminase begin seed or rootlet colonization following the release of tryptophan and other small molecules by seed or root exudates (Bayliss et al. 1997; Penrose and Glick 2001). Thereafter, the bacteria synthesize and secrete IAA (Patten and Glick 1996, 2002) that, together with the IAA synthesized by the plant, can either stimulate plant growth or induce the transcription of the gene encoding the plant enzyme ACC synthase, which converts SAM to ACC. A portion of the ACC that is produced by this reaction has been shown to be exuded from seeds or plant roots, taken up by the bacteria, and subsequently converted by the enzyme ACC deaminase to ammonia and α -ketobutyrate (Bayliss et al. 1997; Penrose and Glick 2001). As a result of this activity, the level of ethylene produced by the plant is lowered and does not reach levels which are inhibitory for plant growth. In addition, ethylene inhibition of IAA signal tranduction is decreased thereby facilitating IAA stimulation of plant growth (Glick et al. 2007a). The main short-term visible effect of seed inoculation with ACC deaminaseproducing bacteria, under gnotobiotic conditions, is the enhancement of root elongation (Glick et al. 1995; Hall et al. 1996; Shah et al. 1997).

2.2.2 Protein Biochemistry

ACC deaminase is a multimeric enzyme (either a dimer or a trimer) that utilizes the coenzyme pyridoxal phosphate as a tightly bound cofactor, with an approximate subunit mass of 35,000-42,000 and a native size estimated to be 100-112 kDa (Jacobson et al. 1994; Hontzeas et al. 2004; Karthikeyan et al. 2004; Klee et al. 1991; Sheehy et al. 1991). In all instances examined, the K_m indicated that the enzyme did not have a particularly high affinity for ACC with Km values typically around 1.5-6.0 mM. While most organisms with ACC deaminase contain a basal level of enzyme activity, ACC is induced by ACC, at levels as low as 100 nM (Jacobson et al. 1994) with full induction requiring up to 10 h. The amino acids 1-alanine, d1-alanine and d1-valine can also induce enzyme activity to a small extent, and α -aminoisobutyric acid can induce activity to almost the same level as ACC (Honma 1983). ACC deaminase activity is optimum at a temperature of 30°C and at pH 8.5 (Hontzeas et al. 2006). The affinity of the substrate ACC and the competitive inhibitors 1-alanine and 1-serine for the enzyme is also highest at pH 8.5.

2.2.3 Occurrence in Nature

Bacterial ACC deaminase activity is relatively common. Genes coding for ACC deaminase have been found in gram-negative (Wang et al. 2000; Belimov et al. 2001; Hontzeas et al. 2005) and -positive (Belimov et al. 2001; Ghosh et al. 2003; Raddadi et al. 2008) bacteria, rhizobia (Ma et al. 2003a, b; Shaharoona et al. 2006; Duan et al. 2009), methylobacteria (Chinnadurai et al. 2009), endophytes (Sessitsch et al. 2005; Rothballer et al. 2008), and fungi (Jia et al. 1999; Viterbo et al. 2010). In one study, 12% of isolated Rhizobium spp. from various sites in southern and central Saskatchewan possessed this enzyme (Duan et al. 2009). In another study, ACC deaminase activity/genes were found in a wide range of bacterial isolates including Azospirillum, Rhizobium, Agrobacterium, Achromobacter, Burkholderia, Ralstonia, Pseudomonas, and Enterobacter (Blaha et al. 2006).

Although, based on sequence similarity, many organisms have putative ACC deaminase genes (Kaneko et al. 2010; Taghavi et al. 2010), sequence homology by itself does not suffice to define these open reading frames as ACC deaminase encoding sequences (Glick 2005) since other pyridoxal phosphate enzymes have sequences similar to ACC deaminase (Todorovic and Glick 2008).

2.2.4 Regulation

Analysis of DNA sequence data for the region upstream of the ACC deaminase structural gene (i.e., acdS) from the PGPB Pseudomonas putida UW4 indicated that this DNA segment contained a CRP (cyclic AMP receptor protein) binding site, an FNR (fumarate-nitrate reduction regulatory protein) binding site (which is a known anaerobic transcriptional regulator), an Lrp (leucine-responsive regulatory protein) binding site, and a gene encoding an Lrp protein (Grichko and Glick 2000; Li and Glick 2001). All of these features were previously shown to be components of the transcriptional regulatory mechanism of the ACC deaminase gene from this bacterium, with the *acdS* promoter under the transcriptional control of the regulatory Lrp protein (encoded by acdR). Subsequent studies have indicated that in a large number of bacteria the *acdS* gene is under the transcriptional control of *acdR* (Ma et al. 2003b; Blaha et al. 2006; Prigent-Combaret et al. 2008; Duan et al. 2009; Sun et al. 2009). Moreover, in strain P. putida UW4, a protein that specifically interacts with ACC, the Lrp protein and the region of DNA upstream of acdS has been identified and characterized (Cheng et al. 2008; Glick et al. 2007b). While transcriptional regulation of acdS by acdR is a key feature of the expression of many bacterial ACC deaminases, in some instances acdS is not under the transcriptional control of *acdR*. For example, an acdS gene from a strain of *Mesorhizobium loti* has been shown to be transcriptionally regulated by the *nifA* promoter (Kaneko et al. 2000), and to be expressed exclusively within legume nodules (Uchiumi et al. 2004). It is possible that the expression of ACC deaminase genes within nitrogen-fixing nodules might decrease the rate of nodule senescence - as nitrogen fixation with its high energy demand could activate stress ethylene synthesis - and thereby effectively increase the amount of fixed nitrogen.

3 Decreasing Plant Stress with PGPB

3.1 Salt

Salinity is an enormous worldwide problem for agriculture, especially for crops that are grown under irrigation. The amount of salt-affected land worldwide is >900 million ha which is approximately 6% of the total global land mass, or about 20% of the world's cultivated area. Moreover, it has been estimated that around half of the land devoted to the growth of irrigated crops is adversely affected by salinity (Flowers 2004).

Plants exposed to excessive salinity are inhibited in their growth and development: seed germination, seedling growth and vigor, flowering and fruit set are negatively affected by salt stress which induces osmotic stress, Na⁺ and Cl⁻ toxicity, plasmolysis, nutrient imbalance, production of reactive oxygen species, and interference with photosynthesis, in a wide range of plants. Notwithstanding other effects, salinity-induced stress in plants is partly ascribed to ethylene production (O'Donnell et al. 1996; Cuartero and Fernandez-Munoz 1999; Blumwald 2000; Mayak et al. 2004a, b; Shibli et al. 2007). For instance, ethylene synthesis was enhanced from two- to about tenfold in tomato (Lycopersicon esculen*tum*) and *Arabidopsis* plants exposed to salinity stress (Richard and El-Abd 1989; Hall and Smith 1995). In addition, in chickpea (Cicer arietinum), salinity induced increases of ethylene, ACC content, and in the activity of the enzyme ACC oxidase (Kukreja et al. 2005). One of the major mechanisms by which PGPB facilitate plant growth under stressful conditions is the reduction of ethylene levels through the enzyme ACC deaminase (Glick 1995, 2004; Glick et al. 1998, 2007a, b).

PGPB isolated from soil samples derived from dried river beds in the arid and salty Arava region in the southern part of the Negev desert in Israel were selected to assess their ability to produce ACC deaminase and, on the basis of preliminary testing, one bacterium, Achrobacter piechaudii ARV8, was further characterized (Mayak et al. 2004a). Tomato plants inoculated with this strain and exposed to salt concentrations up to 200 mM showed increased root and shoot biomass, this effect being related to the reduction of ethylene levels in plants. This pioneering work served as the basis for a spate of experimental work on several different plants by scientists from all over the world. Since then, labs from India, Pakistan, China, Italy and Canada have reported successfully employing ACC deaminase-expressing bacteria to promote plant growth in presence of high concentrations of salt (Saravanakumar and Samiyappan 2006; Cheng et al. 2007; Nadeem et al. 2007; Yue et al. 2007). These studies have included groundnut (Saravanakumar and Samiyappan 2006), maize (Nadeem et al. 2007), cotton (Yue et al. 2007), canola (Cheng et al. 2007), wheat (Nadeem et al. 2010), and cucumber (Gamalero et al. 2010) as experimental systems. A very important point is that the efficacy of the ACC deaminase-producing strains in protecting plants against growth inhibition induced by salinity is evident in the field as well as in a laboratory or greenhouse setting (Saravanakumar and Samiyappan 2006). In the experiments reported by Cheng et al. (2007) and Gamalero et al. (2010), only wild-type *Pseudomonas putida* UW4 and not an ACC deaminase minus mutant of this bacterium protected canola and cucumber plants, respectively, against growth inhibition by salt. These results provide a clear demonstration of the role of ethylene in growth inhibition by high salt.

3.2 Flooding

The main consequence of flooding is oxygen deprivation to the plant roots (Kozlowski 1984; Jackson 1985) which respond with reduced root permeability, water absorption and mineral uptake; closure of stomata followed by a decrease in photosynthesis, alterations in hormone balance, inhibition of stem and root growth, hypertrophy of lenticels, development of aerenchyma and adventitious roots, epinasty, chlorosis, leaf abscission and premature fruit drop (Vartapetian and Jackson 1997). It has been calculated that under normal conditions, about 230 mmol m⁻³ of oxygen is dissolved in water, and when the oxygen level is reduced to 50 mmol m⁻³, hypoxia occurs (Nilsen and Orcutt 1997; Vartapetian and Jackson 1997). In this condition, the remaining available oxygen is quickly consumed by soil microorganisms leading to a worsening of the stress (Frankenberger and Arshad 1995). In addition, hypoxia induces transcription of ACC synthase genes in flooded roots (Drew 1997); however, the oxidation of ACC to ethylene (by ACC oxidase) is thwarted by the lack of oxygen (Abeles et al. 1992; Shiu et al. 1998). Therefore, some of the ACC synthesized by the plant is exuded into the surrounding soil and becomes available to the soil bacteria expressing ACC deaminase (Else et al. 1995) that in turn could support the growth of the plants exposed to flooding. In plants whose roots are not associated with bacteria that produce ACC deaminase, the excess ACC moves up from the plant roots to the shoots where more oxygen is available and ethylene synthesis is thereby enabled. This idea has been demonstrated by the experiments performed by Grichko and Glick (2001a). Tomato plants inoculated with bacteria able to produce ACC deaminase and subjected to 9 days of flooding were taller, healthier, and greener (due to increased levels of chlorophyll) compared to noninoculated ones. In addition, an increased development of adventitious roots and stem aerenchyma was observed in flooded plants treated with ACC deaminase-expressing bacteria compared to flooded plants inoculated with bacteria which did not contain this enzyme or plants that were flooded but not inoculated with bacteria.

3.3 Drought

Soil aridity is one of the greatest limitations being faced by present-day agriculture leading to serious reduction of crop yield. When exposed to drought, plant roots sense the soil drying condition. In order to maintain cellular water status, the roots send chemical signals, including abscisic acid (ABA), via the xylem to the shoots to restrict water use. In addition, drought stress has been extensively associated with other longdistance signals such as ACC. As a consequence of this raised level of ACC, high amount of endogenous ethylene is synthesized by the plant (Graves and Gladon 1985; Kalantari et al. 2000; Mayak et al. 2004a; Sobeih et al. 2004) leading to growth inhibition, premature senescence and abscission, and reduction of crop yield (Davies and Zhang 1991; Dodd 2005). Therefore, together with the promotion of root growth which is beneficial for improving plant water uptake (Reynolds and Tuberosa 2008), the lowering of the ethylene levels could be a strategy to improve the development and health of plants grown under drought stress.

The first demonstration of the efficacy of ACC deaminase-expressing bacteria in supporting the growth of plants cultivated in water scarcity came from Mayak et al. (2004a). In this work, tomato and pepper plant tolerance to water deficit was conferred by the bacterial strain *Achromobacter piechaudi* ARV8, able to synthesize ACC deaminase, resulting in significant biomass increases

when plants were rewatered following several days of drought. Although the relative water contents were unaffected by seed inoculation with the bacterial strain, ethylene synthesis was reduced in inoculated plants compared to controls, with a simultaneous dramatic improved recovery from water deficiency (Mayak et al. 2004a).

In a subsequent study, two pseudomonads, showing ACC deaminase activity, significantly decreased the "drought stress-imposed effects" on the growth and yield of peas (Pisum sativum L.) subjected to drought stress (Arshad et al. 2008). Exposure of plants to drought stress at the vegetative growth stage reduced shoot biomass by 41% in uninoculated plants, and by only 18% in inoculated plants. While plants exposed to drought stress at the flowering and pod formation stages showed a lower grain yield, bacterial inoculation resulted in better grain yields (up to 62% and 40% higher at these two stages, respectively) than the uninoculated ones. Finally, delay of pod ripening observed in inoculated plants was associated with reduction of the endogenous ethylene concentration (Arshad et al. 2008).

A conclusive demonstration of the involvement of ACC deaminase in supporting plant growth under drought stress comes from the recent work of Belimov et al. (2009). Inoculation with Variovorax paradoxus 5C-2, but not with a transposon mutant of this bacterium with decreased ACC deaminase activity, improved growth, yield and water-use efficiency of droughtstressed peas. At a local level (within the root), the bacterial strain enhanced nodulation by symbiotic nitrogen-fixing bacteria, thereby preventing a drought-induced decrease in nodulation and seed nitrogen content. Systemic (within the shoot) inoculation with strain V. paradoxus 5C-2 induced an increased ABA concentration in the xylem, and a reduction of the xylem ACC concentration. The exploitation of ACC deaminaseexpressing bacteria combined with deficit irrigation may be an effective way of more efficiently using water in agriculture while reducing the yield loss induced by soil drying (Belimov et al. 2009).

3.4 Metals

Plants exposed to toxic levels of heavy metals show several macroscopic consequences such as reduced growth (of both roots and above ground parts), leaf chlorosis and necrosis, loss of turgor, a decreased rate of seed germination, cell and plant death (Bingham et al. 1986; Foy et al. 1978). These effects are typically ascribed to ultrastructural, biochemical, and molecular changes in plant tissues and cells caused by the presence of heavy metals. At a biochemical level, the protein profile of plants exposed to heavy metals is modified as a consequence of the reestablishment of the unstressed plant's cellular and redox homeostasis and is an indication of the plant's adaptation to chronic stress (Bona et al. 2007). Other plant reactions are the synthesis of polyamines, accumulation of reactive oxygen species (Mithöfer et al. 2004; Rodrìguez-Serrano et al. 2006), changes in plant nutrient levels (Sandalio et al. 2001) and in water status (Perfus-Barbeoch et al. 2002), a decrease in plasma membrane H⁺-ATPase activity, and a reduction in photosynthesis (Krupa 1988; Hsu and Kao 2004; Bačkor et al. 2007). In addition, independent of the varying amounts of heavy metal mobilization and accumulation, the presence of heavy metals raises plant ethylene levels which lead to subsequent inhibition of root elongation and a stress/senescence response (Deikman 1997). For example, ethylene can: speed up senescence of plants exposed to high concentrations of Cu for a prolonged period of time (Maksymiec et al. 1995; Maksymiec and Baszyński 1996), inhibit cell growth, and increase cell wall rigidity through deposition of lignin (Enyedi et al. 1992).

Despite the fact that heavy metals adversely affect soil microflora (Bååth 1989; Van Beelen and Doelman 1997; Giller et al. 1998; Regvar et al. 2001; Rajapaksha et al. 2004), some beneficial microorganisms typically found living on plant roots can relieve some of the metal toxicity to plants. This may occur as a result of the microorganisms modifying the metal bioavailability (via volatilization, or the sequestering/accumulation of the metal), through the release of chelators, acidification, or redox changes (Abou-Shanab et al. 2008), or by increasing the tolerance to the presence of the metals through the synthesis of the enzyme ACC deaminase (Glick 1995, 2004; Glick et al. 1998, 2007a, b). Several studies have demonstrated the involvement of this bacterial activity in improving the survival and health of plants grown in heavy metal-polluted soils through reduction of the endogenous ethylene levels.

Kluyvera ascorbata SUD165 is a nickel-resistant siderophore-producing bacterial strain that can grow at cold temperatures (5-10°C) and synthesize ACC deaminase (Burd et al. 1998). This organism was originally isolated from nickelcontaminated soil from northern Ontario, Canada. It promoted the growth of Indian mustard, tomato, and canola in the presence of high concentrations of nickel (and other metals) under a range of different experimental conditions (Burd et al. 1998, 2000; Ma et al. 2001) by increasing plant tolerance to the metal through an observed three- to fourfold reduction of the ethylene level. Moreover, the siderophores produced by this bacterium helped the plants to accumulate sufficient iron so that they were no longer chlorotic (and ironstressed, which also can cause the synthesis of stress ethylene).

Several cadmium-tolerant bacterial strains of *Pseudomonas*, *Alcaligenes*, *Variovorax*, *Bacillus*, and *Rhodococcus* spp. that express the enzyme ACC deaminase have been isolated from the rhizoplane of pea and Indian mustard plant that were cultivated in the presence of heavy metal-polluted soil or sewage sludge (Belimov et al. 2001). These strains stimulated the root elongation of canola and Indian mustard cultivated in presence of 300 μ M CdCl₂. In addition, a positive correlation was observed between the measured in vitro ACC deaminase activity and the effect of the added bacteria on plant root elongation (Belimov et al. 2005).

In addition to stimulation of the growth of plants cultivated on heavy metal-polluted soils, in some instances ACC deaminase-expressing bacteria can improve phytoremediation efficacy. In that situation, the plant can accumulate higher levels of heavy metals and/or there may be an enhanced mobilization of metals in the soil due to an increase in the exudation of a variety of organic compounds by plant roots. Recently, the ability of Pseudomonas tolaasii strain ACC23 to increase the biomass of canola plants in the presence of Cd stress (83% for roots and 94% for shoots) and the cadmium uptake per plant (107%), compared to uninoculated plants, was demonstrated (Dell'Amico et al. 2008). Since the specific uptake of Cd into shoots and roots did not change following bacterial inoculation, strain ACC23 did not affect metal availability or mobility; however, plant biomass was increased following the addition of this strain and the total amount of Cd accumulated was also increased (Dell'Amico et al. 2008). Detailed physiological characterization of this bacterium indicated that it had the ability to produce IAA, siderophores as well as the enzyme ACC deaminase. While this study did not show direct involvement of any of these plant beneficial traits in the promotion of plant growth in the presence of cadmium stress, it is nevertheless probable that the success of this bacterial inoculant depended on one or more of these bacterial traits.

3.5 Organic Compounds

Many soil bacteria are able to degrade anthropogenic organic chemicals such as herbicides (Sorensen et al. 2008), pesticides (Wackett et al. 2002; Kim et al. 2004), refrigerants (Parnell et al. 2006; Field and Sierra-Alvarez 2008; Pieper and Seeger 2008), solvents (Coleman et al. 2002; Park et al. 2002; Lee et al. 2010), and other organic xenobiotics. As a consequence of their versatile catabolic properties, bacteria belonging to the pseudomonad group are often well equipped to carry out the biodegradation of complex organic compounds, a process that typically requires the concerted efforts of several enzymes. The genes that code for the enzymes of these biodegradative pathways are often located on large (~50-200 kb) "biodegradative" plasmids (Ghosal et al. 1985; Cork and Krueger 1991). Nonhalogenated aromatic compounds can be enzymatically converted by degradative bacteria to either catechol or protocatechuate, compounds that are readily metabolized by almost all organisms. Halogenated aromatic compounds, which are the main components of most pesticides and herbicides, may also be degraded by these plasmid-encoded enzymes, but with a much slower rate of degradation (Glick et al. 2010). Therefore, microbial degradation might provide a reasonable and effective means of disposing of toxic chemical wastes and inoculation of bacteria with biodegradative capabilities in polluted soils can facilitate the breakdown of the contaminants. While this is especially true under laboratory conditions, successful bioremediation is often more difficult to achieve when bacteria are inoculated directly in environment.

On the other hand, the use of plants to clean up contaminated soils (i.e., phytoremediation) is a viable remediation strategy for environmentally persistent organic compounds, but several limitations hamper its widespread application in the field (Salt et al. 1998; Pilon-Smits 2005; Glick 2003). Thus, plant photosynthesis, respiration, and metabolism may be impaired by organic contaminants leading to low plant biomass and, subsequently, low remediation efficacy (Medina et al. 2004; Pilon-Smits 2005). In addition, plants showing some degree of tolerance to soil contaminants are often characterized by a reduced biomass that does not allow for the efficient and timely degradation of the contaminant. One strategy to overcome these problems is rhizoremediation, that is, the combined use of plants tolerant to the organic pollutant together with biodegradative bacteria having plant growthpromoting activity (Gerhardt et al. 2009). In this case, degradation of organic compounds to nontoxic, or less-toxic, chemicals results from complex interactions involving roots, root exudates, rhizosphere soil, and microbes. Depending on the plant species, age and environmental conditions, root exudation can account for up to 40% of net fixed carbon (Lynch and Whipps 1990). On average, 15-20% of net fixed carbon is released by the roots (Nguyen 2003) into the soil as sugars, amino acids, organic acids, and larger organic compounds (Kumar et al. 2006). These compounds attract a dense and diverse microbial population which can bind efficiently to plant roots and colonize the root surface and often the interior of the plant as well. This provides various benefits to plants, including the synthesis of compounds that increase plant tolerance to stress by lowering plant stress hormone levels and the degradation of contaminants before they can negatively impact the plants (Gerhardt et al. 2009). Besides degrading polycyclic aromatic hydrocarbons, bacteria with hydrophobic external surfaces may make these relatively insoluble compounds more bioavailable (Johnsen and Karlson 2004) through the formation of biofilm on the surface of some crystal-like polycyclic aromatic hydrocarbons.

Several studies have reported the efficacy of improving phytoremediation by the combined use of plants and biodegradative bacteria for removing: petroleum products (Escalante-Espinosa et al. 2005; Huang et al. 2005; Radwan et al. 2005; Alarcón et al. 2008; Al-Awadhi et al. 2009), polycyclic and other aromatic hydrocarbons (Daane et al. 2001; Huang et al. 2004; Sheng and Gong 2006; Sheng et al. 2008; Germaine et al. 2009), and halogenated compounds (Sicilano and Germida 1997; Yee et al. 1998; Leigh et al. 2006; Liu et al. 2007; Uhlik et al. 2009) from contaminated soils. Although most of these studies have been performed under controlled laboratory, growth chamber or greenhouse conditions and some field scale phytoremediation studies have also been reported (Gurska et al. 2009; Kamath et al. 2004).

In general, plants inoculated with PGPB having degradative capabilities withstand the stress of growing in the polluted soils better than noninoculated ones. Gurska et al. (2009) performed a 3-year field test at a Southern Ontario site (~130 gkg⁻¹ of total petroleum hydrocarbons, TPH) used for land farming of refinery hydrocarbon waste for many years. The low molecular weight TPH fractions were removed through land farming and bioremediation; the high molecular weight, recalcitrant fractions remained at high levels in the soil and were remediated by combining plants (annual ryegrass, tall fescue, barley, and fall rye) and the ACC deaminase-expressing strains P. putida UW3 and P. putida UW4. Using this approach, TPH levels in soil decreased from 129.3 to 46.4 gkg⁻¹ over a 3-year period. In addition, the biomass of the plants grown in these polluted soil was enhanced by PGPB by 40%.

Although phytoremediation represents a promising alternative to other, more expensive, remedial biotechnologies, the efficacy of this system may be limited when toxic levels of contaminants hinder plant growth and depress microbial populations. Inoculation of seeds or seedlings with PGPB addresses these problems directly by increasing the plant's tolerance to a variety of xenobiotic compounds.

4 Decreasing Plant Stress with Transgenic Plants

The impact of abiotic stress on plant development represents an advantage both in agriculture and horticulture as well as in more recent developed environmental biotechnologies such as phytoremediation of contaminated soils. Although several transgene products altering ethylene synthesis have been reported, laboratory studies of transgenic plants expressing the bacterial gene encoding ACC deaminase are characterized by an enhanced tolerance to abiotic stresses such as flooding (Grichko and Glick 2001a), salinity (Mayak et al. 2004b; Sergeeva et al. 2006), and metal contamination (Stearns et al. 2005; Grichko et al. 2000; Nie et al. 2002). To our knowledge, no report regarding the growth of transgenic plants expressing ACC deaminase exposed to either drought stress or organic pollutant contamination has yet been published.

The main advantage of this approach is that with genetic manipulation (i.e., lowered level of stress ethylene), plant tolerance to a wide range of different abiotic stresses may be gained. But PGPB are typically more effective in protecting plants against the deleterious effects of various abiotic stresses than the transgenic plants expressing ACC deaminase. This is because PGPB may contribute to plant growth under stressed or natural environments by a number of different mechanisms, including the synthesis of IAA and siderophores, and not solely by reducing plant ethylene levels.

4.1 Salt

A variety of genetic engineering approaches to the improvement of salt tolerance in plants have been reported (Karuna Stree et al. 2000; Shi et al. 2000; Tabaei-Aghdaei et al. 2000; Zhang and Blumwald 2001; Zhang et al. 2001; Zhu 2002), but only one paper dealt with the salt-stress tolerance by transgenic plants expressing the bacterial gene encoding for ACC deaminase (Sergeeva et al. 2006). In this work, the transformed and nontransformed canola plants were cultivated in the presence of 0–200 mM NaCl and several parameters of plant growth and physiology were assessed.

The data are consistent with the possibility that the presence of ACC deaminase confers tolerance to transgenic canola lines under salt stress as compared to the nontransformed canola plants. This protection occurs due to the decreased synthesis of ethylene. Nevertheless, the protection given by the insertion of the ACC deaminase gene in canola plants was not significantly different from that provided by seed inoculation with PGPB able to synthesize ACC deaminase (Mayak et al. 2004b).

4.2 Flooding

Experiments involving ACC deaminase transgenic plants exposed to flooding were conducted on tomato (Grichko and Glick 2001a) and canola (Farwell et al. 2007) as model plants. Preliminary results obtained by Grichko and Glick (2001a) showed that the transgenic tomato plants expressing ACC deaminase gene showed some increased tolerance to flooding stress and were less affected by deleterious effects of root hypoxia than nontransformed plants. Recently, Farwell et al. (2007) compared the growth of transgenic canola expressing ACC deaminase to that of nontransformed canola, inoculated or not with the bacterial strain P. putida UW4, able to synthesize ACC deaminase, and simultaneously exposed to two major stresses, i.e., flooding and nickel contamination. The biomass of transgenic canola, especially when inoculated with P. putida UW4, was significantly

higher than that of nontransformed canola in metal-contaminated soil under low flood-stress conditions. However, as observed earlier, using either transgenic canola or a bacterial strain able to express ACC deaminase, a similar level of tolerance under low flood-stress conditions in the presence of nickel was observed.

4.3 Metals

The use of transgenic plants in the phytoremediation of contaminated environments may provide the plant with some significant growth and survival advantages. Transgenic plants increase the plant's capacity to tolerate, accumulate, and/or metabolize pollutants thereby resulting in the production of large biomass (Krämer and Chardonnens 2001; Pilon-Smits and Pilon 2002). Previously, transgenic tobacco, canola, and tomato plants expressing the bacterial enzyme ACC deaminase have been constructed and tested (Grichko et al. 2000; Nie et al. 2002; Stearns et al. 2005; Farwell et al. 2006). While the use of transgenic plants may appear to be advantageous in comparison with the use of plant growthpromoting bacteria, especially in soils that might inhibit or prevent the proliferation of added bacteria, in practice, when they are tested, these transgenic plants are found to perform quite similarly to plants treated with ACC deaminaseexpressing rhizobacteria.

The fitness of transgenic tomato plants expressing ACC deaminase versus nontransgenic tomato plants was compared in presence of heavy metals (Cd, Co, Cu, Mg, Ni, Pb, or Zn) and measured as metal concentration and ACC deaminase activity in both plant shoots and roots; root and shoot development; and leaf chlorophyll content (Grichko et al. 2000). The transgenic tomato plants accumulated a greater amount of metal within their tissues and, at the same time, were less subject to the deleterious effects of the metals on plant growth than the nontransgenic plants. Similarly, transgenic tobacco plants co-expressing ACC deaminase and tryptophan monooxygenase showed faster growth with larger biomass with a more developed root system, and accumulated a greater amount of Cu^{2+} and Co^{2+} than nontransformed plants (Zhang et al. 2008). Finally, inoculation of transgenic plants expressing ACC deaminase with bacteria able to synthesize this enzyme also led to better plant performance, highlighted by improved growth and higher metal accumulation. Thus, inoculation of transgenic canola plants expressing ACC deaminase in the roots, with *Pseudomonas putida* Biovar B strain HS-2, which is also able to synthesize this enzyme, induced an increase in plant biomass and nickel (Ni) uptake by shoots and roots (Rodriguez et al. 2008).

5 Conclusion and Future Perspective

Soil-based plant-bacterial interactions have likely existed for millions of years. However, it was not until the middle of the twentieth century that scientists selected and tested a wide range of bacteria that actively facilitated plant growth. Many early experiments with PGPB suffered from a lack of reproducibility, probably reflecting the lack of a detailed understanding of the key bacterial traits essential to make this relationship work effectively. However, modern biochemical and genetic techniques have allowed scientists to develop a deeper understanding of the mechanisms used by many bacteria to promote plant growth. Thus, there are a large number of reports of successful field trials of bacteria that were initially selected and tested in the laboratory to produce ACC deaminase, IAA, and siderophores. Notwithstanding the involvement of other bacterial traits, especially in specific circumstances, selection of bacteria for the three above-mentioned traits appears to be one way to short circuit the otherwise long and laborious search for effective and reproducible PGPB. In addition, while all of the evidence is not yet in, it appears that more efficacious and reproducible PGPB may be found amongst bacterial endophytes than amongst rhizospheric bacteria. This is probably a reflection of the fact that endophytes are somewhat protected from the sometimes hostile soil environment.

Finally, given the increasing understanding of the mechanisms used by PGPB that has developed over the past 20 years or so, it would appear that this is a technology whose time has come. Over the next 20 years, commercial use of PGPB is likely to increase dramatically, first in the developing world, where the cost of chemical fertilizers and pesticides is often prohibitive, and then in more developed countries with organic farmers possibly being in the vanguard.

References

- Abeles FB (1973) Ethylene in plant biology. Academic, New York, 302
- Abeles FB, Morgan PW, Saltveit ME Jr (1992) Ethylene in plant biology, 2nd edn. Academic, New York
- Abou-Shanab RA, Ghanem K, Ghanem N, Al-Kolaibe A (2008) The role of bacteria on heavy-metal extraction and uptake by plants growing on multi-metalcontaminated soils. World J Microbiol Biotechnol 24:253–262
- Alarcón A, Davies FT Jr, Autenrieth RL, Zuberer DA (2008) Arbuscular mycorrhiza and petroleum-degrading microorganisms enhance phytoremediation of petroleum-contaminated soil. Int J Phytoremediation 10:251–263
- Al-Awadhi H, El-Nemr I, Mahmoud H, Sorkhoh NA, Radwan S (2009) Plant-associated bacteria as tools for the phytoremediation of oily nitrogen-poor soils. Int J Phytoremediation 11:11–27
- Arshad M, Shaharoona B, Mahmood T (2008) Inoculation with *Pseudomonas* spp. containing ACC deaminase partially eliminates the effects of drought stress on growth, yield, and ripening of pea (*Pisum sativum* L.). Pedosphere 18:611–620
- Bååth E (1989) Effects of heavy metals in soil on microbial processes and populations. Water Air Soil Pollut 47:335–379
- Bačkor M, Vàczi P, Bartàk M, Budòvà J, Dzubaj A (2007) Uptake, photosynthetic characteristics and membrane lipid peroxidation levels in the lichen photobiont *Trebouxia erici* exposed to copper and cadmium. Bryologist 110:100–107
- Bashan Y, Holguin G (1998) Proposal for the division of plant growth-promoting rhizobacteria into two classifications: biocontrol-PGPB (plant growth-promoting bacteria) and PGPB. Soil Biol Biochem 30:1225–1228
- Bayliss C, Bent E, Culham DE, MacLellan S, Clarke AJ, Brown GL, Wood JM (1997) Bacterial genetic loci implicated in the *Pseudomonas putida* GR12-2R3canola mutualism: identification of an exudate-inducible sugar transporter. Can J Microbiol 43:809–818
- Belimov AA, Safronova VI, Sergeyeva TA, Egorova TN, Matveyeva VA, Tsyganov VE, Borisov AY, Tikhonovich

IA, Kluge C, Preisfeld A, Dietz KJ, Stepanok VV (2001) Characterization of plant growth promoting rhizobacteria isolated from polluted soils and containing 1-aminocyclopropane-1-carboxylate deaminase. Can J Microbiol 47:642–652

- Belimov AA, Hontzeas N, Safronova VI, Demchinskaya SV, Piluzza G, Bullitta S, Glick BR (2005) Cadmiumtolerant plant growth-promoting rhizobacteria associated with the roots of Indian mustard (*Brassica juncea* L. Czern.). Soil Biol Biochem 37:241–250
- Belimov AA, Dodd IC, Hontzeas N, Theobald JC, Safronova VI, Davies WJ (2009) Rhizosphere bacteria containing 1-aminocyclopropane-1-carboxylate deaminase increase yield of plants grown in drying soil via both local and systemic hormone signaling. New Phytol 181:413–423
- Bingham FT, Pereyea FJ, Jarrell WM (1986) Metal toxicity to agricultural crops. Metal Ions Biol Syst 20:119–156
- Blaha D, Prigent-Combaret C, Mirza MS, Moënne-Loccoz Y (2006) Phylogeny of the 1-aminocyclopropane-1-carboxylic acid deaminase-encoding gene acdS in phytobeneficial and pathogenic *Proteobacteria* and relation with strain biogeography. FEMS Microbiol Ecol 56:455–470
- Blumwald E (2000) Sodium transport and salt tolerance in plants. Curr Opin Cell Biol 12:431–434
- Bona E, Marsano F, Cavaletto M, Berta G (2007) Proteomic characterization of copper stress response in *Cannabis sativa* roots. Proteomics 7:1121–1130
- Burd GI, Dixon DG, Glick BR (1998) A plant growth promoting bacterium that decreases nickel toxicity in plant seedlings. Appl Environ Microbiol 64:3663–3668
- Burd GI, Dixon DG, Glick BR (2000) Plant growth-promoting bacteria that decrease heavy metal toxicity in plants. Can J Microbiol 46:237–245
- Cheng Z, Park E, Glick BR (2007) 1-Aminocyclopropane-1-carboxylate (ACC) deaminase from *Pseudomonas putida* UW4 facilitates the growth of canola in the presence of salt. Can J Microbiol 53:912–918
- Cheng Z, Duncker BP, McConkey BJ, Glick BR (2008) Transcriptional regulation of ACC deaminase gene expression in *Pseudomonas putida* UW4. Can J Microbiol 54:128–136
- Chinnadurai C, Balachandar D, Sundaram SP (2009) Characterization of 1-aminocyclopropane-1-carboxylate deaminase producing methylobacteria from phyllosphere of rice and their role in ethylene regulation. World J Microbiol Biotechnol 25:1403–1411
- Ciardi JA, Tieman DM, Lund ST, Jones JB, Stall RE, Klee HJ (2000) Response to *Xanthomonas campestris* pv. *vesicatoria* in tomato involves regulation of ethylene receptor gene expression. Plant Physiol 123:81–92
- Coleman NV, Mattes TE, Gossett JM, Spain JC (2002) Biodegradation of *cis*-dichloroethene as the sole carbon source by a beta-proteobacterium. Appl Environ Microbiol 68:2726–2730
- Cook RJ (2002) Advances in plant health management in the twentieth century. Ann Rev Phytopathol 38:95–116
- Cork DJ, Krueger JP (1991) Microbial transformation of herbicides and pesticides. Adv Appl Microbiol 36:1–66

- Cuartero J, Fernandez-Munoz R (1999) Tomato and salinity. Sci Hortic 78:83–125
- Daane LL, Harjono I, Zylstra GJ, Häggblom MM (2001) Isolation and characterization of polycyclic aromatic hydrocarbon-degrading bacteria associated with the rhizosphere of salt marsh plants. Appl Environ Microbiol 67:2683–2691
- Davies WJ, Zhang J (1991) Root signals and the regulation of growth and development of plants in drying soil. Ann Rev Plant Physiol Plant Mol Biol 42:55–76
- Deikman J (1997) Molecular mechanisms of ethylene regulation of gene transcription. Physiol Plant 100:561–566
- Dell'AmicoE, CavalcaL, AndreoniV (2008) Improvement of *Brassica napus* growth under cadmium stress by cadmium-resistant rhizobacteria. Soil Biol Biochem 40:74–84
- Dodd IC (2005) Root-to-shoot signalling: assessing the roles of 'up' in the up and down world of long-distance signalling *in planta*. Plant Soil 274:251–270
- Drew MC (1997) Oxygen deficiency and root metabolism: injury and acclimation under hypoxia and anoxia. Annu Rev Plant Physiol Plant Mol Biol 48:223–250
- Duan J, Müller KM, Charles TC, Vesely S, Glick BR (2009) 1-Aminocyclopropane-1-carboxylate (ACC) deaminase genes in *Rhizobia* from southern Saskatchewan. Microb Ecol 57:423–436
- Else MA, Hall KC, Arnold GM, Davies WJ, Jackson MB (1995) Export of abscisic acid, 1-aminocyclopropane-1carboxylic acid, phosphate, and nitrate from roots to shoots of flooded tomato plants. Plant Physiol 107:377–384
- Enyedi AJ, Yalpani N, Silverman P, Raskin I (1992) Signal molecules in systemic plant resistance to pathogens and pests. Cell 70:879–886
- Escalante-Espinosa E, Gallegos-Martínez ME, Favela-Torres E, Gutiérrez-Rojas M (2005) Improvement of the hydrocarbon phytoremdiation rate by *Cyperus laxus* Lam. inoculated with a microbial consortium in a model system. Chemosphere 59:405–413
- Farwell AJ, Vesely S, Nero V, Rodriguez H, Shah S, Dixon DG, Glick BR (2006) The use of transgenic canola (*Brassica napus*) and plant growth-promoting bacteria to enhance plant biomass at a nickel-contaminated field site. Plant Soil 288:309–318
- Farwell AJ, Vesely S, Nero V, McCormack K, Rodriguez H, Shah S, Dixon DG, Glick BR (2007) Tolerance of transgenic canola (*Brassica napus*) amended with ACC deaminase-containing plant growth-promoting bacteria to flooding stress at a metal-contaminated field site. Environ Poll 147:540–545
- Field JA, Sierra-Alvarez R (2008) Microbial transformation and degradation of polychlorinated biphenyls. Environ Pollut 155:1–12
- Flowers T (2004) Improving crop salt tolerance. J Exp Bot 55:307–319
- Foy CD, Chaney RL, White MC (1978) The physiology of metal toxicity in plants. Annu Rev Plant Physiol 29:511–566
- Frankenberger WT Jr, Arshad M (1995) Phytohormones in soil. Marcel Dekker, New York

- Gamalero E, Glick BR (2011) Mechanisms used by plant growth-promoting bacteria. In: Maheshwari DK (ed) Bacteria in agrobiology: crop ecosystems, pp 17–46
- Gamalero E, Martinotti MG, Trotta A, Lemanceau P, Berta G (2002) Morphogenetic modifications induced by *Pseudomonas fluorescens* A6RI and *Glomus mosseae* BEG12 in the root system of tomato differ according to plant growth conditions. New Phytol 155:293–300
- Gamalero E, Lingua G, Caprì FG, Fusconi A, Berta G, Lemanceau P (2004) Colonization pattern of primary tomato roots by *Pseudomonas fluorescens* A6RI characterized by dilution plating, flow cytometry, fluorescence, confocal and scanning electron microscopy. FEMS Microbiol Ecol 48:79–87
- Gamalero E, Lingua G, Tombolini R, Avidano L, Pivato B, Berta G (2005) Colonization of tomato root seedling by *Pseudomonas fluorescens* 92rkG5: spatio-temporal dynamics, localization, organization, viability and culturability. Micr Ecol 50:289–297
- Gamalero E, Berta G, Massa N, Glick BR, Lingua G (2008) Synergistic interactions between the ACC deaminase-producing bacterium *Pseudomonas putida* UW4 and the AM fungus *Gigaspora rosea* positively affect cucumber plant growth. FEMS Microbiol Ecol 64:459–467
- Gamalero E, Berta G, Massa N, Glick BR, Lingua G (2010) Interactions between *Pseudomonas putida* UW4 and *Gigaspora rosea* BEG9 and their consequences on the growth of cucumber under salt stress conditions. J Appl Microbiol 108:236–245
- Gerhardt KE, Huang XD, Glick BR, Greenberg BM (2009) Phytoremediation and rhizoremediation of organic soil contaminants: potential and challenges. Plant Sci 176:20–30
- Germaine KJ, Keogh K, Ryan D, Dowling DN (2009) Bacterial endophyte-mediated naphthalene phytoprotection and phytoremediation. FEMS Microbiol Lett 296:226–234
- Ghosal D, You IS, Chatterjee DK, Chakrabarty AM (1985) Microbial degradation of halogenated compounds. Science 228:135–142
- Ghosh S, Penterman JN, Little RD, Chavez R, Glick BR (2003) Three newly isolated plant growth-promoting bacilli facilitate the seedling growth of canola, *Brassica campestris*. Plant Physiol Biochem 41:277–281
- Giller KE, Witter E, McGrath SP (1998) Toxicity of heavy metals to microorganisms and microbial processes in agricultural soil: a review. Soil Biol Biochem 30:1389–1414
- Giovanelli J, Mudd SH, Datko AH (1980) Sulfur amino acids in plants. In: Miflin BJ (ed) Amino acids and derivatives, the biochemistry of plants: a comprehensive treatise, vol 5. Academic, New York, pp 453–505
- Glick BR (1995) The enhancement of plant growth by free-living bacteria. Can J Microbiol 41:109–117
- Glick BR (2003) Phytoremediation: synergistic use of plants and bacteria to clean up the environment. Biotechnol Adv 21:383–393

- Glick BR (2004) Bacterial ACC deaminase and the alleviation of plant stress. Adv Appl Microbiol 56:291–312
- Glick BR (2005) Modulation of plant ethylene levels by the bacterial enzyme ACC deaminase. FEMS Microbiol Lett 251:1–7
- Glick BR, Karaturovíc D, Newell P (1995) A novel procedure for rapid isolation of plant growth-promoting rhizobacteria. Can J Microbiol 41:533–536
- Glick BR, Penrose DM, Li J (1998) A model for the lowering of plant ethylene concentrations by plant growth promoting bacteria. J Theor Biol 190:63–68
- Glick BR, Patten CL, Holguin G, Penrose DM (1999) Biochemical and genetic mechanisms used by plant growth promoting bacteria. Imperial College Press, London
- Glick BR, Cheng Z, Czarny J, Duan J (2007a) Promotion of plant growth by ACC deaminase-containing soil bacteria. Eur J Plant Pathol 119:329–339
- Glick BR, Todorovic B, Czarny J, Cheng Z, Duan J, McConkey B (2007b) Promotion of plant growth by bacterial ACC deaminase. Crit Rev Plant Sci 26:227–242
- Glick BR, Pasternak JJ, Patten CL (2010) Molecular biotechnology, 4th edn. American Society for Microbiology, Washington, DC
- Graves WR, Gladon RJ (1985) Water stress, endogenous ethylene and *Ficus benjamina* leaf abscission. Hortic Sci 20:273–275
- Grichko VP, Glick BR (2000) Identification of DNA sequences that regulate the expression of the *Enterobacter cloacae* UW4 1-aminocyclopropane-1-carboxylate deaminase gene. Can J Microbiol 46:1159–1165
- Grichko VP, Glick BR (2001a) Amelioration of flooding stress by ACC deaminase-containing plant growthpromoting bacteria. Plant Physiol Biochem 39:11–17
- Grichko VP, Glick BR (2001b) Ethylene and flooding stress in plants. Plant Physiol Biochem 39:1–9
- Grichko VP, Filby B, Glick BR (2000) Increased ability of transgenic plants expressing the bacterial enzyme ACC deaminase to accumulate Cd, Co, Cu, Ni, Pb, and Zn. J Biotechnol 81:45–53
- Gurska J, Wang W, Gerhardt KE, Khalid AM, Isherwood DM, Huang Xd, Glick BR, Greenberg BM (2009) Three year field test of a plant growth promoting rhizobacteria enhanced phytoremediation system at a land farm for treatment of hydrocarbon waste. Environ Sci Technol 43:4472–4479
- Hall MA, Smith AR (1995) Ethylene and the responses of plants to stress. Bulg J Plant Physiol 21:71–79
- Hall JA, Peirson D, Ghosh S, Glick BR (1996) Root elongation in various agronomic crops by the plant growth promoting rhizobacterium *Pseudomonas putida* GR 12-2. Isr J Plant Sci 44:37–42
- Hao Y, Charles TC, Glick BR (2007) ACC deaminase from plant growth promoting bacteria affects crown gall development. Can J Microbiol 53:1291–1299
- Honma M (1983) Enzymatic determination of 1-aminocyclopropane-1-carboxylate deaminase. Agric Biol Chem 47:617–618
- Honma M (1993) Stereospecific reaction of 1-aminocyclopropane-1-carboxylate deaminase. In: Pech JC,

Latché A, Balagué C (eds) Cellular and molecular aspects of the plant hormone ethylene. Kluwer Academic Publishers, Dordrecht, pp 111–116

- Honma M, Shimomura T (1978) Metabolism of 1-aminocyclopropane-1-carboxylic acid. Agric Biol Chem 42:1825–1831
- Hontzeas N, Zoidakis J, Glick BR, Abu-Omar MM (2004) Expression and characterization of 1-aminocyclopropane-1-carboxylate deaminase from the rhizobacterium *Pseudomonas putida* UW4: a key enzyme in bacterial plant growth promotion. Biochim Biophys Acta 1703:11–19
- Hontzeas N, Richardson AO, Belimov AA, Safronova VI, Abu-Omar MM, Glick BR (2005) Evidence for horizontal gene transfer (HGT) of ACC deaminase genes. Appl Environ Microbiol 71:7556–7558
- Hontzeas N, Hontzeas CE, Glick BR (2006) Reaction mechanisms of the bacterial enzyme 1-aminocyclopropane-1-carboxylare deaminase. Biotechnol Adv 24:420–426
- Hsu YT, Kao CH (2004) Cadmium toxicity is reduced by nitric oxide in rice leaves. Plant Growth Regul 42:227–238
- Huang X-D, El-Alawi Y, Penrose DM, Glick BR, Greenberg BM (2004) Responses of plants to creosote during phytoremediation and their significance for remediation processes. Environ Pollut 130:453–463
- Huang X-D, El-Alawai Y, Gurska J, Glick BR, Greenberg BM (2005) A multi-process phytoremediation system for decontamination of Persistent Total Petroleum Hydrocarbons (TPHs) from soils. Microchem J 81:139–147
- Jackson MB (1985) Ethylene and responses of plants to soil waterlogging and submergence. Annu Rev Plant Physiol 36:145–174
- Jacobson CB, Pasternak JJ, Glick BR (1994) Partial purification and characterization of ACC deaminase from the plant growth-promoting rhizobacterium *Pseudomonas putida* GR12-2. Can J Microbiol 40:1019–1025
- Jia Y-J, Kakuta Y, Sugawara M, Igarashi T, Oki N, Kisaki M, Shoji T, Kanetuna Y, Horita T, Matsui H, Honma M (1999) Synthesis and degradation of 1-aminocyclopropane-1-carboxylic acid by *Penicillium citrinum*. Biosci Biotechnol Biochem 63:542–549
- John P (1991) How plant molecular biologists revealed a surprising relationship between two enzymes, which took an enzyme out of a membrane where it was not located, and put it into the soluble phase where it could be studied. Plant Mol Biol Rep 9:192–194
- Johnsen AR, Karlson U (2004) Evaluation of bacterial strategies to promote the bioavailability of polycyclic aromatic hydrocarbons. Appl Microbiol Biotechnol 63:452–459
- Kalantari KM, Smith AR, Hall MA (2000) The effect of water stress on 1-(malonylamino)cyclopropane-1carboxylic acid concentration in plant tissues. Plant Growth Regul 31:183–193
- Kamath R, Rentz JA, Schnoor JL, Alvarez PJJ (2004) Phytoremediation of hydrocarbon-contaminated soils: principles and applications. In: Vazquez-Duhalt R,

Quintero-Ramirez R (eds) Petroleum biotechnology: developments and perspectives studies in surface science and catalysis. Elsevier, Oxford, UK

- Kaneko T, Nakamura Y, Sato S, Asamizu E, Kato T, Sasamoto S, Watanabe A, Idesawa K, Ishikawa A, Kawashima K, Kimura T, Kishida Y, Kiyokawa C, Kohara M, Matsumoto M, Matsuno A, Mochizuki Y, Nakayama S, Nakazaki N, Shimpo S, Sugimoto M, Takeuchi C, Yamada M, Tabata S (2000) Complete genome structure of the nitrogen-fixing symbiotic bacterium *Mesorhizobium loti*. DNA Res 7:331–338
- Kaneko T, Minamisawa K, Isawa T, Nakatsukasa H, Mitsui H, Kawaharada Y, Nakamura Y, Watanabe A, Kawashima K, Ono A, Shimizu Y, Takahashi C, Minami C, Fujishiro T, Kohara M, Katoh M, Nakazaki N, Nakayama S, Yamada M, Tabata S, Sato S (2010) Complete genomic structure of the cultivated rice endophyte *Azospirillum* sp B510. DNA Res 17:37–50
- Karthikeyan S, Zhao Z, Kao C, Zhou Q, Tao Z, Zhang H, Liu H (2004) Structural analysis of 1-aminocyclopropane-1-carboxylate deaminase: observation of an aminyl intermediate and identification of Tyr294 as the active site nucleophile. Angew Chem Int Ed 43:3425–3429
- Karuna Stree B, Chadalavanda SV, Rajendrakumar A, Reddy R (2000) Aldose reductase in rice (*Oryza sativa* L.): stress response and developmental specificity. Plant Sci 160:149–157
- Kende H (1989) Enzymes of ethylene biosynthesis. Plant Physiol 91:1–4
- Kim IS, Ryu JY, Hur HG, Gu MB, Kim SD, Shim JH (2004) Sphingomonas sp. Strain SB5 degrades carbofuran to a new metabolite by hydrolysis at the furanyl ring. J Agric Food Chem 52:2309–2314
- Klee HJ, Hayford MB, Kretzmer KA, Barry GF, Kishore GM (1991) Control of ethylene synthesis by expression of a bacterial enzyme in transgenic tomato plants. Plant Cell 3:118–1193
- Kloepper JW, Schroth MN (1978) Plant growth-promoting rhizobacteria on radishes. In: Proceedings of the 4th international conference on plant pathogenic bacteria. vol 2, Station de Pathologie Vegetale et Phytobacteriologie, INRA, Angers
- Kloepper JW, Lifshitz R, Zablotowitz RM (1989) Free living bacteria inocula for enhancing crop productivity. Trends Biotechnol 7:39–43
- Kozlowski TT (1984) Extent, causes, and impacts of flooding. In: Kozlowski TT (ed) Flooding and plant growth. Academic, New York, pp 1–5
- Krämer U, Chardonnens AN (2001) The use of transgenic plants in the bioremediation of soils contaminated with trace elements. Appl Microbiol Biotechnol 55:661–672
- Krupa Z (1988) Cadmium-induced changes in the composition and structure of the light-harvesting chlorophyll *a/b* protein complex-II in radish cotyledons. Physiol Plant 73:518–524
- Kuffner M, Puschenreiter M, Wieshammer G, Gorfer MS, Sessitsch A (2008) Rhizosphere bacteria affect growth and metal uptake of heavy metal accumulating willows. Plant Soil 304:35–44

- Kukreja S, Nandwal AS, Kumar N, Sharma SK, Sharma SK, Unvi V, Sharma PK (2005) Plant water status, H₂O₂ scavenging enzymes, ethylene evolution and membrane integrity of *Cicer arietinum* as affected by salinity. Biol Plant 49:305–308
- Kumar R, Pandey S, Pandey A (2006) Plant roots and carbon sequestration. Curr Sci 91:885–890
- Lee EH, Kim J, Cho KS, Ahn YG, Hwang GS (2010) Degradation of hexane and other recalcitrant hydrocarbons by a novel isolate, *Rhodococcus* sp. EH831. Environ Sci Pollut Res 17:64–77
- Leigh MB, Prouzova P, Mackova M, Macek T, Nagle DO, Fletcher JS (2006) Polychlorinated biphenyl (PCB)degrading bacteria associated with trees in a PCBcontaminated site. Appl Environ Microbiol 72:2331–2342
- Li J, Glick BR (2001) Transcriptional regulation of the *Enterobacter cloacae* UW4 1-aminocyclopropane-1carboxylate (ACC) deaminase gene (acdS). Can J Microbiol 47:259–267
- Liu L, Jiang C-Y, Liu X-Y, Wu J-F, Han J-G, Liu S-J (2007) Plant-microbe association for rhizoremediation of chloronitroaromatic pollutants with *Comamonas* sp. strain CNB-1. Environ Microbiol 9:465–473
- Lusk C, Reich PB, Montgomery RA, Ackerly DD, Cavender-Bares J (2008) Why are evergreen leaves so contrary about shade? Trends Ecol Evol 23:299–303
- Lynch JM, Whipps JM (1990) Substrate flow in the rhizosphere. Plant Soil 129:1–10
- Ma W, Zalec K, Glick BR (2001) Effects of the bioluminescence-labeling of the soil bacterium *Kluyvera* ascorbata SUD165/26. FEMS Microbiol Ecol 35:137–144
- Ma W, Guinel FC, Glick BR (2003a) The *Rhizobium leguminosarum* bv. *viciae* ACC deaminase protein promotes the nodulation of pea plants. Appl Environ Microbiol 69:4396–4402
- Ma W, Sebestianova S, Sebestian J, Burd GI, Guinel F, Glick BR (2003b) Prevalence of 1-aminocyclopropaqne-1-carboxylate in deaminase in *Rhizobia* spp. Anton Van Leeuwenhoek 83:285–291
- Ma W, Charles TC, Glick BR (2004) Expression of an exogenous 1-aminocyclopropane-1-carboxylate deaminase gene in *Sinorhizobium meliloti* increases its ability to nodulate alfalfa. Appl Environ Microbiol 70:5891–5897
- Ma Y, Rajkumar M, Freitas H (2009) Isolation and characterization of Ni mobilizing PGPB from serpentine soils and their potential in promoting plant growth and Ni accumulation by *Brassica* spp. Chemosphere 75:719–725
- Madhaiyan M, Poonguzhali S, Sa T (2007) Characterization of 1-aminocyclopropane-1-carboxylate (ACC) deaminase containing *Methylobacterium oryzae* and interactions with auxins and ACC regulation of ethylene in canola (*Brassica campestris*). Planta 226:867–876
- Maksymiec W, Baszynski T (1996) Chlorophyll fluorescence in primary leaves of excess Cu-treated runner bean plants depends on their growth stages and the duration of Cuaction. J Plant Physiol 149:196–200

- Maksymiec W, Bednara J, Baszynski T (1995) Responses of runner plants to excess copper as a function of plant growth stages: effects on morphology and structure of primary leaves and their chloroplast ultrastructure. Photosynthetica 31:427–435
- Mayak S, Tirosh T, Glick BR (2004a) Plant growthpromoting bacteria that confer resistance to water stress in tomato and pepper. Plant Sci 166:525–530
- Mayak S, Tirosh T, Glick BR (2004b) Plant growthpromoting bacteria that confer resistance in tomato to salt stress. Plant Physiol Biochem 42:565–572
- Medina VF, Maestri E, Marmiroli M, Dietz AC, McCutcheon SC (2004) Plant tolerances to contaminants. In: Schnoor JL, Zehdner A (eds) Phytoremediation: transformation and control of contaminants. Wiley, New York
- Mithöfer A, Schulze B, Boland W (2004) Biotic and heavy metal stress response in plants: evidence for common signals. FEBS Lett 56:1–5
- Morgan PW, Drew CD (1997) Ethylene and plant responses to stress. Physiol Plant 100:620–630
- Nadeem SM, Zahair ZA, Naveed M, Arshad M (2007) Preliminary investigations on inducing salt tolerance in maize through inoculation with rhizobacteria containing ACC deaminase activity. Can J Microbiol 53:1141–1149
- Nadeem SM, Zahir ZA, Naveed M, Asghar HN, Arshad M (2010) Rhizobacteria capable of producing ACCdeaminase may mitigate salt stress in wheat. Soil Sci Soc Am J 74:533–542
- Nguyen C (2003) Rhizodeposition of organic C by plants: mechanisms and controls. Agronomie 23:375–396
- Nie L, Shah S, Burd GI, Dixon DG, Glick BR (2002) Phytoremediation of arsenate contaminated soil by transgenic canola and the plant growth-promoting bacterium *Enterobacter cloacae* CAL2. Plant Physiol Biochem 40:355–361
- Nilsen ET, Orcutt DM (1997) The physiology of plants under stress. Abiotic factors. Wiley, New York
- O'Donnell PJ, Calvert C, Atzorn R, Wasternack C, Leyser HMO, Bowles DJ (1996) Ethylene as a signal mediating the wound response of tomato plants. Science 274:1914–1917
- Park J, Kukor JJ, Abriola LM (2002) Characterization of the adaptive response to trichloroethylene-mediated stresses in *Ralstonia pickettii* PKO1. Appl Environ Microbiol 68:5231–5240
- Parnell JJ, Park J, Denef V, Tsoi T, Hashsham S, Quensen J, Tiedje JA (2006) Coping with polychlorinated biphenyl (PCB) toxicity: physiological and genome-wide responses of *Burkholderia xenovorans* LB400 to PCBmediated stress. Appl Environ Microbiol 72:6607–6614
- Patten C, Glick BR (1996) Bacterial biosynthesis of indole-3-acetic acid. Can J Microbiol 42:207–220
- Patten CL, Glick BR (2002) The role of bacterial indoleacetic acid in the development of the host plant root system. Appl Environ Microbiol 68:3795–3801
- Penrose DM, Glick BR (2001) Levels of ACC and related compounds in exudate and extracts of canola seeds

treated with ACC deaminase containing plant growth-promoting bacteria. Can J Microbiol 47:368–372

- Perfus-Barbeoch L, Leonhardt N, Vavasseur A, Forestier C (2002) Heavy metal toxicity: cadmium permeates through calcium channels and disturbs the plant water status. Plant J 32:539–548
- Pieper DH, Seeger M (2008) Bacterial metabolism of polychlorinated biphenyls. J Mol Microbiol Biotechnol 15:121–138
- Pilon-Smits E (2005) Phytoremediation. Annu Rev Plant Biol 56:15–39
- Pilon-Smits EAH, Pilon M (2002) Phytoremediation of metals using transgenic plants. Crit Rev Plant Sci 21:439–456
- Pivato B, Gamalero E, Lemanceau P, Berta G (2008) Cell organization of *Pseudomonas fluorescens* C7R12 on adventitious roots of *Medicago truncatula* as affected by arbuscular mycorrhiza. FEMS Microbiol Lett 289:173–180
- Potters G, Pasternak TP, Guisez Y, Jansen MA (2009) Different stresses, similar morphogenic responses: integrating a plethora of pathways. Plant Cell Environ 32:158–169
- Prigent-Combaret C, Blaha D, Pothier JF, Vial L, Poirier MA, Wisniewski-Dyé F, Moënne-Loccoz Y (2008) Physical organization and phylogenetic analysis of acdR as leucine-responsive regulator of the 1-aminocyclopropane-1-carboxylate deaminase gene acdS in phytobeneficial Azospirillum lipoferum 4B and other Proteobacteria. FEMS Microbiol Ecol 65:202–219
- Raddadi N, Cherif A, Boudabous A, Daffonchio D (2008) Screening of plant growth promoting traits of *Bacillus* thuringiensis. Ann Microbiol 58:47–52
- Radwan SS, Dashti N, El-Nemr IM (2005) Enhancing the growth of *Vicia faba* plants by microbial inoculation to improve their phytoremediation potential for oily desert areas. Int J Phytoremediation 7:19–32
- Rajapaksha RMCP, Tobor-Kaplon MA, Bååth E (2004) Metal toxicity affects fungal and bacterial activities in soil differently. Appl Environ Microbiol 70:2966–2973
- Reed MLE, Glick BR (2004) Applications of free living plant growth-promoting rhizobacteria. Anton Van Leeuwenhoek 86:1–25
- Reed MLE, Glick BR (2005) Growth of canola (*Brassica napus*) in the presence of plant growth-promoting bacteria and either copper or polycyclic aromatic hydrocarbons. Can J Microbiol 51:1061–1069
- Regvar M, Groznik N, Goljevšcek K, Gogala N (2001) Diversity of arbuscular mycorrhizal fungi form various disturbed ecosystems in Slovenia. Acta Biol Slov 44:27–34
- Reynolds M, Tuberosa R (2008) Translational research impacting on crop productivity in drought-prone environments. Curr Opin Plant Biol 11:171–179
- Richard AJ, El-Abd SO (1989) Prevention of salt-induced epinasty by a-aminooxyacetic acid and cobalt. Plant Growth Regul 8:315–323
- Rodriguez H, Vesely S, Shah S, Glick BR (2008) Isolation and characterization of nickel resistant *Pseudomonas* strains and their effect on the growth of non-transformed

and transgenic canola plants. Curr Microbiol 57:170–174

- Rodrìguez-Serrano M, Romero-Puertas MC, Zabalza A, Corpas FJ, Gòmez M, del Rìo LA, Sandalio LM (2006) Cadmium effect on oxidative metabolism of pea (*Pisum sativum* L.) roots. Imaging of reactive oxygen species and nitric oxide accumulation *in vivo*. Plant Cell Environ 29:1532–1544
- Rothballer M, Eckert B, Schmid M, Fekete A, Schloter M, Lehner A, Pollmann S, Hartmann A (2008) Endophytic rootcolonization of gramineous plants by *Herbaspirillum frisingense*. FEMS Microbiol Ecol 66:85–95
- Saleh S, Huang X-D, Greenberg BM, Glick BR (2004) Phytoremediation of persistent organic contaminants in the environment. In: Singh A, Ward O (eds) Applied bioremediation and phytoremediation, vol 1, Series: Soil biology. Springer, Berlin, pp 115–134
- Salt DE, Smith RD, Raskin I (1998) Phytoremediation. Annu Rev Plant Physiol Plant Mol Biol 49:643–668
- Sandalio LM, Dalurzo HC, Gomez M, Romero-Puertas MC, del Río LA (2001) Cadmium-induced changes in the growth and oxidative metabolism of pea plants. J Exp Bot 52:2115–2126
- Saravanakumar D, Samiyappan R (2006) ACC deaminase from *Pseudomonas fluorescens* mediated saline resistance in groundnut (*Arachis hypogea*) plants. J Appl Microbiol 102:1283–1292
- Schmidt W (2008) Inner voices meet outer signals: the plasticity of rhizodermic cells. Plant Sci 174:239–245
- Sergeeva E, Shah S, Glick BR (2006) Growth of transgenic canola (*Brassica napus* cv Westar) expressing a bacterial 1-aminocyclopropane-1-carboxylate (ACC) deaminase gene on high concentrations of salt. World J Microbiol Biotechnol 22:277–282
- Sessitsch A, Coenye T, Sturz AV, Vandamme P, Ait Barka E, Salles JF, Van Elsas JD, Faure D, Reiter B, Glick BR, Wang-Pruski G, Nowak J (2005) *Burkholderia phytofirmans* sp. nov., a novel plant-associated bacterium with plant-beneficial properties. Int J Syst Evol Microbiol 55:1187–1192
- Shah S, Li J, Moffatt BA, Glick BR (1997) ACC deaminase genes from plant growth promoting rhizobacteria. In: Ogoshi A, Kobayashi K, Hemma Y, Kodema F, Kondo N, Akino S (eds) Plant growth-promoting rhizobacteria. Present status and future prospects. Organization for Economic Cooperation and Development, Paris, pp 320–324
- Shaharoona B, Arshad M, Zahir ZA (2006) Effect of plant growth promoting rhizobacteria containing ACC deaminase on maize (*Zea mays* L.) growth under axenic conditions and on nodulation in mung bean (*Vigna radiata* L.). Lett Appl Microbiol 42:155–159
- Sheehy RE, Honma M, Yamada M, Sasaki T, Martineau B, Hiatt WR (1991) Isolation, sequence, and expression in *Escherichia coli* of the *Pseudomonas* sp. Strain ACP gene encoding 1-amino-cyclopropane-1carboxylate deaminase. J Bacteriol 173:5260–5265
- Sheng XF, Gong JX (2006) Increased degradation of phenanthrene in soil by *Pseudomonas* sp. GF3 in the presence of wheat. Soil Biol Biochem 38:2587–2592

- Sheng X, Chen X, He L (2008) Characteristics of an endophytic pyrene-degrading bacterium of *Enterobacter* sp. 12J1 from *Allium macrostemon* Bunge. Int Biodeter Biodegrad 62:88–95
- Shi H, Ishitani M, Kim C, Zhu J-K (2000) The Arabidopsis thaliana salt tolerance gene SOS1 encodes a putative Na⁺/H⁺ antiporter. Proc Natl Acad Sci USA 97:6896–6901
- Shibli RA, Kushad M, Yousef GG, Lila MA (2007) Physiological and biochemical responses of tomato microshoots to induced salinity stress with associated ethylene accumulation. Plant Growth Regul 51:159–169
- Shiu OY, Oetiker JH, Yip WK, Yang SF (1998) The promoter of *LE-ACS7*, an early flooding-induced 1-aminocyclopropane carboxylate synthase gene of the tomato, is tagged by a Sol3 transposon. Proc Natl Acad Sci USA 95:10334–10339
- Sicilano SD, Germida JJ (1997) Bacterial inoculants of forage grasses that enhance degradation of 2-chlorobenzoic acid in soil. Environ Toxicol Chem 16:1098–1104
- Sobeih WY, Dodd IC, Bacon MA, Grierson D, Davies WJ (2004) Long-distance signals regulating stomatal conductance and leaf growth in tomato (*Lycopersicon esculentum*) plants subjected to partial root zone drying. J Exp Bot 55:2353–2363
- Sorensen SR, Albers CN, Aamand J (2008) Rapid mineralization of the phenylurea herbicide diuron by *Variovorax* sp strain SRS16 in pure culture and within a two-member consortium. Appl Environ Microbiol 74:2332–2340
- Stearns JC, Shah S, Greenberg BM, Dixon DG, Glick BR (2005) Tolerance of transgenic canola expressing 1-aminocyclopropane-1-carboxylic acid deaminase to growth inhibition by nickel. Plant Physiol Biochem 43:701–708
- Sun Y, Cheng Z, Glick BR (2009) The role of 1-aminocyclopropane-1-carboxylate (ACC) deaminase in plant growth promotion by the endophytic bacterium *Burkholderia phytofirmans* PsJN. FEMS Microbiol Lett 296:131–136
- Tabaei-Aghdaei SR, Harrison P, Pearce RS (2000) Expression of dehydration-stress genes in the crowns of wheatgrass species (*Lophopyrum elongatum* (host) A. Love and *Agropyron desertorum* (Fisch Ex Link Schult.) having contrasting acclimation to salt, cold and drought. Plant Cell Environ 23:561–571
- Taghavi S, van der Lelie D, Hoffman A, Zhang YB, Walla MD, Vangronsveld J, Newman L, Monchy S (2010) Genome Sequence of the Plant Growth Promoting endophytic bacterium *Enterobacter* sp 638. PLoS Genet 6(5):e1000943
- Todorovic B, Glick BR (2008) The interconversion of ACC deaminase and D-cysteine desulfhydrase by directed mutagenesis. Planta 229:193–205
- Toklikishvili N, Dandurishvili N, Tediashvili M, Giorgobiani N, Szegedi E, Glick BR, Vainstein A, Chernin L (2010) Inhibitory effect of ACC deaminaseproducing bacteria on crown gall formation in tomato plants infected by *Agrobacterium tumefaciens* or *A. vitis*. Plant Pathol 59:1023–1030

- Uchiumi T, Oowada T, Itakura M, Mitsui H, Nukui N, Dawadi P, Kaneko T, Tabata S, Yokoyama T, Tejima T, Saeki K, Oomori H, Hayashi M, Maekawa T, Sriprang R, Murooka Y, Tajima S, Simomura K, Nomura M, Suzuki A, Shimoda S, Sioya K, Abe M, Minamisawa K (2004) Expression islands clustered on symbiosis island of *Mesorhizobium loti* genome. J Bacteriol 186:2439–2448
- Uhlik O, Jecna K, Macknova M, Vlcek C, Hroudova M, Demnerova K (2009) Biphenyl-metabolizing bacteria in the rhizosphere of horseradish and bulk soil contaminated by polychlorinated biphenyls as revealed by stable isotope probing. Appl Environ Microbiol 75:6471–6477
- Van Beelen P, Doelman P (1997) Significance and application of microbial toxicity tests in assessing ecotoxicological risks of contamination in soil and sediment. Chemosphere 34:455–499
- VanLoon LC, Glick BR (2004) Increased plant fitness by rhizobacteria. In: Sandermann H (ed) Molecular ecotoxicology of plants. Springer, Berlin, pp 177–205
- Vartapetian BB, Jackson MB (1997) Plant adaptations to anaerobic stress. Ann Bot 79:3–20
- Viterbo A, Landau U, Kim S, Chernin L, Chet I (2010) Characterization of ACC deaminase from the biocontrol and plant growth-promoting agent *Trichoderma asperellum* T203. FEMS Microbiol Lett 305:42–48
- Wackett LP, Sadowsky MJ, Martinez B, Shapir N (2002) Biodegradation of atrazine and related s-triazine compounds: from enzymes to field studies. Appl Microbiol Biotechnol 58:39–45

- Wang C, Knill E, Glick BR, Défago G (2000) Effect of transferring 1-aminocyclopropane-1-carboxylic acid (ACC) deaminase genes into *Pseudomonas fluorescens* strain CHA0 and its *gacA* derivative CHA96 on their growth-promoting and disease-suppressive capacities. Can J Microbiol 46:898–907
- Yang SF, Hoffman NE (1984) Ethylene biosynthesis and its regulation in higher plants. Annu Rev Plant Physiol 35:155–189
- Yee DC, Maynard JA, Wood TK (1998) Rhizomediation of trichloroethylene by a recombinant root-colonizing *Pseudomonas fluorescens* strain expressing toluene ortho-monooxygenase constitutively. Appl Environ Microbiol 64:112–118
- Yue HT, Mo WP, Li C, Zheng YY, Li H (2007) The salt stress relief and growth promotion effect of Rs-5 on cotton. Plant Soil 297:139–145
- Zhang H-X, Blumwald E (2001) Transgenic salt tolerant tomato plants accumulate salt in the foliage but not in the fruits. Nat Biotechnol 19:765–768
- Zhang H-X, Hodson J, Williams JP, Blumwald E (2001) Engineering salt-tolerant *Brassica* plants: characterization of yield and seed oil quality in transgenic plants with increased vacuolar sodium accumulation. Proc Nat Acad Sci USA 98:12832–12836
- Zhang Y, Zhao L, Wang Y, Yang B, Chen S (2008) Enhancement of heavy metal accumulation by tissue specific co-expression of iaaM and ACC deaminase genes in plants. Chemosphere 72:564–571
- Zhu J-K (2002) Salt and drought stress signal transduction in plants. Ann Rev Plant Biol 53:247–273

New Approaches to Study Metal-Induced Stress in Plants

19

M.C. Cia, F.R. Capaldi, R.F. Carvalho, P.L. Gratão, and R.A. Azevedo

Abstract

The contamination of water, soil, and sediments with toxic metals has been and will continue to be a major environmental problem that needs to be dealt with. In more recent years, the studies with metals have concentrated on the antioxidant stress response characterization in a wide range of plant species. The effects of many different metals have been published, but all have consistently addressed very similar problems and investigated basic parameters. These kinds of studies are important as they allow the verification of the sensitivities of different plant species to distinct metals, eventually indicating specific biomarkers to stressful situations. However, metal-induced stress studies need new approaches that are likely to increase our understanding as how these elements affect plant metabolism and to identify the modifications that are needed to improve plant adaptation and tolerance. New techniques that have greatly improved the identification, localization, and quantification of metals within plant tissues have led to the science of metallomics. This advancement in knowledge should eventually allow the characterization of plants used in the process of phytoremediation of soils contaminated with toxic metals. In this chapter, we discuss the use of new techniques and approaches to study the effects of toxic metals and we propose a more integrated action among distinct areas in the field of metallomics and other "omics."

Keywords

Heavy metals stress • Signaling • Proteomics • Grafting • Cytogenetics • Metallomics

• F.R. Capaldi • P.L. Gratão

Departamento de Genética, Escola Superior de Agricultura Luiz de Queiroz, Universidade de São Paulo, Piracicaba 13418-900, SP, Brazil e-mail: raazeved@esalq.usp.br

R.F. Carvalho Departamento de Biologia Aplicada à Agropecuária, Universidade Estadual Paulista Júlio de Mesquita Filho, Jaboticabal 14884-900, SP, Brazil

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_19, © Springer Science+Business Media, LLC 2012

R.A. Azevedo (🖂) • M.C. Cia

1 Introduction

There is abundant evidence of shifts in temperature, rainfall pattern, species distribution, and biogeochemical cycles in response to global climate change (Parmesan 2006; Allan and Soden 2008; Campbell et al. 2009). This change will probably affect the behavior of contaminants present in the environment. High temperatures can lead to changes in organic matter and increase the availability of the bound toxic metals (Porcal et al. 2009). An increase in the frequency of precipitation can transport toxic metals from a contaminated area to a noncontaminated one (Harmon and Wyatt 2008). Moreover, toxic metals may be continuously introduced into the environment by mining, metalliferous industries, application of fertilizers, sewage sludge, metal-based pesticides, and other anthropogenic activities. Thus, contamination of water, soil, and sediments with toxic metals has been and will continue to be a major environmental problem that needs to be dealt with (Gratão et al. 2005).

Some metals have been classified as toxic, persistent, and bioaccumulative elements. According to the Agency for Toxic Substances and Disease Registry (ATSDR 2007), among the ten most hazardous substances to human health, four are toxic metals: lead (Pb), mercury (Hg), arsenic (As), and cadmium (Cd). Interestingly, Pb and Hg rank first among the most harmful metals to plants, followed by copper (Cu), cadmium (Cd)/arsenic (As), cobalt (Co)/nickel (Ni)/zinc (Zn), and manganese (Mn) (Kopittke et al. 2010). Pb and Hg have been reported as mutagenic agents in plants (Patra et al. 2004). Due to its chemical similarity to phosphorus, As may interfere with several physiological and biochemical processes (Patra et al. 2004). Cd does not appear to have any physiological function, with the exception of the marine diatom Thalassiosira weissflogii, that possesses a carbonic anhydrase with Cd as its metal center (Lane et al. 2005). Cu, Ni, Co, Zn, and Mn are all plant micronutrients. They participate in prosthetic groups and as

cofactors of many proteins and are therefore essential for growth and development, however at high concentrations cause oxidative stress (Hansch and Mendel 2009). Aluminum (Al) is another toxic element with significant implications for agriculture, because 30% of the world's land areas consist of acid soils (Horst et al. 2010). Several other metals have been investigated for their effects on plant metabolism, toxicity, and crop productivity (Becana et al. 1998; Panda and Choudhury 2005; Gomes-Junior et al. 2007).

Exposure to a toxic metal can result in inhibition of seed germination, photosynthesis, plant growth, and consequently causes yield losses. These symptoms are normally related with overproduction of reactive oxygen species (ROS), changes in the permeability and structure of cell membranes, imbalance of mineral nutrients, incorporation of the metal into S-containing molecules, and cell cycle disruption (Schutzendubel and Polle 2002; Benavides et al. 2005). However, the toxicity of metals varies among plant species (Tolrà et al. 2006), cultivars (Liu et al. 2007), according to the characteristic of the soils (Bradl 2004; Li et al. 2009a), and the dose and time of metal exposure (Gratão et al. 2008). Moreover, metals can be found in different oxidation states and form various species. The different oxidation states of a particular metal ion possess different physicochemical properties and vary in their toxicity (Shanker et al. 2004).

The majority of toxic metals are presented as cations under physiological conditions, with a high redox potential. Thus, when exposed to toxic metals, cells can suffer oxidative damage caused either directly by redox-active metals or indirectly by metal-induced metabolic disturbance (Valko et al. 2005). Any current literature search will reveal that a lot of attention has been given to the responses of antioxidant defense systems to exposure to toxic metals (Azevedo and Azevedo 2006). Although these studies have led to a considerable increase in our understanding of the responses of plants to abiotic stress, new methodologies have emerged (Arruda and Azevedo 2009) and will be discussed in this chapter.

2 Studies on Metal-Induced Stress in Plants

Plants respond to metal toxicity in many different ways, according to species sensitivity, metal speciation, mobility within the living organism, intensity and duration of the exposure. A metal can cause the accumulation of an array of metabolites, such as proline and other amino acids (Sharma and Dietz 2006), anthocyanins and other phenolic compounds (Posmyk et al. 2008), phenylpropanoids, terpenoids, alkaloids, and many others secondary metabolites (Mithofer et al. 2004). It is known that metals influence plant development at all level of organization (Prasad and Strzalka 2002).

Metal-induced stress increases the generation of ROS by imbalance of cellular redox status, leading to a condition of oxidative stress and increased cellular damage (Møller et al. 2007). A relationship between redox homeostasis, metal stress, and antioxidant capacity was indicated recently through a link attributed to salicylic acid (Sharma and Dietz 2009). Reactive nitrogen species (RNS) are also important endogenous signals in plant stress; for instance, nitric oxide (NO) can mediate responses to several stimuli. Exogenous NO can protect plants against oxidative stress, acting also in stomatal closure (Neill et al. 2008). The data suggest that NO can induce cGMP production and mitogen-activated protein kinase (MAPK) activity, which are required for abscisic acid (ABA)-induced stomatal closure (Neill et al. 2008).

ROS such as the superoxide radical (O_2^{--}) , hydroxyl radical (HO[•]), hydrogen peroxide (H_2O_2) , and oxygen singlet ($^{1}O_2$) are common products of aerobic metabolism. Their production often leads to a nonspecific oxidation of membrane lipids, carbonylation of proteins, DNA and carbohydrate damage, blocking essential groups, and displacement of essential ions from biomolecules (Schutzendubel and Polle 2002; Mithofer et al. 2004). In fact, during stress situations, ROS production is an essential component of the signals (Gechev et al. 2006; Møller et al. 2007) that plants use to make adjustments in gene expression, cell structure (Shao et al. 2008), and modulation of antioxidant response and programmed cell death (Gechev et al. 2006).

The most important sites of ROS generation and scavenging in plants are the chloroplasts, due to the interaction between electron escape from the photosynthetic electron transfer system and molecular oxygen (Asada 2006; Wu et al. 2008) and mitochondria (Navrot et al. 2007; Wu et al. 2008). In chloroplasts, Cd can induce alterations in organization and shape of the thylakoid membranes with deposition of electron-dense material in double membranes (Ouzounidou et al. 1997; Vitória et al. 2006; Lage-Pinto et al. 2008). Moreover, Cd can cause structural changes in mitochondria; breakdown of the nuclear envelope (Vitória et al. 2006); disintegration of the epidermis; changes in root diameter, cell size, and intercellular spaces (Gratão et al. 2009). These effects may also be found when plants are exposed to other metals such as Cr (Choudhury and Panda 2004; Panda 2007), Ni, Zn (Jin et al. 2008), Cu (Wójcik et al. 2009; Upadhyay and Panda 2009), Pb (Choudhury and Panda 2005; Islam et al. 2008), and Tb (Wang et al. 2009b).

In the last few years, a number of reports have focused on the oxidative stress characterization and antioxidant enzyme response of a wide range of plant species. The effects of many different metals have been published, but all have consistently addressed very similar problems and investigated basic parameters. These kinds of studies are important, as they allow the verification of the sensitivities of different plant species to distinct metals, eventually indicating specific biomarkers to stress situations. However, metalinduced stress studies need new approaches that are likely to increase our understanding of how these elements affect plant metabolism and to identify the modifications that are needed to improve plant adaptation and tolerance (Arruda and Azevedo 2009).

Acquired tolerance can in some cases be observed by growing the plant under increasing metal concentrations, thus indicating the role of antioxidant enzymes and compounds in the adaptation process (Gratão et al. 2008). Chronic exposure of plants is interesting because it can indicate tolerant or bioaccumulator species that will be useful for phytoremediation and rhizoremediation programs to degrade, stabilize, and/or remove soil contaminants. Plant tolerance to metals can occur through a high capacity for sequestration, immobilization or chelation and by exclusion mechanisms (Mora et al. 2009; Válega et al. 2009; Yang et al. 2010). Genes involved in glutathione and phytochelatin biosynthesis have been identified (Sung et al. 2009) and also the contribution of metallothioneins (Guo et al. 2008) and vacuolar membrane transporters (Kawachi et al. 2009) in increasing the response/tolerance to metal stress (Shim et al. 2009). Critical parameters for the use of phytoremediation such as environmental and climatic characteristics, nutrient sustainability, feasible ecosystems, and pollutant characteristics influence the level of technical success for such an approach (McCutcheon and Jorgensen 2008). The rhizosphere plays an important role in phytoremediation, because microorganisms can decisively interfere with and alter metal mobility and availability to plants (Li et al. 2009b; Martínez-Alcalá et al. 2009; Yang et al. 2010). However, more field experiments are necessary to validate the results obtained in the laboratory and greenhouse.

In the field, plants are often exposed to a combination of different stress-causing agents and the response of individual plants species may be unique and should be one of the focuses of future studies (Moffat 2002; Mittler 2006). Molecular and genetical approaches, genome response, transcriptome and proteomic analyses are essential tools to be used more often from now on for the generation of a wide range of information on plant responses to metal stress or any other environmental contaminant.

3 Hormonal Modulation of Stress Signaling

It is not surprising that some of the major components believed to be involved in the responses to stress signaling are the plant hormones. Since these substances regulate almost every step of plant development, it is expected and it has been confirmed that they play an important role in the biochemical program during stress, also influencing the adaptive response.

Although virtually every hormone may be involved in plant defense responses against various biotic and abiotic stresses, the work with brassinosteroid (BR), jasmonic acid (JA), and salicylic acid (SA) has presented new perspectives on stress tolerance studies. For instance, it has been reported that there is an ameliorative influence of BR on drought stress (Behnamnia et al. 2009; Fariduddin et al. 2009), low (Kamuro and Takaysuto 1991) and high (Dhaubhadel et al. 1999) temperature, and salinity (Hayat et al. 2007a; El-Khallal et al. 2009). BR may also reduce the inhibitory effect of metals such as Cd (Anuradha and Rao 2007; Hasan et al. 2008) and Al (Ali et al. 2008), and more recently to have an ameliorative effect on excess manganese-induced oxidative stress in maize (Wang et al. 2009a). JA has been correlated with pathogen defense, but recently it has been shown to be an important component in the mediation of abiotic stresses, such as toxic metals (Maksymiec et al. 2005) and drought (Shan and Liang 2010). SA can also modulate the plant response to several stresses, such as ultraviolet light (Yalpani et al. 1994), toxic metal (Mishra and Chordhuri 1999), chilling (Janda et al. 1999; Kang et al. 2003), heat (Dat et al. 1998), salt (Taşgın et al. 2006), and pathogens (Khaosaad et al. 2007). Recently, Belkhadi et al. (2010) showed that the presoaking of dry flax grains in SA-containing solutions partially protected seedlings from Cd toxicity during the following growth period.

One of the points that had been approached in the above-cited works is that the mutual modifications induced by hormones result in an antioxidant burst, which can explain an adaptive system (Mazorra et al. 2002; Behnamnia et al. 2009). For instance, the activities of the antioxidant enzymes superoxide dismutase (SOD), ascorbate peroxidase (APX), catalase (CAT), and glutathione peroxidases (GPX) were increased in the seedlings of *Raphanus sativus* treated with BR (Anuradha and Rao 2007). In maize (El-Khallal et al. 2009) and tomato (El-Khallal 2007) plants, an increase in the activities of SOD and CAT after pretreatment with BR was also observed. Recently, BR was shown to lead to an increase in the activity of the SOD, CAT, APX, and guaiacol peroxidase (GPOX) in tomato (Ogweno et al. 2008; Behnamnia et al. 2009) as well as the activity of CAT and SOD in *Brassica juncea* (Hayat et al. 2007a, b; Fariduddin et al. 2009). It has been shown that JA can induce an increase in the activities of a wide range of antioxidant enzymes. APX, CAT, glutathione reductase (GR), and SOD were increased with JA treatments in the leaves of *Phaseolus vulgaris* L. var. Top Crop (Clarke et al. 2002). JA was also reported to enhance the antioxidant response of Brassica *napus* roots and shoots, particularly by a higher activity of SOD when compared to the control (Comparot et al. 2002). In other studies, the activities of SOD and CAT have also been shown to increase in JA-treated plants (Hung and Kao 2004; Kumari et al. 2006), as well as in recent reports in which enhanced GR, GPX, APX, and CAT and SOD activities were observed (Cao et al. 2009; Parra-Lobato et al. 2009; Shan and Liang 2010). Equally, activation of antioxidant enzymes induced by the treatment with SA may contribute to its ability to protect plants against stress. SA pretreatment of Zea mays L. caused an increase in GR, SOD, CAT, and APX activities (Janda et al. 1999, 2000; Kang et al. 2003), besides other recent observations in which SOD, CAT, and APX activities were shown to be altered (El-Khallal et al. 2009; Mutlu et al. 2009). Hormonal applications have also been exploited to enhance production of antioxidants in fruits (Kubicka and Zadernowski 2007; Cao et al. 2009), an interesting approach toward the improvement of food quality.

Although in the examples cited earlier it is proposed that exogenous hormones increased the activity of some antioxidant enzymes causing mitigation of oxidative stress, the details of the biochemical and molecular mechanisms involving BR, JA, and SA in these amelioration responses are still unclear. The linear concept in which an antioxidant burst appears as a result of hormone-induced ROS could explain the defense response, although the biochemical pathway from hormone to response seems to be a very complex phenomenon. This complexity is due to four main issues: (1) it has been assumed that hormones act as secondary messengers for the induction of an antioxidant defense system in stressed plants and thereby could effectively scavenge ROS in plants under stress leading to reduced levels (Ogweno et al. 2008; Zhang and Xing 2008); (2) although there are enhanced activities of antioxidant enzymes in plants treated with exogenous hormones, high ROS levels can still be maintained (Hung and Kao 2004; Cao et al. 2009; Parra-Lobato et al. 2009); (3) a wide range of stress-associated genes can be induced by hormones (O'Donnell et al. 2003; Uquillas et al. 2004; Blanco et al. 2005), and therefore become a more acute issue, primarily because the activation of these genes can result in the induction of ROS which can induce stress genes resulting in a complex biochemical network (Moon et al. 2003; Torres et al. 2002; Seo et al. 2007); (4) signaling stress can be accompanied by different forms of biochemical regulation due to exogenous hormones. Thus, when applied at physiological concentrations, these substances may cause a temporary low level of oxidative stress in plants, which acts as a hardening process. This low stress can improve the antioxidative capacity of the plants and help to induce the synthesis of protective compounds. On the other hand, overdosage, repeated or prolonged application can cause irreversible disturbance of plant metabolism. Certainly, these issues fall into the question of how one hormone and the various components of oxidative stress work together interacting with others hormones. The final response to one hormone depends on the crosstalk between the different signaling routes, involving perhaps synergetic or antagonistic interactions inducing a wide spectrum of antistress reactions in plants (Campos et al. 2009).

Considerable attention has been focused on the use of new molecular and genomics tools, which have provided useful information for improving our understanding of the complex interplay between hormones and components of oxidative stress. Over the past decade, several mutants with changed hormone signal perception and transduction have been isolated and well characterized with some of them being useful in helping to improve our understanding of the modulation of ROS signaling by the reactive oxygen gene network of plants. In addition, the cellular localization and coordination of the ROS scavenging pathways of plants, including components involved in mediating the cross-talk between these pathways and hormones have been studied. These often operate synergistically to activate the expression of some defense-related genes after stress, including NO and peptide hormones, such as systemin and hydroxyproline-rich glycopeptides systemins (Glazebrook 2005; Miao and Zentgraf 2007). Additionally, key components of the reactive oxygen gene network have been identified by reverse genetics, and microarray analysis of defense-related genes, which have revealed a significant overlap in the number of genes induced by ROS and hormones (Schenk et al. 2000). Finally, the recent identification of oxidative stress-specific transcription factors (Dietz et al. 2010) represents an additional effort in the elucidation of stress signaling in plants. However, a better understanding of phytohormonemediated plant defense responses is important in designing effective strategies for engineering crops that will be resistant to biotic and abiotic stresses (Overmyer et al. 2000; Mittler et al. 2004; Bari and Jones 2009).

4 The Proteome in the Context of Stress

Dramatic biochemical changes during a wide range of stresses have been shown to be closely related to gene expression (Mousavi and Hotta 2005). Progress in obtaining complete plant genome sequences has not only updated our knowledge but has also provided a substantial platform from which to study plant stress physiology (Cui et al. 2005). Microarray analysis has provided a global examination of gene expression corresponding to stress (Perez-Torres et al. 2009; Soren et al. 2010). It has been shown that a wide range of genes appear to be induced under stress, and the gene expression level for several genes reached a maximum, for example, after cold treatment for 24 h (Rabbani et al. 2003). On 18 cold-responsive rice the other hand, MicroRNAs (miRNAs) were downregulated (Lv et al. 2010). Current data of microarray analysis showed that the RPK1 (receptor-like protein kinase 1), which act as a regulator of ABA, enhances the expression of stress-responsive genes, such as Cor15a, Cor15b, and rd29A, in addition to H₂O₂-responsive genes (Osakabe et al. 2010). Although this gene expression profiling under stress has increased our understanding a great deal, it is still important to know how the transcriptional changes are reflected at the translational level (Cui et al. 2005).

In recent years, proteomic profiling has been used to study the effects of stresses on plants under different conditions. Changes in proteomic profiles at the organelle, cell, organ and tissue levels, induced by several stress have been described in a wide range of plant species and compared to proteomes affected by different physiological conditions (Porubleva et al. 2001). Moreover, the new generation of proteomic techniques facilitates the investigation of the global protein expression profile using efficient protein extraction and identification methods (Cui et al. 2005). This has helped in the identification of specific proteins, elucidation of protein structures and identification of protein functions, proteinprotein interactions, and localization of proteins during stress-signaling cascades.

Today proteomics is boosted by new technologies and equipments/instruments, which have made it a key research field (Agrawal et al. 2006). Oxidative stress proteomic studies have employed the use of 2D-polyacrylamide gel electrophoresis (PAGE) associated with a mass spectrometry (MS) (Agrawal et al. 2006; Rinalducci et al. 2008). The use of these techniques has allowed the recognition of proteins involved in ROS reactions triggered by stresses (Seggara et al. 2007; Ahsan et al. 2008a). This approach has allowed the identification, for example, of predominant proteins related to ROS, and how, what, and why proteins are modified during stress. the Comparative proteomics of salt tolerance revealed

a total of 79 and 32 proteins involved in photosynthesis, energy metabolism, and stress response in *Arabidopsis thaliana* and *Thellungiella halophila*, respectively, showing distinct patterns of protein changes in the two species (Pang et al. 2010).

The answer to these questions had also allowed the understanding of the importance of proteins in ROS and oxidative damage for maintenance of plant function under stress in field condition as well as controlled environments (Salekdeh and Komatsu 2007). Thus, using a proteomic approach, a wide range of reports demonstrated that the expression of various cell compartment proteins was up- or downregulated following exposure to ROS (Shulaev and Oliver 2006). However, ROS can affect the proteome not just by increasing or decreasing the levels of proteins directly or indirectly, but also by modification of protein activity. For example, Irar et al. (2010) compared several shock proteins, enzyme metabolism, and radical scavengers in mature embryos from wheat genotypes, Mahmoudi (salt and drought sensitive) and Om Rabia3 (salt and drought tolerant), suggesting that the protein differential accumulating could be used for the screening of tolerance/sensitivity to drought and salt stress.

Recently, several efforts have been made to generate proteome maps and to investigate the overall response at the proteome level of plants under toxic metal stresses. The results of these studies have shown that plants operate various mechanisms in which proteins play central roles in coping with metal stress (Ahsan et al. 2008a). Thus, a large set of differentially expressed proteins is involved in several cellular functions during metal toxicity. These functions have been analyzed in different species, from algae (Gillet et al. 2006) to trees (Kieffer et al. 2008), and showed that metals, for example, Cd (Rodríguez-Celma et al. 2010), Al (Yang et al. 2007), Cu (Li et al. 2009c), and As (Ahsan et al. 2008a), have a severe effect on proteins involved in photosynthesis and respiration. For example, Visioli et al. (2010) showed the importance of mitochondria and chloroplasts in the response to metal stress, verifying that the proteins which were more abundant in response to Cd were located in

these organelles. Certainly, these stress tolerance mechanisms involve antioxidant molecules, which, based on recent proteomic data, indicate the involvement of enzymes of the ascorbate– glutathione pathway (Lee et al. 2010).

Future progress in proteomics tools will allow the identification and complete characterization of protein markers in a given stressed physiological condition. These findings in association with genomics and metabolomics will result on global cellular responses to oxidative stress, providing data for the biochemical networks involved in functional and/or structural alterations to proteins (Rinalducci et al. 2008).

5 Grafting

Stress signaling is an aspect that still demands more attention for future research. For this purpose, grafting can be a powerful tool to improve our understanding of the stress signaling and specific reponses between plant organs during stress. The grafting technique relies on an old principle and has been extensively used in the horticultural industry for woody species.

Analyses of such plants and their controls may indicate a complex array of long distance or systemic signals, including abiotic stress responses (Turnbull et al. 2002). Recently, the production of grafted plants has been shown to be a useful strategy in increasing the salinity tolerance of *Cucurbita focifolia* (Huang et al. 2010), *Cucumis sativus* L. (Zhen et al. 2010), and salinity resistance in tobacco plants (Ruiz et al. 2005).

However, the use of the grafting technique on plants subjected to toxic metals is rare and only very few studies have been reported. Thus, grafting studies focusing on stress signaling by toxic metals need to be more firmly addressed and explored. For instance, Sugiyama et al. (2007) reported that the seed Cd concentration can be influenced by differences in the translocation of Cd to the seed and in the Cd accumulation capacity of roots among soybean cultivars. In another study, *Solanum lycopersicum* L. cv. *Belladona* F-1 plants were either self-grafted or grafted onto the rootstock "He-Man" showing that cv. *Belladona* was more tolerant to excess Mn than to Mn deficiency in terms of vegetative growth and fruit yield (Savvas et al. 2009). The message is very clear: there is plenty of room for research using this approach which appears to be quite powerful.

6 Cytogenetic

As already discussed, ROS can affect macromolecules and cause severe damage to nucleic acids leading to alterations in the number and structure of chromosomes. This aspect has been poorly explored in plants subjected to metal stress, but the interest in the genetical and cytological effects of toxic compounds was greatly stimulated by the results obtained with cyanide which indicated that hydrogen peroxide was involved in the production of mutations and chromosome aberrations (Kihlman 1957). Furthermore, studies on genotoxicity can be correlated with tissue concentration and plant availability of the toxic metal, having ecotoxicological implications.

A variety of cytogenetic tests have been applied to plants species as bioindicators, evaluating the toxicity and mutagenicity of environmental contaminants such as cyanide, fluoride, organics, and metals (Andrade et al. 2010). Cytotoxic evaluation using root length, weight gain, mitotic index (MI), frequency of micronuclei (MN) and chromosomal aberrations (CAs) was investigated on *Cicer arietinum* L. root tip cells submitted to different concentrations of Pb and Hg ions (Cavusoglu et al. 2009). A high frequency of interphase aberrations was found in Armeria maritima grown in soils with high concentrations of toxic metals or increased salinity (Coulaud et al. 1999). In root tip cells of Helianthus annuus submitted to copper chloride, condensed and sticky chromosomes, scattered chromosomes, and chromatid bridges were observed (Ünceer et al. 2003).

It is now clear that cytogenetical alterations can also be detected in plants submitted to metal toxicity. Cu decreased mitotic index, inhibited meristematic cell proliferation, resulting in changes in morphology of the chromosomes (Posmyk et al. 2008). Cd also caused chromosome aberrations and micronuclei formation in *Allium cepa* (Seth et al. 2008). Finally, it has also been demonstrated that exogenous antioxidants such as anthocyanins limit and prevent the cytogenetical damage induced by Cu in *Vicia faba* (Posmyk et al. 2009).

7 Metallomics

Metal ions are used by biological systems in fundamental processes such as signaling, gene expression, and catalysis and the absence of some metals can result in cellular disorders, whereas the presence of others has frequently been evoked in the context of toxicity (Szpunar 2004). The aim of metallomics is to achieve a better understanding of the function and regulation of metals in biological systems, allowing a comprehensive look at the role of essential and toxic metals. Several original papers have been published on total element determination and, only recently, the research has moved towards elemental speciation analysis, characterizing metals or metalloids in a cell or sample (Muñoz et al. 2009).

The metallomics techniques require the development of versatile measurement tools involved with detection, mapping, and/or quantification of metals associated with proteins in biological systems (Mounicou et al. 2009). Thus, the metallome can give us the information about how the element (metal or metalloid) is distributed among the cellular compartments, in which biomolecule it is incorporated, and the concentrations of the individual metal species present (Szpunar 2004).

Metallomics can elucidate the metal speciation and biological functions of biometals in toxic metal accumulating and hyperaccumulating plant species (Peng and Yang 2006). In contrast to traditional approaches, metallomics allows the investigation of global and multielement interactions in a protein or enzyme system. Although well established, the sample preparation has been subject to change and new strategies have been adopted (Arruda 2007). It is important that the stability of the linkage between the metal and the organic species or the biomolecule is considered. However, the more common methods used for total metal extraction and determination are not able to maintain metal-organic linkages (Magalhães and Arruda 2007). Others methods, such as microwaveassisted extraction (Sussulini et al. 2007), microwave water-assisted extraction (Salgado et al. 2006), and molecularly imprinted solid phase extraction (Xu et al. 2009) have been shown to be efficient.

Two dimensional (2D) electrophoresis such as PAGE, or more recently fluorescence difference gel electrophoresis (DIGE) are particularly useful techniques for a high resolution separation (Timms and Cramer 2008), as well as two dimensional liquid chromatography (Pedrero et al. 2007). After the separation of the proteins/ enzymes, the next step is related to their identification and some techniques like MALDI-QTOF-MS (matrix assisted laser desorption ionization-quadrupole time of flight-mass spectrometry) (Garcia et al. 2009), and ESI-MS (electrospray ionization mass spectrometry) can be applied (Aki and Yanagisawa 2009).

At the last step, some techniques are used for metal identification, as well as sensitivity for metal quantification. Inductively coupled plasma mass spectrometry (ICP-MS) can be used for ultra-sensitive quantification of metal-containing proteins or peptides which are determined down to the low attomole range (Bettmer et al. 2009). ICP-MS can provide higher sensitivity, selectivity, and reliability in molecule tracing (González-Fernández et al. 2008).

Metallomics can be successfully used in traditional biomarker analyses, providing a quantitative analytical approach in proteomics (metalloproteomics), which converts this relatively recent technology to a valuable alternative for biomolecule analysis (González-Fernández et al. 2008).

8 Conclusion and Future Perspective

It has been observed that the majority of the published reports related to metal-induced oxidative stress have shown signs of repetition, by applying similar methods and analysis (Arruda and Azevedo 2009). The measurements of antioxidants as stress markers will remain an essential aspect in assessing the stress response in plants, but other aspects related to an interdisciplinary view should be taken into account, particularly when considering the complexity of the studies that are now required (Arruda and Azevedo 2009).

The information available in this chapter is just an overview of aspects that have not received the attention that they deserve. At the same time, the text provides some new points to the emerging techniques that are available or are being developed, which, eventually may be powerful tools to aid our understanding of the different molecular and cellular mechanisms involved in cell stress responses. Thus, a more comprehensive view has to be considered and must necessarily include studies on gene expression, protein translation, enzyme activity, and metabolite concentrations, all considered simultaneously, wherever possible.

References

- Agency for Toxic Substances and Disease Registry (ATSDR), Agency for Toxic Substances and Disease Registry (ATSDR) (2007) CERCLA priority list of hazardous substances. Department of Health and Human Services, Atlanta, GA, http://www.atsdr.cdc. gov/cercla/07list.html. Accessed 3 Sept 2010
- Agrawal GK, Jwa NS, Iwahashi Y, Yonekura M, Iwahashi H, Rakwal R (2006) Rejuvenating rice proteomics: facts, challenges, and visions. Proteomics 6:5549–5576
- Ahsan N, Lee DG, Alam I, Kim PJ, Lee JJ, Ahn YO, Kwak SS, Lee IJ, Bahk JD, Kang KY, Renaut J, Komats US, Lee BH (2008a) Comparative proteomic study of arsenic-induced differentially expressed proteins in rice roots reveals glutathione plays a central role during As stress. Proteomics 8:3561–3576
- Aki T, Yanagisawa S (2009) Application of rice nuclear proteome analysis to the identification of evolutionarily conserved and glucose-responsive nuclear proteins. J Proteome Res 8:3912–3924
- Ali B, Hasan SA, Hayat S, Yadav S, Fariduddin Q, Ahmad A (2008) A role of brassinosteroids in the amelioration of aluminium stress through antioxidant system in mungbean (*Vigna radiata* L Wilczek). Environ Exp Bot 62:153–159
- Allan RP, Soden BJ (2008) Atmospheric warming and the amplification of precipitation extremes. Science 321:1481–1484

- Andrade LF, Davies LC, Gedraite LS (2010) The effect of cyanide compounds, fluorides, aluminum, and inorganic oxides present in spent pot liner on germination and root tip cells of *Lactuca sativa*. Ecotoxicol Environ Saf 73:626–631
- Anuradha S, Rao SSR (2007) The effect of brassinosteroids on radish (*Raphanus sativus* L.) seedlings growing under cadmium stress. Plant Soil Environ 53:465–472
- Arruda MAZ (2007) Trends in sample preparation. Nova Science Pub. Inc., New York, NY, 304pp
- Arruda MAZ, Azevedo RA (2009) Metallomics and chemical speciation: towards a better understanding of metalinduced stress in plants. Ann Appl Biol 155:301–307
- Asada K (2006) Production and scavenging of reactive oxygen species in chloroplasts and their functions. Plant Physiol 141:391–396
- Azevedo JA, Azevedo RA (2006) Heavy metals and oxidative stress: where do we go from here? Commun Biometry Crop Sci 1:135–138
- Bari R, Jones JD (2009) Role of plant hormones in plant defence responses. Plant Mol Biol 69:473–488
- Becana M, Moran JF, Iturbe-Ormaetxe I (1998) Irondependent oxygen free radical generation in plants subjected to environmental stress: toxicity and antioxidant protection. Plant Soil 201:137–147
- Behnamnia M, Kalantari KM, Rezanejad F (2009) Exogenous application of brassinosteroid alleviates drought-induced oxidative stress in *Lycopersicon esculentum* L. Gen Appl Plant Physiol 35:22–34
- Belkhadi A, Hediji H, Abbes Z, Nouairi I, Barhoumi Z, Zarrouk M, Chaibi W, Djebali W (2010) Effects of exogenous salicylic acid pre-treatment on cadmium toxicity and leaf lipid content in *Linum usitatissimum* L. Ecotoxicol Environ Saf 73:51004–51011
- Benavides MP, Gallego SM, Tomaro ML (2005) Cadmium toxicity in plants. Braz J Plant Physiol 17:21–34
- Bettmer J, Bayón MM, Encinar JR, Sánchez MLF, Campa MRF, Sanz-Medel A (2009) The emerging role of ICP-MS in proteomic analysis. J Proteomics 72:989–1005
- Blanco F, Garreton V, Frey N, Dominguez C, Perez-Acle T, van Der Straeten D, Jordana X, Holuigue L (2005) Identification of NPR1-dependent genes early induced by salicylic acid treatment in Arabidopsis. Plant Mol Biol 59:927–944
- Bradl HB (2004) Adsorption of heavy metal ions on soils and soils constituents. J Colloid Interface Sci 277:1–18
- Campbell JL, Rustad LE, Boyer EW, Christopher SF, Driscoll CT, Fernandez IJ, Groffman PM, Houle D, Kiekbusch J, Magill AH, Mitchell MJ, Ollinger SV (2009) Consequences of a climate change for biogeochemical cycling in forests of northeastern North America. Can J Forest Res 39:264–284
- Campos ML, de Almeida M, Rossi ML, Martinelli AP, Junior CGL, Figueira A, Rampelotti-Ferreira FT, Vendramim JD, Benedito VA, Peres LEP (2009) Brassinosteroids interact negatively with jasmonates in the formation of anti-herbivory traits in tomato. J Exp Bot 60:4347–4361

- Cao SF, Zheng YH, Wang KT, Jin P, Rui HJ (2009) Methyl jasmonate reduces chilling injury and enhances antioxidant enzyme activity in postharvest loquat fruit. Food Chem 115:1458–1463
- Cavusoglu K, Ergene A, Yalcin E, Tan S, Cavusoglu K, Yapar K (2009) Cytotoxic effects of lead and mercury ions on root tip cells of *Cicer arietinum* L. Fresen Environ Bull 18:1654–1661
- Choudhury S, Panda SK (2004) Induction of oxidative stress and ultrastructural changes in moss *Taxithelium nepalense* (Schwaegr.) Broth. under lead and arsenic phytotoxicity. Current Sci 87:342–348
- Choudhury S, Panda SK (2005) Toxic effects, oxidative stress and ultrastructural changes in moss *Taxithelium nepalense* (Schwaegr.) Broth. under chromium and lead phytotoxicity. Water Air Soil Pollut 167:73–90
- Clarke SF, Guy PL, Burritt DJ, Jameson PE (2002) Changes in the activities of antioxidant enzymes in response to virus infection and hormone treatment. Physiol Plant 114:157–164
- Comparot SM, Graham CM, Reid DM (2002) Methyl jasmonate elicits a differential antioxidant response in light-and dark-grown canola (*Brassica napus*) roots and shoots. Plant Growth Regul 38:21–30
- Coulaud J, Barghi N, Lefebvre C, Siljak-Yakovlev S (1999) Cytogenetic variation in populations of *Armeria maritima* (Mill.) Willd. in relation to geographical distribution and soil stress tolerances. Can J Bot Revue Canadienne de Bot 77:673–685
- Cui S, Huang F, Wang J, Cheng Y, Liu J (2005) A proteomic analysis of cold stress responses in rice seedlings. Proteomics 5:3162–3172
- Dat JF, Lopez-Delgado H, Foyer CH, Scott IM (1998) Parallel changes in H₂O₂ and catalase during thermotolerance induced by salicylic acid or heat acclimation in mustard seedlings. Plant Physiol 116:1351–1357
- Dhaubhadel S, Chaudhary S, Dobinson KF, Krishna P (1999) Treatment of 24-epibrassinolide, a brassinosteroid, increases the basic thermotolerance of *Brassica napus* and tomato seedlings. Plant Mol Biol 40:332–342
- Dietz KJ, Vogel MO, Viehhauser A (2010) AP2/EREBP transcription factors are part of gene regulatory networks and integrate metabolic, hormonal and environmental signals in stress acclimation and retrograde signaling. Protoplasma 245:3–14
- El-Khallal SM (2007) Induction and modulation of resistance in tomato plants against *Fusarium* wilt disease by bioagent fungi (arbuscular mycorrhiza) and/or hormonal elicitors (jasmonic acid and Salicylic acid):
 2-changes in the antioxidant enzymes, phenolic compounds and pathogen related-proteins. Aust J Basic Appl Sci 1:717–732
- El-Khallal SM, Hathout TA, El Raheim A, Ahsour A, Kerrit AA (2009) Brassinolide and salicylic acid induced antioxidant enzymes, hormonal balance and protein profile of maize plants grown under salt stress. Res J Agri Biol Sci 5:391–402
- Fariduddin Q, Khanam S, Hasan SA, Ali B, Hayat S, Ahmad A (2009) Effect of 28-homobrassinolide on
the drought stress-induced changes in photosynthesis and antioxidant system of *Brassica juncea* L. Acta Physiol Plant 31:889–897

- Garcia JS, Souza GHMF, Eberlin MN, Arruda MAZ (2009) Evaluation of metal-ion stress in sunflower (*Helianthus annuus* L.) leaves through proteomic changes. Metallomics 1:107–113
- Gechev TS, Van Breusegem F, Stone JM, Denev I, Laloi C (2006) Reactive oxygen species as signals that modulate plant stress responses and programmed cell death. Bioessays 28:1091–1101
- Gillet S, Decottignies P, Chardonnet S, Le Maréchal P (2006) Cadmium response and redoxin targets in *Chlamydomonas reinhardtii*: a proteomic approach. Photosynth Res 89:201–211
- Glazebrook J (2005) Contrasting mechanisms of defence against biotrophic and necrotrophic pathogens. Annu Rev Phytopathol 43:205–227
- Gomes-Junior RA, Gratão PL, Gaziola SA, Mazzafera P, Lea PJ, Azevedo RA (2007) Selenium-induced oxidative stress in coffee cell suspension cultures. Funct Plant Biol 34:449–456
- González-Fernández M, García-Barrera T, Jurado J, Prieto-Álamo MJ, Pueyo C, López-Barea J, Gómez-Ariza JL (2008) Integrated application of transcriptomics, proteomics, and metallomics in environmental studies. Pure Appl Chem 80:2609–2626
- Gratão PL, Polle A, Lea PJ, Azevedo RA (2005) Making the life of heavy metal-stressed plants a little easier. Funct Plant Biol 32:481–494
- Gratão PL, Monteiro CC, Antunes AM, Peres LEP, Azevedo RA (2008) Acquired tolerance of tomato (*Lycopersicon esculentum* cv. Micro–Tom) plants to cadmium-induced stress. Ann Appl Biol 153:321–333
- Gratão PL, Monteiro CC, Rossi ML, Martinelli AP, Peres LEP, Medici LO, Lea PJ, Azevedo RA (2009) Differential ultrastructural changes in tomato hormonal mutants exposed to cadmium. Environ Exp Bot 67:387–394
- Guo WJ, Meetam M, Goldsbrough PB (2008) Examining the specific contributions of individual Arabidopsis metallothioneins to copper distribution and metal tolerance. Plant Physiol 146:1697–1706
- Hansch R, Mendel RR (2009) Physiological functions of mineral micronutrients (Cu, Zn, Mn, Fe, Ni, Mo, B, Cl). Curr Opin Plant Biol 12:259–266
- Harmon SM, Wyatt DE (2008) Evaluation of post-Katrina flooded soils for contaminants and toxicity to the soil invertebrates *Eisenia fetida* and *Caenorhabditis elegans*. Chemosphere 70:1857–1864
- Hasan SA, Hayat S, Ali B, Ahmad A (2008) 28-Homobrassinolide protects chickpea (*Cicer arie-trinum*) from cadmium toxicity by stimulating antioxidants. Environ Pollut 151:60–66
- Hayat S, Ali B, Aiman HS, Ahmad A (2007a) Brassinosteroids enhanced antioxidants under cadmium stress in *Brassica juncea*. Environ Exp Bot 60:33–41
- Hayat S, Ali B, Hasan SA, Ahmad A (2007b) Effect of 28-homobrassinolide on salinity-induced changes in *Brassica juncea*. Turk J Biol 31:141–146

- Horst WJ, Wang Y, Eticha D (2010) The role of the root apoplast in aluminium-induced inhibition of root elongation and in aluminium resistance of plants: a review. Ann Bot 106:185–197
- Huang Y, Bie Z, He S, Bin Hua B, Zhen A, Liu Z (2010) Improving cucumber tolerance to major nutrients induced salinity by grafting onto *Cucurbita ficifolia*. Environ Exp Bot 69:32–38
- Hung KT, Kao CH (2004) Nitric oxide acts as an antioxidant and delays methyl jasmonate-induced senescence of rice leaves. J Plant Physiol 161:43–52
- Irar S, Brini F, Goday A, Masmoudi K, Pages M (2010) Proteomic analysis of wheat embryos with 2-DE and liquid-phase chromatography (ProteomeLab PF-2D) – a wider perspective of the proteome J. Proteomics 73:1707–1721
- Islam E, Liu D, Li T, Yang X, Jin X, Mahmood Q, Tian S, Li J (2008) Effect of Pb toxicity on leaf growth, physiology and ultrastructure in the two ecotypes of *Elsholtzia argyi*. J Hazard Mater 154:914–926
- Janda T, Szalai G, Tari I, Páldi E (1999) Hydroponic treatment with salicylic acid decreases the effects of chilling injury in maize (*Zea mays L.*) plants. Planta 208:175–180
- Janda T, Szalai G, Antunovics Z, Horváth E, Páldi E (2000) Effect of benzoic acid and aspirin on chilling tolerance and photosynthesis in young maize plants. Maydica 45:29–33
- Jin XF, Yang XE, Islam E, Liu D, Mahmood Q, Li H, Li J (2008) Ultrastructural changes, zinc hyperaccumulation and its relation with antioxidants in two ecotypes of *Sedum alfredii* Hance. Plant Physiol Biochem 46:997–1006
- Kamuro Y, Takaysuto S (1991) Capability for problems of practical uses of brassinosteroids. In: Cutler HG, Yokota T, Adam G (eds) Brassinosteroids chemistry, bioactivity and application, vol 474, ACS Symposium Series. American Chemica Society, Washington, DC, pp 292–297
- Kang G, Wang C, Sun G, Wang Z (2003) Salicylic acid changes activities of H_2O_2 -metabolizing enzymes and increase the chilling tolerance of banana seedlings. Environ Exp Bot 50:9–15
- Kawachi M, Kobae Y, Mori H, Tomioka R, Lee Y, Maeshima M (2009) A mutant strain Arabidopsis thaliana that lacks vacuolar membrane zinc transporter MTP1 revealed the latent tolerance to excessive zinc. Plant Cell Physiol 50:1156–1170
- Khaosaad T, Garcia-Garrido JM, Steinkellner S, Vierheilig H (2007) Take-all disease is systemically reduced in roots of mycorrhizal barley plants. Soil Biol Biochem 39:727–734
- Kieffer P, Dommes J, Hoffmann L, Hausman JF, Renaut J (2008) Quantitative changes in protein expression of cadmium exposed poplar plants. Proteomics 8:2514–2530
- Kihlman BA (1957) The production of chromosome aberrations by cyanide and other heavy metal complexing agents. J Biophys Biochem Cytol 3:363–380
- Kopittke PM, Blamey FPC, Asher CJ, Menzies NW (2010) Trace metal phytotoxicity in solution culture: a review. J Exp Bot 61:945–954

- Kubicka E, Zadernowski R (2007) Enhanced jasmonate biosynthesis in plants and possible implications for food quality. Acta Aliment 36:455–469
- Kumari GJ, Reddy AM, Naik ST, Kumar SG, Prasanthi J, Sriranganayakulu G, Reddy PC, Sudhakar C (2006) Jasmonic acid induced changes in protein pattern, antioxidative enzyme activities and peroxidase isoenzymes in peanut seedlings. Biol Plant 50:219–226
- Lage-Pinto F, Oliveira JG, da Cunha M, Souza CMM, Rezende CE, Azevedo RA, Vitória AP (2008) Chlorophyll *a* fluorescence and ultrastructural changes in chloroplast of water hyacinth as indicators of environmental stress. Environ Exp Bot 64:307–313
- Lane TW, Saito MA, George GN, Pickering IJ, Prince RC, Morel FMM (2005) A cadmium enzyme from a marine diatom. Nature 435:42
- Lee K, Bae DW, Kim SH, Han HJ, Liu X, Park HC, Lim CO, Lee SY, Chung WS (2010) Comparative proteomic analysis of the short-term responses of rice roots and leaves to cadmium. J Plant Physiol 167:161–168
- Li F, Shi J, Shen C, Chen G, Hu S, Chen Y (2009a) Proteomic characterization of copper stress response in *Elsholtzia splendens* roots and leaves. Plant Mol Biol 71:251–263
- Li HF, Gray C, Mico C, Zhao FJ, McGrath SP (2009b) Phytotoxicity and bioavailability of cobalt to plants in a range of soils. Chemosphere 75:979–986
- Li J, He M, Han W, Gu Y (2009c) Availability and mobility of heavy metal fractions related to the characteristics of the coastal soils developed from alluvial deposits. Environ Monit Assess 158:459–469
- Liu J, Qian M, Cai G, Zhu Q, Wong MH (2007) Variations between rice cultivars in root secretion of organic acids and the relationship with plant cadmium uptake. Environ Geochem Health 29:189–195
- Lv DK, Bai X, Li Y, Ding XD, Ge Y, Cai H, Ji W, Wu NLH, Zhu YM (2010) Profiling of cold-stress-responsive miRNAs in rice by microarrays. Gene 459:39–47
- Magalhães CS, Arruda MAZ (2007) Sample preparation for metalloprotein analysis: a case study using horse chestnuts. Talanta 71:1958–1963
- Maksymiec W, Wianowska D, Dawidowicz AL, Radkiewicz S, Mardarowicz M, Krupa Z (2005) The level of jasmonic acid in *Arabidopsis thaliana* and *Phaseolus coccineus* plants under heavy metal stress. J Plant Physiol 162:1338–1346
- Martínez-Alcalá I, Clemente R, Bernal MP (2009) Metal availability and chemical properties in the rhizosphere of *Lupinus albus* L. growing in a high-metal calcareous soil. Water Air Soil Pollut 201:283–293
- Mazorra LM, Nunez M, Hechavarria M, Coll F, Sanchez-Blanco MJ (2002) Influence of brassinosteroids on antioxidant enzymes activity in tomato under different temperatures. Biol Plant 45:593–596
- McCutcheon SC, Jorgensen SE (2008) Phytoremediation. In: Jorgensen SE, Fath B (eds) Encyclopedia of ecology. Elsevier Science BV, Amsterdam, Netherlands, pp 2751–2766
- Miao Y, Zentgraf U (2007) The antagonist function of Arabidopsis WRKY53 and ESR/ESP in leaf senescence

is modulated by the jasmonic and salicylic acid equilibrium. Plant Cell 19:819–830

- Mishra A, Chordhuri MA (1999) Effects of salicylic acid on heavy metal-induced membrane deterioration mediated by lipoxygenase in rice. Biol Plant 42:409–415
- Mithofer A, Schulze B, Boland W (2004) Biotic and heavy metal stress response in plants: evidence for common signals. FEBS Lett 566:1–5
- Mittler R (2006) Abiotic stress, the field environment and stress combination. Trends Plant Sci 11:1360–1385
- Mittler R, Vanderauwera S, Gollery M, Van Breusegem F (2004) The reactive oxygen gene network of plants. Trends Plant Sci 9:490–498
- Moffat AS (2002) Finding new ways to protect droughtstricken plants. Science 296:1226–1229
- Møller IM, Jensen PE, Hansson A (2007) Oxidative modifications to cellular components in plants. Annu Rev Plant Biol 58:459–481
- Moon H, Lee B, Choi G, Shin D, Prasad DT, Lee O, Kwak SS, Kim DH, Nam J, Bahk J, Hong JC, Lee SY, Cho MJ, Lim CO, Yun DJ (2003) NDP kinase 2 interacts with two oxidative stress-activated MAPKs to regulate cellular redox state and enhances multiple stress tolerance in transgenic plants. Proc Natl Acad Sci USA 100:358–363
- Mora ML, Rosas A, Ribera A, Rengel Z (2009) Differential tolerance to Mn toxicity in perennial ryegrass genotypes: involvement of antioxidative enzymes and root exudation of carboxylates. Plant Soil 320:79–89
- Mounicou S, Szpunar J, Lobinski R (2009) Metallomics: the concept and methodology. Chem Soc Rev 38:1119–1138
- Mousavi A, Hotta Y (2005) Glycine-rich proteins: a class of novel proteins. Appl Biochem Biotech 120:169–174
- Muñoz AS, Kubachka K, Wrobel K, Corona FG, Yathavakilla SKV, Mutlu S, Atici Ö, Nalbantoglu B (2009) Effect of salicylic acid and salinity on apoplastic antioxidant enzymes in two wheat cultivars differing in salt tolerance. Biol Plant 53:334–338
- Navrot N, Rouhier N, Gelhaye E, Jacquot JP (2007) Reactive oxygen species generation and antioxidant systems in plant mitochondria. Physiol Plant 129:185–195
- Neill S, Barros R, Bright J, Desikan R, Hancock J, Harrison J, Morris P, Ribeiro D, Wilson I (2008) Nitric oxide, stomatal closure and abiotic stress. J Exp Bot 59:165–176
- O'Donnell PJ, Schmetz E, Block A, Miersch O, Wasternach C, Jones JB, Klee HJ (2003) Multiple hormones act sequentially to mediate a susceptible tomato pathogen defense response. Plant Physiol 133:1181–1189
- Ogweno JO, Song XS, Shi K, Hu WH, Mao WH, Zhou YH, Yu JQ, Nogués S (2008) Brassinosteroids alleviate heat-induced inhibition of photosynthesis by increasing carboxylation efficiency and enhancing antioxidant systems in *Lycopersicon esculentum*. J Plant Growth Regul 27:49–54
- Osakabe Y, Mizuno S, Tanaka H, Maruyama K, Osakabe K, Todaka D, Fujita Y, Kobayashi M, Shinozaki K, Yamaguchi-Shinozaki K (2010) Overproduction of the membrane-bound receptor-like protein kinase 1,

RPK1, enhances abiotic stress tolerance in Arabidopsis. J Biol Chem 285:9190–9201

- Ouzounidou G, Moustakas M, Eleftheriou EP (1997) Physiological and ultrastructural effects of cadmium on wheat (*Triticum aestivum* L.) leaves. Arch Environ Contam Toxicol 32:154–160
- Overmyer K, Tuominen H, Kettunen R, Betz C, Langebartels C, Sandermann H Jr, Kangasjärvi J (2000) Ozone-sensitive Arabidopsis *rcd1* mutant reveals opposite roles for ethylene and jasmonate signaling pathways in regulating superoxide-dependent cell death. Plant Cell 12:1849–1862
- Panda SK (2007) Chromium-mediated oxidative stress and ultrastructural changes in root cells of developing rice seedlings. J Plant Physiol 164:1419–1428
- Panda SK, Choudhury S (2005) Chromium stress in plants. Braz J Plant Physiol 17:95–102
- Pang QY, Chen SX, Dai SJ, Chen YZ, Wang Y, Yan XF (2010) Comparative proteomics of salt tolerance in *Arabidopsis thaliana* and *Thellungiella halophila*. J Proteome Res 9:2584–2599
- Parmesan C (2006) Ecological and evolutionary responses to recent climate change. Annu Rev Ecol Evol Syst 37:637–669
- Parra-Lobato MC, Fernandez-Garcia N, Olmos E, Alvarez-Tinaut MC, Gomez-Jimenez MC (2009) Methyl jasmonate-induced antioxidant defence in root apoplast from sunflower seedlings. Environ Exp Bot 66:9–17
- Patra M, Bhowmik N, Bandopadhyay B, Sharma A (2004) Comparison of mercury, lead and arsenic with respect to genotoxic effects on plant systems and the development of genetic tolerance. Environ Exp Bot 52:199–223
- Pedrero Z, Encinar JR, Madrid Y, Cámara C (2007) Identification of selenium species in selenium-enriched *Lens esculenta* plants by using two-dimensional liquid chromatography–inductively coupled plasma mass spectrometry and [77Se]selenomethionine selenium oxide spikes. J Chromatogr A 1139:247–253
- Peng HY, Yang X (2006) Analytical techniques for metalorganic ligand and metallomics study. Chin J Anal Chem 34:1190–1196
- Perez-Torres E, Paredes M, Polanco V, Becerra V (2009) Gene expression analysis: a way to study tolerance to abiotic stresses in crops species. Chilean J Agric Res 69:260–269
- Porcal P, Koprivnjak JF, Molot LA, Dillon PJ (2009) Humic substances – part 7: the biogeochemistry of dissolved organic carbon and its interactions with climate change. Environ Sci Pollut Res Int 16:714–726
- Porubleva L, Velden KV, Kothari S, Oliver DJ, Chitnis PR (2001) The proteome of maize leaves: use of gene sequences and expressed sequence tag data for identification of proteins with peptide mass fingerprints. Electrophoresis 22:1724–1738
- Posmyk MM, Kontek R, Janas KM (2008) Red cabbage extract limits copper stress injury in meristematic cells of *Vicia faba*. Acta Physiol Plant 30:481–491

- Posmyk MM, Janas KM, Kontek R (2009) Red cabbage anthocyanin extract alleviates copper-induced cytological disturbances in plant meristematic tissue and human lymphocytes. Biometals 22:479–490
- Prasad MNV, Strzalka K (eds) (2002) Physiology and biochemistry of metal toxicity and tolerance in plants. Kluwer, Dordrecht, 460p
- Rabbani MA, Maruyama K, Abe H, Khan MA, Katsura K, Ito Y, Yoshiwara K, Seki M, Shinozaki K, Yamaguchi-Shinozaki K (2003) Monitoring expression profiles of rice genes under cold, drought, and high-salinity stresses and abscisic acid application using cDNA microarray and RNA get-blot analyses. Plant Physiol 133:1755–1767
- Rinalducci S, Murgiano L, Zolla L (2008) Redox proteomics: basic principles and future perspectives for the detection of protein oxidation in plants. J Exp Bot 59:3781–3801
- Rodríguez-Celma J, Rellán-Álvarez R, Abadía A, Abadía J, López-Millán AF (2010) Changes induced by two levels of cadmium toxicity in the 2-DE protein profile of tomato roots. J Proteomics 73:1694–1706
- Ruiz JM, Blasco B, Rivero RM, Romero L (2005) Nicotine-free and salt-tolerant tobacco plants obtained by grafting to salinity-resistant rootstocks of tomato. Physiol Plant 124:465–475
- Salekdeh GH, Komatsu S (2007) Crop proteomics: aim at sustainable agriculture of tomorrow. Proteomics 7:2976–2996
- Salgado SG, Nieto MAQ, Simón MMB (2006) Determination of soluble toxic arsenic species in alga samples by microwave-assisted extraction and high performance liquid chromatography-hydride generation-inductively coupled plasma-atomic emission spectrometry. J Chromatogr A 1129:54–60
- Savvas D, Papastavrou D, Ntatsi G, Ropokis A, Olympios C, Hartmann H, Schwarz D (2009) Interactive effects of grafting and manganese supply on growth, yield, and nutrient uptake by tomato. Hortscience 44:1178–1982
- Schenk PM, Kazan K, Wilson I, Anderson JP, Richmond T, Somerville SC, Manners JM (2000) Coordinated plant defense responses in Arabidopsis revealed by microarray analysis. Proc Natl Acad Sci USA 97:11655–11660
- Schutzendubel A, Polle A (2002) Plant response to abiotic stresses: heavy metal-induced oxidative stress and protection by mycorrhization. J Exp Bot 53:1351–1365
- Seggara G, Casanova E, Bellido D, Odena MA, Oliveira E, Trillas I (2007) Proteome, salicylic acid, and jasmonic acid changes in cucumber plants inoculated with *Trichoderma asperellum* strain T34. Proteomics 7:3943–3952
- Seo H, Li J, Lee S, Yu J, Kim K, Lee S, Lee I, Paek N (2007) The hypernodulating *nts* mutation induces jasmonate synthetic pathway in soybean leaves. Mol Cells 24:185–193
- Seth CS, Misra V, Chauhan LKS, Singh RR (2008) Genotoxicity of cadmium on root meristem cells of

Allium cepa: cytogenetic and Comet assay approach. Ecotoxicol Environ Saf 71:711–716

- Shan C, Liang Z (2010) Jasmonic acid regulates ascorbate and glutathione metabolism in *Agropyron cristatum* leaves under water stress. Plant Sci 178:130–139
- Shanker AK, Djanaguiraman M, Sudhagar R, Chandrashekar CN, Pathmanabhan G (2004) Differential antioxidative response of ascorbate glutathione pathway enzymes and metabolites to chromium speciation stress in green gram (*Vigna radiata* (L.) R.Wilczek. cv CO 4) roots. Plant Sci 166:1035–1043
- Shao HB, Chu LY, Lu ZH, Kang CM (2008) Primary antioxidant free radical scavenging and redox signaling pathways in higher plant cells. Int J Biol Sci 4:8–14
- Sharma SS, Dietz KJ (2006) The significance of amino acids acid-derived molecules in plant responses and adaptation to heavy metal stress. J Exp Bot 57:711–726
- Sharma SS, Dietz KJ (2009) The relationship between metal toxicity and cellular redox imbalance. Trends Plant Sci 14:43–50
- Shim D, Hwang JU, Lee J, Lee S, Choi Y, An G, Martinoia E, Lee Y (2009) Transcription factor HsfA4a confer cadmium tolerance in wheat and rice. Plant Cell 21:4031–4343
- Shulaev V, Oliver DJ (2006) Metabolic and proteomic markers for oxidative stress. New tools for reactive oxygen species research. Plant Physiol 141:367–372
- Soren KR, Ali K, Tyagi V, Tyagi A (2010) Recent advances in molecular breeding of drought tolerance in rice (*Oryza sativa* L.). Indian J Biotechnol 9:233–251
- Sugiyama M, Ae N, Arao T (2007) Role of roots in differences in seed cadmium concentration among soybean cultivars – proof by grafting experiment. Plant Soil 295:1–11
- Sung DL, Kim TH, Komives EA, Mendonza-Cózatl DG, Schroeder JI (2009) ARS5 is a component of the 26S proteasome complex, and negatively regulates thiol biosynthesis and arsenic tolerance in Arabidopsis. Plant J 59:802–813
- Sussulini A, Garcia JS, Arruda MAZ (2007) Microwaveassisted decomposition of polyacrylamide gels containing metalloproteins using mini-vials: an auxiliary strategy for metallomics studies. Anal Biochem 361:146–148
- Szpunar J (2004) Metallomics: a new frontier in analytical chemistry. Anal Bioanal Chem 378:54–56
- Taşgın E, Atici Ö, Nalbantoglu B, Popova LP (2006) Effects of salicylic acid and cold treatments on protein levels and on the activities of antioxidant enzymes in the apoplast of winter wheat leaves. Phytochemistry 67:710–715
- Timms JF, Cramer R (2008) Difference gel electrophoresis. Proteomics 8:4886–4897
- Tolrà R, Pongrac P, Poschenrieder C, Vogel-Mikus K, Regvar M, Barcelo J (2006) Distinctive effects of cadmium on glucosinolate profiles in Cd hyperaccumulator *Thlaspi praecox* and non-hyperaccumulator *Thlaspi* arvense. Plant Soil 288:333–341
- Torres MA, Dangl JL, Jones JD (2002) Arabidopsis gp91phox homologues AtrbohD and AtrbohF are

required for accumulation of reactive oxygen intermediates in the plant defense response. Proc Natl Acad Sci USA 99:517–522

- Turnbull CGN, Booker JPB, Leyser HMO (2002) Micrografting techniques for testing long-distance signalling in *Arabidopsis*. Plant J 32:255–262
- Ünceer H, Ayaz S, Beyazğlum O, Şentürk E (2003) Cytogenetic effects of copper chloride on the root tip cells of *Helianthus annuus* L. Turk J Biol 27:43–46
- Upadhyay RK, Panda KS (2009) Copper-induced growth inhibition, oxidative stress and ultrastructural alterations in freshly grown water lettuce (*Pistia stratiotes* L.). C R Biol 332:623–632
- Uquillas C, Letelier I, Blanco F, Jordana X, Holuigue L (2004) NPR1-independent activation of immediate early salicylic acid-responsive genes in Arabidopsis. Mol Plant Microbe Interact 17:34–42
- Válega M, Lima AIG, Figueira EMAP, Pereira E, Pardal MA, Duarte AC (2009) Mercury intracellular partitioning and chelation in a salt marsh plant, *Halimione portulacoides* (L.) Aellen: strategies underlying tolerance in environmental exposure. Chemosphere 74:530–536
- Valko M, Morris H, Cronin MT (2005) Metals, toxicity and oxidative stress. Curr Med Chem 12:1161–1208
- Visioli G, Marmiroli M, Marmiroli N (2010) Twodimensional liquid chromatography technique coupled with mass spectrometry analysis to compare the proteomic response to cadmium stress in plants. J Biomed Biotechnol 201:10p
- Vitória AP, da Cunha M, Azevedo RA (2006) Ultrastructural changes of radish leaf exposed to cadmium. Environ Exp Bot 58:47–52
- Wang HH, Feng T, Peng XX, Yan ML, Zhou PL, Tang XK (2009a) Ameliorative effects of brassinosteroid on excess manganese-induced oxidative stress in Zea mays L. leaves. Agric Sci China 8:1063–1074
- Wang L, Zhou Q, Huang X (2009b) Photosynthetic responses to heavy metal terbium stress in horseradish leaves. Chemosphere 77:19–25
- Wójcik M, Pawlikowska-Palenga B, Tukiendorf A (2009) Physiological and ultrastructural changes in *Arabidopsis thaliana* as affected by changed GHS level and Cu excess. Russian J Plant Physiol 56:820–829
- Wu WH, Song XS, Shi K, Xia XJ, Zhou YH, Yu JQ (2008) Changes in electron transport, superoxide dismutase and ascorbate peroxidase isoenzymes in chloroplasts and mitochondria of cucumber leaves as influenced by chiling. Photosynthetica 46:581–588
- Xu Z, Chen S, Huang W, Fang G, Pingzhu H, Wang S (2009) Study on an on-line molecularly imprinted solid-phase extraction coupled to high-performance liquid chromatography for separation and determination of trace estrone in environment. Anal Bioanal Chem 393:1273–1279
- Yalpani N, Enyedi AJ, Leon J, Raskin I (1994) Ultraviolet light and ozone stimulate accumulation of salicylic acid, pathogen-related proteins and virus resistance in tobacco. Planta 193:372–376

- Yang Q, Wang Y, Zhang J, Shi W, Qian C, Peng X (2007) Identification of aluminum-responsive proteins in rice roots by a proteomic approach: cysteine synthase as a key player in Al response. Proteomics 7:737–749
- Yang J, Ma Z, Ye Z, Guo X, Qiu R (2010) Heavy metal (Pb, Zn) uptake and chemical changes in rhizosphere soils of four wetland plants with different radial oxygen loss. J Environ Sci 22:696–702
- Zhang LR, Xing D (2008) Methyl jasmonate induces production of reactive oxygen species and alterations in mitochondrial dynamics that precede photosynthetic dysfunction and subsequent cell death. Plant Cell Physiol 49:1092–1111
- Zhen A, Bie Z, Huang Y, Liu Z, Li Q (2010) Effects of scion and rootstock genotypes on the anti-oxidant defense systems of grafted cucumber seedlings under NaCl stress. Soil Sci Plant Nutr 56:263–271

Sulfur in the Alleviation of Cadmium-Induced Oxidative Stress in Plants

20

Noushina Iqbal, Nafees A. Khan, Md. Iqbal R. Khan, Rahat Nazar, Asim Masood, and Shabina Syeed

Abstracts

The toxicity of cadmium (Cd) is an emerging environmental problem that has attracted the attention of plant scientists all over the world. It deteriorates soil, plant and human health. Researchers have focused their attention in evolving strategies to reduce its toxicity at cellular, molecular and/or whole plant level. Sulfur is an important plant nutrient that takes part in plant metabolism and provides vigor to plants under stressful environments. This nutrient element could be used in agricultural system for reducing Cd toxicity and increasing sustainability. Sulfur uptake results in the formation of the first stable product cysteine through a cascade of enzymatic reactions. The formation of cysteine leads to the synthesis of glutathione, a nonenzymatic antioxidant known to be involved in Cd detoxification either through quenching reactive oxygen species or formation of phytochelatin that binds Cd and sequester it into vacuole. Manipulation of sulfurassimilating enzymes, cysteine, glutathione and/or phytochelatin content could possibly lead to Cd detoxification. The present work gives insight into the role of sulfur in the alleviation of Cd stress.

Keywords

Sulfur • Cadmium toxicity • Cadmium tolerance • ROS • Antioxidants

• Nutrient availability • Sulfur sink

1 Introduction

Plants experience a range of abiotic stress factors such as drought, cold, salinity, temperature and heavy metals during their life time, which cause extensive loss to agricultural productivity worldwide (Boyer 1982; Bray et al. 2000; Peters et al. 2004). Heavy metal toxicity in particular is one of the major abiotic stresses that lead to reduction

N. Iqbal (⊠) • N.A. Khan • Md.I.R. Khan • R. Nazar
• A. Masood • S. Syeed
Department of Botany, Aligarh Muslim University, Aligarh 202002, Uttar Pradesh, India
e-mail: naushina.iqbal@gmail.com in health and vigor of plants, and also of animals through food chain. The growth of anthropogenic activities associated with development has added high concentrations of heavy metals in agricultural soils and large magnitude to environmental pollution (Sanita di Toppi and Gabbrielli 1999; Rao and Reddy 2008). Heavy metals such as arsenic, cadmium, cobalt, copper, nickel, zinc, and chromium are phytotoxic either at all concentrations or above certain threshold levels. Some of these heavy metals are essential for plant growth when they are present at optimal level, but become toxic at high concentration, and may cause toxic effects on plant growth and yield or even death of plants. Heavy metal toxicity primarily interferes biochemical reactions of photosynthetic apparatus leading to adverse effects on biomass production and leads to progressive senescence (Sobkowiak and Deckert 2003; Alaoui-Sossé et al. 2004; Lin et al. 2005). Of all toxic heavy metals, cadmium (Cd) is of major environmental concern because the discharged rate of Cd in soil is of the order of 22,000 tons per year globally (Nriagu and Pacyna 1988) and it ranks highest in terms of damage to plant growth and human health. The conscious or unconscious addition of Cd in the soil disturbs the plants biological functions as the residence time of Cd in soil is more than thousand years (Alloway 1995). Excess Cd induces complex changes in plants at genetical, biochemical and physiological levels, leading to phytotoxicity. Plants are also able to load part of the Cd taken up from soil into the xylem and transport it to leaves (Chardonnens et al. 1998). Cadmium can interfere with photosynthetic and respiratory activities, mineral nutrition, enzymatic activities, membrane functions, and hormone balance (Chien and Kao 2000).

In fact, the inhibition of Cd stress effects on plant physiological processes is related to its effect on nutrient uptake and distribution. The interactions of Cd and metal nutrients, such as Fe, Zn, Cu, and Mn have been reported in some plant species (Zhang et al. 2002; Wu and Zhang 2002). Most plants are sensitive to low Cd concentrations and show inhibition in plant growth as a consequence of alterations in the photosynthesis rate and the uptake and distribution of macronutrients and micronutrients (Lozano-Rodriguez et al. 1997; Sandalio et al. 2001; Benavides et al. 2005; Dube et al. 2009).

As heavy metals cause deficiency of essential mineral nutrient elements by restricting their uptake and distribution in plant tissues (Siedlecka 1995), better understanding of how the supply of these nutrients may reverse metals toxicity is needed (Pankovic et al. 2000). Optimization of mineral nutrients may probably reduce some of the metal-induced negative effect on plants as mineral nutrients affect the activity and bioavailability of Cd in the soil-plant environment. The study of interaction of Cd with plant nutrients in soil-plant systems is, therefore, of great importance and may be used to minimize Cd accumulation in edible plant parts (Dheri et al. 2007). Among the various macronutrients, sulfur (S) serves a distinguished role in Cd detoxification because it is an integral part of most of the defense compounds involved in Cd detoxification. Optimum S nutrition is helpful in reducing Cd translocation within the plant system (Sarwar et al. 2010). Cadmium induces generation of reactive oxygen species (ROS) that causes oxidative stress in plants (Skórzyńska-Polit et al. 2003/04; Mobin and Khan 2007; Anjum et al. 2008; Khan et al. 2009; Iqbal et al. 2010). These excess ROS cause damage to proteins, lipids, carbohydrates, DNA and ultimately may result in cell death (Foyer and Noctor 2005; Djebali et al. 2008). Cadmium-induced ROS leads to reduction in photosynthesis, plant growth and yield. Iqbal and Khan (2010) reported that Cd leads to reduction in photosynthetic pigments, growth and yield. The present review focuses mainly to gain detailed insight on the effects of Cd toxicity in plants and elucidates the potential of sulfur in modulating Cd-induced oxidative stress.

2 Updates on the Cadmium Toxicity in Plants

Cadmium is a highly toxic heavy metal that negatively affects physiological processes in plants, growth and development, and eventually plant death. The critical concentration at which Cd causes injuries is in the range of 3–10 mg/kg dry mass (Balsberg-Pahlsson 1989). Its accumulation in plants adversely affects photosynthetic processes and diminishes water and nutrient uptake (Sanita di Toppi and Gabbrielli 1999; Maksimović et al. 2007), root and shoots growth and disturbs cellular redox control (Clemens 2001; Schützendübel and Polle 2002).

Cadmium induces significant disturbances in the structural organization and functional activity of the photosynthetic apparatus (Baszynski 1986; Vassilev et al. 1995; Dahlin et al. 2000). The main targets of toxic Cd effects are the pigment apparatus and photosynthetic gas exchange system (Clijsters and Van Assche 1985; Tukiendorf and Baszynski 1991; Lang et al. 1995). The most characteristic symptoms of Cd stress are brown and short roots, chlorosis, fewer tillers, senescence and reduced plant growth and biomass (Arduini et al. 1994; Wu and Zhang 2002; Wu et al. 2003; Cosio et al. 2006). Moussa and El-Gamal (2010) reported that at higher concentrations of Cd (400-1,000 µM Cd), the toxic symptoms of root cells are mainly continued to disintegration of cell organelles, disruption of membranes, withdrawal of plasma membrane from cell walls, and formation of multivesiculate bodies in the cytoplasm. The uptake of Cd from soil seems to occur mainly via Ca²⁺, Mg²⁺, Fe²⁺, Mn²⁺ and Zn²⁺ transporters (Clemens 2006). The best-studied nonspecific transporter is the ZIP IRT1, which is the major transporter responsible for high-affinity iron uptake from the soil. Cadmium exposure rapidly induces apparent Zn deficiency that may be through binding to a Zn sensor protein (Weber et al. 2006; Roth et al. 2006). Cadmium stress inactivates macromolecules and cellular structure (Stroi ski et al. 1990; Jiang et al. 2009) resulting in altered physiology processes and biochemical mechanisms.

The Cd-induced toxicity in plants is because of altered physiological phenomena (Demirevska-Kepova et al. 2006). Cadmium exerts its toxic effect in plants by inducing oxidative stress generated due to overproduction of ROS (Sandalio et al. 2001; Ali et al. 2002; Ranieri et al. 2005; Smeets et al. 2005; Singh et al. 2008). Oxidative stress occurs when there is a serious imbalance in any cell compartment between the production of ROS and antioxidant defense, leading to significant physiological challenges (Foyer and Noctor 2000). These excess ROS cause damage to proteins, lipids, carbohydrates, DNA and ultimately result in cell death (Mittler et al. 2004; Foyer and Noctor 2005; Shulaev and Oliver 2006; Djebali et al. 2008). Evidence that Cd causes the production of ROS in plants arises from the observations that new isozymes of peroxidases were detectable in both root and leaves of Phaseolus vulgaris (Van Assche and Clijsters 1990). Besides, this fact is also confirmed from the detection of lipid peroxidation, increased lipoxygenase activity, chlorophyll degradation and inhibition or stimulation of the activity of several antioxidant enzymes under Cd stress (Dixit et al. 2001; Leon et al. 2002; Skórzyńska-Polit et al. 2003/04; Mobin and Khan 2007; Anjum et al. 2008; Agrawal and Mishra 2009). Cadmium-induced increase in ROS production acts as a cellular signal triggering the stress response (Verbruggen et al. 2009). Stress-responsive MAP kinases seem to be involved in transcriptional responses to Cd as they are activated possibly by ROS under Cd²⁺ excess (Jonak et al. 2004). Another putative player in Cd-induced oxidative stress signaling is AtOS1, a member of the Abc1 family localized in the chloroplasts. AtOS1 does not transport Cd but seems to be crucial for Cd tolerance, possibly through a putative kinase activity (Jasinski et al. 2008). An important source of Cd toxicity is its chemical similarity with essential elements, in particular Zn, but also Ca and Fe, deregulating the homeostasis of the latter elements or causing their displacement from proteins (Verbruggen et al. 2009; Sarwar et al. 2010).

The principal mechanisms of plant response to Cd stress include phytochelatin (PC)-based sequestration and compartmentalization, and additional defense mechanisms based on cell wall immobilization, plasma membrane exclusion and induction of stress proteins (Draži et al. 2006). Cadmium stress leads to the activation of antioxidant defense system. The effects of Cd on antioxidative capacity are dual: on one hand, Cd can induce oxidative stress via the inhibition of antioxidants, but on the other hand it also activate several antioxidative components as a result of a disturbed redox balance and a consecutively induced signal transduction cascade.

A putative chromatin remodeling factor, named OXS3, was recently identified in a screen for Cd tolerance of a Brassica juncea cDNA library in Schizosaccharomyces pombe (Blanvillain et al. 2008). An oxs3 mutant was hypersensitive to Cd and its overexpression improved Cd tolerance. Verbruggen et al. (2009) postulated that OXS3 might protect DNA or alter its transcriptional selectivity. While most Cd are chelated before its transport to the vacuole, Cd²⁺ can be directly transported into the vacuoles by Cd²⁺/ proton antiporters such as CAX2 and CAX4 and possibly also by MHX (Korenkov et al. 2007; Berezin et al. 2008). Korenkov et al. (2007) have reported that the overexpression of AtCAX2 or AtCAX4 in tobacco enhances Cd and Zn transport into root tonoplast vesicles and enhances Cd accumulation in roots of plants exposed to Cd. NRAMP3 and NRAMP4 are responsible for Cd²⁺efflux from the vacuole (Thomine et al. 2003). Their overexpression increased Cd sensitivity in Arabidopsis as a result of the impairment of Fe homeostasis, as NRAMP3 and NRAMP4 are responsible for the release of vacuolar Fe²⁺. Plant cells contain a range of protective and repair systems, which under normal circumstances minimize the occurrence of oxidative damage. There are systems which either react with ROS and keep them at a low level or with antioxidant enzymes such as superoxide dismutase (SOD), catalase (CAT), guaiacol peroxidases (POD), ascorbate peroxidase (APX), glutathione reductase (GR) that quench ROS. The quenching of ROS is supported by nonenzymatic antioxidants, glutathione, ascorbic acid, α -tocopherol and carotenoids (Sairam et al. 2000; Shah et al. 2001) or systems that regenerate oxidized antioxidants (glutathione, mono- and dehydroascorbate) (Markovska et al. 2009).

Genes that are involved in Cd detoxification notably include ATP-binding cassette (ABC) proteins (Hall and Williams 2003; Plaza and Bovet 2008). Members of the ABC transporter family that are known to confer Cd tolerance to plants include MRP3 (multidrug-resistance-related protein) (Kolukisaoglu et al. 2002), ATM3 (ABC transporter of the mitochondria) (Kim et al. 2006) and PDR8 (pleiotropic drug resistance) (Kim et al. 2007). The strong upregulation effects in the roots on genes that are putatively involved in Cd detoxification indicate that the ATPase and the ABC transporters might have been activated by both the increased Cd concentration and the increased availability of sulfur (Fässler et al. 2011). Nicotianamines (NA) are involved in Cd chelation, transport and detoxification in plants. Nicotianamines are ubiquitously present in plants and synthesized from three molecules of methionine by nicotianamine synthase (NAS) (Sharma and Dietz 2006).

Responses to Cd also include the induction of genes involved in sulfur and glutathione metabolism (van de Mortel et al. 2008). Cadmium shows a high affinity for thiols, and therefore the major thiol antioxidant, reduced glutathione (GSH), that is highly abundant in cells is a primary target for free Cd ions. The Cd-induced depletion of the reduced GSH pool (Lopez et al. 2006) results in a disturbance of the redox balance leading to an oxidative environment.

3 Nutrients Availability and Cadmium Tolerance

Plant metabolism is affected by optimal composition of mineral nutrients. Cations such as Ca²⁺, Mg²⁺, Zn²⁺, and Mn²⁺ compete with Cd for uptake by plants (Tlustos et al. 2006) and for exchange site in soils (Degryse et al. 2004). Studies have shown that appropriate application of plant nutrients increases plant biomass and grain yield, and decreases Cd concentration in grain and other edible plant parts by dilution effect (Sarwar et al. 2010). Plant nutrients also help to sequester Cd in vegetative parts by production of PCs and avoid Cd accumulation in grains. They help to alleviate physiological stress caused due to an excess of Cd. However, antagonistic interaction between Cd and content of these nutrients has been reported (Uraguchi et al. 2006). In fact, the toxicity symptoms due to Cd are correlated with disturbances in the uptake and distribution of macro and micronutrients in plants (Gussarson et al. 1996). Cadmium is taken up by plants via cation transport systems normally involved in the uptake of essential elements, such as members of ZIP and Nramp families or Ca2+ channels and transporters (Pence et al. 2000; Perfus-Barbeoch et al. 2002). Zhao et al. (2002) reported that Cd is taken up by the roots via the Ca channels or via Zn and Mn transporters. Its entry through the Ca channel in the leaves disturbs the plant-water relationship (Perfus-Barbeoch et al. 2002) causing stomatal closure in many plants, leading to lower transpiration rate and inhibition of photosynthesis through an adverse effect on chlorophyll metabolism. This subsequently leads to growth inhibition and imbalance in the nutrient level (Sandalio et al. 2001; Chaffei et al. 2004). The effect on growth reduction may also be due to the rate of interference of toxic metals and nutrients with sensitive metabolic reactions (Gupta et al. 1995, 1998; Haag-Kerwer et al. 1999). The decrease of Mn, Fe, Mg, S and P concentrations in Cd-sensitive cultivar's (Brassica oleracea L.) leaves under Cd stress may be the key reason for the restraint of leaf photosynthesis, and the decrease of plant growth (Sun and Shen 2007).

The increase or decrease in plant nutrients under Cd stress may vary in root and shoot. Inhibition may occur at the uptake level or in translocation of nutrients. Different transporters are involved in the translocation of nutrients into the aerial part of the plant at different levels and Cd can inhibit these transporters. Morphological changes of the conducting xylem tissue may also contribute to a limited translocation of nutrients from the roots (Barceló et al. 1988). Zornoza et al. (2002) have reported that cadmium addition reduces P, K, Fe, Mn and Zn concentrations in the shoot and Mn in the root of Cd-treated plant. However, the effect of Cd on the nutrient content depends on the plant species and the experimental conditions. A number of factors, viz. concentrations of ions, pH, abundance of chelators, etc. govern these interactions and thus various types of responses may take place in different experiments using cadmium. It is of general concern that cadmium in certain concentration may cause deficiency of other elements essential for normal

growth of plants. Birch plants grown in the presence of Cd showed a reduction in the concentration of Fe, Mn and Cu in the shoot, while in roots no significant changes were observed, and the content of Zn did not change either in shoots or roots (Gussarson et al. 1996). Thus, the presence of Cd in the plants caused a general decrease in the nutrient content of plants.

Cd stress leads to a decrease in nutrients content, and the severity of Cd toxicity can be reduced through the optimization of these nutrients. Sufficient amount of nutrients may reduce the accumulation of the metal in plants thereby reducing its toxicity. Pankovic et al. (2000) have shown that optimal N supply decreases the inhibitory effects of Cd on photosynthesis of sunflower plants probably by increasing ribulose 1,5-bisphosphate carboxylase (rubisco) activity. They found that it was only at optimum N that the adverse affect of Cd was reduced and thus N nutrition could be manipulated as a means of decreasing Cd phytotoxicity. Potassium may protect plants from Cd-toxicity-induced oxidative damage by reducing the Cd availability to the plants thereby depressing the contents of H₂O₂ and TBARS in the mustard leaves and increasing the content of antioxidative enzymes (Umar et al. 2008). He et al. (2005) suggests that Ca^{2+} can enhance Cd tolerance by increasing the expression of PC synthase gene under Cd²⁺ stress. The deficiency of Mg enhances Cd-induced oxidative stress resulting in damage to chlorophyll, photosynthesis and plant growth. Abul Kashem and Kawai (2007) have reported that Mg decreases the content of Cd in the shoots because of the enhancement of shoot dry weight induced by the detoxifying effect of higher Mg concentration in Japanese mustard spinach.

4 Sulfur as Essential Factor for Cd Tolerance

Among all the nutrients involved in Cd detoxification, the role of S in Cd tolerance is important because of its presence in most of the defense compounds. Sulfur has been accepted as an outstanding factor for the improvement in production and quality of crops in several plant species. The negative relation between Cd and NPK could be mitigated with S nutrition as several studies have shown the synergistic interaction effects of S with N and K in influencing the yield, quality (oil, protein, amino acid, and fatty acid synthesis) and uptake of nutrients by different crops (Tandon 1991; Aulakh and Chhibba 1992). Probably, the improved S nutrition allows a more adequate plant defense response by synthesis of sulfur defense compounds, including PC and GSH. The importance of S in Cd tolerance is discussed emphasizing on the upregulation of various S-assimilating enzymes under Cd stress, which ultimately lead to synthesis of S containing defense compounds.

Sulfur is an essential nutrient, taken up as sulfate from soil, reduced and incorporated into bioorganic compounds in plant cells. The pathway of sulfate assimilation is highly regulated in a demand-driven manner in seed plants. The importance of S as a plant nutrient is becoming more imminent due to its effect on crop productivity and quality. Sulfur is present in amino acids, proteins, peptides, coenzymes and vitamins. Plants and many microorganisms are able to utilize inorganic sulfate and assimilate it into these compounds (Leustek et al. 2000; Kopriva 2006). A large number of studies have reported a marked influence of applied S on the yields of several cereals, pulses, oilseeds, vegetables, forages and other crops (Pasricha et al. 1987; Tandon 1991; Ahmad et al. 1998; Aulakh and Pasricha 1998). Oilseed rape has a high requirement for S and is particularly sensitive to any shortfall in S supply (Ahmad et al. 2005). Yield responses of oilseed rape to S supply have been reported (Walker and Booth 1992). In fact, S is an important nutrient for oilseed rape due to its association with yield and also a range of quality factors. It is required by Brassicas for the synthesis of the S-bearing compound, glucosinolates. Seeds from many plants, especially of legume crop, contain low concentrations of small 2S proteins that are relatively rich in cysteine and methionine (Shewry and Pandya 1999). Glycine max also produces low molecular weight polypeptides that contain disproportionately high methionine content (Paek et al. 2000). In fact, the availability of reduced S

(i.e., cysteine and methionine) is the rate-limiting factor for the regulation of β -conglycinin chains that usually synthesized only during late seed development (Meinke et al. 1981). Sulfur deficiency is an environmental condition that highly upregulates sulfate assimilation in seed plants (Hirai et al. 2003; Maruyama-Nakashita et al. 2003; Nikiforova et al. 2003).

Sulfur is found in soil in the form of sulfate and, through a set of reaction, is converted to sulfide and into an N/C-skeleton forms cysteine or its homologues (Droux 2004). The assimilation of sulfate could be summarized in four steps: (1) uptake of sulfate, (2) activation of sulfate, (3) reduction of sulfate and (4) synthesis of cysteine. Sulfate uptake is facilitated by sulfate transporters; once sulfate is within the cells, it can either be stored or enter the metabolic stream. Metabolism of sulfate is initiated by its activation by the reaction of adenylation catalyzed by ATPsulfurylase. The reaction product adenosine 5'-phosphosulfate (APS) is a branch point intermediate, which can be channeled toward reduction or sulfation (Leustek et al. 2000). The key enzyme of plant sulfate assimilation is the adenosine 5'-phosphosulfate reductase (APR), which reduces activated sulfate as demonstrated by control flux analysis (Vauclare et al. 2002). The sulfate assimilation pathway and APR are highly regulated in a demand-driven manner (Lappartient and Touraine 1996; Leustek et al. 2000; Vauclare et al. 2002; Kopriva 2006; Davidian and Kopriva 2010). In periods of low sulfate availability, plants increase sulfate transport and rate of reduction. Activation of sulfate reduction is the dominant route for assimilation and is carried out in plastids (Leustek et al. 2000; Saito 2000). APS is reduced to sulfite by APS-reductase (APR) (Leustek and Saito 1999; Kopriva and Koprivova 2003), and finally sulfite is reduced to sulfide by sulfite reductase (SiR). Sulfide is then transferred to activated serine by O-acetylserine(thiol)lyase (OAS-TL) to form cysteine. The formation of cysteine is a direct coupling step between S and N assimilation in plants (Brunold 1990, 1993; Brunold et al. 2003). Cysteine is the precursor or S-donor for most other organic S-compounds in plants. In addition, cysteine is the precursor of GSH, a low molecular weight water soluble nonprotein thiol compound which functions in protection of plants against varied environmental stresses (De Kok et al. 2005).

Cysteine synthesis in plants represents the final step of assimilatory sulfate reduction and the almost exclusive entry reaction of reduced sulfur into plant metabolism (Wirtz and Hell 2006). The importance of reduced sulfur is further illustrated by the multitude of functions that are directly or indirectly mediated by the major sulfur metabolites cysteine, methionine and GSH. This step also marks the convergence of S and N metabolism giving rise to regulatory interactions (Kopriva et al. 2002). Further, increased rate of synthesis and accumulation of cysteine occur in order to form GSH during the abiotic defense response of plants to heavy metals or xenobiotics (Rüegsegger and Brunold 1992; Farago et al. 1994).

Sulfur assimilation has an important effect on Cd accumulation. A significant induction in S assimilation has been reported in heavy metal exposed higher plants (Tukiendorf and Rauser 1990). Exposure of plants to Cd induces enzymes involved in the sulfate assimilation pathway (Herbette et al. 2006; Khan et al. 2007). Cadmium stress induces intracellular sulfur sink because of increased PC synthesis. In response, the plant may adapt the expression of genes involved in S assimilation and GSH biosynthesis, allowing an increased flux through the entire pathway when sulfate is not limiting. In S. cerevisiae, very high Cd concentrations (1 mM) led to upregulation of the MET3 gene (ATPS, Fauchon et al. 2002) and concomitantly, ATPS protein levels increased sixfold (Vido et al. 2001). In addition, a microarray analysis of S. cerevisiae under yet another Cd concentration (300 µM CdCl) showed that MET14 (APK) and MET 16 (PAPSR) had increased gene expression of 21- or 6-fold (Momose and Iwahashi 2001).

It has been noted that genes involved in S assimilation pathway are rapidly upregulated such as Sultr1;1 and Sultr2;1 encoding two sulfate transporters which are upregulated after 2 or 6 h of Cd treatment and 12–24 h after sulfate depletion (Takahashi et al. 2000; Herbette et al. 2006; Sarry et al. 2006). Heiss et al. (1999)

reported the upregulation of sulfate transporter(s), ATPS and APR under Cd stress. Hermsen et al. (2010) reported that mRNA levels of APR, SiR2 and SULTR4;1 were increased while ATPS2, SULTR1;1, -1;2 and -1;3 were significantly reduced upon Cd treatment in *P. patens*.

Nussbaum et al. (1988) reported that Cd accumulation increased ATPS and APR activities in Zea mays seedlings. Rüegsegger et al. (1990) have shown that APR activity is induced coordinately with glutathione synthetase in Cd-treated *Pisum* sativum plants. ATPS is regulated as demanddriven homeostatic mechanisms in order to match the rate of cysteine and GSH biosynthesis with the total sulfur needs of the whole plant (Lappartient and Touraine 1996; Lappartient et al. 1999).

ATPS and serine acetyl transferase (SAT) play important roles in heavy metal tolerance and accumulation (Hawkesford 2003; Freeman et al. 2004). Harada et al. (2002) reported the expression of three sulfur assimilation pathway enzyme genes including ATPS increased significantly, and that the total thiol compounds increased threefold under Cd stress in Arabidopsis. These may be due to the need of thiol for increased chelation. Rother et al. (2006) have reported that in heavy metal dependent induction of genes involved in cysteine and GSH biosynthesis, the increased activity of the corresponding proteins and thiol metabolites are the bases for elevated intracellular GSH concentrations. Treatment with Cd²⁺ influenced the transcriptions of genes coding for enzymes involved in sulfate reduction and GSH biosynthesis in time- and concentration-dependent manner. However, transcription of the SAT gene, encoding the OAS-generating enzyme, was hardly affected upon Cd²⁺ stress in *P. patens*.

Krishnan (2005) reported increased OASTL in Cd-treated soybean. Manipulation of critical enzymes in the sulfur assimilation pathway is one of the most promising approaches for increasing the amounts of sulfur containing amino acids in soybean. In fact, the relatively higher expression of GmOASTL4 gene occurred in transgenic plants treated with CdCl₂ than that in untreated transgenic plants, which could explain the clear induction of OASTL activity. Ning et al. (2010) reported that GmOASTL4 overexpression stimulates the antioxidant enzyme system. Howarth et al. (2003) observed an increase in the expression of all SAT genes in response to cadmium in *A. thaliana*.

In A. thaliana, OASTL gene Atcys-3A and the SAT gene Sat-5 are induced in leaf lamina, root and stem cortex and trichomes in response to Cd treatment (Domínguez-Solís et al. 2001; Howarth et al. 2003). Additionally, the overexpression of Atcys-3A in A. thaliana resulted in increased Cd tolerance (Domínguez-Solís et al. 2001). These results suggest that specific OASTL and SAT isoforms may play a role in increasing cysteine production under conditions of heavy metal stress when increased biosynthesis of GSH is required. Plants overexpressing SAT were found to have elevated levels of cysteine and glutathione when compared to the wild type (Blaszczyk et al. 1999; Harms et al. 2000). Thus, the overexpression of S-assimilating enzymes led to the production of GSH and subsequently PC which are involved in Cd detoxification.

5 Sulfur Sinks and Cd Tolerance

Glutathione is a tripeptide thiol (Glu-Cys-Gly) and is a major water soluble antioxidant and redox buffer in plants, performing critical functions in cell cycle regulation, development, sulfur transport and storage, stress responses, and heavy metal detoxification (Maughan and Foyer 2006). It is synthesized from its immediate precursor cysteine formed through reductive sulfur assimilation pathway in a cascade of enzymatic reactions (Kopriva 2006). The synthesis of GSH is accomplished in two sequential ATP-dependent reactions catalyzed by c-glutamylcysteine synthetase (cECS) and glutathione synthetase (GSHS) (Becana et al. 2010). Sulfur supplementation might further help the plants to improve the content of GSH by enhancing γ -ECS enzymes (Schneider and Bergmann 1995; Strohm et al. 1995). Hassan et al. (2008) reported that addition of $(NH_4)_2SO_4$ to Cd-fed plants showed increased glutathione content, which reduced oxidative stress in Oryza sativa plants that were related to more S supply for this N form. As discussed above, the entry of Cd into the cytosol promptly

activates the sulfur metabolism resulting in the production of GSH. Glutathione, besides performing critical functions in regulating plant growth and adaptation to abiotic stresses, acts as an important S sink in the plant system (Leustek et al. 2000; Maughan and Foyer 2006).

Glutathione occurs in reduced form (GSH) in plant tissues and is localized in all cell compartments such as cytosol, endoplasmic reticulum, vacuole, mitochondria, chloroplasts, peroxisomes as well as in apoplast (Jimenez et al. 1998). It is necessary to maintain the normal reduced state of cells so as to counteract the inhibitory effects of ROS-induced oxidative stress (Meyer 2008) and is a potential scavenger of ${}^{1}O_{2}$ and $H_{2}O_{2}$ (Briviba et al. 1997; Noctor and Foyer 1998). In addition, GSH is a substrate for glutathione peroxidase (GPX) and glutathione-S-transferases (GST), which are also involved in the removal of ROS (Noctor et al. 2002). Glutathione-S-transferases can reduce peroxides with the help of GSH and produce scavengers of cytotoxic and genotoxic compounds. An increased GST activity was found in leaves and roots of Cd-exposed Pisum sativum plants by Dixit et al. (2001) and in roots of Oryza sativa and Phragmites australis plants (Iannelli et al. 2002; Moons 2003). Besides, GSH plays a key role in the antioxidative defense system by regenerating another potential water soluble antioxidants such as Ascorbate (AsA), via the Ascorbate-glutathione (AsA-GSH) cycle (Foyer and Halliwell 1976). GSH is a constituent of the AsA-GSH cycle that is a major pathway for the conversion of H2O2 and is stimulated under Cd stress (Smeets et al. 2008; Smeets et al. 2009).

Further, Chen et al. (2010) reported that GSH promotes acclimative and adaptive responses in antioxidant systems of barley seedlings to cope with Cd stress. It not only acts as a directly preformed antioxidant to scavenge ROS but might also act indirectly in modifying the redox balance by pro- or con-activating such antioxidant responses depending on its endogenous levels to alleviate Cd toxicity. They reported that GSH-induced alleviation of Cd stress by stimulating APX (especially cAPX, sAPX and tAPX iosenzymes) and CAT activities. It upregulated root cAPX and leaf cAPX, CAT1 and CAT2 expression

	Increase/							
Thiols	decrease	Plants	Parameters	Response	Reference			
Glutathione (added)	Increase	Brassica juncea	Cd content	-	Zhu et al. (1999a)			
Glutathione (added)	Increase	Glycine max	OH.	_	El-Shintinawy (1999)			
Phytochelatin (added)	Increase	Transgenic Nicotiana tobacum	Cd content	-	Harada et al. (2001)			
Glutathione	Increase	Phragmites australis	Antioxidative enzymes	+	Pietrini et al. (2003) and Ederli et al. (2004)			
Phytochelatin	Increase	Bread wheat	H ₂ O ₂	_	Ranieri et al. (2005)			
Glutathione	Increase	Oryza sativa	MDA	_	Hassan et al. (2005)			
Glutathione	Decrease	Helianthus annuus	TBARS	+	Gallego et al. (2005)			
Glutathione	Increase	Brassica chinensis and Brassica pekinensis	Cd content	-	Liu et al. (2007)			
Glutathione	Increase	Brassica campestris	Cd content, TBARS	_	Anjum et al. (2008)			
Glutathione and total sulfur	Increase	Sedum alfredii	H_2O_2	_	Chao et al. (2008)			
Phytochelatin	Increase	Lycopersicon esculentum	Cd content	_	Ammar et al. (2008)			
Cysteine and glutathione (added)	Increase	Tagetes erecta L.	Cd content	-	Feng et al. (2009)			
Glutathione	Increase	Brassica juncea	H ₂ O ₂ , TBARS, e-leakage	_	Khan et al. (2009) and Iqbal et al. (2010)			
Glutathione (added)	Increase	Solanum nigrum	TBARS, H_2O_2 , superoxides	_	Deng et al. (2010)			
Glutathione	Increase	Cucurbita pepo	Lipid peroxidation	+	Kolb et al. (2010)			
Glutathione (added)	Increase	Solanum tuberosum	Oxidative stress	-	El-Tayeb et al. (2010)			

Table 20.1 Role of thiols in reducing cadmium (Cd)-induced oxidative stress. (+) sign indicate increase and (-) sign indicate decrease

in the sensitive genotype of barley to achieve stimulation. Cuypers et al. (2010) reported that glutathione metabolism played a crucial role in controlling the gene regulation of the antioxidative defense mechanism under Cd stress in the roots of *Arabidopsis thaliana*.

The role of GSH in the antioxidant defense system provides a strong basis for its use as a stress marker (Tausz et al. 2004). Pietrini et al. (2003) reported that the antioxidant activity in the leaves and chloroplast of *Phragmites australis* Trin. (cav.) ex Steudel was associated with a large pool of GSH which resulted in protecting the activity of many photosynthetic enzymes against the thiophilic bursting of Cd. GSH is a precursor of PCs, which plays an important role in controlling cellular heavy metal concentration (Grill et al. 1985). Increased GSH levels are connected with enhanced plant tolerance to stress (Stancheva et al. 2010). Xiang et al. (2001) observed that plants with low levels of GSH were highly sensitive to even low levels of Cd^{2+} due to limited PC synthesis.

Therefore, sulfur plays an important role in Cd detoxification through the synthesis of GSH and PC. Fan et al. (2010) reported that sulfur could alleviate the Cd stress in rice due to S-induced increase of GSH for synthesis of PCs related to Cd tolerance. Studies have shown that thiols reduce Cd-induced oxidative stress (Zhu et al. 1999a, b; Ammar et al. 2008; Feng et al. 2009; Khan et al. 2009; Iqbal et al. 2010; Kolb et al. 2010, Table 20.1).

Phytochelatins are peptides that consist of repetitions of the γ -glutamylcysteine (γ -GluCys)



Fig. 20.1 Representation of sulfur assimilation and cadmium tolerance. *ROS* reactive oxygen species, *LMW* low molecular weight complex, *HMW* high molecular weight complex

dipeptide followed by a terminal glycine. The basic structure being (γ -Glu-Cys)-Gly, where n is generally in the range 2–7. These peptides are synthesized from glutathione and related thiols by PC synthases (PCS, a γ -glutamyl-cysteinyl transpeptidase (EC 2.3.2.15)). PCS mediates a dipeptidyl transfer reaction in the presence of heavy metals, leading to the formation of PCn with a greater number of γ -glutamylcysteinyl units, which has increased the affinity for heavy metals and the capacity to sequester them more effectively (Löffler et al. 1989; Clemens 2006).

Phytochelatins have been involved in metal detoxification, and used as potential biomarkers for an evaluation of metal toxicity (Wang et al. 2008). The levels of PCs were positively correlated with external metal concentrations (Keltjens and van Beusichem 1998; Sneller et al. 1999; Sun et al. 2005). The increase in PCs has been reported to increase Cd tolerance in plants (Martínez et al. 2006; Pomponi et al. 2006; Gasic and Korban 2007). It is an important intracellular Cd chelating molecule that mediates the sequestration of the heavy metal-PC conjugant in the vacuole (Siripornadulsil et al. 2002; Mendoza-Cozatl and Moreno-Sanchez 2006).

Overproduction of PCS from wheat in *S. cerevisiae* enhances tolerance to and accumulation of cadmium (Clemens et al. 1999).

Cadmium leads to activation of sulfur metabolism leading to the production of PC (Sanita di Toppi and Gabbrielli 1999) since biosynthesis of PC is closely dependent on sulfur metabolism (Leustek et al. 2000). An increased expression of genes involved in sulfur assimilation and GSH and PC syntheses in response to Cd treatment has been shown in Arabidopsis (Harada et al. 2002), and Brassica juncea (Heiss et al. 1999). Among the mechanism utilized by the plants to detoxify ROS, production of GSH appears to be more important as this is a substrate for the biosynthesis of PCs, which are involved in heavy metal detoxification (Inouhe 2005). Sulfur assimilation results in GSH synthesis that detoxifies ROS either directly or through the synthesis of PCs. Phytochelatins are involved in Cd removal by chelation. They form low molecular weight Cd-PC complex (LMW) that enters the vacuole and combines there with more Cd and sulfide to form high molecular weight Cd-PC complex (HMW), thus, binding maximum Cd and storing it into vacuole. This keeps the cell free from Cd toxicity (Fig. 20.1).

6 Conclusion and Future Perspective

Cadmium toxicity can be alleviated by modulating various enzymes involved in S assimilation and manipulating the pathway of S metabolism. In fact, the nutritional and environmental conditions may act as molecular signals in the regulation of S assimilation under stress (Nazar et al. 2011). Chen and Huerta (1997) have shown that S is a critical nutritional factor for alleviating Cd toxicity. A positive effect of S on Cd detoxification has also been reported by Popovic et al. (1996). It is most likely that plant type with higher S accumulation capacity is expected to show more tolerance to Cd stress. Overexpression of sulfate assimilating enzymes increases tolerance to heavy metals (Pilon-Smits et al. 1999; Domínguez-Solís et al. 2001). The general alteration of the sulfur metabolic pathways induced by Cd is a possible consequence of an increase in the GSH demand driven by PC biosynthesis. In other words, exposure to Cd would induce an "additional sink" increasing the need for thiol compounds by cells (Tukiendorf and Rauser 1990; Heiss et al. 1999) which could be compensated by application of S.

The mechanisms of alleviation of Cd stress by sulfur may also include synthesis of phytohormones in addition to reduced S compounds. Phytohormones are believed to influence biochemical and molecular mechanisms under optimal and stressful environment. Cadmium stress affects the signaling of phytohormones and its toxicity can also be reverted by phytohormones (Hsu et al. 2006; Celik et al. 2008; Dalcorso et al. 2008; Popova et al. 2009; Stroiński et al. 2010). In this context, it may be said that S could alleviate Cd toxicity by affecting ethylene evolution because S-adenosyl methionine (SAM), an ethylene precursor, contains sulfur. Cysteine is the central precursor of all organic molecules containing reduced sulfur including SAM (a precursor of ethylene), which acts as a signaling molecule in the control of plant developmental processes under optimal and stress conditions (Höfgen et al. 2001). Sulfur nutrition thus may

provide a novel strategy to reduce Cd toxicity through various pathways that are accelerated by its application.

References

- Abul Kashem MD, Kawai S (2007) Alleviation of cadmium phytotoxicity by magnesium in Japanese mustard spinach. Soil Sci Plant Nutr 53:246–251
- Agrawal SB, Mishra S (2009) Effects of supplemental ultraviolet-B and cadmium on growth, antioxidants and yield of *Pisum sativum* L. Ecotoxicol Environ Saf 72:610–618
- Ahmad A, Abraham G, Gandotra N, Abrol YP, Abdin MZ (1998) Interactive effect of nitrogen and sulphur on growth and yield of rapeseed-mustard (*Brassica juncea* L. Czern and Coss. and *Brassica campestris* L.) genotypes. J Agron Crop Sci 181:193–199
- Ahmad A, Khan I, Anjum NA, Abrol YP, Iqbal M (2005) Role of sulphate transporter systems in sulphur efficiency of mustard genotypes. Plant Sci 169:842–846
- Alaoui-Sossé B, Genet P, Vinit-Dunand F, Toussaint ML, Epron D, Badot PM (2004) Effect of copper on growth in cucumber plants (*Cucumis sativus*) and its relationships with carbohydrate accumulation and changes in ion contents. Plant Sci 166:1213–1218
- Ali MB, Chun HS, Kim BK, Lee CB (2002) Cadmiuminduced changes in antioxidant enzyme activities in rice (*Oryza sativa* L. cv. Dongjin). J Plant Biol 45:134–140
- Alloway BJ (1995) Heavy metals in soils, 2nd edn. Blackie, London
- Ammar WB, Mediouni C, Tray B, Ghorbel MH, Jemal F (2008) Glutathione and phytochelatin contents in tomato plants exposed to cadmium. Biol Plant 52:314–320
- Anjum NA, Umar S, Ahmad A, Iqbal M, Khan NA (2008) Sulphur protects mustard (*Brassica campestris* L.) from cadmium toxicity by improving leaf ascorbate and glutathione. Plant Growth Regul 54:271–279
- Arduini I, Godbold D, Onnis A (1994) Cadmium and copper change root growth and morphology of *Pinus pinea* and *Pinus piaster* seedlings. Physiol Plant 92:675–680
- Aulakh MS, Chhibba IM (1992) Sulphur in soils and response of crops to its application in Punjab. Fert News 37:33–45
- Aulakh MS, Pasricha NS (1998) The effect of green manuring and fertilizer N application on enhancing crop productivity in mustard-rice rotation in semiarid subtropical regions. Eur J Agron 8:51–58
- Balsberg-Pahlsson AM (1989) Toxicity of heavy metals (Zn, Cu, Cd, Pb) to vascular plants. Water Air Soil Pollut 47:287–319
- Barceló J, Vazques MD, Poschenreider Ch (1988) Structural and ultrastructural disorders in cadmium-treated bush bean plants (*Phaseolus vulgaris* L.). New Phytol 108:37–49
- Baszynski T (1986) Interference of Cd²⁺ in functioning of the photosynthetic apparatus of higher plants. Acta Soc Bot Pol 55:291–304

- Becana M, Matamoros MA, Udvardi M, Dalton DA (2010) Recent insights into antioxidant defenses of legume root nodules. New Phytol 188:960–976
- Benavides MP, Gallego SM, Tomaro ML (2005) Cadmium toxicity in plants. Braz J Plant Physiol 17:21–34
- Berezin I, Mizrachy-Dagry T, Brook E, Mizrahi K, Elazar M, Zhuo S, Saul-Tcherkas V, Shaul O (2008) Overexpression of AtMHX in tobacco causes increased sensitivity to Mg²⁺, Zn²⁺, and Cd²⁺ ions, induction of V-ATPase expression, and a reduction in plant size. Plant Cell Rep 27:939–949
- Blanvillain R, Kim JH, Wu S, Lima A, Ow DW (2008) OXIDATIVE STRESS 3 is a chromatin-associated factor involved in tolerance to heavy metals and oxidative stress. Plant J 57:654–665
- Blaszczyk A, Brodzik R, Sirko A (1999) Increased resistance to oxidative stress in transgenic tobacco plants overexpressing bacterial serine acetyltransferase. Plant J 20:237–243
- Boyer JS (1982) Plant productivity and environment potential for increasing crop plant productivity, genotypic selection. Science 218:443–448
- Bray EA, Bailey-Serres J, Weretilnyk E (2000) Responses to abiotic stresses. In: Gruissem W, Buchannan B, Jones R (eds) Biochemistry and molecular biology of plants. American Society of Plant Biologists, Rockville, MD, pp 1158–1249
- Briviba K, Klotz LO, Sies H (1997) Toxic and signaling effects of photochemically or chemically generated singlet oxygen in biological systems. J Biol Chem 378:1259–1265
- Brunold C (1990) Regulation of sulfate sulfide. In: Renenberg H, Brunold C, de Kok LJ, Stulen I (eds) Sulfur nutrition and sulfur assimilation in higher plants: fundamental, environmental and agricultural aspects. SPB, The Hague, pp 13–31
- Brunold C (1993) Regulatory interactions between sulfate and nitrate assimilation. In: de Kok LJ, Stulen I, Renenberg H, Brunold C, Rauser W (eds) Sulfur nutrition and sulfur assimilation in higher plants: regulatory, environmental and agricultural aspects. SPB, The Hague, pp 125–138
- Brunold C, Von Ballmoss P, Hesse H, Fell D, Kopriva S (2003) Interactions between sulfur, nitrogen and carbon metabolism. In: Davidian JC, Grill D, De Kok LJ, Stulen I, Hawkesford MJ, Schnug E, Rennenberg H (eds) Sulfur transport and assimilation in plants: regulation, interaction and signaling. Backhuys, Leiden, pp 45–56
- Çelik A, Ünyayar S, Çekiç FÖ, Güzel A (2008) Micronucleus frequency and lipid peroxidation in Allium sativum root tip cells treated with gibberellic acid and cadmium. Cell Biol Toxicol 24:159–164
- Chaffei C, Pageau K, Suzuki A, Gouia H, Ghorbel MH, Masclaux-Daubresse C (2004) Cadmium toxicity induced changes in nitrogen management in *Lycopersicon esculentum* leading to a metabolic safeguard through an amino acid storage strategy. Plant Cell Physiol 45:1681–1693

- Chao YE, Zhang M, Tian SK, Lu LL, Yang XE (2008) Differential generation of hydrogen peroxide upon exposure to zinc and cadmium in the hyperaccumulating plant species (*Sedum alfredii* Hance). J Zhejiang Univ Sci B 9:243–249
- Chardonnens AN, Bookum WM, Kuijper LDJ, Verkleij JAC, Ernst WHO (1998) Distribution of cadmium in the leaves of cadmium tolerant and sensitive ecotypes of *Silene vulgaris*. Physiol Plant 104:75–80
- Chen Y, Huerta AJ (1997) Effects of sulfur nutrition on photosynthesis in cadmium-treated barley seedlings. J Plant Nutr 20:845–856
- Chen F, Wang F, Wua F, Maoa W, Zhanga G, Zhouc M (2010) Modulation of exogenous glutathione in antioxidant defense system against Cd stress in the two barley genotypes differing in Cd tolerance. Plant Physiol Biochem 48:663–672
- Chien H-F, Kao CH (2000) Accumulation of ammonium in rice leaves is response to excess cadmium. Plant Sci 156:111–115
- Clemens S (2001) Molecular mechanisms of plant metal tolerance and homeostasis. Planta 212:475–486
- Clemens S (2006) Toxic metal accumulation, responses to exposure and mechanisms of tolerance in plants. Biochimie 88:1707–1719
- Clemens S, Kim EJ, Neumann D, Schroeder JI (1999) Tolerance to toxic metals by a gene family of phytochelatin synthases from plants and yeast. EMBO J 18:3325–3333
- Clijsters H, Van Assche F (1985) Inhibition of photosynthesis by heavy metals. Photosynth Res 7:40–41
- Cosio C, Vollenweider P, Keller C (2006) Localization and effects of cadmium in leaves of a cadmium-tolerant willow (*Salix viminalis* L.) I. Macrolocalization and phytotoxic effects of cadmium. Environ Exp Bot 58:64–74
- Cuypers A, Karen S, Jos R, Kelly O, Els K, Tony R, Nele H, Nathalie V, Suzy VS, Frank VB, Yves G, Jan C, Jaco V (2010) The cellular redox state as a modulator in cadmium and copper responses in *Arabidopsis thaliana* seedlings. J Plant Physiol. doi:10.1016/j. jplph.2010.07.010
- Dahlin C, Aronsson H, Almkvist J, Sundqvist C (2000) Protochlorophyllide-independent import of two NADPH: Pchlide oxidoreductase proteins (PORA and PORB) from barley into isolated plastids. Physiol Plant 109:298–303
- DalCorso G, Farinati S, Maistri S, Furini A (2008) How plants cope with cadmium: staking all on metabolism and gene expression. J Integr Plant Biol 10:1268–1280
- Davidian J-C, Kopriva S (2010) Regulation of sulfate uptake and assimilation-the same or not the same? Mol Plant 3:314–325
- De Kok LJ, Castro A, Durenkamp M, Kralewska A, Posthumus FS, Stuiver E, Yang L, Stulen I (2005) Pathways of plant sulphur uptake and metabolism – an overview. Landbauforschung Volkenrode 283:5–13
- Degryse F, Buekers J, Smolders E (2004) Radio-labile cadmium and zinc in soils as affected by pH and source of contamination. Eur J Soil Sci 55:113–121

- Demirevska-Kepova K, Simova-Stoilova L, Stoyanova ZP, Feller U (2006) Cadmium stress in barley: growth, leaf pigment, and protein composition and detoxification of reactive oxygen species. J Plant Nutr 29:451–468
- Deng X, Xiay H, Zhang H, Shen Z (2010) Cadmiuminduced oxidative damage and protective effects of N-acetyl-L-cysteine against Cd toxicity in *Solanum nigrum* L. J Hazard Mater 15:722–729
- Dheri GS, Brar MS, Malhi SS (2007) Influence of phosphorus application on growth and cadmium uptake of spinach in two Cd contaminated soils. J Plant Nutr Soil Sci 170:495–499
- Dixit V, Pandey V, Shyam R (2001) Differential oxidative responses to cadmium in roots and leaves of pea (*Pisum* sativum L. cv. Azad). J Exp Bot 52:1101–1109
- Djebali W, Gallusci P, Polge C, Boulila L, Galtier N, Raymond P, Chaibi W, Brouquisse R (2008) Modifications in endopeptidase and 20S proteasome expression and activities in cadmium treated tomato (*Solanum lycopersicum* L.) plants. Planta 227:625–639
- Domínguez-Solís JR, Gutierrez-Alcala G, Romero LC, Gotor C (2001) The cytosolic O-acetylserine (thiol) lyase gene is regulated by heavy-metals and can function in cadmium tolerance. J Biol Chem 276:9297–9302
- Dražić G, Mihailović N, Lojić M (2006) Cadmium accumulation in *Medicago sativa* seedlings treated with salicylic acid. Biol Plant 50:239–244
- Droux M (2004) Sulfur assimilation and the role of sulfur in plant metabolism, a survey. Photosynth Res 79:331–348
- Dube BK, Sinha P, Shukla K, Chatterjee C, Pandey VK, Rai AD (2009) Involvement of excess cadmium on oxidative stress and other physiological parameters of eggplant. J Plant Nutr 32:996–1004
- Ederli L, Reale L, Ferranti F, Pasqualini S (2004) Responses induced by high concentration of cadmium in *Phragmites australis* roots. Physiol Plant 121:66–74
- El-Shintinawy F (1999) Glutathione counteracts the inhibitory effect induced by cadmium on photosynthetic process in soybean. Photosynthetica 36:171–179
- El-Tayeb AE, Yamamoto SY, Habora MEE, Matsukubo Y, Aono M, Tsujimoto H, Tanaka K (2010) Greater protection against oxidative damages imposed by various environmental stresses in transgenic potato with higher level of reduced glutathione. Breed Sci 60:101–109
- Fan J-L, Hub Z-Y, Ziadic N, Xia X, Wu C-Y-H (2010) Excessive sulfur supply reduces cadmium accumulation in brown rice (*Oryza sativa* L.). Environ Pollut 158:409–415
- Farago S, Brunold C, Kreuz K (1994) Herbicide safeners and glutathione metabolism. Physiol Plant 91:537–542
- Fässler E, Plaza S, Pairraud A, Gupta SK, Robinson B, Schulin R (2011) Expression of selected genes involved in cadmium detoxification in tobacco plants grown on a sulphur-amended metal-contaminated field. Environ Exp Bot 70:158–165
- Fauchon M, Lagniel G, Aude JC, Lombardia L, Soularue P, Petat C, Marguerie G, Sentenac A, Werner M, Labarre

J (2002) Sulfur sparing in the yeast proteome in response to sulfur demand. Mol Cell 9:713–723

- Feng Q, Tai P, Li P, Guo Y, Fu S (2009) The role of sulfur in cadmium (Cd) accumulation was investigated in *Tagetes erecta* L. J Plant Nutr 32:919–928
- Foyer CH, Halliwell B (1976) The presence of glutathione and glutathione reductase in chloroplasts: a proposed role in ascorbic acid metabolism. Planta 133:21–25
- Foyer CH, Noctor G (2000) Oxygen processing in photosynthesis: regulation and signaling. New Phytol 146:359–388
- Foyer CH, Noctor G (2005) Oxidant and antioxidant signaling in plants: a re-evaluation of the concept of oxidative stress in a physiological context. Plant Cell Environ 28:1056–1071
- Freeman JL, Persan MW, Nieman K, Albrecht C, Peer W, Pickering IJ, Salt DE (2004) Increased glutathione biosynthesis plays a role in nickel tolerance in *Thlaspi nickel* hyperaccumulators. Plant Cell 16:2176–2191
- Gallego SM, Kogan MJ, Azpilicueta CE, Penã C, Tomaro ML (2005) Glutathione-mediated antioxidative mechanisms in sunflower (*Helianthus annuus* L.) cells in response to cadmium stress. Plant Growth Regul 46:267–276
- Gasic K, Korban SS (2007) Expression of Arabidopsis phytochelatin synthase in Indian mustard (*Brassica juncea*) plants enhances tolerance for Cd and Zn. Planta 225:1277–1285
- Grill EE, Winnacker L, Zenk MH (1985) Phytochelatins: the principal heavy metal complexing peptides of higher plants. Science 230:674–676
- Gupta M, Rai UN, Tripathi RD, Chandra P (1995) Leadinduced changes in glutathione and phytochelatin in *Hydrilla verticillata* (lf) Royle. Chemosphere 30:2011–2020
- Gupta M, Tripathi RD, Rai UN, Chandra P (1998) Role of gluthatione and phytochelatin in *Hydrilla verticillata* (lf) Royle and *Vallisneria spiralis* L. under mercury stress. Chemosphere 37:785–800
- Gussarson M, Asp H, Adalsteinsson S (1996) Enhancement of cadmium effects on growth and nutrient composition of birch (*Betula pendula*) by buthionine sulphoximine (BSO). J Exp Bot 47:211–215
- Haag-Kerwer A, Schafer HJ, Heiss S, Walter C, Rausch T (1999) Cadmium exposure in *Brassica juncea* causes a decline in transpiration rate and leaf expansion without effect on photosynthesis. J Exp Bot 50:1827–1835
- Hall JL, Williams LE (2003) Transition metal transporters in plants. J Exp Bot 54:2601–2613
- Harada E, Yamaguchi Y, Koizumi N, Sano H (2002) Cadmium stress induces production of thiol compounds and transcripts for enzymes involved in sulfur assimilation pathway in *Arabidopsis*. J Plant Physiol 159:445–448
- Harada E, Coi YE, Tsuchisaka A, Obata H, Sano H (2001) Transgenic tobacco plants expressing a rice cysteine synthase gene are tolerant to toxic levels of cadmium. J Plant Physiol 158:655–661

- Harms K, von Ballmoos P, Brunold C, Hofgen R, Hesse H (2000) Expression of a bacterial serine acetyltransferase in transgenic potato plants leads to increased levels of cysteine and glutathione. Plant J 22:335–343
- Hassan MJ, Shao G, Zhang G (2005) Influence of cadmium toxicity on growth and antioxidant enzyme activity in rice cultivars with different cadmium accumulation. J Plant Nutr 28:1259–1270
- Hassan MJ, Shafi M, Zhang G, Zhu Z, Qaisar M (2008) The growth and some physiological responses of rice to Cd toxicity as affected by nitrogen form. Plant Growth Regul 54:125–132
- Hawkesford MJ (2003) Transporter gene families in plants: the sulphate transporter gene family-redundancy or specialization? Physiol Plant 117:155–163
- He Z, Li J, Zhang H, Ma M (2005) Different effects of calcium and lanthanum on the expression of phytochelatin synthase gene and cadmium absorption in *Lactuca sativa*. Plant Sci 168:309–318
- Heiss S, Schafer H, Haag-Kerwer A, Rausch T (1999) Cloning sulfur assimilation genes of *Brassica juncea* L.: cadmium differentially affects the expression of a putative low affinity sulfate transporter and isoforms of ATP sulfurylase and APS reductase. Plant Mol Biol 39:847–857
- Herbette S, Taconnat L, Hugouvieux V, Piette L, Magniette M-LM, Cuine S, Auroy P, Richuad P, Forestier C, Bourguignon J, Renou J-P (2006) Genome-wide transcriptome profiling of the early cadmium response of *Arabidopsis* roots and shoots. Biochimie 88: 1751–1765
- Hermsen C, Koprivova A, Matthewman C, Wesenberg D, Krauss GJ, Kopriva S (2010) Regulation of sulfate assimilation in *Physcomitrella patens*: mosses are different. Planta 232:461–470
- Hirai MY, Fujiwara T, Awazuhara M, Kimura T, Noji M, Saito K (2003) Global expression prowling of sulfurstarved *Arabidopsis* by DNA macroarray reveals the role of O-acetyl-l-serine as a general regulator of gene expression in response to sulfur nutrition. Plant J 33:651–663
- Höfgen R, Kreft O, Willmitzer L, Hese H (2001) Manipulation of thiol contents in plants. Amino Acids 20:291–299
- Howarth JR, Dominguez-Solis JR, Gutierrez-Alcala G, Romero LC, Gotor C (2003) The serine acetyltransferase gene family in *Arabidopsis thaliana* and the regulation of its expression by cadmium. Plant Mol Biol 51:589–598
- Hsu YT, Kuo MC, Kao CH (2006) Cadmium-induced ammonium ion accumulation of rice seedlings at high temperature is mediated through abscisic acid. Plant Soil 287:267–277
- Iannelli MA, Pietrini F, Fiore L, Petrilli L, Massacci A (2002) Antioxidant response to cadmium in *Phragmites australis* plants. Plant Physiol Biochem 40:977–982
- Inouhe M (2005) Phytochelatins. Braz J Plant Physiol 17:65–78
- Iqbal N, Khan NA (2010) Variation in growth, photosynthesis functions and yield of five mustard (*Brassica*)

juncea L.) cultivars under high cadmium stress. Plant stress 4 (special issue 1) 87–93

- Iqbal N, Masood A, Nazar R, Syeed S, Khan NA (2010) Photosynthesis, growth an antioxidant metabolism in mustard (*Brassica juncea* L.) cultivars differing in cadmium tolerance. Agric Sci China 9:519–527
- Jasinski M, Sudre D, Schansker G, Schellenberg M, Constant S, Martinoia E, Bovet L (2008) AtOSA1, a member of the Abc1-like family, as a new factor in cadmium and oxidative stress response. Plant Physiol 147:719–731
- Jiang WS, Liu DH, Xu P (2009) Cd-induced system of defence in the garlic root meristematic cells. Biol Plant 53:369–372
- Jimenez A, Hernandez JA, Pastori G, del Rio LA, Sevilla F (1998) Role of the ascorbate-glutathione cycle of mitochondria and peroxisomes in the senescence of pea leaves. Plant Physiol 118:1327–1335
- Jonak C, Nakagami H, Hirt H (2004) Heavy metal stress. Activation of distinct mitogen-activated protein kinase pathways by copper and cadmium. Plant Physiol 136:3276–3283
- Keltjens WG, van Beusichem ML (1998) Phytochelatins as biomarkers for heavy metal toxicity in maize: single metal effects of copper and cadmium. J Plant Nutr 21:635–648
- Khan NA, Samiullah, Singh S, Nazar R (2007) Activities of antioxidative enzymes, sulphur assimilation, photosynthetic activity and growth of wheat (*Triticum aestivum*) cultivars differing in yield potential under cadmium stress. J Agron Crop Sci 193:435–444
- Khan NA, Anjum NA, Nazar R, Iqbal N (2009) Increased activity of ATP-sulfurylase and increased contents of cysteine and glutathione reduce high cadmium-induced oxidative stress in mustard cultivar with high photosynthetic potential. Russ J Plant Physiol 56:670–677
- Kim D-Y, Bovet L, Kushnir S, Noh EW, Martinoia E, Lee Y (2006) AtATM3 is involved in heavy metal resistance in *Arabidopsis*. Plant Physiol 140:922–932
- Kim D-Y, Bovet L, Maeshima M, Martinoia E, Lee Y (2007) The ABC transporter AtPDR8 is a cadmium extrusion pump conferring heavy metal resistance. Plant J 50:207–218
- Kolb D, Müller M, Zellnig G, Zechmann B (2010) Cadmium induced changes in subcellular glutathione contents within glandular trichomes of *Cucurbita pepo* L. Protoplasma 243:87–94
- Kolukisaoglu HÜ, Bovet L, Klein M, Eggmann T, Geisler M, Wanke D, Martinoia E, Schulz B (2002) Family business: the multidrug-resistance related protein (MRP) ABC transporter genes in *Arabidopsis thaliana*. Planta 216:107–119
- Kopriva S (2006) Regulation of sulfate assimilation in *Arabidopsis* and beyond. Ann Bot 97:479–495
- Kopriva S, Koprivova A (2003) Sulphur in plants. In: Abrol YP, Ahmad A (eds) Sulfate assimilation: a pathway which likes to surprise. Kluwer, Dordrecht, pp 87–112
- Kopriva S, Suter M, von Ballmoos P, Hesse H, Krahenbuhl U, Rennenberg H, Brunold C (2002) Interaction of sulfate

assimilation with carbon and nitrogen metabolism in Lemna minor. Plant Physiol 130:1406–1413

- Korenkov V, Hirschi K, Crutchfield JD, Wagner GJ (2007) Enhancing tonoplast Cd/H antiporter activity increases Cd, Zn and Mn tolerance, and impacts root/shoot Cd partitioning in *Nicotiana tabaccum* L. Planta 226:1379–1387
- Krishnan HB (2005) Engineering soybean for enhanced sulfur amino acid content. Crop Sci 45:454–461
- Lang F, Sarvari E, Szigeti Z, Fodor F, Cseh E (1995) Effects of heavy metals on the photosynthetic apparatus in cucumber. In: Mathis P (ed) Photosynthesis: from light to biosphere IV. Kluwer, Dordrecht, pp 533–536
- Lappartient AG, Touraine B (1996) Demand-driven control of root ATP sulphurylase activity and SO₄⁻² uptake in intact canola. Plant Physiol 111:147–157
- Lappartient AG, Vidmar JJ, Leustek T, Glass ADM, Touraine B (1999) Interorgan signalling in plant: regulation of ATP sulfurylase and sulfate transporter genes expression in roots mediated by phloem-translocated compounds. Planta 18:89–95
- Leon AM, Palma JM, Corpas FJ, Gomez M, Romero-Puertas MC, Chatterjee D, Mateos RM, del Rio LA, Sandalio LM (2002) Antioxidant enzymes in cultivars of pepper plants with different sensitivity to cadmium. Plant Physiol Biochem 40:813–820
- Leustek T, Saito K (1999) Sulfate transport and assimilation in plants. Plant Physiol 120:637–643
- Leustek T, Martin MN, Bick J-A, Davies JP (2000) Pathways and regulation of sulfur metabolism revealed through molecular and genetic studies. Annu Rev Plant Physiol Plant Mol Biol 51:141–165
- Lin Ch-Ch, Chen L-M, Liu Z-H (2005) Rapid effect of copper on lignin biosynthesis in soybean roots. Plant Sci 168:855–861
- Liu CP, Shen ZG, Li XD (2007) Accumulation and detoxification of cadmium in *Brassica pekinensis* and *B. chinensis*. Biol Plant 51:116–120
- Löffler S, Hochberger A, Grill E, Winnacker E-L, Zenk MH (1989) Termination of the phytochelatin synthase reaction through sequestration of heavy metals by the reaction product. FEBS Lett 258:42–46
- Lopez E, Arce C, Oset-Gasque MJ (2006) Cadmium induces reactive oxygen species generation and lipid peroxidation in cortical neurons in culture. Free Radic Biol Med 40:940–951
- Lozano-Rodriguez E, Hernandez LE, Bonay P, Carpena-Ruiz RO (1997) Distribution of cadmium in shoot and root tissues of maize and pea plants: physiological disturbances. J Exp Bot 306:123–128
- Maksimović I, Kastori R, Krstić L, Luković J (2007) Steady presence of cadmium and nickel affects root anatomy, accumulation and distribution of essential ions in maize seedlings. Biol Plant 51:589–592
- Markovska YK, Gorinova NI, Nedkovska MP, Miteva KM (2009) Cadmium induced oxidative damage and antioxidant responses in *Brassica juncea* plants. Biol Plant 53:151–154
- Martínez M, Bernal P, Almela C, Vélez D, García-Augustín P, Serrano R, Navarro-Aviñó J (2006) An

engineered plant that accumulates higher levels of heavy metals than *Thlaspi caerulescens*, with yields of 100 times more biomass in mine soils. Chemosphere 64:478–485

- Maruyama-Nakashita A, Inoue E, Watanabe-Takahashi A, Yamaya T, Takahashi H (2003) Transcriptome profiling of sulfur-responsive genes in *Arabidopsis* reveals global effects of sulphur nutrition on multiple metabolic pathways. Plant Physiol 132:597–605
- Maughan S, Foyer CH (2006) Engineering and genetic approaches to modulating the glutathione network in plants. Physiol Plant 126:382–397
- Meinke DW, Chen J, Beachy RN (1981) Expression of storage-protein genes during soybean seed development. Planta 153:130–139
- Mendoza-Cozatl DG, Moreno-Sanchez R (2006) Control of glutathione and phytochelatin synthesis under cadmium stress. J Theor Biol 238:919–936
- Meyer AJ (2008) The integration of glutathione homeostasis and redox signaling. J Plant Physiol 165: 1390–1403
- Mittler R, Vanderauwera S, Gollery M, Breusegem FV (2004) Reactive oxygen gene network of plants. Trends Plant Sci 9:490–498
- Mobin M, Khan NA (2007) Photosynthetic activity, pigment composition and antioxidative response of two mustard (*Brassica juncea*) cultivars differing in photosynthetic capacity subjected to cadmium stress. J Plant Physiol 164:601–610
- Momose Y, Iwahashi H (2001) Bioassay of cadmium using a DNA microarray: genome-wide expression patterns of *Saccharomyces cerevisiae* response to cadmium. Environ Toxicol Chem 20:2353–2360
- Moons A (2003) Osgstu3 and Osgstu4, encoding tau class glutathione S-transferase, are heavy metal- and hypoxic stress-induced and differentially salt stressresponsive in rice roots. FEBS Lett 553:427–432
- Moussa HR, El-Gamal SM (2010) Effect of salicylic acid pretreatment on cadmium toxicity in wheat. Biol Plant 54:315–320
- Nazar R, Iqbal N, Masood A, Syeed S, Khan NA (2011) Understanding the significance of sulfur in improving salinity tolerance in plants. Environ Exp Bot 70:80–87
- Nikiforova V, Freitag J, Kempa S, Adamik M, Hesse H, Hoefgen R (2003) Transcriptome analysis of sulphur depletion in *Arabidopsis thaliana*: interlacing of biosynthetic pathways provides response specificity. Plant J 33:633–650
- Ning H, Zhang C, Yao Y, Yu D (2010) Overexpression of a soybean O-acetylserine (thiol) lyase-encoding gene GmOASTL4 in tobacco increases cysteine levels and enhances tolerance to cadmium stress. Biotechnol Lett 32:557–564
- Noctor G, Foyer CH (1998) A re-evaluation of the ATP: NADPH budget during C₃ photosynthesis. A contribution from nitrate assimilation and its associated respiratory activity? J Exp Bot 49:1895–1908
- Noctor G, Gomez LA, Vanacker H, Foyer CH (2002) Interactions between biosynthesis, comparmentation

and transport in the control of glutathione homeostasis and signaling. J Exp Bot 53:1283–1304

- Nriagu JO, Pacyna JM (1988) Quantitative assessment of worldwide contamination of air, water and soils by trace metals. Nature 333:134–139
- Nussbaum S, Schmutz D, Brunold C (1988) Regulation of assimilatory sulfate reduction by cadmium in Zea mays L. Plant Physiol 88:1407–1410
- Paek NC, Sexton PJ, Naeve SL, Shibles R (2000) Differential accumulation of soybean seed storage protein subunits in response to sulfur and nitrogen nutritional sources. Plant Prod Sci 3:268–274
- Pankovic D, Plesnicar M, Arsenijevic-Maksimovic I, Petrovic N, Sakac Z, Kastori R (2000) Effects of nitrogen nutrition on photosynthesis in Cd-treated sunflower plants. Ann Bot 86:841–847
- Pasricha NS, Aulakh MS, Bahl GS, Baddesha HS (1987) Nutritional requirements of oilseed and pulse crops in Punjab. Research Bulletin 15. PAU, Ludhiana
- Pence NS, Larsen PB, Ebbs SD, Letham DLD, Lasat MM, Garvin DF, Eide D, Kochian LV (2000) The molecular physiology of heavy metal transport in the Zn/Cd hyperaccumulator *Thlaspi caerulescens*. Proc Natl Acad Sci USA 97:4956–4960
- Perfus-Barbeoch L, Leonhardt N, Vavasseur A, Forestier C (2002) Heavy metal toxicity: cadmium permeates through calcium channels and disturbs the plant water status. Plant J 32:548–639
- Peters DPC, Pielke RA Sr, Bestelmeyer BT, Allen CD, Munson-McGee S, Havstad KM (2004) Cross-scale interactions, nonlinearities, and forecasting catastrophic events. Proc Natl Acad Sci USA 101:15130–15135
- Pietrini F, Iannelli MA, Pasqualini S, Massacci A (2003) Interaction of cadmium with glutathione and photosynthesis in developing leaves and chloroplasts of *Phragmites australis* (Cav.) Trin. ex Steudel. Plant Physiol 133:829–837
- Pilon-Smits EAH, Hwang S, Lytle CM, Zhu Y, Tai JC, Bravo RC, Chen Y, Leustek T, Terry N (1999) Overexpression of ATP sulfurylase in Indian mustard leads to increased selenate uptake, reduction and tolerance. Plant Physiol 119:123–132
- Plaza S, Bovet L (2008) Cadmium detoxification in plants: involvement of ABC transporters. In: Prasad MNV (ed) Trace elements as contaminants and nutrients: consequences in ecosystems and human health. Wiley, Weinheim, pp 449–469
- Pomponi M, Censi V, Di Girolama V, De Paolis A, di Toppi LS, Aromolo R, Costantino P, Cardarelli M (2006) Overexpression of *Arabidopsis* phytochelatin synthase in tobacco plants enhances Cd²⁺ tolerance and accumulation but not translocation to the shoot. Planta 223:180–190
- Popova LP, Maslenkova LT, Yordanova RY, Ivanova AP, Krantev AP, Szalai G, Janda T (2009) Exogenous treatment with salicylic acid attenuates cadmium toxicity in pea seedlings. Plant Physiol Biochem 47:224–231
- Popovic M, Kevresan S, Kandrac J, Nikolic J, Petrovic N, Kastori R (1996) The role of sulphur in detoxification

of cadmium in young sugar beet plants. Biol Plant 38:281–287

- Ranieri A, Castagna A, Scebba F, Careri M, Cagnoni I, Predieri G, Pagliari M, Sanità di Toppi L (2005) Oxidative stress and phytochelatin characterisation in bread wheat exposed to cadmium excess. Plant Physiol Biochem 43:45–54
- Rao ASVC, Reddy AR (2008) Glutathione reductase: a putative redox regulatory system in plant cells. In: Khan NA, Singh S, Umar S (eds) Sulfur assimilation and abiotic stress in plants. Springer, Netherland, pp 111–147
- Roth U, von Roepenack-Lahaye E, Clemens S (2006) Proteome changes in *Arabidopsis thaliana* roots upon exposure to Cd²⁺. J Exp Bot 57:4003–4013
- Rother M, Krauss GJ, Grass G, Wesenberg D (2006) Sulphate assimilation under Cd²⁺ stress in *Physcomitrella patens* – combined transcript, enzyme and metabolite profiling. Plant Cell Environ 29:1801–1811
- Rüegsegger A, Brunold C (1992) Effect of cadmium on γ-glutamylcysteine synthesis in maize seedlings. Plant Physiol 99:428–433
- Rüegsegger A, Schmutz D, Brunold C (1990) Regulation of glutathione synthesis by cadmium in *Pisum sativum* L. Plant Physiol 93:1579–1584
- Sairam RK, Srivastava GC, Saxena DC (2000) Increased antioxidant activity under elevated temperature–a mechanism of heat stress tolerance in wheat genotypes. Biol Plant 43:245–251
- Saito K (2000) Regulation of sulfate transport and synthesis of sulfur-containing amino acids. Curr Opin Plant Biol 3:188–195
- Sandalio LM, Dalurzo HC, Gomez M, Romero-Puertas MC, del Rio LA (2001) Cadmium-induced changes in the growth and oxidative metabolism of pea plant. J Exp Bot 52:2115–2126
- Sanita di Toppi L, Gabbrielli R (1999) Response to cadmium in higher plants. Environ Exp Bot 41:105–130
- Sarry JE, Kuhn L, Ducruix C, Lafaye A, Junot C, Hugouvieux V, Jourdain A, Bastien O, Fievet J, Vaihen D, Amekraz B, Moulin C, Ezan E, Garin J, Bourguignon J (2006) The early responses of *Arabidopsis thaliana* cells to cadmium exposure explored by protein and metabolite profiling analyses. Proteomics 7:2180–2198
- Sarwar N, Naeem A, Saifullah, Bibi S, Malhi SS, Farid G, Zia MH (2010) Role of mineral nutrition in minimizing cadmium accumulation by plants. J Sci Food Agric 90:925–937
- Schneider S, Bergmann L (1995) Regulation of glutathione synthesis in suspension cultures of parsley and tobacco. Bot Acta 108:34–40
- Schützendübel A, Polle A (2002) Cadmium and H_2O_2 induced oxidative stress in *Populus*×*canescens* roots. Plant Physiol Biochem 40:577–584
- Shah K, Kumar RG, Verma S, Dubey RS (2001) Effect of cadmium on lipid peroxidation, superoxide anion generation and activities of antioxidant enzymes in growing rice seedlings. Plant Sci 161:1135–1144
- Sharma SS, Dietz KJ (2006) The significance of amino acids and amino acid-derived molecules in plant

responses and adaptation to heavy metal stress. J Exp Bot 57:711–726

- Shewry PR, Pandya MJ (1999) The 2S albumin storage proteins. In: Shewry PR, Casey R (eds) Seed proteins. Kluwer, Dordrecht, pp 563–568
- Shulaev V, Oliver DJ (2006) Metabolic and proteomic markers for oxidative stress. New tools for reactive oxygen species research. Plant Physiol 141:367–372
- Siedlecka A (1995) Some aspects of interactions between heavy metals and plant mineral nutrients. Acta Soc Bot Pol 64:265–272
- Singh S, Anjum NA, Khan NA, Nazar R (2008) Metalbinding peptides and antioxidant defence system in plants: significance in cadmium tolerance. In: Khan NA, Singh S (eds) Abiotic stress and plant responses. IK International, New Delhi, pp 159–189
- Siripornadulsil S, Traina S, Verma DPS, Sayre RT (2002) Molecular mechanisms of proline-mediated tolerance to toxic heavy metals in transgenic microalgae. Plant Cell 14:2837–2847
- Skórzyńska-Polit E, Drazkiewicz M, Krupa Z (2003/04) The activity of the antioxidative system in cadmiumtreated Arabidopsis thaliana. Biol Plant 47:71–78
- Smeets K, Cuypers A, Lambrechts A, Semane B, Hoet P, Laere A, Vangronsveld J (2005) Induction of oxidative stress and antioxidative mechanisms in *Phaseolus vulgaris* after Cd application. Plant Physiol Biochem 43:437–444
- Smeets K, Ruytinx J, Semane B, Van Belleghem F, Remans T, Van Sanden S (2008) Cadmium-induced transcriptional and metabolic alterations related to oxidative stress. Environ Exp Bot 63:1–8
- Smeets K, Opdenakker K, Remans T, Van Sanden S, Van Belleghem F, Semane B (2009) Oxidative stress related responses at transcriptional and enzymatic level after exposure to Cd or Cu and in a multipollution context. J Plant Physiol 166:1982–1992
- Sneller FEC, Noordover ECM, Ten Bookum WM, Schat H, Bedaux JJM, Verkleij JAC (1999) Quantitative relationship between phytochelatin accumulation and growth inhibition during prolonged exposure to cadmium in *Silene vulgaris*. Ecotoxicology 8:167–175
- Sobkowiak R, Deckert J (2003) Cadmium-induced changes in growth and cell cycle gene expression in suspension-culture cells of soybean. Plant Physiol Biochem 41:767–772
- Stancheva I, Geneva M, Hristozkova M, Markovska Y, Salamon I (2010) Antioxidant capacity of sage grown on heavy metal polluted soil. Russ J Plant Physiol 57:799–805
- Strohm M, Jouanin L, Kunert KJ, Pruvost C, Polle A, Foyer C, Rennenberg H (1995) Regulation of GSH synthesis in leaves of transgenic poplar (*Populus tremula*×*P. alba*) overexpressing GSH synthetase. Plant J 7:141–145
- Stroiński A, Floryszak-Wieczorek J, Woźny A (1990) Effects of cadmium on the host-pathogen system. I. Alteration of potato leaves and Phytophthora infestans relations. Biochem Physiol Pflanzen 186:43–54
- Stroiński A, Chadzinikolau T, Giżewska K, Zielezińska M (2010) ABA or cadmium induced phytochelatin synthesis in potato tubers. Biol Plant 54:117–120

- Sun JY, Shen ZG (2007) Effects of Cd stress on photosynthetic characteristics and nutrient uptake of cabbages with different Cd-tolerance. Ying Yong Sheng Tai Xue Bao 18:2605–2610
- Sun X, Sun XM, Yang ZM, Wang SQ, Li J, Wang SH (2005) Expression of *Brassica napus* L. γ-Glutamylcysteine synthetase and low- and high-affinity sulfate transporters in response to excess cadmium. J Integr Plant Biol 47:243–250
- Takahashi H, Watanabe-Takahashi A, Smith F, Blake-Kalff M, Hawkesford M, Daito K (2000) The role of three functional sulfate transporters involved in uptake and translocation of sulphate in *Arabidopsis thaliana*. Plant J 23:171–182
- Tandon HLS (1991) Sulphur research and agricultural production in India, 3rd edn. The Sulphur Institute, Washington, DC
- Tausz T, Šircelj H, Grill D (2004) The glutathione system as a stress marker in plant ecophysiology: is a stress– response concept valid? J Exp Bot 55:1955–1962
- Thomine S, Lelievre F, Debarbieux E, Schroeder JI, Barbier-Brygoo H (2003) AtNRAMP3, a multispecific vacuolar metal transporter involved in plant responses to iron deficiency. Plant J 34:685–695
- Tlustos P, Szakova J, Korinek K, Pavlikova D, Hanc A, Balik J (2006) The effect of liming on cadmium, lead and zinc uptake reduction by spring wheat grown in contaminated soil. Plant Soil Environ 52:16–24
- Tukiendorf A, Baszynski T (1991) The in vivo effect of cadmium on photochemical activities in chloroplasts of runner bean plants. Acta Physiol Plant 13:51–57
- Tukiendorf A, Rauser WE (1990) Changes in glutathione and phytochelatins in roots of maize seedlings exposed to cadmium. Plant Sci 70:155–166
- Umar S, Diva I, Anjum NA, Iqbal M (2008) Research Findings: II Potassium nutrition reduces cadmium accumulation and oxidative burst in mustard (*Brassica campestris* L.). e-ifc no. 16
- Uraguchi S, Watanabe I, Yoshitomi A, Kiyono M, Kuno K (2006) Characteristics of cadmium accumulation and tolerance in novel Cd-accumulating crops, Avena strigosa and Crotalaria juncea. J Exp Bot 57:2955–2965
- Van Assche F, Clijsters H (1990) Effects of metals on enzyme activity in plants. Plant Cell Environ 13:195–206
- van de Mortel JE, Schat H, Moerland PD, Loren V, van Themaat E, Van Der Ent S, Blankestijn H, Ghandilyan A, Tsiatsiani S, Aarts MG (2008) Expression differences for genes involved in lignin, glutathione and sulphate metabolism in response to cadmium in *Arabidopsis thaliana* and the related Zn/Cd hyperaccumulator *Thlaspi caerulescens*. Plant Cell Environ 31:301–324
- Vassilev A, Yordanov I, Chakalova E, Kerin V (1995) Effect of cadmium stress on growth and photosynthesis of young barley (*H. vulgare* L.) plants. 2. Structural and functional changes in photosynthetic apparatus. Bulg J Plant Physiol 21:12–21
- Vauclare P, Kopriva S, Fell D, Suter M, Sticher L, von Ballmoos P, Krähenbühl U, Opden Camp R, Brunold C (2002) Flux control of sulphate assimilation in *Arabidopsis thaliana*: adenosine 5 -phosphosulphate

reductase is more susceptible than ATP sulphurylase to negative control by thiols. Plant J 31:729–740

- Verbruggen N, Hermans C, Schat H (2009) Mechanisms to cope with arsenic or cadmium excess in plants. Curr Opin Plant Biol 12:1–9
- Vido K, Spector D, Lagniel G, Lopez S, Toledano MB, Labarre J (2001) A proteome analysis of the cadmium response in *Saccharomyces cerevisiae*. J Biol Chem 276:8469–8474
- Walker KC, Booth EJ (1992) Sulfur research on oilseed rape in Scotland. Sulfur Agric 16:15–19
- Wang L, Zhou Q, Ding L, Sun Y (2008) Effect of cadmium toxicity on nitrogen metabolism in leaves of *Solanum nigrum* L. as a newly found cadmium hyperaccumulator. J Hazard Mater 154:818–825
- Weber M, Trampczynska A, Clemens S (2006) Comparative transcriptome analysis of toxic metal responses in *Arabidopsis thaliana* and the Cd⁽²⁺⁾ hypertolerant facultative metallophyte *Arabidopsis halleri*. Plant Cell Environ 29:950–963
- Wirtz M, Hell R (2006) Functional analysis of the cysteine synthase protein complex from plants: structural, biochemical and regulatory properties. J Plant Physiol 163:273–286
- Wu FB, Zhang G-P (2002) Alleviation of cadmium-toxicity by application of zinc and ascorbic acid in barley. J Plant Nutr 25:2745–2761
- Wu FB, Zhang GP, Dominy P (2003) Four barley genotypes respond differently to cadmium: lipid peroxida-

tion and activities of antioxidant capacity. Environ Exp Bot 50:67–77

- Xiang C, Werner BL, Christensen EM, Oliver DJ (2001) The biological functions of glutathione revisited in *Arabidopsis* transgenic plants with altered glutathione levels. Plant Physiol 126: 564–574
- Zhang G-P, Motohiro F, Hitoshi S (2002) Influence of cadmium on mineral concentrations and yield components in wheat genotypes differing in Cd tolerance at seedling stage. Field Crops Res 77: 93–99
- Zhao FJ, Hamon RE, Lombi E, McLaughlin MJ, McGrath SP (2002) Characteristics of cadmium uptake in two contrasting ecotypes of the hyperaccumulator *Thlaspi caerulescens*. J Exp Bot 53:535–543
- Zhu YL, Pilon-Smits EAH, Jouanin L, Terry N (1999a) Overexpression of glutathione synthetase in Indian mustard enhances cadmium accumulation and tolerance. Plant Physiol 119:73–79
- Zhu YL, Pilon-Smits EAH, Tarun AS, Weber SU, Jouanin L, Terry N (1999b) Cadmium tolerance and accumulation in Indian mustard is enhanced by overexpressing gamma-glutamylcysteine synthetase. Plant Physiol 121:1169–1178
- Zornoza P, Vázquez S, Esteban E, Fernández-Pascual M, Carpena R (2002) Cadmium-stress in nodulated white lupin: strategies to avoid toxicity. Plant Physiol Biochem 40:1003–1009

Role of Salicylic Acid in Alleviating Heavy Metal Stress

Losanka P. Popova, Liliana T. Maslenkova, Albena Ivanova, and Zhivka Stoinova

Abstract

Both plant breeders and crop producers have an interest in finding crops capable of tolerating environmental changes with damage as little as possible. In order to develop such crops, the knowledge of plant defense mechanisms and regulatory processes is essential. The study presented in this chapter was performed to analyze the role of salicylic acid (SA) in regulation of plant growth and development, flowering, ion uptake, stomatal regulation and photosynthesis. The role of SA in development of plant resistance to different environmental stresses is described. Besides the physiological functions of SA, the general properties, biosynthesis and metabolism of this plant growth regulator are discussed. The present chapter focuses on the mechanisms of the beneficial effect of SA on maize plants exposed to toxic Cd concentrations.

Exposure of plants to Cd (10, 15 and 25 μ M) caused a gradual decrease in the dry weight accumulation of shoots and roots. Pretreatment of seeds with 500- μ M SA for 6 h alleviated the negative effect of Cd on plant growth parameters. The same tendency was observed for the chlorophyll level. The rate of CO2 fixation was lower in Cd-treated plants, and the inhibition was partially overcome in SA-pretreated plants. A drop in the activities of carboxylating enzymes ribulose-1,5-bisphosphate carboxylase (RuBPCase) and phosphoenolpyruvate carboxylase (PEPCase) was observed for Cd-treated plants. Pretreatment with SA alleviated the inhibitory effect of Cd on the enzymes activity. In vivo the excess of Cd-induced alterations in the redox cycling of oxygen-evolving centers and the assimilatory capacity of maize leaves as revealed by changes in the termoluminescence emission. Pretreatment with SA before imposition of high

· Z. Stoinova

Bulgarian Academy of Sciences, Institute of Plant Physiology, Acad. G. Bonchev str, BL. 21, 1113 Sofia, Bulgaria e-mail: lpopova@bio21.bas.bg

P. Ahmad and M.N.V. Prasad (eds.), Environmental Adaptations and Stress Tolerance

of Plants in the Era of Climate Change, DOI 10.1007/978-1-4614-0815-4_21,

L.P. Popova (🖂) • L.T. Maslenkova • A. Ivanova

[©] Springer Science+Business Media, LLC 2012

concentration of Cd has a stabilizing effect on photochemical reactions. Changes in the activity of several important antioxidative enzymes, namely superoxide dismutase (SOD), ascorbate peroxidase (APX), catalase (CAT), guaiacol peroxidase (POD), glutathione reductase (GR) and glutathione-*S*-transferase (G-S-Tr) were measured. The presence of Cd in the nutrient solution led to disturbances in the activity of the antioxidant enzymes. Pretreatment with SA alleviated the negative effect of Cd on the studied enzymes. Our results suggest that the phytotoxicity of Cd is mainly induced by oxidative stress and SA is involved in the defense responses of maize plants to Cd exposure. This suggestion was consistent with the observed protective role of SA on the lipid membranes of Cd-treated maize plants.

Keywords

Maize • Oxidative stress • Photosynthesis • Thermoluminescence • Salicylic acid

1 Introduction

Salicylic acid (SA) is a widespread phenolic compound, and many of its physiological and biochemical effects have been known for a long time (Raskin 1992a). As a potent signaling molecule in plants it is involved in eliciting specific responses to biotic and abiotic stresses. It has been shown that SA provides protection in maize plants against different stresses, e.g., low-temperature stress (Janda et al. 1999; Tasgin et al. 2003) induces thermotolerance in mustard seedlings (Chen et al. 1997; Dat et al. 1998a, b) or modulates plant responses to salt and osmotic stresses (Borsani et al. 2001), ozone or UV light (Sharma et al. 1996), drought (Senaratna et al. 2000; Singh and Usha 2003), herbicides (Ananieva et al. 2004) or pathogens (Malamy et al. 1990; Durner et al. 1997). Furthermore, SA is also known to be involved in plant protection to heavy metals. SA pretreatment alleviates Pb- and Hg-induced membrane damage in rice (Mishra and Chudhuri 1999) and Cd toxicity in barley (Metwally et al. 2003) and maize plants (Pal et al. 2002).

Cadmium (Cd²⁺) is a highly toxic trace element which enters the environment mainly from industrial processes and phosphate fertilizers. It can reach high levels in agricultural soils and is easily assimilated by plants. Taken up in excess by plants it induces various visible symptoms of phytotoxicity, for example, leaf roll, chlorosis and growth reduction of root and shoot, browning of root tips and finally death (Kahle 1993). A large number of studies have demonstrated the toxic effect of Cd on of photosynthesis through effects on the chlorophyll metabolism and chloroplasts structure (Gadallah 1995), the activity of both photosystem II and the enzymes of photosynthetic carbon metabolism (Krantev et al. 2008; Popova et al. 2009). Cadmium also produces alterations in the functionality of membranes by inducing changes in their lipid composition (Hernandez and Cooke 1997), and this can affect some enzymatic activities associated with membranes such as H⁺-ATPase (Fodor et al. 1995). Once Cd enters the cytosol, other detoxification mechanisms are induced, primarily the formation of complexes between Cd and phytochelatins (PCs) and their subsequent compartmentalization (Vazquez et al. 2006). Other mechanisms that plants have developed to cope with damages caused by cadmium are related with some stress signaling molecules, such as SA, jasmonic acid and ethylene. All these compounds were induced by Cd treatment, which suggest that they are involved in cell response to Cd toxicity (Rodriguez-Serrano et al. 2006).

Although there have been many reports on the photochemical and biochemical events occurring

in photosynthesis during Cd toxicity, a lot of contradictory data can be found in the literature. Probably this is because of the very heterogeneous experimental approaches, including both laboratory-grown conditions and field experiments. Only a limited number of studies have been carried out during the germinating stage of plants. We have attempted to study the effect of exposure of maize plants to Cd during early stages of their establishment, on the physiological and biochemical properties of maize leaves. Our main purpose was to determine the physiological and biochemical changes in maize plants treated by SA during Cd-induced stress, to investigate whether this plant regulator is involved in the induction of defense response and to test the hypothesis that the observed protection of SA on photosynthesis against Cd stress is mediated by its effect on antioxidant defense system.

2 SA: Properties, Structure and Biosynthesis

SA or ortho-hydroxybenzoic acid and related compounds belong to a diverse group of plant phenolics. Salicylates from plant sources have been used in medicines since antiquity. In 1828, in Munich, was isolated for the first time a small amount of salicin, the glucoside of salicyl alcohol, from willow bar. Ten years latter Raffaele Piria named it SA, from the Latin word Salix for willow tree. Aspirin, a trade name for acetylsalicylic acid (ASA), undergoes spontaneous hydrolysis to SA. It is rapidly converted to SA when applied exogenously. Despite the fact that aspirin was not identified as a natural product, it is widely used by many plant scientists in their experiments. The reason is the similarity in their physiological effects (Popova et al. 1997; Hayat et al. 2010).

SA could be actively transported, metabolized or conjugated, and it could also translocate rapidly from the point of initial application to different plant tissues. Salicylates are distributed in many important agricultural plant species. In many plants, such as rice, crabgrass, barley, soybean, etc., the levels of SA has been found to be approximately 1 µg g⁻¹ fresh weight. A survey of SA in leaves and reproductive structures of nonthermogenic angiosperms confirmed the ubiquitous distribution of this compound in plants. Levels of SA varied substantially in the floral parts of seven nonthermogenic species and in the leaves of 27 nonthermogenic plants. The highest levels of SA were determined in the inflorescence with necrotizing pathogens (Raskin et al. 1990).

In the early 1960s it was suggested that in plants SA is synthesized from cinnamic acid by two possible pathways: one involves side-chain decarboxylation of cinnamic acid to benzoic acid followed by 2-hydroxylation to SA. Alternatively, cinnamic acid could be first 2-hydroxylated to an o-coumaric acid and then decarboxylated to SA. These pathways differ in the order of B-oxidation and ortho-hydroxylation reactions and could operate independently in plants. Two key enzymes are involved in SA biosynthesis and metabolism: benzoic acid 2-hydroxylase, which converts benzoic acid to SA, and UDP glucose: SA glucosyltransferase, which catalyzes conversion of SA to SA glucoside. It was shown that the cinnamic acid \rightarrow benzoic acid \rightarrow SA pathway functions in rice seedlings (Sillverman et al. 1995) and tobacco plants (Yalpani et al. 1993). However, recently, genetic studies in Arabidopsis have shown that SA was also produced via chorismic acid - an important intermediate of shikimic acid pathways, and the rate of SA biosynthesis and excretion can be substantial (Wildermuth et al. 2001).

Some publications have suggested that the rate-limiting step in SA biosynthesis is the conversion of *trans*-cinnamic acid to benzoic acid, and that this involves a ß-oxidation pathway in which *trans*-cinnamoyl CoA is the first of four intermediates. However, evidence is increasing for an alternative, as yet undefined, rout to SA that does not involve benzoic acid (Hayat et al. 2010).

Many different types of SA conjugates have been found in plant species. They were mainly formed by glucosylation and less frequently by esterification. Large amounts of SA glucosides were detected in sunflower, oat and bean roots (Yalpani et al. 1992a, b). Methyl salicylate was found in the leaves of oats, red clover, tobacco and in the volatile fractions of fruits (plum, strawberry, black cherry and tomato). Large amounts of volatile methyl salicylate are released from tobacco mosaic virus (TMV)-inoculated tobacco indicating that this compound may be a major metabolite of SA. It is tempting to speculate that methylsalicylate may function as an airborne signal for both intra- and inter-plant communication. SA forms also conjugates with aminoacids. Salicyloyl aspartic acid was identified in wild grapes, but its physiological function is not known (Sillverman et al. 1995).

3 Physiological Functions of Salicylates

The first indication for a physiological effect of SA was the discovery of flower-inducing action and bud formation in tobacco cell cultures (Eberhard et al. 1989). The stimulatory effect of SA on flowering was later demonstrated in other plant species, and this was ground for suggesting that SA functions as an endogenous regulator of flowering. The SA effect was not specific and it promoted flowering in combination with other plant regulators (e.g., gibberellins).

The ability of certain Arum lilies to generate heat in the inflorescences during blooming has provoked studies on the role of SA in heat production. When SA was applied to the inflorescence exogenously, it elicited heat production in spadix. On the day proceeding blooming, the levels of endogenous SA in the spadix increased 100-fold to 1 μ g g⁻¹ fresh weight and returned to control levels at the end of the thermogenic period. By this way SA appears to be calorigen. The heating is believed to be associated with a large increase in the cyanide-insensitive, nonphosphorylating electron transport pathway. It was also shown that SA caused the induction of the alternative oxidase gene and the alternative oxidase protein with molecular mass of 38.9 kDa was isolated and characterized (Rhoads and McIntosh 1991). The effects of SA on the alternative pathway respiration in slices and isolated mitochondria of dormant and dormancy-breaking potato tubers were compared (Wen and Liang 1994). It was found that treatment with $20-\mu M$ SA increased the capacity of cyanide-resistant

respiration in both model systems. The involvement of the alternative pathway was enhanced by SA to a greater extent in dormancy-breaking potato tubers. The thermal effect of SA was abolished with inhibitors of the alternative pathway, such as salicylhydroxamic acid and propyl gallate. It was suggested that, as in thermogenic species, SA increased both the activity of the total respiration and the cyanide-resistant pathway in tobacco leaves, leading to an elevation in surface temperature (Van der Straeten et al. 1995).

There are experimental data indicating participation of SA in signal regulation of gene expression in the course of leaf senescence in *Arabidopsis* (Morris et al. 2000). Moreover, SA might serve as a regulator of gravitropism, and the inhibition of fruit ripening (Srivastava and Dwivedi 2000). SA is an effective inhibitor of ethylene biosynthesis, the effect being pH-dependent.

More widespread interest in SA was generated when it was closely linked to the hypersensitive response, a disease-resistant mechanism in which plants restrict the spread of fungal, bacterial or viral pathogens by producing necrotic lesions around the initial point of penetration. Treatment of tobacco plants with aspirin solution showed that plants have enhanced resistance to TMV and reduced number and size of necrotic lesions. Treatment of tobacco genotypes with SA resulted in the coordinate expression of pathogenesis-related (PR) genes. Several lines of evidence suggest that SA is the endogenous signal involved in induction of PR protein synthesis and systemic acquired resistance (SAR) in tobacco and cucumber. It has been established that during infection by pathogens, a huge number of proteins are induced, including PR protein families, which are thought to play an active role in defense processes.

Resistance to pathogens and the production of some PR proteins in plants can be induced by SA or aspirin, even in the absence of pathogenic organisms. Exogenously applied SA-induced PR proteins mainly at the side of application, in contrast to pathogens that induced PR proteins systemically (Hayat et al. 2010). The levels of SA increased dramatically following inoculation of tobacco or cucumber plants with pathogens. Increases in SA levels have been correlated to changes in gene expression. The production of active oxygen species such as superoxide ions is a rapid response that precedes cell death. It has been proposed that oxygen radicals play a direct role in cell death during the hypersensitive response. H_2O_2 formed during the oxidative burst may also trigger cell death and serve as a diffusible signal for induction of defense-related genes in surrounding cells (Levine et al. 1994). Leon et al. (1995) have demonstrated that treatment of tobacco leaves with H_2O_2 induced accumulation of free benzoic acid and SA. They have suggested that H_2O_2 activated SA biosynthesis.

Wounding the leaves by chewing insects or other mechanical damage induces the synthesis of defensive proteinase inhibitor proteins in both wounded leaves and distal unwounded leaves. Stomatal behavior and regulation are a very important factor for photosynthetic ability. The established effects of SA on stomatal function, chlorophyll content, transpiration rate and respiratory pathways lead to the assumption that SA might possess another physiological function, most probably involved in regulation of some photosynthetic reactions. Since the past 15 years our team had studied the role of SA in regulation of photosynthesis. Pancheva et al. (1996) demonstrated that long-term treatment (7 days) of barley seedlings with SA decreased the rate of photosynthesis and the activity of RuBP carboxylase, and increased both CO₂-compensation point and stomatal resistance. The short-term treatment with SA (from minutes to 2 h) did not affect either the rate of photosynthesis or the capacity of biochemical machinery as compared with untreated control plants. An explanation for these changes could in part be because of stomatal closure and reduced supply of CO₂. However, the Ci-values were not declined in SA-treated plants. This implies that stomatal closure did not restrict CO₂ entry into the leaf enough to reduce internal CO_2 level and the reduction in photosynthesis probably was nonstomatal. It was suggested that one very possible reason for the observed inhibition of photosynthetic ability and RuBP carboxylase activity could be the effect of SA on protein synthesis, including ribulose-1,5-bisphosphate

carboxylase/oxygenase (RubisCO) synthesis. It was also found that treatment of barley plants with SA caused decrease in the level of total soluble protein, in particular in the level of RubisCO. The percentage of inhibition on the small subunit was higher, as a result of which the small/large subunit ratio was lower for the experimental variants (Pancheva and Popova 1998). It was suggested that SA like some other stress factors diminished chloroplast photosynthetic activity as a result of effects on the thylakoid membranes and light-induced reactions connected with them, and in this way it may indirectly participate in regulating the activity of RubisCO. Additional data showed that treatment of barley seedlings with SA caused alterations in leaf anatomy and chloroplast ultrastructure (Uzunova and Popova 2000). The suggestion was that the observed low rate of growth and photosynthesis might be associated with disorders of leaf anatomy and plastid ultrastructure. At the same experimental conditions Maslenkova and Toncheva (1998) observed the inhibitory effect of SA on PSII oxygenevolving reactions. The effect of SA depended on the time of treatment duration – no changes in these parameters were observed when barley seedlings were treated with SA for 2 h, the inhibition appeared 6 h after the start of treatment. These data confirm the suggestion that SA plays different roles based on its endogenous levels in particular plant species under specific developmental and environmental conditions (Pancheva et al. 1996).

The ability of SA and ASA to induce a protective effect on plants under stress received considerable interest during the past few years. It has been shown that SA provides protection in maize (Janda et al. 1999) and winter wheat plants (Tasgin et al. 2003) against low-temperature stress, induces thermotolerance in mustard seedlings (Chen et al. 1997; Dat et al. 1998b) or modulates plant responses to salt and osmotic stresses (Borsani et al. 2001), ozone or UV light (Sharma et al. 1996), drought (Senaratna et al. 2000) and herbicides (Ananieva et al. 2004). SA has been shown to accumulate in plants in response to various oxidizing stresses, such as H_2O_2 (Leon et al. 1995), ozone (Sharma et al. 1996), heat



Fig. 21.1 Physiological effects of SA and factor affecting changes in its endogenous accumulation

(Dat et al. 1998b), and it has been suggested that it is directly involved in signaling different antioxidant responses (Larkindale and Knight 2002). These relationships are presented in Fig. 21.1.

4 SA and Heavy Metal Stress

The environmental quality is endangered by many adverse effects. The most important problem for plant ecosystems is a consequence of the contamination by toxic wastes, biotic stress factors and abnormal climate changes. In addition to industrial pollution (heavy metals, pesticides, oil residues, etc.), the global warming and dehydration as well as pathogen infection are recognized as environmental constraints that the vegetation endures.

Heavy metals are used in various industries and are consequently discharged into the environment. At least 20 metals are known to be toxic, and fully half of these, including cadmium (Cd), copper (Cu), mercury (Hg), nickel (Ni), silver (Ag) and zinc (Zn), are released into the environment in sufficient quantities to present a risk to human health. Soil heavy metal contamination is increasingly severe worldwide and the global average annual emission of Hg, Cu, Pb, Mn and Ni are 15,000 tons, 3.4 million tons, 5 million tons, 15 million tons and 1 million tons, respectively (Popova and Lu 2010).

Cadmium (Cd) is a trace metal, which may accumulate to potential harmful levels in certain crops. It is one of the most toxic pollutants found in air, water and soil. It is highly toxic to plants and animals. Accumulation of cadmium in plant tissue is influenced by the levels of cadmium available in the soil in which the plants are grown. It should be mentioned that phosphate fertilizers additionally increase the level of Cd in soil. Cadmium is easily taken up by plant roots and can be loaded into the xylem for its transport into leaves. Cd induces genetic and biochemical changes in plant metabolism related to general and Cd-specific stress responses. A large number of studies have demonstrated the toxic effect of Cd on plant metabolism, such as decreased uptake of nutrient elements (Sandalio et al. 2001), changes in nitrogen metabolism (Boussama et al. 1999), inhibition of photosynthesis through effects on the chlorophyll metabolism and chloroplasts structure (Gadallah 1995; Stoyanova and Tchakalova 1997; Stoyanova and Merakchiiska-Nikolova 1992), inhibition of stomatal opening (Barcelo and Poschenrieder 1990) and activity of both photosystem II and the enzymes of photosynthetic carbon metabolism (Krupa and Baszynski 1995; Krantev et al. 2008; Popova et al. 2009). The main targets of the influence of Cd are two key enzymes of CO₂ fixation, RubisCO and phosphoenolpyruvate carboxylase (PEPCase). It has been shown that Cd^{2+} ions lower the activity of ribulose-1,5-bisphosphate carboxylase (RuBPCase) and damage its structure by substituting for Mg²⁺ ions, which are important cofactors of carboxylation reactions and also Cd can shift RuBPCase activity toward oxygenation reactions (Siedlecka et al. 1998). Stiborova (1988) and Malik et al. (1992a, b) demonstrated that Cd caused an irreversible dissociation of the large and small subunits of RuBPCase thus leading to total inhibition of the enzyme. The early studies indicated that Cd ions affect the oxidizing side of PS2 and lead to uncoupling of electron transport in chloroplasts (Atal et al. 1993; Mohantly and Mohanty 1988). With regard to site and mechanism of inhibition of Cd, it is generally accepted that the wateroxidizing complex (OEC) of PS2 is affected by Cd by replacing the Ca²⁺ in Ca/Mn clusters that constitutes the oxygen-evolving centers (Sigfridsson et al. 2004) or by some modifications in Qb-binding site (Geiken et al. 1998).

Cadmium ions are known to cause alterations in the functionality of membranes by affecting the lipid composition (Quariti et al. 1997) and some enzymatic activities associated with membranes, such as H⁺-ATPase (Fodor et al. 1995). Evidences suggest that cadmium toxicity induces oxidative stress, as a result of stimulation of free oxygen radical production (Sanita di Toppi and Gabrielli 1999) and by modified activity of various antioxidant enzymes (Hagedus et al. 2001). To avoid Cd toxicity, plants adopt various defense strategies, including phytochelation and sequestration as well as induction of antioxidant machinery and stress proteins (Cobbett and Goldsbrough 2002; Vazquez et al. 2006). The prevailing part of cell Cd is bound to phytochelatins (PC) and transported into vacuole as PC–Cd–S complex through Fe-dependent transporters (Hall and Williams 2003). Survival under stressful conditions depends on the plant's ability to perceive the stimulus, generate and transmit signals, and induce biochemical changes that adjust the metabolism accordingly. Therefore, the search for signal molecules that mediate the stress tolerance is an important step in our better understanding on how plants acclimate to the adverse environment.

Convincing data have been obtained on SA-induced plant protection to heavy metals, demonstrating the involvement of SA in the induction of different antistress programs. SA pretreatment alleviates Pb-induced membrane damage in rice (Mishra and Chudhuri 1999), Hg-induced oxidative stress in Medicago sativa (Zhou et al. 2009), and Cd toxicity in barley (Metwally et al. 2003) and maize plants (Pal et al. 2002; Krantev et al. 2008). Recently He et al. (2010) showed that pretreatment of rice seeds with 0.1-mM SA for 24 h alleviated the negative effect of Cd on germination parameters and early seedlings growth. Kovacik et al. (2009) studying the effect of SA on Cd and Ni uptake in Chamomile (Matricaria chamomilla) reported that it is differentially modulated by SA. In case of Ni, it seems that SA serves as a root barrier in order to prevent Ni from reaching the aboveground organs. The exogenous application of SA conferred Al tolerance to the plants of Cassia tora, exposed to Al toxicity that was mediated by an increase of citrate efflux in roots of the treated plants (Yang et al. 2003). Shi and Zhu (2008) reported that exogenous SA alleviated the toxicity generated in Cucumis sativus by manganese exposure and the response was mediated by reduction in ROS level and lipid peroxidation.

Recent experiments revealed a clear relationship between metal stress and redox homeostasis and antioxidant capacity. It was proposed that SA serves as a link between the degree of plant tolerance to metals and the level of antioxidants (Sharma and Dietz 2009).

5 Role of SA in the Regulation of Cadmium Toxicity in Maize (*Zea Mays* L.)

Different plant species and varieties show a wide rage of plasticity in Cd tolerance, reaching from the high degree of sensitivity to the hyper accumulating phenotype of some tolerant plants. Legume plants are less tolerant to Cd toxicity than cereals and grasses (Metwally et al. 2005). Although there have been many reports on the photochemical and biochemical events occurring in photosynthesis during Cd toxicity, a lot of contradictory data can be found in the literature. Probably this is because of the very heterogeneous experimental approaches, including both laboratory-grown conditions and field experiments. Only a limited number of studies have been carried out during the germinating stage of plants. We have attempted to study the effect of exposure of maize plants to Cd during the early stages of their establishment, on their physiological and biochemical properties. The study reported here is mainly focused on the mechanisms by which SA influences the photosynthetic processes to overcome Cd toxicity in maize plants. In appropriate parts of the work we tried to compare the obtained results with maize to pea plants because they differ in their sensitivity to Cd toxicity. It has been shown that Cd produced a concentrationdependant reduction in the growth of maize and pea. Under our experimental conditions, pea plants did not tolerate Cd concentrations higher than 5 µM without showing any visible toxicity symptoms. This confirms the data by other authors (Belimov et al. 2003) that pea plants can be considered as cadmium-sensitive species.

Maize plants grown for 14 days with 10-, 15and 25- μ M CdCl₂ exhibited a significant inhibition of growth. After 14 days of treatment, plants grown on 25- μ M Cd showed visible toxicity symptoms and the survival rate declined greatly.

The comparison of the growth injuries of both plant species confirmed that maize plants are able to tolerate higher Cd concentrations. In general, it can be said that the sensitivity of given plant species to heavy metal toxicity depends on its concentration, the treatment duration on the plant species, its age and the plant organ examined.

Next data will be focused only on physiological changes of maize plants to Cd toxicity and on the protective role of SA.

Changes in some parameters associated with oxidative stress, namely proline production, lipid peroxidation, CO_2 fixation and the activity of the carboxylating enzymes RuBPCase and PEPCase, were assayed as they are known to be most affected by Cd treatment. The other goal was to test the hypothesis that the observed protection of SA on photosynthesis against Cd stress is mediated by its effect on antioxidant defense system. also In addition we tried to explain the possible effect of SA on Cd toxicity with events leading to detoxification of cadmium.

5.1 CO₂ Assimilation Rate and Carboxylating Enzymes Activity

Here we demonstrated that growth inhibition of maize plants was accompanied by a decrease in the rate of photosynthetic CO_2 fixation expressed on both fresh weight and chlorophyll content bases because Cd treatment caused a reduction in chlorophyll level.

The activities of both carboxylating enzymes (RuBPCase and PEPCase) were also affected by Cd treatment. PEPCase activity was reduced only after exposure to $25-\mu$ M Cd, while RuBPCase activity exhibited a strong reduction at all Cd concentrations applied.

Pretreatment of maize plants with SA before exposure to Cd alleviated the inhibitory effect of Cd and led to nearly a twofold increase in PEPCase activity compared with untreated plants. A very strong protective effect of SA was observed on RuBPCase activity (Fig. 21.2).

5.2 Thermoluminescence Emission in Cd-Stressed and SA-Pretreated Maize Leaves

The illumination of unfrozen dark adapted leaves from nontreated (control) maize plants with two consecutive flashes (2Fl) induced a main TL peak





Fig. 21.2 Photosynthetic CO_2 fixation rate and activities of carboxylating enzymes RuBPCase and PEPCase in maize plants treated with Cd or pretreated with SA before exposure to Cd. Dry seeds were soaked in 500-µM SA (*black bars*) or

water (*white bars*) for 6 h and were germinated for 3 days in moist filter paper. They were grown for 14 days in hydroponic medium without Cd or with Cd (in the respective concentrations). Data are means \pm s.e. (n=3) from three experiments

at 35°C, the so-called B-band (S₂Qb⁻) and a hardly distinguishable shoulder at temperature about 42°C (Fig. 21.3), designated as afterglow (AG). The B-band results from the thermal activated recombination of the trapped electrons and positive charges on the reduced quinone acceptor (Qb^{-}) and the $S_2(S_2)$ oxidation state of the water-oxidizing complex of PSII, respectively (Rutherford et al. 1982). The AG TL emission corresponds to a back electron transfer toward PSII centers initially in the $S_{2(3)}QB$ state (Ducruet 2003). Increasing Cd concentrations decrease the overall TL intensity, the inhibitory effect of B-band amplitude being more expressed in comparison with AG emission (Fig. 21.3). A damping of period-four oscillation of B-band, according to exciting flash number, was also observed (not shown). The characteristic damping of oscillation of B-band intensity can be ascribed to already known effects of Cd on PSII, by increasing misses in charge separation. Short-term application of 500-mM SA to the maize seeds did not exert substantial changes in the investigated TL parameters. However, pretreatment with SA before the imposition of high concentration of heavy metal has a stabilizing effect on photochemical reactions, judging from some restoration of TL bands amplitude (Fig. 21.3) and B-band oscillation pattern (not shown).

5.3 Effect of SA on Chlorophyll Content, Proline Level and Relative Water Content

Chlorophyll content decreased only in the variant treated with $25-\mu M$ Cd and the inhibition was partially overcome in those pretreated with SA seeds. The concentration of the stress metabolite proline increased upon Cd exposure. The most



Fig. 21.3 Changes in thermoluminescense glow curves of maize plants treated with 15- and 25-mM Cd or pretreated with 500-mM SA before exposure to heavy metal.

TL was recorded immediately after two flashes excitation, with a $0.5^{\circ}C \text{ s}^{-1}$ heating rate

prominent effect was observed at 15–25 μ M Cd (a nearly twofold rise compared with the control). SA pretreatment counteracted the Cd-induced increase in proline levels. Exposure of maize plants to Cd led to a slight decrease in leaf RWC. The values of this parameter for the control plants were 95–97% and 92–93% for H₂O- and SA-presoaked plants, respectively. The reduction in RWC was approximately 6 and 11% for H₂O- and SA-presoaked plants, respectively, in the variants treated with 10–15 μ M Cd (Table 21.1).

5.4 Effect of SA and Cd on the Amount of Biomass, Total Lipophilic Extracts and FAME

It is evident that SA treatment increased the amounts of the total lipophilic extracts and the lipid composition (expressed as FAME), but there were no statistically significant differences in the amounts of the leaf biomass (compare samples 1 and 5, Table 21.2). The treatment with Cd led to an increase of the total lipophilic extracts. The moderate Cd stress (10 µM) led to increased FAME, but the higher Cd concentration caused decreased FAME amounts (samples 3 and 4, Table 21.2). The decrease of lipids (respectively, FAME) was observed in Cd-stressed wheat (Malik et al. 1992a, b), barley (Vassilev 2004), tomato (Krupa and Baszynski 1989, 1995, Quariti et al. 1997; Ben Ammar et al. 2005) and mustard (Gaur and Grupa 1994; Nouairi et al. 2006) plants. Pretreatment of maize plants with SA resulted in an increase in Cd-induced changes in lipid content by low and mild Cd stress. Only the severe Cd stress led to a decrease in FAME content, similar to those observed in non-pretreated with SA plants (compare samples 4 and 8, Table 21.2). Probably, by high Cd concentrations in the nutrient solution the SA cannot prevent the effect of Cd on lipid membranes of maize plants.

Variants	RWC (%)	Chlorophyll $a + b$ (µg chl/g fw)	Proline (µmol/g fw)		
Control	95.700	1.7±0.11	1.12 ± 0.08		
10-µM Cd	91.500	1.7 ± 0.09	1.60 ± 0.07		
15-µM Cd	91.820	1.6±0.10	2.38 ± 0.45		
25-µM Cd	90.010	0.9 ± 0.08	2.26 ± 0.37		
Control+SA	92.680	1.57 ± 0.10	0.82 ± 0.09		
10-µM Cd	81.385	1.11 ± 0.09	0.94 ± 0.09		
15-µM Cd	82.900	1.10 ± 0.10	0.91 ± 0.06		
25-µM Cd	88.285	1.37 ± 0.08	1.02 ± 0.07		

Table 21.1 Effect of Cd and SA on relative water content (RWC), chlorophyll content and proline accumulation. Means \pm s.e, n=4

Table 21.2 Effect of SA and Cd on the amount of biomass, total lipophilic extracts and FAME in maize leaves

No.	Variants/samples	Dry weight (g) ^a	Total lipophilic extract (mg g ⁻¹ DW) ^a	FAME (mg g ⁻¹ DW) ^b
1	Control	0.7 ± 0.1	150±6	6.3 ± 0.5
2	10-µM Cd	0.6 ± 0.1	183±8	10.2 ± 0.8
3	15-µM Cd	0.6 ± 0.1	175±7	1.8 ± 0.1
4	25-µM Cd	0.5 ± 0.1	170±7	3.2 ± 0.2
5	Control+SA	0.8 ± 0.1	185±8	10.9 ± 0.9
6	10-µM Cd	0.7 ± 0.1	214±11	11.0 ± 0.9
7	15-µM Cd	0.5 ± 0.1	100±4	10.4 ± 0.8
8	25-µM Cd	0.6 ± 0.1	175±7	4.7 ± 0.4

^aResults ± s.e. from three parallel experiments

^bResults obtained from three parallel preparative thin-layer chromatographic and GC procedures

5.5 Effect of SA and Cd²⁺ on the Fatty Acid Composition in Maize Leaves

Treatment with Cd (samples 2–4, Table 21.3) led to a decrease in the content of the short-chain (C14–C15) and an increase in the long-chain fatty acids (C20–C24). Possibly the Cd treatment caused activation of enzymes, responsible for elongation of C18 – acids. The higher Cd stress (sample 4, Table 21.3) increased the content of saturated acids (16:0 and 18:0) and decreased the content of linolenic (18:3) acid. This is a typical reaction of plant lipid membranes to environmental stress, leading to decreased lipid membranes permeability. The same effect was observed in other plants too (Jemal et al. 2000). The FA changes in maize plants treated with low doses of Cd were apparently smaller. A similar effect was observed in tomato plants (Quariti et al. 1997). The more severe Cd stress led to decreased linolenic acid content, as observed in other Cd-treated plants (Krupa and Baszynski 1989; Jemal et al. 2000; Vassilev 2004; Nouairi et al. 2006). The changes in the content of the oleic (C18:1) and linoleic (C18:2) acids showed a reserve trend. The same was observed in mustard seeds treated with Cd (Gaur and Grupa 1994). The content of saturated acids increased, but the amount of hexadecanoic acids remained constant. Pretreatment of maize plants with SA seemed to have an insignificant effect on their fatty acid composition (sample 5, Table 21.3). There was a slight increase of linolenic acid and a decrease of all saturated FA.

The mild Cd stress (10 μ M) applied on the pretreated with SA plants led to the same changes in their FA composition, as in the nontreated (compare samples 2 and 6, Table 21.1). The

		Fatty acids (wt% of total)*												
No.	Variants/samples	14:0	14:1	15:0	15:1	16:0	16:1	18:0	18:1	18:2	18:3	20:0	22:0	24:0
1	Control	3.3	3.2	1.2	1.9	25.1	3.7	3.1	3.8	14.8	36.3	1.0	1.4	1.3
2	10-µM Cd	0.4	2.2	0.9	1.8	23.4	3.5	2.6	3.7	16.0	42.4	0.8	1.0	1.4
3	15-µM Cd	1.2	2.7	1.0	1.5	27.2	4.0	3.2	3.7	15.8	35.6	1.0	1.5	1.7
4	25-µM Cd	1.0	3.1	1.4	2.2	33.2	3.5	4.2	5.0	14.3	26.3	1.4	2.1	2.4
5	Control+SA	1.5	2.4	0.8	1.8	22.9	3.5	2.6	4.5	16.4	40.4	0.7	1.3	1.1
6	10-µM Cd	1.7	2.2	0.8	1.4	21.5	4.1	2.2	3.4	18.5	41.1	0.7	1.1	1.2
7	15-µM Cd	0.6	1.4	0.2	1.2	20.5	3.1	2.3	4.5	18.4	46.3	0.4	0.6	0.4
8	25-µM Cd	1.8	3.5	1.1	1.6	21.9	3.5	2.4	2.9	13.7	43.9	0.8	1.4	1.6

Table 21.3 Effect of SA and Cd on the fatty acid composition in maize leaves

* Values obtained from three parallel measurements; the s.e. (related to peak proportions on the chromatogram) are as follows: ± 0.3 for 16:0 and 18:3; ± 0.2 for 18:2 and ± 0.1 for the others



Fig. 21.4 Effect of Cd and SA on SOD and APX activity in leaves and roots of 14-day-old maize plants. For variants and treatments, see Fig. 21.2. Data are means of four independent experiments \pm s.e.

amount of the saturated acids decreased, whereas the content of linoleic and linolenic acid increased. The severe Cd stress (25μ M) applied on the pretreated plants with SA led to FA profiles, similar to those of the control plants (compare samples 1 and 8, Table 21.3). In all cases, the changes in fatty acid composition after severe Cd stress in the pretreated plants with SA were opposite, as in the nontreated plants (compare samples 4 and 8, Table 21.3). Antagonistic influences of Cd and SA have been observed also in other plants (Drazic and Mihailovic 2005; Drazic et al. 2006). On the basis of the results presented in Table 21.3, it could be concluded that SA plays a protective role on the lipid membranes of Cd-treated maize plants.



Fig. 21.5 Activity of CAT and POD in the leaves and roots of 14-day-old maize plants. For variants and treatments, see Fig. 21.2. Data are means of four independent experiments ± s.e.

5.6 Effect of SA and Cd on Antioxidant Enzymes Activity

The activity of the antioxidative enzymes was measured in leaves and roots of treated plants. The presence of Cd in the nutrient solution (10-, 15- and 25- μ M Cd) led to disturbances in the activity of the antioxidative enzymes (Fig. 21.4). The superoxide dismutase (SOD) activity of leaves and roots increased at all Cd concentrations. Pretreatment with SA before Cd application kept the same tendency in leaves while SA reduced the activity of SOD of roots when compare with Cd treatment alone. In contrast to SOD, ascorbate peroxidase (APX) activity was suppressed by all Cd concentrations and SA treatment alone with smaller effect on roots.

Catalase (CAT) activity was not affected by Cd treatment both in leaves and roots. guaiacol peroxidase (POD) activity of leaves and roots increased in all Cd concentrations but dropped after SA pretreatment mainly in leaves (Fig. 21.5).

The same tendency was observed and for GR activity – nearly 50% increased in Cd-treated variants and without changes after SA pretreatment. Regarding to the last studied enzyme gluta-thione-*S*-transferase (G-S-Tr) we observed that Cd caused an opposite effect in leaves and roots. Pretreatment with SA alleviated the stimulating effect of Cd in leaves and had no significant effect on roots G-S-Tr (Fig. 21.6).

6 Discussion

Summarizing our results we showed that Cd produced a concentration-dependant reduction in the growth of maize plants (Photo 2). Maize plants grown for 14 days with 10-, 15- and 25- μ M CdCl₂ produced a significant inhibition of growth. In our model system maize plants did not tolerate Cd concentrations higher than 25 μ M, which suggests that maize can be considered as a relatively cadmium-sensitive species. We explain this result


Fig. 21.6 Activity of GR and G-S-Tr in the leaves and roots of 14-day-old maize plants. For variants and treatments, see Fig. 21.2. Data are means of four independent experiments \pm s.e.

with the fact that plants were exposed to Cd at the very early stage of their development. The data showed that chlorophyll content was reduced in Cd-treated plants (Table 21.1). The growth inhibition of maize plants was accompanied by a decrease in net photosynthesis measured as CO_2 assimilation (Fig. 21.2). The activity of RuBPCase (Fig. 21.2) and chlorophyll content (Table 21.1) also decreased with rising Cd concentrations.

These severe alterations on the chlorophyll level, chloroplast photochemistry and carboxylating enzyme activities are ultimately responsible for the destruction of photosynthesis caused by Cd. In addition to the negative effects of Cd on the photosynthetic carboxylation reactions PSII electron transport and especially oxygen-evolving complex were found to be very sensitive to the effect of Cd (Clijsters and van Assche 1985). Different components of electron transport chain were proposed as primary targets of Cd and different mechanisms of actions were discussed. Donor side (Clijsters and van Assche 1985; Krupa et al. 1993) or acceptor side (Atal et al. 1993) was

implicated as a main site of the inhibition. In our attempt to contribute to the understanding the effects of cadmium on PSII reactions, an analysis of the TL glow curve parameters was performed in Cd-stressed pea. Our data showed that higher Cd concentrations affect the B-band oscillation pattern and decrease the TL intensity. These results showed a reduction in the number of PSII reaction centers and an increase of misses in charge separation thus suggesting that the centers cannot reach higher oxidation state S_3 and S_4 . Although the B-TL band is specific for PSII charge recombination (Rutherford et al. 1982), the AG band is thought to originate from a feedback of reducing equivalents from the stroma toward PSII centers initially in TL inactive S2(S3) Qb state (Ducruet 2003). AG to reflect the [NADPH+ATP] assimilatory potential in the chloroplasts when induced by flashes has been proposed (Miranda and Ducruet 1995). The decrease of the assimilatory potential can be as a result of the reduced electron transport. In the case of high concentration of heavy metal, this can be a result of harmful effect of reactive oxygen species on the thylakoid membrane composition and function. Our data showed that Cd toxicity, especially in sensitive pea plants, was linked to free radical processes in membrane components leading to alterations in membrane stability and increasing their permeability (Popova et al. 2009).

Presoaking of maize seeds for 6 h with 500- μ M SA before exposure to Cd had a beneficial effect on growth, photosynthesis, carboxylation reactions, thermoluminescence characteristics and chlorophyll content, and it led to a decrease in oxidative injuries caused by Cd.

Recently we have published that leaves of maize plants contained both the SA-free and bound forms. Cd treatment caused accumulation of free and conjugated SAs, to a greater extent in the bound form (Krantev et al. 2008). A similar effect of cadmium on SA accumulation has been reported by Pal et al. (2005) for maize plants.

In addition, we have reported that Cd content of dry seeds and root tissue was low in the absence of Cd in the growth medium and strongly increased after treatment with Cd (Krantev et al. 2008). SA pretreatment led to an insignificant decrease in the root level of Cd in 15- and 25-µM Cd-treated plants. Our results excluded the possibility that formation of stable SA-Cd complexes has lowered Cd toxicity after SA pretreatment. Cd-SA complex formation in the hydroponic solution could not be the cause for the beneficial effect of SA because the exposure to Cd started 3 days after 6-h SA soaking of seeds. Another cause to exclude the formation of such a complex is that pretreatment with SA decreased the level of root Cd accumulation to a very lesser extent [from 27.23 ng/g fw for the control to 35.79 ng/g fw for SA-treated-alone maize plants (Krantev et al. 2008)].

Accumulation of large amounts of osmolytes (proline) is an adaptive response in plants exposed to stressful environment. Proline is known to accumulate in plants under drought, salt, hypoxia, UV radiation, etc. Proline accumulation appeared to be a suitable indicator of a heavy metal stress. The observed decrease in the level of proline in SA-pretreated seeds indicated partial relief from Cd stress (Table 21.1). Previously we found that SA pretreatment decreased MDA accumulation and altogether with the reported decline in the level of electrolyte leakage in pretreated with SA maize plants (Krantev et al. 2008) led to the suggestion that SA is involved against oxidative damage. Our data are in agreement with those reported by Metwally et al. (2003).

When plants are exposed to various environmental stresses they produce large quantities of ROS sufficient to disturb cellular and metabolic functions of the plants. These oxygen species (O^-, OH^-, H_2O_2) can convert fatty acids to toxic lipid peroxides, destroying biological membranes. Although Cd does not generate directly ROS like other heavy metals as Cu and Fe, it also generates oxidative stress via the interference with the antioxidant defense system (Sanita di Toppi and Gabrielli 1999; Somashekaraiah et al. 1992).

An important ROS-scavenging antioxidate enzyme is SOD. By catalyzing the detoxification of O_2^- (superoxide radical) to O_2 , SOD blocks O₂⁻-driven cell damage. Our results showed that Cd treatment increased SOD activity both in roots and leaves indicating activation of the antioxidative system. The activity of SOD was found to be much higher in roots than in leaves and treatment with 10- and 15-µM Cd caused nearly threefold increase in the enzyme activity (Fig. 21.4). Among the H_2O_2 destroying enzymes it was the POD activity that was stimulated by Cd (Fig. 21.5). Pretreatment with SA lowered the activity of POD. This result fits well with the increased rate of lipid peroxidation reported recently by Krantev et al. (2008) and corresponds to other observations (Shaw 1995). The findings indicate that the activities of these enzymes (SOD and POD) are directly or indirectly regulated by SA, thereby providing protection against Cd stress. It is also known that POD participates in the lignin biosynthesis and by this way might build up a physical barrier against poisoning heavy metals. Earlier data in the literature concerning the CAT and APX response in leaves exposed to Cd stress are completely contradictory since both enzyme activation (Lee et al. 1976) and inhibition (Gallego et al. 1996) have been described. In our experiments Cd did not induce changes in CAT activity (Fig. 21.5), while the activity of APX was decreased (Fig. 21.4). GR is known to catalyze some vital steps of the ascorbate–glutathione cycle. The enzyme maintains high ratio of GSH/GSSG, which is essential for the recovery of ascorbate so as to activate a number of enzymes involved in CO_2 fixation. Here we found that GR activity increased upon Cd treatment but suppressed upon SA treatment. Similar tendency was observed and for G-S-Tr activity and changes were well expressed both in roots' and leaves' tissues (Fig. 21.6).

Summarizing our results, we can conclude that among the H_2O_2 -eliminating enzymes POD responds to Cd stress. The absence of changes in CAT activity suggests a different role of CAT in the heavy metal-induced oxidative stress. At the first glance, it may appear surprising that Cd, which is not a transient metal, may cause oxidative stress. However, Cd binds to thiol groups and thereby inactivates thiol-containing enzymes. That could be the reason for the observed inhibition of APX activity as it is known that the enzyme is sensitive to thiol reagents.

Our data suggest that endogenous SA plays an important anti-oxidant role in protecting maize plants from oxidative stress. SA is a direct scavenger of hydroxyl radical and iron-chelating compounds as well as their generation via the Fenton reaction (Dinis et al. 1994; Halliwell et al. 1995). Data have been presented for a salicylate-iron complex with SOD activity catalyzing the dismutation of superoxide radicals (Jay et al. 1999). Therefore, high levels of SA in maize plants may act directly as a preformed antioxidant to scavenge ROS and/or indirectly modulate redox balance through activation of antioxidant responses as was suggested by Yang et al. (2004) for rice plants. Rao and Davis (1999) proposed two different mechanisms to explain the role of SA in ozone-induced cell death in Arabidopsis. In some cases, SA potentiates the activation of antioxidant defense responses to minimize the oxidative stress induced by ozone, while in other cases, high levels of SA led to the activation of oxidative bursts and cell death. These examples demonstrate that SA is an important component in modulating stress responses and may play pro- or antioxidative roles based on its endogenous levels (Yang et al. 2004). It has been observed that not only SA but also other related compounds, such as oHCA, could provoke protection against abiotic stress (Janda et al. 2000). On the basis of ability of oHCA to quench singlet molecular oxygen, it was suggested that it may play a role in the antioxidative response (Foley et al. 1999).

Several hypothetical explanations may account for the positive effect of SA on Cd-induced stress in maize plants. SA prevented cumulative damage development in response to Cd. The suggestion is supported by the data of the lowered root level of Cd in SA-pretreated maize plants (Krantev et al. 2008). Similar data have been reported by Szalai et al. (2005) in maize and by Popova et al. (2009) in pea plants. Cd is usually accumulated in the roots, because this is the first organ exposed to heavy metals in the soil, but it is also translocated into the shoots. Obviously the lowered root level of Cd in SA-pretreated maize plants reduced the harmful effect of Cd and exerted a beneficial effect on the growth and photosynthesis. SA alleviated the oxidative damages caused by Cd. The values of MDA, electrolyte leakage (Krantev et al. 2008) and proline content of SA-pretreated plants were lower compared with the Cd-exposed plants (Table 21.1). Pretreatment with SA exerted a protective effect on the membrane stability judging by the increased total lipids' level and by changes in their FA composition.

Taken together these evidences support the conclusion that SA may indirectly attenuates Cd toxicity through a development of general antistress response of the plants which probably includes the regulation of antioxidant system and lipid metabolism leading to maintenance of membrane integrity.

7 Conclusion and Future Perspective

Phytohormones play a vital role in plant tolerance against environmental stresses. Plant resistance can be induced by adopting various strategies. One of these exogenous uses of various growth regulators and other chemicals has been proved worthwhile in producing resistance to many stresses in a number of plants. Another important function of plant growth regulators is their agricultural use for improving various physiological parameters and crop yield. Among other regulators SA has been proved to be effective in increasing plant production and quality. Seed priming is a short-term and a very pragmatic approach for agricultural use. Another strategy is foliar application of plant growth regulators, both natural and synthetic. SA response against heavy metal stress is a new study in the field of crop physiology. Excessive use of chemical fertilizers in agriculture industries has appeared as a threat to soil health and yield. Results indicated that seed imbibitions with SA affected physiological processes related to growth and development and photosynthesis in maize plants. An important question of this study was how this short-term treatment with SA (presoaking of seeds for 6 h) affected certain physiological processes, such as plant growth, photosynthesis and antioxidant defense system. The beneficial effect of SA during the earlier growth period may help plants to avoid cumulative damage upon exposure to Cd. Alternatively, SA could be involved in the expression of specific proteins or defense-related enzymes. These results may provide a good background for strategies aimed at manipulating plants for decreased Cd content in order to develop crops capable of tolerating environmental changes with damage as little as possible.

References

- Ananieva EA, Christov KN, Popova LP (2004) Exogenous treatment with salicylic acid leads to increased antioxidant capacity in leaves of barley plants exposed to paraquat. J Plant Physiol 161:319–328
- Atal N, Sardini PP, Mohanty P (1993) Inhibition of the chloroplast photochemical reactions by treatment of wheat seedlings with low concentrations of cadmium. Analisys of electron transport activities and changes in fluorescence yield. Plant Cell Physiol 32:943–951
- Barcelo J, Poschenrieder C (1990) Plant water relations as affected by heavy metal stress: review. J Plant Nutr 13:1–37
- Belimov AA, Safronova VI, Tsyganov VE, Borisov AI, Kozhemyakov AP, Stepanok VV, Martenson AM, Gianinazzi-Pearson V, Tikhonovich IA (2003) Genetic variability in tolerance to cadmium and accumulation of heavy metals in pea (*Pisum sativum* L.). Euphytica 131:25–35

- Ben Ammar W, Nouairi I, Tray B, Zarrouk M, Jemal F, Ghorbe MH (2005) Cadmium effects on mineral nutrition and lipid content in tomato leaves (in French). J Soc Biol 199:157–163
- Borsani O, Valpuestan V, Botella MA (2001) Evidence for a role of Salicylic acid in the oxidative damage generated by NaCl and osmotic stress in *Arabidopsis* seedlings. Plant Physiol 126:1024–1030
- Boussama N, Quariti O, Ghorbal MH (1999) Changes in growth and nitrogen assimilation in barley seedlings under cadmium stress. J Plant Nutr 22:731–752
- Chen Z, Iyer S, Caplan A, Klessig DF, Fan B (1997) Differential accumulation of salicylic acid and salicylic acid-sensitive catalase in different rice tissues. Plant Physiol 114:193–201
- Clijsters H, van Assche H (1985) Inhibition of photosynthesis by heavy metals. Photosynth Res 7:31–40
- Cobbett P, Goldsbrough P (2002) Phytochelatins and metallothioneins: role of heavy metal detoxification and homeostasis. Annu Rev Plant Physiol Plant Mol Biol 53:159–182
- Dat JF, Foyer CH, Scott IM (1998a) Changes in salicylic acid and antioxidants during induced thermotolerance in mustard seedlings. Plant Physiol 118:1455–1461
- Dat JF, Lopez-Delgado H, Foyer CH, Scott IM (1998b) Parallel changes in H_2O_2 and catalase during thermotolerance induced by salicylic acid or heat acclimation in mustard seedlings. Plant Physiol 116: 1351–1357
- Dinis TC, Maderia VM, Almeida LM (1994) Action of phenolic derivates (acetaminophen, salicylate, and 5-aminosalicylate) as inhibitors of membrane lipid peroxidation and as peroxyl radical scavengers. Arch Biochem Biophys 315:161–169
- Drazic G, Mihailovic N (2005) Modification of cadmium toxicity in in soybean seedlings by salicylic acid. Plant Sci 168:511–517
- Drazic G, Mihailovic N, Lojic M (2006) Cadmium accumulation in *Medicago sativa* seedlings treated with salicylic acid. Biol Plant 50:239–244
- Ducruet J-M (2003) Chlorophyll thermoluminescence of leaf disc: simple instruments and progress in signal interpretation open the way to new ecophysiological indicators. J Exp Bot 54:2419–2430
- Durner J, Shah J, Klessig DF (1997) Salicylic acid and disease resistance in plants. Trends Plant Sci 7:266–274
- Eberhard S, Doubrava N, Marta V, Mohnen D, Southwick A, Darviell A, Albersheim P (1989) Pectic cell wall fragments regulate tobacco thin-cell layer explant morphogenesis. Plant Cell 1:747–755
- Fodor A, Szabo-Nagy A, Erdei L (1995) The effects of cadmium on the fluidity and H⁺-ATPase activity of plasma membrane from sunflower and wheat roots. J Plant Physiol 14:787–792
- Foley S, Navaratnam S, McGarvey DJ, Land EJ, Truscott G, Rice-Evans CA (1999) Singlet oxygen quenching and redox properties of hydroxycinnamic acids. Free Radic Biol Med 26:1202–1208
- Gadallah MAA (1995) Effects of cadmium and kinetin on chlorophyll content, saccahrides and dry matter accumulation in sunflower plants. Biol Plant 37:233–240

- Gallego SM, Benavides MP, Tomaro M (1996) Effect of heavy metal ion excess in sunflower leaves: evidence for involvement of oxidative stress. Plant Sci 121:151–159
- Gaur A, Grupa SK (1994) Lipid components of mustard seeds (*Brassica juncea* L.) as influenced by cadmium levels. Plant Foods Hum Nutr 46:93–102
- Geiken B, Masojidek M, Rizuto M, Pompili ML, Giardi MT (1998) Incorporation of [35S] methionine in higher plants reveals that stimulation of D1 reaction center II protein turnover accompanies tolerance to heavy metal stress. Plant Cell Environ 21:1265–1273
- Hagedus A, Erdei S, Horvath G (2001) Comparative studies of H₂O₂ detoxifying enzymes in green and greening barley seedlings under cadmium stress. Plant Sci 160:1085–1093
- Hall J, Williams E (2003) Transition metal transporters in plants. J Exp Bot 54:2601–2613
- Halliwell B, Aeschbach R, Loliger J, Auroma OI (1995) The characterization of antioxidants. Food Chem Toxicol 33:601–617
- Hayat Q, Hayat Sh, Irfan M, Ahmad A (2010) Effect of exogenous salicylic acid under changing environments. Environ Exp Bot 68:14–25
- He J, Ren Y, Pan X, Yan Y, Zhu C, Jiang D (2010) Salicylic acid alleviates the toxicity effect of cadmium on germination, seedling growth, and amylase activity of rice. J Plant Nutr Soil Sci 173(2):300–305
- Hernandez H, Cooke D (1997) Modification of root plasma membrane lipid composition of cadmiumtreated *Pisum sativum*. J Exp Bot 48:1375–1381
- Janda T, Szalai G, Tari I, Paldi E (1999) Hydroponic treatment with salicylic acid decreases the effects of chilling injury in maize (Zea mays L.) plants. Planta 208:175–180
- Janda T, Szalai G, Antunovics Z, Horvath E, Paldi E (2000) Effect of benzoic acid and aspirin on chilling tolerance and photosynthesis in young maize plants. Maydica 45:29–33
- Jay D, Jay EG, Medina MA (1999) Superoxide dismutase activity of the salicylate-iron complex. Arch Med Res 30:93–96
- Jemal F, Zarrouk M, Ghorbal MH (2000) Effect of cadmium on lipid composition of pepper. Biochem Soc Trans 28:907–910
- Kahle H (1993) Response of roots of trees to heavy metals. Environ Exp Bot 33:99–119
- Kovacik J, Gruz J, Hedbavny J, Kllejdus B, Strand M (2009) Cadmium and Ni uptake are differentially modulated by salicylic acid in *Matricaria chamomilla* plants. J Agric Food Chem 57(20):9848–9855
- Krantev A, Yordanova R, Janda T, Szalai G, Popova L (2008) Treatment with salicylic acid decreases the effect of cadmium on photosynthesis in maize plants. J Plant Physiol 165:920–931
- Krupa Z, Baszynski T (1989) Acyl lipid compositionof thylakoid membranes of cadmium – treated tomato plants. Acta Physiol Plant 11:111–116
- Krupa Z, Baszynski T (1995) Some aspects of heavy metals toxicity towards photosynthetic apparatus – direct and indirect effects on light and dark reactions: a review. Acta Physiol Plant 17:177–190

- Krupa Z, Oquist G, Nunner N (1993) The effect of cadmium on photosynthesis of *Phaseolus vulgaris* – a fluorescence analysis. Physiol Plant 88:626–630
- Larkindale J, Knight M (2002) Protection against heat stress induced oxidative damage in Arabidopsis involves calcium, abscisic acid, ethylene, and salicylic acid. Plant Physiol 128:682–695
- Lee KC, Cunningham BA, Paulsen GM, Liang GK, Moore RB (1976) Effects of cadmium on responses of several enzymes in soybean seedlings. Physiol Plant 36:4–6
- Leon J, Lawton M, Raskin I (1995) Hydrogen peroxide stimulates salicylic acid biosynthesis in tobacco. Plant Physiol 108:1673–1678
- Levine A, Tenharen R, Dixon R, Lamb C (1994) H₂O₂ from the oxidative burst orchestrates the plant hypersensitive desease resistance response. Cell 79:583–593
- Malamy J, Carr JP, Klessig DF, Raskin I (1990) Salicylic acid: a likely endogenous signal in the resistance response of tobacco to viral infection. Science 250:1002–1004
- Malik D, Sheoran IS, Singh R (1992a) Lipid composition of thylakoid membranes of cadmium treated wheat seedlings. Indian J Biochem Biophys 29:350–354
- Malik D, Sheoran S, Singh P (1992b) Carbon metabolism in leaves of cadmium treated wheat seedlings. Plant Physiol Biochem 30:223–229
- Maslenkova L, Toncheva S (1998) Salicylic acid induced changes in photosystem II reactions in barley plants. Compt Rend Acad Bulg Sci 51(11–12):101–104
- Metwally A, Finkermeier I, Georgi M, Dietz KJ (2003) Salicylic acid alleviates the cadmium toxicity in barley seedlings. Plant Physiol 132:272–281
- Metwally A, Safronova VI, Belimov AA, Dietz KJ (2005) Genotypic variation of the response to cadmium toxicity in *Pisum sativum* L. J Exp Bot 56:167–178
- Miranda T, Ducruet J-M (1995) Characterization of the chlorophyll thermoluminescence afterglow in darkadapted or far-red illuminated plant leaves. Plant Physiol Biochem 33:689–699
- Mishra A, Chudhuri MA (1999) Effect of salicylic acid on heavy metal-induced membrane deterioration in rice. Biol Plant 42:409–415
- Mohantly N, Mohanty P (1988) Cation effects on primary processes of photosynthesis. In: Singh R, Sawheny SK (eds) Advances in frontier areas of plant biochemistry. Prentice Hall India, Delhi, pp 1–18
- Morris K, Mackerness TS, Page A-H et al. (2000) Salicylic acid has a role in regulating gene expression during leaf senescence. Plant J 23:677–685
- Nouairi I, Ben Ammar W, Ben Youssef N, Ben Miled Daoud D, Habib Ghorbal M, Zarrouk M (2006) Comparative study of cadmium effects on membrane lipid composition of *Brassica juncea* and *Brassica napus* leaves. Plant Sci 170:511–519
- Pal M, Szalai G, Horvath E, Janda T, Paldi E (2002) Effect of salicylic acid during heavy metal stress. Proceedings of 7th Hungarian Congress. Plant Physiol 46:119–120
- Pal M, Horvath E, Janda T, Paldi E, Szalai G (2005) Cadmium stimulates the accumulation of salicylic acid and its putative precursors in maize (*Zea mays*) plants. Physiol Plant 125:356–364

- Pancheva TV, Popova LP (1998) Effect of salicylic acid on the synthesis of ribulose-1-5-bisphosphate carboxylase/ oxygenase in barley leaves. J Plant Physiol 152: 381–386
- Pancheva TV, Popova LP, Uzunova AN (1996) Effects of salicylic acid on growth and photosynthesis in barley plants. J Plant Physiol 149:57–63
- Popova LP, Lu Z (2010) Physiological and pathological effects of heavy metals at an early stage of cereal crop development and after complete maturity of grains. Protective role of nitric oxide. Bilateral project between Bulgaria and China (NTS, 02/81)
- Popova L, Pancheva T, Uzunova A (1997) Salicylic acid: properties, biosynthesis and physiological role. Bulg J Plant Physiol 23(1–2):85–93
- Popova LP, Maslenkova LT, Yordanova RY, Ivanova AP, Krantev AP, Szalai G, Janda T (2009) Exogenous treatment with salicylic acid attenuates cadmium toxicity in pea seedlings. Plant Physiol Biochem 47:224–231
- Quariti O, Baussama N, Zarrouk M, Cherif A, Ghorbal MH (1997) Cadmium- and cooper-induced changes in tomato membrane lipids. Phytochemistry 45:1343–1350
- Rao MV, Davis KR (1999) Ozone-induced cell death occurs via two distinct mechanisms in *Arabidopsis*: the role of salicylic acid. Plant J 17:603–614
- Raskin I (1992a) Role of salicylic acid in plants. Annu Rev Plant Physiol Plant Mol Biol 43:439–463
- Raskin I, Skubatz H, Tang W, Meeuse BJD (1990) Salicylic acid levels in thermogenic and non-thermogenic plants. Ann Bot 66:369–373
- Rhoads D, McIntosh ML (1991) Isolation and characterization of a cDNA clone encoding an alternative oxidase protein of *Sauromatum guttatum* (Schott). Proc Natl Acad Sci USA 88:2122–2126
- Rodriguez-Serrano M, Romero-Puertas MC, Zabalza A, Corpas FJ, Gomes M, Del Rio A, Sandalio M (2006) Cadmium effect on oxidative metabolism of pea (*Pisum sativum*) roots. Imaging of reactive oxygen species and nitric oxide accumulation *in vivo*. Plant Cell Environ 29:1532–1544
- Rutherford AW, Crofts AR, Inoue Y (1982) Thermoluminescence as a probe of photosystem II photochemistry. The origin of the flash-induced glow peaks. Biochim Biophys Acta 682:457–465
- Sandalio LM, Dalurzo HC, Gomes M, Romero-Puertas MC, del Rio LA (2001) Cadmium-induced changes in the growth and oxidative metabolism of pea plants. J Exp Bot 52:2115–2126
- Sanita di Toppi L, Gabrielli R (1999) Response to cadmium in higher plants. Environ Exp Bot 41:105–130
- Senaratna T, Touchell D, Bunns E, Dixon K (2000) Acetyl salicylic acid (aspirin) and salicylic acid induce multiple stress tolerance in bean and tomato plants. Plant Growth Regul 30:157–161
- Sharma SS, Dietz KJ (2009) The relationship between metal toxicity and cellular redox imbalance. Trends Plant Sci 14(1):43–50
- Sharma YK, Leon I, Raskin I, Davis KR (1996) Ozoneinduced responses in Arabidopsis thaliana – the role of salicylic acid in the accumulation of defence-related

transcripts and induced resistance. Proc Natl Acad Sci USA 93:5099–5104

- Shaw BP (1995) Effects of mercury and cadmium on the activities of antiopxidative enzymes in the seedlings of *Phaseolus aureus*. Biol Plant 37:587–596
- Shi Q, Zhu Z (2008) Effects of endogenous salicylic acid on manganese toxicity, elements contents and antioxidative system in cucumber. Environ Exp Bot 63: 317–326
- Siedlecka A, Samuelsson G, Gardenstrom P, Kleczkowski LA, Krupa Z (1998) The "acvivatory model" of plant response to moderate cadmium stress-relationship between carbonic anhydrase and Rubisco. In: Garab G (ed) Photosynthesis: mechanisms and effects, vol IV. Kluwer, Dordrecht, pp 2677–2680
- Sigfridsson KG, Bernad G, Mamedov F, Styring S (2004) Molecular interference of Cd⁽²⁺⁾ with photosystem II. Biochim Biophys Acta 1659:19–31
- Sillverman P, Seskar M, Kanter D, Schweizer P, Metraux JP, Raskin I (1995) Salicylic acid in rice. Biosynthesis, conjugation, and possible role. Plant Physiol 108:633–639
- Singh B, Usha K (2003) Salicylic acid induced physiological and biochemical changes in wheat seedlings under water stress. Plant Growth Regul 39:137–141
- Somashekaraiah BV, Patmaja K, Prasad ARK (1992) Phytotoxycity of cadmium ions on germinating seedlings of Mung bean (*Phaseolus vulgaris*): involvement of lipid peroxides in chlorophyll degradation. Physiol Plant 85:85–89
- Srivastava MK, Dwivedi UN (2000) Delayed ripening banana fruit by salicylic acid. Plant Sci 158:87–96
- Stiborova M (1988) Cd²⁺ ions effect on the quaternary structure of ribulose-1,5-bisphosphate carboxylase from barley leaves. Biochem Physiol Pflanz 183: 371–378
- Stoyanova DP, Merakchiiska-Nikolova MG (1992) Influence of cadmium on the formation of the internal structure of chloroplasts during illumination of etiolated bean plants (*Phaseolus vulgaris* L.). Comp Rend Acad Bulg Sci 45(2):71–74
- Stoyanova DP, Tchakalova ES (1997) Cadmium-induced ultrastructural changes in chloroplasts in the leaves and stems parenchyma in *Myriophyllum spicatim* L. Photosynthetica 34(2):241–248
- Szalai G, Pal M, Horvath E, Janda T, Paldi E (2005) Investigations on the adaptability of maize lines and hybrids to low temperature and cadmium. Acta Agronomica Hungarica 53:183–196
- Tasgin E, Attici O, Nalbantoglu B (2003) Effect of salicylic acid and cold on freezing tolerance in winter wheat leaves. Plant Growth Regul 41:231–236
- Uzunova AN, Popova LP (2000) Effect of salicylic acid on leaf anatomy and chloroplast ultrastructure of barley plants. Photosynthetica 38:243–250
- Van der Straeten D, Chaerle L, Sharkov G, Lambers H, van Montagu M (1995) Salicylic acid enhances the activity of the alternative pathway of respiration in tobacco leaves and induces thermogenicity. Planta 196:412–419

- Vassilev A (2004) Cadmium-induced changes in chloroplast lipids and photosystem activities in barley plants. Biol Plant 48:153–156
- Vazquez S, Goldsbrough P, Carpena RO (2006) Assessing the relative contributions of phytochelatins and the cell wall to cadmium resistance in white lupine. Physiol Plant 128:487–495
- Wen JQ, Liang HG (1994) Comparison of the effects of salicylic acid on alternative pathway in slices of dormant and dormancy-breaking potato tubers (*Solanum tuberosum*). Plant Sci 102:127–131
- Wildermuth MC, Dwedney J, Wu G, Ausubel F (2001) Isohorismate synthase is required to sentisize salicylic acid for plant defence. Nature 414:562–565
- Yalpani N, Schulz M, Davies MP, Balke NE (1992a) Partial purification of a inducible uridine-5-diphosphate glucose: salicylic acid glucosyltransferase from oat roots. Plant Physiol 100:457–463

- Yalpani N, Balke NE, Schulz M (1992b) Induction of UDP-glucose: salicylic acid glucosyltransferase in oat roots. Plant Physiol 100:1104–1119
- Yalpani N, Leon J, Lawthon MA, Raskin I (1993) Pathway of salicylic acid biosynthesis in healthy and vitus-inoculated tobacco. Plant Physiol 103: 315–321
- Yang M, Wang SH, Xu LL (2003) Salicylic acid induced aluminium tolerance by modulation of citrate efflux from roots of *Cassia tora*. Planta 217:168–174
- Yang Y, Qi M, Mei C (2004) Endogenous salicylic acid protects rice plants from oxidative damage caused by aging as well as biotic and abiotic stress. Plant J 40:909–919
- Zhou K, Goo K, Elbaz AA, Yang ZM (2009) Salicylic acid alleviates mercury toxicity by preventing oxidative stress in roots of *Medicago sativa*. Environ Exp Bot 65:27–34

Bioremediation and Mitigation of Organic Contaminants in the Era of Climate Changes

22

Laura Coppola, Edoardo Puglisi, Costantino Vischetti, and Marco Trevisan

Abstract

In the present chapter, we addressed the possible links between climate changes and the fate, degradation, and mitigation of organic contaminants in the environment. Particular interest was devoted to techniques based on plants (phytoremediation, wetlands, and buffer strips), organic biomass residues (biobeds), as well as on bioremediation processes controlled by microorganisms. Climate change scenarios were identified, and the obtained info critically correlated on available info about the effects of climatic parameters (temperature, precipitations, soil humidity, pH, organic matter, and nutrients) on the fate, degradation, and mitigation of contaminants.

We conclude that climate change most probably has a significant effect on the fate and behavior of contaminants, and a more limited effect on bioremediation and mitigation strategies. Since fate and behavior determine the exposure of biological receptor to contaminant toxicity, it will be very important to carry out risk assessment evaluations in the context of climate change. Bioremediation and mitigation will remain as powerful tool to address the ever increasing global pollution. Attention must be devoted to adapt these tools to climatic changes, in order to maintain and, if possible, improve their efficiencies.

Keywords

Climate change • Organic contaminants • Bioremediation • Phytoremediation • Wetlands • Buffer strips • Biobeds

Istituto di Chimica Agraria ed Ambientale, Università Cattolica del Sacro Cuore, Via Emilia Parmense 84, Piacenza, Italy, 29122 e-mail: edoardo.puglisi@unicatt.it

L. Coppola • C. Vischetti Dipartimento di Scienze Ambientali e delle Produzioni Vegetali, Università Politecnica delle Marche, Via Brecce Bianche, Ancona, Italy

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_22, © Springer Science+Business Media, LLC 2012

E. Puglisi (🖂) • M. Trevisan

1 Introduction

A large debate is ongoing about climate change: the main issues including the influence and responsibilities of anthropic activities, the possible predicted scenarios, and the effects on ecological system. Climate change is expected to have a wide range of impacts on the environment, such as direct effects from increased temperatures, changes in precipitation, more intense floods, droughts, hurricanes and storms, a lower air quality and a potential change in the environmental and human exposure to toxic environmental pollutants such as persistent organic pollutants (POPs), metals, and pesticides.

In the present chapter, we will throw light on different scenarios indicated by the scientific literature in relation to their possible effects on the main ecological parameters affecting the persistence and the bioremediation/mitigation of pollutants in the environment.

The chapter begins with a brief introduction on climate change, with description on the main possible scenarios depicted by scientists in terms of possible changes on ecological parameters (temperature, water availability, soil moisture, soil organic matter (SOM), and pesticides utilization); these scenarios are then critically discussed in order to assess their possible effects on the fate and behavior of organic contaminants in the environment and about the main strategies for their bioremediation and mitigation.

2 Climate Change Scenarios

The reconstruction of the historical climate trends of our planet demonstrates how climate is constantly changing, showing peculiar oscillations at different time scales. Climate is far from being constant, even in the short term, as the alternation of glacial and postglacial periods is accompanied by less dramatic shortterm climate oscillations that nonetheless can result in significant alterations of ecosystems and living organism distribution (Dalla Valle et al. 2007). The United Nations Intergovernmental Panel on Climate Change (IPCC 2007) has projected significant warming over Europe by the 2030s, with greater warming in winter in the North, and in summer in Southern and Central Europe. It is expected that the mean annual precipitation will increase in Northern Europe and decrease in Southern Europe. Significant changes to climate variability and extremes are also projected, although many of the studies also refer to the 2050s or 2080s.

In accordance with climate change model predictions, the IPCC has depicted different emission scenarios that are based on predictions about future demographic, economic, and technological developments and related greenhouse gases emissions (Nakicenovic et al. 2000). Six scenarios have been specifically made: A1 assumes a high demographic rise reaching its peak in the middle of the twenty-first century, but also includes the introduction of new efficient technologies. A1 is then divided into three subscenarios based on the relative future reliance on fossil fuels: A1F1 is fossil intensive, A1T is mostly based on nonfossil energy resources, and finally A1B is in between the two. A2 assumes high population growth, slow economic development, and slow technological change, B1 is assumed to be similar to A1, but with a lower technological development, and finally B2 is a scenario that takes into account the introduction of local solutions for economic, social, and environmental sustainability. For each of these scenarios, the ICPP has then made the following previsions of warming estimates at year 2090: 1.8°C for B1, 2.4°C for A1T, 2.4°C for B2, 2.8°C for A1B. 3.4°C for A2. and 4.0°C for A1F1.

The predictions at climate level have then been linked by the IPCC to the ecological level. According to IPCC (2007), it is likely that climate change and associated disturbances such as flooding, drought, and wildfires will affect negatively the resilience of many ecosystems in the world. The situation will be most probably worsened by other global change drivers such as changes in land use, pollution, fragmentation of natural systems, and resources overexploitation. It is also expected that the extinction rates for plant and animal species will increase to approximately 20 to 30% for temperature increases ranging from 1.5 to 2.5°C.

Possible effects on crops productivity are more differentiated. Production is projected to increase slightly at mid-to-high latitudes for local mean temperature increases of up to 1 to 3° C depending on the crop, but it is then to decrease beyond that temperature. At lower latitudes the situation is instead different, since crop productivity is generally projected to decrease for even small local temperature increases (1–2°C).

One of the most important factors affected by climate change will be of course water and hydrogeological cycles. Previsions include changes in precipitation patterns with more intense and extreme events, melting of snow and ice, increases in evaporation and atmospheric water vapor, changes in soil moisture and runoff. A likely increase in typhoons and hurricanes is expected, as well as heavier precipitation and higher peak wind speed (IPCC 2007).

Another very important issue related to climate change is related to ice melting from mountain glaciers and icecaps (Raper and Braithwaite 2006): this will in turn not only raise the global sea level, but also affect the exchange rates of contaminants from air to water, and thus affect their environmental distribution.

3 Effects of Climate Change on Fate and Behavior of Organic Pollutants

Once organic contaminants are intentionally or unintentionally released into the environment, they undergo a series of distribution and fate processes related to their physicochemical properties and the properties and processes of the environment (Fig. 22.1). It is thus clear that the environmental scenarios related to climate change summarized earlier will indeed affect the fate and behavior of organic contaminants.

Chemicals may volatilize into the air, runoff or leach into surface water and groundwater. Volatilization is a major cause of losses (Kurtz and ACS 1990) and the rate of these losses often exceed that by chemical degradation, runoff, or leaching (Taylor and Spencer 1990). Loss of chemicals in the air typically range between 20 and 30% of the applied active ingredient during the application, and around 50–60% after the application and can sometimes reach up to 90% (Gregoire et al. 2009). Once volatilized, pesticides can then be easily deposited with rainfall events (Trevisan et al. 1993). A significant proportion may also remain in the soil as a persistent residue bound to the soil colloids (Calderbank 1989). In this bound state contaminants are usually difficult to extract and characterize, but they also tend to lose their biological activity.

Uptake of chemical from the soil by plants is another important route, and also a likely major source of food chain bioaccumulation and an important way of exposure to humans and animals (Paterson et al. 1990).

Water may disperse contaminants into the environment via foliar wash off, surface runoff, and leaching. Runoff may contribute to pollution of surface water, and leaching to contamination of groundwater.

Runoff may include dissolved, suspended particulate and sediment-adsorbed pollutants. Chemicals that remain at the soil surfaces for longer period of time, because they are strongly adsorbed and resistant to degradation and volatilization, will be more susceptible to runoff, whereas incorporation into the soil will reduce runoff risk (Larson et al. 1995).

Groundwater contamination is mainly due to the leaching through infiltration. The extent to which groundwater contamination occurs will depend among others on chemicals properties, soil characteristics, drainage rate, and water table depth. For many years, chemical mobility has been identified as a key characteristic in assessing groundwater potential pollution. However, mobility alone is not a good indicator of the groundwater pollution potential of a chemical, but rather the combination of mobility and persistence determines whether a compound will be degraded during its residence time in the zone above the groundwater (Jury et al. 1987; Gustafson 1989).



Fig. 22.1 Description of the processes affecting the fate and behavior of organic contaminants in soil, air, water, sediment, and plants/biomass

Table 22.1 Effects of increases in temperature (T) and precipitation (P) on the main processes governing the fate of organic contaminants in air, soil, and water

Effects	Environmental compartment	Expected impact on processes	
↑T	Air	↑ Photodegradation	
	Soil	↑ Volatilization	
		↑ Microbial degradation	
	Water	↑ Volatilization	
	water	↑ Hydrolysis	
		↑ Solubility	
		↑ Microbial degradation	
↑P	Air	↑ Deposition	
	Soil	↑ Soil moisture	
		↑ Runoff	
		↑Leaching	
		↓ Persistence	
	Water	↑ Dilution	

↑ increase, ↓ decrease

Even if concerns for surface water are often separated from groundwater, the hydrolytic cycle provides direct connection between these compartments (Leonard 1990). Therefore, levels of pollution in surfaces water may affect groundwater or in turn be affected by groundwater.

Climate change will affect organic contaminant fate and distribution in several ways: it is possible to distinguish between direct and indirect effects. Direct effects are those affecting the processes shown in Fig. 22.1. As discussed in detail later, it is expected that the increase in temperature and precipitation due to climate change will affect almost all processes involved in organic contaminants environmental fate. As outlined in Table 22.1, most of these processes will be increased, with different effects on contaminant distribution. Indirect effects are instead related to shifts in agriculture systems because of climatic changes. Climate change is indeed likely to affect agriculture by shifting the locations and type of crops grown and the range and magnitude of crop: pesticide use will thus shift in response to these altered cropping patterns and crop pest distribution (Noyes et al. 2009).

A study was carried out in order to assess the possible effects of climate change on the range and severity of the plant disease "phoma stem canker," an important disease of Brassica crops caused by Leptosphaeria maculans, a fungal pathogen (Evans et al. 2008). In this study, scenarios for the future severity of epidemics in the UK were generated by joining weather-based models predicting the development of phoma stem canker epidemics with climate change models. Different parameters such as total rainfall, mean maximum temperature, start of leaf spotting and thermal canker, as well as different climate models were used to develop the multiparametric model. Results showed that increase in temperature as well in changes in the intensity of rainfall events will most probably affect not only the severity of the disease but also spread its geographical distribution, at least in the UK. This type of phenomena will surely affect the global costs and the use of pesticides. Chen and McCarl (2001) carried out a statistical study to assess how the costs of pesticide usage are influenced by temperature and precipitation. The investigation was carried out using pesticide usage and weather data, and considering corn, cotton, potato, and soybean cultivations in the USA. For all four crops, the authors found that increase in precipitation significantly increases pesticide usage cost. Regarding temperature, the results showed mixed effects: corn, cotton, and soybeans costs also increased with temperature but wheat costs decrease. In a following study, Reilly et al. (2003) predicted for US agriculture yield reduction for crops such as corn, potato, soybean, and cotton, and subsequent increase in pesticide production and use.

Climate change is also expected to affect the distribution of pollutants. Dalla Valle et al. (2007) conducted a semiquantitative assessment of the possible influence of climate change on the

distribution and fate of some selected chemicals (namely, polychlorinated biphenyls (PCBs) and PCDDs) in a temperate region, the Venice Lagoon area, which is quite polluted by industrial activities. An environmental fate model was then applied, using experimental data about chemical properties and sources of pollution as model inputs. Three scenarios were then assumed. In scenario A, defined as the "control scenario," environmental conditions have been considered constant along the 50 y of the simulation, with constant temperature (15°C, the annual mean temperature in Venice), precipitation (900 mm/y), solar radiance, and degradation rate. Scenario B assumed a modest and gradual evolution of the initial conditions along the simulation run, with temperature increasing from 15 to 16°C, precipitations decreasing by 5% of the degradation rate increasing by 10%. In the last scenario C, more dramatic variations were assumed, with the mean annual temperature reaching 18°C, precipitations decreasing by 10%, and degradation rate increasing by 30%. Four PCB and PCDF congeners were considered: PCB 118, PCB 180, 1,2,3,4,7,8-HCDF, and 2,3,7,8- TCDF. The net result of the tested scenarios pointed towards a faster reduction of the overall contaminants budget, but at the same time an enhancement of their mobility. This may result in a greater transfer to polar regions, where degradation and removal from the environment is more difficult for such chemicals, causing an increase in bioaccumulation and biomagnification.

Lamon et al. (2009) reviewed a number of environmental factors potentially affected by climate change in relation to their possible effects on POPs environmental behavior. Wind and precipitation patterns can alter the way chemical redistribution in the environment, temperature is going to affect the degradation rate as well as the emission rate from soils, changes in duration and intensity in seasonal precipitation may lead to changes in the spatial and temporal distribution of POPs wet deposition and their degradation products, floods may result in POPs environmental dissemination as contaminated sediments may be redistributed on large uncontaminated soil areas, increase in salinity may lead to a decrease in POPs solubility. The situation is of course very

uncertain, because of the variability of the foreseen scenario and the lack of enough data, but the authors confirm the importance of taking into account climate changes in the monitoring, modeling, and risk assessment of pollutants fate.

Changes in climatic conditions can also have consequences on the ecotoxicity of pesticides. The effect of increase in salinity and temperature on the toxicity towards grass shrimp (Palaemonetes pugio) of two common pesticides, chlorothalonil and scourge (a mixture of resmethrin and piperonyl) has been assessed (deLorenzo et al. 2009). Toxicity of chlorothalonil increased with both temperature (10°C increase) and salinity (10 ppt increase), while the toxicity of the fungicide source was also increased by temperature, but reduced by salinity. These results suggest that standard toxicity bioassays may not be predictive of actual pesticide toxicity under variable environmental conditions, and testing under a wider range of exposure conditions is important in order to improve the accuracy of chemical risk assessments.

Assessment of the effects of increasing temperature on the response of labile and resistant soil carbon pools has been shown by Fang et al. (2005). Experiments carried out on incubating organic matter fractions of different lability at different temperature showed that also the stable fraction is affected by increasing temperature. These results, together with model evidences, indicate according to the authors that by 2100, the SOM loss could be up to 46% in arable soil, 37% in grassland, and 32% in forest. Since organic matter is an important factor affecting the bio availability and degradation of contaminants (Puglisi et al. 2007a), these predicted changes in SOM levels will surely have an influence on organic contaminants fate worldwide.

4 Effects of Climate Change on Strategies for Risk Reduction

A number of techniques can be applied to reduce the risk related to the presence of organic contaminants in the environment. Here bioremediation, natural attenuation, phytoremediation, and mitigation by means of wetland, buffer strips, and biobed systems are presented and discussed in the context of climate changes.

4.1 Bioremediation and Natural Attenuation

Bioremediation is not a new concept since scientists have studied these processes since 1940s, but it became popular after the Exxon Valdez oil spill in Alaska in the late 1989. Bioremediation can be defined as a biological technique to reduce hazardous pollutants contamination in the environment to undetectable, nontoxic, or acceptable levels. Bioremediation is based on the biodegradation process, which can be defined as a natural process whereby bacteria or other microorganisms alter and breakdown organic molecules into other substances (Hoff 1993). Microbial metabolism is probably the most important process of pollutant degradation in soils (Kearney and Wauchope 1998). Indeed, microorganisms have the ability to interact both chemically and physically with a huge range of organic compounds, often leading to a structural change or the complete degradation of the molecule (Semple et al. 2001).

For bioremediation to be successful, it is necessary to have the right microbes in the right place with the right environment factors for degradation to occur (Boopathy 2000).

A particular example of bioremediation is represented by natural attenuation, a remediation strategy that is becoming much popular in the last years. It is possible to define natural attenuation processes as a number of chemical, physical, or biological process that, under favorable condition, act without human intervention to reduce the toxicity, mobility, or concentration of contaminants. The most common among these in situ processes are biodegradation, dilution, sorption, volatilization, and chemical or biological stabilization. In order to identify and understand the possible impacts of climate change on bioremediation strategies, it is however useful to discuss which are the most important limiting factors for biodegradation.

Contaminants	Soil temperature (°C)	Degradation rate	References
Hydrocarbon (gasoline and diesel)	From 1 to 40	Degradation rate significantly increase from 1 to 20°C	(Walworth et al. 2001)
n-hexadecane	18 60	60°C enhanced the degradation rate	(Perfumo et al. 2007)
Rimsulfuron	10 25	Lower half life at higher temperature	(Vischetti et al. 1997)
DDT	15 45	45°C enhanced the degradation rate	(Samuel and Pillai 1989)
Acephate	15 25 35	Degradation was significantly faster at 35°C compared with 15°C	(Chai et al. 2010)
Metribuzin	20 +5/-5	T1/2=20 days; no significant degradation	(Stenrød et al. 2008)

Table 22.2 Examples of the effect of temperature on the degradation rate of organic contaminants

The biodegradation of an organic contaminant does not depend only on the physicochemical property of the chemical, but it is the resultant of a number of factors related to both chemical and environmental conditions. Indeed, the fate and behavior of organic pollutants is usually limited by different factors such as the chemical (pH, temperature, moisture, oxygen, nutrients) and biological site conditions (biomass concentration, population diversity, enzyme activity). These parameters affect the acclimation period of the microbes to the substrate, while the molecular structure and contaminant concentration have been shown to strongly affect the feasibility of bioremediation strategy and the type of microbial transformation occurring (Boopathy 2000).

The pH of soil environments can vary widely. In general, biotic and abiotic factors can be influenced by soil pH and both of them can have thus an influence on contaminants dissipation. Low or high pH levels can inhibit growth and metabolism of the microbial community and microbial degradation process. Many researchers have demonstrated a positive correlation between pH and soil microbial biomass and activity (Singh et al. 2003). In general, most heterotrophic bacteria favor neutral pH, while fungi are less adversely affected by low pH value (Baath and Anderson 2003). Moreover, soil pH can also affect the solubility of pesticides and influence the adsorption and bioavailability of pesticides on organic matter and clay colloids. Commonly, acidic soil tends to increase pollutant solubility and reduce sorption on soil particles (Gavrilescu 2005).

Temperature is one of the important factors in bioremediation strategies since it controls the rate of enzymatic reaction of microorganisms (Worthington 1988), the rate of microbial growth, and it can also modify the physical nature of the contaminants. Many researchers studied the relationships between degradation and soil temperature of different organic contaminants. In general, it has been found that temperature is positively correlated with the degradation of organic contaminants. In Table 22.2, some examples of degradation of some organic contaminants in different temperature conditions are reported and discussed. Walworth et al. (2001) carried out a study to assess the effect of temperature on microbial activity and petroleum hydrocarbon degradation in contaminated cryic soils. They observed maximum microbial activity between 21 and 31°C, while average biodegradation rates increased from 1.7 mg kg⁻¹ day⁻¹ at 1°C to 8.2 mg kg⁻¹ day⁻¹ at 11°C, and to 15.1 mg kg⁻¹ day⁻¹ at 21°C. Average biodegradation rates were similar at 21 and 31°C, and decreased slightly when the temperature was increased to 41°C. The greatest increases of biodegradation rate occurred between 1 and 21°C. The decrease in degradation at 41°C can be explained considering that the study was conducted on a sub-Arctic soil, which most probably harbored a microbial community adapted to low temperatures.

Perfumo et al. (2007) using instead an unpolluted soil spiked with hexadecane found that an increase in temperature up to 60°C resulted in significant increases in degradation, with or without the addition of biosurfactants or degrading strains. Vischetti et al. (1997) assessed the persistence of Rimsulfuron, a sulfonylurea, in a silty clay loam soil, and found that both DT50 and DT90 disappearance rates were increased when the temperature of the soil microcosms was raised from 10 to 25°C. Samuel and Pillai (1989) assessed the effect of raising the temperature from 15 to 45°C on the volatilization, soil binding, and degradation of DDT: at the higher temperature they found a significant increase in volatilization and degradation, but they also found a significant increase in the formation of bound residues. They also compared soils at 75% water holding capacity (WHC) with flooded soils, and found that in the latter both volatilization and degradation were higher.

Chai et al. (2010) studied three different soils and found that the degradation of acephate, an organophosphorus insecticide with high water solubility and low Kow, was significantly faster at 35°C compared with 15°C. Acephate degraded completely at 70 days for soils incubated at 25 and 35°C while periods of up to 160 days were required for complete acephate degradation at 15°C.

Stenrød et al. (2008) assessed degradation and leaching of Metribuzin, an asymmetrical triazine prone to leaching because of its weak adsorption in soils. They compared degradation at 20, 5, and -5° C, and they found a significant degradation only in the first case. They also found that freeze-thaw cycle increased the mobility of the pesticide but had no effect on degradation, which at temperatures below 5°C was found to be always negligible.

In order to describe the effects of temperature on degradation, Walker and Eagle (1983) developed a simplified degradation model, where the equations describing the temperature dependence of degradation are set so that a change in temperature of 10°C changes the half-life by a factor of two.

Another important parameter affecting the organic contaminant disappearance is soil moisture, since water availability has also an effect on microbial growth and enzymes production (Marín et al. 1998). Besides affecting the microbial activity, soil moisture can also influence the contaminant binding and distribution. Indeed, water plays a significant role in contaminant availability for microbial utilization (Atagana et al. 2003) and, by decreasing the sorption, it can also increase the rate of microbial degradation (Sims et al. 1991). Low levels of moisture may limit the microbial activity and the amounts of contaminants in solution, on the other hand wetter soil can affect oxygen diffusion because the free pore space may be blocked by water. Schoefs et al. (2004) have observed that hexadecane biodegradation rate decreases with increasing water content, concluding that the effect of a limitation of oxygen through the aqueous phase appears dominant. While Garcia-Valcarcel and Tadeo (1999) reported that simazine and hexazinone degradation rate increase with soil moisture content.

This different behavior is most probably due to differences in the physicochemical properties of the compounds, mainly water solubility and hydrophobicity (as expressed by the coefficient of partition between octanol and water – Kow): for hydrophobic compounds such as hexadecane the increase in water content may reduce or not affect biodegradation because of low water solubility, while for more polar compounds an increase in soil moisture can result in a higher fraction of contaminants in the water phase more susceptible to microbial attack.

Moreover, Chai et al. (2010) observed that there were no significant differences in the degradation rates of acephate among different soils for the same level of moisture content indicating that the soil type did not markedly affect degradation. However, several researchers reported that moisture levels of 60% of WHC are the optimum for the dissipation of most contaminants, because it represents a sufficient amount of water for microbial processes, enough oxygen to support aerobic processes and solubilization of contaminant.

Microbial degradation rates in soil are often limited by ineffective supply of nutrients to the microorganisms (Morgan and Watkinson 1989). A number of studies on soils clearly demonstrated that nutrient supplementation (C, N, and P) can augment microbial growth rate-limiting components (Graham et al. 1999) and can further enhance biodegradation of organic pollutants (Atlas and Bartha 1972; Zhang et al. 2005; Qui et al. 2009). Adequate nutrients must be available to meet the metabolic and growth needs of the microbial populations performing the degradation reaction (Graham et al. 1999). In other case, nutrient stimulation does not improve the biodegradation rate, and this could be due to inadequate nutritional supplements that advance the activity of nondegrading microorganisms (Steffensen and Alexander 1995). Moreover, Alexander (1994) reported that the time necessary to biodegrade organic compounds containing P and N can be long because the organisms use the inorganic P and N from the environment in preference to the forms which would be released as results of degradation of the contaminant.

However, Davis and Madsen (1996) reported that microbial activity usually show greater results at a carbon:nitrogen:phosphorus ratio of 100:5:1, indicating that in soil containing high level of organic material, the toluene degradation appeared to be limited by the availability of N source in this soil.

Microorganisms such as bacteria and fungi play a key role in bioremediation process and generally they can use two different mechanisms in biodegrading pollutants (metabolic and cometabolic). Depending on the concentration of the compound and the environmental condition, the microorganisms can exhibit different acclimation period. This could be due to different factors such as enrichment of the capable microbial population, enzyme induction, and production of toxic metabolites.

4.2 Phytoremediation

Phytoremediation has been first defined in 1993 as the use of green plants to remove, contain, or render harmless an environmental contaminant (Cunningham and Berti 1993). Phytoremediation has emerged as a promising, cost effective, and solar-driven in situ technology for the remediation of contaminated sites (Vamerali et al. 2010). Most phytoremediation studies are aimed at inorganic pollutants through different approaches defined as phytoextraction (the use of metal-accumulating plants to transport and concentrate metals from the soil into roots and above-ground biomass), rhizofiltration (the use of plant roots to absorb, precipitate, and concentrate toxic metals from polluted effluents), and phytostabilization (the use of plants to reduce the mobility of metals) (Salt et al. 1995). These approaches can however be applied also for organic contaminants: phytoextraction has been used to remediate BTEX (benzene, toluene, ethylbenzene, and xylenes) mixtures, pentachlorophenol, and short-chained organic contaminants; phytostabilization has been successfully applied in the case of phenols and chlorinated solvents such as tetrachloromethane and tricholoromethane; and finally rhizofiltration to degrade different organic pollutants (Susarla et al. 2002).

Plants of course are not the only drivers of the remediation of pollution: rhizosphere-associated microorganisms also play a major role, and the term rhizoremediation thus often used. As for other bioremediation practices, the remediation is mainly due to the expression and activity of enzymes able to degrade, detoxify, or block the organic pollutants and these enzymes can indeed have a microbial or a plant origin. Among plants, many species are able to produce enzymes with a role in transforming organic pollutants. Some hybrid poplars (Populus sp.) produce aliphatic dehalogenases that are capable of degrading chlorinated pollutants such as trichloroethylene (TCE), one of the most polluting industrial solvents. Plants grown in TCE polluted soil are indeed able to extract TCE, transpire it, as well as improve the rhizodegradation by feeding degrading microorganisms in the rhizosphere with root exudates (Meagher 2000). Interestingly, plant dehalogenases are thought to act with an oxidative pathway, while in microbes dehalogenation is only a reductive process. Other pollutants for which plants show good degrading abilities are nitroaromatic compounds such as 2,4,6-trinitrotoluene (TNT), hexahydro-1,3,5trinitro-1,3,5-triazine (RDX), and nitroglycerin (GTN): little is still known about the genetic basis of the degradation of these compounds, although a role of glutathione-S-transferases has been recently suggested by a study conducted on poplar plants exposed to TNT (Brentner et al. 2008). Another example is represented by the phytoremediation of PCB contaminated sites. PCB degradation is mainly a microbial process, as encoded by the biphenyl (bph) pathway: since the bioavailability of these pollutants is low (Puglisi et al. 2007b). Plants can improve the degradation not only by feeding microbes, but also by enhancing the bioavailability of PCB through rhizodeposition processes. A recent study also demonstrated for the first time a role of a plant nitrate reductase from Zea mays in the dechlorination of one representative PCB congener, PCB 153 (Magee et al. 2008).

As for other techniques discussed in this chapter, there are no published works about the possible effects of climate changes on the phytoremediation of organic pollutants. We can thus only speculate using evidence about phytoremediation in arid or wet areas and about general effects of climate change on plant physiology. As discussed in details in other chapters of this book, climate change is indeed expected to have a range of effects on the physiology of plant species and on their geographical distribution (Thuiller et al. 2005). The climatic variables which are considered to have the strongest effects on plant growth and physiology are mean annual precipitation, mean annual temperature, minimum temperature of the coldest month, and moisture availability were identified as the climatic variables influencing more plant growth and physiology. Although changes in these climatic variables will most probably alter the geographical distribution of plants, previsions are difficult to make about phytoremediation abilities. Some speculations of interest can, however, be made about effects of increasing atmospheric carbon dioxide concentrations on the soil-plant interface. There is a substantial agreement in the scientific community in associating increases in CO₂ air concentrations with a higher rhizodeposition and in turn with higher microbial densities and activities (Rogers et al. 1994; Smart et al. 1997; Marilley et al. 1999; Lu et al. 2004). Since organic contaminants degradation is a rhizobacterial activity thought to be increased by CO_2 (Shaw and Burns 2005), it can be even possible to predict a positive effect of some climate change scenarios on phytoremediation. Moreover, it should be taken into account how phytoremediation has proven to be a successful approach on different climatic and pedological conditions, ranging from arid zones (Mendez and Maier 2008; Cook et al. 2009; Zhu et al. 2010) to wetlands and buffer strips, as discussed in the following paragraphs. It is thus expected that climate change will not represent a limitation to phytoremediation: particular attention should be however devoted to the choice of species cultivars most adapted to specific pedoclimatic conditions.

4.3 Mitigation: Wetlands

Wetlands are transitional habitats between terrestrial and aquatic systems, where the water table is usually at or near the surface or the land is covered by shallow water (El-Refaie 2010). The ability of natural wetland to mitigate nonpoint source pollution and to limit surface water contamination has been extensively demonstrated (Salmon et al. 1998; Weaver et al. 2004; Moreno et al. 2007). In order to exploit these mitigation abilities, several artificial wetlands were constructed and studied. Indeed, constructed wetlands offer a cost-effective alternative to the traditional treatment systems to clean up water by organic and inorganic contaminants.

Compared to natural wetlands, artificial wetlands offer different advantages related to the possibility of controlling parameters such as substrate types and concentrations, plant species, and hydraulic connectivity. Within wetlands, organic contaminants are removed as a resultant of chemical, physical, and biological processes such as sedimentation, precipitation, adsorption, plant uptake, and microbial degradation (Brix 1994). Indeed, the main processes in which wetland contribute to pollutant degradation are phytoremediation and bioremediation, which have been discussed earlier.

The active reaction zone for the degradation of contaminants in wetland is always the root zone (or rhizosphere), in which the interactions between plant, microorganisms, soil, and pollutants take place (Stottmeister et al. 2003).

Macrophytes play an important role in the water treatments of wetlands because they have a significant and positive effect on biological removal of organic contaminant. They take up inorganic pollutants, decrease the flow velocity favoring the absorption of solutes, transport gases and solutes between above-ground root zones, release oxygen and carbon compounds into the rhizosphere, and influence microbial diversity and activity (Brix 1994; Tanner 2001; Taylor et al. 2010).

On the other hand, microorganisms play the main role in the transformation and mineralization of organic contaminants. In subsurface flow systems, near roots and on the rhizoplane, aerobic processes are predominant in the zones that are largely free of oxygen, anaerobic processes such as denitrification, sulfate reduction, and/or methanogenesis take place (Stottmeister et al. 2003).

Gregoire et al. (2009) have extensively studied the artificial wetland in European countries and reported that the purification capability of wetlands can be optimized by developing both physical (hydraulic design, soil management, and water pathway) and biological processes (plant and bacteria development).

It is important to assess the possible impacts of climate changes on artificial wetlands in order to adapt and optimize future mitigation strategies. Indeed, the indirect effect of climate change can have an important implication on the efficiency of the systems. Since the change in demand of water, the variability of temperature, and the change in the land use may have an important effect on the fate of contaminants in the environment (Bloomfield et al. 2006; Gregoire et al. 2009).

Because the increasing variability in seasonal rainfall pattern leads to an unstable distribution of water resources, it is important to optimize the hydraulic design of wetlands to the specific site.

Hydraulic retention times are important parameters and commonly too low to permit an optimized adsorption and degradation of pollutants. Thus, the control of the hydraulic design and the use of adsorbing materials can be useful to increase the pollutant residence time and the contact between pollutants and biocatalyzers (Gregoire et al. 2009). Therefore, hydraulic dimensioning of artificial wetland should take into account the groundwater and surface water interaction and the evapotranspiration from wetlands.

Once the contaminants are sequestered in the wetlands, biological treatments can be activated. To improve biological treatments in the climate change scenario it is important to consider two aspects, the adsorption by selected macrophyte and the degradation by microorganisms induced. Thus, the introduction and stimulation of degrading microorganisms and plants adapted to the specific environmental conditions which can degrade and adsorb specific contaminant can be a solution to improve the efficiency of the system.

4.4 Mitigation: Buffer Strips

Buffer strips are filter zones inserted near the edge of the field, between the agricultural fields and the receiving water, permanently vegetated with trees, shrubs, or grasses (planted or indigenous). Buffers can reduce the amount of pollutants carried by runoff, erosion, spray drift, and drainage to the nearby surface water.

Buffer strips can have several different configurations such as grassed waterways, filter strips, vegetative barriers, riparian buffers, contour buffer strips, or field borders (Fig. 22.2). Grassed waterways are very simple and useful buffers to reduce soil erosion and capture most organic contaminants that would wash out of



Fig. 22.2 Examples of different types of buffer strips and vegetated areas for mitigation of organic contaminants

fields into waters. Filter strips are important in protecting water quality, and they are composed of grass or permanent vegetation that protects riparian areas from pesticides, organics, nutrients, and sediment runoff. Vegetative barriers are narrower compared to the other types of buffer, and they are performed by perennial grasses or shrubs planted and are very effective in reducing water erosion which results in sediment or contaminant trapping and water infiltration. Riparian buffers, designed by diverse community of trees, shrubs, and perennial grasses, effective to intercept sediments, nutrients, pesticides, and other materials in surface runoff and reduce nutrients and other pollutants in shallow subsurface water flow. Contour buffer strips are simply strips of perennial vegetation that is interchanged with cultivated strips of crop. These types of strips are most effective when runoff water enters uniformly. Field borders are characterized by perennial vegetation on the edge of a field and they are effective in reducing wind and water erosion.

Due to their environmental benefit in treating non-point-source pollution, buffers are considered as natural solutions for pollutant removal. Several studies have demonstrated that buffers have the potential to improve water quality by reducing soluble nutrient concentrations and removing pollutants from field runoff (Patty et al. 1997; Hefting and de Klein 1998; Schmitt et al. 1999; Borin et al. 2005).

Different mechanisms are involved in contaminant removal such as reducing surface runoff, filtering surface and groundwater runoff, reducing erosion, and filtering pollutants from field (Borin et al. 2010).

Sediment deposition process largely controls the effectiveness of the filter area, whereas infiltration is the controlling factor for contaminant transport. Buffer strip alters flow hydraulics, reducing runoff speed and increasing water infiltration and contaminant deposition before to their export to the water bodies (Misra et al. 1996; Vianello et al. 2005). Wilson (1967) observed that increasing the flow volume and velocity markedly reduced filtration efficiencies. Indeed, buffer strips are most effective when the flow is shallow, slow, and enters the buffer strip uniformly along its length (Barling and Moore 1994).

The presence of vegetation on buffer strips is strongly recommended mainly to limit the amount of surface and subsurface flow and erosion. Misra et al. (1996) determined that the major factor in reduction of herbicide transport was infiltration of inflow into the vegetative buffer strips. Abu-Zreig (2001) demonstrated that the conservation of grass cover in buffer strips prevents the formation of preferential flow channels, while Patty et al. (1997) showed that more than 99% of isoproturon and 97% of diflufenican residues in runoff were removed by grass buffer strips.

Plants also confer higher organic matter content in the filter zone, which increases pollutant adsorption capacity and microbial activity for degradation. An interesting work was presented by Benoit et al. (1999), which observed a rapid degradation of isoproturon in soil from grass buffer strips compared to the cultivated soils. Isoproturon half-live was significantly reduced from 72 days in the cultivated field to 8 days in the grass buffers. In the same study, authors observed that isoproturon sorption was enhanced in the grassed soil compared to the cultivated soil and especially in the surface layer (0–2 cm) containing a high proportion of nondecomposed plant residues.

It is generally recommended that the species to be used in buffer strips should be appropriate to the local soil, climate, and specific requirement, for example, plants with high nitrogen requirement or herbaceous plants with a long growing period, high moisture tolerance, and extensive root system (Delgado et al. 1995).

Traditionally buffer strips are usually relatively narrow, between 5 and 30 m, but different climatic conditions and cropping systems may require different types of buffer strips to assure the best performance (Vianello et al. 2005). However, it was noted that the width of the buffer zone controls the trapping of contaminants up to a maximum after which no improvements were observed. Establishing a buffer width is dependent on different factors including vegetation and hydraulic conditions, topography, geology, and local climatic conditions (Vischetti 2008).

The slope can affect the performance of the buffer strip since higher inclination can increase runoff flow velocity, and thus reduces the contaminant trapping efficiency.

Climate variability also affects the decontamination effectiveness of buffer strips and should be carefully studied to adapt the system to the future scenario. Buffer performance and efficiency could be affected by climate change.

In prevision of high rainfall events that can exceed the infiltration capacity of the soil and this can affect surface runoff and contaminants remobilization, hydrology parameter should thus be carefully implemented. For example, concentrated surface flow through buffer zones or flow through subsurface drains may affect buffer performance. Other consideration can be made on buffer strip width that should be proportioned on the basis of flow length to decrease the pesticides load in receiving water.

High temperature promotes infiltration, since evapotranspiration and soil water requirement are both higher during this period. Moreover, Ledwith (1996) observed that buffer width has an effect on air temperature and relative humidity. It should be taken into account indeed that buffer strips of insufficient width may allow an increase in direct and reflected solar radiation, thus increasing the air temperature and lowering the relative humidity on warm days.

On the contrary, it has been observed that during the winter when temperatures are lower the biological activity and the plant uptake declines, thus nutrient circulation in the environment is encouraged (Delgado et al. 1995). The introduction of plants adapted to the specific environmental conditions, adequate buffer configuration with appropriate buffer width and slope can be a solution to improve the efficiency of the system in different temperature conditions.

4.5 Mitigation: Biobeds

A system for reduction of point sources of organic contaminants was presented by Torstensson and Castillo (1997) with the introduction of biobed.

Biobed is a simple and cheap on-farm construction aimed at minimizing point source environmental contamination by means of pollutants adsorption and degradation.

Point source contamination usually derives from inappropriate managements of organic contaminants especially by the utilization of pesticides in agriculture. Indeed, unsatisfactory pesticides handling such as tank filling, spilling, and residues disposal procedure can be a major cause of pesticides contamination in water and groundwater.

The original biobed design proposed by Torstensson and Castillo (1997) consists of a lined hole in the ground (at least 0.6 m deep and a minimum surface area of 1 m² for every 1,000 L of liquid requiring treatment) filled with topsoil (25%), straw (50%), and peat (25%) which have the ability to retain and degrade contaminants from the water. This biomixture was used in order to ensure maximum binding capacity, moisture control, and to create optimal conditions for microbial degradation (Fogg et al. 2004). As extensively reported by Castillo et al. (2008), straw is the main substrate for pesticide degradation and microbial activity, especially from lignin-degrading fungi; the soil provides sorption capacity and should be rich in humus and sufficient clay content to promote microbial activity; the peat contributes to sorption capacity, moisture control, and abiotic degradation.

Biobeds are also made up of an impermeable clay layer (10 cm) in the bottom part, which decreases the water flow downward and increases the contaminants retention time in the system. Moreover, a grass layer covering the biobed regulates the moisture, enhances evapotranspiration, and produces root exudates to support cometabolic processes. The survival of this grass cover can also be used as an indicator of herbicide spills (Castillo et al. 2008).

Torstensson (2000) demonstrated that biobed can effectively retain and degrade pesticides in short time, thus it generates interest in other countries. Since environmental conditions and agricultural practices vary widely between Countries, the original biobed model has been modified to satisfy site-specific conditions and to utilize locally available organic materials (Fogg et al. 2003; Vischetti et al. 2004; Pigeon et al. 2005; Spliid et al. 2006; Coppola et al. 2007). A series of studies have shown that biomixtures consisting of composted materials instead of peat have high degradation and adsorption capacity (Fogg et al. 2003; Coppola et al. 2007; Fait et al. 2007; Vischetti et al. 2007; Monaci et al. 2009; Kravvariti et al. 2010; Coppola et al. 2011).

Since there are no studies correlating the biobed system with climate change scenarios, it is only possible to predict the activity of the system on the basis of previous work in different environmental conditions.

The future variations in temperature and moisture levels can affect the efficiency of the system. Temperature, as in all biological processes, affects the pollutant degradation, while moisture is a critical factor, which affects contaminants leaching, oxygen availability, and microbial activity.

Castillo and Torstensson (2007) carried out a laboratory study in order to determine the optimal conditions of temperature and moisture for pesticides degradation in a biomixture composed by straw-peat-soil in different proportion. They observed that increasing temperatures increase the microbial and enzymatic activity, which have a positive effect on the degradation of most of the pesticides tested. Moreover, it should also be considered that increase in temperature can also increase the solubility of the organic contaminants, which resulted in higher fraction bioavailable for microbial attacks.

However, increases in temperature have also effects on the degradation of organic matter and this may have drawbacks in terms of necessities for more frequent substitution of the substrate to ensure the system activity. Moisture content in the biomixture should be enough to leave oxygen for aerobic processes and to promote microbial activity and pesticides solubilization.

In the same work, Castillo and Torstensson (2007) observed that moisture at 60% of the

WHC gave the highest dissipation of most pesticides tested while moisture at 30 or 90% limited the microbial activity. As discussed for the wetlands and buffer strips, the most critical effect of climate changes on biobeds would probably be related to water management, while possible increase in temperature (as reported for bioremediation) would probably just result in an increased bioremediation. Given the negative effects of oversaturating WHC on the biobed performances, it will be at least necessary to work with covered biobed systems, to avoid negative effects of increased precipitations, and to promote the relative presence in the biomixture of substrates with higher water sorption capacities such as peat or compost. The development and introduction of closed-cycle drainage systems may also represent an improvement of biobed in order on one hand to reduce the moisture levels, on the other to increase the residence time of the contaminated water on the biomixture.

5 Conclusion and Future Perspectives

In the present chapter, we have addressed a topic still quite neglected in the scientific literature, which is the possible impact of climate changes on the fate, degradation, and mitigation of organic contaminants in the environment. Particular interest was devoted to techniques based on plants (phytoremediation, wetlands, buffer strips), organic biomass residues (biobeds), as well as on bioremediation processes controlled by microorganisms. Since in most cases no specific works on the effects of climate change on bioremediation and mitigation of organic contaminants could be found in the scientific literature, climate change scenarios were identified, and the obtained info critically correlated on available info about the effects of climatic parameters (temperature, precipitations, soil humidity, pH, organic matter, nutrients) on the fate, degradation, and mitigation of contaminants.

As widely known, it was evinced how the future scenarios will be dominated by increasing temperatures, increased precipitation (both in intensity and frequency) in some areas, and decreased precipitations in others. These changes will most probably affect other relevant ecological parameters, such as pH, soil moisture, organic matter levels, microbial activities, and chemicals bioavailability. According to the information gathered in this chapter, these changes will have a quite strong effect on the fate and behavior of contaminants, affecting their distribution in environmental compartments. Major processes controlling the distribution of contaminants will be indeed affected, with probable increases in photodegradation, volatilization, hydrolysis, deposition, runoff, and leaching. Persistence of contaminants may probably be reduced because of increased microbial activity, but on the other hand the reduction of SOM levels predicted by some authors will result in lower degradation and higher bioavailability of pollutants, with possible negative effects in terms of risk assessment.

While effects on fate and behavior can be considered quite relevant, it was found that biodegradation and mitigation will have limited and in some cases positive effects from climate changes. Temperature and moisture will probably increase biodegradation rates, high moisture levels may negatively affect some systems (e.g., wetlands, buffer strips, and biobeds), but this can be easily addressed by promoting more efficient water management systems. In plant-based bioremediation and mitigation systems (phytoremediation, wetlands, and buffer strips), it should be however taken into account how climate change will affect plant physiology and distribution. The solution here is local adaptation, that is the choice of species (in wetlands and buffer strips) or organic materials (in biobed mixtures) that will be more efficient in future climatic conditions. The history of biobed development described in the last chapter is already a successful example of this adaptation possibility, as it shows how a system developed in Northern Europe climatic conditions has been adapted with success to function properly in very different climates.

We can conclude that possible climate change will indeed have a significant effect on the fate and behavior of contaminants, and a more limited effect on bioremediation and mitigation strategies. Since fate and behavior determine the exposure of biological receptor to contaminants toxicity, it will be very important to carry out risk assessment evaluations in the context of climate change. Bioremediation and mitigation will remain very powerful tools to contrast worldwide pollution: attention must be devoted to adapt these tools to climatic changes, in order to maintain and, if possible, improve their efficiencies.

References

- Abu-Zreig M (2001) Factors affecting sediment trapping in vegetated filter strips: simulation study using VFSMOD. Hydrolog Process 15:1477–1488
- Alexander M (1994) Biodegradation and bioremediation. Academic Press, San Diego
- Atagana HI, Haynes RJ, Wallis FM (2003) Optimization of soil physical and chemical conditions for the bioremediation of creosote-contaminated soil. Biodegradation 14:297–307
- Atlas RM, Bartha R (1972) Degradation and mineralization of petroleum in sea water: Limitation by nitrogen and phosphorous. Biotechnol Bioeng 14:309–318
- Baath E, Anderson T (2003) Comparison of soil fungal/ bacterial ratios in a pH gradient using physiological and PLFA-based techniques. Soil Biol Biochem 35:955–963
- Barling RD, Moore ID (1994) Role of buffer strips in management of waterway pollution: a review. Environ Manage 18:543–558
- Benoit P, Barriuso E, Vidon P, Real B (1999) Isoproturon sorption and degradation in a soil from grassed buffer strip. J Environ Qual 28:121–129
- Bloomfield JP, Williams RJ, Gooddy DC, Cape JN, Guha P (2006) Impacts of climate change on the fate and behaviour of pesticides in surface and groundwater-a UK perspective. Sci Total Environ 369:163–177
- Boopathy R (2000) Factors limiting bioremediation technologies. Biores Technol 74:63–67
- Borin M, Vianello M, Morari F, Zanin G (2005) Effectiveness of buffer strips in removing pollutants in runoff from a cultivated field in North-East Italy. Agr Ecosyst Environ 105:101–114
- Borin M, Passoni M, Thiene M, Tempesta T (2010) Multiple functions of buffer strips in farming areas. Eur J Agron 32:103–111
- Brentner L, Mukherji S, Merchie K, Yoon J, Schnoor J, Aken B (2008) Expression of glutathione S-transferases in poplar trees (Populus trichocarpa) exposed to 2, 4, 6-trinitrotoluene (TNT). Chemosphere 73:657–662
- Brix H (1994) Functions of macrophytes in constructed wetlands. Water Sci Technol 29:71–74
- Calderbank A (1989) The occurrence and significance of bound pesticide residues in soil. Environ Contam Toxicol 108:71–103

- Castillo MdP, Torstensson L (2007) Effect of biobed composition, moisture, and temperature on the degradation of pesticides. J Agric Food Chem 55:5725–5733
- Castillo MdP, Torstensson L, Stenström J (2008) Biobeds for Environmental Protection from Pesticide Use A Review. J Agric Food Chem 56:6206–6219
- Chai LK, Wong MH, Mohd-Tahir N, Hansen HC (2010) Degradation and mineralization kinetics of acephate in humid tropic soils of Malaysia. Chemosphere 79:434–440
- Chen CC, McCarl BA (2001) An investigation of the relationship between pesticide usage and climate change. Climatic Change 50:475–487
- Cook L, Inouye R, McGonigle T (2009) Evaluation of four grasses for use in phytoremediation of Cs-contaminated arid land soil. Plant Soil 324: 169–184
- Coppola L, Castillo MdP, Monaci E, Vischetti C (2007) Adaptation of the biobed composition for chlorpyrifos degradation to Southern Europe conditions. J Agric Food Chem 55:396–401
- Coppola L, Castillo MDP, Vischetti C (2011) Degradation of isoproturon and bentazon in peat- and compostbased biomixtures. Pest Manag Sci 67:107–113
- Cunningham SD, Berti WR (1993) Phytoremediation of contaminated soils: Progress and promise. ACS Press, Denver, CO, USA, pp 265–268
- Dalla Valle M, Codato E, Marcomini A (2007) Climate change influence on POPs distribution and fate: A case study. Chemosphere 67:1287–1295
- Davis JW, Madsen S (1996) Factors affecting the biodegradation of toluene in soil. Chemosphere 33:107–130
- Delgado AN, Periago EL, Diaz-Fierros Viqueira F (1995) Vegetated filter strips for wastewater purification: a review. Biores Technol 5:113–122
- deLorenzo ME, Wallace SC, Danese LE, Baird TD (2009) Temperature and salinity effects on the toxicity of common pesticides to the grass shrimp, Palaemonetes pugio. J. Environ. Sci. Health - Part B Pesticides, Food Contamin. Agric Wastes 44:455–460
- El-Refaie G (2010) Temperature impact on operation and performance of Lake Manzala Engineered Wetland, Egypt. Ain Shams Eng J. doi:10.1016/j.asej. 2010.09.001
- Evans N, Baierl A, Semenov MA, Gladders P, Fitt BD (2008) Range and severity of a plant disease increased by global warming. J R Soc Interface 5:525–531
- Fait G, Nicelli M, Fragoulis G, Trevisan M, Capri E (2007) Reduction of point contamination sources of pesticide from a vineyard farm. Environ Sci Technol 41:3302–3308
- Fang C, Smith P, Moncrieff JB, Smith JU (2005) Similar response of labile and resistant soil organic matter pools to changes in temperature. Nature 433:57–59
- Fogg P, Boxall A, Walker A, Jukes A (2003) Pesticide degradation in a 'biobed' composting substrate. Pest Manag Sci 59:527–537
- Fogg P, Boxall A, Walker A, Jukes A (2004) Degradation and leaching potential of pesticides in biobed systems. Pest Manag Sci 60:645–654

- Garcia-Valcarcel AI, Tadeo JL (1999) Influence of soil moisture on sorption and degradation of hexazinone and simazine in soil. J Agric Food Chem 47:3895–3900
- Gavrilescu M (2005) Fate of Pesticides in the Environment and its Bioremediation. Eng Life Sci 5:497–526
- Graham DW, Smith VH, Cleland DL, Law KP (1999) Effect of nitrogen and phosphorus supply on hexadecane biodegradation in soil system. Water Air Soil Pollut 111:1–18
- Gregoire C, Elsaesser D, Huguenot D, Lange J, Lebeau T, Merli A, Mose R, Passeport E, Payraudeau S, Schuetz T, Schulz R, Tapia-Padilla G, Tournebize J, Trevisan M, Wanko A (2009) Mitigation of agricultural nonpointsource pesticide pollution in artificial wetland ecosystems. Environ Chem Lett 7:205–231
- Gustafson DI (1989) Groundwater ubiquity score: a simple method for assessing pesticide leach ability. Environ Toxicol Chem 8:339–357
- Hefting MM, de Klein JJM (1998) Nitrogen removal in buffer strips along a lowland stream in the Netherlands: a pilot study. Environ Pollut 102:521–526
- Hoff RZ (1993) Bioremediation: an overview of its development and use for oil spill cleanup. Mar Pollut Bull 29:476–481
- IPCC (2007) Climate change 2007: the physical science basis summary for policymakers. Contribution of working group I to the Fourth assessment report of the IPCC.
- Jury WA, Winer AM, Spencer WF, Focht DD (1987) Transport and transformations of organic chemicals in the soil-air-water ecosystem. Rev Environ Contam Toxicol 99:119–164
- Kearney P, Wauchope R (1998) Disposal options based on properties of pesticides in soils and water. In: Kearney P, Roberts T (eds) Pesticide remediation in soil and water. Wiley Series in Agrochemicals and Plant Protection, Chichester, pp 35–57
- Kravvariti K, Tsiropoulos NG, Karpouzas DG (2010) Degradation and adsorption of terbuthylazine and chlorpyrifos in biobed biomixtures fromcomposted cotton crop residues. Pest Manag Sci 66:1122–1128
- Kurtz DA, American Chemical Society. Meeting (1990) Long range transport of pesticides. Lewis Publishers, Chelsea, Mich
- Lamon L, Dalla Valle M, Critto A, Marcomini A (2009) Introducing an integrated climate change perspective in POPs modelling, monitoring and regulation. Environ Pollut 157:1971–1980
- Larson SJ, Capel PD, Goolsby DA, Zaugg SD, Sandstrom MW (1995) Relations between pesticide use and riverine flux in the Mississippi River basin. Chemosphere 31:3305–3321
- Ledwith T (1996) Effects of buffer strip width on air temperature and relative humidity in a stream riparian zone. Humboldt State University, Arcata, CA
- Leonard RA (1990) Movement of pesticides into surface waters. In the soil Environment, Madison, WI
- Lu Y, Murase J, Watanabe A, Sugimoto A, Kimura M (2004) Linking microbial community dynamics to

rhizosphere carbon flow in a wetland rice soil. FEMS Microbiol Ecol 48:179–186

- Magee K, Michael A, Ullah H, Dutta S (2008) Dechlorination of PCB in the presence of plant nitrate reductase. Environ Toxicol Pharmacol 25:144–147
- Marilley L, Hartwig UA, Aragno M (1999) Influence of an elevated atmospheric CO₂ content on soil and rhizosphere bacterial communities beneath *Lolium perenne* and *Trifolium repens* under field conditions. Microb Ecol 38:39–49
- Marín S, Sanchis V, Ramos AJ, Magan N (1998) Effect of water activity on hydrolytic enzyme production by *Fusarium moniliforme* and *Fusarium proliferatum* during colonisation of maize. Int J Food Microbiol 42:185–194
- Meagher R (2000) Phytoremediation of toxic elemental and organic pollutants. Curr Opin Plant Biol 3:153–162
- Mendez M, Maier R (2008) Phytoremediation of mine tailings in temperate and arid environments. Rev Environ Sci Biotechnol 7:47–59
- Misra AK, Baker JL, Mickelson SK, Shang H (1996) Contributing area and concentration effects on herbicide removal by vegetative buffer strips. Trans ASAE 39:2105–2111
- Monaci E, Coppola L, Casucci C, Perucci P, Vischetti C (2009) Retention capacity of an organic bio-mixture against different mixtures of fungicides used in vineyards. J Environ Sci Health B 44:724–729
- Moreno D, Pedrocchi C, Comin F, Garcia M, Cabezas A (2007) Creating wetlands for the improvement of water quality and landscape restoration in semi-arid zones degraded by intensive agricultural use. Ecol Eng 30:103–111
- Morgan P, Watkinson RJ (1989) Hydrocarbon degradation in soil and methods for soil biotreatment. Biotechnology 8:305–333
- Nakicenovic N, Alcamo J, Davis G, de Vries B, Fenhann J, Gaffin S, Gregory K, Grubler A, Jung T, Kram T (2000) Special report on emissions scenarios: a special report of Working Group III of the Intergovernmental Panel on Climate Change. Pacific Northwest National Laboratory, Richland, WA (US), Environmental Molecular Sciences Laboratory (US)
- Noyes PD, McElwee MK, Miller HD, Clark BW, Van Tiem LA, Walcott KC, Erwin KN, Levin ED (2009) The toxicology of climate change: environmental contaminants in a warming world. Environ Int 35:971–986
- Paterson S, MacKay D, Tam D, Shiu WY (1990) Uptake of organic chemical by plant: a review of processes, correlation and model. Chemosphere 21:279–331
- Patty L, Real B, Gril JJ (1997) The use of grassed buffer strips to remove pesticides, nitrate and soluble phosphorus compounds from runoff water. Pestic Sci 49:243–251
- Perfumo A, Banat IM, Marchant R, Vezzulli L (2007) Thermally enhanced approaches for bioremediation of hydrocarbon-contaminated soils. Chemosphere 66:179–184

- Pigeon O, De Vleeschouwer C, Cors F, Weickmans B, De Ryckel B, Pussemier L, Debongnie P, Culot M (2005) Development of biofilters to treat the pesticides wastes from spraying applications. Commun Agric Appl Biol Sci 70:1003
- Puglisi E, Cappa F, Fragoulis G, Trevisan M, Del Re AAM (2007a) Bioavailability and degradation of phenanthrene in compost amended soils. Chemosphere 67:548–556
- Puglisi E, Murk AJ, van den Berg HJ, Grotenhuis T (2007b) Extraction and bioanalysis of the ecotoxicologically relevant fraction of contaminants in sediments. Environ Toxicol Chem 26:2122–2128
- Qui Y, Pang H, Zhou Z, Zhang P, Feng Y, Sheng GD (2009) Competitive Biodegradation of dichlobenil and atrazine coexisting in soil amended with a char and citrate. Environ Pollut 157:2964–2969
- Raper SC, Braithwaite RJ (2006) Low sea level rise projections from mountain glaciers and icecaps under global warming. Nature 439:311–313
- Reilly J, Tubiello F, McCarl B, Abler D, Darwin R, Fuglie K, Hollinger S, Izaurralde C, Jagtap S, Jones J (2003) US agriculture and climate change: new results. Climatic Change 57:43–67
- Rogers HH, Runion GB, Krupa SV (1994) Plant responses to atmospheric CO2 enrichment with emphasis on roots and the rhizosphere. Environ Pollut 83:155–189
- Salmon C, Crabos J, Sambuco J, Bessiere J, Basseres A, Caumette P, Baccou J (1998) Artificial wetland performances in the purification efficiency of hydrocarbon wastewater. Water Air Soil Pollut 104:313–329
- Salt DE, Blaylock M, Kumar NP, Dushenkov V, Ensley BD, Chet I, Raskin I (1995) Phytoremediation: a novel strategy for the removal of toxic metals from the environment using plants. Biotechnology 13:468–474
- Samuel T, Pillai MKK (1989) The effect of temperature and solar radiation on volatilisation, mineralisation and degradation of [¹⁴C]-DDT in soil. Environ Pollut 57:63–77
- Schmitt TJ, Dosskey MG, Hoagland KD (1999) Filter strip performance and processes for different vegetation, widths, and contaminants. J Environ Qual 28:1479–1489
- Schoefs O, Perrier M, Samson R (2004) Estimation of contaminant depletion in unsaturated soils using a reducedorder biodegradation model and carbon dioxide measurement. Appl Microbiol Biotechnol 64:53–61
- Semple KT, Reid BJ, Fermor TR (2001) Impact of composting strategies on the treatment of soils contaminated with organic pollutants. Environ Pollut 112:269–283
- Shaw LJ, Burns RG (2005) Rhizodeposition and the enhanced mineralization of 2,4-dichlorophenoxyacetic acid in soil from the Trifolium pratense rhizosphere. Environ Microbiol 7:191–202
- Sims GK, Radosevich M, He XT, Traina SJ (1991) The effects of sorption on the bioavailability of pesticides. In: Betts WB (ed) Bio-degradation: Natural and synthetic materials. Springer, London
- Singh BK, Walker A, Morgan JA, Wright DJ (2003) Effects of soil pH on the biodegradation of chlorpyri-

fos and isolation of a chlorpyrifos-degrading bacterium. Appl Environ Microbiol 69:5198–5206

- Smart DR, Ritchie K, Stark JM, Bugbee B (1997) Evidence that elevated CO2 levels can indirectly increase rhizosphere denitrifier activity. Appl Environ Microbiol 63:4621–4624
- Spliid N, Helweg A, Heinrichson K (2006) Leaching and degradation of 21 pesticides in a full-scale model biobed. Chemosphere 65:2223–2232
- Steffensen WS, Alexander M (1995) Role of competition for inorganic nutrients in the biodegradation of mixtures of substrates. Appl Environ Microbiol 61:2859–2862
- Stenrød M, Perceval J, Benoit P, Almvik M, Bolli RI, Eklo OM, Sveistrup TE, Kværner J (2008) Cold climatic conditions: Effects on bioavailability and leaching of the mobile pesticide metribuzin in a silt loam soil in Norway. Cold Regions Sci Technol 53:4–15
- Stottmeister U, Wießner A, Kuschk P, Kappelmeyer U, Kastner M, Bederski O, Muller RA, Moormann H (2003) Effects of plants and microorganisms in constructed wetlands for wastewater treatment. Biotechnol Adv 22:93–117
- Susarla S, Medina VF, McCutcheon SC (2002) Phytoremediation: An ecological solution to organic chemical contamination. Ecol Eng 18:647–658
- Tanner CC (2001) Plants as ecosystems engineers in subsurface-flow treatment wetlands. Water Sci Technol 44:9–17
- Taylor AW, Spencer WF (1990) Volatilization and vapor transport processes. In: Cheng HH (ed) Pesticides in the environment, SSSA book series no. 2. Soil Science Society of America, Madison, WI, pp 213–270
- Taylor CR, Hook PB, Stein OR, Zabinski CA (2010) Seasonal effects of 19 plant species on COD removal in subsurface treatment wetland microcosms. Ecol Eng. doi:10.1016/j.ecoleng.2010.05.007
- Thuiller W, Lavorel S, Araujo MB, Sykes MT, Prentice IC (2005) Climate change threats to plant diversity in Europe. Proc Natl Acad Sci USA 102:8245–8250
- Torstensson L (2000) Experiences of biobeds in practical use in Sweden. Pestic Outlook 11:206–212
- Torstensson L, Castillo MdP (1997) Use of biobeds in Sweden to minimize environmental spillages from agricultural spraying equipment. Pestic Outlook 8(June):24–27
- Trevisan M, Montepiani C, Ragozza L, Bartoletti C, Ioannilli E, Del Re AAM (1993) Pesticides in rainfall and air in Italy. Environ Pollut 80:31–39
- Vamerali T, Bandiera M, Mosca G (2010) Field crops for phytoremediation of metal-contaminated land. A review. Environ Chem Lett 8:1–17
- Vianello M, Vischetti C, Scarponi L, Zanin G (2005) Herbicide losses in runoff events from a field with a low slope: role of a vegetative filter strip. Chemosphere 61:717–725
- Vischetti C (2008) Agrofarmaci conoscenze per un uso sostenibile. In: Gennari M, Trevisan M (eds) Agrofarmaci-conoscenze per un uso sostenibile. Oasi Alberto Perdisa, Bologna

- Vischetti C, Perucci P, Scarponi L (1997) Rimsulfuron in soil: Effect of persistence on growth and activity of microbial biomass at varying environmental conditions. Biogeochemistry 39:165–176
- Vischetti C, Capri E, Trevisan M, Casucci C, Perucci P (2004) Biomassbed: a biological system to reduce pesticide point contamination at farm level. Chemosphere 55:823–828
- Vischetti C, Coppola L, Monaci E, Cardinali A, Castillo MdP (2007) Microbial impact of the pesticide chlorpyrifos on Swedish and Italian biobeds. Agron Sustain Dev 27:267–272
- Walker A, Eagle DJ (1983) Prediction of herbicide residues in soil for advisory purposes. Asp Appl Biol 4:503–509
- Walworth J, Braddock J, Woolard C (2001) Nutrient and temperature interactions in bioremediation of cryic soils. Cold Regions Sci Technol 32:85–91

- Weaver M, Zablotowicz R, Locke M (2004) Laboratory assessment of atrazine and fluometuron degradation in soils from a constructed wetland. Chemosphere 57:853–862
- Wilson LG (1967) Sediment removal from flood water by grass filtration. Trans ASAE 10:35–37
- Worthington C (1988) Worthington enzyme manual. Worthington Biochemical Corporation, Freehold, NJ, pp 254–260
- Zhang P, Sheng G, Feng Y, Miller D (2005) Role of wheatresidue-derived char in the biodegradation of benzonitrile in soil: nutritional stimulation versus adsorptive inhibition. Environ Sci Technol 39:5442–5448
- Zhu K, Chen H, and Nan Z (2010) Phytoremediation of Loess Soil Contaminated by Organic Compounds. In Application of Phytotechnologies for Cleanup of Industrial, Agricultural, and Wastewater Contamination. Ed. PA Kulakpw, and VV Pidlisnyuk. pp 159–176. Springer

23

Exploitation of Weeds and Ornamentals for Bioremediation of Metalliferous Substrates in the Era of Climate Change

M.N.V. Prasad

Abstract

Weeds not only adversely affect the plant productivity but many of them also cause health hazards in human beings and animals. They are also known to seriously affect the biodiversity. Apart from this negative side, many weeds are known to have beneficial properties in one way or the other and have immense potential as food and fodder, medicinal, aromatic, phytoremediation, industrial, soil and water conservation resources, etc. A very little information is available on the use of weeds for such beneficial purposes. Therefore, this subject of research needs to be explored and expanded. Several of the weeds are utilized for (a) soil and water conservation, (b) alternative livelihood opportunities, and (c) industrial uses. Survey of published literature indicates that there is great scope for application of weeds in bioremediation. More research efforts are required for utilizing weeds for bioremediation of different type of pollutants from air, water, and soil. Ornamental plants have an added advantage of enhancing the environmental esthetics besides cleaning the environment. This approach has several advantages for environmental moderation, cleanup, and generation of revenue. Therefore, this approach will add a new dimension to the field of bioremediation of contaminated aquatic and terrestrial environments.

Keywords

Weeds • Ornamentals • Phytostabilization • Phytoremediation • Metalliferous substrates • Biocontrol

M.N.V. Prasad (🖂)

Department of Plant Sciences, University of Hyderabad, Prof. C.R. Rao Road, Gachibowli, Central University P.O., Hyderabad, AP 500 046, India e-mail: mnvsl@uohyd.ernet.in; prasad_mnv@yahoo.com

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4_23,

[©] Springer Science+Business Media, LLC 2012

1 Introduction

Bioremediation is an emerging and an effective technology for treatment of a wide variety of contaminants. This technology includes plant- and microbe-mediated processes (phytoremediation and rhizoremediation, respectively). Bioremediation approach is currently applied to contain contaminants in soil, groundwater, surface water, or sediments including air. These technologies have become attractive alternatives to conventional cleanup technologies due to relatively low capital costs and the inherently esthetic nature (Fig. 23.1). Quite a variety of plants, natural, transgenic, and/or associated with rhizosphere micro-organisms are extraordinarily active in these biological interventions for cleaning up pollutants by removing or immobilizing (Ma et al. 2011). Climate change will affect the ability of ecological systems that provide a range of essential ecological goods and services, such as food and fiber production; provision of clean and sufficient water maintenance of biodiversity; maintenance of human health; and storage and cycling of carbon, nitrogen, and phosphorus.

Technogenic and anthropogenic sources of metals is subject of importance not only to human health but also in general to the field of biogeochemistry, environment, and medicine (Figs. 23.2–23.4). Smelting and mining processes are the point source of a contamination of metals causing environmental contamination and pollution. Consequently, these pollutants get dispersed in natural resources (soil, water, and air) and ultimately enter the food chain. Physical stabilization (covering the metalliferous waste with geotextiles/ geomembranes, etc.) and chemical stabilization (use of chemical binding agents) to reduce wind and water erosion are not a feasible proposition for

large areas. However, phytostabilization – use of a specific type of vegetation is far more desirable than physical and chemical stabilization (Prasad 2006; Tordoff et al. 2000). Phytostabilization is an effective process of phytoremediation technology.

In order to cleanup large areas contaminated with toxic metals, plants producing very high biomass with limited inputs and simplistic management are desirable. It is a general belief that climate change promotes explosion of weeds in addition to other phenomena (Fig. 23.5).



Fig. 23.1 Bioremediation, a schematic presentation. Bioremediaiton is based on use of the ecosystem services provided by its biotic compartment. Some examples of its application include the reduction and control of pollution

through wetland systems, restoration of degraded natural systems or establishment of Eco-industrial parks, carbon sinks and ameliorating the effects and impacts of climate change



Fig. 23.2 Biogeochemical cycling of heavy metals in a generalized ecosystem



Rhizoremediation

Fig. 23.3 Bioremediation knowledge capital triangle

Non-alien plants when they grow outside their niche become invasive. Invasives are widely distributed in a variety of ecosystems throughout the world. Many invasive alien species support farming and forestry systems positively in a big way. However, some of the alien species become invasive when they are introduced deliberately or unintentionally outside their natural habitats into new areas where they express the capability to establish, invade, and outcompete native species. According to International Union for Conservation of Nature and Natural Resources (IUCN), alien invasive species means, an exotic species which becomes established in natural or seminatural ecosystems or habitat, an agent of undesirable change which threatens the native biological diversity. Invasive species are therefore considered to be a serious hindrance to conservation and profitable use of biodiversity, with significant undesirable impacts on the services provided by ecosystems. Alien invasive species are supposed to have huge requirement and destructive modes of resource acquisition and consumption that would ultimately bring change in soil structure and nutrient composition, its profile, decomposition, moisture availability, etc.

Trace metal contamination and pollution in the environment is increasing due to technogenic and geogenic sources. The flux of trace metals deteriorates the quality of the environment since these are considered to be cytotoxic, mutagenic, and carcinogenic. In order to be healthy, physically and mentally, clean soil, water, and air are



Fig. 23.4 Sources of heavy metals in the environment and various applications of bioremediation for treatment of natural resources and for miscellaneous applications



Fig. 23.5 Climate change induced adaptations and mitigation processes

prerequisites. Trace metal contamination and pollution would exert direct and indirect harmful effects that would eventually deteriorate biodiversity and economic wealth. In developed nations, trace metal contamination or pollution is often highly localized and the pressure to use such contaminated land and water for agricultural food production is minimal. In contrast, the technogenic/geogenic pollution and contamination is widespread in many Asian and eastern European countries and is dramatically increasing in Southeast Asia, India, and China (Cheng 2003; Meharg 2004). In order to contain trace metal pollution in soil, water, and air, phytoremediation is being considered as a low cost solution and a globally recognized technology (Garbisu and Alkorta 2001; Garbisu et al. 2002; McCutcheon and Schnoor 2003; Prasad and Freitas 2003; Macek et al. 2004; Gratão et al. 2005). Ornamental foliage plants have been suggested for the removal of arsenic (Alkorta et al. 2004). Therefore, this approach of involving weeds for bioremediation of metalliferous substrates in the era of climate change would yield desirable results with limited efforts and investments (Lorenzini et al. 2006; Prasad and Freitas 2003).

2 Grasses: Ideal for Phytostabilization

Grasses are tolerant to toxic metals and have played a convincing role in phytostabilization (Prasad 2006; Lai and Chen 2006; Li et al. 2009; Néel et al. 2007; Redondo-Gómez et al. 2011; Shu et al. 2002; Vernay et al. 2007; Wang et al. 2005; Zhang et al. 2010; Atabayeva et al. 2010). Abandoned mine soils and estuarine sediments are phytostabilized against erosion by grass species (Cambrollé et al. 2008; Comino et al. 2009; Mateos-Naranjo et al. 2008). Soil amendments and biosolids accelerate phytostabilization process (Santibáñez et al. 2008; Zhou et al. 2007). Grasses possess thickets of adventitious roots, unique root morphology (Li et al. 2009), high bioproductivity (Liu et al. 2009), therefore have an added advantage for application in phytostabilization. Further, grasses are often associated with mycorrhizal and endophytic fungi (Chen et al. 2008; Deram et al. 2007, 2008; Kuldau and Bacon 2008; Ortega-Larrocea et al. 2010; Punamiya et al. 2010). Grasses together with legume association have helped in situ stabilization of chemical waste (Hartley et al. 2009; Hartley and Lepp 2008). In climate constrained and carbon dioxide enriched era, grasses have physiological advantage (majority being C_{4}) of producing/increasing their biomass. Hence, grasses perform well in phytostabilization process (Wu et al. 2009). In view of their advantageous metabolic processes, hydroponic grass system based on plate or fabric is considered for the treatment of aquacultural wastewater (Pan et al. 2007).

2.1 Lolium perenne (Ryegrass)

It is a perennial exhibits luxuriant growth and produces large amounts of aboveground biomass. It has been used for phytostabilization of abandoned uranium mine (Abrutiga, Portugal) (Fig. 23.6).

2.2 Panicum virgatum (Switchgrass)

It is one of the perennial rhizomatous grasses being developed for the purpose of biomass production. It is a perennial C_4 grass propagated by seed that can be established at low cost and requires very low inputs while giving high biomass yields even on marginal soils. Attributes of switchgrass desirable for bioenergy cropping include its demonstrated high productivity across many environments, suitability for marginal and erosive land, relatively low water and nutrient requirements, and positive environmental benefits. There is need to examine its (a) adaptability across a range of contaminated sites, (b) fresh and dry matter yields.

2.3 *Prosopis juliflora* (Velvet Mesquite)

It is an evergreen phreatophyte, fast growing, drought resistant, widely distributed not only in



Fig. 23.6 Lolium perenne - phytostabilization of abandoned mine and soil profile

India but also in other arid and semiarid tropical countries. It is the only exotic species capable of growing on a wide variety of soils and climatic conditions. It is a valued tree for shade, timber, and forage. It is a thorny, deciduous, large crowned and deep rooted bush or tree which grows up to 10-m height or more, depending on the variety and climatic conditions. It is also widely distributed in the dry tropical and subtropical regions of Central America and Northern South America.

P. juliflora is an ideal species for stabilizing the pegmatitic tailings of mica mines in Nellore district of Andhra Pradesh, India (Nagaraju and Prasad 1998). It is also helpful for reclamation of copper, tungsten, marble, dolomite mine tailings and is a green solution to heavy metal contaminated soils (Varun et al. 2011). It is an appropriate species for rehabilitation of gypsum mine spoil in arid zone and restoration of sodic soils. It outperformed all other tree species in sand dune stabilization (Kailappan et al. 2000; Rai et al. 2003; Senthilkumar et al. 2005). Mycorrhizae are reported to greatly improve the growth of *P. juliflora* on high pH soils. *P. juliflora* was able to grow satisfactorily without amendments up to pH 9. Arbuscular mycorrhizal

inocula isolated from its rhizosphere (low cost agrotechnology) were found to accelerate the growth of other agroforestry and social forestry legumes in perturbed ecosystems (Gardea-Torresdey et al. 2005; Siddhu and Behl 1997; Singh 1995).

3 Ornamentals for Environmental Moderation and Toxic Trace Metal Cleanup

Several ornamental plants have successfully been applied in environmental toxic cleanup. (Belmonth and Metcalfe 2003; Chintakovid et al. 2007; Davies et al. 2001; Madejon et al. 2003; Mazen 2004; McIntyre 2003; Mungur et al. 1995; Murillo et al. 1999; Negri and Hinchman 2000; Niu et al. 2007; Wei et al. 2009, 2010a, b)

Lemon-scented geraniums (*Pelargonium* sp. "Frensham," or scented geranium) accumulated large amounts of Cd, Pb, Ni, and Cu from soil in greenhouse experiments (Dan et al. 2000). Biotechnological interventions through hairy root regenerants are useful in floriculture (Giri and Narasu 2000; Giovanni et al. 1997). Pellegrineschi et al. (1994) improved the ornamental quality of scented *Pelargonium* spp. This plant has pleasant odor that adds scent to the toxic metal contaminated soil.

Vetiveria zizanioides (Vetiver grass): It is known to have multiple uses. This plant had several popular names such as "the miracle grass," "a wonder grass," "a magic grass," "an unique plant," "an essential grass," "an amazing plant," "an amazing grass," "a versatile plant," "a living barrier," "a living dam," "a living nail," "a living wall," "an eco-friendly grass." This extraordinary grass is adaptable to multiple environmental conditions and it is globally recognized as an easy and economical alternative to control soil erosion and to solve a variety of environmental problems. It has been used for restoration, conservation, and protection of land disrupted by man activities like agriculture, mining, construction sites, oil exploration and extraction, infrastructure corridors, as well as used for water conservation in watershed management, disaster mitigation, and treatment of contaminated water and soil. Research at the global level has proved the relevance of vetiver in multiple applications. In Australia, V. zizanioides has been successfully used to stabilize mining overburden and highly saline, sodic, magnesic, and alkaline (pH 9.5) tailings of coalmines, as well as highly acidic (pH 2.7) arsenic tailings of gold mines. In China, it has been demonstrated that V. zizanioides is one of the best choices for revegetation of Pb/Zn mine tailings due to its high metal tolerance (Chen et al. 2004a, b; Chiu et al. 2005, 2006; Chong and Chu 2007; Rotkittikhun et al. 2007; Makris et al. 2007; Pang et al. 2003; Singh et al. 2008; Truong 2000; Wilde et al. 2005).

- (a) Hydrocotyle umbellata (Pennywort): It is a wetland/marshy plant commonly found in many tropical countries. The plant grows very rapidly and serves as an ornamental and decorative purpose. It is reported to remove trace metals from aquatic systems.
- (b) Alternanthera philoxeroides (Alligator weed): It is one of the most common aquatic weed in contaminated/polluted ecosystem. This is native to South America and naturalized in India (Naqvi and Rizvi 2000). Several

Amaranthaceae produce large biomass and are suitable for environmental remediation and toxic metal cleanup (for e.g., *Amaranthaceae retrflexus*, Prasad 2001) (Table 23.1).

(c) Talinum cuneifolium (Portulacaceae): It is a succulent shrub of about 60-cm height with cuneate to obovate leaves, flowers in terminal panicles and purple colored corolla. It flowers and fruits throughout the year. It is widely distributed in India, Arabia, and Africa. Cuttings are a ready means of propagation of these plants. T. cuneifolium is and has been reported to accumulate high levels of copper in its leaves. These plants showed absorption barriers at high soil copper concentrations, indicating limits to uptake of the metal (Tiagi and Aery 1986; Adeniyi 1996).

4 Ornamental Hydrophytes for Phytoremediation

Several ecotechnological opportunities are available for aquatic plants (Lakshman 1987; Outridge and Noller 1991). The use of aquatic plants in water quality assessment has been in practice for centuries. The occurrence of aquatic macrophytes is unambiguously related to water chemistry and using these plant species or communities as indicators or biomonitors has been well recognized and established for in situ bioremediation (Deng et al. 2004). The noable examples are: Azolla filiculoides, A. philoxeroides, Bacopa monnieri, Canna flaccida, Carex juncell, Carex pedula, Carex rostrata, Carex Sp., Ceratophyllum demersum, Chara, Nitella, Cladium jamaicense, Cyperus eragrostis, Distichlis spicata, Eichhornia crassipes, Elodea canadensis, Elodea densa, E. crassipes, Eriocaulon septangulare, Euryale ferox, Elodea nuttallia, E. canadensis, Eloea sptangulare, Eriophorum angustifolium, Eriophorum scheuchzeri, Glyceria fluitans, Hydrilla verticillata, Hygrophila onogaria, Isoetes lacustris, Lemna minor, L. trisulca, L. gibba, L. palustris, H. umbellata, Ipomea aquatica, Juncus articulatus, L. minor, Littorella uniflora, Ludwigia natans, Lysimachia nummularia, Myriophyllum spicatum,

Tolerant to cadmium stress	Ding et al. (2007)
Responds rapidly to shoot removal	Wilson et al. (2007)
Accumulate Cd, Pb, and Zn from constructed wetlands	Liu et al. (2007a, b)
Herbivory, mowing, and herbicides differently affect production and nutrient allocation	Schooler et al. (2007)
Distribution and bioaccumulation of microcystins in water columns: a systematic investigation into the environmental fate and the associated risks with microcystins	Song et al. (2007)
Lead and zinc accumulation and tolerance in populations	Deng et al. (2006)
Exhibit phenotypic plasticity in relation to different water availability	Geng et al. (2006)
Differently respond to biological control	Li and Ye (2006)
Abiotic stress and phenotypic plasticity influenced riparian zone population	Pan et al. (2006)
Growth and reproduction simulated herbivory	Schooler et al. (2006)
Removes Ni(II), Zn(II), and Cr(VI) from aqueous solution	Wang and Qin (2006)
Genetic diversity has been established in Alternanthera philoxeroides in China	Wang et al. (2005)
Suitable for phytoremediation of small-scale oil spills in fresh marsh environments: a mesocosm simulation	Dowty et al. (2001)
Biologically controlled with fungi	Barreto et al. (2000)
Its extract had antiviral effect on epidemic hemorrhagic fever virus in vivo	Peng et al. (1997)
Contain phytochemicals, viz., alternanthin, A C-glycosylated flavonoid	Zhou et al. (1988)
Accumulate monosodium methanearsonate (MSMA)	Anderson et al. (1980)
The economics of its biological control	Andres (1977)
Insects as agents for biological control	Bennett (1977)
The biological control in the USA	Spencer and Coulson (1976)
Water hyacinths and alligator weeds for removal of lead, mercury, silver, cobalt, and strontium from polluted waters	Wolverton and McDonlad (1975a, b)

Table 23.1 Alternanthera philoxeroides (Mart.) Griseb: Potential for environmental remediation and cleanup

This is not an exhaustive

M. alterniflorum, Melilotus indica, Mentha aquatica, Miscanthus floridulus, Miscanthus sacchariflorus, Mougeotia, Najas marina, Nasturtium officinale, Nuphar lutea, Nymphaea alba, Nymphaea violacea, Nymphoides germinate, Potamogeton natans, P. attenuatum, P. communis, Potamogeton crispus, P. filiformis, P. lapathifoilum, P. orientalis, P. pectinatus, P. perfoliatus, P. richardsonii, P. subsessiles, Phragmites karka, Pistia stratiotes, Ranunculus aquatilis, Ruppia maritima, Sagittaria latifolia, Salvinia acutes, Salvinia molesta, Scapania uliginosa, Schoenoplectus lacustris, Scirpus validus, Spartina alterniflora, Spirodela oligorrhiza, Sporobolus virginicus, Typha domingensis, Typha latifolia, Vallisneria americana, Vallisneria spiralis, Wolffia globosa, and Zizania aquatica (Prasad 2007; McCutcheon and Schnoor 2003; Keskinkan et al. 2004; Peles et al. 2002; Hattink et al. 2000; Sheppard and Motycka 1997).

Aquatic plants have been frequently used to remove suspended solids, nutrients, trace metals, toxic organics and bacteria from acid mine drainage (AMD), agricultural landfill, and urban storm-water runoff. In addition, considerable research has been focused on determining the usefulness of macrophytes, as biomonitors of polluted environments and as bioremediative agents in waste water treatments. The response of an organism to deficient or excess levels of metal (i.e., bioassays) can be used to estimate metal impact. Such studies done under defined experimental conditions can provide results that can be extrapolated to natural environment. There are multifold advantages in using an aquatic macrophyte as a study material. Macrophytes are costeffective, universally available with their ability to survive adverse conditions and high colonization rates and are excellent tools for studies of phytoremediation. Rooted macrophytes especially



Fig. 23.7 Cascade model for removal of xenobiotics and treatment of waste streams. Most commonly employed species are *Spartina alterniflora*, Cord grass; *Sporolobus virginicus*, Coastal dropseed; *Salicornia virginica*, Perennial glasswort; *Cladium jamaicense*, Sawgrass;

play an important role in metal bioavailability through rhizospheric processes. Macrophytes (floating, emergent, and submersed) readily take up metals in their reduced form from sediments, which exist in anaerobic situations due to lack of oxygen and oxidize them in the plant tissues making them immobile and bioconcentrate them in their tissues, thus reduced the toxic trace metal bioavailability in the interstitial waters. Rooted and emergent macrophytes make them particularly effective as bioindicator of metal pollution, as they represent real levels present at that site.

In the past research with macrophytes has centered mainly on determining effective eradication techniques for nuisance growth of several species such as *Elodea canadensis, Eichhornia crassipes, Ceratophyllum demersum*, etc. Scientific literature exists for the use of a wide diversity of macrophytes in toxicity tests designed to evaluate the hazard of potential pollutants. Estuarine and marine plant species are being used considerably less than freshwater species in toxicity tests conducted for regulatory reasons. *Lemna, Myriophyllum, Potamogeton, Ceratophyllu, Elodea, E. crassipes* have been exhaustively used in phytotoxicity investigations. Duckweeds have received the greatest attention for toxicity tests as they

Salicornia alterniflora, Vermillon cordgrass; *Scirpus validus*, Great bulrush. A cascade model of treatment system was suggested for removal of radionuclides by rhizofiltration Dushenkov et al. 1995; 1997a, b; 2002 and blastofiltration (seedlings), e.g., sunflower

are relevant to many aquatic environments, including lakes, streams, and effluents.

The most important role of plants in wetlands is that they increase the residence time of water, which means that they reduce the velocity and thereby increase the sedimentation of particles and associated pollutants. Thus, they are indirectly involved in water cleaning. Plants also add oxygen providing a physical site of microbial attachment to the roots generating positive conditions for microbes and bioremediation.

Constructed and engineered wetlands (including naturalvwetlands) are in use for centuries for waste water treatment containing organic matter, nitrogen, phosphorus, (Kadlec and Knight 1996) (Figs. 23.7 and 23.8).

Aquatic macrophytes have paramount significance in the monitoring of metals in aquatic ecosystems (e.g., *L. minor*, *E. crassipes*, *Azolla pinnata*). Aquatic plants are important in nutrient cycling, control of water quality, sediment stabilization, and provision of habitat for aquatic organisms. The use of aquatic macrophytes in water quality assessment has been a common practice employing in situ biomonitors (Sobolewski 1999). The submerged aquatic



Fig. 23.8 Use of arsenic hyperaccumulator ferns and wetland vegetation (Carbonell-Barrachina et al. 1998; Rahman and Hasegawa 2011; Ma et al. 2001) for arsenic removal from water (Elless et al. 2005)

macrophytes have very thin cuticle and therefore readily take up metals from water through the entire surface. Macrophytes with their ability to survive adverse conditions and high colonization rate are excellent tools for phytoremediation. Further they redistribute metals from sediments to water and finally take up in the plant tissues and hence maintain circulation. Benthic rooted macrophytes (both submerged and emergent) play an important role in metal bioavailability from sediments through rhizosphere exchanges and other carrier chelates. This naturally facilitates metal uptake by other floating and emergent forms of macrophytes. Macrophytes readily take up metals in their reduced form from sediments, which exist in anaerobic situations due to lack of oxygen and oxidize them in the plant tissues making them immobile and thus get bioconcentrated in their tissues (Okurut et al. 1999).

A special group of plants may reduce element leakage from mine tailings by phytostabilization. Plants that are tolerant to elements of high concentrations have been found useful for reclamation of dry mine tailings containing elevated levels of metals and other elements. Mine tailings rich in sulfides, e.g., pyrite, can form AMD if it reacts with atmospheric oxygen and water, which may also promote the release of metals and As. To prevent AMD formation, mine tailings rich in sulfides may be saturated with water to reduce the penetration of atmospheric oxygen. An organic layer with plants on top of the mine tailings would consume oxygen, as would plant roots through respiration. Thus, phytostabilization on water-covered mine tailings may further reduce the oxygen penetration into the mine tailings and prevent the release of elevated levels of elements into the surroundings. Metal tolerance can be evolutionarily developed while some plant species seem to have an inherent tolerance to trace metals. Since, some wetland plant species have been found with the latter property, for example, T. latifolia, G. fluitans, and Phragmites australis, wetland communities may easily establish on submerged mine tailings, without prior development of metal tolerance. Some plant species have mechanisms that make it possible to cope with high external levels of elements. Low accumulators are plants that can reduce the uptake when the substrate has high
element concentrations or have a high net efflux of the element in question, thus the plant tissue concentration of the element is low even though the concentration in the substrate is high (Williams 2002; Wood and Mcatamney 1994; Woulds and Ngwenya 2004; Ye et al. 2001).

5 Utilization of Water Weeds for Bioremediation of Metalliferous Substrates

Weeds cannot be eradicated, hence there is a need to find appropriate and sustainable solutions. (Ji et al. 2011; Lin and Liu 2003); Liphadzi et al. 2003) Several of the wetland plants not only effectively purify metal contaminated water effectively (Horne 2000), Zhang et al. 2007 but also establish a dense vegetative cover (Ye et al. 2003).

For successful phytoremediation, plants chosen should have the following attributes:

- (a) Adaptive and tolerance mechanisms.
- (b) Fast growing with high bioproductivity such as duck weeds (Fig. 23.9) *Typha latifolia* (Cattail) and *Phragmites australis* (Ye et al. 1997a, b), *Eichhornia crassipes* (water hyacinth), *Alternanthera philoxeroides* (alligator weed), *Pistia stratiotes* (water lettuce), and *Potamogeton crispus*. The biomass production of these plants often exceeds the yield of most productive agricultural crops.

Further, the dried biomass of many of these aquatic macrophytes is an excellent biosorbent for removal of Cr(III), Ni(II), Cu(II), Zn(II), Cd(II), and Pb(II) (Andre et al. 1999). Wetland plants (water weeds) accelerate the sedimentation in constructed wetland and this being principal process for the removal of heavy metals from wastewater. Also wetland plants act as sites for metal precipitation (Mays and Edwards 2001). Water weeds for treatment of waste water are enumerated in Table 23.2



Fig. 23.9 Duck weeds for waste water treatment and for phytoproducts

Uptake of Zn, Cu, and Cd in metal loaded Elodea canadensis	Nyquist and Greger (2007)
Tolerance and phytoaccumulation of chromium by three Azolla species	Arora et al. (2006)
Tolerance and accumulation of copper and chromium in two duckweed species: Lemna minor L and Lemna gibba L	Ater et al. (2006)
Phytoremediation of chromium by model constructed wetland	Mant et al. (2006)
Wetland grasses for phytoremediation	Czako et al. (2005)
Accumulation of As in <i>Lemna gibba</i> (duckweed) in tailing waters of two abandoned uranium mining sites in Saxony, Germany	Mkandawire and Dudel (2005)
Potential of Azolla caroliniana for the removal of Pb and Cd from wastewaters	Stepniewska et al. (2005)
Lead accumulation in the aquatic fern Azolla filiculoides	Benaroya et al. (2004)
The ability of <i>Azolla caroliniana</i> to remove heavy metals such as Hg ²⁺ , Cr ³⁺ , Cr ⁶⁺ from municipal wastewater	Bennicelli et al. (2004)
Responses induced by high concentration of cadmium in <i>Phragmites australis</i> roots	Ederli et al. (2004)
Capacity of Salvinia minima to tolerate and accumulate As and Pb	Hoffmann et al. (2004)
Phytoaccumulation of heavy metals by aquatic plants	Kamal et al. (2004)
Bioaccumulation of copper from contaminated wastewaters by using <i>Lemna minor</i> (aquatic green plants)	Kara (2004)
Heavy metal adsorption properties of a submerged aquatic plant (<i>Ceratophyllum demersum</i>)	Keskinkan et al. (2004)
Capacity of <i>Lemna gibba</i> (duckweed) for uranium and arsenic phytoremediation in mine tailing waters	Mkandwire et al. (2004)
Accumulation of trace elements by <i>Pistia stratiotes</i> : implications for phytoremediation	Odjegba and Fasidi (2004)
Metal uptake transport and release by wetland plants: implications for phytoreme- diation and restoration	Weis and Weis (2004)
Lead and nickel removal using Microspora and Lemna	Axtell et al. (2003)
Removal of heavy metals from aqueous solution by water hyacinth (<i>Eichhornia crassipes</i>)	Ingole and Bhole (2003)
Removal by marsh macrophytes <i>Spartina alterniflora</i> (cordgrass) and <i>Phragmites australis</i> (common reed)	Windham et al. (2003)
Phytoaccumulation and phytotoxicity of Cd and Cr in Wolffia globosa	Boonyapookana et al. (2002)
Chromium removal from tannery effluents by aquatic plants	Sinha et al. (2002)
Biosorption of cadmium and chromium in duckweed Wolffia globosa	Upatham et al. (2002)
Chromium phytoaccumulation from solution by selected hydrophytes	Zurayk et al. (2001)

Table 23.2 Water weeds for treatment of waste water: Lab, field, and pilot-scale experiments (not exhaustive)

5.1 Ipomoea aquatica

Ipomoea aquatica is a fast growing aquatic plant and has been applied widely to purify eutrophic water. It is a metal accumulator and metal removal potential depends upon levels of metal contamination in the water body in which they were growing. Water is regarded as a limited and susceptible resource, essential for life. It is widely distributed throughout tropical and warm climate regions in the world, especially in China and India. It is a fast-growing herbaceous vine commonly found in creeping on muddy stream banks or floating in freshwater marshes and ponds. Moreover, its leaves have high nutritive value and eaten as vegetables by human beings as well as fish and other grazing animals, and possess medicinal importance. In addition, in recent years, it is also used widely to purify wastewater (Gothberg et al. 2002, 2004; Cao et al. 2006; Hu et al. 2007). Rai et al. (1995) reported the toxic metals Pb, Cd, and Cr in *I. aquatica* accumulated highly from water resources of Eastern Ghats of India.

6 Biocontrol of Invasives Applied in Phytoremediation

Classical biocontrol agents or mycoherbicides are known for biocontrol of the following water weeds (Barreto et al. 2000): Azolla xliculoides, Echinochloa polystachya, Eichhornia azurea, E. crassipes, Egeria densa, Myriophyllum aquaticum, Paspalum repens, Pistia stratiotes, Polygonum spectabile, Salvinia auriculata, S. molesta, and Typha domingensis.

A triad approach of lab, pilot, and field studies are necessary for understanding the limitations and scope of bioremediation potential of weeds (Figs. 23.10 and 23.11).

7 Conclusions and Future Perspective

Screening of weeds and ornamentals capable of accumulating and hyperaccumulating metals for bioremediation of metalliferous substrates in the era of climate change has sufficient scope. Weeds, ornamentals, and grasses possess such properties. Some of these are extensively adaptive in capacity. Compared with crop, they possess adaptive and antistress properties which make them exceptional to grow in metalliferous substrates. With these characteristics, it is possible that weeds exhibit strong tolerance and exceptional functions to heavy metals. Identification of appropriate soil amendments that can enhance biomass production need to be investigated. Selected examples of weeds that might be useful for polishing soils contami-



Fig. 23.10 The triad approach for successful bioremediation



Bioremediation potential of weeds

Fig. 23.11 Scope and limitations of weed application for bioremediation of heavy metals

nated with metals are *Solanum nigrum*, *Rorippa globosa*, *Bidens pilosa*, *Taraxacum mongolicum*, *Conyza canadensis*, and *Kalimeris integrifolia* (Wei and Zhou 2004a, b, 2008a, b; Wei et al. 2005, 2006, 2008a, b, 2009, 2010a, b). *Chromolaena odorata* (L) King & Robinson, (Asterceae), an invasive weed has the capability to phytoremediate soil contaminated with crude oil in the presence of metals (Atagana 2011; Tanhan et al. 2007). Critical functions of weeds for potential applications in cleanup of metalliferous substrates are:

- (a) Translocation property: the contents of heavy metals in shoots should be higher than those in its roots, i.e., TF (transport factor)>1.
- (b) Enrichment factor: the concentration ratio in plant shoots to soils should be higher than 1 (EF>1).
- (c) Use of soil amendments for enhancing the biomass production for removal of metals.

Several potential and promising options for this emerging technology are forging ahead for environmental management., yet certain bottlenecks are to be investigated for wider applications such as (a) many tested plants have bioconcentration factor (BCF) less than 1, (b) usage of soil amendments and chelators may be necessary for achieving the hyperaccumulation and BCF>10.

The regulatory bodies and environmental safety agencies are concerned about chelate and amendment assisted cleanup, as this approach would rapidly mobilize contaminant and increase the area of contamination including leaching of the toxic trace metals into ground water. Costs of the chelate applications need to be assessed. Arsenic hyperaccumulating fern, *Pteris vittata* (an important foliage ornamental), was used in the pilot-scale demonstration phytofiltration project to produce drinking quality water from arsenic contaminated ground water in New Mexico, a classic example of service to mankind (Elless et al. 2005).

The compost generated from the plants used in remediation serves as a compost and can be reused as growing media for production of ornamentals (Abad et al. 2001; Benito et al. 2006; Hernandez-Apaolaza et al. 2005; Hicklenton et al. 2001; Ingelmo et al. 1998) (Table 23.3). These questions have been satisfactorily answered for fostering phytoremediation using ornamentals. Phytoremediation technologies today have reached the site from lab to pilot-scale trials and field applications using aquatic, terrestrial (Prasad 2003, 2004a, b, 2007; Prasad and Freitas 2003) including space ecology (Kozyrovska et al. 2004, 2006). The debris generated from the ornamentals containing the toxic metal residues need to be treated as the biomass would be relatively less in view of high water contents and an appropriate

Table 23.3 Ornamentals for environmental cleanup and ecosystem service

1 2	
Ornamental for environmental moderation and remediation	References
<i>Calendula officinalis</i> and <i>Althaea rosea</i> exhibited higher tolerance to Cd and Pb contamination and could effectively accumulate these metals	Liu et al. (2007a, b)
Ni(II) biosorption by <i>Cassia fistula</i> (its common names are Amaltas, Canafistula, Golden Shower, and Indian Laburnum)	Hanif et al. (2007)
<i>Flindersia schottiana</i> is a tree species used in the ornamental horticulture industry. Urea formaldehyde resin foam (UFRF) product used as a soil amendment. It is proposed to improve the physicochemical properties (viz., water relations and aeration) of the plant root zone. UFRFs are a relatively new class of soil amendment compared with hydro gels. Under plant nursery conditions, incorporation of 30% (v/v) Hydrocell TM into composted pine bark media and also into sand and loam soils led to limited, but significant ($P \le 0.05$) growth benefits (e.g., increased leaflet number) for potted	Chan and Joyce (2007)
Biodegradable chelating agents, [<i>S</i> , <i>S</i>]-ethylenediaminedisuccinic acid (EDDS) and methylglycinediacetic acid (MGDA) assisted. Trace metals phytoextraction was demonstrated in <i>Mirabilis jalapa</i> including the growth of the associated bacterial population	Cao et al. (2007)
Phytoextraction trace metals with <i>Mirabilis jalapa</i> , combinatorial effect of biodegradable chelating agents and on its associated bacteria	Cao et al. (2007)
Crassula portulacea (Crassulaceae), Hydrangea macrophylla (Hydrangeaceae), Cymbidium Golden Elf (Orchidaceae), Ficus microcarpa var. fuyuensis (Moraceae), Dendranthema morifolium (Asteraceae), Citrus medica var. sarcodactylis (Rutaceae), Dieffenbachia amoena cv. Tropic Snow (Araceae), Spathiphyllum Supreme (Araceae), Nephrolepis exaltata cv. Bostoniensis (Davalliaceae), and Dracaena deremensis cv. Variegata (Dracaenaceae) had greatest capacity to remove benzene from indoor air	Liu et al. (2007a, b)
African marigold (<i>Tagetes erecta</i>), scarlet sage (<i>Salvia splendens</i>), and sweet hibiscus (<i>Abelmoschus manihot</i>) were investigated. According to the tolerant indexes, sweet hibiscus (<i>A. manihot</i>) was the most tolerance while scarlet sage (<i>S. splendens</i>) was the least and African marigold (<i>Tagetes erecta</i>) is in between	Wang and Zhou (2005)
Different compensatory mechanisms in two metal-accumulating aquatic macrophytes exposed to acute cadmium stress in outdoor artificial lakes	di Toppi et al. (2007)

(continued)

Table 23.3 (continued)

Ornamental for environmental moderation and remediation	References
<i>Ficus microcarpa</i> can provide useful information about the spatial variations of Ba, Cu, Fe, and Mg contents assuming the spatial differences are high enough. Temporal variations are evident for Al, Cu, Fe, Mg, Pb, V, and Zn. <i>F. microcarpa</i> foliage is not considered as a reliable biomonitor for Pb, Zn, and V in the urban areas	Oliva and Rautio (2005)
<i>Tagetes patula</i> and ornamental arum (<i>Syngonia</i> sp.) as phytoremediators of arsenic	Huq et al. (2005)
<i>Araucaria angustifolia</i> (Brazilian pine). The species is valuable for its wood, edible seeds, and ornamental use, and is today listed as a threatened species. This tree species establishes associations with arbuscular mycorrhizal fungus <i>Glomus clarum</i> , <i>Araucaria angustifolia</i> seedlings inoculated with <i>G. clarum</i> had a high degree of mycorrhizal colonization of their roots (81%). The inoculated seedlings grew significantly more (312% mass increase) than the controls	Zandavalli et al. (2004)
<i>Cyperus papyrus</i> and <i>Miscanthidium violaceum</i> -based constructed wetlands for wastewater treatment in a tropical country – a comparative study	Kyambadde et al. (2004)
Anorthosite is a poor support of the marigold growth, hence bacterial residents of alumino-silicate rocks to leach the plant essential ions from a substrate and therefore improved plant development and identified pioneer plants for a lunar base	Kozyrovska et al. (2004, 2006)
<i>Ipomoea aquatica</i> (water spinach, Morning glory, Convolvulace), is common in Southeast Asia. <i>I. aquatica</i> significantly accumulated toxic metals (such as cadmium, copper, and lead) in the roots, stems, and leaves	Costa-Pierce (1998), Kashem and Singh (2002), Gothberg et al. (2004)
Ornamental hydrophytes, viz., <i>Acorus gramineus</i> Soland and <i>Iris japonica</i> L., <i>Acorus calamus</i> L., <i>Lythrum salicaria</i> L., are suitable for sewage treatment. Performance of <i>Canna indica</i> on domestic sewage was better than <i>Phragmites communis</i>	Liu et al. (2003), Zhao et al. (2003)
Phosphate enhanced arsenic uptake by <i>Lolium perenne</i> , <i>Urtica dioica</i> , and <i>P. vittata</i> . Thus to alleviate arsenic toxicity, phosphate condition has to be managed	Otte et al. (1990), Cao et al. (2003)
Uptake and translocation of plutonium in two plant species using hydroponics. Comparative uptake of plutonium from soils by <i>Helianthus annuus</i>	Lee et al. (2002a, b)
Transgenic hairy roots in ornamentals: recent trends and applications	Giri and Narasu (2000)
Reed beds for water treatment	Lienard et al. (1995)
Nutrient removal from aquaculture wastewater using a constructed wetlands system	Lin et al. (2002)

techno-economic feasible options based on integrated model systems was recently suggested for the appropriate use of *E. crassipes* (water hyacinth) (Malik 2007). Similar solutions need to be worked out for the ornamental plants proposed for toxic metal cleanup, since each ornamental plant is a nonpolluting chemical factory producing a wide range of bioresource for the welfare of the mankind in addition to their ecosystem service, viz., environmental remediation and enhancing the beauty with esthetics and fragrance (Table 23.3).

References

- Abad M, Noguera P, Burés S (2001) National inventory of organic wastes for use as growing media for ornamental potted plant production: case study in Spain. Bioresour Technol 77:197–200
- Adeniyi AA (1996) Determination of cadmium, copper, iron, lead, manganese, and zinc in water leaf (*Talinum triangulare*) in dumpsites. Environ Int 22:259–262
- Alkorta I, Hernández-Allica J, Garbisu C (2004) Plants against the global epidemic of arsenic poisoning. Environ Int 30:949–951
- Anderson AC, Abdelghani AA, McDonell D (1980) Screening of four vascular aquatic plants for uptake of

monosodium methanearsonate (MSMA). Sci Total Environ 16:95–98

- Andre I, Schneider H, Rubio J (1999) Sorption of heavy metal ions by the non-living biomass of freshwater macrophytes. Environ Sci Technol 33:2213–2217
- Andres LA (1977) The economics of biological control of weeds. Aquat Bot 3:111–123
- Arora A, Saxena S, Sharma DK (2006) Tolerance and phytoaccumulation of chromium by three *Azolla* species. World J Microbiol Biotechnol 22:97–100
- Atabayeva S, Sarsenbayev B, Prasad MNV, Teixeira da Silva JA, Kenzhebayeva S, Usenbekov B, Kirshibayev Y, Asrandina S, Beisenova A, Danilova A, Kotuhov Y (2010) Accumulation of trace metals in grasses of Kazakhstan: relevance to phytostabilization of mine waste and metal-smelting areas. Asian Australian J Plant Sci Biotechnol 4:91–97
- Atagana HI (2011) Bioremediation of co-contamination of crude oil and heavy metals in soil by phytoremediation using Chromolaena odorata (L) King & H.E. Robinson. Water Air Soil Pollut 215:261–271
- Ater M, Aït Ali N, Kasmi H (2006) Tolerance and accumulation of copper and chromium in two duckweed species: *Lemna minor* L. and *Lemna gibba* L. Rev Sci Eau 19:57–67
- Axtell NR, Sternberg SPK, Claussen K (2003) Lead and nickel removal using *Microspora* and *Lemna*. Bioresour Technol 89:41–48
- Barreto R, Charudattan R, Pomella A, Hanada R (2000) Biological control of neotropical aquatic weeds with fungi. Crop Protect 19:697–703
- Belmont MA, Metcalfe CD (2003) Feasibility of using ornamental plants (*Zantedeschia aethiopica*) in subsurface flow treatment wetlands to remove nitrogen, chemical oxygen demand and nonylphenol ethoxylate surfactants – a laboratory-scale study. Ecol Eng 21:233–247
- Benaroya RO, Tzin V, Tel-Or E, Zamski E (2004) Lead accumulation in the aquatic fern Azolla filiculoides. Plant Physiol Biochem 42:639–645
- Benito M, Masaguer A, Moliner A, Antonio RD (2006) Chemical and physical properties of pruning waste compost and their seasonal variability. Bioresour Technol 97:2071–2076
- Bennett FD (1977) Insects as agents for biological control of aquatic weeds. Aquat Bot 3:165–174
- Bennicelli R, Stepniewska Z, Banach A, Szajnocha K, Ostrowski J (2004) The ability of Azolla caroliniana to remove heavy metals such as Hg²⁺, Cr³⁺, Cr⁶⁺ from municipal wastewater. Chemosphere 55:141–146
- Boonyapookana B, Upatham ES, Kruatrachue M, PokethitiyookP,SinghakaewS(2002)Phytoaccumulation and phytotoxicity of Cd and Cr in duckweed *Wolffia globosa*. Int J Phytoremediation 4:87–100
- Cambrollé J, Redondo-Gómez S, Mateos-Naranjo E, Figueroa M (2008) Comparison of the role of two *Spartina* species in terms of phytostabilization and bioaccumulation of metals in the estuarine sediment. Mar Pollut Bull 56:2037–2042

- Cao X, Ma LQ, Shiralipour A (2003) Effects of compost and phosphate amendments on arsenic mobility in soils and arsenic uptake by the hyperaccumulator *Pteris vittata* L. Environ Pollut 126:157–167
- Cao JS, Li YR, Chen J (2006) Purification of seriously polluted river by *Ipomoea aquatica* and its allelopathic effect on algae. Water Resour Protect 2:36–41 (in Chinese)
- Cao A, Carucci A, Lai T, Colla PL, Tamburini E (2007) Effect of biodegradable chelating agents on trace metals phytoextraction with *Mirabilis jalapa* and on its associated bacteria. Eur J Soil Biol 43:200–206
- Carbonell-Barrachina AA, Arabi MA, DeLane RD, Gambrell RP, Patrick WH Jr (1998) Arsenic in wetland vegetation: availability, phytotoxicity, uptake and effects on plant growth and nutrition. Sci Total Environ 217:189–199
- Chan CL, Joyce DC (2007) Effects of urea formaldehyde foam soil amendment on growth and response to transient water deficit stress of potted Flindersia schottiana saplings. Sci Hortic 114:112–120
- Chen Y, Shen Z, Li Z (2004a) The use of vetiver grass (*Vetiveria zizanioides*) in the phytoremediation of soils contaminated with trace metals. Appl Geochem 19:1553–1565
- Chen Y, Li X, Shen Z (2004b) Leaching and uptake of trace metals by ten different species of plants during an EDTA-assisted phytoextraction process. Chemosphere 57:187–196
- Chen B, Roos P, Zhu YG, Jakobsen I (2008) Arbuscular mycorrhizas contribute to phytostabilization of uranium in uranium mining tailings. J Environ Radioact 99:801–810
- Cheng S (2003) Trace metal pollution in China: origin, pattern and control. Environ Sci Pollut Res Int 10:192–198
- Chintakovid W, Visoottiviseth P, Khokiattiwong S, Lauengsuchonkul S (2007) Potential of the hybrid marigolds for arsenic phytoremediation and income generation of remediators in Ron Phibun District, Thailand. Chemosphere 70:1532–1537
- Chiu KK, Ye ZH, Wong MH (2005) Enhanced uptake of As, Zn, and Cu by Vetiveria zizanioides and Zea mays using chelating agents. Chemosphere 60:1365–1375
- Chiu KK, Ye ZH, Wong MH (2006) Growth of Vetiveria zizanioides and Phragmities australis on Pb/Zn and Cu mine tailings amended with manure compost and sewage sludge: a greenhouse study. Bioresour Technol 97:158–170
- Chong CW, Chu LM (2007) Growth of vetivergrass for cutslope landscaping: effects of container size and watering rate. Urban Forestry Urban Greening 6: 135–141
- Comino E, Fiorucci A, Menegatti S, Marocco C (2009) Preliminary test of arsenic and mercury uptake by *Poa* annua. Ecol Eng 35:343–350
- Costa-Pierce BA (1998) Preliminary investigation of an integrated aquaculture–wetland ecosystem using tertiary-treated municipal wastewater in Los Angeles County, California. Ecol Eng 10:341–354

- Czako M, Feng X, He Y (2005) Genetic modification of wetland grasses for phytoremediation. Z Naturforsch 60:285–291
- Dan TV, Raj KS, Saxena PK (2000) Metal tolerance of scented geranium (*Pelargonium* sp. 'Frensham'): Effects of cadmium and nickel on chlorophyll fluorescence kinetics. Int J Phytoremediation 2:91–104
- Davies FT Jr, Puryear JD, Newton RJ, Egilla JN, Grossi JAS (2001) Mycorrhizal fungi enhance accumulation and tolerance of chromium in sunflower (*Helianthus annuus* L.). J Plant Physiol 158:777–786
- Deng H, Ye ZH, Wong MH (2004) Accumulation of lead, zinc, copper and cadmium by 12 wetland plants species thriving in metal contaminated sites in China. Environ Pollut 132:29–40
- Deng H, Ye ZH, Wong MH (2006) Lead and zinc accumulation and tolerance in populations of six wetland plants. Environ Pollut 141:69–80
- Deram A, Denayer FO, Dubourgier HC, Douay F, Petit D, Van Haluwyn C (2007) Zinc and cadmium accumulation among and within populations of the pseudometalophytic species *Arrhenatherum elatius*: implications for phytoextraction. Sci Total Environ 372:372–381
- Deram A, Languereau-Leman F, Howsam M, Petit D, Haluwyn CV (2008) Seasonal patterns of cadmium accumulation in *Arrhenatherum elatius* (Poaceae): influence of mycorrhizal and endophytic fungal colonization. Soil Biol Biochem 40:845–848
- di Toppi LS, Vurro E, Rossi L, Marabottini R, Musetti R, Careri M, Maffini M, Mucchino C, Corradini C, Badiani M (2007) Different compensatory mechanisms in two metal-accumulating aquatic macrophytes exposed to acute cadmium stress in outdoor artificial lakes. Chemosphere 68:769–780
- Ding B, Shi G, Xu Y, Hu J, Xu Q (2007) Physiological responses of Alternanthera philoxeroides (Mart.) Griseb leaves to cadmium stress. Environ Pollut 147:800–803
- Dowty RA, Shaffer GP, Hester MW, Childers GW, Campo FM, Green MC (2001) Phytoremediation of smallscale oil spills in fresh marsh environments: a mesocosm simulation. Mar Environ Res 52:195–211
- Dushenkov S, Kapulnik Y (2002) Phytofiltration of metals. In: Raskin I, Ensley BD (eds) Phytoremediation of toxic metals: using plants to clean up the environment. Wiley, New York, pp 89–106
- Dushenkov V, Nanda PBAN, Motto H, Raskin I (1995) Rhizofiltration: the use of plants to remove trace metals from aqueous streams. Environ Sci Technol 29:1239–1245
- Dushenkov S, Kapulnik Y, Blaylock M, Sorochisky B, Raskin I, Ensley B (1997a) Phytoremediation: a novel approach to an old problem. In: Wise DL (ed) Global environmental biotechnology. Elsevier Science B.V, Amsterdam, pp 563–572
- Dushenkov S, Vasudev D, Kapulnik Y, Gleba D, Fleisher D, Ting KC, Ensley B (1997b) Removal of uranium from water using terrestrial plants. Environ Sci Technol 31:3468–3474

- Ederli L, Reale L, Ferranti F, Pasqualini S (2004) Responses induced by high concentration of cadmium in *Phragmites australis* roots. Physiol Plant 121: 66–74
- Elless MP, Poynton CY, Willms CA, Doyle MP, Lopez AC, Sokkary DA, Ferguson BW, Blaylock MJ (2005) Pilot-scale demonstration of phytofiltration for treatment of arsenic in New Mexico drinking water. Water Res 39:3863–3872
- Garbisu C, Alkorta I (2001) Phytoextraction: a cost-effective plant-based technology for the removal of metals from the environment. Bioresour Technol 77:229–236
- Garbisu C, Hernandez-Allica J, Barrutia O, Alkorta I, Becerril JM (2002) Phytoremediation: a technology using green plants to remove contaminants from polluted areas. Rev Environ Health 17:173–188
- Gardea-Torresdey JL, Peralta-Videa JRP, de la Rosa G, Parsons JG (2005) Phytoremediation of heavy metals and study of the metal coordination by X-ray absorption spectroscopy. Coord Chem Rev 249:1797–1810
- Geng YP, Pan XX, Xu CY, Zhang WJ, Li B, Chen JK (2006) Phenotypic plasticity of invasive Alternanthera philoxeroides in relation to different water availability, compared to its native congener. Acta Oecol 30:380–385
- Giovanni A, Pecchioni N, Rabaglio M, Allavena A (1997) Characterization of ornamental *Datura* plants transformed by *Agrobacterium rhizogenes*. In Vitro Cell Dev Biol Plant 33:101–106
- Giri A, Narasu ML (2000) Transgenic hairy roots: recent trends and applications. Biotechnol Adv 18:1–22
- Gothberg A, Greger M, Bengtsson BE (2002) Accumulation of heavy metals in water spinach (*Ipomoea aquatica*) cultivated in the Bangkok region, Thailand. Environ Toxicol Chem 21:1934–1939
- Gothberg A, Greger M, Holm K, Bengtsson BE (2004) Influence of nutrient levels on uptake and effects of mercury, cadmium, and lead in water spinach. J Environ Qual 33:1247–1255
- Gratão PL, Prasad MNV, Lea PJ, Azevedo RA (2005) Phytoremediation: green technology for the clean up of toxic metals in the environment. Braz J Plant Physiol 17:53–64
- Hartley W, Lepp NW (2008) Effect of in situ soil amendments on arsenic uptake in successive harvests of ryegrass (*Lolium perenne* cv Elka) grown in amended As-polluted soils. Environ Pollut 156:1030–1040
- Hartley W, Dickinson NM, Clemente R, French C, Piearce TG, Sparke S, Lepp NW (2009) Arsenic stability and mobilization in soil at an amenity grassland overlying chemical waste (St. Helens, UK). Environ Pollut 157: 847–856
- Hanif MA, Nadeema R, Bhatti HN, Ahmada NR, Ansari TM (2007) Ni(II) biosorption by *Cassia fistula* (Golden Shower) biomass. J Hazard Mater B139:345–355
- Hattink J, de Goeij JJM, Wolterbeek HT (2000) Uptake kinetics of ⁹⁹Tc in common duckweed. Environ Exp Bot 44:9–22

- Hernandez-Apaolaza L, Gasco AM, Gasco JM, Guerrero F (2005) Reuse of waste materials as growing media for ornamental plants. Bioresour Technol 96:125–131
- Hicklenton PR, Philip VR, Warman R (2001) The effectiveness and consistency of source-separated municipal solid waste and bark composts as components of container growing media. Sci Hortic 91:365–378
- Hoffmann T, Kutter C, Santamria JM (2004) Capacity of Salvinia minima Baker to tolerate and accumulate As and Pb. Eng Life Sci 4:61–65
- Horne AJP (2000) Phytoremediation by constructed wetlands. In: Raskin I, Ensley BD (eds) Phytoremediation of toxic metals: using plants to clean up the environment. Wiley, New York, pp 13–39
- Hu C, Zhang L, Hamilton D, Zhou W, Yang T, Zhu D (2007) Physiological responses induced by copper accumulation in *Eichhornia crassipes*. Hydrobiologia 579:211–218
- Huq SM, Joardar JC, Parvin S (2005) (*Tagetes patula*) and ornamental arum (*Syngonia* sp.) as phytoremediators for arsenic in pot soil. Bangladesh J Bot 34:65–70
- Ingelmo F, Canet R, Ibañez MA, García FPJ (1998) Use of MSW compost, dried sewage sludge and other wastes as partial substitutes for peat and soil. Bioresour Technol 63:23–129
- Ingole NW, Bhole AG (2003) Removal of heavy metals from aqueous solution by water hyacinth (*Eichhornia crassipes*). J Water Supply Res Technol Aqua 52:119–128
- Ji P, Sun T, Song Y, Ackland ML, Liu Y (2011) Strategies for enhancing the phytoremediation of cadmiumcontaminated agricultural soils by *Solanum nigrum* L. Environ Pollut 159:762–768
- Kadlec RH, Knight RL (1996) Treatment Wetlands. Lewis Publishers, Boca Raton, 893p
- Kailappan R, Gothandapani L, Viswanathan R (2000) Production of activated carbon from prosopis (*Prosopis juliflora*), Bioresour Technol 75:241–243
- Kamal M, Ghaly AE, Mahmoud N, Cote R (2004) Phytoaccumulation of heavy metals by aquatic plants. Environ Int 29:1029–1039
- Kara Y (2004) Bioaccmulation of copper from contaminated wastewaters by using *Lemna minor* (aquatic green plants). Bull Environ Contam Toxicol 72:467–471
- Kashem MA, Singh BR (2002) The effect of fertilizer additions on the solubility and plant-availability of Cd, Ni and Zn in soil. Nutr Cycl Agroecosyst 62:287–296
- Keskinkan O, Goksu MZL, Basibuyuk M, Forster CF (2004) Heavy metal adsorption properties of a submerged aquatic plant (*Ceratophyllum demersum*). Bioresour Technol 92:197–200
- Kozyrovska NO, Korniichuk OS, Voznyuk TM (2004) Microbial community in a precursory scenario of growing *Tagetes patula* in a lunar greenhouse. Space Sci Technol 10:221–225
- Kozyrovska NO, Lutvynenko TL, Korniichuk OS, Kovalchuk MV, Voznyuk TM, Kononuchenko O, Zaetz I, Rogutskyy IS, Mytrokhyn OV, Mashkovska

SP, Foing BH, Kordyum VA (2006) Growing pioneer plants for a lunar base. Adv Space Res 37:93–99

- Kuldau G, Bacon C (2008) Clavicipitaceous endophytes: their ability to enhance resistance of grasses to multiple stresses. Biol Control 46:57–71
- Kyambadde J, Kansiime F, Gumaelius L, Dalhammar G (2004) A comparative study of *Cyperus papyrus* and *Miscanthidium violaceum*-based constructed wetlands for waste water treatment in a tropical country. Water Res 38:475–485
- Lai H, Chen Z (2006) The influence of EDTA application on the interactions of cadmium, zinc, and lead and their uptake of rainbow pink (*Dianthus chinensis*).
 J Hazard Mater 137:1710–1718
- Lakshman G (1987) Ecotechnological opportunities for aquatic plants, a survey of utilization options. In: Reddy KR, Smith WH (eds) Aquatic plants for water treatment and resource recovery. Magnolia Publishing Inc, Orlando, FL, pp 49–68
- Lee JH, Hossner LR, Attrep M, Kung KS Jr (2002a) Comparative uptake of plutonium from soils by *Brassica juncea* and *Helianthus annuus*. Environ Pollut 120:173–182
- Lee JH, Hossner LR, Attrep M, Kung KS Jr (2002b) Uptake and translocation of plutonium in two plant species using hydroponics. Environ Pollut 117: 61–68
- Li J, Ye WH (2006) Genetic diversity of alligator weed ecotypes is not the reason for their different responses to biological control. Aquat Bot 85:155–158
- Li T, Yang X, Lu L, Islam E, He Z (2009) Effects of zinc and cadmium interactions on root morphology and metal translocation in a hyperaccumulating species under hydroponic conditions. J Hazard Mater 169:734–741
- Lienard A, Duchene P, Gorini D (1995) A study of activated sludge dewatering in experimental reed-planted or unplanted sludge drying beds. Water Sci Technol 32:251–261
- Lin JW, Liu JD (2003) Accumulation of copper by roots, hypocotyls, cotyledons and leaves of sunflower (*Helianthus annuus* L). Bioresour Technol 86:151–155
- Lin YF, Jing SR, Lee DY, Wang TW (2002) Nutrient removal from aquaculture wastewater using a constructed wetlands system. Aquaculture 209: 169–184
- Liuphadzi MS, Kirkham MB, Mankin KR, Paulsen GM (2003) EDTA-assited trace-metal uptake by popular and sunflower grow at a long-term sewage-sludge farm. Plant Soil 257:171–182
- Liu WH, Zhao JZ, Ouyang ZY (2005) Impacts of sewage irrigation on trace metal distribution and contamination in Beijing China. Environ Int 31:805–812
- Liu J, Dong Y, Xu H, Wang D, Xu J (2007a) Accumulation of Cd, Pb and Zn by 19 wetland plant species in constructed wetland. J Hazard Mater 147:947–953
- Liu YJ, Zhou QX, Sun T, Ma LQ, Wang S (2007b) Growth responses of three ornamental plants to Cd and Cd–Pb stress and their metal accumulation characteristics. J Hazard Mater 151:261–267

- Liu X, Shen Y, Lou L, Ding C, Cai Q (2009) Copper tolerance of the biomass crops Elephant grass (*Pennisetum purpureum* Schumach), Vetiver grass (*Vetiveria zizanioides*) and the upland reed (*Phragmites australis*) in soil culture. Biotechnol Adv 27:633–640
- Lorenzini G, Nali GC, Petiti C, Loppi A, Tognotti S (2006) Leaves of *Pittosporum tobira* as indicators of airborne trace element and PM10 distribution in central Italy. Atmos Environ 40:4025–4036
- Ma LQ, Komar KMM, Tu C, Zhang W, Cai Y, Kennelley ED (2001) A fern that hyperaccumulates arsenic. Nature (London) 409:579
- Ma Y, Prasad MNV, Rajkumar M, Freitas H (2011) Plant growth promoting rhizobacteria and endophytes accelerate phytoremediation of metalliferous soils. Biotechnol Adv 29:248–258
- Macek T, Francova K, Kochankova L, Lovecka P, Ryslava E, Rezek J, Sura M, Triska J, Demnerova K, Mackova M (2004) Phytoremediation: biological cleaning of a polluted environment. Rev Environ Health 19:63–82
- Madejon P, Murillo JM, Maranon T, Cabrera F, Soriano MA (2003) Trace element and nutrient accumulation in sunflower plants two years after the Aznalcollar mine spill. Sci Total Environ 307:239–257
- Makris KC, Shakya KM, Datta R, Sarkar D, Pachanoor D (2007) High uptake of 2,4,6-trinitrotoluene by vetiver grass – potential for phytoremediation? Environ Pollut 146:1–4
- Malik A (2007) Environmental challenge vis a vis opportunity: the case of water hyacinth. Environ Int 33:122–138
- Mant C, Costa S, Williams J, Tambourgi E (2006) Phytoremediation of chromium by model constructed wetland. Bioresour Technol 97:1767–1772
- Mateos-Naranjo E, Redondo-Gómez S, Cambrollé J, Figueroa ME (2008) Growth and photosynthetic responses to copper stress of an invasive cordgrass, *Spartina densiflora*. Mar Environ Res 66:459–465
- Mays PA, Edwards GS (2001) Comparison of heavy metal accumulation in a natural wetland and constructed wetlands receiving acid mine drainage. Ecol Eng 16:487–500
- Mazen AMA (2004) Accumulation of four metals in tissues of *Corchorus olitorius* and possible mechanisms of their tolerance. Biol Plant 48:267–272
- McCutcheon SC, Schnoor JL (2003) Phytoremediationtransformation and control of contaminants. Wiley Interscience, Hoboken, NJ, 985pp
- McIntyre T (2003) Phytoremediation of trace metals from soils. Adv Biochem Eng Biotechnol 78:97–123
- Meharg AA (2004) Arsenic in rice-understanding a new disaster for south-east Asia. Trends Plant Sci 9: 415–417
- Mkandawire M, Dudel EG (2005) Accumulation of As in Lemna gibba (duckweed) in tailing waters of two abandoned Uranium mining sites in Saxony, Germany. Sci Total Environ 336:81–89
- Mkandwire M, Taubert B, Dudel EG (2004) Capacity of *Lemna gibba* (duckweed) for uranium and arsenic phyoremediation in mine tailing waters. Int J Phytoremediation 14(66):347–362

- Mungur AS, Shutes RBE, Revitt DM, House MA (1995) An assessment of metal removal from highway runoff by a natural wetland. Water Sci Technol 32:169–175
- Murillo JM, Maranon T, Cabrera F, Lopez R (1999) Accumulation of trace metals in sunflower and sorghum plants affected by the Guadiamar spill. Sci Total Environ 242:281–292
- Nagaraju A, Prasad KSS (1998) Growth of *Prosopis juliflora* on pegmatite tailings from Nellore Mica Belt, Andhra Pradesh, India. Environ Geol 36:320–324
- Naqvi SM, Rizvi SA (2000) Accumulation of chromium and copper in three different soils and bioaccumulation in an aquatic plant, *Alternanthera philoxeroides*. Bull Environ Contam Toxicol 65:55–61
- Néel C, Soubrand-Colin M, Piquet-Pissaloux A, Bril H (2007) Mobility and bioavailability of Cr, Cu, Ni, Pb and Zn in a basaltic grassland: comparison of selective extractions with quantitative approaches at different scales. Appl Geochem 22:724–735
- Negri MC, Hinchman RR (2000) The use of plants for the treatment of radionuclides. In: Raskin I, Ensley BD (eds) Phytoremediation of toxic metals: using plants to clean up the environment. Wiley, New York, pp 107–132
- Niu ZX, Sun LN, Sun TH, Li YS, Wang H (2007) Evaluation of phytoextracting cadmium and lead by sunflower, ricinus, alfalfa and mustard in hydroponic culture. J Environ Sci 19:961–967
- Nyquist J, Greger M (2007) Uptake of Zn, Cu, and Cd in metal loaded *Elodea Canadensis*. Environ Exp Bot 60:219–226
- Odjegba VJ, Fasidi IO (2004) Accumulation of trace elements by *Pistia stratiotes*. Implications for phytoremediation. Ecotoxicology 13:637–642
- Okurut TO, Rijs GBJ, Van Bruggen JJA (1999) Design and performance of experimental constructed wetlands in Unganda, planted with *Cyperus papyrus* and *Phragmites mauritianus*. Water Sci Technol 40:265–271
- Oliva SR, Rautio P (2005) Spatiotemporal patterns in foliar element concentrations in *Ficus microcarpa* L. growing in an urban area: implications for biomonitoring studies. Ecol Indicat 5:97–107
- Ortega-Larrocea MP, Xoconostle-Cázares B, Maldonado-Mendoza IE, Carrillo-González R, Hernández-Hernández J, Garduño MD, López-Meyer M, Gómez-Flores L, González-Chávez MCA (2010) Plant and fungal biodiversity from metal mine wastes under remediation at Zimapan, Hidalgo, Mexico. Environ Pollut 158:1922–1931
- Otte ML, Rozema J, Beek MA, Kater BJ, Broekman RA (1990) Uptake of arsenic by estuarine plants and interactions with phosphate, in the field (Rhine estuary) and under outdoor experimental conditions. Sci Total Environ 98:839–854
- Outridge P, Noller B (1991) Accumulation of toxic trace elements by freshwater vascular plants. Rev Environ Contam Toxicol 121:2–63
- Pan X, Geng Y, Zhang W, Li B, Chen J (2006) The influence of abiotic stress and phenotypic plasticity on thedistribution of invasive *Alternanthera philoxeroides* along a riparian zone. Acta Oecol 30:333–341

- Pan J, Sun H, Nduwimana A, Wang Y, Zhou G, Ying Y, Zhang R (2007) Hydroponic plate/fabric/grass system for treatment of aquacultural wastewater. Aquacult Eng 37:266–273
- Pang J, Chan GSY, Zhang J, Liang J, Wong MH (2003) Physiological aspects of vetiver grass for rehabilitation in abandoned metalliferous mine wastes. Chemosphere 52:1559–1570
- Peles JD, Smith MH, Brisbin Jr IL (2002) Ecological half-life of ¹³⁷Cs in plants associated with a contaminated stream. J Environ Radioact 59:169–178
- Pellegrineschi A, Damon JP, Valtorta N, Paillard N, Tepfer D (1994) Improvement of ornamental characters and fragrance production in lemon scented geranium through genetic transformation by Agrobacterium rhizogenes. Biotechnology 12:64–68
- Peng H, Qu C, Yang Z, Liu Y (1997) Studies on antiviral effect of extract from *Alternanthera philoxeroides* Griseb on epidemic hemorrhagic fever virus in vivo: 3. Antiviral Res 34:A90
- Prasad MNV (2001) Bioremediation potential of Amaranthaceae. In: Leeson A, Foote EA, Banks MK, Magar VS (eds) Phytoremediation, wetlands, and sediments: proceedings, 6th international in situ and on-site bioremediation symposium. Battelle, Columbus, OH, pp 165–172
- Prasad MNV (2003) Phytoremediation of metal-polluted ecosystems: hype for commercialization. Russ J Plant Physiol 50:686–701
- Prasad MNV (2004a) Phytoremediation of metals and radionuclides in the environment: the case for natural hyperaccumulators, metal transporters, soil amending chelators and transgenic plants. In: Prasad MNV (ed) Heavy metal stress in plants: from biomolecules to ecosystems, 2nd edn. Springer, Heidelberg, pp 345–392
- Prasad MNV (2004b) Phytoremediation of metals in the environment for sustainable development. Proc Indian Natl Sci Acad 70:71–98
- Prasad MNV (2006) Stabilization, remediation and integrated management of metal-contaminated ecosystems by grasses (Poaceae). In: Prasad MNV, Sajwan KS, Naidu R (eds) Trace elements in the environment: biogeochemistry, biotechnology and bioremediation. CRC Press, Boca Raton, FL, pp 405–424
- Prasad MNV (2007) Phytoremediation in India. In: Neil Willey (ed) Phytoremediation, methods and reviews. Humana Press, Totowa, NJ
- Prasad MNV, Freitas H (2003) Metal hyperaccumulation in plants – biodiversity prospecting for phytoremediation technology. Elect J Biotechnol 6:275–321
- Punamiya P, Datta R, Sarkar D, Barber S, Patel M, Das P (2010) Symbiotic role of glomus mosseae in phytoextraction of lead in Vetiver Grass [*Chrysopogon zizanioides* (L.)]. J Hazard Mater 177: 465–474
- Rahman MA, Hasegawa H (2011) Aquatic arsenic: Phytoremediation using floating macrophytes. Chemosphere 83:633–646
- Rai UN, Singh S, Tripathi RD, Chandra P (1995) Wastewater treat ability of potential of some aquatic

macrophtyes: removal of heavy metals. Ecol Eng 5:5-12

- Rai UN, Tripathi RD, Vajpayee P, Pandey N, Ali MB, Gupta DK (2003) Cadmium accumulation and its phytotoxicity in *Potamogeton pectinatus* (Potamogetonaceae). Bull Environ Contam Toxicol 70:566–575
- Redondo-Gómez S, Mateos-Naranjo E, Vecino-Bueno I, Feldman SR (2011) Accumulation and tolerance characteristics of chromium in a cordgrass Cr-hyperaccumulator, *Spartina argentinensis*. J Hazard Mater 185:862–869
- Rotkittikhun P, Chaiyarat R, Kruatrachue M, Pokethitiyook P, Baker AJM (2007) Growth and lead accumulation by the grasses *Vetiveria zizanioides* and *Thysanolaena maxima* in lead-contaminated soil amended with pig manure and fertilizer: a glasshouse study. Chemosphere 66:45–53
- Santibáñez C, Verdugo C, Ginocchio R (2008) Phytostabilization of copper mine tailings with biosolids: implications for metal uptake and productivity of *Lolium perenne*. Sci Total Environ 395:1–10
- Schooler S, Baron Z, Julien M (2006) Effect of simulated and actual herbivory on alligator weed *Alternanthera philoxeroides*, growth and reproduction. Biol Control 36:74–79
- Schooler S, Yeates AG, Wilson JRU, Julien MH (2007) Herbivory, mowing, and herbicides differently affect production and nutrient allocation of *Alternanthera philoxeroides*. Aquat Bot 86:62–68
- Senthilkumar P, Prince WSPM, Sivakumar S, Subbhuraam CV (2005) *Prosopis juliflora* – a green solution to decontaminate heavy metal (Cu and Cd) contaminated soils. Chemosphere 60:1493–1496
- Sheppard SC, Motycka M (1997) Is the akagare phenomenon important to iodine uptake by wild rice (*Zizania aquatica*)? J Environ Radioact 37:339–353
- Shu WS, Ye ZH, Lan CY, Zhang ZQ, Wong MH (2002) Lead, zinc and copper accumulation and tolerance in populations of *Paspalum distichum* and *Cynodon dactylon*. Environ Pollut 120:445–453
- Siddhu OP, Behl HM (1997) Response of three Glomus species on growth of *Prosopis juliflora* Swartz at high pH levels. Symbiosis 23:23–34
- Singh G (1995) An agroforestry practice for the development of salt lands using *Prosopis juliflora* and *Leptochloa fusca*. Agroforestry Systems 29:61–75
- Singh S, Eapen S, Thorat V, Kaushik CP, Raj K, D'Souza SF (2008) Phytoremediation of ¹³⁷cesium and ⁹⁰strontium from solutions and low level nuclear waste by *Vetiveria zizanoides*. Ecotoxicol Environ Saf 69: 306–311
- Sinha S, Saxena R, Singh S (2002) Comparative studies on accumulation of chromium from metal solution and tannery effluent under repeated metal exposure by aquatic plants: its toxic effects. Environ Monit Assess 80:17–31
- Sobolewski A (1999) A review of processes responsible for metal removal in wetlands treating contaminated mine drainage. Int J Phytoremediation 1:19–51

- Song L, Chen W, Peng L, Wan N, Gan N, Zhang X (2007) Distribution and bioaccumulation of microcystins in water columns: a systematic investigation into the environmental fate and the risks associated with microcystins in Meiliang Bay Lake Taihu. Water Res 41:2853–2864
- Spencer NR, Coulson JR (1976) The biological control of alligator weed, *Alternanthera philoxeroides*, in the United States of America. Aquat Bot 2:177–190
- Stepniewska Z, Bennicelli RP, Balakhnina TI, Szajnocha K, Banach A, Wolinska A (2005) Potential of Azolla caroliniana for the removal of Pb and Cd from wastewaters. Int Agrophys 19:251–255
- Tanhan P, Kruatrachue M, Pokethitiyook P, Chaiyarat R. 2007. Uptake and accumulation of cadmium, lead and zinc by Siam weed [*Chromolaena odorata* (L.) King & Robinson] Chemosphere 68:323–329
- Tiagi YD, Aery NC (1986) Biogeochemical studies at the Khetri copper deposits of Rajasthan, India. J Geochem Explor 26:267–274
- Tordoff GM, Baker AJM, Willis AJ (2000) Current approaches to the revegetation and reclamation of metalliferous mine wastes. Chemosphere 41:219–228
- Truong P (2000) Vetiver grass technology for environmental protection. In: Vetiver and the environment, the second international vetiver conference. China, America, Thailand
- Upatham ES, Boonyapookana B, Kruatrachue M, Pokethitiyook P, Parkpoomkamol K (2002) Biosorption of cadmium and chromium in duckweed *Wolffia globosa*. Int J Phytoremediation 4:73–86
- Varun M, D'Souza R, Pratas J, Paul MS (2011) Phytoextraction potential of Prosopis juliflora (Sw.) DC. with specific reference to lead and cadmium. Bull Environ Contam Toxicol 87:45–49
- Vernay P, Gauthier-Moussard C, Hitmi A (2007) Interaction of bioaccumulation of trace metal chromium with water relation, mineral nutrition and photosynthesis in developed leaves of *Lolium perenne* L. Chemosphere 68:1563–1575
- Wang XS, Qin Y (2006) Removal of Ni(II), Zn(II) and Cr(VI) from aqueous solution by *Alternanthera philoxeroides* biomass. J Hazard Mater 138:582–588
- Wang XF, Zhou QX (2005) Ecotoxicological effects of cadmium on three ornamental plants. Chemosphere 60:16–21
- Wang B, Li W, Wang J (2005) Genetic diversity of Alternanthera philoxeroides in China. Aquat Bot 81:277–283
- Wei S, Zhou QX (2004a) Identification of weed species with hyper-accumulative characteristics of heavy metals. Progress Nat Sci 14:495–503
- Wei S, Zhou QX (2004b) Identification of weed species with hyper-accumulative characteristics of heavy metals. Progress Nat Sci 14:495–503
- Wei S, Zhou Q (2008a) Screen of Chinese weed species for cadmium tolerance and accumulation characteristics. Int J Phytoremediation 10:584–597
- Wei S, Zhou Q (2008b) Screen of Chinese weed species for cadmium tolerance and accumulation characteristics. Int J Phytoremediation 10:584–597

- Wei S, Zhou QX, Liu R (2005) Utilization of weed resource in the remediation of soils contaminated by heavy metals. J Nat Res 20:432–440
- Wei S, Zhou QX, Koval PV (2006) Flowering stage characteristics of cadmium hyperaccumulator *Solanum nigrum* L. and their significance to phytoremediation. Sci Total Environ 369:441–446
- Wei S, Zhou Q, Saha UK (2008a) Hyperaccumulative characteristics of weed species to heavy metals. Water Air Soil Pollut 192:173–181
- Wei S, Zhou Q, Saha UK (2008b) Hyperaccumulative characteristics of weed species to heavy metals. Water Air Soil Pollut 192:173–181
- Wei S, Qixing Z, Saha UK, Xiao H, Hu Y, Ren L, Ping G (2009) Identification of a Cd accumulator *Conyza* canadensis. J Hazard Mater 163:32–35
- Wei S, Zhou Q, Zhan J, Wu Z, Sun T, Lyubu Y, Prasad MNV (2010a) Poultry manured *Bidens tripartite* L. extracting Cd from soil-potential for phytoremediating Cd contaminated soil. Bioresour Technol 101:8907–8910
- Wei S, Wang S, Zhou Q, Zhan J, Ma L, Wu Z, Sun T, Prasad MNV (2010b) Potential of *Taraxacum mon-golicum* Hand-Mazz for accelerating phytoextraction of cadmium in combination with eco-friendly amendments. J Hazard Mater 181:480–484
- Wei S, Zhou Q, Zhan J, Wu Z, Sun T, Lyubu Y, Prasad MNV (2010c) Poultry manured *Bidens tripartite* L. extracting Cd from soil-potential for phytoremediating Cd contaminated soil. Bioresour Technol 101: 8907–8910
- Weis JS, Weis P (2004) Metal uptake, transport and release by wetland plants: implications for phytoremediation and restoration. Environ Int 30:685–700
- Wie S, Qixing Z, Saha UK, Xiao H, Hu Y, Ren L, Ping G (2009) Identification of a Cd accumulator *Conyza* canadensis. J Hazard Mater 163:32–35
- Wilde EW, Brigmon RL, Dunn DL, Heitkamp MA, Dagnan DC (2005) Phytoextraction of lead from firing range soil by vetiver grass. Chemosphere 61:1451–1457
- Williams JB (2002) Phytoremediation in wetland ecosystems: progress, problems, and potential. Crit Rev Plant Sci 21:607–635
- Wilson JRU, Yeates A, Schooler S, Julien MH (2007) Rapid response to shoot removal by the invasive wetland plant, alligator weed (*Alternanthera philoxeroides*) Environ Exp Bot 60: 20–25
- Windham L, Weis JS, Weis P (2003) Uptake and distribution of metals in two dominant salt marsh macrophytes, *Spartina alterniflora* (cordgrass) and *Phragmites australis* (common reed). Estuar Coast Shelf Sci 56:63–72
- Wolverton BC, McDonlad RC (1975a) Water hyacinths and alligator weeds for removal of lead and mercury from polluted waters. NASA Tech. Memo. No. TM-X-71723. Bay St. Louis, MO, National Space Technology Lab
- Wolverton BC, McDonlad RC (1975b) Water hyacinths and alligator weeds for removal of silver, cobalt and strontium from polluted waters. NASA Tech. Memo. No. TM-X-71723. Bay St. Louis, MO, National Space Technology Lab

- Wood B, Mcatamney C (1994) The use of macrophytes in bioremediation. Biotechnol Adv 12:653–662
- Woulds C, Ngwenya BT (2004) Geochemical processes governing the performance of a constructed wetland treating acid mine drainage, Central Scotland. Appl Geochem 19:1773–1783
- Wu H, Tang S, Zhang X, Guo J, Song Z, Shuai T, Smith DL (2009) Using elevated CO₂ to increase the biomass of a Sorghum vulgare × Sorghum vulgare var. sudanense hybrid and Trifolium pratense L. and to trigger hyperaccumulation of cesium. J Hazard Mater 170:861–870
- Ye ZH, Baker AJM, Wong MH, Willis AJ (1997a) Zinc, lead and cadmium tolerance, uptake and accumulation by *Typha latifolia*. New Phytol 136:469–480
- Ye ZH, Baker AJM, Wong MH, Willis AJ (1997b) Zinc, lead and cadmium tolerance, uptake and accumulation by the common reed, *Phragmites australis* (Cav.) Trin. Ex Steudel. Ann Bot 80:363–370
- Ye ZH, Whiting SN, Qian JH, Lytle CM, Lin ZQ, Terry N (2001) Wetlands and aquatic processes, trace elements removal from coal ash leachate by a 10 year old constructed wetland. J Environ Qual 30:1710–1719
- Ye ZH, Baker AJM, Wong MH, Willis AJ (2003) Copper tolerance, uptake and accumulation by *Phragmites australis*. Chemosphere 50:795–800

- Zandavalli RB, Dillenburg LR, de Souza PVD (2004) Growth responses of *Araucaria angustifolia* (Araucariaceae) to inoculation with the mycorrhizal fungus *Glomus clarum*. Appl Soil Ecol 25:245–255
- Zhang XB, Liu P, Yang YS, Chen WR (2007) Phytoremediation of urban wastewater by model wetlands with ornamental hydrophytes. J Environ Sci 19:902–909
- Zhang X, Xia H, Li Z, Zhuang P, Gao B (2010) Potential of four forage grasses in remediation of Cd and Zn contaminated soils. Bioresour Technol 101: 2063–2066
- Zhao JG, Yang Q, Chen ZH (2003) Studies on root system biomass of the plants in several kinds of wetland. China Environ Sci 23:290–294
- Zhou BN, Blasko G, Cordell GA (1988) Alternanthin, A C-glycosylated flavonoid from Alternanthera Philoxeroides. Phytochemistry 27:3633–3636
- Zhou D, Chen H, Cang L, Wang Y (2007) Ryegrass uptake of soil Cu/Zn induced by EDTA/EDDS together with a vertical direct current electrical field. Chemosphere 67:1671–1676
- Zurayk R, Sukkariyah B, Baalbaki R, Ghanem DA (2001) Chromium phytoaccumulation from solution by selected hydrophytes. Int J Phytoremediation 3:335–350

Index

A

ABA. See Abscisic acid (ABA) Abiotic stress responses atmospheric CO₂ concentration, 15-19 compatible solutes amides, 6-8 amino acids, 6-8 betaines, 4-6 polyamines, 9-10 proline, 6-8 sugars and sugar alcohols, 8-9 oxidative stress and antioxidative responses enzymatic antioxidants, 11-14 nonenzymatic antioxidants, 14-15 ROS, production of, 10-11 Abiotic stress signaling calcium, 327, 347 and MAPKs, 305-306 Abiotic stress tolerance adaptations antioxidant systems, 38-39 genomic approaches, 40-43 ion compartmentation, 33 osmotic adjustment, 35-38 plant responses, 33, 34 redox and energetics, 39-40 succulence, 35 desalination and stabilization, saline soils, 44-45 halophytes, 30-32 halophytes diversity, 32 phytoremediation, 45-47 saline agriculture, 48-49 wetland restoration and revegetation, 47-48 Abscisic acid (ABA) calcium signaling, 346-347 K⁺ nutrition, 34-95 phytohormone signaling, 383-384 plant signaling, 309-310 Acclimatization, 124-126, 130 Adaptation, high temperature, 130 Aluminium, 343–346 Ambient light environment, 59 Amides, 6-8 Amino acids, 6-8

AM symbiosis. See Arbuscular mycorrhizal (AM) symbiosis Anoxia, 332-335 Anthocyanins, 67-71 Antioxidant enzyme expressions, 17 Antioxidants, 431, 432, 436, 437 AR4. See Fourth Assessment Report (AR4) Arabidopsis chilling-sensitive mutants category, 162 functional categorization, 171 mapping of, 167-169 physiological and biochemical responses, 164-168 transcriptome of, 168-170 vs. wild-type, 165-168 Arbuscular mycorrhizal (AM) symbiosis beneficial and nonbeneficial soil microflora, 252 - 253intrinsic vs. extrinsic stress tolerance, 234-236 macro-and micronutrient uptake micronutrient (metal) uptake, 244-246 N acquisition, 240-242 P acquisition, 242-244 and Mycorrhizae, 236-240 plant stress tolerance, 253-257 plant water relations, 246-247 soil structure stabilization metal biosorption, 250-252 soil aggregation, 248-250 Artificially enhanced UV-B radiation, 63-64 AS. See Asparagine synthetase (AS) Asparagine (Asn), 226 Asparagine synthetase (AS), 226 Atmospheric CO₂ concentration antioxidant enzyme expressions, 17 GSH/GSSG ratios, 16 osmolytes, 18 Auxins, phytohormone signaling, 96, 313, 382-383

В

Betaines, 4–6 Biobeds, 480–481 Biochemical responses (*chs* mutants), 164–167 Biocontrol, phytoremediation, 499

P. Ahmad and M.N.V. Prasad (eds.), *Environmental Adaptations and Stress Tolerance of Plants in the Era of Climate Change*, DOI 10.1007/978-1-4614-0815-4, © Springer Science+Business Media, LLC 2012

Bioremediation organic contaminants, 472–475 weeds and ornamentals, 488, 490, 493–497
Biotechnological perspectives genetic engineering, 103–104 natural variability and QTL mapping, 102–103
Biotic environmental stress factors, 234
Brassinosteroids (BRs), 310–311, 387–388
Buffer strips, 477–479

С

Cadmium-induced oxidative stress cadmium toxicity, 430-432 nutrients availability and cadmium tolerance, 432-433 sulfur, 433-436 sulfur sinks, 436-438 Cadmium stress nitric oxide (NO) production, 203-205 plant responses cellular detoxification and repair, 206-207 cell wall metabolism, 207 hydric balance, 208 metal transporters, 207 photosynthesis regulation, 205 protein degradation, 208 signal transduction, 205-206 sulfate and GSH metabolism, 207-208 transcriptional regulation, 205-206 ROS, 201-203 signal transduction, 208-209 toxic effects, 200-201 Cadmium tolerance nutrients availability, 432-433 sulfur, 433-436 sulfur sinks, 436-438 Cadmium toxicity. See Zea Mays L., cadmium toxicity Calcium-dependent protein kinases (CDPKs), 298-300 Calcium signaling aluminium and heavy metal stress signaling aluminium toxicity, 343 calcium homeostasis, 344 cytosolic calcium elevation, 344-345 heavy metal signaling, 346 heavy metal toxicity and uptake, 345 tolerance mechanisms, 345-346 anoxia anoxic injury, 332-333 anoxic stress, 333 cytosolic calcium signaling, 334-335 oxygen deficiency tolerance mechanisms, 333 biotic stress signaling, 349-350 calcineurin B-like proteins, 302-303 calcium-dependent protein kinases (CDPKs), 298-300 calmodulin and other calcium-binding proteins, 300-302 cell types, 329 cold temperature stress signaling acclimatization, 335-336 cold/chilling stresses, 336-338

membrane potential, 338-339 plasma membrane lipid composition, 335 with hormones abscisic acid (ABA), 346-347 auxin. 348 ethylene, 347-348 gibberellins, 348-349 jasmonic acid (JA), 347 organelles apoplastic signaling, 330 chloroplasts, transport, 330-331 mitochondria, transport, 331 nucleus, transport, 331-332 salinity stress agricultural productivity, 339-340 cellular perception, 341 cytosolic calcium signaling, 341-343 injury, 340 sodium toxicity and osmotic stress, 343 tolerance mechanisms, 340-341 sensors for, 326-327 transporters Ca2+ ATPases, 328 calcium channels, 328-329 calcium/proton exchangers, 328 CaM-binding protein kinase (CBK), 287 Carboxylating enzymes activity, 454 Cation proton antiporters, 89 CBK. See CaM-binding protein kinase (CBK) CDPKs. See Calcium-dependent protein kinases (CDPKs) Cellular detoxification and repair, 206-207 Cell wall metabolism, 207 Chilling injuries, 119-120 Chilling-sensitive plants, 122 Chilling tolerance. See also Arabidopsis chilling-sensitive mutants Arabidopsis, 160–162 Arabidopsis chilling-sensitive mutants, 162-164 Chlorophyll content, chilling effects, 166 Climate changes adaptations and mitigation processes, 490 biogeochemical cycling, heavy metals, 489 bioremediation, 488, 490, 493-497 and natural attenuation, 472-475 mitigation biobeds, 479-481 buffer strips, 477-479 wetlands, 476-477 organic pollutants, 468-471 ornamentals, 492-493, 500 phytoremediation, 475-476 phytostabilization, 491-492 root formation and altered precipitation, 180-183 and excess water, 180-185 and high temperature, 177-180 and salinity, 186-192 water weeds, 497-498 CNGCs. See Cyclic nucleotide-gated channels (CNGCs) Cold temperature stress signaling, Ca²⁺ signaling acclimatization, 335–336 cold/chilling stresses, 336–338 membrane potential, 338–339 plasma membrane lipid composition, 335 Compatible solutes, 38 Conventional breeding, 147–148 Cyclic nucleotide-gated channels (CNGCs), 89–90 Cytogenetics, 420 Cytokinins K⁺ nutrition, 96 phytohormone signaling, 384–385

D

DNA, damage and repair, 64–67
DNA repair mechanisms, 64–65 *DREB1/CBF* TFs, 280–283
Drought stress, 180, 181, 183, 235, 246, 247, 250
DREB2, 283–286
miRNAs, 275
plant growth-promoting bacteria (PGPR), 400–401

E

Electrolyte leakage rates, chilling effects, 166 Environmental abiotic stresses annotation criteria, 270-271 K⁺ nutrition cross-talk, NH₄+, NO₃-, Pi, 101-102 high CO₂, 98–99 high temperature, 99 and NH₄+, 101 salinity, 99-101 soil moisture, 99 miRNAs isolation and characterization cold and related stresses, 273-274 drought stress, 275 hypoxia, 275 mechanical stress, 274 nutrient stress, 271-273 salt stress, 274-275 in silico prediction and characterization, 267 - 269sRNA library preparation, 269-270 submergence, 275 nomenclature parameters, 271 stress factors, 234 Environmental moderation and toxic trace metal cleanup, 492-493 Environmental stressor, 62 Enzymatic antioxidants ascorbate peroxidase, 13 catalase, 12-13 glutathione reductase, 13-14 superoxide dismutase, 11-12 Enzyme activity, 114-117 EORCU. See European Ozone Research Coordinating Unit (EORCU) ESTs analysis, 268

Ethylene mechanisms, 397 phytohormone signaling, 385-386 plant growth-promoting bacteria (PGPR) ACC deaminase, 397-399 drought, 400-401 flooding, 400 metals, 401-402 organic compounds, 402-403 salt, 399-400 plant signaling, 311-312 transgenic plants flooding, 404-405 metals, 405 salt, 404 European Ozone Research Coordinating Unit (EORCU), 59

F

Flavonoids, 67–71 Flooding ethylene and abiotic stress tolerance, 404–405 plant growth-promoting bacteria (PGPR), 400 Fourth Assessment Report (AR4), 177 Freezing injuries, 119–120 Freezing stress, 119, 121–124

G

GB. See Glycine betaine (GB) GCM. See Global Climate Model (GCM) Gene expression, modification of, 104 Genes mapping (chs1, chs2, and chs3), 167-169 Genetic engineering, 103–104 Genomic approaches, 40-43 Gibberellins calcium interactions, 348-349 phytohormone signaling, 386 Gln. See Glutamine (Gln) Global Climate Model (GCM), 59 Glutamate synthase (GOGAT) gene systems, 225-226 Glutamine (Gln), 226 Glutamine synthetase (GS), 225-226 Glycine betaine (GB), 37 Grafting, 419-420 GS. See Glutamine synthetase (GS) GSH metabolism, 207-208 GSSs analysis, 268

н

Halophytes, 30–32 Heat shock proteins (HSPs), 283, 286–287 Heat stress. *See also* High temperature stress; Low temperature stress anatomical and developmental responses, 139, 140 growth and phenology, 137–139 management, 147–148 molecular responses, 143–147 physiological and metabolic responses, 139–143 Heavy metal stress. See also Metal-induced stress calcium signaling aluminium toxicity, 343 calcium homeostasis, 344 cytosolic calcium elevation, 344-345 heavy metal signaling, 346 heavy metal toxicity and uptake, 345 tolerance mechanisms, 345-346 salicylic acid, 452-453 High temperature stress anatomical and developmental responses, 139, 140 growth and phenology, 137-139 management conventional breeding, 147-148 genetic variability, exploitation of, 147 molecular strategies, 147-148 seed treatments and plant acclimation, 148 molecular responses dehydrins, 145-146 heat shock genes and proteins, 143-145 senescence-associated genes, 146-147 Stay-green gene, 147 physiological and metabolic responses metabolite synthesis, 142-143 osmolytes accumulation, 141-142 photosynthesis, 140-141 reactive oxygen species and oxidative damage, 141 water relations, 141 Hormone activities phytohormone signaling, 381-382 and ROS, 94-96 H₂O₂ signaling. See Hydrogen peroxide (H₂O₂) signaling Hydric balance, 208 Hydrogen peroxide (H_2O_2) signaling accumulation, chilling effects, 168 chemical properties, 362-363 electron transfer reactions, 362 physiological processes germinating seeds, 372-374 microorganism interaction, 374-375 plant growth, 375 programmed cell death (PCD), 376 stomatal movements, 371-372 production of chloroplasts, 365-366 cytoplasm, 366 mitochondria, 365 peroxisomes (glyoxisomes), 364-365 plasmalemma and cell-wall apoplast, 366 scavenging, 366-368 signal perception, 368-369 transcription factor genes, 370-371 transduction of, 369-370 Hypoxia, 275

I

Inter governmental Panel on Climate Change (IPCC), 177 Ion compartmentation, 33 IPCC. *See* Inter governmental Panel on Climate Change (IPCC) *Ipomoea aquatica*, 498

J

Jasmonates (JA), 312, 388

Κ

K⁺ nutrition arabidopsis model uptake systems, 96-97 uses, 97 biotechnological perspectives genetic engineering, 103-104 natural variability and QTL mapping, 102-103 deficiency, 87 environmental abiotic stresses cross-talk, NH₄+, NO₃-, Pi, 101-102 high CO₂, 98–99 high temperature, 99 and NH₄+, 101 salinity, 99-101 soil moisture, 99 levels inside, plant, 86-87 transport systems post-translational regulation, 93-94 signaling molecules, 94-96 transcriptional regulation, 92-93 uptake systems entities mediating root, characterization, 90-92 high-and low-affinity, 88 molecular entities mediating affinity, 88-90

L

Lethal temperatures, classification of plant, 119–120 Living organism, classification of, 113–114 Low temperature stress, 161

Μ

Macronutrient, 240-242 Maize, 454-459. See also Zea Mays L., cadmium toxicity MAPK kinases abiotic stress signaling, 305-307 MAPKs, 305 MAP2Ks, 304-305 MAP3Ks. 304 Mechanical stress, 274 Membrane transport, 327–329 Metabolism, 117, 118, 128 Metabolite synthesis, 142-143 Metal-induced stress cytogenetics, 420 grafting, 419-420 hormonal modulation, 416–418 metallomics, 420-421

proteome, 418-419 ROS. 415 Metalliferous substrates, 497-498 Metallomics, 420-421 Metals biosorption, 250-252 ethylene and abiotic stress tolerance, 405 plant growth-promoting bacteria (PGPR), 401-402 transporters, 207 Micronutrient, 244-246 MicroRNAs (MiRNAs) abiotic stress cold and related stresses, 273-274 drought stress, 275 hypoxia, 275 mechanical stress, 274 nutrient stress, 271-273 salt stress, 274-275 submergence, 275 annotation criteria, 270-271 isolation and characterization ESTs and GSSs analysis, 268 in silico prediction and characterization, 267-269 potential miRNAs, 268 target prediction, 268-269 nomenclature parameters, 271 sRNA library preparation, 269-270 Molecular entities mediating low-and high-affinity K⁺ uptake cation proton antiporters, 89 cyclic nucleotide-gated channels (CNGCs), 89-90 HAK/KT/KUP transporters, 88-89 HKT transporters, 89 Shaker channels, 88 Morphogenic responses, 71-73 Mycorrhizospher, 236

Ν

Nitrate uptake regulation, 223-224 Nitric oxide (NO) biosynthesis of, 307-308 cadmium stress, 203-205 signal transduction, 308 Nitrogen metabolism, signaling and regulation, 226-227 Nitrogen use efficiency (NUE) concept of, 221-222 physiological and molecular aspects asparagine (Asn), 226 asparagine synthetase (AS), 226 glutamate synthase (GOGAT) gene systems, 225-226 glutamine (Gln), 226 glutamine synthetase (GS), 225-226 nitrate reduction, physiology of, 224 nitrate uptake regulation, 223-224 transport gene systems, 224-225 proteomics approach, 228-229 QTL approach, 227–228 NO. See Nitric oxide (NO)

Nonenzymatic antioxidants ascorbic acid, 14–15 reduced glutathione, 15 a-tocopherol, 15 N pollution, 222–223 NUE. *See* Nitrogen use efficiency (NUE) Nutrients availability, 432–433 Nutrient stress, miRNAs copper maintenance, 273 phosphorus starvation, 272 sulfur deficiency, 272–273

0

Organic compounds, PGPB, 402-403 Organic contaminants climate change scenarios, 468-469 fate and behavior, 469-471 phytoremediation, 475-476 risk reduction biobeds, 480-481 bioremediation and natural attenuation, 472-475 buffer strips, 477-479 phytoremediation, 475-476 wetlands, 476-477 Ornamentals, 492-493 Osmolytes accumulation, 15-19, 141-142 Osmotic adjustment glycine betaine, 37 proline, 36-37 soluble sugars, 37-38 Oxidative stress and antioxidative responses enzymatic antioxidants, 11-14 nonenzymatic antioxidants, 14-15 ROS, production of, 10-11 salicylic acid, 453, 454, 462

Ρ

Peroxisomes (glyoxisomes), 364-365 PGPR. See Plant growth-promoting bacteria (PGPR) Phosphorus starvation, 272 Photosynthesis chilling effects, 166 regulation, 205 salicylic acid, 451, 452 Physiological responses (chs mutants), 164-167 Phytohormone signaling abscisic acid (ABA), 383-384 auxins, 313, 382-383 brassinosteroids (BRs), 310-311, 387-388 cytokinins, 384-385 ethylene, 311-312, 385-386 gibberellins, 386 hormone activities, 381-382 jasmonates (JA), 312, 388 salicylic acid (SA), 312-313, 388-389 strigolactones, 389

Phytoremediation biocontrol, 499 organic contaminants, 475-476 ornamental hydrophytes, 493-497 Phytostabilization Lolium perenne, 491-492 Panicum virgatum, 491 Prosopis juliflora, 491–492 Plant growth-promoting bacteria (PGPR) ACC deaminase model, 397-398 occurrencenature, 398 protein biochemistry, 398 regulation, 398-399 drought, 400-401 flooding, 400 metals, 401-402 organic compounds, 402-403 salt, 399-400 Plant interactions, 374-375 Plant signaling abscisic acid (ABA), 309-310 calcium signaling calcineurin B-like proteins, 302-303 calcium-dependent protein kinases (CDPKs), 298-300 calmodulin and other calcium-binding proteins, 300-302 MAPK kinases abiotic stress signaling, 305-307 MAPKs, 305 MAP2Ks, 304-305 MAP3Ks, 304 nitric oxide (NO) biosynthesis of, 307-308 signal transduction, 308 phytohormones auxins, 313 brassinosteroids (BRs), 310-311 ethylene, 311-312 jasmonates (JA), 312 salicylic acid (SA), 312-313 sugar, 308-309 Polyamines, 9-10 Post-translational regulation, 93-94 Potassium. See K⁺ nutrition Potential mycorrhizospheric interactions, 240 Precipitation, 180-183 Pretreatment strategies, 147-148 Programmed cell death (PCD), 376 Proline, 6-8 Protein activity, modification of, 104 Protein degradation, 208 Proteomics approach metal-induced stress, 418-419 NUE, 228-229

Q

QTL approach, 227–228 QTL mapping, 102–103

R

Reactive nitrogen (Nr). See also Nitrogen use efficiency (NUE) nitrogen removal by crops, 221 N loss in ecosystem, 220 N pollution, agriculture, 222-223 NUE, 221-222 Reactive oxygen species (ROS), 160, 163 cadmium stress, 201-203 cadmium tolerance, 438 cell death, 10 ethylene, 95 and oxidative damage, 141 Redox and energetics, 39-40 Root form and altered precipitation, 180-183 and excess water, 180-185 and high temperature, 177-180 and salinity and function, 188-192 hydroponic vs. soil systems, 186 Na + uptake and accumulation, 187 salt-overlay-sensitive, 188 soil constraints, 186-187 ROS. See Reactive oxygen species (ROS)

S

Salicylic acid (SA) biosynthesis, 449-450 cadmium toxicity, Zea Mays L. antioxidant enzymes activity, 459 carboxylating enzymes activity, 454 chlorophyll content, 455-456 CO₂ assimilation rate, 454 FAME, 456-457 fatty acid composition, 457-458 proline level, 455-456 relative water content, 455-456 thermoluminescence emission, 454-455 total lipophilic extracts, 456-457 and heavy metal stress, 452-453 physiological functions, 450-452 phytohormone signaling, 388-389 plant signaling, 312-313 properties, structure, 449-450 Saline soils desalination and stabilization, 44-45 saline agriculture, 31, 48-49 Salinity stress calcium signaling agricultural productivity, 339-340 cellular perception, 341 cytosolic calcium signaling, 341-343 injury, 340 sodium toxicity and osmotic stress, 343 tolerance mechanisms, 340-341 DREB2 and drought/high salinity stress, 283-286 responses, 41 root formation function, 188-192

hydroponic vs. soil systems, 186 Na⁺ uptake and accumulation, 187-188 salt-overlay-sensitive (SOS) pathway, 188 soil constraints, 186-187 Salt K⁺ nutrition, 100-101 miRNAs, 274-275 PGPR, 399-400 transgenic plants, 404 Scavenging, 366–368 Secondary metabolites, 67-71 Signal factor, 60 Signaling molecules. See also Hydrogen peroxide (H₂O₂) signaling abscissic acid (ABA), 94-95 AtAKT1 activity, 94 auxins, 96 cytokinins, 96 ethylene, 94 jasmonic acid (JA), 95 ROS, 94 Signaling pathways, 73-77 Signal transduction, 205-206, 327, 330, 335 Soil structure stabilization metal biosorption, 250-252 soil aggregation, 248-250 Solar UV-B radiation. See UV-B radiation sRNA library preparation, 269-270 Starch accumulation, chilling effects, 167 Stratification, 115 Stress signaling, 416-418 Stress tolerance mechanism, 34 Strigolactones, 389 Succulence, 35 Sugars and sugar alcohols, 8-9 Sulfate and GSH metabolism, 207-208 Sulfur, Cd tolerance, 433-436

Т

Sulfur sinks, 436-438

Temperature stress enzymes, 115-119 high temperature adaptation to, 130 heat exposure, 126-128 plant resistance mechanism, 128-130 living organisms, classification of, 113-114 low temperature chilling and freezing stress, 119-124 resistance to, 124-127 membranes, 115-119 metabolic processes, 115-119 stratification, 115 vernalization, 114-115 Terrestrial plants, solar UV-B radiation, 59-63 TFs. See Transcription factors (TFs) Thermoluminescence, 454–455 Transcriptional regulation, 92-93, 205-206

Transcription factors (TFs) AtMYC2 and AtMYB2 proteins, 288 DREB1/CBF TFs and cold stress response, 280-283 drought/high salinity stress response and AREB TFs, 284-286 and DREB2, 283-284 H₂O₂, 370-371 HSFs and HSP-related heat stress response, 286-287 NAC gene family, 288 ZFHD proteins, 288 Transcriptome, 168-170 Transgenic plants flooding, 404-405 metals, 405 salt, 404 Transporters HAK/KT/KUP, 88-89 HKT, 89

U

UV-B radiation anomalous signal factor, 60 artificially enhanced UV-B radiation, 63–64 environmental stress factor, 60 life of terrestrial plants, 59 physiological and morphological responses DNA, damage and repair, 64–67 flavonoids and anthocyanins, 67–71 morphogenic responses, 71–73 plant function, 60 and signaling pathways, 73–77

V

Vernalization, 114-115

W

Waterlogging, 183–185 Water weeds utilization, 497–498 Weeds duck weeds, 497 water weeds, 497–498 Wetlands, 476–477

Ζ

Zea Mays L., cadmium toxicity antioxidant enzymes activity, 459 carboxylating enzymes activity, 454 chlorophyll content, 455–456 CO₂ assimilation rate, 454 FAME, 456–457 fatty acid composition, 457–458 proline level, 455–456 relative water content, 455–456 thermoluminescence emission, 454–455 total lipophilic extracts, 456–457